BUDDHIST HYBRID SANSKRIT GRAMMAR AND DICTIONARY

FRANKLIN EDGERTON

Sterling Professor of Sanskrit and Comparative Philology, Yale University

VOLUME I : GRAMMAR

1985

RINSEN BOOK CO.



Reprinted from the original edition of 1953, with permission of Yale Univ. Press, London office.

ISBN4-653-01245-8 / C Yale Univ. Press



REPRODUCED BY RINSEN BOOK CO., KYOTO 1985



BUDDHIST HYBRID SANSKRIT GRAMMAR AND DICTIONARY

VOLUME I: GRAMMAR



TO E.



TABLE OF CONTENTS OF VOLUME ONE

List of Chapters

Prei	lace	XXI
Bib	liography and abbreviations	
1.	Introduction	1
2.	Phonology, consonants	15
3.	Phonology, vowels	23
4.	Samdhi	32
5.	Number	38
6.	Gender	- 39
7.	Case	42
8.	<i>a</i> -stems	48
9.	<i>ā</i> -stems	61
10.	i-stems with i- and in-stems	69
11.	Feminines in -ā and -ī	84
12.	u and a -stems	85
13.	r-stems	90
13.	Diphthongal stems	93
15.	Consonant stems	-94
16.	s-stems	96
	<i>n</i> -stems	.99
17.	<i>n</i> -stems	102
19.	Numerals	102
		108
20.	Personal pronouns	
21.	Generic pronouns	114
22.	Noun suffixes	120
23.	Noun and verb compounds	125
24.	The verb: general remarks	127
25.	Person and number	129
26.	Personal endings	131
27.	Non-Sanskritic treatment of thematic vowel	134
28.	Summary classification of non-Sanskritic presents	136
29.	Optative	141
30.	Imperative	146
31.	Future	148
32.	Aorist, or preterite (including imperfect)	153
33.	Perfect	165
34.	Participles, including gerundlyes	167
35.	Gerund	171
36.	Infinitive	178
37.	Voice	181
38.	Stems in aya (e); causative, denominative, etc.	185
39.	Intensive	197
40.	Desiderative	198
41.	Syntactic addenda to the verb	199
42.	Constructions with ma	200
43.	Synopsis of verb forms	204
2Q.	CARACTER AT ACTE FAIRER	

List of Chapter-subdivisions

Preface	
1. Introduction	1
Languages used in early Buddhism § 1.1	1
An 'original language of Buddhism'? § 1.14	2
Buddhist Hybrid Sanskrit § 1.33	4
Changes in the course of tradition § 1.39	6
Plan and methods of this work § 1.57	9
Sanskrit versus Buddhist Hybrid Sanskrit § 1.76	11
The Prakrit underlying BHS § 1.78	11
The BHS Lexicon § 1.106	13

2. Phonology, consonants	
	15
Assimilation of consonants § 2.5.	15
Semivowels 'lost' in triconsonantal clusters § 2.17	16
ch § 2.18 kş, kşy, ş, kh § 2.23	16 16
Voiced for voiceless stops § 2.28	17
v for p and m , and for $y \notin 2.30$	17
y for intervocalic stop and vice versa $\S 2.32$	17
<i>i</i> for <i>y</i> and <i>y</i> for <i>j</i> § 2.34	17
j for y and y for j § 2.34 Aspirates and h § 2.35	17
t for d for Proto-Indic j^1 (IE \hat{g})? § 2.38	17
Domai and dental stops and nasals § 2.39	17
d(d), l , and $r $ § 2.46	18
Prothetic <i>v</i> § 2.50	18
Sibilants § 2,56	19
Nasals and anusvāra § 2.64 Loss of final (rarely internal) nasal § 2.72	19
Addition of final nasal § 2.74	20
Double and single consonants; loss of final consonants	20
§ 2.77	20
Consonant, generally initial, doubled after short	
vowel § 2.78	21
Reduction of double to single consonants § 2.84	21
Compounds of Skt. nis- § 2.86	21
Forms of dars-, MIndic dass- § 2.87	21
Compounds with Skt. ud- § 2.88	21
Miscellaneous § 2.89	22
Dropping of final consonants § 2.90	22
2 Manufadu variala	23
3. Phonology, vowels	23 23
Long vowel for short before consonant cluster § 3.2	23
Long vowel for nasalized vowel § 3.3	23
Nasalized vowel for short plus double, or long plus	
	20
single consonant § 3.4	
single consonant § 3.4 Short vowel plus double consonant for long plus	
Short vowel plus double consonant for long plus single § 3.4a	23 23
Short vowel plus double consonant for long plus single § 3.4a Long vowels for short § 3.5	23 23 23
Short vowel plus double consonant for long plus single § 3.4a Long vowels for short § 3.5 ā for a § 3.5	23 23 23 23
Short vowel plus double consonant for long plus single § 3.4a Long vowels for short § 3.5 å for a § 3.5 å for final a § 3.6	23 23 23 23 23
Short vowel plus double consonant for long plus single § 3.4a Long vowels for short § 3.5 å for a § 3.5 å for final a § 3.6 å for non-final a § 3.9	23 23 23 23 23 23
Short vowel plus double consonant for long plus single § 3.4a Long vowels for short § 3.5 ā for a § 3.5 ā for final a § 3.6 ā for non-final a § 3.9	23 23 23 23 23 23 23 24
Short vowel plus double consonant for long plus single § 3.4a Long vowels for short § 3.5 ā for a § 3.5 ā for final a § 3.6 ā for non-final a § 3.9 for i § 3.12 i for final i § 3.14	23 23 23 23 23 23 24 24
Short vowel plus double consonant for long plus single § 3.4a Long vowels for short § 3.5 å for a § 3.5 å for final a § 3.6 å for non-final a § 3.9 for i § 3.12 i for non-final i § 3.14 i for non-final i § 3.17	23 23 23 23 23 23 24 24 24 24
Short vowel plus double consonant for long plus single § 3.4a Long vowels for short § 3.5 ä for a § 3.5 ä for non-final a § 3.6 for i § 3.12 i for final i § 3.14 i for non-final i § 3.17 ü for u § 3.20	23 23 23 23 23 23 23 24 24 24 24 24
Short vowel plus double consonant for long plus single § 3.4a. Long vowels for short § 3.5. å for a § 3.5. å for final a § 3.6. å for non-final a § 3.9. for final i § 3.14. i for non-final i § 3.17. å for final u § 3.20. å for final u § 3.21.	23 23 23 23 23 23 23 24 24 24 24 24 24
Short vowel plus double consonant for long plus single § 3.4a. Long vowels for short § 3.5. å for a § 3.5. å for final a § 3.6. å for non-final a § 3.9. for final i § 3.14. i for non-final i § 3.14. i for non-final i § 3.17. ů for non-final u § 3.20. ů for non-final u § 3.26.	23 23 23 23 23 23 23 24 24 24 24 24
Short vowel plus double consonant for long plus single $\S 3.4a$ Long vowels for short $\S 3.5$ \mathring{a} for a $\S 3.5$ \mathring{a} for final a $\S 3.6$ \mathring{a} for non-final a $\S 3.9$ for i $\S 3.12$ \mathring{a} for non-final i $\S 3.14$ \mathring{a} for non-final i $\S 3.14$ \mathring{a} for non-final i $\S 3.17$ \mathring{a} for u $\S 3.20$ \mathring{a} for final u $\S 3.21$ \mathring{a} for non-final w $\S 3.26$ Short vowels for long $\S 3.27$	23 23 23 23 23 23 23 24 24 24 24 24 24 24 24
Short vowel plus double consonant for long plus single $\S 3.4a$ Long vowels for short $\S 3.5$ \mathring{a} for a $\S 3.5$ \mathring{a} for final a $\S 3.6$ \mathring{a} for non-final a $\S 3.9$ for i $\S 3.12$ \mathring{a} for non-final i $\S 3.14$ \mathring{a} for non-final i $\S 3.14$ \mathring{a} for non-final i $\S 3.17$ \mathring{a} for u $\S 3.20$ \mathring{a} for final u $\S 3.21$ \mathring{a} for non-final w $\S 3.26$ Short vowels for long $\S 3.27$	23 23 23 23 23 23 24 24 24 24 24 24 24 24 24
Short vowel plus double consonant for long plus single § 3.4a Long vowels for short § 3.5 \tilde{a} for a § 3.5 \tilde{a} for final a § 3.6 \tilde{a} for non-final a § 3.9 for i § 3.12 i for final i § 3.14 \tilde{a} for non-final i § 3.17 \tilde{a} for final u § 3.21 \tilde{a} for final u § 3.26 Short vowels for long § 3.27 a for final \tilde{a} § 3.27 a for final \tilde{a} § 3.27 a for final \tilde{a} § 3.27 a for non-final \tilde{a} § 3.27	23 23 23 23 23 23 23 24 24 24 24 24 24 24 24 24 24 24 25
Short vowel plus double consonant for long plus single $\S 3.4a$ Long vowels for short $\S 3.5$ \mathring{a} for a $\S 3.5$ \mathring{a} for final a $\S 3.6$ \mathring{a} for non-final a $\S 3.9$ for i $\S 3.12$ \mathring{a} for non-final i $\S 3.14$ \mathring{a} for non-final i $\S 3.14$ \mathring{a} for non-final i $\S 3.14$ \mathring{a} for non-final i $\S 3.20$ \mathring{a} for fon-final u $\S 3.21$ \mathring{a} for non-final u $\S 3.26$ Short vowels for long $\S 3.27$ a for final \mathring{a} $\S 3.27$ a for final \mathring{a} $\S 3.27$ a for non-final \mathring{a} $\S 3.31$	23 23 23 23 24 24 24 24 24 24 24 24 24 24 25 25
Short vowel plus double consonant for long plus single $\S 3.4a$. Long vowels for short $\S 3.5$. <i>ā</i> for a $\S 3.5$. <i>ā</i> for final <i>a</i> $\S 3.6$. <i>ā</i> for non-final <i>a</i> $\S 3.9$. <i>f</i> for <i>i</i> $\S 3.12$. <i>i</i> for final <i>i</i> $\S 3.14$. <i>i</i> for final <i>i</i> $\S 3.17$. <i>ā</i> for <i>i</i> a $\S 3.20$. <i>ā</i> for final <i>u</i> $\S 3.21$. <i>ā</i> for non-final <i>u</i> $\S 3.26$. Short vowels for long $\S 3.27$. <i>a</i> for final <i>ā</i> $\S 3.27$. <i>a</i> for final <i>ā</i> $\S 3.27$. <i>a</i> for final <i>ā</i> $\S 3.27$. <i>a</i> for non-final <i>ā</i> $\S 3.27$. <i>a</i> for final <i>ā</i> $\S 3.37$. <i>i</i> for non-final <i>ā</i> $\S 3.37$. <i>i</i> for final <i>i</i> $\S 3.37$.	23 23 23 23 24 24 24 24 24 24 24 24 24 24 25 25 25
Short vowel plus double consonant for long plus single § 3.4a. Long vowels for short § 3.5. å for a § 3.5. å for non-final a § 3.6. å for non-final a § 3.9. for final i § 3.14. i for final i § 3.14. ä for non-final i § 3.17. å for non-final i § 3.17. å for non-final u § 3.21. å for non-final u § 3.27. å for final a § 3.27. a for final å § 3.37. i for son-final å § 3.37. i for non-final å § 3.37. i for non-final i § 3.37.	23 23 23 23 24 24 24 24 24 24 24 24 24 25 25 25 25
Short vowel plus double consonant for long plus single § 3.4a	23 23 23 23 23 24 24 24 24 24 24 24 24 24 24 25 25 25 25 25
Short vowel plus double consonant for long plus single $\S 3.4a$ Long vowels for short $\S 3.5$ a for a $\S 3.5$ a for final $a \S 3.6$ f for non-final $a \S 3.9$. for $i \S 3.12$ i for non-final $i \S 3.14$ i for non-final $i \S 3.14$ i for non-final $i \S 3.17$ a for final $u \S 3.20$ a for final $u \S 3.21$ a for non-final $u \S 3.26$ Short vowels for long $\S 3.27$ a for final $d \S 3.27$ a for non-final $d \S 3.27$ a for non-final $d \S 3.27$ a for non-final $d \S 3.37$ i for final $i \S 3.44$ Substitution of i $\xi 3.44$	23 23 23 23 23 24 24 24 24 24 24 24 24 24 24 25 25 25 25 25 25 26
Short vowel plus double consonant for long plus single $\S 3.4a$ Long vowels for short $\S 3.5$ \mathring{a} for a $\S 3.5$ \mathring{a} for final a $\S 3.6$ \mathring{a} for non-final a $\S 3.9$. for $i \S 3.12$ \mathring{a} for final $i \S 3.14$ \mathring{a} for non-final $i \S 3.17$ \mathring{a} for non-final $u \S 3.21$ \mathring{a} for non-final $u \S 3.26$. Short vowels for long $\S 3.27$ a for non-final $\mathring{a} \S 3.31$ \mathring{a} for non-final $\mathring{a} \S 3.27$ a for non-final $\mathring{a} \S 3.27$ a for non-final $\mathring{a} \S 3.27$ a for non-final $\mathring{a} \S 3.27$ \mathring{a} for non-final $\mathring{a} \S 3.27$ \mathring{a} for non-final $\mathring{a} \S 3.37$ \mathring{a} for $\mathring{a} \mathring{a} 3.44$ Substitution of \mathring{a} $\mathring{a} 3.44$ u, \hat{a} for o (and au) $\S 3.51$	23 23 23 23 23 24 24 24 24 24 24 24 25 25 25 25 25 26 26
Short vowel plus double consonant for long plus single $\S 3.4a$ Long vowels for short $\S 3.5$ <i>ā</i> for a $\S 3.5$ <i>ā</i> for final <i>a</i> $\S 3.6$ <i>ā</i> for non-final <i>a</i> $\S 3.9$. for <i>i</i> $\S 3.12$ <i>i</i> for final <i>i</i> $\S 3.14$ <i>i</i> for final <i>i</i> $\S 3.17$ <i>ā</i> for <i>i</i> a $\S 3.20$ <i>ā</i> for final <i>u</i> $\S 3.21$ <i>ā</i> for non-final <i>u</i> $\S 3.26$. Short vowels for long $\S 3.27$ <i>a</i> for final <i>d</i> $\S 3.31$ <i>i</i> for non-final <i>i</i> $\S 3.31$ <i>i</i> for non-final <i>i</i> $\S 3.37$ <i>i</i> for non-final <i>i</i> $\S 3.37$ <i>i</i> for non-final <i>i</i> $\S 3.37$ <i>i</i> for o non-final <i>i</i> $\S 3.37$ <i>i</i> for non-final <i>i</i> $\S 3.37$ <i>i</i> for <i>i</i> non-final <i>i</i> $\S 3.57$ <i>i</i> for <i>i a</i> $\S 3.57$	23 23 23 23 23 24 24 24 24 24 24 24 24 25 25 25 26 26 26 26
Short vowel plus double consonant for long plus single § 3.4a Long vowels for short § 3.5 a for a § 3.5 a for non-final a § 3.9. for i § 3.12 i for final i § 3.14 i for final i § 3.17 a for u § 3.20 a for final u § 3.21 a for non-final u § 3.26. Short vowels for long § 3.27 a for final d § 3.27 a for final d § 3.27 a for final d § 3.31 i for non-final d § 3.31 i for final i § 3.37 i for a for a § 3.44 Substitution of i , i for e , ai § 3.47 u , \hat{u} for o (and au) § 3.51 u for a § 3.58	$\begin{array}{c} 23\\ 23\\ 23\\ 23\\ 24\\ 24\\ 24\\ 24\\ 24\\ 24\\ 25\\ 25\\ 25\\ 26\\ 26\\ 26\\ 26\\ 26\end{array}$
Short vowel plus double consonant for long plus single $\S 3.4a$ Long vowels for short $\S 3.5$ <i>ā</i> for a $\S 3.5$ <i>ā</i> for final <i>a</i> $\S 3.6$ <i>ā</i> for non-final <i>a</i> $\S 3.9$. for <i>i</i> $\S 3.12$ <i>i</i> for final <i>i</i> $\S 3.14$ <i>i</i> for final <i>i</i> $\S 3.17$ <i>ā</i> for <i>i</i> a $\S 3.20$ <i>ā</i> for final <i>u</i> $\S 3.21$ <i>ā</i> for non-final <i>u</i> $\S 3.26$. Short vowels for long $\S 3.27$ <i>a</i> for final <i>d</i> $\S 3.31$ <i>i</i> for non-final <i>i</i> $\S 3.31$ <i>i</i> for non-final <i>i</i> $\S 3.37$ <i>i</i> for non-final <i>i</i> $\S 3.37$ <i>i</i> for non-final <i>i</i> $\S 3.37$ <i>i</i> for o non-final <i>i</i> $\S 3.37$ <i>i</i> for non-final <i>i</i> $\S 3.37$ <i>i</i> for <i>i</i> non-final <i>i</i> $\S 3.57$ <i>i</i> for <i>i a</i> $\S 3.57$	23 23 23 23 24 24 24 24 24 24 24 25 25 25 26 26 26 26 26 26

	27
e for o (final) § 3.66 2	27
e for ai § 3.67 2	27
	27
	27
	27
• 101 H, 100 D	27
	27
	27
	28 28
b ioi all g orre recent contractions and a second contraction of the s	10 28
$i \text{ for } a \S 3.89 \dots 2$	28
Mindic representatives of r, and Hyper-Sanskritic r	.0
	28
	28
	29
	29
	29
	29
	29
	29
Epenthetic a § 3.99 2	29
Epenthetic <i>i</i> § 3.102 3	80
	80
Before other consonants than y § 3.107 3	BÒ.
	80
	0
	30
	80
	80
	80
	80
Compressions and the like \$2.115	30
Loss of a syllable § 3.118	81
Loss of a syllable § 3.118 3	
Loss of a syllable § 3.118 3 4. Samdhi 3	81
Loss of a syllable § 3.118	1 12
Loss of a syllable § 3.118	11 12 12
Loss of a syllable § 3.118	11 12 12 13
Loss of a syllable § 3.118	11 12 13 13
Loss of a syllable § 3.118	11 12 13 13
Loss of a syllable § 3.118	11 12 13 13 14 14 14
Loss of a syllable § 3.118	12 12 13 13 14 14
Loss of a syllable § 3.118	11 12 13 13 14 14 14 15
Loss of a syllable § 3.118	11 12 13 13 14 14 14 15 15
Loss of a syllable § 3.118	11 12 12 13 13 14 14 14 15 15 15
Loss of a syllable § 3.118	11 12 12 13 13 14 14 14 15 15 15 16
Loss of a syllable § 3.118	11 12 12 13 13 14 14 14 15 15 15 16 17
Loss of a syllable § 3.118	11 12 12 13 13 14 14 14 15 15 15 16 17 17
Loss of a syllable § 3.118	1 12 12 13 13 14 14 14 15 15 15 16 17 17 17
Loss of a syllable § 3.118	11 12 12 13 13 14 14 14 15 15 15 16 17 17
Loss of a syllable § 3.118	1 12 12 13 13 14 14 14 15 15 15 16 17 17 17
Loss of a syllable § 3.118	
Loss of a syllable § 3.118	
Loss of a syllable § 3.118	
Loss of a syllable § 3.118	
Loss of a syllable § 3.118	
Loss of a syllable § 3.118	
Loss of a syllable § 3.118	
Loss of a syllable § 3.118	
Loss of a syllable § 3.118	
Loss of a syllable § 3.118	11 222333444445 555677777 888 9 9 10000
Loss of a syllable § 3.118	
Loss of a syllable § 3.118	11 222333444445 555677777 888 9 9 100000
Loss of a syllable § 3.118	
Loss of a syllable § 3.118	
Loss of a syllable § 3.118	11 22 23 34 44 44 5 55 66 77 77 888 9 9 100 100 100 11
Loss of a syllable § 3.118	

Nominative § 7.13	42
Nominative absolute or nominativus pendens § 7.13	42
Nominative for instrumental § 7.14	43
Nominative for ablative § 7.15	43
Accusative § 7.16	43
Accusative with nouns and adjectives verbal in	
force § 7.16	43
Blend constructions resulting in two accusatives	
§ 7.17	43
Accusative in precise dating § 7.18	43
Accusative of measure § 7.19	43
Miscellaneous accusatives § 7.20	43
Instrumental-accusative adverbial phrases § 7.27	44
Instrumental § 7.30	44
Instrumental for locative § 7.30	44
Instrumental for ablative § 7.35	45
Miscellaneous uses of instrumental § 7.37	45
Dative § 7.43	45
Dative for genitive § 7.43	45
'Expletive' or 'ethical' dative § 7.45	45
Ablative § 7.46	46
Ablative for instrumental 67.46	
Ablative for instrumental § 7.46	46
Predicative ablative § 7.47	46
Miscellaneous § 7.48	46
Canitive \$753	46
Genitive § 7.00	
Genitive § 7.53 Genitive absolute § 7.53	46
Genitive for ablative § 7.54	46
Genitive for instrumental § 7.60	46
Genitive for dative § 7.63	46
Genitive in predicate with forms of bhavali § 7.64	47
Genitive with miscellaneous verbs § 7.65	47
	47
Locative § 7.80	
Locative for instrumental § 7.81	47
Locative for ablative § 7.82	47
Locative for (dative or) genitive § 7.83	47
Locative for (dative of) generve § 7.00	
8. a-stems	48
	48 48
Stem in -a as oblique case form § 8.3	48
Stem in -a as oblique case form § 8.3 Instrumental § 8.8	48 48
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9	48 48 48
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9	48 48
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10	48 48 48 48
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11	48 48 48 48 48
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12	48 48 48 48 48 48 49
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18	48 48 48 48 48
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18	48 48 48 48 48 48 49
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.10 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -o § 8.18	48 48 48 48 48 49 49 49
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -u § 8.18 Nominative singular -u § 8.20	48 48 48 48 49 49 49 49 49
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -u § 8.18 Nominative singular -u § 8.20 Nominative singular -u § 8.21	48 48 48 48 49 49 49 49 49 49 49
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -u § 8.18 Nominative singular -u § 8.20	48 48 48 48 49 49 49 49 49
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -u § 8.18 Nominative singular -u § 8.20 Nominative singular -u § 8.21	48 48 48 48 49 49 49 49 49 49 49
Stem in -a as oblique case form § 8.3Instrumental § 8.8Ablative § 8.9Genitive § 8.10Locative § 8.10Stem final in composition § 8.12Nominative singular masculine § 8.18Nominative singular -o § 8.18Nominative singular -o § 8.18Nominative singular -a § 8.20Nominative singular -a § 8.21Nominative singular -a § 8.21Nominative singular -a § 8.21	48 48 48 48 49 49 49 49 49 50 50
Stem in -a as oblique case form § 8.3Instrumental § 8.8Ablative § 8.9Genitive § 8.10Locative § 8.10Stem final in composition § 8.12Nominative singular masculine § 8.18Nominative singular -u § 8.20Nominative singular -u § 8.21Nominative singular -u § 8.21Nominative singular -a § 8.22Nominative singular -e § 8.24	48 48 48 48 49 49 49 49 49 50 50
$\begin{array}{llllllllllllllllllllllllllllllllllll$	48 48 48 48 49 49 49 49 49 50 50 50
Stem in -a as oblique case form § 8.3Instrumental § 8.8Ablative § 8.9Genitive § 8.10.Locative § 8.11Stem final in composition § 8.12.Nominative singular masculine § 8.18Nominative singular -0 § 8.18Nominative singular -0 § 8.18Nominative singular -2 § 8.20Nominative singular -4 § 8.20Nominative singular -4 § 8.21Nominative singular -4 § 8.22Nominative singular -5 § 8.24Nominative singular -6 § 8.25Nominative singular -6 § 8.26Vocative singular 8 8.27	48 48 48 49 49 49 49 49 50 50 50 50
Stem in -a as oblique case form § 8.3Instrumental § 8.8Ablative § 8.9Genitive § 8.10.Locative § 8.11Stem final in composition § 8.12.Nominative singular masculine § 8.18Nominative singular -0 § 8.18Nominative singular -0 § 8.18Nominative singular -2 § 8.20Nominative singular -4 § 8.20Nominative singular -4 § 8.21Nominative singular -4 § 8.22Nominative singular -5 § 8.24Nominative singular -6 § 8.25Nominative singular -6 § 8.26Vocative singular 8 8.27	48 48 48 48 49 49 49 49 49 50 50 50
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10. Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -a § 8.20. Nominative singular -a § 8.21 Nominative singular -a § 8.22 Nominative singular -a § 8.24 Nominative singular -a § 8.25 Nominative singular -am § 8.26 Vocative singular -am § 8.27	48 48 48 49 49 49 50 50 50 50 50 50
Stem in -a as oblique case form § 8.3Instrumental § 8.8Ablative § 8.9Genitive § 8.10Locative § 8.11Stem final in composition § 8.12Nominative singular masculine § 8.18Nominative singular -o § 8.18Nominative singular -o § 8.18Nominative singular -a § 8.20Nominative singular -a § 8.21Nominative singular -a § 8.21Nominative singular -a § 8.24Nominative singular -a § 8.25Nominative singular -a § 8.25Nominative singular -a § 8.26Vocative singular § 8.27Vocative singular -a § 8.28Vocative singular -a § 8.28	48 48 48 49 49 49 49 49 50 50 50 50
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.10 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -a § 8.20 Nominative singular -a § 8.21 Nominative singular -a § 8.22 Nominative singular -a § 8.22 Nominative singular -a § 8.25 Nominative singular -a § 8.26 Nominative singular -a § 8.26 Nominative singular -a § 8.26 Nominative singular -a § 8.27 Vocative singular -a, 9 8.28 Nocative singular -a, 9 8.28 Nocative singular -a § 8.27 Vocative singular -a, 9 8.28 Nocative singular -a, -a, -e § 8.28	48 48 48 49 49 49 49 50 50 50 50 50 51
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.10 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -a § 8.20 Nominative singular -a § 8.21 Nominative singular -a § 8.22 Nominative singular -a § 8.22 Nominative singular -a § 8.25 Nominative singular -a § 8.26 Nominative singular -a § 8.26 Nominative singular -a § 8.26 Nominative singular -a § 8.27 Vocative singular -a, 9 8.28 Nocative singular -a, 9 8.28 Nocative singular -a § 8.27 Vocative singular -a, 9 8.28 Nocative singular -a, -a, -e § 8.28	48 48 48 49 49 49 49 50 50 50 50 50 51
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -a § 8.18 Nominative singular -a § 8.20 Nominative singular -a § 8.21 Nominative singular -a § 8.22 Nominative singular -a § 8.24 Nominative singular -a § 8.25 Nominative singular -a § 8.26 Vocative singular -ā § 8.27 Vocative singular -a § 8.28 Accusative singular masculine and nominative-accusative singular masculine and nominat	48 48 48 48 48 49 49 49 50 50 50 50 50 50 50 50 50 50 50 50 50 50 50 50 51
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10. Locative § 8.11. Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -a § 8.20 Nominative singular -a § 8.20 Nominative singular -a § 8.21 Nominative singular -a § 8.22 Nominative singular -a § 8.24 Nominative singular -a § 8.25 Nominative singular -a § 8.26 Vocative singular $\frac{1}{a}$ § 8.27 Vocative singular -a § 8.27 Vocative singular -a § 8.28 Accusative singular neuter § 8.29 The ending -u § 8.30	48 48 48 48 48 49 49 50 50 50 50 50 50 50 50 50 50 51 51
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -o § 8.18 Nominative singular -a § 8.20 Nominative singular -a § 8.21 Nominative singular -a § 8.24 Nominative singular -a § 8.25 Nominative singular -a § 8.25 Nominative singular -a § 8.26 Vocative singular $\frac{1}{a}$ § 8.27 Vocative singular $\frac{1}{a}$ § 8.27 Vocative singular $\frac{1}{a}$ § 8.27 Vocative singular $\frac{1}{a}$ § 8.27 Vocative singular masculine and nominative-accusa- tive singular neuter § 8.29 The ending $\frac{1}{a}$ § 8.30 The ending $\frac{1}{a}$ § 8.31	48 48 48 48 48 49 49 49 50 50 50 50 51 51 51
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -a § 8.20 Nominative singular -a § 8.21 Nominative singular -a § 8.22 Nominative singular -a § 8.25 Nominative singular -a § 8.26 Vocative singular $-a § 8.27$ Vocative singular $-a § 8.28$ Nominative singular -a § 8.26 Nominative singular -a § 8.27 Vocative singular -a, a Accusative singular masculine and nominative-accusa- tive singular neuter § 8.29 The ending -a § 8.31 The ending -a § 8.36	48 48 48 48 49 49 50 50 50 50 50 50 51 51 51
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -a § 8.20 Nominative singular -a § 8.21 Nominative singular -a § 8.22 Nominative singular -a § 8.25 Nominative singular -a § 8.26 Vocative singular $-a § 8.27$ Vocative singular $-a § 8.28$ Nominative singular -a § 8.26 Nominative singular -a § 8.26 Nominative singular -a § 8.27 Vocative singular -a, a e § 8.28 Accusative singular masculine and nominative-accusa- tive singular neuter § 8.29 The ending -a § 8.31 The ending -a § 8.36	48 48 48 48 49 49 50 50 50 50 50 50 51 51 51
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -u § 8.20 Nominative singular -u § 8.20 Nominative singular -u § 8.21 Nominative singular -a § 8.22 Nominative singular -a § 8.25 Nominative singular -a § 8.26 Vocative singular $\frac{1}{2}$ § 8.27 Vocative singular $\frac{1}{2}$ § 8.28 Accusative singular -a, u, e § 8.28 Accusative singular masculine and nominative-accusa- tive singular es § 8.30 The ending -a § 8.31 The ending -a § 8.37	48 48 48 48 49 49 50 50 50 50 50 50 51 51 51
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10. Locative § 8.11. Stem final in composition § 8.12. Nominative singular masculine § 8.18 Nominative singular -a § 8.18 Nominative singular -a § 8.20. Nominative singular -a § 8.21. Nominative singular -a § 8.22 Nominative singular -a § 8.24 Nominative singular -a § 8.25. Nominative singular -a § 8.26. Vocative singular $\frac{1}{a}$ § 8.27. Vocative singular $\frac{1}{a}$ § 8.27. Vocative singular -a § 8.28. Accusative singular neuter § 8.29. The ending -a § 8.30. The ending -a § 8.31. The ending -a § 8.37. The ending -a § 8.37. The ending -a § 8.37. The ending -a § 8.38. The ending -a § 8.38. The ending -a § 8.38.	48 48 48 48 49 49 49 50 50 50 50 50 50 50 50 50 51 51 51 52
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -u § 8.18 Nominative singular -u § 8.20 Nominative singular -u § 8.21 Nominative singular -a § 8.22 Nominative singular -a § 8.24 Nominative singular -a § 8.25 Nominative singular -a § 8.26 Vocative singular $\{x, y, z, z,$	48 48 48 48 49 49 50 50 50 50 50 50 51 51 51 51 52 52
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -u § 8.18 Nominative singular -u § 8.20 Nominative singular -u § 8.21 Nominative singular -a § 8.22 Nominative singular -a § 8.24 Nominative singular -a § 8.25 Nominative singular -a § 8.26 Vocative singular $\{x, y, z, z,$	48 48 48 48 49 49 50 50 50 50 50 50 51 51 51 51 52 52
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -a § 8.20 Nominative singular -a § 8.21 Nominative singular -a § 8.22 Nominative singular -a § 8.25 Nominative singular -a § 8.26 Vocative singular $\{x, y\}$ Nominative singular -a § 8.26 Nominative singular -a § 8.26 Vocative singular $\{x, y\}$ Nocative singular nou, -e § 8.28 Accusative singular not $\{x, y\}$ The ending -a § 8.30 The ending -a § 8.36 The ending -e § 8.37 The ending -e § 8.37 The ending -a $\{x, x\}$ Instrumental singular -end § 8.39	48 48 48 48 49 49 49 50 50 50 50 50 51 51 52 52 52 52
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -a § 8.20 Nominative singular -a § 8.20 Nominative singular -a § 8.21 Nominative singular -a § 8.22 Nominative singular -a § 8.25 Nominative singular -a § 8.26 Nominative singular -a § 8.26 Vocative singular -a § 8.27 Vocative singular -a § 8.28 Accusative singular masculine and nominative-accusa- tive singular neuter § 8.29 The ending -a § 8.30 The ending -a § 8.36 The ending -a § 8.36 The ending -a § 8.37 The ending -a § 8.38 Instrumental singular $-ina$ § 8.39 Instrumental singular -ina § 8.40	48 48 48 48 49 49 49 50 50 50 50 50 51 51 52 52 52 52 52
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -u § 8.20 Nominative singular -u § 8.20 Nominative singular -u § 8.21 Nominative singular -a § 8.22 Nominative singular -a § 8.24 Nominative singular -a § 8.25 Nominative singular -a § 8.26 Vocative singular -a § 8.27 Vocative singular -a § 8.28 Accusative singular -a § 8.29 The ending -u § 8.30 The ending -a § 8.31 The ending -a § 8.35 The ending -a § 8.35 The ending -a § 8.36 The ending -a § 8.36 The ending -a § 8.36 The ending -a § 8.37 The ending -a § 8.38 Instrumental singular -enā § 8.39 Instrumental singular -enā § 8.40 Instrumental singular -arā § 8.41	48 48 48 48 49 49 49 50 50 50 50 50 51 51 52 52 52 52
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -a § 8.20 Nominative singular -a § 8.21 Nominative singular -a § 8.22 Nominative singular -a § 8.24 Nominative singular -a § 8.25 Nominative singular -a § 8.26 Vocative singular -a § 8.27 Vocative singular -a § 8.27 Vocative singular -a § 8.28 Accusative singular masculine and nominative-accusative singular masculine and nominative-accusative singular masculine and nominative-accusative singular masculine and nominative-accusative singular as § 8.30 The ending -a § 8.30 The ending -a § 8.31 The ending -a § 8.33 The ending -a § 8.33 Instrumental singular -ara § 8.40 Instrumental singular -ara § 8.41 Instrumental singular -ara § 8.41 Instrumental singular -ara § 8.41	$\begin{array}{r} 48\\ 48\\ 48\\ 48\\ 49\\ 49\\ 49\\ 50\\ 550\\ 550\\ 550\\ 551\\ 511\\ 552\\ 522\\ 552\\ 552\\ 552\\ 552$
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -a § 8.20 Nominative singular -a § 8.21 Nominative singular -a § 8.22 Nominative singular -a § 8.24 Nominative singular -a § 8.25 Nominative singular -a § 8.26 Vocative singular -a § 8.27 Vocative singular -a § 8.27 Vocative singular -a § 8.28 Accusative singular masculine and nominative-accusative singular masculine and nominative-accusative singular masculine and nominative-accusative singular masculine and nominative-accusative singular as § 8.30 The ending -a § 8.30 The ending -a § 8.31 The ending -a § 8.33 The ending -a § 8.33 Instrumental singular -ara § 8.40 Instrumental singular -ara § 8.41 Instrumental singular -ara § 8.41 Instrumental singular -ara § 8.41	$\begin{array}{r} 48\\ 48\\ 48\\ 48\\ 49\\ 49\\ 49\\ 50\\ 550\\ 550\\ 550\\ 551\\ 511\\ 552\\ 522\\ 552\\ 552\\ 552\\ 552$
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -a § 8.18 Nominative singular -a § 8.20 Nominative singular -a § 8.21 Nominative singular -a § 8.22 Nominative singular -a § 8.24 Nominative singular -a § 8.25 Nominative singular -a § 8.26 Vocative singular $\{a, b, 27,\}$ Vocative singular $\{a, b, 27,\}$ Vocative singular masculine and nominative-accusa- tive singular neuter § 8.29 The ending -a § 8.31 The ending -a § 8.36 The ending -a § 8.36 The ending -a § 8.37 The ending -a § 8.38 Instrumental singular -enā § 8.39 Instrumental singular -anā § 8.40 Instrumental singular -anā § 8.41 Instrumental singular -anā § 8.41 Instrumental singular -anā § 8.41	48 48 48 48 48 49 50 50 50 51 51 51 51 52 52 52 52 52 52
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -a § 8.18 Nominative singular -a § 8.20 Nominative singular -a § 8.21 Nominative singular -a § 8.22 Nominative singular -a § 8.25 Nominative singular -a § 8.26 Vocative singular $-a$ § 8.27 Vocative singular $-a$ § 8.27 Vocative singular $-a$ § 8.28 Accusative singular masculine and nominative-accusa- tive singular neuter § 8.29 The ending -a § 8.30 The ending -a § 8.30 The ending -a § 8.36 The ending -a § 8.37 The ending -a § 8.36 The ending -a § 8.36 The instrumental singular -enä § 8.40 Instrumental singular -asā § 8.41 Instrumental singular -asā § 8.41 Instrumental singular -asā § 8.43	48 48 48 48 48 49 50 50 51 51 52
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -u § 8.20 Nominative singular -u § 8.20 Nominative singular -u § 8.20 Nominative singular -a § 8.22 Nominative singular -a § 8.25 Nominative singular -a § 8.26 Vocative singular -a § 8.27 Vocative singular -a § 8.28 Accusative singular -a, § 8.28 Accusative singular masculine and nominative-accusa- tive singular meuter § 8.30 The ending -a § 8.30 The ending -a § 8.31 The ending -a § 8.36 The ending -a § 8.36 The ending -a § 8.36 The ending -a § 8.36 The ending -a § 8.37 The ending -a § 8.38 Instrumental singular -enā § 8.39 Instrumental singular -asā § 8.41 Instrumental singular -asā § 8.41 Instrumental singular -asā § 8.41 Instrumental singular -asā § 8.43 Dative singular § 8.44	$\begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$
Stem in -a as oblique case form § 8.3 Instrumental § 8.8	488 488 484 499 500 500 500 511 511 522 222 222 555
Stem in -a as oblique case form § 8.3 Instrumental § 8.8 Ablative § 8.9 Genitive § 8.10 Locative § 8.11 Stem final in composition § 8.12 Nominative singular masculine § 8.18 Nominative singular -a § 8.18 Nominative singular -a § 8.20 Nominative singular -a § 8.21 Nominative singular -a § 8.22 Nominative singular -a § 8.25 Nominative singular -a § 8.26 Vocative singular $-a$ § 8.27 Vocative singular $-a$ § 8.27 Vocative singular $-a$ § 8.28 Accusative singular masculine and nominative-accusa- tive singular neuter § 8.29 The ending -a § 8.30 The ending -a § 8.30 The ending -a § 8.36 The ending -a § 8.37 The ending -a § 8.38 Instrumental singular -ena § 8.40 Instrumental singular -asā § 8.41 Instrumental singular -asā § 8.41 Instrumental singular -asā § 8.43	488 488 484 499 500 500 500 511 511 522 222 222 555

List of Chapter-subdivisions

Ablative singular § 8.46	e
	-52
Ablative singular $-\bar{a} $ § 8.46	52
Adiative singular - a \$ 8.40	
Ablative singular $-a $ § 8.49	53
Ablative singular -āla(h), -ālo, -ālu § 8.50	53
Ablative singular -ato, -atu § 8.53 Ablative singular -atta (for -attah) § 8.56	53
Ablative singular -alta (for -altah) § 8.56	53
Ablative singular -asmā § 8.57	
Aplative singular -using \$ 0.07	53
Genitive singular, -asyā for -asya § 8.58	
Locative singular § 8.59	53
Locative singular $-i$ § 8.59	53
Locative singular -amhi § 8.61	54
Locative singular -amse § 8.62	51
Locative singular -asmin, -asmin, -asmi § 8.63.	54
Locative singular -asinin, -asinin, -asini 9 6.05.	
Locative singular -esmim, -esmin (or -esmi) § 8.70	54
Nominative-accusative dual masculine § 8.74	55
Nominative-accusative dual -0 § 8.74	55
Nominative-accusative dual -u § 8.75	55
Nominative-accusative dual $-a \leq 8.76$	55
Nominative-accusative tidal -d. § 6.70	
Nominative-accusative dual neuter, -i for -e § 8.77	55
Nominative plural masculine § 8.78	55
Nominative plural $-\tilde{a} \S 8.78$	55
Nominative plural $-a $ 8.79	55
Nominative plural -e § 8.80	56
Nominative plural $-i$ § 8.81	56
Nominative plural -āyo, m.cāya § 8.82	56
Nominative plural $-\alpha h$, -0 § 8.83	56
Nominative plural -u § 8.84	56
Nominative plural an am \$8.85	56
Nominative plural <i>-ān</i> , <i>-ām</i> § 8.85 Nominative plural <i>-āni</i> § 8.86	56
Nominative plural -ant § 0.00	
Vocative plural § 8.87 Vocative plural $-\bar{a}$, $-a$ § 8.87	56
Vocative plural $-a_{,} -a_{,} 8.87$	56
Vocative plural -āho § 8.88	57
Vocative plural -āvo § 8.89	57
Accusative plural masculine § 8.90	57
Accusative plural masculine § 0.90	
Accusative plural masculine -ām, -ām § 8.90	57
Accusative plural masculine -an § 8.91	57
Accusative plural masculine $-\bar{a} \S 8.92 \dots$	57
Accusative plural masculle -a 90.04	
	57
Accusative plural masculine -ās, -āh § 8.93	57
Accusative plural masculine -ās, -āh § 8.93	57 57 57
Accusative plural masculine -ās, -āh § 8.93 Accusative plural masculine -a § 8.94 Accusative plural masculine -e § 8.95	57 57
Accusative plural masculine $-as$, $-ah$ § 8.93 Accusative plural masculine $-a$ § 8.94 Accusative plural masculine $-e$ § 8.95 Accusative plural masculine $-i$ § 8.96	57 57 57
Accusative plural masculine -ās, -āh § 8.93 Accusative plural masculine -a § 8.94 Accusative plural masculine -e § 8.95 Accusative plural masculine -i § 8.96 Accusative plural masculine -i § 8.97	57 57 57
Accusative plural masculine -ās, -āh § 8.93 Accusative plural masculine -a § 8.94 Accusative plural masculine -e § 8.95 Accusative plural masculine -i § 8.96 Accusative plural masculine -i § 8.97	57 57 57
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h \S 8.93$ Accusative plural masculine $-a \S 8.94$ Accusative plural masculine $-i \S 8.95$ Accusative plural masculine $-i \S 8.96$ Accusative plural masculine $-u \S 8.97$ Accusative plural masculine $-\bar{a}ni \S 8.98$	57 57 57 57 58
Accusative plural masculine -ās, -āh § 8.93 Accusative plural masculine -a § 8.94 Accusative plural masculine -e § 8.95 Accusative plural masculine -i § 8.96 Accusative plural masculine -u § 8.97 Accusative plural masculine -āni § 8.98 Accusative plural masculine -āni § 8.98 Accusative plural masculine -āni § 8.98	57 57 57 57 58 58
Accusative plural masculine -ās, -āh § 8.93 Accusative plural masculine -a § 8.94 Accusative plural masculine -e § 8.95 Accusative plural masculine -i § 8.96 Accusative plural masculine -i § 8.96 Accusative plural masculine -i § 8.96 Accusative plural masculine -i § 8.97 Accusative plural masculine -āni § 8.98 Iccusative plural masculine -āni § 8.98 Iccusative plural masculine -āni § 8.98 Iccusative plural masculine -āni § 8.90	57 57 57 57 58
Accusative plural masculine -ās, -āḥ § 8.93 Accusative plural masculine -a § 8.94 Accusative plural masculine -i § 8.95 Accusative plural masculine -i § 8.96 Accusative plural masculine -i § 8.96 Accusative plural masculine -i § 8.96 Accusative plural masculine -i § 8.97 Accusative plural masculine -āni § 8.98 [Accusative plural masculine -āni § 8.98 [Accusative plural masculine -āni § 8.98 [Accusative plural masculine -āni § 8.90 [Accusative plural masculine -āni § 8.90 [Accusative plural masculine -āni § 8.90 [Accusative plural masculine -āni § 8.00 [Accusative plural masculine -āni § 8.00 [Accusative plural masculine -āni § 8.00 [Accusative plural masculine -āni § 8.100 [Accusative plural masculine plural masculine -āni § 8.100 [Accusative plural masculine plural	57 57 57 57 58 58 58 58
Accusative plural masculine -ās, -āḥ § 8.93 Accusative plural masculine -a § 8.94 Accusative plural masculine -e § 8.95 Accusative plural masculine -i § 8.96 Accusative plural masculine -i § 8.96 Accusative plural masculine -i § 8.96 Accusative plural masculine -ii § 8.96 Accusative plural masculine -iii § 8.97 Accusative plural masculine -āni § 8.98 [Accusative plural masculine -āni § 8.98 Nominative-accusative plural neuter \$ \$8.100 Nominative-accusative plural neuter first \$ 8.100 Nominative-accusative plural neuter first	57 57 57 57 58 58 58 58
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h \S 8.93$ Accusative plural masculine $-a \S 8.94$ Accusative plural masculine $-i \S 8.95$ Accusative plural masculine $-i \S 8.96$ Accusative plural masculine $-in \S 8.96$ Accusative plural masculine $-\bar{a}ni \S 8.98$ Accusative plural masculine $-\bar{a}ni \S 8.98$ Accusative plural masculine $-\bar{a}ni \S 8.98$ Nominative-accusative plural neuter $\S 8.100$ Nominative-accusative plural neuter $-\bar{a}$ ($-\bar{a}h?$) $\S 8.100$ Nominative-accusative plural neuter $-a \S 8.101$	57 57 57 57 57 58 58 58 58 58 58
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h \S 8.93$ Accusative plural masculine $-a \S 8.94$ Accusative plural masculine $-i \S 8.95$ Accusative plural masculine $-i \S 8.96$ Accusative plural masculine $-in \S 8.96$ Accusative plural masculine $-\bar{a}ni \S 8.98$ Accusative plural masculine $-\bar{a}ni \S 8.98$ Accusative plural masculine $-\bar{a}ni \S 8.98$ Nominative-accusative plural neuter $\S 8.100$ Nominative-accusative plural neuter $-\bar{a}$ ($-\bar{a}h?$) $\S 8.100$ Nominative-accusative plural neuter $-a \S 8.101$	57 57 57 57 57 58 58 58 58 58 58
Accusative plural masculine -ās, -āḥ § 8.93 Accusative plural masculine -a § 8.94 Accusative plural masculine -e § 8.95 Accusative plural masculine -i § 8.96 Accusative plural masculine -u § 8.97 Accusative plural masculine -āni § 8.98 [Accusative plural masculine -āni § 8.98 [Accusative plural masculine -āni § 8.98 [Accusative plural masculine -āna? § 8.99] Nominative-accusative plural neuter § 8.100 Nominative-accusative plural neuter -ā (-āḥ?) § 8.100 Nominative-accusative plural neuter -a § 8.101 Nominative-accusative plural neuter -ā § 8.101	57 57 57 57 57 58 58 58 58 58 58 58 58
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h \ $ 8.93 Accusative plural masculine $-a \ $ 8.94 Accusative plural masculine $-i \ $ 8.95 Accusative plural masculine $-i \ $ 8.96 Accusative plural masculine $-\bar{a}ni \ $ 8.97 Accusative plural masculine $-\bar{a}ni \ $ 8.98 [Accusative plural masculine $-\bar{a}ni \ $ 8.99 Nominative-accusative plural neuter $-\bar{a} \ $ 6.101 Nominative-accusative plural neuter $-\bar{a} \ $ 8.102 Nominative-accusative plural neuter $-\bar{a} \ $ 8.102	57 57 57 58 58 58 58 58 58 58 58 58 58 58 58
Accusative plural masculine -ās, -āḥ § 8.93 Accusative plural masculine -a § 8.94 Accusative plural masculine -a § 8.95 Accusative plural masculine -i § 8.96 Accusative plural masculine -i § 8.96 Accusative plural masculine -a § 8.97 Accusative plural masculine -a § 8.97 Accusative plural masculine -a § 8.97 Accusative plural masculine -āni § 8.98 IAccusative plural masculine -āni § 8.99 Nominative-accusative plural neuter § 8.100 Nominative-accusative plural neuter -a § 8.101 Nominative-accusative plural neuter -a § 8.101 Nominative-accusative plural neuter -a § 8.102 Nominative-accusative plural neuter -a § 8.103 Nominative-accusative plural neuter -a § 8.104	57 57 57 57 58 58 58 58 58 58 58 58 58 58 58 58 58
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h \notin 8.93$ Accusative plural masculine $-a \notin 8.94$ Accusative plural masculine $-i \notin 8.96$ Accusative plural masculine $-i \notin 8.96$ Accusative plural masculine $-\bar{a}n \notin 8.97$ Accusative plural masculine $-\bar{a}n \notin 8.98$ Accusative plural masculine $-\bar{a}n i \notin 8.98$ Accusative plural masculine $-\bar{a}n i \notin 8.98$ Nominative-accusative plural neuter $\$ 8.100$ Nominative-accusative plural neuter $\$ 8.100$ Nominative-accusative plural neuter $-\bar{a} \notin 8.101$ Nominative-accusative plural neuter $-\bar{a} \notin 8.101$ Nominative-accusative plural neuter $-\bar{a} \notin 8.102$ Nominative-accusative plural neuter $-\bar{a} $	57 57 57 57 57 57 57 57 57 57 57 57 57 5
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h \notin 8.93$ Accusative plural masculine $-a \notin 8.94$ Accusative plural masculine $-i \notin 8.96$ Accusative plural masculine $-i \notin 8.96$ Accusative plural masculine $-\bar{a}n \notin 8.97$ Accusative plural masculine $-\bar{a}n \notin 8.98$ Accusative plural masculine $-\bar{a}n i \notin 8.98$ Accusative plural masculine $-\bar{a}n i \notin 8.98$ Nominative-accusative plural neuter $\$ 8.100$ Nominative-accusative plural neuter $\$ 8.100$ Nominative-accusative plural neuter $-\bar{a} \notin 8.101$ Nominative-accusative plural neuter $-\bar{a} \notin 8.101$ Nominative-accusative plural neuter $-\bar{a} \notin 8.102$ Nominative-accusative plural neuter $-\bar{a} $	57 57 57 57 57 57 57 57 57 57 57 57 57 5
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h$ § 8.93Accusative plural masculine $-a$ § 8.94Accusative plural masculine $-e$ § 8.95Accusative plural masculine $-i$ § 8.96Accusative plural masculine $-\bar{a}$ § 8.97Accusative plural masculine $-\bar{a}n$ § 8.98Accusative plural masculine $-\bar{a}n$ § 8.99Accusative plural masculine $-\bar{a}na$ § 8.99Nominative-accusative plural neuter \$ \$8.100Nominative-accusative plural neuter \$ \$8.101Nominative-accusative plural neuter $-\bar{a}$ § \$.101Nominative-accusative plural neuter $-\bar{a}$ § \$.102Nominative-accusative plural neuter $-\bar{a}$ § \$.104Nominative-accusative plural neuter $-\bar{a}$ § \$.104Nominative-accusative plural neuter $-\bar{a}$ § \$.105Nominative-accusative plural neuter $-\bar{a}$ § \$.105Nominative-accusative plural neuter $-\bar{a}$ § \$.106	57 57 57 57 57 57 57 57 57 57 57 57 57 5
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h$ § 8.93Accusative plural masculine $-a$ § 8.94Accusative plural masculine $-e$ § 8.95Accusative plural masculine $-i$ § 8.96Accusative plural masculine $-\bar{a}$ § 8.97Accusative plural masculine $-\bar{a}n$ § 8.98Accusative plural masculine $-\bar{a}n$ § 8.99Accusative plural masculine $-\bar{a}na$ § 8.99Nominative-accusative plural neuter \$ \$8.100Nominative-accusative plural neuter \$ \$8.101Nominative-accusative plural neuter $-\bar{a}$ § \$.101Nominative-accusative plural neuter $-\bar{a}$ § \$.102Nominative-accusative plural neuter $-\bar{a}$ § \$.104Nominative-accusative plural neuter $-\bar{a}$ § \$.104Nominative-accusative plural neuter $-\bar{a}$ § \$.105Nominative-accusative plural neuter $-\bar{a}$ § \$.105Nominative-accusative plural neuter $-\bar{a}$ § \$.106	57 57 57 57 57 57 57 57 57 57 57 57 57 5
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h$ § 8.93Accusative plural masculine $-a$ § 8.94Accusative plural masculine $-i$ § 8.95Accusative plural masculine $-i$ § 8.96Accusative plural masculine $-\bar{a}i$ § 8.97Accusative plural masculine $-\bar{a}i$ § 8.98Accusative plural masculine $-\bar{a}ni$ § 8.98Accusative plural masculine $-\bar{a}ni$ § 8.99Nominative-accusative plural meuter § 8.100Nominative-accusative plural neuter $-\bar{a}$ ($-\bar{a}hi$)§ 8.100Nominative-accusative plural neuter $-a$ § 8.101Nominative-accusative plural neuter $-a$ § 8.102Nominative-accusative plural neuter $-a$ § 8.103Nominative-accusative plural neuter $-a$ § 8.104Nominative-accusative plural neuter $-a$ § 8.104Nominative-accusative plural neuter $-a$ § 8.105Nominative-accusative plural neuter $-a\bar{a}n\bar{s}$ § 8.107Nominative-accusative plural neuter $-a\bar{a}n\bar{s}$ § 8.107	57 57 57 57 57 57 58
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h$ § 8.93Accusative plural masculine $-a$ § 8.94Accusative plural masculine $-e$ § 8.95Accusative plural masculine $-i$ § 8.96Accusative plural masculine $-i$ § 8.97Accusative plural masculine $-\bar{a}n$ § 8.98Accusative plural masculine $-\bar{a}n$ § 8.98Accusative plural masculine $-\bar{a}na$ § 8.99Nominative-accusative plural neuter \$ \$ 8.100Nominative-accusative plural neuter $-\bar{a}$ ($-\bar{a}h$?)§ 8.100Nominative-accusative plural neuter $-a$ § 8.101Nominative-accusative plural neuter $-a$ § 8.102Nominative-accusative plural neuter $-a$ § 8.103Nominative-accusative plural neuter $-u$ § 8.104Nominative-accusative plural neuter $-\bar{a}n$ § 8.105Nominative-accusative plural neuter $-\bar{a}n$ § 8.105Nominative-accusative plural neuter $-\bar{a}n$ § 8.104Nominative-accusative plural neuter $-\bar{a}n$ § 8.105Nominative-accusative plural neuter $-\bar{a}n$ § 8.104Nominative-accusative plural neuter $-\bar{a}n$ § 8.105Nominative-accusative plural neuter $-\bar{a}n$ § 8.104Nominative-accusative plural neuter $-\bar{a}n$ § 8.105Nominative-accusative plural neuter $-\bar{a}n$ § 8.106Instrumental plural § 8.107Instrumental plural $-eh$ § 8.108	57 57 57 57 57 57 58
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h$, § 8.93Accusative plural masculine $-a$ § 8.94Accusative plural masculine $-e$ § 8.95Accusative plural masculine $-i$ § 8.96Accusative plural masculine $-\bar{a}$ § 8.97Accusative plural masculine $-\bar{a}n$ § 8.98Accusative plural masculine $-\bar{a}n$ § 8.99Accusative plural masculine $-\bar{a}na$? § 8.99]Nominative-accusative plural neuter \$ \$8.100Nominative-accusative plural neuter $-\bar{a}$ ($-\bar{a}h$?)§ 8.100Nominative-accusative plural neuter $-a$ § 8.101Nominative-accusative plural neuter $-a$ § 8.102Nominative-accusative plural neuter $-a$ § 8.104Nominative-accusative plural neuter $-u$ § 8.104Nominative-accusative plural neuter $-\bar{a}n$ § 8.105Nominative-accusative plural neuter $-\bar{a}n$ § 8.105Nominative-accusative plural neuter $-\bar{a}n$ § 8.104Nominative-accusative plural neuter $-\bar{a}n$ § 8.105Nominative-accusative plural neuter $-\bar{a}n$ § 8.106Instrumental plural $-eh$ § 8.107Instrumental plural $-eh$ § 8.108Instrumental plural $-eh$ § 8.109	57 57 57 57 57 58 59 59 59 59 59 79
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h$ § 8.93Accusative plural masculine $-a$ § 8.94Accusative plural masculine $-e$ § 8.95Accusative plural masculine $-i$ § 8.96Accusative plural masculine $-i$ § 8.97Accusative plural masculine $-\bar{a}n$ § 8.98Accusative plural masculine $-\bar{a}n$ § 8.98Accusative plural masculine $-\bar{a}n$ § 8.99Nominative-accusative plural neuter $\$$ 8.100Nominative-accusative plural neuter $\$$ 8.101Nominative-accusative plural neuter $-\bar{a}$ § 8.101Nominative-accusative plural neuter $-a$ § 8.101Nominative-accusative plural neuter $-a$ § 8.102Nominative-accusative plural neuter $-a$ § 8.104Nominative-accusative plural neuter $-a$ § 8.104Nominative-accusative plural neuter $-a\bar{n}$ § 8.105Nominative-accusative plural neuter $-$	57 57 57 57 57 58 59
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h$ § 8.93Accusative plural masculine $-a$ § 8.94Accusative plural masculine $-i$ § 8.95Accusative plural masculine $-i$ § 8.96Accusative plural masculine $-\bar{a}i$ § 8.97Accusative plural masculine $-\bar{a}i$ § 8.98Accusative plural masculine $-\bar{a}i$ § 8.98Accusative plural masculine $-\bar{a}i$ § 8.99Nominative-accusative plural neuter § 8.100Nominative-accusative plural neuter $-\bar{a}$ ($-\bar{a}h^2$)§ 8.100Nominative-accusative plural neuter $-a$ § 8.101Nominative-accusative plural neuter $-a$ § 8.102Nominative-accusative plural neuter $-a$ § 8.103Nominative-accusative plural neuter $-a$ § 8.104Nominative-accusative plural neuter $-a$ § 8.104Nominative-accusative plural neuter $-an$ § 8.105Nominative-accusative plural neuter $-an$ § 8.104Nominative-accusative plural neuter $-an$ § 8.105Nominative-accusative plural neuter $-an$ § 8.104Nominative-accusative plural neuter $-an$ § 8.104Nominative-accusative plural neuter $-an$ § 8.104Nominative-accusative plural neuter $-an$ § 8.105Nominative-accusative plural neuter $-an$ § 8.104Instrumental plural $-ai$ § 8.107Instrumental plural $-ai$ § 8.108Instrumental plural $-ehi$ § 8.108Instrumental plural $-ehi$ § 8.108Instrumental plural $-ehi$ § 8.110Instrumental plural $-ehi$ § 8.111	57 57 57 57 58 59 50
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h$ § 8.93Accusative plural masculine $-a$ § 8.94Accusative plural masculine $-i$ § 8.95Accusative plural masculine $-i$ § 8.96Accusative plural masculine $-\bar{a}i$ § 8.97Accusative plural masculine $-\bar{a}i$ § 8.98Accusative plural masculine $-\bar{a}i$ § 8.98Accusative plural masculine $-\bar{a}i$ § 8.99Nominative-accusative plural neuter § 8.100Nominative-accusative plural neuter $-\bar{a}$ ($-\bar{a}h^2$)§ 8.100Nominative-accusative plural neuter $-a$ § 8.101Nominative-accusative plural neuter $-a$ § 8.102Nominative-accusative plural neuter $-a$ § 8.103Nominative-accusative plural neuter $-a$ § 8.104Nominative-accusative plural neuter $-a$ § 8.104Nominative-accusative plural neuter $-an$ § 8.105Nominative-accusative plural neuter $-an$ § 8.104Nominative-accusative plural neuter $-an$ § 8.105Nominative-accusative plural neuter $-an$ § 8.104Nominative-accusative plural neuter $-an$ § 8.104Nominative-accusative plural neuter $-an$ § 8.104Nominative-accusative plural neuter $-an$ § 8.105Nominative-accusative plural neuter $-an$ § 8.104Instrumental plural $-ai$ § 8.107Instrumental plural $-ai$ § 8.108Instrumental plural $-ehi$ § 8.108Instrumental plural $-ehi$ § 8.108Instrumental plural $-ehi$ § 8.110Instrumental plural $-ehi$ § 8.111	57 57 57 57 58 59 50
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h$ § 8.93Accusative plural masculine $-a$ § 8.94Accusative plural masculine $-a$ § 8.95Accusative plural masculine $-i$ § 8.96Accusative plural masculine $-\bar{a}i$ § 8.97Accusative plural masculine $-\bar{a}i$ § 8.98Accusative plural masculine $-\bar{a}ni$ § 8.98Accusative plural masculine $-\bar{a}ni$ § 8.99Nominative-accusative plural neuter § 8.100Nominative-accusative plural neuter $-\bar{a}$ ($-\bar{a}hi$)§ 8.100Nominative-accusative plural neuter $-a$ § 8.101Nominative-accusative plural neuter $-a$ § 8.102Nominative-accusative plural neuter $-a$ § 8.104Nominative-accusative plural neuter $-a\bar{n}i$ § 8.105Nominative-accusative plural neuter $-\bar{a}ni$ § 8.104Nominative-accusative plural neuter $-\bar{a}ni$ § 8.105Nominative-accusative plural neuter $-\bar{a}ni$ § 8.104Nominative-accusative plural neuter $-\bar{a}ni$ § 8.105Nominative-accusative plural neuter $-\bar{a}ni$ § 8.104Instrumental plural $-ai$ § 8.107Instrumental plural $-ai$ § 8.108Instrumental plural $-ehi$ § 8.109Instrumental plural $-ehi$ § 8.101Instrumental plural $-ehi$ § 8.110Instrumental plural $-ehi$ § 8.111Instrumental plural $-ehi$ § 8.112	575575588 5585555555555555555555555555555555555
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h$, § 8.93Accusative plural masculine $-a$ § 8.94Accusative plural masculine $-a$ § 8.95Accusative plural masculine $-i$ § 8.96Accusative plural masculine $-i$ § 8.96Accusative plural masculine $-\bar{a}ni$ § 8.98Accusative plural masculine $-\bar{a}ni$ § 8.98Accusative plural masculine $-\bar{a}nai$ § 8.99Nominative-accusative plural neuter \$ \$ 0.00Nominative-accusative plural neuter \$ \$ 0.00Nominative-accusative plural neuter $-\bar{a}$ ($-\bar{a}hi$)§ 8.100Nominative-accusative plural neuter $-a$ § 8.101Nominative-accusative plural neuter $-a$ § 8.103Nominative-accusative plural neuter $-a$ § 8.104Nominative-accusative plural neuter $-\bar{a}ni$ § 8.105Nominative-accusative plural neuter $-\bar{a}ni$ § 8.104Nominative-accusative plural neuter $-\bar{a}ni$ § 8.105Nominative-accusative plural neuter $-\bar{a}ni$ § 8.104Instrumental plural \$ 8.107Instrumental plural $-ai$ § 8.108Instrumental plural $-ehi$ § 8.109Instrumental plural $-ehi$ § 8.110Instrumental plural $-ehi$ § 8.111Instrumental plural $-ehi$ § 8.112Instrumental plural $-abhi$ § 8.113	575575588 5585555555555555555555555555555555555
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h$, § 8.93Accusative plural masculine $-a$ § 8.94Accusative plural masculine $-e$ § 8.95Accusative plural masculine $-i$ § 8.96Accusative plural masculine $-\bar{a}$ § 8.97Accusative plural masculine $-\bar{a}n$ § 8.98Accusative plural masculine $-\bar{a}n$ § 8.99Accusative plural masculine $-\bar{a}na$ § 8.99Nominative-accusative plural neuter \$ \$8.100Nominative-accusative plural neuter $-\bar{a}$ ($-\bar{d}h$?)§ 8.100Nominative-accusative plural neuter $-a$ § 8.101Nominative-accusative plural neuter $-a$ § 8.102Nominative-accusative plural neuter $-a$ § 8.104Nominative-accusative plural neuter $-u$ § 8.104Nominative-accusative plural neuter $-\bar{a}n$ § 8.105Nominative-accusative plural neuter $-\bar{a}n$ § 8.104Nominative-accusative plural neuter $-\bar{a}n$ § 8.105Nominative-accusative plural neuter $-\bar{a}n$ § 8.104Nominative-accusative plural neuter $-\bar{a}n$ § 8.105Nominative-accusative plural neuter $-\bar{a}n$ § 8.104Nominative-accusative plural neuter $-\bar{a}n$ § 8.104Nominative-accusative plural neuter $-\bar{a}n$ § 8.105Nominative-accusative plural neuter $-\bar{a}n$ § 8.106Instrumental plural $-ehi$ § 8.107Instrumental plural $-ehi$ § 8.108Instrumental plural $-ehi$ § 8.109Instrumental plural $-ehi$ § 8.110Instrumental plural $-ehhi$ § 8.111Instrumental plural $-ehhi$ § 8.113Instrumental plural $-ehhi$ § 8.113Instrumental plural $-ibhis$ § 8.114	57557558855555555555555555555555555555
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h$ § 8.93 Accusative plural masculine $-a$ § 8.94 Accusative plural masculine $-i$ § 8.95 Accusative plural masculine $-i$ § 8.96 Accusative plural masculine $-\bar{a}ni$ § 8.97 Accusative plural masculine $-\bar{a}ni$ § 8.98 [Accusative plural masculine $-\bar{a}ni$ § 8.99] Nominative-accusative plural neuter § 8.100 Nominative-accusative plural neuter $-\bar{a}$ ($-\bar{a}h^2$) § 8.100 Nominative-accusative plural neuter $-\bar{a}$ § 8.101 Nominative-accusative plural neuter $-\bar{a}m$ § 8.102. Nominative-accusative plural neuter $-\bar{a}m$ § 8.102. Nominative-accusative plural neuter $-\bar{a}m$ § 8.104. Nominative-accusative plural neuter $-\bar{a}m$ § 8.104. Nominative-accusative plural neuter $-\bar{a}ni$ § 8.105. Nominative-accusative plural neuter $-\bar{a}ni$ § 8.104. Nominative-accusative plural neuter $-\bar{a}ni$ § 8.105. Nominative-accusative plural neuter $-\bar{a}ni$ § 8.105. Nominative-accusative plural neuter $-\bar{a}ni$ § 8.106. Instrumental plural $-ai$ § 8.107 Instrumental plural $-ehi$ § 8.108 Instrumental plural $-ehi$ § 8.110 Instrumental plural $-ehi$ § 8.111. Instrumental plural $-ehi$ § 8.112 Instrumental plural $-ebhi$ § 8.113 Instrumental plural $-ibhi$ § 8.113 Instrumental plural $-ibhi$ § 8.113 Instrumental plural $-ibhi$ § 8.113	57755755885555555555555555555555555555
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h$ § 8.93 Accusative plural masculine $-a$ § 8.94 Accusative plural masculine $-a$ § 8.95 Accusative plural masculine $-i$ § 8.96 Accusative plural masculine $-\bar{a}n$ § 8.97 Accusative plural masculine $-\bar{a}n$ § 8.98 Accusative plural masculine $-\bar{a}n$ § 8.99 Accusative plural masculine $-\bar{a}n$ § 8.99 Nominative-accusative plural neuter § 8.100 Nominative-accusative plural neuter $-\bar{a}$ ($-\bar{a}h$?) § 8.100 Nominative-accusative plural neuter $-\bar{a}$ § 8.101 Nominative-accusative plural neuter $-\bar{a}$ § 8.102 Nominative-accusative plural neuter $-a$ § 8.104 Nominative-accusative plural neuter $-a\bar{m}$ § 8.102 Nominative-accusative plural neuter $-\bar{a}n\bar{s}$ § 8.105 Nominative-accusative plural neuter $-\bar{a}n\bar{s}$ § 8.105 Nominative-accusative plural neuter $-\bar{a}n\bar{s}$ § 8.105 Nominative-accusative plural neuter $-\bar{a}n\bar{s}$ § 8.105 Instrumental plural $-ai\bar{s}$ § 8.107 Instrumental plural $-ehi\bar{s}$ § 8.108 Instrumental plural $-ehi\bar{s}$ § 8.108 Instrumental plural $-ehi\bar{s}$ § 8.110 Instrumental plural $-ehi\bar{s}$ § 8.110 Instrumental plural $-ehi\bar{s}$ § 8.110 Instrumental plural $-ehi\bar{s}$ § 8.112 Instrumental plural $-ehi\bar{s}$ § 8.113 Instrumental plural $-abhi\bar{s}$ § 8.114 Instrumental plural $-abhi\bar{s}$ § 8.115 Dative-ablative plural $-\bar{a}hi\bar{s}$ § 8.116	57 57 57 57 57 58 58 58 588 599 999 590 500 5
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h$ § 8.93 Accusative plural masculine $-a$ § 8.94 Accusative plural masculine $-a$ § 8.95 Accusative plural masculine $-i$ § 8.96 Accusative plural masculine $-\bar{a}n$ § 8.97 Accusative plural masculine $-\bar{a}n$ § 8.98 Accusative plural masculine $-\bar{a}n$ § 8.99 Accusative plural masculine $-\bar{a}n$ § 8.99 Nominative-accusative plural neuter § 8.100 Nominative-accusative plural neuter $-\bar{a}$ ($-\bar{a}h$?) § 8.100 Nominative-accusative plural neuter $-\bar{a}$ § 8.101 Nominative-accusative plural neuter $-\bar{a}$ § 8.102 Nominative-accusative plural neuter $-a$ § 8.104 Nominative-accusative plural neuter $-a\bar{m}$ § 8.102 Nominative-accusative plural neuter $-\bar{a}n\bar{s}$ § 8.105 Nominative-accusative plural neuter $-\bar{a}n\bar{s}$ § 8.105 Nominative-accusative plural neuter $-\bar{a}n\bar{s}$ § 8.105 Nominative-accusative plural neuter $-\bar{a}n\bar{s}$ § 8.105 Instrumental plural $-ai\bar{s}$ § 8.107 Instrumental plural $-ehi\bar{s}$ § 8.108 Instrumental plural $-ehi\bar{s}$ § 8.108 Instrumental plural $-ehi\bar{s}$ § 8.110 Instrumental plural $-ehi\bar{s}$ § 8.110 Instrumental plural $-ehi\bar{s}$ § 8.110 Instrumental plural $-ehi\bar{s}$ § 8.112 Instrumental plural $-ehi\bar{s}$ § 8.113 Instrumental plural $-abhi\bar{s}$ § 8.114 Instrumental plural $-abhi\bar{s}$ § 8.115 Dative-ablative plural $-\bar{a}hi\bar{s}$ § 8.116	57 57 57 57 57 58 58 58 588 599 999 590 500 5
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h$ § 8.93 Accusative plural masculine $-a$ § 8.94 Accusative plural masculine $-a$ § 8.95 Accusative plural masculine $-i$ § 8.96 Accusative plural masculine $-\bar{a}ni$ § 8.97 Accusative plural masculine $-\bar{a}ni$ § 8.98 Accusative plural masculine $-\bar{a}ni$ § 8.99 Accusative plural masculine $-\bar{a}ni$ § 8.99 Nominative-accusative plural neuter § 8.100 Nominative-accusative plural neuter $-\bar{a}$ ($-\bar{a}hi$?) § 8.100 Nominative-accusative plural neuter $-a$ § 8.101 Nominative-accusative plural neuter $-a$ § 8.102 Nominative-accusative plural neuter $-a$ § 8.104 Nominative-accusative plural neuter $-a\bar{n}n$ § 8.105 Nominative-accusative plural neuter $-\bar{a}n\bar{n}$ § 8.104 Nominative-accusative plural neuter $-\bar{a}n\bar{n}$ § 8.105 Nominative-accusative plural neuter $-\bar{a}n\bar{n}$ § 8.105 Nominative-accusative plural neuter $-\bar{a}n\bar{n}$ § 8.105 Instrumental plural $-a\bar{n}$ § 8.107 Instrumental plural $-eh\bar{n}$ § 8.108 Instrumental plural $-eh\bar{n}$ § 8.108 Instrumental plural $-eh\bar{n}$ § 8.108 Instrumental plural $-eh\bar{n}$ § 8.110 Instrumental plural $-ehh\bar{n}$ § 8.112 Instrumental plural $-ehh\bar{n}$ § 8.113 Instrumental plural $-ehh\bar{n}$ § 8.114 Instrumental plural $-abh\bar{n}$ § 8.115 Dative-ablative plural § 8.116 Genitive plural § 8.117	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h$, § 8.93Accusative plural masculine $-a$ § 8.94Accusative plural masculine $-a$ § 8.95Accusative plural masculine $-i$ § 8.96Accusative plural masculine $-\bar{a}i$ § 8.97Accusative plural masculine $-\bar{a}i$ § 8.98Accusative plural masculine $-\bar{a}ni$ § 8.98Accusative plural masculine $-\bar{a}ni$ § 8.99Nominative-accusative plural neuter \$ \$ 8.100Nominative-accusative plural neuter $-\bar{a}$ ($-\bar{d}h$?)§ 8.100Nominative-accusative plural neuter $-a$ § 8.101Nominative-accusative plural neuter $-a$ § 8.103Nominative-accusative plural neuter $-a$ § 8.104Nominative-accusative plural neuter $-a\bar{n}i$ § 8.105Nominative-accusative plural neuter $-a\bar{n}i$ § 8.104Nominative-accusative plural neuter $-\bar{a}n\bar{i}$ § 8.105Nominative-accusative plural neuter $-\bar{a}n\bar{i}$ § 8.104Nominative-accusative plural neuter $-\bar{a}n\bar{i}$ § 8.105Nominative-accusative plural neuter $-\bar{a}n\bar{i}$ § 8.104Instrumental plural $-ai$ § 8.107Instrumental plural $-ai$ § 8.107Instrumental plural $-ehi$ § 8.108Instrumental plural $-ehi$ § 8.109Instrumental plural $-ehi$ § 8.110Instrumental plural $-ehi$ § 8.111Instrumental plural $-ehi$ § 8.112Instrumental plural $-abhi$ § 8.113Instrumental plural $-abhi$ § 8.114Instrumental plural $-abhi$ § 8.115Dative-ablative plural § 8.117Genitive plural § 8.117	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h$ § 8.93Accusative plural masculine $-a$ § 8.94Accusative plural masculine $-i$ § 8.95Accusative plural masculine $-i$ § 8.96Accusative plural masculine $-\bar{a}n\bar{a}$ § 8.97Accusative plural masculine $-\bar{a}n\bar{a}$ § 8.98Accusative plural masculine $-\bar{a}n\bar{a}$ § 8.99Accusative plural masculine $-\bar{a}n\bar{a}$ § 8.99Nominative-accusative plural neuter § 8.100Nominative-accusative plural neuter $-\bar{a}m$ § 8.101Nominative-accusative plural neuter $-\bar{a}m$ § 8.102Nominative-accusative plural neuter $-\bar{a}m$ § 8.104Nominative-accusative plural neuter $-\bar{a}n\bar{a}$ § 8.105Nominative-accusative plural neuter $-\bar{a}n\bar{a}$ § 8.106Instrumental plural $-ai$ § 8.107Instrumental plural $-ai$ § 8.108Instrumental plural $-eh\bar{h}$ § 8.109Instrumental plural $-eh\bar{b}$ § 8.113Instrumental plural $-eh\bar{b}$ § 8.113Instrumental plural $-ebh\bar{s}$ § 8.113Instrumental plural $-ebh\bar{s}$ § 8.113Instrumental plural $-abh\bar{s}$ § 8.113Instrumental plural $-abh\bar{s}$ § 8.113Instrumental plural $-abh\bar{s}$ § 8.113Genitive plural § 8.117Genitive plural § 8.117Genitive plural $-\bar{a}nam, -\bar{a}nam, § 8.120$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h$ § 8.93Accusative plural masculine $-a$ § 8.94Accusative plural masculine $-i$ § 8.95Accusative plural masculine $-i$ § 8.96Accusative plural masculine $-\bar{a}ni$ § 8.97Accusative plural masculine $-\bar{a}ni$ § 8.98[Accusative plural masculine $-\bar{a}ni$ § 8.99Nominative-accusative plural masculine $-\bar{a}ni$ § 8.99]Nominative-accusative plural neuter § 8.100Nominative-accusative plural neuter - \bar{a} ($-\bar{a}h^2$)§ 8.100Nominative-accusative plural neuter $-a$ § 8.101Nominative-accusative plural neuter $-a$ § 8.102Nominative-accusative plural neuter $-a$ § 8.103Nominative-accusative plural neuter $-a$ § 8.104Nominative-accusative plural neuter $-a$ § 8.104Nominative-accusative plural neuter $-ani$ § 8.105Nominative-accusative plural neuter $-ani$ § 8.104Nominative-accusative plural neuter $-ani$ § 8.105Nominative-accusative plural neuter $-ani$ § 8.104Nominative-accusative plural neuter $-ani$ § 8.105Nominative-accusative plural neuter $-ani$ § 8.104Nominative-accusative plural neuter $-ani$ § 8.105Instrumental plural $-ai$ § 8.107Instrumental plural $-ahi$ § 8.109Instrumental plural $-ahi$ § 8.109Instrumental plural $-ahi$ § 8.110Instrumental plural $-ehi$ § 8.110Instrumental plural $-bhi$ § 8.113Instrumental plural $-bhi$ § 8.114Instrumental plural $-bhi$ § 8.115Dative-ablative plural $-\bar{a}na$ § 8.117Genitive plural $-anan$ $-anan$ § 8.120Genitive plural $-anan$ § 8.12	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h$ § 8.93Accusative plural masculine $-a$ § 8.94Accusative plural masculine $-a$ § 8.95Accusative plural masculine $-i$ § 8.96Accusative plural masculine $-\bar{a}i$ § 8.97Accusative plural masculine $-\bar{a}i$ § 8.97Accusative plural masculine $-\bar{a}i$ § 8.98Accusative plural masculine $-\bar{a}i$ § 8.99Nominative-accusative plural neuter § 8.100Nominative-accusative plural neuter $-\bar{a}i$ § 8.101Nominative-accusative plural neuter $-a$ § 8.101Nominative-accusative plural neuter $-a$ § 8.102Nominative-accusative plural neuter $-a$ § 8.103Nominative-accusative plural neuter $-ain$ § 8.102Nominative-accusative plural neuter $-ain$ § 8.104Nominative-accusative plural neuter $-ain$ § 8.105Nominative-accusative plural neuter $-ain$ § 8.104Nominative-accusative plural neuter $-ain$ § 8.105Nominative-accusative plural neuter $-ain$ § 8.106Instrumental plural $-ai$ § 8.107Instrumental plural $-ai$ § 8.108Instrumental plural $-ehi$ § 8.108Instrumental plural $-ehi$ § 8.110Instrumental plural $-ehi$ § 8.111Instrumental plural $-ehi$ § 8.112Instrumental plural $-ehi$ § 8.113Instrumental plural $-abhi$ § 8.114Instrumental plural $-abhi$ § 8.115Dative-ablative plural $-aina$ § 8.120Genitive plural $-aina$ § 8.120Genitive plural $-aina$ § 8.123Genitive plural $-aina$ § 8.123	$\begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h$ § 8.93Accusative plural masculine $-a$ § 8.94Accusative plural masculine $-a$ § 8.95Accusative plural masculine $-i$ § 8.96Accusative plural masculine $-\bar{a}i$ § 8.97Accusative plural masculine $-\bar{a}i$ § 8.97Accusative plural masculine $-\bar{a}i$ § 8.98Accusative plural masculine $-\bar{a}i$ § 8.99Nominative-accusative plural neuter § 8.100Nominative-accusative plural neuter $-\bar{a}i$ § 8.101Nominative-accusative plural neuter $-a$ § 8.101Nominative-accusative plural neuter $-a$ § 8.102Nominative-accusative plural neuter $-a$ § 8.103Nominative-accusative plural neuter $-ain$ § 8.102Nominative-accusative plural neuter $-ain$ § 8.104Nominative-accusative plural neuter $-ain$ § 8.105Nominative-accusative plural neuter $-ain$ § 8.104Nominative-accusative plural neuter $-ain$ § 8.105Nominative-accusative plural neuter $-ain$ § 8.106Instrumental plural $-ai$ § 8.107Instrumental plural $-ai$ § 8.108Instrumental plural $-ehi$ § 8.108Instrumental plural $-ehi$ § 8.110Instrumental plural $-ehi$ § 8.111Instrumental plural $-ehi$ § 8.112Instrumental plural $-ehi$ § 8.113Instrumental plural $-abhi$ § 8.114Instrumental plural $-abhi$ § 8.115Dative-ablative plural $-aina$ § 8.120Genitive plural $-aina$ § 8.120Genitive plural $-aina$ § 8.123Genitive plural $-aina$ § 8.123	$\begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$
Accusative plural masculine $-\bar{a}s$, $-\bar{a}h$ § 8.93Accusative plural masculine $-a$ § 8.94Accusative plural masculine $-i$ § 8.95Accusative plural masculine $-i$ § 8.96Accusative plural masculine $-\bar{a}ni$ § 8.97Accusative plural masculine $-\bar{a}ni$ § 8.98[Accusative plural masculine $-\bar{a}ni$ § 8.99Nominative-accusative plural masculine $-\bar{a}ni$ § 8.99]Nominative-accusative plural neuter § 8.100Nominative-accusative plural neuter - \bar{a} ($-\bar{a}h^2$)§ 8.100Nominative-accusative plural neuter $-a$ § 8.101Nominative-accusative plural neuter $-a$ § 8.102Nominative-accusative plural neuter $-a$ § 8.103Nominative-accusative plural neuter $-a$ § 8.104Nominative-accusative plural neuter $-a$ § 8.104Nominative-accusative plural neuter $-ani$ § 8.105Nominative-accusative plural neuter $-ani$ § 8.104Nominative-accusative plural neuter $-ani$ § 8.105Nominative-accusative plural neuter $-ani$ § 8.104Nominative-accusative plural neuter $-ani$ § 8.105Nominative-accusative plural neuter $-ani$ § 8.104Nominative-accusative plural neuter $-ani$ § 8.105Instrumental plural $-ai$ § 8.107Instrumental plural $-ahi$ § 8.109Instrumental plural $-ahi$ § 8.109Instrumental plural $-ahi$ § 8.110Instrumental plural $-ehi$ § 8.110Instrumental plural $-bhi$ § 8.113Instrumental plural $-bhi$ § 8.114Instrumental plural $-bhi$ § 8.115Dative-ablative plural $-\bar{a}na$ § 8.117Genitive plural $-anan$ $-anan$ § 8.120Genitive plural $-anan$ § 8.12	$\begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$

Locative plural § 8.126	60
Locative plural -isu, -asu § 8.126	60
Locative plural -esű § 8.127	60
9. ā-stems	61
Stem form in composition: -a § 9.6	61
Nominative singular § 9.8	61
Nominative singular $-a \notin 9.8$	61
Nominative singular -ās § 9.10	61
Nominative singular -ūm § 9.11	61
Nominative singular -u § 9.13	62
Vocative singular § 9.14	62
Vocative singular $-i$ for $-e \S 9.14 \dots$	62
Vocative singular -a § 9.15	62
Accusative singular § 9.16	62
Accusative singular -am, -am § 9.16	62
Accusative singular $-\alpha $ § 9.19	62
Accusative singular $-\bar{a} \S 9.20$	62
Accusative singular $-u$ § 9.23	63
Accusative singular $-\bar{a}n$, for $-\bar{a}m \S 9.24$	63
Accusative singular -āram § 9.25	63
Oblique cases of the singular § 9.26	63
Oblique singular -āye and congeners § 9.31	63
-ūe § 9.31	63
-āyi § 9.32	63
-aye § 9.33	63
-āyai § 9.34 -āyc, outside of Mv § 9.35	63
	63 64
-āye in Mv, Instr., Dat., Abl., Gen., Loc. § 9.36 Oblique singular -āya and congeners § 9.42	64
-ayā, Abl., Gen., Loc. § 9.43	64
-āyā, Instr., Gen., Loc. § 9.47	64
-ayam, Instr. § 9.51	64
-äyäm, Abl. or Gen. § 9.52	64
	01
- <i>āua</i> in prose other than My, Instr., Dat., Gen.,	
-āya in prose, other than Mv, Instr., Dat., Gen., Loc. § 9.53	64
Loc. § 9.53	64
Loc. § 9.53 - <i>āya</i> in verses, and Mv prose, Instr., Dat., Abl.,	64 65
I.oc. § 9.53 9.53 -āya in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58 Gen., Loc. § 9.58 9.58 Oblique singular -ā § 9.64 -ā	_
Loc. § 9.53 -āya in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58	65 65 65
Loc. § 9.53 - $\bar{a}ya$ in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58 Oblique singular - \bar{a} § 9.64 Instrumental § 9.65 Ablative § 9.66	65 65 65 65
Loc. § 9.53 . - $\bar{a}ya$ in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58 . Oblique singular - \bar{a} § 9.64 . Instrumental § 9.65 . Ablative § 9.66 . Genitive § 9.67 .	65 65 65 65 65
Loc. § 9.53 . - $\bar{a}ya$ in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58 . Oblique singular - \bar{a} § 9.64 . Instrumental § 9.65 . Ablative § 9.66 . Genitive § 9.67 . Locative § 9.68 .	65 65 65 65 65 65 65
Loc. § 9.53 -āya in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58 Oblique singular -ā § 9.64 Instrumental § 9.65 Ablative § 9.66 Genitive § 9.66 Locative § 9.68 Special forms of oblique cases § 9.69	65 65 65 65 65 65 65
Loc. § 9.53 -āya in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58 Oblique singular -ā § 9.64 Instrumental § 9.65 Ablative § 9.66 Genitive § 9.66 Locative § 9.68 Special forms of oblique cases § 9.69 Instrumental -ena, -ayena § 9.69	65 65 65 65 65 65 65 65
Loc. § 9.53 - $\bar{a}ya$ in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58 Oblique singular - \bar{a} § 9.64 Instrumental § 9.66 Ablative § 9.66 Genitive § 9.68 Locative § 9.68 Special forms of oblique cases § 9.69 Instrumental - $ay\bar{a}ye$ § 9.70	65 65 65 65 65 65 65 65 65
Loc. § 9.53 $-\bar{a}ya$ in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58Oblique singular $-\bar{a}$ § 9.64Instrumental § 9.65Ablative § 9.66Genitive § 9.67Locative § 9.68Special forms of oblique cases § 9.69Instrumental -ena, -ayena § 9.69Instrumental -qyäye § 9.70Ablative - at § 9.71	65 65 65 65 65 65 65 65 65 65
Loc. § 9.53 -āya in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58. Oblique singular -ā § 9.64. Instrumental § 9.65 Ablative § 9.66 Genitive § 9.68. Special forms of oblique cases § 9.69. Instrumental -ena, -ayena § 9.69. Instrumental -ayäye § 9.70. Ablative -āl § 9.71. Ablative -ālas, -āta, -atas, -atu § 9.72.	65 65 65 65 65 65 65 65 65 65
Loc. § 9.53 . - aya in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58 Oblique singular - a § 9.64 . Instrumental § 9.65 Ablative § 9.66 . Genitive § 9.68 Locative § 9.68 Instrumental -ena, -ayena § 9.69 Instrumental -ena, -ayena § 9.69 Ablative - ait § 9.71 Ablative - ait § 9.72 Genitive -asyá (pronom.) § 9.74	65 65 65 65 65 65 65 65 65 65 65 65
Loc. § 9.53 . -āya in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58 Oblique singular -ā § 9.64 . Instrumental § 9.65 Ablative § 9.66 . Genitive § 9.68 . Special forms of oblique cases § 9.69 Instrumental -ena, -ayena § 9.69 Instrumental -ayāye § 9.70 . Ablative -āt § 9.71 Ablative -ātas, -āta, -atas, -atu § 9.72 Genitive -asyā (pronom.) § 9.74 Genitive -asya § 9.75	$\begin{array}{c} 65\\ 65\\ 65\\ 65\\ 65\\ 65\\ 65\\ 65\\ 65\\ 65\\$
Loc. § 9.53 . - $\bar{a}ya$ in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58 Oblique singular - \bar{a} § 9.64 Instrumental § 9.65 Ablative § 9.66 Genitive § 9.67 Locative § 9.68 Special forms of oblique cases § 9.69 Instrumental - ana , - $ayena$ § 9.69 Instrumental - $an\bar{a}$, - $ayena$ § 9.69 Instrumental - $an\bar{a}$, - $ayena$ § 9.69 Ablative - $\bar{a}t$ § 9.71 Ablative - $\bar{a}tas$, - $\bar{a}ta$, - $atas$, - ata § 9.72 Genitive - $asya$ § 9.75 Locative - e § 9.77	$65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\$
Loc. § 9.53 - $\bar{a}ya$ in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58 Oblique singular - \bar{a} § 9.64 Instrumental § 9.66 Genitive § 9.66 Genitive § 9.67 Locative § 9.68 Special forms of oblique cases § 9.69 Instrumental - $ay\bar{a}ye$ § 9.70 Ablative - $\bar{a}t$ § 9.71 Ablative - $\bar{a}ts$, - $\bar{a}ta$, - $atas$, - atu § 9.72 Genitive - $asya$ (pronom.) § 9.74 Genitive - $asya$ § 9.75 Locative - e § 9.77 Locative - e § 9.77	65 65 65 65 65 65 65 65 65 65 66 66 66 6
Loc. § 9.53 . - aya in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58 . Oblique singular - a § 9.64 . Instrumental § 9.65 . Ablative § 9.66 . Genitive § 9.68 . Special forms of oblique cases § 9.69 . Instrumental - $ayaya$ § 9.69 . Instrumental - $ayaya$ § 9.70 . Ablative - $aita$ § 9.71 . Ablative - $aita$ § 9.71 . Ablative - $aaya$ (pronom.) § 9.74 . Genitive - $asya$ § 9.75 . Locative - $ayam$ § 9.79 . Locative - $ayam$ § 9.80 .	65 65 65 65 65 65 65 65 65 65 65 66 66 6
Loc. § 9.53 . -āya in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58 . Oblique singular -ā § 9.64 . Instrumental § 9.65 . Ablative § 9.66 . Genitive § 9.68 . Special forms of oblique cases § 9.69 . Instrumental -ayāye § 9.70 . Ablative -āl § 9.71 . Ablative -āl ş 9.71 . Ablative -asyā (pronom.) § 9.74 . Genitive -asyā § 9.75 . Locative -aşyān § 9.79 . Locative -āyām § 9.80 . Nominative-acusative plural § 9.81 .	$65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\$
Loc. § 9.53	65 65 65 65 65 65 65 65 65 66 66 66 66 6
Loc. § 9.53	65 65 65 65 65 65 65 65
Loc. § 9.53 . - aya in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58 . Oblique singular - \bar{a} § 9.64 . Instrumental § 9.65 . Ablative § 9.66 . Genitive § 9.68 . Special forms of oblique cases § 9.69 . Instrumental - $ayaya$ § 9.69 . Instrumental - $ayaya$ § 9.70 . Ablative - $aits$ 9.71 . Ablative - $aits$ 9.71 . Ablative - $aaya$ (pronom.) § 9.74 . Genitive - $asya$ § 9.75 . Locative - $asya$ § 9.75 . Locative - $ayam$ § 9.80 . Nominative-accusative plural \$ 9.81 . Nominative-accusative plural - a § 9.86	65 65 65 65 65 65 65 65
Loc. § 9.53 . - aya in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58 Oblique singular - a § 9.64 . Instrumental § 9.65 . Ablative § 9.66 . Genitive § 9.68 . Special forms of oblique cases § 9.69 Instrumental - $ayaya$ § 9.69 Ablative - ait § 9.71 Ablative - $aits$ 9.71 Ablative - $aits$ 9.75 Genitive - $asya$ (pronom.) § 9.74 Genitive - $asya$ § 9.75 Locative - e § 9.77 Locative - $ayam$ § 9.80 Nominative-accusative plural s 9.82 Nominative-accusative plural - a § 9.88 Nominative-accusative plural - a § 9.88 Nominative-accusative plural - a § 9.88	$65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\$
Loc. § 9.53	$65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\$
Loc. § 9.53	$65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\$
Loc. § 9.53 . -āya in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58 . Oblique singular -ā § 9.64 . Instrumental § 9.65 . Ablative § 9.66 . Genitive § 9.68 . Special forms of oblique cases § 9.69 . Instrumental -ena, -ayena § 9.70 . Ablative -âl § 9.71 . Ablative -alus, -âlo, -âla, -alas, -alu § 9.72 . Genitive -asya § 9.75 . Locative -asya § 9.75 . Locative -ayām § 9.79 . Locative -ayām § 9.80 . Nominative-accusative plural -ā § 9.82 . Nominative-accusative plural -ā § 9.88 Nominative-accusative plural -ā § 9.88 Nominative-accusative plural -ā § 9.88 Nominative-accusative plural -āye § 9.91 Nominative-accusative plural -āye § 9.92 Nominative-accusative plural -aye § 9.93 Nominative-accusative plural -ave § 9.93	$65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\$
Loc. § 9.53 . - aya in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58 Oblique singular - \bar{a} § 9.64 . Instrumental § 9.65 Ablative § 9.66 . Genitive § 9.68 Locative § 9.68 Special forms of oblique cases § 9.69 Instrumental - $ayaya$ § 9.70 . Ablative - $aits$, - $aita$, - $aits$, - $aits$ § 9.72 . Ablative - $aits$, - $idto$, - $aita$, - $aits$, - $aits$ § 9.72 . Genitive - $asya$ (pronom.) § 9.74 Locative - $agam$ § 9.79 . Locative - $agam$ § 9.79 . Locative - $agam$ § 9.80 Nominative-accusative plural $-a$ § 9.82 Nominative-accusative plural - a § 9.88 Nominative-accusative plural - a § 9.92 Nominative-accusative plural - aya § 9.91 Nominative-accusative plural - aya § 9.91 Nominative-accusative plural - aya § 9.93 Nominative-accusative plural - aya § 9.94 Nominative-accusative plural - aya § 9.94	$65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\$
Loc. § 9.53 . - aya in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58 Oblique singular - a § 9.64 . Instrumental § 9.65 . Ablative § 9.66 . Genitive § 9.68 . Special forms of oblique cases § 9.69 Instrumental - $ayaya$ § 9.70 . Ablative - $ails$ 9.71 Ablative - $ails$ 9.71 Ablative - $ails$ 9.75 Genitive - $asya$ (pronom.) § 9.74 Genitive - $asya$ § 9.75 Locative - sya § 9.77 Locative - $ayam$ § 9.80 Nominative-accusative plural - a § 9.82 Nominative-accusative plural - ay § 9.88 Nominative-accusative plural - $ayga$ § 9.93 Nominative-accusative plural - $ayga$ § 9.93 Nominative-accusative plural - $ayga$ § 9.97 Nominative-accusative plural - $ayga$ § 9.97 Nominative-accusative plural - $ayga$ § 9.92 Nominative-accusative plural - $ayga$ § 9.92 Nominative-accusative plural - $ayga$ § 9.92 Nominative-accusative plural - $ayga$ § 9.93 Nominative-accusative plural - $ayga$ § 9.97 Nominative-accusative plural - $ayga$ § 9.98	$65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\ 65 \\$
Loc. § 9.53 . - aya in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58 Oblique singular - \bar{a} § 9.64 . Instrumental § 9.65 Ablative § 9.66 . Genitive § 9.68 Locative § 9.68 Special forms of oblique cases § 9.69 Instrumental - $ayaya$ § 9.70 . Ablative - $aits$, - $aita$, - $aits$, - $aits$ § 9.72 . Ablative - $aits$, - $idto$, - $aita$, - $aits$, - $aits$ § 9.72 . Genitive - $asya$ (pronom.) § 9.74 Locative - $agam$ § 9.79 . Locative - $agam$ § 9.79 . Locative - $agam$ § 9.80 Nominative-accusative plural $-a$ § 9.82 Nominative-accusative plural - a § 9.88 Nominative-accusative plural - a § 9.92 Nominative-accusative plural - aya § 9.91 Nominative-accusative plural - aya § 9.91 Nominative-accusative plural - aya § 9.93 Nominative-accusative plural - aya § 9.94 Nominative-accusative plural - aya § 9.94	655 6655 66555 655555555555555555
Loc. § 9.53 . - aya in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58 . Oblique singular - \bar{a} § 9.64 . Instrumental § 9.65 . Ablative § 9.66 . Genitive § 9.68 . Special forms of oblique cases § 9.69 . Instrumental - ana , $-ayena$ § 9.69 . Instrumental - $ayayea$ § 9.70 . Ablative - al § 9.71 . Ablative - al § 9.71 . Ablative - ads 9.75 . Locative - $asya$ (pronom.) § 9.74 . Genitive - $asya$ § 9.75 . Locative - $ayam$ § 9.79 . Locative - $ayam$ § 9.80 . Nominative-accusative plural § 9.81 . Nominative-accusative plural - a § 9.86 Nominative-accusative plural - a § 9.88 Nominative-accusative plural - a § 9.88 Nominative-accusative plural - ayg § 9.91 Nominative-accusative plural - ayg § 9.92 Nominative-accusative plural - aye § 9.93 Nominative-accusative plural - ayo § 9.94 Nominative-accusative plural - ayo § 9.95	655 6655 66555 665555666666666666
Loc. § 9.53 . - aya in verses, and Mv prose, Instr., Dat., Abl., Gen., Loc. § 9.58 . Oblique singular - \bar{a} § 9.64 . Instrumental § 9.65 . Ablative § 9.66 . Genitive § 9.68 . Special forms of oblique cases § 9.69 . Instrumental - $ayayea$ § 9.69 . Instrumental - $ayayea$ § 9.70 . Ablative - $alas$, - $ayena$ § 9.69 . Instrumental - $ayayea$ § 9.70 . Ablative - $alas$, - $adas$, - $atas$, - $atas$ § 9.72 . Genitive - $asya$ (pronom.) § 9.74 Genitive - $asya$ (pronom.) § 9.74 Genitive - $asya$ § 9.75 . Locative - $ayam$ § 9.79 . Locative - $ayam$ § 9.79 . Locative - $ayam$ § 9.80 Nominative-accusative plural - a § 9.82 Nominative-accusative plural - a § 9.88 Nominative-accusative plural - $agya$ § 9.91 Nominative-accusative plural - $agya$ § 9.92 Nominative-accusative plural - $agya$ § 9.93 Nominative-accusative plural - $agya$ § 9.93 Nominative-accusative plural - $agya$ § 9.97 Nominative-accusative plural - $agya$ § 9.93 Nominative-accusative plural - $agya$ § 9.95 Nominative-accusative plural - $agya$ § 9.97 Nominative-accusative plural - $agya$ § 9.98 Nominative-accusative plural - $agya$ § 9.99 Nominative-accusative plural - $afya$ § 9.90 Nominative-accusative plural - $afya$ § 9.91 Nominative-accusative plural - $afya$ § 9.92 Nominative-accusative plural - $afya$ § 9.93 Nominative-accusative plural - $afya$ § 9.90 Nominative-accusative plural - $afya$ § 9.90 Nominative-accusative plural - $afya$ § 9.90	655 6655 66555 665555555555555555
Loc. § 9.53	655 6655 65555555555555555555555555
Loc. § 9.53	6655666666666666666666666666666666666
Loc. § 9.53	6556666666666666666666666666666666666

ХI

Table of Contents of Volume One

Genitive plural -āna § 9.106	68
Locative plural § 9.107	
Locative plural -asu § 9.107	68
Locative plural -āsū § 9.108	
•	
10. i-stems with i- and in-stems	69
Stem-extensions (in -a, etc.) § 10.3	69
Stem in composition: confusion of i and I § 10.9	69
i for i § 10.10	69
ī for i § 10.11	69
Nominative singular § 10.13.	70
Nominative singular -i § 10.15	70
Masculine <i>i</i> -stems § 10.15	70
Feminine <i>i</i> -stems § 10.16	70
Feminine <i>i</i> -stems, polysyllabic § 10.17 Feminine <i>i</i> -stems, monosyllabic § 10.18	70 70
Masculine <i>in-stems</i> § 10.19	70
Nominative singular -is, -ih, -ir, etc. § 10.20	70
Feminine i-stems, polysyllabic, monosyllabic	• •
§ 10.21	7 0
Masculine <i>in</i> -stems § 10.22	70
Nominative singular -im or -im § 10.23	71
Masculine and feminine § 10.23	71
Neuter § 10.24	71
Nominative singular -im § 10.25	71
Nominative singular -i § 10.26	·71
Masculine <i>i</i> -stems § 10.27	71
Feminine i-stems § 10.28	71
i-stems with Sanskrit -is § 10.29	71
- i as feminine of <i>in</i> -stems § 10.30	71
-i in neuter in-stems § 10.31	71
Nominative singular -is from i-stems § 10.32	71
Vocative singular § 10.33	71
Vocative singular -i §.10.33	71
Masculine <i>i</i> -stems § 10.34	71
Masculine in-stems § 10.35	72 72
Vocative singular -is (-ir) § 10.36 Vocative singular -i § 10.37	72
Masculine <i>i</i> -stems § 10.38	72
Masculine <i>in</i> -stems § 10.39	72
Masculine <i>i</i> -stems § 10.40	72
Feminine <i>i</i> -stems § 10.41	72
Accusative singular § 10.42	72
Accusative singular -im § 10.42	72
Masculine <i>in</i> -stems § 10.43	72
i-stems § 10.44	72
Neuter <i>i</i> -stems § 10.45	72
Neuter in-stems § 10.46	72
Accusative singular -i § 10.47	72
Masculine <i>i</i> -stems § 10.48	72
Masculine <i>in</i> -stems § 10.49	72
Feminine i-stems § 10.50	72
Polysyllabic feminine <i>i</i> -stems § 10.51	72 72
Monosyllabic i-stems § 10.52 Accusative singular -y § 10.53	72
Accusative singular -i § 10.53	73
Accusative singular -Im § 10.56	73
Masculine <i>i</i> -stem § 10.57	73
Masculine <i>in</i> -stem § 10.58	
Feminine <i>i</i> -stems § 10.59	73
Accusative singular -is (-ir, etc.) § 10.60	73
Accusative singular -is (-ir, etc.) § 10.60 Accusative singular -inam § 10.61	73
Accusative singular -iya § 10.62	·73
Accusative singular -iyam (polysyllabic i and i-	
stems) § 10.63	73
Oblique cases of the singular § 10.64	73
Oblique singular case forms common to all genders	
\$ 10.65	73
Oblique singular -i and i § 10.65	73
Instrumental § 10.66	
Ablative § 10.67	- 13

Locative § 10.68	
	73
Dative singular -ayi § 10.70	74
Ablative singular -ito § 10.71	74
Genitive singular -e § 10.72	
	74
Genitive singular -yus (from feminine -pati) § 10.73	74
Locative singular -yau for -au § 10.74	74
Locative singular - <i>im</i> § 10.75	74
Distinctively masculinc-neuter forms, not normal Sans-	• -
krit § 10.76	74
Instrumental singular -ina § 10.76	74
Instrumental singular -inä § 10.77	74
Genitive singular -isya § 10.78	74
Genitive singular -inah, -ino § 10.81	74
Locative singular -esmim § 10.83	
Locative singular -ismim § 10.84	74
Locative singular -ismi § 10.85	74
	74
Oblique singular -iye, -iye, -iyi § 10.90	75
-iye, Instrumental § 10.91	75
-iye, Dative § 10.92	75
-iye, Ablative § 10.93	75
-iye, Genitive § 10.94	75
<i>iye</i> , Locative § 10.95	75
-iye, Instrumental § 10.96	75
-iye, Dative § 10.97	76
-iye, Ablative § 10.98	76
-iye, Genitive § 10.99	76
-iye Locative § 10.100	76
<i>-iyi</i> , Locative § 10.101	76
Oblique singular -īya, -īyā, -iyā, -iya § 10.102	76
-iya, Instrumental § 10.103	76
-iya, Ablative § 10.104	76
The Comitine \$ 10.105	
-iya, Genitive § 10.105	76
<i>-iya</i> , Locative § 10.106	76
- <i>īyā</i> , Instrumental § 10.107	76
-īyā, Instrumental § 10.107 -iyā, Instrumental § 10.108	76 76
-īyā, Instrumental § 10.107 -iyā, Instrumental § 10.108 -iyā, Ablative § 10.109	76 76 76
-īyā, Instrumental § 10.107 -iyā, Instrumental § 10.108 -iyā, Ablative § 10.109 -iyā, Locative § 10.110	76 76 76 76
-īyā, Instrumental § 10.107 -iyā, Instrumental § 10.108 -iyā, Ablative § 10.109 -iyā, Locative § 10.110 -iya, Instrumental § 10.111	76 76 76 76 76
-īyā, Instrumental § 10.107 -iyā, Instrumental § 10.108 -iyā, Ablative § 10.109 -iyā, Locative § 10.110	76 76 76 76
-īyā, Instrumental § 10.107 -iyā, Instrumental § 10.108 -iyā, Ablative § 10.109 -iyā, Locative § 10.110 -iya, Instrumental § 10.111 -iya, Genitive § 10.112	76 76 76 76 76
-īyā, Instrumental § 10.107 -iyā, Instrumental § 10.108 -iyā, Ablative § 10.109 -iyā, Locative § 10.110 -iya, Instrumental § 10.111 -iya, Genitive § 10.112 Oblique singular -iyo and variants -īyu (m.c.), -iyo	76 76 76 76 76 76
-īyā, Instrumental § 10.107 -iyā, Instrumental § 10.108 -iyā, Ablative § 10.109 -iyā, Locative § 10.110 -iya, Instrumental § 10.111 -iya, Genitive § 10.112 Oblique singular -īyo and variants -īyu (m.c.), -iyo § 10.113	76 76 76 76 76 76 76
-īyā, Instrumental § 10.107 -iyā, Instrumental § 10.108 -iyā, Ablative § 10.109 -iyā, Locative § 10.110 -iya, Instrumental § 10.111 -iya, Genitive § 10.112 Oblique singular -īyo and variants -īyu (m.c.), -iyo § 10.113 -īyo, Instrumental § 10.114	76 76 76 76 76 76 76 77
$\begin{array}{c} -\bar{i}y\bar{a}, \text{ Instrumental } \$ 10.107\\ -iy\bar{a}, \text{ Instrumental } \$ 10.108\\ -iy\bar{a}, \text{ Ablative } \$ 10.109\\ -iy\bar{a}, \text{ Locative } \$ 10.110\\ -iy\bar{a}, \text{ Instrumental } \$ 10.111\\ -iya, \text{ Genitive } \$ 10.112\\ \text{Oblique singular - } iyo and variants - } iyu (m.c.), -iyo\\ \$ 10.113\\ -iyo, \text{ Instrumental } \$ 10.114\\ -iyo, \text{ Ablative } \$ 10.115\\ \end{array}$	76 76 76 76 76 76 76 77 77
-īyā, Instrumental § 10.107 -iyā, Instrumental § 10.108 -iyā, Ablative § 10.109 -iyā, Locative § 10.110 -iya, Instrumental § 10.111 -iya, Genitive § 10.112 Oblique singular -īyo and variants -īyu (m.c.), -iyo § 10.113 -īyo, Instrumental § 10.114 -īyo, Ablative § 10.115 -īyo, Genitive § 10.116	76 76 76 76 76 76 76 77
-īyā, Instrumental § 10.107 -iyā, Instrumental § 10.108 -iyā, Ablative § 10.109 -iyā, Locative § 10.110 -iya, Instrumental § 10.111 -iya, Genitive § 10.112 Oblique singular -īyo and variants -īyu (m.c.), -iyo § 10.113 -īyo, Instrumental § 10.114 -īyo, Ablative § 10.115 -īyo, Genitive § 10.116	76 76 76 76 76 76 76 77 77
-īyā, Instrumental § 10.107 -iyā, Instrumental § 10.108 -iyā, Ablative § 10.109 -iyā, Locative § 10.110 -iya, Instrumental § 10.111 -iya, Genitive § 10.112 Oblique singular -īyo and variants -īyu (m.c.), -iyo § 10.113 -īyo, Instrumental § 10.114 -īyo, Ablative § 10.115 -īyo, Locative § 10.117	76 76 76 76 76 76 76 77 77 77
-īyā, Instrumental § 10.107 -iyā, Instrumental § 10.108 -iyā, Ablative § 10.109 -iyā, Locative § 10.110 -iya, Instrumental § 10.111 -iya, Genitive § 10.112 Oblique singular -iyo and variants -iyu (m.c.), -iyo § 10.113 -īyo, Instrumental § 10.114 -iyo, Ablative § 10.115 -iyo, Genitive § 10.116 -iyo, Locative § 10.117 -iuu, Genitive § 10.118	76 76 76 76 76 76 76 77 77 77 77 77
-īyā, Instrumental § 10.107 -iyā, Instrumental § 10.108 -iyā, Ablative § 10.109 -iyā, Locative § 10.110 -iya, Instrumental § 10.111 -iya, Genitive § 10.112 Oblique singular -iyo and variants -iyu (m.c.), -iyo § 10.113 -iyo, Instrumental § 10.114 -iyo, Ablative § 10.115 -iyo, Genitive § 10.116 -iyo, Locative § 10.117 -iyu, Genitive § 10.118 -iyo, Genitive § 10.119	76 76 76 76 76 76 76 76 76 77 77 77 77 7
-īyā, Instrumental § 10.107 -iyā, Instrumental § 10.108 -iyā, Ablative § 10.109 -iyā, Locative § 10.110 -iya, Instrumental § 10.111 -iya, Genitive § 10.112 Oblique singular -iyo and variants -iyu (m.c.), -iyo § 10.113 -iyo, Ablative § 10.115 -iyo, Genitive § 10.116 -iyo, Genitive § 10.116 -iyu, Genitive § 10.118 -iyu, Genitive § 10.118 -iyo, Genitive § 10.119	76 76 76 76 76 76 76 76 76 77 77 77 77 7
-īyā, Instrumental § 10.107 -iyā, Instrumental § 10.108 -iyā, Ablative § 10.109 -iyā, Locative § 10.110 -iya, Instrumental § 10.111 -iya, Genitive § 10.112 Oblique singular -īyo and variants -īyu (m.c.), -iyo § 10.113 -īyo, Instrumental § 10.114 -īyo, Genitive § 10.115 -īyo, Genitive § 10.116 -īyo, Locative § 10.117 -īyo, Genitive § 10.118 -iyo, Genitive § 10.119 Oblique singular -yā, -ya § 10.120 -yā, Instrumental § 10.121	76 76 76 76 76 76 76 77 77 77 77 77 77 7
-īyā, Instrumental § 10.107 -iyā, Instrumental § 10.108 -iyā, Ablative § 10.109 -iyā, Locative § 10.110 -iya, Instrumental § 10.111 -iya, Genitive § 10.112 Oblique singular -iyo and variants -iyu (m.c.), -iyo § 10.113 -iyo, Ablative § 10.115 -iyo, Genitive § 10.116 -iyo, Genitive § 10.116 -iyu, Genitive § 10.118 -iyu, Genitive § 10.118 -iyo, Genitive § 10.119	76 76 76 76 76 76 76 76 76 77 77 77 77 7
-īyā, Instrumental § 10.107 -iyā, Instrumental § 10.108 -iyā, Ablative § 10.109 -iyā, Locative § 10.110 -iya, Instrumental § 10.111 -iya, Genitive § 10.112 Oblique singular -īyo and variants -īyu (m.c.), -iyo § 10.113 -īyo, Instrumental § 10.114 -īyo, Ablative § 10.115 -īyo, Genitive § 10.116 -īyo, Genitive § 10.118 -iyu, Genitive § 10.118 -iyo, Genitive § 10.119 Oblique singular -yā, -ya § 10.120 -yā, Ablative § 10.121	76 76 76 76 76 76 76 77 77 77 77 77 77 7
-īyā, Instrumental § 10.107 -iyā, Instrumental § 10.108 -iyā, Ablative § 10.109 -iyā, Locative § 10.110 -iya, Instrumental § 10.111 -iya, Genitive § 10.112 Oblique singular -iyo and variants -iyu (m.c.), -iyo § 10.113 -īyo, Instrumental § 10.114 -iyo, Genitive § 10.115 -īyo, Genitive § 10.116 -iyo, Genitive § 10.117 -iyo, Genitive § 10.118 -iyo, Genitive § 10.119 Oblique singular -yā, -ya § 10.120 -yā, Instrumental § 10.121 -yā, Genitive § 10.123	76 76 76 76 76 76 76 76 76 77 77 77 77 7
- <i>īyā</i> , Instrumental § 10.107 - <i>iyā</i> , Instrumental § 10.108 - <i>iyā</i> , Ablative § 10.109 - <i>iyā</i> , Locative § 10.110 - <i>iya</i> , Instrumental § 10.111 - <i>iya</i> , Genitive § 10.112 Oblique singular - <i>iyo</i> and variants - <i>iyu</i> (m.c.), - <i>iyo</i> § 10.113 - <i>iyo</i> , Ablative § 10.115 - <i>iyo</i> , Ablative § 10.116 - <i>iyo</i> , Genitive § 10.116 - <i>iyu</i> , Genitive § 10.118 - <i>iyu</i> , Genitive § 10.118 - <i>iya</i> , Instrumental § 10.120 - <i>yā</i> , Instrumental § 10.121 - <i>yā</i> , Genitive § 10.124	76 76 76 76 76 76 76 76 76 77 77 77 77 7
$\begin{array}{c} -\bar{y}\bar{a}, \mbox{ Instrumental } 10.107	$\begin{array}{c} 76\\ 76\\ 76\\ 76\\ 76\\ 76\\ 76\\ 77\\ 77\\ 77\\$
$\begin{array}{c} -\bar{y}\bar{a}, \mbox{ Instrumental } 10.107	$\begin{array}{c} 76\\ 76\\ 76\\ 76\\ 76\\ 76\\ 76\\ 77\\ 77\\ 77\\$
$\begin{array}{c} -\bar{y}\bar{a}, \mbox{ Instrumental } 10.107	$\begin{array}{c} 76\\ 76\\ 76\\ 76\\ 76\\ 76\\ 76\\ 77\\ 77\\ 77\\$
$\begin{array}{c} -\bar{y}\bar{a}, \mbox{ Instrumental } 10.107	$\begin{array}{c} 76\\ 76\\ 76\\ 76\\ 76\\ 76\\ 76\\ 77\\ 77\\ 77\\$
$\begin{array}{c} -\bar{y}\bar{a}, \mbox{ Instrumental } 10.107	$\begin{array}{c} 76\\ 76\\ 76\\ 76\\ 76\\ 76\\ 76\\ 77\\ 77\\ 77\\$
$\begin{array}{c} -\bar{y}\bar{a}, \mbox{ Instrumental } 10.107	$\begin{array}{c} 76\\ 76\\ 76\\ 76\\ 76\\ 76\\ 76\\ 77\\ 77\\ 77\\$
$\begin{array}{c} -iy\bar{a}, \mbox{ Instrumental } 10.107$	$\begin{array}{c} 76\\ 76\\ 76\\ 76\\ 76\\ 76\\ 76\\ 77\\ 77\\ 77\\$
$\begin{array}{c} -iy\bar{a}, Instrumental $ 10.107$	$\begin{array}{c} 76\\ 76\\ 76\\ 76\\ 76\\ 76\\ 76\\ 77\\ 77\\ 77\\$
$\begin{array}{c} -\bar{y}\bar{a}, \mbox{ Instrumental } 10.107	$\begin{array}{c} 76\\ 76\\ 76\\ 76\\ 76\\ 76\\ 77\\ 77\\ 77\\ 77\\$
$\begin{array}{c} -\bar{y}\bar{a}, \mbox{ Instrumental } 10.107	$\begin{array}{c} 76\\ 76\\ 76\\ 76\\ 76\\ 76\\ 76\\ 77\\ 77\\ 77\\$
$\begin{array}{c} -\bar{y}\bar{a}, \mbox{ Instrumental } 10.107	$\begin{array}{c} 76\\ 76\\ 76\\ 76\\ 76\\ 77\\ 77\\ 77\\ 77\\ 77\\$
$\begin{array}{c} -\bar{y}\bar{a}, \text{ Instrumental } \$ 10.107. \\ -iy\bar{a}, \text{ Instrumental } \$ 10.108. \\ -iy\bar{a}, \text{ Ablative } \$ 10.109. \\ -iy\bar{a}, \text{ Coctive } \$ 10.110. \\ -iya, \text{ Instrumental } \$ 10.111. \\ -iya, \text{ Genitive } \$ 10.112. \\ \text{Oblique singular -}iyo and variants -}iyu (m.c.), -iyo \\ \$ 10.113. \\ -iyo, \text{ Instrumental } \$ 10.114. \\ -iyo, \text{ Ablative } \$ 10.115. \\ -iyo, \text{ Genitive } \$ 10.116. \\ -iyo, \text{ Genitive } \$ 10.116. \\ -iyo, \text{ Genitive } \$ 10.118. \\ -iyo, \text{ Genitive } \$ 10.118. \\ -iyo, \text{ Genitive } \$ 10.119. \\ \text{Oblique singular -}y\bar{a}, ya \$ 10.120 \\ -y\bar{a}, \text{ Instrumental } \$ 10.121 \\ -y\bar{a}, \text{ Ablative } \$ 10.123 \\ -y\bar{a}, \text{ Locative } \$ 10.123 \\ -ya, \text{ Genitive } \$ 10.124 \\ -ya, \text{ Genitive } \$ 10.126 \\ -ya, \text{ Locative } \$ 10.127 \\ -ya, \text{ Locative } \$ 10.126 \\ -ya, \text{ Locative } \$ 10.127 \\ -ya, \text{ Locative } \$ 10.127 \\ -ya, \text{ Locative } \$ 10.127 \\ -ya, \text{ Locative } \$ 10.126 \\ -ya, \text{ Locative } \$ 10.126 \\ -ya, \text{ Locative } \$ 10.127 \\ -ya, \text{ Locative } \$ 10.128 \\ -ya, \text{ Locative } \$ 10.128 \\ -ya, \text{ Locative } \$ 10.130 \\ -ya, \text{ Locative } \$ 10.130 \\ -ya, \text{ Locative } \$ 10.130 \\ -ya, \text{ Locative } 10.131 \\ -xa, \text{ Lotative -} i0, -itas \$ 10.133 \\ -xa, \text{ Lotative -} i0, -itas \$ 10.133 \\ -xa, \text{ Locative -} i0.144 \\ \$ 10.133 \\ -xa, \text{ Locative -} i0.144 \\ \$ 10.133 \\ -xa, \text{ Locative -} i0.144 \\ \$ 10.133 \\ -xa, \text{ Locative -} i0.144 \\ \$ 10.133 \\ -xa, \text{ Locative -} i0.144 \\ \$ 10.133 \\ -xa, \text{ Locative -} i0.144 \\ \$ 10.133 \\ -xa, \text{ Locative -} i0.144 \\ \$ 10.133 \\ -xa, \text{ Locative -} i0.144 \\ \$ 10.133 \\ -xa, \text{ Locative -} i0.144 \\ \$ 10.133 \\ -xa, \text{ Locative -} i0.144 \\ \$ 10.133 $	$\begin{array}{c} 76\\ 76\\ 76\\ 76\\ 76\\ 76\\ 77\\ 77\\ 77\\ 77\\$
$\begin{array}{c} -iy\bar{a}, Instrumental $ 10.107$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
$\begin{array}{c} -iy\bar{a}, Instrumental $ 10.107$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
$\begin{array}{c} -iy\bar{a}, Instrumental $ 10.107$	$\begin{array}{c} 76\\ 76\\ 76\\ 76\\ 76\\ 77\\ 77\\ 77\\ 77\\ 77\\$
$\begin{array}{c} -\bar{y}\bar{a}, \ \text{Instrumental } \$ 10.107\\ -iy\bar{a}, \ \text{Instrumental } \$ 10.108\\ -iy\bar{a}, \ \text{Ablative } \$ 10.109\\ -iy\bar{a}, \ \text{Locative } \$ 10.110\\ -iya, \ \text{Genitive } \$ 10.112\\ 0blique singular -iyo and variants -iyu (m.c.), -iyo \\ \$ 10.113\\ -iyo, \ \text{Genitive } \$ 10.115\\ -iyo, \ \text{Ablative } \$ 10.115\\ -iyo, \ \text{Genitive } \$ 10.116\\ -iyo, \ \text{Genitive } \$ 10.116\\ -iyo, \ \text{Genitive } \$ 10.118\\ -iyo, \ \text{Genitive } \$ 10.118\\ -iyo, \ \text{Genitive } \$ 10.118\\ -iyo, \ \text{Genitive } \$ 10.120\\ -y\bar{a}, \ \text{Genitive } \$ 10.121\\ -y\bar{a}, \ \text{Genitive } \$ 10.121\\ -y\bar{a}, \ \text{Ablative } \$ 10.124\\ -ya, \ \text{Instrumental } \$ 10.121\\ -ya, \ \text{Instrumental } \$ 10.125\\ -ya, \ \text{Genitive } \$ 10.126\\ -ya, \ \text{Locative } \$ 10.127\\ \text{Miscellaneous femline singular oblique endings} \\ \$ 10.128\\ \text{Instrumental } -iyena \$ 10.129\\ \text{Instrumental } -iyena \$ 10.129\\ \text{Ablative } -ilo, -ilas \$ 10.133\\ \ \text{Ablative } -ilo, -ilas \$ 10.133\\ \ \text{Ablative } -ilo, -ilas \$ 10.135\\ \ \text{Ablative } -ilo, -ilas \$ 10.136\\ \ \text{Ablative } -ilo, 131\\ \ \text{Ablative } -ilo, 10.136\\ \ \text{Ablative } -ilo, 131\\ \ \text{Ablative } -ilo, 136\\ \ \text{Ablative } -ilo, 136\\ \ \text{Ablative } -ilo, 136\\ \ \text{Ablative } -ilo, 137\\ \ \text{Ablative } -ilo, 136\\ \ \text{Ablative } -ilo, 131\\ \ \text{Ablative } -ilo, 136\\ \ \text{Ablative } -ilo, 13$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
$\begin{array}{c} -\bar{y}\bar{a}, \ \text{Instrumental } \$ 10.107\\ -iy\bar{a}, \ \text{Instrumental } \$ 10.108\\ -iy\bar{a}, \ \text{Ablative } \$ 10.109\\ -iy\bar{a}, \ \text{Locative } \$ 10.110\\ -iya, \ \text{Genitive } \$ 10.112\\ 0blique singular -iyo and variants -iyu (m.c.), -iyo \\ \$ 10.113\\ -iyo, \ \text{Genitive } \$ 10.115\\ -iyo, \ \text{Ablative } \$ 10.115\\ -iyo, \ \text{Genitive } \$ 10.116\\ -iyo, \ \text{Genitive } \$ 10.116\\ -iyo, \ \text{Genitive } \$ 10.118\\ -iyo, \ \text{Genitive } \$ 10.118\\ -iyo, \ \text{Genitive } \$ 10.118\\ -iyo, \ \text{Genitive } \$ 10.120\\ -y\bar{a}, \ \text{Genitive } \$ 10.121\\ -y\bar{a}, \ \text{Genitive } \$ 10.121\\ -y\bar{a}, \ \text{Ablative } \$ 10.124\\ -ya, \ \text{Instrumental } \$ 10.121\\ -ya, \ \text{Instrumental } \$ 10.125\\ -ya, \ \text{Genitive } \$ 10.126\\ -ya, \ \text{Locative } \$ 10.127\\ \text{Miscellaneous femline singular oblique endings} \\ \$ 10.128\\ \text{Instrumental } -iyena \$ 10.129\\ \text{Instrumental } -iyena \$ 10.129\\ \text{Ablative } -ilo, -ilas \$ 10.133\\ \ \text{Ablative } -ilo, -ilas \$ 10.133\\ \ \text{Ablative } -ilo, -ilas \$ 10.135\\ \ \text{Ablative } -ilo, -ilas \$ 10.136\\ \ \text{Ablative } -ilo, 131\\ \ \text{Ablative } -ilo, 10.136\\ \ \text{Ablative } -ilo, 131\\ \ \text{Ablative } -ilo, 136\\ \ \text{Ablative } -ilo, 136\\ \ \text{Ablative } -ilo, 136\\ \ \text{Ablative } -ilo, 137\\ \ \text{Ablative } -ilo, 136\\ \ \text{Ablative } -ilo, 131\\ \ \text{Ablative } -ilo, 136\\ \ \text{Ablative } -ilo, 13$	$\begin{array}{c} 76\\ 76\\ 76\\ 76\\ 76\\ 77\\ 77\\ 77\\ 77\\ 77\\$
$\begin{array}{c} -iy\bar{a}, Instrumental $ 10.107$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
$\begin{array}{c} -\bar{y}\bar{a}, \ \text{Instrumental } \$ 10.107\\ -iy\bar{a}, \ \text{Instrumental } \$ 10.108\\ -iy\bar{a}, \ \text{Ablative } \$ 10.109\\ -iy\bar{a}, \ \text{Locative } \$ 10.110\\ -iya, \ \text{Genitive } \$ 10.112\\ 0blique singular -iyo and variants -iyu (m.c.), -iyo \\ \$ 10.113\\ -iyo, \ \text{Genitive } \$ 10.115\\ -iyo, \ \text{Ablative } \$ 10.115\\ -iyo, \ \text{Genitive } \$ 10.116\\ -iyo, \ \text{Genitive } \$ 10.116\\ -iyo, \ \text{Genitive } \$ 10.118\\ -iyo, \ \text{Genitive } \$ 10.118\\ -iyo, \ \text{Genitive } \$ 10.118\\ -iyo, \ \text{Genitive } \$ 10.120\\ -y\bar{a}, \ \text{Genitive } \$ 10.121\\ -y\bar{a}, \ \text{Genitive } \$ 10.121\\ -y\bar{a}, \ \text{Ablative } \$ 10.124\\ -ya, \ \text{Instrumental } \$ 10.121\\ -ya, \ \text{Instrumental } \$ 10.125\\ -ya, \ \text{Genitive } \$ 10.126\\ -ya, \ \text{Locative } \$ 10.127\\ \text{Miscellaneous femline singular oblique endings} \\ \$ 10.128\\ \text{Instrumental } -iyena \$ 10.129\\ \text{Instrumental } -iyena \$ 10.129\\ \text{Ablative } -ilo, -ilas \$ 10.133\\ \ \text{Ablative } -ilo, -ilas \$ 10.133\\ \ \text{Ablative } -ilo, -ilas \$ 10.135\\ \ \text{Ablative } -ilo, -ilas \$ 10.136\\ \ \text{Ablative } -ilo, 131\\ \ \text{Ablative } -ilo, 10.136\\ \ \text{Ablative } -ilo, 131\\ \ \text{Ablative } -ilo, 136\\ \ \text{Ablative } -ilo, 136\\ \ \text{Ablative } -ilo, 136\\ \ \text{Ablative } -ilo, 137\\ \ \text{Ablative } -ilo, 136\\ \ \text{Ablative } -ilo, 131\\ \ \text{Ablative } -ilo, 136\\ \ \text{Ablative } -ilo, 13$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$

	•
Genitive -iyam § 10.141	78
Locative -āyān § 10.142	78
Locative -iyām § 10.143	78
Locative -iyam § 10.144	78
Locative -iyam § 10.145	78
	78
Dual instrumental -ibhya § 10.147	78
Nominative-accusative plural § 10.148	78
Nominative-accusative plural -ayas § 10.152	79
Masculine <i>i</i> -stems, accusative plural § 10.153	79
Feminine <i>i</i> -stems, accusative plural § 10.153	79
Nominative-accusative plural -qua § 10.154	79
Nominative-accusative plural -inas (-ino, etc.)	
§ 10.155	79
Nominative-accusative plural -ind § 10.156	79
Nominative-accusative plural -ina § 10.157	79
Nominative-accusative plural ino § 10.158	79
Nominative-accusative plural -ina § 10.159	79
Nominative-accusative plural -ini § 10.160	79
Nominative-accusative plural -ini § 10.161	79
Nominative-accusative plural -yas § 10.162	79
Nominative-accusative plural -ya § 10.163	79
Nominative-accusative plural -yā(s) § 10.164	79
Nominative-accusative plural -in, -im § 10.165	80
Accusative plural -im § 10.166	80
Nominative-accusative plural -is § 10.167	80
Nominative-accusative plural -iyas, -iyo § 10.168	80
(Masculine <i>i</i> -stems, nominative, accusative;	
masculine in-stems, nominative; feminine	
i-stems, nominative, accusative; feminine	
polysyllabic <i>i</i> -stems, nominative, accusative)	
Nominative-accusative plural -iya § 10.169	80
Nominative-accusative plural - <i>iyā</i> § 10.170	80
Nominative-accusative plural -io § 10.171	80
NomInative-accusative plural -iyo § 10.172	80
(Macaulina i stame nominative accurative)	
(Masculine <i>i</i> -stems, nominative, accusative;	
feminine <i>i</i> -stems, nominative, accusative;	
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative;	
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative)	81
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative) Nominative-accusative plural - <i>iya</i> § 10.173	
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative Nominative-accusative plural - <i>iya</i> § 10.173 Nominative-accusative plural - <i>iyā</i> § 10.174	81 81
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative) Nominative-accusative plural - <i>iya</i> § 10.173 Nominative-accusative plural - <i>iya</i> § 10.174 Nominative(-accusative) plural - <i>iye</i> , feminine <i>i</i> -	81
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative) Nominative-accusative plural - <i>iya</i> § 10.173 Nominative(-accusative) plural - <i>iya</i> § 10.174 Nominative(-accusative) plural - <i>iye</i> , feminine <i>i</i> - stems § 10.175	81 81
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative; Nominative-accusative plural - <i>iya</i> § 10.173 Nominative-accusative plural - <i>iyā</i> § 10.174 Nominative(-accusative) plural - <i>iyē</i> , feminine <i>i</i> - stems § 10.175 Nominative-accusative plural - <i>iyē</i> § 10.176 Nominative-accusative plural - <i>i</i> § 10.177	81 81 81
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative) Nominative-accusative plural - <i>iya</i> § 10.173 Nominative(-accusative) plural - <i>iya</i> § 10.174 Nominative(-accusative) plural - <i>iye</i> , feminine <i>i</i> - stems § 10.175 Nominative-accusative plural - <i>iye</i> § 10.176	81 81 81
 feminine i-stems, nominative, accusative; feminine i-stems, nominative, accusative; feminine monosyllabic i-stem, nominative) Nominative-accusative plural -iya § 10.173 Nominative-accusative plural -iye, feminine i-stems § 10.175 Nominative-accusative plural -iye § 10.176 Nominative-accusative plural -iye § 10.176 Nominative-accusative plural -iye § 10.176 Mominative-accusative plural -iye § 10.176 Masculine i-(i-)stems, nominative § 10.180 	81 81 81 81
feminine i-stems, nominative, accusative; feminine i-stems, nominative, accusative; feminine monosyllabic i-stem, nominative Nominative-accusative plural -iya § 10.173 Nominative-accusative plural -iya § 10.174 Nominative-accusative plural -iye, feminine i- stems § 10.175 Nominative-accusative plural -iye § 10.176 Nominative-accusative plural -i § 10.177 Masculine i-(i-)stems, nominative § 10.180 Masculine in-stems, nominative § 10.181	81 81 81 81 81 81 81 81
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative) Nominative-accusative plural - <i>iya</i> § 10.173 Nominative-accusative plural - <i>iyā</i> § 10.174 Nominative-accusative plural - <i>iyē</i> § 10.176 Nominative-accusative plural - <i>iyē</i> § 10.176 Nominative-accusative plural - <i>i</i> § 10.176 Masculine <i>i</i> -(<i>i</i> -)stems, nominative § 10.180 Masculine <i>in</i> -stems, accusative § 10.181 Masculine <i>in</i> -stems, accusative § 10.182	81 81 81 81 81 81 81 81 81
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative) Nominative-accusative plural - <i>iya</i> § 10.173 Nominative(-accusative) plural - <i>iya</i> § 10.174 Nominative(-accusative) plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>i</i> § 10.177 Masculine <i>i</i> -(<i>i</i> -)stems, nominative § 10.179 Masculine <i>i</i> -stems, accusative § 10.180 Masculine <i>in</i> -stems, accusative § 10.183 Feminine <i>in</i> -stems, nominative § 10.183	81 81 81 81 81 81 81 81 81
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative Nominative-accusative plural - <i>iya</i> § 10.173 Nominative-accusative plural - <i>iya</i> § 10.174 Nominative-accusative plural - <i>iye</i> , feminine <i>i</i> - stems § 10.175 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>iye</i> § 10.176 Masculine <i>i</i> -(<i>i</i> -)stems, nominative § 10.179 Masculine <i>i</i> -stems, accusative § 10.180 Masculine <i>in</i> -stems, nominative § 10.181 Feminine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.184	81 81 81 81 81 81 81 81 81 81
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative Nominative-accusative plural - <i>iya</i> § 10.173 Nominative-accusative plural - <i>iya</i> § 10.174 Nominative-accusative plural - <i>iye</i> , feminine <i>i</i> - stems § 10.175 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>iye</i> § 10.176 Masculine <i>i</i> -(<i>i</i> -)stems, nominative § 10.179 Masculine <i>i</i> -stems, accusative § 10.180 Masculine <i>in</i> -stems, nominative § 10.181 Feminine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.184 Feminine <i>i</i> -stems, nominative § 10.184	81 81 81 81 81 81 81 81 81 81 81
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative) Nominative-accusative plural - <i>iya</i> § 10.173 Nominative-accusative plural - <i>iya</i> § 10.174 Nominative-accusative plural - <i>iye</i> , feminine <i>i</i> - stems § 10.175 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>i</i> § 10.176 Masculine <i>i</i> -(<i>i</i> -)stems, nominative § 10.179 Masculine <i>i</i> -stems, accusative § 10.180 Masculine <i>in</i> -stems, accusative § 10.182 Feminine <i>in</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.185 Feminine <i>i</i> -stems, nominative § 10.185 Feminine <i>i</i> -stems, nominative § 10.185	81 81 81 81 81 81 81 81 81 81 81
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative) Nominative-accusative plural - <i>iya</i> § 10.173 Nominative-accusative plural - <i>iyā</i> § 10.174 Nominative-accusative plural - <i>iyē</i> § 10.176 Nominative-accusative plural - <i>iyē</i> § 10.176 Nominative-accusative plural - <i>i §</i> 10.176 Masculine <i>i</i> -stems, nominative § 10.179 Masculine <i>i</i> -stems, accusative § 10.181 Masculine <i>i</i> -stems, accusative § 10.182 Feminine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.184 Feminine <i>i</i> -stems, accusative § 10.184 Feminine <i>i</i> -stems, nominative § 10.185 Feminine <i>i</i> -stems, nominative § 10.186 Feminine <i>i</i> -stems, accusative § 10.187	81 81 81 81 81 81 81 81 81 81
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative) Nominative-accusative plural - <i>iya</i> § 10.173 Nominative-accusative plural - <i>iyā</i> § 10.174 Nominative-accusative plural - <i>iye</i> , feminine <i>i</i> - stems § 10.175 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>i</i> § 10.176 Masculine <i>i</i> -stems, nominative § 10.179 Masculine <i>i</i> -stems, nominative § 10.181 Masculine <i>i</i> -stems, accusative § 10.183 Feminine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.184 Feminine <i>i</i> -stems, nominative § 10.185 Feminine <i>i</i> -stems, nominative § 10.185 Feminine <i>i</i> -stems, accusative § 10.187 Feminine <i>i</i> -stems, accusative § 10.187 Feminine <i>i</i> -stems, accusative § 10.187 Feminine <i>i</i> -stems, accusative § 10.187	81 81 81 81 81 81 81 81 81 81 81 81 81
feminine i-stems, nominative, accusative; feminine i-stems, nominative, accusative; feminine monosyllabic i-stem, nominative) Nominative-accusative plural -iya § 10.173 Nominative-accusative plural -iya § 10.174 Nominative-accusative plural -iye, feminine i- stems § 10.175 Nominative-accusative plural -iye § 10.176 Nominative-accusative plural -i § 10.176 Masculine i-(i-)stems, nominative § 10.180 Masculine in-stems, accusative § 10.180 Feminine in-stems, nominative § 10.183 Feminine i-stems, nominative § 10.183 Feminine i-stems, nominative § 10.184 Feminine i-stems, nominative § 10.185 Feminine i-stems, nominative § 10.185 Feminine i-stems, accusative § 10.185	81 81 81 81 81 81 81 81 81 81 81 81 81
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative Nominative-accusative plural - <i>iya</i> § 10.173 Nominative-accusative plural - <i>iya</i> § 10.174 Nominative-accusative plural - <i>iye</i> § 10.174 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>iye</i> § 10.176 Masculine <i>i</i> -(<i>i</i> -)stems, nominative § 10.179 Masculine <i>i</i> -stems, accusative § 10.180 Feminine <i>in</i> -stems, nominative § 10.183 Feminine <i>in</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.185 Feminine <i>i</i> -stems, accusative § 10.185 Feminine <i>i</i> -stems, nominative § 10.185 Feminine <i>i</i> -stems, accusative § 10.185 Feminine <i>i</i> -stems, nominative § 10.185 Feminine <i>i</i> -stems, nominative § 10.185 Feminine <i>i</i> -stems, accusative § 10.185 Feminine <i>i</i> -stems, nominative § 10.186 Feminine <i>i</i> -stems, nominative § 10.186 Feminine <i>i</i> -stems, accusative § 10.186 Feminine <i>i</i> -stems, accusative § 10.188 Noutter <i>i</i> and <i>in</i> -stems § 10.188 Nominative-accusative plural - <i>i</i> § 10.189	81 81 81 81 81 81 81 81 81 81 81 81 81
feminine i-stems, nominative, accusative; feminine i-stems, nominative, accusative; feminine monosyllabic i-stem, nominative) Nominative-accusative plural -iya § 10.173 Nominative-accusative plural -iya § 10.174 Nominative-accusative plural -iye, feminine i- stems § 10.175 Nominative-accusative plural -iye § 10.176 Nominative-accusative plural -i § 10.176 Masculine i-(i-)stems, nominative § 10.179 Masculine i-stems, accusative § 10.180 Masculine in-stems, nominative § 10.181 Feminine in-stems, nominative § 10.183 Feminine i-stems, nominative § 10.183 Feminine i-stems, nominative § 10.185 Feminine i-stems, nominative § 10.185 Feminine i-stems, nominative § 10.185 Feminine i-stems, nominative § 10.186 Feminine i-stems, nominative § 10.186 Nominative-accusative plural -i § 10.188 (Masculine i-stems, nominative; accusative; masculine in-stems, accusative; feminine	81 81 81 81 81 81 81 81 81 81 81 81 81
feminine i-stems, nominative, accusative; feminine i-stems, nominative, accusative; feminine monosyllabic i-stem, nominative) Nominative-accusative plural -iya § 10.173 Nominative-accusative plural -iya § 10.174 Nominative-accusative plural -iye, feminine i- stems § 10.175 Nominative-accusative plural -iye § 10.176 Nominative-accusative plural -i § 10.176 Masculine i-cli-stems, nominative § 10.179 Masculine i-stems, accusative § 10.181 Masculine in-stems, accusative § 10.183 Feminine in-stems, nominative § 10.183 Feminine i-stems, nominative § 10.184 Feminine i-stems, nominative § 10.184 Feminine i-stems, nominative § 10.185 Feminine i-stems, nominative § 10.186 Feminine i-stems, nominative § 10.187 Feminine i-stems, nominative § 10.186 Feminine i-stems, nominative § 10.187 Feminine i-stems, nominative § 10.186 Feminine i-stems, nominative § 10.187 Feminine i-stems, nominative § 10.186 Feminine i-stems, nominative § 10.187 Nouter i and in-stems § 10.188 Nominative-accusative plural -i § 10.189	81 81 81 81 81 81 81 81 81 81 81 81 81
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative Nominative-accusative plural - <i>iya</i> § 10.173 Nominative-accusative plural - <i>iya</i> § 10.174 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>i</i> § 10.176 Nominative-accusative plural - <i>i</i> § 10.176 Masculine <i>i</i> -(<i>i</i> -)stems, nominative § 10.179 Masculine <i>i</i> -stems, accusative § 10.180 Masculine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.184 Feminine <i>i</i> -stems, nominative § 10.184 Feminine <i>i</i> -stems, nominative § 10.185 Feminine <i>i</i> -stems, accusative § 10.185 Feminine <i>i</i> -stems, nominative § 10.185 Feminine <i>i</i> -stems, nominative § 10.186 Keuter <i>i</i> and <i>i</i> -stems, accusative § 10.187 Nominative-accusative plural - <i>i</i> § 10.188 (Masculine <i>i</i> -stems, nominative, accusative; masculine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; masculine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative;	81 81 81 81 81 81 81 81 81 81 81 81 81 8
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative Nominative-accusative plural - <i>iya</i> § 10.173 Nominative-accusative plural - <i>iya</i> § 10.174 Nominative-accusative plural - <i>iye</i> § 10.174 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>i</i> § 10.176 Masculine <i>i</i> -(<i>i</i> -)stems, nominative § 10.179 Masculine <i>i</i> -stems, accusative § 10.180 Masculine <i>in</i> -stems, nominative § 10.181 Feminine <i>in</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, accusative § 10.185 Feminine <i>i</i> -stems, accusative § 10.185 Feminine <i>i</i> -stems, accusative § 10.185 Feminine <i>i</i> -stems, nominative § 10.185 Feminine <i>i</i> -stems, nominative § 10.185 Feminine <i>i</i> -stems, accusative § 10.185 Feminine <i>i</i> -stems, accusative § 10.185 Feminine <i>i</i> -stems, accusative § 10.186 Feminine <i>i</i> -stems, accusative § 10.188 Nouninative-accusative plural - <i>i</i> § 10.188 Nominative-accusative plural - <i>i</i> § 10.189 (Masculine <i>i</i> -stems, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine	81 81 81 81 81 81 81 81 81 81 81 81 81 8
feminine i-stems, nominative, accusative; feminine i-stems, nominative, accusative; feminine monosyllabic i-stem, nominative Nominative-accusative plural -iya § 10.173 Nominative-accusative plural -iya § 10.174 Nominative-accusative plural -iye, feminine i- stems § 10.175 Nominative-accusative plural -iye § 10.176 Nominative-accusative plural -ige § 10.176 Nominative-accusative plural -i § 10.177 Masculine i-(i-)stems, nominative § 10.180 Masculine in-stems, accusative § 10.181 Masculine in-stems, accusative § 10.183 Feminine in-stems, nominative § 10.183 Feminine i-stems, nominative § 10.183 Feminine i-stems, nominative § 10.185 Feminine i-stems, nominative § 10.185 Feminine i-stems, accusative § 10.185 Feminine i-stems, accusative § 10.185 Feminine i-stems, accusative § 10.185 Feminine i-stems, nominative § 10.185 Feminine i-stems, accusative § 10.185 Feminine i-stems, accusative § 10.185 Feminine i-stems, nominative § 10.186 Isems, nominative, accusative; feminine i-stems, nominative, accusative; feminine i	81 81 81 81 81 81 81 81 81 81 81 81 81 8
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative) Nominative-accusative plural - <i>iya</i> § 10.173 Nominative-accusative plural - <i>iya</i> § 10.174 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>i</i> § 10.176 Masculine <i>i</i> -(<i>i</i> -)stems, nominative § 10.179 Masculine <i>i</i> -stems, accusative § 10.180 Masculine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.184 Feminine <i>i</i> -stems, nominative § 10.184 Feminine <i>i</i> -stems, nominative § 10.184 Feminine <i>i</i> -stems, nominative § 10.185 Feminine <i>i</i> -stems, nominative § 10.186 Feminine <i>i</i> -stems, nominative § 10.186 Feminine <i>i</i> -stems, nominative § 10.186 Feminine <i>i</i> -stems, nominative § 10.187 Nouter <i>i</i> and <i>i</i> n-stems § 10.188 Nominative-accusative plural - <i>i</i> § 10.189 (Masculine <i>i</i> -stems, nominative, accusative; masculine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminin	81 81 81 81 81 81 81 81 81 81 81 81 81 8
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative) Nominative-accusative plural - <i>iya</i> § 10.173 Nominative-accusative plural - <i>iya</i> § 10.174 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>i</i> § 10.176 Masculine <i>i</i> -(<i>i</i> -)stems, nominative § 10.179 Masculine <i>i</i> -stems, accusative § 10.180 Masculine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.184 Feminine <i>i</i> -stems, nominative § 10.184 Feminine <i>i</i> -stems, nominative § 10.184 Feminine <i>i</i> -stems, nominative § 10.185 Feminine <i>i</i> -stems, nominative § 10.186 Feminine <i>i</i> -stems, nominative § 10.186 Feminine <i>i</i> -stems, nominative § 10.186 Feminine <i>i</i> -stems, nominative § 10.187 Nouter <i>i</i> and <i>i</i> n-stems § 10.188 Nominative-accusative plural - <i>i</i> § 10.189 (Masculine <i>i</i> -stems, nominative, accusative; masculine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminin	81 81 81 81 81 81 81 81 81 81 81 81 81 8
feminine i-stems, nominative, accusative; feminine i-stems, nominative, accusative; feminine monosyllabic i-stem, nominative) Nominative-accusative plural -iya § 10.173 Nominative-accusative plural -iya § 10.174 Nominative-accusative plural -iye, feminine i- stems § 10.175 Nominative-accusative plural -iye § 10.176 Nominative-accusative plural -i § 10.177 Masculine i-stems, nominative § 10.179 Masculine i-stems, accusative § 10.181 Masculine in-stems, accusative § 10.182 Feminine in-stems, nominative § 10.183 Feminine i-stems, nominative § 10.183 Feminine i-stems, nominative § 10.184 Feminine i-stems, nominative § 10.184 Feminine i-stems, nominative § 10.185 Feminine i-stems, nominative § 10.186 Feminine i-stems, nominative § 10.187 Nouter i and in-stems § 10.188 Nominative-accusative plural -i § 10.189 (Masculine i-stems, nominative, accusative; masculine in-stems, nominative; feminine i-stems, nominative, accusative; feminine i-stems, no	81 81 81 81 81 81 81 81 81 81
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative Nominative-accusative plural - <i>iya</i> § 10.173 Nominative-accusative plural - <i>iya</i> § 10.174 Nominative-accusative plural - <i>iye</i> § 10.174 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>iye</i> § 10.176 Masculine <i>i</i> -(<i>i</i> -)stems, nominative § 10.179 Masculine <i>i</i> -stems, accusative § 10.180 Masculine <i>in</i> -stems, nominative § 10.181 Feminine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, accusative § 10.185 Feminine <i>i</i> -stems, accusative § 10.185 Feminine <i>i</i> -stems, nominative § 10.185 Feminine <i>i</i> -stems, nominative § 10.185 (Masculine <i>i</i> -stems, accusative § 10.185 Feminine <i>i</i> -stems, nominative § 10.186 (Masculine <i>i</i> -stems, accusative § 10.186 Feminine <i>i</i> -stems, accusative § 10.186 Nominative-accusative plural - <i>i</i> § 10.189 (Masculine <i>i</i> -stems, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine	81 81 81 81 81 81 81 81 81 81
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative) Nominative-accusative plural - <i>iya</i> § 10.173 Nominative-accusative plural - <i>iya</i> § 10.174 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>iye</i> § 10.176 Masculine <i>i</i> -stems, nominative § 10.179 Masculine <i>i</i> -stems, accusative § 10.179 Masculine <i>i</i> -stems, accusative § 10.181 Masculine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.184 Feminine <i>i</i> -stems, nominative § 10.184 Feminine <i>i</i> -stems, nominative § 10.185 Masculine <i>i</i> -stems, nominative § 10.186 Feminine <i>i</i> -stems, nominative § 10.186 Feminine <i>i</i> -stems, nominative § 10.187 Nouter <i>i</i> and <i>in</i> -stems § 10.188 Nominative-accusative plural - <i>i</i> § 10.189 (Masculine <i>i</i> -stems, accusative; feminine <i>i</i> -stems, nominative, accusative; masculine <i>i</i> -stems, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; masculine <i>i</i> -stems, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems; feminine <i>i</i> -stems; masculine <i>i</i> -stems; feminine <i>i</i> -stems	81 81 81 81 81 81 81 81 81 81
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative Nominative-accusative plural - <i>iya</i> § 10.173 Nominative-accusative plural - <i>iya</i> § 10.174 Nominative-accusative plural - <i>iye</i> § 10.174 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>iye</i> § 10.176 Masculine <i>i</i> -(<i>i</i> -)stems, nominative § 10.179 Masculine <i>i</i> -stems, accusative § 10.180 Masculine <i>in</i> -stems, nominative § 10.181 Masculine <i>in</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, accusative § 10.185 Feminine <i>i</i> -stems, accusative § 10.185 (Masculine <i>i</i> -stems, accusative § 10.185 Feminine <i>i</i> -stems, nominative § 10.186 Feminine <i>i</i> -stems, nominative § 10.186 (Masculine <i>i</i> -stems, accusative § 10.187 Noutier <i>i</i> and <i>in</i> -stems § 10.188 Nominative-accusative plural - <i>i</i> § 10.189 (Masculine <i>i</i> -stems, accusative; feminine <i>i</i> -stems, nominative, accusative; masculine <i>in</i> -stems, accusative; feminine <i>i</i> -stems, nominative, accusative; masculine <i>i</i> -stems, accusative; feminine <i>i</i> -stems, nominative, accusative; masculine <i>i</i> -stems, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems; feminine <i>i</i> -stems; masculine <i>in</i> -stems; feminine <i>i</i> -stems; feminine <i>i</i> - stems, monosyllabic, polysyllabic)	81 81 81 81 81 81 81 81 81 81
feminine i-stems, nominative, accusative; feminine i-stems, nominative, accusative; feminine monosyllabic i-stem, nominative Nominative-accusative plural -iya § 10.173 Nominative-accusative plural -iya § 10.174 Nominative-accusative plural -iye, feminine i- stems § 10.175 Nominative-accusative plural -iye § 10.176 Nominative-accusative plural -iye § 10.176 Nominative-accusative plural -i § 10.177 Masculine i-(i-)stems, nominative § 10.180 Masculine in-stems, accusative § 10.181 Masculine in-stems, accusative § 10.182 Feminine in-stems, nominative § 10.183 Feminine i-stems, nominative § 10.183 Feminine i-stems, nominative § 10.185 Feminine i-stems, accusative § 10.185 Feminine i-stems, accusative § 10.185 Keuter i and in-stems § 10.188 Nominative-accusative plural -i § 10.189 (Masculine i-stems, nominative; accusative; masculine in-stems, accusative; feminine i-stems, nominative, accusative; feminine i-stems; feminine i-stems; masculine in-stems; feminine i-stems; masculine in-stems; feminine i-stems; feminine i- stems, monosyllabic, polysyllabic) Instrumental plural -ibhis (etc.) § 10.195	81 81 81 81 81 81 81 81 81 81
feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine monosyllabic <i>i</i> -stem, nominative Nominative-accusative plural - <i>iya</i> § 10.173 Nominative-accusative plural - <i>iya</i> § 10.174 Nominative-accusative plural - <i>iye</i> § 10.174 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>iye</i> § 10.176 Nominative-accusative plural - <i>iye</i> § 10.176 Masculine <i>i</i> -(<i>i</i> -)stems, nominative § 10.179 Masculine <i>i</i> -stems, accusative § 10.180 Masculine <i>in</i> -stems, nominative § 10.181 Masculine <i>in</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, nominative § 10.183 Feminine <i>i</i> -stems, accusative § 10.185 Feminine <i>i</i> -stems, accusative § 10.185 (Masculine <i>i</i> -stems, accusative § 10.185 Feminine <i>i</i> -stems, nominative § 10.186 Feminine <i>i</i> -stems, nominative § 10.186 (Masculine <i>i</i> -stems, accusative § 10.187 Noutier <i>i</i> and <i>in</i> -stems § 10.188 Nominative-accusative plural - <i>i</i> § 10.189 (Masculine <i>i</i> -stems, accusative; feminine <i>i</i> -stems, nominative, accusative; masculine <i>in</i> -stems, accusative; feminine <i>i</i> -stems, nominative, accusative; masculine <i>i</i> -stems, accusative; feminine <i>i</i> -stems, nominative, accusative; masculine <i>i</i> -stems, accusative; feminine <i>i</i> -stems, nominative, accusative; feminine <i>i</i> -stems; feminine <i>i</i> -stems; masculine <i>in</i> -stems; feminine <i>i</i> -stems; feminine <i>i</i> - stems, monosyllabic, polysyllabic)	81 81 81 81 81 81 81 81 81 81

Instrumental plural -ihi § 10.197	82
(Masculine i-stems; feminine i-stems; mascu-	
line <i>in</i> -stems)	~~
Instrumental plural -ibhi § 10.198	
Ablative plural - <i>ihi</i> § 10.199 Genitive plural § 10.200	82
Genitive plural -inām, from in-stems § 10.201	82
Genitive plural -ina § 10.202	
(Masculine and feminine i- and i-stems; in-	
stems)	
Genitive plural -inām, from i and i-stems § 10.203	83
Genitive plural -ina § 10.204	
Genitive plural -iya § 10.205	
Genitive plural - <i>isām</i> § 10.206	
Genitive plural - <i>im</i> § 10.207	83
Locative plural § 10.208	83
Locative plural <i>isu</i> , from <i>i</i> -stems § 10.208	
Locative plural -ișū and -ișū § 10.209	83
11. Feminines in -ā and -ī	84
······································	84
-ī for Sktā § 11.3	84
12. u and ū-stems	85
Forms with intrusive -n § 12.2	85
Stem form in composition § 12.4	85
\tilde{u} for u as stem final in composition § 12.6	85
	85
	85
Nominative singular feminine -u § 12.9	
Nominative singular feminine -v § 12.10	
Nominative singular feminine -ū § 12.11 Nominative singular feminine -um § 12.12	80 95
Nominative singular masculine -u § 12.12	
Vocative singular § 12.15	86
Vocative singular -u § 12.15	86
Accusative singular feminine -u § 12.18	
Accusative singular feminine -um § 12.20 Accusative singular feminine -uyam § 12.21	
Accusative singular masculine -u § 12.22	
	86
	86
Accusative singular masculine -us § 12.27	
Nominative-accusative singular neuter § 12.28	86
Nominative-accusative singular neuter -a § 12.28	86
Nominative-accusative singular neuter -um, -um	86
§ 12.29 Oblique singular, distinctively masculine-neuter forms	
§ 12.31	87
Instrumental singular -ūnā § 12.32	87
Ablative singular -uno § 12.33	87
	87
Genitive singular -usya § 12.35	87
Genitive singular -ūsya § 12.36	87
Genitive singular -o § 12.37	87
Locative singular -usmin or -usmim § 12.38	~-
	87
Locative singular -une § 12.39	87
Oblique singular feminine § 12.40	87 87
Oblique singular feminine § 12.40 Oblique singular -ūye § 12.41	87 87 87
Oblique singular feminine § 12.40 Oblique singular -āye § 12.41 Oblique singular -uye § 12.42	87 87
Oblique singular feminine § 12.40 Oblique singular - âye § 12.41 Oblique singular - aye § 12.42 Oblique singular - aya § 12.43 Oblique singular - aya § 12.43 Oblique singular - aya § 12.44	87 87 87 87
Oblique singular feminine § 12.40 Oblique singular - âye § 12.41 Oblique singular - aye § 12.42 Oblique singular - aya § 12.43 Oblique singular - aya § 12.43 Oblique singular - aya § 12.44	87 87 87 87 87
Oblique singular feminine § 12.40 Oblique singular -āye § 12.41 Oblique singular -uye § 12.42 Oblique singular -āya § 12.43 Oblique singular -uya § 12.44 Oblique singular -āya § 12.45 Oblique singular -āya § 12.45 Oblique singular -āya § 12.45	87 87 87 87 87 87 87 88 88 88
Oblique singular feminine § 12.40 Oblique singular -āye § 12.41 Oblique singular -uye § 12.42 Oblique singular -āya § 12.43 Oblique singular -uya § 12.43 Oblique singular -uya § 12.44 Oblique singular -āya § 12.45	87 87 87 87 87 87 87 88 88 88 88

Nominative-accusative plural -ava § 12.49	88
Nominative plural -uvah § 12.50	88
Nominative-accusative plural -uno § 12.51	-88
Nominative-accusative plural -ūna § 12.52	88
Nominative-accusative plural -ūni § 12.53	88
Nominative-accusative plural -uni § 12.54	88
A computing plural um um \$19.55	88
Accusative plural -um, -um § 12.55	88
Nominative plural -ūs, -ūh § 12.56	
Nominative plural -uh § 12.57	88
Nominative-accusative plural -uyo and -uyo § 12.58	
Nominative-accusative plural $-\hat{u}$ and $-u \S 12.59$	88
-ū, masc. fem. and nt., nominative and accusa-	
tive § 12.60	88
-u, masc. Iem. and nt., nominative and accusa-	
tive § 12.61	88
Vocative plural § 12.62	89
Vocative plural -ūh § 12.62	89
Vocative plural -0 § 12.63	89
Instrumental plural § 12.64	89
Instrumental plural -ūhi § 12.65	89
Instrumental plural -ūbhi § 12.66	89
Instrumental plural -uhi § 12.67	89
Instrumental plural -ubhi § 12.68	
	89
Instrumental plural -ubhir § 12.69	89
Genitive plural § 12.70	89
Genitive plural -ūna § 12.70	89
Genitive plural -unām, -unām § 12.71	89
Genitive plural -unam, -ūnam § 12.72	89
Locative plural -ūșu § 12.73	89
	<i></i>
13. r-stems	90
Stems in -a for -r § 13.3	90
Stems in -ā for -r § 13.7	90
Stem -ā in composition § 13.8	90
Vocative singular -e § 13.9	90
Accusative singular -ām (-ā) § 13.10	90
Oblique singular -āye, -āyam, -āya § 13.11	90
Nominative-accusative plural -ā(s) § 13.12	90
Stems in -ra (-rā, -rī) for -r § 13.14	90
Stems in -ara (-arā, -arī) for -r § 13.16	91
Masculine (and neuter) forms, stem -ara § 13.17	91
Feminine forms, stem -arā (-arī) § 13.18	91
Stems in and for at \$13.19	91
Stems in -āra for -r § 13.19 Stems in -i for -r § 13.20	91
Stems in $-u$ for $-r$ § 13.20	91
Stem in $-u$ in composition § 13.22	91
Nominative singular -uh or more often -u § 13.23	
Nominative singular -un or more often -u § 13.23	
Accusative singular -um (or -u, m.c.) § 13.24	91
Instrumental singular -unā § 13.25	91
Ablative singular -usmā § 13.26	91
Genitive singular masculine -uno § 13.27	91
Genitive plural -unām § 13.28	
Miscellaneous forms of r-stems § 13.29	91
Nominative singular -ās § 13.30	91
Vocative singular -ā § 13.31	92
Accusative singular -r § 13.32	92
Oblique singular -are, -ari § 13.33	92
Instrumental singular -rnā § 13.38	92
Genitive singular -u § 13.39	92
	92
Nominative plural -āra, -ara § 13.41	92
Accusative plural feminine $-rn $ § 13.42	92
Accusative plural -rn, -rm § 13.43	92
Instrumental plural -rhi, -rhi § 13.44	92
Genitive plural -rnām § 13.45	92 92
Locative plural - <i>F</i> su § 13.46	92
Locative plural -754 § 15.40	32
14. Diphthongal stems	93
15. Consonant stems	94
Vocalic stems produced by loss of final consonants § 15.2	
a-stems \$ 15.2	9.1

Feminine stems in -ā § 15.4	94
<i>i</i> -stems § 15.5	94
<i>u</i> -stems § 15.6	94
Consonant-stems made into a-stems by addition of $-a$	94
§ 15.7 Masculine and neuter stems § 15.7	94 94
Feminine stems § 15.9	95
The stem hrdi § 15.14	95
Other, miscellaneous irregularities § 15.15	95
16. s-stems	96
Stem -asa § 16.2	96
In composition § 16.2	96
Nominative singular masculine -asas, -aso, etc.	
	96
Nominative singular feminine -asa § 16.4 Vocative singular masculine -asa § 16.5	96 96
Nominative-accussative singular neuter -asam §16.6	96
Nominative-accusative singular neuter -asa § 16.7	96
Locative singular -ase § 16.8	96
Genitive plural -asānām § 16.9	96
Stem -a for -as § 16.10.	96
In composition § 16.10 Nominative singular masculine -as, -o etc. § 16.12	96 96
Nominative singular feminine $-\bar{a}$ § 16.13	96
Accusative singular masculine -am § 16.14	96
Accusative singular feminine -am § 16.15	96
Nominative-accusative singular neuter -am, -am	~~
§ 16.16 Instrumental singular <i>-ena</i> § 16.17	96 96
Ablative singular -ālo, and m.calo, -ālu, -alu	90
§ 16.18.	96
Genitive singular -asya § 16.19	96
Locative singular -e § 16.20	96
Locative singular -asmi § 16.21	97
Instrumental dual -ābhyām § 16.22	97
Nominative plural masculine and feminine -ās	-
	97
Nominative plural masculine and feminine -ās § 16.23 Nominative-accusative plural neuter -āni, m.cāni § 16.24	97 97
Nominative plural masculine and feminine -ās § 16.23 Nominative-accusative plural neuter -āni, m.cānī § 16.24 Accusative plural -a § 16.25	97 97 97
Nominative plural masculine and feminine -ās § 16.23 Nominative-accusative plural neuter -āni, m.cānī § 16.24 Accusative plural -a § 16.25 Instrumental plural -ais § 16.26	97 97 97 97
Nominative plural masculine and feminine -ās § 16.23 Nominative-accusative plural neuter -āni, m.cāni § 16.24 Accusative plural -a § 16.25 Instrumental plural -ais § 16.26 Genitive plural -ānām § 16.27	97 97 97
Nominative plural masculine and feminine -ās § 16.23 Nominative-accusative plural neuter -āni, m.cāni § 16.24 Accusative plural -a § 16.25 Instrumental plural -āis § 16.26 Genitive plural -ānāņ § 16.27 Locative plural -esu § 16.28	97 97 97 97 97
Nominative plural masculine and feminine -ās § 16.23	97 97 97 97 97 97 97 97
Nominative plural masculine and feminine -ās § 16.23	97 97 97 97 97 97 97 97
Nominative plural masculine and feminine -ās § 16.23. Nominative-accusative plural neuter -āni, m.cāni § 16.24. Accusative plural -a § 16.25. Instrumental plural -ais § 16.26 Genitive plural -ānām § 16.27. Locative plural -ēs § 16.28. Miscellaneous non-Sanskrit forms of as-stems § 16.29. Comparative stems in -yas § 16.34. Stems in -is § 16.36.	97 97 97 97 97 97 97 97 97
Nominative plural masculine and feminine -ās § 16.23. Nominative-accusative plural neuter -āni, m.cāni § 16.24. Accusative plural -a § 16.25. Instrumental plural -ais § 16.26. Genitive plural -ānām § 16.27. Locative plural -ais § 16.28. Miscellaneous non-Sanskrit forms of as-stems § 16.29. Comparative stems in -yas § 16.34. Stems in -is § 16.36. Stems in -is § 16.40. Stems in -us § 16.41.	97 97 97 97 97 97 97 97 97 97
Nominative plural masculine and feminine $-\bar{as}$ § 16.23 Nominative-accusative plural neuter $-\bar{ani}$, m.c. $-\bar{ani}$ § 16.24 Accusative plural $-a$ § 16.25 Instrumental plural $-ais$ § 16.26 Genitive plural $-\bar{anam}$ § 16.27 Locative plural $-eis$ § 16.28 Miscellaneous non-Sanskrit forms of as-stems § 16.29. Comparative stems in $-yas$ § 16.34 Stems in $-is$ § 16.40 Stems in $-us$ § 16.41 Stems in $-us$ § 16.42	97 97 97 97 97 97 97 97 97
Nominative plural masculine and feminine -ās § 16.23. Nominative-accusative plural neuter -āni, m.cāni § 16.24. Accusative plural -a § 16.25. Instrumental plural -ais § 16.26. Genitive plural -ānām § 16.27. Locative plural -esu § 16.28. Miscellaneous non-Sanskrit forms of as-stems § 16.29. Comparative stems in -yas § 16.34. Stems in -is § 16.36. Stems in -is § 16.40. Stems in -us § 16.41. Stems in -us § 16.42. Stems in -vāms, -us § 16.49.	97 97 97 97 97 97 97 97 97 97 97 97 97
Nominative plural masculine and feminine $-\bar{as}$ § 16.23. Nominative-accusative plural neuter $-\bar{ani}$, m.c. $-\bar{ani}$ § 16.24. Accusative plural $-a$ § 16.25. Instrumental plural $-ais$ § 16.26. Genitive plural $-\bar{anam}$ § 16.27. Locative plural $-eis$ § 16.28. Miscellaneous non-Sanskrit forms of as-stems § 16.29. Comparative stems in $-yas$ § 16.34. Stems in $-is$ § 16.40. Stems in $-us$ § 16.41. Stems in $-us$ § 16.42. Stems in $-us$ § 16.42. Stems in $-us$ § 16.49.	97 97 97 97 97 97 97 97 97 97 97
Nominative plural masculine and feminine -ās § 16.23. Nominative-accusative plural neuter -āni, m.cāni § 16.24. Accusative plural -a § 16.25. Instrumental plural -ais § 16.26 Genitive plural -ānām § 16.27 Locative plural -ēsu § 16.28 Miscellaneous non-Sanskrit forms of as-stems § 16.29 Comparative stems in -yas § 16.34 Stems in -us § 16.40 Stems in -us § 16.41 Stems in -us § 16.42 Stems in -us § 16.43 Stems in -us § 16.49 Meak stem in -a, declined as a-stem § 17.2 Nominative singular masculine -aħ, -o etc. § 17.4	97 97 97 97 97 97 97 97 97 97 97 97 97 9
Nominative plural masculine and feminine $-\bar{as}$ § 16.23. Nominative-accusative plural neuter $-\bar{ani}$, m.c. $-\bar{ani}$ § 16.24. Accusative plural $-ais$ § 16.25. Instrumental plural $-ais$ § 16.26. Genitive plural $-\bar{anan}$ § 16.27. Locative plural $-eis$ § 16.28. Miscellaneous non-Sanskrit forms of as-stems § 16.29. Comparative stems in $-yas$ § 16.34. Stems in $-is$ § 16.36. Stems in $-is$ § 16.40. Stems in $-us$ § 16.41. Stems in $-us$ § 16.42. Stems in $-us$ § 16.42. Stems in $-us$ § 16.42. Miscellaneous non-Sanskrit forms of as-stems § 17.2. Nominative singular masculine $-ah$, $-o$ etc. § 17.4 Nominative singular masculine $-u$ § 17.5.	97 97 97 97 97 97 97 97 97 97 97 97 97 9
Nominative plural masculine and feminine $-\bar{as}$ § 16.23. Nominative-accusative plural neuter $-\bar{ani}$, m.c. $-\bar{ani}$ § 16.24. Accusative plural $-a$ § 16.25. Instrumental plural $-ais$ § 16.26. Genitive plural $-\bar{asg}$ § 16.27. Locative plural $-\bar{sg}$ § 16.28. Miscellaneous non-Sanskrit forms of as -stems § 16.29. Comparative stems in $-yas$ § 16.34. Stems in $-is$ § 16.36. Stems in $-is$ § 16.40. Stems in $-us$ § 16.42. Stems in $-us$ § 16.42. Stems in $-us$ § 16.42. Miscellaneous non-Sanskrit forms of as -stems § 17.2. Stems in $-a$, declined as a -stem § 17.2. Nominative singular masculine $-a$, $-o$ etc. § 17.4 Nominative singular masculine $-a$ § 17.6.	97 97 97 97 97 97 97 97 97 97 97 97 97 9
Nominative plural masculine and feminine $-\bar{as}$ § 16.23. Nominative-accusative plural neuter $-\bar{ani}$, m.c. $-\bar{ani}$ § 16.24. Accusative plural $-a$ § 16.25. Instrumental plural $-ais$ § 16.26. Genitive plural $-\bar{anan}$ § 16.27. Locative plural $-eis$ § 16.28. Miscellaneous non-Sanskrit forms of as-stems § 16.29. Comparative stems in $-yas$ § 16.34. Stems in $-is$ § 16.40. Stems in $-us$ § 16.40. Stems in $-us$ § 16.42. Stems in $-us$ § 16.42. Stems in $-us$ § 16.42. Stems in $-us$ § 16.42. Miscellaneous non-sanskrit forms of as-stems § 17.2. Nominative singular masculine $-ah$, $-o$ etc. § 17.4 Nominative singular masculine $-a$ § 17.5. Nominative singular masculine $-a$ § 17.6. Vocative singular masculine $-a$ § 17.7.	97 97 97 97 97 97 97 97 97 97 97 97 97 9
Nominative plural masculine and feminine $-\bar{as}$ § 16.23. Nominative-accusative plural neuter $-\bar{ani}$, m.c. $-\bar{ani}$ § 16.24. Accusative plural $-a$ § 16.25. Instrumental plural $-ais$ § 16.26. Genitive plural $-eis$ § 16.27. Locative plural $-eis$ § 16.28. Miscellaneous non-Sanskrit forms of as -stems § 16.29. Comparative stems in $-yas$ § 16.34. Stems in $-is$ § 16.36. Stems in $-us$ § 16.41. Stems in $-us$ § 16.42. Stems in $-us$ § 16.42. Stems in $-va\bar{s}$ § 16.49. Miscellaneous non-Sanskrit forms of as -stems § 17.2. Stems in $-us$ § 16.42. Stems in $-us$ § 16.49. Miscellaneous non-Sanskrit forms of as -stems § 17.2. Nominative singular masculine $-ah$, $-o$ etc. § 17.4 Nominative singular masculine $-a$ § 17.5. Nominative singular masculine $-a\bar{g}$ § 17.8.	97 97 97 97 97 97 97 97 97 97 97 97 97 9
Nominative plural masculine and feminine $-\bar{as}$ § 16.23. Nominative-accusative plural neuter $-\bar{ani}$, m.c. $-\bar{ani}$ § 16.24. Accusative plural $-ais$ § 16.25. Instrumental plural $-ais$ § 16.26. Genitive plural $-\bar{anan}$ § 16.27. Locative plural $-\bar{ans}$ § 16.28. Miscellaneous non-Sanskrit forms of as-stems § 16.29. Comparative stems in $-yas$ § 16.34. Stems in $-us$ § 16.40. Stems in $-us$ § 16.40. Stems in $-us$ § 16.42. Stems in $-us$ § 16.42. Weak stem in $-a$, declined as a-stem § 17.2. Nominative singular masculine $-an$, $-o$ etc. § 17.4 Nominative singular masculine $-a$ § 17.6. Vocative singular masculine $-a$ § 17.7. Accusative singular masculine $-an$ § 17.8. Accusative singular masculine $-an$ § 17.8.	97 97 97 97 97 97 97 97 97 97 97 97 97 9
Nominative plural masculine and feminine $-\bar{as}$ § 16.23. Nominative-accusative plural neuter $-\bar{ani}$, m.c. $-\bar{ani}$ § 16.24. Accusative plural $-a$ § 16.25. Instrumental plural $-ais$ § 16.26. Genitive plural $-\bar{anam}$ § 16.27. Locative plural $-\bar{asg}$ § 16.28. Miscellaneous non-Sanskrit forms of as -stems § 16.29. Comparative stems in $-yas$ § 16.34. Stems in $-is$ § 16.36. Stems in $-us$ § 16.40. Stems in $-us$ § 16.42. Stems in $-us$ § 16.42. Stems in $-us$ § 16.42. Stems in $-us$ § 16.42. Miscellaneous non-Sanskrit forms of as -stems § 17.2. Nominative singular masculine $-a_{1}$, $-o$ etc. § 17.4 Nominative singular masculine $-a$ § 17.5. Nominative singular masculine $-a$ § 17.6. Vocative singular masculine $-a$ § 17.7. Accusative singular masculine $-a$ § 17.8. Accusative singular masculine $-a$ § 17.8. Nominative singular masculine $-a$ § 17.8. Accusative singular masculine $-a$ § 17.9. Nominative singular masculine $-a$ § 17.8. Accusative singular masculine $-a$ § 17.9. Nominative	97 97 97 97 97 97 97 97 97 97 97 97 97 9
Nominative plural masculine and feminine $-\bar{as}$ § 16.23. Nominative-accusative plural neuter $-\bar{ani}$, m.c. $-\bar{ani}$ § 16.24. Accusative plural $-a$ § 16.25. Instrumental plural $-ais$ § 16.26 Genitive plural $-\bar{anam}$ § 16.27. Locative plural $-eis$ § 16.28. Miscellaneous non-Sanskrit forms of as-stems § 16.29. Comparative stems in $-yas$ § 16.34. Stems in $-is$ § 16.36. Stems in $-us$ § 16.40 Stems in $-us$ § 16.41. Stems in $-us$ § 16.42. Stems in $-us$ § 16.42. Stems in $-us$ § 16.49. Minative singular masculine $-ah$, $-o$ etc. § 17.4 Nominative singular masculine $-a$ § 17.5. Nominative singular masculine $-a$ § 17.7. Accusative singular masculine $-a$ § 17.8. Accusative singular masculine $-af$ § 17.9. Nominative singular masculine $-aff$ § 17.9. Nominative singular masculine $-afff$ § 17.11.	97 97 97 97 97 97 97 97 97 97 97 97 97 9
Nominative plural masculine and feminine $-\bar{as}$ § 16.23. Nominative-accusative plural neuter $-\bar{ani}$, m.c. $-\bar{ani}$ § 16.24. Accusative plural $-ais$ § 16.25. Instrumental plural $-ais$ § 16.26. Genitive plural $-\bar{anan}$ § 16.27. Locative plural $-\bar{ans}$ § 16.28. Miscellaneous non-Sanskrit forms of as-stems § 16.29. Comparative stems in $-yas$ § 16.34. Stems in $-us$ § 16.40. Stems in $-us$ § 16.40. Stems in $-us$ § 16.41. Stems in $-us$ § 16.42. Nominative singular masculine $-an$, $-o$ etc. § 17.4 Nominative singular masculine $-a$ § 17.5. Nominative singular masculine $-a$ § 17.6. Vocative singular masculine $-a$ § 17.7. Accusative singular masculine $-a$ § 17.8. Accusative singular masculine $-a$ § 17.9. Nominative singular masculine $-an$ § 17.8. Accusative singular masculine $-a$ § 17.9. Nominative singular masculine $-a$ § 17.11. Instrumental singular $-ina$ § 17.12.	97 97997997799779977997799999999999999
Nominative plural masculine and feminine $-\bar{as}$ § 16.23. Nominative-accusative plural neuter $-\bar{ani}$, m.c. $-\bar{ani}$ § 16.24. Accusative plural $-a$ § 16.25. Instrumental plural $-a$ § 16.26. Genitive plural $-\bar{anan}$ § 16.27. Locative plural $-\bar{ans}$ § 16.28. Miscellaneous non-Sanskrit forms of as-stems § 16.29. Comparative stems in $-y$ as § 16.34. Stems in $-is$ § 16.36. Stems in $-is$ § 16.40. Stems in $-us$ § 16.41. Stems in $-us$ § 16.42. Stems in $-us$ § 16.42. Stems in $-a$, declined as a-stem § 17.2. Nominative singular masculine $-a$, $-o$ etc. § 17.4 Nominative singular masculine $-a$ § 17.5. Nominative singular masculine $-a$ § 17.6. Vocative singular masculine $-a$ § 17.8. Accusative singular masculine $-a$ § 17.8. Nominative singular masculine $-a$ § 17.8. Accusative singular masculine $-a$ § 17.8. Nominative singular masculine $-a$ § 17.8. Nominative singular masculine $-a$ § 17.8. Accusative singular masculine $-a$ § 17.4. Nominative singular masculine $-a$ § 17.8. Accusative singular masculine $-a$ § 17.4. Nominative accusative singular masculine $-a$ § 17.4. Ablative singular $-inaa$ § 17.11.	97 97 97 97 97 97 97 97 97 97 97 97 97 9
Nominative plural masculine and feminine $-\bar{as}$ § 16.23. Nominative-accusative plural neuter $-\bar{ani}$, m.c. $-\bar{ani}$ § 16.24. Accusative plural $-a$ § 16.25. Instrumental plural $-a$ § 16.26. Genitive plural $-\bar{anan}$ § 16.27. Locative plural $-\bar{ans}$ § 16.28. Miscellaneous non-Sanskrit forms of as-stems § 16.29. Comparative stems in $-y$ as § 16.34. Stems in $-u$ § 16.40. Stems in $-u$ § 16.41. Stems in $-u$ § 16.42. Stems in $-u$ § 16.42. Stems in $-u$ § 16.42. Nominative singular masculine $-an$, $-o$ etc. § 17.4 Nominative singular masculine $-an$ § 17.5. Nominative singular masculine $-an$ § 17.8. Accusative singular masculine $-an$ § 17.4. Instrumental singular $-in\bar{a}$ § 17.11. Instrumental singular $-in\bar{a}$ § 17.12. Ablative singular $-asya$ § 17.14. Locative singular $-asya$ § 17.14.	97 97997997799779977997799999999999999
Nominative plural masculine and feminine $-\bar{as}$ § 16.23. Nominative-accusative plural neuter $-\bar{ani}$, m.c. $-\bar{ani}$ § 16.24. Accusative plural $-a$ § 16.25. Instrumental plural $-a$ § 16.26. Genitive plural $-\bar{anan}$ § 16.27. Locative plural $-e$ § 16.28. Miscellaneous non-Sanskrit forms of as-stems § 16.29. Comparative stems in $-y$ s§ 16.34. Stems in $-u$ § 16.40. Stems in $-u$ § 16.40. Stems in $-u$ § 16.42. Stems in $-u$ § 16.42. Stems in $-u$ § 16.42. Nominative singular masculine $-a$, $-o$ etc. § 17.4 Nominative singular masculine $-a$ § 17.5. Nominative singular masculine $-a$ § 17.6. Vocative singular masculine $-a$ § 17.8. Accusative singular masculine $-a$ § 17.8. Accusative singular masculine $-a$ § 17.9. Nominative singular masculine $-a$ § 17.8. Accusative singular masculine $-a$ § 17.9. Nominative accusative singular neuter $-am$ (- am) § 17.10. Instrumental singular $-ina$ § 17.11. Instrumental singular $-ina$ § 17.12. Ablative singular $-a$ § 17.13. Cocative singular $-e$ § 17.15. Locative singular $-e$ § 17.16.	97 97 97 97 97 97 97 97 97 97 97 97 97 9
Nominative plural masculine and feminine $-\bar{as}$ § 16.23. Nominative-accusative plural neuter $-\bar{ani}$, m.c. $-\bar{ani}$ § 16.24. Accusative plural $-a$ § 16.25. Instrumental plural $-a$ § 16.26. Genitive plural $-\bar{anan}$ § 16.27. Locative plural $-\bar{ans}$ § 16.28. Miscellaneous non-Sanskrit forms of as-stems § 16.29. Comparative stems in $-y$ as § 16.34. Stems in $-is$ § 16.36. Stems in $-is$ § 16.40 Stems in $-us$ § 16.41. Stems in $-us$ § 16.42. Stems in $-us$ § 16.42. Stems in $-a$ declined as a-stem § 17.2. Nominative singular masculine $-a$, $-o$ etc. § 17.4 Nominative singular masculine $-a$ § 17.6. Vocative singular masculine $-a$ § 17.8. Accusative singular masculine $-a$ § 17.8. Accusative singular masculine $-a$ § 17.9. Nominative-accusative singular neuter $-am$ ($-am$) § 17.10. Instrumental singular $-inaa$ § 17.11. Instrumental singular $-inaa$ § 17.12. Ablative singular $-a$ § 17.14. Locative singular $-a$ § 17.16. Nominative singular $-is$ § 17.17.	97 97997997997997997997999999999999999
Nominative plural masculine and feminine $-\bar{as}$ § 16.23. Nominative-accusative plural neuter $-\bar{ani}$, m.c. $-\bar{ani}$ § 16.24. Accusative plural $-a$ § 16.25. Instrumental plural $-a$ § 16.26. Genitive plural $-\bar{anan}$ § 16.27. Locative plural $-\bar{as}$ § 16.28. Miscellaneous non-Sanskrit forms of as-stems § 16.29. Comparative stems in $-y$ as § 16.34. Stems in $-is$ § 16.36. Stems in $-is$ § 16.40. Stems in $-us$ § 16.41. Stems in $-us$ § 16.42. Stems in $-us$ § 16.42. Stems in $-u$ § 16.42. Mominative singular masculine $-a$, $-o$ etc. § 17.4 Nominative singular masculine $-a$ § 17.5. Nominative singular masculine $-a$ § 17.6. Vocative singular masculine $-a$ § 17.8. Accusative singular masculine $-a$ § 17.9. Nominative singular masculine $-a$ § 17.8. Accusative singular masculine $-a$ § 17.9. Nominative singular masculine $-a$ § 17.8. Accusative singular masculine $-a$ § 17.9. Nominative singular masculine $-a$ § 17.8. Accusative singular masculine $-a$ § 17.9. Nominative accusative singular neuter $-am$ (- am) § 17.10. Instrumental singular $-ena$ § 17.11. Instrumental singular $-ena$ § 17.12. Ablative singular $-a$ § 17.13. Genitive singular $-a$ § 17.15. Locative singular $-a$ § 17.16. Nominative accusative plural masculine $-\bar{a}$ § 17.17. Nominative accusative plural masculine $-\bar{a}$ § 17.18.	97 97997997997997997997997999999999999
Nominative plural masculine and feminine $-\bar{as}$ § 16.23 Nominative-accusative plural neuter $-\bar{ani}$, m.c. $-\bar{ani}$ § 16.24 Accusative plural $-a$ § 16.25 Instrumental plural $-a$ § 16.26 Genitive plural $-\bar{asg}$ § 16.27 Locative plural $-\bar{asg}$ § 16.28 Miscellaneous non-Sanskrit forms of as -stems § 16.29. Comparative stems in $-yas$ § 16.34 Stems in $-is$ § 16.36 Stems in $-is$ § 16.40 Stems in $-us$ § 16.41 Stems in $-us$ § 16.42 Nominative singular masculine $-a\bar{h}$, $-o$ etc. § 17.4 Nominative singular masculine $-a$ § 17.5 Nominative singular masculine $-a$ § 17.6 Vocative singular masculine $-a$ § 17.7 Accusative singular masculine $-a$ § 17.7 Nominative singular masculine $-a$ § 17.7 Nominative singular masculine $-a$ § 17.8 Accusative singular masculine $-a$ § 17.8 Accusative singular masculine $-a$ § 17.8 Accusative singular masculine $-a$ § 17.1. Instrumental singular $-ina$ § 17.14 Locative singular $-ina$ § 17.14 Locative singular $-ina$ § 17.16 Nominative plural masculine $-\bar{a}$ § 17.17 Nominative plural masculine $-\bar{a}$ § 17.18 Instrumental plural $-ehi$ § 17.19	97 97997997997997997997999999999999999

List of Chapter-subdivisions

Locative plurai -esu § 17.21	100
Prevocalic weak stem extended by -a § 17.22	100
Stem form in composition § 17.23	100
Stem form in composition § 17.25	
Nominative singular masculine -nas § 17.24	100
Nominative singular masculine -(a)na § 17.25	100
	100
Accusative singular masculine -anamy, -anamy 17.26	
Accusative singular masculine -ana or -ana § 17.27	100
Nominative-accusative singular neuter -(a)nam,	
-(a)nam § 17.28	100
Instrumental singular -(a)nena § 17.29	100
Ablative singular -nātu § 17.30	100
Genitive singular -(a)nasya § 17.31	100
Locative singular -ne § 17.32	100
Nominative plural masculine -nas § 17.33	100
Nominative-accusative plural neuter -näni § 17.34	100
Instrumental plural -nais § 17.35	100
Locative plural -nesu § 17.36	100
Locative plurat shear § 17.50	
Strong stem extended by -a § 17.37	100
Stem in composition § 17.38	100
Nominative singular masculine (-ānas,) -āno § 17.39	100
Nominative singular masculine -ānu § 17.40	100
Genitive singular -ānasya § 17.41	100
Landing singular on \$17.19	
Locative singular -āne § 17.42	100
Locative singular - <i>ani</i> § 17.43	100
Nominative plural masculine -ānās § 17.44	100
Instrumental plural -ānais § 17.45	100
Instrumental plural -ānehi § 17.46	-100
Locative plural -anesu § 17.47	100
The state of second before a \$ 17.49	
Epenthesis of vowel before n § 17.48	100
Miscellaneous non-Sanskritic forms § 17.51	101
Nominative singular masculine -as § 17.51	101
Nonthactve singular masedine as \$17.51	
Vocative singular masculine -ā § 17.52	101
Vocative singular masculine -e § 17.53	101
Accusative singular masculine -a § 17.54	101
Accusative singular masculine -a g 17.54	
Accusative singular masculine -ana § 17.55	101
Nominative-accusative singular neuter -a § 17.56	101
Genitive singular -ānas, -āno § 17.57	101
Genitive singular $-(a)na \S 17.58$	101
Locative singular $-(a)ni $ § 17.59	101
Maminating plugal magaziling and \$17.00	
Nominative plural masculine -ana § 17.60	101
Nominative plural masculine -nas § 17.61	101
Nominative plural masculine -nas § 17.61 Accusative plural masculine -ānah § 17.62	
Accusative plural masculine -ānah § 17.62	101
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63	101 101
Accusative plural masculine -ānah § 17.62	101
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63	101 101
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64	101 101 101
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems	101 101 101 102
Accusative plural masculine -ānah § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2	101 101 101
Accusative plural masculine -ānah § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2	101 101 101 102 102
Accusative plural masculine -ānah § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nta § 18.4	101 101 101 102 102 102
Accusative plural masculine -ānah § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5	101 101 101 102 102
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -ntaḥ, -nto, etc.	101 101 101 102 102 102
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -ntaḥ, -nto, etc.	101 101 101 102 102 102 102
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -ntaḥ, -nto, etc. § 18.6	101 101 101 102 102 102 102 102
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -ntaḥ, -nto, etc. § 18.6 Nominative singular masculine -nta § 18.7	101 101 102 102 102 102 102
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -ntaḥ, -nto, etc. § 18.6	101 101 101 102 102 102 102 102
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -nta § 18.6 Nominative singular masculine -nta § 18.7 Nominative singular masculine -nta § 18.8	101 101 102 102 102 102 102 102 102
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -ntaḥ, -nto, etc. § 18.6 Nominative singular masculine -nta § 18.7 Nominative singular masculine -nta § 18.8 Nominative singular feninine -ntā § 18.9	101 101 102 102 102 102 102 102 102 102
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -ntaķ, -nto, etc. § 18.6 Nominative singular masculine -nta § 18.7 Nominative singular masculine -nta § 18.7 Nominative singular feminine -nta § 18.8 Nominative singular feminine -nta § 18.9 Vocative singular -nta § 18.10	101 101 102 102 102 102 102 102 102
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -ntaḥ, -nto, etc. § 18.6 Nominative singular masculine -nta § 18.7 Nominative singular masculine -nta § 18.8 Nominative singular feninine -ntā § 18.9	101 101 102 102 102 102 102 102 102 102
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nla § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -ntaḥ, -nto, etc. § 18.6 Nominative singular masculine -nta § 18.7 Nominative singular masculine -nta § 18.7 Nominative singular feminine -nta § 18.9 Vocative singular feminine -ntā § 18.9 Vocative singular feminine -ntā § 18.0	101 101 102 102 102 102 102 102 102 102
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nla § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -ntaḥ, -nto, etc. § 18.6 Nominative singular masculine -nta § 18.7 Nominative singular feminine -ntā § 18.9 Vocative singular feminine -ntā § 18.9 Accusative singular feminine -ntāņ, and m.cnta § 18.11	101 101 102 102 102 102 102 102 102 102
Accusative plural masculine -ānah § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -ntah, -nto, etc. § 18.6 Nominative singular masculine -nta § 18.7 Nominative singular femínine -nta § 18.7 Nominative singular femínine -nta § 18.7 Nominative singular femínine -nta § 18.9 Vocative singular -nta § 18.10 Accusative singular femínine -ntām, and m.cnta § 18.11 Nominative-accusative singular neuter -ntam	101 101 101 102 102 102 102 102 102 102
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nla § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -ntaḥ, -nto, etc. § 18.6 Nominative singular masculine -nta § 18.7 Nominative singular feminine -ntā § 18.9 Vocative singular feminine -ntā § 18.9 Accusative singular feminine -ntāņ, and m.cnta § 18.11	101 101 102 102 102 102 102 102 102 102
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahāni- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -ntaḥ, -nto, etc. § 18.6. Nominative singular masculine -nta § 18.7 Nominative singular masculine -ntā § 18.7 Nominative singular feminine -ntā § 18.9 Vocative singular feminine -ntā § 18.9 Vocative singular feminine -ntām, and m.cnta § 18.11. Nominative-accusative singular neuter -ntām § 18.12.	101 101 101 102 102 102 102 102 102 102
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -ntaķ, -nto, etc. § 18.6 Nominative singular masculine -nta § 18.7 Nominative singular masculine -nta § 18.8 Nominative singular feminine -nta § 18.9 Vocative singular feminine -ntā § 18.9 Vocative singular feminine -ntā § 18.10 Accusative singular feminine -ntāņ, and m.cnta § 18.11 Nominative-accusative singular neuter -ntaṃ § 18.12 Nominative singular feminine -stāā § 18.13	101 101 101 102 102 102 102 102 102 102
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nla § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -ntaḥ, -nto, etc. § 18.6 Nominative singular masculine -nta § 18.7 Nominative singular masculine -nta § 18.7 Nominative singular feminine -nta § 18.8 Vocative singular feminine -ntā § 18.9 Vocative singular feminine -ntā § 18.9 Nominative-accusative singular neuter -ntaṃ § 18.11 Nominative-accusative singular neuter -ntāṃ § 18.12 Instrumental singular -ntaṣ § 18.13 Genitive singular -ntaṣya § 18.14	101 101 101 102 102 102 102 102 102 102
Accusative plural masculine -ānah § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -ntah, -nto, etc. § 18.6 Nominative singular masculine -nta § 18.7 Nominative singular feminine -ntā § 18.9 Vocative singular feminine -ntā § 18.9 Vocative singular feminine -ntā § 18.9 Nominative singular feminine -ntā § 18.11 Nominative singular feminine -ntām, and m.cnta § 18.11 Nominative-accusative singular neuter -ntam § 18.12 Instrumental singular -ntena § 18.13 Genitive singular -ntena § 18.15	101 101 101 102 102 102 102 102 102 102
Accusative plural masculine -ānaņ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -ntaħ, -nto, etc. § 18.6 Nominative singular masculine -nta § 18.7 Nominative singular feminine -ntā § 18.9 Vocative singular feminine -ntā § 18.9 Vocative singular feminine -ntāħ, and m.cnta § 18.11 Nominative-accusative singular neuter -ntaħ § 18.12 Instrumental singular -ntena § 18.13 Genitive singular -ntena § 18.15	101 101 101 102 102 102 102 102 102 102
Accusative plural masculine -ānah § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahāni- § 18.2 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -nta ķ 18.7 Nominative singular masculine -nta § 18.7 Nominative singular masculine -nta § 18.7 Nominative singular masculine -nta § 18.7 Nominative singular feminine -nta § 18.7 Nominative singular feminine -nta § 18.9 Vocative singular feminine -ntā § 18.9 Nocative singular feminine -ntā § 18.10 Accusative-accusative singular neuter -ntam § 18.11 Nominative-accusative singular neuter -ntam § 18.12 Instrumental singular -ntena § 18.13 Genitive singular -nte § 18.15 Locative singular -nte § 18.16	101 101 101 102 102 102 102 102 102 102
Accusative plural masculine -ānah § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -nta § 18.7 Nominative singular masculine -nta § 18.7 Nominative singular masculine -nta § 18.7 Nominative singular feminine -nta § 18.8 Nominative singular feminine -nta § 18.9 Vocative singular feminine -ntā § 18.9 Nominative singular feminine -ntā § 18.10 Accusative singular feminine -ntā § 18.11 Nominative singular feminine -ntā § 18.12 Instrumental singular -nta § 18.13 Genitive singular -ntasya § 18.14 Locative singular -nta § 18.15 Locative singular -nta § 18.16 Nominative plural masculine -ntās § 18.17	101 101 101 102 102 102 102 102 102 102
Accusative plural masculine -ānah § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -nta § 18.7 Nominative singular masculine -nta § 18.7 Nominative singular masculine -nta § 18.7 Nominative singular feminine -nta § 18.8 Nominative singular feminine -nta § 18.9 Vocative singular feminine -ntā § 18.9 Nominative singular feminine -ntā § 18.10 Accusative singular feminine -ntā § 18.11 Nominative singular feminine -ntā § 18.12 Instrumental singular -nta § 18.13 Genitive singular -ntasya § 18.14 Locative singular -nta § 18.15 Locative singular -nta § 18.16 Nominative plural masculine -ntās § 18.17	101 101 101 102 102 102 102 102 102 102
Accusative plural masculine -ānah § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -ntah, -nto, etc. § 18.6 Nominative singular masculine -nta § 18.7 Nominative singular masculine -nta § 18.7 Nominative singular feminine -nta § 18.8 Nominative singular feminine -nta § 18.9 Vocative singular feminine -ntā § 18.9 Nominative singular feminine -ntā § 18.10 Accusative singular feminine -ntā § 18.13 Genitive singular -nta § 18.10 Accusative singular -nta § 18.13 Genitive singular -ntasya § 18.14 Locative singular -nta § 18.15 Locative singular -nta § 18.16 Nominative plural masculine -ntā § 18.17	101 101 101 102 102 102 102 102 102 102
Accusative plural masculine -ānah § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nla § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -nlah, -nlo, etc. § 18.6 Nominative singular masculine -nla § 18.7 Nominative singular masculine -nla § 18.8 Nominative singular feminine -nla § 18.9 Vocative singular feminine -nlā § 18.9 Vocative singular feminine -nlā § 18.10 Accusative singular feminine -nlāh, and m.cnla § 18.11 Nominative-accusative singular neuter -nlam § 18.12 Instrumental singular -nlas § 18.13 Genitive singular -nla § 18.15 Locative singular -nla § 18.15 Locative singular -nla § 18.16 Nominative plural masculine -nlā § 18.17 Nominative plural masculine -nlā § 18.18.17	101 101 101 102 102 102 102 102 102 102
Accusative plural masculine -ānah § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahānt- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -ntah, -nto, etc. § 18.6 Nominative singular masculine -nta § 18.7 Nominative singular masculine -nta § 18.7 Nominative singular feminine -nta § 18.8 Nominative singular feminine -nta § 18.9 Vocative singular feminine -ntā § 18.9 Nominative singular feminine -ntā § 18.10 Accusative singular feminine -ntā § 18.13 Genitive singular -nta § 18.10 Accusative singular -nta § 18.13 Genitive singular -ntasya § 18.14 Locative singular -nta § 18.15 Locative singular -nta § 18.16 Nominative plural masculine -ntā § 18.17	101 101 101 102 102 102 102 102 102 102
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahāni- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5. Nominative singular masculine -ntaḥ, -nto, etc. § 18.6 Nominative singular masculine -nta § 18.7 Nominative singular masculine -nta § 18.7 Nominative singular feminine -ntā § 18.9 Vocative singular feminine -ntā § 18.9 Nominative singular feminine -ntā § 18.9 Nominative singular singular neuter -nta § 18.11 Nominative-accusative singular neuter -ntam § 18.12 Instrumental singular -ntena § 18.13 Genitive singular -nte § 18.15 Locative singular -nte § 18.15 Nominative plural masculine -ntā § 18.17 Nominative plural masculine -ntā § 18.19 Nominative plural masculine -ntā § 18.19 Nominative plural masculine -ntā § 18.19	101 101 101 102 102 102 102 102 102 102
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 18. nt-stems Forms of stem mahāni- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -ntaḥ, -nto, etc. § 18.6 Nominative singular masculine -ntā § 18.7 Nominative singular feminine -ntā § 18.9 Vocative singular feminine -ntā § 18.9 Vocative singular feminine -ntā § 18.9 Nominative-accusative singular neuter -ntaṃ § 18.11. Nominative-accusative singular neuter -ntaṃ § 18.12. Instrumental singular -ntena § 18.13 Genitive singular -nte § 18.14 Locative singular -nte § 18.16 Nominative plural masculine -ntā § 18.17 Nominative plural masculine -ntā § 18.18 Nominative plural masculine -ntā § 18.19 Nominative plural masculine -ntā § 18.19 Accusative plural masculine -ntā § 18.19	101 101 101 102 102 102 102 102 102 102
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 Is. nt-stems Forms of stem mahānt- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -nta § 18.7 Nominative singular masculine -nta § 18.7 Nominative singular masculine -nta § 18.8 Nominative singular feminine -ntā § 18.9 Vocative singular feminine -ntā § 18.9 Nocative singular feminine -ntā § 18.9 Nominative singular feminine -ntā § 18.9 Nominative singular feminine -ntā § 18.10 Accusative singular -nta § 18.10 Nominative singular feminine -ntām, and m.cnta § 18.11 Nominative singular -nta § 18.13 Genitive singular -ntasya § 18.14 Locative singular -nta § 18.15 I.ocative singular -nta § 18.15 Nominative plural masculine -ntā § 18.17 Nominative plural masculine -ntā § 18.19 Accusative plural masculine -ntā § 18.19 Accusative plural masculine -ntā § 18.20	101 101 101 102 102 102 102 102 102 102
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 Is. nt-stems Forms of stem mahānt- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -nta, -nto, etc. § 18.6. Nominative singular masculine -nta § 18.7 Nominative singular masculine -nta § 18.8 Nominative singular feminine -ntā § 18.9 Vocative singular feminine -ntā § 18.9 Vocative singular feminine -ntā § 18.9 Nominative-accusative singular neuter -ntam § 18.11. Nominative singular -nta § 18.13 Locative singular -nta § 18.15 Locative singular -nta § 18.15 Locative singular -nta § 18.15 Nominative plural masculine -ntā § 18.17 Nominative plural masculine -ntā § 18.18. Nominative plural masculine -ntā § 18.19 Accusative plural masculine -ntā § 18.19 Accusative plural masculine -ntā § 18.21 Nominative plural masculine -ntā § 18.21	101 101 101 102 102 102 102 102 102 102
Accusative plural masculine -ānaḥ § 17.62 Accusative plural masculine -āna § 17.63 Instrumental plural -ābhis? § 17.64 Is. nt-stems Forms of stem mahānt- § 18.2 Stem -nta § 18.4 Stem in composition and before suffix -tara § 18.5 Nominative singular masculine -nta § 18.7 Nominative singular masculine -nta § 18.7 Nominative singular masculine -nta § 18.8 Nominative singular feminine -ntā § 18.9 Vocative singular feminine -ntā § 18.9 Nocative singular feminine -ntā § 18.9 Nominative singular feminine -ntā § 18.9 Nominative singular feminine -ntā § 18.10 Accusative singular -nta § 18.10 Nominative singular feminine -ntām, and m.cnta § 18.11 Nominative singular -nta § 18.13 Genitive singular -ntasya § 18.14 Locative singular -nta § 18.15 I.ocative singular -nta § 18.15 Nominative plural masculine -ntā § 18.17 Nominative plural masculine -ntā § 18.19 Accusative plural masculine -ntā § 18.19 Accusative plural masculine -ntā § 18.20	101 101 101 102 102 102 102 102 102 102

Instrumental plural -ntais § 18.24	103
Instrumental plural -ulebhi(s) § 18.25	103
Dative plural -ntebhyas § 18.26	103
Genitive plural -ntānām § 18.27	103
Genitive plural -nlānam § 18.28	103
Genitive plural -ntāna § 18.29	103
Genitive plural -ntān' § 18.30	103
Locative plural -ntesu § 18.31	103
Stems in -la § 18.32	
Stem in composition § 18.32	103
Nominative singular masculine <i>-las</i> , <i>-lo</i> § 18.33.	103
Nominative singular masculine -ta § 18.34 Nominative singular masculine -ta § 18.35	103
Nominative singular feminine -ta § 18.36	103
Vocative singular masculine - <i>la</i> § 18.37	$\frac{103}{103}$
Accusative singular masculine -lam § 18.38	103
Accusative singular masculine - <i>Iu</i> § 18.39	103
Accusative singular feminine <i>-tām</i> § 18.40	103
Nominative-accusative singular neuter -lam, -lam	100
§ 18.41	103
Instrumental singular -lena § 18.42	103
Genitive singular -lasya § 18,43	103
Locative singular -le § 18.44	103
Nominative dual -lau § 18.45.	
Nominative plural masculine -lā § 18.46	103
Nominative plural feminine $-l\hat{a}(s) $ § 18.47	103
Accusative plural masculine (-lān,) -lām § 18.48	103
Accusative plural masculine -la § 18.49;	103
	103
Genitive plural -lānām § 18.51	103
Forms based on a stem in -a (minus -nt-) § 18.52.	103
Stem -a in composition § 18.52	103
Stem -am in composition § 18.54	104
Nominative singular masculine -as, -o etc. § 18.55	104
Nominative singular feminine $-\bar{a}$ § 18.57	104
Accusative singular masculine -am § 18.58	104
Accusative singular masculine $-a \S 18.59$	104
Nominative-accusative singular neuter -am § 18.60	104
Instrumental singular -ena § 18.61	104
Genitive singular -asya § 18.62	104
Locative singular feminine -āyām § 18.63	104
Nominative plural masculine -as § 18.64	104
Nominative-accusative plural neuter -āni, -ā, -a	
§ 18.65	104
Substitution of <i>n</i> for <i>nt</i> stem forms § 18.66	104
Strong stem used for weak and vice versa § 18.68.	104
Instrumental singular <i>-ntā</i> § 18.68 Genitive singular <i>-ntaḥ</i> § 18.69	104
Nominative plural $-tas, -to \S 18.70$	104 104
	104
Accusative plural -nlas, -nlo § 18.72	104
Accusative plural -nla, -nla § 18.73	104
	104
	104
Nominative singular masculine, miscellaneous	
	104
Nominative singular feminine -(m)ām § 18.80	105
Vocative singular masculine -ān, -ām § 18.81	105
Vocative singular masculine -a § 18.82	105
Accusative singular masculine -nta, -ntu § 18.83	105
Instrumental singular -ta § 18.84	105
Genitive singular - <i>lu</i> § 18.85	105
Nominative plural masculine -nta, -ntu § 18.86.	105
Nominative plural masculine -nle § 18.87	105
Nominative plural masculine -(v)an § 18.88	105
Genitive plural - <i>tu</i> § 18.89	105
	106
eka § 19.1	106

xv

XVI

catur etc. § 19.13	
	106
pañca § 19.22	106
sas § 19.24	107
sapta § 19.25	107
dasa § 19.26	107
Teens § 19.28	107
	107
Tens § 19.29	107
Ordinals § 19.36	4.07
Nominative forms with other cases and in composition	105
§ 19.37	107
20. Personal pronouns	108
Stem in composition § 20.1	108
Nominative singular § 20.6	108
First person § 20.6	108
Second person § 20.8	108
Accusative singular § 20.11	108
First person § 20.11	108
Second person § 20.14	109
Instrumental singular § 20.17	109
First person § 20.17	109
Second person § 20.20	109
Dative singular § 20.24	110
Ablative singular § 20.25	110
Genitive singular § 20.26	110
First person § 20.27	110
Second person § 20.31	110
Locative singular, first person § 20.36	111
Locative singular, inst person y 20.00	
Nominative plural § 20.37	111
First person § 20.37	111
Second person § 20.42	111
Accusative plural § 20.45	111
First person § 20.45	111
Second person § 20.49	112
Instrumental plural § 20.51	112
First person § 20.51	112
Second person § 20.53	112
Ableticia minural first mercen f 00 55	
Ablative plural, first person § 20.55	112
Genitive plural § 20.56	112
First person § 20.56	112
Second person § 20.61	113
	113
Table of Personal Pronouns § 20.63	
21. Generic pronouns	114
21. Generic pronouns Stems ta (sa), eta (eşa), ya, ka and prominal adjectives	114
21. Generic pronouns Stems la (sa), ela (eşa), ya, ka and prominal adjectives § 21.1	114 114
21. Generic pronouns Stems la (sa), ela (eşa), ya, ka and prominal adjectives § 21.1 Stem in composition § 21.2	114 114 114
21. Generic pronouns Stems la (sa), ela (eşa), ya, ka and prominal adjectives § 21.1 Stem in composition § 21.2 Nominative singular masculine § 21.5	114 114 114 114
21. Generic pronouns Stems ta (sa), eta (eşa), ya, ka and prominal adjectives § 21.1 Stem in composition § 21.2 Nominative singular masculine § 21.5 Nominative singular feminine § 21.8	114 114 114 114 114
21. Generic pronouns Stems la (sa), ela (eşa), ya, ka and prominal adjectives § 21.1 Stem in composition § 21.2 Nominative singular masculine § 21.5 Nominative singular feminine § 21.8 Accusative singular masculine and feminine § 21.9	114 114 114 114 114
21. Generic pronouns	114 114 114 114 114 114 114 114
21. Generic pronouns	114 114 114 114 114 114 114 114
21. Generic pronouns	114 114 114 114 114 114 114 114
 21. Generic pronouns Stems ta (sa), eta (eşa), ya, ka and prominal adjectives § 21.1 Stem in composition § 21.2 Nominative singular masculine § 21.5 Nominative singular feminine § 21.8 Accusative singular masculine and feminine § 21.9 Nominative-accusative singular neuter § 21.10. Oblique singular feminine § 21.13 Instrumental singular masculine and neuter § 21.14 	114 114 114 114 114 114 114 114 114
 21. Generic pronouns	114 114 114 114 114 114 114 114 114 115
 21. Generic pronouns Stems la (sa), ela (eşa), ya, ka and prominal adjectives § 21.1 Stem in composition § 21.2 Nominative singular masculine § 21.5 Nominative singular feminine § 21.8 Accusative singular masculine and feminine § 21.9 Nominative-accusative singular neuter § 21.10. Oblique singular feminine § 21.3 Instrumental singular masculine and neuter § 21.14 Ablative singular masculine and neuter § 21.15. Genitive singular masculine and neuter § 21.16 	114 114 114 114 114 114 114 114 114 115 115
21. Generic pronouns Stems la (sa), ela (eşa), ya, ka and prominal adjectives § 21.1 Stem in composition § 21.2 Nominative singular masculine § 21.5 Nominative singular feminine § 21.8 Accusative singular masculine and feminine § 21.9 Nominative-accusative singular neuter § 21.10 Oblique singular feminine § 21.13 Instrumental singular masculine and neuter § 21.15. Genitive singular masculine and neuter § 21.15. Genitive singular feminine and neuter § 21.16. Stative singular feminine and neuter § 21.17	114 114 114 114 114 114 114 114 114 115
21. Generic pronouns	114 114 114 114 114 114 114 114 114 115 115
21. Generic pronouns Stems ta (sa), eta (eşa), ya, ka and prominal adjectives § 21.1 Nominative singular masculine § 21.5 Nominative singular feminine § 21.8 Accusative singular masculine and feminine § 21.9 Nominative-accusative singular neuter § 21.10 Oblique singular feminine § 21.13 Instrumental singular masculine and neuter § 21.15. Genitive singular masculine and neuter § 21.16 Genitive singular feminine § 21.17 Genitive singular masculine and feminine se, si § 21.18	114 114 114 114 114 114 114 114 115 115
 21. Generic pronouns Stems ta (sa), eta (eşa), ya, ka and prominal adjectives § 21.1 Stem in composition § 21.2 Nominative singular masculine § 21.5 Nominative singular feminine § 21.8 Accusative singular masculine and feminine § 21.9 Nominative-accusative singular neuter § 21.10. Oblique singular feminine § 21.3 Instrumental singular masculine and neuter § 21.15. Genitive singular masculine and neuter § 21.16 Genitive singular masculine and neuter § 21.17 Genitive singular masculine and feminine se, si § 21.18 Locative singular masculine and neuter § 21.20 	114 114 114 114 114 114 114 114 115 115
 21. Generic pronouns Stems la (sa), ela (eşa), ya, ka and prominal adjectives § 21.1 Stem in composition § 21.2 Nominative singular masculine § 21.5 Nominative singular feminine § 21.8 Accusative singular masculine and feminine § 21.9 Nominative-accusative singular neuter § 21.10. Oblique singular feminine § 21.3 Instrumental singular masculine and neuter § 21.15. Genitive singular masculine and neuter § 21.16 Genitive singular masculine and neuter § 21.16 Genitive singular masculine and neuter § 21.16 Genitive singular masculine and feminine se, si § 21.18 Locative singular masculine and neuter § 21.20 Locative singular feminine § 21.23 	114 114 114 114 114 114 114 114 114 115 115
 21. Generic pronouns Stems la (sa), ela (eşa), ya, ka and prominal adjectives § 21.1 Stem in composition § 21.2 Nominative singular masculine § 21.5 Nominative singular feminine § 21.8 Accusative singular masculine and feminine § 21.9 Nominative-accusative singular neuter § 21.10. Oblique singular feminine § 21.13 Instrumental singular masculine and neuter § 21.15 Genitive singular masculine and neuter § 21.15 Genitive singular masculine and neuter § 21.16 Genitive singular feminine § 21.17 Genitive singular masculine and neuter § 21.18 Locative singular masculine and neuter § 21.24 	114 114 114 114 114 114 114 114 114 115 115
 21. Generic pronouns Stems ta (sa), eta (eşa), ya, ka and prominal adjectives § 21.1. Stem in composition § 21.2. Nominative singular masculine § 21.5. Nominative singular feminine § 21.8. Accusative singular masculine and feminine § 21.9 Nominative-accusative singular neuter § 21.10. Oblique singular feminine § 21.13. Instrumental singular masculine and neuter § 21.15. Genitive singular masculine and neuter § 21.16. Genitive singular masculine and neuter § 21.17. Genitive singular masculine and neuter § 21.16. Genitive singular masculine and neuter § 21.20. Locative singular masculine and neuter § 21.20. Locative singular feminine § 21.23. Nominative-accusative dual neuter § 21.24. Nominative plural masculine § 21.25. 	114 114 114 114 114 114 114 114 114 115 115
 21. Generic pronouns Stems ta (sa), eta (eşa), ya, ka and prominal adjectives § 21.1. Stem in composition § 21.2. Nominative singular masculine § 21.5. Nominative singular feminine § 21.8. Accusative singular masculine and feminine § 21.9 Nominative-accusative singular neuter § 21.10. Oblique singular feminine § 21.13. Instrumental singular masculine and neuter § 21.15. Genitive singular masculine and neuter § 21.16. Genitive singular masculine and neuter § 21.17. Genitive singular masculine and neuter § 21.16. Genitive singular masculine and neuter § 21.20. Locative singular masculine and neuter § 21.20. Locative singular feminine § 21.23. Nominative-accusative dual neuter § 21.24. Nominative plural masculine § 21.25. 	114 114 114 114 114 114 114 114 114 115 115
 21. Generic pronouns Stems ta (sa), eta (eşa), ya, ka and prominal adjectives § 21.1 Stem in composition § 21.2 Nominative singular masculine § 21.5 Nominative singular feminine § 21.8 Accusative singular masculine and feminine § 21.9 Nominative-accusative singular neuter § 21.10. Oblique singular feminine § 21.3 Instrumental singular masculine and neuter § 21.15. Genitive singular masculine and neuter § 21.16 Genitive singular masculine and neuter § 21.16 Genitive singular feminine § 21.17 Genitive singular masculine and neuter § 21.20 Locative singular masculine and neuter § 21.20 Locative singular feminine § 21.23 Nominative-accusative dual neuter § 21.24 Nominative plural masculine § 21.29 Nominative plural feminine § 21.29 	114 114 114 114 114 114 114 114 114 114 115 115 115 115 115 115 115 115 115 115 115 115 115 115 115 115
 21. Generic pronouns Stems ta (sa), eta (eşa), ya, ka and prominal adjectives § 21.1 Stem in composition § 21.2 Nominative singular masculine § 21.5 Nominative singular feminine § 21.8 Accusative singular masculine and feminine § 21.9 Nominative-accusative singular neuter § 21.10. Oblique singular feminine § 21.3 Instrumental singular masculine and neuter § 21.15. Genitive singular masculine and neuter § 21.16 Genitive singular masculine and neuter § 21.16 Genitive singular feminine § 21.17 Genitive singular masculine and neuter § 21.20 Locative singular masculine and neuter § 21.20 Locative singular feminine § 21.23 Nominative-accusative dual neuter § 21.24 Nominative plural masculine § 21.29 Accusative plural feminine § 21.30 	$\begin{array}{c} 114\\ 114\\ 114\\ 114\\ 114\\ 114\\ 114\\ 114$
 21. Generic pronouns Stems la (sa), ela (eşa), ya, ka and prominal adjectives § 21.1 Stem in composition § 21.2 Nominative singular masculine § 21.5 Nominative singular feminine § 21.8 Accusative singular masculine and feminine § 21.9 Nominative-accusative singular neuter § 21.10. Oblique singular feminine § 21.13 Instrumental singular masculine and neuter § 21.14 Ablative singular masculine and neuter § 21.15 Genitive singular masculine and neuter § 21.16 Genitive singular masculine and neuter § 21.16 Genitive singular masculine and neuter § 21.16 Genitive singular masculine and neuter § 21.20 Locative singular masculine and neuter § 21.20 Locative singular feminine § 21.23 Nominative-accusative dual neuter § 21.24 Nominative plural masculine § 21.29 Accusative plural feminine § 21.35 	$\begin{array}{c} 114\\ 114\\ 114\\ 114\\ 114\\ 114\\ 114\\ 114$
 21. Generic pronouns Stems ta (sa), eta (eşa), ya, ka and prominal adjectives § 21.1. Stem in composition § 21.2. Nominative singular masculine § 21.5. Nominative singular feminine § 21.8. Accusative singular masculine and feminine § 21.9 Nominative-accusative singular neuter § 21.10. Oblique singular feminine § 21.13. Instrumental singular masculine and neuter § 21.14 Ablative singular masculine and neuter § 21.15. Genitive singular masculine and neuter § 21.16. Genitive singular masculine and neuter § 21.17. Genitive singular masculine and neuter § 21.16. Genitive singular feminine § 21.17. Genitive singular masculine and neuter § 21.20. Locative singular masculine and neuter § 21.20. Locative singular feminine § 21.23. Nominative-accusative dual neuter § 21.24. Nominative plural masculine § 21.29. Accusative plural feminine § 21.30. Accusative plural feminine § 21.35. 	$\begin{array}{c} 114\\ 114\\ 114\\ 114\\ 114\\ 114\\ 114\\ 114$
 21. Generic pronouns Stems ta (sa), eta (eşa), ya, ka and prominal adjectives § 21.1. Stem in composition § 21.2. Nominative singular masculine § 21.5. Nominative singular feminine § 21.8. Accusative singular masculine and feminine § 21.9 Nominative-accusative singular neuter § 21.10. Oblique singular feminine § 21.13. Instrumental singular masculine and neuter § 21.15. Genitive singular masculine and neuter § 21.16. Genitive singular masculine and neuter § 21.17. Genitive singular masculine and neuter § 21.16. Genitive singular feminine § 21.17. Genitive singular feminine § 21.17. Genitive singular masculine and neuter § 21.20. Locative singular masculine and neuter § 21.20. Locative singular feminine § 21.23. Nominative-accusative dual neuter § 21.24. Nominative plural masculine § 21.29. Accusative plural feminine § 21.30. Accusative plural feminine § 21.36. Nominative-accusative plural neuter § 21.36. Instrumental plural masculine and feminine § 21.37 	$\begin{array}{c} 114\\ 114\\ 114\\ 114\\ 114\\ 114\\ 114\\ 114$
 21. Generic pronouns Stems ta (sa), eta (eşa), ya, ka and prominal adjectives § 21.1 Stem in composition § 21.2 Nominative singular masculine § 21.5 Nominative singular feminine § 21.8 Accusative singular masculine and feminine § 21.9 Nominative-accusative singular neuter § 21.10. Oblique singular feminine § 21.3 Instrumental singular masculine and neuter § 21.15. Genitive singular masculine and neuter § 21.16 Genitive singular masculine and neuter § 21.16 Genitive singular feminine § 21.17 Genitive singular masculine and neuter § 21.20 Locative singular masculine and neuter § 21.20 Locative singular feminine § 21.23 Nominative-accusative dual neuter § 21.24 Nominative plural feminine § 21.35 Nominative plural feminine § 21.35 Nominative-accusative plural mexculine § 21.35 Nominative-accusative plural neuter § 21.36 Instrumental plural masculine and feminine § 21.37 Genitive plural § 21.38 	$\begin{array}{c} 114\\ 114\\ 114\\ 114\\ 114\\ 114\\ 114\\ 114$
 21. Generic pronouns Stems la (sa), ela (eşa), ya, ka and prominal adjectives § 21.1 Stem in composition § 21.2 Nominative singular masculine § 21.5 Nominative singular feminine § 21.8 Accusative singular masculine and feminine § 21.9 Nominative-accusative singular neuter § 21.10. Oblique singular feminine § 21.3 Instrumental singular masculine and neuter § 21.15 Genitive singular masculine and neuter § 21.16 Genitive singular masculine and neuter § 21.16 Genitive singular feminine § 21.17 Genitive singular masculine and neuter § 21.20 Locative singular masculine and neuter § 21.20 Locative singular feminine § 21.23 Nominative plural masculine § 21.25 Nominative plural masculine § 21.30 Accusative plural masculine § 21.35 Nominative-accusative plural meuter § 21.36 Instrumental plural masculine and feminine § 21.37 Genitive plural § 21.38 Paradigms § 21.46 	$\begin{array}{c} 114\\ 114\\ 114\\ 114\\ 114\\ 114\\ 114\\ 114$
 21. Generic pronouns Stems la (sa), ela (eşa), ya, ka and prominal adjectives § 21.1 Stem in composition § 21.2 Nominative singular masculine § 21.5 Nominative singular feminine § 21.8 Accusative singular masculine and feminine § 21.9 Nominative-accusative singular neuter § 21.10. Oblique singular feminine § 21.3 Instrumental singular masculine and neuter § 21.15 Genitive singular masculine and neuter § 21.16 Genitive singular masculine and neuter § 21.16 Genitive singular feminine § 21.17 Genitive singular masculine and neuter § 21.20 Locative singular masculine and neuter § 21.20 Locative singular feminine § 21.23 Nominative plural masculine § 21.25 Nominative plural masculine § 21.30 Accusative plural masculine § 21.35 Nominative-accusative plural meuter § 21.36 Instrumental plural masculine and feminine § 21.37 Genitive plural § 21.38 Paradigms § 21.46 	$\begin{array}{c} 114 \\ 114 \\ 114 \\ 114 \\ 114 \\ 114 \\ 114 \\ 115 \\ 115 \\ 115 \\ 115 \\ 115 \\ 115 \\ 115 \\ 116 \\ 116 \\ 116 \\ 116 \\ 116 \\ 117 \\ \end{array}$
 21. Generic pronouns Stems ta (sa), eta (eşa), ya, ka and prominal adjectives § 21.1 Stem in composition § 21.2 Nominative singular masculine § 21.5 Nominative singular feminine § 21.8 Accusative singular masculine and feminine § 21.9 Nominative-accusative singular neuter § 21.10. Oblique singular feminine § 21.3 Instrumental singular masculine and neuter § 21.15. Genitive singular masculine and neuter § 21.16 Genitive singular masculine and neuter § 21.16 Genitive singular feminine § 21.17 Genitive singular masculine and neuter § 21.20 Locative singular masculine and neuter § 21.20 Locative singular feminine § 21.23 Nominative-accusative dual neuter § 21.24 Nominative plural feminine § 21.35 Nominative plural feminine § 21.35 Nominative-accusative plural mexculine § 21.35 Nominative-accusative plural neuter § 21.36 Instrumental plural masculine and feminine § 21.37 Genitive plural § 21.38 	$\begin{array}{c} 114\\ 114\\ 114\\ 114\\ 114\\ 114\\ 114\\ 114$

	Nominative singular masculine imo (ima?, ime?)	
	§ 21.51	
	Nominative singular feminine ima § 21.52	117
	Accusative singular masculine imu, imām § 21.53	117
	Accusative singular feminine imam, ima, imu	
		117
	§ 21.54	117
	Nominative-accusative singular neuter imam, imu,	
	imu, im' § 21.55	117
	Instrumental singular masculine-neuter imena	
	£ 91 56	
	§ 21.56	117
	Instrumental singular masculine-neuter iminā	
	§ 21.57	117
	[Instrumental singular masculinc-neuter emina]	
	§ 21.58	117
	Instrumental singular feminine imäye and imäya	
	§ 21.59	117
	Dative singular masculine-neuter imāya § 21.60	117
	Ablative singular masculine-neuter imato § 21.61	117
		117
	Genitive singular feminine imasyā(s) § 21.63	117
	Genitive singular feminine imäye § 21.64	117
	Locative singular masculine-neuter imasmim,	
	imarmi f 91 65	
	imasmi § 21.65	117
	Locative singular masculine-neuter imamhi § 21.66	117
	Locative singular masculine-neuter imahim § 21.67	117
	Locative singular feminine imasyām § 21.68	
	Nominative plural masculine imi § 21.69	
	Accusative plural masculine ime (imi, ima) § 21.70	117
	Nominative-accusative plural neuter ime § 21.71	117
	Instrumental plural masculine-neuter imais § 21.72	117
	Instrumental plural masculine-neuter inebhih	
	§ 21.73	117
		117
	Instrumental plural masculine-neuter imehir § 21,75	117
	Instrumental plural feminine imabhih § 21.76	118
	Genitive plural masculine-neuter imeșăm § 21.77	118
	Locative plural masculine-neuter imesu § 21.78.	118
	Locative plural masculine-neuter imesu § 21.78.	
	Locative plural masculine-neuter imeşu § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms	118
	Locative plural masculine-neuter <i>imeşu</i> § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79.	118 118
	Locative plural masculine-neuter <i>imeşu</i> § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79 ayam, etc.: miscellaneous non-Sanskrit forms § 21.80	118 118 118
	Locative plural masculine-neuter <i>imeşu</i> § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79. ayam, etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for ayam, <i>idam</i> § 21.85	118 118 118 118 118
	Locative plural masculine-neuter <i>imeşu</i> § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79 ayam, etc.: miscellaneous non-Sanskrit forms § 21.80	118 118 118
	Locative plural masculine-neuter <i>imeşu</i> § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79. ayam, etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for ayam, idam § 21.85 asau, adas § 21.86	118 118 118 118 118 119
	Locative plural masculine-neuter <i>imeşu</i> § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79. ayam, etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for ayam, idam § 21.85 asau, adas § 21.86 22. Noun suffixes	118 118 118 118 119 120
	Locative plural masculine-neuter <i>imeşu</i> § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79. ayam, etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for ayam, <i>idam</i> § 21.85 asau, adas § 21.86 22. Noun suffixes. Suffix aka § 22.2.	118 118 118 118 119 120 120
	Locative plural masculine-neuter <i>imeşu</i> § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79. ayam, etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for ayam, <i>idam</i> § 21.85	118 118 118 118 119 120
	Locative plural masculine-neuter <i>imeşu</i> § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79. ayam, etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for ayam, <i>idam</i> § 21.85	118 118 118 118 119 120 120
	Locative plural masculine-neuter <i>imeşu</i> § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79. ayam, etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for ayam, idam § 21.85 asau, adas § 21.86 22. Noun suffixes Suffix aka § 22.2 Suffix ana, and § 22.7 Suffix áni (ayanī, āyanī, āyanikā, (ā)yinī) § 22.10	118 118 118 118 119 120 120 120 120
	Locative plural masculine-neuter <i>imeşu</i> § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79. ayam, etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for ayam, idam § 21.85	118 118 118 119 120 120 120 120
	Locative plural masculine-neuter <i>imeşu</i> § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79. ayam, etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for ayam, idam § 21.85 asau, adas § 21.86 22. Noun suffixes Suffix aka § 22.2. Suffix ana, anā § 22.7. Suffix ana, anā § 22.7. Suffix āna § 22.11 Suffix āra § 22.11	118 118 118 119 120 120 120 120 120
-	Locative plural masculine-neuter <i>imeşu</i> § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79	118 118 118 119 120 120 120 120 120 120
	Locative plural masculine-neuter <i>imeşu</i> § 21.78. <i>ayam</i> , etc.: phonetic alterations of Sanskrit forms § 21.79. <i>ayam</i> , etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for <i>ayam</i> , <i>idam</i> § 21.85	118 118 118 119 120 120 120 120 120 120 120 121
	Locative plural masculine-neuter <i>imeşu</i> § 21.78. <i>ayam</i> , etc.: phonetic alterations of Sanskrit forms § 21.79. <i>ayam</i> , etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for <i>ayam</i> , <i>idam</i> § 21.85	118 118 118 119 120 120 120 120 120 120
-	Locative plural masculine-neuter <i>imeşu</i> § 21.78. <i>ayam</i> , etc.: phonetic alterations of Sanskrit forms § 21.79. <i>ayam</i> , etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for <i>ayam</i> , <i>idam</i> § 21.85	118 118 118 119 120 120 120 120 120 120 120 121 121
-	Locative plural masculine-neuter <i>imeşu</i> § 21.78. <i>ayam</i> , etc.: phonetic alterations of Sanskrit forms § 21.79. <i>ayam</i> , etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for <i>ayam</i> , <i>idam</i> § 21.85	118 118 118 119 120 120 120 120 120 120 120 120 121 121
	Locative plural masculine-neuter <i>imeşu</i> § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79	118 118 118 118 119 120 120 120 120 120 120 120 120 121 121
	Locative plural masculine-neuter <i>imeşu</i> § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79	118 118 118 118 119 120 120 120 120 120 120 120 120 121 121
	Locative plural masculine-neuter <i>imeşu</i> § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79	118 118 118 118 119 120 120 120 120 120 120 120 120 121 121
	Locative plural masculine-neuter <i>imeşu</i> § 21.78. <i>ayam</i> , etc.: phonetic alterations of Sanskrit forms § 21.79. <i>ayam</i> , etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for <i>ayam</i> , <i>idam</i> § 21.85	1118 1118 1118 1119 1200 1200 1200 1200 1200 1200 1200
-	Locative plural masculine-neuter $imeşu \S 21.78$. ayam, etc.: phonetic alterations of Sanskrit forms $\S 21.79$	1118 1118 1118 1119 1200 1200 1200 1200 1200 1200 1200
	Locative plural masculine-neuter $imeşu \S 21.78$. $ayam, etc.: phonetic alterations of Sanskrit forms \S 21.79$	1118 1118 1118 1119 1200 1200 1200 1200 1200 1200 1200
	Locative plural masculine-neuter $imeşu \S 21.78$. ayam, etc.: phonetic alterations of Sanskrit forms $\S 21.79$. ayam, etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for $ayam$, $idam \S 21.85$	1118 1118 1118 1119 1200 1200 1200 1200 1200 1200 1200
	Locative plural masculine-neuter imegu § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79. ayam, etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for ayam, idam § 21.85	118 118 118 118 119 120 120 120 120 120 120 120 120 120 121 121
	Locative plural masculine-neuter imegu § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79. ayam, etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for ayam, idam § 21.85	1118 1118 1118 1119 1200 1200 1200 1200 1200 1200 1200
	Locative plural masculine-neuter imegu § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79. ayam, etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for ayam, idam § 21.85	118 118 118 118 119 120 120 120 120 120 120 120 120 120 121 121
	Locative plural masculine-neuter imegu § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79. ayam, etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for ayam, idam § 21.85. asau, adas § 21.86. 22. Noun suffixes Suffix aka § 22.2. Suffix ana, anā § 22.7. Suffix dni (ayanī, āyani, āyanikā, (ā)yinī) § 22.10. Suffix dra § 22.11. Suffix itar § 22.12. Suffix itar § 22.12. Suffix itar § 22.13. Suffix itar § 22.13. Suffix ita § 22.18. Suffix ita § 22.18. Suffix ita § 22.19. Suffix ita § 22.19. Suffix ita § 22.20. Suffix kā § 22.21. ka suārthe § 22.23. Diminutive kā § 22.32. Specifying kā § 22.38. Suffix itā § 22.40. Suffix itā § 22.41. Suffix itā § 22.41. Suffix itā § 22.42.	1118 1118 1118 1119 1200 1200 1200 1200 1200 1200 1210 1201 1211 1211 1211 1211 1211 1212 1223 1233 123
	Locative plural masculine-neuter <i>imeşu</i> § 21.78. <i>ayam</i> , etc.: phonetic alterations of Sanskrit forms § 21.79. <i>ayam</i> , etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for <i>ayam</i> , <i>idam</i> § 21.85. <i>asau</i> , <i>adas</i> § 21.86. 22. Noun suffixes Suffix <i>aka</i> § 22.2. Suffix <i>ana</i> , <i>and</i> § 22.7. Suffix <i>ani</i> (<i>ayani</i> , <i>āyani</i> , <i>āyanikā</i> , (<i>ā)yini</i>) § 22.10. Suffix <i>āra</i> § 22.11 Suffix <i>itar</i> § 22.12. Suffix <i>itar</i> § 22.13. Suffix <i>ita</i> § 22.17. Suffix <i>ita</i> § 22.17. Suffix <i>ita</i> § 22.18. Suffix <i>ita</i> § 22.19. Suffix <i>ita</i> § 22.20. Suffix <i>ita</i> § 22.21. Suffix <i>ita</i> § 22.41. Suffix <i>ita</i> § 22.40. Suffix <i>ita</i> § 22.41. Suffix <i>ita</i> § 22.45. Suffix <i>maya</i> § 22.46.	118 118 118 118 119 120 120 120 120 120 120 120 120 120 121 121
	Locative plural masculine-neuter imegu § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79. ayam, etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for ayam, idam § 21.85. asau, adas § 21.86. 22. Noun suffixes Suffix aka § 22.2. Suffix aka § 22.2. Suffix ana, and § 22.7. Suffix ana, and § 22.7. Suffix dra § 22.11. Suffix idar § 22.12. Suffix idar § 22.13. Suffix ida § 22.17. Suffix ika § 22.13. Suffix ika § 22.18. Suffix ika § 22.19. Suffix ika § 22.19. Suffix ika § 22.21. Suffix ika § 22.21. Suffix ika § 22.21. Suffix ika § 22.21. Suffix ika § 22.23. Suffix ika § 22.40. Suffix ia § 22.41. Suffix ia, or alha § 22.44. Suffix niká § 22.45. Suffix niká § 22.47. Suffix ia (alaka) § 22.47.	$\begin{array}{c} 118 \\ 118 \\ 118 \\ 118 \\ 119 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 121 \\ 121 \\ 121 \\ 121 \\ 121 \\ 121 \\ 122 \\ 123 \\ 123 \\ 123 \\ 123 \end{array}$
	Locative plural masculine-neuter imegu § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79. ayam, etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for ayam, idam § 21.85	$\begin{array}{c} 118 \\ 118 \\ 118 \\ 118 \\ 119 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 121 \\ 121 \\ 121 \\ 121 \\ 121 \\ 121 \\ 122 \\ 123 \\ 123 \\ 123 \\ 123 \\ 123 \\ 123 \\ 123 \end{array}$
	Locative plural masculine-neuter imegu § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79. ayam, etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for ayam, idam § 21.85. asau, adas § 21.86. 22. Noun suffixes Suffix aka § 22.2. Suffix aka § 22.2. Suffix ana, and § 22.7. Suffix ana, and § 22.7. Suffix dra § 22.11. Suffix idar § 22.12. Suffix idar § 22.13. Suffix ida § 22.17. Suffix ika § 22.13. Suffix ika § 22.18. Suffix ika § 22.19. Suffix ika § 22.19. Suffix ika § 22.21. Suffix ika § 22.21. Suffix ika § 22.21. Suffix ika § 22.21. Suffix ika § 22.23. Suffix ika § 22.40. Suffix ia § 22.41. Suffix ia, or alha § 22.44. Suffix niká § 22.45. Suffix niká § 22.47. Suffix ia (alaka) § 22.47.	$\begin{array}{c} 118 \\ 118 \\ 118 \\ 118 \\ 119 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 121 \\ 121 \\ 121 \\ 121 \\ 121 \\ 121 \\ 122 \\ 123 \\ 123 \\ 123 \\ 123 \end{array}$
	Locative plural masculine-neuter imegu § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79. ayam, etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for ayam, idam § 21.85. asau, adas § 21.86. 22. Noun suffixes Suffix aka § 22.2. Suffix aka § 22.2. Suffix ana, and § 22.7. Suffix dni (ayani, dyani, dyanikd, (d)yini) § 22.10. Suffix dra § 22.11. Suffix ila § 22.12. Suffix ila § 22.13. Suffix ila § 22.13. Suffix ika § 22.14. Suffix ika § 22.18. Suffix ika § 22.19. Suffix ika § 22.19. Suffix ika § 22.20. Suffix ika § 22.21. ka svärthe § 22.23. Diminutive ka § 22.32. Specifying ka § 22.34. Suffix ia fa § 22.40. Suffix ia, or alha § 22.44. Suffix nikä § 22.45. Suffix nikä § 22.45. Suffix la (alka) § 22.47. Suffixe svant and mant § 22.48. Suffix (d)vin § 22.51.	$\begin{array}{c} 118\\ 118\\ 118\\ 118\\ 119\\ 120\\ 120\\ 120\\ 120\\ 120\\ 120\\ 120\\ 120$
	Locative plural masculine-neuter imegu § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79. ayam, etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for ayam, idam § 21.85. asau, adas § 21.86. 22. Noun suffixes Suffix aka § 22.2. Suffix aka § 22.2. Suffix ana anā § 22.7. Suffix dīa (ayanī, āyanī, āyanikā, (ā)yinī) § 22.10. Suffix dīa § 22.11. Suffix itar § 22.12. Suffix itar § 22.13. Suffix itar § 22.13. Suffix ika § 22.18. Suffix ika § 22.18. Suffix ika § 22.19. Suffix ika § 22.19. Suffix ika § 22.20. Suffix kā § 22.21. ka soārlhe § 22.23. Diminutive ka § 22.32. Specifying ka § 22.38. Suffix tā, or atha § 22.40. Suffix tā, or atha § 22.44. Suffix nikā § 22.45. Suffix nikā § 22.45. Suffix nikā § 22.45. Suffix nikā § 22.45. Suffix nikā § 22.46. Suffix la (laka) § 22.47. Suffix la (laka) § 22.48. Suffix (ā) valama § 22.48. Suffix (a) valama § 22.48. Suffix (a	$\begin{array}{c} 118 \\ 118 \\ 118 \\ 119 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 121 \\ 121 \\ 121 \\ 121 \\ 121 \\ 122 \\ 123 \\ 123 \\ 123 \\ 123 \\ 123 \\ 123 \\ 123 \\ 123 \\ 123 \\ 123 \\ 123 \\ 123 \\ 124 \\ 125 \\ \end{array}$
	Locative plural masculine-neuter imegu § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79. ayam, etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for ayam, idam § 21.85. asau, adas § 21.86. 22. Noun suffixes Suffix aka § 22.2. Suffix aka § 22.2. Suffix ani (ayani, āyani, āyanikā, (ā)yini) § 22.10. Suffix dīa § 22.11. Suffix itar § 22.12. Suffix itar § 22.12. Suffix itar § 22.13. Suffix ita § 22.13. Suffix ita § 22.18. Suffix ika § 22.18. Suffix ika § 22.18. Suffix ika § 22.19. Suffix ika § 22.19. Suffix ika § 22.20. Suffix kā § 22.21. ka soārthe § 22.23. Diminutive ka § 22.32. Specifying ka § 22.38. Suffix ita § 22.40. Suffix ita § 22.40. Suffix ita § 22.40. Suffix ita § 22.41. Suffix ita, or atha § 22.44. Suffix nikā § 22.45. Suffix nikā § 22.45. Suffix (a) or atha § 22.47. Suffix (a) or atha § 22.48. Suffix (a) or atha § 22.48. Suffix (a) or atha § 22.48. Suffix (a) or atha § 22.47. Suffix (a) or atha § 22.47. Suffix (a) or and mant § 22.48. Suffix (a) or and verb compounds. Dual dvandvas § 23.1.	$\begin{array}{c} 118\\ 118\\ 118\\ 118\\ 119\\ 120\\ 120\\ 120\\ 120\\ 120\\ 120\\ 120\\ 120$
	Locative plural masculine-neuter imegu § 21.78. ayam, etc.: phonetic alterations of Sanskrit forms § 21.79. ayam, etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for ayam, idam § 21.85. asau, adas § 21.86. 22. Noun suffixes Suffix aka § 22.2. Suffix aka § 22.2. Suffix ani (ayani, āyani, āyanikā, (ā)yini) § 22.10. Suffix dīa § 22.11. Suffix itar § 22.12. Suffix itar § 22.12. Suffix itar § 22.13. Suffix ita § 22.13. Suffix ita § 22.18. Suffix ika § 22.18. Suffix ika § 22.18. Suffix ika § 22.19. Suffix ika § 22.19. Suffix ika § 22.20. Suffix kā § 22.21. ka soārthe § 22.23. Diminutive ka § 22.32. Specifying ka § 22.38. Suffix ita § 22.40. Suffix ita § 22.40. Suffix ita § 22.40. Suffix ita § 22.41. Suffix ita, or atha § 22.44. Suffix nikā § 22.45. Suffix nikā § 22.45. Suffix (a) or atha § 22.47. Suffix (a) or atha § 22.48. Suffix (a) or atha § 22.48. Suffix (a) or atha § 22.48. Suffix (a) or atha § 22.47. Suffix (a) or atha § 22.47. Suffix (a) or and mant § 22.48. Suffix (a) or and verb compounds. Dual dvandvas § 23.1.	$\begin{array}{c} 118 \\ 118 \\ 118 \\ 119 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 120 \\ 121 \\ 121 \\ 121 \\ 121 \\ 121 \\ 122 \\ 123 \\ 123 \\ 123 \\ 123 \\ 123 \\ 123 \\ 123 \\ 123 \\ 123 \\ 123 \\ 123 \\ 123 \\ 124 \\ 125 \\ \end{array}$
	Locative plural masculine-neuter <i>imeşu</i> § 21.78. <i>ayam</i> , etc.: phonetic alterations of Sanskrit forms § 21.79. <i>ayam</i> , etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for <i>ayam</i> , <i>idam</i> § 21.85. <i>asau</i> , <i>adas</i> § 21.86. 22. Noun suffixes Suffix <i>aka</i> § 22.2. Suffix <i>ana</i> , <i>and</i> § 22.7. Suffix <i>ana</i> , <i>and</i> § 22.7. Suffix <i>ani</i> (<i>ayani</i> , <i>āyani</i> , <i>āyanikā</i> , (<i>āyini</i>) § 22.10. Suffix <i>iar</i> § 22.11 Suffix <i>iaf</i> § 22.12. Suffix <i>ila</i> § 22.13. Suffix <i>ila</i> § 22.17. Suffix <i>ika</i> § 22.18. Suffix <i>ika</i> § 22.19. Suffix <i>ika</i> § 22.19. Suffix <i>ika</i> § 22.20. Suffix <i>iva</i> (<i>aniya</i>) § 22.20. Suffix <i>iva</i> § 22.21. <i>ka svārthe</i> § 22.23. Diminutive <i>ka</i> § 22.32. Specifying <i>ka</i> § 22.38. Suffix <i>tara</i> § 22.40. Suffix <i>tara</i> § 22.40. Suffix <i>ina</i> § 22.41. Suffix <i>ina</i> § 22.42. Suffix <i>ina</i> § 22.42. Suffix <i>iaa</i> § 22.43. Suffix <i>iaa</i> § 22.44. Suffix <i>iaa</i> § 22.45. Suffix <i>iaa</i> (<i>alka</i>) § 22.47. Suffix <i>iaa</i> (<i>alka</i>) § 22.47. Suffix <i>iaa</i> (<i>alka</i>) § 22.51. 23. Noun and verb compounds. Dual dvandvas § 23.1. Inflected forms as prior members § 23.5.	$\begin{array}{c} 118\\ 118\\ 118\\ 118\\ 119\\ 120\\ 120\\ 120\\ 120\\ 120\\ 120\\ 120\\ 121\\ 121$
	Locative plural masculine-neuter <i>imeşu</i> § 21.78. <i>ayam</i> , etc.: phonetic alterations of Sanskrit forms § 21.79. <i>ayam</i> , etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for <i>ayam</i> , <i>idam</i> § 21.85. <i>asau</i> , <i>adas</i> § 21.86. 22. Noun suffixes Suffix <i>aka</i> § 22.2. Suffix <i>aka</i> § 22.2. Suffix <i>ana</i> , <i>anā</i> § 22.7. Suffix <i>ana</i> , <i>anā</i> § 22.7. Suffix <i>iar</i> § 22.11. Suffix <i>iar</i> § 22.12. Suffix <i>iar</i> § 22.13. Suffix <i>ila</i> § 22.17. Suffix <i>ika</i> § 22.18. Suffix <i>ika</i> § 22.18. Suffix <i>ika</i> § 22.19. Suffix <i>ika</i> § 22.20. Suffix <i>iag</i> § 22.20. Suffix <i>iag</i> § 22.32. Suffix <i>iag</i> § 22.40. Suffix <i>iag</i> § 22.40. Suffix <i>iag</i> § 22.41. Suffix <i>iag</i> § 22.41. Suffix <i>iag</i> § 22.42. Suffix <i>iag</i> § 22.41. Suffix <i>iag</i> § 22.42. Suffix <i>iag</i> § 22.43. Suffix <i>iag</i> § 22.45. Suffix <i>iag</i> § 22.46. Suffix <i>iag</i> § 22.47. Suffix <i>iag</i> § 22.47. Suffix <i>iag</i> § 22.48. Suffix <i>iag</i> § 22.51. 23. Noun and <i>verb</i> compounds. Dual dvandvas § 23.1. Inflected forms as prior members § 23.5. Irregular order of parts of compounds § 23.10	$\begin{array}{c} 118\\ 118\\ 118\\ 118\\ 119\\ 120\\ 120\\ 120\\ 120\\ 120\\ 120\\ 120\\ 120$
	Locative plural masculine-neuter <i>imeşu</i> § 21.78. <i>ayam</i> , etc.: phonetic alterations of Sanskrit forms § 21.79. <i>ayam</i> , etc.: miscellaneous non-Sanskrit forms § 21.80 Paradigm for <i>ayam</i> , <i>idam</i> § 21.85. <i>asau</i> , <i>adas</i> § 21.86. 22. Noun suffixes Suffix <i>aka</i> § 22.2. Suffix <i>ana</i> , <i>and</i> § 22.7. Suffix <i>ana</i> , <i>and</i> § 22.7. Suffix <i>ani</i> (<i>ayani</i> , <i>āyani</i> , <i>āyanikā</i> , (<i>āyini</i>) § 22.10. Suffix <i>iar</i> § 22.11 Suffix <i>iaf</i> § 22.12. Suffix <i>ila</i> § 22.13. Suffix <i>ila</i> § 22.17. Suffix <i>ika</i> § 22.18. Suffix <i>ika</i> § 22.19. Suffix <i>ika</i> § 22.19. Suffix <i>ika</i> § 22.20. Suffix <i>iva</i> (<i>aniya</i>) § 22.20. Suffix <i>iva</i> § 22.21. <i>ka svārthe</i> § 22.23. Diminutive <i>ka</i> § 22.32. Specifying <i>ka</i> § 22.38. Suffix <i>tara</i> § 22.40. Suffix <i>tara</i> § 22.40. Suffix <i>ina</i> § 22.41. Suffix <i>ina</i> § 22.42. Suffix <i>ina</i> § 22.42. Suffix <i>iaa</i> § 22.43. Suffix <i>iaa</i> § 22.44. Suffix <i>iaa</i> § 22.45. Suffix <i>iaa</i> (<i>alka</i>) § 22.47. Suffix <i>iaa</i> (<i>alka</i>) § 22.47. Suffix <i>iaa</i> (<i>alka</i>) § 22.51. 23. Noun and verb compounds. Dual dvandvas § 23.1. Inflected forms as prior members § 23.5.	$\begin{array}{c} 118\\ 118\\ 118\\ 118\\ 119\\ 120\\ 120\\ 120\\ 120\\ 120\\ 120\\ 120\\ 120$

Compounds of cype phataphant &	126 126
Thematic presents § 24.4 Stems in long vowels § 24.8	127 127 127
Stems in -ā- § 24.11 Stems in -ī- § 24.12	128 128 128 128
Stems in -o- § 24.14 Non-thematic stems § 24.15	128 128
The dual § 25.1 Confusion of person and number § 25.4 Third singular forms § 25.8	129 129 129 129 129
Present indicative with second singular subject	129 129
Present indicative with third plural subject § 25.10 Present optative with first singular subject § 25.11 Present optative with second singular subject	129 129
Imperative with third plural subject § 25.14	129 129 129 129
Future with first singular subject § 25.16 Future with second singular subject § 25.17 Future with third plural subject § 25.18	129 129 129 130
Preterite with first singular subject § 25.19 Preterite with second singular subject § 25.20	130 130 130
Preterite with third plural subject § 25.22 Perfect with second singular subject § 25.23 Perfect with first plural subject § 25.24	130 130 130
Other than third singular forms § 25.26 First person forms § 25.27 Second person forms § 25.28	130 130 130 130 130
26. Personal endings i for final i § 26.2	131 131
Secondary first singular n for m (m) § 26.4 First plural ma and mas § 26.5	131 131 131 131
First plural matha (mathā) § 26.8 Second plural ta varying with tha § 26.11	131 132 132
Second plural imperative thâ in vss, m.c. § 26.14 Second plural optative in tha § 26.15 Second plural thâh(?), thâ § 26.16	132 132 132
	133 133
27. Non-Sanskritic treatment of thematic vowel <i>ā</i> for <i>a</i> as thematic vowel § 27.1 <i>āsi, āti</i> , second and third singular apparently or sup-	134
posedly with modal force § 27.4 a for \tilde{a} as thematic vowel § 27.9 <i>i</i> instead of thematic vowel <i>a</i> , \tilde{a} § 27.10	134 134 134
28. Summary classification of non-Sanskritic presents	136
§ 28.3 Third singular ati to first singular āmi, and to third plu- ral ati § 28.8	136 136
	136 136

Third singular ali from original (Sanskrit) fifth and	
eighth class presents § 28.15	137
Presents based on past passive participles § 28.19	137
New presents in $ya \S 28.20$ ya after long vowels § 28.20	137 137
ya after consonants § 28.26	138
a for ya presents § 28.29	138
Loss of infixed nasal in thematic presents § 28.31	
Influction of nasal in thematic presents § 28.32 Thematic presents with \bar{a} for a in root syllable § 28.33	138
Long i , \tilde{u} for i , u in root syllable of thematic presents	100
§ 28.34	138
Other new presents of the sixth class type § 28.37.	139
Other new presents of the first class type § 28.38	139
Presents in <i>eti</i> § 28.46 Presents in <i>āti</i> § 28.51	139 139
Presents in other long vowels § 28.56	140
Non-thematic presents § 28.57	140
Strong for weak stem forms, second class § 28.60	
Third class § 28.61 Fifth and eighth classes § 28.62	140 140
Ninth class § 28.63	140
Weak for strong stem forms § 28.64	140
29. Optative	141
Suppression of -ay- in stem, and the like § 29.4	141
Non-thematic middle endings in aya stems § 29.6	141
Heirs of thematic optative § 29.7	141
First singular $e $ 29.8 First singular $i $ 29.9	141 141
Second singular $e \S 29.10$	141
Second singular <i>i</i> § 29.11	142
Third singular $e \S 29.12$	142
Third singular $i \S 29.14$ Third plural $e \S 29.15$	142 142
First plural ema § 29.16	142
First plural e § 29.17	142
Third plural $elsu(h)$, $ensu(h) $ § 29.18	142
Second singular esi § 29.20	142
So-called 'subjunctives' in āsi, āti § 29.22 eya, eyā and cognate endings § 29.23	142 142
eya, eyā, third singular § 29.28	143
eya, eyā, other than third singular § 29.29	143
eyya, eyyä § 29.30	143
eyam as third singular § 29.31eyo, third singular § 29.32	143 144
eyāma(s), eyātha, first and second plural § 29.33.	144
iyā, iya, iyāma(s) § 29.34	144
aya(?), first singular § 29.35	144
Second singular (e)y(y)āsi § 29.36 Non-thematic optatives § 29.39	144 144
Forms of root as 'to be' § 29.39	144
Other athematic optatives in yā, ya § 29.42	144
Miscellaneous athematic optatives § 29.43	145
30. Imperative	146
Spread of second singular ending hi § 30.2	146
ehi or m.c. ehi (with e for Skt. aya) § 30.3 ihi for ehi(?) § 30.4	146 146
<i>ehi</i> (with <i>e</i> for Skt. <i>a</i>) § 30.5	146
āhi or (m.c.) āhī § 30.6	146
ahi or (m.c.) ahi § 30.7	146
[ahe, for ahi, ahi?] § 30.8	147 147
ihi? § 30.10	147
Other non-Sanskrit endings § 30.11	147
Second singular ā § 30.11	147
First plural matha and second plural tha (thā) § 30.12	147
\$ 50.12	147
Third singular to § 30.14	147
Second singular su § 30.15	147

	147
First plural mahe § 30.17	
Second plural dhva § 30.18	147
Second plural dhve § 30.19	147
Third plural ntān § 30.20	147
24 Destaura	148
31. Future isya with thematic presents § 31.1	148
isya with thematic presents g 51.1	149
Future sign affixed to stems in long vowels § 31.3	
Futures containing h for sy (sy) § 31.19	150
First singular forms in eham(?) § 31.21	150
Futures in sya (sya) after consonants, more or less	
un-Sanskritic § 31.23	151
Futures with s for sy § 31.26	151
First singular futures in am (a) § 31.30	151
Second singular futures in sye, isye (isyi) § 31.35	152
Periphrastic future § 31.36	152
Optative and imperative of the future § 31.37	152
Conditional § 31.38	152
32. Aorist, or preterite (including imperfect)	153
Augment § 32.2	153
Confusion of person and number § 32.13	154
Conspectus of preterites § 32.14	154
Descendants of the Sanskrit is-aorist § 32.15	154
The ending \underline{i} \$32.16	154
The ending $i $ § 32.17	154
	155
Special forms of the endings i and $i \S 32.18$	156
Distinctive third plural forms § 32.29	
Third plural isu and congeners § 32.30	156
işu § 32.31	156
isu § 32.32	156
işū § 32.33	156
isu § 32.34	156
işu, işû § 32.35	156
işuh, işus § 32.36	156
su added to quasi stem-final i, or to third	
singular <i>il</i> § 32.37	156
Third, plural imsu and congeners § 32.38	156
(iṃsu, insu, iṃṣū, iṃsū, ilsu[ḥ])	
Other non-Sanskritic personal endings of is-aorist	
type § 32.39	
cype 3 02.00	157
First singular im \$ 32.39	157 157
First singular $im \S 32.39$ First plural $imba \S 32.40$	157
First plural <i>imha</i> § 32.40	157 157
First plural imha § 32.40 ilha, ilhā, illha, ila (mostly 3d singular) § 32.41	157 157 157
First plural <i>imha</i> § 32.40 ilha, ilhā, illha, ila (mostly 3d singular) § 32.41 Third plural or singular <i>ire</i> § 32.43	157 157
First plural <i>imha</i> § 32.40 <i>ilha, ilhā, illha, ila</i> (mostly 3d singular) § 32.41 Third plural or singular <i>ire</i> § 32.43 Other MIndic forms with Skt. <i>is</i> -aorist endings	157 157 157 157
First plural imha § 32.40 ilha, ilhā, illha, ila (mostly 3d singular) § 32.41 Third plural or singular ire § 32.43 Other MIndic forms with Skt. is-aorist endings § 32.44	157 157 157
First plural imha § 32.40 itha, ithā, itlha, ita (mostly 3d singular) § 32.41 Third plural or singular ire § 32.43 Other MIndic forms with Skt. is-aorist endings § 32.44 Descendants of Sanskrit s and sis aorists and similar	157 157 157 157 157
First plural imha § 32.40 itha, ithā, ithā, ita (mostly 3d singular) § 32.41 Third plural or singular ire § 32.43 Other MIndic forms with Skt. is-aorist endings § 32.44 Descendants of Sanskrit s and sis aorists and similar forms § 32.47	157 157 157 157 157 157
First plural imha § 32.40 itha, ithā, itlha, ita (mostly 3d singular) § 32.41 Third plural or singular ire § 32.43 Other MIndic forms with Skt. is-aorist endings § 32.44 Descendants of Sanskrit s and sis aorists and similar forms § 32.47 Type āsi (āsil) § 32.53	157 157 157 157 157 157 157
First plural imha § 32.40 itha, ithā, ittha, ita (mostly 3d singular) § 32.41 Third plural or singular ire § 32.43 Other MIndic forms with Skt. is-aorist endings § 32.44 Descendants of Sanskrit s and sis aorists and similar forms § 32.47 Type āsi (āsil) § 32.53 Type esi (esi, eşil) § 32.63	157 157 157 157 157 157 157 158 159
First plural imha § 32.40 ilha, ilhā, ilha, ila (mostly 3d singular) § 32.41 Third plural or singular ire § 32.43 Other MIndic forms with Skt. is-aorist endings § 32.44 Descendants of Sanskrit s and sis aorists and similar forms § 32.47 Type āsi (āsil) § 32.53 Type esi (esi, eşīl) § 32.63 Other si (sil, etc.) after long vowels § 32.69	157 157 157 157 157 157 157 157 158 159 159
 First plural imha § 32.40	157 157 157 157 157 157 157 158 159
 First plural imha § 32.40	157 157 157 157 157 157 157 157 158 159 159
 First plural imha § 32.40	157 157 157 157 157 157 157 157 158 159 159
First plural imha § 32.40 itha, ithā, ittha, ita (mostly $3d singular$) § 32.41 Third plural or singular ire § 32.43 Other MIndic forms with Skt. is-aorist endings § 32.44 Descendants of Sanskrit s and sis aorists and similar forms § 32.47 Type āsi (āsil) § 32.53 Type esi (esi, eşil) § 32.63 Other si (sit, etc.) after long vowels § 32.69 s-aorists from roots in $r(r)$ § 32.74 Third plural āsu(h), ānsu(h), ansu(h), atsu etc. § 32.75 Miscellaneous s-aorists § 32.78	157 157 157 157 157 157 157 158 159 159 160
 First plural imha § 32.40	157 157 157 157 157 157 157 157 158 159 159 160 160 160
 First plural imha § 32.40	157 157 157 157 157 157 157 157 158 159 159 160 160 160
 First plural imha § 32.40	157 157 157 157 157 157 157 157 158 159 159 160 160 160 160
First plural imha § 32.40 itha, ithā, ittha, ita (mostly 3d singular) § 32.41 Third plural or singular ire § 32.43 Other MIndic forms with Skt. is-aorist endings § 32.44 Descendants of Sanskrit s and sis aorists and similar forms § 32.47 Type āsi (āsil) § 32.53 Type esi (esi, eşīl) § 32.63 Other si (sil, etc.) after long vowels § 32.69 s-aorists from roots in $r(r)$ § 32.74 Third plural āsu(h), ānsu(h), ansu(h), atsu etc. § 32.75 Miscellaneous s-aorists § 32.85 The ending e(t) as preterite § 32.85 Third plural forms in eşu, ensu, etsu(h) etc.	157 157 157 157 157 157 157 157 158 159 159 160 160 160 160
First plural imha § 32.40 itha, ithā, ittha, ita (mostly 3d singular) § 32.41 Third plural or singular ire § 32.43 Other MIndic forms with Skt. is-aorist endings § 32.44 Descendants of Sanskrit s and sis aorists and similar forms § 32.47 Type āsi (āsil) § 32.53 Type esi (esi, eşil) § 32.63 Other si (sil, etc.) after long vowels § 32.69 s-aorists from roots in $r(r)$ § 32.74 Third plural āsu(h), ānsu(h), ansu(h), atsu etc. § 32.75 Miscellaneous s-aorists § 32.85 The ending $e(t)$ as preterite § 32.87 Third plural forms in eşu, ensu, elsu, elsu(h) etc. § 32.95	157 157 157 157 157 157 157 158 159 159 160 160 160 160 161
First plural imha § 32.40 itha, ithā, ittha, ita (mostly 3d singular) § 32.41 Third plural or singular ire § 32.43 Other MIndic forms with Skt. is-aorist endings § 32.44 Descendants of Sanskrit s and sis aorists and similar forms § 32.47 Type āsi (āsil) § 32.53 Type esi (esi, eşil) § 32.63 Other si (sit, etc.) after long vowels § 32.69 s-aorists from roots in $r(r)$ § 32.74 Miscellaneous s-aorists § 32.78 Miscellaneous s-aorists § 32.85 The ending e(t) as preterite § 32.87 Third plural forms in eşu, emsu, etsu, etsu, (t), etc. § 32.95 Miscellaneous optative forms used as aorists	157 157 157 157 157 157 157 158 159 159 160 160 160 160 161
 First plural imha § 32.40	157 157 157 157 157 157 157 158 159 159 160 160 160 161 161
First plural imha § 32.40	157 157 157 157 157 157 157 157 157 157
First plural imha § 32.40	157 157 157 157 157 157 157 157 157 157
First plural imha § 32.40	157 157 157 157 157 157 157 157 157 157
First plural imha § 32.40	157 157 157 157 157 157 157 157 157 157
First plural imha § 32.40 itha, ithā, ittha, ita (mostly 3d singular) § 32.41 Third plural or singular ire § 32.43 Other MIndic forms with Skt. is-aorist endings § 32.44 Type āsi (āsil) § 32.53 Type āsi (āsil) § 32.63 Type esi (esi, eşīl) § 32.63 Other si (sīl, etc.) after long vowels § 32.69 s-aorists from roots in $r(r)$ § 32.74 Third plural āsu(h), ānsu(h), ansu(h), atsu etc. § 32.75 Miscellaneous s-aorists § 32.85 Third plural āsu(h), ānsu(h), ansu(h), etc. § 32.95 Third plural forms in eşu, emsu, ensu, etsu(h) etc. § 32.95 Miscellaneous optative forms used as aorists § 32.105 Relatives of the Sanskrit root-aorist § 32.106 Relatives of Sanskrit thematic aorists (and imperfects) § 32.110 Third singular a, ā (roots drś, gam, vac, etc.) § 32.110	157 157 157 157 157 157 157 157 157 157
First plural imha § 32.40	157 157 157 157 157 157 157 158 159 159 160 160 160 161 161 161 162 163 163
 First plural imha § 32.40	157 157 157 157 157 157 157 157 157 157
First plural imha § 32.40	157 157 157 157 157 157 157 158 159 159 160 160 160 161 161 161 162 163 163

34. Participies, including gerundives	167
Present participle suffixes affixed to past participles	
§ 34.1	167
māna with athematic stems § 34.3	167
āna with thematic stems § 34.4	167
Blend of perfect vāņis and (la)vant § 34.6	167
ita added to thematic present stems § 34.7	167
Spread of <i>ta</i> outside its Sanskrit bounds § 34.11	168
Strong form of root, for Sanskrit weak form, before ta	
§ 34.13	169
Weak form of root, for Sanskrit strong form, before ta	
§ 34.14	169
Participles in ta with active meaning from transitive	
verbs § 34.15	169
Suffix na § 34.16	169
Gerundives in itanya § 34.20	169
Other non-Sanskrit forms in lavya § 34.21	169
Gerundives in aniya § 34.22	170
Gerundives in ya § 34.24	170
Gerundives in ga 304.24	170
35. Gerund	171
Gerunds in am § 35.3	171
Gerunds in <i>im</i> § 35.6	171
Gerunds in $lv\bar{a}$ and $ya \S 35.7$	171
Present stems as 'roots' before lvā and ya § 35.12.	172
From nasal infix presents § 35.13	172
From reduplicating presents § 35.14	172
From ninth-class presents § 35.15	172
From <i>ya</i> -presents § 35.16	172
The type of grahāya § 35.19	172
Gerunds in <i>ilvā</i> § 35.22	173
(in etvā § 35.25)	173
Gerunds in (i)lvāna (°tvanā, °tvana) § 35.29	174
Gerunds in <i>lūna</i> , m.c. <i>tunu</i> § 35.36	175
Gerunds in <i>iya</i> (<i>iyâ</i> , <i>ia</i>) § 35.37	175
From a (not aya) presents § 35.42	175
From aya presents § 35.43	175
Gerunds in <i>iyāna</i> § 35.45	175
Gerunds in <i>i</i> , <i>ī</i> § 35.49	175
Gerunds in <i>tā</i> § 35.52	176
Gerunds in <i>tu(m)</i> § 35.54	177
Gerund (falsely alleged) in dhya § 35.56	177
36. Infinitive	178
Ending <i>itum</i> § 36.2	178
<i>tu</i> for <i>tum</i> § 36.3	178
$t\hat{u} \text{ for } tu = tum \ \S \ 36.5$	179
<i>itum</i> for <i>itum</i> § 36.6	179
atum for itum § 36.7	179
Ending elum § 36.8	179
Miscellaneous non-Sanskrit forms in tum § 36.9	179
Ending tave (tavai) § 36.14	179
Ending anāya § 36.15	170
	173
Ending and s 36.16	179 179
Ending anāye § 36.16	179
Ending ana-tāyai § 36.17	179 180
Ending ana-tāyai § 36.17 Ending anā (ana?) § 36.18	179 180 180
Ending ana-tāyai § 36.17 Ending anā (ana?) § 36.18 Other dative infinitives § 36.19	179 180 180 180
Ending ana-tāyai § 36.17 Ending anā (ana?) § 36.18	179 180 180
Ending ana-tāyai § 36.17 Ending anā (ana?) § 36.18 Other dative infinitives § 36.19	179 180 180 180
Ending ana-tāyai \S 36.17 Ending anā (ana?) \S 36.18 Other dative infinitives \S 36.19 am , $akam$, $im(?)$ \S 36.20	179 180 180 180 180
Ending ana-tāyai \S 36.17 Ending anā (ana?) \S 36.18 Other dative infinitives \S 36.19 am, akam, im(?) \S 36.20 37. Voice Passives in <i>iya</i> , <i>iya</i> \S 37.2 <i>iya</i> passives from <i>aya</i> (e) presents \S 37.6	179 180 180 180 180 180
Ending ana-tāyai \S 36.17 Ending anā (ana?) \S 36.18 Other dative infinitives \S 36.19 am, akam, im(?) \S 36.20 37. Voice Passives in <i>iya</i> , <i>iya</i> \S 37.2 <i>iya</i> passives from <i>aya</i> (e) presents \S 37.6	179 180 180 180 180 181 181
Ending ana-tāyai § 36.17. Ending anā (ana?) § 36.18. Other dative infinitives § 36.19 am, akam, im(?) § 36.20 37. Voice Passives in iya, iya § 37.2	179 180 180 180 180 181 181 181
Ending ana-tāyai \S 36.17 Ending anā (ana?) \S 36.18 Other dative infinitives \S 36.19 am, akam, im(?) \S 36.20 37. Voice Passives in iya, iya \S 37.2 iya passives from aya (e) presents \S 37.6 iya passives from a presents \S 37.7 iya passives from aya (e) presents \S 37.8	179 180 180 180 180 181 181 181 181
 Ending ana-tāyai § 36.17. Ending anā (ana?) § 36.18. Other dative infinitives § 36.19. am, akam, im(?) § 36.20. 37. Voice. Passives in iya, iya § 37.2. iya passives from aya (e) presents § 37.6. iya passives from a presents § 37.7. iya passives from aya (e) presents § 37.8. iya passives from a presents § 37.9. 	179 180 180 180 180 181 181 181 181 181 182
 Ending ana-tāyai § 36.17 Ending anā (ana?) § 36.18 Other dative infinitives § 36.19 am, akam, im(?) § 36.20 37. Voice Passives in iya, iya § 37.2 iya passives from aya (e) presents § 37.6 iya passives from aya (e) presents § 37.7 iya passives from aya (e) presents § 37.8 iya passives from a presents § 37.9 Active for middle (passive) endings § 37.10 	179 180 180 180 180 181 181 181 181 182 182
 Ending ana-tāyai § 36.17 Ending anā (ana?) § 36.18 Other dative infinitives § 36.19 am, akam, im(?) § 36.20 37. Voice Passives in iya, iya § 37.2 iya passives from aya (e) presents § 37.6 iya passives from aya (e) presents § 37.8 iya passives from a presents § 37.9 Active for middle (passive) endings § 37.10 Seemingly passive forms with active meaning and 	179 180 180 180 181 181 181 181 182 182 182
 Ending ana-tāyai § 36.17 Ending anā (ana?) § 36.18 Other dative infinitives § 36.19 am, akam, im(?) § 36.20 37. Voice Passives in iya, iya § 37.2 iya passives from aya (e) presents § 37.6 iya passives from a presents § 37.7 iya passives from a presents § 37.8 iya passives from a presents § 37.9 Active for middle (passive) endings § 37.10 Seemingly passive forms with active meaning and usually active endings § 37.22 	179 180 180 180 181 181 181 181 182 182 182 182
 Ending ana-tāyai § 36.17 Ending anā (ana?) § 36.18 Other dative infinitives § 36.19 am, akam, im(?) § 36.20 37. Voice Passives in iya, iya § 37.2 iya passives from aya (e) presents § 37.6 iya passives from aya (e) presents § 37.6 iya passives from aya (e) presents § 37.8 iya passives from a presents § 37.9 Active for middle (passive) endings § 37.10 Seemingly passive forms with active meaning and usually active endings § 37.22 	179 180 180 180 181 181 181 181 181 182 182 182 183 183
 Ending ana-tāyai § 36.17 Ending anā (ana?) § 36.18 Other dative infinitives § 36.19 am, akam, im(?) § 36.20 37. Voice Passives in iya, iya § 37.2 iya passives from aya (e) presents § 37.6 iya passives from aya (e) presents § 37.6 iya passives from aya (e) presents § 37.8 iya passives from a presents § 37.9 Active for middle (passive) endings § 37.10 Seemingly passive forms with active meaning and usually active endings § 37.22 Medio-passive form apasive stems § 37.26 	179 180 180 180 181 181 181 181 181 182 182 182 183 183 183
 Ending ana-tāyai § 36.17 Ending anā (ana?) § 36.18 Other dative infinitives § 36.19 am, akam, im(?) § 36.20 37. Voice Passives in iya, iya § 37.2 iya passives from aya (e) presents § 37.6 iya passives from aya (e) presents § 37.6 iya passives from aya (e) presents § 37.8 iya passives from a presents § 37.9 Active for middle (passive) endings § 37.10 Seemingly passive forms with active meaning and usually active endings § 37.22 	179 180 180 180 181 181 181 181 181 182 182 182 183 183

38. Stems in aya (e); causative, denomina-	
tive, etc	185
Causatives in ayali § 38.9	185
Causatives in aya from passive stems § 38.10	186
Causatives with long \tilde{a} for a in root-syllable § 38.14.	186
Verbs in ayati (eti) = $ati $ § 38.18	186
Historic causatives with non-causative meaning § 38.22	188
Non-causative forms with causative meaning § 38.24	189
Verbs in ati = ayati (eti) § 38.25	189
Optatives § 38.27	189
Present Indicatives § 38.28	190
Present Imperatives § 38.29	190
Present Participles § 38.30	190
Futures § 38.31	190
Aorists § 38.32	190
Gerunds § 38.33	190
Denominatives § 38.34	190
Causatives in apayati (apeti, aveti) § 38.41	191
Forms in apayali from roots in a § 38.52	192
Causatives in <i>āpaya</i> (<i>āpe</i>) to present stems in a	
§ 38.53	192
Causatives in āpaya (āpe) to present stems in aya(e)	
§ 38.54	193
Causatives from aya presents which are neither	
caus. nor denom. § 38.55	193

Causatives from denominatives § 38.56 Causatives from <i>aya</i> forms which are them-	193
selves originally causative § 38.57	194
Formal causatives in (\tilde{a}) paya without causative	
meaning § 38.58	195
Miscellaneous and irregular forms in payati § 38.64	195
Prakritic forms in aveti § 38.68	196
39. Intensive	197
40. Desiderative	198
41. Syntactic addenda to the verb Special uses of as and $bh\hat{u}$; periphrastic verbal expres-	199
sions § 41.1	199
42. Constructions with ma	200
Prohibitions § 42.5	200
Strong determination § 42.6	200
'Lest' clauses § 42.7	200
Strong negative wishes or hopes for the future § 42.8 Fears for the future (also referring to present and past)	201
§ 42.9	201
Questions § 42.12	202
na for $m\bar{a}$ with imperative(?) § 42.16	203
43. Synopsis of verb forms	204

XIX

PREFACE

EARLY TWO DECADES ago I began tentative studies which have grown into the present work, the first attempt at a description of the grammar and lexicon of Buddhist Hybrid Sanskrit. By no means all, but the major part, of my research activity during these two decades has gone into this work. I know that especially the dictionary falls short of what I wish it were. If I could take the time to repeat the whole process, that is to restudy carefully and minutely all the texts included in my plan, I am certain that these volumes could be enlarged and brought nearer to completion.

One limitation has been self-imposed. I have deliberately excluded words and forms occurring (with the same meanings) in standard Sanskrit. For this reason, in order to interpret BHS texts, one must supplement my grammar and dictionary by Sanskrit grammars and Sanskrit dictionaries, especially the lexicons of Boehtlingk (and Roth), which I treat as containing substantially the Sanskrit word-stock. (They also contain some words and meanings which are not standard Sanskrit but BHS; these are, of course, included in my work.)

I have no doubt that some competent scholars will question the wisdom of this limitation. I ask them at least not to condemn it until they have carefully considered my reasons, which are stated in Chapter 1, especially §§ 1.57 ff., to which §§ 1.33-56 furnish a necessary background. I admit that at first sight it may seem startling that a dictionary of Buddhist Hybrid Sanskrit should contain no article on, for instance, the word nirvāna, because in the texts included in this work I have failed to find any use of that word which is not adequately covered for standard Sanskrit by Boehtlingk. This is not to deny that in certain works of technical Buddhist metaphysics, composed in standard Sanskrit and therefore not considered by me, philosophical aspects of the meaning of nirvana may depart considerably from those recognized by Bochtlingk. But, unless I have erred, the Northern Buddhists who used BHS did not use the word nirvana in any way not known to their brahmanical fellow-countrymen. To this fact (and I believe it is a fact) the silence of my dictionary bears eloquent testimony. It does not, to be sure, guarantee the converse, namely that every standard Sanskrit word is used in BHS in all its Sanskrit meanings; that is far from being true. What I have tried to do is to record all non-Sanskrit elements, grammatical or lexical, found in BHS. The elements common to Sanskrit and to the BHS texts as we have them cannot safely be assumed to be genuine BHS, for reasons explained in the place cited; that is why I have omitted them.

I considered seriously providing the Grammar with an index. No doubt the lack of it may seem a defect to some. It would have added to the already swollen bulk of this work, and I finally decided that it was made unnecessary by the following considerations. The Table of Contents is very detailed, and, it is hoped, analyzes the grammar so clearly as to make an index scarcely profitable. The Dictionary also, to a large extent, replaces such an index.

Pre/ace

The verb-forms, for which an index might seem more needed than for other parts of the grammar, are summarily analyzed by roots in Chapter 43, the Synopsis of Verb Forms. This chapter especially, and also all the rest of the Grammar and the Dictionary, contain many cross-references to sections of the Grammar in which grammatical features, especially less obvious ones, are recorded.

After finishing the Grammar and Dictionary, 1 decided to publish with them, or shortly after, a Reader containing selections from the most important texts, which are hardly accessible except in large libraries. It is hoped that this third volume will facilitate the use of the other two, and prove helpful to Indologists, both for private reading and in conducting university courses.

In my lamented ignorance of Chinese and Japanese, I have often applied for help in dealing with those languages to various Sinological or Japanological colleagues, past or present. Those who have contributed most, in addition to Mr. Chang Kun (who is mentioned below), are Professors Kan-ichi Asakawa (deceased), Chao Yuen-ren, George A. Kennedy, Li Fang-kuei, Lo Ch'ang-pei, and Johannes Rahder. I am deeply indebted to all these gentlemen for their unvarying patience and helpfulness.

Mr. D. R. Shackleton Bailey, of Gonville and Caius College, Cambridge, England, generously put at my disposal some of his notes on the Tibetan version of the Divyāvadāna, supplementing his valuable Notes in JRAS 1950.166 ff.

Professor Paul Mus, at present my colleague at Yale, most generously volunteered to help in the final checking of quotations and references, and did a considerable part of the labor of checking those from the Mañjuśrīmūlakalpa. In doing so he made a number of corrections and at least one notable improvement in the Dictionary; the article *cīraka* is almost wholly his work, and I am proud to be able to include it.

My very able pupil, Mr. Chang Kun, gave me invaluable help by checking the quotations and references from a number of texts, including some of the greatest importance, such as the Saddharmapundarīka, the Gandavyūha, and the Mahāvyutpatti. His knowledge of Tibetan and of his native Chinese, as well as of Sanskrit, made his assistance especially important.

My secretary, Miss Eleanor A. Hill, who also knows Sanskrit, has helped by checking the quotations and references from a number of texts. She has typed the entire work, much of it more than once and some of it more than twice (owing to no fault of hers), with extraordinary skill and efficiency. She has made many other valuable contributions; for example, the subdivisions and typographical arrangement of the parts of the grammar owe much to her intelligence and skill. It would be hard to overstate what she has done to lighten my labors in preparing this work for publication.

Another able pupil of mine, Miss Betty Jean Shefts, cooperated with Miss Hill in reading the first proof of the entire Grammar and Dictionary. These two readers, who showed remarkable competence at this none too easy task, have notably reduced the number of misprints and other errors which would have escaped my own proof reading.

The publication has been very materially assisted by generous financial support from the John Simon Guggenheim Memorial Foundation. To it, and to its Secretary General, Mr. Henry Allen Moe, and Assistant Secretary, Mr. James F. Mathias, I am very deeply indebted.

Even with this help, the publication would perhaps not have been possible without the

XXII

Prc/ace

generosity of the Yale University Press in issuing it, an undertaking which will certainly not be financially profitable. I am especially grateful to Provost E. S. Furniss, at that time Director of the Press, whose influence certainly had much to do with these arrangements.

I cannot close without an expression of gratitude to my friend William Clyde DeVane, Dean of Yale College. He is himself a great humanist, and one of the most effective supporters of humanistic research whom I know. This makes me the more appreciative of the unwavering confidence he has shown in my ability to accomplish the task I undertook in this work—a confidence which has been shown in various ways, and which has had important practical results. If it should prove that this confidence was not seriously misplaced, I know he would feel rewarded.

BIBLIOGRAPHY AND ABBREVIATIONS

THE BHS TEXTS fall roughly into three main classes, as regards their language. These may possibly, if only loosely, represent successive chronological layers in the tradition. It must be emphasized, however, that such a chronological classification does not necessarily apply to the original composition of the texts in question, but only to the forms in which they are now preserved to us. Some of them are known to have undergone changes, always in the direction of greater Sanskritization (see §§ 1.39 ff.; note for example the facts about Vaj, § 1.41, which is put in my third class). Winternitz (Hist. Ind. Lit. 2.314) holds that the Prajñā-Päramitās 'belong to the earliest Mahāyāna-Sūtras... A Prajñā-Pāramitā was already translated into Chinese as early as 179 A.D.' Yet AsP and SsP, the two works of this class which I have included, belong linguistically in my third class, in the forms now preserved to us; they contain practically no verses, and very little non-Sanskrit morphology.

1. The first class consists only of Mv, plus a short citation from the lost Bhikṣuprakīrņaka (in Śikṣ), and a short Jātaka printed in the Appendix to Jm. In these works, the prose parts are thoroughly hybridized, showing as many Middle-Indicisms as the verses. In its 'nucleus', at least, Mv seems to be the oldest BHS text we have; and this linguistic type is surely older than the other two.

2. In the second class, the verses are hybridized, as in the first class, but the prose contains relatively few signs of Middle Indic phonology or morphology. However, the vocabulary, at least, stamps even the prose as BHS. In this class belong SP, LV, Gv (including Bhad), KP, Dbh, RP, Sukh, Samādh, Suv, Ud (contains no prose), and nearly all of Śikṣ (which consists mainly of quotations from older texts, many of which are not otherwise preserved).

3. In the third class, even the verses (if any; some of these texts contain few or none; those which have none could be included in the second class) are substantially as Sanskritized as the prose parts; the entire text of these works resembles linguistically the prose parts of the second class. That is, non-Sanskritic forms are not common; the vocabulary is the clearest evidence that they belong to the BHS tradition. These include MSV, Divy, Av, Prāt, Mmk, Bbh, AsP, ŚsP, Māy, Bhīk, Karmav, Kv, Lańk, Vaj, and Jm; here belong also the vocabulary-lists in Dharmas and Mvy, and most of the miscellaneous short texts collected in Sādh, so far as they are not standard Sanskrit, which not a few of them seem to be.

Except as otherwise stated below, references are to page and line of editions.

AbhidhK. = Abhidharmakoša, transl. LaVallée Poussin, Paris and Louvain, 1923-31; many technical words cited from Index, transl., and notes. References, as in the Index, are to Book (of the original) and page of the translation.

AbhK. = preceding.

- AdP Konow MASI = Sten Konow, Central Asian Fragments of the Astādašasāhasrikā, in Memoirs of the Archaeological Survey of India, No. 69, 1942. Only a few rare and important words.
- AsP = Aştasāhasrikā-prajňāpāramitā, cd. Rajendralala Mitra, Calcutta, 1888.
- Av = Avadāna-šataka, ed. Speyer, 2 vols., St. Petersburg, 1902, 1906; transl. Feer, Annales du Musée Guimet 18 (1891).
- Bbh = Bodhisattvabhūmi, ed. Wogihara, Tokyo, 1930—
 36. Includes the editor's dissertation, Lexikalisches aus der Bodhisattvabhūmi, which is referred to as Wogihara, Lex.
- Bct, Bca.t. = Bodhicarvāvatāra-tikā, cd. LaVallée Poussin, Brussels, 1898. Technical terms cited from Index.
- Bhad = Bhadracari, ed. Watanabe, Leipzig, 1912. Identical with Gv 543.9-548.2. Cited by numbered verses.
- Bhik = Bhiksuni-karmavācanā, ed. Ridding and LaVallée Poussin, BSOS 1, Part 3, 123-143. Cited by page and line of the manuscript as reproduced in ed.
- Buddhac. = Buddhacarita, ed. Johnston, Calcutta, 1935. The language is standard Skt.; a few words, chiefly religious, are cited by chapter and verse.
- Calc. = Calcutta; see LV.
- Dbh = Daśabhūmikasūtra, ed. Rahder, Paris and Louvain, 1926.
- Dbh.g. = id. (Gāthā portion), ed. (Rahder and) Susa, Eastern Buddhist, 5.337 ff. and 6.51 ff., cited from reprint with consecutively numbered pages 1-58; the pages of the reprint are cited first, then in parentheses the pages of the original, then the line on each page.
- Dbs = Daśabalasūtra, in Waldschmidt, Kl. Skt. Texte 4, 209 ff.
- Dharmas = Dharmasamgraha, ed. Müller, Oxford, 1885. Cited by section numbers.
- Divy = Divyāvadāna, ed. Cowell and Neil, Cambridge, 1886.
- Foucaux, see LV.
- Guhyasamāja, ed. Bhattacharya, Gaekwad's Oriental Series 53, Baroda, 1931; contains a few verses with BHS forms; very rarely cited.
- Gv = Gandavyúha, ed. Suzuki and Idzumi, Kyoto, 1934-36; 2d ed., photographically reproduced with corrections, 1949.
- Hoernle, see MR.
- Jm = Jātakamālā, ed. Kern (HOS 1), Boston, 1891. Its language is close to standard Skt., but contains quite a number of BHS words, and even a few forms; see § 1.53, fn. 17. Transl. Speyer, London, 1895. K', see SP.
- Karmav = (Mahā-)Karmavibhanga, and Karmavibhangopadeśa, ed. Lévi, Paris, 1932.
- Kl. Ski. Texte = Kleinere Sanskrit Texte (5 vols.; Kgl. Preuss. Turfan-Expeditionen), Leipzig (Vol. 1 Berlin), 1911-39.
- KN, see SP.
- KP = Kāśyapa Parivarta, ed. Staël-Holstein, Shanghai, 1926. Cited by sections as numbered in edition, and lines.

- Kv = Kāraņdavyūha, in The Hindu Commentator, vol. 5; 1872.
- Lańk = Lańkāvatāra-sūtra, ed. Nanjio, Kyoto, 1923; transl. Suzuki, London, 1932; Index, by Suzuki, Kyoto, 1934; 'Studies' = Suzuki, Studies in the Lankavatara Sutra, London, 1930.

Lefin., see LV.

- LV = Lalitavistara, ed. Lefmann, Halle, 1902; Critical Apparatus, 1908; abbreviated Lefm.; 'Calc.' == ed. of Rajendralala Mitra, Calcutta, 1877; transl. Foucaux, Annales du Musée Guimet 6, and Notes, 19; Tibetan version (partial), with transl., by Foucaux, Paris, 1847; Weller = F. Weller, Zum Lalita Vistara, I. Über die Prosa des Lal. Vist., Leipzig, 1915.
- MadhK == Müla-Madhyamaka-kārikās, with comm., ed. LaVallée Poussin, St. Petersburg, 1903-1910. Only partially excerpted.
- Mahāsamāj. = Mahāsamājasūtra, in Waldschmidt, Kl. Skt. Texte 4.
- Māy = Mahā-Māyūrī. Two figures refer to page and line of ed. by Oldenbourg in Zapiski Vostochnago Otdelenija Imp. Russk. Archeol. Obschestva, Tom 11 (1899), 218-261. A single number refers to the numbered lines of 'Le catalogue géographique des yakşa dans la Mahā-Māyūrī', by S. Lévi, JAs. XI. 5 (1915), pp. 1-138.
- Megh = Meghasūtra, ed. Bendall, JRAS 1880, 288 ff.
- Minaeff, see Mvy.
- Mironov, see Mvy.
- Mmk = (Ārya-)Mañjuśrimūlakalpa, ed. Gaņapati Šāstri, 3 vols., Trivandrum, 1920, 1922, 1925. See Lalou, in the following list of abbreviations.
- MPS = E. Waldschmidt, Das Mahāparinirvāņasūtra. 3 parts; Abhandlungen der deutschen Akademie . . . zu Berlin, 1950, 1951. Arrived too late to be extensively used; references are to Waldschmidt's numbered divisions.
- (Hoernle) MR = Hoernle (general editor), Manuscript Remains of Buddhist literature found in Eastern Turkestan, Oxford, 1916. BHS fragments ed. by Hoernle, Lüders, Pargiter, Thomas.
- MSV = Mūla-Sarvāstivāda-Vinaya; Gilgit Manuscripts, ed. Nalinaksha Dutt, Vol. 3, Parts i (undated), ii (1942), iii (1943), Srinagar, Kashmir. Part iv (1950) arrived late, but its most important words have been included. Unfortunately this valuable series came to hand so late in the progress of my task that I felt it impossible to include Volumes 1 and 2, which seem somewhat less important.
- Mv = Mahāvastu, ed. Senart, Paris, 1882, 1890, 1897. The first volume of Jones's translation (London, 1949) appeared, to my regret, too late to be much used (I received it in 1950).
- Mvy = Mahāvyutpatti. Cited by numbered entries in ed. of Sakaki, Kyoto, 1916, with Sanskrit and Tibetan Indexes, 1925 and 1936; some variants are cited from Mironov (2d ed. of Minaeff, which is cited in BR and pw), St. Petersburg, 1910-11. Tibetan version cited from Sakaki; also translations of many Chinese and Japanese renderings *ibidem*.

Pischel, SBBA 1904 pp. 807-827: Bruchstücke des Sanskritkanons der Buddhisten aus Idykutšari. A few rarer words cited by number of page and manuscript-folio.

Prāt = Prātimokşasūtra des Sarvāstivādins, ed. Finot (with translation of Chinese version by Huber), JAs. XI.2 (1913), 473 fl.

(XXVI)

- RP == Rāstrapālapariprechā, ed. Finot, St. Petersburg, 1901.
- Sādh = Sādhanamālā, ed. Bhattacharya, Baroda, 2 vois., 1925, 1928.
- = Sälistambasútra, (reconstructed) ed. LaVallée Poussin; Bouddhisme, Études et matériaux, Théorie Śāl des douze causes (Gand, 1913), pp. 69-90.
- Samādh = Samādhirājasūtra, (1) chapters 8, 19, 22, ed. Régamey, Warsaw, 1938, cited by chapters and numbered sections or verses; (2) ed. Das and Vidyābhūshan (Calcutta, 1896); references (only a few; text incredibly corrupt) cited by page and lien, thus: 'p. ---, line --
- Samy. Ag. = Samyuktāgama; fragments from Stein mss. ed. LaVallée Poussin, JRAS 1913, 569 fl.; a few references to numbers of fragments (1-15), followed by r(ecto) or v(erso) when necessary, and line of mss.
- Siks = Siksāsamuccaya, ed. Bendall, St. Petersburg, 1897-1902; transl. Bendall and Rouse, London, 1922.
- SP == Saddharmapundarika, ed. Kern and Nanjio, St. Petersburg, 1912, abbreviated KN; supplementary references to ed. of Wogihara and Tsuchida, Tokyo, 1934-35, abbreviated WT; fragments of 'Kashgar' or Central Asiatic recension, ed. Thomas and Lüders, ap. Hoernle, MR 133 ff., 144 ff.; others, ed. LaVallée Poussin, JRAS 1911.1070 ff.; transl. Burnouf (Lotus de la Bonne Loi, Paris, 1852), and Kern (SBE 21, Oxford, 1884). Tibetan citations chiefly from blockprint in my possession, partly from WT. When my work was practically ready for print, my colleague Professor Rahder received, and lent to me, the

- photostatic reproduction of the ms. referred to by WT as K'. It has been cited a very few times. The quotations from it in WT seem to be very inaccurate. ŚsP = Śatasāhasrikā-prajňāpāramitā, ed. P. Ghosa, Bi-
- bliotheca Indica, Calcutta, 1902 ff. Sukh = Sukhāvatīvyūha ('Larger' and 'Smaller'), ed.
- Müller, Oxford, 1883.
- Süträl. = Asanga, Mahāyāna-sūtrālamkāra, ed. and transl. Lévi, Bibliothèque de l'École des Hautes Études, 159 and 190, Paris, 1907, 1911. The language is standard Skt., but the text and Lévi's notes are important for the understanding of many technical terms. Cited by chapter and verse (with commentary after each verse).
- Suv = Suvarņabhāsottamasūtra, ed. Nobel, Leipzig, 1937; Tibetan version, ed. same, Leiden and Stuttgart, 1944.
- Svay = (Vrhat-)Svayambhū Purāņa, ed. Haraprasad Sastri, Calcutta, 1894-1900; of very little use and only rarely cited. Cf., Winternitz, Hist. Ind. Lit. 11, 376, note.
- Ud = Udānavarga; chap. 1-21, cd. Chakravarti (1930; except ch. 4, for which see Lévi, JAs. X. 20, 1912, 203 ff.); ch. 29, ed. Pischel, SBBA 1908, 968 ff.; ch. 30, 31, 32, ed. LaVallée Poussin, JRAS 1912, 369 ff. Cited by chapter and verse.
- Vaj = Vajracchedikā, ed. Müller, Oxford, 1881; fragments of Central Asiatic version, ed. Pargiter, ap. Hoernle MR 179 ff.

Weller, see LV.

Wogihara, Lex., see Bbh.

WT, see SP.

Other abbreviations and works chiefly referred to

AB = Aitareya Brāhmaņa.

- Abhidh-s. = Abhidhammatthasamgaha (Pali).
- abl. = ablative.
- ABORI = Annals of the Bhandarkar Oriental Research Institute (Poona).
- abstr. == abstract.
- acc. = according; accusative.
- Acharya (P.K.), Dict(ionary of) Hindu Arch(itecture), Oxford Univ. Press, n.d. [1927?].
- act. = active; action.
- Add. = Addendum, Addenda (e.g. to BR, pw).

adj. = adjective.

- AIS = Delbrück, Altindische Syntax.
- Ait. Br. = Aitareya Brāhmaņa.
- AKM = Abhandlungen für die Kunde des Morgenlandes. al. = alibi, elsewhere.
- Alsdorf, see Kum.
- Amarak. = Amarakośa.
- AN = Anguttaranikāya (Pali).
- analog. = analogical(ly).
- aor. = aorist.
- ap. = apud.
- Ap. = Apabhramśa; Apadana (Pali text).
- app. = apparently.
- App. = Appendix. ApS. = Apastamba Śrauta Sūtra.
- Apte = Apte, Student's Sanskrit-English Dictionary (1922).
- ASGW = Abhandlungen der sächsischen Gesellschaft der Wissenschaften.
- assoc. = associated.

AV. = Atharvaveda.

AVPariś. = Atharvaveda Pariśista.

- Äyär. = Äyäranga Sutta (AMg.).
- $B_{\cdot} = Buddha_{\cdot}$ B (as in Wh. Roots) = Brāhmaņa(s).
- B. in Tr. (Warren) = Warren's Buddhism in Transla-
- tions, HOS. 3. BEFEO = Bulletin de l'École Française d'Extrême
- Orient.
- BhagP. = Bhagavata Purana.
- Bhav. = Bhavisattakahā (Ap.), ed. Jacobi (1918).
- BhG. = Bhagavad Gitā.
- BHS = Buddhist Hybrid Sanskrit.
- Bhvr. = Bahuvrihi.
- Bibl. Bu. = Bibliotheca Buddhica.
- Bloch (Jules), (L')Indo-Aryen, Paris, 1934.
- Bloch (Jules), (Les Inscriptions d') Asoka, Paris, 1950.
- BR = Boehtlingk and Roth, Sanskrit Wörterbuch.
- Br. = Bråhmaņas (language of).
- BSL = Bulletin de la Société de Linguistique (de Paris).
- BSOS = Bulletin of the School of Oriental (and African) Studies, London.
- Bu. = Buddhist.
- Bull(etin of the) Deccan Coll(ege) Research Inst(itute) (Poona).
- Burnouf, Intr. = Introduction à l'histoire du bouddhisme indien (1844).
- Burnouf, Lotus = transl, of SP, q.v. in preceding list. Burrow (T.), (The languages of the) Kharosthi Documents, Cambridge, 1937.

Bv, Bu.v. = Buddha-vamsa (Pali).

Car. = Caraka.

- caus. = causative.
- cf. = compare.

XXVIII

ch., chap. = chapter. Childers = Dictionary of the Pali Language, R. C. Childers. Chin. == Chinese. Cl. Skt. - Classical Sanskrit. comb. = combination. comm. == commentary. comp. = composition. Compendium of Philosophy (Aung and Rhys Davids, London, 1910). compy. == comparative. corr. = correct(s), correction. corresp. = correspondent, corresponding, corresponds. Corrig. = Corrigenda. cpd. == compound. CPD = Andersen and Smith, Critical Pali Dictionary (Vol. 1), Copenhagen, 1924 fl. crit. app. = critical apparatus. Das = S. C. Das, Tibetan-English Dictionary. Daśak. = Daśakumāracarita. dat. == dative. def. == definition. denom. = denominative. der(iv). = derived, derivative. Deśin. = Deśināmamālā, ed. Pischel, Bombay, 1880. Dhātup. == Dhātupāțha. Dhp. = Dhammapada (Pali); Dhp. c(omm.), commentary on Dhp.; Fausb. refers to Fausböll's ed. of Dhp. Dhs. = Dhanimasamgani (Pali). dial. = dialectic. Dict. = Dictionary, usually referring to my Dict. of BHS; Dictt. = Dictionaries. Dict. of the Fr. Cath. Miss. = the Tibetan Dictionary of the French Catholic Mission. Dict. Hindu Arch. = Acharya, Dictionary of Hindu Architecture. diff. = different. dim. == diminutive. Dip. = Dipavamsa (Pali). DN = Dighanikāya (Pali). DPPN = Malalasekara, Dictionary of Pali Proper Names. Dpv., Dpvs. = Dip. Dutreuil (de Rhins ms.), see § 1.3 with fn. 2. E = Epic (Sanskrit, in citing Whitney's Roots). ed. = editor, edition. em. = emended, emendation. ep. = epithet.Ep. Zeyl. = Epigraphia Zeylanica. equiv. = equivalent. erron. = erroneously. etym. = etymology. esp. = especially.exc. = except. expl. = explained, explanation. f. = feminine; following. fem. = feminine. Fest. = Festschrift. $\mathbf{ff.} = \mathbf{following.}$ fig. = figurative(ly). fn. = footnote. foll. = followed (by); following. fut. = future. gdve. = gerundive.Geiger (W.) = Pali, Strassburg, 1916. Cited by numbers of sections, without the symbol §. gen. = genitive. ger. = gerund. Ger. = German. Git. = Gitagovinda. GobhGS = Gobhila Grhya Sútra. GOS = Gaekwad's Oriental Series. Gr. = Grammar, grammatical, grammarian(s).

haplogr. == haplography. haplol. == haplology. Hem. = Hemacandra, chiefly his Prakit Grammar, ed. and transl. Pischel, Halle, 1877, 1880; see also Deśin. HJAS = Harvard Journal of Asiatic Studies. HOS = Harvard Oriental Series. Hultzsch (E.), Inscr(iptions) of Asoka, Oxford, 1925. $I\Lambda$ = (Lassen) Indische Altertumskunde. 1F. = Indogermanische Forschungen. ifc. = at the end of compounds. IHO = Indian Historical Quarterly. impf. = imperfect. impv. = imperative. incl. = including, inclusive. indecl. = indeclinable. indic. = indicative. Ind. Min. = (Garbe) Die indischen Mineralien (1882). Ind. Spr. = Boehtlingk, Indische Sprüche (2d ed.). inf. = influitive.inst., instr. = instrumental. intens. = intensive intrans. = intransitive. l.St. = Indische Studien (Weber). lt., Itiv. = Itivuttaka (Pali). JA. = Journal Asiatique. Jä., see Jäschke. Jacobi, Ausgew. Erz. = Ausgewählte Erzählungen in Māhārāstrī. See also Bhav., San. JAOS = Journal of the American Oriental Society. Jap. = Japanese. JAs. = JA.(Jä.,) Jäschke = Jäschke, Tibetan-English Dictionary. Jāt. = Jātaka (Pali). JB. = Jaiminīya Brāhmaņa. JM. = Jaina Māhārāstrī. JPTS = Journal of the Pali Text Society. JR (Vikramacarita) = Jainistic Recension of the Vikr., see HOS 26-27. JRAS = Journal of the Royal Asiatic Society. JŚ. = Jaina Śauraseni. Karmadh. = Karmadhāraya. Kāś. = Kāśikā (comm. on Pāņ.). Kauś. = Kauśika Sūtra. Kaut. (Arth.) = Kautiliya (Arthasästra). Kh.p., Khud.p. (comm.) = Khuddaka-pātha (commentary) (Pali). Kirfel, Kosm. = Kosmographic der Inder. KSA = Kāthaka Samhitā, Aśvamedhagrantha. KSS = Kathāsaritsāgara. KŚS = Kātyāyana Śrauta Sūtra. Kull. = Kullūka. Kum. = Kumārapālapratibodha (Ap.), ed. Alsdorf. Kvu. = Kathā-vatthu (Pali). Lalou, Iconographie (des étoffes peintes dans le Mañjuśrimūlakalpa), Paris, 1930. E. Lamotte, Mél. chin. et boud. = Mélanges chinoises et bouddhiques. Lang. = Language (Journal of the Linguistic Society of America). Lāty. = Lātyāyana Śrauta Sútra. l. c. = in the place cited. lect. diff. = lectio difficilior. lect. fac. = lectio facilior. lex. = lexicon, lexical; see also Bbh, in the preceding list of abbreviations. Lin Li-kouang, (L')Aide-Mémoire (de la Vraie Loi), Paris, 1949. lit. = literal(ly); also, literary, found in literature, as opposed to lex. loc. = locative.

M. = Mähäräştri.

m. = masculine.

Mahāv. - Mahāvamsa (Pali).

Mark. Pur. = Markandeva Purana.

masc. = masculine.

math. = mathematics.

Mbh. = Mahābhārata.

- m.c. = metri causa. I include under this term two classes of cases, because in both I find evidence for believing that metrical considerations were either the exclusive or the dominant reason for the use of the form in question. The first includes any cases of a particular form which is never used in prose, or in positions in a verse where the alternative ('regular') form would fit as well. The second includes cases in which the form labelled 'm.c.' is indeed also used (elsewhere) in prose or metrically indifferent positions, but in such limited or specialized ways (which vary in different cases) that I still believe the choice was determined. in the cases under consideration where the term 'm.c.' occurs, by meter. In both types, of course, the alternative ('regular') form would be impossible in a specific context where it is labelled 'm.c.' It is hoped that my manner of presentation will make it easy to distinguish the two types. Frequency of occurrence, and use of corresponding forms in other MIndic dialects, are important considerations to which I have tried to give due weight.
- Mehendale (M.A.), Hist(orical) Gr(ammar) of Inscr(iptional) Pkts., Poona, 1948.

metr. = metrical(ly).

Meyer, Kaut. (Arth.) = J. J. Meyer's translation of Kaut., Leipzig, 1926.

mg. = meaning.

- Mg. = Māgadhī.
- Miln.(p.), Milp. = Milindapañha (Pali). MIndic or MInd. = Middle Indic.
- misc. = miscellaneous.
- MN = Majjhimanikāya (Pali).
- Mrcch. = Mrcchakatikā.
- ms(s). = manuscript(s).
- MS = Maitrāyaņī Samhitā.
- MSL = Mémoires de la Société de Linguistique (de Paris).
- Mus (P.), Barabudur (1935).
- mus. = musical.
- MW = Monier Williams, Sanskrit-English Dictionary (2d ed., 1899).

n. = nominative; name.

n. pr. = proper name.

n. act. = noun of action.

n. ag. = noun of agent.

Nachtr., see Schmidt.

- NAWGött. = Nachrichten der Akademie der Wissenschaften zu Göttingen.
- neg. = negative.
- Neisser, Z. Wbch, d. RV .: W. Neisser, Zum Wörterbuch des Rigveda.

Nep. = Nepalese.

Nett. = Nettipakarana (Pali).

NIA = New Indian Antiquary.

Nidd¹ = Mahā-Niddesa (Pali).

Nir. = Yāska's Nirukta.

nom. = nominative.

nom. act. = noun of action.

nom. ag. = noun of agent.

nt. = neuter.

obj. = object.

obl. = oblique.

onomat. = onomatopoetic.

opp. \approx opposite (of).

opt. = optative.

orig. = original(ly).

- Pān. = Pānini.
- Pañc. Rec. = Edgerton, Pañchatantra Reconstructed.

pass. = passive.

- Pat. = Patañjali. PBr. = Pañcavimáa Brâhmana.
- pej. = pejorative.

perh. = perhaps.

- pers. = person.
- Philologica Indica (reprints of monographs by H. Lüders), Göttingen, 1940.
- Pischel (without further reference) == Grammatik der Prakrit-Sprachen, Strassburg, 1900. Cited as under Geiger.

Pkt. = Prakrit.

pl. = plural.

- postp., postpos. = postposition.
- ppp. = past passive participle.
- prec. = precative; preceding; preceded.
- pres. = present.
- prob. = probably.
- Proc(eedings of the) Am(erican) Philos(ophical) Soc(iety). Przyluski, Aśoka = La Légende de l'Empereur Acoka,
- Paris. 1923.

PTSD = Pali Text Society's Pali-English Dictionary.

- Pug(g). = Puggala Paññatti (Pali).
- Pv. = Petavatthu (Pali).
- pw = Boehtlingk, Sanskrit Wörterbuch in kürzerer Fassung.

qy = query.

- Rājat(ar). = Rājataramgini (ed. Stein).
- Rām. = Rāmāyana (NWRām. = Northwestern Rām.) Ratnach. = Ratnachandra, Ardha-Māgadhi Dictionary (5 vols.).

rec. = recension.

- ref. = reference(s).
- Renou (Louis), without specific reference, or Gr. sctc.. = Grammaire sanscrite, Paris, 1930.
- Renou (Louis), Études de grammaire Sanskrite, Paris, 1936. RO = Rocznik Orjentalistyczny.

- RV = Rig Veda.S (as in Wh. Roots) = Sūtra(s).
- Ś. = Śauraseni.
- San., Sanatk. = Sanatkumāracaritam (Ap.), ed. Jacobi (1921).
- Saundarān. = Saundarānanda, ed. Johnston.
- SB. = Satapatha Brāhmaņa (Mādhy. = Mādhyamdina, K. = Kāņva recension).
- SBBA = Sitzungsberichte der Berliner Akademie der Wissenschaften.
- SBE = Sacred Books of the East.

sc. = scilicet.

- Schmidt, Nachtr. = R. Schmidt, Nachträge zum Sanskrit Wörterbuch ... von Otto Boehtlingk.
- Sdhp. = Saddhammopāyana (Pali).
- Sen = Sukumar Sen, An outline syntax of Buddhistic Sanskrit, in Journal of the Department of Letters, Univ. of Calcutta, (XVII) (1928).

sg. = singular.

- Sheth = H.D.T. Sheth, Paia-sadda-mahannavo (Pkt.-Hindi Dict.), Calcutta, 1923-1928.
- Skt. = Sanskrit; (hyper-)Sktism, Sktization = Sanskri-tism, Sanskritization.

Skt. Synt. = Speyer, Sanskrit Syntax.

Smith, Helmer, Les deux prosodies du vers bouddhique (see § 1.38, fn. 15).

Sn. = Suttanipăta (Pali).

SN. = Samyuttanikāya (Pali).

Speyer: see Skt. Synt., and VSS; also, in the preceding list, s.yv. Av and Jm.

subst. = substantive.

superl. = superlative.

XXX

Sūyag. = Sūyagadangasutta (AMg.).

s, v(v) = under the word(s).

syn. = synonym(ous).

Tagare (G.V.), Hist(orical) Gr(ammar) of Ap(abhramsa), Poona, 1948.

Tatp(ur). = Tatpuruşa.

- TB. Taittiriva Brähmana.
- Therag. = Theragatha (Pali).
- Therig. = Therigatha (Pali).
- Thig. = Therig.
- Tib. = Tibetan.
- Ting (Fu-Pao), Dictionary of Buddhlst Chinese, 16 vols., Shanghai, 1920. In Chinese; a few references provided by Mr. Chang Kun.
- Toch. = Tocharian (: Gram., Tocharische Grammatik, Sieg and Siegling).
- trans. = transitive.

transl. = translation(s).

- Trik. = Trikāņdašesa (a Skt. lexicographer cited in BR; contains many BHS words).
- TS. = Taittiriya Samhitä.
- Turner (R.L.), Nep. Dict. = Dictionary of the Nepali Language, London, 1931. U (as in Wh. Roots) = Upanişads.
- Uhlenbeck (C.C.) = Kurzgefasstes etymologisches Wörterbuch der altindischen Sprache (Amsterdam, 1898-99).
- uncpd. = uncompounded.
- unmetr. = unmetrical(ly).

var(r). = variant(s).

- Varāh.Brh.S. Varāhamihira, Brhat-samhitā.
- vb. = verb.
- Ved. = Vedic.
- Ved. St. = Pischel and Geldner, Vedische Studien (3 vols.).
- Ved. Var., Vedic Variants, see VV.
- Versl. en Med. d. K. Akad. v. Wet. Amsterdam = Proceedings of the Amsterdam Academy.
- Vin. = Vinayapițaka (Pali).
- Vism., Vis.M. = Visuddhi-magga (Pali).
- v.l., vv.ll. = variant reading(s).
- voc. = vocative.

- VS. == Vājasanevi Samhitā.
- vs, vss = verse, verses. VSS = Speyer, Vedische und Sanskrit-Syntax.
- VV = Bloomfield, Edgerton, and Emeneau, Vedic Vari-ants, 3 vols., Philadelphia, 1930, 1932, 1934.
- Vv. = Vimānavatthu (Pali).
- Wack(ernagel), AIGr. = Altindische Grammatik, Göttingen, 1896, 1905, 1930.
- Warren (H. C.), B(uddhism) in Transl(ations) (= HOS 3; 1896).
- Wh. = Whitney; Gr., Sanskrit Grammar; Roots (Verbforms and primary derivatives). Wilson = H. H. Wilson's Sanskrit Dictionary, as cited in
- MW.
- Winternitz, H(ist). I(nd). Lit. = W's History of Indian Literature (English version; vols. 1 and 2, Calcutta, 1927 and 1933).
- w.r. = wrong reading.
- Yājñ. = Yājňavalkya (Dharmaśāstra).
- ZDMG = Zeitschrift der deutschen morgenländischen Gesellschaft.
- ZII = Zeitschrift für Indologie und Iranistik.
- Z.Wbch. d. RV., see Neisser.

Symbols

- : (as in statements of proportional analogy) = 'is related to'.
- () Parentheses, in the Dictionary, enclose articles on words which occur in standard Skt. in substantially the same meanings as in BHS, but which by exception, for one reason or another, seem to need treatment here. The reason is usually a (previous) erroneous interpretation, or sometimes what is felt to be a danger thereof.
- [] Brackets, in the Dictionary, enclose articles on what are considered erroneous readings. They are, however, also used in the interior of an article, in lieu of parentheses which, if used, would fall within other parentheses.

1. Introduction

Languages used in early Buddhism

1.1. Buddhist religious texts have been preserved in at least four Indic languages.

1.2. One is standard Sanskrit, used for example by the poet Aśvaghoşa (attributed to the 2d century A.D.).¹

1.3. Two of the others are Middle Indic. The better known, which contains the largest Buddhist literature now preserved in any Indic language, is Pali, the sacred language of Southern Buddhism. Most scholars now believe that it is, in the main, based on a western, or west-central, Middle Indic vernacular. The other is the dialect of the 'Prakrit Dharmapada' (ms. Dutreuil de Rhins) published by Senart in JAs. IX.12 (1898), pp. 193 ff. It was based on a northwestern Middle Indic.² There is no need for me to discuss these languages further at this point. Especially Pali will be very often mentioned, for comparative purposes, thruout my Grammar and Dictionary.

1.4. There remains the subject of this work, which I call Buddhist Hybrid Sanskrit (abbreviated BHS). Most North Indian Buddhist texts are composed in it. It is based primarily on an old Middle Indic vernacular not otherwise identifiable. But there seems reason to believe that it contains features which were borrowed (originally, or in the course of historical development, or both) from other Middle Indic dialects. In other words, even its Middle Indic aspects are dialectically somewhat mixed. (For that matter, we shall soon see that the same could be said of Pali, and probably of all other Middle Indic dialects of which we have any considerable knowledge.) Most strikingly, however, BHS was also extensively influenced by Sanskrit, from the very beginning of the tradition as it has been transmitted to us, and increasingly as time went on.

1. 'His Sanskrit is faultless, even though it does not always comply strictly with the rules of Pāņini' (Winternitz, Hist. Ind. Lit., vol. 2, 1933, p. 260). Very many brahmanical 'Sanskrit' works likewise fail to 'comply strictly' with these rules.

2. J. Bloch, JAs. 1912, I, p. 331 ff. This text, with a translation of Senart's French notes, and new textual and exegetic notes, was reprinted by B. Barua and S. Mitra, Calcutta, 1921; a new edition of the greater part of the text, with an index of words and bibliography, by H.W. Bailey, BSOS 11.488-512. I know of no evidence that Buddhist religious texts (such as might have formed parts of a canon, or have been ancillary to a canon) have been preserved to us in any other Middle Indic language. Asoka's inscriptions, and most other Buddhist inscriptions, mainly memorial and votive, from the celebrated Piprawa vase (most recently treated by Lin Li-kouang, Aide-Mémoire 227; see note 6 below) onward, are not 'religious texts' in my sense. (See §§ 1.18 ff. for their bearing on dialects used in Buddhism.) The finds from Central Asia (Chinese Turkestan) have revealed such Buddhist texts only in Buddhist Hybrid Sanskrit, and to some extent in standard Sanskrit (to ignore non-Indic languages). The 'Niya' inscriptions (Stein, Ancient Khotan, pp. 321, 364-8), published by Boyer, Rapson, Senart, and Noble, 3 vols., Oxford, 1920, 1927, 1929, seem to contain only two fragMany (especially later) products of this tradition have often, the I think misleadingly, been called simply 'Sanskrit', without qualification.

1.5. There have been attempts, as will be seen later, to identify the Middle Indic dialect on which this language was originally based. This question has also been somewhat blended, or confused, explicitly or implicitly, with the question of what language was used in what is presumed to have been the oldest, or 'original', form of the canonical literature of the Buddhists. Admittedly, no such 'original' canon is preserved to us.

1.6. The nature of the BHS language, and the question of an 'original' canonical language of Buddhism, can be understood best against the background of the attitude of early Buddhism towards the use of various languages as vehicles for its religious teaching. Fortunately that attitude can be quite clearly defined.

1.7. In the Pali canon (Cullavagga 5.33; Vin. ii.139.1ff.) occurs a famous passage which, in abbreviated form, may be rendered thus:

Two monks, brothers, brahmaus by birth, of fine language and fine speech,³ came to the Buddha and said: Lord, here monks of miscellaneous origin (literally, of various names, clan-names, races or castes, and families) are corrupting (*dūsenti*) the Buddha's words by (repeating them in) their own dialects; let us put them into Vedic.⁴ The Lord Buddha rebuked them: Deluded men, how can you say this? This will not lead to the conversion of the unconverted... And he delivered a sermon and commanded (all) the monks: You are not to put the Buddha's words into Vedic. Who does so would commit a sin. I authorize you, monks, to learn the Buddha's words each in his own dialect.

ments of Buddhist religious texts (Nos. 510, 511), both in Buddhist Hybrid Sanskrit, besides one set of niti verses (which have no obvious distinctively Buddhist traits) in what is meant for standard Sanskrit (No. 523). The rest, in what has been called 'Niya Prakrit', seem to be purely secular. According to Konow, BSOS. 8.611, this and the dialect of the ms. Dutreuil de Rhins are 'closely connected forms of one and the same ancient dialect'.

3. kalyāṇavācā seems to mean that their language was cultivated and elegant; kalyāṇavākkaraṇā, that they were eloquent and mellifluous. speakers (comm. madhurasaddā; similary AN comm. ii.40.15 kalyāṇavākkaraṇatā li vacanakiriyāya madhurabhāvo, 'mellifluence in the use of the voice, or of words').

4. chandaso āropema. Not, as some have thought, "into verses", Sanskrit or other; the context makes it quite clear that chandaso means texts in a certain language or dialect; and more precisely in the Vedic language (whether prose or verse). This is the regular meaning of chandas in Pāṇini, and is familiar in Epic and Classical Sanskrit (BR s.v. 3). To these brahmans by birth, the proper language for sacred texts was Vedic (which included the Brāhmana and Upsanişad dialects). Comm., vedam viya, sakkalabhāsāya (not 'Sanskrit'! but 'respectable, elegant') vācanāmaggam āropema. 1.8. The Pali commentator Buddhaghosa takes the last clause to mean rather 'in (the Buddha's) own dialect'; and some moderns (notably Geiger p. 5) follow him.⁵ In my opinion, however, as in that of most scholars, there can be no real doubt of the above interpretation; and it is made doubly sure by several Chinese versions of the same incident.⁶ which never speak of 'the Buddha's dialect'. Some of them specifically authorize use of 'the ordinary languages of each country' or the like. This is clearly a very old tradition, supported by closely parallel texts in both the northern and southern canons. It justifies us in assuming that the Buddha himself and his earliest disciples used only popular dialects as vehicles for their teachings, and refused to employ the venerated 'Vedie' language of the brahmans.

1.9. Taken together, the various reports of this incident seem to me to indicate more than that, and more than has usually been recognized. Let us look at some Chinese translations of lost Indic versions, as given by Lin Likouang (note 6).

1.10. The Vinaya of the Mahišāsaka sect, translated by Buddhajiva in 423-424 A.D., reports that the two originally brahman brothers heard monks reciting the sūtras 'incorrectly'. They ridiculed them, saving: 'Tho they have long since become monks, they recite the sūtras in *this* fashion! not knowing masculine and feminine gender, nor singular and plural, nor present, past, and future, nor long and short sounds (vowels), nor (metrically) light and heavy sounds (syllables).' [A student of BHS might be forgiven for thinking that the writer had this very dialect in mind! But most if not all of these 'errors' can be found in most Middle Indic dialects.] When they appealed to the Buddha, he ordered that the texts be recited 'according to the sounds of the regions, but taking care not to distort the meaning. It is forbidden to make of the Buddha's words an "outside" (non-Buddhist, heretical) language.'

1.11. The Vinaya of the Dharmagupta sect is closer to the Pali, but speaks of only one originally brahman monk, not two. He complained to the Buddha that 'monks of different clans and bearing different names were ruining the sûtras', and proposed to 'arrange them according to the good language of the world', that is, no doubt, Vedic or Sanskrit, the language of culture. In his rebuke the Buddha said that it would ruin the sûtras to use 'the language of heretics', and that 'it is allowed to recite and learn the Buddha's sûtras according to the interpretation of the popular languages of (various) regions.'

1.12. Even more striking for my purpose is the Chinese rendering (between 350 and 431 A.D.) of the lost Vinayamätrkä ('Summary of the Discipline'). Here, the two monks of brahmanic origin say to the Buddha: 'Among the disciples there are men of different clans, of different countries, of different prefectures and sub-prefectures. Their pronuncialions (lit. sounds of language) are not the same. As their languages are not correct, they all corrupt the true meaning of the Buddha.' They wish to compile and regulate the sūtras according to the 'Chando-(vi-)citi-śāstra'. The Buddha replies: 'In my religion, fine language is not recognized. All I want is that the meaning and reasoning be correct. You are to preach according to a pronunciation (lit. sound) which people can understand. Therefore it is proper to behave (sc. in the use of language) according to the countries.'

1.13. There are other, less close reflexes of this old

5. Buddhaghosa says 'the Māgadha language, spoken by the Buddha', which Geiger thinks means real Māgadhi, while Lin (see n. 6), p. 226, supposes that he referred to Pali. It makes little difference; Buddhaghosa was surely mistaken in any case. passage, but they seem to add nothing important for my present purpose. The net result seems to me to show, not only that the Buddha authorized and commanded the recitation of his teachings in local dialects everywhere; but also that from the very start there were clearly perceptible differences in the speech of different members of the Buddhist order, which showed in their ways of reciting the sacred texts. Their pronunciations are not the same,' says the last quoted text; and if no other has been found recording this fact quite so clearly, others, including the Pali, emphasize the miscellaneous origins of the monks and the fact that the Buddha ordained that everyone, monks included, should use his own particular dialect in reciting the sacred texts.

An toriginal language of Buddhism'?

1.14. In what sense, then, is it possible to assume one 'original language' of Buddhism?

1.15. The Buddha himself was an 'casterner'; his family lived at Kapilavastu, in northeastern Kosala (Oudh); his wanderings seem to have been chiefly bounded on the west by Śrāvasti (also in Kosala, tho considerably to the west of Kapilavastu) and on the cast by Rājagfha, the capital of Magadha (Bihar south of the Ganges). All this region belongs linguistically to what is now called modern Bihari (except that Śrāvasti may perhaps be just over the line in Eastern Hindi). Doubtless most of his disciples belonged to the same general region, and we may assume that, during the Buddha's lifetime, the Buddhist texts were mainly, at least, recited in castern dialects. Yet no one knows just what dialect the Buddha spoke; and it seems clear that the dialects of his disciples differed perceptibly.

1.16. According to Buddhist tradition, shortly after the Master's death his chief followers met and tried to establish what we may call an approach to a canon, based on recitations of texts which they remembered. These, and in time other texts, were memorized and handed down by oral tradition, doubtless for centuries before writing came to be used for such purjoses. This 'canon', at first vague and fluid, surely included narrations of the main events of the Buddha's life, and many sayings and sermons, in both verses and prose, which were attributed to him and set forth his principal doctrines; as well as rules for the life of his monkish followers. Other councils, one under Asoka, were convened, which discussed the form and content of the canon.

1.17. Even from the beginning of this tradition, different dialects were used in these recitations, varying with the speech-habits of different monkish reciters. We have seen that such apparently was the custom during the Buddha's lifetime, and that he approved it and rejected any standardization. And as Buddhism spread, the texts certainly tended to become adapted in language to the vernaculars of regions to which they were brought by missionaries. This, too, was in accord with the Buddha's own instructions. In early days the adaptation was easy, and did not necessarily involve what we should dignify by the name 'translation'. Northern and central Indian dialects are predominantly Indo-Aryan, historically related to Sanskrit, even to this day. In late pre-Christian centuries the ancestors of these modern dialects were in the Middle Indic stage. They were still quite close to each other in

6. First recorded (as far as I know) by S. Lévi, JAs. 1915, I, 441 ff., and more recently discussed at length by Lin Li-kouang, L'aide-mémoire de la vraie loi (Paris, 1949), 218 ff. sounds and grammatical forms, and not too distant from their common mother tongue, which we may call 'Sanskrit', using that word in a way which Sanskritists will recognize as somewhat untechnical. So, the Buddha's injunction to make use of local dialects could be carried out with little conscious effort, in many cases. If not the missionary, at least his converts, in repeating the memorized lessons, would make the phonetic and morphological changes called for by the local vernacular, pretty much automatically, altho—N.B.1--we should not be surprised if we find a lack of complete consistency. In places which became thriving centers of Buddhist propaganda, there would in time tend to develop an increasingly crystallized canon, in a language based primarily on the local dialect.

1.18. This is not mere speculation. There is much evidence suggesting the one-time existence of Buddhist texts in a number of different ancient Middle Indic dialects, even tho such texts are actually preserved only in two or three. In an important article of 1912,' Sylvain Lévi showed this on the basis of essentially two types of evidence: first, aberrant (dialectically foreign) forms in Pali, and forms which seem aberrant in BHS (where lack of a formulated standard has made it harder to be sure what to call 'aberrant'); and secondly, proper names and other words occurring in titles and scraps of phraseology in Asoka's inscriptions, and in other pre-Christian inscriptions like those of Bharhut, which allude to Buddhist literature.

1.19. Illustrations of the first kind of evidence could be multiplied. By way of further illustration of Lévi's point, I may note BHS khāyati, Pali ppp. khāyita-, for Skt. (and normal Pali and BHS) khād- 'eat'; and Prakritic causatives in -āveti (for regular -āpeti or -āpayati), sporadically in Pali and BHS (\S 38.68 ff.). But since most Prakrits show loss of most intervocalic stops, and change of intervocalic p to v, these and the analogous cases cited by Lévi certainly prove no 'eastern' influence.

1.20. Some of the words cited by Lévi from early inscriptions lend more support to his conclusion (p. 511) that the (Buddhist Hybrid) Sanskrit and Pali canons are 'the late heirs of an older tradition, recited or redacted in a dialect now lost' which (if I understand Lévi correctly) was 'one of the languages of the land of Magadha.'⁸

1.21. I recognize the interest and value of Lévi's linguistic comparisons drawn from early inscriptions, and admit that some of them point towards the east, as indeed we should expect. Yet the most striking impression they make on me is their great variety. They most emphatically suggest lack of any dialectic agreement.

1.22. Take Läghula = Rähula[•]; l for r does indeed agree with Māgadhī, but gh for h is not normal to any Prakrit; it seems to be a hyper-Sanskritism. So, cy for ty In adhigicya is wholly isolated in Middle Indic. Lévi makes much of the ya-śruti in avayesi (p. 497) = Pali avādesi; but neither Mg. nor AMg. nor any Prakrit recorded by Pischel (187) has the ya-śruti before e; it is, to be sure, perhaps found in the old fragments of Buddhist dramas published by Lüders, Kl. Skt. Texte I, in the dialect which Lüders considers 'old AMg.' (bhumjitaye, inf., p. 39; but the origin of the ending is not entirely clear, cf. Lüders, p. 40 with n. 3). Most of Lévi's examples are widely, if not

7. JAs. 1912, II, 495-512, Observations sur une langue précanonique du Bouddhisme.

8. Surely this is geographically too narrow, in any case. Neither the Buddha's original home (Kapilavastu), nor Śrāvastī, one of his favorite dwelling places, nor Benares, where he preached his first sermon and stayed at other times, nor Kušinagari (Pali Kusinārā), the place of his death, was in Magadha.

9. Läghuloväde musävädam adhigicya, in the Asokan

almost universally, found in Frakrits of all regions; or if not that, they are largely isolated, without clear parallels anywhere, like the forms just noted. In either case they do not help in defining geographically the dialect where they originated.

1.23. Lüders (op. cit. p. 41) believed that an 'old form' of Ardha-Māgadhī was the Buddha's native dialect, and also the language of the Buddhist canon in its oldest form. But he offered, in this place at least, no real linguistic evidence; only an expression of the opinion that 'Ardhamāgadhisms in the Pali canon, which in verses were not so easy to delete in rendering into Pali, are probably much more numerous than has heretofore been assumed.'

1.24. Hiän-lin Dschi also believes that 'Old Ardha-Mägadhi' was the 'original' language of the Buddhlst canon (NAWGött. ph.-hist. Kl. 1944, 136, and especially ib. 1949, 269 ff.). He advances five arguments based on allegedly 'eastern' forms in BHS, specifically in Mv, viz.:

1.25. First, voc. pl. in *-aho*. This is found not only in Mg. (note: not in AMg.!); but also, as *-aho*, *-ahu*, in Apabhramśa (all varieties, and specifically in Western Ap., in all periods; cf. BSOS 8.512; Tagare, Hist. Gr. of Ap., 138 ff.). The short a is, of course, obviously a secondary shortening in Ap. This disproves the exclusively 'eastern' character of the ending.

1.26. Second, 'meistens ist vor eva ein y eingeschoben.' The form yeva after vowels, and sometimes after anusvāra, is extremely common in Pali, where it has every appearance of being a native form. It is much less common than eva, even after vowels, in Mv, and I have noted it in no other BHS work. Since it agrees with Pali, it is certainly no proof of 'eastern' connexions; and AMg. is reported to lack it (unless, perhaps, in the altered form *je*, Pischel 336), tho it is found in Mg.

1.27. Third, the alleged 1st (and 3d) sg. opt. ending -eham (-eha). On this see §§ 31.21, 22, where I discuss all the cases which can with any shadow of plausibility be assumed as verb forms. Dschi does not cite all those treated by me, but cites in addition dadeha My iii.46.14 (vs) and upanayeham 82.14 (vs); these are certainly to be read dade 'ha (= aham; denasalization m.c.) and upanaye 'ham (for which, incidentally, the mss. read upanaye; ham is Senart's emendation, tho it may well be right). Many of the others are textually doubtful. Some certainly contain the pronoun (a)ham. If the ending -eha(m) has any real existence, which seems to me somewhat questionable, I believe it should be regarded as a future, not an optative. The Asokan forms seem more likely to be real than any in My; they are not exclusively found in the eastern versions, tho the western occurrences are customarily explained by the facile assumption of 'Māgadhism'. In summary: it is doubtful whether BHS has a verbal ending -eha(m) at all; if it does exist, it is probably a future, not an optative; and there is no good reason for regarding it as 'eastern'.

1.28. Fourth, $-\bar{a}vo$ for $-\bar{a}yo$ (§§ 1.84; 9.93). Dschi's note 7 on p. 271 fails to convince me that the replacement of y by v should be regarded as an 'easternism'. Most of the cases he cites, like the 3 pl: opt. ending *-vu* for *-yuh*, are not good parallels to $\bar{a}vo$ for $\bar{a}yo$, since the following

edict called by Lévi that of Bhabra, now known as that of Calcutta-Bairãt, Hultzsch, Inscr. of Asoka, xxv, or of Bairãt Bhābrû, Mehendale, Hist. Gram. of Inscriptional Prakrits, Poona, 1948, ix: 'the admonition to Rāhula concerning falsehood' (referring to the text known in Pall as Sutta 61 of MN). Lāghula for Rāhula is quite unparalleled.elsewhere; and adhigicya for Skt. adhikriya shows an apparently unique cy for iy (no other case in Mehendale). vowel u may be suspected of responsibility for the change of y to v, a change which by the way is familiar in Pali (e.g. $\bar{a}nuso$) and elsewhere. But $-\bar{a}vo$ for $-\bar{a}yo$ can hardly be due to assimilation to the following o. Dschi himself notes that interchange between y and v is widespread, if sporadic, from very ancient times. Cf. § 2.31; Vedic Variants II §§ 246 ff.; note the preference of the Taittiriya schoot for v as against y of other texts. There is no good reason for assigning the change of y to v to any one region. Such a change before a following u-vowel should be distinguished from that before other vowels.

1.29. Fifth, the form holi for bhavali, which Dschi, following Michelson, calls a 'Mågadhism'. Michelson's only reason was that while holi occurs in all Asokan dialects, bhoti and bhavati occur only in western Asokan. The form holi (note that it is much rarer than bholi in BHS! Chap. 43, s.v. bhū 2) is, of course, extremely common in Pali, and its equivalents are equally common in most Prakrits. It is general Middle Indic, and is obviously the form originally proper to enclitic use after vowels; intervocalic bh became h, e.g. in the instr. plur. ending -hi from -bhi(s) in all Asokan dialects. The other form, bhavati or bholi, persisted by the side of hoti in most, perhaps all, MIndic dialects; its popularity in some late forms of Pkt. may be due, in part, to Skt. influence. But it certainly persisted in Magadhi and AMg. long after the time of Asoka. According to Pischel 475 bhavai etc. is common in AMg.; the opt. bhave (1 sg. also bhaveam) is the only opt. known in Mg., which also knows a present bhavāmi and other forms based on the stem bhava-. In view of all this, how is it possible to take hoti as a 'Māgadhism', simply because, in the Asokan inscriptions, bh- forms happen to occur only in the west (along with h-forms, be it noted)? The evidence of later Mg. and AMg. shows that the absence of bh-forms in the few eastern Asokan occurrences can only be accidental.

1.30. To me, the opinion of Lin Li-kouang (Aidemémoire 227-8) seems much sounder. He asks:

Is it probable, however, that the Buddhists ever used one sole language, even in the oldest times? Considering the pliant and liberal spirit which they always showed in regard to the use of languages, we may well ask whether there did not exist, from very early times [I would say, 'from the lifetime of the Buddha', as shown above], a multiplicity of Buddhist languages, Mågadhi being only one of them... Why not assume that other centers of ancient Buddhism, Vaiśāli, Kauśāmbī, Mathurā, Ujjayinī, and others, also had their own special dialects, which served as sacred

10. In later times (see Lin Li-kouang, op. cit., 176-187; 194-216) Tibetan traditions attribute to various Buddhist schools the use of 'Sanskrit' (which means BHS as Lin shows), and other Indic dialects (apparently meaning Prakrit, Apabhramśa, and Patśāci). The historic meaning and value of these statements is doubtful, and in any case they are not directly relevant here. The Chinese traditions of Buddhist schools (op. cit. 188-194) seem to say nothing about their languages. (Corrections to Lin: p. 169, last line, tajjakriyā KP 105.8 = tad-ja°, not for Skt. tarja°; p. 199, prayoya Bbh 19.1 is certainly a mere misprint for prayoga.)

11. The ending e, instead of o (Skt. as) and am, is now known to occur in other than eastern MIndic. First, extensively in the far west and northwest: in the Kharosthi Pkt. inscriptions (Konow, CII Vol. II pt. 1, see esp. cxii; Mehendale 314 n. 28: in certain districts o for masc. a-stems, e for nt.; west of the Indus, for the most part, e thruout); see also Burrow, Kharosthi Documents, § 12. Secondly, in the Prakrit inscriptions of Ceylon (Mehendale 134, note 61; Ep. Zeyl. I, 20.7, 62.2; lene 'cave', languages for the communities established in these centers, according to the principle which the canonical texts expressly set down that Buddhist communities should adopt the local dialects?

1.31. Only this theory seems to me to account reasonably for the facts. The languages we know as Pali, BHS, and the Prakrit of the ms. Dutreuil de Rhins, all originated in such local dialects where Buddhist communities were settled. At least these three developed something approaching 'canons'. Many more such local canons may well have existed. A dim reflection of this condition may be seen in the early inscriptions treated by Lévi, l. c_s^{10}

1.32. Since in all such cases the 'canon' was originally brought by missionaries from outside, we should expect to find traces of 'foreign' dialects, in other words some dialect mixture, in any extensive canonical collection, So we do, in Pali and BHS. Since according to tradition the Buddha and many of his leading disciples were easterners, it is not strange that some sporadic forms in Pali seem, or have been assumed, to resemble Magadhi, or Ardha-Mägadhi; but I think their importance has been exaggerated. There are a goodly number of Pali substitutions of l for older r, which is regular in Mg. but also found, at least occasionally, in many other dialects (Pischel 256-7), and even in Sanskrit. (The BHS occurrences of l for r are balanced by a substantial number of r for l. Standard Skt. shows this shift in both directions.) Otherwise there is in Pali the notorious but only very sporadic substitution of e for o (Skt. as; by morphological extension also for am). Similar, and equally sporadic, forms occur in BHS.11 Other 'Prakritisms' in Pali, to the best of my knowledge, cannot be localized as to origin.

Buddhist Hybrid Sanskrit

1.33. As we saw, the most striking pecularity of this language is that from the very beginning of its tradition as we know it (that is, according to the mss. we have), and increasingly as time went on, it was modified in the direction of standard Sanskrit, while still retaining evidences of its Middle Indic origin. In all its texts, even the oldest, at least as shown by our manuscripts and editions, Sanskritisms are constantly presented cheek by jowl with Middle Indic forms, and often with hybrids which strictly are neither one nor the other. These Sanskritisms are much too common to be comparable with stray Sanskrit loanwords or loan-forms which may have been occasionally adopted in many a genuine Middle Indic vernacular.¹⁸

passim, = Skt. layanam, Pali and AMg. lenam); here too the evidence seems to indicate that e was the regular ending of masc. and nt. a-stems, n. and acc. sg. Furthermore, e for these same forms occurs in early Pkt. inscriptions found in every part of India, including west (Karle, Nasik: Mehendale 87-88), center (Meh. 172-173, e.g. Sanchi); and south (Meh. 135). In most of these regions, to be sure, e is not the favored or standard form; but this can be said just as truly of Pali or BHS. As in these languages, the inscriptional forms in e are no doubt largely due to dialect mixture. In the case of Pali and BHS, it may be that they are indeed borrowed from an eastern dialect. But the e-forms are so widespread in inscriptions that caution seems called for. They are not distinctive. As far as Pali is concerned, I do not know that it shows much, if anything, that is truly distinctive in common with Mg. or AMg. And this is certainly true of BHS.

12. A few examples (hundreds could easily be cited) of close juxtaposition of Skt. and MIndic forms in §§ 8.108,110. —Pali contains such borrowed Sanskrit words and forms, e. g. vākya, āroqya, kva (Geiger 53.3), and brāhmaņa (and 1.34. Sinologists date Chinese translations of some BHIS works as early as the 2d century A.D.; and these are not the earliest works in BHS, which must apparently be pre-Christian, by perhaps more than one century.¹³ It seems, then, that in quite early times some north-Indian Buddhists abandoned their original principle of using genuine vernaculars, and partially yielded to the prestige of the classical and learned language of their brahman neighbors. Yet they made no effort to 'translate' into Sanskrit. BHS works, especially the oldest, retain in all parts clear evidences of being based on some form of Middle Indic, only partially, and it seems haphazardly, Sanskritized.

1.35. This mixture can, in my opinion, never have been spoken as a real vernacular. Yet it existed for centuries as a religious language, and seems to have become the prevalent language used by north-Indian Buddhists generally for religious purposes. At least, little else is preserved to us except some works in normal standard Sanskrit.

1.36. The extent of Sanskritization varies greatly in different periods, and even in different parts of some of the same works. Nearly all BHS works are composed in a mixture of prose and verse. In Mv (the Mahāvastu, see note 13), probably the earliest, the Sanskritization is relatively slight and imperfect, and all parts, both prose and verses, are affected by it to about the same extent.

1.37. In many other BHS texts (those of my second class, see Bibliography), c. g. in SP, LV, Gv, Suv, Samādh, the verses are presented in a relatively Middle Indic form, linguistically similar to the Mv; but the prose is far more

its relatives); this last word is certainly a Skt. loan, as shown partly by the initial br- (which is not conclusive), but especially by the \bar{a} before a consonant cluster and the lack of metathesis in hm (cf. Geiger 49.1). Notorious is the Pali gerund suffix -(*i)lvā*, which must be a Skt. loanform. Even Asokan inscriptions, with the single exception of - $lp\bar{a}$ in the west (Girnar), show only -lu, and the only other record of - $lv\bar{a}$ elsewhere in Middle Indic (unless BHS be counted as such) seems to be the dialect of the 'Pkt. Dharmapada' (ms. Dutreuil de Rhins; e. g. ñatva, Senart p. 218; hitva p. 219). But the number of such words and forms is so limited in Pali that it is far from constituting a parallel to BHS. They are no more significant, as dialect mixture, than the so-called Māgadhisms of Pali, mentioned above.

13. According to Winternitz, Hist. Ind. Lit. II (1933). 247, the 'nucleus' of the Mahāvastu, which is commonly and I think rightly regarded as the oldest BHS work we have, 'originated as far back as the 2nd century B.C.', tho it was expanded later, some additions being as late as the 4th century A.D. and perhaps later yet. On stratification in Mv see references in fu. 21, to § 1.81. On classification, in part chronological, of BHS texts, see Bibliography.

14. No other fully preserved work is comparable with Mv in presenting its prose parts, as well as the verses, in a largely Middle Indic guise. We know, however, that there must have been others. In Siks (154.17), which is largely a mosaic of quotations from older works, we find a prose passage cited from a lost work called Bhiksuprakirŋaka, and this prose is precisely like that of Mv, and radically different from the prose of any other work preserved to us. One ms. of Jm contains a short Jātaka story (printed in the Appendix to the Jm ed.) told in the same language and style, and perhaps borrowed directly from an older form of Mv, where it occurs (ii.244 f.); but the Jm insertion contains some passages not found in our mss. and the ed. of Mv, as well as many variants (in part mere corruptions). Sanskritic in appearance, to such an extent that superficially, in its phonology and morphology, it looks like almost standard Sanskrit.¹⁴ However, even the prose shows its Middle Indic base, first, by the fact that it occasionally shows non-Sanskrit forms; and secondly, by the fact that its vocabulary is just as Middle Indic as that of the verses. That is, it contains large numbers of words which never occur, or do not occur with the same meanings, in standard Sanskrit. They are words of the Buddhist, that is a Middle Indic, tradition, even tho they may appear in a Sanskritized garb. These words include, of course, many technical terms of the Buddhist religion, but the great majority are non-religious terms, words applicable in secular language. They stamp the language of the works containing them as based upon another dialect than Sanskrit.

1.38. Even the verses are written, in all our mss. and editions, in a partially Sanskritized manner. This applies to Mv (and to its prose parts) as well as to the verses of other texts. For example, consonant clusters which in all Middle Indic would be assimilated, or otherwise altered, are usually (tho not always) written as in Sanskrit. Wordfinal consonants which would be dropped in all Middle Indic are often written. But a careful study of the metrical structure of the verses has revealed the fact that in some respects, at least, this Sanskritized spelling is mere windowdressing, and misrepresents the actual pronunciation, which was Middle Indic. For example, a consonant cluster at the beginning of a word is proved by the meter to have been pronounced as a single consonant; c. g. a written shilah was pronounced thi-, or $\hbar i$.¹⁵ Internally, meter can give

15. See my article on 'Meter, Phonology, and Orthography in Buddhist Hybrid Sanskrit', JAOS 66.197 ff. This applies to texts preserved in older forms, the first and second classes listed with the Bibliography; not to the verses of the third class.

Professor Helmer Smith ('Les deux prosodies du vers bouddhique', K. Human. Vetensk. Lund 1949-1950, 1; Lund, 1950; 43 pp.) has honored my article on Meter etc., cited above, by a somewhat detailed critique. On a number of important points, I am glad to find, he agrees with me, notably on the purely orthographic and artificial character of initial consonant clusters in the writing (my §§ 15, 39 ff.). On the other hand, he is unwilling to accept many of my cases of syllable-lengthening m.c., especially by nasalization or consonant doubling. On p. 4, top, he expresses fear that I may mislead beginners by my use of the term 'm.c.' Of course I agree with him that such phenomena originated in genuine linguistic developments of doublet forms, each usable at will, and hence both used, according to metrical convenience. He seems, however (if I understand him), unwilling to grant that once such doublets existed in certain categories, analogy could operate to create similar doublets where historically they 'ought' not to exist. To me it seems impossible to doubt the reality of such analogical extensions, many instances of which are used in BHS only in verses where they fit the meter, and where the 'regular' form would not fit. The term 'm.c.' seems therefore appropriate to them. See table of abbreviations for my use of 'm.c.'

One of the two ways in which he seeks to avoid acceptance of my interpretations in this category is to explain otherwise the individual cases mentioned in my article. In reply, I would note, first, that in the JAOS article I cited only a very few examples. Many more are cited in this grammar, but even here my lists are by no means exhaustive. The cumulative weight of the great mass of materials seems to me to make fruitless such efforts to explain some of them away, by pluralistic, and

1•

Introduction

no evidence on this particular matter; Skt. ucyale and MIndic (v)uccali would fit the meter equally well. But since (v)uccali is actually written fairly often, we may reasonably suspect the real linguistic value of orthographic ucyale. And if the verses were demonstrably pronounced, in large part at least, in a thoroughly Middle Indic way, despite partially Sanskritized spelling, is it not at least a plausible guess that the accompanying prose of the same works may have been pronounced similarly, despite much more extensive orthographic Sanskritization?¹⁶

Changes in the course of tradition

1.39. There is evidence to show that, morphologically as well as orthographically speaking, BHS texts have undergone increasing adaptation to Sanskrit norms in the course of handing down by tradition.

1.40. This is particularly clear when, as is true in a number of cases, we have different forms of the same text.

sometimes forced or even impossible, explanations. Let me cite a single instance (p. 4): 'que, seul, midham-gili (o: -gilān Pāņ 6.3.70), epithēte des prāņaka, donnerait un sens à [LV] 197.3'. I think Professor Smith, had he investigated this passage with his usual care and acumen, would not have made this statement. In this LV verse, the Bodhisattva has three dreams about himself; in pāda a, four black and white animals (pranaka has this meaning here, not 'insects' as Mr. Smith may possibly take it, and as it may be used) lick his feet; in pada b, four-colored birds come to him and 'become one-colored' (read bhula = bhūta, with Tibetan, instead of -dbhula); in c-d, he walks on 'mountains of dung' without being soiled. Instead of 'mountains of dung', Mr. Smith would have 'dung-swallowing' (animals, or insects). But: (1) between the pranaka of pada a, and pada c, intervenes pada b, with the unrelated 'birds'; Smith's syntax seems to me impossible. (2) Tib. reads ri 'mountain(s)', for -giri. (3) In Mv ii.137.3 ff. the same three dreams are recorded, in the same order, and in line 11 midha-parvalasya disproves Smith's emendation of LV. (4) To me, at least, it is of some interest that 'A' (on the whole perhaps the best ms. of LV) is cited by Lefmann as midhagbhiri, i. e. midhaggiri; this may be the true reading; it tends to support my view of the equivalence of nasalization and doubling of consonants as means of metrical lengthening; Smith's emendation could not deal with it .--- This case shows how even the greatest of scholars may go astray on an individual case. It consoles me, a little, for the (at least) two errors which Smith's sharp eyes detected in my work; he is quite right (pp. 2-3) on Mv i.70.17 and Lank 268.15, which should be deleted from my §§ 72, 71. I can only express gratitude to him, and chagrin at my own carelessness. (As to malinam, printed in my § 75-for malimam, it was not an 'emendation', as Smith p. 9 naturally supposed, but-I hope-a mere misprint; or else a slip in copying. I am much less inclined to accept most of Smith's other 'corrections' or variant interpretations of passages treated in my work.)

I cannot here deal at length with Mr. Smith's more general considerations. He relies extensively on Pali metrics, and even to some extent on Vedic. I frankly have never understood Pali meter, as a whole. (I hope and expect to profit from Prof. Smith's studies, based on his vast knowledge of Pali, which infinitely surpasses my own.) At certain points I have noted resemblances to BHS, but a great many Pali verses baffle me; they seem to involve principles which I am unable to formulate, but which in any case seem to me, for the present at least, and even after reading Smith, quite different from Most complete texts of extensive works are known to us only from Nepalese and Japanese mss. But some are preserved in other regions; and particularly in Chinese Turkestan, old fragments have come to light of parts of some of the same works known more completely in Nepalese or other later recensions. It was pointed out notably by Lüders (see fn. 16) that the fragments of the 'Kashgar' (Chinese Turkestan) recension of SP show many differences from the Nepalese recension, a fact sufficiently clear even from the critical notes to the very unsatisfactory Kern-Nanjio edition. Lüders emphasizes that this is as true of the prose as of the verses, and that the Kashgar form of the prose often contains Prakritisms, while the Nepalese has Sanskritized the forms (e. g. bhāsimsu: abhāsanla); tho at times the reverse relation exists, so that we must assume some Sanskritization in both recensions, and an original more Middle Indic than either. Lüders is undoubtedly right in taking it for granted that Middle Indic or non-Sanskrit words and forms, in whatever recension they

any in BHS. As to Vedic meter, BHS seems to me radically different in fundamental principles, and I think it dangerous to interpret the latter by the former. BHS meter, in fact, seems to me in some important respects quite individual, despite, of course, many points of resemblance to Pali and Classical Skt. meters. At least provisionally, and at first, I believe it should be studied by itself.

Such study is complicated. First, the text tradition of most BHS texts is wretched. Corruptions abound everywhere, notably in Mv, and (probably near the other end of the chronological scale) in such a text as Mmk. We must collect, for each text, forms, and metrical patterns too, which are attested by considerable amounts of evidence; having done that, we may, cautiously, suggest that apparent deviations may be text-corruptions. In My I have found a very considerable number of cases which agree with the metrical principles I have set up. In view of the known frightful corruption of the mss., I think we may apply the above principle to seeming exceptions. On the whole I am inclined to treat Mmk in the same way. But there are some texts of my class 3 (see the introduction to my Bibliography) where I still hesitate, because of the lack (in their mss.) of a compelling number of cases supporting my formulas. (Divy is an example.) Some of these Class 3 texts may belong, metrically, to a developed, or broken-down, system of metrics, compared to the texts of Classes 1 and 2. In footnote 21 to § 1.81 I call attention to some stanzas inserted very late in the Nepalese version of SP which are metrically very aberrant, and which I cannot analyze satisfactorily. In principle, therefore, I am not averse to recognizing different 'prosodies' in BHS texts as we have them. So far, I remain unconvinced by Professor Smith's particular views as regards types of BHS prosody. There is not room in this already swollen publication to discuss them in detail, nor have I as yet had time to give sufficient study to his (I am sure, very valuable and important) studies in Pali metrics.

16. In Hoernle, Ms. Remains, 161 f., Lüders wrote: 'I am even inclined to believe that the original (sc. of SP) was written in a pure Prakrit dialect which was afterwards gradually put into Sanskrit.' (Cf. the next paragraphs. SP is in no way distinctive among BHS works.) If Lüders had been aware of the above evidence, perhaps he would have been ready to consider with me the possibility that the 'putting into Sanskrit' was in part purely orthographic. I think, however, that Lüders was quite wrong in indentifying the 'original dialect' as Māgadhī, solely on the ground of vocatives in -dho; these are not exclusively Mg. (§ 8.88). 1.46]

occur, are invariably older than corresponding Sanskritizations in other versions.

1.41. Similar results are shown by a comparison of the Central Asiatic fragments of Vaj published by Pargiter (Hoernle, MR 176 ff.), compared with Müller's ed. of Vaj (from Japanese sources). Examples, all prose (first Pargiter. then Müller's ed.): parindiläh . . . parindanayā 179.3, parin-(even in this form the word is not Sanskrit!) 20.4, 5: pralisthihitvā 180.12, pralisthilena 21.9; pralisthihe (3 sg. opt.) 180.14, pratifisthet 21.11; viyubha 180.16, 188.20. viyūha 188.19, for vyūha 27.4, 38.6, 7; u(d)grahesyanti 186.8. 22, and 187.9, udgrahisyanti 33.17, 34.14, 35.5; babhūpa with 3 pl. subject 187.6, abhūvan 35.1; carimikāyām paścimikāyā(m) 187.8, paścimāyāņi (omitting cari") 35.1. In all these the Central Asiatic version is more Middle Indic, Müller's text Sanskritized and so presumably secondary. But the contrary is the case with aradhila and viradhila 187.7, instead of aragita, viragila 35.3.

1.42. It has sometimes been suggested that the verses retain in most texts a more Middle Indic appearance than the prose because the meter made it harder to Sanskritize them. I should not venture to deny that this consideration may have had some weight. Perhaps the greater anra of distinction, secular or religious, which has usually attached to verses in Inlin, may also have been concerned. (The verses alone of the Pali Jätaka are canonical; the prose is mere 'commentary'.) I would, however, point out that we have definite proof that the BHS verses. too, were not exempt from changes in tradition. Chakravarti's edition of Ud cites the text of one ancient and fragmentary ms. (provenience not stated) and of several later ones, mostly. it seems, from Chinese Turkestan. I shall record here a selection of the variants. They are interesting as showing that meter was not really a serious obstacle to change. By patching the meter (often with insertion of a 'patchword', Flickwort), by rearranging the order of words, or otherwise, a new form, Sanskrit or nearer to it, may replace a more Middle Indic one. If anything really important had depended on it, later redactors could probably have made the verses of SP, for example, look as Sanskritic as the accompanying prose. Indeed, this is just what happened in the BHS works of my third class (see Bibliography). In the forms in which they are preserved to us, their verses (so far as they have any; Divy, for example, has a good many) seem as Sanskritized as the prose. Furthermore, some Middle Indic forms could have been Sanskritized very easily without affecting the meter. Thus the Skt. 3 sg. optative ending -el is metrically equivalent to Middle Indic -e. Yet the latter is extensively retained in the verses of most texts of the first and second classes, while the prose almost always has -el, if we may trust the mss. and editions (except in My; even in My -e is much commoner in verses than in prose).-When the same verse is known to exist in Pali, it is sometimes interesting to compare the Pali equivalents of the words cited; this will be done occasionally, tho by no means invariably, in the following list. I cite first the readings of the oldest ms., then of the later ones, which are invariably secondary.

1.43. Earlier and later forms in ms. readings of verses of Ud.

- viii.12 bhāşeya, vihinseya: bhūşcta, vihimsela. Pali bhāseyya, vihimseyya.
- ix.3 bhāyasi: bibhesi.
- x.5 śraddhāya (instr.): śraddhayā.
- x.6 cchindali: chinatti.
- xi.2 parākkrame: parākramet.
- xi.3 samkiliştâ (read °tam?) va (m.c. for vā) yas (read yat) tapah: samkliştam vāpi (note patchword!) yat tapah.
- xi.7 careya: carate; samkalpānam (gen. pl.): °nām.

- xi.10 sålam vå måluvolata (:- måluvà-olatam, for avatatam): sålam vå måluvå yothå. Pali måluvå sålam isolatam.
- xi.11 sthero: sthaviro (which here is unmetrical, even in the later ms, which writes it; its writer doubtless read it sthero (if not thero), the 'Skt.' form being purely orthographic).
- xii.3 yāya (instr.): yayā.
- xil.4 märgånästängikah (i. e. märgåna, gen. pl., ast.) śresthah salyänam cature padah: märgesn astångikah śresthaś calväry ärgägä salpatah. Pali maggån' affhangiko settho sacränam caturo padä, supporting the prior reading, all the forms of which are paralleled in BHS. Chakravarti's emendations are all false.
- xii.5 yadā prajhāya (instr.) pašyai: prajhayā pašyale yadā (order changed to accommodate Sktized form to the meter).
- xv.7 ye suptāķ pratibuddhathu (impy.): suptāš ca pratibudhyalaķ (read %ta; stem and ending both Sktized).
- xvi.1 -daršāvī (n. pl.); -daršino,
- xvi.5 so imām (acc. pl.) bhāsate lokr (acc. pl.); so imam (one fragment imām!) bhāsate lokam. Pali so (i)mam lokam pabhāseti (more nearly agreeing with the later version of Ud; the older version of Ud is probably older than the Pali).
- xvi.5, 7, 9 abhramukto va (= iva): abhramuktaiva (v.1. *kta iva, unmetr.). Pali abhhā mutto va.
- xvi.14 iccheya: icched.
- xviii.3 chindatha, bhavatha (impvs.): chindula, bhavata.
- xviii.5 paduma, vrnihaye: padmani, brmhayel.
- xviii.11 (vyatiro)cati prajňāya (instr.): prajňayā vyatiroca(n)te.
- xviji.13 vipramuñcatha (impv.): °ta.
- xviii.18 prapuspa(kāņi): tu puspakāni (so). Pali papupphakāni.
- xix.5 hirinişevî : hrinişevî hi (patchword). Pali hirinisedho,
- xix.12 bhadram va: bhadram iva (unmetr.).
- xx.1 viprajaheya : viprajahec ca (patchword). Pali vippajaheyya.
- xx.2 prajahe: °hed.
- xx.5 ahiri: ahriko.

1.44. At least equally interesting are parallel passages in different works, sometimes paralleled also in the Pali canon, and in that case presumably going back to extremely early Buddhist tradition. In the statement of the first of the Four Noble Truths, presumed to contain the Buddha's own words, Mv iii.332.4 has the adverbial instr. samksiptena, 'in brief, in a word', not recorded in Skt., but = Pali (Vin. i.10.29 = SN v.421.23) samkhiltena, while the LV version, 417.7, substitutes the regular Skt. samksepäl. In another prose passage Mv ii.283.14 reads purime yāme 'in the first night-watch', but the LV correspondent (344.7) reads prathame yāme. The word purima is not Skt., and is used by LV only in verses, never in prose, according to our mss. Yet in such a passage as this, one cannot help wondering if the obviously secondary prathame was not introduced by some relatively late redactor or copyist; in short, whether the original LV text did not read purime.

1.45. Even in a verse, LV 387.18 substitutes parvikām for Mv iii.305.19 purimām, here meaning 'eastern', a rather rare use of purima in BHS, and nowhere found in LV, I believe. In verses, Mv iii.306.6 purastime 'eastern' = LV 388.8 pūrvasmin vai (note the patchword, m.c.); Mv iii.309.8 Prthioī Padumāvali = LV 391.3 Prthvī Padmāvali tathā (two epenthetic vowels removed, meter again mended by a patchword).

1.46. In general, LV (at least as preserved to us) is secondary and Sanskritizing compared to Mv, in passages

which the two texts have in common. Yet it would be wrong to assume that this is universally true. Each case is a separate problem. Neither text borrowed directly from the other; both incorporated materials that were older than either, and sometimes LV seems to have preserved features which more closely resemble the common original than any parallel version, including Pali versions. In the prose account of the Buddha's first sermon at Benarcs occurs a passage in which he tells his audience, the five monks who had formerly been his followers, how he first realized that he had reached enlightenment: LV 418.13 fL. Mv iii.333.11 ff., and Pali Vin. i.11.18 ff. = SN v.422.31 ff. The two Pali versions are identical, verbatim; the two BHS versions, closely resembling the Pali in essence, diverge from it and from each other in many details, sometimes one, sometimes the other being closer to the Pali. In the first sentence, LV esu caluso (so all mss.; note the MIndic form, for Skt. calursu, in LV prose!) dryasalyesu = Pali imesu calusu ariyasaccesu, but Mv (changing construction) imāni catvāry āryasatyāni. There are many interesting things which a detailed study of the three versions reveals. I shall mention only the verb-form pratijñāsisam, LV 418.15 and 19, an unaugmented aorist. In 15 all mss. lack the augment; in 19 only a few inferior mss. have it. This is a non-Skt. feature, and therefore likely to go back to the oldest times. Yet Pali has an augmented form, paccanñāsim, both times. My substitutes a present, pralijāne 'ham the first time, and prajānāmi (perhaps read pralijānāmi?) the second time; in lines 12 and 15 it has abhyajñāsisam

LV and the Pali. 1.47. I know of no Pali correspondent to the Avalokita (Mv), or (in Siks) Avalokana (or °nā), Sūtra which is reproduced at length in Mv ii.293.16-397.7 (following another similar text with the same title). But many of its stanzas are cited in Siks 297.10-308.12, and a few others 89.15-90.3. The Siks extract is only a small selection, yet includes some verses not found in My; for this and other reasons it is certain that Siks was not citing from Mv; rather, both cite from a common ultimate original. Most of the verses common to the two texts are composed in a sama-vrtta meter known to me otherwise only from LV 229.21-234.10; each of the four pādas has the scheme ---- ×. Both Siks and Mv, especially the latter, are very corrupt in this passage. A combination of the two, with the help of the meter, sometimes makes possible a restoration superior to both. (Omitting two of the four shorts which follow the fifth syllable produces a normal BHS tristubh pada; and this is often the result of textual corruptions, especially in Mv.)

(augmented) in a clause which is differently expressed in

1.48. In these verses, apart from corruptions, the Mv readings are in general more un-Sanskritic, and hence closer to the original, than those of Siks. Many scores of examples would show this; one is phalikha-bhujo, to be read with mss. (except that they have "kham-) Mv ii.379.5, 'with arms like bars' (see 1 ict. s.v. phalikha), for which Siks 303.3 has the normal Skt. parigha-bhujo. Nevertheless, Siks is sometimes non-Skt. and clearly original, or closer to it, and Mv Sanskritized and secondary. Examples are: yatha-r-iva Siks 304.10, yatha Mv ii.383.5 (in meter, also, secondary); kavacita laksanebhih Siks 304.16, varalaksitango Mv 386.10 (in meter, also, secondary); pilhita Siks 307.3, (a)pihitā Mv 393.6; sthita bhonti Siks 307.15, nivasanti Mv 394.19 (but here both are metrically second

17. Among the works treated by me, at most Jm might possibly be questioned. It may be a borderline case, but seems to me to have enough characteristic BHS vocabulary to justify inclusion. It certainly has few traces of Middle Indic morphology. Cf. however, the gerund

ary, lacking two short syllables); sampramuñei (aor., or possibly opt.) Šikş 308.12, so pramuñeed My 395.23.

1.49. We might go much farther in citing such examples, but I trust that these are enough. Let me summarize the facts about BHS as stated so far.

1.50. (1) Buddhist Hybrid Sanskrit tradition, as a whole, starts from, or goes back to, an early Buddhist canon, or quasi-canon, which was composed not in Sanskrit but in a Middle Indic vernacular which very probably already contained dialect mixture.

1.51. (2) Some parts of this old canon, or passages from it, are preserved in BHS, sometimes in more than one form. When this is the case, any non-Sanskrit features of form and vocabulary, wherever recorded, are always closer to the original on which they are based than corresponding standard Sanskrit features, wherever recorded.

1.52. (3) The verses of BHS texts of my classes 1 and 2, as presented in our mss., are on the whole semi-Middle-Indic or hybridized. This means that they represent the BHS tradition in its purest form. In texts of class 2, the accompanying prose parts of these texts are nearly (not quite) Sanskritized in phonology and morphology, according to the mss. In vocabulary, the prose is just as Middle Indic as the verses.

1.53. (4) In all BHS works, as presented in our mss. and editions, there are very many words and forms which are standard Sanskrit. These include many forms which cannot possibly have existed, at any time, in any Middle Indic dialect. They represent alterations in the tradition, later in time than the original Middle Indic canon, at least (not necessarily later than the composition of the specific BHS works in which they are recorded). As time went on, the tendency was in general towards ever in-creasing Sanskritization. Yet the BHS tradition continued to live, apparently for centuries, as a religious language among the Buddhists, or at least some Buddhists, of North India. The hallmark which distinguishes it is the vocabulary, which contains not only technical religious terms, but quantities of purely secular words, which never occur in standard Sanskrif. Very rarely can any serious doubt arise as to whether a particular work should be classed as BHS. Even if its grammar is virtually Sanskrit, or entirely so (tho such a case hardly occurs), its vocabulary will decide.17

1.54. (5) There is clear evidence that some of these Sanskrit words and forms were substituted for older, non-Sanskrit ones, by later copyists or redactors of the individual work containing them: in other words, that some such works were originally more Middle Indic than is indicated by some, or even all, of the mss. in which they are preserved to us. In SP for example, one recension Sanskritizes some words, another recension others; the original text of SP must have been less Sanskritic than either.

1.55. (6) There is, further, evidence that in citing or incorporating older materials, any BHS text (if we accept the evidence of its mss.) may be expected to have introduced some Sanskritizations of originally Middle Indic features.

1.56. (7) It is, however, certain that some Sanskritappearing features are orthographic only; the words were pronounced as in Middle Indic. This is proved by the metrical structure of the verses of IBHS texts of classes 1 and 2. How old this misleading Sanskritic spelling is, we have no way of telling; it appears very commonly, tho

adhiśrāya Jm 35.5 (§ 35.20); the pples. (an-)āstarita 220.14 and (Aiga-)dinna 192.12; sātmy eva bhavanti 95.9 (= sātmībhavanty eva); and the characteristic BHS and Pali locution teṣām ... etad abhūt 41.23. not invariably, in the mss. of all the specified texts. That the same was true of the same or similar features in the

accompanying prose, at least in earlier times, seems a reasonable guess; naturally, there can be no direct proof that the prose was pronounced otherwise than as written.

Plan and methods of this work

1.57. I trust that the preceding statements will justify a procedure adopted in this work, which at first sight may seem surprising. In principle, I have excluded from my grammar and dictionary all forms which are standard Sanskril, and all words which are used in standard Sanskrit with the same meanings.¹⁸

1.58. My work is therefore to this extent incomplete. For certainly some words and some forms were common to the original dialect, as to all Middle Indic dialects, on the one hand, and to Sanskrit on the other. But there is no way of distinguishing such items from late intrusions put in by secondary redactors, copyists, or even modern editors; nor is there any general, objective, and reliable way of distinguishing those words which are Sanskrit in spelling only and were pronounced in Middle Indic fashion. To have included all words and forms presented in Sanskrit guise would have padded the work and, what is more important, obscured the picture. And it is inpossible to draw any line objectively. It was a question of all or none.

1.59. What I have tried to do, then, is to collect and classify the non-Sanskrit forms and words, only, which are contained in BHS. It seems to me that this is what has always been needed; and no attempt has previously been made to supply it.

1.60. My work is based on available printed texts. A few of the editors have provided critical apparatuses giving the readings of their mss., or some of them (see §§ 1.73-75). These are especially valuable and have been constantly utilized in my work. They demonstrate that even the hest editors did not know how to handle the language, for lack of any standard to go by. How can an editor be sure whether a non-Sanskrit word or form found in his manuscripts is just a copyist's error, or represents (accurately or inaccurately) a real form of the language? It was necessary to collect the evidence from all, or nearly all,¹⁹ the published texts, and to classify and systematize the grammar and lexicon. Only then could editors know how to proceed. No existing printed text of any work in this dialect is satisfactory, or indeed could be, without such a work as I have tried to provide. The editors should not be blamed; theirs was an impossible task. Even the best of them print false emendations on almost every page; the worst, we may fear, do the same thing without acknowledging it.

1.61. It is, of course, true that my rule of excluding standard Skt. forms and words is not always easy to apply. Just what is 'standard Sanskrit'? I have tried to include, rather than exclude, doublful cases, indicating the reason for my doubt. Exclusively Vedic words and forms are not 'Sanskrit'; but it is rarely necessary to consider them, since they hardly ever concern BHS. The Mahābhārata is another matter. It contains definite Middle-Indicisms; and I have thought best, for example, to mention in my Synopsis of Verb Forms (Chap. 43) the thematic present dadati, 'he gives', and forms of its type, which are not rare in the Epic

18. Technical terms of Buddhist religion, and proper names, belonging exclusively to Buddhist stories, are of course included in the dictionary, even tho they occur in *Buddhist* works which are composed in standard Sanskrit (\S 1.2). This constitutes no real exception to the above principle. and occur in the Veda. In the Dictionary, 1 occasionally include in *parentheses* a word which seems to be used in standard Skt. in virtually the same way as in BHS, and yet to deserve inclusion for some reason. Compound nouns paralleled in Pali, and not recorded in Skt., I have tried to include, even when the parts are normal Skt. and the meaning of the compound is easily and simply derived from the meaning of its parts. Words and meanings recorded only by Sanskrit grammarians and lexicographers, or perhaps also once or twice in late and artificial literature, are generally treated as BHS. The extent and nature of the Skt. records about them are included with the definitions.

1.62. Naturally, the two great dictionaries of Boehtlingk (and Roth) have been my main reliance for Sanskrit. Schnidt's Nachträge have also been constantly consulted, and occasionally I have found a Skt. word elsewhere; in such cases the word has been included in my Dictionary, at least in parentheses, with a reference to its Skt. occurrence.

1.63. Pali correspondents, when I have found any, are always cited in the Dictionary (cf. § 1.106). If no reference is given, the Pali word will be found in its alphabetical position either in the Copenhagen Dictionary, Volume I, or otherwise in the Pali Text Society's Dictionary. If I know no correspondent in Pali, but have found one in Ardha-Mägadhi, this is cited, from Ratnachandra's Dictionary (without reference), or Sheth. If no Pali or AMg. correspondent is found, I cite (from Sheth, if no reference is given) any Prakrit equivalent found, or any from the Asokan or other inscriptions, from the Deśināmamālā, or from Apabhramśa, or occasionally from a modern Indo-Aryan language. If no correspondent is cited, none is known to me.

1.64. As stated above, my work is primarily based on the printed texts of BHS works listed in the Bibliography, with such critical apparatuses as are provided therein. I have had no direct access to manuscripts. To have tried to use them systematically would have meant, in effect, reediting the texts. New editions are unquestionably needed of most of them, and I trust that my work will facilitate the labors of future editors. But if I had tried to do this task myself, the fifteen years or so which I have given to this work would have been indefinitely extended, and I should certainly not have lived to finish it.

1.65. In general, the dictionary cites nouns, adjectives, and pronouns by stem form, verbs by third person singular present. For purposes of general reference, the grammar also uses these forms, but in the case of verbs it also often uses the 'root' in accordance with the common Sanskrit convention.

1.66. Specific forms cited from any text are in general intended to reproduce the exact spelling of the text cited, unless the contrary is stated, except that when the inflectional form is deemed unimportant, the bare stem alone may be cited. The chief further exception concerns alternative spellings, which vary at random, with either anusvāra or a nasal consonant before another consonant. This seems to be a purely orthographic matter, and as a rule not worth recording. Such writings as final $-\delta n$ are probably of no more significance. See on this whole subject §§ 2.65-67.

• 1.67. Often I adopt a reading stated to be that of all mss., which an editor emended (in my opinion wrongly).

19. A few works have been used only partially, or not at all, because they appeared in print, or became accessible to me, only after my work was nearing completion; or because they seemed to me of minor importance.

Technical terms of Duddhist ----

In such cases I sometimes write '(mss.)' or '(so mss.)' after the form in question, not always citing the emendation which the editor put into the text. In the case of the ending -etsu(h), 3 pl. aor. or opt., which occurs many hundreds of times in Mv and is invariably replaced by -ensu(h) in Senart's text (§ 1.88), I cite the reading of the mss. without calling attention to Senart's emendations individually. What seem to me clearly misprints (and, of course, still more what I believe are copyist's errors) are corrected, but (I hope) always with indication of the fact, if only by a phrase like '(so read)', and oftener with quotation of the text as actually printed.

1.68. I need hardly say that I have read with care all the texts on which my work is based (with the qualifications noted under certain works in the Bibliography), tried to understand them, and noted all words and forms which seemed not to be standard Sanskrit. Naturally, in the case of the commoner words and forms, only samples could be included in the dictionary and grammar, but I have tried to make them typical and abundant enough to give an approximate notion of what the totality would show.

1.69. Unfortunately, the state of most of the textual tradition is deplorable. While many passages, and consequently many words and forms, have become clear as a direct result of my collection and collation, I am only too well aware that many textual and exegetic problems have remained unsolved. To help future investigators, I have tried to include all dubious words and forms as far as possible. It is unfortunately not always possible to do so; very often the true word-division depends on the interpretation, and cannot be established in default of the latter.

1.70. Parallel passages in BHS itself are, of course, the most important aids to interpretation, as in all philological work. Next come parallel passages in Pali, which often contains close equivalents of BHS verses or prose passages, and still oftener phrases and locutions which correspond to those of BHS. The Copenhagen ('Critical') Pali Dictionary has proved extremely valuable in helping me to discover them. If that monumental and magnificent work of scholarship had progressed further than the first volume, I should doubtless have found more than I have. The other Pali dictionaries are much less useful.

1.71. Tibetan translations, so far as available, are extremely helpful. Their notorious literalness usually makes possible a confident decision as to what their BHS originals read, and how the translators understood it. Unfortunately I had access, while the work was in progress, to Tibetan versions of only a few of the BHS texts. After the work was completed, Yale University acquired a set of the Kanjur (Lhasa ed.), as a gift from His Holiness the Dalai Lama; but this arrived too late to be useful to me.

1.72. I know no Chinese, and so have had to depend on others for such help as I have been able to get from Chinese Buddhist works. Finot, for example, furnishes a very helpful French translation (by Huber) of the Chinese version of Prat. Various kind colleagues, present and past, Sinologists and Japanologists, have given me valuable

20. To the kindness of my colleague Professor Johannes Rahder I owe the following note. A Japanese work called Bombun-butsuden-bungaku no Kenkyū (Studies in Sanskrit biographies of Buddha), by Taiken Kimura (died 1930) and (his pupil) Tsūshō Byōdō (799 pp., Tokyo 1930), deals extensively with the Mahāvastu on pp. 565-668. Particularly noteworthy is a comparison of the contents of Mv i and ii (vol. iii is not treated only 'for lack of time', not because of lack of parallels) with the contents, especially, of the Chinese Fo-pên-hsing-chi-ching (not lact

help, especially in interpreting the Chinese and Japanese equivalents of the BHS and Tibetan entries contained in Mvy; occasionally also in dealing with other works.

1.73. Perhaps the most difficult and corrupt, as also probably the oldest and most important, of all BHS works is the Mahāvastu. So far as is yet known, no translation of it exists in either Tibetan or Chinese.²⁰ It was edited by Émile Senart in three slout volumes, 1882-1897. Senart's extensive notes often let the reader perceive the despair which constantly threatened to overwhelm him. It must be remembered that around 1880 comparatively few Pali texts had been published; even by 1897 many were still inaccessible; and as to BHS, few texts had been edited in Senart's day, and those few badly. Confronted with the many hundreds, perhaps thousands, of corruptions in My, Senart had little choice except to let them stand as the manuscripts present them, or to try to rewrite them himself. He did sometimes one, sometimes the other. It goes without saying that in a great many cases his emendations were successful; for he was a very great scholar. And yet, as is generally the case when an editor attempts a vast number of emendations, many of Senart's have not proved sound. We must, however, acknowledge with the utmost gratitude and appreciation the meticulous care with which he recorded the actual readings of his six mss. as far as i.193.12, and thereafter of two typical ones. With the far more extensive aids now available, Senart's careful and scholarly critical apparatus has enabled mo to recognize as correct many readings of his mss. which he rejected; and I fully believe that many others will be recognized by my better-informed or more perceptive successors.

1.74. Johannes Nobel has edited Suv with equally praiseworthy care, and also published its Tibetan translation. The fragments in Hoernle's MR are all admirably edited. There are other able and sound editions. Lefmann's edition of the important LV seems to be quite meticulous in giving the readings of his rather numerous mss., which makes his work very valuable, even the his editorial judgment in constituting the text cannot be rated very highly. On the other hand, the Kern-Nanjio edition of SP, which is also a text of prime importance, the it professes to cite the readings of a number of mss., is sadly unreliable (see W. Baruch, Beiträge zum Saddharmapundarikasūtra, Leiden, 1938, pp. 7-12). And the later edition of SP by Wogihara and Tsuchida has little value; for example it constantly repeats, without note or comment, indefensible emendations of the KN text.

1.75. Still less helpful are the texts, printed usually in the orient, which do not even try to record the readings of the mss., or do so only very sporadically and unsystematically. Many of them are obviously printed very badly, with numerous misprints, which at times are not easy to distinguish from scribal errors of the underlying mss. Yet I have hesitated to refuse to use a printed text on such grounds alone; some of them contain important materials. I have, to be sure, tried to exercise restraint and caution in relying on the evidence of such texts, of which the edition of Mmk is a rather extreme example.

than the end of the 6th century). This work seems to follow rather closely the outline of Mv, omitting a great deal, much of which is suspected on other grounds of having been added to Mv in late times. Byödö believes that it is either a translation of an older version of Mv, or that both it and Mv were based on an older Indian work. Fuller knowledge may possibly compel us to qualify the usual statement that the Mv was 'never translated into Chinese' (so e. g. Lin Li-kouang, L'Aide-mémoire, 174).

Sanskrit versus Buddhist Hybrid Sanskrit

1.76. Many scholars, even down to the present day, refer to BHS simply as 'Sanskrit'. Louis Renou, in his excellent Grammaire Sanscrite, includes (e. g. on p. 350) references to some, tho relatively very few, forms of BHS; on p. i he notes that 'on a été à la frontière du sanscrit en signalant les faits de langue mixte représentés par le Mahāvastu et le Lalitavistara'. Of course all have recognized that, if this language is 'Sanskrit', it is a peculiar kind. But there are other peculiar kinds of what is still usually called 'Sanskrit'. The language of the Mahābhārata, for instance, contains Middle-Indicisms; yet few would hesitate to describe it as fundamentally a kind of 'Sanskrit' (tho it does not follow Pāņini very closely).

1.77. The great lexicographer Boehtlingk included in BR and pw many BHS words (especially from LV, Kv, Mvy, Vaj, Divy, and Jm). But in the preface to the last volume of pw, Bochtlingk refers to such BHS words as 'hardly to be called Sanskrit'. The publication date of this volume was 1889; that of the first volume of Senart's Mahāvastu was 1882. Yet Bochtlingk never mentions Mv, and does not cite a single word from it. Had he not seen Senart's publication before finishing his work on the pw? If he had seen it, and deliberately ignored it, I should have expected him to state his reasons for doing so. He could, in my opinion, have given very good reasons. If the rest of BHS literature were like Mv, in presenting prose as well as verses in Middle Indic or hybrid forms, I can hardly believe it would ever have been called Sanskrit, or that its vocabulary would have been included in Sanskrit dictionaries. One might as well include Pali and Prakrit words in a Sanskrit dictionary. I believe it is a fact, and if so it is significant, that nearly all BHS words included in BR and pw are taken from the prose, not the verses, of such works as LV. The form of this prose is such that it is easy to mistake it for Sanskrit. In my opinion, however, it all belongs to a different linguistic tradition, and should be excluded from works professing to deal with Sanskrit.

The Prakrit underlying BHS

1.78. Various attempts have been made to identify the underlying Prakit with some known Middle Indic dialect, on the basis of specific resemblances between such a dialect and Middle Indic elements in BHS. On careful examination, such points of agreement are usually found to be not sufficiently specific; that is, the feature in question turns out to occur in other Middle Indic dialects besides the one with which identification is proposed. Furthermore, any such point of agreement will always

21. For stratification in Mv, see especially Windisch, Die Komposition des Mahāvastu, ASGW, ph.-hist. Kl., 27, Nr. 14, p. 476 ff.; Oldenberg, NAWGött. ph.-hist. Kl., 1912, Heft 2, p. 124 ff.; also id., Studien zur Geschichte des buddhistischen Kanons, ibidem 156 ff. Dschi, NA WGött. ph.-hist. Kl. 1949, p. 245 fl., Die Verwendung des Aorists als Kriterium für Alter und Ursprung buddhistischer Texte, finds a much greater number of aorists in the older than in the younger strata of My, and extends this criterion to other Buddhist texts. Some of his data are interesting and valuable, but at times I think he exaggerates the validity of his inferences. In particular, he seems to me to fail to distinguish between aorists of MIndic, or semi-MIndic, type, like abhūși, and thoroughly Sanskrit aorists like pravikşat (Divy; Dschi p. 261). The former may reasonably be considered relatively old in BHS; but forms like praviksat are late, and belong to the Sanskritized stage of be found to be more than counterbalanced by points of disagreement. Hence, no doubt, scholars in recent times have become wary of such identifications. One of the latest to be published, as far as I know, is that of Lüders in Hoernle MR 162, who thought the original dialect of SP, at least, was Māgadhī, solely on the ground of voc. pl. forms in *iho* (but see \S 1.38, fn. 16). For the views of Hiän-lin Dschi see \S 1.24 ff., 1.97, and fn. 21 to \S 1.81.

1.79. I find no reason to believe that the Prakrit chiefly underlying BHS, or any substantial part of its tradition, was an eastern dialect. I know no way of localizing it geographically at all. Complete dialectic unity, indeed, could not reasonably be expected, and will certainly not be found, in so large a body of texts, obviously of quite different dates.

1.80. I am not thinking of the varying degrees of Sanskritization, referred to above. If we limit our attention to non-Sanskrit forms, we still find variation: some forms which are common in certain works occur rarely, or perhaps not at all, in others; and often in the same work we find forms which may plausibly be taken to show dialect mixture. As is well known, Pali also shows linguistic differences between the gathas, canonical prose, later prose, etc. (Geiger p. 1 f.), and dialect mixture in all of them. I should add that, as in the case of Pali, I tind no reason to question the essential dialectic unity of the BHS Prakrit, Such differences as occur are minor compared to the great mass of resemblances.

1.81. In some cases, chronological layers in the same book are more easily detected by stylistic or metrical criteria than by morphological or phonological ones.²¹ I have not felt it possible, in this work, to consider style or meter except as they seemed to me directly reflected in phonology or morphology. Nor have I attempted anything like a full collation of parallel passages (for examples see above, §§ 1.43 ff.), either within BHS itself, or between these texts and Pali or other Buddhist texts. Generally speaking, I have referred to such parallels only when I have found in them something useful for the interpretation of a BHS word or form. There is great need for much more extensive study of such parallels than has yet been made; I hope my work may help future workers in this field, but it does not claim to anticipate such work to any noteworthy extent.

1.82. My work aims to be descriptive rather than historical or comparative. Nevertheless, under each non-Skt. form recorded in the grammar I shall usually refer briefly to such correspondences in other Mindic dialects as I have noted. These references are very far from exhaustive, and are not intended to be that. In the now following sections I shall mention a few which are of special interest at this point, because they are not general

the language. It is significant that Dschi can cite no correspondent to prāviksat in Divy 39.19, 25, from the parallel passage in Pali. Sometimes such forms may indeed replace old (MIndic) aorists, as on p. 200 of Divy (Dschi p. 262), but this cannot be taken for granted; a late Sanskritizing author is also quite capable of introducing regular Sanskrit aorists in a BHS text, when an older form (if there was any) of the passage had none .-An example of how meter may give evidence of relative age is found in SP chapter 11, verses 42 ff. The meter here is obviously quite unlike the usual meter of the verses of SP and similar BHS texts. It alone proves that the passage is an interpolation. This fact is confirmed by the omission of all that follows verse 41 of this chapter in certain Chinese translations (see the KN ed., 256 note 5), and in the Kashgar ms. reported by La Vallée Poussin, JRAS 1911, p. 1074.

MIndie or even 'general Prakrit'; some are even quite unknown elsewhere.

1.83. So, first and foremost, the loc. sg. of a-stems in esmin (or esmim: also esmi), on which see \$\$ 8.70-73. It occurs very frequently, in the verses of most texts, and in the prose of My. Yet Senart systematically excluded it from his printed text of My, Many other editors have followed his example and relegated it to their critical apparatus. Yet it is very easily explained, and must certainly be accepted. Since it occurs in no other known dialect, and is so very common here, it alone is enough to prove that the BHS Prakrit is not exactly the same as any other known to us. In passing it may be noted that asmin or asmin (also asmi) is likewise common (as in Pali), but that the alternative ending amhi (cf. Pali amhi; also in some Pkt. mss., according to Pischel 366a by corruption) is extremely rare; it can scarcely have been a form native to the underlying Prakrit. I have noted, in fact, only a single noun form in appli; there are a very few pronouns. The locative ending i (which Pischel recognizes only for Ap.) occurs very often for e in verses of most texts, m.c.; it is not common in My, the cases are found. The ending amse, recalling regular AMg. amsi, is recorded just once, in a verse of My (according to the mss.).

1.84. Another ending unknown elsewhere is dvo in the nom.-acc. pl. of [em. d-stems. It occurs at least half a dozen times in My; nowhere else. It also occurs even in a mase, pl. voc., dvusdvo, § 8.89. It is probably somehow related to the commoner dyo (== Pali id., and probably Pkt. do; otherwise Pischel); nont-acc. ph dyd and dye are perhaps to be recognized, tho they are very rare. On the y as against Prakrit zero cf. the oblique fem. dye (as in some Asokan dialects), clearly corresponding to general Pkt. de (which is very rare in BHS); dye is almost limited to My, but commoner there than dya, which is the regular ending in (verses of) most texts and is also familiar enough in My. The oblique fem. ending d (bare stem), known in Pali, is rather rare in BHS.

1.85. The endings of fem. i (and *i*-)stems are partly analogous to those of \bar{a} -stems; but the *i* may be short (without correlation with the historic quantity of the stem vowel), as in Pali oblique sg. $iy\dot{a}$ (but Pkt. regularly ie, ia). Thus the usual oblique sg. fem. endings in My, including prose, are *iye*, *iye*; in other texts these occur chiefly in verses; and in most texts, even in verses, *iya*, *iya* are commoner; *iyo*, *iyo* also occur, rarely, and chiefly in My; also $y\ddot{a}$, ya, as generalized oblique endings, chiefly in verses. Fem. u-, \ddot{u} -stems (much fewer in number) are roughly analogous to *i*-, *i*-stems.

1.86. Another ending not recorded elsewhere is a general oblique form of *ar*-stems in *arc* or *ari*, fairly well attested, but only in Mv, and always in prose. It is found from both m. (*pilare*, *pilari*) and f. stems, but I presume that it is analogous to the (regularly fem.) oblique endings *iye*, *uye*, of *i*- and *u*-stems.

1.87. The gen. pl. sānam (cf. the general Pkt. gen. sg. se), 'of these' (to stem sa-: la-), is not clearly recorded elsewhere. In BHS it is limited to Mv, but quite frequent there.

1.88. The aorist and optative use a 3 pl. ending elsuh or elsu; also, rarely, alsu(h), ilsu(h), and lsu(h) after other vowels. These are extremely common in Mv, tho Senart regularly emends them to -nsu(h); elsewhere they are rare. Outside of Mv, the usual 3 pl. aor. ending is *isu* (also found in verses of Mv), for Skt. *isuh*; it is precisely paralleled only in Asokan inscriptions (especially Shahbazgarhi; oftener *isu*); cf. Pali *isum*. Also, less often, we find *imsu* = Pali and AMg, id.

1.89. Only in Mmk, we find a 3 pl. preterite (middle) ending *ire* (cf. Pali *are*); and, in perfect forms, *ure* for Skt. *ire*.

1.90. The root gam forms a future gamsali etc., quite common in My, not noted elsewhere in BHS, and scemingly not exactly paralleled in other dialects; but for possibly related forms see Bloch, Indo-Aryen 88 (after H. Smith).

1.91. Very often in Mv, and occasionally in other texts, the augment *a*- is prefixed to present tense forms, especially of the root *bhās* 'speak'. Even an optative form is thus provided with the augment. Cf. for Pali s. v. ^{1}a - and *amāpeli* in CP1).

1.92. The regular, and in most texts extremely common, aorist of the root $bh\bar{u}$ is *ubhūşi* (rarely Sktized as *ubhūşil*). No other known MIndic dialect has such a form, nor any form resembling it, unless we count the Ašokan 3 pl. *husu* (and once 1 sg. *husan*).

1.93. Entirely unparalleled, so far as 1 know, is the present *shhiati*, from root *shhā* (§ 28.33). It is however quite common, especially in Mv, and common enough in other texts, so that it seems reasonable to suppose that a Middle Indie form (**fhihati* or **fhihati*?) on which it is based was the regular present of *shhā* in the dialect. Once there is trace of a similar present **-dhihati* (aor. *-dhihe*) to *dhā*.

1.94. The occurrence of e instead of o, as representing Skt. final as (ar, ah), is found in the adverbs *pure* (fairly common) and *anle* (rare), and sporadically in the n. (and voc.) sg. m. (also nt.) of *a*-stem nouns and pronouns. It is also found, to about the same extent, in Pali. See § 1.32 with fn. 11.

1.95. Very common is u for (chiefly final) o (almost always representing Skt. as). For the most part it is limited to verses, probably metri causa. For its morphological range see §§ 3.51 ff. In Mv it is quite rare, and seems not to occur in prose. According to Pischel, it should be recognized only for Ap, among the dialects which he treats. It is however found in northwestern Prakrit (Dutreuil de Rhins; Niya, Burrow op. cit. § 12, in adverbs), and in all varieties of Ap. (Tagare, Hist, Gram, of Ap., Poona, 1948, p. 27). But BHS a is commoner than u; it is very common in Ap. but also known in various other dialects, at least AMg. and Mg. (Pischel 364). In BHS it seems to be nearly or quite restricted to use m.c.; it is hardly found in prose. Less common, but well established, is \tilde{a} ; it even occurs, tho rarely, as acc. sg. It is recorded in Asokan, and in all local varieties of Ap., according to Tagare p. 27 (not recognized by the grammarians).

1.96. For final am or am, not only as acc. sg. m. and nom.-acc. sg. nt. of a-stems but everywhere else (e.g. in aham, ayam), BHS may substitute u, apparently only in verses, m.c. The morphological range of the substitution is summarized in § 3.58. In BHS it seems to me that this *n* is not to be regarded as a reduction of *o*, or in any way related to o. We find BHS o for am only in a very few cases (§ 8.36) of acc. sg. m. or nom.-acc. sg. nt. of a-stems, where it seems clearly due to morphological confusion (nom. for acc., m. for nt.). The situation seems therefore quite different from that of the northwestern Prakrits of the Niya and Khotan documents and the 'Prakrit Dharmapada' (ms. Dutreuil de Rhins), where both o and u seem to occur for final am generally (evidence summarized by Dschi, see § 1.97). For final am we find u also in Dhakki Prakrit (Pischel 351) and especially in Ap.; Jacobi (San. XXVIII) tentatively proposed to regard u as characteristic of Western Ap., as against a of Eastern Ap.; Tagare shows, however, that while a is commoner in Eastern than in Western Ap., u is found quite commonly in all varieties of Ap. (see his Hist. Gram. of Ap., pp. 108, 111 ff. for a-stem nouns, 208, 242, et alibi, for pronominal forms). Since Tagare's work it seems to me clear that in Ap. generally, as in BHS, the phonetic change of final am to u must be recognized, the in BHS it is used only in verses m.c.

1.97, In NAWGott, ph.-hist, KL, 1944, Nr. 6, pp. 121 - 144, Hiän-lin Dschi discusses 'Die Umwandlung der Endung -am in -o and -u im Mittelindischen'. He regards o and u for am as equivalent, which is not true for BHS (he considers that here o 'has almost disappeared', p. 133, implying a one-time existence). nor for Asokan (Shahbazgarhi, where only o is found, not a). He refers (136) to a then unpublished monograph of his own (since published, see § 1.24 ft.) in which he thinks he has proved that the older parts of both the Pali and BHS canons are based on an old canon in an eastern, 'old Ardha-Mågadhi' dialect. I have shown above why I do not agree with him. He believes, however, that u for um is a dialectic feature of northwestern Middle Indic, following Jacobi in the now untenable assumption that it did not exist in Eastern Apabhramsa. He explains its frequent occurrence in BHS as secondary, and due to relatively late influence of some northwestern Prakrit. The only specific evidence he cites for this is a collection of about 50 cases in which, according to the KN edition of SP, the Kashgar recension reads am (or a) where the KN text has u, with some Nepalese mss. He assumes that these cases are typical and that they prove that the older (Kashgar) form of SP had am or a generally, and that *u* was introduced later in the 'Nepalese recension'. His arguments do not convince me, for several reasons. First, in many of his instances, some or even most of the Nepalese mss. of SP are reported as reading u(m), like the Kashgar recension; for such readings, Nep, can not be said clearly to support the supposedly 'secondary' u. Secondly, we have as yet only very scanty information about the Kashgar recension of SP; Dschi's few cases do not prove that the Kashgar recension avoided a for am. Indeed, it would be very strange if it did so; we should expect it to agree with northwestern Prakrits, which are precisely the ones which according to Dschi show If the precisely the ones which according to posen show u for an (tho he is wrong in denying this coange to Eastern Apabhrama with Jacobi). Thirdly, the change of am to u is common in the verses of most BHS texts, and it would take much more evidence than Dschi adduces to prove it a late or secondary feature. It is true (tho not noted by Dschi) that it is rare in My. But, the My is probably the oldest BHS text we have, not all the positive and negative formal differences between it and later texts can reasonably be explained as due to secondary changes in the latter. And u for am is not entirely unknown in My; perhaps in its original form it was much commonor, and has for some reason been using eliminated in the course of tradition. The same may be true, e. g., of u for v, and of the loc. i for e, which are also rare in our mss. of Mv. I have noted that the mss., and Senart's edition, of Mv very often read final -am in verses in positions where it is metrically impossible, because a short syllable is required. In all such cases the original text must have read either a or u for am; no one can say which.

1.98. Peculiar to BHS as far as 1 know is the very common substitution of σ for final σ in verses, m.c. It is not a phonetic change but due to morphological analogy, §§ 3.79–88.

1.99. Reference was made above to the voc. pl. ending aho, of masc. *a*-stems (cf. Mg. aho, Ap. aho, ahu), which occurs not only in Mv (fairly often) but also in SP, and even in its prose, according to the Kashgar recension. It occurs once in a fem. *ā*-stem, and there is even a case of *deviho*, voc. pl. of *devi* (both these in Mv).

1.100. The personal pronouns are, as in most dialects, extremely varied and confused: see the chapter dealing with them. Some forms have few or no correspondents elsewhere. The chapter on the generic pronouns will also present some curiosities.

1.101. The confusion of person and number in verb inflection transcends by far anything noted elsewhere; §§ 25.1 fl. It is even found in the prose of texts which present a superficial appearance of pretty complete Sanskritization, such as Divy, SsP, and Lańk, not to mention the prose of LV.

1.102. The gerund ending i, which I believe is historically derived from Skt. ya by 'samprasāraņa', seems to be known otherwise only in Ap.

1.103. Constructions with the negative particle $m\bar{a}$ have seemed to me so peculiar as to deserve a special

chapter. Most of them can, to be sure, probably be paralleled elsewhere, even in Epic and other aberrant forms of Sanskrit. I do not know of any parallel for the fairly common use of må in questions, especially as equivalent to Skt. *kaccin na*, when a negative response is desired or hoped for, as in måsi dåsi, '(I hope) you aren't a slavewoman, are you?'

1.104.. The above list could be indefinitely extended by including morphological features which are more widely paralleled in Middle Indic, and on the other hand some which, while not widely paralleled, are scantily or dubiously recorded in BHS.

1.105. Summarizing the results which BHS grammar as a whole seems to indicate as to the dialectic relations of the underlying Prakrit, we find:

(1) In a substantial number of cases, BHS features are either unique, or virtually so. Attention has been called above to the most important of these.

(2) While it has some features in common with Pali, on the whole its morphology is definitely unlike Pali in many important respects.

(3) It has a few special features in common with Apabhranisa: but its general character is clearly older than Ap., and more in line with Prakrit as a whole. The features which constitute this 'general character' are for the most part such as can be described as 'common Prakrit', or even 'common Middle Indie' (e.g. treatment of consonant clusters). In my opinion they do not justify the assumption of specially close relations between the Prakrit underlying our dialect and any other specific dialect known to us. I now believe that I was wrong in seeing special relations to Ardha-Māgadhī (BSOS 8.501 ff.). Nearly all the features which I there listed as common to BHS and AMg, are also found in other Prakrits, or Pati, or Both.

(4) No one, to my knowledge, has suggested identifying the BHS Prakrit with Ap. Since identification with AMg, has been suggested, it seems worth while to ist some striking differences between the two: BHS loc. sg. csmip (also loc, ending i representing c_i anse, c_i . AMg, anisi, recorded just once): nom.-acc, pl. fem. avo; oblique sg. fem. aye, iye, etc. (AMg, and general Prakrit have, to be sure, similar forms lacking the y), and ayo, iyo; oblique sg. are, ari, from stems in ar; gen, pl. sānam to stem sa-; 3 pl. aor. etsu(h) and isu: gamsati as fut. of gam; abhāsi, the regular aor, of $bh\hat{a}$; sthihadi, common present of $sth\hat{a}$; u for final o and am, both extremely common; o for final a m.c.; voc. pl. aho; gerund ending i.

(5) The Prakrit underlying BHS was certainly not identical with any Middle Indic dialect otherwise known to us.

The BHS lexicon

1.106. The non-Sanskrit words used in BHS, aside from proper names, correspond to Pali words in the overwhelming majority of cases. Many even of the proper names have Pali correspondents. Naturally, certain linguistic alterations are more or less regularly involved, and sometimes the meaning varies. But in the great mass of cases the meaning is the same, and the changes in form are self-explanatory; so c. g. ātmabhāva, 'body', = Pali attabhava. It may, however, be worth while to cite from the Dictionary, which should be consulted for the facts regarding each word, a few Pali-BHS correspondences which are not quite so simple, either formally or semantically: anyātaka, (a-)sāmpreya (= Pali [a-]sappāya; both forms obscure), kilāsin, kisara, kusīda and kausīdya, prahāņa, etc.; see the Index at the end of the Dictionary. Not infrequently a BHS form is helpful on doubtful, or at least disputed, points of Pali lexicography. So, e. g., BHS prasrcstha, 'very superior', supports Pali pasettha (Pv ii.9.75) against the PTSD emendation pasallha (for which the proper Pali form is pusattha = Skt. prasasta).

1.107. As stated above (§ 1.63), when a BHS word has a Pali correspondent known to me, my Dictionary cites that; and as a rule it cites no other. In the rare cases where corresponding words in other dialects give additional help towards understanding the BHS word, such words are also cited. When, however, I have found no correspondent in Pali, I have always tried to find one in some other Mindic source. In quite a few such cases a correspondent has been found in Prakrit, especially Ardha-Māgadhī, the canonical language of the Jains, and the vehicle of the oldest extensive literature preserved to us in any Prakrit dialect.

1.108. The following BHS words, on which see my Dictionary, have (more or less close) correspondents at least in AMg., often also in other Prakrits, but not in Pali: agārustha, anavarāgra (closer to AMg, than to Pali correspondent), anuragatam, andhara (Skt., Pali andhakara), anvādijati 'commands', aparādhyati 'disappears', abhyangeti, alinda, avasya (osa), asila 'inauspicious, offensive', fsi(t), utkrośayati 'exalts', utłuka(?), uttima, utthala, upapela, urasa, ulla, ūrmi, ekāhalyu, eltiya (also etlaka = Pali id.), em (for Skt. evam), eluka, kantheguna, kandita, kalāva, kallavāla, kasaļļa, kīrlika (hyper-Skt., cf. AMg. killiā = Skt. kritika), korpara and kaurpara, kosikara, khallika, khu and hu (Skt. khalu; Pali kho, before vowels khv-), -khuttam or -khutto, gomukhi, gosthika, gaura-khara, carpalaka(?), chodayali or chor°, tayin, liriccha, luna, lehika, lharatharāyale, diši, daumya, nakula (a musical instrument), nandī(-mukhā), nayuta, nišrņoti, niskuļa, parisāmanta, (and congeners), bhramarikā, miņdha, murava, moham, ballari, vāsīcandanakalpa, vidhāna, veşli, vaisranaņa, saniscara, hadi, hastala. The list does not claim to be complete, but is thought to contain a good proportion of the clear and important cases. Cf. the Index at the end of the Dictionary.

1.109. There are also words not found in Skt., Pali, or AMg., but with correspondents, more or less close, in some other Prakrit, Ap., or the Deśināmamālā: avidham or avidhā, ārāgayali, āvārī, upalagna, edānīm, kāhala, kşanika, guhmila, cakoraka, limayali, dandavāsika, dūsika, 1 dvīpika, nirvišisla, palāna, paligodha, popphala, musaņa, viparoksa, vella. 1.110. Finally, there are quite a good many words

recorded nowhere else than in BHS. Many of these are proper names, mostly of transparent formation. Many others are equally transparent and simple in formation; the lack of record of them elsewhere may often be accidental. Not a few, on the contrary, are completely obscure in formation, and sometimes of uncertain meaning; even the reading may frequently be questioned. The mss. of the Mahāvastu, especially, contain many dubious forms, which are extensively emended by Senart, often successfully, but I fear often unsuccessfully. It has not seemed to me profitable to list such problematic or obscure forms at this point. Nor, on the other hand, shall I present here formations which are made with elements and in ways normal to either Sanskrit or Middle Indic, so that the fact that I have failed to find them elsewhere may be only an accident.

1.111. I shall, however, mention a few samples of specific BHS vocabulary, tending to prove my thesis that we are dealing with a real language, not a modification or corruption of any other dialect on record, and as individual in its lexicon as it has been shown to be in its grammar. These words seem textually reliable, not mere corruptions. For the most part they are not wholly obscure in etymology. And I have found no record of them elsewhere. One of the most interesting is parindami, or parindāmi (also anu-pa°), 'l hand over, present, deliver'. It seems clearly related to Skt. paridadati, of like meaning; BHS has anuparilla, 'handed over' (but I find no anupari-dā in Skt., Pali, or Pkt.). Others, on which see the Dict., are: 1 akhila 'ungentle, harsh'; anvali, 'goes'; 2 adhyālambali, 'grasps'; anyalra (1), 'on the contrary'; apakşāla, 'fault'; abhichādayali, 'presents'; abhisāra, 'gift, honorarium'; ulplāvayali, 'leads astray'; also ulplāvaka; rg-ili, rig-ili, rf-ili, 'instantly'; rla(ka), 'prizefighter'; kākhorda, an evil spirit (Iranian Ioan?); chāyika or °kā, 'ashes' (cf. Pali chārikā, AMg. chāri(y)a, °yā); jilam, interj. of astomishment; jihma in the meanings 'obscure, deprived of light, dull' and 'disappointed, depressed', with derivatives; nirmädayali, 'washes'; palikuñc(ik)a, °cikä, ((thatched) hut (of straw)'; pudini, 'pool'; pūri, pūri, 'fulfilling, full measure'; pralivaluali, 'opposes, rejects, disobeys', and n. act. prativahana; prativibudhyate, 'wakes up', especially common in the ppp. prativibuddha; bhil(t)vara; yad ula; vidungikā; sukhila, 'happy', and duhkhila, 'unhappy'; sumbhaka, 'bowl', and sumbhalika. This list could be considerably extended.

2. Phonology, Consonants

2.1. Of the many traces of Mindic phonology affecting consonants in the written tradition of BHS, the majority fall under the head of assimilation between adjoining consonants in Skt.

2.2. In an article on Meter, Phonology, and Ortho-graphy in BHS, JAOS 66.197-206, I showed that the metrical structure of BHS verses in the older texts proves that such assimilation was much commoner than one would judge from the writing found in our mss. and editions. The meter can be understood only on the assumption that initial consonant clusters were regularly simplified to a single consonant. This implies previous assimilation, as in MIndic. We must therefore assume that in the underlying dialect such assimilation took place; and if initially, presumably also medially, tho this cannot be revealed by the meter. Note also, in the (prose) spelling lesson LV 127.5 fl., such indications of Mindic assimilation as ñakāre jñāpanašabdah (16; so mss., obviously pronounced ñāp°), thakāre thapanīyapraśnaśabdah (17), thakāre thāma .. sabdah (19); also such spellings as traya-timiad LV 61.16 (prose).

2.3. Similarly, final consonants (except generally nasals) were or could be dropped, as meter proves. In this case even the writting gives more extensive support. And in some other points, as the following will show, there is evidence that the consonantism of MIndic was, on the whole, characteristic for the dialect underlying BHS.

2.4. In general, we shall note here only Mindic processes reflected in the written tradition. Any other procedure would seem arbitrary and subjective. Occasionally, however, we shall refer to metrical facts, when they seem to help in making comprehensible the forms recorded.

Assimilation of consonants

2.5. On a rather wide scale we find evidence of assimilation of a stop or other consonant to an adjoining stop. A list of examples, certainly not exhaustive, and chiefly limited to those recognized in writing, follows. When no special reference is given, the Dictionary will show the occurrences.

2.6. kl(h) altered to ll(h): suyulta = suyukla; madhusitthena, cf. Skt. siktha.

ky to kk: śakkila, ppp. to Pali sakkali = Skt. śakyali, °le. kr to kk, initially k: koda- = kroda-; kosluka 'jackal' = kro°; kimi = krimi (or krmi; § 3.91); even in root-initial in a derivative of a cpd. verb, prakāntā = prakrāntāh.

On ks to kh, see § 2.25.

2.7. gr to gg: aviggrhila = Skt. a-vigr° Samadh p. 49 line 26 (vs). Most cpds. of root grah, in weak-grade forms, are written in our texts with single g. But in verses the meter regularly requires a long preceding syllable; suggesting that the true pronunciation was with gg as in Pali and Pkt. (analogically carried over from full-grade forms with Skt. -gr-). And this in turn implies that in full-grade forms, -gr- in compounds of grah- was assimilated to gg (JAOS 66.201; § 44). Some examples: pari(g)grhita SP 89.8; LV 158.12; 190.14; Siks 43.6, 8; prati(g)grhita LV 411.21; prali(g)grhilvā LV 387.7; prali(g)grhņe Mv i.220.13 = ii.22.14; anugrhitā SP 166.6 (so Nep. mss.; ed. with Kashgar rec. anugrahitā, perhaps false Skt. for original anugg°?); in Divy 401.18 text nirgrhita, which is certainly false Skt. for ni(g)g° = Pali niggahila, Skt. nigrhila, 'checked, humbled'.

2.8. cy to cc, initially c: vuccati, vuccanti, pravuccati (§ 2.51), uccati, etc., = ucyate etc.; ruccati = rucyati; paccale = pacyale; cavilvā = cya°; -culi = cyuli.

jy to jj, initially j: bhajjali = bhajyale; jestha- = jye°; sa-jotisa; su-jota-; lujjati.

jv to jj, initially j: jalūsilā = Skt. jvalosilā, jvāl°; jālayişyati — jvāl°.

2.9. dy to dd: kudda 'wall' = Pali id., Skt. kudya. lk to kk: ukkāsila = utk°; ukkāsali, ukkāsana; ukkaļļali; Ukkala; ukkarikā = utk°; ukkārika (nt.?) (read ukkar°?).

tm to tt: attātamiya = Pali attattaniya, Skt. ātmātmīya. tr to tl, initially t: krttima = krtrima; 1 citta = citra (Skt. cailra), n. of a month; cillaratha = cilraratha; 2 li- = tri-; lāyana, °na = trāyana, 'saving'. Is to tth, initial th: tharu = Pali id., Skt. tsaru.

dg to gg: uggami (v.l. udgami) Mv ii.91.17; uggiramāņā My ii.412.1 (mss. ugguru°, udguru°; to udgirati; but perhaps read ugguru"?).

dv to dd: aviddasu.

dhr to ddh: grddha(kûla) = grdhra°; vaddha- = Pali id., Skt. vadhra or vardhra.

2.10. pl to tl: gutti = gupti; atitti, perhaps = atrpli; āņatti(kā) = Pali id., Skt. ājňapti, also āņatta, etc.; vulta = Pali id., Skt. upta.

py to pp: palappanti = pralapyanti, which is read in a repetition.

pr to pp, initially p: pamallabandhu = pra°; panidhi; patrician propanti propanti provinci provinc

rl to ll: adhivallali, v.l. for adhivarlali, q.v. Dict.

rdh to ddh: addhaliya = Pali id., for ardha- (Dict.). rdh to ddh: omuddhaka = avamūrdha(ka); vaddhāpayitvā, °petvā, °pita, = vardh°. rp to pp: appehi = Skt. arpaya.

2.12. sc, sch to cch: pacche or paccha = pascat, pacchimaka (v.l. paści°); tiraccha, tiracchāna = Pali id. (cf. Skt. tiraścina), also tiriccha, tiricchāna; ducchana (read prob. °channa).

st to tth (th): vighuttham (°stam); samvimattha = -mrsta; simplified to th after long vowel in veth-ayati etc. = vest-. A hyper-Sktism is adhyusia, q.v. in Dict., Mvy 8172, for AMg. addhullha (Pali addhudaha).

sk, skh to kkh, initially kh: khandhävåra = skandhåvāra; khalitam = Pali id., Skt. skha°; khali (see khalati); prakhalamānair = pra-skha°.

st, sth to (11h,) initially th (cf. next): thambhanas = sta°; [hapeti etc., see Chap. 43, s.v. sthā (9), and cf. thakāre thapanīyaprašnašabdah LV 127.17. Hyper-Skt. puşkarāstika, vanāstika = Pali pokkharaļļha, vanaļļha.

st, sth to tth, initially th (cf. preceding): pallatthikā (Pali id.) = paryastikā (so Mvy 8544); -vittharikām = vistarikām; ultharya (? to ud with str); thapeti etc., see Chap. 43, s.v. sthā (9); thāpayişye, see ibid. s.v. sthā (8); cî. lhakāre lhāma . . . śabdah LV 127.19; lhera 'old man' = sthavira; Ihala = sthala; Ihandila = stha°; Ihina. Hyper-Skt. sista (Dict.) for Pali sittha (Skt. siktha).

st to t, initially: tabdha = stabdha (prose: Pali thaddha). sp to p, initially: pršati and other forms of sprš, see Chap. 43, s.v. sprš (3): $prh\bar{a} = sprh\bar{a}$.

sph to ph, initially: (priti-)phutā = °sphutā; pharati etc. = sphar°, also pharaņa. Note the hyper-Sanskritism sphāla = Skt. phāla 'plowshare' Myy 5613 (surely not an inheritance from prehistoric initial s-).

2.13. Hypersanskritic substitutes for Skt. double stops are rare; cf. some of the preceding \$\$; in My i.20.2, mss. present the strange form marj(j)a- for majja(n). 'marrow', see Dict.

2.14. As in MIndic, assimilation of a dental and a following y may yield a double palatal: praccaya = Pali paccaya, Skt. prayaya: khijjati = khidyate; khajjati = khidyate; khajjati = khidyate; note khajjati Mv i.360.1, repeated as khidyanti ibid. 3; also khajjaka; vijjhati = vidhyati; anumajjhima = anumadhyama; bhijje, Chap. 43, s.v. bhid (2). See also raccha(? -ā) for rathyā, § 2.18. For hyper-Skt. back-formations on this model, see Dict. s. vv. (abhi-)dhyāyati (Pali jhāyati); vadya(te).

2.15. jñ to $n\bar{n}$, initially \bar{n} ; for this ny may be written; also n, nn. Sporadically we meet spellings pointing to $n\bar{n}$, or initially \bar{n} , for j \bar{n} as in Pali. Cf. § 2.2; anyātaka = aj $n\bar{a}ta(ka)$; a $n\bar{a}nam = aj\bar{n}anam$; in Nv 1.351.14 mss. somanyā, a way of writing something like Pali soma $\bar{n}n\bar{a}\bar{n}$, for sam $aj\bar{n}a\bar{a}$; similarly anyāsi, mss., Mv iii.386.18, for aj $n\bar{a}si$ (which Senart reads) or $\bar{a}j\bar{n}a\bar{s}i$, aorist of ($\bar{a}j\bar{n}a\bar{a}$. In Mv 1.247.10 (vs) Senart em. sam $aj\bar{n}\bar{a}$ for mss. samanyite, a puzzling form, in a line which is unmetr. and clearly corrupt. In Mv iii.284.4, Senart reads $a\bar{n}n\bar{a}\bar{a}ya$ with the Pali equivalent (see Dict. s.v.) for mss. anyāya. For Skt. $aj\bar{n}\bar{a}$, $\bar{a}j\bar{n}apla$, 'ti, and related words, Nv repeatedly reads $an\bar{a}$, $\bar{a}natla$, 'li($k\bar{n}$), etc. as in Pali. Isolated is sanniyate for sam $j\bar{n}\bar{a}yate$: § 37.3 n.1; for the nasal see Pischel 276.

2.16. Other MIndic consonantal assimilations, occurring in writing sporadically:

 n_i^a to n_i : sanna = Skt. sanda, 'thicket', Gv 408.4; see Dict. This type of assimilation is found in the NW Prakrit of the ms. Dutreuil de Rhins, Senart, JA Sept.-Oct. 1898, 212, top. I have failed to note any other case in BHS.

[my to mm: sammā = Pali id., Skt. samyak, assumed by Senart My i.211.7, but only by a false em., see Dict.]

mr to mm, initially m: -makşilā = mrakşilā. ry to yy: ayyaka = Pali id., Skt. āryaka; payyaka =

Pali id., Skt. *prāryaka. ry to ll: pallallhikā = Pali id., Skt. paryaslikā; pallānati 'harnesses', based on Skt. paryāņayali, denom. to paryāņa 'saddle'.

rl to ll: dullabha = durl^o.

ly to II: sallayati, denom. to salya.

vr to (initially) v: pra-vajila = ^ovraj^o (except one doubtful Pali form pavajamāna, see PTSD, only bb and vv are recorded in this base in Pali and Pkt.). sm to (initially) s: sarati = smarati.

Sibilant plus r or v to theoretically double, but initially single, sibilant: samsati = sramsati; sutā = śrutā; pra-savanti = pra-sravanti; a-saddadhāno = a-šrad°; sāpateya = Pali id., Skt. svā°; setaka = Pali id., for śveta-ka; prati-śaya = prati-śraya (śr initial in root).

Semivowels 'lost' in triconsonantal clusters

2.17. Doubtless really assimilatory in character, but perhaps in part mere errors of tradition, are cases in which y, r, or v is lost after (less commonly before) a consonant cluster, or at least what is written as such.

Loss of y. On ks: ksy see § 2.23. Otherwise note kāmsa- and kamsa- 'brass', for kāmsya: antaka (cf. Pali anta, Skt. antya).

Loss of r. BHS repeatedly presents bhals- for Skt.

bhurts-. For vardhra, even Skt. records vadhra, which is also found in BHS, with vadhri; these are probably semi-MIndic forms. Pure MIndic (Pali vuddha) is vaddha-palikā for va(r)dhra-. Also ucchethā, prob. for ucchrethā = ucchrayatha or better ucchrayadhvam, 'arise!', see Chap. 43, s. vv. chid (3) and śri; jantu- = jantra, yantra-, jantitā = yantritā.

Loss of v, after ks: praksedā, praksedita = °ksved°.

ch

2.18. As in Mlndic, ch replaces not only Skt. ic (§ 2.12), but also ks and is; rarely sk (chambh- = Skt. skambh-). Examples of ks: lacchila, lacchaka (laks-): ruccha = rūksa (in Pali only lūkha: AMg. lūha and rukkha; no ruccha recorded in Mlndic for this word); kaccha = kakşa. - Is: ucchanga- (pāda, or -caraṇa), one of the 32 lakṣaṇa; variants ucchakhu-, ulsaṅga- (Pali ussaṅkha-); ucchada(ka); ucchahali, °le, = ulsahale; kucchanli = kulsa(ya)nti; nirbhacchila = nirbharlsila (Pali nibbhaccheli); (a-)macchara = malsara; maccha = malsa; ucchava (AMg. id., but Pali ussau).--Also for lhy: raccha = (or read) racchā (as in Pali) = ralhyā.

2.19. Note also the hyper-Skt. utsista for ucchista; jugutsu for jugupsu (MIndic cch falsely Sanskritized as ts instead of ps); utsta for ucchrita.

2.20. In Skt. ch is always a long or doub'e consonant, whether written cch or not. In BHS the meter shows inconsistency. At the beginning of a word, it is to be sure always single; that is, when a short vowel precedes, that syllable is short. So also in cpds.; e. g. not only kileśachedani LV 53.13, and achidra (short first syllable) Bhad 17, but pratichādā (short antepenult) RP 46.18, and in verb compounds, ucclosyatu Suv. 53.10; vicchidyatu Suv 53.12, etc. But even in the middle of a word the syllable before (c)ch may be short, as in *icchali* LV 45.11; gacchata LV 36.19. On the other hand, it is long, as in Skt., in *icchali* LV 46.3, 9; various forms of gacch- SP 113.7; LV 50.7; 74.6, 10; 78.15; 81.11, etc.

2.21. Sporadically, as in Pali, s appears for (c)ch in a (Skt.) consonant cluster: kisara = Pali kasira (beside kiccha) = krcchra (Geiger 59.2).

2.22. Merely graphic corruption seems to me concerned in writings of *cch* for *llh*; see Dict. s.vv. *illhalva*, *ucchihilvā*, *nipacchila*(?).

kş, kşy, ş, kh

2.23. We find both ks written for usual ksy, and vice versa. This probably is at least in part a matter of phonetic uncertainty (perhaps imperfect Sanskritization), interchange, or corruption. But it may also have morphological bearings. I have recorded ksy for ks only in present forms of root iks (see Chap. 43, s.v.); these may be 4th class presents (§ 28.28). Conversely, futures containing (k)s instead of normal (k)sy could be classed with other futures lacking y (§ 31.26). Then we need not regard as purely phonetic such forms as prativaksanti LV 88.14-15 (so read for pratipakşam pakşanti, cf. Weller 23, 41); pravekşi for °ksye LV 223.4 (vs, both edd., no v.l.); draksase LV 237.10 (all mss., only Calc. draksyase); less certain is LV 396.8, where Lefm. reads by em. naksyate ... vinaksyate (the corrupt mss. all lack y; I conjecture (vi-)namksale = Skt. (vi-)nankşyale; fut. of naš).
2.24. But at other times kş for kşy can hardly be

2.24. But at other times kş for kşy can hardly be anything but a purely phonetic (or else graphic) matter: nirikşa LV 341.18 (so all mss.; both edd. nirikşya; ger.); asamikşa-kāriņaš Mv 1.90.5 (for asamikşya, ger., which is read in one inferior ms., out of six).—antarikşa, as adj., 'atmospheric,' LV 266.1, and antarikşa, ibid., LV 367.7, probably represent Skt. āntarikşa (or °ikşa, rather than antarikşı, RV., or *āntarikşu or the like). for Skt. kş, as in śekheli, to śaikşa; khudrāka = kşudraka; sukhama, sukhuma = sūkşma; prākharati or prakharati = prakşarati. There are false Sanskritizations such as prakhyālana, for MIndie (Pali) pakkhāl-, Skt. prakşālana; and ukşa, for Skt. ukhā (Pali and AMg. sporadically ukkhā); akṣaṇa(-vedha) for Pali akkhaṇa = Skt. ākhaṇa (Diet.); saṇkṣayati (Diet.), perhaps for MIndie saṇkhāyati = saṃkhyāti. For kāṃkṣāṃ My i.162.7, two mss., including one of Senart's two best, read kāṇkhāṃ (Skt. kāṅkṣāṃ). In lāha (AMg. id.) = rākṣā we have Prakritic replacement of this kh (BHS also lākhu) by h.

2.26. There is some replacement of $k_{\$}$ by $s_{,\$}$ which perhaps means the same sound as kh (cf. the references in Renou, Gr. Scte. p. 4): sampresale, all mss., certainly meaning sampreksate (confirmed by Tib.); pravesye, for praveksye; abhinivesyali, for °veksyali, to °viádi; in part thru confusion with forms of *sis* (see Dict. s.v. *sisyale*), *sisyā* = *sikşā*; *sisyāpada* = *sikşāpada*; *sisyāpayati*. A special case is abhispam or abhispam (so all mss.) = abhikspam; the original text evidently had a half-Sanskritized form of the Mindic word found in Pali abhinham. And by the converse (hyper-Skt.) process, preksane = presane ('sending forth') LV 432.18-19.

2.27. On ky to cch see § 2.18.

Voiced for voiceless stops

2.28. In the manner of Prakrit, BHS occasionally has voiced for Skt. voiceless stops. Pali shows the like sporadically (Geiger 38), sometimes in the same words: so the root vedh = Skt. vyalh appears in both BHS and Pali, but BHS also has wyadh; and niryādayati, once for niryāt° (Pali usually niyyādeli). In other cases the BHS forms occur in Pkt. the not in Pali; but some are peculiar to BHS, so far as my information goes. Sometimes BHS has a Sanskritic form where Pali has a Prakritic one; see pratikriya (Dict.) for Pali paligacca. Besides the above I have noted: sadha (or sadha = AMg. id.) = salha; pradiksanle = praliksanle; uposadha, posadha, and derivatives, = Pali (u) posatha (forms with dh are also recorded in Jain Skt.); daga = (u)daka, in dagodara; parijaya = paricaya; $p\bar{a}r\bar{a}jika$ (= Pali id.) if this is derived from a form of parāňc-; rugina-(varņa) 'gold' = rukma-; udaya (AMg. id.) = ulaja; ulpādu 'portent' = Pali uppāda, Skt. ulpāla; upaga = upaka is not purely phonetic, see Dict.

2.29. By hyper-Sktism, or mere corruption, t is recorded seemingly for original d in *utacchigam*, Dict.; perhaps also in *pratara* = Pali *patara* for *pratara*. Anomalous, but seemingly acceptable, is *phalikha* (AMg. *phaliha*) = Skt. *parigha*.

v for p and m, and for y

2.30. As in Prakrit, so sporadically in BHS, we find v for Skt. p. It occurs in quite a group of causative forms, showing -veti instead of -peti (§§ 38.68-70). Other cases: pravanna = AMg. pavanna, Skt. prapanna; suvarna = suparna, and suvarnin = suparnin; anupråvetsuh (to anupråp-); vi perhaps for (a)pi; dandaväsika (Pkt. ^oga, ^oya) for Skt. ^opäśika; vaiyāvrtya regularly for ^oprtya, as in Pali veyyāvacca; perhaps by hyper-Sktism, p for v in pithi = vithi, and prajāpali (^ovali); also v for m, śravana = śramana (and conversely once śramana for śravana); cf. Anu-vaineya, a village of the Maineya people; samwrli (hyper-Skt.) = Pali sammuti (root man).

2.31. Rarely v seems to occur for intervocalic y (Geiger 46, Pischel 254): *jivati* = *jiyate* 'is conquered'; *āvusa* (Pali 'so), *āvus* (? = *āyus*). This is also presupposed by pubba = Skt. paya, 'pus', thru *pava, *puvva, with Palilike change of vv to bb, according to Senart; but the true reading may be puvva, see Dict. Cf. also the ending $avo = \ddot{a}yo$, § 1.28: 9.93.

y for intervocalic stop and vice versa

2.32. As regularly in Pkt. and sporadically even in Pali, we find (rarely) y for an intervocalic stop: tâyin, regularly for Pali tâdin, AMg. tâi; khâyati, khâyita = khâdati, khâdita (as in Pali khâyita beside khâdita); āchâya = āchâda 'gift'; (na-)yānāti = (na) jānāti; indaya (AMg. id.) = uļaja; ulacchiyam for ud-acchidat(?). In viprotyaniya, etc., for °nika, the change of k to y involves assimilation to suffixal -iya.

2.33. By hyper-Sanskritism, and attraction to forms in suffixal k (Pkt. y or zero), we find k for original y, in -bharikā 'wife', from Mindic bhariyā = Skt. bhāryā; harmika = harm(i)ya; ralanāmikām 'made of jewels', fem. to *ralanāmaka = BHS rat(a)nāmaya, Pkt. rayaņāmaya, Skt. ratna-maya; and for original j in ātmakā — ātmaja,

j for y and y for j

2.34. For y, we occasionally (the rarely) find j as in Prakrit, both initially, as in *janta-* \approx yantra-; *jantilā* = yantrilā; *jakrt*; *jāpayali* and *jāpanā* \approx yāp; and medially in *peja* (Pkt, *pejja*) = *peya*; *niryāla* and *nirjāla* are partly blended. By hyper-Sanskritism, or by dialectic Prakritism (Pischel 236), y for j occurs in *aneya* for Pali *aneja*, and in *sārāyaņi(ya)*, Dict.; probably also in *Yanbhaka*. A bizarre form is *anārjaņ*, apparently for *anāryaņ*; see Dict.

Aspirates and h

2.35. Sporadically, as also in Pali, h is found substituted for aspirate stops, as regularly in Prakrit. So in some compounds of root $dh\bar{a}$, especially $antar(a)h\bar{a}gate$ etc., also dahati, prahāņa; jalāhara = jaladhara; gāha = gādha; lāha (AMg. id.), also lākha, = Skt. rākşa; lahu etc. = laghu; moham = AMg. id., Skt. and Pali mogham; guhmitā, to Pkt. gumh-, Skt. and Pali gumph-; vihu, for vidhu or vibhu; samhāta for samghāta; vahati = vadhati. As in Pkt. (Pischel 267), h may be replaced by gh after uasalization: samghata = samhala = samhala Gv 55.1.

2.36. By hyper-Sanskritism, viyūbha is written for normal BHS v(i)gūha, Skt. vyūha. But idhu (Pali id.) is older than Skt. iha.

2.37. The Pali form of the word for 'dog', sunakha, occurs repeatedly in Mv. It is commonly assumed that kh is here a modification of k.

| for d for Proto-Indic j^1 (IE, \hat{g})?

2.38. Extraordinary interest, but also some dubiety, attaches to the form bhisalka, 'physician', recorded with-out v.l. thrice in SP, = Skt. bhisaj, Pali bhisakka, BHS once, at least, *bhisailka* LV 285.1 (vs). The j of *bhisai* is historically from IE. \hat{g} , but seems in Vedic and Skt. always to be analogically assimilated to the other j, from IE. velar or labiovelar (Wackernagel I pp. 161, 174). If we could rely on bhisalka, were it only as a hyper-Sktism for MIndic bhisakka, it would be, to the best of my knowledge, the only record in Indic forms of this word showing the historically expected consonantism. Unfortunately, Senart Mv i note 367, says that for the last two passages cited from SP, the Paris ms. used by Burnouf reads bhişaka (metrically impossible) and bhisanka (intending bhişanka, as in LV 285.1). Yet one cannot help wondering where Kern and Nanjio got their reading bhisalka, allegedly found in all their mss. There seems to be no known Skt. form which could have given a basis for analogical introduction of *l*.

Domal and dental stops and nasals

2.39. Our mss. often write n for Skt. n, e. g. sonita = sonita LV 262.2, 3 (all mss.); anumātram for aņu° LV 261.18 (all mss.; Pali often reads anu for aņu, but CPD

calls it a false reading; the Pali vs corresponding to this one, Sn 431, reads anumaltena, with v.l. anu-). To some extent this may be due to over-correction of MIndic n, which in many dialects replaces Skt. n. (So also e.g. BHS and Pali -pona = Skt. pawana.) It may partly be due to errors of tradition. According to Senart Mv i, p. xii note 1, in Nep. mss. hn and hn are indistinguishable; and cf. *ibid.* p. xiv, infra. I have not, in general, collected variants of this sort.

2.40. A case of Prakrit double nn for nn is -pravannaih (= prapannaih), to be read Mv i.84.3; AMg. pavanna.

2.41. Domal *t* for *t* occurs in $pa | \bar{a} k \bar{a}$ 'banner', well attested, perhaps influenced by pa | a; in $va dhy a - gh \bar{a} | a | ka | = -gh \bar{a} ta (ka); gh \bar{a} | in; cf. the vv.ll. <math>gh \bar{a} | tha$ at Mv ii.169.6; $gh \bar{a} | \bar{a} | ha$ other related words, all for $^{\circ}gh \bar{a} | c$, seemingly not paralleled in other dialects; perhaps blended with root gh a | c for similar changes see Geiger 42, Pischel 218 fl. (but usually where Skt. had *r* in the vicinity).

2.42. As in MIndic, th sometimes results from assimilation of st(h), by the side of th; see § 2.12.

2.43. Domal for dental due to preceding Skt. r, as in MIndic; valla (Pali id.) = vrlla, and cf. -daha below.
2.44. Domal d for d has been noted in gadā 'club';

2.44. Domal d for d has been noted in gadā 'club'; this spelling does not seem to be recorded elsewhere. Also in (*Deva*-)daha, 'pool', for Pali, AMg. daha, Skt. hrada (Lex. draha); the domal d. tho historically quite normal, seems to be recorded nowhere else.

2.45. The converse of this, dental d for domal d, occurs at least once in the mss. of Mv, with $n\bar{a}da$ for regular nada or nala. Once, in LV 158.5, acc. to Lefmann all mss. read kathina for kathina (so Calc.). No such reading seems to be recorded elsewhere.

d (d), l, and r

2.46. Very commonly, domal d is written for normal intervocalic Skt. l. This reverses the regular Pali change of intervocalic d to (domal) l. Most of the words which show it are etymologically obscure, and there is nothing to disprove a suggestion that d was more original than lin them. Only one, *gugala* (for which LV 337.9 has *gugada* in most and the best mss. and Lefm.), seems pretty surely to contain an IE l (and even here one might conjecture some influence of the Ap. suffix da, Pischel 599). Note also *jada* = *jala* 'water' LV 372.15, but the IE etymology of this word is not certain. Others: *argada* (and *nir-a*°), *āvadi*, *upalādana*, *upalādayali*, *kadatra*, *kadevara*, *kāda*, *gadila* (and *vigadila*), *cakravāda*, *coda*, *tāda*, *nadinī*, *nāda*, *pravāda*, *Badi*, *ladita*, *lād*, *ludali*, *viradīkrla* (to *virala*), *velāda*, *sydāa* Conversely, *l* appears for Skt. d in cpds. of Skt. di, 'fly'; see Dict. s.v. 1 -*liyali*. 2.47. There is also contusion between domal d and r.

In Pkt. padi- (corresponding regularly to Skt. prati-) has been shown to occur as replacement for Skt. pari-, see Emeneau, JAOS 51.33 ff. I have found a few cases that appear as cpds. of prati-, in which BHS itself regularly has pari-: pratinirvāti, pratisphuļa and pratišodhayalī (°yitavyam), instead of regular parinirvāti, parisphuļa, parifodhayali (these forms occur in the vicinity of the prati- forms). With these belongs prālicāraka (read prati°?), 'attendant', = AMg. padiyāraga, but Skt. and Pali pari°; pațipăti (Pali id.), Skt. pari°; perhaps praliframa = pari°. All these may well be Sktized forms of Pktic. *padi-, altered from pari-. Perhaps paristhila (for pralisthila, once) shows the reverse change. Note also the variant forms parihājaka and °hāraka, some sort of ornament; they are surely forms of what is fundamentally the same word, and a Pkt. *°hādaka may mediate between them; but popular etymology is likely to be concerned also, cf. Skt. hātaka, 'gold', and hāra(ka), 'necklace'. In SP 151.9, KN

read -achodila, 'set, inlaid, covered', which if correct would replace an *achorila related to Skt. (KSS.) achurila; but no ms. has the reading of KN, and most of them read achadila, 'covered', which should probably be adopted. WT, however, keep the KN reading; their note cites Tib. as bris pa 'painted' or the like. **2.48.** The Pali anamatagga, Pkt. anawadagga or

2.48. The Pali anamatagga, Pkt. anavadagga or anavagagga, appears regularly in BHS as anavarāgra, apparently based on the Pkt. form, with r for d (influenced by thought of avara?).

2.49. Interchange occurs between r and l, in both directions from the Skt. standpoint. The Dict. will supply references on the following: (l for regular Skt. r) aňkula, kala (= kara 'hand'), Kubela, vicálana, paňjala, pali-(= pari-; various items), pāudala, piňjala (?), luiyale etc., lūha, lūkha = rūksa, valganā (?), višālada, sarkalā, šikhala; in several words regularly derived from root car-, Lańk substitutes corresponding forms of cal-, which in itself is familiar in Skt. but is not used there in these derivatives; see Dict. s.v. cal-; (r for regular Skt. l) abhinīra, kāra (= kāla 'time') and (derivative) -kārīka, vikāra (s.v. vikāla), tiraka, narada, paripārayati (= °pāl[°]), palvara, raghu = laghu, rajasvara (= °la?), vatsara, varāhaka, vargu, virolayati, šītara, śrgāra, sakara.

Prothetic v

2.50. Before weak-grade forms in u- from certain roots in va-, namely vac, vah, vas 'dwell', and vap, a v is prefixed, as in Mindic. In my opinion these are blend forms, primarily based on Skt. weak-grade stems (Skt. uc- or uk-, uh-, us-, up-), but with v prefixed by influence from the full-grade (Skt. vac- etc.). (Pischel and Geigee-have different explanations.) The forms concerned are, in other respects, sometimes straight Mindic, sometimes normal Skt., sometimes neither (imperfectly Sanskritized). They are specially common in Mv but occur sporadically elsewhere (chiefly in verses except for Mv).

2.51. Root vac: vuccati = ucyale (frequently with v.l. -cy- for -cc-) Mv i.22.8, 9; 26.14; ii.69.4; 240.3; 461.16; 463.16 (twice), etc.; KP 81.14; vucyati Mv i.361.21; 362.10; 365.19; vuccanti SP 131.6; vucyanti Mv i.362.9; iii.92.3; vucyāmi, 'I am told', Mv i.362.19; pravuccati Mv ii.358.1; pravuccate SP 130.4; pravucyate SP 255.9; vutta = ukta Mv ii.437.13 fl.; sampravulta SP 129.10, 12 (so ed. em., Nep. mss. sampravulta, hyper-Skt., \S 3.95); vukta KP 14.14; 126.4.

2.52. Root vah: vuhyali = uhyale Mv iii.384.12; vuhyasi Mv iii.105.12; vuhyanli Mv iii.314.12; vuhyale (mss. ⁹to) Mv iii.453.15; vuhyanlī (pple.) Mv iii.166.10; vuhyanlīyē = uhyantyām Mv i.342.15; vuhyantasya Mv iii.105.8; vuhyanto id. 11; vuhyanlam iii.431.6; vuhyamānam iii.421.1.

2.53. Root vap: vulta = upla Mv iii.360.14.

2.54. Root vas, 'dwell': vusta, ppp. (doubtless an imperfect Sanskritization of the MIndic form exemplified by Pali vuttha, 'dwelt'; cf. Epic Skt. usta, prec. by vāsam (mss.), Mv iii.433.2; upavusta (mss. opa°; BHS also upoşita) Mv iii.216.5 (= Pali upavuttha, in equivalent vs DN 2.244.3); dvustam, so read for avustam LV 388.13 (vs, but first syllable metrically indifferent; = Pali āvuttha); in Mv iii.185.9 Senart reads samvutthā (rather n. sg. °ttho?), the Pali form (mss. vusto or samvutto); vustā Mv iii.325.11 (so mss., may be kept, as gerund, see § 35.52; Senart em. vusto).

2.55. Senart assumes $v\bar{u}dagra = udagra$ Mv i.83.8 and 154.7. But in 83.8 the mss. read *ananlavud*^o, which I believe means *ananlav(ān)* ud^o; in 154.7 they read *bhava* ud^o (without any v). Meter requires long initial syllable both times; read, doubtless, $\bar{u}dagra$ m.c. But there is no reason to assume prothetic v.

Sibilants

2.56. Interchange between the three sibilants is common even in Skt. mss. Especially between f and s, in both directions, it is very much commoner in BHS; so common that it would be fruitless to collect many examples, especially since corruptions in tradition are very much to be suspected in this case. They are generally ignored even in the Dict. A few examples only are here cited.

2.57. ś for regular Skt. s: pariveśilum and °śakair Divy 86.18 and 20, for °veş°; anuśakłam Divy 422.14, mss., for °şak° (so ed. em.); samudeśyałe Sikş 102.10, 12; 103.7, for samudeşyałe (fut. of sam-ud-i), cited from Gv 482.16, 18; 483.2, where text samuddeşyałe, samudeşyałe (2d ed. °dveşyałe), and samuddiśyałe, respectively.

2.58. ś for regular Skt. s: (in forms and derivatives of root sru common even in Skt.) praśravanii LV 76.16 (all mss. and both edd.); 251.7, 8; mrdukasugandhaśravā(sya) LV 49.20; äśrava (see Dict.) is perhaps the prevalent spelling for the equivalent of Pali äsava, e.g. anāśrava LV 242.11; pāmśu LV 74.16; 190.15; iūkṣma LV 163.3; 395.20 (here both edd. sū°, but all Lefm.'s mss. śū°); 422.15 (here both edd. with all mss. śū°); Mv i.158.9; (nih)śarana LV 175.2 (here Lefm. em. °sar°); 180.14, 15; 346.2; apāśya = apāsya, 'getting rid of', LV 48.7; adhyavašitāh = °sitāh LV 207.7 (all mss.).

2.59. s for regular s (not very common): -kilesām = klešā $n \perp V$ 164.9 (all mss. but one); krsā $nga = krs^{\circ} \perp V$ 188.8 (all mss.); parāmrsati, 'touches', = ° mrs° Mv i.284.5; parāmrsya Mv ii.412.5 (twice).

2.60. s for regular s (rare): nyașit LV 271.19 (all mss.), 'set down, provided' = nyasit (so only Calc.; aor. of ni-as); more doubtful is nirașta, ppp., Dict.; avabhāsāta = °sāt Divy 92.29. Note that s is not usually substituted for Skt. s after e, o when these are Mindic for aya, ava. But occasionally this substitution seems to occur; see Dict. s.v. oșila, apparently for avasita; kirleşy (alitân, so divide with WT and Tib.) SP 63.9.

2.61. A special case is \$! for \$!, in $anu(!)tra\$!d\hbar = anultras!d\hbar$, 'unalraid', LV 12.16 (both edd., no v.l.); a\$!amga, n. of the western mountain (Skt. asta-, Pali althamga) LV 390.8, prob. used here for the sake of word play on a\$!am artham in the next line (a\$!a, ppp. of al, attain); see Dict.

2.62. s for regular \$ (not very common): saknosi for $\hat{s}i$ Divy 129.2, 13; 279.23; bhesyanti Sikş 45.12; 47.9; bhesyāmo 46.16; bhavisyāmo 46.18, all for $\hat{s}y^{\circ}$ (fut.). A hyper-Sanskritism, which reverses the process of § 2.61, is found in vusta for Skt. usfa, § 2.54.

2.63. s for regular s (extremely common): (a)drsu (to drs) LV 27.22; yoniso LV 37.12; vasa- = vasa- LV 45.11; Mv i.156.1; vasam Mv i.129.1; pesma LV 49.7; sayānām LV 56.19; -yasā(h) LV 59.4; āsā = āśā LV 65.2; 221.21; 222.22; My i.156.18; sungibhūta LV 76.10; nasyet LV 81.19; nikāsa LV 92.12; sabala LV 95.1; nihsvasati = nihšvasati LV 103.22 (all mss.); 104.2 (most and best mss.); 104.4 (best mss., but majority "ivas"; Lefm. "ivas" all three times); socami = śocāmi LV 111.7; mahāsāla LV 117.17; siri = sri LV 122.16, 18, and often; sobhate LV 122.19; sobhami LV 137.16; silibhāvah LV 125.3; vyupasamād LV 129.3; sāntam LV 153.18; samatha LV 156.5; 181.19; sākiya = Šākya LV 133.5; 135.7; svašrau ... svašure LV 139.11 and svašrūm ... svašuram 157.11; prsisyali LV 153.18; sosisye = šos° LV 164.8; sveta- LV 188.8; prasnāna = prainānām, so read with nearly all mss. for pramlāna (talse em.) LV 370.16; vasihi, to vasi, 'ax', (but vasi sometimes is read in Skt.) Mv i.5.4; sunakha, 'dogs', Mv i.5.1 etc.; sarad(a) Mv i.74.8; samka Mv i.151.5 = 224.16 = ii.27.16 (in the last Senart reads sanka with one ms.); vāgīsena (: īśa) Mv i.163.12; sisya Mv i.305.18.

Nasals and anusvara

2.64. There is much confusion in writing between the anusvara sign, which I transliterate m, and both m and n. esnecially final, but also in medial position before consonants. In endings, n is also written for historic m, and vice versa (next §). When a consonant follows, or in absolutely final position, I cannot discover any evidence of real linguistic significance in these variations. They seem to be purely random and meaningless. Before vowels, on the other hand, we shall see that final m (or n) was not interchangeable with m; the latter made a syllable closed and hence metrically long. (The mss. also sometimes write mm, which seems to me only a typical careless writing. I doubt its correspondence to any phonetic reality, such as is assumed, solely on this ground, by H. Smith [see my § 1.38, fn. 15], p. 3.) But this occurs only at the end of words, not internally. There is, however, no evidence in our texts suggesting differences between what are called 'nasalized vowels', anusvāra, and anunāsika (cf. Wackernagel I.1 §§ 223-4; Pischel 178 ff.). We shall use the single term anusvara and write m; to do anything else would be purely subjective and speculative.

2.65. The regular replacement in Mindic of final nasals by anusvāra (§ 2.68), except sometimes before an initial vowel, leads in BHS now and then to final *n* instead of *m*, and vice versa, at least in the orthography of the mss. Cf. JAOS 66.202 (§ 51). So *iraddhān* LV 294.17 (end of a line) for *iraddhām*; ramantān LV 43.6 (end of a line) for ramantām, 3 pl. impv.; tām (for tān, sc. adhyāṣiayān) udāhare Mv i.77.6 (Senart reads tām, without ms. authority; some mss. tāmm); dipyantām (mss., before i-) Mv ii.28.6, for °tān, acc. pl. masc. In Mv i.169.18; 170.5 Senart regards kalpakotīm asamkhyeyām as acc. pl., but they may be acc. sg.: 'for a numberless crore of kalpas'?

2.66. A reflex of anusvāra for nasal consonants may be the frequent writing (here generally ignored as purely orthographic and sporadic, and linguistically meaningless) of n for Skt. anusvāra, in such words as mānsa Mv 1.7.6; LV 49.22 etc.; pānsu Mv i.94.16, and often; also hansye = hansye, fut. of han, LV 311.18. Cf. Senart i p. xvi, and JAOS 66.202 (§ 51).

2.67. In this same place Senart notes the frequent occurrence of t for anusvära (or BHS n) before s, which he nowhere accepts in his edition, writing always n (for either n or t of ms.), or η . Senart suggests that this t for n before s may be merely a graphic error, but adds nevertheless that it may be connected with the Vedic development of t between internal n and s. It may occur in any word containing Skt. ηs , as matsa = mansa = Skt.mansa, v.l. in Mv iii.269.8; *vihalsyase*, to be read with best mss. for Lefm. *vibhartsyase* (em.) LV 335.1 (Skt. *vihamsyase*, fut. of *vi-han*; § 31.24). It is extremely common in 3 pl. aorist forms in -(su(h) for -nsu(h) and - $\eta su(h)$; for these Senart, wrongly as it seems to me, never admits the reading -(su(h) in his text. It is less common, but occurs sporadically, in other texts than Mv. See §§ 32.96 ff.

2.68. As in MIndic generally, anusvāra is often used instead of any final nasal. This seems to be more than a merely orthographic matter. For it occurs before vowels, in what must have been close juncture, and in prose as well as verse. In prose it is, to be sure, chiefly Mv that writes anusvāra before vowels: yam asti i.3.13; abhātam abhyākhyānam 45.11; prathamam adāsi 48.15, etc. (very common). But note also gaccham (= gacchan) upakārī bhavati Bbh 31.19-20, prose (for normal Skt. gacchann; close juncture), in a text which on the whole is quite thoroughly Sanskritized; and elsewhere.

2.69. Most texts make use of this practice in verses for metrical convenience. It is absolutely standard practice in all verses to use final m before a following initial vowel

2

nasalization of a short vowel.¹ In our language, which in this respect seems to differ from the general run of MIndic, such unhistorical doubling of consonants seems to occur only when meter requires a long syllable, and is furthermore very nearly limited to word-final plus initial position, including, of course, the seam between parts of compounds. It is balanced by a corresponding simplification of double consonants.

2.78. Consonant, generally initial, doubled after short preceding vowel, m.c.: so caj jino SP 25.11, read thus (or, so co jino) for KN sā co jino (em.; text jano); WT so co jino with their ms. K'; all KN's mss. are reported as reading ca (unmetrical) except one caj; the same reading must be adopted SP 193.3 and 9 (both edd. so cā jino, without ms. support), as well as in SP 68.7 where both edd. read so caj- (tho the mss. vary); same also Samādh 8.28; so cal tahī SP 294.5, read thus with 3 mss., both edd. cā with 1 ms.; kadāci pij (= api) jāti- SP 48.8, read thus with WT and all Nep. mss. for KN °pi° (unmetrical).

2.79. Common in many texts is dasad-disa-(-disi, etc.); the mss. and editions show inconsistencies, but the overwhelming weight of evidence proves that when a long syllable is required in this combination, even if the mss. read $dasa-di^{\circ}$, doubling of the d is to be assumed. A few examples: dasad-disasu SP 32.14; 194.9 (Kashgar rec. $dasand^{\circ}$); 222.3; 251.2; Gv 475.19, 21; Dbh.g. 29(55).25 (so, acc. to note, Tib. and Chin.; mss. fatad-dissu); dasaddise SP 190.13; 217.13; dasad-disi Bhad 1, 9, 10, 13, 14, 15, 47; Sikş 259.14; Gv 57.21, 23; 215.6, 13; 230.21; 254.25, etc.; also dasasud-disasu (= dasasu di^o) SP 55.11; 207.11 (here KN dasasu di^o without v.l., unmetr., but WT -ddwith their ms. K'); 222.7. The unmetrical dasa-disah of Gv 34.14 is, therefore, probably to be read dasad-disah; but dasā-disah would also be possible, of course, as in LV 219.13.

2.80. In LV 197.3 text midhamgiri, v. l. (ms. A) intends midhaggiri, both = midhagirin, 'mountains of dung'; in Sikş 103.2 trimalattamopaham (so read for ed. "pyaham) = trimala-tamo", in citation from Gv 482.23 which reads trimalam-(in ed. separated from the following)tamoharam (nasalization instead of doubling of consonant); Sarvajjagābhimukharūpa Gv 285.9 (2d ed. Sarvaja", unmetr.; n. of a Jina, = Sarvajagadabhi"); triyadhvanayappatheşu (misprinted "spatheşu) Gv 313.26 (= tryadhvanayapath-).

2.61. When the initial consonant is a sibilant, it may be doubled by itself, as in kanlakabhasmalrnaisayanānām Sikş 332.7 (from Ratnolkādhāranī), for 0 tra-say 0 , 'of those making their beds on thorns, ashes, or grass'. More often, visarga is written before the sibilant. The two speilings obviously have essentially the same linguistic meaning. hantah (= hanta, interjection) śrnusva LV 200.6 (so text, correctly, with ms. A only; most mss. hanta, unmetr.); diso daśah sarva bhaveyu pürnā SP 32.2, 'if ali the ten directions were filled'; all mss. daśah, kept by WT but explained as f. nom. pl. (1); KN em. daśa (unmetr.); jinahsakāse LV 393.9, so (or ${}^{\circ}$ te) read with best mss. (Lefm. jinasakāse, unmetr.); buddhahsatā, unmetr.); -sialahsahasrā Dbh.g. 41(67).15 (m.c. for -satasah^o); dharmahsamudra-GV 314.7 (m.c. for dharmasam^o).

2.82. In the middle of words, I have noted samuddita, seemingly m.c. for samudita (Dict.), and otherwise only rjju for rju; in Mv ii.63.15 the reading indicated somewhat corruptly by the mss. is certainly asa/h' (or asadh') mss. asadh') rjubhūto; and in Mv ii.80.13 for the un-

1. The corresponding MIndic conditions are perhaps best set forth in Geiger 5, 6, 24 (but I question the pertinence of word-accent, which Geiger assumes here), 32, metrical rjubhūtah we should doubtless read rjju^o. But Pali ujju (beside uju) seems not to be limited to occurrence m.c.

2.83. Different in character is the writing of yy after e, as often in Pali. This is nothing but a special case of the 'law of morae'; long e plus y is replaced by short e plus yy, but long and short e are written alike. Hence the writing of *-eyya* etc. in optative endings, regular in Pali (and in Pkt. changed to *-ejja* etc.), and sporadic in BHS, § 29.30. Similarly peyya, 'beverage' = Pali id., Skt. peya.

2.84. Reduction of double to single consonants. The reverse of such doubling, namely reduction of a double to a single consonant, also occurs in verses m.c. In my judgment, we must include here not only cases in which a Sanskrit double consonant is thus reduced, like ulapla for ullapla; but also those in which Sanskrit would have a cluster of dissimilar consonants, which only by MIndic assimilation would be alike (or different at most in that the second may be aspirated), as in upida for MIndic uppida, Skt. ulpida. In other words, I assume that such a form as upida can be understood only as derived directly from a MIndic uppida, not from the Skt. ulpida. That MIndic treatment of consonants underlies our dialect is suggested by its sporadic appearance even in orthography, shown above. It is more impressively demonstrated by the obvious metrical structure of the verses, in which what is usually written as a consonant cluster was nevertheless regularly pronounced as a single consonant, since a preceding syllable ending in a short vowel is prosodically short (§ 2.2).

2.85. Such reduction occurs at least sporadically, and apparently only m.c., in Pali (Geiger 32.2, end). Fairly common is Pali dukha for dukkha (see Childers s.v. dukkho); to be sure the analogy of sukha may have helped, but dukha seems to occur only metri causa. Our language also has dukha, sometimes written so, sometimes written dukha but shown by meter to have been pronounced dukha; also dukhin, dukhita.

2.66. Other examples (all vss, m.c.): Compounds of Skt. nis-: niguna for nirg°, i. e. MIndic nigg°; vinigala for vinirg° (Pali vinigata); nikaruna for nisk°, Pali nikk°; nisamisayam = nihsamisayam; nimulayitum = nirm°; nisattva = nihs°; vinibhāga (but in prose vinirbhāga); nivyāpara, pronounced nivā°, for nirvyāpāra.

2.87. Forms of dari-, MIndic dass- (or dams-), appearing as dai- with single i: dinadaji must be read LV 362.13, with several mss. including A, the best (text dinadarii, unmetr.); in Mv i.43.12 = 242.22 read (akhandam acchidram akalmas' avranam) vyākare arthadaii (mss. dari, unmetrical) matimām; daiayi Dbh.g. 42(68).19, for darisyi, which is an unmetrical y.l.
2.88. Compounds containing Skt. ud-: upala prob. =

2.88. Compounds containing Skt. ud-: upala prob. = ulpala; ulisiha, to be read LV 397.1 and 10 with best ms. A, for ultisiha of other mss. and both edd.; uplda for ulpida; read upādo Samādh 19.22 and upanno 19.25, for text with mss. ulpāda, ulpanno (unmetr.); so also read upa° for unmetr. ulpannu, °no, °nah, Gv 254.24; 255.2, 3; samupādayelsuh, samupādanīyam, for samup°; in Siks 103.3 text unmetrically samujvodlayisyati, perhaps read samujāl°, but Gv 482.24, whence Šiks cites the line, reads in ed. corruptly dharmadīpam ayu jāl°, where jāl° is for Skt. jvāl°; the corrupt ms. of Šiks also contains maitra-ya, which spoils the meter; ayu or aya = ayam seems the only possible reading; maitra perhaps got in by error from the preceding line; Tib., cited in note to Šiks, supports

33; in the seam of compounds some such cases (anuddayd, palikkūla, etc.) seem to me due to analogical influence (JAOS 41.462 fl.).

2*

if meter requires a short final syllable, but m if a long is required. No editor has seen this clearly; all editions are confused and inconsistent in this respect. So are the mss. to some extent; but they follow the rule in an overwhelming majority of instances, and there can be no question of its original validity; the exceptions are mere corruptions of tradition.

2.70. Examples are found everywhere in great abundance. A few are cited JAOS 66.202 (§ 50); I add here a few more, which might be multiplied indefinitely. In all the following, meter clearly demands a long syllable where final m is written before a vowel: SP duhkham idam 86.9, 14: trānum aham 90.3: ekam imu 90.10: sūtram abhi° 93.4: sūtram imam 93.7; sūtram idam 94.1; tvam ida (for idam; v.l. ima) 97.14; sütram upa- 98.8; kalpam ahu 99.5; LV, paramam ud- 43.11; ratnākaram a- 45.18 (so all mss. but one; Lefm. °karam a., unmetrical); pithitum a- 46.7; pajārahām ališayam (no v.l., but meter proves °yam must be read) abhi- 47.9; vyākaraņam rseh 111.14; param ašokam 163.19; tvam adīna- 170.15; Mv, divyanayanam ani- i.66.4 (mss. divyam nay°, unmetrical, em. Senart); nirvrtim upa-66.9; tam a- 82.11, 12; asampham idam uvāca 112.2 (both m and m metrically correct); nirbhayam idam 177.6; saranam upenti 256.4 (one ms. saranam, unmetrical); satatam alamkrto ii.44.16; Dbh.g., satatam a- 7(343).16; atuliyam a- 11(347).9; Sukh, agatya ca (read ca or co) ksetram idam udāram 52.15.

2.71. Not only is the variation between final m and m, before vowels, utilized for metrical purposes. Furthermore, as in Pali and Pkt. (cf. Geiger 32.2; Pischel 181; 182, 350), a final nasal (very rarely a non-final, § 2.73) may be lost, or unhistorically added (i. e. a final vowel may be nasalized). This is certainly connected with the rule of Pāņini 8.4.57 which permits optional nasalization before a pause of any absolutely final a, i, or u vowel, long or short. It seems to have been characteristic of Mindic generally (except that Mindic had no long nasalized vowels). Only scant traces of 'spontaneous' nasalization are found in our Skt. records (Wackernagel I § 259), despite Pāņini. In BHS both loss and addition of final nasalization occur chiefly in verses, being utilized for metrical convenience. It must be noted, further, that some secondary final nasalizations are due, at least in part, to morphological analogies rather than phonetic developments. This is also true in MIndic; such nasalizations of final vowels as are illustrated by Pali manam, liriyam etc. (= Skt. manak, liryak; Geiger 66, end) are clearly assimilated to adverbial forms in final -am.

Loss of final (rarely internal) nasal

2.72. Loss of final nasal occurs in many endings, usually m.c.: in the acc. sg. m. and n.-acc. nt. of a-stems (§§ 8.31 ff.), and the acc. sg. of consonantal stems (e. g. §§ 17.55; 18.83); in pronominal forms such as mahya, iya, ida (§§ 20.27; 21.79); in 1st person verb endings, such as aor. abhūva and opt. pravadeya (§§ 32.109; 29.29). Also in indeclinables: katha SP 314.4; LV 233.4; eva = evan; evarūpa = evanvūpa is very common, and occurs even in the prose of LV, as well as of Mv (and of Pall). In SP 71.3, prose, KN print ālmāna, which WT em. to ālmanah (the next word is sanjānatām), but ālmānam (acc. sg.) may quite as well be intended, with denasalization. I suspect a misnrint. however.

misprint, however. 2.73. Denasalization for the sake of meter in the interior of a word is extremely rare, but I think it must be recognized in Mv i.305.21, where (as Senart's note indicates, tho his text reads *injilena*) we must read *prāsādiken' ijilena*, m.c. for *iñjilena*. (The mss. have °kena *ijilena* or °kenejilena; they agree in lacking the nasal.) It must be noted, however, that nasalless forms of this root are recorded where meter cannot be concerned; see Dict. s.vv. anijya, injale.

Addition of final nasal

2.74. Conversely, a final anusvāra may be added unhistorically as a means of lengthening the syllable, equivalent to lengthening of the vowel or doubling the following initial consonant ('law of morae', §§ 3.1-4). So, in verses, metri causa: mam' iham (= iha) irnohi SP 94.2; nam hi LV 329.22 (v.l. nu hi, unmetrical); sādhum (= sādhu) ti SP 55.12; susthum khalu (read khalū with WT and ms. K') SP 114.10; śiri ham asmim manujeşu sammatā Mv ii.57.2, 'I am glory, highly regarded among men' (asmim = asmt 'I am'; so mss., Senart em. asmi, unmetrically); repeated with śraddhāham asmim 57.18; divim (= divi) galah (or gatāh) Mmk 618.26; 625.7; 627.14; anubhūya divim sukham, 'having experienced bliss in heaven', Mmk 629.16 (here only a loc. is construable, proving that we should not emend divim, q.v. in Dict., in the other cases to divam); in these Mmk passages always in verses where meter requires a long, hence likely to be m.c.; paratram ca Ud vii.6 (later ms. paratrásau; possibly the ending of param may have contributed some blending influence).

2.75. And in the seam of cpds.: dasandisasu, v.l. of Kashgar rec. SP 194.9 for ed. dasaddisāsu; citrakathām bahumvidhām, Kashgar rec., SP 283.13, for Nep. mss. citrakathān bahūmi ca (note masc. endings!), em. in both edd. to °kathā bahūś ca; kršamtanuh, 'lean-bodied', LV 157.21 (Calc. krsatanuh, unmetr., without ms. support in Lefm.); svetamiiro LV 188.8 (all mss.; Calc. śvetaśirāh, unmetr.); midhamgiri LV 197.3, 'mountains of dung', (ms. A intends midhaggiri; most mss. midhagiri, unmetr.); ugramleja- LV 221.2 (most mss. ugraleja-, unmetr.); alavi-puram-prakājam LV 231.3 (so read, as cpd., having the appearance of a jungle-town'); ksayamniruddha(h) (so read, as cpd.; no possible construction for separate ksayam as Lefm. prints it) LV 420.14; dhutamguna Šikş 328.2 = dhutaguna; samparivāram Šikş 343.15 for sa-p°, m.c.; śīlamśrutajñānasusthito KP 36.7 (m.c. for śila-śruta-, pronounced "sula-, or at least with single initial cons.); dharmacakraratanampurojavā Gv 54.16 (so read, as one cpd. word; text prints purojavā separately); corambhayā, 'dangers from thieves', Gv 213.12; trimalamtamoharam Gv 482.23 (text trimalam, as separate word, la°, which is senseless), but the line is cited Siks 103.2 with trimalattamo- (doubling of initial cons., \$\$ 2.78 ff.); sphatikammayam Lank 308.1, 'made of quartz or crystal', so read with ail mss. (ed. °ka-mayam, unmetr.); not strictly a cpd. since -maya is better called a suffix, but of the same essential character.

2.76. Such forms seem to occur in prose only in close compounds in which the prior member originally ended in a consonant or a long vowel. In other words, these cases are applications of the 'law of morae' by which, without regard to meter, internal nasalized vowel may be substituted for long vowel or for short vowel plus consonant cluster (double consonant). Examples: -māiramjñātā (read 'jňatā?) LV 430.3-4, for māirajña(tā, which also occurs), Skt. *māirā-jña(tā), Pali mattaññutā; mahangata(-tā) for mahadgata (which also occurs), Pali mathaggata. See § 3.4 for similar cases in the interior of words.

Double and single consonants; loss of final consonants

2.77. By the 'law of morae' (§§ 3.1-4a), in our language as in MIndic generally, double consonant after short vowel may interchange with single consonant after long or nasalized vowel. As a corollary to this, in metrical passages, doubling of a consonant (unhistorically) is employed where a long syllable is required, on a par with lengthening or some form or cpd. of *jvālayati*; utapta (also °ta-vatī) = uttapta.

2.89. Miscellancous: tiryaggata and °ti occur several times with single g, m.c. (Dict. s.v. (irya); fayādo, 'from the bed', abl. to fayyā; visandati = Pali vissa°, Skt. visya°; madharmam RP 17.15, Finot's em. for ms. mudharmam (assumed to be m.c. for maddh°, and reported to be confirmed by Chin.); read sadharmam Bhad 26 with several good mss. (ed. säddharmam, so!) for sad-dh°; read fanimnah Sikş 106.13 for ed. lannimnah, unmetr.; catutha = caturtha; kama = karma(n), MIndic kamma; budhita, see Chap. 43, s.v. budh (2), for buddhita; bhajisyati, see ib. s.v. 2 bhaj (2); (a)thi, to be read in LV 46.18 with 2 mss., for MIndic atthi = Skt. asti: pratipūrna thi ye, 'who are filled (with all virtues)'; the only other metrically possible reading in the mss. is iy (meaning?); Lefm.'s em. siddhye is impossible. (In the preceding line read certainly nirorti ye, sc. icchanti.)

2.90. Dropping of final consonants. The dropping of any final consonant other than a nasal, regular in Mindic, is common in the writing of BHS, in verses of all texts, when meter demands that the final syilable be short. It also occurs commonly in the prose of Mv, and occasionally in metrically indifferent positions in verses of other texts. Many examples will be found in the morphology, where a Skt. final consonant of an inflectional form is often dropped. This leads, for example, to the creation of new vocalic stems, based on consonantal stems from which the stem-final consonant (also word-final in the n. sg.) was lost, as jaga- from Skt. jagal., § 15.2, tiryth from tiryañe, tiryak, etc.

2.91. Such cases may be supplemented here by a few examples, out of very many, of indeclinables in which a final consonant is lost. So in prose in Mv: puna i.5.9; 349.3; yāva i.18.14; 20.4; kāni ci i.328.8; kasya ci i.364.7, etc.; and in verses, paścā = paścāt (see Dict.) ii.391.2; punāyam iii.367.2 (so mss., to be kept; for punar ayam). And in verses of other texts, almost always where meter

demands a short final syllable: *lāva* SP 92.10; 355.9 (here in a metrically indifferent position); *kai ci* SP 34.10; 93.1; Samādh 8.25; LV 37.21; *kā ci* SP 126.9; LV 36.16; Suv 40.12; *ke ci* Samādh 19.23; Bhad 1, 14, 15; *kadā ci* SP 96.2, 8; Suv 61.1; *kva ci* Suv 40.8; RP 4.3; *kutra ci* Suv 51.5; *paica* (*-kāle*, cf. *paicā* above) Samādh 8.31; *puna* LV 12.7; Samādh 22.6, and in SP 284.1 read *puna* for *punar*, m.c. (unless the quite different reading of the Kashgar rec. be adopted); *dhi* = *dhik* SP 351.6 (read *hā hā dhi muşyanli hi sarvasatīvā*); *samya* = *samyak* Dbh.g. 20(356).15.

2.92. The loss of final s or its samdhi equivalent (r, visarga, etc.) is so common, and so abundantly illustrated in the morphology, that it hardly needs illustration here. (See e. g. §§ 8.22; 10.15; 16.31, etc.) It is by no means limited to cases of metrical convenience, tho extensively utilized in verses for such purposes. Even in prose, and even before vowels, such cases are frequent, particularly in the Mv, but also elsewhere. Note e. g. babhāvatu (3 dual perf.) utlara- Divy 435.4 (prose; no v.l.). According to Speyer's note on Av i.16.11 (prose), the mss. of that text regularly write catu-rddhipāda which Speyer always emends to caturrddhi².

2.93. On the other hand, the following cases are evidently to be interpreted as use of final visarga, even before a vowel or a voiced consonant, to make the syllable closed and therefore prosodically long as required by meter, just as a final nasal may be changed to anusvära under the same circumstances and for the same reason (\S 2.74): lalilagatih anavadyagátrasamdhih Mv 1.205.4 = ii.8.18 (so Senart with both mss. the first time and one of them the second time, where the other reads "gati ana", unmetrically; same line LV 55.8 lalilagatir drahavajra"); kalpahvikalpápagalo KP 136.7, where kalpah- (in a cpd.) certainly represents a metrical lengthening for kalpa- (more often this is accomplished by lengthening the stem-final vowel, kalpa- or kalpa-, or anasalization of vowels; see above).

3. Phonology, Vowels

'Law of morae'

3.1. The excellent statement on this subject in Geiger 5, 6 reveals that to some extent Mindic shows what look like free variations between long vowel plus single consonant, nasalized vowel plus single consonant, and short vowel plus double consonant. Occasional, tho relatively few, cases of this sort are recorded in BHS. Thus:

3.2. Long vowel for short before original consonant cluster: $v\bar{a}ka$, 'bark', (Pali id.) for vakka = Skt. valka; ntharati = Pali id., Skt. nirharati; velt-ayati etc. must be interpreted as for $v\bar{v}\ell\bar{t}/h$ = Skt. vesl-, with lengthening of the *e* originally shortened before a Mindie double consonant; *limisaka*, 'dark', appears to be based on Mindie (AMg.) *limissa* = Skt. *lamisra*; *purākrta* = *puraskrta*, Pali *purakkhata*; BHS yātaka, lātaka (also $^{\circ}uka$; yältaka, *lātlaka* may be blends) are unparalleled elsewhere, and perhaps belong here as representing the equivalent forms yatlaka, *latlaka* (= Pali id.). Probably here belong also ähata, ühanati, and sam-ūh-. See also § 3.73.

3.3. Long vowel for nasalized vowel: $s\bar{a}hartavya = samh^{2}$; mailrāsa-tā for •maitrāmisa(tā), MIndic °amisa°, mailra plus amisa, see Dict.; $abhis\bar{a}hita$, prob. for abhisamihita; sārajyati = sam^{0} ; sālekhika to samlekha; sārambha and sālambha; sārāyanī(ya); rājanīya = $ranj^{0}$; gābhīra-tā, probably for gambh°; pişati etc., for pimṣ-; if Pali Vangīsa, n. pr., is more orig, than BHS Vāgīša, the latter would be an etymologizing hyper-Sktism. Also ū occurs once for the iufinitive ending um; vodutī (= vodumī) yalas le na dhuram samarthāħ KP 82.9 (vs). Or is this to be regarded as a lengthening m.c. of voduu = vodum? Cf. the apparent (tho rare) use of ū in the n. sg. m. of a-stems, which seems to be a lengthening of u, itself a reduction of o, § 8.21.

3.4. Nasalized vowel for short vowel plus double consonant, or long vowel plus single consonant: pumgala = pudgala (Pali puggala); vanka (Pali id.) for vakka (also Pali) = Skt. vakra; munga = mudga; manku (= Pali id.) = mangu = madgu (it this is the original form); bhisanka, 'physician', = Pali bhisanka, BHS bhisafka (§ 2.38); alambu, or a° , or °lämbu = alābu, 'gourd'; kumsana, also written kunsana (§ 2.64) = *kussana for Skt. kulsana; samunācita for samuccila. On -mātramījnātā (Pali mattañnūtā), mahamgala (also mahadgala; Pali mahaggala), see § 2.76.

3.4a. Short vowel plus double consonant for long vowel plus single consonant: $alliyali = \bar{a}liyale$; ularanti = olaranti, § 3.54; cf. also § 2.83, where e was surely long before y, short before yy.

Long vowels for short

3.5. a for **a**. Lengthening of historically short a seems to be almost wholly a matter of metrical convenience. It occurs chiefly in verses, in positions where meter requires length; rarely in prose, except for sporadic cases which are mostly explainable in special ways. It is particularly common with final a, including stem-final in composition. These are, as a rule, not listed in my Dictionary.

There are also not a few cases of lengthening of the thematic vowel in verb forms, and other, miscellaneous, internal *a*. The following examples are nearly all from verses, and nearly all interpretable as m.c. Prose examples will be specifically identified.

3.6. \tilde{a} for final a. In indeclinables: $c\tilde{a}$ for ca SP 57.16; 83.4 (so with Nep. mss. and WT); 97.10; 113.1; 204.10; 294.6; LV 125.3 (divide $c\tilde{a}$ nirodha); 220.9; 223.2; 271.9 (divide $c\tilde{a}$ kalpa-); RP 54.22; Dbh.g. 41(67).3; kil\tilde{a} for kila SP 220.14 (so read with WT); hantā for hanta LV 110.17; 133.7; Dbh.g. 7(343).5; vā för iva (Dict. s.v.); ivā LV 175.22; evā for eva LV 216.21 (naivā); ihā for iha LV 324.5; RP 27.10; nā for na Gv 214.15 (divide nā viprayogu). Even punā ca occurs Gv 476.9 (vs), instead of punaš ca; this could be regarded as a case of the 'law of morae', like § 3.2; Pali also has punā for puna (besides puno).

3.7. In noun and pronoun inflection, lengthening of final a occurs, for the most part but not always metri causa, chiefly in forms of the a-declension. The final a of the stem is frequently lengthened in compounds, sometimes even in prose (§§ 8.15-17). In the nom. and acc. sg., the final a, as representative of historic as or o etc., is sometimes lengthened (§§ 8.24, 38). In the voc. sg. the lengthening is very common but does not appear to be primarily a metrical matter (§ 8.27). For instr. sg. end, inā see §§ 8.39, 40, and for inā in pronouns § 21.14; for dat. sg. dyd, § 8.44; for gen. sg. asyd § 8.58. Note also the personal pronoun forms mamā and tavā, §§ 20.29, 33.

3.8. In verb inflection, lengthening of final a occurs in the 2 sg. imperative (§ 30.11); and in the 2 pl. ending *lhā* for *tha*, imperative or optative (§§ 26.14, 15); also indicative, including e. g. future, as: *lapsyathā* RP 18.14 (vs). The gerund endings ya and *iya* also appear as yā and *iyā*, usually m.c., but sometimes apparently in metrically indifferent positions (§§ 35.10, 39, 40).

3.9. a for non-final a. Lengthening of non-final a occurs, first, in the case of the thematic vowel; on this see §§ 27.1 ff. We have counted as 'final' the stem-final a of nouns at the end of prior members of compounds. Nearly all the rest are, or could be, cases of metrical lengthening. They are, moreover, far less common than lengthenings of final a. Geiger 24 cites similar cases in Pali, which he finds particularly in the first syllable of words and attributes to the influence of word-accent. This theory seems to me doubtful.

3.10. The clearest prose cases I have noted are cāraņa for caraņa; pāripāri (also Pali), pārišuddhi (also Pali) for pari^o, and (not in Pali) pāripāraka, pārihāņa, pārihāņi, pārihāraka, all for pari^o; durāvagāhatoāt LV 424.11; so all mss.; Calc. durava^o, see Dict.

3.11. In verses, presumably m.c.: (SP) anābhibhūta; anābhibhū; ānubhāva (regular in Pali in prose as well as verse); (LV) ālireka; Anāvalapta; svālamkrtam (sv-a°); bhāvisyanli (etc., s.v. bhāvali); śayāni for śayane; sāvāsanā, for sa- (associative prefix); nirāparādha for nirapa°; sāhodhā for sa°, i. e. saha-ū°; samāya(laḥ) probably for samaya-laḥ; (Mv) pratibhāņilo for °bhaṇilo; pāripūrņa for pari°; ānušásti for anu°; (Suv and Gv) nir-ābhiramya for nir-a°; (Samādh) ān-abhiramyāħ for an-; bhāvāmi for bha°; (RP) nirānuraktā for nir-anu°; probably acāra-, 'constant, unchanging', for acara-; (Gv) sāmabhilasatyā for sam^o. In verses, but metrically indifferent, pāļirāja for Pali paļi^o, Skt. prati^o.

3.12. I for i. Lengthening of historic *i* to *i* has been recorded only in verses, and almost exclusively in positions where a long syllable is required. By exception it occurs in metrically neutral positions, namely at the end of pādas or lines where quantity appears to be indifferent; for example in bhonti (= bhavanti) SP 326.10; Siks 329.9; 331.1, 3 (in the same Siks passage short vowels are presented at the end of lines of the same meter, c. g. in 328.1, 3, 5, 6, and bhonti also 331.1); or in *lvayī*, for *lvayi*, used as instr. sg. of the 2 personal pronoun, Gv 489.14 (end of a rathod-dhatā line; other such lines in the same passage end in short vowels).

3.13. We shall see (§§ 10.9, 11) that the stem form of what are in Skt. short *i*-stem nouns, in composition and before such suffixes as *-mant*, often show *I*. This is certainly not chiefly a matter of meter, since it appears in prose. But these cases seem to me to deserve no more than mention here. They appear to be part of the farreaching morphological blend of *i*, *i*, and *in*-stems, which appears through the declension. That is, the lengthening is not, or at least not primarily or exclusively, a phonological matter.

3.14. *i* for final *i*. In indeclinables: prati LV 119.7; hi SP 97.4; Gv 240.19; yadī SP 305.13; LV 242.22; apī LV 46.3; 109.19; 194.5; pi (= api) Sukh 51.3. Indeclinables as prior members of compounds: vimati; vitatha; -alijaya-. Also when the vowel is 'final' only thru MIndic loss of an original final consonant: koci Mv i.166.10 (so Senart with good mss. support; followed by *iha*: vv.ll. °ei, °eid, both bad metrically).

3.15. In noun and pronoun inflection, we find such (usually metrical) lengthenings as advani for advani (§ 17.59); nom.-acc. pl. nt. forms in -ni for -ni, as kuntalāni LV 49.20, triņi SP 33.4; instr. pl. of a-stems in -ehi, -ebhi instead of -ehi (which is very common in BHS) and -ebhi; besides tvagi for tvagi, above § 3.12. There is also the curious loc. sg. loki Gv 253.5, isolated so far as I know; see § 8.59. We may compare the apparent sporadic occurrence of $-\bar{u}$ and (with qualifications) $-\bar{d}$ (§§ 8.21, 24) in the nom. sg. of a-stems, instead of -u and -a which themselves replace -o (-as) chiefly in positions where meter requires a short final.

3.16. In verb inflection, any final i of an ending may appear as i, but chiefly if not exclusively in verses m.c. See § 26.2 for examples.

3.17. I for non-final *i*. All the following examples occur in verses where meter demands a long in the syllable in question. Some are hardly to be separated in principle from the cases of 'final' I for *i*. Thus the instr. pl. ending appears as -bhir, before a vowel: bahubhir acintyair SP 385.3. Before suffixal ka, what are normally or at least historically *i*-stems show *i*, just as before suffixal mant and in noun compounds (see § 10.12): prāñjalīka; anu-padhlka; edrikā for cārikā; saņskrta-galīkam anāgalīkam. Cases like sukundalīnam Suv 246.7 (for °linam, 'having fair earrings') have been included under the *i*-*i*-in declension.

3.18. More sporadic, and hardly of morphological significance, are the inf. ksayitum Siks 345.10; a-parājita for °jita; su-pithita = °pithita, 'well closed, guarded'; fonita, 'blood', for sonita (see Dict.).

3.19. Even epenthetic *i* may, it seems, be lengthened m.c.: *ilriye* LV 54.8 (vs), for *iiriye* (which to be sure most mss. read, unmetrically) = Skt. *iriye*.

3.20. 0 for u. Lengthening of historic u to \bar{u} , like that of a and i, is mostly a matter of metrical convenience. Of the examples which seem to belong here, only $p\bar{u}_{la}$, $(pari)sph\hat{u}_{la}$, and s $\hat{u}rala$ occur in prose. And I have noted

no others in metrically indifferent positions in verse. As in the case of *i*-stems, *u*-stems in composition with lengthened \tilde{u} (§ 12.6) are not regarded as primarily pertinent here.

3.21. \hat{u} for final u. In indeclinables: $t\hat{u} \text{ SP 97.7}$; $susth\hat{u}$ SP 213.4; $kh\hat{u}$ (for khu = khalu) SP 308.9; $ma h\hat{u}$ (v.l. ° $kh\hat{u}$; $= m\hat{a}$ khalu) LV 141.3; $s\hat{a}dh\hat{u}$ LV 110.13 (most mss. $s\hat{a}dhus$); 220.15 (most mss. $s\hat{a}dhu$, unmetrically); 276.2 (no v.l.); $j\hat{a}t\hat{u}$ LV 120.9 and 10.

3.22. The indeclinable su lengthened to sũ in cpds.: sūrata-, which, however, occurs in prose and in metrically indifferent positions, as does also the equivalent surata (Pali seems to have only sūrata, sorata in this sense of 'mild, gentle'); sū-sukhitah Samādh 19.28, 31; sū-pariśuddhe Samādh 19.35; sūgali-Gv 54.25; Sūdhana- (n. pr., otherwise Sudhana) Gv 208.11; 215.14.

3.23. In noun inflection, the loc. pl. ending su appears m.c. as sū: grhesū LV 57.1; diśāsū SP 53.11; vidhisū Gv 55.3; galīsū SP 9.6; 48.3; pañcasū RP 59.6; salsū SP 9.6; 48.3; daśasū SP 53.11 etc. (§ 19.27). The imperfect tradition often records forms with short u where meter demands ū; so in LV 152.1 Lefmann properly reads diksū, tho all his mss. are reported as reading diksu.

3.24. Once \bar{u} seems to occur for nom. sg. u (= o, as) of a-stems (like \bar{a} for a in the same ending), § 8.21.

3.25. In verb inflection, final u is lengthened m.c. in the impv. endings tu, ntu (§ 30.13); in 3 pl. aorist endings in -su, -su (§§ 32.33, 35, 38); as to vodhū for inf. vodhum see § 3.3.

3.26. \ddot{u} for non-final *u*. This is not common. It is probably m.c. in: *an-ūpalipla* (cf. *an-opa*°; Pali also *an-ūpalitla* m.c. for *anu*°); *srūņute* LV 74.9 (m.c. for *sruņute*, § 3.96, semi-Mindic for *srņute* which several mss. read unmetrically); kūhanā and (*a*-)kūhaka- = kuh^o. But pūļa (cf. pūļi), which seems related to Skt. puja (and puļi), occurs in prose; also sphūļa and parisphūļa, for (*parisphū*ļa, for (*parisphū*ļa).

Short vowels for long

3.27. a for ā. a for final ā: in indeclinables, only in verses m.c.: lada SP 68.1, 2; Mv i.111.16; ii.300.6; 318.21; 338.16; Suv 147.2; Gv 214.4; latha SP 84.8, 13; 87.1; 90.12; 96.3, 16; 97.12; LV 49.8, 14; Mv iii.136.15; Suv 23.10; 37.16; 60.4; Samādh 19.18; sada SP 91.2; 93.13; 97.11; 98.5; 155.3 (so read with all Nep. mss. and WT); LV 41.20; Bhad 15; Sukh 9.8; Mv ii.376.11; 389.1; Samādh 8.26, 31; 19.28, 31, 34, 36; RP 5.16; 11.2; yada SP 147.14; Mv i.154.15; yatha SP 89.13; 96.9; 99.1; LV 41.16; Mv i.70.6; 166.14; ii.81.1, 7; 88.13; 135.16; 321.22; 334.2, 6; 340.9; Suv 60.2; 146.7; RP 6.1; 11.1; Bhad 55; milliya LV 38.1; pa (= mā) LV 11.1; 120.13; read va for vā Mv i.155.7; ma (= mā) LV 123.5; 141.3 (ma, v.l. mā; followed by hū, v.l. khā, = khalu); 237.1; 288.4 (v.l. mā); RP 59.7.

3.28. In composition: yatha-bhūtam (read as cpd.) LV 188.18; samudayāntv Divy 399.16 (m.c. for samudā-yāntv; note āyāntu in next verse, 399.19). In RP 52.9 Finot reads akarşaya and understands it for ākarşayan, with a m.c. for preverb ā, but the correct reading is akarkafa, as in the Sikş 320.9 citation of this passage, supported by Tib. acc. to Bendall. On acchindati for āchindati see below.

3.29. In an indeclinable which in Skt. ends in $\bar{a}t$, after MIndic loss of t, a for \bar{a} m.c.: pasca (= pasc $\bar{a}t$) Suv 52.1.

3.30. In noun inflection, final \bar{a} is shortened to a, chiefly but not always m.c. (sometimes in prose, at least according to the tradition), in stem final of prior members of cpds., and in various case endings, particularly those of the \bar{a} -declension. As in *paica* for *paical*, above, these include cases in which a final consonant followed the \bar{a} in Skt., or even in which a nasalization followed it in normal MIndic, as in the acc. sg. $\bar{a}m$ (MIndic am). For \bar{a} -stem forms see §§ 9.6, 8, 9, 16–19, 86, 87. The same shortening

occurs, in verses m.c., with mahā-, composition form of mahat-: mahabuddhaghosam LV 116.20; mahadaksiniyah Mv ii.300.4; maha-abhijña-upāya-jñāne Gv 477.1, 'in the knowledge of the great abhijñās and upāyas'. In other declensions a occurs for ā final (Skt., or Mindic after loss of final consonant). So, in verses m.c., in the n. and acc. pl. of a-stems (§§ 8.79, 94); in the nom. sg. of r-stems (§§ 13.6, 15) and n-stems (§ 17.6); in the instr. sg. of ntstems (§ 18.84) and other consonant stems (§ 15.16). On gen. sg. -asya for fem. asyās (not limited to verses m.c.; rather substitution of masc. for fem. form) sce §§ 9.75, 76.

3.31. a for non-final ā. These fall into two very distinct groups. In the first, the shortening occurs only in verses and seems definitely m.c. Here belong the numerous cases of shortening of long thematic vowel in verb forms (ending ami etc., § 27.9). There are also a fair number of miscellaneous cases: abhinadita, 'caused to sound' = abhinādita; Syamu for Śyāmah; prodyayāmāna = prodyāyamāna; duyamanu = dūyamānah; bhasi = (a)bhāşi (2 sg. aor. of bhāş, 'speak'); -samsarī = samsārinah; svākarāi = svākā-rāi; acc. to Senart mahāpakam Mv i.104.12 (= mahāpākam; but see Dict.); nivyāpara = nirvyāpāra; ananlariya = ānanlariya; ajīvika = ājīvika; kakhorda = kā°; karaka, 'doer, maker', in LV 340.4, Dbh.g. 27(53).11.

3.32. In the second group, on the contrary, we have MIndic shortening of \hat{a} to a before a consonant cluster. Such forms may be definitely MIndic in all respects: khajja, khajjaka, khajjali = khādya etc.; kilanta, pari-kilanta, etc., = (pari)klānta, §3.109; acchindati as in Pali, = āchinatli; ayyaka = Pali id., Skt. āryaka; payyaka = Pali id., Skt. $\delta rparyaka; jambūnada = j\hat{a}^\circ$.

3.33. There are also cases in which, as in Mindic, the shortening takes place before an original Skt. consonant cluster which has been broken by Mindic epenthesis: $kariya = k\bar{a}rya$; $ariya = \bar{a}rya$; $\bar{a}cariya = \bar{a}c\bar{a}rya$. Cf. the like with *i* for *i*, *u* for \bar{n} , §§ 3.38, 45. There is also a hyper-Sanskritic bharikā, Sanskritization of Mindic bhariyā, 'wife' = Skt. bhāryā.

3.34. In other cases the forms may be called semi-Mindic. So vasyati, 'cries, shrieks' (= Pali vassati) for Skt. väiyate; uparddha for upärdha, halt-Sanskritized from Pali upaddha; margati = märgati; jalya, perhaps for (a)jälya; attättamiya = Pali attattaniya, Skt. ätmätmiya; samajää = samajää, Pali samaänä; utsaryati, pass. to utsårayati; pradanti, Chap. 43, s.v. dä (4).

3.35. In LV 419.8 (vs) atma is read in both edd., for $\bar{a}tm\bar{a}$ (the final *a* is m.c.), without v.l. in Lefm.; Foucaux quotes one ms. as reading $\bar{a}tma$ (implying that the other two which he had at his disposal read atma with Calc.; to be sure, the footnote in Calc. quotes the text as $\bar{a}tma$). Unless the form is an uncorrected misprint in both edd., which seems unlikely, it appears that we have a semi-Sanskritzed form of Pali atta. So atmānam Ud xvii.10; atmānam, atmā xix.12. In badly printed texts like Gv, isolated cases of this sort are apt to be misprints. So (an-)attamanas in Gv 411.8 may be a misprint for $(an-)\bar{a}tta^{3}$, the it coincides with the Pali form.

3.36. Instead of Skt. *tathātva*, BHS has usually *tathatva* = Pali *tathatla*, of which *tathatva* is probably a semi-Sanskritization. Note that Mvy 134 has the correct Skt. *tathātva*. The equivalent *tathātā* is only found in BHS. There is a similar *mithyatva* = Pali *micchatla* (Skt. *mithyātva*), antonym of *tathatva*. The form *tathatā* also occurs in Pali, but is noted only once in PTSD (Vism. 518.18); *tathatīa* is much commoner. In BHS *tathatā* is not noticeably rarer than *tathatva*. But, being clearly a Buddhist word, its form is probably influenced by the semi-MIndic *tathatva*; or rather it was a direct inheritance from the common MIndic original of BHS and Pali, with pre-suffixal a due to influence of *tathatta*. To be sure, Pali has an adjective *a*-stem *tatha* (not in BHS). It can hardly be anything but

a secondary back formation from *lathalla* (*lathalā*), or perhaps from *vitalha*, or partly from both sources.

3.37. i for I. Metrical considerations, unsupported by others, play a smaller rôle here. Particularly as regards original final *i* for older *i*, elther as absolute final or in the seam of cpds., it hardly exists except in nouns of the *I*-declension, where it is certainly not wholly, probably not primarily, a metrical matter; see $\frac{55}{10.9}$, 10. The Skt. *idānīm* appears, as in Pali, commonly as *dāni*, in prose (at least of Mv) as well as verse; rarely (in vss) as *idāni*. Before a vowel we find *dānim*, only in verses; and Mv several times has *dānim* before consonants, in verses where a long final is required. We also find *ladāni* (Skt. °*nīm*) at least once, in prose, Mv i.253.6.

3.38. In non-final position, i before a consonant cluster is sometimes shortened by the standard MIndic change. So praticchila, ppp., = prati°, Pali palicchati; iryā, iryati etc. (Skt. ir°); irsyā = irsyā; but iranta(h) Gv 372.13 (vs) for irayanlah seems to show purely metrical shortening; niriksya for niriksya (one ms. niriksam); and udiksisu for udiksisu, twice, 3 pl. aor. of udiksati (ud-iks); adhisfo for adhisfo. And viriya = Pali id., Skt. vīrya (vss, m.c.); here MIndic epenthesis has broken the originally following consonant cluster (cf. the same with a for \bar{a} , u for \bar{u} , (§§ 3.33, 45).

3.39. Before single consonants, the following seem most likely to be cases of metrical shortening pure and simple: jdniydl, 3 sg. opt. of jnd, SP 31.5; a-pdniyd, 'waterless'; suilla = susilla; bibhatsa, bibhatsana, = bt° ; apidila = apidila; salila = sa-lila; usnisa = usnisa; ksilija = ksilija; gabhira = gabhira, gambhira.

3.40. In other cases, even the many of the occurrences of *i* are favored by meter, there is reason to suspect that other considerations are concerned. The form *krinati* for *krinati* is not rare, is found also in Pali, and is certainly not at bottom a metrical matter; see § 28.5. The adjective *alika* = Skt. *alika*, 'false', also occurs in Pali; acc. to Geiger 23 the shortening is due to MIndic accent on the first syllable; to me it seems more likely a matter of adaptation to suffixal forms in -ika.

3.41. The Pali duliya, latiya, and their Pkt. equivalents, to which BHS dviliya, triiya correspond, are also explained by Geiger 23 as due to accent conditions. Pischel 82 and Jacobi Kalpas. 103 explain them as derived from prehistoric *doilya, *triiya (cf. AMg. docca, ducca). To me it seems much more likely that we have here analogical modification of Skt. dviliya, triiya, adapted to the numerous forms in suffixal iya, compared with which forms in Iya were sporadic and seemed anomalous. In BHS, however, the forms are rare and seem to occur only as required by meter, hence in a sense (selectively) m.c.

3.42. The influence of forms in suffixal iya seems also likely to be present in some forms ending in *-aniya* for Skt. *-aniya*, gerundives and others (cf. § 22.20). In every case meter requires a short i: darfaniya, fāsaniya, -premaņiya.

3.43. Analogical influence of a different sort is shown in forms of the root ni showing *i* for Skt. I (ppp. -*nila*- and ger. -*niya*). Tho such forms occur only in verses and could be called 'm.c.', they are frequent, even in prose, in Pkt. Pischel 81 explains them as due to accentual conditions, which I do not believe. They are simply analogical to the Mindic present -*netii*, on patterns like *katheti: kathila*-: *kathiya*. Our language shows this shortening only in compounds of *ni* (so also prevailingly in Pkt., Pischel l.c.). See §§ 34.7; 38.3; Dict. s.v. samudānana.

3.44. u for \bar{u} . Aside from the morphological confusion of noun stems in u and \bar{u} (§ 12.7 ff.), I have no cases of shortening of final \bar{u} .

3.45. Occasionally MIndic forms containing medial u for \hat{u} which is, or in Skt. was, followed by a consonant cluster, are found in BHS: omuddhaka = avamūrdha(ka);

sukhuma, sukhama = sūksma; suriya = sūrya; luriya = tūrya. Cf. the like with a for \bar{a} , i for \bar{i} , §§ 3.33, 38.

3.46. Otherwise I have noted only shortening of internal \bar{u} , m.c., with the possible exception of pureya = $p\bar{u}r^{o}$, see Dict. s.v. pura(ya)li. It is specially frequent in bhula for bhūla and cpds., as in Ap., where huya is common e.g. in the verses of San. (see Jacobi's Glossary): bhula, udbhula, sambhula, lūsnibhula. Others: bhuyo, bhuyu = bhūyas; duyamanu = dūyamānah; rupa = rūpa; puja = pūjā; śruyāl (precative of śru), Chap. 43, s.v. śru (8); Virudhaka, Virupākṣa = Virūdhaka, Virūpākṣa

Substitution of *i*, *i* for *e*, *ai*

3.47. We find *i* for *e* chiefly as metrical shortening, and for the most part in certain morphological categories, where examples will be found: in the loc. sg. of *a*-stem nouns (\S 8.59, 60); in the instr. sg. of *a*-stem nouns and pronouns ($in\bar{a}$ for *ena*, \S 8.40; 21.14); in *mi*, *ii*, *si* for *me*, *ie*, se of pronouns (\S 20.29, 33; 21.19); in the n. and acc. pl. of generic pronouns (\S 21.25, 32); in personal verb endings (\S 26.3; in part these may be regarded as replacement of medio-passive by active endings). Cases of samdhi in which final *a* is elided before initial *i*, leaving *i* for Skt. *e*, could also be regarded as pertinent here; see § 4.23.

3.48. Other cases are few and scattering. In vss m.c. we find saci for sacel (besides saca); Maitriyo for Maitreyo; pranitar for pranetar; pranidhisi for pranidhesi (2 sg. pres.). In LV 74.22 (vs) sthāpile sma māya mūrdhni pāni, 'Māyā would put her hand on the head' (of those sick ones; cf. the prose account 72.4), it seems that sthāpile must represent sthāpele (or °ti, Skt. sthāpayati). It is not likely to be loc. of the ppp., which would involve taking pāni as loc. (note the v.l. pānim, which is unmetrical but suggests my interpretation). In LV 54.15, however, read with v.l. upasthapitonā (ger.) for text upasthapisva nā.

3.49. Of different character are a few sporadic cases of *i* for *e*.before a consonant cluster. Here we seem to have Prakritic *i* for *e* (which may = Skt. *aya*), which, in MIndic, is automatically shortened (and in Pkt. often written *i*, Pischel 84) before two consonants: pranidhintāna Mv i.124.18 (vs; so 5 mss., Senart with 1 inferior ms. °dhentāna; gen. pl. pres. pple. of pranidheti); yācintyā(ħ, before a-) Mv i.133.1 (prose; kept by Senart; for Skt. yācayantyāħ, or rather yācantyāħ, since yācayati is only causative in Skt.).

3.50. Once vira, 'enmity', is recorded for Skt. vaira, Pali vera. On i for Skt. ai cf. AMg. dhiraya = dhairya (Ratnach.), and Ap. dhira = id., Jacobi, San. 513.1.

u, ū for o (and au)

3.51. For the most part, this too is a matter of metrical shortening. It occurs in *ahu* for the interjection *aho* SP **62.4.** Otherwise the *o* replaced by *u* is oftenest a (generalized) samdhi representative of Skt. *as*; but sometimes also a Mindic form of Skt. *ava.*

3.52. For as = o, we find u (once apparently \hat{u} by secondary lengthening, § 8.21) in various noun and pronoun endings: nom. sg. m. of *a*-stems (and, by transfer, of what in Skt. are consonantal stems; also in nom. pl. m. and nom. sg. fem., by transfer from nom. sg. m.); in the abl. sg. forms in -tu ($\hat{a}tu$, atu, itu, itu; in iyu for iyo, gen. sg. fem. of i(i) stems, and u for o (as) in the gen. sg. of consonantal (nt) stems; in the nom.-acc. sg. of *s*-stems; and once in nu for no = nas, 1st pers. pl. pronoun, enclitic. For cases of samdhi between final a, \hat{a} , and initial u, yielding u where Skt. would have o, see § 4.24.

3.53. Further, in miscellaneous adverbs (all in vss): bhūyu = bhūyas SP 95.1; LV 190.20; 231.8; RP 11.8; bhuyu, id., \S 3.46; śreyu = śreyas LV 57.3; sarvaśu = śarvaśah Samādh 19.29; for sarvaśah SP 255.13, La Vallée Poussin JRAS 1911.1073 reads survasub, at the end of a line, metrically indifferent; tatu = tatas LV 164.22; 169.22; puratu = puratas LV 81.11; sarvatu = sarvatas Gv 43.6.

3.54. As in the case of *i* for $e(\S 3.49)$, *u* also occurs for Mindic o = Skt. and before a Skt. consonant cluster. This, as in Pkt., is a representation of, or substitute for, short o in a closed syllable: *ukrami = okrami*, *avakrami*; manujña = Pali manuñña, for manojña; *utlaranti = *otlaranti* or otaranti (so one ms.), for avalaranti, 'arrive at, betake themselves to'; in this last we have doubling of consonant besides shortening of vowel, by the 'law of morae' (§ 3.4a and Geiger 6.2); ucchādya for avachādya; ullokanaka for olok^o.

3.55. In Dbh.g. 6(342).21 $aia|h'\hat{u}|\hat{n}ak\bar{a}i$ (ca), 'free from deceitfulness and contempt', see Dict. s.v. $u|\hat{n}a-ka$, implies an $*u|\hat{n}\bar{a}|$, or rather perhaps something like $*\hat{u}\hat{n}\hat{a}$ or $*\hat{u}n\hat{a}$ (cf. $\hat{a}n\hat{a}$, Dict.; since the \hat{u} , even as the result of a plus u in samdhi, could be long originally only before a single consonant), = Pali $u\hat{n}\bar{n}\hat{a}$ = Skt. $ava|\hat{n}\hat{a}|$ (with suffixal ka in Bahuvrihi cpd.). Further, in $ukirati = ok^{\circ}$ for $avak^{\circ}$, o for Skt. ava before a single consonant is shortened to u m.c. And sporadically \hat{u} occurs, as in Pali, for o of other origins; so $-r\hat{u}payali$ for $-rop^{\circ}$, Chap. 43, s.v. ruh (5).

3.56. A few times we seem to find u for Skt. au, which presumably was first reduced to MIndic o. Thus twice in dual forms, antu and imu for antau and iman, § 8.75 (but v.l. in both a for u); and once in stuli, m.c., em., for slauti of ms., but see Dict. for other possible explanations. These cases are uncertain; if sound, they seem due to metrical requirements. Different is $duv\bar{a}rika = Pkt. duv\bar{a}riya$, which Hem. 1.160 derives from Skt. dauvārika. However, Skt. also has $(r\bar{a}ja-)dv\bar{a}rika$ once at least; perhaps we should derive the Pkt. and BHS forms from this, with epenthetic u, cf. $duv\bar{a}ra$.

u for a

3.57. As in Pali (Geiger 18.1) and Pkt. (Pischel 104), a is sometimes changed to u after (and perhaps before) labial consonants; see Chap. 43, s.v. -vultati (=vartati) and (under root vr, cover) -vurati (Pali -purati), for -varati; sruvate = sravate, § 28.37.

u for am (ām)

3.58. This phonetic change, on the dialectic relations of which see § 1.97, seems limited to verses, m.c. It is abundantly illustrated in the acc. sg. and nom.-acc. sg. nt. of a-stems (§ 8.30; also by transfer in the acc. sg. of \bar{a} -stems and of thematized consonantal stems). It also occurs in the gen. pl. dnu for dnam thru MIndic dnam (§ 8.125), and in that of pronouns (lesu for lesām, etc., § 21.40). Further ahu or hu is used for aham, vayu for vayam, ayu for ayam (§§ 20.7, 38; 21.79). It is rare in Mv; examples there are asavalu (mss.) il.299.14 and paresu (= paresām) i.286.9.

e for i. i

3.59. In some MIndic forms e occurs for i and l of Skt., as Pall edisa = $id_f i_a$; and possibly Pall ediaa, perhaps cf. Skt. iyatlaka; Pkt. ketti(y)a, perhaps cf. Skt. kiyat (Pall kittaka); and eyāņim. These appear in BHS as edria (also kedria), ettaka, kettaka, edānīm; (more doubtfully) eha. The frequent MIndic writing of (short) e for i before a consonant cluster may account for vesti (AMg. velthi) = Skt. vişti; and cf. Lecchavi, below, § 3.67.

3.60. In other cases *e* seems more or less clearly to occur as metrical lengthening for *i*. A very clear case is *lokadhātv eyam* (for *iyam*) SP 306.7 (vs; so both edd., all mss.). Similarly *lokadhātv emām* (for *tmām*) SP 355.4.

3.61. In verb endings, te and se instead of ti and si may be interpreted as middle for active endings, §§ 37.24, 25. There is however little doubt that generally they are used for the sake of meter. So paiyate Mv ii.359.17, 19; 360.11, 13, etc., where paiyati would be metrically bad; other cases are e. g. bhavale, jahate Mv ii.382.16 and 17; pravadase iii.346.2; and many other cases l. c.

e for aya, ayi; and avi

3.62. This common MIndic change is very frequent in BHS. Numerous illustrations can be found in our treatment of causatives and denominatives. In other verb forms it is also common; e. g. praneti (read °li with Nep. mss. and WT, m.c.) = pranayati SP 62.12; neti Mv ii.81.9. Also in nouns: lena = layana; janetrī, 'mother' = janayitrī, Pali janetlī. For avi, e occurs in bheşyati = bhavişyati, the regular BHS future of $bh\bar{u}$; and in thera = Pali id., Skt. sthavira, 'old'; half-Sanskritized as sthera, 'an elder' (religious).

3.63. Hypersanskritic aya for MIndic e may occur in such forms as *-thayati* for *-dheti*, etc., § 28.48.

Metrically short e

3.64. Instead of i, the usual metrical shortening for e, the texts sometimes write e, especially in word-flnal (cf. Pischel 85), in positions where meter requires a short: tě anyam-anyam SP 85.12, 13; tř nityakālam SP 131.4; tě buddhajāsanam SP 152.11; karişyě 'ham LV 223.5 (Lefm. em. karişya ham); gacchamānakē LV 235.8; yě hi LV 324.15 (Lefm. em. yahi); tuşitālayě LV 364.7 (Lefm. em. °ya); tě imu LV 393.6 (v.l. ta imu); in LV 400.18 read probably brahmann ili satata yě śrotavantah; dahyamāně Mv i.69.1; uktě 69.13; pārśvě 71.5; tě 80.16; sarvajňatāyě 82.13; puruşasimhatāyě ā- 83.6; mě (before consonant) ii.8.17; dešž-deše Suv 58.4; šilě, dhyāně Samādh 19.35; optatives, § 29.13.

3.65. In medial position, rare: phěna- LV 176.2, cited Šikş 237.11 (Lefm. em. phana-, see Dict.); tasya višėsu LV 365.14; bhanēma LV 422.6.

e for o (final)

3.66. This so-called 'Magadhism' (see § 1.32 with fn. 11), found also in AMg. and occasionally in Pall, occurs in the adverb pure (= Pali, AMg. id., Skt. puras), chiefly in vss. The nom. (voc.) sg. m. (and nt.) of a-stems in e, nouns and pronouns, may also be classed here, §§ 8.25, 28, 37; 21.7. It is questionable whether the isolated puni, LV 151.8 (vs), for punar, represents an otherwise unrecorded *pune. But ante (as in Pali, AMg.; also in cpds.) occurs for Skt. anlar.

e for ai

3.67. We also find e for ai as in MIndic, for instance in eli = aili; metra = maitrā; sarveşana- = sarvaişana-(sarva-eşana-) LV 34.14 (prose; all mss.); <math>celiya = Paliid., Skt. cailya; $eneya = ain^\circ$; lecchavi = *laicchavi (adj. to licchavi; even when used of a single person probably a vrddhi formation, yet possibly a case of e for i as suggested by Senart, note on Mv i.255.9, see § 3.59); erāvana = $airā^\circ$ (prose); dvemāsika, tremāsika; ve = vai.

e(o) = MIndic e(o) of secondary (analogical) origin

3.68. In gredha, also parigredha, 'greed', we see a hyper-Sktization of Pali gedha, paligedha, which are analogically based on Pali giddha etc. = Skt. grddha etc., on patterns like Skt. nisiddha: nisedha; Pali viddha: vedha, etc. Similarly paligodha (Asokan ib.), to paligudha, $g_{3.92}$; mosa(-dharma) = Pali mosa, to Pali musā = mṛsā.

ai by Hypersanskritism for e

3.69. Possibly in *vainu* for *venu* (if not a mere corruption; cannot be vrddhi-formation).

ai, vrddhi to MIndic (epenthetic) i

3.70. As in Pali gela $\tilde{n}\tilde{n}a$, vrddhi formation to gil $\tilde{a}na =$ Skt. gl $\tilde{a}na$, BHS shows gall $\tilde{a}nga$, sometimes juxtaposed with the 'correct' gl $\tilde{a}nga$ (which is actually not recorded in normal Skt.).

o for \bar{u} , chiefly as lengthening of u

3.71. In a number of cases u is lengthened to o_1 where Pali generally has \hat{u} (m.c.) instead. In BHS these cases seem generally determined by meter, but not always (cf. Pischel 77; Geiger 24, but with wrong explanation). Rather common is anopalipla, 'unstained', for an-upalipla = Pali anupalitta and (in verses, CPD; presumably m.c.) anūpa[°] (also found in BHS); it occurs not only m.c. but in prose and in metrically indifferent positions. Similarly, in verses m.c., anopapanna == anupa°; anopalambha = an-upa^o; niropadhi = nirupadhi (Pali also, in vss, nir \bar{u}°); duropagama = dur-upa°. The verb form opalipyase (vs, m.c.) stands for upa° (influenced by anopalipta, above?); and vopasamati (vs, m.c.) = Pali vūpa°, Skt. vyupa°. Twice the 3 sg. impv. ending u is changed to o m.c.: § 30.14. In Gv 212.16 (vs) ovacu = uvaca; the o is certainly m.c. for u. Once the adverb sādhu, 'please', appears as sādho m.c.: LV 41.14. (It would be far-fetched and implausible to understand a voc. sādho, 'good sir!'.) Mv iii.370.1 (vs) has so, m.c. for Pali su = svid; the same verse in Pali has sū. The bird-name konāla may be m.c. for kunāla. In Mmk samodita, 'originated', or perhaps, 'united, colletted', occurs repeatedly for samudia, and likewise samodaya, probably 'origin', for samudaya. Both seem to occur only at the end of anustubh lines, so that o may be considered m.c. for u. But samodita occurs in Pali prose, which seems to indicate that something more than mere metrical convenience is involved.

3.72. Finally, as cases of o for older \bar{u} , we may note anopa for an $\bar{u}pa$, 'marsh'; sāropya, 'suitable', for the usual sār $\bar{u}pya$; yosa for Skt. yūsa (Pali yūsa, Pkt. jūsa).

o for u before consonant cluster

3.73. At least once we find o for u written before ll(cf. Geiger 10.2, Pischel 125), oltānaka (cf. Pkt. oltāna) =ultānaka; the adverbs oltarena, and $pralyol^\circ$, 'after', are probably also to be recognized; perhaps we should count here oddeli, which = Pali both udd° and odd° . Possibly the form written oltānaka, above, was intended as olānaka, by the 'law of morae', with substitution of long vowel plus single consonant for short vowel plus double consonant (§ 3.2). In one other case this is what we find written: olariyali, pass. of ul-lar- 'conquer', representing a more nearly 'regular' ultarīyali; but the form is not certain.

Metrically short o

3.74. As in the case of e (§ 3.64), final o is sometimes written in positions where meter requires a short; e. g. kofiyö Mv i.61.10, 13 (in 13 preceded by trayõ; see Senart's note, p. 422); durbalõ alpasthāmo LV 188.6 (so mss., ed.

em. durbala; before this word purușõ is also read by most mss., ed. purușa with a minority). Others, § 8.19.

o for ava

3.75. This regular Mindic change is very common here. Forms of the root $bh\bar{u}$ are frequent; rarely holi, the regular Mindic form, for which BHS usually has bholi etc.; these occur on almost every page in verses. For examples see Chap. 43, s.v. $bh\bar{u}$ (2). It should be noted that sometimes *ava* is written where meter proves that a must have been pronounced. So Ud xvi.16 *bhavali* (no v.l.); the line can be read metrically only with *bhoti*.

3.76. Very commonly o represents the preverb ava. Some of the following are omitted in the Dict. when the form with ava- is Skt. So in prose, chiefly of Mv: oruddha (= avaruddha) i.22.11; ovādena ovadanti 104.9; Anotaplagālra (= anaval°) 141.10, 15; onamati 158.15; osrstā ... obhasilam i.205.10-11 = ii.9.6 (in the latter avabhasilam); obhasenlam i.213.13, 15 = ii.16.17, 20 (in 20 printed as vs); obhāso i.227.6; osakta- 231.12, etc.; also in Mvy 5524 ovidhyana-khā = avavi°; and in Bhik 28b.5; 29a.1, 4 (prose) osāraņā = Pali id. (Mvy 9306 avasāraņam). In verses everywhere: opatantam SP 83.2; otaritvā 114.5; onahali 126.2; anolino 278.9 (an-aval°); abhyokiri 325.4; obhāșito LV 29.2; okrānta 47.9; olambate 134.18; oruhyā 271.12, 14; orodhe My i.143.7; 200.3 = ii.3.21; olari Bhad 29, 30; okramatām Gv 1.16 (here, in an āryā, ava- would be metrically as good); ovrstå Ud iii.9 (so oldest ms.; later var. avavrsta, unmetrically); etc.

3.77. Other, miscellaneous forms (in vss): lona = labana; prona = pravana; (prose) upoșali.

o for au

3.78. This regular MIndic change occurs often in our texts, and is not limited to verses. A few examples: $orasa = aur^\circ$; $opalambhika = aupa^\circ$; airosit LV 143.13; 300.11-12 (both prose), and similar forms passim; $osadha = au^\circ$; $kroñca = krauñca; kopina = kau^\circ$; $odumbara = aud^\circ$.

o for final a, m.c.

3.79. Frequently, in verses we find o for word-final a metri causa. This has been supposed to represent the particle u added to the form in a; so Kern, Introduction to SP transl. (SBE 21) xvii; La Vallée-Poussin, MadhK 166 note 4; Hultzsch, Inscr. of Asoka Ixxiii. The same scholars connect our co (= ca) with Asokan cu, which appears to be adversative, 'but'; it is replaced in the Girnar Asokan inscriptions by tu, and was explained by Michelson (IF 23.261) as a blend of ca and tu. Whatever its origin and meaning, I do not believe it has anything to do with our co, for reasons set forth JAOS 66.203. It appears only as cu (there is one doubtful and conjectural reading cū), whereas our texts never have cu, only co. As to meaning, while our co, like Skt. ca, could sometimes be translated 'but', there are many cases where this would be impossible. The Asokan inscriptions contain no other form in u (or o) for final a, while many such forms in o exist in BHS; in most of them a particle u could make no sense. Finally, the assumption that the particle u is concerned is in itself highly implausible. Even in the Veda, u is almost never attached to ca (only once in RV acc. to Grassmann, 6.66.3; never in AV acc. to Whitney); but co for ca is extremely common in our vss. In Classical Skt. u occurs, it seems, only after na, atha, and kim (BR, s.v.; Renou, Gr. p. 520), while in BHS virtually any final a may be replaced by o, but only in verses m.c.

3.80. The true explanation of this o for final a is to be sought in the frequent variation between final a and o (for Skt. ah etc.), see § 4.32, in adverbs and particles

as well as noun forms and even verb forms (ma: mas, mo in 1 pl.). Since this variation was constantly utilized for metrical convenience in BHS verses, it led to the metrical use of o for any final a as an analogical back-formation.

3.81. The frequency of o for a varies strikingly in different texts, tho it is found at least occasionally in all older texts which contain many verses. In SP it is extremely common, tho there are often ms. variants with unmetrical a, and frequently KN follow the latter (their notes prove that they did not understand the meter of SP; I shall not trouble to note these errors in the following list). It is also common in Gv, but less common in LV, and rare in Mv.

3.82. It is specially frequent in adverbs and particles: co SP 13.15; 14.12; 63.11; 95.12; 96.15; 112.2; 119.9; 129.3; 130.9; 131.1, 10; 190.7; 204.12; 208.2; 323.12; 358.5; 364.12; 369.5; 1.V 43.3; in Suv mss. and Nobel frequently (e. g. 22.7; 24.1) read unmetrically ca where Siks, quoting from Suv, has the metrically correct co; Gv 213.17; 254.18; 255.5 etc.; Siks 260.10. Also adyo = adya SP 118.13; 169.14; LV 92.8; 283.2 (Lefm. em. adyd); ivo = iva (?) LV 333.16 (but see Dict. s.v. oruṇḍa); iho Siks 4.15; Gv 384.8; hanto = hanta Dbh.g. 7(343).14 (cf. hantā line 5 above); 21(357).16.

3.83. Numerals: pañco SP 14.11; Šikš 243.13 (so read with 2 mss. reported p. 412, note; for pañca, sc. abhijňäh; wrongly interpreted by Bendall whose ms. has pañcyo, phonetic corruption); sapto Gv 255.23; daśo SP 67.14; 98.10 (mss. and WT); 165.2; 198.7; Dbh.g. 40(66).24.

3.84. In noun declension, o occurs rarely for a as stem in composition, and as ending of voc. sg.; these may be morphological rather than phonetic (nom. sg. form used by transfer, §§ 8.12, 28).

3.85. Instr. sg. eno for ena: teno, one ms. for tenđ SP 27.11 (certainly not tenđkušalena as suggested by Burnouf's note, cf. 22.6); 93.3 (so one ms. and WT); 119.1; 195.4; 325.12; yeno SP 92.2; Gv 231.16; keno SP 175.5.

3.86. Gen. sg. asyo for asya: nivefanasyo SP 114.2 (mss.); lasyo SP 118.12; 150.4; 175.8; 192.5; 194.4, 7; 213.5; 295.2; Siks 112.3; Mv ii.377.9 = 378.3 (with mss.); 393.21 (so with v.l.); *jinasyo* SP 194.7 (with v.l. and WT); Gv 255.22; yasyo SP 165.2.

3.87. Personal pronoun: mamo = mama SP 326.4 (with v.l.; or mama, WT); Gv 231.15.

3.88. Verb forms: gaccho hi Gv 208.11, for gaccha hi; anušiksatho Gv 481.22 (2 pl. impv., for usual °thā = °tha).

i for a

3.89. In suffixal forms, for Skt. (l)ama, we find (l)ima as in MIndic; see §§ 22.13-16; on *i* for thematic *a* in verbs, § 27.10.

Middle Indic representatives of r, and Hypersanskritic r for other vowels

3.90. a for r. On addaśi and similar aorist forms of drś see §§ 32.25, 111. On a as stem-final for r in r-stem nouns see §§ 13.3 fl. Other cases are sporadic: a-hasila-keśa (so mss., to be kept; Senart em. sahila^o), 'with hair not standing on end', cl. Pali hasila = Skt. hrşila, 'bristling' (of hair); vaiyāpatya = °prtya, °vrtya; tālavanļaka 'fan' = Pali 'vanļa, Skt. °vrnta; amata = Pali id., Skt. amrta; samvimatlha Dbh.g. 22(358).19 = *samvimrşta (which ed. vimrşta(-ka); kata-punya (?) = krta^o, Pali katapuñña; sajati, and utsajati = (ut)srj^o; kasina (Pali katapuñña; sajati, and utsajati = (ut)srj^o; kasina (Pali katapuñña; Vaji = Vrji; gaha Bbh 231.11, 26; 232.7 (prose), possibly for grha (Pali gaha), but this is questionable; the context, placing it between stūpa and kūta (or kūlāgāra 231.26),

indicates that it denotes a part of a shrine, specifically a part below the $k\bar{u}|a$ or top; the Chinese character given for it in Woglhara's glossary means 'pagoda, shrine'; also (perhaps originally?) 'box, container'; does this point to graha (root grah)? note that graha is sporadically recorded for grha even in Skt. (hyper-Sanskritism for MIndic gaha?); gahastha = Pali gahaitha, Skt. grhastha; valta = Pali id.. Skt. vrlta (also ali-valla? Dict.); vadda, a MIndic form of Ski. wrddha; perhaps acchafa (Dict.).

3.9i. ! for **r**. On *-i* for stem-final *r* in *r*-stem nouns see § 13.20. Otherwise only sporadic, and not always certain textually: asadisa, see Dict. s.v. sadrsa; perhaps alilling for *alplingis* (cf. Pali *atiliti*); perhaps *upadisyali* = "*dris*" (§ 31.1, s.v. *pad*); perhaps *ultini(karetsu)* for *ultrnic*; *iddhi* (? perhaps *riddhi*) for *rddhi*; *masina* (read °*na*?) = AMg. *masina*, Skt. *masrna*; *Isidalla* for *Rsi*?; for Skt. *krecha*, our dialect seems to be isolated in showing *kisara* (Dict.); cf. Pali kasira and kiecha; ohila = apahrla; *idyin*; *nipaka*(1).

3.92. u for r. On -u for stem-final -r in r-stem nouns see §§ 13.21 ff. Otherwise sporadic and usually (as in MIndic) after a labial, or when u occurs in an adjoining syllable: pucchati (Pali, Pkt. id.) = pr° ; vukka = wrka (not recorded in MIndic; Pali vakka); vuddhi = Pali, Pkt. id. or vuddhi, Skt. vrddhi; paliguddha; janetukā = *janay-itrkā (not recorded in MIndic); ujju (= Pali ujju and uju, also uj(j)uka), Skt. rju; utu (Pali id.) = Skt. rtu, implied in sarvotuka; anušamsa (q.v., Dict.), etc., if derived from δur° .

3.93. For *abrdha*, q.v. in Dict., Divy 84.10 text *arūdha*, intending doubtless *abūdha* (= Pali *abbūlha*); the r is originally long, Wackernagel I § 28.

3.94. ri for r. Always after preceding final vowel, but mostly in prose: riddhi = rddhi; $ridhyale = rdh^{\circ}$; risi = rsi; Risidalta; rilu = rtu; riksa = rksa.

3.95. Hypersanskritic forms concerning r. Hypersanskrit r for a: anuddhrla = Pali (and Skt.) anuddhala; perhaps upahrla for upahala.

Hypersanskrit r for i: mrla- (for other texts mila-) vārsika MSV iii.124.3, 5; Tib. thun nu, 'short', = mila; kritas, cf. Pali kittaka?

Hypersanskrit r for u: perhaps in sampravrlla, intending °vulla (to vac) at least as to sense; ilivrlla(ka); samvrli (§ 2.30).

Hypersanskrit r for ri: ucchrla and ulsrla = ucchrita; kartma; Irdhā or (v.1.) trvidham, also Irbhava, for tri-; rkla = rikla; mryali, °le, mryyali, °le, = mriyale; lādrnah (read °nah?), gen. sg., = Pali tādino, which to be sure historically represents lādrś-, so that the r is historic. On dhātrbhih for dhātribhih see Dict. s.v. dhātr; phonetic hyper-Sktism is doubtless involved, but surely also riming assimilation to mātrbhih.

3.96. A possible hyper-Sanskritism is *pralikrusla*, 'bad, inferior, low'. It seems to be falsely Sktized from Pali palikultha = palikiltha, both historically perhaps from BHS and Skt. Lex. pratikrsta. But see both words, and pratikrofa, 'stati, in Dict.

Should such forms as *srunuya*, *sruno*, *srunali* etc., and even (§ 3.26) *srunute*, be classed here, as Sktizations of MIndic suno-, sunu-, suna-? Cf. Chap. 43, s.v. *sru* (1 and 3). Altho Skt. *srno-*, *srnu-* doubtless derive from *sruno-*, *srunu-* by dissimilation, I do not believe BHS has a direct inheritance of those prehistoric forms.

Vowel assimilation

3.97. I have noted little that calls for comment here, after § 3.57. Our language shows a few of the forms noted for Pali by Geiger 16, 17, such as ucchu = Skt. iksu. Contrary to Pischel 101, I regard seyyä (Pali id., Pkt. sejjä) as showing change of Skt. a (isuyä) to e owing to

the following yy. It is not at all likely that *i* in Skt. *iijumāra* (popular etymology?) is really older than *u* in BHS *juiju*⁰, AMg. *susu*⁰, Pali *sumsu*⁰. Geiger treats Pali *timissā* (BHS *timisrā* etc.) as having *i* by assimilation from Skt. *lamisrā*, but Skt. has *i* in the obscurely related *limira*; are the Pali and BHS forms blended with this? Some other alleged cases are hardly certain enough to be worth recording here; e. g. Geiger assumes that Pali *matlhalunga* (BHS *masla-, maslaka-lunga*) = Skt. *maslulniga* has *a* in the antepenult by assimilation to *a* of the first syllable, but it seems quite as likely that if assimilation was concerned in any of these forms, it produced Skt. *u* (instead of *a*) in the antepenult by influence of the *u* of the penult. On *i* and *u* for Skt. *r*, sometimes presumably determined by the quality of a following vowel, see above. Like Asokan, BHS has *udupāna* for *uda*²; and for *udapāsi* (Dict.) the variant *udu*⁰ occurs.

Epenthetic vowels

3.98. MIndic epenthesis, usually in words occurring in identical form in Pali, is not rare, but almost limited to verses m.č., except for Mv which also has it in prose; even in Mv, however, it is commoner in verses.

3.99. Epenthetic a. Pali ralana (= AMg. etc. rayana), for raina, occurs as such or in cpds. and derivatives in the prose of Mv not rarely, elsewhere in verses and chiefly (the not always) m.c. Even in Mv note ratana repeatedly in vss i.108.8 ff., while rainani occurs in prose 108.5. Prose examples of ralana and compounds or derivatives: My i.31.5; 54.7; 62.19; 63.9; 89.14; 137.5; 138.10; in 1.193.16 f. (prose) stems raina and ralana occur side by side, apparently at random. Verses where raina would be metrically possible but text reads ralana: SP 353.2 (beginning of triştubh; ralanehi 'hypermetric', ralnehi somewhat more regular); LV 109.8 ralana- in särdülavikridita, ralna- would be more regular; anantaratano My i.55.9 = 56.15 (in āryā; °ralno equally good); ralanaparvalo Mv i.113.10 (id.); ralanajinam My 1.119.9 (id.). Verses where meter favors ralana: Prabhūtaralano SP 252.5 (in adjoining prose °ratna); ralana- SP 11.4; 16.6; 212.11; 213.3; LV 49.14; 59.7; 167.9 etc.; in LV 292.3-5 raina- and ratana- both occur repeatedly, as meter requires; Mv 1.118.16; 119.10; 186.17; Gv 256.7; 257.11; 258.13; 259.1; 284.23; 285.12, 17; Suv 149.11 (Ralanoccaya, cf. Ralnoccaya 147.4 etc.); 23.10; and many others. RP

3.100. Another case is araha(n)t-, araha- (Pali id.), for Skt. *arhan*-, and the related adjective araha- = Skt. *arha*- 'worthy'. For the latter see Dict.; the former as araho = arhan Mv i.37.6 (prose); in verse where the epenthesis is not metrically necessary Mv i.247.12; where it is necessary LV 45.10; 47.9; 354.13; (in 376.11, 378.11 by em.;) Mv i.143.5; (and by em. i.101.8).

3.101. The remaining cases of epenthetic a are sporadic. and not always certain. Clear are garahati = Pali id., Skt. garhati; it may be called m.c.; altāttamīya = Pali altatlaniya, Skt. ālmātmīya; kisara = krcchra (ci. Pali kasira, and Geiger 59.2); probably also manojana = manojña (vs; may be m.c.), altho no form of this word with epenthesis seems recorded in MIndic. In oblique cases of the n declension which in Skt. would show no a before n, we find (in vss, m.c.) not only murdhana == Pali id., Skt. mūrdhnā, but also nāmana - nāmnā (Pali and Pkt. apparently only nāmena); see § 17.49. Others: ruśami = Skt. raśmi, m.c.; no epenthesis is noted in MIndic in this word; sukhama (Pkt. id. and suhama) = sūksma; supana probably for svapna, but possibly for svapana; otherwise the BHS equivalent of svapna is only supina, § 3.111; pratisrulakā or $^{\circ}ka$, perhaps = pratisrulkā with epenthesis, but may be a ka-extension of *pralisrula = AMg. padisuya, masc., Rāyapaseņi Sutta 40, acc. to Ratnachandra (1 have no access to this text), or of *pralisrulā = pralisrul (§ 15.9) with shortening of ā m.c. (the nieter is obscure to me). In Šikş 4.10, 11 (vss) jālaya-, apparently for jālya-, is probably an error for jāliya-.

3.102. Epenthetic i. Before y: As in Mindic generally, this is specially common before y. There is no correlation with the quantity of the preceding syllable ('Sievers's Law'); it is just as common after a single consonant preceded by a short vowel as otherwise. It appears standardly in certain morphological categories, which as a rule will not be dealt with at this place but in the morphology. See under oblique cases of fem. *i* and *i* stems (-*iy*-for -*y*-, § 10.86 etc.); gerunds in -*iya*, (§ 35.11 etc.); passives in -*iya*-(§ 37.4 etc.). In general it is limited to verses, but Mv shows it frequently in prose, and I have noted a case in the prose of Bbh (235.1 pariyesitalith, for parye^o). Prose cases will be identified as such, here or in the Dict. Once, at least, we find a Prakritic spelling -*ia* instead of -*iya*: *acintia*.

3.103. Very common in all texts is the adjective written acintiya, even in prose in Mv and in verses elsewhere (note contrariwise, c. g., acintya SP 7.6-7, in prose). In prose of My, e. g. ii.362.11. In verses: SP 30.15; 62.6 (acintiyālah = acintyāl); 68.1; 176.12; LV 54.1, 4; 171.6; Divy 79.21 ff. = 469.5 ff.; Suv 23.7; 28.7; Samādh 22.29; Bhad 28; Gv 24.18 etc. Other adjectives in iya = ya are so extremely common that I am not sure they are all recorded even in the dictionary. The following, and quite a number of others, will be found there: aganiya, avivarliya, anivartiya, aprativartiya, atuliya, amāpiya, karmaņiyā, akampiya, nindiya, prasamsiya. And noun stems in ya: śâkiya = śākya, common in prose of Mv. e. g. i.351.14; 352.15; and in vss, Mv i.43.15 = 337.11; ii.24.5; LV 76.3; 94.9, 10; 133.5; brahmacariya, cariyā, pāricariyā, vīriya-(vant), harmiya. Also with MIndic shortening of a long vowel in a preceding syllable: kariya, ariya = arya, ācariya, viriya (note also vīriya, above!), suriya, turiya, §§ 3.33, 38, 45.

3.104. In compounds (either noun cpds., or forms and derivatives of compound verbs) in which the prior element ended in *i* which in Skt. appeared as *y* before following initial vowel: $viy\bar{u}ha = vy\bar{u}ha$, besides citations in Dict., LV 74.12; 81.13; Mv ii.275.17 (prose); Bhad 6, 34; $viy\bar{u}hana;$ triyadhva- = tryadhva-; pariyanta; $viy\bar{u}karoti;$ pariyesate.

3.105. One doubtful case before the present suffix ya: apadiyema Mv iii.74.5 (prose; for apadyema, opt. to apadyati; but v.l. apadyiyāmah).

3.106. The reverse of epenthesis occurs, possibly m.c., in kyant for kiyant, 'how many'; and (not m.c.) in $ny\bar{a}ma = niy\bar{a}ma$; cf. § 3.112.

3.107. Before other consonants than y: these cases, too, are almost limited to verses, except that Mv has them also frequently in prose. In prose of other texts I have noted only Supinānlalokamuni Mmk 130.4 (n. pr., for Svapnā^o). All our examples concern words which show the same epenthesis in Pali, except (sam)sparisa which shows it in Pkt. (cf. AMg. pharisa, sampharisana).

3.108. Before r: *firi*, *firi* (also spelled with s-) = *fri* LV 122.16, 18 (read *hirisiribharite*, one word); 123.2; 135.10; 272.4; 362.21; 391.4; Mv i.62.10; 142.11; 276.14; 297.11; *ii*.57.2, 8; 363.15; GV 235.24; *iirimanl* (*sirimal*) Mv i.295.21; 305.18; *Mañjuśiri* (= °*srl*); *hiri*, *hiri* = *hri* LV 122.18; 139.6; f58.20; 292.17; 362.21; 391.4; Mv *ii*.59.22; 61.19 etc.; 64.3 (prose); *ahirika*; *hirinişevi* Ud xlx.5 (so oldest ms., later var. *hrinişevi hi*, Sanskritizing the form and patching the meter with *hi*); *ahiri* Ud xx.5 (so oldest ms., later var. *ahriko*, cf. preceding); *vajira* (Pali id.) = *vajra*.

3.109. Before I: gilāna, gilānaka-tva; kilānta (Pali kilanta) SP 363.12; LV 220.6, 8; Mv i.227.16 (prose); akilāntaka LN 359.20; kilanta (with MIndic short a) Mv iii.118.20; parikilanta Mv i.217.16; kilamāmi Mv i.343.5 (twice; prose); kilamatha (Dict.); kilefa (Pali kilesa) LN 46.3; 48.5; 53.11; 276.12; Mv i.116.16 (dhuta°); ii.115.2; -upakilefa Mv i.228.11 (prose); kilisfa LV 131.17; Mv iii.446.1; samkilisfā va yas tapah Ud xi.3 (oldest ms.; later var. samklisfam vāpi yat tapah, with Sktization and patching of meter); filisfa (= filisfa, Pali siliffha); pilaksa (= plaksa, Pali pilakkha); milāyati (Pali id., = mlā°), milāyila, pramilāyila, milāna, Mv ii.393.9; āmilāla and saŋımilāyati, saŋımilāta.

3.110. Before sibilant: spariša, samspariša = (sam)sparša.

3.111. Before nasal: in oblique case-forms of rājun, § 17.50; tasinā (Pali id.) = trsņā; kasina (Pali kasiņa) = krisna; supina (Pali id., Pkt. suviņa) = svapna, common.

3.112. The reverse of epenthesis, hyper-Sanskritic dropping of a historic interconsonantal *i*, seems to occur in *vyavaslah*, apparently m.c. for *vyavaslah*; cf. § 3.106.

3.113. if inally we have **prothesis** of *i* in forms of *strī*, as in Pali and Pkt. *itthi*, *itthī*. The following 'stems' are noted in our texts: *istri* or *istri* (also *istrigāra*); *istrikā* (Pali *itthikā*, AMg. *itthiyā*). Peculiar to our dialect are the stems *isți* etc. and *isțikā*, with domal *sț*; no MIndic **itțhi(kā*) or the like scems recorded.

3.114. Epenthetic u. This occurs here only (as in Mindic usually) before or after labials. The labial precedes only in forms of -apunoti (prapunoti SP 294.8; 371.8, vss) or more usually -apunali (Chap. 43, s.v. ap), = Pali pāpuņāti etc. Otherwise the labial always follows. Most of the words concerned show the same epenthesis in Pali and Pkt. So paduma = padma, very common; in prose, Mv 1.228.1; 229.8; 308.5; ii.109.4, etc.; Šikş 75.8, 10; in verses, SP 313.3; LV 137.21; 367.3 (read padume with best mss.); Ud xviii.5 (so oldest ms.; later var. padma); etc. Equally common in Mv, even in prose, are suve (also suve as in Pali) and derivatives, and duve (Pali and Pkt. id.) = dve 'two' (§ 19.3). Outside of Mv it is rare: trimsaduve '32' Gv 406.15 (vs). Other forms: -duvāra = dvāra; madhuwāsava = Skt. and Pali madhvās^o; lriyadhuvā = tryadhvā Dbh.g. 51(77).8 (here u actually spoils the meter!); sukhuma = sūksma. The 3 pl. verb form śrnvunti appears as srnuvanti LV 236.10; Dbh.g. 13(349).1 (both vss and may be m.c.).

Samprasarana and the like

3.115. As in MIndic (Geiger 25; Pischel 151-2), and mostly in words and forms which also appear there, we find substitution of i, ī, e for ya, yā, and of u, o for va, vā. BHS presents little that is new in principle. Both Pali and Pkt. often have vili- for vyati- at the beginning of compound verbs and their derivatives. So here vilikrama == Pali vilikkama, and others (Dict. s.vv. vili-). The gerund ending i (i) is, in my opinion, best derived from ya (§ 35.49). It occurs otherwise, apparently, only in Ap. The stem cari, cari 'conduct' is, I think, certainly a phonetic equivalent of carya; it seems not to be recorded anywhere in MIndic, not even in Ap. It is aimost limited to verses, very rare in prose, and is regularly and very commonly used m.c. for caryā. Quite similar are īrsi = īrsyā; sayi = sayyā; celi = cailya. In Šikş 325.13 (from Vimalakirtinirdesa) acintisu certainly means acintyesu and seems to imply a stem acinti = acint(i)ya; acint(i)yeşu would be metrically impossible. MSV has abhişûyase for abhyas^o (Dict.). On liriccha (implying tiri for lirya), and lirthikara, perhaps for lirthya°, see Dict. For styana occurs thina (Pali id.) and semi-Skt. slina. Textually uncertain are dvihika, trihika, with i for ya.

3.116. The verb *vedhali* = Pali id. is certainly a MIndic equivalent of *vyalh*. In Mv i.156.17 (vs) tapetu

seems to stand for (unmetr.) tapyatu (so Senart's note, with alleged parallels, mostly doubtful, being variant readings of only a few mss. and not favored by meter). The same may be recognized in Gv 481.1 (vs) *itksisenti*, for *"syanti* (unmetr.), 3 pl. fut. **3.117.** As in MIndic, *du*- occurs especially in the Mv,

3.117. As in MIndic, du- occurs especially in the Mv, tho not commonly, for dvi-. Similarly suda (for sudam = svidam); supana, supina = Pali id. for svapna, and the verb supili etc. (Pali supati), also sopita (cf. Pkt. sovai), both to Skt. svapili (svapati), Chap. 43, s.v. svap, with osopati and osopana; sosiika = svaslika; doşa 'hatred', for Skt. dveşa, is like Pali dosa a homonym of Skt. doşa 'fault'; sora m.c. for svara.

Loss of a syllable

3.118. Pischel 149, 150 end, and Jacobi, Ausgew. Erz. xxvi, §16.3, call attention to the occasional loss of

syllables, perhaps by contraction of vowels after loss of intervocalic stops, in MIndic. So andhāra = AMg. id., Skt. and Pali andhakāra; manāpu = manaāpa; angulikā, prob. for anguliyakā; em = AMg. id., Skt. evam. An example, unknown in MIndic, seems to occur in mrilikama = a theoretical Pkt. *mallikama for *mallikamaya or "maa = Skt. mrttikamaya (for mrttikāmaya, m.c.; Skt. has such cases of mrttika-). Another plausible case seems to be samanvi, apparently for Skt. samanvilah, implying MIndic loss of t and vowel contraction. Like Pali, BHS has yāgu or yāgū for Skt. yavāgū; and BHS only writes repeatedly yvägu, °gū, instead of yägu (in vss always two syllables); this seems to be only a strangely perverted attempt to Sanskritize the Mindic yāgu in writing (cf. Senart Mv i note 393). Monier Williams cites a Skt. Lex. yväguli or °lyä, 'sour scum of boiled rice', which I have failed to find in BR, pw, or elsewhere.

4. Saındhi

(32)

Loss of initial vowels in samdhi

4.1. In BHS, as in MIndic (see e. g. Geiger 70.2b), either of two originally adjoining vowels could be lost in samdhi, with or without 'compensatory lengthening' of the remaining vowel.

4.2. In the case of some particles or common 'light words', the form without initial vowel acquired, by analogical extension, a kind of status as an independent word, tho usually the form with initial vowel continued to exist. Geiger 66 lists as such forms in Pali va = iva and eva, pi = api, ti = iti, $d\bar{a}ni = id\bar{a}nim$, $hellh\bar{a} = adhastal$, and the pronoun stem na = ena. All these occur in BHS also ($hellh\bar{a}$ in the form $hesl\bar{a}$, and other related forms). The fact that such forms have come to be used independently of vocalic samdhi is shown by their use after a word ending in a nasal (ordinarily written m; no other final consonant is normally tolerated in MIndic, and we very seldom find any other preceding such forms in BHS), or even at the beginning of an utterance.

4.3. We find after anusvāra, pi for api (chiefly in verses except in Mv): tuham pi SP 67.11; dattam pi 95.12; ekām pi 98.12 (? most mss. ekā-pi); pūrņam pi 99.5; asmākam pi 147.5; aham pi 237.1; Av i.198.6 (prose); LV 138.6; ekosatlvam pi SP 254.10; tvam piha 294.10; muhūrtam pi 336.4; in prose of Mv, anyešām pi 1.16.12; 18.5; mānsam pi snāyum pi ... sarvam pi 18.11-12; nirayam pi 102.17 (Bodhisattvas may go 'even to hell'; Senart wrongly inserts na in preceding); talhedam pi 158.13; vayam pi ii.218.14; jñātīm pi iii.91.5 (? v.l. jñātī pi); and in vss of Mv, sarvaratnam pi 1.185.15; vayam pi 186.16, etc.; nāmarūpam pi Samādh 22.25.

4.4. After another consonant: *abravit pi* Mv ii.135.14 (vs; one ms. omits *pi*, unmetrically).

4.5. ti for ili, after anusvāra (sometimes written n; chiefly in verses except Mv): bhāsitam ti SP 62.11; mamāyam ti 113.10; krtam ti 292.6; aham ti LV 308.11; others Mv i.8.11, 14 (prose); ii.205.11 (vs); tuvam ti ii.216.4 (prose), etc. At the beginning of a line of verse, Mv i.104.18, meter seems to demand ty- for mss. ity-ena (Senart ity-ena). Certainly there is no samdhi with the preceding line, which ends narottamah (all mss.).

4.6. va occurs for iva (rarely eva), after anusvāra, see Dict.; after another consonant: -nidhir vā LV 80.15 (vs; so all mss.). dāni (also dānim) is common for idānim in Mv, often after anusvāra, and even in prose; in other texts only in vss, not common. So also hestā, heste, hesti, hestena, hestima, etc. (also written with th for t), often after m and at the beginning of sentences and lines, in prose of Mv and in verses elsewhere. The pronominal stem na occurs at the beginning of a prose sentence and of a line of verse (§ 21.48) and is not restricted as to a preceding final.

4.7. In other words than the above, loss of an initial vowel after final anusvāra is extremely rare, and some of the apparent examples are textually doubtful: vagū-hayel = ava° Mv i.304.15 (vs; but v.l. °mālā for °mālām preceding); (ghrānabalam vipulam) smi = asmi Gv 231.5 (vs; metrically correct); also si = asi, after consonants in prose, see § 28.64, where both these forms may belong; dārikām ham = aham Gv 413.23 (vs, metrically correct);

dhigama = adhigama Lank 144.10 = 324.5 (vs, after rālryām, but in 144.10 v.l. rālryā).

4.8. In Mv i.112.14 *śiti* (see Dict.) seems to be used for *aśiti* at the beginning of a line of verse.

4.9. After a final vowel, an initial vowel is quite commonly dropped, but mostly in verses where it could be called a matter of meter; so in all the following examples except where the contrary is stated.

4.10. When the preceding final vowel is identical with the initial, of course it makes no difference whether we consider t. z initial or the final vowel dropped. Such cases have been classified as showing elision of the *final* vowel (§ 4.20 ff.), except when the following word is one like api (pi), iva (va), etc., which is known to be otherwise used as a monosyllable.

4.11. Initial a dropped: tatha pi = tathapi SP 26.9; dcśa pi SP 31.11; adhimukti (for °tir, n. sg.) pi SP 235.10; māŋsacakşu sya, diuyacakşu sya SP 355.8, 9; ca pi Mv i.204.9 = ii.8.4 (Senart cāpi with one ms. in i.204.9, but against meter; mss. in ii.8.4 are corrupt but do not support cāpi); hi si = hy asi Mv iii.8.14 (vs, but hy asi would be metrically as good, no v.l.); ksetra pi Samādh 19.16; kārī pi Mmk 135.14 (kāry api would be unmetrical); su-laņkritkā = sv-al° LV 322.5; api nityā (for apy anityā) LV 324.9; paśyanti nanya- Dbh.g. 17(353).16; satyeşu Jiâdnu (for aj°) Dbh.g. 27(53).13; jñāyati ranyagato (= araņ°) Sikş 329.13.

4.12. After final \tilde{a} for $\tilde{a}s:$ -cittā pi SP 84.1; ustrā tha (for atha) SP 94.11; kubjā tha SP 95.6; damarā tha SP 96.3; parvatā pi Av li.136.10; anutpannā niruddhāj (for ani^o) Samādh 8.5; devā pi id. 8.23; abhāvā naksarāh (for ana^o) id. 8.6; nairayikā pi id. 19.17; sahitā 'nāgārāḥ Divy 27.22. In prose of Divy: varsā tikrāntāh 15.2; tīrthyā bhihitāh 164.1; and in prose of Av repeatedly, but regularly 'emended' by Speyer, see his note on i.20.15.

4.13. Initial \bar{a} dropped: in the word $\bar{a}lman$ and derivatives this occurs after final e, o in early and epic Skt. (cf. Edgerton, Language 19.116 note 64), and occasionally here: bhavate 'lmabhāvah SP 353.1; 370.10; read māno 'lmikā LV 215.13-14 with best mss. as meter requires. And rarely in other words: raye 'ryā (= āryā) Divy 399.12 (vs, confirmed by meter).

4.14. Initial *i* dropped after vowel: *ti* for *iti* after final -0 SP 26.6; Mv i.4.5 (prose); 112.1; 114.12; mimāmsaye *ti* Av i.94.3 (prose; so read with mss.; see § 38.21); sthero *ti* Ud xi.12, changed in a later v.l. to sthaviro, omitting *ti* and so preserving meter; supino *ti* Samādh 19.26; va for *iva*, Dict.; others, ye mam (= *imam*) LV 57.3; defayisye mām (= *imām*) Suv 29.1 (so best ms., as required by meter; text *imām*, unmetr.).

4.15. Initial i dropped: bhavisyase 'svaravarah (= isvara°) LV 326.19.

4.16. Initial u dropped: kuksau papannah (= upa°) LV 56.8; kleśa-padrulām (cpd., = kleśopa°) LV 178.22; anyalamo pāsakas (= upā°) Divy 56.7 (prose; hardly a cpd.); in Mv i.189.16 (vs) read probably gaccha tvarsayo (= tvam, reduced to tva, rsayo; two mss. tva rsayo) pagamya (= upa°); in RP 59.5 ulpādayecchu is m.c. for ulpādaye(l)ucchu or -icchu (Skt. iksu).

4.17. Initial e dropped: kuksigatā va (= eva) Mv

i.144.13 (prose), and others, some uncertain, Dict. s.v.

va (2). 4.18. With compensatory lengthening of preceding vowel, elision has been noted only with ili, appearing as ti. It is however not restricted to cases where meter requires a long; indeed it even occurs in prose in gacchatū ti Mv i.362.10; bhavatū ti ii.210.11; iii.193.8; sādhū ti ii.491.12. In verses, but in the first two cases in metrically indifferent positions: sādhā ti Mv i.75.5; 76.3; 149.7; 182.10, etc. In verses, of course, any final vowel may be lengthened m.c.

4.19. Instead of lengthened final vowel, doubling of the following consonant perhaps occurs in thi for (i)ti, as in Prakrit (Pischel 143), in vahisyāma (so read with mss., which have vah° for vadh°; Senart vadhişyāmi) tli (mss. nti) Mv i.17.2 (prose).

Loss of final vowels in samdhi

4.20. This occurs only before an initial vowel. When the two vowels are identical, it makes no difference whether we consider the final or the initial vowel lost, cf. § 4.10. We have included here all such cases except those where the second word has a known form lacking the initial vowel in other positions, such as pi for api, va for iva, etc.

4.21. Final a is lost before initial a (compounds included), chiefly in verses, m.c., very often: pañc' anūnakāh SP 44.8; c' anucamkramanti SP 84.8; c' anekā SP 91.1; sahasr' anekā SP 96.9; ast' anūnakāh SP 335.8; c' avocan LV 50.6; yatr' ayam Mv 1.67.7; c' anāgala 70.1; sahasr' arehi iii.118.4 (compound); c' aham 121.5; agr' amālyo 365.2; -lakşan' alamkrtena Suv 37.3 (Nobel em. -lakşanāl°, unmetr.); var' abhijña RP 5.17; tasy' adhimuktih RP 20.5, 6; moh' avidya- Gv 482.9 (cited Siks 102.3, where note calls it 'a curious form'! it is entirely normal). Uncertain is SP 113.5 (vs) so cā dhanī, hoth edd.; acc. to KN note, all mss. ca, followed in Nep. mss. generally by dhvani (lam svaku putra drstva), 'and seeing him, his own son, on the road' (c' adhvani); only Kashgar rec. dhani (before which ca would be unmetr.), but this is supported by Tib. phyug po, 'the rich man', and by 114.3; if this is the true reading, cã (or co), or cad-, must be read before it.

4.22. In prose, drstvān' aho Mv i.8.11; c' aparādhā-(printed ca parādhā-) for ca plus aparādhā- Mv i.132.10 without v.l.

4.23. Final a or \bar{a} may be lost before initial *i*, chiefly in verses m.c. Such cases could be regarded as metrical shortening of theoretical e to i, but are probably better treated as elision of the prior vowel in samdhi. A few examples out of many: sthilv' iha SP 12.5; v' imās 15.5 (vā, or eva, imās); mam' idam 90.2; c' iho 90.8; sukhāpanārlhāy' iha 92.12; mam' iham 94.2 (mama-iha); ady' ina 1964; c' ime 303.7; may' iha LV 57.7 (mayā-iha, not mayi ha with Lefm.); yen' imau Mv i.68.4; yasy' iyam i.73.11. 15; śrutv' imu Samādh 19.33; yasy' imu Bhad 49. In a vs, but not m.c.: yen' icchakam Mv ii.484.8 (= Pali id., elsewhere BHS yenecchakam).

4.24. Final a or \ddot{a} may be lost before initial u or \ddot{u} , chiefly in verses m.c.; but also, at least in compounds, in verses, and occasionally in prose, before a consonant cluster, where meter cannot be concerned. It seems that they are best regarded as instances of elision of final a before initial u or \hat{u} , rather than reduction of theoretical (Skt.) o to u. Note particularly ekūna (= Pali id., Skt. ekona, for eka-ūna) Mv ii.494.10, 18; iii.177.7 (all prose). In verses, but of course not m.c., e. g. padumullarāhi Mv ii.192.16; varuttamam ii.194.10, for Skt. padmottarābhih, varottamam; but the real process is probably symbolized better as padum'-utt°, var'-utt° (despite § 3.54). Cf. Geiger 67, 70. In verses m.c.: sthapetv' upāyam SP 91.8; -udyān' upaśobhilam SP 196.8; buddhabalen' upeto SP 207.11; c' udāram SP 214.3; c' udagrāh My i.83.7 (so read with most mss.; Senart ca ud°, unmetrically); prajň' upáyatah RP 7.13; prajñ-upāya RP 5.16; etc., common everywhere.

4.25. Final i is lost before initial i, in vss m.c.; sobhal' iyam SP 9.1; prakāšayişyan! imam ekayānam SP 53.8; bhon! imi SP 92.3 (= bhavan!i-ime); naivās! iha SP 93.8; ābharanān' imāni SP 96.14; jvalan! ime 'dya SP 165.2, so read with all Nep. mss. (which must intend this; cited as jvalanli megha in KN note); edd. with Kashgar rec. jvalanti adya, which leaves jvalanti without a subject and is a lect. fac.; muñcām' imam Mv ii.183.20; deśan' ito vā Bhad 54 (so analyze, ed. desayito va).

4.26. Also before dissimilar vowels, in verses: vrajal' ayu LV 173.14 = vrajuly ayu (which is metrically impossible), 'life passes'; citrani (so read with mss.) yajñanayulān' api yasta pūrve (adverb) LV 340.15 (here, to be sure, stany api would be equally good metrically, but there is no v.l., and citrani proves the form intended as onquitani).

4.27. So also final u, before initial u, in verses, m.c.: bah' udikşişu LV 364.11 = bahu udikş^o, 'they have looked longingly for ...'; Tib. sdod. (= wait) cin mchis; snigdhamrd' upadisyate Mv i.85.16 (so 4 of 6 mss.; Senart with v.l. °mrdupa°, inferior metr.).

4.28. Final e may be similarly lost before an initial vowel, possibly thru an intermediate reduction to q or i: sarv' (= sarve) imi SP 117.3; śrņuť (= śrņule) ekagālhām SP 351.1; meşa = me eşā Gv 257.5 = 26 (vs, m.c.).

4.29. Not only an original or theoretical Skt. final vowel may be elided before an initial vowel, but the endings (Sanskrit) as, ās, ān, am, ām (all of which in BHS may be reduced to a, at least in verses, §§ 4.32, 33) may be similarly treated. Examples are rarely found in prose, chiefly in verses when meter requires the elision.

as: aprapl' imam ultamam agrabodhim SP 92.11 'he has not attained this ...; kurvant' imam SP 194.4; marut' ägala LV 114.20 (marulas-ägalās); bhavant' (most mss. °nlo, unmetr.; ed. bhavann) anukampya LV 115.10. In prose: anālhapindadāha Av i.37.3 = anālhapindada (n. sg.) āha.

ās: dharm' ime LV 175.21; 176.19; śānt' ādinirmalāķ Samādh 8.6; parvať aneke Samādh 19.19. ān: (a)ntarakalp' anekān SP 94.4; kalp' anantān Mv

ii.337.13.

am: yānanānālv' upadaršayanli SP 46.12; im' evarūpam SP 95.12; 97.6; 115.3; 291.10; sarv' asesatah SP 115.9; märg' anultaram SP 116.4; Irisähasr' ima loka-dhātum SP 157.12; phal' evarūpam SP 325.8; drumarā-jamūl' upagatah LV 292.18; mahy' adya Mv 1.186.9 (so all mss., Senart ādya); śaky' antarikşam Mv ii.348.4; nûn' aham Suv 226.3 (text aham, meter demands aham or, with 3 mss., nunam aham, two shorts for long); 237.8; sary' adhimucyami Bhad 3 (= sarvam, or sarvah?); viriy' ārabhante Dbh.g. 13(349).8; udān' udānayi (so divide) Gv 489.11. Once in prose SP 390.1, KN print im-evamrapam, probably by misprint (tho not corrected in Corrigenda); WT read imam ev°, stating that their ms. K' reads so.

ām: nānādhimuklān' iha prāņikoļinām SP 45.3; balān' aho SP 87.6; teşa käye aśraddadhantān' ima buddhabodhim SP 95,8; teş' anuśliksamānāh SP 203,10; maharddhikān' astavimoksadhyāyinām SP 205.12; prakāśayantān' iha SP 229.9; cary' (= caryām) anuśiksamānāh SP 310.4; sugatān' iha (WT ihā, m.c.) SP 359.3; bodhisativān' iha LV 311.15.

4.30. Once the ending -ānām, gen. pl., seems to be reduced before a vowel to -a (or, with 'hiatus-bridging' m, v.l. -am): daśāna co antarakalpa (v.l. °pam) atyayāt SP 190.7 'and after the lapse of 10 intermediate kalpas'.

Lengthening of vowel after loss of preceding final

4.31. In MIndic (Geiger 67, 70), the unelided vowel in such cases may be lengthened. Certain instances of

4.311

this are rare in our language. In SP 428.11 (prose!) KN read yeniyam (= yena-iyam) without v.l., but WT with their ms. I.' yeneyam. In SP 85.9 (vs) the Kashgar recension is said to read jaläşila (Nep. mss. mostly samtūşila); WT em. joälāşila; the meaning is certainly 'burnt by flames', and the correct reading involves samdhi of jala (= jvala, a doubtful stem) or jvālā (or an equivalent) with uşila. In Mv i.69.8 (vs) read vaya (m.c.; mss. vayam) pi khādya (so mss.; = khu, for khalu, plus adya; Senart vayam apy adya, which is unmetrical). Also a theoretical (Skt.) final am, presumably first reduced to a, may be elided with lengthening of the following initial vowel: var' ūpagatu LV 135.6 = varcm upagalo.

a in samdhi for as, ās, am, ām, an, ān, e

4.32. Final Skt. as, in MIndic regularly appearing in the generalized samdhi form o, is in our language very commonly reduced to a (as well as u, §§ 3.52, 53), but almost exclusively in verses m.c. Occurrences in case endings of nouns, nom. sg. masc. of a-stems etc., are illustrated in the appropriate sections of the morphology. In verb inflection it occurs especially in the variation of ma with mas (mo) in 1 pl. forms, where it is certainly not wholly phonetic but partly morphological in origin (§§ 26.5, 6). Here we shall mention a few cases of adverblal forms, all from verses and apparently m.c. They may perhaps be interpreted as representing Skt. ah with MIndic loss of visarga. But since all such forms could also contain o. without regard to samdhi, this correspondence led, in our language, to the introduction by analogy of o as a metrical lengthening of final a (§§ 3.79-88). Examples: sarvata SP 92.10; bhūya SP 96.2; LV 49.16; Mv ii.319.20; purata LV 57.7; kuta yam = kuto 'yam LV 177.3; yoniśa LV 188.19; tata Mv ii.345.2 (= lato; wrongly interpreted in Senart's note).

4.33. Similarly for a or \bar{a} which in Skt. was final except for a following nasal, BHS often shows unnasalized a (or \bar{a}). Examples will be found in the sections of the morphology where such endings are treated.

4.34. The a which, according to the preceding sections, may represent Skt. as, \hat{as} , am, \hat{am} , an, \hat{an} , or also e (which even in Skt. becomes a before any other vowel than a), may be treated in samdhi exactly like any (Skt.) final a. Thus we have already seen that it may be elided before any initial vowel (§ 4.29). It may also be fused with such an initial vowel, producing the same result which Skt. final a would produce. So we find, first, e and ai resulting from theoretical as or ah, \bar{as} or \bar{ah} , plus i and e respectively (chiefly in verses m.c.):

bhiksaveti (= °oah-ili) SP 10.2; makareva (-iva) LV 175.11; jvalaneva (-iva) LV 175.12; megheva (-iva) LV 339.3; jälamäireha (-iha) Divy 389.20; pavaneva (-iva) Samädh 19.29.

samāhitaikāsani (°hitaḥ-ekāsane) SP 193.10; marutaivam (°taḥ-evam) LV 115.3; pravistaikas (°taḥ-ekas) Divy 391.25.

(Sanskrit -āh:) tasyaiva (= tasyāḥ-eva) rātryā(ḥ) Mv 1.231.3 (prose); yasyeha (= yasyāḥ-iha) Mv 1.199.9 = 11.3.7. (vs).

11.3.7. (vs). **4.35.** Similarly o results from as(ah) plus u: tatottari LV 154.7 (prose; the only v.l. cited is tatontari, which evidently intends tatottari; Calc. tata uttaram, obviously an em.) for tatah-uttari; paratopasargah Divy 566.15, 17 (vs, printed as prose), for paratah-upa^o; (in vss) itocchriah (itah-ucchritah) SP 235.14; pranāyakopāya (pranāyakahupāyam) SP 196.5; narādhipovāca (^opah-uv^o) Mv i.151.12; socyate (sa-uc^o) Samādh 8.6; kutottaro (kutah-ut^o) RP 6.13; usaisodgato (= usaisah-ud^o) RP 6.18. Cf. also ā resulting from as (ah) plus ā in anāthapindadāha (prose, § 4.29).

4.36. Further, Skt. final am, am plus a yields a, in

verses m.c.: leşâgralah SP 98.4, 6; 99.6 = leşâm agralah; evâprameyā SP 158.5 = evam apr°; evâpramānā SP 330.11; sarvakāyādrjāsi LV 195.15 = °yam ad°; viryāpi LV 340.22 = viryam api (v.l. viryam ca). And original am, ām combines with i, e, u to e, ai, o; once in prose, parameti LV 102.9 for paramam ill, all mss.: 'certainly, he said'. In vss: pramāņaisām SP 305.1 = °nam esām (WT °n' esām with v.l.); leşeli LV 195.7 = leşām ili (most mss. draksyanli, but Tib. de dag supports the pronoun); vāyuvegavidhuteva Mv i.68.20 (= °lam iva), so mss. (except °vidhū°, unmetr.); desopa° Suy 155.7 = desām upa°.

4.37. Similarly final e, thru intermediate a, fuses with initial \hat{a} to o in $r\hat{a}jatorna$ RP 7.1 (vs, for $r\hat{a}jate$ plus $\bar{u}rn\bar{a}$); and perhaps in $sac\bar{a}sya$ Mv i.323.21 (prose, = sace or sacet plus asya; but note that saca occurs, tho rarely, for sacet, before consonants and not m.c.).

o for as before voiceless, before pause, and for ar

4.38. Very common is generalization of o for as, the presonantic form, as in MIndic, before voiceless consonants and pause. It occurs even in the prose of LV: ojopali LV 278.10 (Lefm. em. ojāpali); kumāro stri^o LV 187.5 (all mss., Lefm. em. kumāra; voc. is impossible). And in the prose of Mv: mano pra- i.3.14; bhūyo pari- 15; adhośirā 5.4; manokarmā; 134.6; etc., very common. In most texts only in vss, as bhūyo pari- SP 84.4; -klamalho sprfen SP 258.8; also at the end of stanzas, sañgo Samādh 19.31. Very common.

4.39. As in Pali, o may represent Skt. final ar: antonagarāto Mv iii.185.4 (prose); same word in line 12 antarna° Senart, but mss. antena°, antana°; antogatehi (Pali antogata, Skt. antargata) Mv i.237.10 (prose); puno bhomy SP 324.6; puno 'tha LV 125.12; puno adya LV 215.18 (vs; so Lefm.; but reading uncertain; perhaps read with v.l. punār adya, for punar adya m.c.). Cf. next.

Other cases of final r in samdhi

4.40. On the other hand, final ar may appear as a, with loss of r, on the analogy of final as, before a non-a vowel: *bharta* (= *bhartar*) *iti* LV 227.6 (prose; so all mss. and Lefm.).

4.41. Final -is of an ending, before initial r, is written as unlengthened i in all mss. in -patākābhi (for °bhis) raina-LV 367.9 (prose).

4.42. Ålso, r may be retained as such before a consonant, even a volceless, if the mss. can be trusted; punar punah (but v.l. punah punah) Mv iii.294.12; prådurkare Mv ii.280.15; so also one ms. in 16 and 18, where Senart reads prådurakare as v.l. intends; all are in prose; adhar-dis, for adho-dis, see Dict. Once the text of Mv (ii.89.6) even writes bhikşavar (= °vas, °vo) yuşmākam without note of variant; and so äyurkşayā (v.l. dyuh°) Mv iii.65.17, where r represents s.

Miscellaneous vocalic samdhl

4.43. Reduction of final e to a before initial a seems to occur in *sprsisya* (= °ye) an° Mv ii.404.2 (vs, probably m.c.).

4.44. Final d shortened before initial vowel. This is a doubtful change, which has no parallel in MIndic so far as I know, and for which I have noted only a single case: prakampila upary- SP 414.2, in prose. The preceding noun is $iyam \dots lokadhdluh$, so that the pple. should be prakampild, which one ms. reads; other vv.ll. are "lah, and prakampild (which seems clearly lect. fac.). WT keep the form without note.

4.45. If the mss. may be trusted, it seems that \bar{a} and even short a plus r sometimes result in $\bar{a}r$. In LV 234.18 (vs) Lefm. reads -mandilā (voc. sg., so frequently

4.59]

in this context) rsi-, which is unmetrical; no ms. has the 'regular' -mandilarsi-, but some have -mandilārsi-, which is metrically possible. In LV 290.14 (prose) both edd. and apparently all mss. yenārddhy-, for yena rddhy-: Weller does not notice this. In the text of Mv sarvārluka occurs (once) for Skt. sarvarluka, derived from sarva-rlu-; the Pali sabboluka (sabba-ulu) also occurs as sarvoluka or 'duka (Dict.).

4.46. Once, at least, the mss. record $\bar{a}y$ for final ai before a vowel: tasmāy aham LV 403.6 (prose; so all mss.).

4.47. Final vowels which in Skt. would be uncombinable (pragrhya) may undergo combination with a following vowel (rarely): $j\bar{a}y\bar{a}padii$ Divy 259.3 (prose; $j\bar{a}y\bar{a}padi$, dual, plus *iti*).

4.48. Samdhi of the pronouns sa, esa. These often appear, as in MIndic, in the generalized samdhi forms so, eso, in verses of most texts, and in the prose of Mv. So before voiced consonants, e. g. RP 12.12, and before voiceless, e. g. Bhad 55; both are very common.

4.49. On the other hand, there are also recorded occurrences of sa, eşa in which the final a is fused in samdhi with a following vowel (as in the Rigveda): esaiva = eşa eva SP 201.7 (prose; so KN with Nep. mss.; WT with Kashgar rec. esa eva); sātivibuddha (= so 'ti') LV 415.4 (vs); sopagupla; (= sa upa') Divy 386.14 (prose); sārāmikas (= sa ār') Av i.37.12 (prose). We saw above (§§ 4.29, 32) that any final as may be treated as a.

4.50. Note also sah Säkyamunis Suv 152.11 (vs); meter requires a long syllable for Skt. sa, but the normal (MIndic) so would do as well.

Hiatus

4.51. Like Pali (Geiger 67), BHS very often keeps unchanged, with hiatus, two adjoining vowels in the seam of compounds; and a fortiori between separate words. This phenomenon is extremely common in verses everywhere, and in the prose of many texts, particularly Mv, but also others, such as LV (especially, it seems, when the second vowel is r). Texts which show it little or not at all in prose may be suspected of having been adapted to standard Skt. in the course of tradition. At any rate, the constant occurrence in verses is regularly confirmed by meter. Such orthography in violation of meter is very rare; but cf. cyuli-upapalli- Mv i.9.3, where meter seems to demand cyulyupapalli- (or cyul'-upa^o).

4.52. First, in compounds. It is to be noted that editors often divide wrongly, not recognizing cpds. intended in the text, or writing as cpds. what are in fact separate words. Such cases are corrected silently in the following.

4.53. In prose: -pūrvavideha-aparagodāniya-uttarakuruşu Mv 1.6.2; chinna-iryāpalhā (Senart em. iryā⁹) 22.11; mahā-udyāneşu 32.4; -bhojaniya-āsvādaniyena 38.7; -vanaudyānam 216.3; deva-rdahiye... rāja-rdahiye 223.9; in a series of parallel formulas, nila-usnīsā, pilosnīsā, mamjisiha-usnīsā, lohita-usnīsā, iselosnīsā, haritosnīsā, vyāyukta-usnīsā, 259.14 to 261.6; mahā-udyānam 299.19; rājarsih 353.9; eka-ekā (āmredita) 359.16; šyāma-rşisya li.213.1, 3; -simha-ukkāsitam 281.12 fl., etc., common in Mv; prahāna-rddhi LV 8.5, 15; 181.18; 426.7-8; -vyāghra-rķsa-306.7; rāja-rddhyā Av 1.338.3; svabhāva-rjukaḥ Šikş 324.6; abhikṣnāpatti-āpadyanatā KP 119.2; mahā-rṣayai (instr. pl.) aneka-mahā-rşistala- Mmk 18.20.

4.54. In verses, always in accord with meter: nirodha-ulpāda- SP 117.3; cāritra-ācāra- 129.1; vihāra-udyān' upaiobhitam 196.8; mahā-abhijāāsu 204.11; 205.9; anantaaupamya- 222.1; bodhi-arthikah 295.5; jāsvala-uccheda-Lank 25.5 (cf. jāsvatoccheda- in another vs, 22.10); udyāna ārāma- LV 28.4; -praišānta-upaišānta- 46.4; paramarūpaupetā (read as one cpd., and m.c. -apitā for -upetā) 54.7; sa-indrakāh 54.13; ksamā rajo-anākulā 75.19; su-ākarā 114.15; āraņi-ucchritena 116.14; purima-rşibhi 163.20; vişakrta-işuņā 166.5; -trşņa-avidya- 173.15; ratisaukhyaanvitah 215.4; gajabhujasamnibha-ūrunikām 322.21; bodhianģāh 340.20; su-utlhitā 389.8; vanadeva-anila- Mv i.171.2; nandi-āvarta ii.304.17 (Senart em. nandiyāv°); varaagrabodhim 341.7; sa-indrakā 405.13; citta-ulpādo Samādh 22.43; aneka-āyāsa-sahasra-ākulā Suv 39.1 (so citation of passage Šikş 218.2; mss. of Suv °vyākulā or the like, avoiding the last hiatus); varadharma-ulkā Suv 62.5; catudoīpa-išvarah 146.11; vīrya-upāyaih RP 4.4; bodhianga- 11.10; -cari-adhimukti- Bhad 7; faikşa-ašaikşa- 9; padma-āsane Gv 25.18; buddha-āsane 34.6, etc.

4.55. In separate words hiatus is equally common in verses everywhere, and in prose of some texts, especially Mv, but also LV. In the prose of most texts it is rare, probably owing to Sanskritization in the course of tradition. A few prose examples (in Mv and LV they could be multiplied indefinitely) where no pause in speech between the words is conceivable: -simhena agra- Mv i.3.9; tasya abhūsi 13; etena upāyena 15; pārsni (acc. sg.) upādāya 6.4; te evam 8.1; tadā āsi 48.16; eva rddhau LV 14.20-1; eva rddhyā 15.13; -pari antah- 15.17-18; cakravarīt ajsaratnena 16.7; dhvajāgre ucchrāpayibā 16.22.

4.56. In verses, always in accord with meter: dharmeşu anāsraveşu SP 61.12; -dhāluki agra-, and dejayişyāmi anāgate 61.13; dharmeşu aho 62.2; -jnānātu aci^o 62.6; drştvā ca anyān 62.10; samdhāya etal 62.11; vyākurvosi agra- 63.4; ca āsin 63.5; na eşa 64.2; vividhās ca sattveha (= sattvā[h] iha) ayam ca agnir (so with v.l., KN agni) 86.10; 'smi (so with Nep. mss. and WT for KN 'smi) iha 87.3; putehi (WT em. vultehi) aputrakasya, ihāgninā ime 87.4; smara ananta- LV 11.16; mā upekşasva 11.17; paramā (voc. sg.) upekşakā 53.16; ye idam Suv 5.3; jinasya antikam Samādh 8.26; ca artham 8.31; śrutva adharmām (with mss., acc. pl.) 19.33; ca akşaya- Bhad 4, 31; ca aham

'Inorganic' samdhi-consonants or 'Hiatus-bridgers'

4.57. On this phenomenon in Pali cf. Geiger 73. In BHS the situation is approximately the same. Both m and r are very commonly so used; d is rare but seems certainly to occur in a few cases; n is still rarer and less certain. The use of y in this way is rare and special.

4.58. The samdhi-consonant is very often inserted before an initial vowel when the preceding vowel has become final by Mindic alterations. It also occurs between parts of a compound word, as well as between separate words. The commonest in verses, it is found also in prose, especially that of Mv.

4.59. m as samdhi-consonant. This is much commoner than any of the others, and occurs more or less everywhere, tho more commonly in verses than in prose. Prose passages in the following are so marked. dadanti dānān' ima-m-evarūpā SP 14.1 (ima = imāni); dvihaslamäträ-m-anucankramanti 84.8 (Kashgar rec.); santäpitä agnina-m-alpapunyäh 85.11 (Kashgar rec.); svaka-m-atra putran 86.5 (acc. pl.; so with mss.); antarakalpa-m-eva ca 91.4 (so ed. with Kashgar rec.; acc. pl.); ihaiva-m-agatah 115.8 (so ed. with Nep.). - tayaiva-m-rddhyā Lank 12.5-6 (prose), = layaivardahyā; cl. layaiva šobhayā 12.10. — pañca-m-anūnakāni (text °ri) LV 28.19; sarva-m-anubhoktikriyā (acc. pl. fem.) 45.12; šakra-m-api ca (nom. sg.) 92.9, 13; kila-m-itah 110.11; lakşanacitrilänga-m-anagho (nom. sg.) 130.16; yatha-m-agnikhadā 174.1 (so prob. read with best ms. A); iha-m-ākulam 178.21; sthila-m-abhûd gopā (n. sg. fem.) 194.7; prihivī prakampila-m-abhûe 194.9; sarve ābharanā vikirna patilā-m-uhyanti te vārinā 194.17 (confirmed by Tib. khyer bar gyur ba 'become carried away'; note bhūşanā uhyamānā 195.19); vijahya-m-āntahpuri

3

211.7 'abandoning the women of the harem'; iha-m-āgamisye 231.19; yalha-m-iva 242.5 (so ed. with best ms., others yathā iva, unmetr.); dharmarāja-m-anaghe (loc. sg., -a for -e) 281.15; vidu-m-āhuh 324.1 'the sages say' (so Tib.): ca-m-anubhaviyā 329.3 (= ca-anubhūya); varteşyi-m-asya 337.5 (so with best mss.; = vartayisye-asya); pāņina-masim pragrhitva 338.12 'with the hand', instr.; yalhā-manugutā 353.7; saraņu tava-m-upema 365.18.—In vss of My, girāya-m-anupūrvašaķ i.10.7 (instr.); esa-m-abhiyo 44.1 'this Abhiya' (so most niss., v.l. eso a° , unmetr.; Senart em. esa a°): raja-m-aham-iha-m-ud- 108.8 (so mss.; Senart em. rājā aham idam ud-; meter requires aham, otherwise read with mss.); iva-m-aham 116.13; vijānati-m-ālapantam 155.4 (so all mss., Senart em. °ti ā°); bodhyitvā-m-atulām 170.9; sthihiya-m-antarato 203.12; aparimeya-m-ito kalpe 251.16 (Senart em. 'ye-ito); tava-m-etat 279.20; jalpa-m-acintya-284.14 (jalpa = jalpan); sthila-m-apsarānām ii.316.13 (Senart em. sthitā, unmetr.); nayutā-m-anantā 341.5 (so mss., acc. pl.); buddhitva-m-uttamām bodhim 354.4 = iii.278.8 (so Senart second time with both mss.; and so one ms. first time); lalra-m-akāsi iii.139.11; viditva-m-ud° 139.12; 140.6; krtvāna-m-anjalim 345.18. In prose of Mv, hrada-m-iva accho i.237.12 (n. sg. m.); alīla-m-adhvāne 271.19; 283.14; ii.209.9 (loc. sg.); gaccha-m-anujānāpehi i.256.14 (gaccha = gacchan); brāhmaņaparisāya-m-eva sakāšāto 311.8 (gen. sg.); so eva-m-eşo kumāro iiN10.17 'this prince is none but he' (eva, not evam); gāthāya-madhyabhāşali 207.4 (instr.; v.l. gālhāye adhya°); jāne-maham 214.9 (so with mss.); bhūmī-m-adhigantavyā 280.6 (so with 1 ms.); adya-m-aham 404.18. - upapariksya-mupaparikşya Divy 230.9 (prose; falsely emended p. 707); śruta ca vaca-m-api 390.18 (m.c. for vaca); kimcana-m-iti 534.23 (end of line of vs; ed. prints double danda after nāvagāhanto-m-anadhimucyamānāḥ KP kimcanam!). 139.3; te-m-anadhimucyamānā(h) 139.5 (both prose; Tib. ma mos, confirming the neg.). - prahinakrodhasya-manāsravasya Ud xix. 3 (Pali equivalent Dhp. 94 pahinamänassa anāsavassa). — atīta-m-adhvanī Samādh 8.21 (for atīte 'dhvani); pramāņam kalpa-m-ākhyāto 22.52 (for kalpa-h). — vācā-m-anu- Suv 151.15; 152.20. sahilpam-asesam RP 21.5; tyakta-m-anguli udārā 24.1; tena-maprayukto 25.16 (printed tena ma pra°); taih kāñcano bhāram-ivapaviddhah 35.16 'as it were a load (bharah) of gold cast off'. - -kalpa-m-akhinnah Bhad 25, 44 (-kalpa for -kalpān). — dharmagaņdī-m-ākoļyamānā Kv 36.17 (prose; n. sg.). - punyagandha-m-anantaka Mmk 4.26 (prose; voc. sg.); -vigrahā-m-abhilekhyāh 62.6 (prose, n. pl.); devaputrau-m-abhilekhyau 64.6 (prose; dual); buddha-m-ivotpannāh 74.5 (prose) 'originated like the Buddha' (n. sg.); nātyāśi-m-alpabhoji vā 103.19 (n. sg.); cāpi-m-āruhet 119.2; nu-m-ālikhet 132.4; pürvāyām diśi-m-āśritya 205.26.

4.60. Between parts of compound words, -m- is used in the āmreditas anyamanya- = Skt. anyonya, ekameka- = ekaika (see Dict.), and anga-m-ange LV 223.15 (n. pl.). Other examples: rddhi-m-asangata-maitri-balam LV 287.9 (so Tib.; wrongly divided in Lefm.); yuga-m-anlarasmi 338.11 (= yugāntare); devā-nāga-m-abhistuto 402.12 'praised by devas and nagas'; eka-m-ante = Pali id., Skt. ekānte (Dict.); ašoka-m-āhvaya, see Dict. s.v. ašokāhvaya; asubhakarma-m-alipravannaih Mv 1.84.3 (so read, see Dict. s.v. pravanna); raśmi-praloda-m-usnisā 259.16; 260.1; dibasa-m-dhāreņa 302.6, 8 (prose) 'daily food' (or is dibasam an adverb? cf. daivasikam, line 14); šākyarāja-m-udyāne ii.204.10 'in the Śākya-king's park'; sarva-m-anavadyagātro 220.10 (cf. Skt. sarvānavadyānga); tandula-m-āhāratāye 231.10 (prose; in parallel phrases just before, kolāhāratāye, tilāhāratāye; Senart wrongly prints °āharatāye each time, with v.l. in some instances); muni-m-isvaratve 341.9 'in the sage's power' (so read with 1 ms., the other corrupt; Senart em. violently); mūla-phala-m-akrsiaumcha-vrttiko ili.382.17 (prose); rājatva-m-aiśvarya-priyatā Suv 100.16; varşafala-m-anekaih RP 32.9; salyārjavakşayam-upāya-nayaih (read °rjavākşaya°, i. e. °rjava-akşaya°) 52.11; vīrya-dhyāna-m-upāya-fodhilam 54.19; puņya-jñānam-upāya-prajňalo 55.13; svabhra-kedāra-m-auşarye Mmk 113.11; pilla-m-ulthānād 146.7; loka-m-agrais 148.17; adham-ūrdhwa- Gv 518.4 (prose; for adha-h); Lank 28.1 (text °ūrdhāf).

4.61. r as samdhi consonant. The less common than m, r is also familiarly used in this way, chiefly in verses; prose examples are so marked. As in Pali, ati-r-iva is common for Skt. aliva, but only in My (Dict.). Also, but only in verses, yatha-r-iva, which is cited once in PTSD, rendered 'just as', but analyzed as yatha plus iva; the meaning in both Pali and BHS suggests eva rather than iva (to be sure BR cite a few cases of yathā and iva both in the same clause, but never in this order and always attached to different words). Occurrences of yatha-r-iva 'as, just as' are (all verses) SP 14.9 °va Gangavālikāh, 'just like the sands of the Ganges' (in number); 32.11 id.: LV 55.10: 167.4 °va purimā, 'just as the former ones'; Sukh 8.11; Siks 303.9 and 304.10 (both passages in My, ii.382.5 and 383.6, but My has different meter and reads only yatha both times). This meaning is clearly demonstrated by tatha-r-iva, paralleled by tathaiva (with eva, not iva), 'just so, also', iu LV 240.17 tatha-r-iva dvāri tathaiva vātavāne. All of these passages are in verses, so that one might think -iva used for -eva m.c.; but the Pali yatha-r-iva occurs in prose DN i.90.17. I believe that only the meaning of eva, never that of iva, fits in every case.

4.62. Other examples: labdhvā (cited labdhā) ca mūrdhnā-r-iha dhārayeta SP 99.2 (so prob. read with v.l. for text labhitva ca mūrdhani dhār°, unmetrical); drsto na jälu-r-asmäbhir 306.13; srnu-r-asya LV 36.17 (? but should be pl.; two mss. -śrnuntasya, Calc. śrnutāsya; read the latter or °thāsya?); bodhihetu-r-aprameya 170.14 (°hetu, adverb); (jagi-r-ūrmiculi LV 173.13, so ed., but read jagi janmacyulih as in citation of this line Siks 204.1;) giri-r-iva acalam LV 329.20 (acc. sg.: 'him, immovable as a mountain'; but girir may be nom. used as acc., § 10.60); c' osarir-ātmabhāvā LV 357.3 (read so, or with v.l. c' ośarī ātma°, for Lefm. em. svojarir; 3 sg. aor. of ava sirati); samdhisu-rayam Mv i.97.17 (so read with most mss.; Senart omits r by em.); asu-r-eva 217.18, 218.6; repeated ii.19.19, 20.5, where Senart also asu-r-eva by em., but mss. asu-n- and āśu eva(m; with hiatus) respectively; paranirmitavašavartina-r-iti i.229.16 (prose; so one ms.; Senart with v.l. ona iti); bhagavata-r-ardha- iii.301.2 (prose; v.l. °to ardha-; gen. sg.); sattvā-r-āryadharmacakram 320.6 (prose); sādhu-rarth- 345.20 (so mss.; Senart em. sādhu arth-); bhoti-rupetah Bhad 52; pañca-r-aştābhih Mmk 59.26; vyomni-r-iva 111.3 (prose); tri-(text tr-)sapta-r-asta ekam vā 120.10; niśi-r-eva 213.22, 25; devais ca-r-asurais cāpi 598.26; so 'pi-r-atyanladharmavān (text °vām) 631.2.

4.63. Between parts of a compound: hri-r-apatrāpya, q.v. in Dict., also s.v. apatrāpya; aja-r-edake MSV iv.228.16. Much more questionable is anuragatam 'welcome', used repeatedly (Dict.) in Mv along with svagatam, in both prose and verses. In some passages the mss., or some of them, read anuragam, and in some Senart prints the latter in his text, tho his note on i.35.6 (p. 400) appears to recognize anuragatam as the true form. This is confirmed by the AMg. equivalent anurāgayam, which is explained by an ancient comm. (cited by Pischel), as well as by Pischel 353, Sheth, and Ratnachandra, as = Skt. anvāgatam (repha agamikah, Ratnach.). Neither this Skt. word nor any normal Mindic descendant of it ever has any such meaning; if our word really goes back to it, the r is likely to be due to semantic confusion with anuraga 'affection'. It is still more likely, in my opinion, that anurāgalam and AMg. aņurāgayam are an alteration by popular etymology (influenced by association with anuraga) of the word found in its original form in Pali *a-dur-āgatam* 'not un-welcome', which is used exactly like *anurāgatam* (CPD). Kern, IF 31.196, made the connection between the BHS and Pali words, but failed to note the AMg. form, which proves that the BHS is more than a mere corruption.

4.64. d as samdhi consonant. yathā-d-iva Mv i.15.15 (vs); so all mss., Senart em. yatha-r-iva (see § 4.61). In My iii.54.1 ff. (prose) occurs a whole series of such forms: khu-d-ayam, mrdukā-d-ayam, masinā-d-ayam, etc. The forms in final ā are n. sg. fem. — na-d-upeli khedam KP 37.6 (vs) 'does not become weary'.-In LV 133.5 (vs) text sākiyasulā (n. pl. m.) abhinişkramanti; but best ms. A "sutā-d-abhi", which as lect. diff. is likely to be correct. In LV 340.13 (vs) bāhum prasārya yatha vidyud ivā nabhasthā-d-ābhāsale is probably to be read, 'stretching out his arm like lightning as it were in the sky, he spoke' etc. All mss. have -d-, tho Lefm. omits it. - krivā-d-angulibhih Mmk 360.9 (vs); krtvā-d-ubhau Mmk 375.8 (vs).-In Ud xi.10 probably interpret dvisa (= dvisan)-d-icchali, rather than assume neuter dvisad used as masc.; the Pali parallel (Dhp. 162) reads icchati diso (= Skt. dvişah).-In Karmav 158.3 (prose) occurs adyādagre 'from now on', = adyāgre in the next line; the usual form is adyāgreņa (see Dict. s.v. agrena); in Pali we find ajjatagge and ajjadagge, derived by CPD from *ajjato agge, which leaves the form with -dunexplained, unless it be taken as a case of Prakritic voiced for voiceless consonant. Pali ajja-d-agge could contain 'Hiatus-bridging d' (so the Pali commentators take it, CPD) and our adyādagre might go back to an equivalent *adya-d-agre, with \hat{a} by influence of the more usual adyāgre(na).

4.65. n as samdhi-consonant. Rare, and perhaps questionable. Since n appears to vary freely with m, at

least in orthography, perhaps the few seeming cases of hiatus-bridging n are connected with this confusion, which I consider orthographic. Note that in one case, kim imd-nedria parivrājakā (mss. pravrajyā, perhaps to be kept) Mv iii.102.11 (prose), one ms. reads imā-m-. Otherwise I have noted agrhna devā-n-adhivāsanam ca LV 7.13 (vs), 'I accepted (the request of the gods), and the gods (accepted) my consent'; see § 8.85 for an alternative interpretation: vasumati-n-iya mahya sāksi LV 340.14 (vs) 'this earth is my witness'; here all mss. as in Lefm., except two which omit the syllable ni (Calc. °mali-m-iya°). The variant āśu-n-eva occurs (instead of āśu-r-eva) according to the mss. in Mv ii.19.19, see § 4.62. Senart on Mv iii.82.14; 402.8 assumes upa-n-eli = upaili (upa plus eli), but I am not convinced. In 402.8 the true reading is probably param upanamati.

4.66. y as samdhi-consonant. Like Pali, BHS shows (in Mv only) a number of cases of yera (see Dict.) for eva, usually after vowels, but in a few cases after final anusvāra. Probably this was originally a matter of samdhi, Geiger 66.1, but with extension of the use of the form. Some of the cases mentioned by Geiger 1. c. are certainly of different origin; e. g. Pali yillha (= BHS yasla), where y is due to morphological analogy (strong-grade form of ya).

4.67. h as samdhi-consonant? Cf. Geiger 67, 73.7; Pischel 338. So far as I know there is no case of this in our language. In SP 111.4 (vs) KN read digviditasu hance. But WT read añce, and report aññe from their ms. K'; the Kashgar rec. has anvate, which may well be the true reading. Even if hance be accepted, the h- might, as WT point out, represent the particle ha (or, I would add, iha). There is no valid reason for assuming -h- as samdhiconsonant. 5.1. The use of (historically) singular for plural forms, and vice versa, occurs on a large scale, in nouns and verbs alike. For the most part it seens to me a matter of morphological confusion. Therefore it has been treated under the various morphological categories where it occurs. Note the charge of confusion of singular and plural brought by learned monks against their brethren, reputedly in Buddha's lifetime, § 1.10.

5.2. Occasionally I have noted what seem to me plurals of respect. So bhagavān (sc. Dīpamkara) ... prakramensuh (v.l. °melsu; Senart em. prakramel) Mv 1.231.3 (prose); eşatā, twice, RP 39.7, 2 pl. impv., addressed by gods to Bodhisattva Punyaraśmi; yūyam karetha Mv iii.178.9 (prose), addressed by Anuruddha to his elder brother Mahānāma.

5.3. In Mv i.166.13 (vs) occurs a singular of a partitive genitive, instead of the normal plural (which occurs in a parallel phrase in the very next line, filoccayānām yalha meruparvataḥ): tvam eva śresiho sakalasya (mss. samathasya) prānino, 'you alone are the best of every living being (= of all living being)'.

5.4. A collective noun in the singular, or a sing. noun accompanied by an adjective denoting plurality, may be construed with a plural verb: nāriganas... pralibodhayişu LV 173.9 (vs); mārah... saparivāro... ulsrjanli sma LV 317.4-7 (prose).

5.5. Similarly there may be agreement between nouns and adjectives or participles of different numbers: bahūni samdhāvacanehi coklam SP 59.4, 'and many things by means of comprehensive statements it has been said' (quasi-collective); trişu apāyi (for apāye; also no doubt felt as collective); LV 92.15, 'in the three (triple) evil form(s) of existence'; so also deve manuşyeşu ca trişv apāye Suv 59.7 (but paācasu kaşāyakāle LV 257.21 belongs in § 23.11, which see for LV 116.13-14 too); nagaram vyākulu bhitairastamanaso LV 193.10, 'the city was confused, (the people) with minds frightened and alarmed'; sa śalhah sasainyah ultrasta bhinnahrdayo prapalāna sarve LV 319.13-14, 'the evil one with his host, frightened, heartsmitten, fleeing one and all' (cf. LV 317.4-7 in preceding paragraph).

Dual number

5.6. Duals are very extensively replaced by plurals. It is evident that the dual was not really alive in the underlying dialect, as it is not in MIndic generally. Even when a dual noun form is retained (or restored by secondary Sanskritization), plural adjectives or participles and verbs (less often singular verbs) may be used with it. See §§ 25. 1-3.

6. Gender

6.1. Gender in BHS is in about the same condition as in MIndic generally; and perhaps specially close to Apabhramśa, in which Hemacandra 4.445 flatly declares that 'gender has no rule' (*lingam alantram*). Cf. Jacobi Bhav. p. 31* (§ 22); in this text the distinction between masc. and nt. a-stems ('die allein in Betracht kommen') is 'soweit verwischt' that Jacobi makes no attempt to distinguish them in his Glossary, marking only the feminines with 'f'. He notes further that nouns with nt. forms are used with attributes that are definitely fem. in form; 'but the commonest case is that fem. stems assume masc. endings (seldom the reverse)'. Even in Pali examples of all these things are not rare, as Geiger 76 clearly states. Note that confusion as to gender was charged against monks, reputedly in Buddha's lifetime, by their more learned brethren, § 1.10.

6.2. To be sure, the question may be raised (and has been raised by Pischel 359) whether Hemacendra's dictum is not something of an overstatement, even for Ap. Actually Jacobi, in a later statement (San. \S 13), seems a bit more conservative than in his edition of Bhav. I have felt it desirable to retain the traditional classification of nouns by the three Skt. genders, which is after all justified by the great majority of BHS forms. At the same time I wish to emphasize that the confusion of gender in BHS is very far reaching and widespread: more so than has been commonly recognized, I think. I suspect that in MIndic generally it has also been rather understated than overstated.

6.3. This confusion manifests itself in two ways (as in MIndic, cf. the references cited above). First, endings appropriate to one gender are transferred to a noun belonging to another. Second, pronominal or adjectival attributes with endings appropriate to one gender are used in agreement with nouns belonging to another. Often, of course, both types occur together.

6.4. I must refer here to an opinion held by the late lamented Heinrich Lüders (Philologica Indica 279 fl.; originally from SBBA 1913.988-1028), and accepted by Dschi, NAWGött. 1949, 272 fl., that in 'old Ardhamågadhi' (also, Lüders thought, in Pali and later Pkts.) the acc. pl. m. of vocalic stems (and perhaps fem., Lüders 285 f.; more definitely Dschi l. c.) ended in '-ni with lengthening of the final stem-vowel', i. e. in -āni, -īni, -ūni, while the corresponding nom. pl. lacked this -ni. Both these scholars admit, however, that the n. pl. of m. nouns also occurs ('by extension', they think) in -ani, in Asokan and later. To me the whole theory seems unacceptable even for Asokan, and certainly for BHS. The endings in question are simply neuter plurals, due to the confusion of endings here treated. No significant difference in this respect is perceptible between nom. and acc.; if the acc. forms in -ni are commoner than the nom., this cannot be fundamentally important. At most, perhaps neuter forms may have been more favored in object than in subject position. No other origin than the Skt. nt. ending is conceivable for this -ni after a, i, u, in the plural. The confusion of gender-forms is, moreover, not limited to the plural; it is found also in the sg. It is also not limited to stems ending in vowels. It applies to all nouns and their adjectival and pronominal epithets, sg. and pl. Indeed, it is not even limited to the nom. and acc.; it occurs in other cases also; but in them masc. and nt. are indistinguishable in Skt., so that confusion can be noted only between them and the fem. Such forms as yavanti, catvari, idam, and adjectives in -am, all agreeing with n. (pl. or) sg. m. nouns, cannot be accounted for by the Lüders-Dschi theory. Moreover, these scholars ignore the use of masc. forms instead of fem. and nt., and fem. instead of masc. and nt., all equally familiar. The difficulty for Lüders was caused by his assumption that 'change of gender' in the noun would have to be assumed, in default of his theory. It is not a case of 'change of gender', but of incipient breakdown of formal distinctions of gender.

Endings of one gender transferred to nouns of another

6.5. The first of the two types of confusion mentioned above will be perhaps sufficiently, tho certainly not exhaustively, illustrated under Noun Inflection. Some cases, where it seems that 'change of gender' in the commonly accepted sense (as frequently in Sanskrit) may reasonably be assumed, are noted in the Dict. (It must be admitted that here the line cannot be strictly drawn between transfer of nouns to a different gender and transfer of endings from one gender to another.) Certain or likely cases of transfer to one gender of endings originally appropriate to another will be found as follows.

6.6. In nouns of the *a*-declension (m. and nt.): $-\bar{a}$ as n. sg. m. and (also $-\bar{a}m$) as acc. sg. m., at least possibly borrowed from fem. \bar{a} -stems; -am as n. sg. m. (from nt., or from acc. m.? probably both); instr. sg. $-ay\bar{a}$, $-\bar{a}ya$ (f.); dat. $-\bar{a}ye$ (f.); n. and acc. pl. m. $-\bar{a}ni$ (nt.); n. and acc. pl. nt. $-\bar{a}$ (also -a), best regarded as from m.; n. and acc. pl. nt. $-\bar{a}m$ ($=-\bar{a}n$), from m. acc. pl. See §§ 8.24, 26, 38, 42, 45, 86, 98, 100, 101, 102.

6.7. In (fem.) \bar{a} -stems: n. sg. -u (for -o = -as or -am, m. or nt.); inst. -ena; abl. $-\bar{a}l$; gen. -asya (in pronouns); loc. -e; n. pl. -e (in pronouns); acc. pl. -e (MIndic m.) and $-\bar{a}n$, $-\bar{a}m$ (m.); n. and acc. pl. $-\bar{a}ni$ (nt.); voc. pl. $-\bar{a}ho$ (MIndic m.); inst. -ais, -ebhih, -ehi. See §§ 9.13, 69, 71, 75-77, 94-96, 98-100, 103-105.

6.8. In *i*-stems: n. sg m. and f. -im, -im, § 10.23 (may be partly borrowed from MIndic nt., as well as from Skt. acc. m. and f.); oblique (f.) -iya, -iya, -iye, -iye, -iyo, used with m. nouns, § 10.89; n.-acc. pl. -inas, as f. (§ 10.155); -ini (nt.) as m. and f. § 10.160; -yas (f.) as m. § 10.162; -in, -im (m. acc. pl.) as f. § 10.165; -is (f. acc. pl.) as nom. m. (as well as f., § 10.167); -iyas, -iyo, iyas, -iyo (f.), as n. and acc. pl. m. §§ 10.167); -iyas, -iyo, iyas, -iyo (f.),

as n. and acc. pl. m. §§ 10.168-172. 6.9. In *u*-stems: -*ūni* (nt.) as n.-acc. pl. m. and f., § 12.53.

6.10. I shall add here a few striking cases which do not make on me the sort of impression made by what are commonly called cases of 'change of gender' in Sanskrit. Some of them also illustrate the second type of confusion: pranidhi, masc. in Skt. (to be sure not used there in its Buddhist sense), has also, and commonly, fem. forms and modifiers (LV 167.15; 175.13; 283.2; 364.10; 415.21; Mv i.54.6, etc.), but sometimes nom. sg. in im, \S 10.23, which may be regarded as nt.; maranāya (v.l. °nāye) pāram Mv i.165.8 'to the farther shore of death', as if to *maranā, fem.; cf. LV 175.11 marano, masc. (but supported only by the best ms. A; Šikş 206.9 cites the line with maranām as in the other mss.); ratna has masc. forms, vividhāmi ca ratnān SP 364.5 (ed. em. °dhā ca ratnā); ratanās (all mss., ed. em. °nā) trayah LV 421.2; utpannā ratanās (all mss., ed. em. °nā) triņi loke paramadurlabhāh (all mss., ed. em. °bhā) LV 421.6; mudrā, recorded only as fem. in Skt., Pali, and Pkt., is often masc. or nt. in Mmk, as 422.8; 423.7; 424.9; 472.8 f. etc. (Dict.), tho still oftener fem.; sometimes forms of different genders occur in close proximity, as padmamudrena 424.18, padmamudrayā 20; antrāyām 'in the intestines' Divy 409.15 (Skt. antra and Pali anta recorded only as nt.).

Attributes formally of one gender agreeing with nouns of another

6.11. The second type, already illustrated in some of the preceding cases, may be exemplified further. It seems that numerals (notably the words for 'three' and 'four') are especially apt to show this confusion. In not a few of the following cases, it is impossible to draw any sharp line between forms listed here and those listed in the morphological categories referred to in \$\$ 6.6–9.

6.12. Masc. modifier with nt. noun: trayas ca yānāny SP 33.4 (Kashgar rec.; ed. with Nep. mss. trīņi); anuvyaňjana ye ca . . . asilipūrnah pravara višistah SP 62.3; lān api ... vividhāni šabdāni SP 358.14 (šabda, tho masc. regularly, and even in lines 8 and 12 above on this page, sometimes has nt. endings in BHS); sarīra vaistārika . susakkio (all noms.) SP 69.2; jyolişânî palantah (so all mss.) LV 195.13; read mangalā (nom. pl. nt.) ye LV 214.12 for ed. mangalāyai (senseless); tapo . . . pratapto LV 336.17; abhibhavate rula (acc. pl. nt.) sarvi bhāşamānām (= °nān) LV 366.19; nakşatrāņi . . . ye (nom. pl.) LV 390.14; satyāni caturo My 1.188.5; eko akși (so mss., Senart em. akși; prose) My iii.376.1; višesayuktāms ca ratnāni Divy 397.23 (prose); ubhau jänumandale Av i.2.11 etc., see Speyer's note (Pali ubho used for all genders); but contrariwise, antu (=antau, dual masc.) ubhe (nt.!) ... eti (= ete, dual nt.) Samādh 19.32; imi praņidhānā (nom. pl.; regularly nt.) ... sarvi ... samagrāh Bhad 58.

6.13. In LV 438.17 (prose), (maheśvara . . .) -pramukhá asiadasa śuddhāvāsakāyikā devapulra-sahasrāni, the masc. adjectives are construed 'ad sensum' with 'gods', *devapulrāh, not mechanically with 'sahasrāni. Weller 32 (on 285.20) wrongly takes the epithet ending 'kāyikā as compounded with the following devapulra'; he is refuted sufficiently by the parallel adjective ending 'pramukhā, which cannot be compounded with the following word. In 285.20, also, 'kāyikā should be separated from deva' and interpreted in the same way.

6.14. Neuter modifier with masculine noun: yāvanti ... saltvāh SP 9.5; calvāri ca lokapālāh SP 55.3; calvāri dharmā imi evarūpā SP 292.15; māsān pi calvāri SP 373.4; sarve mahānagnabalair upetā vistīrņahastī navaratnavanti LV 28.5; drstam (nom. sg.) kumāra ... nişkramanto LV 135.5 (all mss. drştam except one, unmetrically, drsta); catvāri lokapālāh LV 185.7; ogha catvāri LV 195.18; saroe chinnā legu trīņy apy apāyāh LV 196.8; no ca kimcid api dešu vidyale LV 237.4 'and there was no place found'; mahāsāla iva mūlachinnam (all mss.) LV 334.10; rāgamadanamakaram trspormijalam kudrstisamgrāham (all adjectives with next) samsārasāgara (nom. sg. masc.; separate from next word) maham (= mahān) samlirno (note masc. ending!) LV 374.17-18; idam dharmam (acc, sg.) LV 396.1 (prose); krcchrena me'yam (sc. dharmah) samprāptam alam ... LV 397.19 (but v.l. samprāptam alam, perhaps read so, understanding samprāpta plus 'hiatus-bridging' m? metrically indifferent); ayam ghoso mayā śrulam Mv i.28.15; 29.2, 5 (in 2 Senart reads śrulo, with 2 mss. out of 6); sarvam (2 mss. sarva-) samsāro Mv i.77.14 (prose); ayam lokah ... sarvaśūnyam (all mss.) bhavişyali My 1.126.7-8 (prose); so imām (= imān) calvāri mahādvipām (= °pān; there follow other acc. pl. masc. adjectives) Mv 1.193.18-19, and same phrase ii.158.18 (prose); calvāri lokapatino Mv i.204.12 = ii.8.7; putrā ... trīņi paņditāh Mv i.274.8; calvāri putrāh Mv i.352.13; yāvanti sabdās Samādh 8.25.

6.15. Masc. modifer with fem. noun: parşāś ca caluāra(h) SP 9.1; prabhāsayam (= °yan, pres. pple. nom. sg. masc.) SP 24.2 (agreeing with sā raśmir in preceding line); vedanā (regularly fem., exceptionally nt.) vitivritā ime (masc.) LV 214.6; (apsarāh, line 21) te dadantikāh LV 236.22; other cases of te with fem. nouns, LV 240.18; 328.19; similarly ye LV 193.14; 328.19; rddhi vipulo LV 194.6 (printed as a cpd. by Lefm., but must be two words, 'the abundant prosperity'); Sujāla (m.c. for °lā) malimām (1 ° man) LV 271.21 'the wise Sujätä' (so also Calc., with gloss matimati); caluro dišah (acc. pl.) LV 282.9; sarve prabhā (nom. pl.) LV 291.2; gāthā imi (= ime) LV 355.18; trayo koliyo Mv i.61.13; trayo paksiyo (fem.) niryāntāyo Mv i.272.8; catvāro dhātrīyo Mv ii.423.6 (prose); sarve (of women) Mv ii.425.7; te devalā Mv iii.304.3; kanyāḥ . . . catvāro Suv 106.2; pūrayi tām kriya sarvi asesām Bhad 44 (three formally masc. epithets of kriya = kriyah; even if a nt. stem kriva be admitted, like kiriva reported for Pali, masc. would not fit).

6.16. Neuter modifier with fem. noun: parşās ca catvāri SP 294.11; 393.4; kecil tu catvāri (sc. jālih, 'births') alikramitvā SP 330.7; idam avaci giram LV 329.21 'spake this word'; divyāni sampatli (Senart em. °ti) Mv i.31.11 (prose; acc. pl.; cf. 32.9 divyām sampatlim, acc. sg.); le dāni devadevatām (1) namasyanti svakasvakāni Mv i.245.7 (prose); divyāni ca samgīli (nom. pl.) Mv ii.160.19 (prose); catvāri devalā Mv ii.232.17; trīņi vā jātim catoārim vā jātim (so mss.) Mv ii.284.8-9; catvāri nītayo Mv ii.404.19; puşkariņiyo ca nānāprakārāni acchāni etc. (series of adjectives, all nt. in endings) Mv iii.288.7 ff.; catvāri dhātriye (or °iye, mss.; Senart em. °iyo) Mv iii.405.6; catvāri koļi pramadānām RP 23.17; svakasvakāni strīyo (nom.) Mv iii.69.9; vāpīyo ca puşkariņīyo ca sukhasalilānī (and other adj. in -āni; nom.) 69.14; sarvāņi cāstādaśa śreņī (nom.)

6.17. Masculine and/or neuter modifiers with fem. nouns (i. e. ambiguous forms, or both in the same passage): calvar' (could be = °ra(h) or °ri) ima parsa SP 15.11 (vs); 16.8 (vs); caturna parşāņa SP 237.13; 294.6 (and according to both edd. 295.8, but here the mss. point rather to calasrna, fem.); mahyam bhāsalo LV 78.4 of me speaking' (Māyā speaks); yāvanti kecil lipi (nom. pl. fem.) LV 146.5; calurnam parisadam Mv i.8.11 and calurnām parsānām i.27.11 (both prose); pasyati anekām (= °kān, masc.) brāhmanānām sahasriyo (to sahasrī, fem.) samāgalāni (nt.) Mv i.310.1 (prose); leşām ceļīnām Mv ii.172.10; 173.6; te, nom. pl. (referring to these cetis) My ii.172.11; cf. cețini 15; lehi cețihi Mv ii.173.8; yāvanli senā ... sarve (sc. senāh) Mv ii.341.8-9; tesām (women) Mv iii.265.18; tasmān nagarīto Divy 222.6-7 (prose); tasmād rddhitaķ Divy 223.28 (prose); pralyavasrleşu bhikşublikşunişu Divy 235.27-28 ('ad sensum', common gender?); Vāsavadallayā lobhākrştena Divy 353.18 (prose).

6.18. The instr. masc.-nt. mahalá is very commonly used in Mv, and occasionally elsewhere, with the instr. of nouns like rddhi, vibhūsā, and synonyms. This seems to have been a widespread usage, for it occurs in Prakrit, at least AMg. Cf. Pischel 396, AMg. mahayā iddhīe etc.; Jacobi, Ausgew. Erz. 33.26 (AMg.) mahayā vibhūle (wrongly printed as cpd. by Jacobi). Examples: mahalā rājarddhyā LV 119.20 (prose); mahalā rājarddhīye (sometimes °iye) Mv i.216.18; 227.10; 258.19; 259.11; ii.180.7 etc.; mahatā samrddhīye Mv ii.75.6; 100.6, 10; 111.8; mahatā vibhūsāye mahalā samrddhīye Mv ii.111.12; ma° rājarddhīye ma° viyūhāye ma° samorddhīye ma° vibhūsāye Mv ii.156.17, etc.

6.19. Fem. modifier with masc. noun: pālāš calasro LV 233.14 'the four (loka-)pālas'; daksiņasyām dišo bhāge LV 389.6 (the only v.l. is daksiņasyā; Lefm. em. dakşiņesmin). One might suggest influence from the fem. gender of diś. But 389.10 dakşiņesmin diśo bhāge (Lefm., but many mss. °ņasyām or °ņasyā); pūrvasmin (d° bh°) 388.8; paścimesmin (3 mss. °masyām) 390.4; ultaresmin 391.2. Probably the reading of the mss. is to be kept. catasro mallākā (mss. mallā) Mv i.65.15 (in next line balavantah agreeing with the same masc. noun).

6.20. Fem. modifier with nt. noun: calasrah calurašitikoļivarsasahasrāņi Mv i.59.4 (prose). Uncertain is anutlarām jādna (m.c. for jādnam) gaveşamānāh SP 10.5, without v.l. in KN; but WT anultaram with their ms. K'. In MSV ii.88.18 f. (prose) occurs (sādhu) bhikşavah śraddhādeyasya (see this in Dict.) mātrā-paribojidāyāḥ kālapari°, then three other cpds. in pari°, which, it seems, can hardly be called anything but a gen. sg. f. of a ppp. agreeing with a nt. noun, 'a gift-of-faith enjoyed with moderation' etc.; there is nothing more in the sentence; no simple em. occurs to me, and the form is recorded five times.

7. Case

7.1. On the whole, the functions of the cases of nouns and pronouns are not markedly different from Skt., except for a certain amount of morphological confusion which is treated in the morphology: for example, the use of common oblique-case forms of fem. long-vowel stems; and the absorption by the genitive of the functions of the dative, resulting, as a reflex, in the use of old dative forms in gen. functions, particularly in the pronouns (both as in MIndic).

7.2. Here will be noted examples of some uses of the cases which, tho not always unknown in Skt., are at least not usual in the regular classical language.

7.3. Often no clear distinction can be made between morphological and syntactic confusion. There is no doubt of the existence of both, and frequently it seems to make little difference whether we say e. g. 'this ablative form is used as an instrumental', or 'this instrumental construction is replaced by an ablative.' Especially in the pronouns there is no doubt of extensive morphological confusion between the case-forms.

Syntactic agreement between different case-forms

7.4. Sometimes there is syntactic agreement between different case-forms in one and the same passage, usually (probably always) when both cases may be used in the same sense.

7.5. So, acc. and instr.: kalpāna koļyo bahubhīr acintyair SP 385.3 (vs), 'thru many, inconceivable crores of kalpas'. Most mss. read koļi, but either this or koļyo can only be acc. pl.; the following instr. adjectives agree with it. Both cases are used to express extent of time.

7.6. Acc. and loc. (others § 7.18): diso (acc. pl.) dasāsu (m.c. for dasasu) LV 314.8; and dasasū disato (apparently acc. pl.) LV 167.11; see § 9.97.

7.7. Instr. and gen.: (after comparative) mama ... kāruņikayā Divy 13.1, see § 7.35.

7.8. Instr. and loc.: tasminn eva kşanalavamuhūrtena SP 8.3 (prose), 'at that very instant, moment, and hour'. Either case could be used. No v.l. in KN. But WT report their ms. K' as reading muhārte, and adopt this reading. na ca cyavāmī ilu grdhrakūļāt, anyāsu šayyāsanakojibhiś ca SP 324.12 (vs), so both edd.; the only v.l. is one of KN's mss. 'kaļişā. Perhaps this should be classed below under § 7.32, -kajibhiś being describable simply as an instr. form used as loc.

7.9. Dat. and gen.: dharmeśvarāya... dharma bhāsalo (v.l. bhāş-) LV 171.14 (vs), 'to the Lord of Dharma as he is preaching the Dharma'. Possibly, also, lokanāthasya tāyine (text tāpine) Mmk 599.20; 600.17; but same phrase with "nah 499.19; perhaps "no is intended in this (very corrupt) text.

7.10. Gen. and loc. (wholly or chiefly 'absolute' expressions): mahya (vv. ll. mahyam, mama) parinirerte SP 237.1 (vs), 'when I have entered nirvāņa'; mayi nirerte (v.l. mama nirertasya) SP 252.3, and nirerte naranāyake SP 256.4, both undoubtedly attempts to correct the original reading preserved in the Kashgar rec., JRAS 1911.1071 and 1074, respectively mama nirertasmi, and nirertasya vināyake; sugatasya nirerte SP 392.9 (one ms. sugatesmi, obviously a secondary correction, adopted WT); mama... äyühalyäm niryühalyäm (loc. sg. fem. pres. pples.) Gv 199.24 (2d ed. °lyäh, twice; by em.?).

7.11. In Lank 9.11 ff. occur first three loc. forms, samananlaraprativibuddhe pardurttäšraye (so read, text °vrtā°) svacittadršyamātrādhigame, then gen. (a)vikalpapracārasthitasya, all of which agree with the next following noun, printed lankādhipale, for which we must certainly read °patch, gen.; after it follows a long series of other gens. in agreement with the same noun, ending in 10.2, where follows the verb ašrauşit (of which the intended subject can only be Lankādhipali).

7.12. In Mv iii.94.3 (vs) $c\bar{a}rik\bar{a}m$ pratipannasya sikhisya lokanāyake, 'when the World-guide Sikhin had begun his pilgrimage', is gen.-loc.' absolute. The words sikhisya lokanāyake are repeated many times in the passage 94.2-97.7, usually in similar absolute construction. In a few cases (94.15, 17; 96.22) only a possessive gen. seems interpretable, tho the loc. lokanāyake still remains in agreement. On the other hand, in 97.1 carante (loc.) replaces sikhisya, before lokanāyake. In a similar group of verses 98.12 fl., the phrase carante lokanāyake (as in 97.1) recurs many times; it is clearly absolute, at least in most occurrences; and in many, beginning 98.17, it is preceded by $c\bar{a}rikām$ pratipannasya, this gen. being in agreement with the loc.

Nominative

7.13. 'Nominative absolute' or nominativus pendens. (Treated for the Bråhmana language by Hanns Oertel, Syntax of Cases, 1. The Disjunct Use of Cases, Heidelberg, 1926.) This occurs rather frequently: priyajanaparivārās ca màram utsrjiya dinamukhā

priyajanaparivārāš ca māram utsrjya dīnamukhā ekānte gatvā pradhyāyantam apazyat, LV 301.17 'his friends and attendants abandoning Māra, and going to one side with downcast faces—lie saw them (sc. priyajanam, or parivāram) in pensive meditation.'

talo sudhanu kumāro pilarena subāhunā bandhanāgāre bandhilvā aham visarjilā Mv ii.111.2 'then Prince Sudhanu --by his father Subāhu, imprisoning (Sudhanu) in prison, I was dismissed,'

samanantaravyākrto punar jyotipālo bhiksur ... iyam mahāprthivī ... kampe Mv i.336.11, 'immediately the monk Jyotipāla (was) prophesied ... this earth trembled.'

sa khalv aham ... āhāram āharanto ayam me kāyo ... abhūsi Mv ii.126.16, 'I now, taking nourishment, this my body became ...'

na sattvah... yah samartho 'calāyā... sahadaršanena klešā na vyupašamam gaccheyuh Gv 173.11-13, 'there is no creature who would be able, immediately on seeing Acalā, his impurities would not become stilled.'

atha ... sudhanah ... yena ... shāvarā ... tenopasamkrāmato dašaprthivīdevalāšalasahasrāņy ... vācam udīrayām āsuh Gv 220.26 f., 'then Sudhana, as he was going to where Sthāvarā was, the 1,000,000 earth goddesses said.'

idānīm tvam evāgamya caksuh pratilabdham Divy 17.18 'now your very self, having come—sight has been got back (by us).'

krtsnā me rajanī jūtā (ed. em. yātā) mrtyum evānucin-

layan, Divy 422.6 (vs) 'the whole night passed for me, (as I was) meditating (nom. sg.) on death alone.'

sa Buddha ily asrulapūrvam šabdam srulvā sarvaromakūpāny āhrsfāni Divy 35.4-5 'he, hearing the word Buddha, never heard before, all (his) hairs stood on end.'

7.14. Nominative for instrumental. We may group by themselves some instances of the logical subject in the nom., followed by a passive construction which would require the instrumental:

In LV 186.7, read: na khalv apy ayam (so mss.) kumāro (or ^orah, so best mss., incl. 'l' of Weller 28) kadācid udyānabhūmim abhinirganlavyam, 'not, to be sure, this prince (nom.), is it ever to be gone forth (by him) to the park-ground.'

bhagavān dīpaņkaro meghasya ... anuttarāye samyaksaņbodhaye vyākrtaņ Mv i.239.4-6 (Senart em. vyākārşit), 'the Lord Dipamkara ... (by him) for Megha prediction was made unto supreme complete enlightenment.'

so kāndo lasya rājño pādamāle sthilvā pādaphalakam khandakhandākriam Mv ii.82.15, 'this arrow, stopping at the feet of the king. (by it) the footstool was smashed to pieces.'

so dāni rājā ... paribhogāni cu dinnā (sc. rājāā) Mv ii.443.16--18. And so 443.20--144.2 so dāni rājā ... dhīlā ... bhāryā dinnā.

te dāni udyānapālā ... āmravanam alamkrtam (sc. udyānapālaiķ) Mv ii.451.10—11.

tāni (sc. māmsāni) ca... aurabhrikā, bhaksyānili krlvā, mūlyahelor vikrīyanle, Lank 246.7-8, 'and these (unclean meats; nom. or acc. nt.), butchers (nom.), representing them as fit to eat, they are sold for money.'

7.15. Nominative for ablative. Instead of the abl. of comparison after *anyalra* 'except', a nom. is sometimes substituted (as if subject of a syntactically independent statement):

āhārasamjñā ca na tatra bhesyati, anyatra dharme rati dhyānaprītih, SP 206.1 (vs), 'and no trace of food will be there, except (that the food consisting of) delight in dharma and love of contemplation (will be there).'

anyalra bhūri (v.l. sūri, read so) bodhisallva (nom.) brahmakalpasamnibhe (nom.) LV 74.4 (vs), 'escept it be the Bodhisattva ...'

Accusative

7.16. The accusative with nouns and adjectives treated as verbal in force. (I. Sen 12-13; many of his examples, particularly those depending on forms in -anāya, I regard as infinitives (§ 36.15). It will be noted that sometimes the governing noun is the prior member of a compound. In LV 68.8-9, read saltbān pralisammodana-kušalo (cf. Weller 22), 'capable of greeting creatures'; dharmam eva gocarī LV 76.4 'devoted to Dharma alone'; sarvasaltbān dharmadānena sanıtarpunārtham LV 275.12 'to gratify all creatures by the gift of dharma'; pûjanā (inf., see §§ 36.18) hilakaram LV 282.8; śakyā kartum candrādi-tyau tamatimira vitimirakaran nabho 'dya mahīlalam LV 337.14, 'it might be possible to make obscure with darkness the moon and sun, which today make free from darkness sky and earth' (Tib. seems to have had a different Skt.).

Skt.). 7.17. Blend constructions resulting in two accusatives with verbs: anujānāti (usually governs either dat. or gen. of person and acc. of thing, or acc. of person and dat.-gen. or infin. of thing), Mv i.132.14 (vs) dešikam saha caurebhyah anujānāmi jivitam, 'to the guide with the thieves (apparently dat.-abl. ending for instr., but see § 7.46) I grant life' ('the guide ... I permit to live'); avakirati, usually construed with acc. and instr., 'bestrew (something) with (something),' takes two accs. in LV 317.17-18, muklasukusumāni ca mahīm avakiranto, 'strewing loose flowers upon the earth.'

7.18. Accusative in precise dating. The acc. of time is often used (not only of 'extent' but) where we should expect the loc., in precise dating. To be sure the like is not unknown in Vedic and Skt. (Speyer, VSS § 28). So kulpānu šalasahasram Mv i.35.13 (prose), '100,000 kalj.as ago'; bhaveyam anāgalam adhvānum lathāgalo Mv 1.238.14; 330.1; 'may I in (some) future time become a T.'; with parallel loc. anāgalam adhvānam aparimite asamkhyeye kalpe Mv i.239.6; alītam adhvānam Mv i.286.22; 288.13; seeming blend, atitam adhvane (perhaps to be interpreted as atita-m-, with 'hiatus-bridging' m) My ii.48.16 (so mss.); 64.14 (so text with one ms., v.l. 'te-m-adhoāne); 67.19 and 69.11 (both without v.l.); nāsli mamu adya rātrīm kṣaṇo My iii.37.6 'tonight 1 have no time (liberty)': yam (or yat; kalam 'what time' = 'when', quasi-conjunction, common. Mv ii.210.14; iii.144.13, 15: 145.7: 272.9: yatra kālam = prec., Mv iii.295.13, 15, 16; yām velām = prec., Mv i.362.11; in My more commonly yam velum (MIndic), e. g. i.361.3: 362.2, 17. 20; ii.172.15; 173.18; 210.17; 211.4, 12; 242.13; iii.76.1; 145.5; 163.4; 291.8, 20 (with correlative tam velan) in 292.1); note yam velam . . . tage velam iii.287.11, resumed in 13 with yatra kale . . . tatra kale; also yatra (ca) velam ii.65.14.

7.19. The accusative of measure: not unknown in Skt., tho called 'rare' by Speyer VSS § 27. It is fairly common here; the few examples given by Sen p. 10 could be greatly extended. Those interested will find some others in my Dict. s.vv. *āvedha* and *udvedha*; e. g. (yaşti ...) *dvādašayojanāni udvedhena*, (a stick) 'twelve leagues in height'.

7.20. Miscellaneous accusatives, for more usual loc., dat., or other cases.

7.21. Partiples of root budh with the goal in acc.: buddhe ca bodhim naranāyake 'smin SP 190.4, 'upon this Lord of Men, enlightened into enlightenment' (such cases are less unusual with finite forms, cf. Sen 6 under 'cognate acc.'); cira budhyamānena anultaram padam SP 190.6, 'for a long time becoming enlightened into the highest place (== enlightenent, or nirvāna).'

7.22. Other 'cognate' or 'inner' accs.: aham eva tubhya yali gacchayami (= lava galim gacchāni) LV 223.1, 'I am going your course (i. e. where you go)'; viryabalam parākramā LV 178.18, 'display the force of heroic energy'; āsām (= ājām) oralam sidhyalām LV 221.21, 'let (your) yow succeed (according) to (your) desire!'

7.23. For expected loc.: aparamālaram prasakto Mv i.244.7-8, 'devoted to his ''second mother'''; shilā calasro ... calurdišam Mv ii.56.17, 'four maids stationed in (each of) the four directions'; purimām dijām tişhasi devale loan'id. 19, 'thou standest, goddess, in the eastern quarter'; caraņau (? so probably read with several mss.; Lefm; caraņo; only Calc. caraņe) ..ipatya LV 416.4, 'falling at his feet'; svabhavanāni śmaśānasamjñām utpādayām dsuh LV 278.7, 'conceived their own (divine) dwellings to be (like) graveyards' (the normal construction with samjñām utpād dayati 'form a notion, conceive acthing' is loc., as e. g. SP 319.8 na ca tahhāgate durlabhasamjñām utpādayeyuh; here I believe the analogy of root man, which takes two accus., has operated, cf. LV 280.20, the verse equivalent to 278.7, vimānānt śmašānāniva menire).

7.24. For expected abl.: astamām bhūmim prabhrti 'from the 8th stage' Mv i.105.11, 13 (prose), 15 (vs; here read with mss. prabhrtim, m.c., Senart em. ${}^{\circ}ti$). Noted by Sen 14.

7.25. A curious apparent use of adverbial acc.: ayam ca... lokadhāluh şadpikāram aşlādasamahānimittam abhāl LV 410.21-22 (prose), 'this world became in sixfold fashion eighteen-great-omen-wise' (there follows the stock list of akampal prākampat etc., six groups of three verbs each). 7.26. Asyndetic acc. (sentence not completed): kim puno adya mām hinasamsevalas trpti gacched aham LV 215.18 (vs), 'how much less today to me, from cultivation of the low,—could I go to satisfaction?' Since aham cannot be construed otherwise, trpti must be acc. with gacched (used as 1 sg.). Cf. BhG. 11.26, with my note, HOS 38.186.

Instrumental-accusative adverbial phrases

7.27. Like Pali, BHS has a common adverbial phrase kālena kālam, which means (again as in Pali) both (1) 'from time to time', SP 276.7, 10; Av i.241.9; and (2) 'continuously', LV 100.13; Divy 62.10; 71.5; 298.25; probably Kv 28.14. Sometimes it is not clear which meaning was intended: SP 225.5; Mv iii.154.14; Divy 10.27; Suv 94.3.

7.28. A few other similar expressions occur: ulsange-notsangam LV 76.20 'by myriads' (ulsanga); vrksena vrksam paryalanti LV 82.15 'wandering from tree to tree', it. 'tree-by-tree' (followed by vanād vanam caikramya-mānā, cf. § 7.29, 'from forest to forest', abl.-acc.); grāmeņa grāmam anucankramantah SP 112.4 'wandering thru village after village'; ärämenärämam udyänenodyänam parvalapāršvena parvalapāršvam MSV i.287.21. Apparently intensive is duren duram SP 287.13 'very far'; and certainly sarvena sarvam 'altogether, completely'. This (see Dict. s.v. sarva) is extremely common. It corresponds to Pali sabbathā sabbam, to which the precise equivalent sarvathā sarvam also occurs in BHS, but as a rule following sarvena sarvam in one long phrase (also sarvena sarvam sarvatha, sarvena alone, and possibly other variations). Sen (who fails to note sarvena sarvam) lists (24-25) several other cases, which are dubious or definitely not of the same sort. Thus LV 143.20 (Sen cites 163) mayā sārdham samarthah silpena silpam upadarsauitum means 'able to display art for art with me' i. e. 'to rival me in art'; cetasā cetahparivitarkam (a common phrase, see Dict. s.v. parivilarka) has cetah as prior member of a compound.

7.29. On a somewhat similar but rare use of abl. and acc. adverbial forms see § 7.51, besides vanād vanam (prec.), which seems not abnormal.

Instrumental

7.30. Instrumental for locative. In BHS, historic instr. forms are extensively used in loc. function (also abl., see below). This is specially true of plural forms, particularly of *a*-stems. It can hardly be doubted that it is connected with the regular use of the endings $(c)h\bar{i}$ and $(a)h\bar{l}$ in Ap. as loc. as well as instr.; in the pl. of vocalic stems, the two cases are completely fused in Ap. (Jacobi, Bhav. 33° etc.). In BHS it is not only MIndic endings like *chi* or the like which are often used in traditionally loc. meanings. Doubtless by analogy with them, forms in *ais* are used in the same way.

7.31. It is, however, hard to draw a line between these cases and others, including instr. sing, forms, which also show what are loc. functions in normal Skt., yet are not so readily explainable by reference to the Ap. usage just mentioned, tho they also may be fundamentally MIndic. Thus, the instr. is often used to denote the place where, as well as the time when. Some such phrases are very common in BHS, and equally common in Pali.

7.32. Place where: yena ... tena... 'where, ... there' (Pali id.; Sen 23) is so constantly used that illustrations seem superfluous, e. g. yena sa pilā lenopasamkrāmann SP 75.3. This usage, or at least yena 'where', is not quite unknown in Skt. (at any rate epic): prādravad yena vai sarah Mbh. (Crit. ed.) 3.137.15. Others: karņaśrolābhyām ... praksipya ... mukhadvāreņa praksipya LV 257.8, 10; dvihi kulehi ... bodhisatīvā jāyanti, kşatri-

yakule brāhmaņakule vā Mv i.197.12 (note -kule, resuming -kulehil); satsu gatihi Mv i.42.17 (vs) repeated as satsu galişu 53.14, and in prose 337.5 as salsu galisu; so dirghāyukehi devehi upapanno 'he is born among the D. gods' My 1.51.14; tehi brahmanasahasrehi 'among these thousands of b.' Mv i.314.3; etchi evam āsanehi nisamnānām 'of them thus seated on these seats' Mv i.329.18; caturhi padehi pralibhuko 'surety on four points' My ii.141.4, repeated in next line with padeşu caluruşu (better calurşu with v.l.); etāhi šivikāhi nişīdāhi 'sit in these litters' Mv ii.216.17 (similarly 227.6); devihi upapannä 'born among the goddesses' Mv iii.375.14; anyāsu sayyāsanakolibhis ca SP 324.12 (only one ms. "kolişū cu, out of at least six) .- In slight extensions of the 'place where' notion: prāņehi ahimsam 'non-injury to 'iving beings' My i.145.8 (loc. sense proved by repetitions of same line with pranisu i.202.5, prāņesv ii.6.2); survehi bhūlehi nivārya daņdam Mv iii.412.21 'restraining violence in reference to all beings' (the same vs Divy 339.28 has sarvesu bhūtesu nidhāya, and so also the Pali equivalent Dhp 142 sabbesu bhūtesu); with expressions of pity (loc. in Skt., Speyer, Skt. Syntax § 148), tehi vikrośamānehi salvehi karuņām pratilabhilvā Mv 1.128.9 'conceiving pity for these creatures who lament'; with expressions of desire, lasyā kākīye rājabhojanena dohalako Mv iii.125.16, 'the female crow had a pregnancy-longing for the king's food'; grddho grhi na (so read) tathā kāmair yādrše pravrajitva te grddhāķ RP 29.11, see Dict. s.v. grhina. For others see Sen 18, infra.

7.33. In time expressions the instr. in Vedic and Skt. supposedly denotes the time 'within which' or 'during which' the action takes place (Speyer VSS § 42). In BHS as in Pali it is commonly used simply to denote the point of time 'at which', a usage perhaps not so rare in Skt. as is commonly assumed. The expressions *tena kālena, tena samayena* 'at that time' are common in both Pali and BHS; sometimes temporal locs. occur in the vicinity. So *tena* ... samayena SP 19.11; 20.4; shortly after the latter *tasyām velāyām* 20.8. Further, aparena kālasamayena LV 189.9 'at another time'; in Mv i.50.15 *ekasmim kalpe nādrākşil, kalpasahasrena* (v.l. 'srehi) buddham loke pasyati; loc. and instr. seem equivalent. Also *tasyaiva rātryā atyayena* Mv i.231.3 'at the end of that very night', etc. Unusual is MSV i.47.13 yo 'sau Vipasyi-tathāgatena gaņawācako..., 'that school-teacher at the time of Vipa-syin Buddha'.

7.34. The loc. absolute is replaced by the instr., and not only in the possibly ambiguous forms ending in ehi (cf. § 7.30); raiña simhahanuna kalagalena suddhodanena rājyam pratilabdham Mv i.355.20, 'when King S. was dead, S. obtained the kingship'; mayā phalahāragatena Mv ii.65.15, 'while I was gone to gather fruits' (not 'by me ...'); nāsti mayā mŗtena teşām jivitam Mv ii.214.7, 'there will be no living for them when I am dead'; yam aham āryaputreņa duņkhitena ... bhojanāni bhumjeyam Mv ii.233.4-5, 'that I, while my lord is tormented . should enjoy luxury'; sarvair varnai(h) slutimangalaih pratyupasthilair LV 119.11, 'all praises, lauds and benedictions having been presented'; mayi parinirvrte imehi ca śrāvakehi parinirvrlehi imasmim dharmäkhyāne 'nlarahile My i.50.13 (note -ehi, pl., parallel with loc. sg.); candramasūryehi loke aprajňāyanlehi Mv i.339.2, and parallel forms in chi in the sequel, until line 5, where in the last parallel we find mäsärdhamäsesu loke aprainävamānesu; mānsehi ksīyamānehi Mv ii.239.5, in the same verse in which LV 262.4 reads māmsesu ksīyamānesu (dental n); sarvena arthena samudre vinastena Mv iii.350,12, 'all his wealth having been lost in the sea' (no other possible construction); paripūrņehi ca dašahi māsehi Mv i.148.1 (prose), 'and when ten months have been completed'

(= pralipūrņe dašame māse i.148.4, vs); aprahiņehi klešehi Mv i.153.7; asmehi jivanlehi Mv i.313.8; etc.

7.35. Instrumental for ablative. Instr. forms are also used in functions that are normally ablative, as in Pali (Geiger 77, end). In expressions of comparison, the instr. is not unknown even in Vedic and Skt. (Speyer, VSS § 39). We may note here ultamu sarvadevaih LV 119.5, 'supreme over all gods'; *lvayā ali-r-ivāpi* (so read with mss., see Dict. s.v. *ali-r-iva*) Mv i.129.6, 'even in excess over (in comparison with) you'; in Divy 12.29-13.1 the text presents, after a comparative, a gen. pronoun with an instr. adjective agreeing with it: *kim mama kāruņikayā lvam eva kāruņikalarah*, 'are you really more merciful than me, merciful (as I am)?'; *imehi kila so višişlalaro* Mv i.306.18, 'it appears that he is more distinguished than *lhese!*'.

7.36. Otherwise. I have noted only plural instr. forms in abl. functions. I take this to be related to the Pali and Pkt. use of plural instr. forms in abl. meaning (Geiger 77 end; Pischel 369, 381). But in BHS it is not limited to -bhis (-hi) endings; doubtless by analogy with them, -ais forms from a-stems are also so used, so that KN in note on SP 252.14 are not justified in describing the variant reading ksetrasatair anekaih 'from many hundreds of fields' as 'wholly wrong'. This v.l., read here by most Nep. mss., is adopted by WT, while the variant of the Kashgar rec. (and one Nep. ms.) ye agalah kselrasahasrakolibhih 'from crores of thousands of fields' is adopted by KN, with the parallels 35.13 and 331.2, where no v.l. is recorded. Others: karacaranalalebhi yāvad ūrdhvam (text ūrddham) LV 49.15, 'from the palms of her hands and the soles of her feet up to the top (of her head)'; dhyanasukhebhi bhrastáh LV 138.5, 'fallen from the bliss of medita-tion'; mocehi (Lefm. soc°, misprint) jagu vividhakileśaih LV 164.4, 'free the world from manifold impurities'; payu tava sravi tada karatalacaranaih LV 166.2, 'then milk flowed from your hand-palms and feet'; tunavinasughosakādibhih sabdo niscarate LV 177.14, 'a sound came forth from ...'; sabd' evarūpas (so read) turiyebhi niscarī LV 182.19 (as prec.); netrebhi niścarati vidyud LV 339.18, 'from their eyes lightning comes forth'; (ye āgalā) dašadisebhir LV 416.2, 'from the ten directions'; dhyānehi ca abhijñāhi ca bhrasto Mv i.354.5, 'fallen away from dhyānas and abhijnās'; anychi pi adhişlhānchi ... āgacchali Mv ii.74.8-9, 'and from other places too ... came'; kāmehi avyapakŗslakāyā Mv ii.121.3, 'whose bodies are not withdrawn from lusts'; kacchehi svedā muktā Mv ii.124.3, 'sweat poured from (my) arm-pits'; mukhalo (ca) nāsikā-śrotrehi ca Mv ii:124.10, 11, 13 'from mouth and nose and ears' (note mukhalo, abl. sg.1); Irājāmi (text Ira^o) sarvaja-nalām vyasanair anekair Gv 212.26, 'I save all folk from many disasters'.

7.37. Miscellaneous uses of the instrumental. By analogy with adjectives meaning 'like', visama 'unlike' is construed with an instr.; lokena visamam Mv i.107.1, 'different from the world'. Otherwise Senart's note.

7.38. A predicative instr., the nearest approach to which in Speyer's material is the instr. with variate (Skt. Synt. § 67, Rem. 1), is found with driyate: kasya pārvanimillena mārisā adya driyate SP 168.1, 'as first omen of what is seen today (that which we see)?' Cf. 167.13 kasya khalv idam pārvanimillam bhavişyali, 'of what, pray, will this be the first omen?' So also with the gerund āhāya: puruşo dūlenāhāyoktah Av i.325.12 'a man, (by the king) having summoned him as messenger, was addressed'. So read with ms.; Speyer em. dūlgenā', referring to Skt. Synt. § 237. But the abstract noun is daulya, not dūlga. Cf. BhG. 6.33 yo 'yam yogas loayā proklah sāmyena, 'this discipline which is declared by thee as indifference'. The predicative instr. is equated with the subject; Speyer's § 237 deals with a different idiom. **7.39.** The instr. of the stem *ālman* is used with an ordinal as an 'analytic form' of *ālma*- in composition with the ordinal; see Speyer Skt. Synt. § 300, quoting *ālmanā trliyena ganlavyam* 'you must go with two others' (= *ālmalrliyena*). So *imam kumāram ālmanā calurtham* ('with three others,' = *ālmacalurtham*) ... nehi Mv ii.108.15; similarly 18, and 110.16.

7.40. Doubtful is alpähäratayä pralipadyeyani LV 254.2; 255.7. Weller 30 assumes that the instr. is used with the function of a dat., comparing LV 255.14 and 18 bhäratäyai pralipadyeyam. I may add that pralipädayati, caus., in BHS means 'presents (with)' and is construed with the instr. On the other hand, in view of alpähäratayä suddhim manyante 254.1 (just before the above phrase), we might assume that pralipadyeyam is used without expressed goal: 'I must undertake (to win my goal) by eating little food'.

7.41. The instr. of the logical subject is sometimes used when the following verb (usually rather remote in place) is active, logically demanding a nom.: jyolipälena bhiksunä ... bhagavantam ... abhyakiletsuh (1 so mss.; Senart em. abhyokiresi) Mv i.335.9-10; täye däni syämäye so mrtako puruso gandhodakena snäpayitvä ... (other gerunds) ... krivä lesäm celinäm äha Mv ii.173.5-7, 'now this Syämä had the corpse bathed (etc.) ... and said to the maidservants'; anyähi devihi ... sudarsanäm agrato krivä oksatä Mv ii.450.1-2, 'the other queens ... putting S. in front, descended'; similarly SP 316.2-4.

7.42. By a similar blend (cf. Weller 34 f.) an instr. of the logical subject is followed by a nom. of a past participle which goes logically with it: bhagavatā śākyamuninā... samyaksambodhim abhisambuddha iti SP 316.2-4 (prose); iha mayānuttarām samyaksambodhim abhisambuddha LV 351.16-17.

Dative

7.43. Dative for genitive. As in MIndic generally, and to a considerable extent even in Skt., the dat. is extensively replaced by the gen., except for the infinitival dat. of purpose. A reflex of this is the occasional use of dat. forms in definitely gen. function. So especially with personal pronouns, e. g. bhāşitam agra mahyam SP 93.6, 'my supreme word'; mahya ime 'nyaśrānakā (mss.) SP 93.9, 'these my other disciples'; upāyakauśalya... mahyam SP 93.13 'my clever device'; ima sūtra mahyam SP 95.6, 'this my sūtra'.

7.44. But also with nouns: tebhyah samnipatitebhyas cittāny avalokya Lank 2.14, 'beholding the hearts of those gathered together' (ed. suggests reading gen.); several times the dat. bodhāya (see Dict. s.v. 1 bodha) is used in SP in definitely gen. function (KN always emend to bodhāya, but the mss. must be followed), as in te sarvi bodhāya abhāşi lābhinaņ 49.14, 'they have all become attainers of enlightenment'; yam (284.2 yatha) śrutva bodhāya (all mss.) bhaveyu lābhino 171.1; 284.2. 7.45. 'Expletive' or 'ethical' dative. This old in-

7.45. 'Expletive' or 'ethical' dative. This old inherited construction is mentioned here, by exception, because of its rarity and interest, not as being non-Sanskritic. See on Vedic and Skt. Delbrück in Brugmann's Grundriss III.298, and AIS 206 (add *le*, twice, Rigveda 1.97.4); Holtzmann, Gram. aus dem Mbh., § 285; Renou, Gr. scte. p. 297 top (where some of the Mv forms below are cited). For Latin examples, and English parallels, see Gildersleeve-Lodge, Latin Grammar, § 351; Hale and Buck, Latin Grammar, § 372. Senart Mv i note 456 compares Pali vo (= vas), Jät. i.267.20 passāmi vo ham at-tānam, on which the comm. says 'here vo is a mere particle', ettha vo li nipātamattam. Buddhacarita 8.58 has: katham bala svapsyali so 'dya me vrali, 'how, alas, will he sleep, subject to an (ascetic) vow, today ...'; Johnston rightly

Ablative

7.46. Ablative for instrumental. As instr. forms are used with abl. meanings (§ 7.36), so also the reverse happens with abl. pl. forms, perhaps by false Sanskritization of MIndic endings in *-hi: venubhyo ... ranante* LV 163.13. 'they make a noise with pipes': *ādharşilā ca vadhabandhanutādanebhyo* My i.84.2, 'violated by slaying, imprisonment, beating': *dešikam saha caurebhyah anujānāmi jivilam* My i.132.14, 'to the guide with the thieves I grant life' (or is *caurebhyah* understood as dat., *saha* being an independent adverb, the construction blended? see § 7.17).

7.47. Predicative ablative. Like the instr. (§ 7.38), the abl. is used predicatively with expressions of recognizing, declaring, and the like: it is to be rendered in English by 'as ...': yo 'lyayam alyayado drsfwâ pradideśayady LV 379.13, 'who. recognizing his sin as a sin, confesses it': the passage Mv i.173.1 ff. contains a series of such expressions, in which noms, with following iti vary with the abl. forms: nāham ... ye dharmā anityās le nityato deśayāmi, 'I do not teach impermanent dharmas as permanent', and many like it. In lines 6-7 nāpi ye dharmā rūpiņas te arūpiņo deśayāmi, the word arūpiņo is probably meant for acc. pl., tho it could, of course, be abl. sg. See Senart Mv i note 515, on this passage, referring to the like in Pali (sāram ca asārato ňalvā Dhp.12).

7.48. Miscellaneous. With yārat 'until', the abl. is sometimes used instead of the normal acc. (or loc.), doubtless by analogy of ā plus abl. 'until'. The usage occurs sporadically in epic Skt.: Speyer, Skt. Synt. § 169; VSS § 90.14. Exx.: yāvat tasya ... -nirrāņakālasamayāt SP 160.5 f. (vv.II. °samaye, °samayo 'bhūt); yāvad brahmālokāt SP 164.1 (no v.l.); yāvad bhavāgrād SP 365.1 (no v.l.); yāvat krkāļikāto Mv i.17.10; 19.7 (yāva k°); yāvad abhiniskramaņato Mv 1.157.15.

7.49. A double $y\bar{a}val$ (each, or at least the second, usually followed by ca) is used in the sense of 'from ... up to ...' (Dict.). The following nouns are commonly acc. or nom., but at least once abl.: $y\bar{a}vac$ ca nadyā nairaājā yāvac ca bodhimandāules (Calc. with v.l. °mandād) LV 273.9, 'from the River N. up to the bodhimanda'.

7.50. On the partitive abl., not unknown to Skt., see § 20.55.

7.51. In Divy 182.28–29 occurs a curious use of adverbial abl. plus acc. forms, apparently with intensive meaning. This reminds ane of the not-dissimilar use of instr. and acc. forms mentioned § 7.28: samphät sampham pügät pügam samgamya samägamya, 'assembling and coming together in crowds, in quantities'.

7.52. Unparalleled, as far as I know, is the following: pratinivrita-to dāsyatha MSV i.103.6. 'you will give it after I have returned'; -tas added to stem of the ppp. Tib. cited slar (wrongly printed sla ra) hons na.

Genitive

7.53. Genitive absolute. This is far commoner than in Skt., but since it is well known in Skt. too (Speyer, Skt. Synt. p. 286 ff.; VSS \$211 f.), it perhaps hardly requires illustration. It is by no means limited to use anādare (Pān. 2.3.38); for example, this sense is not present in sthitamātrasya buddhasya Lank 4.9; jālasya mahyam LV 118.21; elasya nirgalasyā LN 202.7; lesām ... ágacchantám (so mss.) My i.18.10, in parallel passage i.21.5 teşu ... ágacchanteşu.

7.54. Genitive for ablative. The genitive is used for a regular Skt. ablative in a number of constructions; to some extent the same is true of Skt. (Speyer, Skt. Synt. §§ 125–128; VSS §§ 73–74). So the abl. of time: cirasya Mv i.237.16–18; 246.7 etc., common: nucirasya Mv i.232.2.

7.55. With verbs of fearing: leşām paksinām bhilā Mv 1.7.7; saced bhāyasi (later ms. bibheşi) duļkhasya Ud ix.3.

7.56. With verbs of separation, departure, deprivation: for the Pali phrase agārasmā (or agārā) anāgarigam pabbajati, the corresponding BHS also uses abl. forms of agāra, usually agārā (Mv also 'rāto, 'rato) anagārigam ('kām etc.)..., in most texts: Mv i.128,10 (agārebhyo 'nagārigam pravrajilvā); ii.69.1; 161.5 ff.; iii.176.2; 378.4; 408.2; Av i.136.6; 234.1; LV 18.8; 101.19; 103.20; Divy 17.16-17; 37.12; 141.1; Samādh 8.15; Bbh 26.12; Bhīk 10b.1. But in Mv the gen. agārasga is quite as common. to cours Mv i.322.15 agārasgārigam pravrajas; 323.1; ii.117.18, 20; 140.3; 271.8; iii.50.11-12; 213.2-3, at least.

7.57. Others: spayunkriänäm karmanäm paläyilum Mv ii.214.18 f.; 217,4 'to escape from the deeds done by oneself': ašravanäd dharmasya parihiyate 1.V 403.10 (and read so for ašrulavän dh' in 403.6); ... 'yante 404.9, 'is (are) deprived of the law, thru not hearing it'.

7.58. With the prepositions \tilde{a} and $y \tilde{u} v a t$ (cf. § 7.48) 'up to, until': \tilde{a} muhūrtasya (cd. prints as one word) Mv iii.296.18 'up to a moment', i. e. 'before a moment had passed': yūvat paramasambodhiprāptasya Mv i.45.11,

7.59. After comparatives: na ca me (Kashgar rec. matto) kaścid viśistataro 'stili SP 134.11 (prose), 'and there is no one more distinguished than 1'; mama ultari yo (text ultariyo) LN 119.3 (vs), 'who is beyond me'; naitasya (text ne°) ... ultari LN 125.5 (vs), 'not higher than he'; mama kârunikayā etc. Divy 13.1, see § 7.35.

7.60. Genitive for instrumental. The gen. occurs sporadically in place of a Skt. instrumental. With sårdham (as also occasionally in Pali with sadhim. Childers, PTSD): dhiluh sårdham My i.36.14 'with the daughter'; acc. to Senart also måluh sårdham My ii.26.1, but the mss. here read bodhisallno ca mäyä (Senart em. mäyäya) ca måluh sårdham sivikā samårūdhāh (Senart em. 'dho). We must understand måluh as nom. sg. (§ 13.23): 'The B. and Mäyä his mother mounted together on the litter'.

7.61. Of price or value: imasya salasahasrasya kesaram Mv i.38.4, 'kesara for the price of this hundred thousand'.

7.62. Another associative use: le samelya vararûpadhārinām Mv i.81.18, 'they, joining the Bearers of the Best Form (Buddhas)'; the parallel 48.1 reads upelya ... °dhāriņo (acc. pl.).

7.63. Genitive for dative. The gen. in practically all the senses of the dative (excepting only that of purpose) is so familiar even in Sauskrit (and so universal in MIndic) that it seems unnecessary to illustrate it. I shall only cite a few instances in which the gen., apparently sub-stituting for a Skt. dat., is used of the 'indirect object' or remoter goal (a person): samprāpayāmi (read mi, with WT) ima teşa prāņinām SP 45.5, 'I bring it (sc. dharma) to these creatures'; radhakām (text ba²) sa lavā upanelī (so read, for text "tri; most mss. paneli) LV 168.18, 'he brought slayers to thee'; jinamātur (or $\circ tuh)$ upagatā drastumanā My i.146.7 = 203.8 = ii.7.4, 'came to the Jina's mother, intending to see (her)'; kasya vā garahāmy aham Mv ii.480.15, 17; iii.16.5, 7 'or to whom shall I make complaint?' (gen. for dat. of the person complained to: otherwise Senart's note).—With verbs of speaking: bhagavām veņuvane bhiksuņām āmantrayati Mv iii.63.15, 'addresses the monks'; sārthavāho tesām avaruddhānām vānijakānām āha My iii.73.6; evam caisām vadali SP 79.7.

7.64. Genitive in predicate with forms of bhavati. Very familiar is the idiom gen. plus etad (rarely evam, or omitted) bhavali = Pali holi (usually preterite verbforms in both languages), 'this thought occurred to him: he thought as follows', generally with following direct quotation. All texts have this idiom, and in narration it is extremely common. A few examples: mahasattvasyaitad abhûl SP 7.4; teşām sarveşām etad abhavat SP 33.7; teşām ... etad abhūt LN 26.4; tasyaitad abhūt LN 101.5; teşām evam bhavati Mv i.30.4; mahāmaudgalyāyanasyu ... etad abhavat 54.11-12; tasya (dâni) etad abhūsi 232.7, 9; tasya rājňo bhavali 272.2, 'that king thought'; rsisya bhavali ii.32.7 'the rsi thought': asyā etad abhavat Divy 67.7: teşām ... evam bhavali 68.6, etc.; teşām ... etad abhūt Jm 41.23; śakrasya . . . etud abhavat Mmk 663.14; tasyailad abhavat Suv 6.7.

7.65. Genitive with miscellaneous verbs. Most of these verbs seem rarely, if ever, to be construed with gen. in Skt. Some of them are noted by Sen.

7.66. Verbs of intellectual perception, and the like (some, such as *budh*, are so used in Skt.): *lesám manasi-karonti* SP 87.2, 'they take note of these things'; gunānām ātmani prajānile LV 141.9, '(womenfolk) make claim or profession of virtues in themselves' (but v.l. gunān).

7.67. Verbs of desiring, seeking (Speyer, VSS § 67): sprhenlu tāvan mamu daršanasya SP 324.10; tava pradiksante (= pratikşante) LV 185.7 'gaze (longingly) toward thee'.

thee'. **7.68.** śru 'hear': sthavirasya mahāmaudgalyāyanasya śrutvā My i.8.15 (see Senart's note); śrnatha mama bhāşalah 10.8; sthavirasya śrutvā 29.16; 30.14; tasya grāmikasya śrutvā 302.3; âyuşmato ānandasya pratyaśrosi 317.18, 'agreed to'. — asmāka (n.c. for °kam), in SP 116.7 śrutvā ca asmāka, may be considered an acc., see § 20.48.

7.69. Verbs meaning 'praise, honor' and the like: lasyā nabhe vrajanto stavayimsu apsarāni (so read with most mss.) LV 223.8 (vs), 'Apsarases, moving in the sky, praised him'; cl. also LV 195.10, § 20.16; lasya me ... paramayā pūjayā pūjayet paramayā ca prašamsayā prašamse Mv ii.119.2-3; parikīrtaya (for paryakīrtayat) rājaputrasya Mv ii.205.6 (vs).

7.70. Verbs of asking (cf. Sen 33) may take the gen. (1) of the person questioned, as well as (2) of the thing asked about. (1) mätäpit²näm ... preche LV 231.17; bhiksänäm prechati Mv i.300.5, 12; amätyänäm prechati ii.155.14; rsikumäränäm prechati ii.355.1; lesäm mrgaräjänäm prechati 360.14; lesäm purusänäm prechati ii.208.6; --(2) bodhisattvasyänyonyam pariprechanti sma LV 20.11 'asked one another about the B.'; sarveşa lesa (so text with best mss., v.l. sarveşu lesu) pariprechati LV 303.2, 'asked about all these things'; pariprechanti sütrasya vinayasya mälrkäyäh Divy 18.15.

7.71. The root bādh 'oppress' (cf. Sen 33, also Pān. 2.3.56): na ca kasyacil satlvasya rāgo bādhate LV 52.12 (prose); no ca tasya rāgadosa naiva moha bādhate LV 75.9 (vs); bhikşusamghasya bādhitum Mv i.314.9 (prose); yo bādhate pūrvakarisya rājno Mv ii.184.12 (vs; mss. °kārisya rājam, Senart em.).

7.72. The root han 'slay' seems sometimes to take the gen., tho this is not entirely certain: haneyu mahyam LV 232.1 (vs), 'they might kill me' (but possibly mahyam may be understood as acc., by analogy with mama which is certainly so used; § 20.12, and cf. § 20.16 $te \dots$ nihanişyâmi); nihanilvana māracamūnām Samādh 22.5, 'having slain the hosts of Māra' (two mss. out of four °camūni; possibly read so, despite the fact that $cam\bar{u}$ is a fem. noun). (Pān. 2.3.56 prescribes the gen. with *ni-pra-han*.)

7.73. abhivādayati is used of salutation, particularly, tho not exclusively, at departure. Is this the reason for the (isolated?) use of the gen. in My i.354.14? It so this should perhaps be included under ablatival genitives, above: mātāpitņnām abhivādilvā (mss.), 'having taken respectful leave of their parents'.

7.74. ava-man 'despise': brāhmaņānām avamanyitvā Mv i.309.11, 'scorning the brahmans'.

7.75. vi-smi 'be astonished' (in Skt. with instr., abl., loc.): $r\bar{a}j\bar{a}$ rsisya vismilo Mv ii.32.1. (the king was astonished at the rsi' (pple. as periphrasis for finite verb).

7.76. The root suc 'grieve (for)': asmākam eşā ekaputrasya soculi Mv ii.174.19 f., 'she grieves for our only son' (just before, 174.18-19, socurti governs acc., as in normal Skt.).

7.77. upa-sthā 'wait upon': lesām mālāpilīmām upasthihati My ii.211.17.

7.78. pari-rakş 'protect': teşām šakuntakānām parirakşati Mv ii.251.4.

7.79. It is alleged (e. g. by Senart on Mv i.123.13, and by Sen 41) that verbs of ruling govern the gen., as in Vedic and rarely in later Skt. (Speyer, Skt. Synt. § 118). But the instances noted are not convincing: *idsid idsali pāqinām* Mv i.123.13 may mean 'the Teacher of creatures rules'; in Mv i.349.3 the gens. are partitive, depending on *kimcil*; in Mv ii.420.7 the gen. is objective, depending on the noun *rāyyam (kārayali)*, 'he does the ruling at 60,000 cities'.

Locative

7.80. The uses of the loc. in Skt. itself are so free and extensive that little enlargement of its scope seems possible, and I have not noted much.

7.81. Loc. for instr.: Senart Mv i n.603, assumes this for *pitari* i.310.17; but see \S 13.33 fl.; *pilari* and *pitare* are common oblique-case forms and there is no reason to consider them primarily loc. But in Mv ii.38.15-16 *nišāmya duātrimsailaksanesu upāgatam, lathā anuoyañjanehi*, 'perceiving him attended by the 32 marks and the subsidiary signs', the loc. pl. form seems to be equivalent to the instr., and even parallel with the following instr. in *-ehi*. This is doubtless a reflex of the use of *ehi* forms as locs., § 7.30.

7.82. Loc. for ablative: alo mahānarake nirmuklāh Mv i.6.16, 'freed from this great hell' (cf. Senart's notes on this and the next); (for abl. of comparison) nagne vinagnatara te vicaranti loke LV 158.22 (vs), 'they go about in 'the world more completely naked than (lit. 'in reference to?') a naked man'; (with verbs of fearing) dandesu sămtrāsila tatra-tatra SP 94.10, 'they are frightened, in every place, of ('in regard to'?) clubs' (Kashgar rec. and WT dandena); ulpājane 'ham na bibhemi ... netradvayasya Divy 411.24, 'I am not afraid of the plucking out of my two eyes'.

7.63. Loc. for (dat. or) gen.: prānaharesv api lesu abhayam datlvā Mv i.132.12 (prose), 'having granted security to them, murderers tho they were' (indirect object); srutvā svastiku vāca nāyake suruciramadhurām LV 287.13 (vs), 'Svastika, having heard the charming sweet voice of (lit. 'in'?) the Guide' (possessive gen.; some mss. unmetrically nāyaka).

8. A-stems

8.1. For the most part the forms to be noted here are fairly simple and obvious products of MIndic, or semi-MIndic, phonology and morphology. They are arranged under the several cases, except that we shall first treat the use of the uninflected stem form for various oblique cases, and follow this with certain exceptional and un-Sanskritic forms assumed by, or substituted for, the stem in the prior part of compounds.

8.2. Attention may further be called here to the borrowing of endings from the s-declension (instr. sg. $-as\bar{a}$, nom.-acc. pl. nt. $-\bar{a}\eta(si)$, and to the sporadic use of characteristically feminine endings, borrowed from the \bar{a} -declension, in various cases (acc., inst., dat. sg.; nom, and perhaps voc. and inst. pl.). We shall see that fem. \bar{a} -stems show the converse of this, § 9.4. This is an aspect of the confusion between forms of different genders, on which see §§ 6.1 ff. In § 6.6 are listed the pertinent sections of this chapter on a-stems.

Stem in -a as oblique case form

8.3. In HJAS 1.66 ff. I showed that the stem in a is used in LV for any oblique singular case-form, citing inst., abl., gen., and loc. instances. In JAOS 59,369 ff. I pointed out that the same thing is recognized for Prakrit (Mähärästri and other dialects) by some Prakrit grammarians. I should have added that Hemacandra 4.344, 345 prescribes 'loss of ending' in Ap. for nom. and acc. sg. and pl., and 'generally' for the genitive.

8.4. For our dialect, more evidence, from a number of other texts, has now come to light. No doubt the phenomenon is commoner even than the following list suggests. I have included only cases where the form in -a could not, in my opinion, reasonably be interpreted as nom. or acc., or as stem in composition. There are many ambiguous cases where interpretation as an oblique case in -a would be quite possible.

8.5. It is evident that this development of a sort of generalized singular form, identical with the 'stem' used in composition, resembles the condition found in most modern Indo-Aryan vernaculars, which have for many nouns only one form in the singular, for others only two forms, one of which is used as 'inflectional base' before postposition's.

8.6. Only the ablative forms, which are found only in verses, may be regarded as shortenings of -d(l) m.c. All the forms are found mostly in verses but occasionally in prose.—My pupil Mr. Andrew H. Yarrow suggests that in the locative, -a might be a generalized inheritance from the Skt. samdhi-form of final -e before vowels (other than a-). It may be an accident, but it is true that these -a forms seem commoner as locs. than otherwise.

8.7. An alternative explanation for a few forms of this sort is suggested by Senart Mv i.499, note on Mv i.155.17 (vs), which Senart reads *bherimrdaiga yadi śańkhasahasraśabdam* (kuryāt) etc. Senart thinks we must understand *bherimrdaiga-śańkhasahasra-śabdam* as a single noun cpd., which is 'divided' by the word yadi. In fact, however, bheringdanga (read bheri°, m.c.) and śańkhasahasra are noms., parallel subjects of kuryál, and śabdam is its object. No such division of a cpd. as Senart suggests is possible, even in BHS. Moreover, even if it could be assumed, it would not suffice to explain quite a number of the following cases, such as the first.

8.8. Instrumental: ārogya dhig vividhavyādhiparāhatena LV 191.3 (vs), 'fie upon health, which is afflicted with all sorts of diseases'. Note the adjective -parāhatena, agreeing with ārogya; and the parallel dhig yauvanena in preceding line. Possibly sankalpa (q.v., Dict.) in LV 419.17.

8.9. Ablative: tal sādho puravara ila šīghram niskramyā LV 163.20 (vs) 'so, O saint, go forth quickly from this excellent city' (Skt. puravarād); ceļīya sakāša jāto KP 81.9 (vs) 'born from a slave-girl' (sakāša = °šāl); duķkhasamudra vimocayi satīvā Suv 51.15 (vs); anyatra karma sukŗtād LV 37.7 (vs) (karma, abl. of a-stem for n-stem, § 17.13).

8.10. Genitive: yāvaj jarāmaraņa cāntakarah prasūlah LV 47.20 (vs) 'and before he is born as Ender of old age and death'; mahāpadma yathodbhavah 97.15 (vs) 'as (there is) springing up of a great lotus'. So the best ms. A; both edd. mahāpadmo, which makes no sense. This is one of the cases interpreted by Senart (above) as a 'divided compound' (= Skt. mahāpadmodbhavah); suvyaktam jinaratna jambunilaye dharmākarasyodbhavah 109.4 (vs) 'clearly in the home of the jambu there is birth of the Jina-jewel, the Mine of the Law.' Note gen. of appositional epithet, agreeing with jinaratna. — duhkhāni samsāra acintiyāni Suv 61.9 (vs). Tib. renders as gen., 'the miseries of the samsāra.' — sattva (v.1. adds -m-, Hiatus-bridger) ašeşata nişiha tathaiva Bhad 46. Here sattva parallels nabhasya of the preceding pāda, and can only be gen.

8.11. Locative: na khalu punar bhikşavo māyā devī bodhisattva kuksigate gurukāyatām samjānīte sma LV 71.5 (prose) ('while the B. was in her womb'); tahi nrpa 166.6 (vs), all mss. and both edd., = tasmin nrpe; nidhi drsta (one ms. and Calc. drsti) yathā hi palāyati ko ci naro 323,11 (vs) 'as some man might run away when a treasure is seen'; drsta eva dharma 409.12 (prose) (so mss.; Lefm. em. foolishly, see Weller ad loc.; Weller would emend to dharme). — mūla My i.204.16 (vs); so mss., Senart em. müle, which is unmetrical; parallel ii.8.11 müli, which is possible; na tailabindu jinastūpa dattvā ii.385.15 (vs) 'not having given a drop of oil at a stupa of a Jina'; so mss.; Senart em. ^ostūpe, which is unmetrical; vāpiyodaka (v.l. vāpī^o; Senart em. ^odake, which is unmetrical) snāyitum (mss. snapitum) iii.12.5 (vs) 'to bathe in the water of the pond'; antara (mss.; Senart em. antare, unmetrical) tisthati iii.23.4. — kalpa anāgala bodhi careyam Suv 53.7 (vs) 'in a future age'. - nirvrtau ca sthili dharma yādršī RP 6.9 (vs) 'and of what sort is the situation in the state (of) nirvāna,' dharma and nirvrlau in apposition; druma puspaphalädhye 26.19 (vs) 'in a tree rich in flowers and fruit'. — tasmin saddharma kşine SP 67.8 (prose), 'when that good law had perished'; ksina can only be an adj. and saddharma a separate word; cf. the verse version 69.1 saddharmi (so Kashgar rec. but Nep. mss. °ma) kşiņe; unless we em. to saddharme, I see no real alternative to °ma as loc.

Stem final in composition

8.12. Several times in verses, a nom. sg. form in -ah or -o is used instead of the 'stem' in -a, as prior member of a compound. In some of these, where -ah is followed by a sibilant, the visarga could perhaps be interpreted as doubling of the following sibilant m.c.; cf. lengthening of a to \bar{a} in the same circumstances, below. Or, since as we shall presently see -u is also substituted for stem-final -a in composition, both -ah (-o) and -u might be analogical to the frequent variation of word-final, and especially nom. sg., -ah with -a and -u. Yet the occurrence of similar nom. sg. forms of u-stems in composition (§ 12.4), and of dual and plural noms. used similarly in compounds (see §§ 23.5, 6), inclines me to believe that we are dealing with the morphological peculiarity first suggested. See also § 23.7 for one or two sporadic cases which might be similarly interpreted, but permit an alternative explanation ('hiatus-bridging' consonants r and m, instead of nom. endings). aksayomati SP 447.2, 4 (vss); for aksayamati, so always in prose. — ākāśahsamatulyamānasā LV 325.16 (vs) 'with minds quite the same as ether'. -- Varalakşanahśiri (= Varalakşanaśri) Gv 284.24 (vs); Vajrāšavogiriśirī (= Vajrāšayagiriśrī) 285.13 (vs); Dharmahsamudra (divide thus) 285.16 (vs); same in longer cpds., see Dict.

8.13. Occasionally also a form in -u, evidently a transfer from the nom. sg. (which very commonly ends in -u as well as -a) by analogy, is used in verses as stem in composition, as in: Gandharvakāyu-prabharāja Gv 284.21 (vs), name of a Buddha; Samušarīru (for Samašarīraḥ) 285.10 (vs).

8.14. A neuter nom.-acc. form in -am is also sometimes used in the prior member of a cpd.: e.g. LV 429.8 (prose) prānālipāta-vairamaņyam-parasaltvasamādāyanatvād, and 9 °ņyam-guņavarņasamprakāfanatvād; 429.13 (prose) sattvasamgrahakaušalyam-sušiksitatvād, 'thru his being very learned in the skill (kaušalya; so Calc. reads, but all mss. °yam) of drawing-in (attracting) creatures (by the 'net' of the 4 samgrahavastūni).' No other construction seems possible, but perhaps the anusvāra should be deleted, despite the mss., three times.

8.15. Very common is the lengthening m.c., in verses, of a to \bar{a} as stem-final in the prior member of compounds. (Cf. Pali, Geiger 33.) A few examples, which might be multiplied indefinitely, will be furnished presently. There are, however, also cases in prose, at least well supported in the mss. Are these somehow related to the rhythmic lengthening of stem-final (and other final) a to \bar{a} in Vedic texts (Ved, Var. II, chapter XI, esp. § 466)? Or are they to be interpreted as nom. pl. forms substituted for the 'stem' in composition? See § 8.12. kämävacarā-devebhyah LV 83.6 (prose; all mss., confirmed by Weller's ms., Weller 23); (sarvarāgadveşamohadarpārativisāda-) bhayā-lobherşyāınātsaryavigatāh 86.5 (prose); so all mss.; Calc. °bhaya°; [asamkhyayā-kalpakoļinayutašalasahasraih 86.15 (prose) can scarcely concern us here; read possibly with 2 mss. asamkhyeyā- (which would then be an example belonging here), but more likely asamkhya- with two mss. and Calc., or possibly asamkhyayā as a separate word, inst. of a-samkhyā, without number';] kulikā-šalāni 95.8 (prose) 'hundreds of noble youths (kulika)'; cf. Weller 24; note that the same phrase occurs, out of place, in 94.3 reading kulikaśatāni; šakaļā-cakrāņi 381.14 (prose) 'cart-wheels'; all mss.; Calc. sakaļa-; sarvabodhipākšikā-dharmaratna° 424.12 (prose); so nearly all mss.; one ms. and Calc. "pāksika"; asaktābaddhā-muktacittatvāt 424.13 (prose); all mss. have -ā here, only Calc. -a; samgrahā-vaslujñānena 437.15 (vs, but hardly a m.c.; initial in anuştubh line); so both edd., no v.l.; the compound samgrahavastu is commonplace.

8.16. The adjective which in Skt. is *ratnamaya* 'made of jewels' occurs repeatedly in Mv, even in prose (e.g.

i.31.5; 32.3, 5, 10; ii.109.3 twice, 5), as ratnāmaya or ratanāmaya. Such forms have been noted in other texts only in verses, e. g. SP 11.4, 89.1, 91.5, 340.13 (here however not required by meter, and two mss. read ratna^o); LV 59.7. Cf. in Pkt. (JM.) -rayaņāmao Jacobi, Ausgew. Erz. 39.8 (prose).

Erz. 39.8 (prose). 8.17. Other cases, in verses, where ā is required or strongly favored by the meter are (a few among very many): dharmā-mukhā SP 53.5; šalāpadi 84.2; 86.1; candanāmayām 119.9; jinā-nayam 145.9. — chaltrā-patāka LV 43.17; divyācakşuşa (so, as cpd.) 108.16. — acalā-dhrlih Mv 1.98.11, repeated i.100.1; samāsamo = sama-samah ii.488.12. — narā-surānām Suv 24.5.

Nominative singular masculine

8.18. Nom. sg. -o. As in MIndic dialects, the ending -o, used for -as only before voiced consonants and a- in Skt., is generalized, and often occurs before other vowels, voiceless consonants, or a pause. In most texts such forms are recorded only in verses, but in Mv also very commonly in prose. A few examples (the Mv ones all prose except the last; those not from Mv are verses): māro sa SP 63.6; ratho i- 91.2; samarpito k- 111.6. — sabdo LV 80.15 (end of line, before pause); nirdhāvito k- 92.11. so prativišişlo (before pause); nirdhāvito k- 92.11. so prativišişlo (before pause); jino ca- 3.12; bodhisatīvo s-, and kālajīto samayajīto p- 4.1; bhāvayanto sugato c- 4.3; -bhirato 109.6 (vs, end of line, before pause).

8.19. Occasionally the writing -o is kept even where meter requires a short (§ 3.74): $smrlo \ samprajano \ Mv$ i.207.10 (vs); both -o's short. In repetition $smrla \ (mss. \ ^{0}t)$ $samprajana \ ii.11.21$. Read as separate words, nom. sg., not with Senart as cpd. Also in pronouns, e. g. ko ci SP 99.1 (vs); ko puni LV 151.8 (vs).

8.20. Nom. sg. -u. A shortening of this -o is to be seen in the ending -u, which is extremely common in most texts, but only in verses, and almost invariably in positions requiring a short; rarely in metrically indifferent positions, as at the end of a line. This ending is rare in Mv, and apparently never used in Mv prose, tho at least once it is found at the end of a line of verse. Cf. Senart's notes, i.478 and 537; my §§ 1.95, 3.51 ff. The ending is reported by Pischel 346 only for Ap. (where it is very common) and Dhakki. It is also common in the dialect of the 'Prakrit Dhammapada' (Dutreuil de Rhins ms.). See further under -u for -am, § 3.58. A few examples: kalpu SP 68.1; salkāru 93.3; utpannu 177.9 (initial in anuştubh, not m.c.). priyu LV 46.13; gajavaru 56.16; samayu 162.10 (at end of a line; no metrical ground for shortening; no v.l.). My (only in verses, and not common) esu My i.131.4; grddhu i.179.20; sthilu i.204.14 (but in repetition sthila ii.8.9); abhiratu ii.326.17; kavacitu ii.371.14 (Senart em. ⁹to; meter obscure to me); satkrtu ii.372.7 (v.l. ⁹ta); suruciru prašastu ii.372.12 (v.l. °ra, °ta); vipāku ii.393.21; vegajātu ij.399.5: -kovidu ij.405.6 (end of line; no metrical reason for shortening; is the form influenced by the synonym vidu? see Dictionary s.v., vidu). — utpannu Samādh 8.21; abhāvu 8.23; tāvantu, svaru 8.25, etc. (very common). - drstu RP 4.5; niketu 16.6; arthiku 17.1, etc. (common). - nisannaku Bhad 3; dhārmiku, pradaksinu 15; -smaru 16; pešalu, pariksayu 19, etc. (common). – - sthitu Gv 34.6; prasthitu 54.1; niścayu 54.2; -ghoşu 241.7 (common). — sadrśu Sukh 7.17; gabhiru vipulu suksmapraptu 7.18 (read sū°?); etc. (common). (Not noted in Divy.)

8.21. Nom. sg. - \bar{u} . Once we find - \bar{u} instead of -o as a metrical lengthening of the -u of the preceding section; cf. -a (for -as) occasionally lengthened to - \bar{a} m.c. (? § 8.24): mārgū visodhilu jinānām Gv 258.20 (vs). No other interpretation is possible; cf. line 25 below, mārgu visodhilo. Of course some might emend to mārgo. But the analogous forms in - \bar{d}

may support $-\tilde{u}$ indirectly. Both -u and (perhaps) -u attained enough general currency to stand on their own feet, so to speak, and to be lengthened to \tilde{u} and \tilde{a} when meter required a long.

8.22. Nom. sg. -a. Even commoner than -u is the ending -a for -as, as in Ap. and m.c. in AMg. Mg., Pischel 364. In our dialect it occurs almost exclusively in verses m.c.; even My seems hardly to show it in prose but almost only in verses, and even there not very commonly. At least twice, however, LV has prose occurrences: maraputra brahmä LV 44.11 (absurdly printed as cpd. by Lefm.; Calc. ^oputro; no other v.l. cited; can only be nom. sg.); mahāvaipulya (so divide) bodhi- LV 438.20. In two other prose cases there is seeming fusion of -a for -as with a following initial a-: madāpagatah LV 86.7 (for mado 'pa°) and dharmadhigatah 239.7 (for dharmo 'dhi°). The only case I have noted from prose in Mv is i.132.2, where all mss. prāpya khalv ayam; Senart em. prāpyo. For another erroneously alleged case in prose (LV 377.10) see HJAS 1.69,-Cf. the use of -a as oblique case-ending, above, § 8.3 fT.

8.23. The rest all come from verses. Only a few out of an enormous mass of cases are cited: saddharma SP 68.12; uccāra 85.14; vimukta 92.8; etc. -- samartha LV 29.3; nāsti satīva yo daridra 76.1; dvija 165.17; etc. ākānksamāņa Mv i.204.15 (in repetition ii.8.10 °ņā, Senart em. °no, unmetrically); sakra ii.54.2; artha ii.54.5; üditya ii.55.12 (separate from the following word); sukhila ii.371.20; niścaya iii.253.18; pralikūla iii.367.19. In all these Senart allows the -a of the mss., favored by the meter, to stand. In many others, however, he emends to -o (or sometimes -as), contrary to the meter; of course the reading of the mss, should in such cases always be restored. Such cases are: abhavya i.292.9; nrpa ii.13.8; avalīrņa ii.91.8; uccāru ii.325.15; sphula ii.336.6 and 374.3; sakya ii.347.17 (but in 348.2 and 6 Senart keeps sakya of the mss.); parivara ii.378.3 and 387.1; nisanna ii.408.15; jälyandha ii.409.9. buddha Samādh 8.6, 31 etc. — -samudra Suv 53.11 fl. etc. -- -citrita RP 3.6; gunasāgara 4.5; šakra 6.16, etc. padma ... sūrya Bhad 20; pravista 32; etc. —Also in pro-nouns, e. g. ya = yah LV 165.12 (vs).

8.24. Nom. sg. -ā. As we found -ù, apparently m.c. for -u (= -0, -as), § 8.21, so too, and rather more frequently (as in Asoka's language, with nouns and pronouns: Hultzsch lxiv sā; lxxvi etc., and in Ap., Tagare p. 27), -ā occurs as nom. sg., chiefly in verses where meter requires a long. Perhaps confusion with the nom. pl. is concerned; it appears that nom. sg. forms are used in the pl., see §§ 8.83, 84; or, as in some other case forms, confusion with (n. sg. of) fem. ā-stems. It seems to occur several times even in the prose of Mv, according to the mss.; anuvirājilašarīrā (aştā-) Mv i.38.14 where Senart em. °šarīro; yuvarājā abhişiñcilavyā 1.51.5 (Senart em. °lavyo; is final -ā here mechanically carried over from yuvarājā?); ayam evarūpo cetaso parivilarkā (Senart em. °ko) udapādi i.51.9. The rest are all from verses: apūrvadharmā śruņitavyu adya, āścaryabhūto hi tathāgatānām SP 308.11; the form °dharmā seems certainly sing.; three mss. odharma, but this is unmetrical. — asahāyu naro vrajate 'dvitiyo svakakar-maphalānugatā vivašah LV 175.10 (all mss. °gatā; Calc. galo); ksanā upasthilah 219.4 'the moment has arrived' (so all mss., this time kept in Calc., while Lefm. em. ksano). - LV 219.22 could be interpreted, with HJAS 1.69, as anuttarā yo bhava (= abhavah) rāga-sūdanah; but the Tib., which contains bar chad 'impediment', makes it probable that the true reading is anantarayo. -- ko gunā iha Mv i.69.7 (so, or gunah, all mss.; Senart em. guno); -kuśalā abhi° 1.122.6 (Senart em. -kuśalo); -satvā ili 1.122.7 (Senart em. -satvo); devasamghā k- 1.143.11 = 200.7 = 11.4.4 (Senart em. °samgho or °samghah each time; mss. samghā, v.l. first time ogha, or osamghāh; certainly nom.

sg. is meant). — yada nişkramî sa dupadendrā Gv 259.20 (so read) 'when that king retired from the world'; ed. padanişkramî sadu padendrā (l); abhyudgatā jinu 284.12 (can only be nom. sg.; name of a Buddha); leşa lābha paramā acintiyas 488.5.—Also in pronouns: saplarātra bhaņamānu gopike sā viyūha napi šakya kşepitum, yā viyūha abhu ... LV 237.13-14 (vs) 'tho described for a week, that pomp could not be exhausted, which pomp was ...' yā is read by all mss. (except one which reads svā) and sā by all but two; however, there is a fem. vyūhā, Dict.; yā (for ya = yas) icchatī ... LV 46.1 (vs, no v.l.).

8.25. Nom. sg. -e. The ending -e (§ 1.32 and fn. 11) scems to be rare, judging by the printed texts. However there are quite a number of cases in which the mss. of Mv show -e; but Senart always emends to -o; this time I have not taken the pains to collect many. The form ullare in SP 313.8, LV 364.18, etc. is probably an adverb; see Dict. anyatra sūri (so read with v.l. for ed. bhūri; nom. sg.) bodhisattva (nom. sg.) brahmakalpasamnibhe (no v.l.; can only be nom. sg.) LV 74.4 (vs). — sampratijāte sugate ... avaliş/he(l) Mv i.220.9 = ii.22.10 (vs), all mss. both times; Senart em. 'jālo sugato; narottame ii.193.19 (vs), mss.; Senart em. °mo. — Pronouns (see also se, te § 21.7); na me 'sti etādrša bhogu (v.l. bhāga, bhāgu) kecit (v.l. kimcit) SP 115.2 (vs) 'I have not any such enjoyment (or share)'. See BSOS 8.514; but a plural seems improbable. Cf. also §§ 8.28, 37.

8.26. Nom. sg. -am. In view of the very frequent interchange of masculine and neuter endings (§§ 6.1 ff.), the occurrence of nom. sg. forms in -am from stems which are regularly masc. hardly calls for special comment here. But influence of the acc. sg. masc. is also conceivable. It may be viewed as 'change of gender', or better, as signalizing a tendency towards breakdown of the system of formal gender distinctions, especially as between masc. and nt. Yet I shall cite one specially curious case: nāyam mārgam bodhāya Mv ii.130.13 (prose) 'this is not the path to enlightenment'. So Senart with all mss. Note the masc. pronoun ayam; to be sure we find ayam for idam, nt. (§ 21.82); but would it be realistic to speak of 'change of gender' here? The word marga (magga) seems to be recorded only as masc. in Skt. and MIndic. Could the ending (mārg)am have been mechanically carried over from the modifying pronoun (ay)-am?

Vocative singular

8.27. Voc. sg. -ā. Very common is the ending -ā. which is not recorded in Pali but is general Prakrit and is explained by Pischel (71) as due to pluti. Tho rare in prose (once in Divy mss.; otherwise no prose case noted outside of Mv. and few there), it is fairly frequent in positions where meter does not demand lengthening, notably at the end of lines or pådas in which short final vowels are otherwise frequent, as in all the verses of SP. The following examples, except those marked as prose, are all from verses, but none of them can properly be described as having -ā metri causa. (See HJAS 1.70 for mārsā in the prose of LV, erroneously taken as voc. sg. by Weller 44.) jināna ullamā SP 36.9; šalapuņyalaksaņā 162.12. — nāthā LV 12:10; in 234.11 ff., a long series of vocatives, nearly all in -ā, and many at the ends of lines, in Gopā's apostrophe to the departed Bodhisattva. -- jinasulā Mv 1.76.9; mahāyaśā 77.4 and 187.10; naravarātmajā 84.13; jinaputrā 126.6 (prose); -samnikāśā 156.19; puruşādhamā 185.10; māņavā ii.92.22 (v.l. °vah, but voc.); vānararājā (so mss., Senart em. °ja) 108.14 (prose); chandakā 165.14 (prose; here kept in ed.); acintiyā 296.4; putrakā iii.8.14; vīrā 124.5. — mahārājā (ed. em. ^oja) Divy 387.28 (prose). — RP (many cases) nāyakā RP 5.15 and 9.12; pāramingatā 5.16; narottamā 6.2 and 8.15; -vikramā 7.5; -dāyakā, dāntamānasā 7.11; others 8.16; 9.5, 9, 10, 12, etc.; in 50.9 ff. a long list, addressed to a Buddha, many at the ends of lines or pādas. — Gv (frequent) -maņdalā Gv 54.13; udāgatā 54.14; sambhrlā 54.15; purojavā 54.16; sudhīrā 230.14; suvišuddhanetrā 334.14; etc.

8.28. Voc. sg. -o, -u, -e. Furthermore, we find occasionally nom. sg. endings (-o, -u, perhaps -e) used as voc., at least according to the mss. Generally the editors emend. Only in verses; illustrated here by a few samples only. Cases of -o could be classed with -o m.c. for final -a, §§ 3.79 ff. sûru or jūru LV 171.8 'O herol', seems the true reading despite some variation in mss., and can hardly be anything but voc.; sārathivare 193.16 (vs) voc. acc. to Tib. and Foucaux; but perhaps loc. 'in (the company of) the excellent charioteer'; on citritake and susobhanake, LV 321.21-22, either voc. sg. m. or nom. sg. nt., see Dict. s.v. citritaka, -- (In HJAS 1.70, sahāyakas LV 235.7 was taken as voc.; but now I believe rather that the adjoining bhartu is nom. sg., and that sahāyakas agrees with it: 'your companion, my husband.') - śiriprabho Mv ii.235.5 (mss.; Senart em. "bhā; voc. of "prabha). — sudhano Gv 208.17 (can only be voc. of sudhana); sthiro (= sthira in prec. line; but 2d ed. sthirā) Gv 412.16.

Accusative sg. masc. and nom.-acc. sg. nt.,

8.29. These forms can be grouped together, since they are identical in our dialect as in Sanskrit. The classical forms, -am or -am, are very common, but in verses are not differentiated as in Skt.; the ending with anusvāra occurs quite standardly even before vowels for metrical reasons (when a long syllable is required), and -am only when the syllable is short. In absolutely final position the mss. regularly show -am; but many mss. of standard Skt. works regularly write anusvāra for final m. This in itself, therefore, cannot be regarded as conclusive evidence that in our dialect final -am regularly became -am, altho it can scarcely be doubted that it did (as in all MIndic).

8.30. The ending -u is common in the verses of most texts. It has not been noted in prose; and seems hardly to occur in My, where it is much rarer even than the same ending in the nom. sg. masc. One might be tempted to take it as a nom. (masc.) form used as acc., and then transferred to the nt. But -u is also found for any other final -am (e. g. ahu for aham; § 3.58). Hence, with Pischel 351, and H. L. Dschi, § 1.97, we must probably consider it a phonetic development of -am, -am. Apparently it is used only m.c.: āśayu SP 62.14; jñānu 93.10; dharmu 93.12; etc., common. - buddhakāryu krtu LV 48.13; dānu dallu 53.1; vişūdu 91.17; naru 168.16; etc., common. asavalu brahmacaryam Mv ii.299.14 (vs); so mss.; Senart em. asabala-brah^o. This is the only case I have found in my records from Mv, but it may be that I have over-looked some. — śāsamānu RP 7.12; avasthānu 18.9; hāsyu 28.3; pramāņu 54.22, etc. — abhāvu Samādh 8.22 and 23; krtu 8.22; āśayu 8.28; kālu, ciru 19.24; drstu 19.27; pravartitu 19.30; dukhu 19.32; cittu 22.26; nirdešu 22.35, 37; etc., very common. — pāpu Bhad 8; cakru anultaru 10; sameitu 12 and 61; nityu 16; pramāņu 45; agru 48; kşipru 49, 51, 53; sujīvitu 50; māru sasainyaku 53. — -dvāru Gv 55.2; -jāānu 122.14. — anantu Sukh 7.14. — Pronouns: etu SP 57.7; 97.4; 113.1; LV 287.12; imu Samādh 8.28; 19.27, 29.

8.31. The ending -a is, in most texts, commoner than the preceding. It is practically limited to verses; there are a very few prose cases in LV and fewer yet in Mv. There are, to be sure, a few other cases which, tho occurring in verses, are in metrically indifferent positions, so that they cannot be said to be metrically demanded.

8.32. The prose cases are as follows: dharma desitam LV 404.10 and 19, printed by Lefm. as one word; Calc. and

one ms. dharmam de° in line 19; vitāna samalamkrtam, gaganalala samalamkrtam 413.4; and dharmacakra pravartayatv ... 413.6-7. Certainly dharmacakra, and pretty surely vitāna and gaganalala, must be independent noun forms, not compounded with the following words as in Lefm. In 380.4 Weller 36 would read sukha as a separate word (text kadācil sukhaprāplam pūrvam); I suspect the true reading is sukham prāplapūrvam (misplacement of anusvāra in text). In LV 19.12 Weller 18 wrongly takes jāli, jarā, vyādhi, and maraņa as separate nouns; jāliprajādyale etc. are denominative verbs ('there is, arises, consciousness-of-birth', jāliprajād, etc.). — rddhiprātihārya (so mss., Senart em. °yam) drsįvā Mv i.266.17.

8.33. I have noted the following cases in verses where meter does not demand a short syllable: dlmabhdva prabhdsvaram SP 237.9; virya 254.3, according to ed. with Nep. mss.; but Kashgar version, La Vallée-Poussin JRAS 1911, 1073, rearranges the order in such a way that virya is metrically required. — nadikūla ivā savālukam LV 175.22 'like a sandy river-bank'; the adjective savālukam proves that -kūla has its usual neuter gender; but -kūlam ivā would be metrically just as good (even better, as avoiding hiatus); saukhya agra^o 76.6, as prec., but here Calc. saukhyam, and v.l. ⁹yam (unmetr.). — ddinava My i,184.14, all mss.; Senart em. ⁹vam; metr. indifferent; meruśriga a- ii.404.7 (as in LV 175.22, above). 8.34. In the rest meter favors or demands a short

syllable: śarira SP 26.8; dharma 177.3, 236.14; jñāna 302.6; etc., common. - apunya LV 42.11; sukha 42.15; sugata 46.9; -rasa (divide here) 165.6; etc., common.-In Mv Senart often lets the form in -a stand, but about equally often emends to -am, contrary to metrical requirements; I mark with an asterisk the cases in which Senart reads -am unmetrically; in practically all of these he has emended, tho a few times there is a v.l. supporting him. In Mv i.47.6 = i.81.15, read: (mānasam) jñānasāgar' avatāra nāyakā(h), 'a mind that penetrates (plunges; or, if causative, immerses) into the ocean of knowledge'. Further, -kāsiha i.68.19; suparigīla (or, with most mss., °nila) 70.6; yacanaka 84.6; samupacila 120.3; read vighna 156.7 with v.l. (m.c.); anantara 204.6 = ii.8.1* (adverb; note that Senart emends the second passage but keeps the mss. reading in the first, tho the two are identical); ādīnava i.359.6; rsivara ii.53.19; pratirūpa 92.22*; duķkha 93.16*; putra 133.15; 134.16; phala 134.4*; kuśala 220.14*; bhrngära 299.10*; ekämsa 320.15*; -sahasra 322.4*; sphuja 327.6, 7; nidhāna 334.2*; vimāna 334.10*; prāsāda 334.18*; rula 337.9*; sphula 344.1; cilla 368.8*; ulpida 370.5; dharma 371.13, 17; aparimila (separate word) 373.2; samgha 373.3; paijunya 384.8; grhila 396.13; prācīna iii.386.12. — trisahasra Samādh 8.22; adhyavasāna 22.4,6; etc.: (seems less common than -u here). - kāya Suv 18.4 (so read with v.l., text kāyam, unmetr.); sarva 24.8 (so read with v.l. for impossible sarvi; acc. sg. nt.); nilya 37.17 and 39.18; dharma 42.11; ksetra 52.2; sammukha 52.5; etc., common. - dharma RP 5.11; rāgadosaja 6.3; jihma 6.15 (separate word); mailra ... šīla 7.13; mārga ... ksema 7.16; śāthya 10.11; dharma śrestha 11.7; etc., common. - punya Bhad 9; pranidhāna 23 and 61; vibudhyana, darśana 35; viśisla 48; etc., common. --- Pronouns: ta buddhaksetram SP 149.14; eta SP 93.4, 7; Mv i.72.18.

8.35. The entire ending may be lost or fused with a following vowel: viry' *ārabhanti* (for viryam \bar{a}°) Dbh.g. 8(344).4.

8.36. The nom. sg. masc. ending -o seems to occur also in acc. sg. and nom.-acc. sg. nt. forms, at least in LV: ko 'tra javi ti vismayo janeyā ... karolha gauravo 'smin LV 152.6 (vs) 'who here would make astonishment (be astonished) that he is swift? ... pay respect to him!'; were it not for gauravo, which seems to confirm vismayo as acc., we might take ko ... vismayo as nom. and janeyā

51

4

as a syncopated opt. passive (cf. Pkt. vejja, lahejja, acchejja, Pischel 535), 'what astonishment would be produced?' Tib. seems to have had a quite different text; naişkramyašabdo 'nuvicārayanti LV 219.18 (vs) 'meditate on the word (sound) of (the Bodhisattva's) departure from the world'; Calc. śabdā but all mss. śabdo; see Dict. s.v. anuvicārayali; na ca mana (so all mss. in 165.16, but in repetitions manu) kşubhito LV 165.16 (vs, end of line); repeated 166.1, 6, etc.; nagaram vyākulu bhitatrastamānaso LV 193.10 (vs) 'the city was perturbed, with mind frightened and alarmed.'

8.37. The ending -e may possibly be used as nom. sg. nt. in *citritake* and *susobhanake*, LV 321.21-22; see Dict. s.v. *citritaka*. But they could also be voc. sg. m., § 8.28.

8.38. Acc. sg. masc.-nt. -ā, -ām. In HJAS 1.72, I cited svargamrtadvaram uttamam LV 178.8 (vs), very dubiously, as a masc. acc. sg. with the fem. ending $-\bar{a}m$, read by all mss. and Lefm., only Calc. uttamam, which I there found it hard to doubt was the true reading. Now I am less sure. Not only have I discovered many masc. and nt. forms used in the fem. \bar{a} declension (q.v.); but also a few other clearly masc. or nt. forms in -a or -ām, acc. sg., have showed up (and cf. -ā in the nom. sg.). It may be that these were actually used by our authors occasionally, and signalize the beginning of a breakdown between masc., nt., and fem. forms; cf. §§ 6.1 ff. vilumpatha imām LV 335.13 (vs) 'despoil him (the Bodhisattva)!' All mss. are said to read imām, and even Calc. has imā, glossing imam; läv api ca balīvardhau . . . šakaļā prakarsilau LV 382.1 (prose) 'those two bulls drew (§ 34.15) the cart'. Certainly sakatā cannot be abl. There is no v.l. in Lefm., but Foucaux, Notes 196, cites v.l. °fan. It is true that more than one cart was concerned in the story (five according to 385.14), so that conceivably sakata is meant as plural; this is not disproved by Tib.'s rendering śin rta, without specifically plural suffix. But would the text speak of a pair of bulls drawing (all) 'the carts'? - - asatpurusā Mv i.132.1 (prose), acc. sg., all mss.; Senart em. °sam.

Instrumental singular

8.39. Inst. sg. -enā, with lengthened final vowel, only m.c. in verses, common especially in LV. -varenā LV 57.21; -cireņā 78.5; vīryeņā 162.4; -ratenā 168.6.— Also in pronouns, e.g. yenā SP 53.2. In our language any final short vowel may be lengthened in verses m.c.; this is only one special case.

8.40. Inst. sg. -inā for -ena, as in Ap. -ina (Pischel 363; Jacobi, Bhav. 33*, San. 12). Rare and probably only in verses m.c. in nouns; in fact only one instance has been noted. In pronouns and pronominal adjectives there are other cases, including some in prose: § 21.14. mārdhinā Mv i.65.14 (vs), best understood as m.c. for mārdhena = mūrdhnā (which has taken on a-stem endings). Senart takes the *i* for an epenthetic vowel and derives directly from mūrdhnā; but only mūrdhanā or mūrdhnanā are found in the mss. at Mv i.67.11 (vs), where Senart also reads mūrdhinā; while a-stem forms like mūrdhena are well established from n-stems.

8.41. Inst. sg. -asā, as in Pali (Geiger 79.1) and AMg. (Pischel 364). Not common; noted only in Mv, but there in prose. The two forms recorded are balasā and padasā, both found in Pali, and the former at least in AMg. padasā occurs Mv ii.199.8; iii.115.11; 437.17; 443.17 (all prose except the first), and balasā iii.169.7 (prose). — According to Kern, SP Preface p. viii, balasā is found in the Kashgar version of SP (Nep. balātkāreņa). — Cf. the nt. nom-acc. pl. ending -āmsi, § 8.105.

8.42. Inst. sg. -ayā and -āya, and perhaps -ā, fem.

oblique forms, used as instr. masc., as conversely -ena is used in fem. ā-stems (§ 9.69). pūrvapakşayā Lank 348.12, m.c.; °pakşeņa metrically impossible here. There is no stem *pakşā. — prathamacittolpādāya āvŗtāni Mv i.104.4 (prose) 'blocked by the first origination of the thought (to become a Bodhisattva).' The mg. is clear and inst. certainly required; there is no fem. stem *ulpādā. In the plural there is some formal confusion between the endings inst. -ebhis (-ehi) and dat. -ebhyas (or representatives), but this seems hardly related to that phenomenon.— Apparently ending -ā: ūrņapāšā (v.l. ūrņakošā), LV 357.9, see Dict. s.vv. ūrņā(koša).

8.43. Inst. sg. -a, stem-form, see § 8.8.

Dative singular

8.44. Dat. sg. -aya and -āyā occur sporadically in verses as purely m.c. variants of the regular -āya. sarvasattvahitayābhiprasthita Gv 54.18 (vs) (for °hitāya + abhi°).—hitāyā (for hitāya) Dbh.g. 51(77).3.

8.45. Dat. sg. -aye. Almost limited to the Mv, but not rare there in prose as well as verse, is this ending; occasionally but rarely written $-\bar{a}yai$ in the mss. See HJAS 1.73 for a few forms in LV interpreted by Weller (44) as showing the ending -ayai; I still think the LV evidence inconclusive. I find it hard to understand the form except as a borrowing of the ending of fem. a-stems. The only reasonably clear case outside of Mv is: (cārakagahanagalān sallvān) samāśvāsayām āsa, bandhanam vipramoksāye Gv 353.19 (prose). If text is correct, the form can only be a dat. of purpose from vipramoksa. Cf. 362.14 below, in the verse account of the same incident, vipramoksaya nrpam yayace 'asked the king to release them'. - vipramoksäye Mv i.4.4 (prose); only one inferior ms. °ya; but in the next line vipramoksāya; tāye bodhāye (Senart bodhaye) i.60.11 (prose); bodhāye 80.7 (prose; so Senart here with 2 mss., 3 mss. bodhāya; repeated several times in the sequel with similar variations); bodhaye also 120.5, 7 (prose); samyaksambodhāye pranidhenti 127.1 (prose), all mss.; in l. 3 below °dhāya; ihāgamanāye (Se nart em. iha ga°) 226.3 = ii.29.6 (vs), all mss.; but parallel i.152.9 °nāya without v.l.; arthāye (= arthāya) with gen., 'for the sake of', i.233.11; ii.103.18; 158.12; 242.16; 251.1; 253.9 etc.; 425.16 (mostly prose); in cpd. darsanärthäye ii.35.19 (vs); darśanāye vandanāye paryupāsanāye (v.l. for each °nāya) 1.320.2 (prose); darśanāye ... paryupāsanāye repeated lines 4 and 6, without v.l.; gamanaye 1.362.15 (prose), quasi-infinitive; same ii.101.1 (prose); in i.361.21 gamanaya in practically identical phrase; samyagduhkhakşayāye ii.119.5 (prose); same passage in LV 239.15 kşayāya; jfānāye daršanāye sambodhāye ii.121.7 (prose); repeated in the sequel; same passage in Pali, MN i.241.8, has forms in -āya; abhinirvarlanāye 121.9; prahāņāye 124.1; daršanāye 127.6 f.; 128.11 f.; 129.13 f.; 153.13; 156.5; 449.7; iii.57.5; 6; 407.8 (all prose); alam ... gamanāye ii.151.8; 153.3; 155.13 (all prose); palāyanāye 172.8 (prose); piņdāye (once v.l. °āya) 198.4; 200.15 (both vss); moksāye 204.9 (vs); gamanāye 233.16 (prose); -jñānāye 284.7 (prose); dharşaņāye 320.8 (vs); -vijayāye 399.14; 400.3 (both prose); -vihārāye iii.48.12 (prose); -parichedāye 48.13 (prose); -hilāye 124.14 (vs); virāgāye, nirodhāye, śrāmaņyāye, sambodhāye, nir-vāņāye 331.5 (prose), repeated 8-9; ekamekāye hastāye 358.5 (prose); anugrahāye 374.9 (vs); pralobhāye 387.9 (vs).

Ablative singular

8.46. Abl. sg. -ā, MIndic for, -at, as in Pali, AMg., M., JM. Rare outside of Mv, but not at all uncommon there, in prose as well as verse. Meter, of course, cannot be concerned here, yet the few occurrences in other texts than Mv are almost entirely found in verses; ihere is only one rather doubtful prose occurrence: manusyān-

dhatvā jadā LV 19.19, for manusyā(h) andhatvāj jadā, which Weller 18 would put into the text by emendation; in view of the following j- this is not impossible.

8.47. In verses I have noted: anyalr' upāyā SP 46.12 'except for a means'; ajātatvā 281.6 (Kashgar mss. 'lvād); merutalā 'nupātayad 449.1.—nānyatra istamadhurā LV 42.19.—nacirā RP 38.4; duspathā 38.18.—likhanā Mmk 70.9; daršanā 70.10.

8.48. In Mv, however, the forms are common; the following list is far from exhaustive. Prose passages are marked as such: kuņapāto . . . narakotsadā Mv i.7.3 (prose) 'from the subordinate hell kunapa'; asipatravana 12.2; karmapratyayā 13.13; 16.5; talā 26.3 (prose); jīvitā 101.9; adinnā 145.9; kāranā 170.9; vadanā 171.5; 172.6.—In 204.7 = ii.8.2 the mss. readings, altogether, suggest to me (contrary to Senart) this text: vālam pi yeşa calilam śruniya krodhāt (or krodhā, abl.) samutpatati agnisamā, 'of whom even the wind (breath), when they hear something move, flies up from (as a result of) their fire-like anger.'-sukhasya ca prahānā duķkhasya ca prahānāt (!) i.228.8 and ii.132.4 (prose); kāraņā i.244.6 (prose) and ii.232.6; bādhā (mss. vādhā) i.295.1 'from plague' (Senart em. badly); -vāditā 326.16 (prose); uccasayanā 326.17 (prose); uparikosihakā 327.15, 16 (prose); prahmakāyikā devanikāyā (abl. with yāvad) 333.7 (prose); priyā 358.23; vacanā ii.102.18; 105.4; hastināpurā 112.14 (prose); ekolibhāvā 131.18 (prose); puravarā 134.9; bandhanā 181.21; dānā 183.19; -samucchrayā 197.2; kulā 199.17 (= Pali, same form, Sn 423); nivāpā, kālapāšokāsā (= °pāśāvakāśāt) 252.2 (prose); gagaņā 404.4; vijitā iii.7.15; -pralyayā 65.13 ff. (prose); āyurkşayā (v.l. āyuh°) 65.17; karmakşayā 18; manasikārā 332.13 (= °kārāt 16); etc.

8.49. Abl. sg. -a, 'stem form'; here perhaps by metrical shortening of the preceding ($-\bar{a}$ for Skt. $-\bar{a}t$); has been noted only in verses; see § 8.9.

8.50. Abl. sg. $-\bar{a}ta(h)$, $-\bar{a}to$, $-\bar{a}tu$. The regular ending of most Prakrit dialects, where it appears as $-\bar{a}o$, $-\bar{d}oc$; representing $-\bar{a}(t) + tas$. Note that it is not recorded in Pali, which uses the pronominal ending $-asm\bar{a}$, $-amh\bar{a}$. Most texts show such forms only in verses. The Mv however uses them constantly in prose, and regularly in the form $-\bar{a}(o)$, without regard to samdhi. Otherwise, in prose, I have noted only a series of compounds ending in *-grahāto*, Māy 220.13 fl., and repeated later in the same text.

8.51. The following are samples, a few out of many, from the Mv; all are taken from prose, tho many could also be cited from verses: talāto Mv i.6.11; kukkulāto 7.1; kuņapāto 7.3; kuddāto 25.14 ff. (repeatedly); tamāto, apāyāto, vinipātāto 27.8, 9; akţtapuņyāto 28.6-7; -dāruņāto mahāprapātāto 34.8; -madhyāto 40.10; devanikāyāto 55.15; dvipāto 158.13 (mss. vary, probably correct); sukhāto ... duķkhāto ... aiubhāto etc. 173.3 ff., mingled with forms like nityato line 2 (in some cases mss. vary); tuşitabhavanāto 196.21; 197.7, 10; padminīvanāto udyānāto 223.9-10; -yojanāto 231.14; anuhimavantāto 232.11; 284.11; sakāiāto 243.17; 245.16; 273.11; adhişthānāto 244.10; sopārakāto 245.3 (so read with mss.); luņdaturikāto parvatāto 245.11; bhadrakalpāto, dipamkarāto 248.6; sudaršanāto nagarāto 262.2, 3; amukāto °koļļarāto (text °koļa°) 273.15; prānātipātāto partivirato 326.14, followed by parallel formulas with ablatives some in -āto, some in standard -āt.

8.52. Otherwise -dio (diah, -dia, -diu) occurs only in verses. We should expect -dia, -diu to occur only where the meter demands a short syllable; but strange to say this is not universally the case. The various forms seem to be used interchangeably in such texts as SP and LV when the meter is indifferent, and also independently of samdhi; the initial of the following word has no bearing. Examples: asaingajñanatu acintiyatah SP 62.6; traidhátu-kāto bhayabhairavātah (Kashgar mss. °tu) 91.10; amukātu nagarātu 115.8; sadevakāto (end of line; so Nep. mss.,

except one sadevalokalo, which is also metrically sound; Kashgar mss. sadevakātu; ed. em. °kātaḥ, without reason), lokāt samārātu sabrahmakātaḥ (Kashgar mss., by lect. fac., samārāc ca sabrahmakāc ca) sarveşa satlvāna ca antikātaḥ (Kashgar mss. °kātu) 119.2-3. — nabhātu LV 194.11; sthānātu 194.22; saṃskrtātaḥ 195.12; 196.2; śayāto 230.11; puravarāto 235.10. — mārapathāto Bhad 20. — -balātaḥ Siks 3.17; 4.1. — -mandalātaḥ Sukh 51.13.

8.53. Abl. sg. -ato, -atu, with short a in the penult, also occurs, in the form -alo, in prose in the Mv. In this form it might be considered a standard Skt. form (Whitney 1098b), or not a great deviation from standard Skt., which however would hardly tolerate such phrases as *ābhāsvarād devanikāyato* Mv 1.52.6-7 (prose); yāvad abhiniskramanato i.157.15 (prose). Moreover Mv prose passages present -alo forms mixed with those in -ālo: *suddhāvāsato devanikāyāto* i.55.15; *nityato* i.173.2, and others like it, parallel with *sukhāto* and other forms in -ālo.

8.54. Not mingled with -āto forms, but with definitely local-ablative (not loosely adverbial) sense ('from', or 'as a result of'): vipākato Mv i.16.11; -kşayato 267.9; grāmato (v.l. -āto) 301.8. And others in prose.

8.55. Outside of Mv such forms have been noted only in verses and m.c.: naciralu LV 230.14; nacirato 237.10. — kāyatu Suv 49.22; rājakulato 232.4; nagaravarato 232.6 (mss. °purato). — puņyatu, jrānatu RP 4.4. — kāyatu Bhad 1, 8, 23, 43; rāgatu dvesatu 8; karmatu kiešatu 20, 46; cetanato 23; jrānatu rūpatu, varnatu gotratu 52. And -ata, in verses m.c.: ašesata dharmata Bhad 3 (separate word; ašesata often in Bhad, e. g. 46); asangata 10.

8.56. Abl. sg. -atta (for -attah), equivalent to -dia(h) above? (Cf. the abl. ending -atto cited by Pkt. grammarians, Pischel 365, end.) Possibly in LV 329.13 (vs), where the mss. readings seem to suggest original lajji hirotrapalla (abl.), 'being ashamed, thru modesty and bashfulness' ... The abl. construction seems confirmed by Tib. no tsha skyens nas (a cpd. abl. gerund) 'from being shameembarrassed'.

8.57. Abl. sg. -asmā, the pronominal ending, as in Pali; only in Mv i.104.8 (prose), where for text agārasthā anagāriyam pravrajanti we must certainly read agārasmā, the regular Pali form used in this formula. The usual BHS form is agārād (also, in Mv, agārasya, agārāto, etc.).

Genitive singular

8.58. Aside from the 'stem-form' in bare -a (above, § 8.10), I have noted only an occasional lengthening of the final a of -asya, in verses, m.c., as npasya LV 80.14; ratanasyā iva (probably read iha) ... yasyā (for yasya) 109.8; yasyā also 111.10; 113.17 etc.; lasyā SP 96.8; LV 111.11; 193.16; Mv i.113.2 (so most mss., required by meter; Senart lasya; in i.156.10 meter also requires tasyā, Senart keeps tasya of mss.); Divy 591.23; asyā LV 152.5. —The isolated maranāya (v.l. °nāye), gen., Mv i.165.8, has been noted in § 6.10, as perhaps a case of change of gender, f. for nt.

Locative singular (for locs. in -a see § 8.11)

8.59. Loc. sg. -i for -e (only in Ap. acc. to Pischel 366a, end) is common in verses of most texts, almost invariably m.c., that is in positions where meter demands a short syllable. It occurs only very rarely in metrically indifferent situations like SP 341.7 ksayakāli ca dešayet (second half of śloka line). Occasionally, however, -e is written, according to mss. and editions, even where meter demands a short $\cdot e$ (§ 3.64). On the other hand, at least once, Gv 253.5, a text presents -i (loki) in what appears to be a

53

4*

loc. sg., as if -i were lengthened m.c. to -i! But the meter of this line is troublesome and doubtful.

8.60. Examples of -i in verses m.c.: loki SP 64.6; 85.14: 92.12: 93.13; saddharmi 69.1; ākāši 87.12; etc., common.— prāsādi dharmoccayi ... simhāsani LV 27.17; tribhavi 46.1; modi prapta 53.16 (so, separate words); etc., common .-- In Mv rare, as Senart says in notes on i.42.16, 70.11; deśi i.70.11; loki 105.1; mūli ii.8.11 (parallel i.204.16 mss. mūla); loki ii.92.10 (v.l. loka); supini 134.11, 12; nabhi 344.5; avikali 346.4. - kalpi Samādh 8.21; -kandari sánti 19.35; -pūliki kāgi asāšvali, jīvili 22.4; etc., common. — loki Suv 40.8; 54.4; etc., common. — šīli RP 11.17; parilyāgi 12.12; jñāni 20.3; etc., common.— -agri Bhad 3, 28; kșetri 28; etc., common. - ekaromi Gv 34.5, 7, 9; 241.13; ekakşetri 34.11, 18. - arthi Dbh.g. 7(343).25. loki Sukh 7.17. -- kāle lathotsāhi krtam ca dānam Divy 405.8 (rare in this text). I take it that this stands for -utsahe, coordinate with kale. Less plausible would be a nom. sg. nt. of -ulsahin; the position of cu is against it, as well as other considerations. - triloki Sadh 16.1 (rare).

8.61. Loc. sg. -amhi, straight MIndic (from pronominal Skt. -asmin), = Pali -amhi (cf. Pkt. -ammi). Noted only once in a noun or adjective: nikhilamhi loke KP 129.7 (vs) 'in the entire world'. The same ending occurs with pronouns (*lamhi*, *imamhi*, §§ 21.21, 66).

8.62. Loc. sg. - amse, also a MIndic form of pronominal Skt. -asmin; cf. -amsi, the usual AMg. ending. Noted only once, in the reading of the mss. at Mv 1.268.4 (vs, end of first half of śloka line) stūpamse (cmended by Senart to stūpasmim).

8.63. Loc. sg. -asmin, -asmim, -asmi. The pronominal ending of Skt. transferred to nouns, as in Pali -asmim (-amhi; AMg. -amsi, -ammi). Occurs commonly in verses of most texts, also in the prose of Mv (but nowhere else as far as I have noted). But some texts, whose verses are in general normal BHS, do not show this ending frequently; thus it is absent from Bhad, and rare in Gv. Everywhere where -asmin etc. occur, the mss. present -esmin, -esmim, -esmi about equally often; in the critical apparatus of such editions as furnish any, there is often, perhaps usually, variation in the mss. between -asmin etc. and -esmin etc. (see § 8.70). The writing of final -n or -m is doubtless to be regarded as purely orthographic and immaterial; scribes who wrote -n no doubt did so under the influence of Sanskrit conventions. Once, at least, a text probably contained -m before an initial vowel (Mv i.207.14, vs, supinasmim asyā, read asya with mss. as in ii.12.4 and 18; Senart with v.l. "smim, which is metrically bad).

8.64. Forms in *-asmi* (as also in *-esmi*), without final nasal, usually occur only where meter requires a short. No sure occurrence has been noted in the prose of Mv, the only text where *-smi(n, -m)* forms have been recorded in prose. Nevertheless there are a few cases where the editions or mss. present nasal-less forms in *-smi* in metrically indifferent positions—at the end of lines or pâdas, or for example in the first part of anustubh pādas where *-smin* would be metrically just as good. Examples of such metrically indifferent occurrences are: *sāsanesmi* Mmk 97.23; *pathasmi* Samādh 19.17. In Mv ii.462.17, a prose passage, Senart prints *grāmasmi*, but one of his two mss. reads *grāmesmim* (or *grāme 'smim*), which is probably to be adopted.

8.65. Examples of -smi where meter favors a short: yāmasmi SP 25.14; dvārasmi 86.4; hastasmi 147.12; kālasmi 254.13; 255.12; 273.5 (in all three of these the Kashgar fragments, La Vallée Poussin JRAS 1911. 1073, 1076, support kālesmi, which is reported from one Nep. ms. in the first two passages).—grhavarasmi LV 50.9 (2 mss. -esmi); gagaņatalasmi 50.10; puravarasmi 54.9. svargakāyasmi Mv ii.193.7 (v.l. -esmiņ, unmetr.); jam budvīpasmi 366.3 (v.l. as preceding); lokasmi 367.8; drumavarasmi 407.19 (v.l. -asmim, unmetr.).

8.66. In the following metrical passages of Mv, Senart emends to *-asmi*, but *-esmi* must be read; meter requires short final: buddhakselresmi ii.350.4; svakārthesmi 409.20; gangālīresmi iii.186.15. And others like these. *--raņasmi* Gv 54.20. *--- kālasmi* Samādh 8.31 (2 mss. *-esmi*); bodhivaļasmi 22.5 (1 ms. *-esmi*); nāmarūpasmi 22.26, 27 (here the meter is obscure to me).

8.67. Pronominal forms in *-asmi* also occur, but have been noted only in vss where meter opposes the regular *-asmin*. So tasmi SP 68.12; 87.5; 1.V 75.20; Mv ii.388.3; yasmi LV 75.5; sarvasmi Mv i.166.12; svasmi Samādh 19.26; imasmi SP 116.8; 359.9; asmi ('smi after e, o) SP 31.4 (but prob. read lokesmi with WT); 44.11; 64.5; LV 50.14; 360.13.

8.68. Forms in *-asmin* or *-asmim* (which, as purely orthographic variants, can be grouped together) are common in the verses of most texts, and not rare in the prose of Mv. Very often, when the editors quote ms. readings, it turns out that some or all of them read *-esmin* or *-esmim*; see § 8.70. In the following examples, there is at least reasonable prima facie ground for accepting the *-asmin* or *-asmim* form. *aşlāpadasnim* SP 146.3; *nirvf-asmin* 253.15. — *nabhatalasmin* Gv 255.8. — *antikasmim* 29.3; *kalpasmim* 10 *asamkhyeye* 48.15 (prose); *pretasmim* 184.10; *naravŗsabhasmim* 237.4; *anavarāgrasmim* 282.6; *grāmasmim* 364.5 (prose); *nyagrodhasmim* 366.2; *vin-dhyasmim* 13.7.4; *lokasmim* 36.3, etc.

8.69. As a sample of the confusion regarding these endings as presented in the mss., note Mv (prose) i.312.8 prathamasmim (v.l. "esmim) divasasmim, but id. 10 dviliyesmim (so mss., Senart em. "asmim), 11 triliyasmim divasasmim (no v.l.), 11-12 caturthasmim (v.l. "esmim) divase.

8.70. Loc. sg. -esmim, -esmin, -esmi. There are however many cases where the prima facie evidence is equally strong for -esmim, -esmin (or -esmi, see § 8.66). This ending was never definitely recognized as a form of our dialect until HJAS 1.74 f. Senart (Mv Introduction i p. xvii) systematically excluded it from the text of Mv despite its very common occurrence in his mss., and altho he noted its occurrence in LV also. To be sure the Calc. edition of LV usually prints -e 'smin, as if the pronoun asmin were present, rather than a case-ending. This is often done in other printed texts. E. g. SP 114.2, printed palikuñcike 'smin; but note 10 recommends emending to ^okuñcikasmin. Nobel, in his ed. of Suv, e. g. on 142.7 (printed visayasmin), likewise assumes that the v.l. visayesmin could only intend visaye 'smin. As recently as 1938, Régamey in his ed. of Samādh (p. 17) says: 'As for the frequent endings esmin, I consider them, following Senart . . . as erroneous and awkwardly Sanskritized forms. I have corrected them into asmin in every case."

8.71. It is true that *-esmin* or the like seems not to be recorded in any MIndic dialect. But it is easily explainable as a blend form of the regular Skt. *-e* with the MIndic (and Skt. pronominal) *-asmin*; and there can be no doubt that it was a genuine form of BHS. It is far too common to be emended out of existence. And there are very many cases where reading *-e 'smin* (for the pronoun *asmin*) is impossible.

8.72. It is very hard to dissociate the list of forms in -esmin etc. from those in -asmin etc., because, as stated above, when editors favor us with citations of readings from a number of mss., it is apt to appear that some of them read a and some e before -smin etc. This has been illustrated above, and will be further illustrated below.

8.79]

It seems clear that both forms were extensively used in our language. The following samples are all cases where I think the evidence favors -esmin etc., and in many of them all mss. agree. All are from verses except those marked as prose; the latter are from My only. cuccaresmin SP 88.1, so read, apparently with all Nep. mss. (the KN note seems to quote cacaresmin); Kashgar rec. cu(c)care, unmetrical; tatra niveśanesmin 114,13 (with Nep. mss.; ed. with Kashgar mss. °asmin; tatra makes pronoun asmin implausible); anāsravesmin 26.5; bodhimaņdesmin 30.16 (only Kashgar rec. °asmin); sadevakesmin 127.11 (with Nep. mss.: Kashgar rec. °kasya; KN em. °kasmin). --- ekarathavaresmin LN 80.9; gaganesmin 81.3; kşititalesmin 153.10 (so all mss. but one, end of line; text °smi); dharanitalesmin 194.15; gehesmin 201.12; nabhesmin 233.16; puresmin 136.7; etc. In the last, one ms. has purasmim; in most of the cases cited all mss. agree. — yame ca pascimesmin Mv 1.4.9 (vs, printed as prose; all mss.); the mss. also obviously intend madhyamakesmim in i.4.8 (prose); samghatesmim 13.11 (with 5 mss., one °tasmim); rauravesmim 14.5 (with 5 mss., one °vasmim); lapanesmim 14.15; rsivadanesmim 43.15 (5 of 6 mss. end -esmim; meter requires °smi); grddhakūfesmim 193.8; vyākaraņesmim 241.14; paralokesmim 270.10; lokesmim 281.5; -śāsanesmim 293.6; rsivadanesmim 337.11; anosaresmim 365.5 (prose); sariresmim ii.34.17; hastināpuresmim 105.9; sopānesmim 172.9 (prosc); udgatesmim 194.3; dharanitalesmim 343.14; yuddhesmim iii.20.13; antariksesmim 96.15; 100.16; dhavilesmir. 184.8 (prose); daksinesmim 307.7 (= LV 389.6, same pessage, same form); paścimesmim 308.7 (Senart with v.l. °asmim; = 1.V 390.4 where no ms. has °asmim); rājagyhesmim 440.4. -- mahāvanavaresmin Suv 226.11; Nobel reads °vare 'smin, but the pronoun is implausible (cf. preceding tatra); tatra mahāvanesmin 236.6 (Nobel °ne 'smin). — grāmesmin Siks 46.9; Bendall interprets as grame 'smin 'in yon village', which seems to me nonsense in the context; occurs in a passage giving purely general vows of Bodhisattvas; -sāsanesmin 305.3, ms.; corresponds to Mv ii.388.1, where śāsanasya is read; gagancsmin 345.14, ms. (ed. em. gaganasmin; pronoun (a)smin quite impossible). — bhojanesmim Mmk 104.7, bhaktesmim 9. (Such forms seem to be rare in Mmk.)

8.73. Some forms in -esmi have been quoted above under -asmi (§§ 8.66). They do not certainly occur in prose, even that of Mv; and usually, at any rate, meter demands a short final syllable. Other examples are: buddhaksetresmi Mv 11.350.4; svakārthesmi 409.20; gangātīresmi iii.186.15. -lokesmi KP 38.6; printed loke smi, but (a)smi(n) is scarcely possible here, and is certainly not represented in Tib. -śāsanesmi Mmk 97.23. I believe kālesmi is probably to be read in SP 254.13, 255.12, and 273.5, rather than kālasmi of ed.; in the first two a v.l. kālesmi is quoted (from only one ms. to be sure; but the critical apparatus of KN is notoriously unreliable), and the Kashgar fragment quoted by La Vallée Poussin, JRAS 1911.1073, 1076, reads kālesmi all three times.

Nominative-accusative dual masculine

8.74. Nom.-acc. dual -o (?). Occasionally the mss. of Mv show forms in -o, possibly standing for -au, dual. Senart regularly emends to $-\bar{a}$ (plural); and indeed plurals generally prevail, for duals, in the context. E. g. vayam (referring to a pair of persons) elasya agraśrāvako (all six mss.; Senart em. °kā) bhaveyāma Mv 1.44.8 (prose).

8.75. Nom.-acc. dual -u, m.c. for -au (-0). Rare; I have recorded only: antu Samādh 19.32 (vs) = antau, 'two extremes'; here clearly m.c.; but there is a variant anta; note modifying pronouns ubhe eti (= ete), dual neuter (or acc. pl. masc.? see § 8.77); and ye imu dāraka dvau mama putrau Suv 52.7 (vs); also m.c. but the majority of mss. ima (see next).

8.76. Nom.-acc. dual -a. A few times -a occurs in verses, referring to pairs, and associated closely with dual forms. See Suv 52.7, just quoted, with $d\bar{a}$ raka (and possibly ima instead of imu?). Also: candrāsūrya nabhātu būmi patitau sajyotisātamkrtau (the two participles modify candrāsūrya) LV 194.11 (vs); hastau chinna talhaiva chinna caranau 194.13 (vs). Obviously meter requires a short final syllable in these forms. In HJAS 1.75 f. ! have taken them as metrical shortenings of -ā, the plural form used for dual. Certainly it would be an error to connect them with the old Vedic dual ending -ā. There is considerable use of plurał forms for dual, and mingling of both in the same passage, in our dialect.

Nominative-accusative dual neuter

8.77. -i for -e: cli, m.c. for ele, occurs Samādh 19.32 antu (v.l. anta) ubhe api eli jahilvā, cf. § 8.75. The parallel ubhe suggests that it is duak. But since plural forms are often associated with duals, it might be interpreted as acc. pl. (masc.).

Nominative plural masculine

8.78. Nom. pl. -ā, without regard to the following sound, and even before a pause. The regular Mindic ending, very common in verses in most texts; also in prose, commonly in Mv and sporadically (but rarely) in some other texts. Most editors deal with this very inconsistently, sometimes allowing -ā to stand, but sometimes emending to whatever Skt. samdhi requires for theoretical -as. Prose examples are marked as such. -śrāvakā SP 93.9; durbalā 95.9 (in these two ed. em. -āh against all mss.); kāņā 95.6, daridrakā 9; -kāyā 97.12; -gulmā 127.2; trptā 222.12, sukhilā 13; anarthikā 273.11 (no case noted in prose). — visarjilā sma LV 67.22; -pramukhā sva- 159.17; yathartukā prā- 186.9; in fusion with following a-, manusyāndhatvā 19.19 (for °syā an°); pañcakāpi bhadra- 408.22 (but here best mss. omit api); (all these from prose, where it is rare; the following are from verses) -gandharvā 11.19; -opapelā 29.4; višālaprajītā 29.11; isļidārakā suduķkhilā (so divide) 74.15; etc. - Mv, very common in prose and verse: parvalā prajvalitā ādīptā sajyotibhūtā i.6.13 (prose; end of sentence); kulalā ca grdhrā ca kākolūkā ca 7.4 (prose); paribubhukșită sma 8.2 (prose); duratikramă 9.10; -mrakșită 12.1; devā 30.5 (prose); dešilā 73.13; -ādhyāšayā 77.6; -sadršā i.151.3 = ii.27.14; koliyā šākiyā ca iii.93.20; etc. buddhadharmāpy acintiyā Divy (rare) 79.21 = 469.5 (vs; = °dharmā apy acintyāh). — RP (not in prose but common in vss, and usually, but not always, em. by Finot): -mānasā RP 10.10; vimohitā 10.17; sthitā 11.1; prayuktā 15.3; svasutā 21.15; -parigrddhā 29.14; ulsadā 35.11, etc. (in the 1st, 3d, 4th, and 5th emended in ed., not in the others). ramaniyā Samādh 19.21, etc. — etādršā Suv 37.1; read paņavā 40.1 (ed. em. °vāh); aciniyā 62.16.—Av, not uncommon in ms., usually but not consistently em. by Speyer; e. g. in a set of verses ii.136.10-137.8, he leaves palikrtā (with a 'sic') 136.10 but adds visarga to -sabdā 137.1 and to mandaghoşakā 137.3.

8.79. Nom. pl. -a, as in Ap.; mainly in verses m.c., metrical shortening of the preceding. Often the final vowel is fused with an initial vowel of the next word; it makes no difference whether we regard these forms as containing $-\tilde{a}$ or -a; e.g. saturcha SP 86.10, etädrisätra SP 86.13. The editions frequently print forms in -a as if compounded with a following word, even when this is manifestly impossible. On the whole the mss. of most texts are pretty reliable in writing a short -a where meter requires it; but some are unreliable, and those of Mv cspecially present -a not infrequently where -a must be

read (malla i.66.4, read malla; pralisthila 71.4, read °la; vighnakarāh 156.20, Senart prints °rā but notes that it must be scanned °ra; etc.). However, some even of the best editors actually do violence to meter by emending to -ā where the mss. have correctly -a. Some such errors by Senart will be noted below. vimukta SP 92.9, mukta 10; kālaka 94.7; etc., very common. — trpta LV 37.3; aksana apāya (śodhilāḥ) 53.6; nagna 74.16; jina 164.21; etc., very common. — lokanātha Mv i.69.21; anāgata 70.1; devasamghu 72.20 (so read with 1 ms., the rest °gho or °ghā, Senart the latter); skandha 73.8 (so correctly all mss., Senart em. °dhā, unmetr.); kāma 73.13; samartha 155.16; ye dustacilta i.204.9 = ii.8.4 (in the first passage only Senart em. °cillā, unmetr.); -sampha i.204.10 = ii.8.5 (in the first passage only, Senart em. °gho, unmetr.); roşaka ii.52.22 (read as separate word); lokapāla 134.2: devarāja 135.1; saltva 333.18; 348.11; asva 337.15; hata 342.16; srestha 375.21; pūjila 387.18; prahrsja 415.1 (mss., Senart om. °/ā, unmetr.); prasthita iii.141.6; sphuta ii.350.15 = iii.274.17 (in the former passage sphulā is read, tho metr. inferior): duhšila ii.355.18 = iii.280.3 (here miss. thus both times, Senart wrongly em. °lā the first time only); and many others. - brahmasvara Suv 23.4; saltva 23.11, 15; upapanna 24.9; nirbhaya 39.6; etc., common. — vilakşana Samādh 19.16; -gala, yamalokika deva 19.17; nara 19.28, 31, 32; etc., common. - buddhasuta RP 5.10; dharmakośadhara 6.10; bāla 10.17; dānta sānta 11.2; etc., common. pūjila, alīlaka (Gv 544.7 °ku), anāgala, pūrņamanoralha Bhad 13; pariśuddha 14; pāpaka 19; etc., common. śrunamāna Gv 34.18; bodhisatīva 34.19, 57.17; rašmimegha 241.3; etc., common.

8.80. Nom. pl. -e, the pronominal ending, transferred to nouns. Not recorded in MIndic for the nom. (but regular in acc. pl.), and only rare and sporadic here. Chiefly in verses. CL next paragraph. *le saniskāre na santi tattvatah* LV 176.13, 'these compounds do not exist in reality.' Cf. also chidyanti anga-m-ange 223.15, 'various limbs were cut off'; so (except chidyanta) Lefm. with his ms. A; the other mss. metrically impossible. While anga is regularly nt., we have here the masc. (pronominal) ending. — alitārthe Mv i.122.18 'having missed their aim'. So all mss.; Senart em. °ādhve. — na te virājite RP 6.16 'they are not resplendent.' Finot's note suggests reading virājate for virājante, which is unnecessary aud implausible.

8.81. Nom. pl. -i, metrical shortening of the preceding. For the same form in nt. nom.-acc. see § 8.103, and in pronouns (*li*, eli, yi etc.), see § 21.25: $k_{say}antadharmi sarvi$ bhāvu LV 173.3, 7, 'all states of being have a nature thatends in destruction.' — mānı ca (read co or cā, m.c.)apaśyanti janenti trşnām SP 324.2 (vs); note suggestsemending to apaśyanta, as nom. pl., 'and not seeing me,they generate ardent desire (for me).' But the form of allmss. may be kept, as nom. pl. pple. (cf. § 18.19) or even asfinite verb, 'and they do not see me' (Renou, Gr. Scte.p. 175).

8.82. Nom. pl. -āyo, m.c. -āya, fem. nom. pl. ending transferred to masc.: *le ca tatra kukkule dahyamānāyo janā* pradhāvanti Mv i.6.16 (prose); hetubalopagatāya mahātmā Šiks 4.5, 6 (vs; cited from Ratnolkādhāranī). Can hardly be anything but nom. pl. masc., unless we emend.

8.83. Nom. pl. -ah, -o, nom. sg. ending used for nom. pl.; cf. next paragraph. Confusion between these two case forms results from the fact that both have come frequently to have the same ending -a, as in Ap. Sporadic, and chiefly in verses; also in the prose of Mv. jarā ca vyādhi mrţyu enti dāruņo mahābhayā LV 173.5, 'age, disease and death go on, cruel and terrible'; all mss. dāruņo; Calc. dāruņā; ima evarūpās luriyebhi šabdah 183.11; so all mss. (Calc. śabdāh); the modifiers show that a plural is meant; hā dhik priyair viyogo nājarangasvabhāvasamibhā anilyā 235.16; here either viyogo must be pl. or the following two adjectives singular. — nirmukto kukkulam avagāhişu Mv i.11.1; so 3 mss.; two nirmuktoh, one °lah; none °lāh which is Senart's em.; advārakā gharo (all mss.; Senart em. gharā) praliyaltā bhavanli i.24.13 (prose); lenānivarliyas (3 mss. °layas; Senart em. °liyās) lesām adhyāšayā ili smŗtāh i.87.12; nirmito bhikşuh i.192.10, all mss.; certainly pl.; Senart em. °lā bhikşūh; ime... āšramo (all mss.; Senart em. °me) udake samcaranli iii.150.13 (prose). — atra bāla abudho vimohitā RP 10.17. Or is abudho derived from the stem abudh, recorded in Upanişads? Not likely.

8.84. Nom. pl. -u, metrical shortening for preceding; once, however, if the text (Šikş 347.4) is right, not m.c., tho in a verse. — $bh\bar{a}vu$ (see § 8.81) LV 173.3, 7; no v.l.; *tyaktu tyāga* (so, separate) *dustyajū* 170.7; here ms. A *tyakta*, but below, 170.14, all have *tyaktu* (*dustyajā tvayā*). — bhāsilu dharma tena bahuprakārā vividhā anantāh SP192.3 (?); certainly nom. pl., but bhāsila is the reading ofthe Kashgar rec. and one Nep. ms. —*jinu*Gv 34.11; canonly be n. pl. — Šikş 347.4 (see Dict. s.v. palāta); if thereading is correct, palātu asurendrāh can only be for palātau(= palātāh) asu^o. The form (with -v in samdhi) is sobizarre that 1 am inclined to emend to*palāt*(for °*iāh*) asu^o.

8.85. Nom. pl. -ān, -ām (the difference is ortho-graphic and of no consequence). The acc. pl. ending used as nom.; a reflex of the general MIndic confusion between the two cases. In verses only; sporadic, but hardly to be disposed of by emendation: ramanti tan kridanakapramaltān SP 86.6 (vs; so mss.; KN em. te ... °ttā, so WT without note); adhyeşanām devaganasya tūsņīm, agrhņa devān adhivāsanam cu LV 7.13, 'I silently accepted the request of the throng of gods for instruction, and the gods (accepted) my assent;' all mss. devān; only a nom. pl. is possible; -n- possibly samdhi-consonant (§ 4.65); hayagajarathapaltīm (v.l. °tlī) sainya śrīmadvicitrām, dvāri sthita nrpasyā śrūyate uccayhoşāh 80.13-14; no acc. construction is possible; the forms in -am and -i(m) certainly intend noms.; all mss. and both edd. -vicitram, which (as well as sthita!) Calc. note construes as gen.; abhinadita manojñām (v.l. °jñā) hamsa kroncān mayūrān (v.l. °rāh) 80.21; all forms can only be noms.; pravarșamānān 220.3, all mss.; Lefm. em. °nä; can only be nom.; in LV 357.3 read °sa-hasrāms c' osarī(r) ūtmabhāvā, 'and hundred-thousands (of rays) came forth from his body(?)'; here all mss. are said to read the acc. pl. form; it could be construed as nom., subject of osari(r). But more likely the latter means 'sent forth' (the Buddha being the subject), with osari for osiri or osiri; see Dict. s.v. avasirati (3), which is used in this sense; the spelling with a for i occurs elsewhere in mss. sarve le- niravašešām (v.l. °şam; Senart em. °şā) abhūd buddhasya śrāvakān (Senart em. °kā) Mv i.250.20.

8.86. Nom. pl. -āni, the nt. ending, with masc. nouns (see § 6.4): -putrāņy ... udyatāni Mv 1.155.15; putrāņi mahārājasya sadršāni iii.167.3; bahūni bālāni caranti loke ii.92.7; yāni vānarāņi ... okaļļitāni (mss.) iii.29.17, te vānarāņi 18; pādāni 30.11; košāni 42.9 (but koša also nt. Skt. Lex.); masakāni (= maš°) 101.10; tāni mrgapaksiganāni saņtrasanti 146.17; (sahasram) dīpavrksāņi niši pradīpyanti 177.7; krlāvakāšāni vadetha yūyam 368.10, 'do you (ascetics) speak, being granted opportunity'.

Vocative plural

8.87. Voc. pl. $-\bar{a}$, -a. The endings $-\bar{a}$ and (chiefly m.c.) -a may, of course, be used in the voc. as in the nom. pl.: $y\bar{a}dr\bar{s}atatlvabh\bar{u}t\bar{a}$ LV 47.5 (end of a line). — kulaputra SP 86.9 (addressed to a group; not to only one member of it, as suggested by Kern, note to Transl.); $m\bar{a}rsa$ 175.3. $m\bar{a}risa$ Mv iii.292.10; in a verse, but at the end of an anuştubh line, hence not m.c.; yet certainly plural; $m\bar{a}risa$ iii.334.14, seemingly plural. **8.88.** Voc. pl. -āho, a special MIndic voc. pl. ending. Taken by Lüders, in Hoernle MR 162, as of Mg. origin and so as evidence that the protocanonical Pkt. was Mg. In BSOS 8.512 I have pointed out that it is to be identified with the Ap. ending -aho, -ahu. It is not limited to verses but occurs also in prose, not only of Mv but of SP, at least in the Kashgar rec.: kulaputāho SP 253.1, 255.11; acc. to Kashgar rec. *abusāho* 378.1; kumārakāho 73.6; kulaputāho 270.12; all prose; ed. with Nep. has forms in -ā(h). — amareivarāho LV 47.5. — (mostly from prose): vāsisthāho Mv i.38.4; 257.13, 15, 19 etc.; amātyāho ii.436.12; 437.2; kumārāho 436.18; 437.8; bhavantāho (mss. °toho) 479.15; jambūdvipakāho (69.8 °pikāho) udyānavarāho (79.11 adds ramyāho; in 79.11 mss. °dvipakāhā, and so v.l. in 83.3) iii.69.8; 71.13; 79.11; 83.3; viparyastāho 292.10; 293.12; 294.4; mānavakāho 378.19 (v.l. °kā). Cf. §§ 9.100.

8.89. Voc. pl. - $\bar{a}vo$. In Mv i.317.15, 16 occurs $\bar{a}vus\bar{a}vo$, voc. pl. (see Dict.); this rare ending is otherwise known only with fem. \bar{a} -stems as nom.-acc. pl., and may be borrowed from them (§ 9.93), like nom. pl. - $\bar{a}yo$, - $\bar{a}ya$ above, § 8.82.

Accusative plural masculine

8.90. Acc. pl. -ām, -ām. The writing -ām for -ān is very common in the mss. (and editions) of most texts, both before consonants and before pause. It is hardly necessary to cite instances, but some from SP and LV are given HJAS 1.78; from Mv, i.153.6 kāmām, i.115.16 pādām. Somewhat more noteworthy are spellings with -ām before a vowel. These may perhaps be regarded as representing -ā (next paragraph) plus hiatus-bridging m. E. g. äyuhsamskäräm ulsrjanlänäm Mv i.125.19 (vs; 2 of 6 mss. °ram, Senart em. °ram; Pali äyusamkhära is regularly plural); ślokām imām (Senart em. ślokān imān) agāsi i.130.6 (prose); susamyatām (Senart em. °tām) rsayo darśanāye iii.366.20 (vs). — -pratyekabuddhām abhibhavati KP 84.5 (prose). - vanditva pādām amitaprabhasya Sukh 50.6 (vs). — yām (= yān) atītajina yām (= yān) anāgatā ... bhāskarāms ca yān (text erroneously caryān) Gv 55.7 (vs). — ya imām evam bhadrikām sūtrānlām pratiksepsyanti LV 88.14 (prose; no v.l.; cf. 89.7-8 imān evamrūpān sūtrāntān); nirmitām abhinirminvanti sma 350.19, probably 'they create-by-magic magic personages'.

8.91. Acc. pl. - an, m.c. for -ān: sarvāš (read °vāmš?) ca drsti- (should read °tī, m.c.) -gatan utsrjam[ti] KP 18.8 (vs).

8.92. Acc. pl. -ā, fairly common, chiefly in verses; but also in prose of Mv, and occasionally elsewhere, it seems. It is best regarded as the nom. form used as acc.; cf. the following paragraphs, and the acc. ending used as nom., above. buddhā ca bodhim ca (read na with WT?) prakāšayāmi SP 47.12; istāmayā (all mss., ed. em. °yān) 50.9; cakravādā 355.5. — anātmā niriksathā yoniso imā dharmā LV 37.12 (vs); varņā guņām (so read) 47.11; and in prose, län ... aprameyäsamkhyeyä gananäsamalikränlakalpälikräntän buddhä (all mss., Lefm. em. °än) bhaga-vanto 'nusmaranti sma 4.7–8 (cf. Weller's note, 15, which rightly suggests taking °samkhyeyā as a separate acc. pl. but fails to note that the mss. have likewise buddha); aprameyāsamkhyeyā gaņanāsamalikrāntān bodhisattvān 30.22; possibly šakatā 382.1 (§ 8.38). — Mv (very common, in prose and verses alike; the following examples all prose): daša kusalā karmapathā ye hi samādāya vartanti i.3.1; adrākşīl salvā narakeşu anubhavalā (so with mss.) .. ūrdhvapādā adhoširā ... 5.3; tām nairayikā 17.9; bhītā 26.15; a series of accs. in -ā 30.2; dharmā 61.4; vašibhūtā 74.21; etc. - saltvā Suv 51.15 (vs). In this and most texts common in verses.—In Divy rare; but mandilakā paktvā 258.9, mss., ed. em. °kān, which the mss. present later, lines 12, 14.

8.93. Acc. pl. -ās, -āh; the nom. form used as acc., cf. the preceding. Not limited to verses; not exactly common, but the examples are numerous enough to establish the usage. Those cited are from verse except when marked as prose. -pūrņāh SP 9.3; praskanda samsāri niruddha durge magnāh punah (so both edd.!) duhkhaparamparāsu 54.8 (the three adjectives all agree with sattvan, preceding; KN note calls them 'ungrammatical'); sarvāś (so all mss., KN em. sarvā, which WT keep without note) ca cakravādā sa paśyali 355.5. - dravyāmbarāś (Calc. °rāmś, without mss. authority) ca puruşān LV 42.21; deva paśyi mānusāś ca mānusā amānusām 75.15, 'the gods looked upon men, and men upon demons'; tyaji tvayi ... dhanamanikanakāh 165.9 (so all mss. and Calc.; Lefm. em. °ka); buddhadharmaś (all mss. and Calc.; Lefm. em. °māņs) cāmukhīkaroli 180.5 (prose; not noted by Weller). — apsaravargās ca (praligrhņa) Lank 6.5; ed. suggests em. to °gāms. - drumā pasyanti śobhanā, harilān (mss. write °lāṃ) palrasaṃchannās lān ayanti ... Mv i.11.9-10; note variety of acc. pl. forms; -ā twice in line 9, -ān twice and -ās once in 10; badhyās (Senart em. °yāms) ca samgopāÿanti i.96.8 (prose); -dharmā usādhāranāh (acc.; Senart em. °ņā) pra- 142.5 (prose). parvalacakravādapariyantāķ (calvāra dvīpa ... ānasati sarvān; 2d ed. 'sarvā; can only be acc.) Gv 254.18 (vs, end of line).

8.94. Acc. pl. -a, shortening of -ā (see above) m.c.; only in verses, but common there. Sometimes, when a vowel follows, a hiatus-bridging consonant (§ 4.57 ff.) is inserted: SP 86.5 svaka-m-atra putran (so all Nep. mss.; Kashgar rec. different). aśva edakān SP 10.12; dvipa lathaiva 11.10; sampidila 48.10. — jihma vipasyatha divya ālmabhāvām LV 49.12; priyasula 165.10; guņa 167.3. daśāngula Mv ii.8.11 (so mss., Senart em. °lam, unmetr.); bodhyang sapla 324.2; vimāna sarvām 333.10; pašyilvā stūpa buddhānām 370.17 (so read, cf. ms. C); kalpa 370.21; in 393.19-20 read with mss. (line 20) nirmalya (for °yam of text) so apanaye; kula 395.13 (or neut.?); aśva iii.23.20 (mss., Senart em. aśvā, metrically inferior). - jina sarva Suv 45.10; etc., common. - varșa acinliya Samādh 19.24. -sańgha Gv 236.2; etc., common. — rātriprašānta ... saltvān) RP 3.13; sattva 10.12; etc., common. — jina Bhad 7, 25; kalpa 22; etc., common. 8.95. Acc. pl. -e. The regular Pali and Pkt. ending;

not common, but not so rare as represented in HJAS 1.79, tho I have not noted any case in SP. (Originally pronominal; for its use in pronouns see §§ 21.30, 31.) Only in verses except for Mv. Cf. also next. (Also as fem., § 9.95.) śleşayilvā krame LV 94.8; pūrve nimittasupine imi adršāsi (pron. add°) 196.12; mukhapuşpake 201.19. - vikirne (so read) Mv i.42.17 = 53.14 = 337.5 (Senart °nām; mss. vinikīrņo, vinikīrņe, vikīrņe); samghe ca te na bhindanti 101.10 (all mss., Senart em. samgham; the pl. makes excellent sense, 'they do not disunite assemblies of monks'); paryesanli tirthe 170.8; sadevake loke 215.2 = ii.17.18 (with mss. ii.17.18; in i.215.2 mss. sadeva-loke, unmetr., Senart em. °kam lokam); vinīvaraņe ca dharme abhisameli i.312.9–10 (prose); vikrame sapta vikramate ii.20.19; avahāyāsvachandake 166.9, 'abandoning the horse and Chandaka'. - sa imām (= imān) bhāsate loke Ud xvi.5, 'he illumines these worlds'; later var. sa imam bhāsate lokam, but one later fragment keeps imām. — sarva prayuñji (see note) svakasvaka dharme Siks 343.2, 'all would carry out each his own duties.

8.96. Acc. pl. -i, m.c. for preceding, only in verses and not common. evamvidhā supini adrši (pron. add^o) LV 197.17 'saw such dreams'; so Tib. (better than as loc., 'saw such things in a dream'); naragaņi 233.9 (so Lefm.; certainly acc. pl.; but several mss. with Calc. ^ogaṇa); gehi 240.18 (all mss.; Calc. geham, unmetr.). Cf. also imi, LV 196.12, § 8.95.

8.97. Acc. pl. -u, rare, and only in verses, m.c.;

fundamentally a nom.-acc. sg. ending, transferred to pl. as in the nom. pl. (§ 8.84). kaşaļu tān SP 44.11 (WT's ms. K' kaşaļa); nirgatu 88.1, Nep. mss.; Kashgar rec. and WT nirgata; KN em. nirgati (!); certainly acc. pl. with dārakān. — śrulu' imu dharmām Samādh 19.33 'having heard these conditions'. Régamey em. imu dharmam, acc. sg., and in next pāda he also em. adharmam for adharmām. But this double emendation is obviously impossible; Tib. cited p. 81 n. 138 indicates pl. for both; therefore imu must also be acc. pl., unless we emend gratuitously. — pariorlu nāthām Bhad 25; (buddhām) nişanaku (no v.l.) 28.

8.98. Acc. pl. -āni, nt. ending with masc. nouns (cf. § 6.4): sarvānimāni kumārakāny SP 73.2 (prose; so all mss., except Nep. sarvānimām; note domal nl); etāni brāhmanāni Mv i.309.10 (prose); dāsāni iii.42.1 (prose); putrāni 79.3 (vs); varmitāni ca purusāni catvāri Suv 106.1 (vs); many other examples from Mv in Dschi, NAWGött. 1949, 273 ff.

[8.99. Acc. pl. -āna? In HJAS 1.79f. I admitted -āna or -ānā (in verses) as an acc. pl. ending on the basis of one form in SP and one in LV, supported by one SP form in -īna from an i-stem. The LV passage, 196.15 (d)hadānā, is found in no ms.; some mss. (d)hadānām; others with Calc. (karām caranā) višālam, which had better be adopted. This leaves only sarvāms ca saltvāna (lathaiva cāham) SP 323.13, which can hardly be anything but acc. pl.; WT em. to saltvān pi, but no sound scholar will agree to such an easy but violent evasion. The photostat of ms. K' reads: sarvāms ca saltvān aham adhisthihāmi; omitting the metrically impossible (Sktizing) syllable ma, we get, with sth for sth, ... sattvān ah' adhisthihāmi, which is probably the correct reading; on mg. see Dict. s.v. adhitisthati 3.]

Nominative-accusative plural neuter

8.100. Nom.-acc. pl. nt. -a.(-ah?). This is common, as (at least sporadically) in Pali, AMg. and other Prakrits. Pischel and Geiger regard it as the Vedic ending. I think Hemacandra 1.33 was better advised in considering such forms 'masculine'; that is, the masc. nom.-acc. pl. ending is also used with neuters, as vice versa (§§ 8.86, 98). Chiefly in verses; also in prose of Mv, and a few (sometimes doubtful) cases elsewhere; these are specifically noted below. balā SP 30.11, 62.2; koļišatā e' anekā (v.l. °ke) 91.1 etc., common. — rosavākyā LV 43.3; -nayutā ... vinītā 48.13-14; vähanä 78.18, 79.6; etc., common in verses. Weller 45 lists half a dozen cases in LV prose, but most of them are to be taken otherwise (HJAS 1.80); at most parijñālā saltacarilā 351.11 may be admitted as a fairly plausible instance. — duķkhā Mv i.5.3 (prose); asthīni avašesitā instance. — (3 mss. °lāh) 15.3; vaihāyasā 18.9 (prose; taken by Senart as adverb, blend of vihāyasā and vaihāyasam); vividhā duhkhāni 27.3 (prose); indhanā (mss.) vinā 67.8; šāsanā ca 121.10 (v.l. šāsanāni, Senart em. °nam ca); pairā ca phalā ca (repeated six times) 194.7 fl. (prose); imā (v.l. imā-m-) utpalāni 233.3; trņā (in same line trņānil) 328.17 (prose); etc., common. — dvipāni saratnapūrnā niryātitā Suv 146.1-2; etc. — lādršā yādršā laksanā Samādh 22.15; balā 22.16. — jālimaraņā RP 7.17. — anantā Bhad 47; yā . agrā 56 (see Dict. s.v. pariņāmana). — kulā baddhā Divy 76.8 (prose). — Once, at least, the mss. even present the ending -ah, at the end of a line: sarvakāryāh LV 313.20 (vs), so best mss., Lefm. em. va (v,l. vam, but plural verb).

8.101. Nom.-acc. pl. -a, shortening of preceding m.c., as in the masc.; only in verses. Common here, as in Ap.; Alsdorf 57 records Kumārap. as showing it 26 times to only 4 times each of -aī and -āī. saplāha trīņi (WT °ņī, m.c.) paripūrņa SP 54.13; rātrimdiva 62.9; yānaka 87.7; etc.— -sahasra LV 11.20; puspa 49.8; vāhana krta sajjā (so divide) 79.16; padma rucirā (so divide; followed by mlānāni) 193.20; etc. — cīvara Suv 4.13. kşetra pi tāttaka Samādh 19.16; dvādaša yojana 19.21. kşetra svaka RP 5.12. — kalpa Bhad 11; bala 38; praņidhāna 44; etc. — Pronoun ya: bhojana pāna daditva ya (ed. daditvaya) labdhā šiks 338.16, 'giving food and drink which have been got.'

8.102. Nom.-acc. pl. -ām. Not only in the acc., but apparently even in the nom. (as happens in masc. nouns, § 8.85), the masc. acc. pl. ending $-\bar{a}m$ (= $-\bar{a}n$) occurs a number of times in what, it seems, are at least normally neuters; cf. -ani in masc. nom.-acc. pl. This has been recorded only in verses. supinantaranidrsa LV 195.3 (vs) and svapnantaranidrsa 8 (vs), 'the contents of (my) dreams are such'; so Lefm., but nearly all mss. both times -idršām; can only be nom. pl. — elāni ... narakām (or masc.? acc. pl.; no v.l.) Mv i.12.9; trīņi samyojanām tyaktvā (two mss. samyojanā) 192.7; yadā satyām (no v.l., = salyāni) prakāšayet 251.1. — skhalitām (= °tāni) gavesī RP 18.10; ratnāni . . . sušubhām (acc.) 52.16.—bala sarvām (acc. pl.) Bhad 38; saukhya viśistām (v.l. °tam) 47; āraranām ... sarvām 57 (acc. pl.; in Gv 547.17 both printed as ending in -ān).

8.103. Nom.-acc. pl. -e, or m.c. -i (§ 8.81), the masc. ending, in a word which ordinarily, at least, is neuter: angam-ange (nom. pl.) LV 223.15; yasyā laksani kāyu citritah purimašubhaphalaih 123.3, 'who has the signs (and) a body adorned by the fruits of former merit'. But most mss. read laksana. (Even laksani should not be taken as instr., correct HJAS 1.81.) And pronouns, te nirarthāni ulsrjya Mv ii.206.3 (vs), 'abandoning these (i. e. tapā = tapāmsi) as useless'; LV (vss) karmā ... ime 42.7; ima (i. e. ime) āsana 115.9; ceļasyābharanāni deth' ime 123.7; ime (sc. bhājanāni) 383.12.

8.104. Nom.-acc. pl. -u, as in the masc. nom. and acc. pl.: jānayi sarvi vikurvilu teşām Bhad 45. No v.l. for vikurvitu, which can only be acc. pl. So alamkrtu (with kşetra, pl.) Bhad 47.

8.105. Nom.-acc. pl. -āmsi, the ending of nt. asstems. Not recorded for MIndic a-stems; but cf. the instr. sg. ending -asā (and in Pali also loc. sg. -asi), § 8.41. svasarirāmsi Gv 539.12 (prose). — para-mantrāmsi Mmk 366.13; hrdayāmsi 496.5 (vs); -pretāmsi 504.20 (vs). In LV 307.7 (prose) Lefm. kešāmsi, with all mss., tho keša is normally masc.; Weller 33 suggests influence of preceding urāmsi (mss. urasāmsi or udarāmsi). See Dict. s.v. kešas for possible support for the s-stem.

8.106. Nom.-acc. pl. -ānī. The normal ending -āni may appear with final i, m.c.: kuntalānī LV 49.20 (most mss. read ^oni; end of pāda, but regularly long in this meter).

Instrumental plural

8.107. Inst. pl. -ai? This is read quite often by Lefmann in his edition of LV; but usually the majority of his mss. do not support him (see e. g. his Crit. App. on 236.9 and 18, 237.7-8), and I regard this as merely a matter of bad editing. In LV 93.2 and 5 gandhodakai is, indeed, read by most mss., but A (perhaps the best) has ^{k}ke , and a loc. sg. would be syntactically possible. (Delete paragraph 2 under Instr. pl. in HJAS 1.81; if laksani be read in LV 123.3 it should be taken as nom. pl.; see § 8.103.)

8.108. Inst. pl. -ehi. Very common in both nouns and pronouns. In Mv it prevails even in prose, where it seems to be decidedly commoner than the Skt. -ais; but in other texts only in verses. It is the regular ending in Pali; in Pkt. -ehim is regular, for which according to Pischel 368 -ehi occurs only in verses m.c., and sometimes in prose before enclitics. It is curious that -ehim seems never to occur in our dialect, even when meter requires a long final. In that case either -ebhis is used, or more rarely the final i is lengthened (see below). Once, only. I have recorded imehir (followed by a vowel), in My i.90.6, a prose passage; probably the -r is to be classified as samdhi-consonant ('hiatus-bridger'). dantehi SP 85.12; putrehi 87.4; anyamanyehi arthehi 125.14; -putrehi 278.11 (quoted as -putrebhi Šiks 47.15); etc., common.-valamsakehi LV 172.6; punyehi 235.4; etc.-Mv (examples here cited only from prose; extremely common thruout the work): (a)prameyehi kalpehi i.1.14 f.; aparimāņehi lalhāgalehi 2.2; āyasehi nakhehi 5.5; kalamehi 120.8; yogācārehi 9; paripūrņehi ca dašahi māsehi 148.1; etc. Such forms jostle the (seemingly less common) ones in standard -ais: imehi ... adhyāšayaih 1.89.6; ākārehi 89.9, resumed in 12 by ākāraiḥ; viviktaṃ kāmehi viviktaṃ pāpakair akušalair dharmaih 228.3.-laksanehi Samādh 22.19; bhāvehi 22.30, 31; apramāņehi dharmehi, akalpitehi dharmehi 22.51. buddhehi Suv 24.2; 38.2; vrksehi 41.6; lokapālehi 133.6; 134.1, 3. — -bāndhavehi Gv 214.14; lakşaņehi 240.19; etc. etehi tribhi sthänehi Ud xx.16.-hinehi Sukh 74.3.-Pronominal stems, besides some cited above: lehi SP 279.1; 280.5; Mv i.193.9 (prose); Samādh 19.32; Sukh 51.1; yehi tehi Dbh.g. 22(358).21; yehi SP 89.5; Šiks 4.3; Gv 255.1; sarvehi Mv i.187.1; anyehi SP 96.3; kalamehi Mv i.170.12 (prose).

8.109. Inst. pl. -ehī, rare, and only in vss., m.c. for -ehi. (Generally -ebhis or its samdhi representatives are used instead.) Only pronominal forms have been recorded, viz. tehī SP 194.8 (but Kashgar rec. tebhih), and yehī SP 9.1; 49.9; 52.9; 97.9; LV 222.20; Sukh 51.2.

8.110. Inst. pl. -ebhis; the regular Vedic ending, on which -ehi (above) is based; here probably by secondary Sanskritization of the MIndic form. Often used in verses, at the end of a pada, (as -ebhir) before vowels, and before consonants when a long final syllable is required. In prose very rare; I have noted only -nirdesebhih KP 29.2; imebhir Mmk 4.20; and in the Kashgar rec. of SP 328.9 (Thomas ap. Hoernle MR 135), -samebhiś ca (bodhi)-sattvebhi(r), for ed. -samais ca bodhisattvair; cāraņebhih six times, and cāraņaih six times, in the same two passages (almost, but not quite, in regular alternation with one another), MSV i.34.11-15 and 35.11-16. The rest are in verses: -dhvajebhih SP 89.2; nafebhir jhallamallebhir 280.6.--simhāsanebhih LV 80.19; -guņebhir 111.13; ābharaņebhir 123.4; -priyebhir 230.14. - raudrakarmebhih Mv i.9.11; end of a pada; -ehi would do just as well metrically, but no v.I. is recorded. This ending is extremely rare in Mv; in fact I have failed to note another case (but cf. next paragraph). — -caritebhiš Suv 4.12; varņitebhih 44.7; -sahasrebhir 160.5; sarvebhir bodhisattvebhis 161.11; vicilrebhiś ca puspebhir vicitrebhih phalair api 165.2. — satebhih RP 20.17; jinebhih 27.17; -praharebhih 32.15. — rulebhir Gv 231.3. — jinebhih Bhad 3, 14, 56; rulebhih 4, 18, 30; etc. - dustebhih Mmk 494.18; mudritebhis 494.19. (Rare in this text.) — Pronouns: lebhis Suv 160.1; yebhih (-ir) SP 14.10; LV 343.10; Suv 161.15; MSV i.276.20; 277.1.

8.111. Inst. pl. -ebhi, chiefly in verses as Sktization of -ehi above, or as shortening m.c. of preceding (yet cf. next paragraph). But occurs in Mv prose, tho rarely. In most texts much rarer than the regular MIndic -ehi. anyebhi sūtrebhi na lasya cintā SP 99.3, read with Kashgar rec. — gunebhi LV 46.18; kramatalebhi (so, one word) 94.17; pratyayebhi 177.17; devebhi 224.3. — ayovişkambhanebhi Mv 1.8.2 (prose; in the same sentence, line 3, salejobhūtehi); äryadharmebhi (so mss.; Senart em. -ehi) 337.1 (prose; forms in ais in line 4). — etebhi ślokebhi Suv 44.7; -punyebhi 44.12; -varebhi 49.3; -caurebhi 57.4.—puspebhi Gv 43.12. kuśalebhi Dbh.g. 8(344).10. — -varebhi Bhad 5, 6; -putebhi 6; -galebhi 14, 56; -sutebhi 14, 25; -rutebhi 18. — Pronouns: tebhi SP 286,2 (so read with Kashgar rec., ed. em. tehi); Bhad 23, 24; yebhi LV 171.3; 343.8; sarvebhi LV 138.4; anyebhi SP 99.3 (so Kashgar rec.). 8.112. Inst. pl. -ebhī, would seem to be interpretable only as metrical lengthening for -ehhi, like -ehī for -ehi. As we saw -ehī is not common, and -ebhī seems to be even rarer (and only in verses). Its existence can, however, hardly be denied; the independent standing (apart from metrical requirements) of -ebhi is confirmed by its occurrence in the prose of Mv (above), and metrical lengthening of its final *i* is quite typical for our dialect. turiyebhī LV 169.22. So most mss.; the meter is troublesome; the variants (turiya, turiyabhī) do not seem to improve it. svarebhī Dbh.g. 18(354).12; followed by e-. Note that here a long final can be produced only by lengthening the vowel, since -bhir before a vowel would still be short. — -rulebhī (at the end of a dodhaka line) Šiks 329.5 (cited from Ratnolkādhāranī).

8.113. Inst. pl. -abhis? This and the next two are sporadic and doubtful. In LV 172.4 and 215.10 ed. ratnabhih; all mss. end -abhih; in the first passage at least, and probably in the second, meter demands a short penult. With this ending we may compare the common Ap. ending -ahim (Pischel 368). In HJAS 1.82 it is suggested that -abhih may be analogical to the n-declension (nāmabhih etc.), which in Mindic and BHS is often in-flected according to the a-declension.

8.114. Inst. pl. -ibhis, m.c. for -ebhis? Possibly in Mmk varnibhih 134.12 (vs; doubtfully recorded in Dict. under varnin, q.v.) and laksapramänibhih 139.25 (vs) 'measured by lacs' (also, possibly, based on an otherwise unrecorded stem pramänin; see Dict.). 8.115. Inst. pl. -äbhi? If acceptable, this would be

8.115. Inst. pl. -ābhi? If acceptable, this would be the fem. ending transferred to the masc.-nt.; sarvendriyābhi LV 189.15 (vs), most mss. (the variants, ^oyabhiḥ and ^oyibhiḥ, cf. the preceding two, are metrically bad); Lefm. em. ^oyebhi, perhaps justifiably.

Dative-ablative plural

8.116. Only regular Sanskrit forms noted, except -ebhya m.c. for -ebhyah: narakebhya Kashgar rec. for °keşu SP 94.5.

Genitive plural

8.117. Gen. pl. -āna. The Pali ending is -ānam; in Pkt., especially M. and AMg., this and -āna occur side by side. In our dialect -āna is the commonest (non-Sanskrit) ending, before both vowels and consonants. In Mv it is not infrequent even in prose, but in all other texts it has been noted only in verses, and only in positions where meter favors a short ultima.

8.118. Before vowels it is not exactly common (-ānam, or even -ān', being apparently commoner), and has not been noted in prose. Examples from verses: pañcāna SP 83.3; bālāna 99.4. — tatropavişāna LV 27.19; mānuşāna 115.1. — manujāna Mv i.73.4; kāmāna 184.13; buddhāna (but v.l. °nam) 269.5. — dharmāna Samādh 8.22; buddhāna 19.22; lokanāthāna 22.17. And, with fusion with following initial vowel: mārgānāşlāngikah Ud xil.4 (= Skt, mārgānā mag²).

8.119. Before consonants (prose passages from Mv are so marked), very common: sugatāna SP 61.8; bālāna 87.6; mrdukāna 89.5.—šākiyāna LV 76.3; puņyārthikāna 223.10.—Mv, mss. ravantāna or °nam (Senart em. °nām) i.6.1 (prose); -mahisāna sūkarāņa kukkuļāna 17.1 (prose); kalpāna 35.13 (prose); jināna 62.2, 64.2; lokanāthāna, agrapudgalāna 80.11 and 13 (prose; the more striking because preceded by parallel phrases containing regular forms in -ānām); āryāņa iii.419.16; 420.8 (in Pali version of same verse, Jāt. iii.354.18, ariyānam, unmetrical!); etc., very common.—krtapuņyāna Suv 4.10; sattvāna 28.11. dharmāņa Samādh 8.23, 24; kalpāna 8.29; marutāna 19.21. — nāuakāna RP 12.8. — -dharmāna Gv 24.17; vrksāna A-slems

43.12. — jināna Bhad 2, 4, 5 etc.; -sutānu 3, 28. --- adhanāna (so divide) Sukh 22.8; buddhāna 49.4.

8.120. Gen. pl. -ānam, -ānam. This ending, in the form ānam, occurs only before vowels, and I have not noted any certain occurrence in prose, even in Mv (but note Mv iii.114.19 below). dvipadānam SP 53.8; manujānam 68.10; sarvajinānam 193.1.—satīvāņam LV 219.3. — bodhisatīvānam Mv i.91.6; 107.1; dvipadānam 316.13; tridašānam ii.35.18 (mss.); 52.12 (here Senart with v.l. °āna); trayastrimšānam iii.1.17; narānam 108.2; dvipadānam utlamo 114.19 (printed as prose; I am not certain that it is not originally a bit of verse); tridašānam 367.3. — narāņam Samādh 8.21. — -ratānam Gv 473.10. — śuddhakakālakānam (edakalomnām) Mvy 8397 (?), so Kyoto ed.; formula cited in Pali with suddha-kālakānam (see Dict. s.v. kālaka 1). But Mironov reads °kālakānām.

8.121. In the form $-\bar{a}nam$ it occurs at least once before a vowel, in a prose Mv passage, guhānam ašīti i.54.5; so at least all mss. read; Senart emends to guhānām, keeping the anusvāra (which is the only questionable feature of the reading of the mss.). Otherwise this, the regular Pali and Prakrit ending, occurs only before consonants, and is very rare in our texts; we may however reasonably suspect that later copyists have changed not a few $-\bar{a}nam$ readings to $-\bar{a}n\bar{a}m$, and even that careless editors have done the same without recording the fact (as responsible editors like Senart sometimes do, with proper record of the fact). Only the following have been noticed: ravantānam or ^ona, mss. (Senart em. ^onām) Mv i.6.1 (prose); devānam ii.352.1 (vs); tānam devānām, mss.^o (Senart em. devaānām, allowing tānam to standl) iii.304.3 (prose). — satyānam cature Ud xii.4 (vs).

8.122. Gen. pl. -ana, only m.c. for -āna: anyana (= anyeşām) RP 21.4 (vs).

8.123. Gen. pl. -ān', before vowels; only in verses, as demanded by meter, with MIndic samdhi; that is, for -āna (from -ānam = -ānām) with elision of final vowel before initial vowel: prakāšayantān' iha (for Skt. prakā-sayatām) SP 229.9; sugatān' iha (WT ihā, m.c.) 359.3. — hitāya bhātān' abhibhur Mv 1.167.9; buddhān' ādilyaban-dhunām ii.232.11. — buddhasutān' atha Bhad 9; sarvajinān' anu- 17.—Read manujān' upapaltim Dbh.g. 7(343).6.

8.124. Gen. pl. -an, -am. We find also occasionally forms written -an, -am before consonants, and even -am before a pause (at the end of a line of verse), which seem certainly intended as genitives plural. They can hardly be connected with the preceding cases of -an before vowels. Rather, we must assume that they are based on the analogy of Skt. consonantal stems, which in our dialect, as in MIndic generally, often show both the old consonantal endings and those of *a*-extensions. Thus doublets arose like *vanijanām* (LV 387.10, prose) beside *vanijām*, both gen. pl. of *vanij(a)*. This then led to spor-

adic forms like nayanām, gatām, by the side of regular nayanānām, gatānām. Cl., as a sort of converse process, Pali tesānam (= tesām) etc., Geiger 105; our language has such forms as etesānām, § 21.44. Writings in final -n instead of -m (-m) are purely orthographic, or may even be treated as scribal errors. Examples, all from verses (meter proving the soundness of the tradition): preksatu (for °tām, see next paragraph) nayanān na cāsti trptim (nom. sg.) LV 49.16, 'and there is no satiating of the eyes as they gaze'; dašadīggatām (read °di-gatām with most and best mss., m.c.) jinotlamānām 163.14, 'of the noble Jinas in the ten directions'; bodhāya cillam atulam

noble Jinas in the ten directions'; bodhāya cillam atulaņu pranidhāya saltvām 356.11, (Bodhisattvas speaking) 'having vowed our minds in an unequalled degree to the enlightenment of creatures' .--- anyatha citryamanam (so read with most mss. for ed. citra°) hi lobhaniyam hi bālišām (so mss.; ed. °sān, but an acc. cannot be construed) Lańk 365.14. 'for if otherwise variegated, it would be a thing coveted of fools' .--- bodhāya prasthitām sattvām sadā siddhir udährlä Mmk 59.22, 'for creatures that have set out for enlightenment, success is always declared'; duhillam (here meter requires duķsīlānām! probably a monosyllabic word has been lost) jugupsitām 66.24; nānyesām kalhitā siddhih bālišām strīsu mūrchitām 101.22; duhkhinām sarvalokānām dīnām dāridryakhedinām, āyāsoparalām kli-sļām lesām arthāya bhāsitam 104.19, 20 (all the forms preceding lesām are gen. pl.); śrāvakapungalām 108.23; 109.1 (final in line; can only be gen. pl.).

8.125. Gen. pl. -ānu. Since -u appears as phonetic reduction of -am, -am (-ām) (§ 3.58), -ānu can stand for the MIndic resultant of Skt. -ānām. Cf. such pronominal gen. pl. forms as teşu, yeşu, pareşu (for Skt. teşām etc.), § 21.40, and, from consonantal stems, forms like prekşatu LV 49.16 (in preceding paragraph). Only in verses: read trşitānu SP 324.10 with most mss. (only one [°]āna with text). — sugatānu sutāh Gv 476.22, 'sons of the Sugatas'. — bahupunyānu hetavah Mmk 119.23 (ed. as one word, but I cannot believe in a stem *anuhetu); grhya-m-aksāņu sūtritam 121.16 (? ed. as one word; interpretation uncertain); pillakāntānu dehinām 148.15 (this seems a clear and certain case; for -kāntānāmām); perhaps śasyānu KP 44.6, see Dict. s.v. anuvarşa.

Locative plural

8.126. Loc. pl. -işu, -asu. Extremely rare, and only in verses m.c., are reductions of the ending -esu to -isu or even -asu. I have noted only: sarvişu (m.c. for sarveşu) Siks 344.15 and 17 (cited from Ratnolkādhāranī). sarvasu Mmk 377.4. — sarvasu janmasu Gv 544.14; = Bhad 16, which however reads sarvasu (for sarvāsu) jālişu.

8.127. Loc. pl. -eşü, m.c. for -eşu: grheşü LV 57.1.

9. Ā-stems

9.1. The non-Sanskrit forms of the \bar{a} declension include, to begin with, a number of definitely Middle Indic forms, recorded as such in Pali or Prakrit or both: Acc. sg. -am for $-\bar{a}m$

Oblique singular in $-\bar{a}ya$ and $-\bar{a}$ besides $-\bar{a}ye$, see below Abl. sg. in $-\bar{a}tas$ (and phonetic variations)

Loc. sg. in -ayam

Nom.-acc. pl. in $-\overline{a}$ and $-\overline{a}yo$ (and -e, see below) Inst. pl. in $-\overline{a}hi$

9.2. There are also one or two endings which are clearly MIndic but not recorded, at least in Geiger or Pischel, as occurring in Pali or Prakrit. Most notable of these is the nom.-acc. pl. $-\bar{a}vo$. The oblique sg. $-\bar{a}ye$ may also be included here, since it is recorded in Pischel only without -y, tho it is evidently essentially the same as the standard Prakrit $-\bar{a}e$. It is however found in Asokan inscriptions.

9.3. Purely phonetic developments from either Skt. or Mindic endings, chiefly in verses and m.c., and in large part recorded (under the like conditions) for MIndie dialects, are:

-a for $-\bar{a}$ as stem-form in composition, as nom. sg., voc sg. (originally nom. used as voc.), acc. sg., oblique sg., and nom.-acc. pl.

-i for -e as voc. sg.

-am for -am as acc. sg. (also -an, orthographic)

-ayām for -āyām as loc. sg.

-abhi for -abhih (or for MIndic -ahi, above) as inst. pl. -ana for -anam as gen. pl.

-asu for -āsu as loc. pl.

-āsū for -āsu as loc. pl.

9.4. Originally masc. and nt. endings are used in fem. nouns on a rather large scale. Such forms occur in Prakrit (Pischel 358, 359) and Pali (Geiger 76); they are probably commoner there than Pischel, in particular, seems inclined to admit. At any rate they cannot be emended out of our texts, tho most editors have tried to do so; they are, in fact, pretty surely commoner than our editions suggest. We find -u (for -o = -as and perhaps for -am) in nom. and acc. sg; -ena (and -ayena) inst. sg; -dl abl. sg; -asya gen. sg. (pronouns); -e loc. sg.; -e nom. (of pronouns) and acc. pl. (Mindic masc.); -ani nom.-acc. pl.; -an, -am acc. pl.; -dho voc. pl. (MIndic masc.); -ais and -ebhih, -ehi (MIndic inst. pl. Compare the reverse process, fem. \bar{a} -endings used with masc. and nt. a-stems (§ 8.2). Cf. §§ 6.1 ff.

9.5. And finally there are a number of morphological analogies which do not seem to be recorded in Pali or Prakrit: $-\bar{as}$ and $-\bar{am}$ as nom. sg.; $-\bar{a}$ and $-\bar{aram}$ (in bhāryāram) as acc. sg.; $-\bar{aya}$ and $-\bar{aye}$ (a little doubtful) as nom. pl. and -a (doubtful) as acc. pl.

Stem form in composition

9.6. Often shortened to -a, almost always in verses and clearly m.c. But also in prose: -prajña-prabha-raśmi-LV 9.11; both edd., no v.l.; noted by Weller 16, and supported by his ms.; velajñas ca 218.20; no v.l.; not noted by Weller or others; associated with kālajňa and samayajňa; vela- for velā- perhaps by attraction to them. So also vijňa-prāpla, beside vijňā°, see Dict. s.v. vijňā; kālajňo velajňo samayajňo kālam ca volam ca samayam ca āgamayati Mv ii.233.15; see under acc. sg. -am for -ām, below; kānkşa-prahāņam RP 8.10 (kānkşa as masc.? see § 9.13).

9.7. In verses, common everywhere, where meter requires short: vicikitsa-prāptam SP 64.3; krīda-ratīsu 88.5. — cinta-manah LV 41.16; māya-deni 75.5. — susrūga-kāri (su°) Mv i.156.11; gaiga-vāluka-sameşu iii.253.1. — Compounds containing pipāsa- Suv 39.7; krpa- RP 3.2; Gv 54.11; prabha- RP 3.9; Gv 241.1; prajīa- Bhad 27; 37.

Nominative singular

9.8. Nom. sg. -a, often in verses m.c. I have noted at least one case in prose, eşa ... pravrajyopasampadā Mv iii.329.13, and two cases at the end of lines of verse where there is no metrical reason for the shortening: yena na śodhila carya viśuddhā Samādh 19.22 (for śodhilā caryā viśuddhā). The first two m.c., but there is no metrical obstacle to viśuddhā. Apparently not misprint, since Régamey cites it p. 16. pūja jane sadātra cala sidhya RP 30.4, which I take to be for pūjā ... (a)ealā sidhyā (i. e. siddhyā, see Dict. s.v.; or read siddha, siddhā?), 9.9. Otherwise, only where meter requires short a,

9.9. Otherwise, only where meter requires short a, but common everywhere: $l_{f\bar{\eta}a}$ SP 92.6, pralisla 94.3; $p\bar{u}ja$ -nantikā 341.8 (= $p\bar{u}j\bar{a}$ anant^o). — kanya LV 28.17; pralir $\bar{u}pa$ 28.18, 20; 29.12; vasudha 297.2. — sampašyamāna Mv 1.155.3; sthila ii.339.15; (sā ...) prāpla bodhih 347.3; saha 380.18 (so mss.; Senart sahā, spoiling the meter). — ukla avidya Suv 61.4. — naivātmasamjīna hhavate 'pi na jīvasamjīnā RP 14.5; pativrata lyakta saputrā 22.17. — nişha (thrice) Bhad 46. — upajāta Gv 212.14. Pronoun: eşa ... rddhibhāvanā Mv 1.66.12.

9.10. Nom. sg. -as (?). Sometimes the mss. add visarga (or other representatives of s) to a nom. sg. in \bar{a} . The editors regularly emend; often, I suspect, without even recording the reading of the mss., so that it is hard to be sure how commonly the forms occur. It is, of course, possible that such readings are mere corruptions. Yet they might have been used by the original authors, as hyper-Sanskritisms. The MIndic nom.-acc. pi. ending -a is common in our texts, and was certainly used by many authors. Since this made nom. sg. and pl. alike, the nom. pl. -as might easily have been used also in the sg. by analogy. The fact that other vocalic stems so often end in -s might have helped. Two examples, both from verses: kşipram bhohī (3 sg. fut., with 2 sg. fem. subject) prili-prāmodyalabdhāh LV 196.4 (so all Lefm.'s mss.; em. °labdhā; Calc. °lābhāh, without note, presumably em.); pranidhī iyam evarūpāh 415.21 (so all Lefm.'s mss.; both edd. °pđ; certainly nom. sg. fem.).

9.11. Nom. sg. -ām. The use of the Skt. acc. sg. form as nom. is quite common in the mss. of Mv, and at least sporadic elsewhere. As in the preceding ($-\bar{a}s$ for $-\bar{a}$), it may have been actually used as a hyper-Sanskritism, induced by the identity of MIndic nom. and acc. forms, both sg. and pl. Note that the converse, $-\bar{a}$ as acc. sg., is also common, and often accepted by the editors, sometimes even by Senart, who rejects $-\bar{a}m$ as nom. Meter, of course, cannot be concerned here; but it is to be noted that -a for Skt. - $\bar{a}m$, acc. sg., is about as common, where meter favors it, as it is for nom. sg. - \bar{a} . dharşita (for ° $t\bar{a}$) mārasenām (for ° $n\bar{a}$) LV 353.2 (vs); so Lefm. with most and best mss.; Calc. with some poor mss. ° $n\bar{a}$.

9.12. In My the mss. very commonly present such forms, not only here, but in other vowel declensions, both masc. and fem. Senart systematically emends to the regular Skt. forms. A few examples out of many: In My i.14.5 and 17: 15.9 (vss) Senart reads oruddhā (15.9 an/a) *janalā bahu*. Once all six mss., and twice all but one, read *janalām* (some corruptly $^{\circ}n\bar{a}m$); nearly all likewise bahum or bahum; esām ... praņidhim Mv i.61.2 (prose), all mss.; Senart em. esā ... praņidhi; certainly nom. sg.; gālhā hi dharmasamhilām 92.16 (end of vs), all mss.; Senart em. $^{\circ}samhilā$; adhivāsanām (mss., Senart em. $^{\circ}n\bar{a}$) ... vijnādā 263.4 (prose). These cases are typical of Mv and could be multiplied many times. In most texts, however, such forms rarely occur in our mss., if the reports of the editors can be trusted.

9.13. Nom. sg. -u. Rarely, and only in verses, the texts seem to show forms in -u (also in acc. sg., § 9.23), as in masc. and nt. a-stems for Skt. -as, -am. If not mere corruptions of tradition, they are best regarded as the masc.-nt. forms taken over into the fem. owing to Mind. breakdown of distinctive gender forms, § 9.4. The clearest cases noted are: anusmrli bhāvanu šabda nišcarī LV 182.21. There can be no doubt that bhavanu (both edd. and all mss. -nu) represents a nom. sg. of bhāvanā. Ivayi mali pralibalu amrlarasa dadā LV 165.6 'in thee is competent wisdom; give (them) the essence of nectar.' Calc. Sola, but Lefm.'s mss. all "balu (for "bala"). Or (?): 'in thee is wisdom; give them essence of nectar that is competent' (to slake the thirst mentioned in the preceding line); but this seems to me forced and unlikely..anupraplu ... bodhih LV 243.7. But here bodhi may be masc., as it may be in Skt.; to be sure in Pali it is cited only as fem. by PTSD, but Childers gives both genders. In our dialect it is usually fem. In this same line there is doubt about the form of a modifying pronoun; Lefm. reads seli (= sā-ili), but several of his mss. read so ti (masc.). — parivartati tiksņu dhārā Gv 213.6. bodhi-kānksu mama vidyate mune RP 9.13. There is no evidence for a masc. or nt. *kanksa (or Pali *kankha), unless the cpd. känksa-prahānam (in prose of RP 8.10, § 9.6) be taken as such. imu bhaktih RP 27.9. - ruju bhūya vardhale (both edd., no v.l.) 'their disease increases more' SP 96.2. No masc. noun *ruja is recorded in Skt. or Pali, But ruju could stand for rujo (rujas), nom. pl. of ruj, 'diseases' (altho subject of vardhale, sg. in form).

Vocative singular

9.14. Voc. sg. -i for -e. Instead of the regular ending -e, forms in -i occur, but only in verses and certainly m.c. Any final -e is likely to be treated thus. E. g.: gopi, for gope, voc. of Gopã, LV 236.4; 237.9 and 11.

9.15. Voc. sg. -a. This occurs rarely and only in verses. Like Pkt. vocs. in -d (Pischel 375; apparently Pali has none except the like of amma, Geiger 81.2), it is doubtless to be understood as a nom. form used as voc. agrasativadhāra LV 76.6 'O thou (Māyā) that bearest the Supreme Being!' — devala Suv 60.3, voc. to devalā. But the best ms. reads daivala, voc. to daivada, and this may be the true reading (or, on the other hand, it may be merely a Sanskritization).

Accusative singular

9.16. Acc. sg. -am, -am. In most texts, this ending seems to occur only in verses, regularly in the form -am (not -am), before a word beginning with a vowel, and usually where meter requires a short syllable. Thus: dagdhaulkam abhi^o Mv i.65.18; so all mss., and so meter demands; Senart em. "ulkām, spoiling the meter. $-p\bar{n}jam$ idršīm SP 15.3; imam eva cintām 61.11. — tām pārvām gira varam (= giram varām) anucintyā LV 164.11; gira varam anuravi talu varaturiyaih 164.22 (so read; cf. notes of Lefm. and Foucaux, H p. 133; supported by Tib.) 'they continued to sound the excellent words with the excellent musical instruments'; chinnām jālikam a- 194.20; śriya kriyam atulām (so read) 361.18. — Amitābhapurim varam eva Bhad 62. — caryam imām RP 16.12. — ksematam abhapupti Dbh.g. 4(340).10. But in Mv i.305.9 (vs), the meter is indifferent, and in fact the mss. read tam enām avadad rājā (Senart em. tam enam, both are fem.).

9.17. Apparent cases of -am before a consonant, except in Mv, are questionable. Thus in SP 63.5 (vs) prathamam giram, praihamam is an adverb, not an adj. with giram (Kern, with Chinese, SBE 21 p. XLI line 34). And in Suv 234.17 (vs) tam bodhi gambhīram udāram istam, most mss. read istva, which makes good sense; if this be adopted, bodhi may be understood as masc. (cf. § 9.13), and lam would then be normal Skt. But in My there are clear cases of -am before consonants, where shortening of the \bar{a} to a cannot be m.c.; moreover it occurs in prose. It is the normal MInd. ending, Pali and general Prakrit; the strange thing is not that Mv has it, but that it has been virtually eliminated in all other texts, judging from the editions, except where meter compels its use. Cf. the loc. sg. -āyam for -āyām, § 9.80. Senart often, but not always, emends. Examples: kalpakofim asamkhyeyam Mv i.169.18 and 170.5; before consonant; so all mss. both times; Senart em. 'yām; he regards the form as plural; yam velam 361.3, 362.2, and elsewhere, 'what time ...'; here Senart keeps the mss. reading, as in: kālajňo velajňo (§ 9.6) samayajňo kālam ca velam ca samayam ca āgamayali ii.233.15. The frequent collocation of velā with kāla and samaya makes this word peculiarly susceptible to admission of apparently masculine forms. - seyyam (for śeyyām) ii.234.10 (prose), mss.; Senart em. "yām. On the other hand, in iii.411.14 (also prose), before a vowel, Senart keeps the reading of the mss., seyyam api; also in iii.126.8 āņallikam (the stem is °kā); but in iii.130.12 āsilam, mss., Senart em. °lām. -In an *āryā* verse occurring thrice, Mv i.145.8, 202.5, and ii.6.2, with variants, Senart prints the form ahimsam or avihimsam, before a consonant, but either ahimsa or avihimsa, for °sām, is to be read each time.

9.18. As prior member of a tatpurusa cpd., prabhamkara, 'light-maker', especially as epithet of a Buddha, is widespread. It represents Pali pabhamkara, Skt. prabhākara (of sun and moon), and is a clear case of MIndic -am for -ām. It seems even to have led to the creation, once (LV 123.1), of a masculine stem prabha, 'light'; see Dict. on both these items.

9.19. Acc. sg. -a. Occurs only in verses, m.c., but very common. Cl. Mv i.145.8 etc., § 9.17, end: krlva ... pāja jineşu Mv iii.140.23. — krīdāratim sarva vivarjayilvā SP 13.7; pilrsamjňa kuryāt 287.7. — yo māya drşivā LV 28.10; ima mahi (= imām mahīm) 165.15; prabha muñcamānāh 296.15. — vedana ... dukhām anubhonti Samādh 19.20 (Régamey p. 14 misunderstands dukhām, which is an adjective, with vedana for vedanām). — idrša bherīm pašyami svapne Suv 51.7; idrša dešana talra śrpomi 51.8. pūja krīva RP 5.11; trşņa utsrja 59.9. — pūja Bhad 25; kānkşa 54. — puja karoti Gv 215.3; pravartayi pūja jinānām 231.18.

9.20. Acc. sg. -ā. In this form, of course, meter cannot be concerned, altho it is actually recorded oftener in verses than in prose. It is not recorded for Pali or any Prakrit. And I have failed to note any occurrences in other texts than LV and Mv (except, perhaps, *anyatarā* MSV iv.227.20, prose). This may be due partly to oversights on my part; it is pretty certain that it is partly due to unacknowledged emendations by editors. The form is by no means startling;

it is simply the nom. form used as acc. The forms of the two cases have become virtually interchangeable. The following list is doubtless not complete, but too long to be emended out of existence.

9.21. In LV (verses, except as noted): (pūjanām . . .)anantā 152.4 (end of a verse); (yantra)yuktā varākapratimām 155.18 (prose; printed as a cpd.; cf. Weller 27); (lyaji . . .) mahi (= mahim) sanagaranigamā 165.10, 'the earth with its towns and villages'; śriya (= śriyam) kari ... purimā 167.4 (2 mss. °mām; end of a verse); krpa karuņā janiyā (= janayilvā) 168.8; kariyati (= kāryate 'he is caused to make') anantapüjä 185.6; nagnā drši ātmanam 194.13 'she saw herself naked' (perhaps interpret as nagnādrśi, i. e. nagna-[for nagnām]-adrsi?); trsņā tanum cu kari 200.9 'and he diminished his longing'; (imām . . .) sayanagatā virocalīm 214.16; ekā gira (= ekām giram) ... na bhasi 235.9; drstvā vikāravikrtā nanuces tu senā 339.19 (Calc. °krtām ... senām, but without support in Lefm's mss.); (?) anullarā (Calc. °rām; mss. all °rā; Lefm. prints as if cpd. with next, which may be right, see § 23.8) samyaksambodhim abhisambuddhasya 439.20 (prose); [imā gālhā (abhāşanla) 290.20, prose, taken by Weller 33 as sing., because only one stanza follows; but Tib. has plural. In this standard situation gathas, plural, are regularly announced; the author is not thinking of a precise count.]

9.22. In My this ending is as common in prose as in verse. It is regularly emended by Senart; yet he sometimes allows it to stand, for reasons not always evident to me: adhimātrām vedanā (kept by Senart, with all mss.) vedayanti i.19.9; adhimātrā (kept by Senart, with all mss.) vedanām (2 mss. °nā) vedenti 20.4 (prose); samkhyā (Senart em. yām) 62.13 (vs); irsyā 145.12 (vs) = 202.9 = ii.6.6 (acc. sg., mss. all °yā except one in ii.6.6; Senart em. °yām); anuddhatām ... girā (acc.; Senart em. girām) i.166.3 (vs); utsāraņā karonti 223.8 (prose) = ii.26.2 'make the transportation' (the noun is utsarana); so all mss. both times; Senart em. °ņam; sugandhā, salkrtā, supuspilā, sevilā ii.53.21–22 (vss; so mss., all accs., Senart em. all to -ām); dhītā (Senart em. °ām) 73.17 (prose); ghosanā (Senart em. °ām) 73.18 (prose); bhāryā (kept by Senart, perhaps understanding a plural? but cf. the modifying bhadrikām) kalyāna bhadrikām ii.367.21 (vs); āņā (so mss.; read mamāņā; Senart em. ānām) pratikrośe (=°śet) iii.7.16 (vs).

9.23. Acc. sg. -u. See § 9.13 for the same ending as nom. sg. Only in verses. rajani vigatu (all mss. and Calc. vinirgatu; Lefm. em. m.c.) jñātva LV 240.11 (vs) 'knowing the night to be spent'. — Pronouns: imu bhadracaryām Gv 212.17. — imu carim abhivīkşya RP 17.1. — imu bhadracarim Bhad 51.

9.24. Acc. sg. -ān, for -ām, probably merely orthographic; reflex of the MInd. anusvāra for all final nasals. Frequent, but not worth listing. Examples: cārikān for cārikām Gy 241.9, 16 (both at end of verses).

9.25. Acc. sg. -āram. aram. Repeatedly in Mv the stem bhāryā 'wife' shows an acc. sg. bhāryāram or (m.c.) bhāryaram; variants bhāyy', bhājy' are recorded, suggesting that more Prakritic forms may have been used in the original. No such form is recorded in Pali or Pkt, and Senart i.476 can make nothing of it; he emends here to bhār(i)yām. But it is clearly an analogical creation, using the r-stem ending of bhartr 'husband', acc. sg. bhartāram (or m.c. bhartaram, the like of which is recorded in Pali, Geiger 90 note 1), and of svasr 'sister', acc. svasāram. And later, iii.503, Senart withdraws the emendation. Mv i.129.2 (vs), read bhāryāram with mss.; 233.17 (vs), read bhāryaram; mss. 'yāram, 'yeram, unmetrical; 234.8 (vs), read bhāryaram with the one ms.; iii.8.6 and 9.1 (both vss, metr. indifferent) bhāryāram with both mss. in 9 (short penult required metr.); Senart em. bhāryarām; in line 7 the mss. are corrupt.

Oblique cases of the singular

9.26. For all of these (aside from certain endings peculiar to abl. and loc. respectively) Pali has only -aya (or its 'contraction' -a, Geiger 27.2), and normal Pkt. generally -ae (m.c. -ai; -aye in some Asokan dialects); only Mähārāstri, acc. to Pischel 374 f., seems to show also -aa, resembling Pali -aya.

9.27. Besides special endings for inst., abl. and loc. respectively, and the much rarer ending $-\bar{a}$ (as in Pali), our language shows standardly $-\bar{a}ya$ and $-\bar{a}ye$ for all sg. cases from inst. to loc. The former is evidently identical with the regular Pali ending, and similar to the alternative M. $-\bar{a}a$; while our $-\bar{a}ye$ is closest to normal Pkt. $-\bar{a}e$. The distribution of these among the several texts is, however, peculiar.

9.28. -aye (very rarely -ae, as in normal Pkt.; once also -ayi, m.c.; occasionally -ayai is written, by Hypersanskritism?; whether -aye with shortened a should be recognized is doubtful) is almost restricted to Mv; only a few sporadic cases occur in other texts. Yet in Mv it is much commoner than -ayar Both occur in Asokan.

9.29. However, -aya (for which -aya and -aya are found m.c.) is also quite common in Mv, in prose and verse alike. In the other texts it very rarely occurs in prose, but is common in the verses.

9.30. It seems best to present $-\bar{a}ye$ and its variants first, for all oblique cases together; then $-\bar{a}ya$ and its variants, similarly; then $-\bar{a}$ which, the quite rare, also serves for all oblique cases; then endings which seem to be peculiar to one of the oblique cases of the sg. Pronominal forms with these endings are included.

9.31. Obl. sg. -āye and congeners. First, the ending occurs in its normal Pkt. form -āe twice, on the same page of Mv: ii.432.1 hrstatustāe (here inst.; one ms. ⁹tustā me, clearly corrupt); and 9 imāe (here loc.; one ms. imāya). And again, Mv iii.180.12 ehibhiksukāe (one ms. ⁹kāye, the other actually ⁹koe).

9.32. Secondly, as -ai occurs in Pkt. m.c. for -ae, so -ayi is found m.c. for -aye, but extremely seldom. I have noted only: Mv ii.59.6 (vs), where the reading of the mss. seems to be asayi (inst. of aia), which is certainly demanded by the meter rather than Senart's em. asaye.

9.33. Occasionally the mss. read -aye, but Senart always emends to -aye, probably rightly. The forms occur either in prose, or in doubtful metrical surroundings such as: -kridaye (inst.; mss. -kridaye) Mv iii.15.2; in the first half of what seems meant for an anustubh line, but so bad that I despair of interpreting it metrically. The short -a, however, seems metrically poor. — Doubtful is LV 117.4 (vs) janataye, Lefm. with his ms. A only; other mss. janayadā, apparently error for janatayā (m.c. for °tāya); Calc. °tāyai, unmetr. A dat. or gen. of janatā must be intended, and the penultimate a must be short for metrical reasons; but whether the original ended in -taye (-tayai?) or -tayā is uncertain.

9.34. -ayai, when found in the mss., may be only a Hypersanskritism for -aye (except, of course, when it may be a dative case form). The following seem to be instrumentals: akarsanatāyai Mv i.314.3; vyaktatāyai suvibhaktatāyai gambhīratāyai akhandaśresihatāyai ii.43.7; anāhāratāyai (one ms. °tayai) 129.5.

9.35. Next we shall present the few scattered -āye forms which I have noted in other texts than Mv: abhaye(inst.) LV 122.20 (vs). Tho no ms. reads precisely so, it can hardly be doubted that Lefm. is right in adopting this reading. The variant $\bar{a}bh\bar{a}ya$ is metrically impossible; $\bar{a}bh\bar{a}yaih$ and $\bar{a}bheya$ intend $\bar{a}bh\bar{a}ye$. — Appendix to Jm 240.20 (corruptly corresponds to Mv ii.244.12) mānuşikāye vācāye (inst.). — Amrtāye (gen.) Māy 245.1 (prose). Surrounded by parallel forms in - $\bar{u}e$; from line 3 on, where other gens. of \bar{a} -stems occur (e. g. $M\bar{a}tang\bar{a}ya$ etc.), all end in $-\bar{a}ya$, not $-\bar{a}ye$. — cary $\bar{a}ye$ Bhad 23 (vs; gen. or loc.); Bhadracariya samanlasubh $\bar{a}ye$ (loc.) 44 (vs), 'in the perfectly beautiful Bhadracari'. Note the collocation of forms in -ya and -ye; no v.l. is cited for either. — parigavesal $\bar{a}ye$ (loc.) KP 97.3 (prose).

9.36. There remain only forms in -āye from Mv, where they are extremely common, in prose as well as verses.

9.37. Instrumental: The following are samples of what seem to be instrumentals—a few out of a couple of hundred. I have cited chiefly from prose; cases from verses are so marked. gäthäye Mv i.66.13; 255.12; ii.84.11 etc. (common); anulomatâye i.63.12; atirekapûjâye 89.16 (construe with preceding sentence); yâye prabhâye 205.10 = ii.9.6; anyâye ca janatâye i.231.15; manujâye vâcâye 274.3; svakâye 357.16; sudhâye ii.57.5; 59.1 (vss); vibhûşâye 100.6 etc.; imâye duşkaracārikāye 130.12; lâye syâmâye 173.8; agrâye paramāye pûjâye 259.9; senâye 410.5; lrşnāye 460.10; anyatarāye vrddhāye iii.15.7; elâye ganikāye 38.4; -nāvāye 67.17; nāsáye 128.18 (vs); dārikāye 155.1, 6; anyatarāye mānavikāye 184.10; upekşāye 421.16 (vs); yâye iii.290.18 (relative); elâye i.309.11 etc.; tâye i.348.14, 17 etc. 9.38. Dative: These are, as in all declensions,

9.38. Dative: These are, as in all declensions, much rarer, but the following seem reasonably clear: anuttarāye (samyaksambodhaye) Mv i.44.5; 233.9 (prose) et alibi; sarvajňatāye 82.13 (vs); purusasimhatāye 83.6 (vs); narasimhatāye 83.8 (vs); purusasimhatāye 83.6 (vs); narasimhatāye 83.8 (vs); pararajyāye ii.193.19 (vs); āharanatāye 399.7, 15 (prose); paricaryāye 426.17 (prose); raksāye iii.48.12; nirvidāye 331.4; sahavratāye iii.223.5, 6, 9, 11; 322.12 (all prose; the form sahavratāyai also occurs, e. g. Mv ii.118.3 = LV 238.16; see sahavratā in Dict.); tāye i.60.11.

9.39. Ablative: these are still rarer: paścimāye Mv i.6.9, dakşiņāye 10, utlarāye 10 (but in the parallel i.6.8 pūrvāya); māyāye i.356.13, abl. of comparison, 'than Māyā'; Uruvilvāye ii.207.1 'from Uruvilvā'.

9.40. Genitive: these are quite common; I have recorded about 150 cases, of which the following are a sprinkling. All are from prose except those marked as from verses: lohitikāye Mv i.49.14; māņavikāye 233.14; anyāye ca janatāye anekasahasrāye parisāye 255.6; Ganāye 261.17 et alibi; dārikāye 353.2; bhāryāye ii.65.3 et alibi; sujātāye grāmikāye 131.10; gaņikāye 168.14 et alibi; syāmāye 173.8; vadhukāye sudaršanāye 445.12; kubjāye 459.5 (prose), 6 (vs); prajňāye iii.33.20; avidyāye ... trşnāye 65.19; anāthāye 132.15; devalāye 165.13; anyāye velāsikāye (for vai^o) janatāye 223.2; sadevamānusāye parisāye 375.10; parivrājikāye 389.18 et alibi; pravrajyāye 450.14; tāye ii.65.12; 110.5; etāye ii.177.1, 2.

9.41. Locative: somewhat less common than the inst. and gen. but by no means rare; there are at least several score of cases in Mv. Again my citations are all from prose, except as specified: paścimāye Mv i.6.8, pūrpāye 9, ultarāye 10 (but in 11 daksināyamıl two interior mss. reading °nāye); sudharmāye devasabhāye 32.13; karuņāye 175.4 (vs); tāye velāye 237.15; 267.11 and often (also iha velāye ii.108.9); dārikāye i.342.15; kanyāye ii.72.18; prabhātāye 112.9; 167.7; 449.10; sītalāye jambuchāyāye 208.17; urupilvāye 231.7; 263.15; kalamāye nişidāye (Dict.) 432.8; vīņāye iii.34.18; gangāye 163.10; 166.6; parişāye samāgatāye 393.16; bhiksāye labdhāye 420.10; tāye ii.65.13.

9.42. Obl. sg. -āya and congeners. This ending is evidently identical with Pali -dya, and corresponds to M. -da (alternative to -de). It is common in verses of most texts, extremely rare in prose except that of Mv, where it is about as common as in verses (but in both much less common than -dye). It may be noted, however, that there is some reason to believe that originally other texts than Mv used it more extensively in prose also; the Kashgar recension of SP, for example, shows it where the Nepalese text has Sanskritized.—The variant -aya occurs only m.c.; but $-\bar{a}y\bar{a}$ seems, if the mss. can be trusted, to be found also in prose. It will be recalled that Skt. has a long vowel in the ultima; this may be older than $-\bar{a}ya$ with shortened -a, that is, an independent phonetic development from Skt. $(-ay\bar{a}), -\bar{a}y\bar{a}x, -\bar{a}y\bar{a}m$. — All oblique cases may use this ending, and we shall subdivide by what seem to be such syntactic distinctions.

9.43. The few occurrences noted of $-ay\bar{a}$ are clearly due to metrical considerations only.

9.44. Ablative: read with Kashgar rec. -pīdayā SP 117.8 'from oppression' instead of text with Nep. mss. -pīdilāh, which can hardly be interpreted. Tib. gzir las supports abl.

9.45. Genitive: śubhakarmanişihayāḥ (two mss. including the best ⁹yā) LV 29.2 (vs); Gangayā Sukh 8.11 (vs); tad eva ... phalam ... himsayā (v.l. himsaya) Mv i.184.10 (vs); gen. acc. to Senart (could it be inst., 'the fruit (obtained) by a harmful (violent) act'?).

9.46. Locative: dakşiņayā SP 191.5 (vs). See also -ayām, § 9.79.

9.47. The ending $-\bar{a}y\bar{a}$ (see § 9.42) has been noted as follows. Note that, except in the inst., the regular Sanskrit forms would be just as good metrically.

9.48. Instrumental: samo vimšaligangāyā SP 304.3 (end of an anustubh first pāda; followed by p-; gangāya occurs in the vicinity; tho in a vs. not m.c.); bhāryāyā sārdham Mv ii.443.8 (proše, no v.l.); certainly inst. (misprint for bhāryāya or bhāryayā? Elsewhere Senart sometimes emends such forms; see below); -locanāyā Av i.265.5 (end of tristubh; the first ā, but not the second, metrically necessary); pramudilāyā Gv 256.11 (end of line; as preceding).

9.49. Genitive: mallikāyā SP 362.4 (vs, end of pāda, before t-; meter does not demand final -ā); gopāyā LV 159.10, prose, all mss., also Weller's ms. 1 (Weller 27); next word begins ś-; prajñāyā 164.13 (vs), m.c.; nairañjanāyā 271.8 (vs), m.c.; may be loc.; trṣṇāyā 204.21, prose, all mss.; followed by s-; may be inst., or abl.
9.50. Locative: kālakriyāyā (WT °yām with ms.

9.50. Locative: $k\bar{a}lakriyay\bar{a}$ (WT °yām with ms. K') samupashhilāyām SP 384.2 (vs); parisāyā Mv i.74.21 (prosel but v.l. °ye; Senart here em. °āyām; contrast next); gangāya ca yamunāyā ca antarā Mv iii.363.19 (prosel so Senart with one ms., but the other has yamunāya, matching gangāya; this is probably the true reading. Could also be taken as gen.).

9.51. Chiefly before a vowel, the ending appears as -āyam, the m being perhaps a samdhi-consonant. However, paddha-patikāyam seems used in Mv iii.258.16; 260.12 as instr. of °paţikā (Dict.) before consonants. Other instrumentals: girāya-m anupūrvašah Mv i.10.7 (vs); gāthāya-m (v.l. gāthāye) adhyabhāşati Mv ii.207.4; prabhāya-m abhigphnişu (Senart em. prabhāya atigrhyate) 307.7 (vs).

9.52. Of quite different nature is the ending -ayam, before a consonant, once in vahyalo (= ba°) devasabhāyām Mv i.32.12 (prose) 'outside of the gods' assembly'. Here the regular Skt. loc. is used as abl. or gen., a retroflex of the common use of -aya for all oblique cases. Cf. § 10.141.

9.53. The only occurrences of -aya noted in prose, in texts other than Mv, are:

9.54. Instrumental: samdhāya SP 34.4; so text with Kashgar rec. and 1 Nep. ms.; paribhāşāya SP 60.14 'with censure'; so Kashgar rec., to be adopted; Nep. mss. vary; text paribhāşanayaiva; samudravelāyo° (for °velāya plus u°) Divy 234.13, according to mss., ed. em. °velayo°; mānuşikāya vācāya Jm App. 240.14.

9.55. Dative: samprakāšanalāya, Kashgar rec. and Lūders ap. Hoernie MR p. 152, twice, for ed. of SP 270.11 and 271.1 °tāyai (with Nep. mss.); abhijāāya LV 416.18, certainly dat., and corresponding to Pali abhiāfāğya; in the dharmacakra-pravartana, cf. Vin. i.10.16.

9.56. Genitive (or Dative): Mātangāya Māy 245.3,

and a series of others in the sequel. In a long list of gens. (or dats.) fem.; all the ā-stems end in -āya except Amŗlāye 245.1.

9.57. Locative: devānām sabhāgatāyopapatsyante SP 478.4 and (°syate) 11; in this idiom loc. is regular, see Dict. s.v. sabhāgatā.

9.58. Further examples of $-\bar{a}ya$, all in verses except those (from Mv only) marked as prose, are:

9.59. Instrumental: Irsnäya SP 48.2; ksudhaya 86.2 (parallel is dähena); anukampäya 147.4. — mäyäya LV 28.18; prabhāya 198.10 et alibi; väcāya ... garjitāya . . samudgatāya . . . subhāvitāya 419.1–3. — ehibhiksukāya Mv i.2.15; ii.234.2 (both prose); bhujāya Mv i.217.16 = ii.19.17; not dat. of bhuju (Senart on i.149.14) but inst. of bhujā; in i.149.15 (the same line) all mss. bhujābhir, which must be kept and which proves my interpretation; mānuşikāya vācāya i.271.18 (prose); iii.72.20 (prose); bhāryāya ii.64.19 (prose); gālhāya 78.20; 207.7 etc. (prose); lākhatāya 208.2 (prose); bubhuksāya iii.158.10 (prose); yācāya madhurāya 266.8 (vs); imāyānumodanāya 426.6 (prose); tāya i.244.7 (prose); iii.294.15 (vs); in both v.l. läye. --- sumägadhäyopanimantrilah Divy 402.6. Meter requires -āy-; in prose 402.1 sumāgadhayā. — virocanatāya Suy 49.8. - - prabhāya RP 55.13. - vācāya Ud vii.10, in the oldest text form; replaced in later recension by vācā su-; vācayā would be metrically impossible; prajňāya xviii.11; in later rec. prajňayā, with change in order; śraddhāya x.5 (in later rec. śraddhayā; meter indifferent) and xix.2. — alulāya Sukh 50.10. — prabhāya Gv 254.13. 9.60. Dative: durgāya subhairavāya SP 196.2

(could be other obl. cases); $s\bar{u}n\bar{q}ug$ subnaradaya 37 190.2 could be other obl. cases); $s\bar{u}n\bar{q}ya$ 280.4 (could be other oblique cases). — lokānukampāya Mv 1.40.4 (prose). In an oft-repeated formula, e. g. Mv 1.50.7; 60.9; 229.18; 239.2; 330.8; 336.2; some mss. so at i.331.8, v.l. °pāyair, Senart °pāyai; at i.332.12 mss. also °pāyai, no v.l.; at i.333.3, 16, mss. °pāyair or °pāyaiħ, Senart °pāyai; at i.336.11 °pāyai without v.l.; anultarāya samyaksambodhaye i.46.11 (prose); pravrajyāya iii.176.13. — siddhikarāya etc. (a long series of forms with this ending) Suv 110.12–111.14. nirvidāya KP 126.10, 15.

9.61. Ablative: paramparāya SP 352.2 (cf. paramparālah in 351.11). — pūrvāya bhillīye 'from the east wall' Mv i.6.8 (prose; cf. lalālo in line 11); anukampāya (could be inst.) 187.16, where read sānukampāya (sa, pronoun, plus anu°) nāyakaḥ, with mss.; Senart wrongly em. spoiling meter as well as sense; śraddhāya ii.62.18. — .deśanalāya, -yācanalāya Bhad 12, could be inst., but v.l. -āyālı, suggesting abl.

9.62. Genitive: ūrņāya SP 8.11; parsāya 24.8; 272.12; mritikāya 51.1; gangāya 253.3; 255.4. — māyāya LV 29.5; karuņāya 224.1; senāya 311.11. — Mv (all prose except the last) jīvamānāya (agrees with mālinīye) i.311.16; abhayāya (in same line abhayāye) ii.26.5; pāragāya 210.10; hrştāya 430.18; dehalāya (in 17 Senart °āye with v.l.) 437.14, 15, 17; gulikāya iii.7.19; nisaņnāya 257.15; elāya i.311.15; ii.65.15; yāya iii.2.18 (vs). — mātur ... janetukāya (vv.ll. °ye, °yam; Nobel em. °yāh) Suv 238.1 (vs. end of pāda). — trṣnāya Ud xvii.9; the vs is cited Divy 56.13, with rearrangement, reading tṛṣnāyāh.

9.63. Locative: tasyām cīrņāya caryāyām SP 30.15; purimā-dišāya 9.3; parşāya 36.9; kofisahasrāya ca dhāraņīye 330.4 'and in the dhāraņī that makes 1000 crores (of revolutions)'; cf. SP 327.8; misunderstood by Burnouf and Kern; parisāya 353.5; agnikhadāya 448.5. mayi ... abhiratāya susamprtāya LV 42.9; mahāvyūhāya sthitah samādhiye (fem.!) 60.4. — nāvāya MV i.264.14 (vs); kalyāņadharmāya pativratāya, prahāya (chandam? so Jāt., om. in MV) (su-, MV, perhaps omit) kuladhītarāya ii.58.1-2 (vss); these forms are found in the mss. and wrongly emended by Senart; the Pali Jāt. v.403.4-5 proves that locs. are meant; trisāhasrāmahāsāhasrāya (lokadhātūya, see mss. reading) ii.281.16 (prose); rudamānāya 427.17 (prose); nişidāya (Senart em. nişadāya) 432.9, 10 (prose). samalāya Samādh 19.34 (v.l. °ye, but meter demands short). — jinendraghoşāya Suv 147.1.—cariyāya Bhad 45. — catuthāya Dbh.g. 16(352).8.

9.64. Obl. sg. -**ā**. Likewise common to all oblique singular cases (none have been noted for the dative; but see § $36.18 \ p \bar{u} j a n \bar{a}$) is the ending - \bar{a} . It is much rarer than - $\bar{a} y a$, tho in most texts no rarer than - $\bar{a} y a$ (this being little used outside of Mv). It is found in Pall, Geiger 81.1; Geiger regards it as a contraction of - $\bar{a} y a$ (27.2). Apparently not known in Prakrit. Occurs only in verses except for one prose case in Mv. Several times (SP 92.6; 93.7; 191.5) the \bar{a} is shortened m.c.

9.65. Instrumental: paramparā eva tathānyamanyam SP 209.5. Certainly an inst. is meant, but Kashgar rec. paramparāya (meter would require °yā!) tatha anyamanyam; abhijňa jñālvāna (ed. jñātbā na) SP 93.7 (vs; so Tib., mion par šes pas, instr., dig nas). — (avikopitayā) caryā LV 259.14 (so all mss.); kāyaprabhā (must be inst.) 291.2; maitrā vijitya ... mārapaksām 343.6 'having conquered Māra's partisans by love'. — tvarāgatya Suv 233.6 (no v.l.) 'coming hastily'.

9.66. Ablative: dviliyā bhūmilo Mv i.91.1 (prose) 'from the second stage'. One good ms. dviliyāyā, but the other five dvilīyā; certainly abl.

9.67. Genitive: upagami līru niraājanā (no v.l.) narendrah LV 243.12 'the prince went to the bank of the N. (river)'; gangā yathā vālikā(h) 283.20; 284.1; 360.19 'like the sands of the Ganges'; *Irsna* SP 92.6 (vs), see Dict. s.v. samudāguma (2).

9.68. Locative: evam diśā dakşinayāpi latra, alha paścimā hestima ultarasyām SP 191.5. The form ultarasyām (not to speak of latra) establishes diśā, paścimā, and hestima (short a m.c.) as locs. — paśya deva kumāro 'yam jambuchāyā hi dhyāyati LV 132.7; '... for he meditates in the shade of a jambu tree'. Lefm. °chāyāhi, presumably as inst.-loc. pl.; but sing. is far more natural. Most mss. °chāyām hi, but A (the best ms.) text. In LV 421.17, read with best mss. dašadišala (yi = ye). Perhaps loc. sg., with short a m.c. as in hestima SP 191.5 above. It might, however, perhaps be acc. pl. 9.69. Special forms of oblique cases. Finally, a

9.69. Special forms of oblique cases. Finally, a few forms which seem to be peculiar to particular cases. In the instrumental, there are to be noted only some curious blend formations. First, two borrowings of the masc. ending *-ena* (cf. below under abl. and loc.), once substituted for the fem. *-ayā*, and once blended with it into *-ayena*! Monstrous as the latter seems, it appears to be well authenticated and guaranteed by the meter: ksulfrşnayena parilaplās RP 32.16 (vs), 'tortured with hunger and thirst'. Only trşnā 'thirst' can be concerned, and there seems no reasonable ground for questioning the reading. *— animisatena* (all mss.) Divy 222.22 (prose); can only intend inst. of animisatā, depending on anyatra, 'except for the nonwinking-ness (of Indra)'.

9.70. Another form, recorded in the mss. of Mv i.158.13 (prose), seems to be a blend of the regular Skt. -ayā with the Mindic -âye, producing -ayāye: caluranginīyā senayāye (so five mss., including the two best; one inferior ms. senāyā; Senart em. senāye).

9.71. Parallel to the inst. in *-ena* and loc. in *-e*, we occasionally find ablatives in (masc.-nt.) *-dt* from fem. \bar{a} -stems. Note that in Prakrit such stems may have abl. *-allo*, according to the grammarians; this is explained by Pischel 375 as analogical borrowing from masc. and nt. nouns. *sayyāt* LV 139.14 (vs), both edd. and all mss. except A which is hopelessly corrupt; mudrāt 176.15 (vs), both edd. and all mss. but one; but mudra m.-nt. is common in Mmk (Dict.).

9.72. Much commoner are ablatives in -ālas (-ālo,

65

9.72]

and m.c. -ālu, -atas, -alu). This is clearly a Sanskritization of the regular Prakrit ending -ao (-ado); Geiger does not speak of it as occurring in Pali. It is hardly found in prose except, as usual, in the Mv; only one other prose case has been recorded: svapratijhātas cyutah RP 58.7. Prose occurrences from Mv are: sasthilo bhûmilo i.127.14; saplamāto 136.4; nāvālo 217.7 'from the boat'; abrahmacaryāto 326.15; pišunavācāto ii.99.9; taksasilāto 167.1; 175.7; diśāto 426.5; iii.210.15 f.; puşkiriņikāto ii.438.5; uttarāto iii.210.15 f.; parisāto 393.6.

9.73. Examples from verses: paramparatah SP 351.11: dvitīyāto Mv i.91.7; pratisamvidāto Šiks 109.6.

-āta, m.c.: trsnāta LV 420.4.

-alas, -ato m.c.: avidyatas Suv 60.6; 61.2 (1 do not understand why Nobel calls this gen.); prajňalo RP 55.13. -atu, m.c.: daša-dišatu LV 421.13; vácatu Bhad 23.

9.74. In the gen. sg., final visarga may be omitted

before a voiceless consonant: tato 'syā khalu (so all inss., Calc. 'syah) LV 28.16 (vs). (Pronominal form.)

9.75. Further, the masc.-nt. ending -asya occurs repeatedly in pronouns which refer to fem, nouns. I have failed to note any similar nominal forms; but the use of -asya here is evidently parallel to the other masc.-nt. endings noted in fem. a-stems. In some cases, to be sure, the final -a is doubtless only shortening for $-\bar{a}(h)$, in verses m.c., usually, tho not always, before a vowel or voiced consonant; so in the following group: niścarałe 'sya ātmabhāvāt LV 49.18; °śravāsya mūrdhajāni 49.20; tathāsya netre 49.21; yasya (followed by ahu) 79.21; tasya (followed by s-) 139.11.-Mv i.203.13 and 19 = ii.7.9 and \$15 pramadāy' imasya (yalha) etc.; in i.207.14 = ii.12.4 and 18, read asya each time with mss. (in i.207.14 Senart em. asyā), and read for the next word sarva with v.l. on ii.12.4 (instead of sarve or sarvam, unmetrical). Not metrically required is (udgatā-)sya at the end of a line of vs LV 50.1, where Lefm. cites no variant, not even that of Calc. which reads -syāh; since it is Lefm.'s habit to report Calc. readings, may he have failed to record this reading in some of his mss. too?

9.76. However, there are quite a number of prose occurrences in the mss. of Mv, tho Senart almost always emends to -asya(h). The following list is certainly far from complete, but long enough to raise doubts of the propriety of emending the form out of existence: Mv i.353.2 tasya guhāye; il.459.5 lasya kubjāye; ili.56.10 lasya (kept by Senart here) dāni šārīye; lasya dāni mrgīye 150.21; lasya dāni ... 163.4; imasya 162.1; asya 163.4.

9.77. In the loc. sg. we find fem. a-stems showing the masc.-nt. ending -e. Cf. the inst., abl., and gen. forms -ena, -āl, -asya above; also AMg. giriguhamsi for °guhāe, stem -guhā, Pischel 375. This AMg. form, as Pischel notes, is associated with masc.-nt. forms in -amsi and perhaps gets its ending by attraction to them. So in the following case, upekse may be attracted to the neighboring kārunye: kāruņye varamudila upekse LV 164.15 (vs), 'thou (hast) well delighted in compassion, in renunciation'. Can only be a loc. of upeksā. There is a v.l. upeksaye (metr. impossible). Similarly dhyane prajne LV 162.4 (vs); prajne for prajñāyām, perhaps assimilated to dhyāne.

9.78. But there is no parallel form in -e adjoining the following: vanalate antarhite Mv i.342.4 and 7 (prose). Can only intend loc. of °lalā. In LV 291.2 (vs) dašadiše 'in the ten directions' we have the option of taking dise as loc. sg. of disā, or regarding it as for disi (stem dis) with e for i m.c. (§ 3.60).

9.79. The regular loc. ending has penultimate \bar{a} shortened m.c. at least once: rāmakaksayām (stem °kaksā) Māy 83 (vs).

9.80. The loc. ending appears with Mindic shortening of the ā af the ultima, as in regular Pali -āyam (not in Prakrit); but I have noted this only in Mv. The following

examples are all from prose except those noted as vs: daksināyam Mv i.6.11 (parallel with paścimāye, pūrvāye, ultarāye); bodhisattvacaryāyam 105.3 (with mss.); bubhuksayam aneti 188.17 (vs; here meter is improved by the shortening); -krtayam ii.38.11 (vs; meter not concerned); parāhatāyam 282.13; lekhāyam 423.15; 434.10; samkhyāyam ... gaņanāyam ... mudrāyam 434.11; ekamātā jam 450.18 (mss.: Senart em. °ya); the same occurs as v.l. in the repetition 453.6 where Senart adopts "mātāya with one ms.; parisāyam iii.139.20 (vs, but end of line; so mss.; Senart em. [°]yām); nāvāyam 149.17 (tollowed by ā-); maājāsāyam 167.1; sayyāyam 275.21 (vs).

Nominative-accusative plural

9.81. As in Skt. the nom. and acc. pl. forms of this declension are identical for the most part in our dialect too. It seems best to juxtapose nom. and acc. forms showing the same ending; but we shall separate the two cases in listing examples.

9.1 2. Nom.-acc. pl. -ā. The regular Pali and Prakrit ending -a, lacking s or visarga, is quite common in the mss.; editors often emend. Meter, of course, cannot be primarily concerned here; yet in most texts (Mv is an exception, as usual) forms in $-\bar{a}$ seem to occur much more often in the verses than in prose, according to the mss. So far as I can see, there is no special tendency to drop the s before any particular consonants (as e. g. sibilants); -ā is quite common in sentence final. This -ā may be, in verses, fused in samdhi with a following vowel initial; examples below.

9.83. Examples from prose, nominative: sarva-ratikrīdās copasamhartavyā strī⁰ . . . LV 193.1 (so all mss., only Calc. °hartavyāh). — aparyantāh bodhisattvena pāramilā (2d ed. °lāh; by em.?) paripūrayitavyāh Gv 460.6. ---prāptā Mv i.147.13 (end of sentence); yatlikā ... kanyā ii.149.21 (pl.: both followed by voiceless stop); antahpurikā ca sarvā nirdhāvilā 449.13 (all pl.).

9.84. From verses, nominative: gopānasī vigadila tatra sarvā SP 83.5 (pl.); yā kāci medinyām jātā osadhayo bhavet 126.9; yā sativakoļyo 330.12 (yā all mss., both edd. em. yāh); upāsikā (tatra) 384.10, apparently read by all mss. but one, should therefore be kept; both edd. upāsikās. evam bahuprakārā sam° gāthā LV 13.4 (so all mss.; Calc. bahuprakārāķ . . . gāthāķ); jātakānksā 49.7 (all mss.; end of line); pramadā s- 164.19; rutaghosā t- 167.12; -bhūșitā p- ... nāriyo praharșită 170.3-4 (end of line); sarvā pramadānu^o 183,12. — samnipatitā My i.217.18 = ii.19.19 (end of line); prasannasamkalpā ii.19.20 (end of line); but in the same line i.218.1 °samkalpah. -- sarvā striyo Suv 42.4. - pūjā Bhad 6º (pl.); yā (= yāh) 6º, 6º, 7; udára 7.

9.85. From verses, accusative: kalpakolyo acintiya SP 30.15 (end of line); tārakā 139.3, probably to be read with Kashgar rec.; Nep. mss. tārakān (with masc. ending); ed. em. tārakāh (below, 139.11, Kashgar rec. reads tārakāh; Nep. mss. again °kan); yady abhijñabhinirhareh 141.11 (for abhijñā(h) abhinir°; otherwise WT); -gocarā 279.5 (so read with Kashgar rec. and WT at end of line, for KN °rām; acc. pl.) -- gāthemā LV 163.15, for gāthā(h) imā(h); pramadānuśiksitā 183.12, for pramadā(ḥ) anu°; kāmakrīdā ca- . . . vijahitvā 184.7.

9.86. Nom.-acc. pl. -a. A shortening of the ending -ā (for -ās) to -a occurs only in verses where meter demands a short syllable, but is quite common there. Examples of the nominative: parsa calasra SP 25.1; vigadila 83.5; ksudrīka (osadhīyo) 129.7; sarva (osadhīyo) 129.8. — vividha LV 164.5; pramudita ratikara pramadā (pl.) 164.19; gātha 167.11; 175.16; prasannacilla premajāla nāriyo 170.4; dhīta 170.13; abhijňa pañca ... labdha 172.20. — yāttika vālika Samādh 19.16. — abhinirhrta pūjā(h) Bhad 6³; anuttara pūja 7.

9.87. Accusative: dišo daša (WT dašā, m.c.) sarva SP 91.8; abhijāa jāātvāna (so read) 93.7. — ima LV 162.8; smarāhi ... lā ca pūja 172.17; rathya bharila ... krivā 240.18; māya 326.18; daša disala (yi; so read; stem °tā) 421.17. — vālika gaigāye Mv ii.370.21 (so read; mss. vārika; Senart em. vālikā, metr. poor).

9.88. Nom.-acc. pl. -āyo. The ending -ayo (regularly in this form, without regard to the following sound, and even final in sentence or line of verse; -ayah noted only SP 15.4, perhaps editorial error?) occurs fairly often in Mv (nearly all the examples are from prose) and a few times (in verses only) in SP: if it is found in other texts I must have overlooked it. It is evidently identical with the same ending in Pali (Geiger 81.3 and note, 'analogical to *i*-stem forms *ratliyo*, *kumāriyo*'). Prakrit has regularly -ao, which might also be connected with this, as the BHS oblique sg. -aye, -aya are surely related to Pkt. -ae, -aa. Pischel 367, 376, to be sure, identifies the fem. nom.-acc. pl. -ao with the same ending in the masc. (where however it is much less common than in the fem.!) and derives both from -a(s) plus -as, the ending of the consonant stems. Possibly the Pkt. -ao had more than one origin in the fem.

9.89. Nominative (all from Mv except the first four): yebhir disāyo daśa šobhitāyaļı SP 15.4; disāyo 31.11 (at end of a line); analpakāyo 251.11 (at end of first pāda, before s-; WT analpikāyo with v.l.); anvitāyo (WT with ms. K' añcitāyo) 306.11 (before p-). — anekāyo Mv i.6.7; -samkalikāyo 22.2, by Senart's em.; mss. -āyā; see \$ 9.91; edakāyo 25.10; nipatitāyo 26.3; devakanyāyo 216.10 = ii.19.7; tāyo dirghikāyo channāyo i.249.9 (others in lines 10, 11); niryāntāyo (mss. nirvā°) 272.8; vivāhitāyo 351.9; -kanyāyo ca dinnāyo 355.11; dārikāyo jyeşihatarikāyo 356.13; dārikāyo 356.15, 19; sarvāyo 356.19; iii.442.12; karņikāyo ii.36.7; anekāyo ... samāgatāyo 74.10-11; šabdāpitāyo 110.14; upasthāpitāyo 433.11; anlahpurikāyo kuśalāyo iii.162.13; antahpurikāyo 165.7; 166.2; jyeşihāyo 167.19; bhāryāyo 222.11; chinnāyo 358.5; lāyo (besides i.249.9 above) i.6.12; ii.54.15; iii.287.3, 15, 18; etāyo iii.72.14.

9.90. Accusative (all from Mv): divyāyo i.32.16; iii.323.17; dišāyo ii.62.11; antaķpurikāyo iii.166.1; bhāryāyo 222.2; -striyāyo 283.5.

9.91. Nom.-acc. pl. -āyā? As stated § 9.89, in Mv i.22.2 (prose) the mss. read -*samkalikāyā*, nom. pl., which Senart emends to -*āyo*. I have noted no other case of the ending -*āyā* as nom. or acc. pl. Yet in the *i*-*i* declension there seem to be nom.-acc. forms in -iyā, -iya (as well as -*iye*, -*iye*; cf. -*āye* below). Possibly -*āyā* also should be admitted here.

9.92. Nom.-acc. pl. -āye? Cf. the preceding; some support for the reading of the mss. in the following cases seems to be furnished by the -*iye*, -*iye* forms of *i*-stems, §§ 10.175 f., tho they are none too common. Nominatives (all from Mv prose): prānasahasriyo (but one ms. °srāye!) ... uparuddhāye (so mss., Senart em. °āyo) ii.98.17; rainamayāye (Senart em. °āyo) puşkariņīye (Senart em. °īyo) ii.177.16; but here it is not certain that a nom. pl. is meant; it may be an oblique (loc.?) sg.; panditāye (Senart em. °āyo) iii.393.11; the noun is striyo or, one ms., striye, nom. pl.

9.93. Nom.-acc. pl. -åvo. Unparalleled elsewhere is the nom.-acc. pl. ending -åvo, occurring a number of times in Mv. It seems necessary to regard it as somehow secondary to -ayo, that is a dialectal variant of the latter. On -v- for -y- cf. Geiger 46, Pischel 254; and for Vedic and Sanskrit, Wackernagel I 188c, Ved. Var. II 246 ff. (the Taittiriya school seems to favor v for y of other schools, particularly in suffixal forms with vin for yin); my §§ 1.28,84. --Senart began by emending to -ayo, but in his note on iii.149.12 wisely retracted the emendation. All the forms are from prose, and all nom. except the last (acc.).—Once a masc. form, avusavo, used as voc., is recorded: § 8.89. lavo (twice; = las; one ms. laro) Mv iii.149.12; anlardyakarāvo 149.13; samnipatitāvo 364.1; samāgatāvo 392.7; dhātavah tişhamānāvo (note the preceding °vahl but no such form accounts for the others) tişhanti iii.65.11; (accusative:) tāvo antaļpurikāvo iii.357.18.

9.94. Nom.-acc. pl. -e (-i). Rare, but seems to occur, perhaps as nom. (at least of pronouns), and more probably as acc., pl. fem.; doubtless a transfer from the masc. a-stems, like -ena, $-\hat{al}$, -e (loc. sg.) etc. The following, and others like it in § 6.15, may be treated as confusion of gender-forms: ye vā istriya māya... LV 193.14 (vs) 'what women's wiles' (nom.); no v.l.

9.95. Accusatives in -e are perhaps a bit better attested; possibly devale LV 120.12 (vs), which is certainly acc. pl., may be associated with devalā rather than daivala. Clearer than this are: -janate LV 170.1 (vs), for janatā(s); most mss. janata, which is unmetrical; best ms. (A) janate iha drsįvā 'seeing the people here'; dašadiše (sarve) LV 291.2 (vs), probably acc. pl., for -dišāh sarvāh, 'thruout (all) ten directions'; so also (or m.c. dašad-diše or the like) SP 217.13; LV 299.11; 365.2 (all vss; in the last read prob. dašad-diše and before it krito with many mss.); cf. dišebhir § 9.104. — sarve (or, with v.l., sarvām) vāpīyo Mv ii.449.5.

9.96. Apparently -i, m.c. for -e, occurs as either nom. or acc. pl. (either in relative clause, or object of prajāni, aor. to prajānati), in caitasika-samjāi (to -samjāā) LV 151.13 (vs).

9.97. Nom.-acc. pl. -o? Twice the text of LV seems to present (in verses) a form disalo, apparently acc. pl. of disalā = dis(ā) 'direction' (see Dict.). Influence from the ending -as of consonantal stems? obhāsesī daša ima disalo LV 162.8, 'thou illuminest these ten directions'. So both edd.; most mss. disilo; all ending in -lo. alha dašasā disalo jinalejair gālha vicilrā ... ravisā LV 167.11, 'then by the power of the Jinas throughout the ten directions various verses resounded'. So Lefm.; all mss. except A (the best) dašadisalo (one syllable short); A dasuša (intending dašasu? should be 'sā m.c.). If Lefm.'s text is right, a loc. numeral modifies an acc. (?) noun (§ 7.6). With this cf., for the acc. case, dišo dašāsu LV 314.8 'in all ten directions'; but here Tib. phyogs-phyogs dag-tu, with no numeral, suggesting dišo-dišah, or some form of an āmedita cpd. of the word for 'direction' (dišo-dišāsu, dišā-dišāsā).

9.98. Nom.-acc. pl. -āni. The neuter ending of the nom.-acc. pl. of a-stems is repeatedly recorded, in Mv prose, in unmistakably fem. ā-stems: bodhisaltbacaryāņi Mv i.1.2 (nom.; no v.l.); in next line °caryā, nom. sg., several times; anyāni te rājakanyāni yaltakāni jalpasi tattakāni ānayişyāmi, ii.101.2-3 (acc.) 'I will bring you other princesses, as many as you say'; etāni šivikāni (nom.) 107.4; nişadyāni śayyāni (nom.) 163.18; -krīdāni (acc.) 171.5; catbāri ca pratisamvidāni (acc.) iii.67.2-3 and 4; bahāni māyāni (acc.) Mv ii.174.12; tasyā nabhe prajanto stavayimsu apsarāņi (nom.) LV 223.8 (so read with all mss. but one; Lefm. apsarāņām, uninterpretable) and others, § 16.24; jatāni (nom.) Mv iii.147.2; -mallikāni (nom.) Mv iii.80.4; others in Mv, Dschi, NAWGött. 1949, 273 fl.; see § 6.4.

9.99. Acc. pl. $-\bar{a}n$, $-\bar{a}m$. The acc. pl. of \bar{a} -stems is repeatedly recorded with the masc. ending $-\bar{a}n$, $-\bar{a}m$ (the difference is probably purely orthographic; $-\bar{a}m$ is a semi-MIndic writing for $-\bar{a}n$). Of course this cannot be concerned with meter. In fact it occurs in prose, even of other texts than Mv, and is commoner than most other masc. endings in the fem. \bar{a} -declension. saśramanabrāhmaņīn prajān sadevamānusān LV 3.6 (prose), both edd. The -i- in °brāhmaņīn (two mss. °ņān) shows that prajā is still a fem., and there is no reason to emend to sing. forms in -m. *işķikān* 'women' LV 79.20 (vs); *lipino* (see § 10.155) 'kşara-

9.99]

5

68

drsyarupām (acc. pl.) 125.12 (vs); disām (stem disā) cālokya cature (for °ro, or rather catasro) 252.21 (vs); svām (so all mss., both edd. em. svā) duhilfn (or duhilari, mss.; acc. pl.) 320.1 (prose) .-- In LV 322.16 ff. (vss), every line down to 323.9 incl. ends in a fem. adjective or noun in -ikām, usually read by all mss. and kept by Lefm. All are certainly plural, referring to the daughters of Mara. So Tib., and cf. 323.2 imās, all mss., em. wrongly to imā by Lefm. Cf. also the following: yah (all mss.; Lefm. ya) souyam āgalikān (Lefm. with v.l. āgaliko, which seems uninterpretable) na hi bhuājasi kāminikām 323.14 (in continuation of preceding passage), 'who dost not enjoy charming women that have come of their own accord to thee' (Skt. svayam āgalāh, acc. pl. fem.; cf. also next); tām (all mss. and Lefm.) māraduhitīrn 323.20 (prose); tāms ca 378.16 (prose); so all mss., both edd. tās; refers to daughters of Mara. - deva-devalam Mv i.245.7 (prose; dvandva cpd.); samudradevatām 245.9 (prose; tatpuruşa cpd.); (kanyā, for kanyās, nom. pl.), tām sarvām kumārasya upadarśayāmi ii.149.21 (prose); tādršīm kanyām (plur.!) 441.5 (prose); sarvām vāpiyo 449.5 (prose; or sarve, with Senart and v.l.); anlahpurikām (once v.l. °kān) 449.11; iii.163.12; 165.15 (all prose); (dhilaro) suvarnasahasramanditām iii.24.8 (prose). — tān (sc. pūjāh) Bhad 62, 63, 7 (vss; — (buddhavyūhān) buddhavrsabhilān before vowel). (buddhavikriditam) Gv 17.24 (prose). Possibly read ^evrsabhitam (acc. sg.)? Note preceding pl., but following sg.

Vocative plural

9.100. Voc. pl. -āho: as in masc. a-stems, § 8.88. Possibly this is to be regarded as a transfer from masc. to fem. declension; Pischel does not cite -aho in the fem. \bar{a} declension from any Prakrit dialect. But note deviho, voc. pl. of devī (§ 10.192). antaḥpurikāho Mv ii.452.11 (prose).

Instrumental plural

9.101. Inst. pl. -ābhi. The final consonant of the ending -ābhis may be dropped in verses, m.c., as in: ślaksnābhi vägbhir Mv i.84.5 (vs); gāthābhi ii.343.1 (vs).

9.102. Inst. pl. -āhi. On the other hand, the regular Pali (and Prakrit, beside -āhim, cf. Pischel 376 and our § 8.108) ending -āhi is fairly common in Mv, in prose as well as verses. I have failed to record it from any other text. The following examples are all from prose except those marked (vs): dālpiāhi samprajvalitāhi sajyotibhūtāhi Mv i.22.2; kartarikāhi 24.14; valgugirāhi 163.16 (vs); sthūlāhi 192.12 (vs); paripūrņāhi 192.14 (vs); citrāhi daršanīyāhi 194.4, 19; 249.5; (su)madhurāhi vācāhi 200.12 = il.4.9 (vs); dišāhi sarvāhi i.201.3 = il.4.19 (vs); rathacakramātrāhi karņikāhi 1.227.19; dirghikāhi 249.5; śvikāhi ... bhūşilāhi 261.11; fivikāhi 262.14 (vs); devatāhi 327.13 (prose); il.349.21 (vs); abhijāthi 1.354.5 (here used as ablative); sarvāhi kanyāhi il.48.5; tāhi 102.12 (vs); 110.3; -vicitrāhi nāvāhi 109.5; gatāhi 110.15; vividhāhi 126.15; 128.2; gāthāhi 157.10; -muktāhi 191.5 (vs); devakanyāhi ... padumutarāhi 192.15-16 (vs); etāhi šivikāhi 216.17 (prose); 227.6 (parņa-šivi°, vs); madhurāhi girāhi 226.10 (vs); kşirikāhi 248.16 (so read with mss., Senart °kehi); akşamātrāhi dhārāhi 349.17 (vs); chalradhvajapatākāhi 349.22 (vs); apsarāhi 359.22 (vs); sarvāhi ca antahpurikāhi 448.15; antahpurikāhi 463.16; -kubjāhi iii.4.19 (vs); tāhi ... sayitāhi 74.13; pramatāhi 79.2 (vs); nāvāhi, with adjectives agreeing, 149.18 f.

9.103. Inst. pl. -ais. As in other cases, the endings of masc. a-stems intrude with fem. nouns, in prose as well as verses. Thus, -ais: Lank (only in vss): vāsanair (from stem vāsanā) 285.5; 293.1; 296.5, 6 etc.; abhijāair vašitaiš (ca) 309.8; 318.8; stems abhijāā, vašitā. Cf. also Dict. s.v. devala (= devatā), where such forms are recorded.

9.104. Inst. pl. -ebhih (as in masc. a-stem nouns): da a-di sebhir LV 416.2 (vs), in ablative sense, 'from the ten directions.' There is a v.l. di sobhir (as if from the lexically cited stem di sas'). I assume ster. i d s a, with masc. ending; cf. § 9.95.

9.105. Inst. pl. -ehi: imehi sārūpyehi gāthehi Mv iii.379.20 (prose; so mss.; Senart em. -āhi); anyamanyehi (so mss., Senart em. -āhi) parivrājikāhi 390.5.

Genitive plural

9.106. Gen. pl. -āna. As in the masc. and nt. a-stems, the gen. pl. is often reduced to -āna, but only in verses m.c. A few examples: galāna avsarānām LV 49.5; -kanyāna 80.7; devatāna 221.15. — .nuktāna Mv ii.187.11; vaidūryamisrāna 187.12; divyakanyāna 192.1. — pramadāna RP 26.16.

Locative plural

9.107. Loc. pl. -asu: I have noted only one case of reduction of -asu to -asu in a verse, m.c.: sarvasu falişu Bhad 16 (Gv 544.14 reads sarvasu janmasu, which would make sarvasu stand for sarveşu).

9.108. Loc. pl. - āsū. Less rare is the metrical lengthening of final u, e. g.: diśāsū SP 53.11.

10. I-stems with \overline{i} and in-stems

10.1. In most of the declension there is such extensive confusion between i, i, and *in*-stems that clear separation between them is scarcely possible. The same holds in large part for the distinction between the three genders. It will be most convenient not to try to set up separate complete paradigms, tho in some parts of our combined list of endings, special groupings will appear; namely in the oblique cases of the singular, where some endings are characteristic of feminine nouns. When it seems possible and desirable, our lists of examples will separate words which ln Skt. belong to different declensional types. Even in Skt. there is some fluctuation, so the lines impliedly drawn by such divisions must not be taken too seriously.

10.2. As with *a*-stems, the bare stem in *i* or *i* seems sometimes to be used in the functions of various oblique cases; see $\frac{59}{10.65}$ ff.

Stem-extensions (in -a, etc.)

10.3. Even in Sanskrit (Whitney 441b, 1209c) instems are sometimes made a-stems by addition of -a to -in-. This occurs more extensively in MIndic: Geiger 95.2, Pischel 406. Undoubtedly it started from the acc. sg. -in-am, analyzed as -ina-m. The following have distinctly MIndic aspect. First, a stem-form in composition: hastinaprsihe Mv iii.178.15 (prose) 'on the elephant's back'.---Nom. sg. m.: maisarina iva daridrapurusah (no v.l.) LV 333.8 (vs); -gāminah RP 7.5 (vs); ālāpinah Ud xix.1 (vs; note that the corresponding Pali vs, Dhp. 144, also has ātāpino, but by further changes it is there made nom. pl.); dharmarājakulavamšadhāriņo bodhisattvakulavamšavardhi-nah Gv 487.5 (vs); hitakāriņah Mmk 32.24 (vs); -utlarāsanginah 45.13 (prose); šilpinah 58.3 (vs); jāpinah 103.3 (vs). Once even the stem kali, which is m. and seems not to be recorded as kalin, shows a nom. sg. kalinas, Lank 364.5 (vs).—Nom.-acc. sg. nt.: krlavedinam (so read for text *tam) ca tat kulam bhavati, achandagāminam etc. LV 24.1-3 (prose; with adjectives as in My i.197.20 f., ii.1.10 f., but always °gāminam; see § 10.31); vihāyasāgāminam vikurvanādharmiņam (hastiratnam, asvaratnam; cf. Mv i.108.16 f., § 10.46) LV 16.1, 10 (prose) .- Gen. sg.: vajrinasya Mmk 48.16 (prose); -karinasya 57.2 (prose); silpinasya 58.4 (vs). -Cf. -inā as nom. pl. of i-stems, § 10.156.

10.4. Yet more anomalous from the Skt. standpoint are a-extensions of monosyllable I-stems, resulting in stems in -iya-. They may be analogical to the acc. sg. śriyam etc., analyzed as śriya-m. They have been noted only in Mmk, the text of which, as presented in the ed., seems otherwise bizarre, even for BHS. For the proper name Mañjuśri this text repeatedly shows a stem mañjuśriya: as stem in composition, mañjuśriya-galadrsti 64.25, 65.3 etc. (prose). And case forms, nom. sg. mañjuśriyo mahāvīrah 60.21 (vs); gen. sg. mañjuśriyasya 1.5, 64.21, 69.1 etc. (these all prose).

10.5. Even for the stem *stri* 'woman' a stem *striya*is implied by *striyasya puruşasyāpi* Mmk 54.13 (vs); the meter is quite correct and seems to guarantee the form. It may be suspected that the following *puruşasya* was concerned in the creation of this nonce-form.

10.6. Similarly, and less surprisingly, such stems are

made to end in $-y\bar{a}$, keeping feminine gender; see the Dict. s.vv. *śriyā*, *striyā* (= *śrī*, *strī*); *Hiraŋyavatī* (°*tyā*); these may be Sktizations of MIndic forms.

10.7. Likewise only in Mmk occur a couple of forms in which the stem rsi (maharsi) is replaced by rsaya-, an *a*-stem, apparently based on the nom. pl. rsayas: maharsaya-siddha-sevilam Mmk 64.23 (prose), 'inhabited by great<math>rsis and siddhas' (stem in comp.); and even instr. pl. rsayai(h), in mahārsayai[r] anekamahārsi- Mmk 18.20 (prose; note hiatus in the compounds, and faulty samdhi between the words!). Cf. the nom. sg. ending -avah from an u-stem, also in Mmk; § 12.5.

10.8. The irregular stem sakhi, 'friend', has a stem sakhāya (analogical to Skt. acc. sg. sakhāyam), see Dict. s.v.

Stem in composition: confusion of -i and i

10.9. The above-mentioned confusion between stems in *i*, *ī*, and *in* is strikingly signalized by the substitution of *i* for *i* and *i* for *ī* as stem-final in composition (and before *-manl*, *-vanl*). Since Skt. *ī*-stems are nearly all fem., the latter change naturally affects only Skt. feminines. The former, however, is by no means limited to fem. nouns; it affects also masculines (*mani*, regularly masc.) and neuters (*asthi*), also masc. *in-stems* (*pakşin*, etc.) which in Skt. are indistinguishable from *i*-stems in compounds. (Note also the same change in indeclinables: *vimati* Samādh 19.35; *vitalha* Gv 55.3, etc., in verses.) Most of both types occur in verses, and could be interpreted as due to metrical requirements; but there are also prose occurrences. The latter are separated in our llsts.

10.10. i for I: prose, vaijaganti- LV 295.15, Lefm. with all mss., for °li-; padmini-vana Mv i.216.3 (v.l. padmini°). — Verses: mahipati SP 35.13; jagatišrestho and jagatipālās LV 81.8 and 9 (ed. prints jagati both times as separate word); rākşasidoīpam 169.2; padmini-(vanam, vane; v.l. padmini° both times, unmetr.) Mv 1.215.14, 16; nāri-(samgha-) ii.328.17; dharani-raso Suv 115.7; vyāghrisutān RP 22.5; sukhāvatikşetra Bhad 57 (in Gv 547.18 unmetrically sukhāvati°); etc., common in verses of most texts.

10.11. I for i (incl. in-stems): prose, jāānīvad LV 289. 14, all mss., also Foucaux's and Weller's; both edd. em. to jāānivad; certalnly jāānin. is the stem concerned; paksījāltnām Mv i.272.16 (to paksin); jālisahasrāni ii.169.3 (to jāti); hitaişī-anantakārī-karaņatāyai ii.264.15 (to hitaişin, -kārin). — Verses: kṣāntībalā SP 13.5; drşīf- 62.13 and 14 (but in 15 drşī;-, where meter demands a short); muktāmaņījālika- (maņi, normally masc.) 89.2; smrtimalīgaliupela LV 74.18; asthī-danta (to the neuter asthi) 208.17; matīmām Mv i.43.12, so read with all mss. (the line must be read viyākare arthadašī matīmām; Senart makes wrong emendations); jālī-maraņasya 256.6; rddhīmām ii.220.10; -gatīgala Samādh 19.17; drsīfgatam Suv 61.13; etc., common in verses of most texts.

10.12. Probably we might just as well include here forms in suffixal -ka added to what at least in normal Skt. would be short *i*-stems, which here show *I* before the suffix, where meter demands a long. I have cited such examples in $\S 3.17$.

(69)

5*

Nominative singular

10.13. In Pali, all *i* and *i* and *in*-stems, masc., fem., and neut., may have the ending -i; besides this, -i is found with masc. *in*-stems, and (more commonly than -i) with original fem. *i*-stems; and -im occurs with neuters. By contrast, in Prakrit, all masc. and fem. *i*- and *i*-stems, and masc. *in*-stems, have the ending -i (but Ap. -i, Jacobi, Bhav. § 27 etc.). Pischel knows no -i for any of these, except that 405 recognizes Ap. -i, and AMg. -i m.c., for -i of an *in*-stem; but it is scarcely conceivable that it should not also exist, m.c., for -i of other stems. Prakrit neuters have -i, but also -im; and -im (nasalized -i) is also recognized for the nom. of masculines by some grammariaus, according to Hem. 3.19 (Pischel 379 regards it as a phonetic equivalent for -i; is it not rather the acc. masc. and alternative nom, nt. used as nom. masc?).

10.14. In our dialect, the endings -i, -i, and -im seem to be used indifferently for all masc. and fem. i and i stems, and at least -i (as in Skt. neuters) and -im for neuters (where -I also occurs, but rarely). In most texts, to be sure, the 'irregular' forms are recorded only in verses, but this may be due only to secondary regularizing of the prose to fit standard Sanskrit. The prose of Mv, as usual, is not different from verse, and occasionally the MIndic forms occur in prose of other texts; moreover, they occur in verses in positions where meter is indifferent, notably at the end of lines, so that they cannot be explained as due to metrical considerations. Furthermore, a short final -i, for -is or -i, often precedes a vowel, where -ir would be just as good metrically; tho sometimes it is changed to -y in such cases, even when it is derived from a Skt. short i-stem and so represents Skt. -ir, as in: galy esam (v.l. gali ca) SP 177.3; bodhisallvasyaivam bhavali pratikrly evam dharmavihary ... abhūt LV 182.1 (prose), 'thus is the defense (against evil) of the B.; thus he dwelt in dharma;' all mss. pralikrly evam; only Calc. °krlir°; bhūmy ālamkrta (note ā-, m.c.) LV 276.17 (vs) 'the earth, adorned ...'; -visuddhy atha Bhad 43 (in same line the 'regular' visuddhih). In the following, all examples are from verses except as specifically indicated.

10.15. Nom. sg. -i. Masc. i-stems: avici SP 94.3; durmali mahya putrah 112.1. — rşi jirno LV 102.12 (prose; acc. to Lefm. with all mss. but one; Calc. rşir; Lefm. prints rsijirno as one word, which, I agree with Weller 24, is hardly possible. Weller erroneously assumes the like in LV 19.12, prose, vyādhiprajāāyate, which is a denom. verb from a cpd. noun, 'consciousness-of-disease-arises', and 34.16, prose, salyasamādhidharmālokamukham, which is a cpd.); nrpati 81.13; rājarşi 170.21; pali 171.1 (at the end of a pada in a meter where a long syllable is normally required; no ms. has patih but perhaps it should be read); sākyamunili nāmnā 383.13 (with fusion with following vowel) .- Mv, very common even in prose, from which are taken the following examples, a few out of many: agni i.21.4 and 5; 22.11; avici ill 26.14; kuksi 221.4 = ii.23.9; thapakarni (in-stem?) grhapati i.245.3; rsi, -vyādhi 284.6; sāli 341.17; etc. - maharşi Samādh 8.21 (end of trişţubh pāda, meter indifferent).-And many other cases in most texts.

10.16. Fem. i-stens: upašānti SP 96.8; oşadhi 129.10, 12 and 130.14; nirwrii 198.6; -bodhi 310.12 (this word is regularly fem. in our dialect; here, at the end of a pāda, meter not concerned); etc., common. — smrti pratilabdhā LV 86.8 (prose; Lefm. prints as cpd., which seems impossible, Weller 23; cf. in 86.7 madāpagatah, i. e. mada-apa^o, for Skt. mado 'pa^o); avabuddhā sattvavyādhi 351.11 (prose; so mss.; both edd. 'dhli; vyādhi regularly masc. but here the modifying adj. is fem.); and in verses, bodhi 12.11 (so best ms., required by meter; Lefm. bodhī); kirti 29.6; rddhi 37.14; mati 45.15; etc., common.—Mv, common in prose and verse; our examples from prose: bhūmi i.77.16 (followed by a-); nīti 135.4; -lipi 135.5 ff. (repeatedly, at end of cpds.; Senart always -lipi; mss. vary but -lipi predominates); yaşii 196.15; malsyajāti 245.2 (followed by ā); aiguli iii.358.3 (v.l. 0 lim); etc. vīmati Samādh 19.35; bodhi 22.15. — sampatti Suv 40.12; jāti 51.6; etc., common. — sthiti RP 6.9; smrti 7.13. bodhi Bhad 54. — prīti Gv 212.14 (end of pāda, meter indifferent; followed by u-); etc., common. — ruci ili (mss., ed. em. rucir) Divy 236.1 (prose); kşali na (mss., ed. em. kşatir) 404.13 (prose).

10.17. Fem. i-stems, polysyilabic: almost equally common in most texts. Read elädrsi SP 214.1 with WT and v.l. of KN for KN °sa; yādrsi 329.13; etc., common. yyapatrāpiņi LV 28.13; devi 49.14; nāri 54.7; māyadevi 75.5 and often; nadi 175.9; sāmagri 177.6; etc., common. prakrti māņavikā Mv i.232.16 (prose), 'the girl that has been mentioned, that is under discussion' (prakrll, for °lā), cf. 233.14; upasthāpana- (or upasthāna-)kāri iii.37.5, 8 (prose); 'a woman who serves a man sexually'; cf. ?kārikā id., 37.12; etc. — Madri Rl⁵ 22.17. — maitri Gv 230.13.

10.18. Fem. *i*-stems, monosyllabic. The paucity of examples here following certainly does not indicate any avoidance of -*i* in the nom. sg. of these stems. It is due to the relatively small number of such stems, combined with the fact that in my selection of examples from reading, I failed to pay sufficient hecd to getting samples precisely of this declensional type. -- *istri* (= *stri*) LV 42.17; 236.5; *sri yā* 215.14 (so read, as two words; Lefm. *śriyā*); *cakravartiśri yam* 253.3 (= *°srii igam*; Lefm. *°sriyan*, which is senseless). -- *śiri* Mv ii.94.2 and 353.17 (both in vs, but not required by meter); same in prose ii.94.9, 11 (with v.l. *śiri*); an iii.371.12, in a metrically obscure verse, with v.l. *śiri* (uncertain).

10.19. Masc. in-stems. Prose instances recorded only from Mv (and here marked as such); the others all from verses. — ananyathāvādi jino SP 32.16. — -vighdi LV 92.21; tapasvi 109.18; cakravarti 111.1 and 12; yogi 176.22; nirapekşi 241.16; doşadarši 330.4. — śresthi abhāşi Mv i.47.16; cakravarti 62.18; 158.12; ii.237.19; śresthi ii.175.1 (all prose). And verses: -kāri i.156.11; -rūpi 207.8; -cakravarti 209.15 (end of line, not m.c.); asamprakampi 292.15 (end of line); savabhutānukampi tam (so read and divide) 295.9: etc. — sarvadarti RP 24.8: -pramalhi 51.11.

divide) 295.9; etc. — sarodarii RP 24.8; -pramalhi 51.11. 10.20. Nom. sg. -is, -ih, -ir, etc. The regular Skt. ending of masc. and fem. short *i*-stems; occurs with fem. *i*-stems and masc. *in*-stems, occasionally in prose, even outside of the Mv. (Prose cases marked as such.) In the verses of most texts it is specially frequent in the form -*ir* before a following vowel, altho -*i* is often allowed to stand in that position; there is no prosodic difference; the -*r*has the effect of a 'hiatus-bridger' (cf. §§ 4.61 ff.). When not followed by a vowel, these endings are prosodically equivalent to -*i*.

10.21. Fem. *i*-stems, polysyllabic and monosyllabic: *rājadhānih* LV 22.1 (prose; no v.l.; only ^onī is recorded in Skt. and Pali; not noted by Weller); *jananir iyam* 49.13 (to *jananī*). — *sirir* (v.l. *siri*) *nāma* Mv il.89. 19 (prose). — Ratanarājaširih Gv 285.4 (end of line of verse; not m.c.). — *visamjiāvalir* Mvy 7974 (so also Mironov), certainly from a stem ^ovali; text of LV 148.9 (prose) *visanjiāgali*, read ^ovali (with pw).

10.22. Masc. in-stems: ananyathāvādir SP 92.4; 308.13 (followed by vowel both times). — śaśir iva LV 162.8; 362.9; 397.2. — śresthis Mv i.37.10; samitāvir 48.17; 50.2 (v.l. °vī both times; once plus cons., once plus vowel); krkiś ca 309.9 (in 14 below, also prose, krkī ca); hastir ii.458.5 (plus cons.; v.l. hasti) (all prose). And verses: šikhir iv° Mv i.67.8; antevāsis ca 269.8; -vašis 282.19; šikhir yalhā 358.10; pakşir vā ii.232.9; read vašir iddhiprāplo (or vaši riddhi°) 322.1; kešarir vā 332.20. — Ratnaśikhih Śikhih Mmk 68.26, 27 (prose); cakravartis 140.25. ---cakravartir (plus vowel) Gv 254.10.

10.23. Nom. sg. -im or -im. Masculine and feminine. As we saw above (§ 10.13), this is recognized by some Pkt. grammarians, according to Hemacandra, as a nom. sg. ending. We find it so recorded in our texts, tho the editors generally tend to emend it (and probably the more carcless ones often emended silently). I regard it as the acc. masc. or alternating MIndic nom.-acc. nt. ending used as nom. masc. and fem. (otherwise Pischel, see § 10.13). My examples are chiefly from LV (mostly verses) and Mv. All concern short i-stems, mostly fem. -- na cāsli trplim LN 49.16 (end of pada; metr. indifferent); aranim yatha collararanim 177.1 (v.l. for both "rani, but this is unmetr.); udyāna-bhūmim (nom.; Tib. represents udyānam, omitting bhūmi, which however all Lefm.'s mss. have) upasobhitam 187.12 (prose); read bhāvitā nitya kṣāntim 196.5 (with nearly all mss.; Lefm. kşānti with 1 ms.; end of line); nāsti le bhayam apāya durgatim 237.2; (labhyale ...) bodhim ullama° 287.18, all mss. and Calc.; meter requires long final, hence Lefm. em. bodhi ut°; rather read bodhim ul⁹7 (cf. §§ 2.69, 70); muñjamuşlim iva (nom.) 334.6; trimsalim ca malināni 372.5. — tam... mūlapranidhim Mv i.45.8 (prose); dvitīyapraņidhim ladāsi 54.3 (prose; all mss.; Senart cm. dviliyo prunidhi; the word is definitely fem. in 54.6, 7); eşâm ca. . . pranidhim salalasamilă abhūsi 61.2 (prose), 'and this vow was eternal'; so all mss., Senart em. esa ... pranidhi; caturasitim 117.2 (end of line of verse); calurāšītim (§ 19.35) 119.5; na ca tam vyādhim praliprasrabhyali 253,10 (prose), 'and the disease did not abate' (no v.l.; kept by Senart; parallels in same context noms., and no other construction is possible); tehi tam prakrtim sarvam āciksitam 355.2 (prose); read with mss. rātrim ca prabhātā (v.l. °tāye) ii.108.7 (prose); smrtim abhūși 264.10 (prose; so read with mss.); -guplim iii.329.11 etc., see Dict. s.v. gupli for others. - landulaphalaśālim Divy 216.8-9 (prose; kept in ed.).

10.24. Neuter. As in Pali and Pkt. (Geiger 85, Pischel 377, 379), -im may replace the regular Skt. -i, sometimes but not always when meter requires a long syllable. — vārim SP 127.8; pūtim ca vaktram 352.11 (in these may be m.c.). — puruşakāramalim (kulam) LV 24.7; prose; so mss. (Lefm. em. °mati); also identical passage Mv i.198.1 and ii.1.12; ardharālrim LV 210.2 (prose), cf. Pali addharalli, which is fem. in the only unambiguous passage recorded in CPD. — ašucim Divy 409.16 (prose; kept in ed. with mss.; in 409.1 ašuci).

10.25. Nom. sg. -im? $bh\bar{u}lakolim$ akolim ca (but some mss. °kolim, which is equally possible metrically, and akolim) LV 437.1 (vs). Note also in Mv i.66.21 (vs) $pr\bar{a}h\bar{u}jalim$, three mss., for (text with one ms.) °li (v. 1. °li, °lim); in Mv ii.280.6 (prose) $bh\bar{u}m\bar{i}$ adhi°, Senart, but mss. $bh\bar{u}m\bar{i}$ adhi° or $bh\bar{u}m\bar{i}m$ adhi° (the latter perhaps with 'Hiatus-bridging' m?).

10.26. Nom. sg. - \overline{i} . In Skt. regular for masc. *instems* and fem. polysyllabic *i*-stems; here used in what in Skt. are *i*-stems, masc. and fem. (Skt. nom. -*is*), as well as in fem. (and even masc.) stems which in Skt. have -is; and in neuter *in*-stems, perhaps by transfer from masc. *in*-stems. In prose rare except in Mv. But, altho other texts present such forms wholly or chiefly in verses, meter can hardly motivate most of them, unless in the sense that short -*i* would usually spoil the meter. Before consonants and when final in a line, the normal Skt. -*is* (-*ih*, -*ir*, etc.) would give the same prosodic result, as would Skt. -*is* always. However I know of no reason for supposing that short -*i* was any more 'original' as a nom. sg. ending in our dialect, even of *i*-stems, than -*i*, the

10.27. Masc. i-stems. yāvān avīcī SP 9.5; maharşī 89.11, 293.3 (end of pāda, followed by consonant); pūlī (... gandhah) 95.7; krpābuddhi ca prāņisu 343.6. — praņidhī LV 161.19; nrpalī 198.18; maharsī 231.13; namucī 302.21. — avīcī (v.l. °cim) Mv i.26.17; lecchavī 255.9, 256.7; kulapalī 273.10; dišāņipalī iii.204.8; angulī 358.2 (all prose). Verses: narādhipalī i.109.7, etc. — ślaksņachavī (kāyah) RP 51.5. — cyulyupapaltī (v.l. °li) Bhad 16, 'having (attaining) rebirth thru fall', nom. sg. masc. (? but possibly loc., in falls and rebirths; so Leumann translates; § 10.68). — Šivī (before consonant) Lank 251.9 (prose).

10.28. Fem. i-stems. -rasmī prasrtā SP 16.1; rasmī vipulā 16.2: ca ratī ca (v.l. caratīva) 87.6; udyānabhāmī 96.11 (at end of same line bhāmiļi); rddhī 161.13 (followed by iyam); adhimuktī 235.10; etc. — nirvrtī LV 38.1; durgatī 196.4; dharmardī sada tasya ratīhā (= ratir iha), dhyānaratī amtārtharatis ca 313.10; prddhī 398.19, etc. — āņatlī Mv ii.95.8; bhāmī 171.9 (v.l. min); prajňaptī 273.12; iii.297.8; aratī iii.281.15; 284.12; 285.5; āhutī 426.16, all prose. Verses: kirtī ii.225.5, etc.

10.29. *i*-stems which in Skt. have nom. sg. in -*is.* That is, monosyllabic stems and *laksmi*. Nearly all fem.; occasionally masc. compounds ending in -*sri*, viz. *Syāmakaširī* Mv ii.218.6 (prose), Mañjuširī Bhad 55. Feminines: *laksmi* LV 129.21, 130.5; *śri* 130.18, 21; *hirī* (= *hrīh*) 158.20; -*śirī* (= -*śrīh*) 222.18 (only in verses). *širī* Mv ii.91.4 (vs); *śrī* (followed by vowel) iii.10.15 (prose).

10.30. -i as fem. of in-stems, for Skt. -ini. Rare, but there are a few seemingly unquestionable cases. They are to be understood as extensions of the masc. ending to the fem., due to the regular identity of the two in i and i stems in our dialect (and assimilation of in-stems to i-stems). — dhyānasukha-samangī (as one word; there is no stem -samanga in normal use; here fem.) LV 56.20 (vs); sāksī (both fem., for sāksiņī) 340.14 and 341.2. balabalī āšā Divy 134.20 (prose; but reading and interpretation both uncertain).

10.31. -i in nt. *in*-stems. Not common; sometimes determined by meter. The same occurs in Pali (Geiger 85.3) and Pkt. (Pischel 377, 379; cf. 358, where i nut. *i*-stems is assumed to mean change to fem: gender, to my mind an absurd interpretation). — sukhī sarvam yalhā jagal LV 97.5 (in an anuştubh verse, but can hardly be called m.c.; sarvam guarantees neuter gender); vešmam na šobhī 231.3 (here m.c.). — achandagāmī ca tan kulam bhavali, adosagāmī ca amohagāmī ca abhayagāmī ca Mv ii.1.10 f. (prosè), mss.; Senart em. all to °gāmi, which is read in i.197.20 f., but the last two forms show there a v.l. °gāmī. The LV parallel 24.1-3 has °gāminam; see § 10.3.

10.32. Nom. sg. -īs, from i-stems. Only in verses and rare. The only certain cases seem to have -*ir* for -*ir* m.c., before initial vowel. Masculine: *devarsir asilo* LV 108.11. Feminine: no ca *lrptir abhūt* LV 215.8 and 17. Doubtful is *apāyabhūmīs tada ulsadāsi* SP 170.1; so Nep. mss.; Kashgar rec. °bhūmi sada°, which is unmetrical (intending bhūmis tada?); KN em. bhūmī, which WT keep without note.

Vocative singular

10.33. Voc. sg. -i. This, the Skt. ending of the *devi* declension, is used in the voc. of masc. i and in stems; in most texts only in verses, and at least usually where meter requires a short; but in Mv in metrically indifferent positions and in prose.

10.34. Masc. *i*-stems (the fact that my collectanea happen to show no fem. *i*-stems with this ending can certainly only be accidental; fem. proper names more often have stems in $-\overline{i}$, and hence regular Skt. vocs. in -i): *n*pali LV 56.7; 57.1; 95.3; 114.19; *narapati* 165.8; muni 360.2 (separate mudila); atidyuli smrlimati 361.19 (both masc., the latter according to Tib. cpd. of smrti and mali, 'having mentality characterized by smrti'); mahāmuni 414.1. — grhapati Mv i.37.11; 245.6; ii.272.11; dharmaruci i.246.6; nrpati ii.145.9; sāralhi 152.15; yūthapati 254.4; upāli iii.180.5 (ali prose). Verses (but often in places indifferent as to meter) bhūmipati Mv i.180.2; mahābuddhi 191.7 (masc., not required by meter; note mahādyude in preceding line); maharsi ii.337.12 (end of a line); etc. — mahāmuni Suv 35.10. — dānapati RP 7.11.

a mey, etc. — manamina Suv 53.10. — anappat RP 7.11. 10.35. Masc. in-stems: -nighāti LV 365.16; svāmi (so Lefm.; many mss. svāmin) 407.9 (prose). pararūpadhāri Mv i.156.14 (v.l. °dhārī); -gāmi 157.3, 165.7 (in the second, v.l. -gāmī); brahmacāri ii.205.5; iii.419.5; udāyi iii.106.3 fl. (examples all from verses, but meter indifferent in most). — Upāli Divy 21.24 (prose; the stem in Divy is upālin, and the voc. upālin occurs elsewhere); pakşi 476.14, 18 (prose).—(The regular Skt. ending -in may be written -im, as in -svāmim LV 243.7; this purely orthographic matter I ignore.)

10.36. Voc. sg. -is (-ir), the nom. ending used as voc. Or should the following be interpreted as a case of ending -*i* plus 'hiatus-bridging' r (§§ 4.61 f.)? There may be others but I have failed to record any. Use of nom. forms as voc. is, in itself, nothing unusual. *nrpatir adya* LV 41.14 (vs).

LV 41.14 (vs). 10.37. Voc. sg. -I. Occurs from stems of all types, as in Prakrit, but chiefly in verses; in Mv also in prose.

10.38. Masc. i-stems: maharse SP 162.12, text with Kashgar rec.; but Nep. mss. maharsi. — vrllakaļī LV 235.2 (stem °kaļi; masc.); svacarī 360.3 (?), but this may be nom.; I have assumed that it is a cpd. of the noun cari (see Dict.), but it might stand for -cārin (ă shortened m.c.).

10.39. Masc. in-stems: lokahitānukampī SP 62.5 (end of pāda). — achambhī LV 284.18; pramocī 285.2.

10.40. Masc. *i*-stems (cpds. of monosyllabic *i*stems which in separate occurrence are fem.): text mañjusrih (voc.) SP 8.5 (prose), but five Nep. mss. mañjuśri; mañjuśri 8.10 (so ed.; Nep. mss. mañjuśri, Kashgar Rec. 'siri, both metr. deficient); mañjuśiri 11.9.

10.41. Fem. i-stems: devi LV 91.21. — devi Mv i.131.3; 218.3 = ii.20.2 (all verses); ii.432.7 (prose; no v.l.); also iii.12.2, in a verse, but here devi would be metrically better; kimnari ii.97.17 (end of line of vs, not m.c., no v.l.); bhagini iii.69.8 (prose, no v.l.); gaulami 256.15 (prose, no v.l.):

Accusative singular

10.42. Acc. sg. -im. (Optionally before vowels, otherwise -im.) The Skt. ending of short *i*-stems, occurs here also in *i*-stems of all types and in masc. *in*-stems.

10.43. Masc. in-stems. In verses of most texts, and prose of Mv (also occasionally of Mmk). amrtagamim (dharmam) LV 38.10; -dhārim 132.21; vyāpim pradeša gala 336.22 (so divide; gone to an all-prevading [all-covering] place'); prasantagamim 355.4 (v.l. °gāmī). — sresįhim (before vowel) Mv i.37.11 (prose); -nirghātim 113.4; -vicārim 131.13 (prose); šresthim 186.11; vādim 186.14; cakravartin (followed by ti) 210.1 = ii.13.21 (prose; but in 1.210.3 = ii.14.2, a verse, the mss. read cakravarti, or unmetrically "li; Senart em. "lim); -samangim 1.210.2 (= il.14.1, where however mss. "ngi; end of line, indifferent metrically); upajīvim 279.18; bhuktāvim 312.15; 325.13 (both prose); pūrvopakārim (before vowel) ii 180.1 (prose); svāmim (v.l. svāmi; before vowel) 246.16 (prose); dvipim 411.2 (prose; so read with mss.; Senart em. dvipim, as acc. pl.); hastim iii.41.18 (prose), etc. - hastim (before vowel) Mmk 24.15 (prose); Ratnaśikhim 63.22. - vihārim Gv 212.9 (end of verse); arthim 411.25 (before vowel; 2d ed. artham). --- alpabhāņim Ud xxix.55(45) (for °bhāninam; the vs = Pali Dhp. 227 where mitabhāninam, which is hypermetric).

10.44. *i*-stems. Usually fem. and polysyllabic; also *sirim* (e. g. Mv i.276.14, end of line of verse) = *siriyam*, from *šri*; and in masculine bahuvrihis ending in originally fem. *i*-stems, as madhuravāņim (sc. bodhisatīvam) LV 294.18 'sweet-voiced' (to vāņī). Otherwise, in verses and Mv prose: pāramim gato SP 451.5. — devim Mv i.218.7 = ii.20.6 (before vowel); vaišālim i.258.6, 10 (prose); vārāņasīm (twice) 287.12 (prose); cetim ii.169.17 (prose); aşļamim caturdaši (or °šī, mss.; Senart em. °šim) paīcadaši (so mss., Senart em. °šim; but cf. § 10.51) 424.20 (prose); sunakhim vā šrgālim vā 483.15; prajāpatim iii.9.2 (so read with mss.; 'queen'); nārim 41.20; etc. — sarasvatim Bhad 30 (before vowel); amitābhapurim 62 (before cons.; for "purīm; so also Gv 548.2). — paācamim Dbh.g. 20(356).11, 15; aşļamim 38(64).16 (all three before vowels).

10.45. Neuter *i*-stems: *vārim* SP 128.13 and LV 234.4 (both may be m.c.). — *aslhim* (several mss. *asthi*) Mv 1.20.1 (prose); *surabhim* (adverb) *pravānti* 294.5, 7 'smell fragrant' (here could be m.c.). — *sucim* (... *khā danīyam*) Divy 64.25 (prose); so mss.; ed. em. *šuci* (which is read 66.4). — *asthim bhaiktvā* Kv 69.17 (prose).

10.46. Neuter in-stems: khagapathe-gāmim gajavararatanam Mv i.108.16 and -keśarim. . . turagaratanam 17–18; both acc. sg.

10.47. Acc. sg. -i. As in the nom., common in the verses of all texts and in all types of inflexion; rare in prose outside of Mv. (Prose passages will be so marked.)

10.48. Masc. i-stems: anusamdhi SP 394.1; pāņi (v.l. pāņim, unmetr.) LV 74.22; praņidhi Mv i.112.4 (with adj. param); muni ii.335.2 (required by meter; Senart wrongly em. munim); sāli RP 27.4, 6, 9.

10.49. Masc. in-stems: vighāti (= vighātinam) vaidyottamām LV 91.21; mss. bhuktāvi or °vī (Senart em. °vīm) Mv i.38.8 (prose); samangi ii.14.1 (end of line of verse; Senart em. °im with i.210.2); in ii.246.16 (prose) v.l. svāmi (text °im); dharmasvāmi (= °svāminam) RP 7.12.

10.50. Fem. i-stems: bodhi SP 35.1; 194.9; caturaślii dhārayel 254.11 (so read with most mss. and WT; KN unmetrical); durgati 314.3; etc. — prili vetsyāmaḥ LV 60.21 (prose); Lefm. as one word!; only one ms. with Calc. prilim; not noted by Weller; cari 161.21 (read cari purimām); 164.16; samgīli 193.12; śānti 287.4; etc. pārsni Mv i.6.4; 17.10 (prose; before vowel); bhūmi (mss.; Senart em. °im) 136.4 (prose); yoni (mss.; Senart em. yonim) 144.3 (prose); etc. — bodhi Bhad 10; nirvrii 11. — bodhi Gv 54.1; gali 54.24; cari 57.17; etc. — -cari RP 32.18.

10.51. Polysyllabic fem. i-stems: mahi SP 96.5 (= mahim); medini $(= ^nim)$ 394.4; etc. — tüli LV 75.1; mahi 165.10; 167.7 (167.7 also sanagari); nadi 271.12, 14; etc. — girinadi Mv ii.222.4; vasumali 342.2, 8; 344.2. In all these meter proves mss. (there is sometimes an unmetrical v.l. °l) correct; Senart em. -im. In i.42.8, 53.9 meter proves dharmabheri the only possible reading; mss. mostly °bheri, Senart °bherim. There are other such cases; the copyists, and Senart after them, seem to have gagged at admitting the acc. sg. ending -i in i-stems even more often than in i-stems; but there is no doubt whatever about both. See also ii.424.20, above, § 10.44. — paācami Dbh.g. 20(356).7 (= °mim, before vowel; in same phrase below, lines 11, 15, paācamim).

10.52. Monosyllabic *i*-stems: *siri* Mv ii.353.11 (mss., supported by meter; Senart *sirim*). Same form in mss. ii.379.2 and 22, before a vowel (so that it and Senart's em. *sirim* are metrically equivalent).

10.53. Acc. sg. -y. May occur for -i as acc. sg., before a vowel, to suit metrical convenience: māyādevy (read māya^o m.c.) abruwan LV 91.16; aprāpya bodhy ullhihet 284.6; sarvanyādhy apanesi 363.5. 10.54. Acc. sg. -I. As in the nom.; not so common here, but occurs even in a masculine *in*-stem, as well as in feminines (cf. under -*Im* below): cakravarti Mv i.210.3 = ii.14.2, mss., should be kept (v.i. °ti is unmetrical; Senart em. °tim). 10.55. The rest are from fem. stems, and (probably

by accident) I seem to have recorded only I-stems, or at least no forms which would necessarily be i-stems in Skt. Rare in prose except in Mv. kaţasi ca vardhenti SP 48.4 (Pali katasim vaddhenti, see Dict. s.v. katasi); buddhanetri kşipi 94.12; irşilām dharanı (all mss., both edd. em. onim) tarpet 126.14. — yasavati LV 50.4; krsi karoti 133.17 (stem normally krsi, but krsi also occurs in Skt.); śuska-nadi 174.10 (Leim. em. 'im); (?) -viśeşanāsadrši lokādhipaleyalām 179.20 (prose; so all mss.; Lefm. prints as cpd. with next word, which may be right, § 23.8; Weller 28 separates); idrśi vibhūtim 316.7 (no v.l.); medini 357.14; maitrī bhāventi 375.19; ratnapātrī 387.4 (Lefm. em. °īm). - divyalańkāpuri ramyām Lank 5.12 (in ed. printed as cpd.; acc. of goal, 'let the Buddha come to ...', cf. line 18 below). - caranti dharani My i.79.4 (no v.l.); sūci ii.87.12, 88.6 (no v.l.; could be plur.); dharmanetri 373.5; °bhirūpām strī (Senart em. strīm) ālmānam...iii.291.2 (prose). - stri RP 25.2; 29.10.

10.56. Acc. sg. -im. The regular ending of the devi declension, here found also with *i*-stems (and not only m.c.; even in prose), not only fems., but masc. *i* and *in* stems. Cf. Epic Skt. prātikāmīm (stem prātikāmin) Mbh. 2.60.1c (Crit. ed.). When followed by a vowel, -im is sometimes required by meter rather than -im (but -imwould always have the like result metrically, §§ 2.69-70).

10.57. Masc. i-stem: añjalim pragrhya (all mss.; both edd. em. °(*im*) SP 161.4 (prose). — añjalim pragrhitvāna (no v.l.) Mv ii.308.6.

10.58. Masc. in-stem: krkim (or krki, mss.; stem krkin; Senart em. krkim) kāširājānam Mv i.324.3 (prose).

10.59. Fem. i-stems: -bhūmim SP 68.7 (before vowel; m.c.); bodhīm 96.8 (before vowel; m.c.); drstīm 204.4 (before consonant; all mss.). — -bhūmīm Mv ii.150.6 (prose; before vowel; no v.l.). The stem bhūmī seems to be Vedic only, at least not Classical. In Mv 1.102.6 and 8 we also find bhūmīm, before consonant (but v.l. °im); cf. bhūmīşv 1.102.9. In i.105.7 (prose) Senart also reads bhūmīm, but by em.; mss. bhūmī or bhūmim. — ratīm (before vowel; m.c.) Samādh 19.26 (vs). 10.60. Acc. sg. -is (-ir, etc.), the nom. sg. used as

10,60. Acc. sg. -is (-ir, etc.), the nom. sg. used as acc. Rare. (amara) municisu kusumaarsiin LV 217.1 (so ed.; vs, printed as prose; all mss. -in or -ir, before j-; only Calc. -im), can only be acc. sg.; girir iva acalam 329.20 (here -r- may be taken as 'hiatus-bridger', after giri, acc. sg.); buddhākoļi-sahasra caikanavalih pūrvam smarāmī aham 402.13 (seems the most likely reading; 'I remember 91 thousand crores of Buddhas of old'; the number is confirmed by Tib.; Lefm. with v.l. naikanavali, bad in sense and in meter, since the last syllable must be long).

10.61. Acc. sg. -Inam. For -inam as acc. sg. of a stem in -in; isolated and m.c. sukundalinam (= °linam, m.c.) Suv 246.7 'having fair ear-rings'.

10.62. Acc. sg. -iya, m.c. for -iyam in monosyllabic *i*-stem. *śriya* (= *śriyam*) *kriyam atulām* (most mss.) LV 361.18.

10.63. Acc. sg. -iyam. The ending of monosyliabic i-stems in Skt., transferred to polysyllable i and i-stems. janetriyam Mv i.199.8 (acc. sg. to janetri = janayitri 'mother'; the parallel ii.3.6 has janetriyah, acc. pl.); -*sreniyam* (v.l. *sreniyam*) Mv iii.68.6 (prose), to *sreni* 'row, series'.

Oblique cases of the singular

10.64. Whereas in the nom., acc. and voc. forms there is very little trace of distinction between genders,

less than between original Sanskrit *i* and *i* declensional types, in the oblique cases gender is quite as important as between masc.-neut. *a*-stems and fem. *d*-stems. There is, to be sure, some confusion, as there is between *a*- and *ā*-stems; but it is clearly sporadic; on the whole, masc.-neut. nouns on the one hand, and fem. nouns on the other, go their separate ways. We shall accordingly treat them separately, in general. There are, however, some forms which seem to belong equally to both. To the latter class belong genitives in *-e* for *-eh*, and various oblique case forms in *-i*, *-i*; datives in *-ayi*; mc. for *-aye*; perhaps ablatives in *-igo* the like (but see § 10.133); and a doubtful locative in *-yau* (probably only an error for *-au*).

Oblique sg. case forms common to all genders

10.65. Oblique sg. -i and -i. Apparently the endingless stems are used for various oblique singular cases (instr., abl., loc.). Cf. the use of forms in -a from a-stems, §§ 8.3 ff. (I should certainly not connect such forms with Vedic instrumentals and locatives in I.) While not numerous, the cases seem, in part, hardly subject to doubt. They occur mostly in verses; Mv prose occurrences do not clearly decide whether -i or -i (or both) should be regarded as the basic ending, the other being (possibly) a change motivated by metrical requirements. The following are the most likely cases noted; all from verses except those stated to be prose. (Cf. also -im, § 10.75.)

10.66. Instrumental: sadršo 'sti na te kutottare (so read) šilasamādhi tathaiva prajňayā LV 364.18, 'no one is like you, much less superior, in morality, concentration, and also wisdom.' No v.l. for samādhi; šīla may also be taken as a separate word in the same construction, which prajňayā seems to define as instr.; prakţli (no v.l.) imi nirātma sarvadharmāh 393.12, 'all these states of being are self-less by nature.' — vašavartli (v.l. 'tī) ca devaputrena Mv i.230.13 (prose). So mss.; certainly instr. of a masc. in-stem; Senart em. to vašavartinā. — In Mv i.298.20 (vs) read with mss. mahājano prīli ('with affection'; Senart em. prīlo, unmetr.) karoti punyam.

10.67. Ablative: nagakanyaham bhadram te avici iha āgatā Mv ii.181.11, 'I am a nāga-girl, bless you!, come here from the nether world.' In a verse, but the syllable -icould be either long or short; the stem is normally avici, and masculine.

10.68. Locative: pāņi grhitvana LV 114.18, 'taking in (or 'with', instr.?) the hand' (object accus. in preceding line); two mss. pāņi, which is unmetrical; Calc. interprets by pānau; na ca ulthitu āsani no ca bhumī (so Calc. and Lefm., m.c., for mss. *bhūmī*) 155.11, 'and he did not arise in his seat nor on the ground'; āsani is certainly loc., and bhumi (for bhumi) must apparently be loc. if it is the true text; however, Foucaux (Notes 130) assumes that the true reading was $yami = (\bar{a})yami$ 'making an effort', on the basis of Tib. brtsol ba ma mchis su 'not making an effort'; nidhi drsta (one ms. and Calc. drsti) yathā hi palāyati koci naro 323.11, 'as some man might run away when a treasure was seen'; unless we em. to drstva (ger., 'having seen a treasure'), it seems necessary to assume a loc. absolute; drsta may be kept, as loc. (§ 8.11); rsipalanam upetya vārāņasī vartilo 421.16, '(by the Buddha) going to Rsip. (the wheel of the law) has been turned at Benares'; so Tib., clearly and plausibly; it would be implausible to take vārānasī as a second acc. with upelya, as supplement to - Possibly Bhad 16 (§ 10.27). rșipalanam. –

10.69. Nearly a score of times the Mv has (in prose) a formulaic phrase printed by Senart as follows: nagare vārāņasi kāšijanapade (1.271.19; 286.22; il.48.16; 64.14; 67.19; 77.5; 82.7; 177.9; 209.9; 241.13; 250.20; 420.6; ili.33.11; 143.11; 286.16; list is not complete). The variant vārāņasi, with short final vowel, occurs in one or another ms. in quite a number of these cases. The meaning seems certain: 'in the city (of) Benares, in the Kāśi country'. Without emendation, it seems scarcely possible to avoid taking $v\bar{a}r\bar{a}nasi$ (or °si) as loc. sg. And $k\bar{a}\dot{s}i$ might be interpreted in the same way, tho the fact that it precedes *janapade* makes it possible with Senart to regard it as part of a cpd.

10.70. Dat. sg. -ayi, only m.c. for *aye*. The form bodhayi, for bodhaye, occurs in verses LV 392.2; Bhad 12; Gv 488.8; Siks 4.8 and 9 (cited from Ratnolkādhāranī).

10.71. Abl. sg. -ito; for this (mostly from fem. stems, a few from stems which are at least normally masc.), see § 10.133.

10.72. Gen. sg. -e: for the gen. -es (-eh, -er etc.), the mss. sometimes write -e, and are (hesitatingly and not consistently) followed by the editions. It is, indeed, possible that lack of the final visarga is only a corruption of tradition; yet it could be interpreted as normal MIndic phonology, tho gens. sg. in -e seem not to be recorded in Geiger or Pischel for Pali or Pkt. A few examples: nrpate LV 114.2 (before bh-; one ms. and Calc. °ler); tasyā mune 422.5 (before v-; here Lefm. and Calc. mune, but best ms. muner). — lankādhipate Lank 9.12 (prose; before p-). — (mahā-)mune Mv i.65.13; 66.18; 67.12, 18; 68.1, 15; iii.380. 11; bhāme i.70.18 (mss.); -adhipate iiii.440.17.

10.73. Gen. sg. -yus from stem -pali, f. (as if initating pali, m., in mg. 'husband', Wh. Gr. 343d): Mahāprajāpatyur haste MSV ii.144.17 (prose).

10.74. Loc. sg. -yau for -au? I.e. blend of -au and -yām? Probably to be excluded; only SP 370.4 (prose) where KN ye cādhastād avīcyāv ūrdhvam ca. But apparently only one ms. reads avīcyāv (tho WT keep it without note). Observe also that avīci is, at least normally, masc. Probably avīcāv (or with v.l. avīci-pūrvam?) should be read. The Kashgar recension is not cited in the note.

10.75. Loc. sg. -im. Assumed as loc., avicim Mv i.26.14, 16 (prose) and aslamäbhümim i.107.7, Senart i note p. 391, citing Kaccāyana ii.1.18 (ādim, bārāņasim). G. the ending i, I §§ 10.65 ff.

Distinctively masculine-neuter forms, not normal Skt.

10.76. Inst. sg. -ina, only m.c. for -inā: agnina SP 85.11; nrpalina LV 166.5; pāņina-m-asim LV 338.12 (with 'hiatus-bridging' m); vyādhina Šikš 330.6 (cited from Ratnolkādhāranī).

10.77. Inst. sg. -Inā, only m.c. for -inā. Cf. AMg. forms like aļļhīņa (Pischel 379): daņdakīnā Mv iii.365.16.

10.78. Gen. sg. -isya, as in Pali and Pkt. -issa. Much commoner in My than anywhere else; in other texts only in verses and infrequently. Only with masc. and nt. stems; possibly a solitary fem. is sāhamjanisya Mv iii.151.4 (v.l. °isya), a name of a locality. Even when applied to compounds ending in monosyllabic *i*-stems which when uncompounded are fem., the cpds, are masc.; rahulasirisya My i.128.13; śyāmakaśirisya il.215.7, 13 etc.; kāśyapaśirisya iii.244.7. Occurs alike with i- and in-stems, which I have tried to list separately, but the distinction is sometimes impossible and always factitious in this dialect. Occasionally we find, at least in the mss., the Sanskritizing form -isya, but it is doubtful whether it has any real standing. In Samādh 19.36 namucişya is printed, but perhaps only by misprint; it is cited as namucisya p. 18. In Mv i.253.11 the mss. read maskarişya, in 305.8 krkişya (but krkişya 303.5, 7; 305.2), in ii.153.2 vyādhişya, and in iii.76.13 keśisya; Senart em. always to -sya, probably rightly.

10.79. *i*-stems: vārisya SP 126.3; lokādhipatisya 313.4; 331.8. — šākyamunisya Mv i.47.15; grhapatisya 245.5; lecchavisya 257.14, 16; vārisya 308.4; kanakamunisya 318.13; šālisya 343.19; rsisya ii.31.13, 15, etc.; vardhakisya 466.3; upālisya iii.179.10 fl. (all prose); etc., common; and equally common in verses, e. g. vajrapāņisya i.183.10; drāhamatisya iii.355.12; sucisya 418.8. — kācamaņisya KP 91.7.

10.80. in-stems: śresłhisya Mv i.36.8; 44.10 etc.; samitāvisya 49.9 etc.; bhuktāvisya ii.116.14; -samongisya 178.16; vipašyisya 271.10; vijitāvisya iii.47.9; šikhisya 94.1; kālodāyisya 103.9; brahmacārisya 163.17; ašvakisya 337.5 (all prose), etc., common. — ratnašikhisya Suv 146.8; 152.10.

10.81. Gen. sg. -inah, -ino. The regular ending of in-stems, applied here to masc. i-stems, as in Pali-Pkt., and to neuters in Skt. Cf. -unas, -uno with u-stems, § 12.2. Strangely, there is one prose instance in the Divy: sucinah 237.21 (adj., with āhārasya). Even more strangely, all the other cases I have noted are only from verses-even those of My: if My prose uses the form I have failed to note an example. The form -ino is generally used without regard to samdhi, and even before a pause; but at the end of a line -inah is found Mv i.219.8 (repeated with -ino ii.21.10) and ii.304.5. A cpd. ending in -sri shows kāsuapasirino My ili.429.14. Other examples: grhapatino SP 113.1. — narapa-tino LV 230.5; munino 275.20; namucino 341.14. — munino Mv i.68.11; 167.18 etc.; varabuddhino 170.16; 208.17 = ii.11.5; i.215.7 = ii.18.4; nrpatino i.199.17 (so read with mss.. Senart °nām); maharsino 251.7; 267.21; 304.13; ili.434.2; amilabuddhino i.314.18; mahipatino ili.37.15; ameyabuddhino 44.14; abuddhino 409.5 (so with mss.; Senart abuddhi). - munino Mmk 594.3.

10.82. Loc. sg. -e. The ending of a-stems, transferred for the nonce; noted only in the phrase avice mahānarake 'in the great hell Avici' Mv i.230.5 = 240.15 (prose). In the parallels Mv i.41.10 and iii.341.17 avicismim. The form mahānarake has induced avice.

10.83. Loc. sg. -esmim. An ending of a-stems, occurs once in an *in*-stem; vipasyesmim Mv 1.294.19, mss.; Senart emends to $^{\circ}yismim$, but there is no more reason for doing so than for emending to avicismim in the preceding.

10.84. Loc. sg. -ismim. As in Pali (and cf. Pkt. -immi, AMg -inisi), analogous to -asmim (-esmim) in the a-stems. Extremely rare except in Mv; indeed outside of that text I have noted only a case or two of the form -ismi, for which see next paragraph. In the form -ismim it occurs in metrically indifferent situations and in prose; -ismi only occurs in verse where meter requires a short ultima (and not always then in the mss.).

From in-stems: hastismin Mv ii.73.8 ff.; 423.16; 434.11; iii.184.7 (all prose); kālodāyismin iii.103.7.

i-stems: agnismim Mv i.6.1; iii.426.15 (prose), and i.14.6, 15.15 (vss); avīcismim i.41.10; 337.5; ii.162.14 (prose) and i.42.16 (vs, end of line); kuksismim i.206.15; i.213.11 = ii.16.14; i.303.5; iii.404.5 (all prose) etc.; in i.207.10 = ii.11.21 (vs) meter demands short final, before a vowel, i. e. kuksismim or °smi; the mss. read kuksi the first time, kuksismim (adopted both times by Senart) the second; -kanakamunismim i.294.20; atithismim (before vowel; °im better metrically) ii.50.11 = 51.1 (vs); -girismim iii.60.1; 441.15; 443.14 (all prose); fākyamunismim iii.238.

10.85. Loc. sg. -ismi, only in verses m.c., for the preceding: ekamuşlismi SP 253.13, Kashgar rec., La Vallée-Poussin, JRAS 1911.1072; edd. with Nep. ekamuşlim tu, obviously a lect. fac., inferior as to syntax (depending on nikşipet). — samādhismi Mv i.164.12; 165.1; sākyamunismi 294.21 (v.l. ^osmin, but meter seems to require ^osmi). — narakagatismi Sukh 23.6 (probably bahuvrihi, 'in (reference to) one that is subject to existence in hell', masc.).

Oblique singular feminine

10.86. In MIndic generally, all fem. i and i stems standardly have endings which are common to all oblique

singular cases, and are roughly parallel to the corresponding endings of \bar{a} -stems: Pali - $i_i d$ (Geiger 86, 87); Pkt. -ia, -ie, less commonly - $i\bar{a}$, -ii (Pischel 385, 386); of these -ie is 'die gewöhnliche Form in allen (Pkt.) Dialekten'. Forms with short *i* for *i* also occur, at least in Ap. (ibid. 385, end). Pischel (386) also records the Ap. ending -i (-i) for instr. and loc. (§§ 10.65 ff.), and certain special endings for particular cases. Pischel found no examples of -ia, -ieetc. for the ablative, but the Pkt. grammarians include that case with the others, and in our dialect it shows in general the same endings with the other cases (in addition to the distinctive -ilo, -ilo, which to be sure are commoner). The *i*, when short, before *y* may be considered epenthetic (§ 3.102).

10.87. Our dialect resembles Mindic in general, but is even more varied in the forms it presents. Our arrangement will follow the general lines of our statement of \bar{a} -stem forms. We shall begin with the ending -*iye* (cf. Pkt. -*ie*) and its congeners (-*iye*, -*iyi*). Then -*iya* (cf. Pali -*iyā*, Pkt. -*ia*, -*iā*) and its congeners (-*iyā*, -*iyā*, -*iya*); next -*iyo*, -*iyu*, -*iyo*, much rarer; next -*yā*, -*ya*; and finally various forms which fundamentally pertain to particular oblique cases, and are not (like the preceding) common to them all. We shall attempt to group the forms under each ending by cases, according to the apparent syntax; but there are many instances where the same form could be attributed to more than one case; the plain fact is that the formal distinction between the cases is obliterated in these categories.

10.88. All these endings appear to be applicable to all *feminine* i and i-stems without distinction. The Skt. differences between the *gali*, devi and sri types are not maintained. This is of course not meant to suggest that distinctive forms of those types do not occur in the texts, at least in our mss. and editions; it only means that, as far as non-Sanskritic forms are concerned, they seem to occur equally freely with all fem. types.

10.89. They even occur sporadically with stems which are, at least in Skt., masculine. This is part of the general blurring of gender-distinctions in MIndic as a whole. Some of the nouns in question have feminine forms in other MIndic dialects. Examples are anjali, m. in Skt. and Pali, but also fem. in AMg.; vidhi, not recorded as fem. in Skt., Pali, or Pkt., except that PTSD (not Childers) attributes fem. gender to it (but its citations do not support the allegation); samadhi, only m. in Skt. and acc. to Childers in Pali (PTSD gives no statement of gender), also in AMg. (Ratnach), but acc. to Sheth both m. and f. in Pkt.; vyādhi, m. in Skt., Pali, AMg., but in JM. (Jacobi, Ausgew. Erz., Glossary) vāhi also fem. Distinctively fem. forms of these stems are found in our texts: añjaliya (v.l. [°]iya, instr.) Mv ii.299.17; añjaliya (Senart em. [°]-iye, instr.) Mv ii.395.15, 19; vidhiye, vidhiye Mv i.308.16; ii.2.14; ii.208.11; i.261.13; i.209.9 = ii.11.16; samādhilo (? perhaps not to be considered fem. ending) SP 25.3; My ii.209.2; 241.2; samādhiye My i.160.13; samādhilo, ku ho.d.; vijādhilo (? cf. under samādhilo) Mv i.308.8; Suv 38.11.—There seems even to be an instr. agnyā from the indubitably masc. stem agni (§ 10.121). Here, certainly 'change of gender' is out of the question; it is rather adoption of an originally fem. ending in a masc. noun. - On kukşi see § 10.100.

10.90. Obl. sg. - \overline{iye} , -iye, -iyi. The last rare and only in verses m.c. In Mv, prose as well as verses, -iyeis extremely common; indeed it seems to be the regular ending, far commoner than -iyu ($-iy\bar{a}$, etc.), and commoner also than -iye, tho this is by no means rare and occurs not infrequently in prose, so that the short *i* cannot be due solely to metrical requirements. Often Mv mss. vary between -iye and -iye. Both -iye and -iye are virtually limited to verses in other texts than Mv; note however a considerable list of forms in *-iye* (gen.) in prose Māy 226.15 ff., 244.33 ff. (On *-iyai*, dat., see §10.131.) 10.91. -iye, Instr.: (i-stems) kşāntiye I.V 162.3;

10.91. -iye, Instr.: (*i*-stems) ksānliye 1.V 162.3; 164.5, 6; silacariye 168.3 (or loc.?); (*i*-stems) mailriye 162.5 (... sphuļa, 'filled with love'); siriye 185.5—Mv, very common in prose and vss; examples from prose: (*i*-stems) rddhiye Mv i.35.3; 248.18, 20; iii.357.9 etc.; praktīye i.238.6; rājānattīye 274.2; 350.2; ii.48.6; angulīye i.339.10; jātiye ii.161.7; sampatīye 271.1; drsfiye iii.281.5, 6; gatīye 376.6; buddhīye 393.12 (v.l. °iye): (*i*-stems) vipākanīye, vipācanīye (see Crit. App.), grahatīye 307.12 etc.; mrģīye 363.7; iii.143.17 etc.; kimnarīye ii.97.11 etc.; lādršiye laksmīye 157.9; brāhmanīye 210.3; devige 422.8 etc.— (*i*-stem) smrtīye Gv 231.2; (*i*-stem) siriye 235.24,—Masc., or doubtful as to gender: vidhīye Mv i.308.16; ii.2.14; 208.11 (above).

10.92. -iye, Dat.: (*i*-stems) -gupliye Mv i.208.6 (prose); v.l. guplaye, which is read without v.l. in the parallel ii.10.18; but i.208.8, 9, 15; in modulations of the same phrase, have -gupliye without v.l.; gupliye (v.l. °iye) iii.48.12 (prose); in ii.256.10 -vrddhiye and in ii.315.2 -sambodhiye are read by Senart, but v.l. °iye both times; both prose. --- cariye (*i*-stem) Bhad 42, 56.

10.93. -īye, Abl.: (i-stems) bhillīye (in prec. line bhillīya) Mv i.6.9 (prose); bhūmīye i.6.11 (prose; also v.l. of 3 mss. i.26.3, prose, where Senart [°]iye with 3 other mss.); (i-stem) vārāņasīye ii.180.8 (prose).

10.94. -iye, Gen.: (i-stems) vimukliye Mv i.160.13 (prose); bodhiyaşliye (v.l. ?i; intending ?iye? or to § 10.65?) ii.264.5 (prose); rddhiye (v.l. °iye; meter indifferent) ii.289.6; bodhiye 317.19 (prose); (i-stems, all prose) deviye i.215.12; 217.6; 223.4 etc.; puşkaraniye i.227.18; gośriąłye 263.3; 271.14, 17; vaiśāliye 270.11; 283.7, 8 etc.; ulākīye, śukiye 272.18; māliniye 307.4, 6, 11 etc.; gurviņiye mrgīye 361.20; strīye prajāyamānīye ii.163.2; etc., very common. — cariye Bhad 45. — sukhāvaliye Sukh 45.9; 46.5. — -samatīye Māy 226.15 fl. (prose; read doubtless samitliye, to samili 'assembly', but repeated many times); gauriye 244.33 fl. (prose), and a series of genitives (? datives) in -iye. — Masculine, or doubtful in gender: samādhīye Mv i.160.13.

10.95. -īye, Loc.: (i-stenı) nirvrliye SP 198.10 (ed. with Nep. mss.; Kashgar rec. °liyā; end of pāda; read visrānta jñātvāna ca nirvrliye (or °yā), sarvajñajñāne upanenti sarvan, 'and knowing that they are rested, they lead all unto nirvāna, unto the Omniscient's knowledge'); (i-stem) kolisahasrāya (loc.) ca dhāraniye 330.4, 'and in the magic formula which makes 1000 crores (of revolutions)', referring to SP 327.8 (misunderstood by Burnouf and Kern) .--- (i-stem) raliye LV 169.9; (i-stems) lumbiniye 81.8; medinīve 195.9; 368.16.--(i-stems) -bhāmiye Mv ii.45.5; 130.14 (prose); rātrīve ii.112.10; 167.7, etc.; ksāntīye iii.360.17 (vs); upapattīye 404.11 (prose); (i-stems, all examples chosen from prose) dipavative rajadhaniye i.194.16; 199.19; 232.10; vārānasīye 325.16; 329.8; ii.167.6, 12; 176.18 etc.; chādyamānīye i.328.1, 16 (Senart em. °āye); vuhyantīye 342.15; otarantīye ii.101.11; nadīye 264.4; 302.7; iii.163.10; 166.6; mahaliye ii.282.10; kamsapatriye 282.11; deviye 427.11; padminiye 448.18; -bāhiriye iii.298.1, 2; etc. — dharaņiye Suv 230.4; 231.7. — -drsiye (i-stem) KP 95.5 and 6 (prose). - buddhabhūmīye Gv 57.25 (printed °mī ye; i-stem). — -bhūmīye Māy 36.

10.96. -iye, Instr.: (i-stem) siriye LV 54.8 (= sriyā; both i and i m.c.).—Mv, much less common than -iye; largely in verses, but also in prose: (i-stems) rddhiye Mv 1.65.4 (vs); 216.18 (prose, v.l. °iye); ii.366.4 (vs, v.l. °iye); iii.116.1 (prose, v.l. °iye); 303.14 (prose); 402.18 (prose; in some of these at end of cpds.); buddhiye i.68.16 (vs); ii5.7.6 (vs); 447.5 (prose); viparītāya drsīţiye i.101.12 (vs, so read, cf. Senart's note); jātiye ii.199.16 (vs); kşāntiye 332.18 (vs); (i-stems) prakrliye (to prakrli, fem. of ppp. prakrta) i.238.3 (prose); mrgiye...amucyanliye 363.7 (prose, no v.l.): -pariphriye ii.205.16 (vs, Senart em. °yd); parydhindanliye iii.155.1 (prose); striye 296.7 (prose); pauriye 322.2 (prose): icchanliye 393.16.--(i-stem) pralipatliye RP 36.12.--(i-stem) priliye Sukh 50.10.--Masculine, or doubtful as to gender: vidhiye Mv i.209.9 = ii.11.16 (here v.l. °iye; meter doubtful); i.261.13 (prose, no v.l.).

10.97. -iye, Dat.: (i-stems) bodhiye upajanenli mānasam Mv i.81.19 (vs), mss., to be kept; Senart em. bodhaye. An identical passage i.48.2 mss. bodhāya (one 'ye), unmetr.; Senart em. bodhaye, but perhaps bodhiye may be the true reading here too. — -yrddhiye (v.1. °iye) Mv ii.256.10 (prose); -sambodhiye (v.1. °iye) 315.2 (prose).

10.98. -iye, Abl.: (*i*-stems) chaltrodgatam dharaniye LV 196.21, 'a parasol sprung up out of the earth'; so Tib. definitely, sa las (abl.); yasyā lejatu pungataś ca śiriye 275.17 (the parallel nouns in -tu, -ta\$ prove śiriye abl.). – bhūmiye (v.l. °tue) Mv i.26.3 (prose), i-stem.

abl.) — bhāmiye (v.l. °iye) Mv i.26.3 (prose), i-stem. 10.99. -lye, Gen.: (i-stem) rāiriye LV 236.1; (i-stems) deviye LV 49.19; 78.16; mediniye 341.9.—(i-stems) bodhiyaşliye Mv ii.181.13 (prose); bodhiye 362.1 (vs); rālriye Mv ii.228.15 (prose; repeating a passage of line 11 where rātrīye; perhaps loc.); (I-stems) kaļījantiye Mv i.217.7 (mss.; prose; follows and agrees with deviye); janentiye 218.14 (vs, but meter indifferent); prakfilye 233.14 (prose; 'mentioned', stem prakrlī, fem. to °la); širiye (v.l. °iye) ii.90.5 (prose); niskramantiye 101.13 (prose); istrije (may be instr.) iii.26.21 (prose); and in verses where short i is required by meter, jinajanetriye i.219.13 = ii.24.15 (also i.218.13 where mss. °iye, unmetr.); mānusiye prajāye i.295.2 (Senart em. wrongly); purimabhavajanetriye ii.206.15 (so read, mss. °trīye; followed by bhavanighātī).

10.100. -iye, Loc:: (i-stem) keci sthitā dhāraniye parāyām SP 330.3 (KN dharāyām, but WT var⁶, and so Burnouf must have read, 'la meilleure').—(i-stem) kuksiye LV 75.6 (kuksi masc. and fem. in Pali, only fem. in AMg.; fem. in Skt. Lex.); (i or i-stem) vilhiye 240.20.—(i-stems) bhūmiye Mv i.7.14; ii.435.15 (both prose; v.l. in both [°]iye); rddhiye i.44.9 (prose; so most mss., Senart [°]iye with one poor ms.); -yaştiye (v.l. [°]iye) ii.268.2 (prose); kṣāntiye 340.18; 368.2; 371.8 (vss); rātriye iii.9.15; 296.19; 297.1 (all prose); bhūmiye 121.13; 451.11 (vss); (i-stems) sayantiye Mv ii.11.20 (vs; inferred to be loc. from ii.9.1 ff.; or gen.?); kumbhadāsiye (so read, Dict.) 58.3 (vs); vāravātiye (v.l. [°]iye) 94.11 (prose); pūrnamāsiye 306.19 (vs); 409.4 (vs; Senart em. [°]sye, wrongly).—(i-stem) rājadhāniye Suv 147.1 (end of pāda; v.l. [°]iya; both metrically possible).

10.101. -iyi, Loc.: only in verses, m.c. for -iye (-iye); so to be interpreted, more likely than by transfer from the (alternative) loc. ending of Skt. monosyllabic *I*-stems (*śriyi*). *rātriyi* LV 194.8 'in the night'; so Lefm. with best ms., all other mss. *rātriya*; *bodhiyi* Bhad 19 (or gen.). Both *i*-stems.

10.102. Obl. sg, -īya, -īyā, -iyā, -iya. These endings are much less common in Mv than -iye etc., but interchangeable with the latter, and at times juxtaposed in the same passage. Thus Mv ii.98.12 (prose) sudhanasya kumārasya kimnarlya (so mss., Senart em. °iye, unnecessarily) udāram premam nipatitam, kimnariye pi sudhanasya premam nipatitam. Or Mv iii.116.15–16 mahāprajāpatīye gaulamīye, while shortly before, 116.7-8, we find mahāprajāpalīya gautamīya. Other texts than Mv show these endings (like -iye etc.) almost exclusively in verses; but cf. mahārājāiya Mmk 24.18 (prose). The verses of SP, however, where -iye and its congeners are extremely rare, show -iya and its congeners much more frequently. In the verses of most texts, I believe, the two types are about equally common. The forms with short iare pretty rare, and even in Mv are hard to find in prose (but jālabhūmiya Mv iii.93.2, prose).

10.103. -iya, Instr.: (i-stems) snytliya SP 158.10; dyuliya 175.4; rddhiya 309.15; prakrtiya 371.8 (cf. 359.13 prakrtya, m.c. for ⁶tyd). --- (i-stem) ārāliya (Lefm. em., m.c.; mss. aratiya) ratiya saņuase na ca sārdham LV 325.11 (wrongly divided in ed.; 'and I do not dweil together with displeasure or pleasure', so Tib.); (i-stems) LV deviya 28.18; ŝiriya 241.6.--(i-stems) pratipattiya Mv ii.357.20 (vs); trpliya 370.3 (vs); buddhiya (v.l. [°]iya) 434.14 (prose); (I-stems) Mv kimnariya (so mss., Senart [°]ye) ii.96.9 (prose); agramahişiya 422.2 (prose); mahādeviya 457.6; deviya 450.7; širiya iii.407.8 (prose; v.l. [°]yo; Senart em. [°]yā!). --- bhadracariya (i-stem) Bhad 41. --vicaratiya (i-stem; pres. pple. fem.) Gv 301.17 (vs).

10.104. -Iya, Abl.: (*i*-stem) bhilliya (no v.l.; in same line bhilliye) Mv i.6.8 (prose); (*i*-stem) siriya ii.62.18 (abl. of comparison).

10.105. -iya, Gen.: (i-stcms) kşāntiya SP 273.10; aham ca badhiya (WT bodhāya with ms. K') vadāmi varnān SP 55.6, 'I declare the aspects (or, speak the praises) of enlightenment'; rātriya yāmasmi ha madhyamasmin 25.14 'in the mid watch of the night' (wrongly Burnouf and Kern); paācāšatiya 304.2 (or instr.?); (i-stems) prihivīya SP 53.10; 363.6; jātiya 362.4.--(i-stem) kşāntīya LV 183.1. --(i-stem) bodhiya mūle Mv i.3.1 (prose) 'at the foot of the Bo-tree'; (i-stems) Mv deviya i.204.16 and 207.16 (vss); i.223.10 (here Senart em. °iye); ii.66.8, 10 (in 10 v.l. °iye); 441.3; iii.7.4; 25.11 (all prose); striya (so mss.; see Dict. s.v. varnadhātu) ii.60.3 (vs); mahāprajāpatiya ... gautamīya iii.116.7-8 (prose); ceļīya 128.3 (prose).--(i-stem) carīya Bhad 24.--(i-stem) mahārājājīya Mmk 24.18 (prose).

10.106. -iya, Loc.: (i-stems) adhimuktiya SP 31.6 (so ed. with Kashgar rec.; Nep. mss. °muktau hi, lect. fac.); nirvfliya 33.2 (so ed. with Kashgar rec.; Nep. mss. nirvfau hi, lect. fac.).—(i-stem) bhūmiya LV 178.14 (v.l. °iye, metrically poor).—(i-stem) rätriya Mv iii.365.13 (vs); (i-stems) paācadašiya Mv i.164.20 (vs), 'on the 15th day'; vārānasiya (mss., Senart em. 'iye) ii.188.21 (prose); karkariya (in 2 v.l. °iya) 249.2, 5 (prose); prthivīya 414.20 (vs); devīga 427.17 (prose); padminīya 448.17 (here mss. padmanīya, Senart em. padminīye), 19; 450.11 (Senart em. °nīye), all prose; vāpīya 451.1 (prose; Senart em. 'iye). dharanīya Suv 235.8 (i-stem).— bhadracariya Bhad 44 (i-stem).

10.107. -iyā, Instr.: dyuliyā (m.c.) SP 164.12 (istem); caluranginīyā senayāye (on this see § 9.70) Mv 1.158.13 (prose; i-stem).

10.108. -iyā, Instr.: (i-stem) prīliyā Mv iii.374.16 (vs); (i-stem) širijālimantiyā ii.57.8, 12 (end of pāda).— (i-stem) pralipalliyā RP 11.1 (vs).—(i-stem) buddhiyā Gv 240.23 (vs).—(i-stems) maliyā, galiyā Sukh 22.16 (vs). 10.109. -iyā, Abl.: (i-stem) dūre vayam ultama-m-

10.109. -iyā, Abl.: (i-stem) dūre vayam ultama-magrabodhiyā SP 203.9, 'we are far from supreme enlightenment'. So ed. with Kashgar rec.; Nep. mss. 'bodhaye, the Skt. dative, which is not easy to construe.

10.110. -iyā, Loc.: (i-stems) upapalliyā Mv i.282.18 (vs, meter indifferent); subhūmiyā ii.423.12 (vs, i m.c.).

10.111. -iya, Instr.: gender doubtful: añjaliya (v.l. [°]Iya; meter troublesome, i prob. right) Mv II.299.17 (vs). —Other (fem.) i-stem: kşāntiya (so mss., Senart em. kşāntyā) Mv III.441.1 (vs). (For śiriya, m.c. for śriyā, see below, § 10.125.)

10.112. -iya, Gen.: (i-stems) jālabhūmiya (no v.l.) Mv iii.93.2 (prose). — buddhiya (m.c.) Dbh.g. 3(339).7 (vs).—Gender doubtful: samādhiya (m.c.) Samādh 19.27 (vs).

10.113. Ob1. sg. -iyo and its variants -iyu (m.c.), -iyo. Rare, but there is too much evidence to be ignored. Neither Pali nor Pkt. seems to know such endings. It may be that the Skt. ending -iyas, abl.-gen. sg. of monosyllabic i-stems, is at the bottom of -iyo, while -iyo (-iyu) could be a blend of this with -iye, -iya etc. Furthermore, -iyo is much used as nom.-acc. pl. ending of (mostly fem.) i and i-stems in our dialect (as are -io in Pkt. and -iyo in Pali). The identity in Skt. of the ending (-iyos) of the nom.-acc. pl. and the abl.-gen. sg. of monosyllabic i-stems may have helped in the extension of -iyo first to the abl.-gen. sg., then to the other oblique cases. The formula would be śriyo (nom.-acc. pl.): śriyo (abl.-gen. sg.) = śriyo (etc.): x. This x is śriyo, etc., abl.-gen. sg. Cf. also the ending -yas etc., § 10.138.

10.114. -iyo, Instr.: Masc., or gender doubt[u]: añjaliyo Mv li.395.15, 19 (vss); so mss.; Senart em. °iye; parallel Šikş 308.6, 8, reading añjalibhir.

10.115. -iyo, Abl.: nābliyo Mv ii.152.13, 16 (prose; i-stem; Senart ^oye, but both mss. ^oyo in 13, and one ms. in 16).

10.116. -iyo, Gen.: (i-stems) nāriyo Mv ii.3.7 (vs: repetition of i.199.9 where nāriye); kumbhadāsiyo iii.264.10 (prose; so read for mss. °vāsiyo, v.l. °vāsiyo, ed. cm. °vāsīye).

10.117. -iyo, Loc.: (i-stem) padminiyo udaka°, or padminiyo daka° Mv ii.450.17 (prose), mss.; Senart em. °niye udaka°.

10.118. -iyu, only in verse, m.c. for iyo; Gen. (i-stem) gatiyu Gv 253.13.

10.119. -iyo, Gen. In verse, but in the only occurrence noted the meter is indifferent as to quantity of *i*. (*I*-stem) deviyo Mv i.183.19 (vs; so all mss. except one "ivo; Senart em. deviye).

10.120. Obl. sg. -yā, -ya. The ending -ya is the regular Skt. ending of the instr. sg. of *i*- and polysyllabic *i*-stems. Except for lack of final visarga (or s), it is also an ending of the abl.gen., and except for lack of final nasal (anusvāra), of the loc. Such endings are not common in our texts, and might be supposed to be due to corrupt tradition. But, especially since we also find (in verses, and doubtless only m.c.) the ending -ya, with shortened a, it is probable that genuine forms, rather than corruptions, are concerned. Prose occurrences are marked as such.

10.121. -yā, Instr.: The regular Skt. ending of feminines. A probable case of its transfer to a masculine: vināpi cāgnyā paramam subhairavam SP 86.13 (vs), so all Nep. mss.; KN with Kashgar rec. cāgneh (probably lect. fac.), kept by WT.

10.122. -yā, Abl.: anyatra devyātiguņānvitāyā LV 29.10, 'except the queen . . .'; in Skt. anyatra governs only the abl.; here samdhi makes the quantity of the final indeterminate; jātyā parimoksante 104.16 (prose) 'are (will be) freed from birth'; no v.l.; the following parallel sentence has ablative. Similarly jātyā (all mss.; here Calc. jātyāḥ) parimoczyisyati 226.19 (prose).

10.123. -yā, Gen.: tīre hi nadyā sthitah LV 271.18; devyā (before p-; no v.l.) Mv i.177.14 (prose).

10.124. -yā, Loc.: dasabhūmyāham Suv 30.1 and 32.5, certainly to be read, m.c. for ^omyām aham. The best ms. has ^omyāham in 32.5, ^omya aham in 30.1. Nobel seems to me to misunderstand line 30.1, which is to be read sthāsyāmi dasabhūmyāham dasaratnākare vare; the forms in -e are loc. sg. (not acc. pl. with Nobel), in apposition with dasabhūmyā(m). The Chin. translations cited by Nobel himself confirm this.

10.125. -ya, Instr.: prakrtya samtişthati śrotram etat SP 359.13, 'this hearing-faculty exists by nature'; one ms. prakrti (unmetrical); cf. prakrtīya in similar phrase 371.8; drştya Gv 485.24.—The form śiriya, m.c. for śriyā, LV 241.10, Mv iii.115.8, may be included here, tho it might also be classed with forms in -iya, § 10.111. I assume that it is based directly on the Skt. form, with epenthetic *i* in the first syllable.

10.126. -ya, Gen.: bhūmya Dbh.g. 19(355).23 (before vowel, see § 10.138) and 20(356).21 (before consonant);

girinadya saman LV 173.14 (or instr.? Lefm. prints as one cpd. word).

10.127. -ya, Loc.: -nadya Suv 35.11; in Mv i.72.8 (vs) mss. kāšipūryām; Senart ^opuryām; but meter requires a short final syllable; ^opurya would be an acceptable emendation (the Skt. ending was put in by a later copyist).

10.128. Miscellaneous fem. sg. oblique endings. Finally, certain endings are peculiar to particular fem. sg. oblique cases, tho not found in Skt. Very rarely they seem to be used with masculines, by transfer. They are here classified under the several cases.

10.129. Instrumental -iyena (cf. Abl. -yātas). The fem. ending -iye with the masc.-nt. -na added: balena ca rddhiyena ca jūānena ca Mv ii.76.14 (prose). Note the surrounding forms in -na, which are evidently concerned in the blend formation.

10.130. Dat. -ye for -yai: bodhi-sambhāra-pūrye Dbh.g. 52(78).11; confirmed by -pūryai 55(81).13. Perhaps mere corruption of tradition, and to be emended.

10.131. Dat. -īyai, for -īye: equivalent to Skt. -yai; a blend of these two, apparently. yoginīyai bhīsmabhaginīyai (1) Sādh 404.8 (prose); vikīrņakešīyai...°dhāriņīyai 9.

10.132. Abl. -īto, -ītas, etc. Usually -*ilo* without regard to samdhi; chiefly in Mv (prose as well as verse), and not limited to Skt. *ī*-stems; thus from *bhāmi* (Vedic *bhāmi*), *bhāmīlo* Mv i.136.4 (prose), and iii.376.7 (prose); to be sure v.l. *bhāmīlo* both times. Other stems: *prthivīlo* Mv i.217.10 (prose); *vaišālilo* 259.7, 12 (prose); *aārāņasīlo* 287.13; 311.1; 313.11 etc. (all prose); *anāgulīto* ii.73.1; ii.358.9 (here v.l. °*ilo*; both prose); *anāgulīto* (v.l. °*ilo*) ii.244.8; *mrgīlo* iii.151.11 (prose). — *tasmān* (1) *nagarīlo* Divy 222.6-7. — *nadīlo* Jm 240.7 (App.; prose; parallel to Mv ii.244.8). — *śrāvastīlo* Av i.326.5 (prose; v.l. °*ilo*). — *-hāriņīlaḥ* Māy 237.20 ff. (prose); long series of cpds. having this final.

10.133. Abl. -ito, -itah, etc. Generally -ito without regard to samdhi. Chiefly in verses except for Mv. Not limited to i-stems; from i-stem, vaišalito Mv i.255.2; ii.207.15 (both prose, no v.l.). These circumstances seem to me to justify us in regarding -ito as essentially Prakritic, despite the rather extensive occurrence in Skt. of -tas ablatives from all sorts of nouns (Whitney 1098b). It seems to belong naturally with -ito, which I have noted only from fem. nouns; -ito, however, occurs also from nouns that are usually masculine, and perhaps might better be placed above; see § 10.71. Other cases: bhamito LV 75.1 (vs); Mv i.91.2; 127.14 (both prose); aiguitto (v.l. °ito) Mv iii.358.4 (prose); tasmād (1) rddhitah Divy 223.28 (prose).--Masculine, or doubtful in gender: samādhito SP 25.3 (vs); Mv ii.209.2; 241.2 (both prose); vyādhito Mv i.308.8 (prose); Suv 38.11 (vs).

10.134. Abl. -itu, m.c. for -ito: -tafitu Gv 213.16 (vs). 10.135. Abl. -itu, m.c. for -ito: pratipattitu Šikş

4.16, 17 (vss). 10.136. Abl. -yātas: nonce blend of -yā (for -yās) and -tas; cf. Instr. -īyena, § 10.129. tasmād (!) devanagaryātas Divy 223.7 (prose); cf. tasmān nagarīto and tasmād rddhitah, above.

10.137. Abl. -ihi: as in Ap. (Alsdorf, Kum. § 24 for fem. stems; as gen. for masc. and fem. stems ibid. §§ 23, 24, and Jacobi Bhav. § 28, San. § 17). durgatihi durgatim gacchanti Mv i.27.9 (prose), 'they go from one evil fate to another'. That an abl. sing. (not pl.) is meant is proved by the sense and by adjoining parallels in -dlo from *a*-stems.

10.138. Gen. -yas, -yo, instead of -yas (cf. also -ya, -ya above), occurs in both *i* and *i* stems; but the only unambiguous cases are in Gv (by misprint? this text is full of misprints). If genuine, however, the form could be explained as analogical to the ending -yas of the nom. pl. (devyas). Even in Skt. monosyllabic *i*-stems

have the same ending in the nom. pl. and abl.-gen. sg., and in our dialect this condition is more widespread; cf. above, § 10.113. All in prose: -galyo (to gati) Gv 106.20; vāsantyo (to °ti) 223.16; 224.15, 17; prasāntarutasāgaravalyo (to °vatī) 286.23. 25; śrīmatyas ca (to śrīmatī or °ti) 466.11, 13, but 2d ed. °tyā (by em.?). The only other possible case is bhūmya ākārān Dbh.g. 19(355).23 (vs) where bhūmya might stand for °yah in samdhi, but probably intends the ending -ya, § 10.126, hiatus being freely allowed in verses; note that bhūmya occurs before a consonant on the next page, 20(356).21.

10.139. Gen. -aye. A reflex of the later Skt. and general MIndic use of gen. forms in dat. function is the occasional appearance of the reverse, the use of Skt. datives (or MIndic derivatives thereof) in gen. function: nāsli smrtaye hānih Mv i.160.12 (prose), 'there is no loss of memory' (one of the 18 āvenika buddhadharma). All mss. smrtaye; there is no doubt of the gen. function nor of the stem smrti. Senart emends to smrtiye. But the following forms in -ayi confirm smrtaye.

10.140. Gen. -ayi, only in verse, m.c. for the preceding. agrabodhayi kāranāt Šikş 100.17 (vs, cited from Samādh), 'for the sake of supreme enlightenment'. ---bodhayi citta KP 28.8 (vs), 'thought of enlightenment'.

10.141. Gen. -Iyam. A MIndic form of the locative ending (§ 10.144), but according to the mss. at Mv ii.444s1 i 12 it is also used by analogical extension in genitive function (-Iya, -iye etc. being used for any oblique case including loc.; hence the loc. form is also used for another oblique case). The phrase is lasyā...rājadhītuņ... krīdayāntīyam ramanlīyam pravicārayantīyum etad abhūsi. Senart emends all three to ⁹iye. Cf. the ending -āyām as Abl.-gen. of ā-stems, § 9.52.

10.142. Loc. -āyām (also -āya or -āyā; -āye, -uye). The ending of a-stems, taken over into i-stems sporadically by influence of an immediately preceding and coordinate a-stem loc.: anultarāyām samyaksambodhau SP 328.2, 3, 5, 7, 8 (prose) in both edd.; but the Kashgar rec. (Thomas ap. Hoernle, MR 134 f.) reads °sambodhāyām, carrying over the ending of anullarayam. A very similar case, tending to confirm the reading -āyām, occurs in: anullarāyām samyaksambuddhāya, or °āyā Mv i.40.6; one of the latter must be read. Senart emends to anullaraye, and keeps 'sambuddhāya, but wrongly takes it as a form (presumably dat.) of a neuter stem sambuddha, said to = sambodhi. There is no such stem; the SP passages cited in Senart's note p. 404 contain -buddhalvam, not -buddham. We must certainly read a loc. fem. from either sambodhi or sambuddhi; the latter would be closer to the mss., and sambuddhi exists in Pali. The seemingly dat. form (samyaksam)bodhaye, on which see Dict. s.v. vyākaroti, may really represent -aye and be meant for the loc. which is normal with that verb.

10.143. Loc. -iyam. Has been noted only once, in Mv i.147.6 (prose) jātiyām; the next-line reads jātau. Should we read jātiyam, with the following group?

10.144. Loc. -iyam (cf. § 10.141 for the same as genitive). MIndic for $-y\bar{a}m$ (cf. Pali -iyam beside -iyā, the usual oblique-case ending; acc. to Pischel no -iamor the like, loc. in final nasal, seems to occur in Pkt.). It is almost limited to Mv; in other texts I have noted only dharaniyam Suv 227.7 (vs), where Nobel emends to "niyām. In Mv however it is fairly common, in prose as well as verse; the following examples are all from prose, except as marked (vs): (*i*-stems) lipīyam Mv iii.184.6 (same, with v.l. "iyam, ii.423.15); yoniyam ii.350.14 = iii.274.16 (vs); bhūmīyam ii.435.15 (same, with v.l. "iyam, iiI.357.9); rātriyam iiI.197.15; (*i*-stems) śrāvastīyam i.4.12; dīpavatīyam ... rājadhānīyam 195.7; 196.15; vaišaīyam 290.6; vārānasīyam 303.5; 326.3; ii.66.8; iii.145.19 etc.; prthivīyam i.347.11; ii.208.12; 282.9, 13; iii.410.2, 3 etc.; ujjeniyam ii.30.7; puşkariniyam ii.95.16; 172.10; calurdasiyam 273.10; dhāraniyam 434.11; vilhīyam iii.35.17; striyam 291.4; sankhamedhīyam 328.14; dvādašiyam 340.16; samjayantīyam 364.4.

10.145. Loc. -iyam, = prec. Only in Mv, but in prose as well as verse; not common. pravenjajam (i or i-stem) Mv i.226.10 (vs), repeated ii.29.13 where mss. ⁹ya, Senart em. ⁹ye; but cf. ii.32.3 (prose) and 38.11 (vs) where mss. pravesiyam, senart em. pravenjyam, rightly; vaišāliyam (i-stem) ii.207.14 (prose); (i-stems) durgatiyam (? reading and interpretation doubtful) ii.225.13 (vs); tipiyam ii.434.11 (prose); bhūniyam iii.114.16, 18 (in 18 v.l. ⁹iyam); 357.10 (all prose); subhūniyam 184.4 (v.l. ⁹iyam) = 405.10 (vs, meter doubtful).

Dual nom,-acc. masc. -inau

10.146. From a mase. *i*-stem; the ending of *in*stems taken over. *yūthapalinau* Mv i.360.17 (prose).

Dual instr. -ibhga, m.c. for -ibhyām

10.147. pāņibhya grhya LV 153.13 (vs). There is a v.l. pāņibhyām, unmetrical.

Nominative-accusative plural

10.148. In our dialect a complete blend has taken place between all endings of the nom. and acc. pl. of all i and i declensions, in all three genders. That is, any ending which may be used in the nom. or acc. pl. of any i, i, or in stem, of any gender, is likely at any moment to occur in such a form, either acc. or nom., of any other such stem. It is true that there are differences of frequency in such occurrences. Thus the Prakritic ending -iyo is much commoner with fem. stems, tho it unquestionably occurs with masculines too. It is, of course, always true that in all our texts, as presented in the editions and presumably in the mss., regular Skt. forms are common; in most texts, certainly, much commoner than the dialectic forms which are the exclusive topic of this work. This does not importantly affect the main point, which is that all the endings of all i, i and in-declensions in these two cases are fundamentally interchangeable. It is safe to assume that a large proportion of the regular Skt. forms are the result of secondary regularization in the course of tradition.

10.149. In this respect our language seems to go somewhat farther than any MIndic dialect as described in standard grammars. For Pali, Geiger (82-86) does indeed abandon any attempt to separate nom. from acc. forms in any *i* or *i* stem; he gives aggayo and aggi for both cases alike, and so also *jātiyo* and *jāti*. Pischel's 387 goes equally far for the fems., but his masc. paradigm 377 is less explicit; yet 381 recognizes 'dialectisch fast alle Formen des N. auch im Acc.'. Some interchange between endings originally peculiar to particular types is, of course, recognized in both Pischel and Geiger. Indeed this process, as well as confusion of nom. with acc. pl., has got a good start even in normal Skt., and still more in Epic Skt.

10.150. I suspect that future work in Mindic may reveal more mixture of these forms than Geiger and Pischel suggest; that is, a closer approach to the state of things in our dialect. In any case, it seems to me that realism demands that we make endings, rather than Skt. declensional type or gender or case, the primary basis of our classification of all nom.-acc. pl. forms.

10.151. Without attempting any rigid scheme (which would only confuse our statement), we shall in general begin with endings which exist in standard Skt. as either nom. or acc. of some *i*, \bar{i} , or *in*-declensional type, showing how they are here extended to other types or to the

other case. We shall then follow with endings which do not exist at all in Skt. in these cases. But this order will be violated whenever convenient grouping of forms demands it.

10.152. Nom.-acc. pl. -ayas. The regular Skt. ending of the nom. pl. in short *i*-stems. Twice, at least, it is used as nom. pl. of a masc. *in*-stem, viz.: *dvipayo* (for *dvipino*) hanyanti Mv ii.213.7; 217.12 (both prose).

10.153. It is also frequently used as acc. pl., as sometimes in Epic Skt. (Edgerton, Builetin Deccan College Research Institute, 5, 1943-4, 2, rasayah Mbh. 2.48.9d, Crit. ed.) as well as in MIndic. My examples concern m. and f. short *i*-stems only; probably by oversight I have failed to record cases of *i*-stems (some stems, like ratri, are perhaps ambiguous). Prose examples (all marked as such) occur in Mv and once each in SP and KP. The SP occurrence is noteworthy, since MIndic forms have mostly been eliminated from the prose of that text, especially as presented in the Nep. mss.

Masculine i-stems, acc. pl.: - $r\bar{a}sayo$ SP 50.10; dundubhayas (or fcm.; no v.l.) 69,11 (prose); trayo 198.3. — apāya trayo (so divide) LV 300.21; sarvām (so read with best mss.) sativa kareya ekamatayalı 335.19 'one might make all beings of one mind'. — $r\bar{a}sayo$ Mv i.107.16 (prose); lecchavayo 262.2 (prose); rsayo ii.217.15 (prose); iii.45.4 (vs); 364.4, 6 (both prose); $j\bar{n}dlayo$ iii.86.8; 292.7. — -vanaspatayo KP 30.1 (prose). — samādhayah Lank 24.16.

Feminine *i*-stems, acc. pl.: râtrayo SP 91.3 (at end of a jagatī pāda; rātrīr, making a tristubh, would be just as satisfactory metrically); niruktayaś 374.2. kāmaratayah I.V 192.5; 327.10, 19. — bhûmayo Mv i.102.11.

10.154. Nom.-acc. pl. -aya, only in verse, m.c. for the preceding: *vyūdhaya bhonti* SP 96.15 (nom. pl.; WT °ayu with ms. K').

10,155. Nom.-acc. pl. -inas (-ino etc.). The nom.acc. pl. ending of *in*-stems in Skt. is used also with *i*-stems, chiefly masc., but once or twice even fem. In most texts rare and only in verses; in Mv also in prose.

Masc. nom.: durbuddhinas ca SP 272.1. — asino (mss. asilo, em. Seuart) Mv i.17.6 (prose); samabuddhino 134.13 (prose); nirmānaratino 240.4; ii.16.4; -patino i.216.16 = ii.19.13 (vs); i.361.7 (prose); -buddhino iii.457.2.

Fem. nom.: varamālyavastradhāriņo pramadā Mv i.149.2.

Fem. acc.: (vetti...) lipino 'kşaradrsyarūpām LV 125.12. The stem lipi is fem. even in LV; cf. shortly after, 125.18-19, katamām...lipim.

10.156. Nom.-acc. pl. -inā. Instead of -ino, the mss. of Mv read -inā at least twice, as nom. pl. masc. of an *i*-stem. Senart emends to -ino both times. The form may however be real; it would merely imply an *a*-extension of an *in*-stem (§ 10.3) and then transfer to *i*-stems. asinā (mss.) Mv i.10.13 (vs); lokapatinā (mss.) i.204.12 (vs). To be sure, the latter passage is repeated with °patino ii.8.7.

10.157. Nom.-acc. pl. -ina, only in verses, m.c. for -inas. The following are nom. pl. of in-stems: prânina SP 89.12; -cârina RP 17.9.

10.158. Nom.-acc. pl. -īno. Once at least in the mss. of Mv for -ino, nom. pl. of a stem in -(v)in: medhāvino (mss.) Mv i.274.3 (prose). Senart em. medhāvino; but cf. next.

10.159. Nom.-acc. pl. -ina: like the preceding, for -inas, once in the mss. of Mv as nom. pl. from a stem in -(v)in: -caritāvīna (mss.; before b-) Mv ii.218.14 (prose). Senart em. ^ovino; but cf. prec. and next.

And once as acc. pl. of a (masc. or fem. adj.) *i*-stem, in a verse, where it may well be m.c. for *-inas: upalam-bhadrş!ina* SP 383.12 (certainly acc. pl., with *bhikşu* or *bhikşunī*).

10.160. Nom.-acc. pl. -ini: the regular nt. ending, is used with indubitably masc. and fem. nouns, i and stems. Sometimes one might speak of 'change of gender', but this can hardly be accepted as describing neuter forms from stems like rsi and cefi. The following examples seem inexplicable except on the supposition that formal gender-distinctions were breaking down (see §§ 6.1 ff. and especially 6.4). trnakutīparnakutīni krtvā Mv i.284.1 (prose), 'making grass-huts and leaf-huts'. One ms. °kulliparnakuțțini. În Mv i.338.5 șanņavati (mss. °tim) pratyekabuddhakofini (nom.) nirvrtāni svayambhuno. In ii.211.16 (prose) mss. Irnakutti- (v.l. °kuti-) parnakuttini (Senart em. kufini; acc.) samsthāpeti. — ceļini (acc.) prechati ii.172.15 (prose; masc. or nt. pronouns tesām, te, are used of these same maidservants in lines 10, 11 of the same page); etāni vanaspatini 178.7-8 and 9 (prose) 'these trees', and etani ca puşkarinini (prose; mss. oina, cf. preceding paragraph) but here elāni and nirodakāni make Senart's em. attractive) nirodakāni id. 10 (all these nom.); rsiņi bhojeti iii.42.14 (prose) 'feeds the rsis'; -kutini Mv iii.168.14 (nom.; so Senart, mss. -kuți or kuțțini); dhānyajātini 178.3 (nom.); others (acc.) in Dschi, NAWGött. 1949, 276 ff.; trini with m. and f. nouns, §§ 6.14, 16.

10.161. Nom.-acc. pl. -īnī, with lengthening of the final vowel of -ini m.c.: trīņī ca yānāny... SP 33.4 (vs). Kashgar rec. trayas.

10.162. Nom.-acc. pl. -yas, the nom. pl. ending of polysyllabic fem. *i*-stems, is used at least once in a masc. *i*-stem (as nom.): *-vanaspalyo* SP 126.10 (vs). One ms. has °palayo, which is metrically less attractive (end of odd pāda of anustubh).

It is also used, much more commonly, as acc.; the following concern fem. i-stems. Frequent is kolyah (°yo, etc.) as acc. of koli (which seems to be regularly an i-stem in our language, so far as it can be said to recognize such distinctions at all): SP 30.15; 63.9; 335.8; LV 223.17 (but most mss. °yāh); 299.2; Mv i.174.14; Samādh 8.16 (prose) etc., 22.23 (vs); Suv 23.16.—Others: bhiksunyo varjayen SP 279.5 (so read with Kashgar rec. and WT; KN with Nep. bhiksunīm); kumāryas ca vivarjayet SP 279.12; krandantyo 'paśyat LV 301.22 (prose).

10.163. Nom.-acc. pl. -ya, in verses, m.c. for the preceding. nadya LV 197.5; bhūmya Dbh.g. 38(64).13 (acc., not loc.; read te bhūmya sapta suvišodhita-).
10.164. Nom.-acc. pl. -yā(s), perhaps a blend of

10.164. Nom.-acc. pl. -yā(s), perhaps a blend of -yas with the nom.(-acc.) ending $-\bar{a}s$ of a (and \bar{a}) stems; or analogical to the abl.-gen. sg. ending $-y\bar{a}(s)$, the reverse of the process suggested in § 10.138. Cf. the endings $-iy\bar{a}$, $-iy\bar{a}$, §§ 10.170, 174. Instead of kolyas (above) we find repeatedly kolyās (or $^{\circ}y\bar{a}$ before voiced sounds). The following are acc. pl. kolyā(s), all in verses: SP 13.10; LV 113.17; 299.11 (and most mss. in 223.17 kolyāh, Lefm. with ms. A kolyah).

In LV 78.18 (vs), for hayagajarathapanktyā (vāhanā), two mss. (including the best, A) are said to read °paktyā; this must surely have intended °pattyā(s), acc. pl. to (masc.) patti, 'footsoldiers', the fourth division of the army (the other three precede it); doubtless read so. In the sequel all four divisions are referred to in more detail, the footsoldiers in 79.7-9, which confirms my interpretation. Cf. also LV 80.13. Tib. seems to have had a quite different text (I cannot find a basis for Foucaux's transl.).

More dubious is kşāntyās te jagi vividha duruktāh LV 164.5. Foucaux 'Tu as... supporté avec patience bien des paroles dures', which seems to imply kşāntās. Foucaux renders the Tib. 'supporte avec patience de nombreuses paroles mauvaises', which suggests interpretation of kşāntyās as a gerundive (as if kşāmyās); Tib. text khyod ni... bzod, which if I am not mistaken may mean 'you have endured'. The form is doubtful, and in any case hardly belongs here; a rendering 'thy various (instances of) patient endurance(s) in the world are hard to declare', as if nom. pl. of kşânti, would ignore the standard meaning of durukta and Pali durutta (see Childers s.v.).

nidrākşyā, n. pl. f., 'sleepy-eyed', Mv ii.203.1 (vs; mss., see Dict. s.v. nidrākşa).

10.165. Nom.-acc. pl. -in, -im. The Skt. acc. pl. ending of masc. *i*-stems may be written -*im*; in our dialect an orthographic variant for -*in*. It seems to occur at least once as nom. pl. masc.: hayagajarathapattim sainya śrimadvicitrām, dvāri sthita nrpasyā śrūyate uccaghosāh LV 80.13 (vs). Several mss. °patti or °patti, but all vicitrām; both edd. as text. No possible construction as acc.; the forms must intend nom. pl. with sthita.

It is also used as acc. pl. fem., once even written -in(for jātim [?] see § 10.166): tebhyaś ca vaidāryamanihārajātebhyo dirghapanktīn kşatantīn apašyat Gv 518.25 (prose). Here kşatantī- is unmistakably fem., and we seem clearly to have the masc. ending on fem. stems. (A singular would be implausible in the context and would require -Im.)

Finally, -in is used in the acc. pl. of in-stems, for Skt. -inas (in verses, and in prose of Mv): adhimānin SP 279.3 (stem adhimānin); prāņin 280.3 (stem prāṇin). — -chandaka-kālodāyin (v.l. °yīm; followed by ā-) Mv ii.233.16 (prose), 'Chandaka and Kālodāyin' (acc.). — medhavīn Gv 25.13 (in anustubh vs, to be read medhāvīn ananlamatīn, stem medhāvin; but 2d ed. medhāvino 'nanla°). vašīņis ca Dbh.g. 51(77).2 (stem vašin).

10.166. Acc. pl. -im. Seems clearly recorded a few times in Mv. If the readings be accepted, it may be one more example of confusion between nom. and acc. endings sing. and pl.; cf. -ih, etc., as nom. pl., § 10.190, and see § 8.83. Note, however, that it might be merely a phonetic (MIndic) form of Skt. -in, often written -im in our texts; the shortening of the vowel would be normal in MIndic when the nasal is retained. hastim Mv ii.453.14 (mss.; prose); a plural acc. seems clearly required; Senart em. hastīm (Skt. hastinah); ekām vā jātim dvau vā jatim trayo vä jätim catvāri vā jātim etc. Mv i.228.16 fl. (prose). In this list one ms. sometimes reads jatih, but apparently no form with long I is recorded; Senart prints jatim, tho surely a plural must be meant. In a similar list Mv ii. 284.8 ff., Senart reads always jätim (cf. § 10.165); but here too the mss. mostly read jātim without v.l. (a few times v.l. jātim). Cf. § 12.55. 10.167. Nom.(-acc.) pl. -is; the ending of fem. i

10.167. Nom.(-acc.) pl. -is; the ending of fem. *i* and polysyllabic *i*-stems in the acc. pl., is used as nom., not only with fem. nouns (short *i*-stems as well as *i*-stems), but even with masculine *i*-stems. Examples, all from verses:

Fem.: (i-stem) bhittis ca SP 83.5.—(i or i-stems) bheris c- LV 194.2; yajñakolih 223.18.

Masc.: krlāňjalih ŠP 57.3. — vanaspalis ca Gv 334.12.

10.168. Nom.-acc. pl. -iyas, -iyo (usually the latter, without regard to samdhi); Pali -iyo; the Skt. ending of monosyllabic (normally fem.) *i*-stems, in both nom. and acc. pl. Here it is used with *i* and *i* stems of all types, masc. and fem., and even with masc. *in*-stems. Cf. \S 10.172.

Masc. i-stems, nom.: osadhīvidhiyo (? as fem., § 10.89), so probably read Mv i.115.1 (vs) (Senart osadhividhīyo; most mss. osadhividhiyo, unmetrical); prānjaliyo 167.14 (vs; the syllable $pr\bar{a}$ omitted in mss.), and the same 191.3 (vs); *lecchaviyo* (v.l. °iyo) 256.9 (prose); -paliyo (v.l. °iyo) ili.43.2 (prose).

Masc. i-stem, acc.: rāšiyo (mss. corruptly ramhiyo) My ili.78.4 (vs).

Masc. in-stems, nom.: -dvīpiyo (stem dvīpin) Mv ii.62.4 (vs); hastiyo (stem hastin) 458.2 (prose).

Fem. i-stems, nom.: oşadhiyo SP 130.11; 131.1

(vss). — bhittiyo (v.l. °iyo) Mv i.24.13 (prose); arciyo (§ 16.39; Pali acci, acci, fem.!) 25.14 ff. (prose); dundubhiyo (or masc.) ii.39.15 (vs). — şaṇṇavatiyo Gv 181.15 (prose).

Fem. i-stems, acc.: oşadhiyo LV 151.7. — gatiyo Mv i.102.14 (prose); -samāpatliyo (v.l. °iye) 127.5 (prose); bhūmiyo (mss., Senart em. °ayo) 192.17 (vs); -śreniyo ili.78.3 (vs).

Fem. polysyllabic *i*-stems, nom.: *jhallariyo* SP 52.1 (vs). — *koliyo* (often with v.l. ^o*iyo*) Mv i.59.9 and 11 (prose); 61.10, 13; 117.1; 119.4; ii.353.9, 10 (these all vss); *sravanliyo* ii.92.15 (vs); *dhātriyo* (v.l. ^o*iyo*) 433.11 and 13 (prose); *nāriyo* ii.8.19 (vs).

Fem. polysyllabic *i*-stems, acc.: koliyah or ^oyo, common in verses, also in Mv prose: SP 334.6; 335.4; 336.5, 8; LV 53.1; Mv i.47.3, 7: 57.8 (prose); 243.13 (prose, v.l. ^oiyo); RP 7.15; 9.1; 38.2; -sahasriyo Mv i.5.13; 6.4, 7, 12; 310.1 (all prose); nāriyo iii.451.5 (vs).

10.169. Nom.-acc. pl. -iya, noted only in verses, m.c. for the preceding.

Polysyllabic *i*-stem, nom.: *nidaršayantiya* (= Skt. °*tyas*) LV 326.18.

Masc. in-stem, acc.: dvipiya Mv ii.222.12 (so read with mss.; = Skt. dvipinas; Senart em. dvipayo; meter requires short final).

[10.170. Nom.-acc. pl. -iyā?, probably only by error of tradition. In Suv 28.9 (end of line of vs) text -koļiyā, acc. pl., but v.l. -koļayah; read either this, or -koļiyah, or -koļiyo. Cf. -iyć § 10.174.]

10.171. Nom.-acc. pl. -70, a regular and very common ending for the nom. and acc. pl. of all *i* and *i*-stems in most Prakrits, especially for fem. nouns, but also masc. (Pischel 377, 380, 387). In our dialect its place is almost wholly taken by -*iyo* (next section), which is substantially the same form, but once at least -*io* is recorded: $r\bar{u}pyamayio$ (n. pl., *i*-stem; no v.l.) Mv iii.228.16 (prose).

10.172. Nom.-acc. pl. -iyo (without regard to samdhi; but also -iyas etc. with Skt. samdhi) is not recorded in Pischel or Geiger, but is much commoner in our dialect than (Pail) -iyo or (Pkt.) -io, of which it may be called the normal representative. It is used for both nom. and acc., chiefly with fems., both *i* and *i*-stems, much more rarely with masc. stems. In most texts only in verses; also in prose of Mv, where the mss. often vary between -iyo and -iyo (see under the latter §-10.168).

Masc. i-stem, nom.: -dānapalīyo (mss.; Senart em. °iyo) Mv iii.43.8 (prose).

Masc. i-stem, acc.: rāšiyas (v.l. [°]iyo; construe as acc. with preceding *adrāksīt*) Mv iii.318.5 (prose).

Fem. i-stems, nom.: osadhiyo SP 127.6 (here $^{\circ}yah$, final); 129.7; 130.6 (in these v.l. $^{\circ}yd$); 131.8. — saktiyo (v.l. $^{\circ}iyo$) Mv ii.88.3 (vs but meter indifferent); -jdtiyo ii.95.14, 17; 99.17 (all prose); osadhiyo 353.12; durgatiyo 387.10; gatiyo iii.377.18 (prose); sreniyo 442.8 (prose).

Fem. *i*-stems, acc.: *dundubliyo* (or masc.) SP 331.5 (object of *ninādayanto*; misinterpreted as nom. by Burnouf and Kern). — *sampattiyo* Mv i.32.16; 33.9 (both prose); *durgatiyo* ii.348.13; 385.21.

Fem. i-stems, nom.: bhiksuņiyo SP 26.9; koļiyo ca 303.3.—(examples all prose) sapatniye yo Mv i.16.8, mss., read sapatniyo ye (or yā, nom. pl. fem.); nadiyo 21.9; 360.20; koļiyo 59.7; 248.12 and 14; rūpyamayīyo kinkiņiyo 195.6; 196.6; sadršiyo dhātriyo 227.8; vāpīyo, puşkariņiyo 308.4; gurviņīyo 308.9; parivādaniyo (mss., Senart em. "diniyo) vallakīyo 308.11; mātriyo (so read, see Dict. s.v. mātri) bhaginiyo 351.4, 8; etc., common.

Fem. i-stems, acc.: $\delta at d p a d i y o$ SP 86.1; n a d i y o 196.9; -ko f i y o 253.6, 8. — n a d i y o Mv 1.28.10; i i i .55.5; 323.17; v d p i y o ii.449.4, 5, 7; d d s i y o iii.42.1; p u s k i r i n y o 288.7; ko f i y o 1.174.16; 175.1 (vss; may be nom.; but in both meter favors v.l. ko f i y o); n d r i y o iii.2.17; 451.9; etc.

Fem. monosyllabic i-stem, nom.: striyo (in 130.4 v.l. striyo) Mv ii.126.10; 130.4 (both prose).

10.173. Nom.-acc. pl. -iya, only in verses, presumably for -iyo m.c.; (fem. i or i stems) -koliya SP 251.11; oşadhiya LV 75.20.

10.174. Nom.-acc. pl. -iyā, as nom. or acc. pl. of a fem. i-stein, uncertain. (Cf. -iyā, § 10.170.) SP 237.3 (vs) bhiksuniyā ca, ed. with Nep. mss.; but Kashgar rec. bhiksunikāš ca. - pisācaniyā Mv iii.292.1 (prose), mss., nom. pl. (see Dict. s.v. pisācinī, °canī); Senart em. °īyo. 10.175. Nom.-acc. pl. -īye, cf. the ending -āye in

10.175. Nom.-acc. pl. -īye, cf. the ending -aye in the nom. pl. of a-stems (§ 9.92). Rare and somewhat uncertain.

Fem. i-stems, nom. pl.: vijaniye Mv i.222.7 (vs; but Senart em. °iyo, with the parallel ii.24.15); ulthāpaniye gāthā (mss.) ii.26.13 (prose); Senart em. °iyā, as if from stem °niya; puşkariniye 177.16 (prose), mss. (Senart em. °iyo); but it may be an oblique sing. (loc.?); °iye to be kept in any case; -koliye (mss.; Senart em. °iyo) 302.7 (prose); hasantliye (Senart em. °iyo) 425.6 (prose); paņditāye (so mss., Senart em. °āyo) striye (Senart with v.l. °iyo) iii.393.11 (prose); dhātriye (v.l. °iye, cf. next; Senart em. °iyo) 405.6 (prose).

10.176. Nom.-acc. pl. -iye; possibly only in verses m.c. for preceding, but cf. v.l. in Mv iii.405.6, at end of the preceding. Nom. of *i*-stem: -kumāriye Mv iii.307.7 = 308.7 = 309.7 (vs). Meter requires short *i*; in the first occurrence v.l. °*ikā*, in the second v.l. °*iyam*; in the third no v.l.; in all three Senart em. °*iya*.--More than doubtful is vimsatiye SP 115.7 (vs); so text; if correct, acc. pl. of extent of time, 'for twenty (years)'. But doubtless read vimsati ye (§ 21.31).

10.177. Nom.-acc.pl. $-\overline{i}$, one of the regular Pali and Pkt. endings for nom.-acc. pl. of all *i* and \overline{i} and *i*.stems, of all genders. Geiger 82.1 considers it (in masc. *i*-stems at least) analogical to the *a*-declension. Without denying the possibility of such influence, it seems to me that the main source of the ending is the acc. pl. (Skt. $-\overline{i}s$) of fem. *i* and *i*-stems, which was transferred to the nom. and to both acc. and nom. masc. (and nt.).

10.178. In our language this ending is very frequent in all declensional types and genders. In most texts, to be sure, it is restricted to verses, or practically so; in Mv, as usual, it is also common in prose. Prose passages in the following are marked as such.

10.179. Masc. *i*-stems (or, in some compounds, *i*-stems), nom.: paripūrņacārī ca SP 13.4; krtānjalī 28.9; *iuci* 46.3; 59.1; maharşī 56.6, 198.3; vinasļadrstī 129.2 (followed by s_7 ; in same line višuddhadrstayayh); etc., common. — nāgāvalī LV 193.5 having rows of elephants' (so Tib.); krtānjalī 236.12; ratnapānī 299.1; dharmacarī surapulrāh 364.4; etc. — jiātī Mv i.220.19 = ii.23.4 (vs; so meter demands, but v.l. first time and all mss. second time jiātii; Senart em.); (nila)yaştī i.259.14, 18; 260.3, 5, etc. (all prose; v.l., and sometimes all mss. °yaşti); vanaspatī (v.l. °lih) ii.353.12 (end of line of vs); dānapatī iii.45.20 (*i* required metr.); sāralhī 120.13 (end of line, v.l. °l); jitārī 141.2 (end of line, v.l. °i); guhyakādhipatī 380.11 (end of line, v.l. °l). — asangabuddhī RP 15.2. — bālabuddhī Dbh.g. 7(343).14. **10.180.** Masc. *i*-stems, acc.: durmatī SP 272.4

10.180. Masc. i-stems, acc.: durmali SP 272.4 (confirmed La Vallée Poussin, JRAS 1911.1076); sūryarasmī 126.5 (in both of these meter indifferent as to quantity of final). — midhamgiri LV 197.3. — jñāti Mv i.359.14 (in line 12 jñātlīm).

10.181. Masc. in-steins, nom.: kathamkathī SP 35.4; na mohavādī (but WT with Chin. namo ha vādī, n. sg. m.) 56.6; lajjī 59.1; hastī 111.8; paksī 331.4; etc. — hastī LV 28.5; 193.5; sukhī 92.15; prašāntacārī 219.11; dhyānadhyāgī 297.10; paksī 341.18; etc. — vipratisārī MY 1.37.4 (prose); paragaņī 74.3; -dhārī 150.13; -vicārī 151.4 =

10.182. Masc. in-stems, acc.: dharanitalesmim śayi I.V 194.15 (for śayinah). — läbhi Mv i.31.1 (prose); 32.1 (prose); in 31.1 läbhino also occurs, in same construction; -kumbhatūni ii.150.4 (prose); paksi 221.18 (v.l. paksi, hut meter favors °i).

10.183. Fem. in-stems: The same form is even used as fem. of (original) in-stems: duhkhi Mv iii.285.15, vipratisārī 16 (with māradhītaro); nom. pl.

10.184. Fem. i-stems, nom.: oșadhi SP 129.8. -prănajăli (v.l. °li) Mv ii.99.2; śreni iii.161.14 (prose).

10.185. Fem. i-stems, acc.: dvalrimsali SP 68.11; 69.1 (antarakalpa; in the first both edd. °lim, against all mss. according to KN, who in the very next verse, nevertheless, keep °li in the same phrasel); nananirukli 130.8. kamarali LV 283.12; vajräsani (for °sani) 339.4. — apäyabhūmi Mv ii.333.19 (v.l. °mi; meter doubtful).

10.186. Fem. i-stems, nom.: śatāpadī SP 84.2; dāsī 111.9; kojī 305.4 (four times); raināmayī 331.7; bhiksunī 384.12, etc. — sarvanadī ca LV 77.10 (prose; all mss. are reported showing °nadī, before a voiceless stop; Calc. °nadyas); evarūpa gātha tūryaniscari 170.6 (so read, cpd.: 'such verses were proceeding from the musical instruments'); -koļī 196.7; šaklidhārī 202.1. — koļī Mv i.338.7; kimnarī ii.97.21 (prose); ceļī 172.2, 15 (both prose); bherī 186.1; mrgī 235.2 (prose); sadršī iii.222.1 (prose); rākṣasī 292.1 (prose); nārī 387.9; puṣkarinī 438.20. pilhī Divy 221.3 (prose, mss.; before a-; ed. em. vilhyaḥ). grāmacārinī Suv 16.7 (end of line; no ms. has -ḥ). — nadī Samādh 19.18.

10.187. Fem. i-stems, acc.: -koți SP 130.4; văpī nadiyo 196.9 (both acc.); bhiksuni 383.12 (end of line). koți LV 223.11; pudini 298.3. — koți Mv i.125.17; ii.319.22 (both v.l. koți, meter indifferent or doubtful in both); năgi ii.188.7.

10.188. Neuter i and in-stems: nakşatrā saisā (stem -šašin) satārakaravī (stem -ravi) LV 284.3. In Skt. and Pali nakşatra (nakkhatla) recorded only as nt., so also AMg. acc. to Ratnach. (but Sheth gives both m. and nt.). bhikşusatāni dhyānalābhī utthāya KP 138.4 (prose). Masc. form because of bhikşu? But in 139.2 the same phrase has "dabhīny.

10.189. Nom.-acc. pl. -i, is much less common than *i*. It occurs, however, not infrequently, in metrically indifferent positions and in the prose of Mv, from *i*, *i*, and *in*-stems alike. Senart often, but not consistently, emends to -*i*. The mss. themselves often vary; under the ending -*i* some such variants are cited.

Masc. i-stems, nom.: rājāna ye mahipali (369.3 kşitipati) cakravartino SP 35.13; 369.3. — anurāgabudahi Mv i.79.6 (all mss., end of line of vs; Senart em. °i); rşi 189.17 (vs, but meter indifferent); nirmāņarati 212.15 (prose; wrongly printed as if part of a cpd.); prašānladrşi yalhā i.151.8 = 224.19 = ii.27.19 (so mss. always; in first two places Senart em. °drşipathā; in the third he rightly keeps mss. reading); -pāņi iii.366.2 (mss., end of line of vs; Senart em. -pāņī).

Masc. i-stem, acc.: sakhi LV 165.10.

Masc. in-stem, acc.: loka-vicāri Śiks 330.13 'roaming the world'.

Fem. i-stems, nom.: lipi LV 146.5. — vrsli Mv ii.92.16 (but meter seems to require vrsli!); samgili 151.15; 153.10; 160.19 (all prose). — -drsfi leşu bhonti RP 11.15; anguli 24.1.

Fem. i-stems, acc.: nirukti SP 238.1. — daśānguli Mv i.204.16 (so with v.l., required by meter); jāli iii.27.14 (adverbial acc. pl.); divgāni sampatli i.31.11 (prose; Senart with 1 inferior ms., out of 6, ${}^{\circ}li$; his note assumes neut. gender for sampatli; rather, confusion of gender-forms results in divgāni going with fem. noun; el. RP 23.17 below).

Fem. i-stems, nom.: $n\bar{a}ri$ LV 202.1 (and in 80.11 n $\bar{a}ri$ by em., required by meter, for mss. $n\bar{a}r\bar{i}$); $t\bar{u}ryakoli$ LV 222.7; -koli Gv 254.16.

Fem. i-stems, acc.: dhāraņi SP 294.8; ānlahpuri LV 211.7 ('women of the harem'); calvāri (cf. Mv i.31.11 above) koļi pramadānām RP 23.17.

10.190. Nom. pl. -is, -ih, etc. The nom. sg. ending of short *i*-stems. It seems to occur as nom. pl., doubtless as a result of the fact that the endings -*i*, -*i* are used alike for nom. (and acc.) sg. and pl.; hence by analogy the Skt. nom. sg. ending comes to be used as pl. In fact the mss. show such forms more commonly than the editions admit. *isir ivā gagaņe supūrņāh* LV 296.13 'like full moons in the sky'. For *isinah*; but here the form might be interpreted as *isis* plus hiatus-bridging -*r*. — caluro nidhih Mv ii.436.9; 438.1, etc. (prose, no v.l.; repeatedly in the context). — apramāņāh saltvavimuktir anuvartayilavyāh, Gv 458.9 (prose). In preceding line -dhātur is similarly used as nom. pl., § 12.57.

Vocative plural

10.191. Voc. pl. -I, ending of the nom. used as voc.; masc. in-stem: chandakakālodāyi Mv ii.233.17; 234.2 (both prose). Dvandva cpd., but hardly to be interpreted as dual; plural verbs go with the form.
10.192. Voc. pl. -iho. Cf. -āho from a and ā-stems,

10.192. Voc. pl. -iho. Cf. -*āho* from a and *ā*-stems, **§§** 8.88; 9.100. Fem. *i*-stem: *deviho* (v.l. *devi*) Mv ii.449.16 (prose).

Instrumental plural

10.193. The commonest non-Skt. ending is -*ihi*, very common in Mv prose as well as verse, and occasionally in verses elsewhere. It is the regular Pali and Pkt. ending of all *i*, *i*, and *in*-stems. Less commonly, but in Mv not rarely, -*ihi* (with short penult) appears; it is recorded for Pali (Geiger 83.8) and Ap. (Pischel 387). Forms in -*ibhi(s)* also occur, even from short *i*-stems, as In Pali ('archaic', Geiger 83.7); and -*ibhi* for -*ibhis* occurs in verses m.c. Forms with final nasal (ending in -*him* or -*bhim*) do not seem to occur, at least as clearly established readings, tho occasionally they are recorded as variants of individual mss. (e. g. Mv ii.464.14 v.l. *bhaginihim* for text '9*i*). 10.194. Inst. pl. -*i*hi, found with all types of *i*, *i*,

10.194. Inst. pl. -Ihi, found with all types of i, \bar{i} , and *in*-stems; only in vss. except for Mv, where it is common also in prose (such examples are marked).

Masc. i-stems: munihi Mv i.72.10; yūthapatihi 360.6; 361.14 (both prose); jhälihi il.154.9 (prose); rsihi (v.l. °ihi) il.216.18 (prose); gajapatihi (v.l. °ihi) il.311.2 (prose); manihi (vs. but i not required metr.) 360.10; asthi 326.14; iil.19.21; anjalihi il.352.20; ksuraptlihi (masc. adjective) iil.180.8 (prose); agnihi 445.12.

Neuter i-stems: akşîhi Mv i.363.18 (prose); ii.234.17 (prose, v.l. °*i*hi).

Masc. in-stems: viparītasamjnīhi SP 281.11. paksihi Mv ii.211.1, 3, 7, 11; 212.13, 14 (all prose); vikālacārihi iii.328.18 (here prob. in sense of loc., cf. § 7.30).

Fem. i-stems: nānāniruktīhi SP 45.5. — dvātrimšatīhi Mv i.50.2 (prose); bhūmīhi (v.l. °ihi) 128.9 (prose); navatīhi 186.19; sastīhi 197.14 (prose); vikrtīhi ii.126.15; 128.2; iii.145.11 (all prose); -jātīhi ii.248.16 (prose); mustīhi (v.l. °ihi) 452.14 (prose); šaktīhi (v.l. °ihi) iii.19.20; šrenīhi 111.20 (prose). Fem. i-stems, monosyllabic: strihi Mv iii.149.11 (prose); istrihi iii.84.14.

Fem. i-stems, polysyllabic: valihi SP 313.9.—(all Mv citations prose) vāsīhi ... kuļhārihi Mv i.16.14; dronīhi 22.2; sakhīhi 217.3; ceļihi 302.16; ii.171.11 (v.l. ⁹ihi); dāsīhi i.350.1; kimnarīhi ii.97.6, 8; 110.3; puşkarinīhi 171.3; 180.12; ratnāmayīhi 180.12; dhātrīhi 423.9; bhaginīhi (v.l. ⁹him) 464.14; nārīhi iii.41.19; rākşasīhi 75.7; 287.4; 290.8. — -pūļīhi Sukh 50.9.

10.195. Inst. pl. -ībhis (etc.), the ending of Skt. *i*-stems, here applied to *i*-stems, in verses where meter requires *i*; but also twice, from a masc. *in*-stem, in prose of Gv, pratisamvedibhir (stem °vedin) 377.12 and 19. Others: Masc. *i*-stems: krtāňjalibhih LV 297.15; maņibhih

Mv ii.319.11.

Fem. i-stem: -ratibhih (°bhis) LV 37.2; 184.15. 10.196. Inst. pl. -ībhi, for the preceding; only in verses, m.c.: jātibhi SP 330.6; -koţībhi RP 18.15; asucībhi RP 43.18.

10.197. Inst. pl. -ibi, noted only in Mv, where it is not very rare, the much rarer than -ibi. Examples are all from prose, except those labelled vs. Chiefly from *i* and *in*-stems; but, according to the mss., twice from fem. *i*-stems; in both Senart reads -*ibi: vāsihi* Mv *i*.5.4 (5 out of 6 mss.; from vāši 'axe'); anvantihi iii.357.10 (mss.).

Masc. i-stems; katihi Mv 1.89.8; jñālihi il.187.9 (vs, i required metr.); maņihi il.311.6 (mss.; Senart em. °īhi); patlihi il.433.20 (vs; i required metr.).

Fem. i-stems: ašīlihi Mv i.32.7; 38.14 (et alibi); -koțihi 49.12; bhūmihi (i m.c.) 192.12 and 14 (vss); paňklihi 194.4; 249.14 (mss. in the first, and v.l. in the second, paklihi); vimšalihi 307.19 (in line 16 vimšalihi).

Masc. in-stems: dvīpihi Mv i.24.16; sarvaprānihi (repeated with °bhih 82.4) 48.8 (vs, end of line); hastihi 349.18; varmihi (v.l. °ibhih) iii.22.9 (vs, end of line).

10.198. Inst. pl. -ibhi, only in vss, m.c. for -ibhis: -rşibhi LV 163.20 (em.; mss. ^obhis); buddharasmibhi Mv iii.274.6.

Ablative plural

10.199. As in Pali and Prakrit, the originally inst. ending -*i*hi is also used for the abl.: *durgatihi vimokşitā*h Mv i.175.2 (vs); *virakto* ... °*ratihi* ii.145.7 (prose).

Genitive plural.

10.200. The most frequent departures from normal Skt. forms concern irregular quantities of the penultimate *i*. In Skt., all *i* and *i*-stems have the ending -*i*nām, while *in*-stems have -*i*nām; in our dialect each is used often for the other, and not only in positions where metrical requirements determine, in fact in prose, and -*i*nām in other texts than Mv. Furthermore, but only in verses m.c., -*a* may replace the final -ām, resulting in endings -*i*na and *i*na, both of which are found with both *i* (*i*) and *in*-stems. Similarly in monosyllabic *i*-stems, -*iya* may replace -*iyām*, m.c. Finally, we seem to have a couple of occurrences of nonce-forms in -*işām* and -*im* as gens. pl.; see below. Prose examples are marked as such.

10.201. Gen. pl. -inām, from in-stems: grhinām SP 272.5 (not required metr.); gaņinām mahāgaņinām (no v.l.) 298.5-6 (prose); maņdalinām 362.8. — paranirmilavašavarlinām LV 46.21; 150.4 (both prose); bhīlavādinām 91.8 (prose; but v.l. °dini, perh. rightly; passage seems confused); sarvaparapravādīnām 275.13 (prose); kešariņām 311.14. — grhiņām Mv i.75.17 (not required metr.); vijitavāsinām 276.1; paksiņām ii.212.11 (prose); chandakakālodāyinām iii.92.11 (prose; in same passage ii.234.6 °ginām). — śresthinām Divy 143.18, 22 (both prose); dhanīnām 143.22 (prose). — grhiņām RP 29.16. — 10.209]

khadginām Mmk 122.7. -- hilaisinām Dbh.g. 4(340).26. ---vaśavartinām Gv 341.1 (prose).

10.202. Gen. pl. -ina, only in vss. m.c. for -inām. Masc. and fem. i and i-stems: kofina SP 30.13. lipina LV 146.6; svapalina 159.2; rsina 222.20. -- nrpalina Gv 213.4.-In Mv iii.365.8 I would read saptana (so mss.) ratrina; the ed. reads sapla- (em.), then with mss. ratriyena, which is impossible in meter and at least bizarre In form (but cf. §§ 10.129 and 9.69).

in-stems: prāņīna SP 38.3; nimiltasamjāln' iha °īna-iha) 57.7; ayuklayogīna 93.11; nimiltacārīņa (for 120.3; viparitasamjñina 326.5; balacakravartin' (atha) 362.8. - cakravartina LV 311.14 (so, one word). [- grhina RP 29.11 (but see Dict. s.v.).]

10.203. Gen. pl. -inām, from i and i-stems. Noted in prose only in Mv.

Masc. i-stems: bālabuddhinām SP 57.7; maharsiņām 70.5. — rsiņām LV 402.11; kāśinām 406.10, 12, 14. dhurabuddhinām Mv i.86.2 (mss.; Senart dhuta^o); vipulabalavarakīrtinām 136.13 (prose); varabuddhinām 168.5; maharsinām 169.11 (here meter demands "inām!); ii.194.14; iii.434.9 (prose); kāšinām ii.184.2; -sārathinām iii.241.1.

Fem. i-stems: -kolinam SP 12.10; 45.3 (or l-stem). - samgīlinām LV 109.7; lipinām 125.7. — calurašīlinām Mv ii.424.10; asīlinām iii.126.7 (both prose). — -kolinām Samādh 8.31 (or i-stem).

Fem. i-stems (cf. koft or koți, above): nărinăm LV 179.5; 240.17; kinnarinām (so!) 214.18.

10.204. Gen. pl. -ina, m.c. for -inām.

i-stems: munina (separate from the following ye) LV 421.18; dundubhina Mv 1.71.8.

in-stems: dhyāyina agrā LV 169.5, 'O best of

meditaters!'; -samjāina sattvānām Šiks 324.16. 10.205. Gen. pl. -iya, m.c. for -iyām, in monosyllabic i-stems: istriya LV 193.14 (separate from following māya; 'women's wiles').

10.206. Gen. pl. -işām, once in Mv prose, bhāryāpalisam iii.148.13 'of wife and husband'. It is not strange that Senart queries the form. It might be due to influence of s-stems, like the inst. balasā, padasā for balena, padena, (see § 8.41); to be sure is-stems (like jyotis, havis) are much less common than as-stems, and I do not think of any that would be specially apt to have influenced pati. Or could the -sam ending of pronouns be involved? Pischel 370, 381, explains the Ap. gen. pl. ending -haii (regular in all declensions) as borrowed from that source.

10.207. Gen. pl. -im, once, Gv 478.16 (vs), for -inām, from an in-stem: sarvajinānām samkhyām cintenti anavaranacintim, 'they reflect on the number of all the Jinas that have unobstructed thoughts'; stem -cinlin. The regular form -cintinām would be hypermetric. Cf. -ān, -ām as gen. pl. ending of a-stems, § 8.124. --Once this ending seems even to be elided (before i-): udyana (q.v. in Dict.) dhāraņ' ita ... Dbh. g. 20(356).11, 'a garden of dhāraņīs'.

Locative plural

10.208. Loc. pl. -īşu, regular in Skt, with i-stems, is common here also with *i*-stems, but in most texts only in verses where meter requires a long penult. As usual My has also prose instances, which are marked here as such.

Masc. i-stems: -yonişu Mv i.27.3, 12 (prose; in 4 and 10, same vicinity, also v.l. in 12, -yonisu); upadhisu iii.444.15.

Fem. i-stems: -bhillişu SP 51.3, 7; -ralişu 88.5; yonişu 97.2; 358.13. - - ratişu LV 323.6; durgatişu 372.19. - gullişu (for guplişu) Mv i.20.6; 23.1 (both prose); galişu i.53.14 (the same passage printed as prose i.337.5 reads galişu; if verse, meter requires galişu); bhūmiso- 1.102.9 (not m.c.; cf. bhumim in line 8); sampattisu ii.141.10 (not m.c.); jātīsu 220.2; 363.2; 365.16; durgatīsu 328.22; ratīsu aratīşu ca iii.59.2. — ratīsu Samādh 19.21; -gatīşu Suv 48.1, 2, 4; Bhad 16, 20; Gv 236.10.

10.209. Loc. pl. -işū and -īşū, ū only m.c.: vidhisū Gv 55.3; galisū SP 9.6; 48.3.

11. Feminines in $-\bar{a}$ and $-\bar{i}$

11.1. There are cases of fems. in $-\overline{a}$ where Skt. would use $-\overline{i}$, and vice versa.

11.2. The use of $-\bar{a}$ for Skt. $-\bar{i}$ corresponding to masc. stems in -a has been noted chiefly in ordinal numerals. All the ordinals from 4 to 10 present \bar{a} -stem forms, where Skt. uses only \bar{i} -stems. Perhaps this is due to the analogy of Skt. prathamā, dvitīyā, and trītīyā. AMg. has (besides \bar{i} forms) cautilīā, paācamā, and chaițihā (Pischel 449). Here we find catuthāya, m.c. (?) for caturthāyām, loc. sg., Dbh.g. 16(352).8; paācamā Mv ii.240.5; iii.306.21 (but in the same vs LV 388.19 paācamī); sasthāyām, loc. sg., Mv i.127.1,4; sasthāto, abl. sg. fem., Mv i.127.14; saptamā Mv ii.240.6, "mām id. 1.127.14; "māyām, loc. sg., id. i.127.2, 4; astamā Mv ii.240.6; navamām Mv i.142.2; dašamām Mv i.142.3; 193.12 (here Senart em. °mā).

Further, tādrśā appears for Skt. tādrši: tādršiye laksmiye tādršāye vibhūsāye Mv ii.157.9 (note that the form of the adjective follows that of its noun); tām tādršām mama vadhūm LV 138.18 (so all mss., except one tādršā).

And there are even feminines in \bar{a} to *-ant*, *-mant*, and *-vant* stems; see § 18.9, 11, 36, 40, 47.

11.3. We find -I for Skt. -ā in pres. mid. pples: araņyakuļikāye chādyamānīye Mv i.328.1 and 16 (prose; so mss. both times; Senart em. °nāye); kaļļiyamānī Mv ii.429.15; yudhyamānī lii.371.3 (vs) 'fighting' (in same vs in Pali, Jāt. v.142.32, yujjhamānā).

In past pples., suffix -la: prak: li, fem. to °ta, 'mentioned, under discussion': prakrti Mv i.232.16 (for °ti, nom. sg.); prakrliye id. 233.14, gen. sg.; °tiye id. 238.3, instr. sg. All prose.

orddhi? Nom. pl. *orddhyah*, RP 41.18 (prose); meaning not clear, perhaps 'skilled, adept' (sc. in erotic arts?). But the text may be corrupt.

Miscellaneous: vayasyi, to vayasya: $^{\circ}yiye$, gen. sg., Mv ii.249.8 (prose); uttari = uttarā: LV 28.18 (vs) kuttottari (text $^{\circ}ntari$), i. e. kuta(h) uttari, 'how (much less a woman) superior?'; acetani = Skt. $^{\circ}nā$: nom. pl. $^{\circ}nyo$ LV 119.22 (prose); anopami = Skt. anupamā, Mv i.166.12 (vs). Even the noun devatā appears also as devatī repeatedly in Sādh (see Dict.). And vikurvī for the more usual vikurvā, 'miracle', may occur once (Dict.).

12. U and \overline{u} -stems

12.1. The declension of u- and \hat{u} -stems is similar to that of the *i*- and *i*-stems thruout, only somewhat simpler. They are, of course, less common; distinctions between nouns of different genders, and between those which in Skt. belong to different declensional types, are perhaps even harder to discover. The most important distinction is in the oblique singular cases, where fem. nouns rather often show the Mindic endings -ûya, -uya, -ûye, -uye, or the like; they are rarely found in masculines and neuters. But between u- and \bar{u} -stems the distinction (by no means rigid even in Skt., Wackernagel III \S 101) can no longer be demonstrated. We shall include here MIndic *u*-stems which in Skt. belong to other declensional types, us-stems (cakşu, etc.) or others such as vidu (vidvāms: vidus), maru (marul). However, u-stem forms from r-stems will be treated under r-stems, §§ 13.21-28.

Forms with intrusive -n-

12.2. Neuter u-stems in Skt. contain, in the oblique cases of the singular, an intrusive -n- (e. g. abl.-gen. madhunas), analogical to nt. *i*-stems, which got the -nfrom in-stems. In BHS this -n- spreads to masc. u-stems. They show, first, abl.-gen. sg. forms in -uno, -unah (as masc. i-stems also show -ino, -inah, § 10.81); and then, by analogical extension, nom. pl. -uno, and acc. sg. -unam or (m.c.) -una (i-stems also have nom. pl. -inas, -ino, § 10.155; perhaps also a stray nom. sg. -inas, § 10.3; I have failed to note any acc. sg. -inam except from instems, cf. § 10.3). Pali has sporadically a masc. acc. sg. bhikkhunam, -bandhunam (as well as gen. bhikkhuno, and rarely nom. pl. -uno). Geiger (83.1) calls it analogical to in-stems, but notes with surprise that Pali (like BHS, as I noted above) does not seem to show *agginam or the like in masc. i-stems. This suggests to me that the influence of in-stems on these forms is only indirect, as indicated above. The acc. sg. -unam is analogical to abl.-gen. -uno. which is taken over from the Skt. neuters.

12.3. Finally, it appears that this acc. sg. ending. analyzed as -una-m, has begotten a loc. sg. -une, as if to a stem -una (cf. the stem -ina, instead of -in, § 10.3).

Stem form in composition

12.4. Twice, at least, we seem to find -us (in the forms -ur, -uh) for -u as stem in composition. This may be a reflex of the frequent use of -u as stem final instead of -us, in composition (and otherwise). Cf. however the use of -as (-ah) for -a in composition, § 8.12. I have not noted any use of -is as stem final of i-stems. dirghabāhurgarvitah LV 310.20 (prose), n. of a son of Mära, 'Proud-of-longarms'; so all mss. and both edd.; hetuhpratyayatah 324.22 (vs), all mss. and Calc.; Lefm. em. hetu-pra^o, but meter demands a long syllable, and pr-, as initial, would hardly satisfy this need. The word is certainly a cpd.

12.5. Once, in Mmk, occurs a nom. sg. masc. of an u-stem in -avah: siddhihelavah Mmk 495.11 (vs) (= °heluh). It obviously belongs with the forms in 'stem' -aya from i-stems found in the same text (§ 10.7).

12.6. ü for u as stem-final in composition: only noted in verses, where required by meter (but this is

probably not significant): bahūvidha- SP 274.2; 334.10; drstāntahetūnayutāna 9.12; 116.6. — bandhūjanena LV 188.13; -mrtyūbhayaih 216.1; madhūpāyasam 271.7. guggulūrasam Suv 104.8 (so read with v.l. for ed. "lu").

12.7. u for ü as stem-final in composition: only noted in verses, m.c. (cf. the preceding): amaravadhunibhām Mv ii.4.1. Note that in the same line i.143.8; 200.4 the mss. have "vadhu"; Senart emends to "vadhu" m.c.

Nominative singular

12.8. The commonest ending is -u, as in Pali. Less common is -ū, which is standard in Pkt. Forms of clearly fem. nouns, and particularly of ü-stems, are not common; they can not be said to demonstrate any clear preference for -ū as against -u (nor the contrary). Most of the examples I have noted are masc. Some of these are masc. cpds. ending in (originally fem.) monosyllabic -ū stems, but these even in Skt. normally become u-stems (type svayambhu). My fem, examples are so few that I shall cite them all together. I shall include dhātu and its cpds.; it has either masc. or fem. gender here, as in Pali (acc. to Childers; PTSD only fem.). And one or two other words which in Skt. may be either masc, or fem, are included. The forms in which meter demands or favors a short or long syllable are marked m.c.; but note that the Skt. ending -us is as good metrically as -u, and so is -us or its equivalent before a consonant.

12.9. Nom. sg. fem. -u: kandu SP 96.16 (vs, m.c.; 'gewöhnlich kandū f.', pw); phalgu 44.14 (vs, metr. in-different). — camu LV 352.21 (to camū; vs, m.c.). (In LV 30.3, prose, Weller 18 would read lokadhātu as a separate word, nom. sg. But I would read with Foucaux's ms. A calurmahādvīpako, see his Notes 94, for the preceding word, otherwise with Lefmann's text.) - isu RP 7.13 (vs, m.c.; masc. or fem.).

12.10. Nom. sg. fem. -v, in samdhi before vowel: (anuştubh vs) jarjarā lokadhātv eyam (= iyam) SP 306.7. Cf. (anuştubh vs) pašyate lokadhātv (acc. sg.) emām (= imām) 355.4.

12.11. Nom. sg. fem. -ū (as with masc., not limited to vss): ākāšadhātū ca SP 330.14 (vs). — camū LV 317.5 (prose; all mss., only Calc. camur; not noted in Weller); (prose; an mis., only care, canar, not noted in wener); 337.16; 360.13, 14 (all vss). — *basist*û (followed by a-) Mv li.444.6 (prose). — *vadh*û MSV il.68.16 (vs). 12.12. Nom. sg. fem. -um (?), perhaps to be read

with Kashgar rec. in SP 31.9 sacaiva sarva iya lokadhatum (end of pāda), pūrņā bhavet ... 'if this whole world system were full ...' Certainly nom. sg.; Nep. mss. lokadhātuşu; KN em. lokadhätu, kept in WT without note. The form is precisely parallel to -im in i-stems, and must be explained similarly; see § 10.23. Cf. picum § 12.30.

12.13. Nom. sg. masc. -u: SP (only in vss., m.c.): vidu SP 25.4; 26.5; pāmsu 158.2; -abhibhu 190.1; -hetu 190.14; śiśu 313.7. - vāyu IV 340.17 (vs; all mss. and Lefm.; Calc. vāyur, which is more in accord with custom in meter; occurs at end of a pada, but in this meter that syllable is generally made long); (others only m.c.) vidu 46.16; meru 53.10; 308.8; 330.3; vindu 74.5; jara vyādhir mrlyu 191.8, read as separate words; -kelu 292.3; svayambhu 6*

(85)

420.16. — vāyu (followed by u-) Mv i.16.15 (prose); guru 27.6 (prose); bhiksu 35.13; 244.12, 15 (all prose), etc.; pānsu 94.16 (= pāmsu; vs but not m.c.); bandhu 156.3 (vs, m.c.); li.229.5 (vs, not m.c.; followed by a-); valgu 1.194.12, 14; 196.8 (all prose); subāhu li.100.14 (prose) etc.; sudhanu 103.15; 110.1, 9, 18 etc. (all prose); krşnabandhu 338.11 (vs, end of line, not m.c.); ikşvāku iii.4.7 (vs, not m.c.); hetu 43.11 (prose); bindu 87.4 (vš, m.c.). — guru RP 22.1 (vs, m.c.); āśukelu 24.14 (vs, m.c.).

12.14. Nom. sg. masc. -0: mostly from historic short u-stems; even forms from -abhibhu, -vibhu probably not to be correlated with the Skt. (original and) alternative -bhd. Except in Mv, only noted in vss m.c.: gurd SP 166.6; lokavidū 166.10; bhikşū 282.3; 284.11. — merū LV 120.9. — sarvābhibhū Mv i.38.13; 39.1, 3, 7 (all prose; followed by vowel; the reading seems well supported tho there are vv.ll., "bhūr, "bhu, and others); same iii.118.8 (vs, m.c.; the following word is recorded in mss. as sarvavidu, but meter demands "vidū, which Senart reads); vibhū i.118.15 (vs, m.c.); sumerū i.301.2 (vs, end of line, not m.c.); višuddhacakşū ii.383.1 (vs). — vidū Gv 208.16 (vs, m.c.).

Vocative singular

12.15. Voc. sg. -u, in Mv (and Bhikşuprakirnaka, cited in Śikş; this text contained prose resembling that of Mv) in prose, otherwise in vss m.c. All masc. bhikşu LV 242.22; naramaru 362.4 and 11. — mahābāhu Mv i.191.7 (vs but not m.c.); svayambhu 247.11 (vş, may be m.c.); bhikşu 330.12; ii.297.9, 10; iii.180.13 (all prose); krşnabandhu ii.238.16; 335.23 (both vss but not m.c.); reņu iii.206.17, 18 (prose). — bhikşu Šikş 154.17, 18 (prose, from Bhikşuprakirnaka).

12.16. Voc. sg. -ū, only recorded in vss, in most texts m.c. but in Mv not always. Mostly masc. but also -oadhû (for which however the Skt. voc. is oadhu). Also (masc.) cpds. of -bhrû. lokavidû SP 162.5. — vikasitabhrû LV 210.7; subhrû 235.1; gajahastorû 235.3; abhtrû 284.18. — vibhû Mv i.112.14 (vs, m.c.); 175.1 (vs, not m.c., end of line); -vadhû i.147.1 = 203.16 = ii.7.12, also i.147.5 (mss. -vadhûr-ili) = 204.3 (v.l. -vadhûh) = ii.7.18 (v.l. -vodhuh; vs, not m.c., end of line); pramādabandhû ii.334.5, repeated 335.3 and 11 (vss; the mss. vary between -û, -u, and -um; Senart reads always -u; but meter seems to require -û).

Accusative singular

12.17. Like the nom. sg., the acc. follows roughly the pattern of *i*-stems. But the only 'irregular' form much used is -u, identical with the same form in the nom. and doubtless transferred thence. It occurs with both masc. and fem. nouns. On the other hand -u, fairly common in the nom., has not been noted in the acc. with masculines and hardly occurs with feminines. As with the nom., I shall list first the few forms that occur with fems.

12.18. Acc. sg. fem. -u: $dh\bar{a}tu$ SP 99.1. — $k\bar{a}madh\bar{a}tu$ LV 45.15; isu 357.14 (masc. or fem.); canu 366.1 (a-stem in Skt.). — $y\bar{a}gu$ Mv i.298.10 (vs, end of line, not m.c.; Senart em. $y\bar{a}gum$); priyaigu ii.210.10 (prose, no v.l.; in Skt. masc. or fem., but in Pali fem.).

12.19. Acc. sg. fem. -0: candrārkatārāgaņa bhū pateta LV 253.18 (vs) 'moon, sun, and host of stars might fall to earth'; m.c.? (the regular Skt. bhuvaņ would spoil the meter). — -karşū Mv ii.409.7 (vs; v.l. °u; long syllable required metr.; Senart em. °ām; Skt. ā-stem); vadhū ii.444.7 (prose; v.l. vadhu; Senart em. vadhūm).

12.20. Acc. sg. fem. -um, of stems regularly ending in -a in Skt.: camum Mv ii.260.10 (prose); śvaśrum ii.445.19; 449.3 (both prose).

12.21. Acc. sg. fem. -uyam (? textually not certain,

see below), in a polysyllabic \bar{u} -stem; cf. the oblique fem. endings -uya, -uye etc., and the ending -iyam in (fem.) i and \bar{i} stems (§ 10.63), by which this (like the oblique -uya, -uye) may well have been influenced. *švašruyam* Mv iii.10.13 (prose; to *švašrū*). So Senart with one ms.; the other reads *švašruram*, which might also stand, as a form based on the analogy of nouns of relationship (mātaram etc.). The one certain thing, in my opinion, is that the text should not be emended to the Skt. form.

12.22. Acc. sg. masc. -u, well attested, and in My prose; otherwise only noted in verses m.c. bhikşu SP 98.10; 280.12; pāmsu 235.12; -heiu 313.6; merum sumeru sarvā (mss. sarvās) ca 355.5 (most mss. sumerum, which is inferior tho possible metrically, in the first pāda of an anuştubh). — meru LV 134.5; 202.19 (in the latter, most mss. merur-ulpálya, perhaps to be interpreted as containing 'hiatus-bridging' r); 297.3; kleśaripu 156.6; pāmšu 190.15. — bāhu My i.18.3 (prose); kişu 21.6 (prose); hetu iii.139.17 (vs, m.c.). — laru RP 26.20.

12.23. Acc. sg. masc. -ūm, from a (masc.) cpd. of -bhū: sarvābhibhūm Mv i.38.7, 9 (prose).

12.24. Acc. sg. masc. -unam, as in Pali (Geiger 83.1 bhikkunam, -bandhunam); see § 12.2: dharmaketunam (stem -ketu) Mmk 365.17 (vs).

12.25. Acc. sg. masc. -una, m.c. for preceding: bhiksuna viksya ca gunādhyam RP 29.9 (vs).

12.26. Acc. sg. masc. -uvam, the encing of monosyllable ü-stems, occurs once in a normal u-stem: mahāviryam prabhavisnuvam Mmk 135.2 (vs) 'the heroic Lord'. The meter is bad; regular prabhavisnum would give eight syllables, but wrong cadence (in the even pāda of an anuşţubh).

12.27. Acc. sg. masc. -us; cf. -is in i-stems (\S 10.60); either the nom. sg. used as acc., or transfer from the us-declension. senapatim namuci simhahanus ca namna (...pariprechati) LV 303.1 (vs), 'and Namuci (Māra) asked his general named Simhahanu'. Only Calc. has "hanum, tho all mss. agree on senapatim! Cf. also the v.l. merur for text meru in LV 202.19, \S 12.22, which however (if adopted in the text) could intend meru plus 'hiatusbridging' r.

Nominative-accusative singular neuter

12.28. Nom.-acc. sg. nt. -ū, occasionally for -u in verses m.c. -bhaisajya bahū analpakam SP 13.12; bahū puņya bhaveta tesām Sukh 46.7.

12.29. Nom.-acc. sg. nt. -um, -um, much more commonly used than the preceding when, in verses, a long final is required; also in prose and in metrically indifferent positions. Like *-im*, *-im* with *i*-stems, common with neuters in Pali and Pkt. Occurs also with original us-stems, particularly $\bar{a}yu = \bar{a}yus$ (cf. § 16.42). Thus, $\bar{a}yum$, nom.-acc. sg., SP 330.10 (vs; so read with v.l. of KN and with WT; KN ed. $\bar{a}yu$, unmetrical); 336.1 (vs, but here meter is indifferent; in 336.6, vs, ed. $\bar{a}yur$ acinityam but several mss. $\bar{a}yum a^{\circ}$); 336.12 (vs, so ed., but here mss. mostly $\bar{a}yu$; meter indifferent); LV 110.3 (vs); Mv i.106.3; 117.2; 119.5 (all vss, but meter requires long final only in 117.2). *— dharmacakşum* LV 179.4 (vs). *cakşum* Mv i.167.15 (vs); ii.374.16 (vs); iii.177.8, 10.

12.30. In all the preceding, of course, the regular Skt. form in -us (-ur, etc.) would be prosodically equivalent to -um. The following concern u-stems; all verses except as specified: bahum SP 111.7; 229.8, 13; laghum 192.4 (at end of line; Kashgar rec. laghu). — laghum LV 46.4, 16; 91.21; 314.13 (in these at end of lines); 109.16; 322.3; 358.19; bahum 203.7 (end of line); madhum 387.4; rohi-tavastum 406.19 (prose; acc. sg.; cpds. of -vastu are regularly nt.; followed by a-). — bahum Mv i.13.15; 14.12; 37.7; 86.7; 101.13 (all these either prose or at end of

lines or otherwise in metrically indifferent positions); abhiyapastum (nom.) 45.16 (prose); kapilapastum (nom.) 47.15 (prose); laghum 75.8, 12 (end of lines); vastum352.3; iii.413.16 (both prose); madhum (before a-) i.341.7 (prose); asādhum etam ii.272.9 (prose) 'this is bad'; lahum(v.l. laghum) iii.2.17 (end of line); kešasimasirum (before a-) 92.5 (prose); kālavastum (nom.) 360.1. — agarum(nom.) šikş 65.14 (prose). — tat pieum samšadhayitavyam (nom.; pieu in Skt. masc.) Mmk 55.21 (prose). — alarum(before a-; nom.) Gv 133.25 (prose). — The best ms. at Ud vii.5 (end of line) reads (kušalam) bahūm, which if genuine would be isolated, as far as I know (§ 12.23 is different); it is probably an error for bahum.

Oblique singular: distinctively masculine-neuter forms

12.31. Essentially the same conditions hold here as in the oblique singular of the *i*-stems, q.v. (§ 10.64). We present first forms which are primarily masculine or neuter, and appropriate to particular cases; then those which are primarily feminine, and for the most part applicable to any oblique singular case.

12.32. Inst. sg. -ūnā. The ending -ūnā, instead of -unā, from the stem sarvābhibhū, seems well attested (despite some variation in the mss.) in sarvābhibhūnā Mv. 40.6, 12; 41.1 (prose). (Cf. the gen. -ūsya below). Otherwise it has not been noted except in Lefmann's em. isūnā LV 168.7, where \bar{u} seems demanded by the meter, tho all mss. and Calc. read isūnā (or °nā).

12.33. Abl. sg. -uno, the ending of neuters in Skt., is found (as in the gen., see next) with a masc. noun, hetuno Mv i.120.11 (prose) 'by reason' (all mss.; Senart implausibly em. hetus'o).

12.34. Gen. sg. -unah, -uno, as in the preceding, and as in Pali and Pkt. -uno. Chiefly in verses, but once at least in Mv prose. See above, \$12.2: -bandhunah, °no, Mv i.185.18; ii.319.13; svayambhuno i.357.8; -vidhuno i.220.12 = ii.22.13; jugulsuno iii.438.2 (prose); bhiksuno 452.14. — mrtyunah Divy 68.20 = 300.22 = Av i.5.7 etc. (vs). — jantunah Ud iii.5; bhiksuno vi.13.

12.35. Gen. sg. -usya, parallel to -isya from i-stems, §§ 10.78-80. The Sanskritizing orthography -usya (like -isya for -isya) occurs a number of times in Mv (where Senart keeps it, contrary to his practice with isya, which he emends to -isya) and in SP (sometimes with v.l. -usya); some miss. are reported to read so also in Bhad 42, 43; even so it is relatively rare. The ending -usua is common in prose of Mv, and occurs once in that of Gv; it is also cited in Siks from prose of the Bhiksuprakirnaka, which contained the same sort of prose as Mv. The SP and Bhad examples are from verses. I have failed to note any from other texts. lokaviduşya SP 31.7; agurusya 50.7; devadārusya 50.8; lokaviduşya 118.9; agarusya 331.3; māmsacakşuşya 355.10 (v.l. °sya); agarusya 362.5 (1 ms. °sya). -- -caksusya Mv i.159.2 (prose); -vastusya 193.13 (prose); bitksusya i.36.11; 329.21; 330.10; 335.3; ii.273.12 (all prose) etc.; simhahanusya i.352.13; 355.19 (both prose); ii.37.17 (vs), but simhahanuşya ii.76.4 (prose); dhanuşya ii.76.10 (prose); brahmāyusya 77.12; 78.16 (prose), but brahmāyusya 78.17 (prose); subāhusya 95.5; 98.5; 101.4 etc. (all prose); sudhanusya 99.15, 17; 110.13; 112.16 etc. (all prose); ketusya ii.354.21 = iii.279.5 (vs); mahāvastusya ii.397.7 (prose); subandhusya 420.9; 421.1, 11 (all prose); ikşvākusya 425.14 (prose); agurusya iii.96.5 (vs). — vidusya (v.l. °şya) Bhad 42, 43 (vss). - bhikşusya Sikş 154.20; 155.1 (prose, from Bhiksuprakirnaka). - bimbahusya Gv 106.16 (prose).

12.36. Gen. sg. -ūsya appears well attested in Mv from the *ū*-stem sarvābhibhū; cf. the inst. -*ūnā* above. In Mv i.36.6 Senart reads sarvābhibhusya, but 2 mss. °bhūsya; the mss. are more favorable to °bhūsya in i.37.8, where Senart still has *°bhusya*; but in 37.11, 38.5, 44.8, 45.2 Senart adopts *°bhūsya* with preponderant ms. evidence.

12.37. Gen. sg. -o, MIndic for -os, -oh (or perhaps corruption?). bhikso Mv i.39.8 (prose), followed by idam; kept by Senart; valaheto ii.222.12, followed by d-, mss., Senart em. °hetor.

12.38. Loc. sg. -usmin or -usmin, like -ismin, -ismin from i-stems (§ 10.84; and cf. Pali -usmin), is well attested in Mv, prose as well as verse; otherwise it has been noted only once in a verse of SP. The nouns are only masc. and nt. with one exception; camusmin Mv ii.173.6 (prose), apparently from the stem which in Skt. is camâ, and which seems to be fem. even in Mv (etân camun ii.173.12); see Dict. pânsusmin SP 236.6.—Mv (all prose except the first): -vidusmin i.235.16 (vs); kapilavaslusmin 230.7; 352.10: ii.3.17; 31.6; 48.4; 74.4, etc.; bhikşusmin i.323.10; iii.03.7, etc.: dhanu-tsarusmin ii.73.8; tharusmin 74.3; dhanusmin 73.16; 74.2; 434.12; iii.184.7; vastusmin iii.91.17; lsarusmin 184.8 (so Senart em., mss. sarusmin).

12.39. Loc. sg. -une, as if to a stem una; see § 12.3. I have recorded only $(sri-)mah\overline{a}vastune$, Mv ii.206.19 (prose, no v.l.); also one ms. in ii.237.16 (omitted by Senart with v.l.), and mss. in i.1.2, where Senart emends to °vastuye with the mss. at i.2.14. It would have been better to emend the latter to °vastune; I know of no parallel to such a form as -vastuye (quasi-feminine) from this stem. Aśokan has loc. -une: Hultzsch cxvii (otherwise Bloch, Aśoka, 62, 167 n. 10).

Oblique singular feminine

12.40. As in MIndic generally, fem. u-stems have in all the oblique sing. cases endings modelled on those of the *i*- and *i*-stems. The regular ones are Pali -uyã, Pkt. $-\hat{u}a$, $-\hat{u}e$; in addition the loc. often has -am for the final vowel. The number of instances I have noted in BHS is, naturally, much smaller than in the case of the far commoner *i*- and *i*-stems, but runs a similar gamut; see § 10.87. We shall present first forms in $-\hat{u}ye$, -uye; then $-\hat{u}ya$, -uya; then $-\hat{u}yo$, and finally $-\hat{u}yam$, primarily belonging to the loc. alone.

12.41. Obl. sg. - $\mathbf{\hat{u}ye}$. This and -uye (see next) have been noted only in Mv. Both are found in prose (all examples of - $\hat{u}ye$ are prose); the mss. sometimes var⁴ between \hat{u} and u, and there is no evidence to discredit either (just as -iye and -iye are both well established with *i*, *i*-stems). Inst.: tejodhätäye Mv i.357.17.—Gen.: yvägäyeMv. ii.84.10.—Loc.: lokadhätäye Mv ii.301.12; and in line 9 above one ms. reads so, Senart with v.l. °tuye; id. Mv ii.314.10 (v.l. °tuye).

12.42. Obl. sg. -uye. Besides those noted under -*ūye*, we find the following, all but one from prose (all Mv): Instr.: *svasruye* ii.454.17; 455.12.—Gen.: *svasruye* iii.9.15; *durbalacaksuye* (fem.) iii.132.15.—Loc.: *lokadhātuye* i.166.10 (vs; u could be m.c.).—On mahāvastuye, mss. and ed. at Mv i.2.14 (prose), apparently loc. of mahāvastu, cf. § 12.39.

12.43. Obl. sg. -ūya, unlike - $\ddot{u}ye$ (-uye), is found in various texts besides the Mv; but, except for Mv, only in verses. It may be noted that in the *i*, *i*-stems the corresponding - $\ddot{u}ye$ etc. is the prevailing form in Mv, while - $\ddot{u}ya$ etc. seem preferred in some other texts. Inst.: dadr $\ddot{u}ya$, kand $\ddot{u}ya$ SP 112.6 (stems dadr \ddot{u} , kand \ddot{u}).—Abl.: lokadhāt $\ddot{u}ya$ SP 126.1.—Gen.: jamb $\ddot{u}ya$ LV 134.2 (separate mārdini); lokadhāt $\ddot{u}ya$ Sukh 46.5.—Loc.: lokadhāt $\ddot{u}ya$ Mv ii.353.4 = iii.277.7 (vs); and read so in Mv ii.281.16 (prose) with one ms. (v.l. °dhātau-ya) for Senart's em. °dhātau ye.

12.44. Obl. sg. -uya: Inst.: śvaśruya Mv ii.445.17; 454.5; iii.9.3 (all prose). **12.45.** Obl. sg. - 0yo (cf. -*iyo*, § 10.113). Gen.: lokadhâtûyo Mv ii.350.3 (vs, û metrically necessary).

12.46. Obl. sg. -üyam, -üyam, primarily loc., but like -Iyam (§ 10.141) also used in gen. sense. Loc.: lokadhâtûyam (v.l. °yam) Mv i.214.12 (prose), before vowel. Note the shortened a before consonantal m, even in prose. -Gen.; śvaśrûyam Mv ii.444.6; 445.5 (both prose), so mss. both times (one °uyam the second time); Senart emends the first time to °uye, the second to °ûye.

Nominative-accusative plural

12.47. While the number and variety of forms here recorded is naturally much smaller than in the corresponding section of *i*, *i*-stems, the same considerations (\S 10.148) impel me to group together the endings of these two cases in all genders.

12.48. Nom.-acc. pl. -avah (-avo), the Skt. nom. pl. masc. and fem. ending, is also used as acc. pl. (examples noted only in verses, and all masc.): *flavo* SP 91.3; (*buddhān*) *bahavo* 207.10. — *bhikṣavah* LV 421.7 (with *krtvā*). *suadān bahavo* RP 15.4. — *viijanti sarva ripavo* GV 213.9.

12.49. Nom.-acc. pl. -ava, only in verses m.c. for the preceding; examples (nom. pl. masc.): bahava LV 421.22; bhiksava RP 17.14.

12.50. Nom. pl. -uvah, as in monosyllabic stems: madgupo, n. pl. f., MSV ii.86.2 (prose).

12.51. Nom.-acc. pl. -uno, analogical to -inas, -ino in i-stems, and to -uno abl.-gen. sg. (cf. § 12.2). So rarely in Pali, Geiger 83.5, but quite commonly in Pkt., Pischel 380. In BHS rare. Noms.: bhiksuno Mv i.192.10 (vs; so we must read with Senart's em. for bhiksun na, unmetrical reading of mss.); svayambhuno 338.5 (vs).

12.52. Nom.-acc. pl. -ūna, m.c. for -unah (preceding). Cf. -īna, \S 10.159. iathaiva bhikṣūṇa ca bhikṣuṇi ca upāsikās cāpi mi (read with WT cāp' imi = cāpi ime) mahya sammukham SP 384.12 (vs). The parallel nouns prove that bhikṣūṇa is nom. pl. (not partitive gen. with paācašatā in preceding line).

12.53. Nom.-acc. pl. -ūni, with m. and f. nouns: bahūni (bālāni, nom. m.) Mv ii.92.7; (māyāni, acc. f.) 174.12; tāni işūņi (acc.) iii.362.9. Cf. § 6.9.

12.54. Nom.-acc. pl. -uni, is reportedly written rather frequently in the mss. of Mv for -ûni, neut. nom.acc. pl.; Senart always emends (rightly?). Thus mrduni Mv li.470.13 (prose) mss.; Senart mrdüni.

12.55. (Nom.-)acc. pl. -um, -um corresponds to -im in i-stems, § 10.166. Usually emended by Senart, and doubtless by other editors. In Mv i.328.9 (prose) Senart keeps tām bhikṣum (= tān bhikṣūn) with mss.; but in i.334.10 and 16 he emends to bhikṣūn, and in i.335.1 (prose, before a vowel) he emends bhikṣūm to bhikṣūm (with anusvāra for -m). Again in i.337.1 mss. bahum or bahu, Senart em. bahūm; and in ii.186.2 Senart em. venū for mss. benum or venu. All acc. -

12.56. Nom.(-acc.) pl. - $\mathbf{u}s$, - $\mathbf{u}h$, the Skt. ending of the acc. pl. of fem. u- and \bar{u} -stems (also of the nom. sg. of \bar{u} -stems), is used as nom. pl. of masc. as well as fem. u-stems. Cf. the similar use of - $\bar{i}s$ with *i*-stems, **f** 10.167, and of - $\bar{u}s$ as voc. pl., § 12.62.

Masculine: spayambhūh (v.l. °bhū) SP 47.10 (vs); pašūš caiva SP 111.8 (vs); bhiksūh My i.328.3, 6 (prose). Feminine: dhātūš ca RP 44.15.

12.57. Nom. pl. -uh. Cf. the similar use of -ih, § 10.190, and see under -ū, § 12.60, for which -uh often occurs as v.l. in mss.: sancinvanti ... viduh (all mss.; Senart em. vidūh) Mv i.87.8 (end of line of vs); ye latra nirmita bhiksuh na caite bhiksuno matā (v.l. matah) 192.10 (vs), mss.; Senart em. nirmitā bhiksūh. Certainly nom. pl. is meant. — apramāņāh satlvāšayadhātur anupraveştavyāh Gv 458.8 (prose). In next line -vimuktir (§ 10.190) is similarly used as nom. pl.; the two forms confirm each other.

12.58. Nom.-acc. pl. -uyo and -uyo. Seems to occur only with fems.; parallel to (and imitations of) -Iyo, -iyo in *i*-stems (which however also occur with masc.), see §§ 10.168 ff. Not many examples noted, but enough to demonstrate the reality of both endings. Mv 1.80.11 and 13 has in close proximity lokadhâtuyo and 2'uyo, acc., in prose; no v.l. for the first; only two inferior mss. 2'uyo for the second. The form lokadhātuyo as acc. again in Mv ii.315.9 (prose), lokadhātuyo as nom. ii.350.3 = iii.274.5 (vs), both without v.l.; in verse, metrical conditions might be concerned. — dhenuyo (acc., no v.l.) Mv iii.42.3 (prose).

12.59. Nom.-acc. pl. - \mathbf{u} and - \mathbf{u} ; the really common non-Skt. endings are these, parallel to -i and -i in the *i*-stems, qq.v. §§ 10.177 ff., 189. Perhaps - $\hat{\mathbf{u}}$ is somewhat commoner than -u, tho this is not so clear as with -i in comparison with -i. In any case both seem well established in the prose of Mv and either in prose, or at least in metrically indifferent positions, of various other texts. No preference between the two seems to be shown by any particular gender or declensional type. Prose passages will be marked as such.

12.60. Masc. nom. -ū: bhiksū SP 11.13: 26.9: 272.10: 273.8; bahū 26.9; 86.11; avidvasū 45.9; 204.6 (both end of pādas, not m.c.); lokavidū 47.10; 193.1; marū 208.9; riū 324.3; vidū 325.4 (end of pāda, not m.c.).-The form bhiksū, nom., is extremely common in Mv, tho not so common in the mss. as in the edition of Senart, who usually substitutes it for mss. readings bhiksu (see below), bhiksuh, bhiksur etc. (see § 12,57). The mss. themselves often vary between these forms. A few instances, in which there is at least some ms. support for bhiksū (but often vv.ll. as above), are (all prose) Mv i.267.4; 317.18; 328.15, 16 (Senart em. °ūnām); ii.48.14; 82.4; 83.13; 89.12, 14; 166.15, 17; 177.4, 7 (in 4 one ms. bhiksū-m, with 'Hiatusbridging' m before vowel); 209.4, 8; 234.10, 13; 241.10, 11. -Other forms: bahū Mv i.10.3 (vs); khānū ii.350.18 (vs, but metr. indifferent); bandhū iii.68.16 (prose); vibhū 101.2 (vs). - bahū Samādh 22.18 (vs. but metr. indifferent). - bhikşū RP 30.10. — spayambhū Gv 258.15.

Masc. acc. $-\hat{u}$: valg \hat{u} ... sabdān SP 358.8. klesasatrā LV 195.14. — bahā Mv i.42.11; bhiksā i.323.8; iii.102.12 (both prose; in both v.l. bhiksu-m, before vowel, with 'Hiatus-bridging' m, or as in § 12.55).

Fem. nom. -ū: nāgavadhū LV 284.9; lokadhātū Sukh 45.12.

Nt. nom.-acc. $-\hat{u}$: aśrū Mv ii.217.16 (prose; v.l. aśu, intending aśru?); bahū Samādh 19.21.

12.61. Masc. nom. -u: vidu SP 56.7; paramānu 158.6; samantacaksu 176.10; bahu 255.7; 341.14 (both vss but meter indifferent); mrdu 324.3; 325.9. - maru LV 81.7; naramaru 165.5; dirghanakhaśmaśru 211.12 (? cf. Weller 28; prose; this should be read as a separate word, and is either nom. pl. masc., or acc. sg. nt. adverb); vimalaketu 296.17: vidu-m-āhuh 324.1 ('Hiatus-bridging' m; Tib. mkhas pas smras, 'the sages say'); prthu 330.14.-Mv: the form bhiksu, nom. pl., is very common in the mss.; Senart usually, but not always, emends to bhiksū (cf. above). There follows a small selection of examples, all prose, mostly emended by Senart; none of the mss. read bhiksū in these (one or two have v.l. -ur); i.246.8; 319.1, 6; 328.11; 329.19; ii.67.15, 17; 69.6; 77.1; 83.15; 94.15, 17.-Others: bahu i.15.1; veņu 235.14; maru 241.18 (also iii.96.17; 100.18, v.l. maruļ); ašruvindu ii.213.17 (prose); mrdu 263.10 (prose). In several of these Senart em. °u. - bahu RP 31.16. - vidu Dbh.g. 7(343).10, ripu 20.

Masc. acc. -u: bahu SP 54.1; 62.10; bhikşu 383.12. — (svakān api) bāhu (one ms. bāhū) prasārilān na pašyanti LV 51.14 (prose; both edd. print bāhuprasāritan as a cpd., which is impossible; cf. Weller 19, whose solution is not the right one); te madhularpanam iksu likhilakāmš cādāya 382.10 (prose; Lefmann again prints iksulikh[°] as a cpd.; not noted in Weller; immediately preceded by likhilakāš cekşavah, nom. pl.; there are variants but the reading is well supported). — bhiksu KP 141.2, 5, 6 (all prose).

Fem. nom. -u: maruvadhu I.V 81.3; -dhātu Mv i.126.12; ii.352.12 (both vss, but metr. iudifferent; in the second Senart em. °tû).

Fem. acc. -u: sarvadhātu nityā Mv i.90.1 (prose) mss., Senart em. ^odhātū anityā (-ū is uncalled-for).

Nt. nom.-acc. -u: jānu LV 50.1 (could be called dual); kariyā bahu LV 169.4; cakşu nipatanti Mv ii.72.18 (prose).

Vocative plural

12.62. Voc. pl. - ūh (?). Cl. same ending in nom., § 12.56: bhiksūh (v.l. ^ouh) Mv ii.357.1 (prose).

12.63. Voc. pl. -o, apparently the ending of the voc. sg. used as voc. pl., patterned on the extensive identity of nom. and acc. forms in sg. and pl. Noted only in *bhikşo* SP 146.4 (vs) 'O monksi' So Kern translates without comment; Burnouf apparently assumes a gen. sg. KN suggest em. to *bhikşân* or *bhikşavo* (which would be possible metrically). WT keep without note.

Instrumental plural

12.64. The endings $-\hat{u}hi$ (rarely $-\hat{u}bhi$) and -uhi (rarely -ubhi) are quite parallel to the corresponding *i*-stem forms (§§ 10.193 ff.). As with the *i*-stems, we find such forms with short -u- only in Mv, in spite of regular Skt. -ubhis. In Mv prose; otherwise only in verses. 12.65. Inst. pl. $-\hat{u}hi$: masc. and nt. *u*-stems: hel $\hat{u}hi$

12.65. Inst. pl. - $\hat{u}hi$: masc. and nt. *u*-stems: *hetühi* SP 45.6; 63.7; *bahühi* 111.12. — *parašāhi* Mv 1.5.4 and 8 (both prose); iii.455.4 (vs); *bahūhi* ii.97.6; 98.10; 109.17 (all prose); *bhikšūhi* ii.166.15; 209.4; iii.431.4 (all prose; v.I. in first two [°]ubhi, in third [°]uhi); *cakšūhi* ii.445.8 (prose; in same context 444.14 mss. [°]ūhi and [°]uhi; 444.20 [°]uhi only).

Fem. u and ū-stems: vadhūhi Mv i.203.12 (vs; repeated ii.7.8 with vadhūbhi, mss.; ū-stem); jambūhi ii.250.3 (vs, metr. indifferent; v.l. ^ouhi; u or u-stem); dhātūhi (u-stem) iii.144.6 (prose).

12.66. Inst. pl. -ūbhi: hetābhi LV 176.5; in LV 364.20 marūbhi (or °hi) seems required by meter; text marubhi without v.l.; vadhūbhi Mv li.7.8 (vs), see preceding §; gurūbhi Šiks 112.13.

12.67. Inst. pl. -uhi, only in Mv and mostly prose (vss marked): samgrahavastuhi i.3.12; svayambhuhi 10.11 (vs; meter requires u, but 5 of 6 mss. °ūhi; repetition iii.455.5 °uhi without v.l.); tarakşuhi i.25.1; jānuhi 360.13 (locative in sense); bhikşuhi ii.48.10; aśruhi 429.13 (prose); iii.3.3 (vs, metr. indifferent; Senart em. °ūhi in both); cakşuhi ii.444.14, 20 (v.l. °ūhi in 14); dhenuhi iii.42.3; maruhi 267.16 (vs).

12.68. Inst. pl. -ubhi: marubhi Mv i.100.9 (vs). 12.69. Inst. pl. -ubhīr, before vowei, for -ubhir, m.c.: bahubhir acintyair SP 385.3 (vs).

Genitive plural

12.70. Gen. pl. -ūna, only m.c. for -ūnām: lokadhātūna SP 166.7; marūņa LV 337.10; bhiksūņa santike (v.l. bhiksuņam antike) Mv ii.188.10 (end of anustubh line); marūņa ib. 322.1.

12.71. Gen. pl. -unäm, -unäm, not only m.c. but (like -inām from i-stems, § 10.203) also in prose of Mv: bhiksunām SP 56.10. — marunām LV 129.19. — bhiksunām (before vowel) Mv i.329.16 (prose); bhiksunām ii.258.15 (prose); iii.63.15 (prose); bandhunām ii.232.11 (vs); once bhiksunām in prose of RP, 56.13 (misprint?).

12.72. Gen. pl. -unam, -ūnam, with short a of ultima m.c., followed by vowel. Only in verses, in final of anustubh lines: bhiksunam antike Mv i.282.5; ii.93,19; iii.137.11 (and v.l. of ii.188.10, § 12.70). The quantity of the u makes no difference metrically; in the first passage Senart em. bhiksūnam. — bhiksūnam antike Mv ii.237.6; iii.89.22.

Locative plural

12.73. Loc. pl. -ūşu, only m.c. for -uşu: bahūşu SP 10.9; dhātāşu 51.9; 324.1; bhikšāşu 94.2; lokadhātāşu 158.3. — lokadhātāşu Samādh 22.40.—In Mv i.70.7 Senart properly em. marūşu; mss. maruşu, unmetrical. 13.1. Our language reflects MIndic conditions here as elsewhere. In nouns of the Skt. r-declension, MIndic phonetic and morphological processes lead to sweeping alterations in the system. In various ways such nouns, in MIndic and in our dialect, often are converted into a, i, and (especially) *u*-stems. It seems best to present each group of such cases as a unit, and we shall begin with them. Afterwards we shall present other 'irregular' forms, case by case; they include both phonetic and morphological developments, some of which are very interesting, and not all of which are paralleled in known MIndic vernaculars.

13.2. Since this method of presentation results in separation of different forms of these nouns in the same case, there is here appended a conspectus or key by which those interested may discover the different ways, not recognized in normal Skt., in which each case may be formed.

Stem in composition, §§ 13.4, 8, 19, 20, 22 nom. sg., 13.4, 5, 6, 15, 19, 23, 30 voc. sg., 13.9, 31 acc. sg., 13.10, 18, 24, 32 nom-acc. sg. nt., 13.17, 32 general oblique sg., 13.11, 15, 18, 33 inst. sg., 13.15, 17, 25, 38 abl. sg., 13.26 gen. sg., 13.4, 15, 27, 39, 40 gen. dual, 13.15 nom. pl., 13.12, 13, 18, 41 acc. pl., 13.12, 13, 15, 17, 18, 19, 42, 43 inst. pl., 13.17, 44 gen. pl., 13.18, 28, 45 loc. pl., 13.46

Stems in -a for -r

13.3. Sporadically, as in Pali (Geiger 90.4, 91.3) and Prakrit (Pischel 391), we find *a*-stem forms replacing *r*-stems, thru simple replacement of stem-final *r* by *a* (as prior member of compounds), or of final -*ar* (as in the vocative) by -*a*, both Mindic phonetic processes.

13.4. As stem in the prior member of compounds, we thus find rapta = naptr in Mv i.348.9 (prose) naptapranapti-kāye 'in the crowd of his near and remote descendants'. So read with mss.; Senart misunderstands and emends erroneously; on pranapti (= pranaptr) see § 13.20. Similarly Dbh.g. 7(343).1 śāsta-caraṇaṃ = śāstr^o. The name Māndhātr, which occurs in that form in BHS as well as in Skt. (see Dict. s.v.), also appears as Māndhāta: nom. sg. Māndhātah, Mvy 3558; °to Mv i.348.9; Divy 210.20; 214.20 (all prose); note that r-forms occur in the same Divy context (°tā nom. sg. 210.23; 214.21; cf. °tuh gen. sg. 278.13; 576.10); gen. sg. māndhātasya Mv i.348.9; Divy 210.21; 213.23 (all prose); Mmk 609.1 (vs; cf. Māndhātā nom. sg. 608.22, vs).

13.5. Perhaps here belongs mahāyajňayaşļah LV 112.11 (prose) 'offerer of great-sacrifices', if yaşļah is nom. sg. of yaşļa = yaşļr. (There is in our dialect a yaşļa = işļa 'sacrificed'; but I do not see how it could be concerned here. The variant mahāyaşļayajňah of most mss. seems questionable, but perhaps could mean 'a great one-that-has-offeredsacrifices'.) 13.6. In verses, nom. sg. forms in -a may stand for -a m.c., rather than for -ah, -as etc. A few examples: kartā na karta iti LV 337.2 (vs; note preceding kartā from stem kartr); in Mv i.69.6 (vs) sāstā is written in mss. but last syllable should be short; Senart em. sāstu; perhaps rather šāsta?—In Sukh 23.2 (vs) šāsta is read in text with 1 ms.; the others sāstu, cf. § 13.23.

Stems in -ā for r

13.7. Rather different is the passage of r-stems into the \bar{a} -declension, or at least use of \bar{a} -declension endings with them. This phenomenon (also familiar in MIndic, Geiger 91.3, Pischel 389 II.) is nearly (yet not wholly! see below) limited to feminine r-stems, and unquestionably started from the nom. sg., where such a fem. form as Skt. mālā was naturally reinterpreted aş an \bar{a} -sterr. (Note the converse influence of r-stems on the word bhāryā 'wife', § 9.25.) Consequently we find:

13.8. Stem -ā in composition: mātāpitaram (acc. sg.), mātāpitmām (gen. pl.) Mv i.348.16, 17 (prose) '(motherand-father =) family, telatives, home-folks' (and adj. mātāpitrka, see Dict.). And even with a masculine noun: sāstābhūtam Gv 495.15 (prose), 'acting like a teacher' (sāstr; said of bodhicittam). Cl. Wackernagel, II.1 § 19 f. (p. 47).

13.9. Voc. sg. -e: kuladuhile, Kashgar rec. and Lüders ap. Hoernle MR 147, for SP 264.9 (prose) kulaputri (v.l. bhagini) of both edd. with Nep. mss.; äryadhile (no v.l.) Mv ii.172.16 (prose). Aud even with a masculine, dharmabhrāte Mv iii.350.18 (prose; in lines 19, 20 dharmabhrātā as voc.). Pali also has such forms as sallhe (= śāstar), Geiger 90.5.

13.10. Acc. sg. -ām (-ā): mātām duhitām LV 29.4; (bodhisattva-) mātām Mv i.212.4, 13, 14; 213.4; iii.132.12 (all prose); dhītām Mv i.356.12; ii.65.6; 73.5 (all prose) etc. And (as in acc. sg. of ā-stems) mātā LV 50.4.

13.11. Oblique $\overline{g}_{a.:}$ mātāye Mv ii.469.20; iii.176.16 (both prose, inst.); id. (gen.) Mv iii.131.14; 358.3 (both prose); ekamātāyam Mv ii.450.18; 453.6 (loc.; both prose; in 453.6 v.l. °tāya, which Senart reads both times); -dhītāye (gen., loc.; note -duhituh 263.17) Mv ii.66.1; 88.15; 263.16 (all prose).

13.12. Nom.-acc. pl. $-\bar{a}(s)$; i. e. $-\bar{a}$, not always interpretable as samdhi-form for $-\bar{a}s$; also -a, m.c.: grāmikaduhitā ābhih LV 265.6 (prose); mātā svasā pitara ... bhrātā LV 341.19 (vs; all nom. plurals); -dhīta (nom. pl., m.c. for $-\bar{a}$) LV 170.13 (vs); pitaram me kausalyam prechesi mātu-svasā pi Mv ii.189.13 (prose; so mss., acc. pl. dvandva).

13.13. And with masculine stems (besides bhrátá, just above): anubhoktá (before n-; nom. pl.) LV 324.18 (vs); matapitá (before m-; acc. pl. [dual?]) RP 55.7 (vs).

Stems in -ra (-rā, -rī) for -r

13.14. More often, r-stems are brought into the a-declension by extension. That is, the 'stem' adds a (or fem. \bar{a} , \bar{i}) to r, ar, or \bar{ar} ; for it may be thus added to all these forms of the Sanskrit stem.

13.15. Of these, the rarest are extensions in -a, or rather in fem. $-\overline{a}$, $-\overline{i}$ (in the only cases noted; see § 19.8 for *lisra*-), added to the zero-grade stem, resulting in stem.

finals in (fem.) $-r\tilde{a}$, $-r\tilde{i}$. The former (possibly blend of stems in -tr- and MIndic $-t\tilde{a}$ -) seems to exist in stems $m\tilde{a}tr\tilde{a} = m\tilde{a}tr$, $dh\bar{t}tr\tilde{a} = (dh\bar{t}t\tilde{a}, dh\bar{t}tar =) duhitr$; the latter has been recorded only in stem $duhitr\bar{i} = duhitr$. Also with a, m.c. for \tilde{a} .

neha mātra na pitā nu bāndhavā RP 38.9 (vs), 'there is here no mother or father nor kinsmen'; mātra m.c. for mātrā = mātā, p.vn. sg. (note that *māta would do just as well metrically!).

kula-(ddhi, below line)-trāya, Mahāparinirvāņa Sūtra in Thomas ap. Hoernle MR 94.2, read (kula)-dhitrāya, general oblique (inst.) sg. of stem dhilrā.

The stem duhitri for duhitr is recorded several times (Dict.), in the inst. and gen. sg., gen. dual, and acc. pl.

Stems in -ara (-arâ, -arî) for -r

13.16. Commoner, with nouns of relationship (which in Skt. have $-ar_*$, not $-\bar{a}r_*$, as stem final in the strong cases), are forms implying a stem -ara. The starting point doubtless was the acc.'sg. -aram, which in fem. nouns may appear in our dialect as $-ar\bar{a}m$; these two forms then lead to other forms implying stems -a, $-\bar{a}$. (On fem. nouns with masc.-nt. endings see § 13.17.) Prakrit has forms of this sort (Pischel 391 f.), but they seem not to be recorded in Pali (cf. Geiger 91.1). Cf. the general oblique forms in -are, -ari(§§ 13.33 ff.), which seem to be different in character. All our examples are from prose except those specified as vs.

13.17. Masculine (and neuter) forms, stem -ara: Inst. sg.: *bhrālareņa* Mv ii.82.11; *pilareņa* Mv ii.111.2; 113.11; 437.8; iii.405.13.

Acc. pl.: bhrātarām (i. e. °rān) Mv iii.428.9.

Nt. nom.-acc. sg. (dual dvandva): mālāpitaram

Bbh 118.25, 27 'mother-and-father'. Once, perhaps, from a noun of agent in -lr (Skt. strong stem -lār): pranetarena, inst. sg. (but v.l. °lārena) Mv ii.218.3.

Even feminine nouns may have masc.-neut. endings. These are perhaps analogical to the acc. sg. málaram (etc.), which resembles a masc. or nt. a-stem: mālarair Mmk 136.20 (vs; = mālrbhih, which would be as satisfactory metrically); dhilarena Mv iii.39.17 (v.l. dhilaro, which seems not construable); sapla dhilarām Mv i.356.17 (= °ān; repeated line 18 as dhilarol); dhilarāms caiva ii.367.21 (vs); and, it seems, even n. sg. dhilaro Mv iii.88.18, 20; 89.12 (here Senart em. °lā), 14.

13.18. Fem. forms, stem -arā (-arī): But more often fem. nouns imply stems in -arā (or possibly, once, -arī), starting with the acc. sg.: mālarām Mmk 33.12 (vs; before consonant; mālaram equally good metrically); dhīlarām Mv iii 146.4; same form (v.l. °laram) iii.284.3; in 284.17 -dhīlarām should also be read with one ms. (the other has completely altered the passage; Senart emends [or misprints?]-dhīlāram).

Oblique sing.: -dñitarāya Mv ii.58.2 (so read with mss., cf. Jāt. v.403.4-5; loc.); dhītarāye (inst.) ii.111.14.

Nom. pl.: dhitarāh LV 53.1 (vs); mālarāh Mmk 21.12; 44.10, 20 (all prose).

Acc. pl.: dhitarā (tor °ās; before s-) Suv 63.3 (vs); svām duhitarī LV 320.1 (mss. so, or °tarīm or °tīn; acc. pl.; Lefm. em. °tīr; prose).

Gen. pl.: dhītarāņam Mv i.356.6 (MIndic for °rāņām).

Stems in -*ūra* for -r

13.19. In both Pali and Pkt. (Geiger 90.3; 91.1; Pischel 389 ff.) we find more commonly 'stems' in $-\bar{a}ra$, from nouns (chiefly of agent) with Skt. strong stems in $-\bar{a}r$. The Skt. acc. sg. in $-\bar{a}ram$ was the starting-point. In our language however they are not very common. Stem -āra in composition: šāstāra-saņijāa LV 232.5 (vs); šāstāra-vādin Mv i.254.1 (prose); šāstārapratijāa- Mv iii.383.17.

Nom. sg. -āro (m.c. -āru): trātāru SP 451.4 (vs); pratigrahetāro Mv iii.54.16 (prose).

Acc. pl.: rāja-karlārām (= °ān) Mv iii.207.9 (prose).

Stems in -i for -r

13.20. In Prakrit (Pischel 389 ff.), stems in -i as phonetic replacement of -r are not rare. In our language they hardly occur. I have noted only napla-pranapti-käye Mv i.348.9 (prose; so mss.; § 13.4), where napla = naplr, pranapti = pranaptr; and cf. jāmātiko, for jāmātr-kah, Mv iii.146.5, 9, 12 (and jāmātuka, § 13.22).

Stems in -u for -r

13.21. On the other hand, stems in which -u similarly appears as phonetic replacement of -r, which are common in Pali and Pkt. (Geiger 90.2; Pischel 389 ff.), are also quite common here. The able-gen in -u for Skt. -uh is of course not counted here, tho it very likely had some influence in creating the apparent 'stem' in -u.

13.22. Stem in -u in composition: śāstu-pūgam (mss. °pūjā) Mv i.126.3 (vs); śāstu-bimbam Mmk 131.26 (vs, but metr. bad). Cf. also jāmātuka = °tr-ka, AMg. jāmātuka: jāmātukasya Mv iii.24.18 (prose); pravestuka Mmk 49.12, seemingly for *pravestr-ka, 'one who enters' (Dict.).

13.23. Nom. sg. -uh or (more often) -u: mama bharlu saháyakas LV 235.7 (vs; wrongly Interpreted by me in HJAS 1.70); pilu mama śuddhodaneli nāmnā 243.3. bharlu rājā Mv i.207.13 (vs; text not certain; probably read śruņi for śrņuya, in which case meter seems to require bharlū); mātu ii.14.12; 26.1 (both prose), in 26.1 read with mss. bodhisaltvo ca māyā ca mātuh sārdham śivikā samārūdhāh, § 7.60. — śāstu Sukh 23.2, perhaps to be read with v.l. for text śāsta, § 13.6.

13.24. Acc. sg. -um (or -u, m.c.): mālāpitum Mv ii.230.13 (vs); šāstum 361.6 (vs); mātum iii.130.12 (prose); pitum 406.15 (prose). — šāstu Mmk 134.12, 16 (vss).

13.25. Inst. sg. -unā, once from a feminine stem: pitunā Mv ii.101.4 (prose); kuladuhitunā Sūramgamasamādhisūtra, Thomas ap. Hoernle MR 126.1 (prose).

13.26. Abl. sg. -usmā (?), apparently for -usmāl, with masc.-nt. pronominal ending, but in a fem. stem: mālusmā Mv ii.38.12 (vs); so Senart with one ms.; the other seems to have a mere corruption of this. But the syntax is not too clear.

13.27. Gen. sg. masc. -uno (as in u-stems, § 12.34). All verses except those marked prose: idstuno Mv i.68.18 (note idstu, gen., line 16); 306.19; ii.297.1 (°nah, prose); iii.401.15; Mmk 103.13; 431.7; 586.10; Gv 34.14; bhartuno Mv i.207.11; ii.12.15; pituno Mv iii.105.13.

13.28. Gen. pl. -unām (as in u-stems, § 12.71, in vss; the short u may be m.c. only): sāstunām LV 172.17. sāstunām Mv i.70.3, so read with most mss.; Senart em. sāstuh.

Miscellaneous forms of r-stems

13.29. The remaining forms are miscellaneous; they are derivable from Skt. *r*-stem forms either by Mindic phonetic alterations, or by analogic morphological processes which are in part also paralleled in recorded Mindic. Some of these latter are however unexampled elsewhere, and of no little interest.

13.30. Nom. sg. -ās (?). Once, in Mv i.312.9 (prose), the mss. present a nom. sg. ending in -ās; is it analogical to masc. -as stems? śāstāś ca (Senart em. śāstā ca). Cf. § 17.51. 13.31. Voc. sg. -ā, borrowed from the nom. So also in Pali (Geiger 90.5) and Prakrit (Pischel 391, see especially end, Culanīpiyā). bhrātā Mv ii.460.18; iii.69.8; dharmabhrātā Mv iii.350.19, 20 (in 18 °bhrāte). All prose.

13.32. Acc. sg. - τ , neuter dvandva; nahi bhūµn mätrpitr draksyati putradārām LV 190.20 (vs), 'he will not again see mother and father, son and wife'. Most mss. read so, but A, the best, mātapita, which would belong above, § 13.3. It seems to be meant for a nt. sg. dvandva, tho this would be unusual in Skt.; the individuals, rather than a collective group, are meant.

13.33. Oblique singular -are, -ari. Without precedent in any Mindic known to me are forms in -are, -ari (both used only in prose, the variation hence not due to meter) which seem to be used for any oblique singular case-form. The forms which occur are pitare, pitari, matare, mātari, and perhaps dhitare. Senart, note i.603, regards pilare as 'a mere orthographic variant of pilari', and the latter as an instance of loc. used for inst. But neither suggestion is satisfactory. The ending -e is commoner than -i in these forms, and the loc. is not likely to be the startingpoint; the inst. usage, tho common, is not exclusive, even with the addition of the loc., for there are clear genitives, perhaps also one dative. It seems to me that we have here common oblique-singular forms of the r-declension, analogous to -aye, -ayi, -iye, -iyi, -uye, -uyi, or with short penultimate vowels, especially -iye, -uye, etc., in the fem. a, i and u-stems; and probably formed in imitation of them, just as in Skt. the acc. pl. endings - rn, - rs, and the gen. pl., -rnam, were formed in imitation of -in, -is, -inam, -un, -us, -ûnâm. The exclusively short a in the penult would be consistent with the non-occurrence of -ar- in the nouns of relationship concerned. It is true that masc. as well as fem. r-stems are involved; but in other masc. r-stem forms, \$\$ 13.9, 12, 13, and even in the case of i and u-stems, there are occasional occurrences of the regularly fem. oblique case-forms in masc. words.

13.34. The forms occur only in the prose of Mv, as follows. Inst.: parityaktā pitare i.311.3; parallels with pitari 310.17; 311.4, 7; mātare kapitavastum visarjitā 354.9; yathā tāye mātari samdişfā 354.20, repeated 355.2; pitare punah-punar uccati ii.101.2; pitare pi (mātare pi, not in mss.) 103.4; pitare kumāro ... visarjitah 103.15; mātare vuccati 463.16 'was spoken to by her mother', and so also 464.14; 465.16 (so v.l.; text with 1 ms. mātaye); 467.5; 473.2; 474.17; 486.18; pitare (preşitāni) 469.16; mātare iii.131.20 (samāgacchatu, 'let him join his mother'); tāye mātare 132.6; 155.9, 10, 12; (tāye) ca mātare mgāye 150.18; mātare 179.9; pitare ca kālagatena 184.9 (note the inst.

13.35. Dative or Loc.: malare agatoa 'coming to her mother' My 1.310.18; so mss.; Senart em. malaram. Note that pilare and pilari occur in the same context, and are kept by Senart who regards them as locs.; malare could surely be kept as a loc., as well as pilare; the loc. could be used of the goal of *dgatoā*; but malare may also be intended as dative.

13.36. Gen.: (eşā) pilare ālokam nisrstā i.313.14, 'she is released unto the sight of her father'; aham elam pilari vacanam ... nivedayişyāmi ii.437.12, 'I shall explain this our father's word'. 13.37. Gen. or Loc. (ubsolute): laye rajadhilare (v.l. °dhilaye)... galaye ii.65.17, 'when that princess had gone'. As loc. could pass for masc. *a*-stem form, § 13.17.

13.38. Inst. sg. -rnā. This ending (admitted for neuter *r*-stems in Skt.) occurs sporadically in prose and verse. It is, of course, analogical to the inst. of *i* and *u*-stems. *pitnā* Mv i.309.10 (prose); Mmk 604.10 (vs); bhrātnā Suv 215.6 (vs).

13.39. Gen. sg. -u. This ending, the regular MIndic resultant of the Skt. -uh, occurs here as in Pali (not in Pkt.); chiefly in verses, but in Mv in metrically indifferent situations and occasionally in prose (these cases are noted). mātu LV 133.12; 411.20 (but in 133.12 read pituh, as metr. required, with all mss. but one, for pitu); pitu 199.2 (with Corrigenda); sāstu 357.10. — sāstu Mv i.68.16, 73.1; ii.364.11; 371.4, 9; 373.5; iii.54.15, 16 (the last two prose); 305.6 (vs but not m.c.); mātu i.125.3, 4 (both vss, but metr. indifferent; mss. vary, Senart once mātu, once mātuh); ii.8.19 (prose); iii.132.7 (Senart prints as stem In comp., which is possible) and 8 (both prose); jāmātu ii.23.14 (prose); pitu (before vowel; v.l. pitur) 124.19 (vs. not m.c.).

13.40. Gen. sg. -aro. No such form seems to be recorded in MIndic; but cf. Pali inst. málarā; both are doubtless formed by analogy to the loc. mālari (= padi: padā; pada); mālaro stanam Mv ii.210.17 (prose; no v.l.) 'the mother's breast'.

13.41. Nom. pl. -āra, -ara. Only in verses m.c. for -āraḥ, -araḥ. Not recorded in Pali or Pkt. sāstāra SP 287.6; pitara LV 341.19.

13.42. Acc. pl. fem. $-\overline{r}n$. A few times fem. *r*-stems are recorded with the masc. acc. pl. ending. (All prose.) $t\bar{am}$ māraduhilīm LV 323.20 (so all mss., Calc. °tā °līm); $t\bar{am}$ svaduhilīm 330.6 (all mss., both edd.); and perhaps 320.1, where some mss. duhilīm, others duhilari (§ 13.18; Lefm. em. duhilīr).

13.43. Acc. pl. -rn, -rm (?). Twice, at least, the mss. of Mv are reported as writing a short r in the acc. pl. ending; Senart keeps the first, emends the second: $m\bar{a}l\bar{a}$ pilf \bar{n} ca (for Skt. $pilf\bar{m}s$ ca) desenti 1.170.2 (vs); pilf \bar{m} ubhau 185.2 (mss.; vs; Senart em. pilf \bar{n} ; meter requires long syllable, but this requirement is met by final anusvara in our dialect).

13.44. Inst. pl. -rhi, -rhi. Parallel to -ihi, ihi, -uhi, - $\overline{u}hi$ in i and u-stems, we find both -rhi and - $\overline{r}hi$. The latter is certainly analogical to -ihi, - $\overline{u}hi$; its occurrence is independent of meter, which is nowhere concerned. All but one of the following are prose. bhrātrhi Mv i.352.20; pitrhi (v.l. pitrhi) ii.86.15 (vs, but metr. indifferent); mālāpitrhi 210.13 (v.l. °pitrhi); iii.383.6; 394.7; bhrātrhi 433.1.

13.45. Gen. pl. -rnām. Only occurs certainly in vss, where it may be m.c. for -rnām: in SP 70.2 Kashgar rec. duhśrāddheyam idam dharmam deśilam adya śāstrnām; but Tib. supports vināyaka of ed. instead of śāstrnām (reading rnam hdren). — bhrātrnām LV 387.6.—In prose: bhrātrnām Mv ii.434.3, 8; v.l. °(rnām in both.

13.46. Loc. pl. - **F**şu. Only in vss, m.c. for -*rşu.* mātāpitīşu Mv ii.220.18; 223.2 (v.l. °*rşu*, but meter requires long penult).

14. Diphthongal stems

14.1. The stem nāvā, an ā-extension of Skt. nau, is familiar in Pali, AMg. and other Prakrits (Geiger 88.2; Pischel 394). It also occurs here: as stem in composition, nāvā-yānena Mv i.217.6; nāvā-yānehi ibid. and i.227.12, 15 (all prose); and acc. sg. nāvām, LV 216.5 (vs); Mv ii.59.7 (vs). The only record of this stem in Skt. is the RV.

άπ. λεγ. nāvayā (instr.), on which see Ved. Var. 3 § 289, Wackernagel III p. 224; a better classification of it might be to call it a morphological Prakritism.

be to call it a morphological Prakritism. **14.2.** In Mv i.156.1 and $6^{\circ}(vss)$ the meter requires a long syllable in the final of *divi*, which apparently all mss. read; probably *divi* is the true reading.

15. Consonant stems

15.1. 'Root' consonant stems, and those which do not belong to special types like the s, n, and nt-stems, show the same two types of MIndic treatment found in the latter. They are made into vocalic stems, either by loss of the final consonant, leaving the preceding vowel as stem-final; or by addition of -a (with fem. nouns usually -ā) to the final consonant. Such processes, particularly the latter, are familiar even in Skt.: cf. in general Wackernagel III.319 ff., and (in cpds.) II.1.61 ff. ('Stammverstümmelung' 64), 108 ff. On MIndic see Pischel 395 and Geiger 75, 89 with note 1. Pischel refers to thematically extended forms only for fems., as sariā = our sarilā. He mentions the stems with loss of final consonant, like our jaga- for jagat-, which is common in Pkt. This stem seems not to occur in Pali, but for its type see Geiger ll.cc., also 66.2b where he mentions the 'new stem' parisā for Skt. parisad; he might have added upanisa = upanisad. Pali also has such stems as vijju- (beside vijjulā-) for vidyul-, maru- for marul-, Pasenadi- for Prasenajil-. Extended stems, in Pali as in Pkt., seem to be nearly limited to Skt. fems. The stem lvac- is fem. in Skt.; in Pali it is recorded as nt., either laco (based on nom.-acc. pl. tvacas?) or taca-. So also sarada- (apparently masc.; or nt.?) = sarad, presumably based on the nom.-acc. pl. saradah; apa- = ap-, ap- 'water'. Examples of other Pali stems, keeping the fem. gender of Skt., are nibbid \hat{a} = nirvid, sampadā = sampad, vijjutā, samidhā, saritā, pațisambhidă = pratisamvid (only BHS), giră. Some of these are cited as Skt. in the lexicons, but not in literature (except BHS).

Vocalic stems produced by loss of final consonants

15.2. a-stems. Almost the only a-stem of this sort noted is jaga (so also Pkt., above, but not Pali), for jagat. It is very common, but so far as my collections show, found only in verses; even Mv prose seems not to contain it, unless I have failed to record occurrences. Note that Kauş Up. 1.3 has jagāni, which Wackernagel III.263, 324 explains as analogical to lat : lāni; but no lāni or similar form occurs in the context, and it seems best to take it as merely a MIndic form. Examples (all in vss):

Stem jaga in composition: LV 158.8; 161.21; 162.3; 360.1; Mv ii.26.14; Lank 364.2.

jagam, nom.-acc. sg., LV 215.20; Mv ii.25.8; 46.10, 12, 14; Bhad 21 (but Gv 544.23 jagat); Lank 313.14. jaga, m.c., nom.-acc. sg., RP 7.12 (read as separate

word).

jagu, m.c., nom.-acc. sg., LV 164.4; 167.8; Gv 253.7, 11.

jagasya, gen. sg., LV 116.19; 134.15 (but in 188.22 read sarvajagato 'sya with mss. for Lefm. sarve jagasya); Bhad 9, 11, 15, 18, 21, 30, 61.

jage, loc. sg., LV 45.10; 46.13; 50.8; 111.17; Suv 31.5; RP 3.4, 6; 6.4; Mmk 59.21; 376.9.

jagi, m.c. for prec., LV 41.18.

jagasmin, loc. sg., Gv 255.25.

jageşu, loc. pl., RP 3.11.

15.3. Otherwise note tirya (Dict.), repeatedly for liryañc, liryak; and what seems to be a nt. sg. bahuvrihi ending in -sampa for -sampad, in a vs (m.c.): jñānasampa jagatopajivitam (so divide) agrayūnam Gv 56.14, 'the supreme vehicle, characterized by a treasure-store of knowlege, upon which the whole world lives."

15.4. Fem. nouns in which a precedes the Skt. final usually become a-stems, as in Pali parisa, upanisa (prose passages marked):

parișă (= parișud), stem in comp., Mv i.133.16 (prose); parisā, nom. sg. Mv. i.158.3 (prose); ii.446.16, 17 (prose); parisām, acc. sg. LV 361,10; 363,13; Mv i.171.12, 16; 354.21 (prose); mahāparisā, acc. sg. (mss.; Senart em. °șām) Mv ii.419.1 (prose); parişâyuh (read °yāh), abl.-gen. sg., SP Kashgar rec., La Vallée Poussin JRAS 1911.1074 (KN 267.10 parșadi without v.l.); parișāya, oblique sg., SP 98.7; Mv i.310.6 (prose); *purişāye*, obl. sg., Mv iii.10.15 (prose); *purişāyām* loc. sg. Mv i.75.3; 171.14 (in the same context 172.9 parisadi); parisāhi inst. pl. Mv iii.53.1 (prose).

parsā (= parisad, parsad): parsāyc obl. sg. Mv 1.310.5 (prose); parşās nom. pl. SP 9.1; parşāņām gen. pl. Mv i.27.11 (prose; one ms. parşadānām); parşāņu SP 237.13; parșāsu loc. pl. SP 354.12.

upunișă: °șâm, acc. sg., SP 333.7 (prose; v.l. °șadam); RP 59.16 (prose); KP 159.17 (prose); in AsP 72.4; 98.11 (both prose) upanisām apy upanisadam (or °sadam) api (! as if the two were quite distinct words); spelled upanisam Sukh 31.9 (prose); Dbh 66.26 (prose); Gv 542.3 (prose). Most texts have, either alone or by the side of this, the stem upanisad.

Once, at least, before the suffix -vant(a), the prefinal Skt. a remains short even in one of these feminines: parșavantâna tâyinâm SP 303.13 (both edd., no v.l.). Tho found in a verse (anustubh), the short a cannot be m.c., for the second syllable could quite well be long.

15.5. i-stems: Stem in comp.: dasadi-gatām LV 163.14 (vs), so read with most and best mss., Lefm. daśadig-gatām (unmetrical); abhiji- (for abhijit-) Mv i.2.5 (prose).

Nom. sg. -i: abhiji Mv iii.236.14 (prose).

Voc. sg. -i: abhiji Mv i.2.3.

15.6. u-stems, only vidyu for vidyul, maru (usually 'god', = deva) for marut; both very common; muru, at least, is used in prose (unmarked examples are from vss):

Stem in composition: vidyu- LV 174.17; Mv i.146.8 = ii.7.5; ii.337.5; RP 45.5; maru- SP 12.13; LV 82.15 (prose); 80.19 (read ca marusahasrair); Mv i.71.21; 72.11; 90.18; ii.328.5; Av ii.176.12.

vidyur, nom. sg.: vidyur iva Mv ii.181.5 (so mss.; Senart em. vidyud iva). - vidyu, nom sg.: LV 173.14. vidyum, acc. sg.: Mv ii.397.15. — marū, nom. pl.: SP 208.9. - maru, nom. pl.: Mv ii.299.5. -- naramarū, acc. pl.: Mv i.72.15 (Senart em.; mss. °ru; meter obscure). vidyubhih, inst. pl.: LV 214.7; marubhi My i.100.9; maruhi Mv iii.267.16 (here in its old meaning, the Maruts). --marûņām, gen. pl. LV 12.13; 370.14; Mv i.113.15; 143.16; 268.15; marunām (m.c.) LV 129.19.

Consonant stems made into a-stems by addition of -a

15.7. Masculine and neuter stems. As we saw above, these are recorded for Pali and Pkt., at least in 15.16]

Pischel and Geiger, only with Skt. feminines (in words of this general type). In our language, however, Skt. masculines and neuters may be treated in the same way. Thus:

vaņija- (see Dict.), stem in comp., LV 385.13, 16 (vss); vaņijāķ nom. pl. id. 208.6 (prose); ^ojānām gen. pl. id. 387.10 (prose).

jagala- (for jagal, but far rarer than jaga-); jagalasya Mv 1.120.3 (vs).

maruta-, stem in comp., LV 95.5 (vs); marutais inst. pl. SP 69.2 (vs); marutāna gen. pl. Samādh 19.21 (vs).

15.8. There is even, once, a stem vanika- (in comp.), which may be formed by addition of -a to the nom. sg. vanik: Mv i.1.8 (prose); see Dict. for another possibility. Senart i.367 interprets it as vani (MIndic for vanij) with suffix -ka, but this seems implausible. He cites a parallel from SP, bhisaka from bhisaj; but (aside from the fact that this might be interpreted as I have suggested interpreting vanika) the reading of KN for the SP passage, 292.11 (vs), is bhisaika without v.l. (so also WT). See Wackernagel III.326 (§ 166b) for stray inflectional forms based on the nom. sg., which are not dissimilar.

15.9. Feminine stems of this type usually add -ā and retain fem. gender, as normally in Pali and Prakrit (§ 15.1). Examples: par(i)şadā: parişadāyām Suv 21.14 (vs), mss., Nobel em. parşa°, m.c. (loc. sg.); parşadānām, gen. pl. Mv i.29.13 (prose); samidhā (= samidh): samidhāhāraka (stem in comp.) Divy 487.14 fl. (prose); puspasamidhānām Divy 70.6 (prose), gen. pl. (so mss.; ed. em. °samidhām); vidyulā (cf. § 15.10): vidyulām LV 214.16, acc. sg.; praisamvidā (cf. § 15.11): dharmapralisamvidā Bbh 89.25 (prose), etc. (nom. sg., four times); prati-saņvidānām SP 204.11 (vs), gen. pl.; °dāsu 205.12 (vs), loc. pl.; °dāto Šikş 109.6 (vs), abl. sg.; nirvidā (Pali nibbidā) = nirvid: nirvidāye Mv iii.331.4 (prose), dat. sg. (cf. LV 416.18 nirvidē); nirvidāya KP 126.10, 15 (vss), dat. sg.; girā (only Lex. in Skt.) = gir: girā, LV 360.15 (vs), nom. sg.; girām SP 152.2 (vs); girām Mv ii.143.23 (vs), acc. sg.; -girāhi Mv i.163.16 (vs), inst. pl.; sarilā: sa-nagara-nigamasaritā... vasumalī Mv i.83.3 (vs), nom. sg.; -saritālayam (stem in comp.; 'home of rivers' = 'ocean') Mmk 66.9 (vs); sampadā: nom. sg. °dā SP 462.6 (vs); Mv ii.164.16; 208.15 (both prose); acc. sg. or pl. °dā Mv i.89.2 (vs; read with mss.); inst. sg. °dāya SP 46.5 (vs) te āšaya-sampadāya visuddharūpāya samanvitā 'bhūt, 'they were endowed with a store of mental disposition pure in form' (wrongly Burnouf and Kern); inst. sg. °daye Mv ii.293.8.

15.10. Sometimes, however, a stem ending in short -a is found in such words. In verses, where meter demands short -a, this may be merely a matter of metrical license: viduuta-prajňa LV 414.11 (vs); and possibly taca-naddhāh 324.11 (vs), which might represent tvacā (inst. of tvac) naddhāh; sarada-ghana-samā (= sarad-gha°, but see § 15.12) 328.1 (vs); in 287.10 (vs), read probably: sa-pralisanwidasatya-balam ca (mss. very confused; pralisanwida might be m.c. for °dā).

1. Probably giri also occurs for gir 'voice, speech' as stem in composition: LV 393.3 (see Dict. s.v. Giri); 15.11. But there are some cases where meter cannot explain the short a. Thus Mv iii.321.14-15 has four times the stem *-pratisamvida-* in prose: artha-pratisamvida- $pr\bar{a}pt\bar{a}$ etc. Clearly the stem is here either masc. or nt.; nt. if we may trust Mv iii.67.2-3, also prose: catuari ca pratisamvidain sāksīkare (same phrase repeated line 4).

15.12. So the stem -tvaca- appears in that form in composition in Mv iii.71.16 (prose); 75.3 (prose); 83.12 (vs); Divy 630.8 (prose). Note that Pali also has taca-(taco? 15.1). Since Pali also has a stem sarada (15.1), it may well be that LV 328.1 (15.10) intends sarada-, not sarada-.

15.13. Such a stem is apa = ap, ap 'water' (also in Pali, Geiger 89 note 1). It occurs in acc. sg. dpam Mv ii.92.5 (vs), and as stem in composition: dpaskandham(acc.) SP 126.7 (vs), 'mass of water'; adha-apaskandham(to be treated as a single cpd.) LV 64.12 (prose), 'mass of water below' (i. e. under the earth, cf. the following); heşlâpaskandha (acc. sg.) caranaih pratigrâhyamānāh LV 298.20 (so read; vs), 'being caused to receive, presented with, the mass of water below'. (they were vajra-bodied, hard, impervious); heşli satasahasram yāvatas cāpaskandho LV 368.19 (vs; contrasted with dharanitala, next line; all these passages clearly refer to subterranean waters).

15.14. There appears to be one anomalous case of addition of *i* to a consonantal stem, viz. *hrdi*, in composition for *hrd*, 'heart'; see Dict. s.v.¹

Other, miscellaneous 'irregularities'

15.15. The nom. sg. (endingless) seems to be used as acc. sg., by analogy with the more or less constant assimilation of the forms of these two cases to each other: $b\bar{a}hum$ prasārya yalha vidyud ivā nabhasthā LV 340.13 (vs), 'stretching out his arm like lightning, as it were, in the sky'. It can only be the arm that is compared to lightning. But perhaps nom., 'as lightning (is) set in the sky'(?).

15.16. Inst. sg. -a, m.c. for -ā: vāca manena Bhad 1, 8; vāca manasā LV 139.8.

Abl.-gen. sg. -a, m.c. for -as, -o: kāyatu vāca manasya Bhad 43.

Nom.-acc. pl. -a, m.c. for -as, -o: vidišo (or, with v.l. on 16, vidišā) difa LV 177.4, 16 (vss; so read with most mss. on 16; ed. diši both times); diša Mv il.321.11; 336.5 (vss), acc. pl.

Nom. pl. with elision of ending -as, -o, or fusion with following \bar{a} -: marut' $\bar{a}gala = maruta(h) \bar{a}gal\bar{a}h$ LV 114.20 (vs).

? Nom. pl. endingless, = nom. sg.: paripūrņadvādaiasahasra marud višuddha(h) LV 113.19 (vs). The form marud is either nom. pl., or part of an irregular compound (§ 23.10).

Acc. pl. used as nom. pl. (weak for strong stem form): tiraico Sikș 80.1 (nom.; not liraicyo as suggested in note).

RP 47.1 (vs) giri-vara sahilarlha, '(Buddha's) excellent speech has sensible meaning.'

16. S-stems

16.1. Most 'irregular' as-stem forms noted fall into two categories. Both involve transfer to the a-declension, in the one case by addition of -a to the stem in as, in the other by loss of s, or substitution of -a for -as as stem final. Both are familiar in Pali and Prakrit, and there as here the latter are much commoner than the former (Geiger 99; Pischel 409, 410). Even in Sanskrit such forms are familiar (Wackernagel II.1 pp. 96, 113; III p. 286). But there they occur mostly in compounds; and they can certainly be called essentially MIndic when they appear on the scale found in our texts, which show them extensively in uncompounded words also, and by no means only in verses.

Stem -asa

16.2. In composition: vrata-tapasa-gunena LV 151.20 (vs).

16.3. Nom. sg. masc. -asas, -aso, etc.: LV (all vss): krpakarunāmanaso LV 167.22; sudinamanasah 194.5; pramuditamanasah 287.14 (note, in same line, attamanās ca); prasannamanaso 315.10; viraktamanaso 330.15. pritamanaso Mv 1.224.12 = ii.27.12 (vs).

16.4. Nom. sg. fem. -asā: apsarasā (repeatedly) Kv 3.9 ff. (prose). 16.5. Voc. sg. masc. -asa: hatatamasa LV 169.20

(vs).

16.6: Nom.-acc. sg. nt. -asam: krsnaširasam LV 16.9 (prose); ujjvalilalejasam 21.4 (prose); vratalapasam 212.11 (vs). — ālmacelasam Mv ii.232.5 (vs); manasam (= manas, not in comp.) 447.6 (prose).

16.7. Nom.-acc. sg. nt. -asa, in verse: oralalapasa LV 168.20 (vs), all mss.; Lefm. em. °sā; I am not certain that the meter (rolā or lolā) requires a long syllable in this position (5th syllable of the pada); it is often short in this passage, to judge from Lefmann's mss.

16.8. Loc. sg. -ase: vayase LV 174.9 (vs); virajase 227.1 (prose).

16.9. Gen. pl. -asānām: anantayaśasānām Mv i.172. 17 (vs).

Stem -a for -as

16.10. In composition, and before such suffixes as vant. Note that even a fem. stem-like apsaras may be treated in the same way; see below. yasakāmanāmnā SP 27.10 (vs; for yaiask°, which is used in the prose version but would be metrically impossible here); ananyamana-samkalpo 335.4 (vs). — yasavati (or yaśa°) LV 50.4 (vs); ura-ganda- (cf. Skt. ura-ga) 72.2 (prose); oja-vinduko, -vindu etc. 73.22 fl. (vss); mana-buddhi- 177.9 (vs); lama-raja-dhūmakelubhih 217.3 (vs); lejavatlamāš ca (all mss.) 301.20 (prose); apsara-koļi^o 313.7 (vs).—Mv (all prose): apsara-(sahasrehi, šala-) Mv i.32.7; 145.1; mana-karmeņa 128.3; ekāgra-mana-sampannā il.292.15. — varcāhāra Av 1.253.2; 254.2; 255.11 (all prose), for varca(s)-āhāra, 'feeding on dung' (in 1.253.2, only, best ms. followed by text reads varcohāra, which is surely only a ms. blunder); also varca-ghala i.252.1, 'pot of dung'; varca-dhānāni i.254.5 (vs) 'dung-heaps'. In the same context varcahsadršim i.253.1, with 'regular' stem varcas. - Suv (all

vss): tama-cetasā Suv 26.6; manendriyam 56.12; širajo 244.10, 'hair'. - apsaravargās ca Lank 6.5 (vs; one ms. apsaro°, metrically inferior).

16.11. Note on apsara(s). Several times above this stem appears as apsara-, even in Mv prose. Below we shall see that inflected forms with neuter endings (apsarani. apsaraih, §§ 16.24, 26) are recorded. It is clear that the stem was capable of inflection as a neuter a-stem, just as neuter s-stems are. No such forms seem to be recorded for MIndic, so far as I know.

16.12. Nom. sg. masc. -as, -o etc. ugratejo LV 39.13 (prose); samalilavayo (v.l. °yā) 174.17 (vs); ananlalejo 240.9 (before s-); virajah 288.2; 393.1 (vss); mūḍhamano 323.12 (vs).-In 442.1-2 (prose), text avahilah śrotah, Weller (39) em. avahila-śrolah, which must be right in sense; but the (best) ms. A has -šrolrah, so that it is not certain that an original s-stem (śrotas) is concerned (rather than śrotra). — mahāyaśo Mv 1.67.9 (vs); atitejah (mss. abhi°) 122.6 (vs); durmano 42.3 = 230.10 (prose; in 240.20, same phrase, durmanā); galavayo ii.41.2 (vs).

16.13. Nom. sg. fem. -ā: āttamanā My i,213.13 = ii.16.16 (prose). Cf. sreyā, § 16.35.

16.14. Acc. sg. masc. -am: anantayasam LV 45.10 (vs). 16.15. Acc. sg. fem. -ām: bodhi virajām (before vowel) LV 337.3 (vs).

16.16. Nom.-acc. sg. nt. -am, -am: rajam SP 157.12 (vs); paramāņurajam 316.7 (prose) and (Kashgar rec.) 316.8 (prose; ed. with Nep. mss. "jah). - śiram LV 56.9 (vs); rajam 114.11 (vs); punyaleja śiram (nom.) ürdhvavilokilam ca 116.6 (vs; so probably read) 'his head has glory of merit and an upward gaze'; firam api 165,11; vratatapam (before \bar{a} -) 211.13; 244.1 (both prose); vacam 336.5 (vs); frolam (v.l. *tram) 409.10 (prose). — manam Mv i.146.13 = 203.14 (vs; in ii.7.10, same vs, manah); manam in prose ii.151.15; 157.13; iii.379.10; lejam i.215.9 = ii.18.6; nabham i.222.20 (here mss. nabhe) = ii.25.6; ojam (haranti; subject ojo-hārakā) 1.253.3 (prose); padma-saram "saras) ii.97.12 (prose); uram 154.9 (prose); tapam 232.6 (vs); siram anu 282.6 (prose; so divide, 'along his head'); śiram 412.5 (prose); vayam (= vayas) anuprāpto iii.206.18 (prose); vacam 370.11 (vs)

16.17. Inst. sg. -ena: ojena SP 175.4 (vs). - tejena LV 122.21; 132.8, 14 (in 14 read tejena with most mss. for ed. tejo nu); 157.22; 385.15 (all vss). - manena Mv i.161.10; 176.11 (vss); lejena 210.14 == ii.14.13; ii.219.2 (all prose); yaśena ii.35.2 (vs); tapena 231.16 (vs). -- pādarajena Divy 606.28 (vs). — yaśena Suv 37.2; 42.2 (vss); manena 49.22 (vs). — manena Bhad 1, 2, 8 (vss). 16.18. Abl. sg. -āto, and m.c. -ato, -ātu, -atu:

tamāto Mv i.27.8; ii.215.10 (both prose; so Senart, rightly; mss. namāto in i.27.8, and corrupt in ii.215.10); nabhato ii.18.14 (vs). — nabhātu LV 194.11 (vs); tejatu 275.17 (vs).

16.19. Gen. sg. -asya: aparimitayaśasya Mv ii.219.18 (vs); manasya Bhad 43; nabhasya Bhad 46.

16.20. Loc. sg. -e: nabhe va (so divide) LV 49.21 (vs); sire (read the next word with v.l. abhivandate) 135.12 (vs); sire 190.15 (vs). - ure Mv 1.130.15 (vs); nabhe 167.2; 186.11; 216.11 = ii.19.8; i.222.4, 6 = ii.24.12, 14 (all vss); rahe i.274.12 (vs); sire ii.282.12 (prose). — nabhe RP 7.3 (vs).

16.21. Loc. sg. -asmi (m.c. for -asmim): lapasmi LV 331.2 (vs).

16.22. Inst. dual -ābhyām: -śrolābhyām LV 257.8 ff. (prose), several times, stem Skt. srolas (root sru). Note that in line 11 the best mss. read -śrolrobhi, which probably intends -srolobhir (inst. pl.); ed. -śrolābhyo, implausibly.

16.23. Norm pl. masc. and fem. $-\tilde{a}s$: (also $-\tilde{a}$, in defiance of Skt. samdhi): *tuşţamanāţ* SP 87.10 (vs); *tuştātlamanāt* ca 89.10 (vs; so read with WT for KN tuştānta^o). — muditamanāţ LV 163.12 (vs); sumanāţ 283.7 (vs); ekaširā dviširā triširāt ca yāval sahasraširā (before b-) 307.20 (vs); durmanās 397.21 (prose). — mahāyašāţ Mv i.101.9 (several mss. °yašā; vs, end of line); mahātejās ca 103.7 (prose). — ātlamanās Divy 66.24 (all mss.; prose; ed. em. °manasas). — muktamanā (before v-) Samādh 19.34 (vs).

16.24. Nom.-acc. pl. nt. -āni, m.c. -ānī: rajastarāņī LV 372.9 (vs) (for °larāņisi, stem taras, 'impetuosity', or 'current'); apsarāņi 223.8 (vs; so read, as nom. pl., with nearly all mss.; Lefm. apsarāņāņ, which I cannot interpret); cāpsarāņī (v.l. °ņi; all mss. one or the other) 233.19. On apsara, guasi-neuter stem, for apsaras, see §§ 16.11, 26. — nislejāni (for °jāņisi) Mv i.41.12 = 230.7 = 240.17 (prose).

16.25. Acc. pl. -a, m.c. for -ā == -āni: vratatapa vividhā LV 287.22 (vs).

16.26. Inst. pl. -ais: *jinolejair* LV 164.17; 167.11 (vss); *apsarai*(*h*) 237.8 (vs). — *dharanī-ajair* Suv 36.6 (vs). — *apsaraih*, °air Lank 4.9; 7.1 (vss; in 7.1 one ms. *apsaro-hāsyalāsyādyaih*, as cpd.; this is clearly a Sanskritizing 'correction'). **16.27. Gen. pl. -ānām**: *apsarāņām* l.V 49.5, 11 (vss);

16.27. Gen. pl. -ānām: *apsarāņām* l.V 49.5, 11 (vss); *dhularajānām* Mv i.113.2; 119.4 (vss; Senart em. *dhūla*°). **16.28. Loc. pl. -eşu**: *sroleşu* (v.l. *srolaņsu*) Suv

15.1 (vs).

Miscellaneous non-Skt. forms of as-stems

16.29. Masc. as-stems, as in Pali and Pkt., may end in the nom. sg. in $-\bar{a}$ even before voiceless consonants or pause: LV (all vss): -upaśāntamanā LV 46.4; mahāyasā 59.4 (for °śāh, which some mss. read); -vismitamanā 109.13; dīplatejā 220.4, most mss. (Lefm. with ms. A °jah, as if a-stem); durmanā 221.18. — ātlamanā Divy 440.24 (prose).

16.30. Apparently in analogy to this form, we even find, according to the printed text, in Gv 528.19 (prose) an acc. sg. in *-ām*, and that too where a neuter should be found: *mā partīttamanām utpādaya*, 'produce no limited mind', i. e. 'don't become fainthearted'. Cf. sumanas 'flower', fem., which has a stem sumanā-(mukha)-.

16.31. In verses, the nom. sg. (and acc. sg. nt.) of as-stems of all genders may end in -a, presumably m.c. Thus, masc.: punyateja LV 45.2; ugrateja LV 173.6. Fem.: punyateja LV 54.7. Nt.: na ca mana (all mss.; Lefm. em. manu) kşubhito LV 165.16 (cf. \S 8.36); nadisrota 175.8; teja 340.17; ura (acc.) 341.13. Prose cases are rare: mana (mss.) Mv ii.157.3.

16.32. Or, instead of -a, -u may appear, also in verses m.c.: payu LV 166.2; raju 338.7.

16.33. As in other declensions, -hi may replace -bhis of the inst. pl.: apsarohi My ii.192.18 (vs).

Comparative stems in yas

16.34. These are treated exactly like ordinary asstems. They may be made into a-stems by addition of -a to the weak stem in -yas, or (more often) by dropping the final s of that stem. Of the former we find the voc. sg. masc. $p\bar{a}p\bar{i}yasa$ Divy 202.5 (prose); and acc. sg. masc. kaniyasam (bhrālaram) Mv ii.82.9 (prose); $p\bar{a}p\bar{i}yasam$ LV 315.2; 318.2 (both prose). More usual, in the mss. and edd., is pāpiyāmsam, e. g. I.V 314.22 (prose; to be sure with v.l. °yasam).

16.35. Of the other type may be noted: stem in composition, and before suffix *-lara: sreyatara-* (see Dict. s.v.) Mv ii.54.6, 10 (vss); *nihśreyasampannā* Mv ii.292.3 (prose).—Nom. sg. fem., *śreyā* Mv ii.54.8, 18 (vss).—Nom. sg. nt., *nihśreyam* (grham) Mv ii.101.8 (prose).—Inst. sg. kaniyena Šikş 125.1 (prose; from Dharmasam-gitisūtra).

Stems in -is

16.36. Stems in *-is* resemble those in *-as* in that they may become either *a*-stems by adding *a* to the stem in *s* (yielding stem-final *-isa*), or vocalic (in this case *i*) stems by the loss of the *s*. (So also sometimes in Skt., Wackernagel 111.290 ft.). Examples of the first process, which here as in the *as*-stems is less common, are *jyotişā* LV 123.1 (vs) and *jyotişānī* (m.c. for $^{\circ}ni$) LV 195.13 (vs), both nom-acc. pl. nt., for *jyotiṣānī palantah*, for Lefm.'s em. *jyotişā nīpalantah*.)

16.37. The other type is shown by frequent use of -*i* instead of -*is* (-*ir*. -*ih*) as stem-final in cpds., and also by inflected forms from *i*-stems. Both are found in a prose phrase occurring as *jyotiarthiko jyotigaveşi jyotim* (acc. sg.) paryeşamāno Mv ii.121.8; repetitions 122.3 and 18 read *jyotyatho* and ^aathiko, and *jyoti* instead of *jyotim*. The same phrase occurs in LV 246.17 in the form agnyarthi *jyotirgaveşi* (one ms., not usually considered a good one, *jyoti-ga*^o) *jyotim* (only the same single ms. *jyoti*) parye-samānah; and the last two words (only) are repeated (with paryeşala iti) 247.7 where most mss. read *jyoti*.

16.38. Other cases of stem in -i in composition, and before suffix vani: sarpir (note this!) yalha sarpimande LV 159.16 (vs); arcival (to arcis) 176.1 (vs); sarpi-hutārciteja-sadršā 329.16 (vs); sarpimandam 386.5 (prose); sarpidohanam 386.18 (vs).

16.39. Other inflected forms of such stems: in LV 353.8 (vs), read: tulyah samo'si yatha sarpina sarpimandaih, 'thou art equal and similar (to us), as ghee to the scum of ghee' (cf. 159.16, cited above; Lefm. sarpini, but nearly all mss. sarpina, which stands m.c. for sarpinam, nom. sg. nt. as of an i(n)-stem, see § 10.3). — arciyo Mv i.25.14 ff. (prose), nom. pl. fem. (§ 10.168); jyotisya ii.121.10; 123.2 (both prose), gen. sg. (§§ 10.78 f.).

Stems in -us

16.40. Aside from commonplace metrical variations like divyācakṣuṣa LV 108.16 (vs; for divya-cakṣuṣā; Lefm. prints as two words), the 'irregular' forms of us-stems fall into the same two broad groups as the *is*-stems, but are much commoner than they. That is, they may be made a-stems by addition of -a, or u-stems by loss of -s. (So occasionally in Skt., Wackernagel III.290-292.)

16.41. Stems in -uşa are not common, and seem to be found only in compounds (cf. Mbh. Crit. ed. 1.169.24b sacakşuşam, and the like in Skt.; Wackernagel II.1.114 -āyuşa): LV (all vss): rūpam vaišravanātirekavapuşam 130.13 (nom. sg. nt.); amitāyuşas 199.18 (nom. sg. masc.); acakşuşānām 224.5 (gen. pl.). — tesām acakşuşānām Mv ii.223.5 (vs; mss. °şonām); māmsacakşuşam iii.177.8 (nom. sg. nt.; prose), so with v.l. for text °cakşum.

16.42. Stems in -u from original us-stems_are very common. They participate generally in all the peculiarities shown by *u*-stems in this dialect. In so far as such forms are not recognized in the declension of Skt. *u*-stems, it has seemed most practical to list them along with the corresponding *u*-stem forms, above. We refer to our chapter on *u*-stems for the following forms: -*cakşū* nom.

sg. masc., § 12.14; äyum, cakşum, nom.-acc. sg. nt., § 12.29; -cakşuşya, -ayuşya, °sya, gen. sg., § 12.35; -cakşuye, oblique fem., § 12.42: -cakşu, nom. pl. masc., and nom.-acc. pl. nt., § 12.61: cakşūhi, °uhi, inst. pl., §§ 12.65,67.

16.43. This list will be completed by listing here forms of us-stems which in Skt. would be regular derivatives of u-stems. Stem -u in composition, and before suffixes (-mant, -ka): cakşumān SP 192.1 (vs). — divyacakşuprādurbhūtena LV 84.7 (prose). — caksumām (= caksusmān) My 1.122.13; ii.143.4; 166.3 (all vss); pañcacaksu-samanvāgatā i.158.1 (prose); divyacaksu-darsana- 228.12 (prose); bhavacakşukaih 337.4 (prose).

16.44. Voc. sg. masc. -o: visuddhacakso LV 370.20 (vs).

16.45. Acc. sg. masc. -um: hatalejavapum 1.V 174.10 (vs); kanakaoapum Mv 1.210.2 = ii.14.1 (vs). 16.46. Nom.-acc. sg. nt. -u: māņsacaksu SP 355.8

(vs), divyacakşu 9; āyu 68.9; 218.1 (both vss) .-- LV (all vss): āyu 79.17; cakşu 235.13; 421.10, 12. — mānsacakşu Mv i.158.10; prajñācakşu 337.4 (both prose and v.l. °şur).

16.47. Nom.-acc. pl. nt. -ūni: cakşūņi My i.3.14; 158.3, 5 (all prose; mss. largely ^ouņi). 16.48. Gen. pl.: viśuddhacakşūņām Mv i.167.15 (vs;

mss. °cakşuņām, unmetrical).

Stems in -pāms, -us

16.49. The stem Skt. vidvāms, vidvat, vidus, most commonly appears as vidu, on which see Dict., and u-stems, above. This seems to me clearly based on the weak stem vidus-, precisely as ayu, caksu etc. The same MIndic stem appears as Pali vidu, Pkt. viu; 1 do not believe it is a Vedic inheritance (Pischel 411 note 1 and Ved. St. 2.236), tho the Vedic form cited by Pischel may indeed be an early morphological Prakritism.

16.50. Again as with cukșuș-a and the like, vidușa-ka occurs in the same sense: My i.134.11 (prose).

16.51. The preconsonantal weak stem vidval appears Prakritically as vidva, as stem in composition, vidvamadena LV 366.22 (vs) 'with pride of learning', which might be described as m.c. for vidvan-madena, but for the fact that the same line contains avidvām (= van), clearly as acc. pl. masc. of a stem avidva: na ca bhava atimanyase avidvām, and you, Sir, do not disdain the ignorant.

16.52. Further, the form vidusas (or m.c. vidusa), the Skt. acc. pl., is used as nom. pl.: -viduşaś ca bhavanti Mv i.133.16 (prose); pravacana-viduso Mv i.176.7 (prose); prajňaviduşa LV 242.12 (vs; m.c. for °sas).

17. N-stems

17.1. The vast majority of non-Skt. forms of n-stems are adaptations to the a-declension.¹ They are made in three ways. (1) The pre-consonantal weak stem in -a, used even in Skt. regularly as composition stem form, is inflected like any a-stem. (2) To the pre-vocalic weak stem in -n or -an is added stem-final -a, and this stem in -na or -ana is thus inflected. (3) The same is done with the strong stem in -an, producing a stem -ana. All three types are found in Pali (Geiger 92.2) and to some extent in Pkt. (notably for the resultants of alman, Pischel 401; see also the adjoining §§). In our language, the first type is much commoner than the other two. After listing forms of these three types we shall mention first a few forms showing MIndic epenthesis of a or i before n in weak-grade forms, and then conclude with others, more heterogeneous in nature.

Weak stem in -a, declined as a-stem

17.2. This is the normal form of stem used as prior member of compounds in Skt.; it is also extensively used in Skt. as final member of compounds, under conditions set forth by Wackernagel. Pischel 400, however, notes that Prakrit often reverses its general tendency by using an n-stem at the end of cpds. where Skt. would substitute an a-stem. The same may be said of our language; e. g. ye irāvaka dharmarājāah SP 146.9 (vs), 'the disciples of the King of Dharma'; misinterpreted by Kern and Burnouf as nom. pl.; cf. asya (= tathāgatasya) śrāvakā(h) 148.13 (prose); the Skt. would be dharmarajasya, Wack. II.1 p. 121, § 52d.

17.3. We find very many a-stem forms of this type from *n*-stems, not only with the regular Skt. a-stem endings, but with various other endings which a-stems may take in our language. A few examples follow; they could be multiplied many times. In the majority of texts only or chiefly in verses; prose passages are marked.

17.4. Nom. sg. masc. -ah, -o etc.: murdho LV 276.7 (so read; best mss. mūrdhah; other mss. and Calc. mūrdhni, mürdhim; Lefm. em. mürdhnam; the stem mürdhan is masc. in Skt. tho it may have nt. endings in BHS); rāja idam Suv 149.8 (can be understood as representing rājah by normal samdhi).

17.5. Nom. sg. masc. -u, only m.c. for preceding: rāju LV 76.3; 326.19 (v.l. rāja); 327.2.

17.6. Nom. sg. masc. -a, also only m.c.; but it is open to question whether it represents -a or -ah (-as, -o etc.); whereas the preceding -u surely implies an a-stem: raja LV 43.11; 170.15; 241.7; brahma 81.11; 92.9; 118.22; 240.22; ālma 393.10. — rāja Suv 150.6. — brahma RP 6.16; ālma 10.16; 12.11.

17.7. Voc. sg. masc. -a: rāja My i.128.19; 129.7; iii.457.3 (all vss, but metr. indifferent as to final); viśvakarma Divy 59.14 (prose; so mss.; ed. em. °man).

17.8. Acc. sg. masc. -am: mahatmam LV 47.9; 57.18; mardham 357.10 (end of line). — brahmam Mv i.245.9 (prose).

17.9. Acc. sg. masc. -a: avacişu rāja LV 240.21 (vs), 'said to the king'.

17.10. Nom.-acc. sg. nt. -am (-am): namam SP 27.10; sthāmam 53.16; 380.2-3 (prose); karmam 114.11. premam LV 41.20; 410.1 (prose); sthāmam 153.3 (prose); 234.7 etc.; veśmam 231.3; romam 311.9; bhasmam 312.20 (end of line); 315.22 (twice, once at end of line). - karmam a- Mv i.104.17; karmam 169.6 (in same line karma; meter indifferent both times); nāmam 227.4; 303.7; ii.210.11; iii.144.17 (ali prose), etc.; veśmam ii.28.14. - - carmam Av i.265.4 (end of pada, not m.c.). - - karmam Suv 12.8 (v.l. karma; prose). — nāmam a- Kv 47.2; carmam u-69.16 (both prose). — nāmam Sukh 46.6.

17.11. Inst. sg. -ena: nāmena SP 23.10; 217.10; 222.3; 383.11; karmena 166.9; 175.7; 353.5; mürdhena 336.7; 343.12. — nāmena LV 28.8; premena 232.6. nāmena My i.51.7 (prose); 112.1; 114.12 etc.; -karmeņa (repeatedly) 128.2ff. (prose): *fleşmena* 213.9 = ii.16.13 (prose); *mürdhena* ii.26.6, 7 (prose); *sthāmena* 74.12 (prose); premena (dental n) 100.2, 3; 111.18 (all prose); carmena 213.9, etc. — tejena Suv 149.2. — nāmena Mmk 104.17.

17.12. Inst. sg. -inā: mūrdhinā Mv i.65.14, m.c. for mürdhena, § 8.40. Senart considers the i epenthetic, MIndic for murdhna. But this stem seems not to show epenthetic i in MIndic as in Pali rājinā, Pkt. rāiņā; in Pali, at least, epenthetic i in n-stems seems to occur only in forms of Skt. rajñ-, and this seems to be true of our language, § 17.50. (In Mv i.67.11 mūrdhinā is Senart's em. for mūrdhanā.)

17.13. Abl. sg. -a, m.c. for -āl, or 'stem-form', cf. § 8.9: anyatra karma sukrlād LV 37.7, 'except deed(s) well done'. No v.l.; meter correct as it stands; no other construction seems possible.

17.14. Gen. sg. -asya: karmasya SP 325.8; LV 36.17; Mv i.16.8 (prose); iii.26.18 (prose). - -nāmasya Mv i.54.5 (prose); carmasya ii.222.12; rājasya (not in cpd.) Suv 148.13.

17.15. Loc. sg. -e: adhve SP 53.5; 61.13; 205.1 etc. --janme LV 27.20; brahme 393.14. — karme Mv ii.463.2 (prose); veśme Mv iii.105.2; - ātme RP 19.6. - śirse KP 132.3 (contrast 132.1, prose, śirasi).

17.16. Loc. sg. -i, m.c. for preceding: -janmi Gv 487.12.

17.17. Nom. pl. masc. -ās, -ā: mahāimā LV 29.11 (before s-); romāh 310.1 (so both edd., no v.l.; roman nt. in Skt., and acc. to Childers and PTSD in Pali); -manokarmāś ca (masc. Bhvr.) Mv i.134.6 (prose).

17.18. Nom.-acc. pl. nt. -ā: mārakarmā LV 12.10; karmā Mv i.12.13; nāmā Mv ii.433.15, 18 (kušamišrāņi nāmā kriāni; both prose; in 18 v.l. nāmāni)

17.19. Inst. pl. -ehi: karmehi Mv 1.109.16. 17.20. Gen. pl. -ānām: In SP 273.1, ed. rājeşu, Kashgar rec. (La Vallée Poussin, JRAS 1911, p. 1078) rājānām. — karmāņām Mv i.16.13; 26.9, 13 (all prose); 178.12; 188.6; rājānām il.169.13; 247.19. — karmānām Suv 141.17 (vs, not m.c.).

1. Cf. Wackernagel II.1 pp. 115-121; III.265, 321; but our forms are obviously MIndic in character and show no relation to the Sanskrit data except in so far as these reflect MIndic conditions, cf. the last reference.

7

17.21. Loc. pl. -eşu: rājeşu SP 273.1 (so ed., see preceding); brahmeşu 129.6. — karmeşu Mv i.134.12 (prose); Mmk 11.27; 29.21 etc. (prose). — adhveşu Gv 28.23, etc.

Prevocalic weak stem extended by -a

17.22. Type of stem murdhna, âlmana. The a before n is found in âlmana, parikarmana, carmana, parvana. Tho much less common than the preceding type, it is well authenticated in prose as well as verse.

17.23. Stem form in composition: mūrdhnābharaņā- LV 432.16 (prose), 'ornaments for the head'; so all mss.; both edd. mūrdhā^o; carmaņārthāya Mv ii.213.7 (prose), 'for the sake of the hide'; the cpd. mūrdhnābhisikta, 'besprinkled (as king) on the head', is common in Divy prose, e. g. 557.19, and in Av, see Speyer's note on i.69.8 (where it occurs); mūrdhna-prāptāħ RP 15.5; but Finot treats mūrdhna as a separate word, which is also possible (for mūrdhnam, m.c.; § 17.27); ātmana-hetur RP 31.3.

17.24. Nom. sg. masc. -nas: mahārājňah Mmk 24.13 (prose).

17.25. Nom. sg. masc. -(a)na, m.c. for preceding: so tīrņa-ātmana (read as one cpd. word; v.l. °ātmanah, unmetrical) parān api tārayeyā LV 300.17 'may he, himself rescued, rescue others also'.

17.26. Acc. sg. masc. -anam, -anam: dimanam: nagnā drši ālmanam LV 194.13, 'she saw herself naked'; LV 298.8 (before u-); Mv iii.25.2 (prose; so mss.; Senart em. ātmānam); Suv 62.2, read ālmanam, mss. mostly Atmanā (unconstruable); Nobel ālmānam (metrically impossible).

17.27. Acc. sg. masc. -anu or -ana, m.c. for preceding: ātmanu LV 341.12 (so Lefm. with ms. A; most mss. ātmana); LV 420.16 (so Lefm.; v.l. ātmana).

17.28. Nom.-acc. sg. nt. -(a)nam, -(a)nam: murdhnam LV 316.11 (nom.); tho regularly masc. in Skt. (and apparently in Pali), murdhan also has nt. forms in our tanguage; Mv ii.109.1 (prose; v.l. murdhnim; before vowel; Senart em. °nel; acc.); premnam 168.14 (prose), 17, 19 (vss); 169.4 (prose); 300.12 (vs); iii.148.7, 13 and 375.15 (all prose); parvanam (nom.; see Speyer's note) Av i.302.6 (prose); parikarmanam (= °rma) KP 92.8 (vs).

17.29. Inst. sg. -(a)nena: mūrdhnena SP 93.2; Mv ii.113.10, 11 (prose); premnena (dental n) Mv i.349.11 (prose); ii.171.19 (prose); 221.4 (vs); 460.9 (prose); alipremnena Mv ii.73.10 (prose); ätmanena Mv ii.329.18. Cf. iii.375.15 (prose) premnakena (Dict. s.v.).

17.30. Abl. sg. -nätu, m.c. for -nätas: mūrdhnätu LV 340.12.

17.31. Gen. sg. -(a)nasya: mahārājňasya Mmk 24.18 (prose); ātmanasya Suv 61.13.

17.32. Loc. sg. -ne: mûrdhne Mv ii.32.4 (prose; v.l. mûrdhnane; Senart em. mûrdhani); 109.2 (prose; v.l. °ni); 335.21; 359.20.

17.33. Nom. pl. masc. -nās: garuda-rājāās Mmk 18.26 (prose).

17.34. Nom.-acc. pl. nt. -nāni: -mūrdhnāni Mv il.307.5.

17.35. Inst. pl. -nais: vidyā-rājñaih Mmk 10.6 (prose).

17.36. Loc. pl. -neşu: -mūrdhneşu SP 405.5 (prose; both edd., no v.l.).

Strong stem extended by -a

17.37. Type of stem -ana; starting doubtless from acc. sg. masc. where alman-am can be analyzed alman-am. Like the preceding, less common than the first type, but well authenticated in prose and verse. See Wackernagel III p. 321, who correctly ascribes to Mindic influence some similar forms in late Skt., notably *ivāna* (which also occurs in our dialect). The form *adhvāna*-(karşila-) Mbh. Calc. 3.13397 (for which Bomb. *adhvani*) occurs in a spurious chapter, evidently an extremely late one, for it is not mentioned even in the Appendix to Sukthankar's ed.; apparently, therefore, it is not found in *any* of Sukthankar's mss.

17.38. Stem in composition. In the first six examples from Mv, Senart prints rājāna as a separate word (understanding a gen. pl., doubtless). But there is no doubt whatever that it is a stem-form, compounded with the following. Senart recognizes this in the precisely similar cases from ii.410.11 on. Mv (ail prose): koļtarājāna-sahasrehi i.234.16; rājāna-sahasrāni 271.15; rājāna-satāni ii.98.8; rājāna-satehi 98.10; rājāna-udānām 141.1; 158.1; sodna-mukhā 410.11; rājāna-sahasra-udānām mārga (Pali adhānā-maga) Suv 72.11 (prose); Prāt 510.4 (prose). — biksurājāna-ksohas šis 45.9. — sodna-tulya KP 107.30 'dog-like'.

17.39. Nom. sg. masc. (-anas.) -ano: dimāno 'vatīrya ... aham tārayişye etc. LV 216.5, 'descending myself (into the flood of the samsāra) I shall rescue' etc. (However, the meter seems to demand a short syllable for -o and a long for 'va-!) More normal Skt. usage would seem to suggest svayam; but the nom. ātmā is used in ways which at least come close to this. — ivāno KP 107.7. -taksāņo (see Dict.).

17.40. Nom. sg. masc. -ānu, m.c. for preceding; na ca yatra svānu (v.l. ścānu, intending śvānu) bhavati, 'and not where a dog is' LV 258.7.

17.41. Gen. sg. -ānasya: rājānasya Mv iii.158.2 (prose, no v.l.); taksāņasya (Dict.).

17.42. Loc. sg. -ane: adhvane Mv i.60.3; 271.19; ii.209.9; iii.138.9 (all prose).

17.43. Loc. sg. -āni, chiefly m.c. for -āne, but also sometimes, apparently, in positions where short final is not required: adhvāni SP 46.6; 48.12; 58.14; 67.11; 145.8; 207.8; 217.9; 336.8, 11; all vss, and generally m.c.; but in 336.8, 11 at the end of prior anustubh pādas, where -e would be metrically as good as -i; in 336.8, in fact, Kashgar rec. adhvāne; in 11 only two Nep. mss. are reported as reading one (Kashgar rec. not reported).

17.44. Nom. pl. masc. -anās: śvānāh SP 83.12 (v.l. śvānā).

17.45. Inst. pl. -ānais: -rājānaih Mmk 38.20; 129.22 (both prose).

17.46. Inst. pl. -ānehi: Mv (all prose): rājānehi Mv 1.267.6; ii.485.6, 16, 20; iii.20.2; pratirājānehi ii.485.3; iii.44.4.

17.47. Loc. pl. - aneşu: rajaneşu Mv ili.24.13 (prose).

Epenthesis of vowel before n

17.48. In a few cases we find epenthesis of a vowel, a or i, before n of weak-grade endings, in places where it would not occur in Skt. but can be paralleled from Mindic.

17.49. First a couple of forms with epenthetic a, only in verses and presumably m.c.; cf. Pail muddhanā (Geiger 92.1). Only inst. sg. -anā, -ana: mūrdhanā Mv i.67.11 (vs), v.l. ²dhnanā; Senart em. murdhinā (see § 17.12); nāmana Bhad 43 (m.c. for nāmnā).

17.50. The epenthetic vowel *i* seems to occur, as in both Pali and Pkt. (Gelger 92, Pischel 399), only in weak forms of rājan, for Skt. rājā-; and again only in verses, perhaps m.c.: dharmarājinām Mv i.76.6, gen. pl.; kāširājāno 305.6, mss., which is metrically impossible, read probably with Senart °jino (or alternatively °jano; gen. sg.); dharmarājinam iii.89.17, mss., read perhaps °jinām, or keep the reading of mss. (MIndic short a), rather than °jino with Senart (gen. pl. or sg.); rājino iii.386.9 (gen. sg.). On the alleged instr. mūrdhinā see §§ 17.12, 49. 17.641

Miscellaneous non-Sanskritic forms

17.51. Nom. sg. masc. -ās (on the ending -a, which may represent -ā, m.c., see above § 17.6): -ās, perhaps analogical to masc. s-stems, as in r-stems (§ 13.30); the Skt. ending -ās of s-stems appears in MIndic as -ā, like restored', by Hypersanskritism, in the following: asampramoșadharmāh SP 318.12 (prose), mss.; ed. em. -dharmā.

17.52. Voc. sg. masc. -ā, nom. used as voc., or for -a? kāśirājā Mv ii.225.6 (vs), so one ms., required by meter; Senart with v.l. °ja.

17.53. Voc. sg. masc. -e, as in Pali, analogical to fem. stems with nom. ā; so also in r-stems, § 13.9: Mv: brahme iii.316.21; 365.5; 419.5; mahābrahme 359.12 = 360.7 (these in vss); iii.215.1 (mss.; prose).

17.54. Acc. sg. masc. -ā, apparently nom. used as acc.: dadāmi samghe ... ātmā kunālam ca ... Divy 403.30 (vs) 'I give to the order ... myself and Kunala.' I see no other possible interpretation; ed. prints ālmākunālam as cpd. Meter requires a long final in ātmā.

17.55. Acc. sg. masc. -ana, m.c. for -anam: rajana SP 115.5; ātmāna 324.4.

17.56. Nom.-acc. sg. nt. -ā, m.c. for -a (?): nāmā SP 68.2, Kashgar rec., for Nep. mss. nāmnas; KN em. nāmnā, kept by WT without note.

17.57. Gen. sg. -ānas, -āno, as if from strong stem;

but probably only m.c. for -anas: mahātmāno, °nah Mmk

59.23; 225.1 (both vss). 17.58. Gen. sg. -(a)na, for -(a)nas, -(a)no, m.c.: ātmana SP 63.13; rājňa I.V 109.16.

17.59. Loc. sg. -(a)ni for -(a)ni; (?) so c' adhvani SP 113.5 (see § 4.21); alita-m-adhvani Samādh 8.21 (here at end of jagati päda; no v.l.).

17.60. Nom. pl. masc. -ana, m.c. for -anas: rajana SP 35.13.

17.61. Nom. pl. masc. -nas, acc. pl. used as nom.: -rājňas Mmk 20.10 (prose).

17.62. Acc. pl. masc. -anah, nom. pl. used as acc.: jälidharmänah sallvän LV 226.19 (prose; so all mss.; only Calc. °dharmān); rājāno Mv i.276.9 (vs); ii.141.3 (prose); Divy 248.1 (prose; so mss.; ed. em. rajño); pratyekarajano Mv ii.270.10 (prose).

17.63. Acc. pl. masc. - āna, for preceding, m.c., but also once in prose: ātmāna samjānatām SP 71.3 (prose; but perhaps misprint for "nam or "nah; ms. K' "nam; WT em. ātmānah); paribhāvitātmāna jinendraputrān 12.9 (vs); älmäna nivarlayanlah 196.6 (vs). 17.64. Inst. pl. -ābhis? m.c.? nāga-rājebhi Suv 157.11

(vs), so Nobel; mss. °rājābhih or °rājais ca; the latter is evidently a secondary correction, and we must probably read "rājābhih m.c. for "rājābhih. The final visarga makes no difference metrically, at the end of a pada.

18. NT-stems

18.1. Very common are a-stem forms based on an extension in a of the strong stem in nt, that is in -nta. These are common in Pali and Prakrit (Geiger 96 ff., Pischel 397). Much less common are similar a-extensions of the weak stem in -*l*, giving stem -(a)la (not recorded in Geiger, but occasionally found in Pkt., Pischel 398). From stems in both -ant(a)- and -at(a)- are even made feminines in -antā, -atā (instead of -antī, -alī). Thirdly, but not very commonly, we find a-stems from which nt is completely lost, leaving e. g. stem arha for arhant; they are also known, tho not common, in Pali and Pkt. (Geiger 96.2, 97.2; Pischel 398); the fem. is again -a. There are moreover a few cases in which an nt-stem is replaced by an n-stem. After these we shall list some cases of confusion between strong and weak stems (with consonantal endings), and then some miscellaneous 'irregularities'. Among these are included the few and rare confusions between Skt. mant or vant endings and ant endings (nom. sg. masc.).

18.2. The stem mahānt, which in Skt. has long \tilde{a} in the strong forms, frequently has MIndic shortening of this a. Most of the examples are otherwise MIndic and will be recorded in their proper places below. We note here the acc. sg. masc. mahantam (= mahāntam) SP 69.11 (prose); Mv i.27.11 (prose). Often the mss. vary between d and a; SP 72.3 (prose) mss. mostly mahantam or mahāntam, nt., ed. mahat with 2 Nep. mss.; 82.12 mahantam, Nep. mss. mahā⁶; also nom. pl. 84.11 mahantāś, Nep. mss. mahā⁶, etc.

18.3. The composition form of this stem, in Skt. almost always mahā-, may appear as mahā', as rarely in Skt., BR s.v. (where mahadbhaya is cited). In the following (all from verses) it is impossible to interpret this form otherwise than as part of a cpd.: sumahadbhaye LV 325.18; mahadbhayamkarā Mv i.10.1; mahalsvarāh i.100.8; mahadphalāni (Bhvr.) i.291.19 (= Pali mahapphalāni, Kh.p. 6.6); mahadbhayo ii.226.7; shānam mahadbhayam Lank 257.1, 'place of great danger', cited with mahābhayam Sikş 132.7.

Stem -nla

18.4. Very common in all parts of Mv, prose as well as verse. In most texts common in verse but occasional in prose.

18.5. Stem in composition (and before suffix -lara): asanlagrāhālu (= asadgrāhāl) SP 92.9 (vs); asanlabhābam 48.6 (vs); arhanlaśabda- 56.12 (vs); mahanla- 110.14 ff. (vs; ed. with Kashgar rec.; Nep. mss. mahānla-). arhanlaghālakā Mv 1.26.11 (prose). — ojovanla-larā Suv 164.8 (vs) 'stronger'. — anutpadanta-pūrvāni RP 59.18 (prose; so, to be taken as one cpd. word; ed. separates), 'not arising before' (... citlāny utpannāni). — balavanlatarā KP 83.3, 7 (prose) 'more powerful'; filavanla-pralirāpaka 134.1, 6 etc. (prose); sanlāsanla-vicilra- Lank 151.3 (prose), 'existent and non-existent'.

18.6. Nom. sg. masc. -ntah, -nto etc. SP (common, only in vss): anucańkramanio SP 61.10; vicintayaniah 62.7; chādayanio 126.2; saniah 111.1 'being'.—LV (only vss): kampayanio LV 56.9; sanio 126.18; mahanio 169.20; sumahāniah 287.3; bhāsanio 438.12. — gaveşanio Mv i.3.10 (prose); bhāvayanto 4.3 (prose); santo 53.2 (prose); vilokayanto 143.7; rudanto 156.11; mahanto 246.3 (prose); il.142.20. — mahatyāgavanto Suv 225.10 (vs; so read with v.l. [°tlyā°], m.c.; text mahā°). — prašamanto Bhad 21. — santah Sukh 22.16 (vs).

18.7. Nom. sg. masc. -nta, in vss, m.c. for preceding: mārganta SP 115.8; nirnādayanta 126.4. — earanta RP 27.11, 18.

18.8. Nom. sg. masc. -ntu, as preceding, m.c. for -nlas: locantu SP 111.4; damayantu 118.6; gacchantu tişthantu 128.11. — tyāgavantu sthāmavantu LV 170.20; prārthentu 223.14. — carantu Mv ii.375.12.

18.9. Nom. sg. fem. -ntā: bhaktimantā Mv ii.235.3 = bhaktimatī; sparšavantā iii.54.6 = sparšavatī. (Both prose).
18.10. Voc. sg. -nta: arhanta SP 147.3 (vs).

18.11. Acc. sg. fem. -ntām; and m.c. -nta: pibantām ca kanyām Mmk 56.8 (prose), 'and the girl while she drinks'; ulkām paiyali niskramanta nagarāt LV 194.19 (vs), 'she saw a meteor going out from the city'. So Tib.: groň khyer dag nas me sgron phyir byuň.

18.12. Nom.-acc. sg. nt. -ntam: mahantam SP 72.3, read so (with Kashgar rec.) or mahāntam (with most Nep. mss.) for ed. mahat; sarvāvantam 72.7 (prose); opatantam (= avapatal) 83.2 (vs); mahantam (= mahat, no v.l. anywhere) 85.3 (vs); 231.9 (prose); 406.8 (prose); (KN) na sattam nāsattam (sc. traidhātukam) 318.10 (prose), but read with WT na santam nāsantam; kiyantam kušalam bhavet 345.6 (vs). — sarvāvantam LV 4.14; 291.9, 21 (all prose).—Mv (prose): prajāāvantam, šilavantam (kulam; note that the LV parallel, 23.19 f., has filavat, prajāāvat) i.197.19 = ii.1.9; mahāntam ii.85.6; ii.45.8; ākāravantam ii.469.18. Also (vs) mahantam ii.81.1; etc., common. sarvāvantam Mvy 6328 (misprinted sarvāntam; correctly Index and Mironov). — tāvantam dānam (nom.) Divy 317.1 (prose); vyasanam mahāntam (nom.) 321.14 (prose). — mahāntam Suv 96.14 (prose). — mama santam svāpateyam Karmav 28.9 (prose) 'the property belonging to me'.

18.13. Inst. sg. -ntena: gaņayantena SP 305.2 (vs); prayujyantena ghaļantena vyāyamantena Mv i.246.4 (prose); ajānantena Suv 27.15 (vs).

18.14. Gen. sg. -ntasya: SP (all vss): prakājayantasy' ima(m) SP 150.3; 154.16; udāharantasya 152.2; bhāşantasya 236.12; read with Kashgar rec. tahim ca viharantasya 237.11. — pravifantasya Mv i.235.10 (vs); abhiniskramantasya (v.l. °matasya) ii.150.10 (prose); niryāntasya 150.12 (prose); bharantasya 213.2. — filavantasya KP 136.2 ff. (vss).

18.15. Loc. sg. -nte: arhante Mv i.111.11 (vs); praviante 235.14 (vs; so mss., submetrical line; Senart em. °ntasmim); abhinişkramante ii.163.20 (prose); sante 345.11 (vs); mahānte iii.121.11 (vs). — sagarutmante Mmk 457.20 (vs); garutmante 458.10 (vs).

18.16. Loc. sg. -nti, only in vss, and probably only m.c. for -nle (it could also be described as a strong-stem loc. in -l, instead of Skt. -ati): vrajanti SP 383.10; nişkramanti LV 237.14; pravišanti or višanti Mv i.237.4, mss., at end of line, submetrical by one or two short syllables; Senart's em. pravišantasmim is metrically impossible; probably a monosyllable was lost after pravišanti; lujjanti

18.53]

or *lujyanti* Mv ii.371.3 (so read with mss.; Senart's em. *lujjante* is unmetrical).

18.17. Nom. pl. masc. -ntās (with normal Skt. samdhi): SP (all vss): vicintentā (v-) 35.4; karontās (ca) 83.14; mahantāš (ca; Nep. mss. mahā^o) 84.11; parivarjayantāh 98.1; dhrtimantās (c-; in same line smrtimanto) 303.7; rddhimantā (v-) 306.6. — upanīyantā vikrošantā a- Mv 1;132.9 (prose); abhinandantāh 153.3 (vs); kathentā (r-) 215.8 (corruptly repeated ii.18.5). — ajānantāh abu(d)dhyantāh KP 106.2 (prose). — vicarantā (y-) RP 29.2 (vs).

18.18. Nom. pl. masc. -ntā, Mindic for -nlās: smarantā (k-) SP 274.3 (vs), so ed. with Nep. mss., Kashgar rec. (also La Vallée Poussin, JRAS 1911.1076) smaranta (metr. inferior); dhārentā (s-), 281.2 (vs), Kashgar rec. dhārenti (metr. inferior). — saņsarantā ca Mv i.101.14 (vs); kathayantā (end of line) 215.9 = (corruptly) ii.18.6 (vs).

18.19. For a possible nom. pl. masc. ending -nli, for -nle, see § 8.81.

18.20. Acc. pl. masc. -ntān (-nlām, before a vowel also -ntām): prakāšayantān SP 9.12 (vs); vivarjayantān 13.9 (vs). — namantām LV 197.14. — vedayantām Mv i.6.4 (prose); paribhujantām 27.5 (prose), pibantām 6; cyavantān (so, or °tām, mss., Senart °tām) upapadyantām 228.13 (prose); dipyantām (i-) ii.28.6 (vs), so mss., Senart °tām. — iyantān MSV iii.21.4 (prose). 18.21. Acc. pl. masc. -ntā: anubhontā Mv i.5.14

18.21. Acc. pl. masc. -ntā: anubhontā Mv i.5.14 (prose); kuļļīyantā (a-, with hiatus) 6.5 (prose).
18.22. Nom.-acc. pl. nt. -ntāni: Wv (all prose):

18.22. Nom.-acc. pl. nt. -ntăni: Mv (all prose): patantâni Mv ii.30.17; abhipatantâni 18; krīdantāni 109.9; ākāravantāni 463.5.

18.23. Nom.-acc. pl. nt. -ntā: mahantā SP 87.8 (vs; with yānaka = yānāni; but Nep. mss. muhāntāh, masc.); jala (= jalāni) lolayantā LV 196.16 (vs).

18.24. Inst. pl. -ntais: rddhimanlair Suv 160.5 (vs). 18.25. Inst. pl. -ntebhi(s): arhanlebhi (v.l. °bhis) ca LV 388.13 (vs; either reading is possible metrically).

18.26. Dat. pl. -ntebhyas: arhanlebhyali Mv i.158.7.

18.27. Gen. pl. -ntānām: arhanlānām Mv i.59.7, 11 (prose); vasantānām 125.1, cyavantānām 2, and others (all in vss) lines 3, 8, 9, 10, 13, 14, etc.; samkramantānām 136.4 (prose); pralikramantānām 158.14 (prose); mahantānām ii.260.15, 16 (prose; v.l. both times mahā⁶).

18.28. Gen. pl. -ntānam, Mindic for preceding: rabantānam (so, or °tāna, mss.; Senart em. °tānām) Mv i.6.1 (prose).

18.29. Gen. pl. -ntāna, in vss, m.c. for -ntānām: parşapantāna SP 303.13; vīryavantāna 306.1. — prakāšayantāna Sukh 45.11.

18.30. Gen. pl. - ntān', before vowel, m.c. for -nlānām: teşa ... aśraddadhantān' ima ... SP 95.8.

18.31. Loc. pl. -nteşu: āgacchanleşu Mv 1.21.6 (prose); gunapanteşu 277.7 (vs).

Stem in -la

18.32. Stem in composition: mahala-janaih LV 241.9 (= mahājanaih; vs; Lefm. prints as two separate words); tāvala-nislha (one word, not two) Bhad 46.

18.33. Nom. sg. masc. -tas, -to: rddhimalo narah SP 111.13 (vs); smaralo 448.6 (vs; repeated in the following vss).—LV (all vss): nidhyāyado LV 110.20; vasado (= vasan) 167.1; smaralo 167.3; vipaśyado 177.11; yāvalaś (c-) 368.19, 20. — bhagavalo Mv i.104.20 (vs); samanveşato 199.19 (prose; so with mss.; Senart em. °nlo); vapuşmalo 221.16 = ii.24.3 (vs; so certainly read in both passages, mss. vary). — nirikşalas (= nirikşan) RP 54.14 (vs).

18.34. Nom. sg. masc. -ta, m.c. for preceding: prayujyata RP 21.7 (m.c. for °tah = prayujyan). 18.35. Nom. sg. masc. -tu, m.c. for -to: krandatu

18.35. Nom. sg. masc. -tu, m.c. for -lo: krandalu LV 341.13; dharma śrnolu (for śrnolas = śrnvan, pple. formed in MIndic fashion on stem śrnont-, weak śrnot-, § 28.62, extended by -a; so Tib. chos mñan pas kyan; not 3 sg. impv.) na vindali trplim 364.5. — ośiralu Gv 255.14 'sending forth'.

18.36. Nom. sg. fem. -ta, m.c. for $-l\bar{a}$, fem. to -la(s): yāvala Bhad 46 (twice; with nislha, nis $lh\bar{a}$).

18.37. Voc. sg. masc. -tä, for -la (see § 8.27): paramalidhunalā LV 362.4 (vs, end of line; dhunalā for dhunala =

dhunan, voc. of pres. pple. dhunanl, to Skt. dhunāli). 18.38. Acc. sg. masc. -tam: bandhumalam rājānam

(so mss.; ed. em. ^omanlam) Divy 284.1 (prose). 18.39. Acc. sg. masc. -tu, only in vss m.c.: dhyāyalu

bodhisallvam LV 134.12; bodhisallvam jvalalu 241.6. 18.40. Acc. sg. fem. -tām: lişļhalām Mv 1.305.9

(so mss.; vs, metr. indifferent; Senart em. lis/hantim). 18.41. Nom.-acc. sg. nt. -tam, -tam: vasumalum

18.41. Nom.-acc. sg. nt. -tam; -tam; vasumalum (for °mat) Mv i.35.14 (prose), name of a city (always a-stem in this story); prasūyatam = prasūyat Lank 9.2 (vs), 'that which is being born', pres. pple.; in cpd. vandhyā-prasūyatam, 'the offspring of a barren woman'.

18.42. Inst. sg. -tena: (kanthakam) pallanalena Mv

ii.160.4 (prose; so mss., to be kept), pple. of pallānali, q.v. Dict.; šilavalena Mmk 76;27 (prose).

18.43. Gen. sg. -tasya: bhagavalasya Mv i.74.6 (vs); pravišalasya Mmk 463.17 (but meter is imperfect).

18.44. Loc. sg. -te: vasumale Mv i.36.3, 10, etc.; 45.6 (all prose).

18.45. Nom. dual -tau: rakşalau Mv ii.219.1 (prose; so the sole ms.; lacuna in the other); subject mālāpilarau, '(are) guarding', a kind of periphrasis for a finite verb. No need to emend with Senart.

18.46. Nom. pl. masc. -tā, MIndic for $-l\bar{a}s$, $-l\bar{a}h$, which latter may be intended Mv i.153.3; in the others here cited, $-l\bar{a}$ is followed always by voiceless consonants: mahatā prākāra LV 193.6 (vs). — mahatā parvatā Mv i.13.11 (vs); abhikānkşatā iti 153.3 (vs, end of line), so read with mss., = Skt. abhikānkşantah; Senart em. unnecessarily to "kşitam (as acc. sg., with the object); äharatā 340.7 (prose; so one ms., v.l. āhāratā; Senart em. äharatā; followed by t-).

18.47. Nom. pl. fem. -tā(s): bhūmayo daša jināna śrimalā Mv i.64.2 (vs; so 5 mss., 1 śrīmalo; Senart em. śrimalām; but construe with bhūmayo).

18,48. Acc. pl. masc. (-tān), -tām: ksīyatām Mv i.5.4 (prose) 'being destroyed', so 4 mss. (Senart with 2 mss. ksīyantā); (buddhām) bhagavatām Mmk 63.17; 229.23 (both prose).

18.49. Acc. pl. masc. -tā, MIndic for preceding (nom. used as acc.): anubhavalā Mv i.5.3 (so all mss.; prose; Senart em. °vanlā).

18.50. Nom.-acc. pl. nt. -tāni; iyalāni Divy 112.16 (proșe).

18.51. Gen. pl. -tānām: -pralivahalānām Bbh 157.6 (prose).

Forms based on stem in -a

18.52. Stem -a in composition: arha-gana (Corr.; for $arhad^{\circ}$) LV 436.1 (prose). — ananlav-udagrah Mv i.83.8 (so mss.; vs; stem ananlava, final vowel lost in samdhi; Senart em. $^{\circ}\dot{u}dagro$, \ddot{u} being demanded by meter; for $ananlavad-u^{\circ}$). — arha-parivaro Divy 148.9; 159.15 (so mss. both times, prose; ed. em. $arhad^{\circ}$ in 148.9).

18.53. The stem samprajāna, for $^{\circ}na(n)t$, = Pali sampajāna, occurs in composition in KP 123.2 (prose) samprajāna-cārī 'behaving (or walking) circumspectly', and especially in samprajāna-mṛsāvāda 'a conscious lie' (Pali sampajāna-musāvāda), Mvy 9266 (so read with Mironov, no v.l.; Kyoto text samprajānan mr°, citing °nain note); Mv ii.131.6, 7; 240.18; Prāt 502.15 (all prose). In Divy 70.16; 71.18 samprajānan-mṛsāvāda (ed. prints as two words, wrongly), altho elsewhere (618.4) Divy has (in prose) samprajānā as nom. sg. fem. On garulma-(stham) see below, § 18.66.

(stham) see below, § 18.66. 18.54. Stem -am in composition (as if nt. sg. for -ad!): iyamduhkhā hi ... nārakāh (or nar°) Divy 375.15, 21; 376.1, 9 (all prose) 'the hell-inhabitants (hells) have torments to this extent'. The cd. suggests in note reading iyaddu°. But cf. the Skt. use of idam- as 'stem' in comnosition.

18.55. Nom. sg. masc. -as, -o etc. samprajanas, or -no (the latter often without regard to Skt. samdhi, especially in Mv), for Skt. °nan 'conscious': SP 29.1 (prose) °nas (so read with WT and virtually all mss.; KN °jñānas); Mv i.150.6 (vs); 206.4 = ii.9.20 (prose); i.218.10 = ii.20.11 (prose); ii.298.9, 15 (vss; in all these °no); ii.366.11 (vs) °nai ca; 383.1 (vs) °no, etc., passim; Ud xv.5 (vs) °nah. The regularized nom. samprajānan occurs, however, even in Mv (e. g. ii.283.9 °nam, prose) as well as elsewhere (e. g. LV 343.19, prose).—Other forms: Mv araho (= arhan) i.37.6; 38.11 (both prose); arcimo i.216.18 (prose); 225.6, 16 (vss); bandhumo ii.271.5 (prose); bandhumas (v.l. °mano, see § 18.67) iii.231.1 (prose)

18.56. There are also, in verses, forms ending in -a, like bhava, samprajāna, certainly m.c., which might be considered to represent -as, -a, but which may also be treated as m.c. for -an or MIndic -am, below, § 18.79.

treated as m.c. for -an or MIndic -am, below, § 18.79. 18.57. Nom. sg. fem. -ā: hrīmā (for °matī) LV. 28.13 (vs); samprajānā Divy 618.4 (prose).

18.58. Acc. sg. masc. -am: -sthāmavam LV 110.21 (vs); arham... pralijānīse 405.22 (prose; so certainly read with Lefm.; the mss. all corrupt; mostly they read aham, but the parallel 406.4 *jinam* proves the reading intended. --Mv (all prose): arcimam Mv i.215.11; maham ca ogham ii.260.11 (repeated 261.11 with mahāntam instead of maham); pāpīmam ii.270.19; 287.7 (here mss. pāpīmām); 404.21 (in line 20 nom. sg. pāpīmām); bandhumam iii.230.16.

18.59. Acc. sg. masc. -a, m.c. for preceding: dandu ... irima LV 194.16 (vs).

18.60. Nom.-acc. sg. nt. -am: kulam suddha susamprajānam LV 27.19 (vs); vyādhimaranam na kupitarupavam LV 327.14 (vs; m.c. for "rūpavam = rūpavat; so we must read with Lefm. for mss. kupitam rupavas, or rūpavas, impossible in both meter and sense); srīmam Mv i.209.9 (end of line of vs; with gagaņam; = ii.11.16 where śrīmad). C1. iyam-duhkha, § 18.54, stem in composition.

18.61. Inst. sg. -ena: Mv (all prose): arcimena Mv i.227.4, 12; 231.8; gatimena smrtimena dhrtimena matimena i.229.9; tena yāyena i.232.6 'by him as he was golng', for yāyatā (Skt. yātā); so mss. (one jāyena). Senart em. yāyinā; gatimena smrtimena matimena dhrtimena dyutimena ii.133.10. — samprajānena Hoernle MR 5.1 fl. (prose), repeatedly; Hoernle wrongly says this stands for samprajāānena.

18.62. Gen. sg. -asya: arcimasya Mv i.200.1 (prose), and (vss) i.200.3; 201.5, 7; 221.18; bandhumasya Mv ii.271.7 (prose); samprajānasya Gv 202.2 (prose).

18.63. Loc. sg. fem. -āyām: bandhumāyām (= °matyām) Mv i.51.7 (prose); samprajānāyām Mv i.205.8 = il.9.4 (prose).

18.64. Nom. pl. masc. -as: samprajānāķ (no v.l.) Suv 9.14 (prose); smrtīmāķ (= smrtīmantaķ) Gv 476.22 (vs).

18.65. Nom.-acc. pl. nt. -āni, -ā, -ā: rajala-śrīmāņi (= śrīmanli) Mv il.36.6 (vs); vilatha dukha (so read m.c. for duhkha) mahā LV 362.7 (vs; read thus as three separate words, text as one cpd.), something like 'falsities and great evils', object of tyaji; for Skt. vilathāni duhkhāni mahānti.

Substitution of n for nt stem forms

18.66. There is some, the scanty, evidence for this substitution. Such forms may be analogical to the nom.

sg. in $-\bar{\alpha}$ (lacking final nasal) which is regular for *nt*-stems in Pali (cf. § 18.77); and/or to the composition stemform in $-\alpha$ (above, § 18.52 f.); cf. garutma-stham below. Of printed texts available to me, I have found forms of this type only in Mmk, in Chapters 40 (latter part) and 41, from pages 455 to 469. Here occurs the stem-form in composition garutma-(stham) 456.6 (vs); and the nom. sg. garutmā 460.6 and 11 (prose); these are the two possible sources for *n*-stem forms that have occurred to me. There are further inst. sg. garutmanā 455.13 (vs), dat. sg. garutmane 467.13 (vs); and gen. pl. garutmanām 460.9 (prose).

18.67. Attention may also be called to a variant at Mv iii.231.1 (prose), where Senart reads in the text bandhumas, nom. sg. (for Skt. bandhumān, or—since *t*- follows—^omāŋıs), with one ms.; but the other ms. reads bandhumano, which, if original, could only be nom. sg. masc. of an *n*-stem extended by -a (cf. § 17.24).

Strong stem used for the weak and vice versa

18.68. Inst. sg. -ntā: ajānanlā Suv 25.5; aprajānanlā Suv 26.1, both vss; reading uncertain but seems to me (following Nobel) probable.

18.69. Gen. sg. -ntah: bhagavanlah (but WT °valah with ms. K') śākyamuneh SP 390.5 (prose); (?) mahānto Mv ii.137.11 (v.l.; Senart with one ms. mahalo); in LV 396.7 (prose) and fl., read almost certainly talhāgalasyārhantah, repeatedly, with most mss. including the best, for Lefm. and Calc. °rhatah; mahaduḥkham paħcaskandham dharanto LV 191.7 (vs), either 'a great misery is the 5-skandha group for one who bears (it)' or 'there is great misery for one who bears the 5-skandha group'. Tib. seems to support the general sense: hon kyan phun po lha hdzin sdug bshal che yin na.

18.70. Nom. pl. -tas, -to, from weak stem: LV (all vss); kalhayato LV 47.11; bhāsato 74.14; vicarato 114.11; (?) praharsayato 415.3 (seems to mean 'causing dėlight', agreeing with preceding nouns, subjects of lyakta = tyaktāḥ). — āvrmhato Mv i.13.2 (vs), so read with mss.; nom. pl. with -patrā (preceding line), which to be sure is normally neuter; read also with mss. -mānsa-rudhiram pā; render 'tearing off the very skin, flesh, and blood, would remove them'. — dhāvatah Divy 218.23; 219.1, 9 (all prose), '(are) running', a sort of periphrastic present for dhāvati. — prabhāsatah RP 5.12 (vs); prayujyato 27.17 (vs). — vikurvayato (= vikurvantah) Gv 267.3 (vs). 18.71. Nom. pl. -ta, m.c.: yāvala (= yāvantah)

Bhad 1, 14, 15, 58. 18.72. Acc. pl. -ntas, -nto, from strong stem:

18.72. Acc. pl. -ntas, -nto, from strong stem: (tān...)buddhā (Lefm. em. buddhān) bhagavanto'nusmaranti sma LV 4.8 (prose); iyanto bodhisaltvān 31.6 (prose); bruvantah 197.16 (vs); rodantah krandantah 302.4 (prose). — viruvantah RP 55.7 (vs). — kāmšeid abhisfavantah Gv 519.22 (prose).

18.73. Acc. pl. -nta, -ntu, in verses m.c.: susyanta ... sthapeyam SP 128.2; sīdanta ... uddharisye RP 45.3; kurvantu pūjā (read pūja m.c.; acc. sg.) LV 222.15.

18.74. Gen. pl. -ntam, MIndic for -ntam = -tam: lesām dāni āgacchantam (Senart em. āgacchatām) sabdam karonti Mv i.18.10 (prose).

Miscellaneous non-Sanskritic forms

18.75. Nom. sg. masc. Here we shall not record spellings with final anusvāra for n, like kathayam for kathayan LV 157.19, galīmām (so all mss.) for galīmān (i mc.) LV 145.11 (ed. em. gatīnām; read gajendrah with ms. A). This I regard as purely orthographic; it is, in any case, a regular way, in our mss., of representing a final nasal, particularly before a consonant.

18.76. We might anticipate confusion between nom. sg. forms in -an, from the pres. pple. type in *ant*, with

18.89]

those in $-\bar{a}n$, from mant and vant stems. But actually such confusion seems hardly to occur. I have noted only one or two doubtful cases: (at end of a verse) arahān LV 326.17, according to Lefm.; some mss. read arhān, the others (with Calc.) arhan; bhagavam (mss., Senart en. ° νdm), n. sg., Mv iii.65.1; cf. § 25.7.

18.77. It might also have been expected that we should find many nom. sg. forms in $-\bar{a}$, since that is the regular Pali ending of mant and vant stems (so far as they do not have -anlo), and is found also in participial stems (Geiger 96, 98). It occurs also, but only to a very limited extent, in Pkt. (Pischel 398 end, AMg. arahā, arihā; M. Hanumā). In our dialect, however, such forms hardly occur. I can cite only mahā for mahān and garutmā for ^omān. The former seems probably to occur LV 193.9 (tho one excellent ms. reads mahām), and Mv i.48.8 = 82.4; 167.4; ii.70.12 (all vss). The latter has been noted only in Mmk in the passage where an *n*-stem garutman is otherwise recorded, perhaps as a direct result of analogy from the nom. garutmā; see § 18.66.

18.78. Once, in *dvisad-icchati* Ud xi.10 (vs), the form *dvisad* seems to occur as masc. Instead of regarding it as a nt. form transferred to the masc., we should probably regard *d* as samdhi-consonant; see § 4.64. If so the form is *dvisa* and belongs in the next paragraph.

18.79. Chiefly in verses, m.c., we find nom. sg. forms in -a, which may be regarded as for Skt. -an (tho they might also represent the quasi-MIndic -as, -o which we have met above, § 18.55). samprajānau LV 135.20, both edd., but all of Lefmann's miss. samprajānam or (A, probably the best) samprajāna, which latter fits the meter and must certainly be read (no ins. has 'onu); lāda 341.13 'beating', = lād(ay)an; bhava 366.8, 22; 367.3 (so with v.l. for text lava); 369.1, for bhavan 'your worship', all m.c. — namasya Mv iii.430.11 (vs; m.c. for 'syan); and (the only apparent occurrence of such a form in prose) samprajāna ii.132.2, mss. (Senart em. 'nam, for 'nan; perhaps rightly? but cf. voc. in -a § 18.82). — akarşaya RP 52.9, supposedly m.c. for dkarşayan, but see § 3.28.

18.80. Nom. sg. fem.: once the masc. form in "mām (= "mān) is clearly used as fem.: Sujāta (m.c. for "tā) matimām LV 271.21 (vs; no v.l.; so also Calc. glossing matimati). 18.81. Voc. sg. masc.: the nom. ending -an (or -an) is used as voc.: *irimān* LV 12.6; 13.2 (both vss; all mss.; only Calc. ^oman). — bhagavāņ bhagavāņ Mv i.245.13 (prose; so, or ^ovān, mss. seem to intend; Senart em. ^ovaņ ^ovaņ); pāpīmāņ ii.238.16 (vs; so mss.; Senart em. ^omaņ).

18.82. The voc. ending -a, which may be regarded as the nom. (above) used as voc., or as based on the stem in -a (§§ 18.52 ff.), occurs in maha LV 170.19 (vs). This is certainly not m.c.; it occurs at the end of a line, in a meter where a long syllable is usually found in that position. Yet all mss. apparently read maha.

18.83. Acc. sg. masc. -nta and -ntu, m.c. for °ntam, in vss: mahānta (all mss.; KN em. mahanta, kept by WT) varşam SP 8.12. — tişthantu SP 323.14; dhyāyantu LV 132.14; jvalantu I.V 241.10.

18.84. Inst. sg. -ta, m.c. for -tā: hitavala (so read, with Tib. phan dan ldan pa khyod kyis; mss. hita-lava; Lefm. em. hita bhava) tvayi (inst.) maru- (Lefm. em. wrongly naru)-slhalapathi slhapilo LV 166.9.

18.85. Gen. sg. -tu, m.c. for *-las*, *-lo: varşalu* SP 130.9; *nişkramatu* LV 115.22; *bhavatu* LV 366.3. In Divy 214.4 (prose) is printed the monstrous form *kārayaluļi*, which I can only suppose to be a misprint or error of mss., probably for *kārayatah*, or less likely for *kārayituh* from a *tr*-stem.

18.86. Nom. pl. masc. -nta, in vss, m.c. for -ntah: smrtimanta SP 68.5; 97.7 (WT °ntu with ms. K'); 131.3; balavanta SP 89.7; bhāsayanta RP 12.5.

18.87. Nom. pl. -nte, 'Māgadhism', in bhavanle (?): uncertain. The mss. read bhavanle at Mv i.152.5 (except one, an inferior one, ^{o}nlo); but Senart reads bhavanlo in accord with the mss. at i.225.21 and ii.29.2 where the line is repeated; and perhaps he is right.

18.88. Nom. pl. masc. -vān, i. e. nom. sg. used as pl.: pratibhānavān sada bhavanli... RP 15.2 (vs). The entire vs has only pl. verbs, nouns, and pronouns. So also mss. Mv i.133.11 (prose; Senart em. ^ovantās).

18.89. Gen. pl. -tu for $-t\bar{a}m$, m.c.: preksalu nayanān na cāsti trptim LV 49.16 (vs) 'and there is no satiating of the eyes as they look'. See § 8.124.

(106)

19.1. eka-, 'one', see under pronominal adjectives. 19.2. dva-, dvi- etc., 'two'. This numeral has the following non-Skt. forms. Plural forms (adjectives and verbs) often accompany the numeral (§§ 25.2, 3).

19.3. Nom.-acc. all genders, due, due (= Pali, both; Pkt. duve, be), and (m.c.) dvi (Ap. bi, Pischel 437). Only noted in Mv, except trimia-duve Gv 406.15 (vs). — duve, masc.: (prose) Mv i.38.1 ff.; 44.5; 359.18; 362.15; 363.6; ii.101.18; iii.26.20; 163.15; 176.8; 415.9; (vss) i.122.14, 16; fem. i.222.12 (here mss. iva) = ii.24.20 (vs); in ii.23.6 (vs) read dvi, with i.221.1 (same vs), for duve, metr. impossible; nt. (prose) i.238.3; (vss) i.116.17; ii.44.18; 169.12. — dve, masc. (prose): Mv i.231.19; 361.7; 362.1 — dvi, only in vss, m.c.: fem. dvi vāridhārā Mv i.221.1 (repeated ii.23.6 with unmetr. duve).

19.4. Inst. dvibhis, dvihi, dvihi (= Pali id.), duvehi (dvehi; this is merely a normal plural form to duve, dve; cf. Pali ubhayehi, Geiger 114; Pkt. Ś. duvehim, Pischel 437; Asoka duvehi, Hultzsch cxix, Pillar Edicts); dubhi, duhi (Ap. duhi, Jacobi, San. 15). — dvibhis: LV 240.13 (vs); SP 330.7 (vs); Mv i.277.21 (vs; mss. dvibhūs!). — dvihi: (prose) Mv i.25.6; 51.1 (v.l. dvehi); 79.10; 195.5; (mss. dvehi dvihi 196.4, Senart em. dvihi dvihi;) 197.12; iii.227.14; 228.15; (vs) 247.16 dvihi (-m- eva). — dvihi: (prose) Mv i.233.6 (mss. dvipihi, em. Senart); ii.1.3 (repetition of i.197. 12, where dvihi); (vs) iii.187.13 (i not m.c.). — duvehi Mv i.362.3 (prose). — dvehi, see above under dvihi. dubhi, Mv i.127.3, 4, 8, acc. to Senart; in 3 only one ms. duhi, but in 4 and 8 most mss. duhi, only one dubhi.

19.5. Gen. pl. duvinnam, dvinnām (= Pali duvinnam, dvinnam); dvinnā? — duvinnam Mv ii.49.18 (vs). Genuinely MIndic form, with short a in final syllable. — dvinnām Mv i.195.10 (prose); mss. dvinnā; repeated four times in lines 11-13, the final m being recorded only once in one ms.; i.249.7 (prose); iii.228.2 ff. (prose, several cases; one ms. regularly lacks m at the end, both mss. lack it once); 370.9 (vs). Should we recognize dvinnā as a genuine form?

19.6. Further, dve- is prefixed to other numerals in a multiplicative sense (as in Pali, see Childers s.v. duve, dve; e. g. doe-caturanga 'twice fourfold' = 'eightfold'): dvepañcayojanikāni Mv 1.42.2 'of twice five yojanas' (contrast dvikrofikāni in preceding line); dvecaturašili (once duve^o) Mv 1.259.8; 261.15; 271.15 (here duve^o; all prose) 'twice 84', as a large number.

19.7. As in Pali, both ubhe (m.c. ubhi) and ubhaye occur as nom. and acc. masc., in analogy with duve, doe: ubhi, m.c. for ubhe (which one ms. reads), Suv 52.9 (acc.). — ubhaye, nom., Mv ii.119.2 (corresponding passage LV 239.11 has ubhāv api); 171.16 (plural verb).

19.8. tri- etc., 'three'. The stem *tisra*-, based on the fem. *tisr*- (cf. § 13.15), occurs in composition with a masc. noun, perhaps m.c.: *tisra-loki* (=*triloke*) [LV 54.8 (vs); *tisra-lokesu* 56.5 (vs).

19.9. The nom. masc. trayo is used as acc.: Mv ii.115.9; iii.428.9 (both prose). The nom.-acc. nt. may be trini m.c.: SP 33.4.

19.10. In the inst., trihi and trihi occur for tribhis; the mss. often vary between the two (cf. Pali lihi, Pkt. lihim and lihim, the latter m.c.? Pischel 438): trihi, v.l. trīhi, Mv i.79.12 (prose), in same line with tribhir, v.l. trībhir; iii.249.7 (vs, but meter indifferent as to i). — trīhi: Mv i.238.4 (prose, v.l. trihi); 321.18 (prose); ii.40.21 (vs); iii.188.4 (vs); 249.10 (vs, meter indifferent, v.l. trihi).

19.11. Further, trayebhi occurs, LV 177.1, 13 (vss; cited Šikş 240.1; 241.1), apparently as numeral = tribhis; it is an analogical creation to Skt. gen. pl. trayānām.

19.12. In this, as in other numerals, there seems to be exceptional confusion of the originally distinct forms of the three genders; see §§ 6.11 fl.

19.13. catur- etc., 'four'. As in Pali and Pkt., the stem appears as *calu*- in composition, and not only in verses: *calu-tāta*- (no v.l.) Mv ii.313.5 (prose); *calu-varņa* LV 197.2 (vs). Cf. *caluhi*, *caluşu* below.

19.14. As in other numerals, gender-forms are confused; §§ 6.11 ff.

19.15. For the nom., the acc. cature (°ras) is often used (as in MIndic); also cature (so-called 'Mågadhism', or originally pronominal ending?), and caturi which represents the latter in verses m.c. — caturo (all vss) LV 12.20; 236.13; 372.2 (caturaí ca); 373.13; 385.7; Mv i.114.16; 157.6 (caturaí ca); 208.11 (repeated ii.10.19 where mss. catvāro, unmetrical; Senart em.); RP 24.16; 55.9. cature (vss) LV 343.9; Mv ii.92.11 (mss.); cature padāh Ud xii.4 (: Pali caturo padā, Dhp. 273; of course not to be emended with Chakravarti to catvārah, which in fact is metrically impossible). — caturi (vss, m.c.) LV 50.5; 80.20; 81.9; 197.1; Mv i.218.6 = ii.20.5.

19.16. Conversely, calvārah (and m.c. $^{\circ}ra$) is used as acc.: calvāro dvīpām Mv i.80.8 (prose); i.168.10 (vs); calvāra dvīpa susamŗddhā ... āvasati sarvān (2d ed. sarvā) Gv 254.19 (vs; probably to be divided thus; ed. calvāra-dvīpa- as one word, which is conceivable, calvāra- as stem in comp.).

19.17. As in the nom., calure also occurs in the acc.: disām cālokya calure LV 252.21 (vs; with fem. noun, tho it has a masc. acc. pl. ending).

19.18. For inst. caturbhis, there occurs catuhi (so Pali), MIndic form based on stem catu- (above, and in loc. catusu below): Mv i.239.11 (prose; Senart em. caturhi but mss. catuhi). Commoner are caturhi and caturbhi, especially the former. caturhi Mv i.3.12 (v.l. catuhi); 6.2; 38.15; 120.8; 120.13 (v.l. caturbhir); 230.12; 237.10; 263.17; 318.17; 335.14; ii.33.5; 141.4; 164.17; 423.9 (fem.); all prose. — caturbhi Mv i.120.7; 292.15 (both v.l. caturhi).

19.19. For gen. caturnām, caturna is read in vss m.c.: SP 237.13; 294.6 (with fem. noun, parsāņa).

19.20. For loc. catursu, we find catusu LV 418.13 and 17 (prose; cf. Weller 38; § 1.46); and read so Mv ii.141.5 (prose), with v.l., for Senart caturusu (unparalleled).

19.21. In LV 130.19 (vs) Lefmann's reading *caluşū* is doubtless correct (\bar{u} m.c.), tho most mss. and Calc. read *calurşu*, unmetrically.

19.22. pañca, 'five'. For inst. Skt. pañcabhis, puñcahi is fairly common in Mv: i.31.5; ii.25.14 (both prose) etc. It is doubtful whether any other non-Skt. form really occurs. In Mv i.60.14 Senart prints pañcehi, but 3 mss. have pañcahi. Cf. however -ehi forms § 19.28. In ii.257.7 mss. pañcāhi or pañcabhi; Senart em. pañcahi.

19.23. The loc. pl. appears m.c. as pañcasū RP 59.6.

19.24. şaş, 'six'. The loc. pl. occurs as salsā m.c. SP 9.6; 48.3.—A strange form sadi occurs for 'six' twice in LV, apparently as 'stem' in composition. I cannot explain the final -i, unless it is analogical to the ending of dvi-, tri-. (Cf. Ašokan sadu-visadi, where u has been explained as analogical to catu-; Bloch, Asoka, 161 n. 3.) In LV 414.19 (vs) Lefm. prints sadi pāramite (cirarātru vivardhilu košn toayā); we must certainly divide pārami from te: 'the six supremacies are thine'. Then sadi could be either nom., or stem in composition (sadi-pārami). But in 420.1 (vs) only the latter seems possible. The ed. reads sadi-ningtiar nipatilo (ili sparša uktah), but this is metrically impossible; read sadi-indriyair with A (the best ms.), 'occurred by reason of (caused by) the six senses' (in the pratilya-samulpāda).

19.25. sapta, 'seven'. The inst. pl. saplahi is not rare in Mv: i.110.5; 128.9 (v.l. saplehi); 194.2. 4; 323.13; iii.92.14 (all prose). Gen. saplanām, Crit. App. ou Mv ii.74.8, but saplānām in text 75.21.

19.26. daśa, 'ten'. Several times the text of LV seems to present daśah for daśa, always after diśa, as acc., 'into (all) ten directions' or the like. It occurs at the end of a line of verse LV 311.4 and 387.16 (no v.l. cited); but in prose, followed by prapalāyamānam, in 301.10, where Lefm. to be sure reads daśa with one inferior ms., while all the other mss. and Calc. have daśah. The Tib. confirms the numeral in 387.16; phyogs (= diś) bcu (= 10). But in the two other places it reads phyogs phyogs = diśo-diśah, 'in all directions' (as also in LV 314.8 where text diśo daiāsu). I cannot explain the form daśah and suspect that it is only an error for (diśo-) diśah, tho apparently in one passage (387.16) this error crept into the text before the Tib. translation was made.

19.27. Otherwise we note, as usual, inst. pl. daśuhi, Mv i.148.1 (prose); and loc. pl. daśasū, m.c., SP 53.11; 342.12; Suv 42.8 (Nobel $^{\circ}sn$ with his mss., but Śiks 219.6 cites the verse correctly with $^{\circ}s\overline{u}$).

19.28. Teens. We have met above, §§ 19.22, 25 (and cf. trayebhi § 19.11), inst. pl. forms in -ehi, at least as variants in mss. Some of the teens show similar forms, better attested: dvādaśchi Mv i.196.21 (prose). — tridaśchi id. i.204.14 = ii.8.9 (vs). And sodaśchi occurs as v.l. for sodaśahi of Senart's text in Mv i.198.13 = ii.2.15; ii.179.7 (all prose). If some of these forms in -ehi are genuine, as seems likely, they are casily explained as analogical to the regular Skt. gens. pl. trayānām, pañcūnām, suptānām, dašānām, etc., which look like normal a-stem forms; hence we need not find surprising the creation of inst. plurals which, in this dialect, are normal in a-stems.

19.29. Tens. As in Epic Skt. (Whitney 475e), the endings -at and -ati are interchanged, c. g. vimśac ca (for vimśali) SP 114.12 (vs); trimśali (v.l. trimśam; for trimśal) Mv ii.132.18 (prose); pañcāśali SP 95.1 (vs).

19.30. Further, any of the tens from 20 to 50 inclusive, whether ending regularly in -a or -ai, may be replaced by a stem in -a, which may be used in that form (apparently) as an independent word (but sometimes, possibly always, this is to be construed as prior member of a cpd.); or it may be treated as an *a*-stem, appearing as nt. nom. -an (perhaps then to be identified with the collectives of this form, Wackernagel III p. 419), or as nom. pl. -ah. Thus:

19.31. Forms in -a: putrāh priyā orasā viņša (m.c. for viņšat or viņšam?) bālāh SP 88.3 (vs); triņša koļyo (cpd.? or read triņšat with 3 mss.?) Mv i.174.14 (vs, but metr. indifferent); triņša-gosthikā Mv iii.375.13 fl. (prose) '30 companions', repeatedly; printed as cpd. by Senart, who in 376.15 adopts the v.l. triņšad-go°; paīcāša koļīyo (cpd.?) Mv i.174.16 (vs, but metr. indifferent); in Mv i.117.15, at end of line of vs, text pañcāśaņ but v.l. pañcāśa.

19.32. Forms in -am: viņšam, triņšam My ii.284.9 (prose); paācāšam My i.115.13 (vs); 117.15 (vs, cf. above); ii.133.1; 284.10 (both prose; one ms. has calvāriņšam before paācāšam in the last passage).

19.33. Forms in -āh: calure ca viņšā(h) LV 343.9 (vs), '24'.

19.34. Far rarer, indeed hardly to be found, is the addition of -a to such a stem as *lrimsal*; it seems to be implied by *dvātrimsalākāra*, 'of 32 forms' (*'sala-ākāra*) My ii.278.21; 279.2; 280.12. The only alternative would seem to be emendation to *'sad-ākāraŋ*: Senart notes no v.l.

19.35. The compounds calurāšili, 84, and şadāšili, 86, appear in this form, with lengthened \hat{a} (see Dict.), which cannot be m.c.; the forms occur in prose and in metrically indifferent positions, as likewise in Pali calurāsīti, chalāsili; cf. also AMg. caurāsītiņ (Pischel 446). Evidently the forms are due to the analogical influence of compounds in which the prior numeral ended in a_{∞} producing $-\hat{a}$ - by samdhi with the initial a-, such as paūčašili, saplāšīti, asglāšīti. — We even find calvāri āšili for 84 in Dbh.g. 44(70).25; but in this broken form \hat{a} - may possibly be m.c.

19.36. Ordinals may be formed, as in Pali and Pkt. (Geiger 118.2; Pischel 449) by adding -mu to the cardinal: ekanavatime '91st' Mv ii.271.5 (prose). Tho not standard Skt., such formations occur in epic and 'late' Skt., Wackernagel III p. 409. A blend of *paācāšatima (Pali paātāšatima, Geiger 118) with Skt. paācāša seems to be paācāšima, 'fiftieth', SP 351.1 (vs). Analogical to such forms in (-i)-ma are šatima, '100th', and sahasrima, '1000th' (both Pali; Dict.).

Nominative forms with other cases, and in composition

19.37. Whitney 486c notes the use of pañca, sapla, śałam, sahasam with nouns in any case. Renou, Gr. scte. p. 385, cites nom. forms of numerals in *-li* sporadically used with other cases (e. g. safsasfir gält, \overline{ApSS}). Since this usage is hardly normal Skt., I cite a few similar cases in our dialect: ašilis tälän uccaistvena LV 278.13 (prose), '80 tal-trees in height'. — viņšatiļt kalpakoļyo (adverbial acc. of time) Samādh 8.16 (prose), 'for 20 crores of kalpas', and: viņšatif ca buddhakoļir ārāgayām āsa, ib., 'and propitiated 20 crores of Buddhas'; kalpāna koļya paripūrņa viņšatif 8.29 (vs; so 2 mss., text viņšati), 'for 20 full crores of kalpas'. — saştiš ca kalpān Suv 44.6 (vs); viņšatif gajān 188.3, 6, 10 (prose; v.l., to be sure, viņšati).

19.38. Such nom. forms in -tir are even used as prior members of cpds., probably in analogy with cpd. numerals like trayo-dasa, trayastrimsal (cf. also satam and sahasram used in comp., Wackernagel II.1 p. 47): asitir-yojanāvrtam LV 280.16 (vs; read so, or °āyatam, with mss., for the unmetrical oavrtlam of Lefm.). Less clear is the construction in LV 413.3 (prose), vipulo vistīrņaķ saptayojana-šatānyāyāmo (so Lefin.) vistāreņa. If this text is correct, -salāny is compounded with -āyāmo: 'immense, extensive, having a length of 700 yojanas in extent'. But several good mss. read ayama-(vistarena); with this reading -salany would be a separate word, '700 yojanas in length and breadth'. Since āyāma 'length' is often contrasted with vistāra 'breadth', this seems likely to be right. Cf. asilir-yojanany äyämena vislärena LV 307.14 (prose), which seems to mean 'eighty yojanas in length and breadth'; here asilir may be either nom. of a separate word, or compounded with yojanani; but yojanani seems clearly not compounded with ayamena.

20. Personal pronouns

Stem in composition

20.1. Once, in LV 364.10 (vs), occurs either $tv\bar{a}$ - or tvdm- (so the mss. that have the syllable) compounded with -*pranidhi*; for so obviously we must interpret the text, the Lefm. prints $tv\bar{a}$ as separate word. An extension of the use of $tv\bar{a}$ - $dr\bar{s}$ etc.?

20.2. More curious is *lvam*-, *tvam*- in composition: *tvamsādršakehi* SP 31.11 (vs) 'like thee'; *tvamīdršair* LV 325.13 (vs) 'with (beings) like thee'. Is this analogically based on *aham*, used rather freely in Skt. as prior member of compounds? (Wackernagel III p. 437, cf. 456).

20.3. Purely phonetic is the occasional loss of the final *-l* of the regular Skt. 'stems' mat-, asmat-: asma-matena (for asman- m°) LV 338.13 (vs). — mu-dharmam (ms.) RP 17.15 (vs), Finot em. ma-dh° (for mad-dh°); it is said that the Chinese supports this in meaning; cf. § 20.56.

20.4. The 'stem' in -l, -d is once analogically affixed (before the suffix -iya) to *ālma*-, forming an adjective *ālmadiya*, '(one's) own', on the model of *asmadiya*, yuşmadiya, etc. (Dict.).

20.5. We shall present the forms noted in the usual order of cases, first singular and then plural. Under each case we shall cite first all 1st person forms noted, then all 2d person forms.

Nominative singular

20.6. First person. Only phonetic variants of aham have been noted in the 1st person, viz. ham, ham, ahu, aha, hu, ha; all in verses only, not in prose. Pischel 417 records ham in Pkt. only after final vowels; in our texts it is also recorded several times after anusvāra. Both (a)ha and (a)hu show phonetic results of final -am which are common in our dialect; they are not recorded in Pischel (cf. Ap. hau from *ahakam).

20.7. ham, ham: after vowels, imena ham SP 28.3; em eva ham (with most mss. and WT) SP 89.11; ca ham SP 324.5; sirī ham Mv ii.57.2; va $(= v\bar{a})$ ham Gv 489.10.

After anusvāra: *lesām ham* Mv i.10.7 (so with 3 mss.; Senart with 3 mss. *lesām aham*, metrically poor); *dārikām* ham Gv 413.23; campāyām ham samutpannah Lank 364.15.

ahu, common: SP 11.13; 62.15; 92.12; 97.5; 99.5; 324.4; LV 78.7; 79.21; 119.5; 327.15 tāv' (= tāvat) ahu; 393.9; Suv 52.6; RP 27.5, 18; Bhad 1, 4, 6², 8, 9, 10, 12, 16, 18, 33, 41, 59, 60.

aha, not common, much rarer than ahu; sometimes as v.l. of mss. where edd. have ahu, above: ahu LV 78.9, so ed. without ms. support; read aha with ms. A; others aham, unmetr. Here may also be counted such cases as dade 'ha dānam Mv iii.46.14 (so divide; probably 'ha does not represent iha, cf. larpaye 'ham in next line); sihilvāha Suv 248.13; māha Sukh 22.5.

hu, after final -a; may also be considered examples of ahu with elision of preceding final -a: putrăna hu (v.l. ha) SP 90.9; yena hu SP 195.5; caramāna hu (= °māno 'ham) RP 26.7.

ha (cf. under aha, above), after vowels: yānāni ha sampradāsye SP 88.10; bodhāv upāyena ha (WT with v.l. and Tib. °yen' iha) sarvi nemi SP 195.4.

20.8. Second person. This is more varied. Besides

tva, m.c. for tvam (rare: sa tva mañjuśiri precha sūdhana Gy 488.23, 'do thou, O Sudhana, ask Mañjuśri'; ed. prints sattvamañjusiri, one word), we find: (1) tuvam, a form well known in Pali, and authorized by grammarians (but not known to Pischel, 420, from literature) in Prakrit; this is fairly common in Mv, prose as well as verse; also its m.c. equivalent tuva; (2) lumam, the regular Prakrit form (Pischel 421), or its m.c. equivalent luma; but these are very rare; (3) tuham (modelled on aham), only once, SP 67.11 (vs); Pischel 421 records it only for Dhakki; eastern Pkt. grammarians attribute it to Ap., see Jacobi, San. p. xxv, and cf. Alsdorf, Kum. p. 59 (it is written in Kum. 5 times instead of regular tuhū); (4) tvayi, thrice as nom. in the same context of LV; cf. Ap. tai, pai, acc. as well as inst. (so also tvayi in BHS) and loc., and as nom. according to Pingala 1.5b (ap. Pischel 421); any acc. is likely to be used as nom. in MIndic; (5) lvayā, apparently once as nom., probably by analogy with the preceding which serves also as inst. in BHS as well as in Ap., hence inst. *loayā* also as (acc. =) nom. In the Pkt. of the Kharoşthi Documents from Chinese Turkestan (Burrow, § 51), 'the inst. tends to be confused with the nom.' (which = acc.), even with nouns.

20.9. *tuvam*, in verses: Mv 1.180.18; 243.1; 305.10; ii.39.4 ff. (repeatedly); 52.18; 54.2; 71.13; 92.9; 226.11; 229.4; 235.12; 397.4, 21; iii.3.5, 7, 17; 4.2; 120.7; 295.3. And in prose: Mv 1.233.10; 302.6; 343.9; 364.14; ii.70.3; 216.4; 400.13 (note here *tvam pi* and *tuvam pi* in the same prose line); 427.4 f.; 428.2; 448.18; 480.4, 9; 493.10; iii.151.6; 404.11.

tuva, only m.c. in verses: SP 93.9; 1.V 343.2.

tumam, probably in Mv i.43.13 (vs), mss. cited as *buddhottamām*, Senart em. *buddho tuvam* (with the repetition i.243.1); but it is likely that *buddho tumam* was intended.

tuma, m.c. for tumam, in verses: LV 165.13 and ff., repeatedly; 168.15; Samadh p. 45 line 19.

luham, see above.

(108)

loayi (all vss of LV) laghu tvayi narapati vibhajahi amrtam 165.8, 'quickly do thou, Lord of Men, deal out nectar'; tvaji tvayi... dhanamanikanakā 165.9, 'thou hast given away...'; syamu puna rşisulu tvayi puri yad abhū 166.3, 'when, again, of old thou wast the rşi's son Syāma'.

loayā: toayā (2 mss. tayā; Nobel em. tuvanı) hi prahlādakari hitankari Suv 245.5 (vs). There can be no doubt of the meaning, nor, in my opinion, any serious doubt of the reading.

20.10. [According to Leimann's text in LV 253.8, *tvām*, the Skt. acc., would be used as nom.: *kaisā ati tvām karunam rudāsi*. Some mss. and Calc. read *kaisālīva karunam rudāte*, which is unmetrical. But Tib. supports a 3 pers. verb, with no 2 pers. subject. A reasonable and metrically correct reading would be *kaisā atīvā karunam rudāti*, or the like; and it seems improbable that there was any 2 pers. pronoun here.]

Accusative singular

20.11. First person. The only non-Skt. forms clearly established are original genitives (datives) — mama, mahyam, me, and phonetic variants of these — used as

accusatives. Pkt. (and occasionally even Skt., Wackernagel III p. 473) uses the enclitic *me* as acc.; both Pali and Pkt. have maman, a sort of blend of mama and mām. (Pischel 418 notes Mg. mama in verse, Mrcch. Stenzler 129.4; he would read mamā.) In our dialect, however. maman (except as emendation by editors) has been noted only in mamam ca, reading of WT with ms. K' for mamaine SP 325.4 (vs; both acc. sg. and equally acceptable). It may be guessed that me came first to be used for acc. as well as dat.gen. (perhaps by analogy with no, no == nas, nas, for all three cases); and that mama and mahyam then followed the analogy of me. **20.12.** mama: mama drstpa (so read with v.). for

20.12. mama: mama dṛṣļva (so read with v.l. for Lefin. dṛṣļi) sapāriṣadyam LV 304.15 (vs).--Mv prose: mama... adhyabhāṣeran Mv i.56.13, 'addressed me'; yadi mama bhāryām upādilyasi 233.7, 'if you take me as your wife'; mama erābhikāṅkṣati ii.69.9 (repeated 72.13; cf. bodhisatīvam erābhic 69.3, 5); muma palim varehi 70.4, 10, 'choose me as husband'; muma gṛhŋāhi 70.8; others, iii.174.9; 265.8 (here Senart em. mamanı); 291.10; mama uddiiya āgalo 351.9 'came back (with reference) to me'; verses i.143.5 (Senart em. mām, without metrical or other good reason); ii.71.5, 11, 19; 194.4. -- mā tāvan mamāpiechasi ghālagilum Divy 380.11 (prose), 'you don't want to kill me too, do you?' (but cf. § 7.72). -- mamāgamya Av i.321.13; ii.96.8 (both prose) 'thanks to me'; in both Speyer em. mām āgamya.

mamā, m.c. for mama, in verses: mamā hi paśyet SP 237.14. No v.l. given by KN; WT cm. mamam (their ms. K' reads mama); yasnin mama (meter requires long final; doubtless read mamā) praņayase tvam ihādya amba LV 119.4, 'to whom you, O mother, are conducting me here today'.

mahyam (?): haneyu mahyam LV 232.1 (vs), 'they might kill me'. Perhaps gen.; see § 7.72.

20.13. me: Mv prose: adrāksit me Mv i.56.10 (with acc. sg. āgacchantam in agreement): mahāprabhāvam (or other adjectives in repetitions) vala me samānam mā haiva me sramaņo gautamo abhibhavisyatīti ii.277.2 fī. (Māra speaks); 428.2; šāpena me saparivāram bhasmīkareya iii.158.6; 169.8; 180.11; 291.14; verses: tnayā ca me pravāreyā ii.88.11, so read (compare readings of mss. with the Pali Jāt. iii.284.4), 'he would gratify me with you' (the rest of the line is troublesome but for Senart's prattam read certainly yam ca [Mv mss. pamca]; after it possibly cayilam); mā ca viprajāhi me 142.18; 319.14; 484.5; iii.3.16; cf. Mv ii.119.2-3, § 7.69.

Weller 47 alleges two cases of me acc. in LV prose, but neither is convincing. In 396.1 me is dative: ahum ced idam dharmam deśageyam pare ca me na vibhāvayeyuh, 'if I taught this dharma, the rest would not understand (it) at my hands'. In 404.11 ed. me ha, but read meha = mā-lha.

mi, m.c. for me (cf. ti for te): so doubtless read (or me? § 3.64) for text me in Mv iii.366.17; meter requires mi as much as ti in line 15 (the question to which 17 is a response). In line 18 me also acc., but here meter is correct.

20.14. Second person. We find *tava*, *tubhyam*, and *le* (*li*), gen.-dat. forms, corresponding to the 1st person forms. But here we also find *tvam*, *tuvam*, and even (before vowels) *tvam*; the first two may be regarded as having MIndic short *a* for \tilde{a} in closed syllable; in the last this shortening may be due to metrical requirements, or it may be simply the nom. form used as acc.

20.15. tvam, before consonants: Mv i.175.14 and 15; iii.386.19 (all vss). In all Senart emends tvam to tvām. tvam, before vowels: aham tvam adhyeşāmi SP 38.2

(vs); lvam u grhyā LV 169.10 (vs). luvam: upādiyāmi (? mss. corrupt) luvam (so mss.; Senart em. lava, perhaps because of 234.8 upādiyāmi lava adya bhāryaram) bhāryām (mss. bhāryā, which may be kept) Mv i.233.11 (prose), 'I take thec as my wife'; Mv iii.122.2 (vs).

20.16. tava: Mv prose: tava āgamya Mv i.365.12, 'thanks to you'; nāham tava palim iccheya ii.70.6; edṛšam tava pūrvopakārim 180.1, 'such (as) you, a former benefactor'; tava... drastum 427.7; and verse: upādiyāmi tava adya bhāryaram i.234.8 (so read with v.l., see § 9.25), 'I take thee today as wife'.

tubhyam: LV (all verses): sarve tubhyam pājya šresthām karonti 195.10 (so divide), 'all, having honored you, make you (into) a noble person' (so Tib. clearly: kun gyis khyod mchod gtso mor byed par hgyur); but cf. § 7.69; te tubhyam na samartha bodhiviļapāc cāletu kampetu vā 283.21; similarly 360.18.

tubhya, in.c. for prec.: tubhya prekşami sayāni suplikām LV 236.3 (vs), 'I look on thee asleep on the couch'.

te: LV (all verses): jayāmi te LV 262.13; chinadmi te 338.14; vandima te 364.6; širasā vandima te lathāgatam 364.19 and 365.5 (note acc...noun agreeing with te)...-Mv (prose): vayam te... sammanyema i.348.2 (cf. 347.18 tam sammanyemah); aham te neşyāmi (in 4 nemi) ii.249.1 and 4; te... nihanisyāmi 270.2 fl. (repeatedly; cf. § 7.72); iii.169.7, 8; and verses: te vande śāntam te dāntam i.98.7; ii.484.2; iii.452.11.

li, m.c. for prec.: My iii.366.15 (vs).

Instrumental singular

20.17. First person. Besides maya in verses m.c. for mayā, we find, first, maye, which corresponds to regular Pkt. mae; it has been noted only in Mv, but is not rare there. Then mayi (also as acc.), as in Ap. (mai, mai; also Mg. mai, verse, Pischel 418). Also me as occasionally in Skt. (Wackernagel III p. 473), and more often in Pali and Pkt. (also m.c. mi); and (presumably by analogy with me) mahyam.

20.18. maya, in verses, m.c. for maya: LV 373.3, 10; 393.10, 11; Mv ii.55.16; 245.4 (so read with v.l., as required by meter; Senart maye); Gv 255.18 (read sa maya for samaya).

maye, only in Mv, but the following examples all from prose except ii.320.8, and all without v.l.: i.37.6; 60.13; 243.6; 286.18, 19, 20, 21; 288.8, 9, 11; 290.5, 6; 363.14; 365.3; ii.77.2, 3; 82.5; 241.11; 255.9; 257.3; 320.8 (vs); 420.3; iii.144.5.

mayi, noted only in verses (rare in Mv): mayi pūrva lyakla LV 210.15; mayi...dānu damo...sucīrņā LV 287.5; mayi bho krlābhyanujñā Mv i.151.13, having permission granted by me; mayi lyakla-m-anguli udārā RP 24.1; Bhad 8, 12, 61; Dbh.g. 11(347).18, read mayi with Susa's mss. (cited in note as mapi), for text me; pūjā krlā mayi jinasyo Gv 255.22.

20.19. me: na me śrutā vā drşļā vā sūci elādrśā (mss. °si, °šam) mayā Mv ii.88.17 (vs; note repetition of me by mayā); na me svayam drşlo... 110.12 (prose; response to question kim tvayā svayam drşlo...) 'I did not see him myself'; me... agāram adhyāvasatā iii.214.15 (prose; note inst. agreeing with me). — ajānantena me Suv 27.15 (vs). — nrpeņa satā me RP 22.12 (vs); caratā me 23.8 (vs); and others in this context. — me... āyūhalā niryūhalā etc. (inst. forms with me) Gv 69.23 fl. (prose).

mi, m.c. for prec.: mi bodhi prāptā Suv 153.16 (vs). mahyam: pūrve mahyam dānu dattam, 'in former time by me charity was given' LV 196.5 (vs); Tib. nas, inst.;

niryālilā...mahyam Suv 146.2 (vs) (Tib. has); mahyam... Iyaktam Suv 146.4 (vs). 20.20. Second person. Here all the preceding 1st

person forms are paralleled: toaya (m.c.); toaye (but this is very rare); toayi (also toayi); te (and m.c. ti, also tybefore vowel); and tubhyam (or m.c. tubhya). In addition, the form *beam* seems to be used as inst., as also as gen. (see below); tho both hardly occur except in one passage of LV, and there before the word *iha*, which raises a suspicion. Are we dealing with an obscure nonce creation of a single author, perhaps involving peculiar phonology (sandhi?)?

20.21. tvaya, m.c., verses: LV 178.19; 243.7; Mv ii.140.7.

tvaye: My iii.32,15 (vs).

tvayi (not noted in My; only vss): krta tvayi hilakara bahuguna I.V 165.1; 166.1, 9 (read in 9 hilavata, for cta, § 18,84); 167.20; tvayi caratā 168.5; (168.6, sec § 20.23:) 168.11; 169.17; 178.3, 7.

toayi, for prec.: bhotu me lughu samāyamas toayi Gv 489.14, 'coming together with thee'. At the end of a rathoddhatā line; can hardly be called m.c., since in the same passage other lines are allowed to end in a short vowel.

20.22. te: LV, no quite certain cases in prose: te...dafsanavisesah säksälkytah 409.4 (prose) is cited by Weller 47, but as Pischel observes 421, te depending on such participles may be considered gen. The same could be said of some of my examples following:—Verses: LV 169.11; 171.9, 11; ksipta harsitena te 171.16; 195.9 ff. (te 9 times); yaşlās te caratā ... 284.2; tyaktā (v. l. °tās) te... caratā 361.2. — mrgabhūlena te Mv i.365.11 (prose), 'by you who are a deer'; mama te jīvitam dattam sarvam te (sõ mss., Senart em. me) rājyam samšayālo mocitam ii.491.3-4, 'you have given me life; you have freed (my) whole kingdom from risk.' — te pralobhyamānena ... nolpādayitanyā Divy 115.8 (prose).

ti, and before vowel ty, m.c. for prec.: nihalā ti mārakarmā LV 12.10 (Tib. khyod kyis, 'by thee'); ti . . . kurnatā 172.15-16; ti 172.19 and 20; 184.1; yam ty upasthilu jage hilamkaro 237.15, 'in that the one who works weal in the world has been respectfully approached by thee'.

tubhyam: buddhaśatasahasra sampūjitā pūrvi tubhyam mune LV 365.13 (vs), 'hundreds of thousands of Buddhas have been honored by thee of old, 0 sage.'

tubhya, in vss, m.c. for prec.: LV 53.1, 5, 7, 9, 11, 13; 170.7; tubhya śişyamāņā 184.6, 'taught by thee'.

20.23. tvam (see § 20.20); only vss: krtu svaguno tvam ihā jagalo 'rthe dhyānaratenā LV 169.7, 'by thee thine own quality was made (perfected) for the sake of the world here, by thee delighting in meditation'; haladamasa (read 'sā with v.l., m.c.) tvam ihā krtu loke arthu mahanto 169.20; in 168.6 Lefm. with most mss. krtu tvam ihā (mss. iha)... śilaratena (Lefm. em. 'nāl); certainly an inst. is meant; but ms. A tvayi for tvam ihā. Since all mss. are corrupt (and metrically imperfect), the true reading remains conjectural.

Dative singular

20.24. In LV 232.6 (vs) lvaya(-m) may be taken as dat., or possibly loc.: premena mahyam lvaya-m-api vartisyante, 'they will behave with love towards you also (as) towards me.' The -m- is presumably 'hiatus-bridger'. In Skt. vrl- in this sense regularly takes a loc., but also dat. or acc.; mahyam here points to a dat. Since tubhyam and mahyam may be used as inst. (and in our language as in Ap. acc., inst., and loc. of personal pronouns seem confused—indeed it would hardly be going too far to say that all cases are confused), it is perhaps not rash to permit Skt. tvayā (in a form altered m.c.) to pass as dat.

Ablative singular

20.25. [In a line of verse repeated four times Mv ii.148.18; 149.2, 6, 14, Senart assumes mamātu = Pkt. mamāo, abl. sg. But a gen., not abl., is needed here. Probably read mamā (= mama, m.c.) tu ratir na bhavet, 'but I should find no pleasure'. In 149.14 one ms. reads mamātra;

if this reading were adopted the sense would be, 'I should find no pleasure therein (atrai).] With unlengthened -a before -to: manato Mv ii.272.7 (prose; one ms. mama); so Asokan mamate (Hultzsch evi).

Genitive singular

20.26. The use of dative forms as genitives is not only standard Mindic, but occurs often enough in some Skt. texts (especially epic; cf. Wackernagel 111 p. 460). In our dialect it is very common, particularly in the personal pronouns, and hardly needs extensive illustration; but we shall cite a few examples of *mahyam* and *lubhyam* which show this construction very clearly.

20.27. First person. Otherwise, we find mahya for mahyam (I have noted it only in verses, but doubt whether it can be called strictly 'm.c.', since it occurs in some situations where mahyam would be as good metrically); mamā and mi, which occur only or chiefly m.c. for mama (tho mamā occurs in mss. Mv i.362.12. prose, and in Aśoka's inscriptions, both Kalsi and Dhauli-Jaugada, Hultzsch lxxviii and cvi) and me respectively; and—barely possibly, but very rarely if at all-mamam, on which see below.

20.28. mahyam (see also mahya, below) as gen.: mahyam bhāşato LV 78.4 (vs); jātasya mahyam (gen. abs.) 118.21 (vs). — mahyam rahogatasya Mv 1.51.8 (prose); mahyam pitā 178.17 (vs), 'my father'; prşihato mahyam ii.104.17 (vs), 'atter me'.

mahya (only in vss, but not invariably demanded, instead of mahyam, by mcter): mahya imasmi SP 25.15 (note that mahyam would be quite as good metrically); 61.7; 62.1; 86.8; mahya... aputrakasya 87.4; mahya ime 93.9; 197.6; 203.4; 237.1 (metr. indifferent). — mahya LV 159.8; (mss. mahyam, Lefm. em. m.c.)... äirilasya 189.7. — Mv i.186.7; 277.12 = 279.21 = 281.19; ii.39.21; 40.1, 3; 56.3, 5; 61.11; 133.19; 204.17; 226.16; iii.122.13. — Suv 51.5; 249.7. — mahyadam Murcanam Mmk 104.24.

20.29. mamā: SP 97.4; 55.8 and 220.9, in both of which mamā must be read with Nep. mss. and WT, against KN with Kashgar rec. mama (unmetrical); LV 42.17; 119.8; Mv i.48.9, Senart prints mamā at the end of a line, without report of v.l.; the repetition i.82.5 reads mayā, and inst. seems better than gen.; is mamā at i.48.9 a mere misprint? See also under mamam below.—Suv 52.2; Gv 54.2; 214.17, etc.

mi, only in vss m.c.: SP 31.8; 35.8; 63.3; 64.5; 152.3; 236.7. — LV 38.10; sā mi kuruta 79.22 (ed.) or (ms. A) sā ca kuru mi: 132.14; 189.6; 195.3. — Mv ii.5.9 (in parallel i.201.14 mss. me, unmetr.: Senart emends another word); ii.140.12. — Sukh 22.3, 7, 11.

20.30. mamam? Tho this is familiar in Pkt. beside mama, it can hardly be said certainly to occur in our texts. In Mv i.48.7 and 223.16 Senart introduces mamam, m.c., but the mss. have mama, and a better em. would be mamā. On the other hand, in Mv ii.273.4 (prose) Senart reads mama *li*, where the mss. seem to point to mamam *ti* (mamanti), at least as a possibly original reading. I have not recorded another; nor is mamam established as acc. in our dialect.

20.31. Second person. Here we find *lubhyan* and *lubhya*, *lavā*, and *li* (ly) corresponding to mahya(m), mamā, and mi. We also find, in Mv only, *luhyam* and *luhya* (cf. Pali *tuyham*, and similar forms in Pkt.), analogical to mahya(m). Further, there is one probable case of *luma* (analogical to mama; cited only by grammarians for Pkt., but cf. AMg. *lumam*, Pischel 420 f.); and several apparent cases of *lvam* as gen. (cf. *tvam* as inst.; mostly in the same LV context as the latter, but once *lvam* in Mv according to the mss.).

20.32. tubhyam, tubhyam: tubhyam antike SP 212.7 (vs), 'in thy presence'; LV 169.4; 189.1 (both vss).

tubhya, only vss, but not always m.c.: SP 274.6; *tubhya* (WT with K' *tava*) *eşa* LV 221.20 (not m.c.); LV 223.1; 367.17; Mv i.156.20; *tubhya asti* Mv ii.321.3 (not m.c.).

tuhyam, only in Mv: i.226.4 = ii.29.7 (in the latter v.l. tubhyam); ii.4.11 (v.l. tubhyam) = i.200.14, where Senart em. tubhyam, read tuhyam (mss. bhujyam, guhyam); i.252.1; 277.9; 326.10 (prose; v.l. tuhya; before eva, but tuhyam, not "yam); ii.92.22; 105.9; iii.45.12; 346.5; 451.11 (twice; v.l. tuhye, tuhya).

tuhya (cf. under tuhyam; chiefly m.c.), only in Mv vss: i.166.5; 255.20; tuhyānukampāya ii.52.20 (= Jāt. v. 391.20 tav' eva atthāya); 57.4, 20; 60.1; 184.13 (mss. tujya; jya and hya are hardly distinguished, Senart i p. xii); 221.10; 229.16; 238.8; iii.124.18; 136.15, 17.

20.33. tavā, m.c. for tava: LV 168.18; 175.14, 18; 211.5.

li, probably only in vss, m.c. for te: vayam ti SP 270.4, so Kashgar rec., confirmed Lüders ap. Hoernle MR 158 (KN and WT vayam pi with Nep. mss.); vayam ti sütram 271.7 with Kashgar rec., confirmed La Vallée Poussin JRAS 1911.1076 (KN vayam tadā te); in 271.8 read with La Vallée Poussin l. c. subhairawe (ms. subheravi) kāli kṣayāt ti, partially with Kashgar rec. in KN note. — In LV 76.6, read kidṛsam ti kāyi saukhya, 'what sort of joy is in thy body?'; 164.4, read silenā na ti sadṛsu; 167.15; 168.18; 170.9, 12, 18; 243.2; 285.15; 362.13; 416.7. — In Mv prose i.45.1 Senart assumes siyā ti; but read siyāti, 3 sg. opt. In Mv vss; ti i.145.17 = ii.6.11 (in another repetition i.202.14 te, or actually tre, is read); i.155.14; 156.12; 202.2 = ii.5.18; i.233.18. — rūpam ti (1st ed. printed rūpanti) Gv 253.22.

ty, m.c.: sarve ty abhinandante LV 12.9 (vs), 'all desire for thee (on thy behalf)'.

20.34. luma: this is probably to be read LV 53.8 (vs): maitracitla luma devamānusāh, 'gods and men (have become) friendly-hearted towards you.' Lefin.'s best ms. A reads lumā (meter requires short final); the other mss. and Calc. unmetrically bhūta; Lefin. em. bhuta.

20.35. lvam: ayu samayo lvam ihā narasimhā nişkramaņāya LV 168.22, 'this is the time for thee, Man-lion, here to depart from the world'; ayu samayo lvam ihā varasaltvā (voc.) dhyānavikurvā (voc.) 169.8; ayu samayo tvam ihā nrpasāno (so read with v.l.) nişkramaņāya 169.16; ayu samayo lvam ihā varabuddhe nişkramaņāya 170.2 (all examples vss.)—The usage seems confirmed by Mv i. 174.12 (vs) where mss. lvam (two out of six lam; Senart em. te) prathame phale.

Locative singular

20.36. First person. hammi, or ahammi, seems to be the true reading Mv iii.216.7 (vs) tato hammi prådur ahi (so read with mss., § 32.28), 'then before me appeared'. There is a v.l. ham pi, which makes no possible sense. Senart is right in assuming a loc. and comparing the grammatically authorized Pkt. forms amhammi, mahammi (Pischel 415).

Nominative plural

20.37. First person. vaya and vayu occur as phonetic reductions of vayam, in verses m.c. Only once I have noted asme (= Pkt. amhe), and once asmo by em. (cf. Pkt. amho, cited in grammarians, Pischel 415). About equally rare, and not too certain, are cases of no, mo, and me, which will be discussed below.

20.38. vaya (m.c.): SP 117.5; LV 133.9.

vayu (m.c.): LV 368.21 (Lefm. with most mss.; ms. A vaya).

asme: SP 272.4 (vs), so KN with Kashgar rec., confirmed by La Vallée Poussin JRAS 1911.1076 asmai. asmo: LV 115.15 (vs); mss. asmi, asmin, asmim, but the em. asmo seems fairly plausible.

20.39. no: Divy 200.16 (prose) yan no dharmaśravaņābhiratā bhavişyāma ity. Note attributes this form to Pali (from Childers). It is not recorded in Geiger (nor in Pischel for Pkt.). But the Saddanīti (xii.51; Smith I p. 295 line 20 fl., 296.1-3) does indeed authorize it as nom. pl.; this is the ultimate source of Childers's citation.

20.40. mo: Hemacandra iii.106 recognizes this as nom. pl. It occurs in our language oftener as acc. and gen. pl. (§§ 20.47, 58; Senart i note p. 601). But it seems quite clearly nom. pl. in Mv iii.9.10 = 10.1 (vs) iha mo tamasamgrahe vasāmo andhakārake (10.1 samvasāmo 'ndha^o). Noteworthy in this connexion is the form mu, evidently a MIndic shortcuing of mo, as gen. pl. in the 'Prakrit Dhammapada', Senart JA IX.12 (1898).293. Cf. next.

20.41. me: lato me etāhi strīhi uddhrtā Mv iii.72.8 (prose), mss. Senart emends me to mam; if any emendation were to be made, mo would be more plausible (cf. prec.). But perhaps me may stand; cf. Pkt. ne beside no as acc. pl. (me would be to mo, above, as ne to no). Cf. Ašokan ve, nom. pl. (see below under vo, vah, nom. pl.).

20.42. Second person. Besides $y\bar{u}ya$ m.c. for $y\bar{u}yam$, we find yusme, = Vedic id. (and cf. Pali-Pkt. tumbe, which presupposes it); as well as yusmi, m.c. for this (and possibly yusma? see LV 314.15, below). The regular Pali-Pkt. form, written tumbe, occurs a few times in the text of Mv, always with a v.l. tusphe (once written tubse), which (if the true reading) recalls Asokan tuphe (Dhauli-Jaugada and Pillar Edicts), nom.-acc. pl. (Hultzsch cvi, cxviii). We also find vo as nom. pl.; cf. Asokan (Maski) ve (Hultzsch cxxvii), nom. pl., and no, mo, above. (In fact ve occurs once here as a v.l. for vo, nom. pl.) Twice a v.l. $v\bar{a}$ occurs for vo, and, doubtless by analogy with vo, there is also yusmakam as nom. (as well as acc.) pl.

20.43. yūya, in vss, m.c.: SP 91.9, 11, 12.

yuşme (not limited to verses; prose marked): SP 45.12; 378.2 (prose, so read with Kashgar rec. for yūyam). — LV 152.1. — Mv i.186.8; 313.8, 9 (both prose); ii.54.21; iii.85.19. — Sikş 354.7. — KP 146.1 (prose), twice.

yuşmi, in vss, m.c. for prec.: SP 195.3, Kashgar rec. yuşmi sarve for text sarvi yūyam. — LV 314.16.

yuşma? Cf. asma, acc. pl. Pischel 420 quotes tumha as given by grammarians only; it may possibly occur in Ap. (Jacobi, Bhav 37*). The only possible occurrence I have noted is yuşmaiva LV 314.15 (vs), nom. pl., if it stands for yuşma plus eva; but perhaps yuşme-eva would yield the same result.

tumhe: occurs in text, with one ms. (out of two used), Mv ii.173.14; 174.4; iii.73.9 (all prose), and ii.184.15 (vs). In the last three there is a v.l. *tusphe*, in the first *tupse*, probably intending the same or *tupphe*.

20.44. vo (once v.l. ve, cf. above): yena vo (mss. vah, vā) liram ānīlā Mv iii.78.17 (vs), 'since you have been brought to the shore'; tam vo upetha śaranam 85.18 (vs), 'go ye to him for refuge' (here one ms. ve, cf. above); kim vo āgatā 91.16 (prose), 'why have you come?' (one ms. vā āgatā, the other vo āgato). This usage is found even in RV. according to G. Liebert, Lunds Univ. Arsskrift N. F. Avd. 1, Bd. 46, Nr. 3 (1950); but he is hardly convincing.

yuşmākam: jīvanto yuşmākam (printed yuşyākam) athavā mrtāh Kv 32.4 (prose), 'are you alive or dead?'

Accusative plural

20.45. First person. The nom. vayam is used as acc. In a verse, asma occurs; cf. Pkt. (M.) amha, acc. (according to grammarians, also nom., cf. yuşma above) pl. Erroneous is the isolated $n\bar{a}$, despite $v\bar{a}$ above as v.l. for vo in nom. pl. There is also mo, a modification of no, with the initial m of the sg. (like Pali mayam for vayam), which we met above as nom. pl. and shall meet (more often) as gen. pl.; and also mam, which is not recorded outside of our dialect in the pl. but occurs here more often and more certainly as gen. pl.; is it a transfer to the pl. from acc. sg. (MIndic) mam (= Skt. mām), first as acc. pl. and then also as gen. pl. (by analogy with no, mo, etc., which are used for both these cases)? By this same analogy we also find asmākam, the Skt. gen. pl., or asmāka m.c., used as acc. pl.

20.46. vayam: vayam ca na samanvāharisyati Mv iii.163.8 (prose), 'and he will not be interested in us'. asına: tam cāsma lokācariyah... upekşate SP 118.1

(vs). [$n\bar{a}$: premagauravam upasthapisva $n\bar{a}$ (end of line) LV 54.15 (vs), could only mean 'cause us to enter into (i. e. to manifest) love and respect'; apparently all Lefm.'s mss. read $n\bar{a}$; Calc. nah. However, several mss. read °pitva for °pisva, and the true text is undoubtedly upasthapitvan \bar{a} , gerund, which makes good sense.]

20.47. mo: only in Mv; and some of the cases cited by Senart are emendations. The following acc. pl. cases seem reasonably reliable; gen. pl. forms are more frequent: Mv iii.84.2, 15 (vss); na mo bhagavan rājā śuddhodano icchali drastum Mv iii.102.18 (prose).

mam: tā mam (v.l. tā-sam) samyak prativartanti (probably read with one ms. prativartetsu) Mv iii.84.9 (vs). Note in line 15 below: tā mo samyak prativartensu.

20.48. asmākam, or m.c. asmāka: asmākam... dāyādān samsthāpayati SP 109.7 (prose); asmākam smārāyali talhāgalajāānadāyādān 110.4 (prose); śrutvā ca asmāka (m.c.; WT °ku; or perhaps gen.?) 116.7 (vs); saced asmākam api bhagavān, yatheme 'nye mahāfrāvakā vyākrtā, evam asmākam api talhāgatah prthak-prthag vyākuryāt 206.6 (prose); yad bhagavān asmākam vyākuryād anuttarāyām samyaksambodhau 215.7–8 (prose). — nehi tvam asmākam Mv ii.219.13 (prose), 'conduct thou us'; asmākam tārehi iii.75.19 (prose), 'save us'; asmākam ca drştvā 102.19 (prose); jahāsi asmākam 216.3 (prose). —? asmākam api samanvāharelhāh Av i.211.2; but samanvāharati appears to be used elsewhere with indubitably gen. forms, see Dict.

20.49. Second person. yūyam (like vayam) may be used as acc. pl. We also find in a verse yuşmi, evidently m.c. for yuşme, a Vedic form which underlies the regular Pkt. tumhe; and yuşme itself is very likely the true reading (instead of yuşmäkam) at SP 379.3 (prose). Like asmäkam, yuşmäkam is used as acc. pl. Sen 33 regards some, perhaps all, such cases as syntactic confusions (gen. instead of acc.); some of his long list are doubtless best understood thus, but it seems to me better to consider the cases listed here formal confusions, i. e. to call them accusatives.

20.50. yūyam: pašyāmi yūyam ... sarvān SP 198.1 (vs). yuşmi (on yuşme see next): LV 324.21 (vs).

yuşmākam; nāham (...) yuşmākam paribhavāmi SP 378.1, 6; 379.3; 380.1; 382.5 (all proše), 'I do not revile you'. I doubt the use of the gen. with paribhavāmi. In 379.3 the Kashgar rec. reads yuşme, which is quite likely the true reading here (possibly in the other cases also?). — prativartensu yathā yuşmākam Mv iil.84.15 (vs); jahāmi yuşmākam (v.l. [°]ka) Mv iil.216.8 (vs). — yuşmākam āgamya Divy 405.10 (prose), 'due to you'.

Instrumental plural

20.51. First person: asmābhi and asmabhi occur in verses, evidently m.c. for asmābhis. There is also asmehi, once with v.l. aspehi, corresponding to amhehi which is regular in Pali (Geiger 104) and alternative to amhehim in Pkt. (Pischel 419); also asmair (asmais) is probably to be read once, cf. Pkt. amhe (gram., Pischel 415) and AMg.

tubbhe (see below under tusphehi). Finally, usmākam is used (not only as acc. but also) as inst., at least once.

20.52. asmābhi, in vss m.c.: SP 117.8; 118.9, 10.

usmabhi, id.: LV 327.20 (mss. asmabhih, unmetrical).

asmehi: asmehi (mss. astehi) jivantehi Mv i.313.8 (prose), 'while we are alive' (see § 7.34 end); ii.228.10 (vs); 469.2 (prose); iii.79.2 (v.l. aspehi), 4 (both vss); 390.10 (prose, v.l. ātmehi). — Sukh 51.3 (vs).

asmair: SP 110.13 (vs), Kashgar rec. asmair, text with Nep. asmābhir (both, because of other changes, are metrically acceptable).

asmākam: lair asmākam bhaklāgre raņam utpāditam Divy 335.24 (prose), 'we produced a fight in the refectory.' Followed by: te vayam bhaktāgre raņam utpādayilvā iha ... upapannāh.

20.53. Second person: tumhehi occurs; it is the regular Pali form and occurs beside tumhehim in Pkt. (Pischel 422); also the semi-Sanskritized equivalent yuşmehi. We find also yuşme; Pischel does not list *tumhe, but he lists (from grammarians) the corresponding amhe (cf. asmair above). Cf. also AMg. tubbhe, which with tubbhehim (for both see Pischel 422) and the Aśokan (Dhauli-Jaugada) tuphehi (Hultzsch cvi), are to be compared with our tusphehi or tuspehi. Finally we find tubhy-amhi, which is the dat.-gen. form provided with an inst. ending.

20.54. tumhehi: My ii.78.2 (prose).

yuşmehi vinilehi (as loc. abs.; cf. § 7.34) Mv i.198.15 (prose).

yuşme: kalham punar yuşme dharmam srutam KP 146.4 (prose).

tusphehi, or tuspehi: Mv ii.2.17 (prose), Senart em. tuphehi (the Asokan form, above); but mss. either tusphehi (so critical note at foot of page) or tuspehi (so note on p. 497); cf. tusphe as nom. pl., § 20.42.

tubhyamhi: Mv 1.254.1 (prose), so mss.; Senart em. tubhyehi.

Ablative plural

20.55. First person. The form asmad, instead of asmad, is read 11 times in Mv ii.79.4 ff. (vss), tasmā hi asmād daharo na mryyati, 'therefore from among us (our family) no young person dies'. On the abl. in partitive sense, cf. Speyer VSS 56. The mss. usually read asmākam, which is metrically impossible. The Pali parallel, Jāt. iv.53.9, has amham.

Genitive plural

20.56. First person. asmāka and asmāku occur in verses m.c. for asmākam. There is also asma = Pkt. amha (Pischel 419, in M.JM., beside the commoner amham which is also Pali). Not uncommon in Mv is mo, for which see §§ 20.40, 47, and mam (cf. § 20.47) occurs often enough to seem not open to question. Once, apparently, nu occurs for no = nas, in a verse, doubtless m.c.; cf. us for vas, and the 'Pkt. Dhammapada' form mu (gen. pl.) for mo = no (§ 20.40). (The ms. reading mu- RP 17.15, see § 20.3, can hardly be right, unless it is used as sg., for which no evidence has been noted; Buddha, speaking of his own dharma, would hardly say 'our' for 'my'.) According to Senart vayam is once used as gen., but the passage is doubtful.

20.57. asmāka, in vss m.c.: SP 116.1, 7 (in 7 WT °ku); 117.2, 10; 118.2. In the last two, some or most Nep. mss. asmāku.

asmāku, see preceding.

asma: asmāpi ksetram siya evarūpam Sukh 50.12 (vs), 'may we also have a field like that'.

20.58. mo: only in My; and a number of the cases listed in Senart's Index are conjectural. The following

20.63]

(not all listed by Senart) seem to be reasonably reliable instances of gen. use: prose, i.351.7; 355.4; verses, i.306.15; yathaiva mo śakra tathaiva so tuvanı ii.54.2, 'just as Śakra is to us, so shall you be'; yo mo prathamo 70.14, 'who first of us'; 79.5, 19; iii.84.11; 86.8; 367.6.

20.59. mam: only in Mv; some cases listed in Senart's Index are conjectural and unreliable. yo mam prathamataram sväkhyätam dharmavinayam (Senart assumes here lacuna for a verb form) tena aparasya ākhyātavyam Mv iii.59.16 (prose); kim mam (? mss. mām) samjayinā... drstena 63.2 (prose), 'what's the use of our seeing Samjayin?'; ye pi mam pultakā āsi 84.18 (vs), 'our sons'; le mam adhigalā iha 293.19 (vs), 'they are found (attained) for (of, by) us here'; 310.10 (prose, less certain; text doubtful and boldly emended by Senart).

20.60. nu: (nirvrto pravaralakşanadhāri) yo nu sāstā

First

(mss.) . . . Mvi .69.6 (vs; so more likely than interpretation of nu as particle).

[vayam: according to Senart, note on Mv ii.480.4; but the passage is otherwise dubious, and likely to be corrupt in some obscure way.]

20.61. Second person. La Vallée Poussin JRAS 1911.1073 furnishes from a Kashgar ms. the form lus for edd. vas SP 255.11 (vs); cf. yu for vo, vas in the 'Prakrit Dhammapada', Senart JA IX.12 (1898) pp. 203, 210.

20.62. Otherwise I can only record tusme for yuşmākam (presumably gen. pl.), cited by Kern, preface to SP, vii as occurring in the Kashgar recension. No reference is there furnished and I have failed to observe it in the critical notes of KN. It would correspond to a Pkt. lumhe which Pischel does not record as gen. pl., tho he does record amhe in that sense (419).

20.63. Table of Personal Pronouns

Sg. tvā(m)-, tvam-Sg. ma- (mu-?) tva, tuva(m), tuma(m), tuham, tvayi, tvayā ham (hanı), ahu, aha, hu, ha mama, mamā, mahyam(?), me, mi maya, maye, mayi, me, mi, mahyam tvaya(-m?) [?mamātu] mamato mahya(m), mamā, mi, mamam(?)

Loc. (a)hammi

Pl. asma-

Stem form

Sg. Nom.

Acc:

Inst.

Dat.

Abl.

Gen.

Pl. Nom. vaya, vayu, asme, asmo, no, mo, me

Acc. vayām, asma, nio, mam, asmākam

- Inst. asmābhi, asmabhi, asmehi, asmair, asmākam Abl. asmād
- Gen. asmāka (asmāku), asma, mo, mam, nu, [?vayam]

tvam (tvam), tuvam, tava, tubhya(m), te, ti tvaya, tvaye, tvayi, tvayi, te, ti (ty), tubhya(m), tvam tubhya(m), tuhya(m), tavā, ti (ty), tuma, tvam

Second

yūya, yuşme, yuşmi, yuşma(?), tumhe (or tusphe?), vo (vā?), uusmākam yūyam, yușmi (yușme), yușmākam tumhehi, yuşmehi, yuşme, tusphehi (or tuspehi), tubhyamhi

tus, tusme(?)

21. Generic pronouns

Stems ta (sa), eta (eşa), ya, ka, and pronominal adjectives

21.1. In so far as the peculiarities of these stems are shared by, and equally 'irregular' (from the standpoint of Sanskrit) in, noun stems in a or \tilde{a} , they do not need separate treatment and have been included under such nouns. Here will be treated only forms peculiar to pronouns as such or, at any rate, forms which would not be normal in Skt. pronouns and do not seem to be as characteristic of nouns as of pronouns in our dialect. In the appropriate sections of noun inflection will be found some examples of those which have no special significance for pronouns as such; attention will be called to these by

21.2. Stem in composition. *ta*- occurs where SRt. would use *tad*- once, in a verse, SP 24.9, presumably m.c.: *ta-śrāvakānām gaņanā na vidyate*, 'of their (or his, i. e. Buddhas' or the Buddha's) disciples there is no counting.' So we must read with four Nep. mss. and the Kashgar rec.; 2 Nep. mss. *le*; KN em. *tahi*, kept by WT without note.

21.3. lam- for lad-: lamvarnā lambhaksā ladāhārā (note this last)) Mv 1.340.7 (prose, no v.l.); lam-sodhanārlhāya 'for the purpose of cleaning that (privy)' SP 113.14 (vs); Kashgar rec. lacchodh^o, but all Nep. mss. lam; text prints as separate word, misunderstanding the form.

21.4. The stem in -d is not used in Skt. with pronominal adjectives, except anyad- (Wackernagel III p.592). Once, at least, katamad- is so used in LV 383.5 (prose) katamadvidhaih pātraih, 'with bowls of what sort?'

21.5. Nom. sg. masc. As in MIndic generally, so and eso occur as generalized forms for Skt. sa and esa, without regard to samdhi. In most texts these are limited to verses, but in Mv they are common in prose. A few out of many examples:

Before voiceless consonants, so: SP 86.4; 87.3, 5; 88.11; 89.9; LV 135.7; eso: LV 151.8.

Before voiced consonants, so: SP 69.3; 88.1; 96.2, 7, 14; Mv i.3.8, 9, 10 (prose); eso LV 57.21; Mv ii.134.6.

As in Pali, we also find sv- before a vowel: sv-aya devadevah, or (with all mss. but one) sv-ayam eva devah LV 119.10 (vs) (cf. Pali $sv\overline{ayam} = so 'yam$); istrigara<math>sv-ayam LV 213.19 (vs) 'this harem' (despite neuter gender of istrigara); $esv \ agacchati'$ here he comes' LV 221.16 (vs).

We even seem to find occasionally su or eşu for sa or eşa before a consonant: teşv eşu śikşitu LV 124.18 (vs). And even in prose, su ca Mv i.131.15, according to all the (six) mss. (Senart em. so ca).

21.6. On the other hand, eşa occurs in final position, at the end of a line of verse, where Skt. would have eşah, in LV 385.17 and 19. And contrariwise, eşah is used before s- where Skt. would have eşa, in SP 15.13 (vs); this may be m.c. in a sense, but eşo would do as well; perhaps eşah is a late, blundering attempt at Sanskritization of an original eşo; but no v.l. is given.

21.7. Finally, se seems to be occasionally used instead of so: Mv ii.81.6 (vs; Senart em. so); Divy 180.9 (vs; one inferior ms. so). And even te, with t- instead of s-, is presented by all the mss. Mv i.104.20 (vs); Senart emends to so. On the ending -e cf. § 8.25, where other pronominal

forms in -e are cited. On pronominal forms in -o, -a (m.c.), and $-\hat{a}$, see §§ 8.19, 23, 24.

21.8. Norm. sg. ferm. The ending $-\alpha$ occurs in pronouns as well as nouns, § 9.9.

21.9. Acc. sg. masc. and fem. The nom. may be used for acc.: grhana su Gautamam LV 337.7 (vs), 'seize this G.I', where su = so for tam: so Tib., dge sbyon hdi zun zhig, 'seize this monkl' Foucaux, and app. Lefmann (who prints su-gautamam as one word), understand the prefix su-, which in Māra's speech is hardly possible and is contradicted by Tib.

Once eşām is read (in the mss.) for elām, by analogy with eşā: Mv i.365.14 (see Dict. s.v. śakyā).

21.10. Nom.-acc. sg. nt. The masc. so replaces tad, e. g. SP 62.7 (vs); cf. se in AMg. (Pischel 423) and su in Ap. (Sanatk. 501.3 su bhavanu; ehu etc. for etad, Jacobi, Sanatk. p. 15).

21.11. The nominal ending -m (-m) replaces -d (-t), as commonly in MIndic. Instances are very numerous, in most texts chiefly in verses, but also occasionally in prose, and not only in Mv; these are marked as such.

tam, pronoun and adverb: SP 118.1; 213.2; 254.4; LV 188.16 (read hi tam for hitam, with Tib.); 236.3, 5; 330.20; in prose, LV 16.20; 105.5; 140.9; 209.4; 269.19; 291.9; Mv (prose) i.36.14; 39.6; 40.3, 12; 158.11; 197.15 etc., very common; Divy 501.18 (prose); Bhad 8, 9 (in these correlated with yac ca), 57.

etam: SP 273.10; LV 79.14; asādhum etam Mv ii.272.9 (prose) 'this is a bad thing'.

yam, pronoun and conjunction: yam ca te cintayanti SP 124.6 (prose; repeated with other verbs in the next two lines); 212.8; 374.3; LV 47.12; 78.4; 259.2; Mv (prose) i.3.10, 13; 4.9; 7.6; 37.1; 46.1; 207.2 = ii.10.12, etc.; Divy (prose) 77.11; 87.9 (note: 'The MSS. often read yam for yal'); 204.11; 466.7 (ed. wrongly 'yam). Pronominal adjectives: katamam LV 26.3 (prose);

Pronominal adjectives: katamam LV 26.3 (prose); anyam Mv i.60.10 (prose); 224.21 = ii.28.1.

On forms ending in -a, -u, see §§ 8.30, 34.

21.12. For kim of the interrogative, a short syllable is required LV 91.18 (vs); we should probably read ki, altho mss. and both edd. have kim.

21.13. Oblique sg. fem. For forms in $-\bar{a}ye$, $-\bar{a}ya$, see §§ 9.37-41, 59, 62. Peculiar to pronouns, and very rare, is the ending *-asyāye*. Only one instance has been noted; *lasyāye velāye* (loc.; no v.l.) Mv iii.139.1 (prose). It is evidently a blend of *lasyā* (= *tasyām*) and *tāye*. It is confirmed by the similar Pali gens. *tissāya*, *etissāya* (Geiger 105.1), blend of *tissā* (= *tassā*) and *tāya*.

21.14. Inst. sg. masc. and nt. Besides *-enā* for *-ena* m.c. (§ 8.39), we find *elinā* and *ekinā*, in prose as well as verse. This is probably an ending primarily belonging only to pronouns; a single instance, in a verse and probably m.c., has been noted in nouns (§ 8.40; Senart interprets it otherwise). Such forms as eiħā, imiāā occur in Pkt., and imiā (or amiāā, but apparently no such forms from other pronouns; Geiger 108) in Pali; but it is not certain that Pischel (128, 428) is right (following Lassen) in taking them as analogical to Pkt. kiāā for kena. In our dialect, at any rate, no *kiāā has been noted. We

find (besides *iminā*, § 21.57): *elinā* Mv ii.126.12; 127.16; 129.3; 130.6 (all prose, in phrase *elinā evam-lūhaprahānena*); Dbh.g. 39(65).18; *ekinā* LV 151.8 (*ekināi*so, in vs); Mv li.103.16 (prose); 217.15 (prose; mss. *kinā*, but em. seems certain); iii.12.8 and 13.1 (vs; read the next word *striyo* with mss. in 13.1, and probably in 12.8 where *striyā* is an em.: 'begotten by one man from a woman'); 235.11 (prose).

21.15. [Abl. sg. masc. and nt. Thrice, in the text of Gv as printed (44.8, 9; 96.22, all prose), occurs the strange form *ekaikasyād*. In spite of the repetition, it is probably only a misprint (or error of mss.? the edition is full of glaring misprints, however) for *ekaikasmād*, which occurs in 97.5; cf. *ekasmād* in 96.26. If sound, it would be a sort of blend of abl. and gen. endings. The second ed. of Gv corrects the first and third occurrences to "kasmād, leaving the second uncorrected. It reads *ekaikasmād* in 96.26.]

21.16. Gen. sg. masc. and nt. On *-asyā* m.c. for *-asya* see \S 8.58.—The interrogative pronoun has *kisya* (only in Mv) = Pali *kissa*, Pkt. *kisa* (Geiger 111.1; Pischel 428). It is (as in Pali-Pkt) specially common in the adverbial sense of 'why?', but also used as regular pronoun in the gen. sense: *kisya khu nāma krtena* Mv ii.222.14 (vs); cf. ii.52.6 (vs) *kisya hetu* 'because of what?' or 'for what reason?', an em. for *kima* (v.l. *kim asya*) *hetuh*, but confirmed by the Pali parallel Jät. v.390.16 *kissa hetu*. Other instances, practically all in the sense of 'why?': prose, Mv ii.74.17; 75.6; 98.17; 464.15; 465.17; 476.19; 485.19; iii.25.12; 37.17; 173.15; 258.4; 330.4; vss, ii.149.2, 6, 14; iii.9.13; 294.21.

21.17. Gen. sg. fem. Adjectives which in normal Skt. have pronominal endings in this case-form may have nominal endings: anuttarāyāh samyaksambodher lābhino SP 41.8 (prose).—On -asya in fem. \hat{a} -stems see § 9.75.

21.18. Gen. sg. masc. and fem. se, si. As is well known, Prakrit has a gen. sg. demonstrative se, masc. and fem., which Pischel (423) regards as belonging to the sa-stem; his theory seems supported by the gen. pl. sdnam of our dialect. It is of course analogous to me, te. We find se chiefly in Mv; the following seem pretty certain: (prose) i.233.13; 342.2; 346.3, 4; ii.145.2; 434.6; 479.10; 480.8; iii.32.6; (verses) ii.231.13, 14; iii.2.18; 244.15, 17. Also, KN assume se by em. at SP 237.11 (vs) tahim ca se viharato; all their Nep. mss. so; Kashgar rec. ca viharanlasya; WT's ms. K' reads so, they em. to sya (= asya). The above cases are masc.; but fem. examples also occur: Mv (prose) ii.15.1; 172.14; 427.17; 459.2.

21.19. Furthermore, we find once si (= Pkt. se, Pischel 423) for this same se, in a vs m.c., Sikş 302.5. This passage has a correspondent Mv ii.375.17, where the meter is secondarily changed and si is eliminated.

21.20. Loc. sg. masc. and nt. On *-asmi*, m.c. for *-asmin*, see § 8.67.—Nominal endings (-e, and m.c. -i) occur in stems which in Skt. would show pronominal endings: sarve jage LV 111.17; anye Mv ii.422.12; Mmk 58.12; 75.6 (all prose); anyatarānyatare Mv iii.73.2 (prose); eke Sukh 10.10 (prose); eki Gv 384.17 (vs, m.c.).

21.21. Once Mv 1.217.11 (prose) has tamhi (so mss., Senart tamhi) dvipe, as regularly in Pali tamhi (beside tasmim). See § 21.66, imamhi, and § 8.61 nikhilamhi.

21.22. Common, on the other hand, are the forms tahim, yahim, kahim (cf. also imahim, \S 21.67); the byforms tahi, kahi, ekahi also occur, but apparently only in verses, usually where meter requires short final, and never in Mv, which in prose and verse seems to know only tahim etc. These forms are known, chiefly as locative adverbs, in Pali and Pkt., and are most commonly so used here, tho like Skt. tatra they are also used in the place of pronouns. Examples are all from verses except as indicated. lahim: SP 33.4; 237.11; Mv i.180.3; 231.19 (prose, v.l. larhi); 232.13 (prose), etc., very common; lahimtahim Mv i.234.6, 'here, there, and everywhere'.

yahim: SP 112.7; 358.4; yahim kule Mv i.253.5 (prose) 'in which family' (= yasmin); yahim deve lli.68.2 (prose); KP 13.15, 110.6.

kahim: (often with ci-t) SP 11.9; 47.5; 91.7; 113.2; kahim cit 128.9 'in reference to anyone'; akahim LV 176.22 'nowhere'; kahim LV 214.22; 215.2; Mv i.155.7; 186.15; 198.16 (prose, kahim before vowel) = ii.2.18 (where Senart kahim upa-, but v.l. kahim); i.221.8 = ii.23.14 (prose); i.300.1, 5 etc. (prose); ii.198.10, 12; 425.12 (prose); Samādh 19.28, 29; kahimcit Mv ii.33.2 (prose); 340.8 (vs; v.l. karhi cit, which is unmetr.).

tahi: SP 9.6 (in plural sense, = teşu); tahi sāsane 26.12; 68.10; 228.12 (at end of line, not m.c.); tahi jambumūle LV 134.10; 166.6, 21; 296.18; tahi lokadhātau Samādh 8.25; Bhad 59; Dbh.g. 39(65).11 (text ta hi).

kahi: LV 230.6; 341.20.

ekahi: Samādh 19.16.

21.23. Loc. sg. fem. Nominal ending substituted for pronominal: layam velayam: Thomas ap. Hoernle MR 89.20 (prose; from Bhadrapāla: Sūtra; Pali also has tayamvelāyam); katamāyām (v.l. °yam) Mv ii.73.3 (prose); anuttarāyām samyaksambodhau SP 64.11; 67.2; 69.6; 70.12; 261.15 etc., all prose, common; for 261.15 the Kashgar fragment Lūders ap. Hoernle MR 144.1 has anuttarasya(m), which would be the regular Skt. form (tho the masc.-nt. loc. of uttara may be nominal, Wackernagel III § 268c).

21.24. Nom.-acc. dual nt. On eti, m.c. for ete, probably acc. dual nt., see § 8.77.

21.25. Nom. pl. masc. For regular -e, as for any final -e, -i is common, but only in verses m.c. A few examples: ti: SP 310.3. - eti: SP 90.2; 162.6; LV 171.3; RP 12.11; Gv 481.3, 6, 8. - yi, probably to be read SP 83.11 for KN vi(bhaksayantah); Kashgar rec. ye bhaksayanti; WT ca bhaksayantah with their ms. K'. There is otherwise no cpd. vi-bhaks; the unmetrical ye and the variant vi- together seem to suggest yi. - sarvi SP 62.2 etc.; LV 74.14 etc.; Mv 11.53.19 (v.l. sarva); Suv 38.8 etc. - pūrvi SP 176.12. - anyi LV 171.3.

21.26. The nominal ending $-\bar{a}s$ occurs in pronominal adjectives but rarely (cf. $-\bar{a}(s)$ as acc. pl.): sarvā (followed by g-) LV 295.17 (prose); agrees with yāvanto... aśruta-pūrvāh, lines 13-14. So all mss., only Calc. sarve.

21.27. But quite common, tho only in verses, is the ending -a. It seems clearly m.c., and since otherwise -e is apparently not reduced to -a, we must doubtless understand this as representing -ās, the nominal ending. ta: sarvi ta (no v.l.) nirmila LV 307.19 'all these were formed by magic'; Gv 256.12. — ya: ya (WT with K' ye) fobhayisyanti SP 149.14; others, LV 341.5; Śiks 329.12 (read sahanti ya; from Ratnolkādhāranī); ya keci jināh Gv 286.1. — [ka: ye ka-cil Śiks 109.9, so text; but probably misprint; meter requires ke.] — sarva SP 191.2 (?possibly acc. pl. with vimāna); LV 74.10, 17; Dbh.g. 17(353).2. — anya SP 325.4; LV 153.12.

21.28. The nom. sg. esa seems to be used as nom. pl. in esa nirortim upema ihaiva Mv 1.69.12 (vs), 'we here approach nirväna right in this life'. The only alternative would be to understand it as for "esām = etām, acc. sg. fem. (analogical to esā nom.). On sg. for pl. nom. forms see \S 8.83, 10.190.

21.29. Nom. pl. fem. On (e)tayo see § 9.89.

21.30. Acc. pl. masc. As regularly in MIndic, the ending -e is used as acc. as well as nom. It has been noted only in verses in most texts, but in prose of Mv, e. g.: nāham...ye dharmā anityās ie nityato dešayāmi Mv i.173.1-2 (prose), 'I do not declare states which are impermianent to be permanent', and parallel phrases in the sequel (note how easily, by attraction to the nom. pl. ye,

8

an acc. le could be introduced in such a context): My ii.141.3; 427.18.

21.31. Otherwise in verses: le SP 45.3; 87.11; LV 172.17; le (sc. kāmān; at beginning of pāda, hence not enclitic 'of thee' with Senart) vai jane My i.180.5 (read in preceding °jñas ca with mss., going with subject of jane); Mv i.189.9; 278.11; Suv 45.10; RP 19.7; Gv 362.14. — ete SP 209.7. — ye SP 115.7, anye c⁴ ato vimisti ye mi drstah, 'and for 20 more (years) than that during which he has been seen by me' (text vimialiye, less likely, as acc. pl. of vimsati). - anye SP 10.8; 115.7 (see preceding); LV 49.11. — sarve tava samkalpām (= °pān) Mv 1.202.15 = ii.6.12. — aneke SP 53.5; 96.9 (all but Kashgar mss.). pūrve LV 196.12 pūrve ... supine imi ad(d), sāsi.

21.32. And -i, m.c. for preceding: sarvi SP 195.4; 351.4; LV 201.11; 308.7; 366.19; Bhad 1, 10 etc.; Samādh 8.7; Gv 241.11 (see Dict. s.v. āmukhīsthila).

21.33. -a, and m.c. -a. Cf. §§ 8.92-94; probably transferred from nom. pl. ending of nouns; rare: sahasranekā (but Nep. mss. all °ke) nayulāms ca bhūyah SP 96.9 (vs); kāyā (mss. kāyahl) sarva paneti LV 330.13 (vs), 'he evaluates all bodies'(?) .--- Weller takes sarva LV 3.14 (prose; sā sarvā śuddhāvāsān) as acc. pl.; but it may well be nom. sg. fem. with sā, sc. raimih.

21.34. On sanam, once as acc. pl., see § 21.45.

21.35. Acc. pl. fem. tan, tam as fem., see § 9.99. 21.36. Nom.-acc. pl. nt. Endings -ā, -a (m.c.), and -e, as in nouns, see \$\$ 8.100, 101, 103.

21.37. Inst. pl. masc. and fem. On the endings -ehi, -ehi, -ebhis, -ebhi, and (fem.) -āhi, see §§ 8.108-111; 9.102.

21.38. Gen. pl. -esa, only in vss m.c. for -esām; usually before consonants, but also before vowels, eteşa (v.1. °su; WT ete sam-) SP 303.6, 8. Others: tesa SP 68.10; 90.9; 95.7 etc.; LV 54.1; 114.9; 115.2 etc.; Samādh 19.16, 35; Suv 41.1; 45.9 etc.; Gv 256.25. - elesa SP 306.2. - keşa-ei(d) Samādh 19.25. - yeşa SP 97.10; 304.9 (WT var.); LV 37.3; 46.7; Mv i.204.7 = ii.8.2; Samādh 19.29; 22.5. — sarveşa Samādh 8.30. — pareşa SP 94.6 (WT with Nep. mss. °\$u).

21.39. -esam, only in vss m.c. for -esam, and only before vowels: tesam SP 86.9; RP 9.2; 28.2.

21.40. -eşu, -eşu (before vowel), also only in vss m.c. for -eşdm: teşp SP 203.10, acc. to WT with their ms. K', tesp anu° (KN tes' anu° with Kashgar rec., Nep. mss. te anu°); tesu moksa (so with v.l., m. c.) nacirena bhesyate LV 54.14, 'they will get salvation soon'; My i.151.7; Bhad 4, 5, 6, 6¹, 19, 25 etc. — yeşu LV 54.1 (v.l. yeşa). — pareşu dāram Mv 1.286.9, 'the wife of others'.

21.41. -- anam, for -anam, the nominal ending; MIndic form, not m.c. (in prose and before consonant): lanam devānam Mv iii.304.3, mss. (Senart em. devānam to devatānām, but he leaves tānam as in mss.)

21.42. -ana, only in vss. m.c. for -anam, the nominal ending. elāna SP 83.8; tāna Bhad 56 (seems to be probably the true reading, with v.l., for text laya, but Gv 547.16 also tāya); sarvāņa SP 89.9; Samādh 8.22, 24.

21.43. -ana, m.c. for same: anyana RP 21.4 (= anyesām).

21.44. -eşāņām (also written -eşānām), blend of -eşām and -ānām, as in Pali (-esānam, Geiger 105.1; 113.1); noted only in vss of Mmk: etesanam Mmk 499.25; 511.2; sarveşānām Mmk 387.2; 500.1 (in the latter spelled ondm).

21.45. sānam occurs many times in Mv, apparently as gen. pl. of the pron. stem ta- (sa-); in one case, i.7.13, an acc. pl. seems required (note that the following aviddhānām of the text is an emendation). See Senart's note i.373; he refers to Weber, Bhagavati (1866), 422 note, where Weber doubtfully quotes the same form from this AMg.text, querying whether it intends Skt. svānām. (Pischel 423 seems to ignore the form.) It happens that at My i.5.9 svānam is actually read in the mss.; but this is isolated. Other variants noted in the mss. are sana, sana, manam, salam, solam; but they seem all to be mere corruptions, and it seems impossible to doubt the reality of the form sanam. Nearly all the following occurrences are prose, and in the great majority no v.l. is reported. Mv 1.5.6, 9; 7.4, 10, 12; 8.2, 5, 6; 18.14; 20.4; 26.6; 27.6; 31.4; 340.8; 341.2, 14; 342.9; 345.5, 10; 346.1; 347.19; 355.6; ii.27.18; 121.4; 122.7, 15; 123.4; 172.11; 214.6; 234.5 = iii.65.6 = iii.92.10; ii.282.16, 17; 412.12 ff. (repeatedly); iii.28.18; 70.18; 72.5; 87.8; 114.7, 8; 282.11; 283.2, 7; 329.13; 379.16; 426.17; 427.6 ff.; 430.16; 432.3, 4.

	21.46. Paradigms. Stem in comp.: <i>la-, lam-; kalamad-</i>									
Masc.				Nt.		•	Fem.			
	so, sv, su, uu, etc., ela		Sg. NomAcc.	so, tam; yam etc.; etu etc., ta, eta; ke, ki (?)	Sg.	Nom.	eşa			
	÷	Inst.	yenā, elinā, ekinā			Obl.	tāya,	tāye,	etc.;	tasyāye
		Abl.	[ekaikasyād?]						•	
		Gen.	tasyā etc.; se, si; kisya			Gen.	tasya	; se;	anutta	ayâh
×.,		Loc.	tasmi etc.; tamhi; tahim kahim; tahi, kahi, ekahi						inte La constante La constante	•

Pl. Nom. ti, yi, etc.; sarvās; ta, ya etc.; esa

Sg. Nom.

Acc. le, ye, sarve etc.; sarvi; anekā

> Inst. tehi etc.; tehi, yehi; tebhis, yebhis; tebhi etc.; sarvebhi

Gen. teşa etc.; teşam; teşu etc.; tānam; lāna etc., anyana; eleşānām, sarveşāņām; sānam

Pl. Nom.-Acc. yā etc.; ya, te

Pl. Nom. (e)tāyo

> lān, lām Acc.

Inst. (e)tāhi **21.47.** In Class. Skt. the forms of *ena* are enclitic only (on Vedic see Wackernagel III pp. 524, 525). Here they were certainly not always enclitic. In AsP 227.9 (prose) *enām* begins a sentence and is emphasized by *eva*; *enām* eva also id. 255.7, 12, and elsewhere in this text.

21.48. Like Pali and Pkt., our language contains a stem na, apparently a reduced form of ena. Even this occurs at the beginning of a line of verse, nesām SP 48.9 (gen. pl.). This (which = Pali nesam) is the only form I have noted other than acc. sg. of which the fem. nām occurs Mv II.247.8, 10 (to be sure with v.l. tām the first time, tam—but it must be fem.!—the second). Otherwise there is only nam, once acc. sg. nt., Mv i.57.3, otherwise acc. sg. masc. This occurs at least once at the beginning of a sentence, in Mv ii.168.7–8 (prose), gacchaiha nam atimuktake (v.l. °kam, Senart em. °ka-) smašāne netvā jivašūlakam karotha, 'go, take him to the A. cemetery and make him to be impaled on a stake.' Other instances of nam, acc. masc.: Mv i.211.18 f. = ii.15.18 f.; i.212.1 = ii.16.1; i.301.16; ii.247.6; 426.4; 447.2; lii.409.12 (these all prose); verses ii.81.6; 187.8 (? drstvā nam).

ayam etc.: stem ima- inflected throughout

21.49. Besides relics of various Skt. forms, MIndic generally has a full pronominal inflexion based on the stem *ima*, which in Skt. is used only in the acc. sg. (masc. and fem.) and in the nom. acc. dual and plural. This stem is used in Pali in all cases except the nom. sg. masc. and fem. (Geiger 108), and in Pkt. in all without this exception (Pischel 430). Our language agrees with Pkt. It even uses the stem *ima*- in a compound. These forms are not limited to verses, tho commonest there (except, as usual, in Mv where they are equally common in prose). Once, at least, a Kashgar fragment of SP has *imasmim* in prose where the printed edd. read *asmin* without report of variant.—As in the preceding sections, we shall refer to the treatment of a- and \bar{a} -stem nouns for endings which concern nouns as much as pronouns.

21.50. Stem ima- in composition: ima-guna (for $^{\circ}n\tilde{a}$, m.c.) vacanā RP 47.8 (vs), '(Buddha's) voice has these good qualities.' — imidršāni SP 120.2 (vs), perhaps for im(a)- id° , 'such things as these' (Kashgar rec. wholly different).

21.51. Nom. sg. masc. imo (ima? ime?): Mv (all prose) imo ii.64.17; 152.15; iii.301.11; imāham (= ayam aham, so 'ham) ii.244.12, perhaps for ima (or imo) aham. — ime, as in AMg. (Pischel 430); Mv ii.82.12 (prose), mss. ime, Senart em. imo.

21.52. Nom. sg. fem. imä (also imu, see § 9.13): Mv i.66.8 (vs); ii.65.6 (prose, mss. imäm); 174.14 (prose).

21.53. Acc. sg. masc.: See §§ 8.30, 38, for the forms imu, imām.

21.54. Acc. sg. fem.: see §§ 9.16, 19, 23 for the forms imam, ima, imu. Especially ima is common in verses m.c.

21.55. Nom.-acc. sg. nt. imam (and m.c. ima, imu, im', as in nt. a-stem nouns); very common instead of idam; chiefly in verses in most texts, but there are prose cases in SP and LV as well as Mv, viz., prose: imam samkäradhänam sodhayatā SP 106.15 (samkāradhāna 'privy' is nt. just before, 105.12-13); imam samyagjāānam LV 400.15; others Mv i.346.6, 7; ii.110.1; 169.15; 195.14; iii.22.21, etc. A few out of many examples from verses: imam SP 93.7; 228.7; LV 178.14; 392.3; te viditam imam ca param ca Mv iii.381.10, 'known to you is this (world, life) and the beyond'; ima SP 95.6; imu SP 90.10; bhuw' (= abhuval) imu (nom. nt.) LV 222.11, 21; Bhad 42 etc.; im' evarūpam (= idam evaŋrūpam, nom. nt.) SP 95.12. --In Mv ii.345.11 (vs) text imo, v.l. ime, but meter requires short final; read imu? (or imô?). **21.56.** Inst. sg. masc.-nt. imena: SP (vss) 28.3; 90.9; Mv (prose) i.301.13; ii.107.1; 180.18; 256.5; iii.290.15; (vs) ii.104.16; Mmk 618.25 (vs).

21.57. Inst. sg. masc.-nt. iminā, the regular Pali form; also Pkt. imiņā (Ś.Mg., and, beside imeņa, also JM., Pischel 430): Mv i.132.15; iii.53.8, 10 (all prose). Cf. § 21.14, etinā, ekinā.

21.58. [Inst. sg. masc.-nt. eminā: aham na (14... naiva) kasyacid emināpi acchindāmi (Senart em. ācchi^o) Mv iii.403.12 and 14 (prose), so Senart, but read... kasyaci demi nāpi, see Dict. s.v. acchindati.]

21.59. Inst. sg. fem. imäye and imäya: imäye Mv iii.163.7 (prose); imäya 426.6 (prose).

21.60. Dat. sg. masc.-nt. imāya: Suv 44.5 (vs; interpretation not entirely certain).

21.61. Abl. sg. masc.-nt. imāto: Mv ii.253.7 (prose).

21.62. Gen. sg. masc.-nt. imasya: SP 149.8 (vs);

LV 314.16 (vs); Mv i.38.4; 246.10; 364.2 (all prose); Suv 65.6; 66.9 (both prose; in both most mss. asya).

21.63. Gen. sg. fem. imasyā(s): imasyā (m-) Mv i.363.14 (prose).—On imasya assfem. see § 9.75.

21.64. Gen. sg. fem. imāye: My i.223.4 (prose).

21.65. Loc. sg. masc.-nt. imasmim, common: SP, acc. to Thomas ap. Hoernle MR 133.4 (prose), for KN 327.3 asmin; Mv (prose) i.50.14; 320.19; 336.5 etc. For imasmi (m.c.), see § 8.67.

21.66. Loc. **eg.** masc.-nt. imamhi (= Pali imamhi), cf. taphi and nikhilaphi **§** 21.21 and 8.61, occurs repeatedly in the mss. of Mv and should certainly be admitted. Senart admits it at i.185.20 (vs); see his note on this; yet this particular passage could be read imam (= idam) hi. But in other places, strangely, Senart rejects the form and emends to imahim (see below). Thus: imamhi (v.l. imāhi) āgamanaprayojanam ii.107.6 'reason for coming here'; aham pi imamhi evam (v.l. eva) osişyam ii.478.7, so mss., Senart āsişyam; read vasişyam, with what appears to be the same phrase ii.463.2, aham pi imahim (so Senart with one ms., v.l. imamhi) vasişyam, 'I too will dwell here'; ii.480.12; iil.154.2; 167.3; 290.2 (here one ms. imehi, which can be interpreted, 'with these'); 290.4. All these, except 1.185.20, are prose, and except as stated both mss. read imamhi; Senart always reads imahim except in 1.185.20, but without good reason.

21.67. Loc. sg. masc.-nt. imahim (cf. lahim, yahim, kahim, § 21.22): Mv iii.102.19 (prose; v.l. imehi); and perhaps ii.463.2 (v.l. imamhi, see just above).

21.68. Loc. sg. fem. imasyām: Mv iii.153.19 (v.l. *imasyā*, in that case gen.).

21.69. Nom. pl. masc. iml, in vss m.c. for ime, common: SP 86.8; 92.5; 93.6; Samādh 19.30; Dbh.g. 7(343).14. (In Bhad 58 imi pranidhānā must be an error for ime, required by meter, tho no ms. is quoted as reading so.)

21.70. Acc. pl. masc. ime (imi, ima), nom. for acc., as with other pronouns, $\frac{55}{21.30-33}$: ime, Mv ii.147.2; 435.11 (both prose); imi (vss), SP 33.1; LV 196.12; 385.7; ima(?), SP 281.14, so KN with their Nep. mss. as reported; WT with their K' imi; Kashgar rec. reported ime (unmetrical).

21.71. Nom.-acc. pl. nt. ime, see § 8.103.

21.72. Inst. pl. masc.-nt. imais: Suv 105.5 (vs); Mmk 23.3 (prose); 480.15 and 492.5 (vss). In all the vss, *ebhis* would be metrically satisfactory.

21.73. Inst. pl. masc.-nt. imebhih: My i.110.9 (prose; v.l. imehi; before s-); Mmk 4.20 (prose).

21.74. Inst. pl. masc.-nt. imehi, noted only in Mv; all examples but one from prose: Mv i.17.4; 50.14 (in loc. sense); 80.1; 89.6; 198.9 = ii.2.6; i.306.18 (vs; in abl. sense); 364.5; ii.293.11; 485.16; iii.73.17.

21.75. Inst. pl. masc.-nt. imehir, Mv i.90.6 (prose) imehir ast^o (-r- 'hiatus-bridger'?).

8*

21.76. Inst. pl. fem. imābhiķ: Suv 132.13; 177.5; 243.9 (all prose); and imāhi, Mv iii.269.16 (vs).

21.77. Gen. pl. masc.-nt. imeşām: Mv 1.273.16; 329.18; 335.1; 347.14; 350.11; 365.10; il.434.19; iil.55.10 (all prose); iil.84.3 (vs).

21.78. Loc. pl. masc.-nt. imeşu: Sikş 16.6 (prose; from Prašāntavinišcayaprātihārya-sūtra).

ayam etc.: phonetic alterations of Sanskrit forms

21.79. Hardly worthy of more than passing attention at this point are the following phonetic matters:

yam for ayam, nom. sg. masc., after a vowel in a verse (really a matter of samdhi); tam eva yam SP 28.6; vyakla yam LV 81.14 and 16.

ayu (after a vowel yu, cf. preceding) for ayam, only in vss m.c. (any final -am may become -u, § 3.58): SP 10.2; 16.4; 27.13; 113.9; 114.4 etc.; LV 50.13; 81.20; 236.4; prāpla yu kālu (= prāplo 'yam kālo) 326.16; ayu dharma 393.15.

iya for iyam, nom. sg. fem., in vss m.c.: prabhrtiya (= prabhrtir iyam) SP 63.13; 92.13; LV 174.11; 292.6. ida for idam, nom.-acc. sg. nt., in vss m.c.: SP 97.14;

98.8; LV 57.3. For asya as gen. sg. masc., asya and asya as gen. sg.

fem., asmi as loc. sg. masc., see §§ 8.58; 9.74-76; 8.67.

ayam etc.; miscellaneous non-Sanskrit forms

21.80. There remain a few odds and ends. First, Senart My i.417 note, cites a form aham = ayam (nom. sg. masc.), referring to Hemacandra 3.87. But what Hem. prescribes is not aham but aha; and he equates it not with ayam but with asau, adas (all three genders). According to Pischel 432, all the alleged occurrences of this aha may be interpreted as representing the adverb atha. Anyhow, some of Senart's cases for the alleged aham = ayam are unconvincing. Thus, in 1.53.2, aham means 'I'; the Buddha shifts to the first person, summing up his experiences in past births. In i.322.3, also, it means 'I'; a speech of Jyotipala follows. In ii.371.2 also aham is 'I'; read the verb bhomi (with one of two mss.). There remain: (1) i.5.2, where however the mss. read aha; the passage seems corrupt, but emendation to aham = ayam is rash. (2) i.348.7 tenäham räjä (no v.l.). (3) iii.49.6, not listed by Senart: na câham kumārako..., repeating the same phrase from 48.19 where nāyam ku^o is read. In the last two, aham 'I' is senseless; but especially the last looks like a mere corruption for ayam; perhaps the other is, too.

21.81. In the nom. sg. fem., Pali and Pkt. use ayam,

the masc. form, freely instead of, or beside, *iyam*. So also our dialect; I have noted instances only in Mv, but many there. Prose: l.214.12 = li.17.10; l.338.20; 344.8; *ayam vela* li.172.8; ll.428.15, 16; 459.17; ll.54.1 fl.—Verses: l.203.15 = li.7.11 (in the same line li.146.14, *iyam*); l.165.10; l.207.3 = ll.10.14; ll.18.4 (in the same line l.215.7, *iyam*); l.247.18; 304.17; ll.194.18.

21.82. For idam, nom.-acc. sg. nt., we find, rarely, both Skt. masc. and fem. forms substituted. Both are so used in Aśoka's inscriptions: Hultzsch lxv, lxxix, xciii, etc. (ayam, iyam, iyo); ayam also in AMg. (Pischel 429). ayam avalokilam nāma vyākaranam Mv ii.258.11 and 13 (prose; repetition of 257.13 which has imam, v.l. idam); ayu LV 173.10 (vs) 'this (thing; indefinite)'; m.c. for ayam; but the ms. A reads amu (§ 21.86). (The preceding word *lāryaravau*, all mss., is troublesome; Tib. sil sñan sgra las, suggesting abl., "ravā(d)? Or "ravai(r)? Tib. certainly does not make ayu agree with this word, as Calc. does, reading "ravo.) tad iyam Mv ii.54.7 (vs), apparently = idam.

21.83. In the inst. sg. masc.-nt. there are a few cases of ena; so M. and Ap. ena (Pischel 429; Jacobi, Bhav. 38°): LV 223.11; 224.18, 19; 237.7; 355.7 and 8 (all vss).

21.84. [If we could trust Senart's printed text, we should have to accept sya and sya, as gen. sg. masc. and fem. respectively, after consonants. But 1 think Senart is wrong.

(a) Mv 1.278.3 (vs) is printed rāstram sya dubbalam bholi... But the mss. read (with slight corruptions): rāstrasya dullabham bholi chidram bhūlam samantatah. The sense is good except that a negative seems required. Read perhaps rāstrasyādullabham: 'of his kingdom it is not hard to find a weak spot, a real one (? or, 'one that has occurred'), on any side.' Also in line 5, two lines below, read with 1 of the 2 mss.: rāstrasya (Senart rāstram sya with v.l.) sthāvaram (mss. °ro) bhavati rādham..., 'his kingdom has solid wealth, abundant' etc. (sthāvara neuter noun, not adj.).

(b) Mv i.199.9 (vs) reads yasyeha parisesam sya narlye fivitam bhavet; but in the repetition ii.3.7 the mss. have syan, which is the correct reading; it is the adverblal sydt, 'perchance' (see Dict.) The pronoun asyah would, in fact, be scarcely interpretable here; the statement is a general one and does not refer to a specific woman. Senart assumes $yasyd(h) \dots asyah$ together as indefinite, referring to Childers p. 602, 2d column; but in this idlom, I think, the form of ya- is always immediately followed by that of ayam.]

21.85. Paradigm for ayam, idam Stem in composition: ima-

Nt.

imam, ima, imu, im',

ayam, ayu, iyam

ida,

Fem.

imā, iya, ayam

imam, ima, imu imāye, imāya

asyā, asya, imasyā(s), imāye, imasva

Loc. asmi, imasmim, imamhi, imahim

Masc.

imasya, asyā

imo, ime, ima (?),

yam, ayu, [aham (?)]

Pl. Nom. imi Acc. ime, imi Inst. imais, imebhih, imehi(r) Gen. imeşün Loc. imeşu

imābhiķ, imāhi

[21.76

Acc. imu, imām Inst. imena, iminā, ena Dat. imāya Abl. imāio

Sg. Nom.

Gen.

21.86. Practically nothing non-Sanskritic concerning this pronoun has been observed. In LV 111.21 *amau* is printed, but (tho not corrected in either of the two Corrigenda) it is certainly a misprint for *asau* as in Calc. (no v.l. is cited). In LV 173.10 (vs) the best ms. A has *amu*, for *ayu* ($\frac{5}{2}$ 21.82) of the other mss.; the word is nt., 'this (thing)'; if *amu* is the true reading, it would be (? m.c.) for *amum* which is used in Pkt. as nt. nom.-acc. (trans-

ferred from the Skt. acc. sg. masc.; Pali uses adum). The same *amu* is given by Nobel in Suv 51.13 (vs), where it appears to be acc. sg. masc. (understand *svapnam*); some mss. *amum*; meter is difficult with either reading, -u seems required. In the following line 14 Nobel also reads *amu*, on the basis of Tib. but against all his mss. which have *leşu*; the same metrical difficulties would apply here too.

22. Noun suffixes

22.1. Here I shall refer to extensions of, or additions to, Sanskrit usage in respect of noun suffixes, both primary and secondary. Some of them are used nearly or quite as in Sanskrit; such cases are, in general, recorded here when the suffix clearly remains productive in BHS, in a way which makes certain or likely the post-Sanskritic origin of some of its examples (as in the case of the suffix ana added to present stems used as 'roots'). In other cases new suffixes, or new uses of Sanskrit suffixes, will be noted.

Suffix aka

22.2. The suffix aka is used with rather specialized verbal force, in a way not to my knowledge quite paralleled in Skt., referring both to the future (= 'intending to, for the purpose of'), and to the past when such forms are hardly more than periphrases for preterite verbs. What may be interpreted as adverbial accs. of such stems are used in a way that approximates gerunds. All these forms may perhaps be regarded as containing secondary -ka affixed to stems in -a; see § 35.5.

22.3. Referring to future: bhagavalo pādavandakā(h) Mv i.300.4 (they came) 'for the purpose of saluting the Lord's feet' (but found him gone); i.300.12; ('ko) ii.271.17; see Dict. for preksikā, nirīksikā; (pāniyam) pāyako 'to drink water'; (pānikhālanadī)snāyako 'to bathe'; vadhaka 'intending to kill'.

22.4. Referring to the past: pañcāśa varşā pi tadā palāyakah SP 112.1 (vs), 'even for fitty years he then was a fugitive' (Kashgar rec. palāyida; WT with ms. K' palānakah); kim nu khalu purimakā samyaksambuddhā bhājanapratigrāhakā utāho pāņipratigrāhakā (mss. ^ohato), bhājanapratigrāhakā ... Mv iii.304.7 (prose), 'did the former Buddhas (in first breaking the seven-weeks' fast after enlightenment) take a bowl (of food), or water? They took a bowl.'

22.5. Quasi-gerunds in -akam: -kārakam, -hārakam (§ 35.5); a list of such forms (incl. -kārakam) occurs in Mvy 8577 fl. and in parallels (Dict.): niścārakam, prihakkārakam, apahārakam, chedakam, sphoļakam, avalehakam, samdhunakam. Pali has similar forms in the corresponding section of the Pätimokkha.

22.6. More in line with the normal Skt. use of aka as primary suffix (Wh. Gr. 1181) is jānaka 'knowing, wise' and a-j^o (Dict.), apparently based on the present stem jāna-ti used as quasi-root.

Suffix alha, see Suffix tha Suffix ana, anā

22.7. Nouns of action, including infinitives. As a noun of action suffix, ana or anā remains freely productive in BHS. Nouns are formed with it from present stems (e. g. nisīdana = Pali maīñanā; spršana = Pali busana; manyanā or °na = Pali maīñanā; paiyana; šişyaņa from passive fişyale, etc.), which in BHS take the place of verbal roots. In a few exceptional cases they seem even to be formed from noun or adjective stems (kūļanā 'fraudulent practice', to kūļa; ākūlana = ākūla 'wish, intent'; niyatana 'determination, fixation', to niyata; vijäänanä and samjäänanä, Dict.). But these are no doubt the result of special analogies. In general, ana is a primary suffix, here as in Skt. Datives from ana stems function quite often as infinitives (§ 36.15). Occasionally, the abstractforming secondary suffix $l\bar{a}$ is added, without change of meaning, to stems in ana; § 22.41.

22.8. The following action-nouns, including stems of infinitives, will be found recorded in the Dict., besides those mentioned above: adhimucyanā, anutapyana, anupaśyanā, anubidhyana, abhisambuddhana, abaranyanā, avasrjana, āpadyana, ālikhana, utpadyana, utsrjana, uddiśana, ovidhyana, krudhyana, grhnana, citranā, jāgrana, tudanā (°na), trāyana, parikarmana (?), paritasyana (°nā), paritrāyana, pithana or °nā, prajahana, patibadyana, mušcana, musana, moksāyana (from denom. moksāyati), raiyanā, rundhana or °nā, tilapyana (from intensive stem), vijahana or °nā, vipašyanā, vibuddhana and vibudhyana, srunana, samāpadyana, sampralicehana.

22.9. Nouns of agent. As in Sanskrit, the suffix is also productive of new nouns of agent (or adjectives with meaning resembling pres. pples.) in BHS, but they are less common than nouns of action; examples: djñāpana, dadana, muñcana (?), vijnāpana, vibhrāmana.

Suffix ānī, and feminine suffixes ayanī, āyanī, āyanikā, (ā)vinī

22.10. Renou, Gr. Scte. § 212 a, notes the BHS fem. Sākiyānī, which follows an ancient Skt. model; cf. kşalriyāņī. As names of scripts, yonānī (cf. Skt. Gr. yapanānī) and šakānī are recorded. On Skt. araŋyānī, Renou § 204, may be directly based BHS banānī, 'vast forest'. More isolated (cf. Renou § 212 b) are the feminine forms Sākyayanī, Sākyāyanī (or ^oyinī) and ^onikā; and sākunikdyinī, 'female bird-catcher'.

Suffix ara (for ala?)

22.11. Pischel 595 notes the secondary suffix *āla* in Pkt. The same or a related suffix seems to occur in *polāra*, 'boat' = Skt. *pola*. See also *polalaka*, § 22.47.

Suffix (ā)vin, see Suffix vin

Suffix ilar

22.12. The agent suffix (*i*)lar is added, in this form, to the present stem of a verb, functioning in MIndic fashion as 'root', in *vibhajilar*, which would be in Skt. *vibhaklar*. The result looks like a set for an anit form, but is actually, of course, only a late MIndicism.

Suffix (i)ma

22.13. The suffix ma was used in Skt. in forming contrastive adjectives with superlative force, and some ordinal numerals (Whitney 474, 487d, e). One isolated superlative has been noted in BHS, from a word of a type which in Skt. would use *lama* instead: spasta-ma 'most clear', in a verse, guaranteed by the meter. In addition,

BHS has two or three contrastive adjectives in which ma is affixed to certain or possible bases ending in *i*, like Skt. *ādi-ma*, but which are not Skt.: *upari-ma* (Pali id.), 'upper, topmost', to *upari; prsthi-ma* (unparalleled elsewhere), 'rear, hindside', which may be derived from BHS prsthi rather than Skt. prstha; and hestima (Pali hetthima), 'lowest, bottommost', a semi-MIndie form, cf. BHS heste, hesti, AMg. hetthi. 22.14. In normal Skt., ordinals from the cardinal

22.14. In normal Skt., ordinals from the cardinal tens, viņiali etc., use lama rather than ma. But in MIndic, ma is used instead, affixed to stems in i: Pali visatima, liņsatima, etc. This usage is reflected in a few late Skt. forms, such as lriņiatima, saplatima, navatima (Wackernagel-Debrunner III.409). In BHS I have noted ekanavalima, '91st', and several blend-forms in which ima has been affixed by 'clipping' to stems which did not end in i, under the influence of such forms: pañcāiima '50th' (cf. § 19.36), fatima (Pali also satima) '100th', and sahasrima (Pali sahassima) '1000th'.

22.15. This same ima also forms contrastive adjectives, being applicable to stems not ending in *i*, originally without doubt by analogy of forms in stem-final i plus ma. This seems to me indicated by Skt. paścima (com-mon) and antima (rare), both 'last', which I believe were formed in imitation of their antonym ādi-ma, 'first' (perhaps RV. agrimá 'first' also belongs here); no base *paściexists, but only pasca(t), and antima is certainly a derivative of anda 'end', not of the old adverb anti. This usage spread in MIndic. So we find BHS adharima, 'lowest', to Skt. adhara; abhyanlarima = Pali abbhan°, to abhyanlara; uparisfima = Pali uparifihima, cf. Skt. uparisfát; pārima (Pali id.), 'further' (only applied to words for 'bank, shore'), from Skt. pāra 'further shore', and its opposite a-pārima; purastima (Pali puratilhima), 'eastern', to Skt. purastāl, BHS purasta, purastalāh, adv., Pali puratilhām; purima (Pali id.), also ^omaka, 'eastern, fore, front', likewise 'former, ancient, first', respectively to Skt. puras and purā or MIndic equivalents, cf. BHS pure (Pali id.) and m.c. puri; pratyantima (Pali paccantima), 'of the border, frontier', to Skt. pratyanta. There is also a non-contrastiveadjective samcārima (AMg. id.), 'moving, movable', to Skt. samedra, which resembles the Skt. forms cited in Whitney 1224a and may perhaps be based on one or another of them, or on their general pattern.

22.16. Finally, as in MIndic, we find a number of forms in which Skt. contrastive adjectives in (a-)ma replace the vowel a before ma by i; cf. Geiger 19.1, Pischel 101, 602. I do not believe with these scholars that this is a regular phonemic change, determined by original accentual conditions. Rather, these forms illustrate the further spread of ima as contrastive suffix. They are adapted to imitate the endings of such forms as Skt. paicima, BHS and Mindic pärima, purima, etc. The cases noted are ullima (Pkt. id., Pali with Skt. ultama), 'highest'; kalima (Pali Id., Mähäräştri kaima), 'which (of several)?' for Skt. kalama (not to kali); carima and 'maka (Pali Id., both), 'last', for Skt. carama; madhyima (Pali majfhima), 'middle, middling', for Skt. madhyama, and (with Mindic consonantism) anumajjhima (Pali Id.). Once parima is recorded, but it is clearly a corruption for pārima, above (not = Pall parima, Skt. parama).

Suffix ila

22.17. This secondary suffix (cf. Whitney 1227a; Pischel 595) occurs in a few adjectives unknown to Skt.; it seems to be fundamentally MIndic: kuşihila 'leprous', to kuşiha'; sukhila and duhkhila(-lå), to sukha and duhkha.

Suffix ika

22.18. A secondary suffix of this form, with the vyddhi of secondary derivation, occurs in *sailika*, 'made

of stone' (in a verse, but metrically indifferent), to Skt. *šilā*. Cf. Whitney 1186c (but these forms are hardly comparable).

Suffix ina

22.19. The word *śdlina* occurs as an adj., epithet of *odana*, 'porridge'. Its derivation is not certain, see Dict.; it seems to mean 'rich, fine'.

Suffix iya (aniya)

22.20. The suffix iya is still alive in BHS, making new formations unknown to Skt., not only gerundives in an-Iya, but others. Some of the latter, tho based on nouns in ana, can hardly be called gerundives; some are not based on such nouns at all. The use of the suffix does not differ in principle from that of Skt. as described in Whitney 1215. The forms have meanings like 'connected with, characterized by, conducive to' whatever the primary noun means. Sometimes they interchange with equivalent stems in ika (-yānīya = -yānika; āveņiya for usual āveņika). So far as I have noted, BHS shows more extension of the iya suffix than any MIndic dialect. For example, BHS -bhdgive does not seen to occur elsewhere, and is represented in Pali by -bhāgiya; -yānīya corresponds to Pali -yāniya or -ydnika (not, or course, used in the same sense as BHS -yāniya). Should sahiya, 'associated' be classed here? In § 34.23 will be found cited from LV 52.6 ff. and 411.8 ff. a list of (an)lya formations, some but not all of which may be called gerundives. Among them is premaniya 'lovely', which is fairly common and which might be analyzed as preman- plus iya. We find also ksemaniya 'healthy' (and as noun 'weal'?), which must be somehow connected with ksema 'weal, comfort', perhaps blended with ksamaniya, q.v. in Dict.; another possible guess would be that it imitates premaniya. See further, in Dict., anulapyaniya, kaliya, kulsaniya, jananiya, jälijarāmaraniya, parāyanīya, bhīsanīya, rajanīya, vartmīya, -liriva-ka. vijnāpaniya, śrāvakiya, samvartaniya, samharşaniya, harsantua.

Suffix ka (see also aka)

22.21. New adjective formations in ka, of the type referred to in Whitney 1222c, are extremely rare. I have noted frāmaņaka (= Pali sām^o; with vrddhi), 'monklsh', and possibly sthitika (= Pali *hitika*), 'characterized by stability' (but this latter is perhaps better analyzed otherwise, see Dict. s.v.); perhaps saumanasyaka (next paragraph).

22.22. One or two peculiar uses of the suffix have been erroneously alleged. Senart i.522 on Mv i.182.15 claims an abstract-forming force. But his saumanasyaka, if a noun, = Skt. saumanasya (if an adjective, see Dict., it belongs to the preceding paragraph); kimkaranlyaka Mv i.211.3 = Pali and BHS kimkaranlya; and his other examples are all explainable without recourse to such a theory. Again, in i.557, on Mv i.232.7, Senart assumes that ka has specifically future meaning; but dgalaka here is an instance of 'specifying ka', § 22.39, and I find no evidence to support Senart's theory.

22.23. ka swarthe. Most of the BHS ka-formations, not recorded for Skt., belong to ka swarthe, that is, used in the same meaning as the primary word to which ka was added. This usage is well recognized in Skt., and is still commoner in Pali and Pkt. (in the latter, the suffix usually appears by phonetic change as ya or a, rarely ga or even ka).

22.24. Use was certainly made of this practice for metrical purposes in verse, not only in BHS but also in Skt. Many of our examples are from verses, and these could be assumed to be m.c. But prose examples are also extremely common. It has not seemed to me worth while to separate them systematically; certainly meter is not primarily the reason for the extensive use of such forms. It may, however, be noted that in a few cases forms which regularly end in ka seem to be used without ka, possibly for the sake of meter; yamānia = regular yamāniaka, repeatedly in Mmk; kešondu = kešonduka, and sivapathi = sivapathikā, each once in Lank; asecana for"naka; istā = istakā.

22.25. Strange is Mv i.181.18 tām yācamānām prānjalikām, the first half of a śloka line; to be sure, prānjalīm in itself would not be as good, metrically, as prānjalikām in this position; but since five syllables precede, the addition of the suffix makes the line as a whole hypermetric and decidedly bad.

22.26. The following examples of ka svārihe, not an exhaustive list, are taken from my Dictionary. Nouns: ajinaka, antevāsika, ašruka, āsanaka, upakanļhaka, kandaraka, kiņkaranīyaka, kaušalyaka, khalaka, gatikā, grhapatika, cīvaraka, caityaka, jñātika, dukūlaka, dvīpaka, patāšaka, pindāraka (plant name), pītaka, prāsādaka, plīhaka, balaka, bhājanaka, landaka, vedayitrka, śresţhika, sakhikā (endearing dim.?), sārathika, sārthavāhaka, sūkarikā, soāmika.

22.27. Adjectives, general, uncompounded: acintyaka, Irsyāluka, kevalaka, g(i)lānaka, jīrnaka (pitying dim.?), diārksuka, dīrghaka, duskaraka, pašcimaka, pāļalaka, prthuka, pradaršanaka, pradīrghaka, balika, vartulaka, vāmaka, vidusaka, šītaka, šītalaka (also as noun), šūnyaka, sobhanaka, fresihaka, sarvaka, sūcanaka, saumyaka, sauvarņaka.

22.28. Comparatives in tara: utkrsfataraka, udārataraka (see AsP 373.3 fl., prose, for a group of such forms), ciratarakena (adv.), jyesthataraka (°ikā), drāhapraharanataraka, nihinataraka, (prativišistataraka § 22.39,) bahutaraka, vrddhataraka, śūrataraka.

22.29. Present active participles in BHS frequently take extensions in a; to this a our ka sourthe may be added (and once, if the text is right, *i* is substituted for this a; gacehanlaka, *itshantika* (! for expected °aka), dadantaka (°ika), nudantaka, rodantaka.

Present middle participles: gacchamānaka, bhāşamānaka (°ikā).

Gerundive: roditavyaka.

Past passive participles: a-jālaka, atilaka, adhirūdhaka, abhiyuktaka, abhirucitaka, āgataka, āmuktaka, āsannaka, nisannaka, pramatlaka, pramuktaka, prākrtaka, mūrchitaka, mraksitaka, lakhaka, likhitaka, vinyastaka, sansaktaka (pej.?), samprasuplaka, sammūdhaka, sahilaka, suplaka (°ikā), sthitaka.

22.30. This 'meaningless' ka is specially common at the end of bahuvrihi compounds, even in Skt. The following do not seem to be recorded in Skt.: -figatika, -ficdryaka, -ddharsaka, -fipattika, -dlambaka, -indraka, -utpattika, (ntr)-upddätrka, -updhanaka, -ullocaka, -rddhika, -ojaska, -krtyaka, -ksäntika, -gatika, -cittaka, -daśaka (daśd 'tringe'), -drşika, -drawyaka, -nemika, -parşatka, -pradharşaka, -buddhaka, -buddhika, -brāhmanaka, -bháryðka, -māradharşaka, -mūrdhaka, -wijnānatka, -ijīnānaka, (Hiranyadhanya)-širika (v.l. -śirika; from śri), -sūcika, sainyaka, -svabhāvaka.

22.31. Indeclinable: kadācikāt = kadācit, Mmk 340.12 (verse); bhavet kadācikāt siddhih, 'success may sometimes ensue'. Perhaps merely m.c.; or quasi-diminutive, see Dict.

22.32. Diminutive ka. Very hard to distinguish clearly from ka sparthe is the use of the same suffix in various shades of diminutive meaning. The same is true of Sanskrit. As usual I quote here only words not found recorded in Skt. in the same meanings. It may be that a number of cases conservatively listed by me under ka sparthe really had diminutive meaning of some kind. In general I have included here only what seem to me fairly clear cases; but the interpretation is necessarily subjective.

22.33. Simple diminutives, without any very special emotional tinge, may be seen in gharakehi Mv 1.22.10 (prose) 'in little huts' (so Senart); rathaka 'toy-cart', also aśvaka, hasiika, 'toy-horse, -elephant'; siûpaka.

22.34. In the numerous ka-forms which decorate the verses LV 321.19 fl., verses spoken by the daughters of Māra in their attempt to seduce the Bodhisattva, it seems reasonable to see endearing diminutives. They are applied to all sorts of nouns and adjectives; not only to the speakers themselves and the Bodhisattva, but to the spring season (321.19), and in short to everything concerned in what they hope to make into an amorous situation. In the ten stanzas the following occur (in Lefm.'s readings): suvasantake, āgatake, phullitapādapake, sušobhanake, citritake, su-samsthilikāh, su-samtutikāh, su-yauvanikam, mānasakam, su-(a)lamkrtikā, āgatikā, sositajīvitako (this form perhaps pitying dim.?), -gandhinikā, -ānanikā (three times, 322.10, 11, 13), -su-locanikā, (-nibh-)ādharikā (adhara), -dantinikā, rati-lālasikām, udgatikām, su-sundarikām, su-vittharikām, su-kāminikām, - Ēruņikām (see § 22.45), -bāhanikām (or -bāhun°), -su-maņģitikām, dāsinikām, -gāminikām, -bhāsinikām (read -bhāș°), -su-bhūșinikām, su-pandilikām, -su-sikșitikām (°śikș°), -su-rūpiņikām, -su-lālasikām, su-vañcitako (pitying?), ajānaku and ajānanako (pitying?), āgatiko (323.14, read °kām), kāminikām: forty cases. An average of four ka-forms per stanza is not likely to be accidental or meaningless.

22.35. Probably bhadraka Divy 38.15, 23, 30, snigdhaka 16, and snehaka 24, 31, are affectionate diminutives in tone: 'good, kindly, affectionate, gentle'. With change to -ika, Pūrnika and Supūrnika are hypocoristic diminutives of Pūrnamukha (2 in Dict.), MSV ii.105.4.

22.36. The pitying diminutive, associated with affection, was exemplified several times above. It may be present also in Mv ii.173.10 so asmākam ekaputrako kālagalo, 'our one poor dear son is dead'. Pitying force seems also present in duhkhilaka Divy 84.1; 86.22; bubhukşitaka, 'hungry (poor fellow!)' Divy 88.7; muşitaka 'miserably robbed' Divy 175.23; probably also in Divy 334.1, 3 utpānātipānaukah kršāluko durbalako miānako (similarly Divy 571.11, 18, 20, 21, in 11 omitting the first word, in other instances reading durbalo and/or miāno; in 334.1 mss. kršālako, which is the reading Mvy 8812); possibly also nagnakā (of ghosts, preta) Mv i.28.8; jivantikā Mv i.353.19 '(abandoned) alive (poor creature!)'.

22.37. A clear case of pejorative ka is sahāyaka in AsP 417.20 fl., 'inferior companion', contrasting with pratirūpah sahāyo, 'a suitable companion', see Dict. Similarly Gv 500.5 (prose) frāvakapratyekabudha-kāca maņikān, 'wretched glass jewels', contrasted with sarvajāatācittotpādānarghamahāmaņiratna- (lines 2-3); LV 326.21 kim muniveşakena bhavalo, 'of what use to you is an ascetic's wretched garb?'; muņdaka iramanaka, 'wretched shaveling monk' (Dict.); krşnaka, 'nikā, '(repulsively) black' (of complexions; definitely derogatory); perhaps paribhūtaka 'wretchedly despised'; asāraka, 'worthless, empty, vain'; atiriktaka, 'quite empty' (with contemptuous implication).

22.38. Specifying ka. In quite a number of cases I am certain that the ka-suffix adds to the primary word a sense of individual specificity or definiteness: something like 'the one who is'. The same usage is found in Pali, e. g. Jat. i.222.3 macchānam cinianaka-bako nāma, 'such a thing as a heron that is one who concerns himself about (the welfare of) fish'. I have not observed this usage in Skt. It is, indeed, the one and only clear extension of the use of the suffix ka known to me in BHS (and Middle Indic).

22.39. It is often used in a supplementary reference to a condition previously mentioned; at the first mention of it, the same form without ka may be used. A good BHS

> **A** 1997 - 1997 - 1997 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 -1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 -1997 - 1997 -

22.48]

example is Mink 82.25-26: sarvāmānusadusļadārunagrhileşu ālmano hastam aslašatābhimantritam krivā grhilakam mastake spriset; svastho bhavali, 'in the case of people seized (possessed, by demons, grhitesu) ... one should touch the possessed person (grhilakam) ... on the head, and he will get well'. Also nagarājā... ahitundikena grhitako, tam änetha Mv ii.179.9, 'the näga-king, the one that has been captured ..., him bring'. Similarly nirmitakā(v) KP 142.4; 143.4 fl., 'the two ones that had been fashloned by magic' (as previously described in 141); samyaksambuddhena yo nirmilako nirmilo AsP 442.15, 'a magically created individual that has been magically created by a Buddha'; sa nirmitako yasya krtyasya krtaso (= °so) nirmilas tat krtyam karoti ibid. 443.1, 'that magically created one (form; nirmitako) does the act for the sake of which he was magically created (nirmito)'; others in Dict. s.v. nirmitaka. Similarly Mv i.353.12-13 sā šākyakanyā guhāyam nihilikā, 'the Sākya-girl that had been put away in the cave' (as described in 352.21); in i.353.13 tam pāmsurāšim pādena apakarsitam, but in 15 tat pāmsu apakarşitakam drştvā, 'secing that dirt which had been removed'. Others (see Dict.): anupari(l)laka, anupradallaka, āgalaka (Mv i.232.7), ābaddhaka, āhalaka, uddistaka(?), upanişannaka, kāritaka(?), khāditaka, gataka, glānaka, dattaka, daştaka, nimantritaka, niyuktaka (?), niryälitaka, purimaka, prajňaplaka, pratipannaka, pratipädilaka, pra-tivišistataraka, pratyuptikā(?), prathamaka, pravistaka, pravritaka, vrddhalaraka, šayilaka, sampannaka, siddhaka, sthāpitaka. It is possible that some otner cases, listed by me under ka svärthe above, may really belong here. Like the diminutive ka, this usage is not always easy to distinguish, and at times the distinction is bound to seem subjective.

Suffix tara

22.40. The suffix *tara*, in Skt. comparative, is used in BHS as superlative. Perhaps this use began with anyatara, which in BHS commonly means 'other' or 'one' of more than two (= Skt. anyatama), and katara 'which' of more than two (= Skt. katama). But once, at least, other adjectives in *tara* are used with definitely superlative meaning: kumārānām daršanīyataro prāsādikataro Mv ii.446.9, 'the handsomest and most attractive of the (500) princes'.

Suffix tā

22.41. The abstract-forming suffix tā may be added in BHS to nouns which are themselves abstract, so that there is no change of meaning. (The same occurs in Pali; to the common abstract issariya, nt., 'lordship', is formed also issariya-tā, id.) It is specially common after nouns in -ana. Attention may be called particularly to the dharmālokamukha section of LV (31.8-36.5), which contains many such forms. To be sure, a good many, perhaps most, of the forms in this section occur at the end of compounds, and it would often be possible to interpret them as abstracts in which -tā is added to a Bahuvrīhi compound. For example, amoghadarsanatā 32.17 is probably better analyzed as amoghadarsana plus ta, 'state of having unfailing vision', rather than as amogha plus darsanatā = darsana. But this section, like others, contains some forms where this interpretation is impossible, e. g. jñānalā 32.18; arthanatā LV 393.13, which are not at the end of compounds. And the usage is confirmed by the fact that tā is added to other abstracts than those in ana, often not at the end of compounds. While, therefore, some of the following list may be better interpreted in the alternative way just mentioned, anyone who looks up the occurrences (all listed in Dict.) will have no doubt of the real and frequent use of $l\bar{a}$ 'svärthe' after abstract noun stems.

22.42. The following cases occur after ana stems: adhivāsanatā, adhyavasānatā, adhyālambanatā, anubodhanalā, anubhavanalā, anurakṣaṇalā, anulomanalā, anuvarlanalā, apakarṣaṇalā, (an-)abhiniweśanalā, abhibhāvanalā, abhimanyanalā, abhiśraddadhānalā, arthanatā, avanāmanalā (? see s.v. unnāma), avamanyanalā, ākarṣaṇalā, ākramaṇalā, ādhāraṇalā, āsanalā, āharaṇalā, (an-)ulkarṣaṇalā, jugupsanalā, jñānalā, darianatā (? see above), deśanatā, parijñānalā(?), (a-)parilasyanalā, paripācanatā, parivarjanalā, pratifaraṇalā (? LV 32.11, 33.2), pratyavekṣaṇatā, pravartanatā, prasādanatā, bhāvanatā, yācanatā, varlanatā, vidhamanatā, (a)vimardanatā, (a)visaṃvādanatā, (a)śraddadhānatā, saṃdaršanatā, (a)saṃdhūkṣaṇatā, saṃprakāšanatā, sevanatā.

22.43. In the same way $l\bar{a}$ is added to other, miscellaneous abstract nouns, without changing the meaning so far as I can see: avidyalā (= avidyā), ādhipateyalā, kāruņyalā, khedalā 'lassitude' (= kheda), gauravalā(?), trplitā 'satiation', dainyalā, daurmanasyalā; probably also himsatā (= himsā).

Suffix that or atha

22.44. This primary suffix is used precisely as in Skt. (Whitney 1163c), forming nouns of action. It seems to have remained productive in MIndic; most of the new forms noted in BHS are paralleled in Pali: klamatha (Pali kilamatha), 'fatigue'; damatha (Pali id.), 'control'; vadatha (unknown elsewhere), 'speech'; vanatha (Pali id.), 'desire'; samatha (Pali sa°), 'tranguillity'.

Suffix nikā

22.45. Several times in the passage LV 321.19 ff., referred to above § 22.34, we find (included in the list l.c.) feminine forms in suffixal nikā, used like the adjoining ka-words with endearing diminutive force: $-\bar{a}runik\bar{a}$, 'having... thighs', to $\bar{a}ru$; $-b\bar{a}hanik\bar{a}$ or $-b\bar{a}hunik\bar{a}$, 'having... arms'; dāsinikā = dāsī, 'female slave'. They seem to be formed by 'clipping', in imitation of fem. forms in (*i*)nikā, from fems. in *-inil* to masc. stems in *-in*; the context contains quite a number of such, as *-gandhinikā*, *-dantinikā*, kāminikā.

Suffix ma, see Suffix (i)ma

Suffix mant, see Suffixes vant and mant Suffix maya

22.46. This suffix, in Skt. denoting material, 'made of', occurs once in a peculiar meaning, 'made, performed by', in *stri-maya*, (with *tûrya*, 'instrumental music'), 'made, played by women'.

Suffix min, see Suffix vin Suffix la (and laka)

22.47. On the basis of AMg. hallhala, thief', I assume that haslalah should be read for the corrupt hanlulah, haslalah should be read for the corrupt hanlulah, haslalah should be read for the corrupt hanlulah, haslalah, of Divy 646.21. It seems to derive from hasla with suffix la, 'handy' in a curious pejorative sense, probably originally slang. In polalaka, °ikā, 'young (of an animal)', = Skt. pola(ka), we may have a simple diminutive use of la(ka); but cf. AMg. poāla, pogāla, same meaning, and go-polālikā in BR 7.1740; also Suffix āra, $\S 22.11$.

Suffixes vant and mant

22.48. The suffix mant seems to be added to an instem without change of meaning in (artha-, bhūta-)daršimant; cf. Pali (attha-)dassima(t), CPD, the same word. Senart's note on Mv i.184.16 is misleading. He says that both Skt. and Pali have only daršivant. It is doubtful whether either has any form of that stem. The Skt. daršivān, n. sg. (only form recorded), is derived by BR from daršivāms-; and Pali probably has no dassivā at all (some editions read bhaya-dassivā Dhp. 31, but others °dassi vā, which seems more likely), and certainly no other form of *dassiva(!). Whether the *i* in putrimant (= Pali putli[°]), My ili.417.15 (in a verse = Pali Sn 33), is in some way due to the influence of daršimant, or of other (Skt.) stems in mant preceded by *i*-, may be doubtful; in any case I cannot accept Geiger's theory (19.1) that *i* is a phonetic change from *a*.

22.49. The suffix vant is substituted for regular Skt. mant in bhänuvant LV 357.9, and in mativanta subuddhivanto LV 299.7 (vs), immediately following pratibhänavanta, and perhaps influenced by it. Note, however, Ap. maivanta and vuddhivanta, Jacobi, Bhav., Glossar.

22.50. Like mant in daritmant, above, a few forms in vant seem to be equivalent in meaning to the primary word without vant. Very common is sarvávant 'entire' = Pall sabáva(t); but it is not quite equivalent to sarva. Kern SBE 21, xvil cites it as occurring in SB 14.7.1.10 (= BrhU 4.3.10); but there it means rather 'containing everything' and is a regular vant formation, not in the least equivalent to the BHS and Pall form. In SP 341.10 asfavat- (in cpd.) seems = asfa 'eight'. In LV 283.7, repeated in 19 (vs), arhavant = arhant; there is no v.l. In several Gv prose passages we find forms in -vali (fem.; sometimes written -vali, perhaps by mere textual corruption), apparently without change in the meaning of the primary noun: abhijňávali apparently = abhijňá: -simá-

vali- (read °vali°? = sima); sarvadharma-sāgaravali- (apparently = °sāgara; read °vali?)-garbheņa. Note that in the last the underlying noun is masc.

Suffix vin (or āvin), and min

22.51. The suffix vin, before which stem-final a is regularly lengthened (often also in Skt., Whitney 1232b), has a productive life in BHS as in Pali; it is particularly (the not exclusively) added to past participles in la, in such a way that -lavin takes the place of Skt. -lavant, also as a sort of periphrastic perfect. (It may be noted that LV 101.13, 136.16, prose, Sanskritizes regular BHS and Pali vijilävin into vijilavani; similarly Lank 241.3 caritavani = caritävin.) In a number of cases the same words found here are also found in Pali. So far as I can find, no such usage is recorded in Pkt. The forms noted are: after pple. stems in la, anujñātāvin, abhisamitāvin, ājñātāvin (Pali aññā°), krtāvin (Pali kat°), caritāvin, parijnātāvin (Pali parinnātāvin), paryupāsitāvin, bhuktāvin (Pali bhultāvin), rutāvin, vijitāvin (Pali id.), samitāvin (Pali samo, so also BHS as n. pr.), *srulāvin*; other forms, aghāvin (Pali id.), daršāvin (Pali dassāvin; common both here and in Pali). Once a form in min replaces a regular Skt. form in vin: sragmin, thrice in Mmk (Dict.), 'garlanded', for sraavin.

23. Noun and verb compounds

Dual dvandvas

23.1. Dual dvandvas, like other duals, are often declined as plurals: $m\bar{a}l\bar{a}pil\bar{r}n\bar{a}m$ LV 231.17 'of (my) father and nother'; candrasūryānām LV 310.21 'of the moon and sun' (cf. 17, just before, candrasūryau); śakrabrahmānah LV 316.13 'Sakra and Brahman'; candrasūryāh LV 30.18; candrasūryānām LV 390.9 = Mv iii.308.19; candrasūryehi Mv iii.310.1.

23.2. Neuter sg. dvandvas are sometimes used tho composed of masc. and fem. nouns, and not necessarily meant collectively; plural adjectives may agree with them. So: na hi bhūyu mālrpit draksyati pulradārām LV 190.20, 'he will not see again nother or father, son or wife'; mālāpitr nitya varnilā (so divide) pûjanīyāh (mss., Lefm. em. 'niyāh) LV 235.14, 'mother and father are declared to be always worthy of reverence'; chandaka-kālodāyi Mv iii.91.14 (prose), 'Chandaka and Kālodāyin', here nom. sg.; same as voc. sg. iii.91.16; 92.3. Note also mālāpilaram, nom., Bbh 389.20 (prose); cf. next paragraph.

23.3. But, as in non-standard Skt. (Wackernagel II. 1 § 70), a singular dvandva may also have the gender and inflection of the last component: *ārohapariņāho*... ^ophalopelah SP 246.8 (prose), 'the height and circumference were provided with... fruits'; *hiri-dhrll-kāruņyamaitryā* (so read, 4-member dvandva, instr. sg. fem.) LV 292.17; candrāsūryaš ca LV 316.13; *mātāpilaraņ* acc. sg., Mv i.109.1; ii.211.17; 212.2, 3; Gv 520.23; ambātāto Mv i.181.3 (with sg. verb); caļila-sphuļilakāh prādurbhūtān blivy 22.27 'cracks and fissures appeared' (but in 23.6 the more normal caįilasphuįilakāni prādurbhūtāni); uccāraprasrāvah Divy 194.15 'dung and urine'; dhanuskalāpakaņ Mvy 5005, 'bow and quiver (of arrows)'; aja-r-edake (Dict.). A common dvandva of this type is avavādānušāsanī; see Dict. s.v. avavāda.

23.4. Contrariwise, note-the nt. pl. mālāpilīņi (mss. ^otrņi) Mv ii.173.17; nānādrumam ašokātimuktacampakapriyangušālāni iii.79.20 (vs).

Inflected forms as prior members

23.5. A prior member sometimes appears in the nom. dual or plural, instead of stem, form (cf. āpomaya 'made of water' ChU. 6.5 ff.): apsarasaḥkoļiniyulāšatasahasrasamgilisāmpracalitā LV 10.22 (apsarasaḥ- can only be part of the cpd.; no v.l.; not mentioned by Weller: '... the song of myriads... of apsarases'); -varuņavāsavāśvinaunāga- LV 249.16 (in a long cpd. listing many deities, including 'Varuṇa, Văsava, the Aśvins, nāgas' etc.; Lefm. leaves a space after -āivinau, but the word can only be construed as part of the cpd.); vāmadakṣiṇau-pānibhir LV 295.4 (so ms. A, the best; Lefm. with other mss. vāmadakṣiṇe, doubtless to be understood also as dual, fem. or nt.) 'with the left and right hands' (must be cpd., tho Lefm. separates after °ne; pañcabhir nilavāsasoyakṣāstair Divy 238.23 'by 500 Nilavāsas (-named?) yakṣās' (cpd., tho ed. separates after °bāsaso).

23.6. Even a bahuvrihi may contain such a nom. pl. form as prior member: *pañcayojanaśalāny-ārohaparināham*...simhāsanam SP 246.9 (prose) 'a throne having a height and circumference of 500 yojanas'. Note the gender of

āroha-parināham; it is one of a series of neuter modifiers of *simhāsanam*; what precedes it can only be part of the same cpd.

23.7. The occasional use of nom. sg. forms as 'stems' in composition has been noted under a-stems, § 8.12, and u-stems, § 12.4. One or two sporadic cases which could be interpreted similarly may be added here; but they may also be understood as containing 'hiatus-bridging' consonants, instead of nom, sg. endings. Thus, hrirapatrāpya, for more regular. hry-apatrāpya; it occurs several times, but may contain 'hiatus-bridging' r (for hrī-apa^o), see § 4.63. So also in LV 430.8, as part of a long cpd., -*šite* (loc.) -usnodakam-usae (loc.)-*šitodakachāyālapa* (loc.)-rtusukha-paribhogānupradāna- etc., 'gift of seasonal enjoyments in shade and sunshine, consisting of warm water in cold and cold water in heat', etc.; here m at the end of usnodakam may be 'hiatus-bridging' m.

23.8. A few times, in feminine karmadhārayas, the prior member (an adjective stem) seems to be made fem. (interpretable either as stem or as nom. sg.) as if agreeing with the final member (cf. Wackernagel II.1 p. 52): prasravanli-lālābhih LV 206.7 (prose; ignored by Weller) 'with flowing saliva'; perhaps -visesanā-sadrši-lokādhipale-yatām LV 179.20 (prose; Weller 28 separates after -drši, taking the latter as acc. sg., \S 10.55); perhaps anutlarā-samyaksanbodhim abhisambuddhasya LV 439.20 (prose; Weller 39 again, as in prec., reads anutlarā as a separate word. acc. sg.; see \S 9.21).

word, acc. sg.; see § 9.21). 23.9. Compounds with inflected form (other than nom. sg., see above) as prior member, not occurring in Skt. (cf. Whitney 1267 c, 1269 b), include *sürangama* (Dict.); *-svabhāvaŋsamdarśanavibhavacakram* LV 422.19, 'wheel of destruction thru showing the nature ...' (so apparently all mss., only Calc. *-svabhāva-san*^o); *-sugandhatailamūrdhniparisiñcanam*, LV 432.15, 'sprinkling on the head with perfumed oil' (so Lefm. with best mss.; v.l. "mūrdhna^o; in preceding line, mūrdhnā may be taken as separate word, tho Lefm. prints it as if cpd. with the following); perhaps yāvad-trīni-varşa-janmika, 'up to three years old', see Dict. s.v. janmika.

Irregular order of parts of compounds

23.10. Cf. Whitney 1314d; not all the examples there listed are sound. Seems to occur in the following:

šilosnadve vāridhāre LV 83.22 (prose) 'two streams of water, cold and hot'. So all mss., confirmed by Weller's (23); Calc. *šilosne*. The cpd. *šilosnadve* is repeated in the verse account of the same incident, 93.3.

câmarasubhe LV 84.18 (prose) = subhe câmare. Not noted by Weller.

rșijirno LV 102.12 (prose); Weller 24 implausibly suggests separating rși, as MIndic nom. sg.

? marud-visuddhā(h), for visuddha- (or visuddhāh) marulah LV 113.19 (vs); but see § 15.16.

sarveyam prihivi prakampila-m-abhūc śailā (so Lefm.) sakūtāvatī LV 194.9 (vs) (read °vatī with v.l.); I can only understand śailāsakūtāvatī as a cpd. (śailā- for śaila-, m.c.), 'all this earth including the mountains-along-withtheir-peaks ...'; normal order sakūta-śailāvatī. Tib. supports this mg.: sa hai thams cad rise mor idan pahi ri dan bcas le gyos par gyur.

gandharvabahubhih Mv i.145.6 (vs) (Senart em. °bhi, m.c.) 'by many gandharvas'.

Adjectives agreeing in sense with prior member

23.11. Sometimes an adjective agrees, rather bizarrely, with a prior member of a cpd.: sadevamānusāsu-rasya lokānukampām upādāya I.V 6.20-21, 'taking pity on the world consisting of gods ...', as if lokasya were read (Weller 15); cāsya (! delete preceding punctuation?) putrasahasram bhavati sūrānām vīrānām etc. (sc. putrānām) LV 18.6; closely similar are LV 101.16-17; 136.19; and My i.193.17–18; patnīšuddho 'parikrstasampannāyāh (sc. patnyāh) LV 26.6-7; ādīpta sarvatribhavam tribhir agnitaptam, samkalparāgavisayāraniucchritena (so, one word) LV 116.13-14 (vs), 'the whole triple world is aflame, blazing with the three fires (sc. agnibilit, in agni-), increased (sc. agninā, in the same, here understood as a collective unit, cf. § 5.5) by (or, arisen from) the firesticks of ...'; tāsām (sc. strīnām) caturašīteh (so read, Weller 27) strīsahasrāņām LV 157.8; pañcasu kasāyakāle LV 257.21, 'in the time of the five corruptions', the numeral here attracted to the case of *-kāle* tho agreeing in sense with kaşāya-; lokottareşu catusatyapathe vidhijño LV 125.1, 'knowing the rules in the sphere of the Four Truths, (in the Truths) which are superworldly' (again with attraction of case to that of -pathe); so cilladhāra jagatām vividhā vicitrā (sc. cillāni) LV 125.9, 'he holds firmly the thoughts, various and manifold, of creatures'; jñātisaṃghaparivrtaṃ sarvai rudadbhiḥ krandadbhiḥ etc. LV 190.10 ff., 'surrounded by a throng of relatives, all weeping' etc. (some inferior mss. 'samphaih pari'); trimsadbuddhakoliyo śākyamunināmadheyanām (sc. buddhaānām) Mv 1.57.8, '30 crores of Buddhas named Sākyamuni'; asfau gangānadīvālukāsamā bodhisattvā(h) SP 297.1 (prose), 'eight Ganges-river-sand-like Bodhisattvas' i. e. eight times as many B's as there are sands of the Ganges.

Compounds of type phalāphala

23.12. BHS has some compounds of the type of Pali phalaphala 'all manner of fruits'. Their interpretation is disputed, cf. Geiger 33.1. That they started with compounds of positive and negative forms, e. g. phala-aphala 'fruit and non-fruit', seems supported by -koly-akoli- RP 4.15 'crores upon crores', lit. 'crores and non-crores'. The others, like all noted in Pali, are ambiguous in that *a*-stems are concerned. (But *kālākāle* Šikş 167.1 'in and out of season' surely contains a negative second member.) Some are identical with Pali cpds.: *phalāphalam* (Pali id.) Mv ii.475.13 (prose) etc.: *kriyākriya* = Pali kiccākicca; also mañcakāmañcaka, chidrāchidra, svarāsvara.

23.13. The nearest approach to this idiom which I have found in standard Skt. is *colācala* 'unstable' (also in BHS; Ślkş 231.4; Divy, Index), interpreted by BR (following Hindu grammatical authorities) as a primary derivative of a reduplicated form of *cal*. The such a reduplicated stem is otherwise unknown, this may be the true explanation. No certain case of an adjective of this type is known in Pali or BHS. In LV 193.19 (vs) *jihmājihma* '(very) depressed' may have \bar{a} m.c., or *jihmā* may be an independent nom. pl.

independent nom. pl. 23.14. In LV 234.21 (vs; by Lefm.'s em., m.c., for rasa-rasāgra), and elsewhere (Dict.), there is trace of a form rasārasāgra; but probably rasa-rasāgra is the true form (Dict. s.v. rasāgra).

Verb compounds

23.15. The strange form sātmy eva bhavanti Jm 95.9 (vs), 'become quite habitual' (for sātmībhavanty eva), is without parallel. It is noted by Renou, Gr. scte. p. 152, referring to Franke, IFAnzeiger 5.33.

23.16. In one passage, Av i.223.11, we seem to have a clear case of Vedic 'tmesis', separation of a preverb from its verb; na lāvad ul te lekhā langhayilavyā, 'you must not cross over the line' (for ul-langhayilavyā). So Speyer interprets, and I see no other possibility.

23.17. As occasionally in Skt. (Renou, Gr. scte. p. 175), a privative is (rarely) compounded with finite verbs: mām ca apaiyanli SP 324.2 'and they see me not'; an atikramāmo Mv ii8.0.8 'we do not transgress'; sā dāni strī dharmalabdhena anicchiyati (mss. anicchayanti, aniccleyati), Mv iii.295.18, 'now this woman is not desired by Dh.' Senart assumes negative of passive, which is perhaps the simplest conjecture; the form is uncertain tho the meaning is clear.—R. Mitra, LV Calc. Introd. 35, cites (without reference) ajanehi 'do not cause' from LV. He doubtless refers to Calc. 293.6 = Lefm. 237.1, interpreted by Mitra as *šīghrašīghram ajanehi*, but by Lefm. (correctly, in my opinion) as *šīghrašīghram ma* (m.c. for 'ghram mad) janehi.

24. The verb: general remarks

24.1. In general, the entire verb system of BHS, as of all Middle Indic, is based on the present stem. This serves as 'root' for all verb forms. Cf. Geiger 120, end, and Pischel 452, end; but these writers understate the facts, laying this principle down only as a possibility (of frequent occurrence, to be sure). It is the regular process, the it has exceptions.

24.2. That it is, as a rule, precisely the present stem, rather than any other form of the verb, which serves as base for the other forms, seems indicated by the constant use in this way of unmistakable present bases which cannot be confused with Sanskrit 'roots'. All sorts of non-present forms (and noun derivatives, see e. g. $\frac{5}{5}$ 22.7, 12) are made from such stems as pasya-, manya-, iccha-, sida-, tişiha-, jaha-, which originally existed only in the present system; also from Skt. passive stems in -ya-, etc. It is true that other forms, notably past passive participles (in -ila-) and futures, were also concerned in new verb formations. But generally they seem to have operated by first creating new present stems, which then served as bases for other verb forms.

24.3. In Chapter 28 I shall try to classify the non-Sanskritic present stems of BHS, noting the various ways in which they seem to have originated, where possible. The great majority are thematic presents, that is, stems ending in $-a^-$; this $-a^-$ is regularly dropped before endings outside of the present system. There is also a group of forms where the present stem ends in a long vowel, most commonly $-e^-$, occasionally $-\bar{a}^-$, quite rarely other long vowels; these contrast with the -a-stems in their way of making certain non-present forms, notably the aorist. Other types of present-formation hardly exist, except as directly inherited from Sanskrit, and then as relic forms (tho some are very common).

Thematic presents

24.4. These, as we said, constitute the vast majority of verbs in the language. Many are inherited from Sanskrit; but many others do not occur, or occur only as sporadic irregularities, in standard Sanskrit. Historically, most of them can be explained by analogical processes; the origin of some remains obscure to me. Striking is the variety of the analogical processes involved, tho they all result in thematic presents. We may describe it as multiple morphologic 'drift': all the processes tend towards a like goal, but by different paths.

24.5. The regular thematic verb had a present indicative resembling the Skt. present active, with three persons in the singular and three in the plural; an optative (largely undifferentiated as to person and number), an imperative (2 and 3 person sg. and pl.), and two participles, with stems usually in *-anta-* (also *-ala-*) and *-amāna-*. Original middle forms occur, but (except the participle) seem to have been sporadic, and not differentiated in meaning from the active. Even the passive (below) commonly had active endings.

24.6. There was also a future, inflected like the present; an aorist, generally undifferentiated as to person and number (but there are distinctive 3 pl. and occasionally 1 sg., 1 pl. and 2 pl. forms); a past passive participle; a gerund (with four or five interchangeable endings); an infinitive; a gerundive; a passive (inflected more or less fully, like an independent stem); and a causative (even more completely inflected). All these, as we said, were based on the present stem.

24.7. Ignoring sporadic or more exceptional forms, for which see my treatment of each category, we may set up the following paradigm of the regular thematic verb. I choose the BHS stem *iruna*-'hear'. Most, tho not all, of the forms cited actually occur from this verb. Present indicative: sg., *irunami*, *irunasi*, *irunati*; pl.,

śrunāma(s), śrunatha, śrunanti Optative: śrune(t), śruni, śruneya, ^oyā (all used for all persons and both numbers)

Imperative: 2 sg. śruna, śrunāhi; 3 sg. śrunatu; 2 pl. śrunatha; 3 pl. śrunantu

Present participle: *srunanta-* (also *srunata-*), *srunamāna-*Future: *sruņisyati* etc. (like present)

Aorist: śruni, śruni, śrune(!) (all used for all persons and both numbers); 3 pl. also śrunisu, śrunimsu, śrunelsu(h) Past passive participle: śrunita-

Gerund: śrunitvā, śrunitvāna, śruniya, śruniyāna, śruni Infinitive: śrunitum

Gerundive: srunilavua-

Passive: śruniyati, śruniyati etc. (inflected even outside the present system)

Causative: śrunāpayati, °peti etc. (inflected thruout)

Stems in long vowels

24.8. The only other productive type of verb inflection shows stems in a long vowel: most commonly e, frequently ā, more rarely ī or ū or o. Their regular inflection differs from that of the thematic verbs, in that they keep the long vowel before most endings. When in thematic verbs the ending begins with -i-, in most forms this -i- is replaced by, or lost after, the long vowel of the stem. (But the optative, past passive participle, passive, and causative have as a rule no distinctive forms.) So from sthapemi 'I place' we find impv. sthapehi, -sthapetha, etc., fut. sthapesyati etc., ger. sthapetva; from -kramemi, inf. -krametum; from *graheti (fut. graheşyati etc.), gdve. grahetavya-. In the aorist there is the further distinction that after a long vowel the ending contains a sibilant, which is lacking in the aorist of thematic stems: thapesi, -kramesi, -grahesid, and the like. The same holds good in Pali (cf. Geiger 163, 165.2, 167) and in AMg., the only normal Pkt. which retains the aorist, tho Pischel 516 does not formulate it so.

24.9. There is some confusion between forms appropriate to the two types of verb inflexion. Especially is this true of verbs based on presents in *-eti*, many, perhaps most, of which are historically causatives and denominatives. We shall see that, as in MIndic generally, they are not infrequently paralleled by *-ati* forms, and vice versa (\S 38.2, 3, 18 ff., 25 ff.). So it happens that our language may show e. g. a 2 sg. impv. in *-ehi*, or an aorist in *-esi*, from a verb which otherwise shows only *-ati* forms. Such cases, where they are fairly numerous, will receive special

attention in our treatment of the imperative, the aorist, or other verbal categories.

24.10. Stems in -e-. Most commonly this -e- is phonetically derived from Skt. -aya- (or -ayi-; from -aviin fut. bhesyati = bhavisyati). Probably the majority are derived from causative or denominative formations; for others see \$5 28.46 ff. We may regard the following as the regular paradigm, but nothing like a full set of forms is quotable from any verb. Present indic. sg. demi, desi, deli; pl. dema, (detha,) denti

(Opt. indistinguishable from -ati forms: -krameyā etc.)

Impv. 2 sg. -nehi (cf. Skt. dehi); 3 sg. detu; pl. detha, dentu Pres. pple. oddenta-

Fut. -dheşyati etc.; bheşyati (for bhavişyati, special case) Aor. -dhesi

(Ppp. indistinguishable from -ali forms, and usually from corresponding Skt. pples.; cf. -dhita- etc.)

Ger. sthapetvā, -lobhetvāna

Inf. -krametum, -metum

Gdve. grahetavya-

24.11. Stems in -ā-. These are much less common than the preceding group. In the present they follow the type of Skt. yāti 'goes'. Their miscellaneous origins will be summarily discussed below, §§ 28.51-55. The following forms seem to belong here:

Pres. ind. dāmi, dāsi, pradanti (MIndic a for ā?); -dhāmi, -dhāti; prādurbhāmi (to bhū); -sthāmi, sthāti

Impv. bhāhi (to bhī), -sthāhi, apacāhi (§ 28.54; on dadāhi cf. § 28.61); -sthālu; śraddadhādhvam

Pres. pple. sthat- (weak stem)

Fut. adattādāsyatha, jāsyate (passive; Epic Skt. jāsyati), -dadhāsyasi, -dadhāsye, svādhyāsyati

Aor. (a)kāsi (to kr), agāsi, adāsi and osil, udapāsi (to pad!), -bhāsi (to bhā, or to bhās with ending i?), nirvāsu, -vāmsu, asthāsi and °sit

Ppp. -dadhāta-, irad-dhāta-, palāta-, palāna-

Inf. -dadhātum

Gdve. svādhyālavya-; -dadhānīya-; (duḥ-śrad-)dadheya-(?) Pass. opt: vikrāyelsuh (to vi-krī), fut. vikrāyişyali and vikrāsyati (also jāsyate, above); opt. svādhyāyela

24.12. Stems in -I-. Few new presents of this type in -I- have been discovered. There is a form of nI, samudanimahe, and a gdve. a-viniya; cf. Skt. adhila-: adhile; on this model the ppp. nila- might have created nile; or, more remotely, on the model yata-: yati etc., and the like. On bhaviti (?) see Chap. 43, s.v. bhū (7).

Skt. has not only adhite but more rarely adhiyate, which is regular in our language and MIndic. By analogy with this pair a future adhisyali is created, matching the regular adhlyişyati. And so fut. patlşyanti, to pa(l)liyati, for more usual patliyişyanti; and with passive meaning nişyati, to present passive niyati (°le). Cl. also dişyati, tarihasi, sthapişyati, § 31.3. Similar are allişyati, fut., and allipayati, caus., to more usual alliyati; related to Skt. li-, ppp. lina (note BHS allina). We may also note -grhisyāmi (also Epic Skt.), for Skt. grahişyāmi, on which (rather than on ppp. grhila-) -grhisyāmi may be based (with r for ra). With this future may be grouped the inf. -grhitum and gdve. -grhilavya-.

There is an aorist abravisi, evidently a modification of abravil to fit this inflectional pattern; the long vowel called for the aorist ending -si. Similarly, perhaps, aor. -grahisi(d), to grahi-(syali, etc.); also 3 pl. forms, based on Skt. imperfects in it, asisu (v.l. asitsu) and abravitsu (§ 32.37).

Gerunds are formed from weak 9th-class present stems: abhinirminīya, grhņiya, grhņitvā, grhniyāna; and from weak 3d-class present stems, abhinirmimiya, jihriya. From such stems other non-present forms occur. So fut., grhnişyati, perhaps krinişyali (§ 31.7); aor. udgrhnişu (? § 32.37); pass. pple. abhinirmimiyanlam (§§ 37.15, 37).

24.13. Stem in -u-. Here I know only of forms of the root bhū. There is no present of this type. But Skt. had a preterite (root aorist) abhul, which BHS inherits (also as abhū, abhu, ahū etc.). And to it was formed. presumably on the model of adasi to ada(t), a new s-aorist abhūși (abhūși, °șil) which is extremely common; it was obviously the regular aorist of this root in the dialect, altho not recorded in any other form of MIndic (except perhaps a few Asokan forms, 1 sg. and 3 pl., which may be parallel, § 32.72). There is also recorded, cuce, a future 1 sg. pralibhüşyam (? possibly denominative). 24.14. Stems in -o-. Here too we find practically

nothing but forms of root bhū; otherwise only an isolated and doubtful -haroti (on which see §§ 28.18, 58), an equally dubious aor. -karosi (§ 32.70), and aśroși, a direct MIndic inheritance from Skt. asrausil 'heard'.

There is a normal and very common present 3 sg. bhoti (= bhavati; 1 sg. bhomi, 3 pl. bhonti), with impv. bhohi; the form hoti (regular in Pali) also occurs but much more rarely.

To it is formed, first, a regular future bhosyali (? doubtful, and certainly much less common than bhesyali; in Pali, also, -bhossati and -hossati are rare, apparently only in cpds., while hessati is regular, Geiger 151.3, 154.2). Further, with h for sy, fut. 2 sg. bhohi, hohi, 3 sg. hohili, hohati, 3 pl. hohanti.

And there is a rare and doubtful aorist bhosisu (3 pl.; § 32.78), of the same type.

Non-thematic stems

24.15. Except for stems in long vowels, treated in the preceding sections, these exist only as relic forms inherited from Sanskrit, or virtually so. They do not spread outside of the present system. In so far as they are used in that system in a different way from Sanskrit, they will be discussed in §§ 28.57 fl.

25. Person and number

The dual

25.1. The dual was surely not alive in the language. Where it occurs, it is generally to be assumed that it has been introduced as a Sanskritism.

25.2. Plural, or sometimes singular, verb forms are used with dual nouns and pronouns. So yuvām adya gacchathā sādhu dārakau SP 462.8 (vs, with varr., mostly pl. verbs). In SP 267.2 (prose) ed. abhāşetām (subject two persons severally named), but 3 mss. abhāsala, and La Vallée Poussin JRAS 1911.1074 bhāşinsuh (with insertion of pronoun te, referring to the two subjects); in the sequel even KN use plural pronouns and verbs of the same pair. The edition prints dual werb forms in SP 249.2; 250.5; and 459.1, but all, or virtually all, mss. are recorded with either sing. or pl. forms, tho the subjects are recorded as dual in form. In SP 105.10 ff. (prose) dual nouns and pronouns are presented regularly as subjects of verbs which, in KN, are also regularly dual; sometimes the mss. seem to support such forms, to judge by the Crit. App.; but often they indicate plurals. So plural verbs with dual subjects occur in other texts. as LV 222.4 šakro brahma ubhau ca ... darsyanti; Mv (vss); duve na jäyante ... nararşabhau i.122.16; icchema pralisrjyanta 181.1; ubhāv etau marişyanti 181.5; yāpenti nirmitau 181.15; also Ud viii.1 ubhāv etau (ms. atau) pretya samau bhavanti.

25.3. Singular verb with dual subject (as with plural subject, below): duve mahāimānau utpadyate iathāgatau Mv i.122.14 (vs; one inferior ms. utpadyante, Senart em. ^oyete; parallel passage, line 16, reads jāyante, see above); vibhāti (kramau; Senart em. vibhānti) Mv i.68.11; candra-sūryau pated LV 310.17; šaširavi (for ^ovi) prapate LV 330.3; avocat, with dual subjects, Mv i.327.11, 18; 328.14; 329.2 (all prose); duve...anvati Mv ii.101.18 (prose; Senart em. ^onli).

Confusion of person and number

25.4. There is widespread confusion of person and number, usually in that 3 sg. forms are used for any person and either number. This usage perhaps started with the optative (ending -e) and aorist (endings -i, -i), where—largely by phonetic loss of endings—confusion set in in Middle Indic (cf. especially Pischel 459). Geiger 141.1 records the frequent use of atthi (= asti) with plural subject, but otherwise this usage seems to be virtually ignored in the books on Pali or Pkt. except as indicated above.

25.5. In BHS the phenomenon is specially common, indeed standard, in the optative and the aorist, where the Mindic endings e, i, \bar{i} , and even the Sanskritized el, il, are used very commonly as 1 and 2 sg. and 3 pl. (rarely as 1 and 2 pl.). Much less commonly, historic 3 pl. and 1 or 2 sg. forms are used with subjects differing in person and number, especially 3 sg. Examples of these usages, with non-Skt. endings, will be found under the opt. and aor. In the following list will be included formally regular Skt. 3 sg. optatives and aorists, used with other than 3 sg. subjects.

3 sg. subjects. 25.6. It seems that any 3 sg. form may occasionally be used with subjects of other persons and of plural number. And to some, the a much slighter, extent the confusion extends to other than 3 sg. forms. Some examples will now be cited, mostly from verses; prose examples will be so marked.

25.7. A special case is the use of 3 sg. verbs with vocative subjects (so sometimes in Vedic, Ved. Var. I § 332, and in late Skt., Emeneau, Jambhaladatta's Vet. xxii): parinirvātu bhagavan parinirvātu sugata LV 377.10 (prose; no em. needed; note in Weller 35 is otiose); parigrhnātu bhagavann... LV 382.13, 20 (prose); mama bhadre patim varet (mss. car⁹) Mv ii.71.5, 11 (vss), 'choose me as husband, fair one!'; imam bhagavan madhutarpanam paribhumjatu Mv iii.304.6 (prose). Cf. § 25.29.

Third singular forms

25.8. Pres. indic. with 1 sg. subject: paśyali Mv i.51.13 (prose; Senart em. °āmi, tho he keeps adrāksīl, parallel, and equally 1 sg., in the preceding line); aham... yujyale vā viyujyale vā SsP 265.1-2 (prose); (aham...) na vindāmi nopalabhale SsP 504.3 (prose).

25.9. Pres. indic. with 2 sg. subject: tvam... bhavati Divy 504.6 (prose).

25.10. Pres. indic. with 3 pl. subject: asti SP 58.8; 87.7; 92.1; 139.4; 363.9; 371.1; LV 235.18; Mv i.177.11; 328.8; eli SP 131.11 (with mss.); 307.2 (WT enli with K'); desayale SP 359.4; samjänite LV 60.1 (prose; v.l. °nate); samjanayali LV 66.21 (prose); śrūņute LV 74.9; śrūyate LV 80.14; sampvidyate LV 106.8 (prose); bhāsate LV 123.2; upety Mv 1.83.17 (Senart em. °nty); parityajati Mv i.105.10 (prose; Senart em. °nti); bhavati Mv i.233.3 (prose); śaknoti Mv ii.45.15 (prose); 286.5 (in both these Senart em. śaknonti); drśyate Lank 8.6; Ikşate RP 10.14; sambudhyate RP 15.8; tyajate RP 16.2 (in same line, with same subject, vicaranti; this and the others in RP may be m.c.); kurute RP 31.6.

25.11. Pres. opt. with 1 sg. subject: bhavet SP 336.7; tārayet (so with all Nep. mss.) SP 336.8; utsahet LV 60.17 (prose; so with v.l. and Weller 20 f.); gacched aham LV 215.18; bhakşayet Mv 1.92.15; kuryāt Mv 11.144.9 (prose); aham syāt Divy 110.13 (prose); aham bhavet Divy 250.4 = 252.22; jived aham ... paśyed aham Divy 638.21 (prose); aham... mimānsed iti Av 1.175.17 (prose; so ms.); praveiayed Suv 237.11 (all mss. have -d-).

25.12. Pres. opt. with 2 sg. subject: toap ... utthihet LV 284.6 (no v.l.); udīrayed Mv i.84.16; jānīyād Divy 284.15 (prose); yadi toam anujānīyād, 'if you permit', Divy 243.12 (prose, no v.l.).

25.13. Pres. opt. with 3 pl. subject: pravised Mv 1.150.16; pated LV 284.3; bhavet SP 32.6; 126.9; 280.6; likhet... likhāpayed vā SP 51.4.

25.14. Imperative with 3 pl. subject: $\bar{a}st\bar{a}m$ My i.152.5 = 225.21 = ii.29.2.

25.15. Imperative with 1 (or 3) pl. subject: bhavatām RP 44.8.

25.16. Future with 1 sg. subject: aham . . . abhisambholsyate LV 269.2-3 (prose; so Lefm. with best mss.; v.l. °tsye); kutrāham upanesyata iti LV 118.18 (prose).

25.17. Future with 2 sg. subject: (tram ...) chelsyati Divy 322.22 (prose).

(129)

25.18. Future with 3 pl. subject: dásyate LV 12.20.

25.19. Preterite with 1 sg. subject: āsīt (usually with aham expressed) SP 28.2; 195.3; 384.8; RP 23.14; Divy 402.19 (prose); avocal SP 257.8 (prose); LV 238.20 (prose); 239.6, 8 (prose); Mv ii.118.7, 13, 17 (all prose); adrāksīt Mv i.51.11-12; ii.123.17; iii.50.18; 75.1 (all prose); LV 265.21 (prose; v.l. °kşam); aham nāirausid Divy 422.3; prāviksat LV 240.2 (prose); abhāt LV 265.22 (prose, but y.l. abhāuam); akarod SP 258.7; aham adāt Gv 427.24.

25.20. Preterite with 2 sg. subject: mā... pariprākşil Karmav 27.17, 21 (both prose); (lvam ...) adhyāgamad Divy 617.25 (prose).

25.21. Preterite with 1 pl. subject: asil (vayam expressed) SP 212.8.

25.22. Preterite with 3 pl. subject: \bar{asil} SP 51.10; 176.13; 192.6; 384.5, 10; avocat Mv i.226.2 = ii.29.5; i.300.8 (prose); 328.7 (prose); ii.259.2 (prose); LV 4.17; 6.15 (both prose, ed. em. avocan); $adr\bar{a}ksil$ Mv i.55.3; 56.10 (both prose); $abh\bar{a}l$ Mv i.193.16 (prose); ii.294.10; kurula (i. e. ak°) LV 81.16; akampat präkampat etc. LV 352.2-4 (prose; a long series of verbs; subject sarvalokadhātavo; mss. discordant but the best ones regularly have singulars); upasamkrāmad LV 364.16; 366.12 (both prose); abhāsataLV 120.8 (prose; but two mss. abhāsantahl); adhyabhāsataRP 46.6, 12 (both prose).

25.23. Perfect with 2 sg. subject: tvam eva so tādrjako babhūva SP 28.2.

25.24. Perfect with 1 pl. subject: vayam savihāram dadau Karmav 97.2. Here Lévi divides sa vihāram, making an unspecified male person the subject of dadau. But this makes no sense. The obvious meaning is that 'we' (three nuns) in a former birth gave a gift which included a vihāra. The verse is identical with Therig. 518, where adāsimha confirms my interpretation.

25.25. Perfect with 3 pl. subject: $uv\bar{a}ca$ SP 196.3; Mv i.150.14; 190.10; 300.15; in i.152.8 = avocat i.226.2 and ii.29.5 (same vs); 1.328.16 (prose); ii.96.6 (prose); iii.323.7 (prose); $\ddot{a}ha$ Mv i.8.1 (prose, in same line dhansuh); pratyujigadma Divy 408.25 (subject paurdh).

Other than third singular forms

25.26. Much rarer is the use of 1 or 2 sg. or 3 pl. forms with inappropriate subjects.

25.27. First person forms. Optative forms in *-eyam*, identical with the Skt. 1 sg., are used as 3 sg. (\S 29.31). Otherwise note the curious form *bhesye*, which looks like a 1 sg. fut. 'I shall become', but seems clearly to mean 'you will become' in LV 196.10 (vs) ksipram bhesye prlii-

prāmodyalābhī. Tib. supports this construction. Others like this, § 31.35.

25.28. Second person forms are also rarely used in other ways than as in Skt. In SP 176.12 (vs) kalpā atllāh pūrvi ye abhūh (one ms. abhūt) can only mean 'kalpas which have passed by in former times'; the verb certainly has a 3 pl. subject.—We seem to find 2 sg. forms used with 2 pl. subjects in utlistha LV 17.4 and jāniyāh LV 274.18 (both prose; all mss.).—In LV 300.16 (vs) kuru prayatnam is taken by Foucaux as 3 sg., 'let him (the Bodhisattva) make an effort'. But it is clearly addressed to Māra and is a true 2 sg. impv.; so Tib. (contrary to Foucaux: bsgrim pa gyis, not 3 person).

25.29. A special case is the use of 2 person verbs with the nom. of the stem *bhavant*, regularly used with 3 person verbs but as a substitute for the 2 person of direct address; mā *bhavanto vişidatha* Mv i.108.1; mā *bhavān ikşvākuh... anutapyāhi* Mv ii.428.3. Cf. the converse § 25.7.

25.30. Third plural forms. Note that AMg. uses 3 pl. aorists as 1 and 3 sg. (Pischel 516 end). Our language also occasionally uses 3 pl. forms, mostly preterites, for other persons and singular number. So, as 3 sg., forms in -etsu(h); upasamkrametsu (v.l. ^ssuh) Mv i.328.19 (prose; Senart em. ^omesi); abhyakilelsuh Mv 1.335.10 (prose; so mss., read ^okirelsuh? Senart em. abhyokiresi); prāpuņelsuh Mv iii.338.20 (prose; Senart em. "nesi). --- Other preterites: avacisu LV 240.21; 241.8, 19 (all vss; no v.I.; unquestionably sing. subjects); pravisisu Mv i.98.12 (vs; subject the Bodhisattva; Senart takes it as a desiderative adjective, which is impossible); avacimsu(h) Suv 233.7 (vs); abhistawinth is impossible), abachisal(i) Suv 253.7 (vs); abilisativity vinsu or 'fsu Suv 243.10 (vs); abhavan (subject viryam) LV 164.7 (vs).—Perfect: ovácu 'he said' Gv 212.16 (vs; o is m.c. for u, and if the final u is not corrupt, it must represent -uh, the 3 pl. ending attached to an otherwise 3 sg. form).-Also presents: rājā paricintayanti Suv 237.3 (vs); prekşanti Suv 233.1 (vs); all mss. read so both times, with unmistakable 3 sg. subjects.

25.31. With 2 pl. subject: prapateyu (v.l. °ya) yūyam 'you would fall down' LV 304.6 (vs).

25.32. With 1 pl. subject: vayam... åsan Divy 243.16-17; 335.23-24; 336.21-22 (all prose); (vayam)... säksätkuryuh Divy 347.18 (prose); vayam smo (= smasl)... abhävan Av 1.257.3-4 (prose); vayam ca... abhävan SP 108.17 (prose; so doubtless to be read with 3 mss., one abhävam, Kashgar rec. åsit, text abhäma).

25.33. With 1 sg. subject: aham asann evamnama LV 345.13 (prose, no v.l. in Lefm.; Foucaux, Notes 185, cites one of his mss. as asam, doubtless lect. fac.; Weller 34 regards asann as a phonetic equivalent of asam = asam, implausibly).

26. Personal endings

26.1. For the most part, personal endings will be found treated under the various mood and tense systems. Here we shall deal with a few endings which are of wider application than any one such category. 26.2. I for final i. The endings mi, si, li, nli, and

26.2. I for final i. The endings mi, si, li, nli, and impv. hi all appear frequently with i m.c. See under Imperative for examples of $h\bar{l}$. A few cases of the other endings are quoted here.

1 sg. mī: anucintayāmī (so with all Nep. mss.) SP 61.11; (a)smī (with Nep. mss.) SP 87.3; samdišāmī SP 97.5; samtarpayāmī SP 128.13; manyamī LV 55.21; pašyamī LV 56.18; jānamī LV 56.20; socamī LV 111.7 (= śocāmi); stavamī Bhad 4; pūjayamī Bhad 7; pratidešayamī Bhad 8; anumodayamī Bhad 9; nāmayamī Bhad 12, 42, 55, 56.

2 sg. si: janesi LV 57.17; 91.21 (= janayasi). 3 sg. li: samādapeti SP 23.6 (so with all mss and WT!); vineti SP 92.2 (so read with WT for vineti of KN with Kashgar rec.; Nep. mss. vinetri; translate: ... all hear my Buddha- [v.l. dharma-] rule, [namely] this clever device of the Jina, whereby he trains many Bodhisattvas'); bhoti SP 94.3; bhavati SP 95.11; Mv ii.378.3 and 4 (so read with mss.); onahati SP 126.2 (= avanahati, for °hyati); carişyati SP 145.10; vinašyati SP 177.5; deśesyati SP 207.12 (Kashgar rec. °te); icchati LV 46.1; bhesyati LV 54.5; roditi LV 111.3; jánati Samādh 8.6.

3 pl. nti (for some cases of bhonti at the end of lines, where length is not demanded by the meter, see § 3.12): stronti (so read with 2 mss. and WT with their ms. K') SP 91.6; stuvanti LV 56.18; bhonti (= bhavanti) Dbh.g. 17(353).12; Sukh 46.4; karonti Sukh 50.18.

26.3. i for e. In 2 and 3 sg. forms this change, the it may be essentially m.c., involves substitution of active for medio-passive endings; see § 37.12. In 1 sg. (and rarely 1 pl.) forms it cannot be so interpreted, but seems to be exclusively a matter of meter; when a short syllable is required, *i* replaces e. It must be noted that sometimes, especially in Mv, final e is still written (at least in the mss.) even under these circumstances: Mv i.72.5 moksayisyě ham (meter also requires ha). Cf. § 3.64. Examples:

bheyi (= bhavişye) SP 64.6; Mv i.234.6; loşayi SP 45.6; niryātayi SP 115.9; smari 'I remember' LV 55.21; anušikşayi LV 138.6; śrunisyi LV 222.13; dadi LV 233.11; vasi 'I dwell' LV 325.10; pranipati LV 361.19; 363.10 (probably not aorist as I assumed JAOS 57.21, but for "pate, 1 sg. mid., with 1 pl. subject in 363.10); vartayişyi Mv 1.72.9 (Senart's note p. 433 seems to say that the mss. read "yişi, intending "yişyi); püjeşyi Mv 1.234.10; ii.300.9; mocişyi upapadyatām Mv ii.354.3 (? here meter does not require short -i; but note v.l. mocayişyopapadyatām) 'of them, coming to birth with the thought "I shall free (creatures)"; construe with teşām in next line (contra Senart); vandāmahi LV 360.16 (present).

Secondary first singular n for m (m)

26.4. The writing of *n* for m(m), before a consonant, is probably purely orthographic. It may be illustrated by *apasyan* 'I saw' (for *apasyam*, before *s*-) LV 114.9.

First plural ma and mas

26.5. In Skt. mas is regular as the primary 1 pl. ending, ma as the secondary. (In the epic the rule is often violated.) In Pali ma is used for both, exclusively so far as appears from Geiger 121-2. In most Prakrits, on the contrary, mo is regular; according to Pischel 455, ma should be regarded as genuine only in verses, presumably m.c. (but without restriction to secondary endings).

26.6. In our language, naturally, meter often requires one or the other in the positions where they occur. But in metrically indifferent positions, and even in prose, we frequently find ma where Skt. would use mas, and vice versa. No principle governing their distribution has been discovered; but ma for Skt. mas is much commoner than mas for Skt. ma (tho this may be in part due to careless writing; it is easier to omit than to add visarga). In the following I shall cite examples either from prose or from metrically indifferent positions in verse.

ma for Skt. mas: sma SP 100.9, 10; 101.8; 206.5, 6; 210.4; 222.12, 13; Mv 1.8.2 (twice); Gv 326.6 fl. (7 times); pradarśayāma SP 116.6; vicintayāma SP 116.11; dadāma SP 177.13; bhesyāma SP 223.1; vadhisyāma (read with mss. vahi^o, followed by nii, read tli = iti; Senart em. vadhisyāmi) Mv 1.17.2; gacchāma Mv 1.183.20; djānāma Mv 1.246.11; dema niryātema Mv 1.299.20; rocayāma Mv 1i.79.10; śrnoma gacchāma jānāma Mv 1.172.11; bhavisyāma Divy 252.28 (mss., ed. em. ^omah).

mas for Skt. ma: paśyāmo 'let us see' (subj.impv.) SP 242.1 (prose; so mss.; ed. em. °ma); syāmo (opt.) SP 70.7; saņmanyemah Mv i.347.18; mucyemah Divy 231.23; bhavemah Divy 252.20 (mss., ed. °ma); labhemo Divy 340.13; caremo 340.15.

First plural mha?

26.7. Senart reads paricaremha 'we would revere' in Mv iii.282.5, 11; 283.1, 2, 7. The mss. vary greatly; the Pali parallel, SN i.124.26, has paricarema. There are readings paricarema, °care 'ham, and perhaps °caramhe. Cf. the Pali ending mhe, Geiger 122, and the Pkt. ending mha, Pischel 455. But the mss. are too confused to allow any reliable inference as to what our author really wrote. Cf. however akarimha, § 32.40, which is better attested.

First plural matha (matha)

26.8. A few times in Mv and SP we seem to find 1 pl. forms in which the syllable *tha* (or m.c. *tha*) is added to the ending *ma*. Some are clearly present indicatives; some, indistinguishable in form from the latter, seem to have imperative force and may really be historically subj.-impv. (since *mas* and *ma* interchange freely, the distinction between these and the indic. no longer exists in the regular forms); some are optatives, not necessarily injunctive in force but also potential; and some futures.

26.9 I find it hard to explain these forms. Sometimes it is possible to assume that *-lna* is only the particle *alna* (all forms but one occur in verses, and *alna* would automatically appear as *lna* after *-ma* if this syllable must be short); but this seems impossible, or extremely forced,

(131)

9

in most of the cases here listed. If all the forms had the force of imperatives, one might suppose that the 2 pl. impv. ending had been added to the 1 pl. (indic.) as in Russian *idjomlji*, *paydjomlji* 'let's gol' (felt as an injunction to the person or persons addressed). But this could hardly explain an optative used in the potential or conditional sense, still less indicatives which are plain statements of fact¹.

26.10. I find no trace of such an ending in Mindic. The Pkt. ending -mha (§ 26.7) can hardly be related. Without explanation, the form is recognized by Senart, Mv, notes on i.29.11; 69.8; and ii.204.15; and by KN on SP 118.13, 177.15, referring to other Mv forms. All our forms are from vss except one, marked prose:

Future: samirāvayisyāmatha (possibly particle atha; Kashgar rec. °ma im-) SP 118.13; upasamharisyāmatha Mv ii.204.15; (?) draksyāmatha (em.) iii.112.7.

Imperative, or Present indicative with hortative or injunctive force: $s\bar{a}dhu$ (read $^{\circ}\bar{u}$ with K' and WT) gavesāmatha etam artham, 'please let us (all) investigate this matter' SP 164.13; also Kashgar rec. gaveşāmatha 171.12; sroomathā sammukha lokanātha 314.2 (WT with ms. K' sammukhi lokanāthā, as abl.), 'let us hear from the Lord of the World in his presence' (with WT), but the particle atha is here possible; vijahāmatha Mv 1.69.8; labhāmatha (obscure) ili.5.9; ramāmatha 6.2, clearly 'let us enjoy ourselves'.

Present indicative, in statements of fact (no impv. force): labhāmatha Mv iii.9.17 and āsāmatha 19, we get, we sit', precisely equvalent to labhāmahe, āsāmahe (v.l. in both ^omatha) in lines 7, 9 above; jivāmatha 11.15; vayam ca elāye gäthāye artham na vijānāmatha (v.l. ^omithal aorist?) 192.7, 'and we do not understand (have not understood?) the meaning of this verse'; next sentence begins tato 'so, therefore', and particle atha seems impossible; mayam le pūrayāmatha 215.13, 'we (will) give you your fill', and 14 dhārayāmatha.

Optative: nadematha Mv i.29.11 (so mss., Senart em. nadāmatha; but meaning very obscure); bhavematha iii.11.17, potential, 'we would be'; anupaśyematha, 23.2 'let us investigate', injunctive optative; possibly particle atha, but not likely; sprśematha Kashgar rec. v.l. SP 177.15.

Second plural to varying with tha

26.11. As in Pali and Mindic generally, the Skt. primary 2 pl. ending tha very commonly replaces the secondary ending ta, particularly in impv. and opt. forms. The reverse, that is, use of ta where Skt. would have tha, is extremely rare, but seems to occur at least once, in LV 209.8 (prose) kim evam kathayata, 'why do you speak thus?' So Lefm. with his best ms. (one of Foucaux's mss., Notes 148, also reads thus); the other mss. and Calc. kathayate, which could at best be a middle form with passive meaning, 'why is it spoken thus?'

26.12. In the impv., 2 pl. tha (or thā, almost invariably m.c.) is extremely common. In most texts (not in Mv) this is commoner in verses than in prose. Yet most other texts also use it in prose, and most of our examples will be chosen from prose. Instructive as to changes in text tradition (cf. § 1.43) are Ud xv.7 (vs), where pratibuddhatha of the oldest ms. is altered in later mss. to pratibuddhatha of the oldest ms. is altered in later mss. to pratibuddhatha and xviii.3 (vs), where old chindatha, bhavatha are similarly changed to 2a.—The final a is, of course, capable of being lengthened, only in verses, and usually m.c.; but note Dbh.g. 19(355).17 päjayathd, at the end of a line,

1. Hiän-lin Dschi briefly lists the forms in a note in Indian Linguistics 11 (1949).1-4 and attributes them to the 'eastern dialect' which in his opinion is found in Mv not m.c.; and there is a v.l. *janethā* for ^otha in Bhad 54, also at the end of a line, metrically indifferent; cf. § 26.16, perhaps to be identified with these cases.—Thematic a before tha may also be lengthened (§ 27.2).

26.13. Second plural imperative tha in prose. In SP 43.13, read with most Nep. mss. prajanilha; agacchatha 74.8 (read with most mss.); pibatha 321.5. Frequently (e. g. 188.2) the Kashgar rec. has forms in tha not adopted in ed.—ulpādayatha LV 90.22 (best mss.; v.l. °ta); nivartayatha 228.8. — chindatha, Mv i.18.2, utpāţetha 3, 4; aśnutha khādatha pibatha detha ... karotha ... caratha 196.14; ulpadyatha 198.13; karotha 245.7; udiretha 245.14. etc. — gacchatha Divy 292.21, ganayatha 22; gacchatha 293.6; 296.18 (in these four ed. em. °ta, but in the rest ed. keeps the reading of the mss.); agacchatha 340.18; nisidatha 350.13; ānayatha 372.25, pravetayatha 26; utpāļayatha 411.8, 12; citrayatha 547.8. — varjayatha Šiks 61.14 (so ms., ed. em. °ta, which is read in ms. in the following parallel verb forms). - pratisaratha KP 64.3; parimargatha (read ^omärg^o) 105.1. — bhuñjatha pibatha mälikramatha, Sädh 532.11, bhavatha 12. — praliyatha Sukh 97.5, 11, etc.; mäkänkşayatha 99.6. — kridatha ramatha paricārayatha Mmk 663.11. — mā vilambathāgacchatha Megh 306.16. — gacchata bhavanto vicārayatheti MSV ii.65.16.

26.14. Second plural imperative thā in vss m.c. irnulhā SP 87.7; nirdhāvalhā 87.9; parigrhnalhā 90.10; jahalhā 326.3. — rakşalhā LV 79.10, pravethā 12 (= pravayala); apanethā 123.5; nirbandhalhā 193.13. abhiyujyalhā RP 33.2.

26.15. Second plural optative in tha. Also numerous; examples from prose: rocetha LV 15.11; kurydtha Mv ii.233.18; prochegătha Mv ii.102.18; from verses (some with final d, mostly m.c.) dadyatha Mv ii.105.3 (§ 29.43); bhavetha SP 198.8; bhavethā LV 80.3, daršayethā 'dya 4; śocayethā 231.18; upanāmayethā 383.13; gacchethā 389.15; sevetha RP 32.18, etc.

Second plural than (?), tha

26.16. In LV (only) we seem to find a number of 2 pl. forms, all impv. or opt. in injunctive sense, ending in *thāh*, or sometimes *thā*. The latter might be regarded as m.c. for *tha*, except that it may occur in metrically indifferent positions. Nevertheless, since the forms in *thāh* are not numerous, possibly they may be taken as mere corruptions, and the forms in *thā* as equivalent to *tha*, even when not strictly required by the meter; we have indeed conceded this possibility in § 26.12.

LV 36.14 ff. contains an address by the Bodhisattva to the Tuşita gods, with many 2 pl. forms, mostly regular (as gacchata 36.19). But among them are pratipadyathåh 36.21; carethåh 37.9; anusmarethå (one good ms. °thåh; followed by dh-) 37.10; nirikşathå (but some mss. °thå; followed by y-) 37.12; ärabhethå yathå ca vadathå tatha karothå (here line ends) 37.19; samanusmarathå (followed by p-) 37.22; vidhamathå (followed by s-) 38.6; avatişthathå (followed by n-) 38.9.

In LV 138.12 and 14, at the end of lines, Lefmann varelhā 'choosel'; in 12 all mss. but one (A, the best) with Calc. "thāh; in 14 all mss. "thā, only Calc. "thāh. Foucaux and Calc. understand a 2 sg. middle (ending thāh), presumably opt., varethāh for Skt. varayethāh, addressed to the king alone (as in line 18, varayasva). This is indeed possible. But it may also be 2 pl., addressed to the Sākya counsellors as a group.

(see my §§ 1.24 ff.), but attempts no explanation. He quotes W. Krause as considering it an IE. ending (= Gk. $-\mu\epsilon\theta\alpha$), which seems to me fantastic and highly unlikely.

Second plural middle dhvam for dhve, and vice versa

26.17. In Divy 511.22, 26 (prose) dhvam is used in the 2 pl. present, for dhve: kim lajjādhvam, 'are you ashamed?' Anomalous also is the long thematic vowel (§ 27.3).

The opposite of this, primary *dhve* for *dhvam*, occurs in the imperative, § 30.19.

Third plural u for us

26.18. For Skt. us in the 3 pl. endings, principally optative and aorist, u occurs very commonly, and by no means only when in verses the meter requires a short final. In the aorist, *işu* may fairly be called the regular 3 pl. ending; *iŋsu* vies with it; when meter demands a long final, the texts generally write \hat{u} , much less often uh or another samdhi representative of Skt. us. But forms ending in u, without s, are also found in prose. This is equally true of aorist forms in *dsu*, *ānsu*, etc.,

eşu, ensu, etsu, etc., and in such isolated forms as abhūnsu, abhūnsu, abhūnsu, etc.; all occur in prose, at least in that of Mv, and are usually matched by corresponding forms in final uh, apparently without any distinguishing principle (possibly only by secondary adaptation to Skt.; but they are very common in the mss.). These forms, and a few other stray cases like adrsu = adriuh, § 32.118, are all dealt with in the chapter on the Aorist, §§ 32.29-38, 75, 95 fl. (esp. 100).

26.19. In the 3 pl. optative, also, (y)us appears as (y)u, chiefly to be sure in verses, but not always where meter demands a short final. Thus for ramayeyu rahogatāh of the ms. in Suv 16.11 Nobel actually reads ramayeyū by em.; the meter is however indifferent and ^oyu may stand. Other examples are grhnīyu SP 47.14; bhaveyu SP 59.1; 82.13; 116.4; 171.1; 326.12; Bhad 58; dahyeyu naśyeyu SP 86.8; bhrameyu 125.12; vrajeyu 171.1, hāyeyu 2; nipateyu LV 153.9, 10; prapateyu 216.20; śrnuyu Mv i.72.1; vrajeyu Sukh 53.4.

9*

27. Non-Sanskritic treatment of thematic vowel

a for a as thematic vowel

27.1. The lengthening of the thematic vowel a is common, but almost limited to verses and to positions where meter demands it. Examples are: bhavdii (with Kashgar rec.) SP 96.6; sahāti LV 81.20; labhāti 137.22; mocaydil 184.3, 4; disāsaydil 231.10; bhandil 233.5; cardii 240.22; rudāsi 253.8; yajāti 271.6; āmantraydii 302.22;. (bhavāsi Mv 1.207.17, so Senart, but mss. bhavasi; prob. read bhavāhi, impv., with repetition ii.12.7;)pratibhanātii Mv 1.246.16; hanāti ii.81.6; viharāti Gv 208.15; bhandāti 212.21; anusmarāti 214.21; cintaydil Dbh.g. 12(348).21; prodyayāmāna = prodyāyamāna Dbh.g. 51(77).25.

27.2. A special case, by no means limited to metrical lengthening, is the 2 sg. impv. ending $\bar{a}hi$; see §§ 30.2,6. Parallel to this there are also cases of 2 pl. impv. in $\bar{a}tha$, all in verses and probably m.c. However, the existence of $\bar{a}hi$ as a standard 2 sg. impv. ending (independent of meter), which might have exerted an influence on the 2 pl., makes this a bit less certain than the preceding cases. Examples: parivarayaitha LV 43.21; nirīkṣayātha LV 50.10; bhanātha Mv i.207.14 = ii.12.4, 18; pasyātha Sukh 51.5, 7.

51.5, 7.
27.3. Isolated, so far as I know, is the 2 pl. mid. lajjādhoam = lajjadhoe Divy 511.22 and 26 (prose): kim la^o 'are you ashamed?' Perhaps the ā here is due to influence of the noun lajjā.

27.4. āsi, āti, 2 and 3 sg., apparently or sup-posedly with modal force (called 'subjunctives'). Senart, My i.499 and elsewhere, recognizes certain forms in asi, ali as subjunctives, following Pischel's earlier interpretation of certain forms in these same endings in Pali. Geiger 123 admits a few such forms, 'where subjunctive (imperative) meaning is clearly present and the possibility of lengthening (of thematic a) metri causa seems excluded.' But, contrary to Geiger's statement, none of the cases accepted by him can qualify by his own criterion, since they all may have d m.c. This is equally true of practically all the My cases. The only certain exceptions are My iii.268.8 (prose) bhandii, apparently = 'he shall say, he is to say, let him say'; and ii.256.4 (prose) yadi atmamamsam paritya*jdsi*, 'if you give up your own flesh'. Particularly in this latter passage, it is doubtful whether the form is other than a present indicative. As to bhanati, it must be noted that bhanādi and similar forms from this root occur in certain Prakrits as indicatives. Pischel 514 interprets them as analogical to 9th class forms.

27.5. There is even, once at least, a form ending in asi, normal 2 sg. pres. indic., which is just as modal ('subjunctive' or 'imperative') as any of Senart's forms in dsi, dti: mdidm eva ca guhasi Mv iii.4.1 (vs), which can only mean 'and wind a garland instead!' (Senart's note calls it 'indicative for subjunctive'.) To be sure the meter requires a short penult, and the ending could be described as shortening of (supposed 'subjunctive') dsi, m.c. But the fact is that indicative forms are often used with modal values in our dialect, as indeed in Skt. and Vedic (Vedic Variants I p. 65 fl.).

27.6. I am not much impressed by the proposal to connect this d(-si, -ti) with the old Vedic subjunctive.

For the most part I believe it is mere metrical lengthening. The few prose cases might be modifications of indicative forms in *asi*, *ali* (often used in ways hardly distinguishable from optatives, whether injunctive or potential-conditional) on the pattern of such endings as AMg. *ejjasi*: *ejjāsi*, both familiarly used as optatives (Pischel 461).

27.7. I shall, however, now list all the forms known to me in *āsi* and *āti*, where injunctive (or, in a few specified cases, potential-conditional) meaning either seems probably present, or has been assumed by Senart. All the textually reliable cases are verses (and most could have \vec{a} m.c.) except those mentioned above.

27.8. paiyāsi (mss. paiyāmi, but Senart's em. seems certain) Mv i.155.21; gacchāsi 157.3; pralipadyāsi 279.19 (vs. but ā not required by meter; however, mss. read 'yosi; and 'yesi, cf. §§ 29.20–21, would be at least as likely an emendation); saced bhadanle gacchāsi ii.194.11, conditional; in 248.7 (prose), Senart mā parilapyāsi, but v.l. 'yāmi, 1st person, which is probably the true reading: '(don't harbor longing further) lest I suffer for it (as your husband)'; the crocodile speaks to his dohala-stricken wife; parilyajāsi 256.4, cited § 27.4; sobhāsi (= śobhasi) 321.17; bhanāli 384.7 (?potential, if not indicative); iii.268.8, cited above.

a for ā as thematic vowel

27.9. Shortening of a thematic vowel which is long in Skt. is also common, but occurs exclusively in verses m.c.: adhyeşami SP 38.2; Bhad 10; adhyeşama (tor °āmah) LV 50.14; prajānami SP 220.13; jānami LV 56.20; jānami Gv 212.22; manyami LV 55.21; pasyami LV 56.18; pasyami Suv 51.7; Gv 230.21; gacchamo LV 78.15; °ma LV 133.7; 134.6; socami LV 111.7; vrajāma LV 113.18; kathayami LV 138.14; prekşami LV 236.3; muñcami LV 236.4; dadami LV 236.6; °mi id. 241.14; budhyami LV 287.20; vandamo RP 5.7, 8, 18; 6.14 etc.; vandami RP 5.16; Bhad 1, 7; adhimueyami Bhad 3; stavami Bhad 4; pūjayami Bhad 7; pratidešayami Bhad 8; anumodayami Bhad 9; abhiyācāmi Bhad 11; nāmayamī Bhad 12, 42, 55, 56; prabhāsami Gv 230.14; chindami Gv 231.20; yujyami Dbh.g. 7(343).25; bhavamo Dhb.g. 23(359).1; bhavişyami

i instead of thematic vowel a, a

27.10. In certain Pkt. dialects (AMg., M.; Pischel 108, 455; Bloch, Indo-aryen, 235 f.) 1 pl. forms may end in *imo*, *ima* instead of *dmo*, *āma*. A few such forms, including vandima (AMg. M. vandimo), are found in our language. But also, if the tradition is to be trusted, a few other forms show *i* instead of regular thematic *a*: one 3 dual present ind., and several imperative forms (3 sg. and 2 pl. active, 2 and 3 sg. middle). I am not impressed by Pischel's explanation of the *imo* forms by association with shift of accent (Pischel himself admits countercases). Of different character are futures like *hohiti*, *kähiti* (§ 31.12) where the *i* is doubtless connected with the y(a) of the original future sign; and aorists in *i*(1)tha etc. (§ 32.41), where the *i* is derived from the old *is*-aorist. But it is,

perhaps, not impossible that our forms, or some of them, may have been made in analogical imitation of some such supplier or a orists, or of old presents in *i-li* of the type svapili, rodili. The forms recorded are:

1 pl. ima, imo: vandima (AMg. M. vandimo) LV 364.6, 19; 365.5; ramimo LV 321.20; preksima LV 367.13 (Lefm. em. preksiya); půjima Dbh.g. 19(355).9 (but here i possibly m.c. for e = aya, to půjayati). 3 dual ilah: krláitah Divy 404.18, 21, 23 (twice).

3 sg. impv. itu: prāvaritu 'let him put on' Divy 215.29 (to prāvarati).

2 pl. impv. itha: samnipatitha Mv i.329.17 'assemble!'. parallel to nisidalha in the same context.

2 sg. impv. mid. işu: mā kānkşişu 'don't hesitate!' Mv ii.55.21 (see § 30.16).

3 sg. impv. mid. *ilām: anubaddhilām* LV 45.4, for anubandhalām, 'let him attach himself to, follow' (on the present stem baddha-li see Chap. 43, s.v. bandh).

28. Summary classification of non-Sanskrit presents

28.1. In principle, and descriptively, all thematic verbs form a single morphological category, inflected identically thruout. From the descriptive standpoint, therefore, there is no reason to classify them at all, unless by pointing out that some of them have alternative stems. Thus stems in -ya- vary with those in simple -a. There are pairs like karati:kurvali, śrunati: śruvali, etc. Sometimes there is a richer variety: lişlhati, sthihati, and at least in compounds -sthahati, besides long-vowel presents sthät and (in cpds.) -stheti, and still other rarer types, are all quite interchangeable presents of the Sanskrit root sthā.

28.2. In the following rough and ready grouping of verb formations, I have tried to take note of descriptive groupings so far as they suggest themselves. But at the same time, and particularly in subdivisions of the thematic verbs, I have called attention to what I believe to be the historic origin of all new (non-Skt.) verb stems, so far as seemed possible.

Third singular ali based on Sanskrit third plural anti

28.3. In many cases the MIndic present in ali is based on a Skt. 3 pl. in anti, non-thematic. Some of these occur sporadically in Skt., particularly epic; and it is probable that the same process produced regular Skt. presents like muñcali, siñcali (which do not fit the original IE phonemic pattern: Edgerton, Lang. 19.109 ff.), tho there is no record of *munakli, *sinakti. In MIndic the process has completely eliminated the non-thematic 7th class of Skt. (Geiger 144; Pischel 506).

28.4. BHS forms from 7th class presents are añjali (only ppp. añjita-), chindati, bhañjati, bhindati, bhuñjati (in Skt., Wh. Roots, U +), yuñjati (E +), riñcati, rundhati (E +); and pişati (sporadic in Skt.; AMg. pīsai) which is a phonetic equivalent, by the Morengesetz, § 3.3, of Pali piņsati (sporadically Skt. piņşa-) based on 3 pl. piņşanti to pinaşii; see Chap. 43 s.v.

28.5. Ninth class presents also develop forms like krina-li to 3 pl. krinanli; but here the Skt. 1 sg. krinami (cf. § 28.8) will also have been influential. Similar forms are found in Pali and Pkt., which however also retain forms of 9th class type (iu contrast to the lack of 7th class forms). BHS has krinati, grinati (Wh. Roots, E.), jinati (in this dialect a synonym of jayali), jänati (U.E.), and punati (in caus. gdve. opunăpayilavya-). Instead of krinati. BHS has also krinati, like Pali kinati, with i for I; Geiger 21c suggests analogical influence of mināti, minati. The form may, however, be historic. Oldenberg, RV Noten, would read krināti for kri° RV 4.24.10 (the only finite form in RV), on the analogy of metrical evidence on other 9th class roots recorded with long final vowels (the isolated kri° is metrically indecisive), cf. Old. Proleg. 477 f.

28.6. Similarly, on the basis of 8th class 3 pl. kurvanli, we find very commonly kurvati (common also in Pali and Pkt., kubbati, kuvvai), and to 5th class śravanti, 3 sg. śravati.

28.7. Finally, to 2d class 3 pl. vidanti is formed thematic vidati (also sporadically in Skt.); and to 3 pl. stuvanti, *stuvati (AMg. thuvai, Ratnach.), indicated by ger. stuviya, which prob. read for supiya LV 50.4, see Dict.; also probably abhistuva (§ 32.115). And there is a 3 sg. opt. yeya (or with mss. yeyam), once, which seems to imply *yali 'goes', to 3 pl. yanti.

Third singular ali to first singular *āmi*, and to third plural ati

28.8. In 9th class verbs, as we noted above, the 1 sg. of type krīņāmi no doubt helped the 3 pl. krīņanti to create a 3 sg. krīņati and other thematic forms.

28.9. The 1st sg. \bar{ami} alone, without help from the 3 pl., was responsible for regular Skt. *lishali*, if (as I think probable; there is another theory) it represents a prehistoric **lishali*. Skt. *jighrali* is also the regular present to ghr \bar{a} , tho in the Veda sporadic 3d class forms occur. And other verbs of this type show in BHS (and MIndic) similar forms which may be considered fundamentally MIndic, tho they all occur at least sporadically in epic Skt., some even in Vedic. BHS has *dadali*, *dadhali*, and *jahali*.

28.10. Once we find *śraddhale*, 3 sg., apparently to the MIndle *śraddhāmi* by this same analogical process (*śrad-dhāli* occurs, cf. § 28.51). Perhaps **ava-dhyali* (pres. pple. *avadhyanto*) is similarly based on *-dhyāmi*; and **samākhyali* (aor, *samākhyalisu*) on *-khyāmi*.

28.11. Since such forms as *dadali* are identical with the old non-thematic 3 plurals (to 3 sg. *dadāti* etc.), we find in BHS such old 3 plurals in *ati* used likewise as 3 sg. even when there is no 1 sg. in *āmi: jāgardi* (Skt. 3 pl., as 3 sg. 'S.E.' in Skt., Whitney, Roots, and in BHS), to regular Skt. *jāgarmi, jāgarti*; and *juhvali* (Skt. 3 pl., BHS 3 sg.) to *juhomi, juholi*. (For 3 sg. BHS *juhati* see § 28.18.) We may assume a proportion: *dadati* (3 pl.): *dadati* (3 sg.) = *jāgrati, juhvali* (3 pl.): x (*jāgrati, juhvali*, 3 sg.).

Present ali based on Skt. preterites in al

28.12. On the basis of Skt. preterites, mostly historic aorists, in -at, BHS and Mindic form presents in ali. So gamali to agamal (cf. also fut. gamisgati); druhali (AMg. duhali, duhai) to Vedic and Epic adruhal; darśati and drśati to adarśat, adrśat (both pre-classical in Skt. but paralleled in Mindic); lipati to alipat; sicati to asicat; chinati (aor. chini, ger. chinitvā) to impf. acchinal (acchinas Pān., acchinam Mbh.), Chap. 43, s.v. chid (2). And vacati to *avacat (Pali avaca) is doubtless a modification of Skt. avocat, blended with related forms in vac.

Present ali based on future isyali

28.13. In a number of other cases it seems probable that *ali*-presents have been formed to Skt. futures in *isyali*, on such models as *bhavisyali* to *bhavali* and many similar cases. This seems to be the source, particularly, of several such presents from roots in r (ar), notably the common karati (to karisyali) instead of karoli; also jāgarali (once in Mbh.), *dharali, marati, starati* (this is sporadic in Skt.). Note that Skt. has carali, larati, sarati, smarali, *harati*, by the side of carisyali, larisyali, etc. But varati (for *vrnoli*) can hardly be based on *varisyali* (only Gr. in Skt.). Perhaps it is based on the ppp. *vrla*-, on the model

of harati, sarati, smarati (BHS karati, etc.) to hrla-, srla-, smrla- (krla-, etc.). And -vurati, to the same root (Pali -purati), seems to be a phonetic modification of this varati, with u for a by assimilation to the labial consonant; see Chap. 43, s.v. 1 vr (3). Is the intensive cankramati based on the Vedic cankramisyati (\$39.1)? Cf. \$28.30.

28.14. Other ali presents probably based on futures are hanali (Epic and late Skt.), to hanişyali; bandhali to bandhişyali; vadhali (no pres. in Skt.) to vadhişyali. Perhaps also asate 'eats' (Pali asamāna-) may be based on asişyali. Since slavişyali is only Vedic, it is not likely to be the source of BHS slavali, Pali thavali.

Third singular ali from original (Skt.) fifth and eighth class presents

28.15. In Pali often, in Pkt. regularly, these become 9th class, or more usually (like the original 9th class) thematic presents (Geiger 147-8, Pischel 502). No explanation of the change from 5th to 9th class has ever been offered, and I am unable to offer any. There are a few roots (like str) which show both 5th and 9th class types in Skt., but most of them show neither in MIndic. Cf. however BHS and MIndic *dhunali* to Skt. *dhunoti*, which shows a few sporadic 9th class forms in Skt.

28.16. In BHS, distinctively 9th class forms from original 5th or 8th class presents hardly exist. I have noted only viornisva, once (Whitney, Roots, records aornidhvam, AV); and ger. abhinirminiya, seemingly based on the weak 9th class stem -minile.

28.17. On the other hand, thematic presents from such roots are not uncommon. In view of the Pall evidence, it may perhaps be assumed that these presuppose the intervention of 9th class forms: e. g. srnoti > *srunāti >srunati. We find -apunati, -cinati, -tanati, dhunati (nirdhunet cited once from Mbh.), -minati, srunati. And (sam-, abhisam-)bhunati (once "bhunoti; in Pall often 9th class, but not 5th; also "bhavati) 'attains' is clearly analogical to its near-synonym Pall pāpunati (see Chap. 43, s.v. bha 5). Probably we should also count manati here, deriving it from manute, which is well attested even in Classical Skt. (Pall munāti; AMg. manati and munat).

28.18. The substitution of karali for karoli is different; there is no reason to suppose that a *kardii, 9th class, ever existed. The form is explained above (\S 28.13). But probably on karali = karoli is based BHS and Pali juhali = juholi; forms based on it are commoner than those from juhvali, on which see \S 28.11. BHS and Pali dhunati is probably based on dhunoli (dhundli-forms are rare in Skt.). Conversely, we find once -haroli for harali, apparently (if genuine) analogical to karoli = karali. On kurvali, frnoali, see \S 28.6.

Presents based on past passive participles

28.19. In Language 13.116 I pointed out that a present buddhali is formed to root budh, 'doubtless a sort of denominative to the stem buddha'. This has now been confirmed by the discovery of a number of similar forms: rddhali, baddhali, -ruddhali, and perhaps *viddhali. And above all we find lagrali, from the na-participle lagna, abundantly attested; possibly also laggali, with MIndic assimilation of consonants. This gives particularly conclusive evidence that buddhali etc. are really based on past participles. The stem laggali also occurs in Pali, and laggal in Pkt.; Pischel 488 and Geiger 136.2 assume a

1. The addition of ya to a stem in a would, in Skt., Pali, and BHS, yield -aya-, which in MIndic and BHS often becomes -e-. Such stems are of course very common. But they belong to a different pattern from those considered here. ya-present, *lagyali; but our lagnali proves them wrong. With -aya- suffix, ārūdhayali; perhaps prajňaplāpilam, § 38.56.

New presents in ya

28.20. ya after long vowels. According to Hemacandra 4.240 (cf. Pischel 487), any Prakrit present-stem ending in a vowel other than short a^1 may optionally take an added ya. This means, in practice, that stems in \bar{a} , and occasionally in other long vowels, may be thus extended. Nearly all the new (MIndic) ya-stems recorded in Geiger 136-8 and Pischel 487-8 belong in that category. Those which do not are in part phonetic developments from Skt. ya-stems; others (e. g. Pali laggati, § 28.19) are not really ya-stems at all.

28.21. As we shall soon see, BHS differs from Pali and Pkt. (at least as presented by Geiger and Pischel) in showing a much larger number of new stems in ya after consonants. After vowels, however, the conditions seem to be much the same. We find ya extensions chiefly of root-class presents in \bar{a} -($ih_{2,0}^{A}$ -occasionally also in other long vowels. Sanskrit has such pairs. Sometimes the Skt. - $d\bar{a}i$ form seems to be the more original (so väli, sndli, as compared with vāyati, snāyate), or there may be no - $\bar{a}yati$ parallel in standard Skt. (as māti, yāti, khyāti). But in other Skt. cases, we find the reverse; so trāyate (trāte sporadic in RV, trāti Epic), dhyāyati (B, dhyāti E +), gāyati (gāti extremely rare), mlāyati (mlāti only once in BR, from MBh). To the same pattern, descriptively, belongs jāyate 'is born' (there is no *jāti unless we count AMg. jāi, cf. § 31.3). 28.22. BHS has a number of presents in $\bar{a}yati$ not

28.22. BHS has a number of presents in $\bar{a}yati$ not found in Skt. Most of them are paralleled in Pali or Pkt. or both; for details see Chap. 43 under each root. When, as often happens, Skt. has a present in dil, we may suppose analogical influence of such pairs as vdti, $tr\bar{a}ti: vdyati$, $tr\bar{a}yate$. (The reverse also occurs: § 28.54.) But in some such cases it may seem equally reasonable to assume influence of a ppp. in -dia, or a future in $-\bar{a}syati$; e.g. $tr\bar{a}ta_{-}$, $tr\bar{a}syate$ (°ti), to present $tr\bar{a}yate$. More than one such influence may, of course, have contributed. The pair jda_{-} : jdyate seems to me particularly likely to have played a rôle in creating new presents in -dya- to participles in $-\bar{a}ta_{-}$. The presents antara-dhāyati and "hāyati seem to be based on the old Skt. 3 sg. aor. pass. antara antaradhāyae.

28,23. There are even cases in which Skt. had neither -*dli* nor -*āyati* presents, but both were created analogically in MIndic and BHS. Thus *bhāyati* and *bhāti* 'fears' are both based on the quasi-antonyms *trāyate* and *trāti*, and/or the quasi-synonym *palāyati* 'flees', which in BHS has ppp. *palāta*, *palāna* (§ 28,54); Pali has *bhāyati*, and Pkt. *bhāi*. Other special influences may exist in individual cases.

28.24. The following presents in -dyati, not normal in Skt. and not obviously denominative, have been noted:^{*} dkhydyati (Skt. dkhydyate, passive), ghrdyati, jdyate (= jayati; cf. vdyati), dhdyati (root dhd; chiefly anlara- dh°), bhdyati ('fears', see above), bhdyati ('shines'), mdyati, ydyati, vdyati (= vayati, cf. jdyate, above; ppp. vdyati = Pali id.), sthdyati (based on BHS sthdti?). There is also hdyate, a passive to root hd (§ 37.38), and vikrdyati is equal to the passive to vi-kri (§ 37.39). On hriydyati see just below.

2. A few of the forms recorded here and in the next section seem to be based on historic passives; see § 37.23. The same may well be true of some cases listed in § 28.28.

28.25. After other long vowels than d, note adhiyati (= Skt. adhiyate, rarely for adhile). According to one view, pattlyati (pati°, prati°) is ultimately based on the same root i, compounded with prati. The forms of hrl are varied and confusing: hriyāyati (and perhaps hriyā°; denominative?), jihriyati, jehriyate (with intensive-type reduplication), and perhaps hriyati (which might be based on ppp. hrita-, like niyate: nita-, but the form is guestionable, see Chap. 43). BHS seems to have shared with Pali an irregular desiderative of *iru*, *juirūyati* (recorded as *śuśu*°; Pali sussūyati beside sussūsati); here y replaces s, doubtless by adaptation to the form of some ya-present. The original passive to ā-dā, ādīyati (also ādiyati), is used (like Pali ādiyati) with active meaning, 'takes'. And once āhūyati, to the Skt. passive āhūyate, is used as an active. See § 37.23.

28.26. ya after consonants. These are surprisingly numerous. They are doubtless heterogeneous as to origin. Such explanations as have occurred to me are suggested under the various roots in Chap. 43. Sometimes they seem to be based on Skt. passives, with active endings and meanings; so *sakyati* (also Epic. Skt., see § 37.22). Sometimes they may be due to phonetic confusions, possibly errors of tradition; so -yuñjyate for -yuñjate (y after palatal). (Is this the explanation of dariyanti, LV 222.4, vs, acc. to Tib. = darśayanti, possibly by way of *dar*santi*? same perhaps Lank 312.9.) Sometimes they may be hyper-Sanskritisms, analogical (unhistorical) reconstructions from MIndic forms; so *srjyati*, perhaps for MIndie *sajjet*); i = Skt. *sarjayati*; *nisedyati*, q.v. in Dict.; and *vrşyate*, perhaps unhistorically based on Pali *vassati*. 28.27. After all explanations that I have thought of,

28.27. After all explanations that I have thought of, there is left a considerable residuum for which I have thought of none. In other words, I see at present no way of avoiding the hypothesis that in BHS the ya-present was more or less freely productive, even after consonants. 28.28. The list follows. I omit the asterisk before

forms not actually recorded in the present system.

işyali, -eşyati (to root iş), -lkşyali, iryali (to root Ir), -kramyali (in part prob. denom., see Dict. s.v. nişkramyati), gaveşya(ya)ti, -gāhyale, tyajyati, -dišyali, dršyati (probably based on passive, cf. § 37.23; Chap. 43 s.v. drf 4), -nadyati, -bādhyati(?), bhāşyati, -bhāsyati(?), bhidyati, -madyati, -yunjyate(?) and -yujyate, rucyati and ruccali, ruhyati, -langhyati, -lambyate, vahyati(?), -visyati (see Chap. 43, s.v. vis 2), vrşyate, -vrajyati (see Chap. 43), śakyati (°te), -śrabhyati (°te), saryati, -srjyati, svapyati (Pali suppati, soppati), -hindyate, hişyati.

a for ya presents

28.29. The converse of the preceding also occurs. A present *straii = stugati is implied by ppp. stotta- (AMg. stoiya-) and ger. stoitod (cf. AV -stoya). It might be explained as a phonetic equivalent of MIndic stovati (Pali sibbati, AMg. stovai) = stoyati, or as analogical to noun forms like stoana, stoaka. And *harsati, implied by aor. harsi, could be a hyper-Sanskrittic substitution for MIndic hamsati (hassati?) = hrsyati. For Skt. klāmyati, klāmati occurs (so Skt. Gr.).

28.30. Perhaps back-formations from past participles in -*ila*- are **kşubhali* (caus. °*bhayali*) = *kşubhyali* (*kşubhila*-), and -*śamali* = *śāmyali* (*samita*-, to caus. *šamayali*). If the intensive ppp. *cańkramila*- exists, *cańkramali* (rare in Skt., very common here) = *cańkramyale* may be based on it (cf. § 28.13). And similarly *jājvalali*, intens. to *jval* (also occasionally in Epic Skt.), may be based on ppp. *jājvalila*- (recorded in BR only from Rājatar.). I have no explanation for -nahali = nahyali (cf. Pali ger. samnahileð). But manali (AMg. M. manai) probably derives from a Skt. 5th or 8th class present (§ 28.17) and has no relation to manyale. Not very certain is (abhi-sam-)budhati, implied in pple. ⁹budhāna (perhaps m.c. for ⁹budhāna) and fut. ⁹budhisyali; ⁹budhati occurs as v.l. for ⁹budhyali once in Mvy (see Chap. 43 s.v. budh 2 and 4). In RP 59.18 (an-)utpadanta- seems to be a pres. pple. stem to ul-pad-, regularly a ya-present (error or misprint? see § 18.5).

Loss of infixed nasal in thematic presents

28.31. We have treated lipali = limpati and sicali = siñcali as based on aorists alipal, asical. This is not likely to account for mucali (once, in vs, Chap. 43; m.c.? but AMg. mucai, muyai) = muñcali, since amucat (and mucali) seem to be Vedic only and not common. So once vidanti 'they find'; cf. early avidal (but vidali 'knows' occurs in Vedic and Epic Skt.). Stray forms pointing to bhajali (Chap. 43). The ger. -yujilvā for -yuāj- is of doubtful authenticity. Probably subhati 'smite' belongs here; Pali has only sumbhali, and AMg. vi-sumbh-; perhaps cf. Vedic sumbhanti KS 10.7 (132.14); TS has asubhana. On ricitvā (RP) see Chap. 43, s.v. riñcati.

Infixation of nasal in thematic presents

28.32. The converse of the preceding occurs a few times. So damiali (caus. damsāpayati), occurring once in Skt. (pw; = Pali damsati, AMg. damsai); -bhuñjati 'bends', perhaps by analogy of the homonym meaning 'enjoys'; -vrmhati 'tears', perhaps due to the homonym vrmhati (brmhati) 'increases'; and vindhali (AMg. vindhai), vin-dhayati = vidhyati, root vyadh (obscure as to origin); possibly samj- for saj-ati, q.v. Dict.

Thematic presents with a for a in root syllables

28.33. There is a little group of forms which differ from Skt. thematic presents only in lengthening radical a to d. Perhaps analogical influence of nouns containing vrddhi root-forms may be concerned. They are: niskāsati (Dict.); jāyate = jayati and *vāyati = vayati, § 28.24; -nāmati = namati; (vyā-)yāmati = yamati (rare in Skt., regular in Pali; Chap. 43, s.v. yam 2); sphālati or sphāleti = sphalati (Gram.), Pali phāleti; (-hārati or 'te, prob. denom.; Pali id. = harati;) -hvāyati (Chap. 43) = hvayati. On bhāyati = bhayati see § 28.23.

Long *i*, *u* for *i*, *u* in root syllable of thematic presents

28.34. The passive of sel-roots in r, Skt. -irya-, appears in Mindie normally with -ira-, so that we find forms pointing to *lirati, *dirati for Skt. passives tiryate, diryate. But in the case of kr 'scatter', this Mindie passive kirati (Skt. kiryate) is used (in cpds.) in an active sense, in both BHS and Pali, and has the appearance of being a modification of kirati, with I for i. Other forms with active meaning are sometimes based on original passives, §§ 28.24, 25 (dkhydyati, dhyati, ddlyati and dat°).

28.35. The presents *pişali* (sporadic in Skt.) and *pişayali* may contain *i* as equivalent to nasalized *i* (Pali *pimsali*; *pimsali* is rare in Skt.), § 28.4.

28.36. Different in character, and without explanation so far as I know, are hişati (cf. Deśi hisamana-, noun) and hişati (Skt. heşati; Dict.); düşyate (düşyant- once in Skt.), formally = duşyate; and müşati (Pkt. mūsai) = Epic Skt., BHS muşati, Pali musati (Skt. muşnāti), blended with mūs, mūşikā? Cf. also *timayati (ger. timayitbā; Ap. ppp. limia-) 'makes wet', to a root appearing in Skt. as tim (ppp. timita-). An isolated līpyase (root lip) may be corrupt.

Other new presents of the 6th class type

28.37. In the last section were mentioned *hisati* and *muşali*; the latter (also in Epic Skt.) may be a back-formation from ppp. *muşila*-, which is standard in Skt. and old (RV). Here might be classed also sruvate for sravate, with u for a probably by assimilation to v (§ 3.57), if not corrupt for sravate; -vurati (Pali purali), to vr (cover', in my opinion based on varati (§ 28.13) and also due to assimilation to the (preceding) v; pari-vişati (Pali d.), see Chap. 43; *ghrşati (Pkt. ppp. oghasiya-) = gharşati.

Other new presents of 1st class type

28.38. These are miscellaneous, and in most cases I have little or nothing to suggest as to their origin.

28.39. Root-final semivowels: kşayati 'destroy' (once in Rām.; cf. caus. kşayayati, 'E + ' in Whitney, Roots; fut. kşayişyati, rare, and kşita-, rarer than kşina-); -cayati (-ceti), cf. cayate V.B. in Whitney's Roots; layati (Gr., and layate V.B., Whitney, Roots); javati (Vedic javate; Chap. 43, s.v. ju); lavati (cf. Pali lavāpeti); stavati (Pali thavati); -galati or -garati (rare in Skt.; AMg. galai); -varati (Pali id.; § 28.13); spharati (Skt. Gr.; Pali pharati), and pharati, phalati.

28.40. Roots containing semivowel plus consonant: karlali 'twists' (a rare root; ppp. once karlila- in Epic Skt.); -deśale (perhaps denom.); -dveşale (probably denom.); lehati (Pali id.), doubtless blend of Skt. ledhi and lihati; sparšati (Pali phassati, cf. AMg. sam-phāse), see Chap. 43, perhaps back-formation from caus. sparšayali?; harşati (hrş-).

28.41. Presents based on original s-futures (or aorists) used in Pali as presents: *-drakşali, *-śakşali = Pali dakkhali, sakkhali (both presents). Our forms are of course hyper-Skt. back-formations.

28.42. sthahati = Pali thahati, doubtless analogical to Pali dahati = dadhāti (with Geiger 132.4).

28.43. Unknown anywhere else is *sthihati*, which is very much commoner than *sthahati*; indeed it is so common as to suggest that it may represent the regular present of the root *sthā* in the original dialect. It seems to be a blend-form based on *sthahati*, with the vowel of the first syllable influenced by *sthila*, *sthibā*, or perhaps *listhati*. **28.44.** Parallel to *sthihati* in formation, and probably

28.44. Parallel to shihati in formation, and probably based on it, is $(sam)^*$ dhihati, implied by the aor. sam dhihe (once), to root dhā.

28.45. ācikşali 'tells' and abhyācikşali 'accuses' seem to be hyper-Skt. back-formations from Pali (abbh-)ācikkhali, AMg. (abbh-)āikkhai, with form influenced by ācakşali, which is a synonym of ācikşali, q.v. in Dict.

Presents in eli

28.46. The MIndic phonetic change of aya to e results in many such presents which in Skt. ended in *-ayati*, such as *jeli*, *neli*, etc., besides the numerous causatives and denominatives. Our language also has many such forms; and they serve as base for forms outside the present system (§ 24.10). We may note also the isolated *pranidhyenti*, apparently for *prani-dhyāyanti* (root *dhyā*), but possibly a corruption for *prani-dhyāyanti* (root *dha*, below). **28.47.** Here may be listed a small group of presents

28.47. Here may be listed a small group of presents in *-eli* which cannot be derived from *-ayati*. It is interesting to note that several of them have, in our language, parallel forms in *-ayati*. But there can be no doubt that these are either corruptions or secondary hyper-Sanskritisms, based on the *-eli* forms.

28.48. The most important are -(s)theti, -dheti, deti (also -sthayali, -dhayati, -dayati, hyper-Sanskritic), from the roots sthā, dhā, and dā, which parallel each other in many of their forms. The forms of sthā seem certainly based on the ppp. sthila-; those of dhā and dā either directly on -(d)hila- and -dila- (cf. Tedesco, JAOS 43.388), or patterned on -(s)lheli to sthila-. Similarly from mā, tho no present *meli appears recorded, the inf. upametum is doubtless based on the ppp. upamila-, on some such pattern as jetum to jila-; cf. Pali ger. upametvā. Probably the gerund forms implying -cayali, -ceti (Chap. 43, s.v. ci 2) are newly formed to ppp. -cila, not really inherited from Vedic cayate.

28.49. To the root krī a present vi-kreti (= AMg. vikkei) and a ger. vi-kreya occur; they are probably based on the ppp. *-krīta-*, on such a pattern as *neti* (Skt. *nayali*) to ppp. *nīta-*; or possibly on the inf. *-kretum*, as in the following (?).

28.50. More anomalous would seem the impv. 2 pl. uc-chethā, if it meant 'cut off!' (? but see Chap. 43, s.v. chid 3); and the gerund acchetvā (for \bar{a} -ch°; Pali chetvā), from chid. AMg. has acche (with abbhe from bhid, Äyär. 1.1.2.5), interpreted by Pischel 466 and 516 as an old aorist (Ved. a -cchet), but used as an optative. I have the impression that such stray anomalies may have started from the infinitive, Skt. chetlum, which appears in Pali as chetum (PTSD) and, interpreted as che-tum, may have led to ger. che-lvā, and then (possibly) to forms implying pres. *che-ti.

Equally anomalous are the inf. setum and the gdve. setavya, both in MSV, from root siv, which otherwise shows no certain forms or derivatives containing e except before v; but see Chap. 43, s.v. siv (2).

Presents in *āli*

28.51. Like stems in *-eli*, and parallel to them, these form approximately a whole new conjugation (§ 24.11). Some of the presents that occur are based on Skt. root aorists: $d\bar{a}mi$, $-dh\bar{a}mi$, $-sh\bar{a}mi$, on $ad\bar{a}l$, $adh\bar{a}l$, $ashh\bar{a}l$. Further *bhāhi* 'fear' is modelled on *trāhi* (§ 28.23), and perhaps prādurbhāmi 'I appear' (tor "bhavāmi) on its antonym antardhāmi (°adahāmi) 'I disappear', but see Chap. 43, s.v. bhū (7).

28.52. To the strong present stem dadhā- are formed not only 2 pl. impv. *srad-dadhādhaam*, but futures -dadhāsyasi and -dadhāsye, ppp. -dadhāta-, inf. -dadhātum, and gdve. -dadhāniya-.

28.53. Peculiar is the future adaltādāsyalha (2 pl.: mādaltādāsyalha 'do not steall'), as if from the present °dāli; perhaps a kind of irregular denominative from adaltādāna 'theft', with thought of the regular Skt. future (a-)dāsyali; cf. ādiyali, which tho passive in form is active in meaning here and in Pali. In Pali adinnam ādiyali 'steals', adinnam is an unchangeāble adverb; cf. Vin. iii.44.30-1 rañīo dārûni adinnam ādiyissali 'will steal the king's wood', id. 35 (tvam ...) rañīo dārûni adinnam ādiyi 'you stole the king's wood'.

28.54. As we found above (\S 28.22) ya-presents substituted for root-presents in $\bar{a}(-li)$, some traces of the reverse process also occur. The impv. apachii (= apacaya) implies an *apacati = apacayati. The future jäsyate 'will be born' (Epic Skt. jäsyati) is based on the regular passive jäyate on a like model (e. g. träsyate to träyate). To svådhyäyati (denom.) are formed, similarly, fut. svådhyäsyati (besides svådhyäyisyati), gdve. svådhyätavya-, and a passive opt. svådhyäyeta 'would be studied', all implying a stem svådhyä-, pres. *svådhyäti. From paläyati are formed past participles paläta- and paläna- (\S 34.12, 19) besides regular paläyita-. Problematic are these passive forms of vi-krī 'sell': opt. vikräyetsuh, and fut. vikräsyati and vikräyisyati (with the former cf. jäsyate above), §§ 31.3; 37.39.

28.55. These ā-stems of course, like e-stems, form aorists in -si; adāsi and asthāsi may be based on dāmi and -sthāmi, nir-vāsu (-vāmsu) on -vāli, and -bhāsi 'shone' on bhāli (if it is not rather connected with bhās, with ending -i); while agāsi 'sang' is based either on the Vedic aorist agāsil, or on the rare Skt. present gāli. From kr occurs (a)kāsi (= Pali and AMg. id.), from Skt. akārşīl. And fairly common in Mv is the interesting blend form udapāsi 'came into being', based on Skt. ud-apādi 'was produced', which has been drawn into the orbit of these ā-stems by changing final -di to -si (§ 32.60).

Presents in other long vowels

28.56. The few sporadic cases of new presents (and other verb forms) based on stems in other long vowels than e or \bar{a} have been sufficiently treated in §§ 24.12-14.

Non-thematic presents

28.57. New non-thematic stems hardly occur, except for the long-vowel stems mentioned in the preceding sections.

28.58. A 5th class impv. 2 pl. asnutha 'eat' occurs once; a couple of such forms of the same root occur sporadically in Skt. (Chap. 43, s.v.). An apparent 8th class pres. of h_r , abhyuddharoli, is recorded once, but may be a corruption for 'reti; see § 28.18 and Chap. 43 s.v. h_r (3). Note the anomalous gumugumunti, 3 pl. (Dict.).

28.59. Otherwise I know of non-thematic presents only as inherited from Skt. Many of these, however, must have been fully alive in the language, since to them are created new, non-Sanskritic forms showing confusion between strong and weak stems. See Geiger 140 ff. for the same phenomena in Pali; in Pkt. they appear to be much rarer, doubtless because non-thematic forms as a whole are less frequent.

28.60. Strong for weak stem forms. 2nd Class: Root as: 1 pl. asmo SP 147.9 (vs; v.l. asma, also asme, which would be a 1 pl. personal pronoun = vayam); LV 115.15 (vs; so both edd., but against all Lefm.'s mss.; very doubtful); opt. asyā etc., see § 29.41.

Root i: 3 pl. enti (to eti) LV 74.11; 173.5; Mv i.299.7; RP 5.10; Gv 34.12 (all vss); impv. 3 pl. samabhyupaintu Divy 399.9 (vs; vv.ll. °entu, °antu; °pa-yantu would be unmetr.); entu Mv iii.206.15 (prose).

Root brū: impv. 2 sg. bravihi SP 93.11 (Kashgar rec.; ed. with Nep. mss. vadesi); Mv i.150.15 (mss.); 277.17; 280.7; RP 42.12 (all vss).

Root vid: 2 pl. vettha LV 125.7 (vs; so Calc. and all mss. except A, vittha, which Lefm. reads). Root si: 3 pl. senti (Skt. serate) Ud xvii.4; impv.

Root *ši*: 3 pl. *šenti* (Skt. *šerale*) Ud xvii.4; impv. 2 sg. *šehi* (or *sehi*; so Tib., rmi, sleep; mss. *mehi* or *mehi*) LV 196.10. But these forms may belong to (Epic) Skt. *šavati*.

28.61. Third class: Root $d\bar{a}$: 2 sg. impv. dadāhi SP 88.8 (vs) etc.; common, but may be classed as belonging to thematic stem dada- with ending -āhi. And others of its type.

Root $dh\bar{a}$: 2 pl. impv. *sraddadhādhvam* SP 44.3 (prose); both edd., no v.i. But the stem $-dadh\bar{a}$ - is also used outside of the present system (§ 28.52).

28.62. Fifth and Eighth classes: Root ap: 3 pl. pråpnonti Mv i.160.5 (vs)? So Senart, as sense seems to require; mss. pråpnoti (perhaps 3 sg. used as 3 pl., § 25.10).

Root kr (chiefly vss): 3 pl. karonti SP 84.6, 14; 85.7; 87.2; 95.11; LV 195.10; 260.4; 282.12, etc.; 1 pl. karoma SP 116.9; 2 pl. (usually impv.) karotha SP 48.9; 87.9 (Kashgar rec.); 197.1; LV 275.20; ^othā LV 37.19; prakarotha LV 43.15, etc.; 3 pl. namaskaronti KP 84.9 (prose); impv. karohi (v.l. karotha) SP 113.14; karontu Mv i.295.1; ii.93.7; pple. nom. pl. karontáš SP 83.14; — karoma is used (for Skt. karawāma) as 1 pl. impv. in LV 47.14, 19 (in the latter read kapilāhvaya, acc., lam karoma), 21. Also impv. 2 sg., § 30.9.

Root ci (vss): 3 pl. upacinonti Šiks 4.5; read upacinonti Šiks 4.4 (metr. required, for °cinvanti).

Root vr (vss): 2 sg. impv. apävrnohi SP 191.10; 2 pl. impv. prävrnohi LV 80.2.

Root śak: 3 pl. šaknonti Mv iii.74.6; 1 pl. šaknoma(h, before vowel) Mv iii.427.2; 2 pl. šaknotha Mv i.313.9 (all prose).

Root śru (mostly vss): 3 pl. śrnonli (m.c., with v.l. and WT; KN \mathcal{U}) SP 91.6 (note śrpwanti 92.1); 282.7; 325.7; 2 pl. indic. or impv. śrnotha SP 128.3; 197.6; 292.14; LV 47.5; 151.21; 296.2 (ni-ś^o); 1 pl. śrpoma(h, before vowel) Mv i.289.1 (prose); śrnoma (before consonant) Mv ii.79.9; 3 pl. mid. śrponta (for \mathcal{U} e, before vowel) Gv 236.11; impv. 2 sg. śrponi SP 93.14; 94.2; 352.6; LV 235.22 (prob. read niśrponi with Lefm.); Mv i.184.1; ii.135.16; 256.13; 327.19; iii.453.12; Suv 17.11.

28.63. Ninth class (all forms could, however, be interpreted as derived from thematic stem with lengthened thematic vowel; cf. § 30.6; only vss):

Root grah: 2 sg. impv. pratigrhndhi SP 177.13 (Kashgar rec.; ed. °grhna; both metrically good owing to a further change).

Root jñā: 2 pl. jānātha LV 314.16; 2 sg. impv. jānāhi SP 130.7 (v.l. jānāsi); prajānāhi SP 91.7.

28.64. Weak for strong stem forms (rare). Root as: 1 sg. smi Gv 231.5 (vs, after m); 2 sg. si (for asi, after consonant) Mv iii.164.15; 207.8; both prose. May be enclitic forms; but cf. § 4.7.

Root i: 3 sg. upeti common, c. g. Mv iii.110.1, 6; 1 sg. abhyupemi Mv ii.105.7. May well be regarded as a phonetic Prakritism, e for ai.

Root kr (vss): 1 sg. kurumi LV 56.11; kurumi LV 221.1. The 1 pl. indic. is used as impv.: kuruāma LV 47.12.

Root brū: 1 sg. brūmi, to be read Mv i.280.14 etc. for kruhi (see Dict. under this; Senart sometimes reads so, e. g. ii.50.2; iii.110.15; iii.457.3; in the last two, one ms. reads brumi).

Root stu: 3 sg. stuti (?), Dict.

29. Optative

29.1. Most optatives are based on thematic present stems, which as usual may be MIndic in origin. The following summarizes our finds, aside from regular Skt. forms:

1 sg. bhane, m.c. bhavi; vrajeyā, bhaveya, bhaveyyā; dešaya (?)

2 sg. prcche, m.c. himsi; abhijāneyā; bhavesi; smareyāsi; satkareyyāsi

3 sg. bhave, m. c. bhavi; bhaveya, bhaveyā, bhaveyyā, bhaveyya, bhaveyam, bhaveyo(?); bhaviya

 pl. māpaye; bhaveya; bhaveyāma, parilyajeyāmaḥ; śruņiyāma, parivrajiyāmaḥ

2 pl. bhavelha; bhaneyālha

3 pl. bhaveyu; bhave; bhidyeya; bhavetsuh, manyetsu, paribhavensuh, syanditsuh

On the endings 1 pl. (e)matha, 2 pl. tha for tu, and 3 pl. (y)u for (y)us, see $\frac{1}{2}$ 26.10, 15, 19. On the alleged 1 sg. opt. ending eham, $\frac{1}{2}$ 31.21, 22.

29.2. The really common and standard endings are e (or m.c. i), and eya, eya. Both are used for 1, 2, 3 sg. and 1 and 3 pl.; but there are also specifically 1 and 2 pl. forms of the same type. Sometimes, as in Pali, -yy-for -y-occurs in the eya-type; and sometimes i replaces e in the same. There is also a specifically 3 pl. ending etsu(h) or the like; and there are specifically 2 sg. forms in esi (rather common) and eyasi, eyyasi (as in Pali, but not common here).

29.3. After listing and discussing these forms we shall treat the remains of the athematic optative, which is commonest in forms of root as 'be'. As in Pali, this shows two types, siyā (also siya, syā) and asyā, asya (Pali assa); details below. Sporadically, other athematic optatives appear minus the final consonant (dadyā etc.), and in verses may shorten the vowel ā to a (kurya etc.). And there are a few other, miscellaneous forms belonging to the same general type.

Suppression of -ay- in stem, and the like

29.4. Ellipsis or telescoping of the syllable -ay- occurs in optatives from aya verbs, as in other present forms of such verbs. They are mostly causatives and denominatives, and as such will be treated under those headings; examples are vacet for vacayet SP 342.4 (vs), katheya for kathayeya SP 283.15 (vs); others will be included below at various places, and in § 38.27. Occasionally other presents in -aya- show similar forms: vineya, for vinayeyam, 1 sg. opt. of vi-ni, Mv i.43.4 = 54.2 = 337.8 (vs, in the last passage printed as prose and emended to vineyam by Senart). Somewhat similarly, utthi m.c. for utthe, 2 sg. opt. (impv.?) to pres. uttheti.

29.5. On a possible but very doubtful case of similar telescoping of the MIndic passive sign -*iy*-, see §§ 8.36, 37.33: LV 152.6 janeyā (for *janīyeyā?).

Non-thematic middle endings in aya stems

29.6. The use of the athematic ending in the 3 sg. middle opt. especially of *aya* presents is well known in Skt., and occurs sometimes in our language too: *parivarja-yila* Siks 114.18. See Whitney 1043 c (also 738 b), and

Renou, BSL 41(1940).9-17. And *vilakşayila* SP 193.9 seems to be a form of this same sort, but used as a preterite (§ 32.105).

Heirs of thematic optative

29.7. As in Pali and Prakrit, a form ending in e (from Skt. el, es) is very common, most often as 3 sg., but also with other persons and numbers, especially 1 and 2 sg. Tho meter cannot be concerned in the substitution of this for regular Skt. et, es, the ending e is much commoner in verses than in prose even in Mv; it occurs also in the prose of Mmk, but otherwise seems to be limited to verses, according to our editions. It may be shortened to i in verses, almost invariably where meter requires a short syllable (the few exceptions are suspicious). But, in the mss. and editions of some texts, the spelling e is sometimes retained even where meter demands a short. These need not necessarily be regarded as text corruptions; e could denote a short vowel, § 3.64. On the alleged ending eham, taken by some as 1 sg. opt., see §§ 31.21, 22. On 2 sg. forms in esi, see §§ 29.20, 21. Occasionally the Skt. ending et is found with other than 3 sg. subjects (§§ 25.11-13). All examples are from verses except those marked prose.

29.8. 1 sg. e: parāhaņe Mv i.42.8 (same by em. 53.9, where mss. unmetrically [°]haņeyam); in a parallel i.337.3, printed as prose, Senart prints [°]haņeyam, but mss. [°]hate, intending [°]hane; śraddadhe (could be 1 sg. indic. mid.) 178.18; bhaņe 366.7 (as preceding); samname ii.93.9, parivartaye 10; bhumje 164.14; parāmrše iii.8.7, 9; prapāraye Suv 30.4 (v.l. [°]yet, [°]ya); paśye Bhad 25.

29.9. 1 sg. i: bhavi SP 11.2; 252.9; LV 199.16, 17, 18; Bhad 16, 27; read in Mv i.201.11 = ii.5.6 (vs) fayanavaram āruhi (1 sg.; or ger.?) yatra kumudavasanasamnibham vimalam; ulsimci Mv ii.92.6; prafami Sukh 23.5, prafāmi 6 (both for prafamayeyam or prafām^o); upanayi 23.10; defayi Suv 51.14; vimocayi 51.15; defayi Bhad 18; virāgayi 24; olari 29, 30, 34; ahhinirhari 34; upasamkrami 35; pūrayi 41; jānayi 45 (to jānayati = janayati, not to root jāā with Leumann); kari 58.

29.10. 2 sg. e: The forms noted here are, I believe, all injunctive (rather than conditional, 'potential') in meaning. More than once they are closely parallel to imperatives in the same context. The question may be raised whether they, or some of them, might not better be regarded as imperatives from e-presents, rather than historic optatives. Note specially vade 'speak thou!' SP 116.4; the present vadeti is familiar both here and in MIndic generally; also utthi for utthe, to uttheti, § 29.11. The usual 2 sg. impv. from e-presents ends, however, in chi (also in Pali and Pkt.). The question is not of much importance, since the injunctive optative cannot really be distinguished from the imperative in meaning, even in Skt. Even in negative commands, the use of mā means nothing, for it is freely used with opt. forms in our language. Cf. the Ap. forms in e or i, called impv. by Hem. 4.387, whereas Pischel 461 and Jacobi, Sanatk. § 23 regard them as historic optatives.

vade SP 116.4; vidhamaye LV 117.6; preche 231.17; anuvicintaye Lank 11.13 (prose); nivartaye Mv i.75.19; ii.142.3; vade i.77.5; udāhare 77.6; unukirtaye 77.8 (cf. anukirtaya 77.3, most mss. $^{\circ}y\bar{a}$); vyāhare 276.11 (v.l. $^{\circ}rel$); nigrhņe, pragrhņe 278.6, saṃgrhņe 7; upasaṃkrame 330.15 (prose; repetition of 330.13 where mss. $^{\circ}kramya$, Senart em. $^{\circ}krame$); nivedaye ii.37.12; āgache (parallel with presehi, impv.) 90.2 (prose); mā luaṃ ... gacche (parallel with preșehi, as in prec.) 169.17 (prose); mā ... krtaṃ na jāne, 'do not fail to recognize (gratefully) what I have done (for you)' 184.8; jaye 322.7; jane (= janaye[ħ]) 336.7; prārthaye 483.15; darśaye (parallel with ehi) iii.2.17; mā ... kilāmaye (= klāmayeħ) 19.15; dahāpaye (v.l. $^{\circ}yet$) 20.8, kāraye 9, āropaye (mss. $^{\circ}yet$) 10; ghālaye 387.13; ālape Šikş 100.16 (from Samādħ); mā ... gile Ud xxxi.31.

29.11. 2 sg. i: utthi arise!', m.c. for utthe, 2 sg. opt. (impv.?) to present uttheti; I.V 236.4: 322.3; Gv 488.18; probably also in Gv 488.16 read utthi ho (= aho) for text utthiho (cf. \S 32.7); himsi Mv ii.236.6; yāci iii.419.5, mss.; vs but not metrically required; Senart em. yāce; loam cari (so ms., Finot em. cara) RP 39.4; vardhayi Gv 488.14.

29.12. 3 sg. e: upasthape (Kashgar rec. and WT) SP 88.12; bhāse 98.7; dhārage 98.12; dhāre (= dhārayel) 255.10; 293.8; preche 279.9; kṣame 334.15. — bhave LV 56.1; 138.12; name 120.9; manye 'would esteem' (Foucaux wrongly) 208.22; suşya (= suşyel) udadhih... prapate 330.3; in 386.17 read with best mss. āhāram upanāmye 'yam, 'let him be offered (passive) food'. — ācare Mv i.12.14; viroce 70.7; bhave 155.9; gacche 274.18; ii.169.2; iii.373.17; tyaje i.310.12; ārāgaye ii.120.1 (prose; mss.); āgacche 121.8 (prose); iii.425.19 (prose); abhinipide (mss. °şide) ii.124.6 (prose); āgame (tò 'aorist' stem āgama-) ii.141.7, 8, 9, 10; iii.306.4; bhave ii.146.13 (prose); bhāse 257.15 (prose); 259.3 (prose); anunade anurane 282.12; 412.9 (prose); janaye 295.14 (prose); etc. bhave Suv 31.12 (all mss. Nobel em. bhaven); bhave (ms., Finot em. bhavel) RP 13.14; vimuhye (v.l. °yet) Bhata 19; abhilikhe Mmk 63.25; 65.3 (both prose); ālikhe 66.21; abhiligacche 82.9-10 (prose); brīņhaye Ud xviii.5 (later v.l. brīmhayel; possibly 2 sg. with Chakravarti); prajahe (later v.l. °hed) xx.2.

29.13. With e where meter requires short syllable (cf. § 3.64); labhe Mv ii.319.15, 17, 18 (3 pl.), 20, 21; 341.2; jane (= janayet) 378.15; 390.23; apanaye 393.13, 17, 20; name 395.11.

29.14. 3 sg. 1: anumodi SP 58.6; praji 111.2; padi 98.7; bhāşi 236.8, and so La Valke Poussin JRAS 1911.1074 at end of a pāda bhāşi (i not required metrically) for 256.6 bhāşel of both edd., no v.1.; preşayi 113.6; kārāpayi 114.11; etc., common. — bhavi LV 42.11; 330.19 (read bhavi yah); 356.2, 3; pravarli 48.11; cintayi 314.18; rami 325.8. — vibuddhi (= vibudhyel) Mv 1.155.19; jānayi ii.324.19; sthihi 329.18; kari 374.13; nirvāyi (or aorist?) 375.8; upagami 387.15; apanayi 393.5; varşayi iii.252.18. — icchi Samādh 8.7; virāji RP 54.20; pravartayi, dharşayi Bhad 53; dhārayi vācayi deiayi (i)ko vā 54 (ed. deiayilo as one word); drši Šikş 341.5; prayuñji 343.2 (see note); adhişthihi 343.13, 15; darfayi 343.14; pravraji Sukh 22.12.

29.15. 3 pl. e: bhave My il.319.18; 328.18; parāmrşe 366.12 (for °*ie*).

29.16. 1 pl. ema: in LV 422.6 bhavema, both edd., no v.l.; the meter seems to demand short e in the penult, cf. § 3.65. On the very dubious paricaremha see § 26.7.

29.17. 1 pl. e: (vayam...) māpaye Mv i.351.19 (prose; so mss., Senart em. °yemaḥ).

29.18. 3 pl. etsu(h), ensu(h): a 3 pl. form, corresponding to the 3 sg. e(l), and according to the overwhelming testimony of the mss. ending in *etsuh* or (less often) *etsu*, occurs often in the Mv with optative meaning. The same form is also used as a orist, but most likely originated in the opt.; see §§ 32.95 ff. Rarely the mss. read ensu(h) or emsu(h); but Senart always reads one of these, usually the former; see my discussion l.c. A few

examples of optative use of the form follow; all are taken from prose except those marked vs.

manyelsuh Mv i.39.4; 40.1; 60.8; 239.1 etc.; śrunelsuh 42.12 (vs) = 53.7 = 337.2; bhavelsuh (twice), varielsuh 61.3-5; nyākarelsuh 197.9; vidyelsuh 273.2; npasamkramelsuh, samyuiyelsu (v.l. °tsuh) 314.6; manyelsu, hāyelsu, abhivardhelsu 330.7-10 (repeated 331.7-10; there are vv.ll. °lsuh, and once °nsuh); abhisincelsuh 350.12; pralishāpayelsuh ii.64.18; bhavelsuh 178.10; 241.20; iii.107.5; 162.3 etc.; abhyokirelsuh ii.131.6; sammiläyelsuh (root mlā; one ms. °nsuh) 178.8; vahelsuh 180.2 (v.l. °nsuh; in verse version of same passage 185.5 both mss. °lsuh); in 241.20, read vikräyelsuh (pass. of vi-krī, see §§ 31.3; 37.39); vihanyelsu iii.36.12 (v.l. °lsuh); budhyelsuh (v.l. °nsuh) 74.2; vuccelsuh (pass. to vac) 194.12; ulpadyelsuh, dešayelsuh, häyelsuh, abhivardhayelsuh (repeated twice; sometimes v.l. n for 1) 199.1 fl.; chindelsuh 18; prajäyelsuh 317.8, prajňāyelsuh 9; paribhavensuh, ulpādayensuh (v.l. in both °lsuh) 393.14.

29.19. The mss. also may show the ending *ilsuh* (cf. the same in the aorist, § 32.38); *syandilsuh* (Senart em. °*nsuh*) My iii.317.8 (prose).

29.20. 2 sg. esi: Senart, Mv i.558, note on i.233.8, recognized a 2 sg. opt. ending esi, and explained it correctly as based on historic 2 sg. e (for es) exactly as Pali eyyäsi, Pkt. ejjäsi, is based on eyya, ejjä (cf. \S 29.37); the 2 sg. pres. indic. ending is of course the source of the addition. The form seems not to be recognized elsewhere. Yet it occurs in Pkt., at least in Säkäri: bhanesi 'you are to say', four times, Mrcch. ed. Stenzler 21.8 and 20.

29.21. These forms in esi (always with dental s) are well authenticated in our language, even in prose in My. and in verses of other texts. Prose examples are so marked. prakāśayesi SP 98.2; vadesi 93.11; 98.4, 6 and 99.6; paśyesi 98.4; upadarśayesi 98.8. (Note in parallel formulas samśrāvayes 98.12, śrāvayes 97.8, 10, 14, bhanes 97.12.) In 99.4 perhaps read prakāśayesi with Kashgar rec. (ed. prakāśayer idam by em., Nep. mss. "yed idam, which need not be emended); in 209.8 perhaps read kathayesi with most Nep. mss. for ed. with Kashgar rec. kathayāhi. - bhavesi (parallel to bhaveyam same line) Mv i.233.8 (prose); vadesi 254.16 (prose), prcchesi ... vadesi 17 (prose); jānesi 275.17; vandanam vadesi 'speak a greeting!' 323.19 (prose); gacchesi ii.42.17, carayesi 18 (mss. var°); icchāmi ... yam ... mama bhavanam pasyesi 179.18 (prose), 'I desire that you should see my house'; vijānesi (answer to 1 sg. vijāneya, line 6) 182.15; prcchcsi 'askl' (but v.l. prcchasi) 189.13 (prose); vadesi 194.12; iii.292.7 (repeated 18; 293.4, 9; 294.1); ägamesi ii.199.20; kathayesi 200.1; gaechesi 214.15, and prechesi 16 (both prose; clearly injunctive; v.l. gacchāsi); bhanesi 224.7, 8; upasthapesi 247.10 (prose); na ... yācesi 406.1, conditional, 'you would not be begging'; jānesi 449.6; 454.1; 457.16 (all prose); kārāpayesi 486.5 (prose); upasthihesi 491.12 (prose); khādesi lii.7.4, 8 (prose); bhavesi 27.14 (prose); āgacchesi (Senart em. °ehi) 159.5 (prose); yadā bhagavam . . . bhavesi (10) ladā . . . paribhuñjesi, 311.9-10 (prose), 'when, Lord, you become ... then make use of . . .'; bhāşesi 384.16; rakşesi 387.7 (v.l. rakşasi, but opt. proved by parallel kuryāsi line 6 and opt. forms in same vs in Pali, Sn 702); agacchesi 406.5 (prose); paśyesi Śiks 354.11.

29.22. So-called 'subjunctives' in $\bar{a}si$, $\bar{a}ti$. On such forms, apparently or supposedly with modal (sometimes potential) force, see § 27.4 fl.

eya, eya, and cognate endings

29.23. From thematic present stems, our language very often shows optative forms (mostly 3 sg.) ending in eya, $ey\ddot{a}$ (also eyya, $eyy\ddot{a}$, $eya\ddot{m}$, iya etc.; forms with double

y are relatively rare). They are obviously to be connected with Pali eyya (Geiger 128), Pkt. $ejj\bar{a}$ (Pischel 459). In Pali, no *eyyā is recorded, while in Pkt. $ejj\bar{a}$ seems to be the regular form; according to Pischel ejja only in verses, presumably where meter requires a short final, and in prose before enclitics.

29.24. To derive this ending directly from nonthematic 3 sg. forms in uat, with Pischel, seems to me impossible. Geiger's vlew is much more reasonable, that it is analogical to 1 sg. ey(y)am (Skt. eyam), and (perhaps also) 3 pl. ev(u)u (Skt. evuh). But the athematic opt. endings Skt. yām, yās, yāi (Mindic yam, yā, yā; also, at least in our dialect, ya for all three persons) provided the pattern on which 2 and 3 sg. $ey(y)\bar{a}$, ey(y)a were created to match 1 sg. e(y)yam. To this extent Pischel's explanation contains an element of truth. There are even traces, tho rare and doubtful ones, of a semi-Sanskritic ending eyat. For SP 102.9 (prose) *ācakşed*, the Kashgar rec. reads *ācakseyād*; and at My ii.135.17 Senart reads *bhaveyāt*, at the end of a line of verse (but v.l. bhaveyā). Cf. also the endings 2 sg. yasi and 3 sg. yati, for Skt. yas and yat, in original non-thematic optatives, and also preceded by the thematic mode-sign -e- (§§ 29.36-38, 41).

29.25. This theory would seem to imply that (as in Pkt. according to Pischel) the final a was originally long (tho in Pali it seems to be always short according to Geiger). In our dialect, however, both a and a seem to be used interchangeably. That at any rate is true of the prose of Mv, where both are common; and in some other texts (such as KP and Ud), short a occurs either in prose or in metrically indifferent positions. On the other hand, some texts, notably SP and LV, which as presented in our editions do not use either form in prose, always or practically always use eyā in metrically indifferent positions (notably at the end of lines in SP), and eya only where meter requires a short final. The only exception I have noted in these two texts is SP 284.1 nirdiseya, at the end of a line of verse; but the Kashgar rec. has nidarsayet (with other changes).

29.26. From present stems in *aya*, the syllable -*ay*-may or may not appear before the -*ey*- of optative endings; cf. §§ 29.4; 38.27.

29.27. In the following examples, prose passages are marked as such; except Mv, most texts record the forms only in verses, or nearly so.

29.28. eya, eyä, 3 sg: SP: (length of ultima metrically determined) vijñapeyā 284.8; bhaveyā 313.7; — bhaveya 53.3; 195.7; bhaneya (v.l. 'yya) 58.6; ksapeya 111.14; dadeya 119.11; 229.4; likheya, dhāreya, śruneya 229.6; vadeya 229.10; 287.1; 352.3; deśeya 284.2; prakāšeya 296.2; kalpeya 344.2; tarpeya 351.4; śruneya 352.5; — (ultima metrically indifferent) chādayeyā 228.15; samslaveyā 229.11; prakāšeyā 237.2; vadeyā 254.6; iccheyā 278.8; deśeyā 280.9; katheyā 283.15; vijňapeyā 284.8; prakāšayeyā 284.11; vadeyā 293.6, etc.; — nirdiśeya 284.1 (ct. above).

LV: (metrically determined) bhaveyā LV 57.3; phaleyā 81.19; janeyā 152.6; niskrameyā 201.21; abhiniskrameyā 202.6; — jareya (tor jarayeya) 74.3; bhaveya 141.4; nyaseya 151.8; daršeya 284.5; dadeya 288.4; caleya 330.3; kareya 335.19; ganeya 338.7; dešeya 393.2(tor dešayeya); — (ultima metrically indifferent) shiheyā 42.18; āruheyā 80.10; in 80.12, read śruneyā (or [°]yyā) for śunesyā of text; bhaveyā 200.7; gaccheyā 201.14; vrajeyā 201.18; vindeyā 201.22; āgameyā 330.17.

Mv (prose) bhaveyā Mv i.60.8; 272.3; 311.18; 356.4; ii.65.16 etc. (in i.60.8 Senart em. °yam; in i.272.4 bhaveya); ägaccheyā 1.255.1; 284.10; (the first passage repeated with °eya i.256.13); samyujyeyā i.295.18 (mss. sampūjy°); pragrhneyā 347.18; mueyeyā ... hanyeyā 364.2; anujāneyā ii.119.15; ākrameyā 146.12 (twice), hareyā 13, vipravskeyā 14; rajyeyā kalpeyā pramadyeyā 147.13, etc.; — bhaveya i.44.7; 61.6; 199.1; 273.12; ii.2.19 etc.; gaccheya i.258.13; deieya 316.17; upasamkrameya 320.8; khajjeya (for khādy°, passive) ii.78.1; āgaccheya 102.17; samanusmareya 103.8; dāleya sampradāleya chindeya 125.4-5; budhyeya 160.5; hareya, bandheya 178.16; upasthiheya 214.6; iobheya iii.10.16; parivarteya 32.2, etc.;—(vss, final determined by meter) pateyā ii.324.21; 329.17; abhyuddhareyā 393.8, dadeyā 18, etc.; — rdhyeya i.299.14; iccheya 359.10, 14: parikseya ii.67.7; ioseya 329.7; muñceya 331.1, etc.

(Prose) vindeya (text bi°) and (?) vāpeya KP 158.3; śruņeya 159.17, parikšipeya 18.

bhaveyā Mmk 168.9, 10 (dodhaka vs).

bhāșeya, vihimseya Ud vili.12; iccheya xvi.14; viprajaheya xx.1. All in vss, but only in xx.1 does meter require short final. All are found in the most ancient ms.; later mss. replace all by forms in *eta* or *et.* Cf. § 1.43.

29.29. eya, eyā, other than 3 sg.: As usual, these forms are used for other persons and numbers. It may be noted that eya as 1 sg. could be interpreted as historically identical with the Skt. 1 sg. opt. middle ending. I doubt whether this would be a realistic view to take of the ending, in view of the relative rarity of middle endings, and the frequency of the use of 3 sg. forms as 1 sg. and also of the phonetic alteration of final am to a. Note e. g. Suv 37.6 ff., where we find a long series of 1 sg. opt. forms (bhaveya etc.), in eya where the meter favors a short final, but in eyam where it favors a long (37.19 anusmareyam, 38.3 vivarjayeyam). As this suggests, the fact that any final am may become a, at least in verses m.c., may have played a part in the use of eya as 1 sg. (for Skt. eyam). But note that eyd is also so used, and even in prose.

1 sg. eyā: vrajeyā LV 78.7 (vs); bhāşeyā Mv ii.180.2, jāneyā 3 (both prose); šakyeyā 249.1 (prose); nikhaneyā (Senart em. °yam) 436.3 (prose); labheyā Bhad 59 (vs).

1 sg. eya (vss except those marked prose): āgaccheya Mv i.186.10; iccheya ii.70.6 (prose); gaccheya 170.4 (prose); 205.18; vijāneya 182.6; bhaveya (Senart em. bhave, m.c.) 299.13; careya 299.14, varjeya, seveya 15 (Senart em. varje m.c.; he keeps seveya, to which there is just as much metrical objection); grāheya 330.18; kareya 424.17 (prose). 2 sg. eyā: abhijāneyā Mv i.75.10 (vs); parirakseyā

277.8 (vs). 1 pl. eya: vayam pi bhaveya Mv iii.138.9 (prose; so

1 pl. eya: vayam pl bhabeya MV 111.138.9 (prose; so mss., Senart em. bhabema; to be sure mss. have bhabema in the same sentence, line 11).

3. pl. eya: chidyeya bhidyeya Sukh 45.13 (vs).

29.30. eyya, eyyä: In mss. and sometimes in printed editions the y in this ending may be written double, as in Pali (and Pkt. ejjā). The misspelling eryyā, eryyā is also recorded. ākrameyyā (apparently all mss. have yy; v.l. 'eryyā) LV 199.15 (vs); bhaveyyā (v.l. 'eṣyā) Mv i.331.8 (prose); prayojeyyā (v.l. 'eyā) 279.16 (vs); anujāneyyā 351.18 (prose); krīdeyyā rameyyā (v.l. 'eyā) pravicāreyyā (v.l. 'eyā) ... kareyyā ii.144.9-10 (prose); kareyyā 144.16 (prose); bhasmikareyyā (Senart em. 'eyā) 178.4 (prose); vadeyya (v.l. 'eyyā, both times):233.13, 18 (prose); bhaveyyā (v.l. in all 'eyā) Bhad 8, 23, 24, 46 (vss); bhaveyya (v.l. 'eya) 45; vadeyya Dbh.g. 12(348).9.

As 1 sg., prakāśeyyā (v.l. eyā) Mv i.336.20 (vs); in same vs i.53.4 mss. eyyā, Senart em. eyyā; in i.42.11 = 53.6 = 337.1 (vs, in 337.1 printed as prose), prob. read in all nivešeyyā (with Senart in 337.1, and most mss. in 53.6); read so also in i.337.4 (vs, printed as prose); bhaveyyā Bhad 16 (vs; v.l. eyd).

29.31. eyam as 3 sg.: The historic 1 sg. ending eyam is, in the mss. of Mv at least, fairly often used as a 3 sg. Senart always emends to eya, but I doubt whether he is right. The ending eyam as 3 sg. could easily be an analogical reflex of the use of eya as both 3 and 1 sg., the forms of the two persons being interchangeable to a

large extent. At any rate I record here a number of cases. mostly in prose (others are marked vs): ulpadyeyam Mv 1.44.15; bhaveyam i.145.17 = 202.14 (vs; here meter seems to favor bhaveya; yet the same vs ii.6.11 reads bhaveyā, kept by Senart); yeyam ii.233.3; upapadyeyam iii.43.11; avatārayeyam 92.5; bhasmīkareyam 158.6; anuśāseyam 215.2; kathayeyam 390.10; samyujyeyam 406.12; vaseyam 432.19; deseyam 437.8; bhaveyam 456.21 (vs).

29.32. eyo 3 sg. (?): More doubtful is the ending euo, sometimes found in the mss., usually as 3 sg., e. g. My ii.70.2 bhaveyo (Senart em. °ya); prapaleyo (Senart em. vā; both these are prose) ii.179.20; and others. It might represent *eyas, a theoretically 2 sg. form (which would be analogical to 1 sg. eyam, on the model of imperfect indicatives), used as 3 sg. But perhaps Senart is right in taking it to be a mere corruption.

29.33. eyāma(s), eyātha, 1 and 2 pl.: To this same pattern belong the 1 pl. ending eyāma(s) and the 2 pl. eyātha (Pali eyyāma, eyyātha): in SP 163.2 (vs), read with Kashgar rec. anuprāpuneyāma for ed. anuprāpnuyāmo 'tha (the patchword 'tha proves that the semi-Sanskritized form is secondary); bhaveyāma Mv i.44.8 (prose); parilyajeyāma (v.l. °iyaje) 313.6 (prose); parityajeyāmah ii.105.13 (end of line of vs). — prccheyātha (v.l. °yatha) Mv ii.102.18 (prose); bhaneyatha 105.4 (vs).

29.34. iyā, iya, iyāma(s): Any of the preceding endings may occasionally be read with i instead of e before the y. Usually such forms appear in verses where meter requires a short syllable instead of e, so that i may be considered a shortening of e, m.c. But sometimes My mss. have such forms even in prose, as in the two My citations below. All the others here cited are from verses. lişthiya SP 251.6; bhaviya LV 78.11; śruņiyāma ('tha) 364.11; pravrajiyāmah Mv ii.234.1 (prose; Senart em. eyāmah); in iii.30.8, read with mss. okaļļiyāti (or 'ļiyā; Senart em. okaddheyā); paśyiya Suv 52.5 (em., mss. mostly paśyami, the best one pasiya); paśyiya Bhad 28, 33, 57 (in 28 as 1 sg.); vibudhyiya 41 (1 sg.).

29.35. aya (?), 1 sg.: In Mv i.43.3 = 54.1 = 337.7 (vs, in 337.7 printed as prose) devamanuşyāņām dešaya dharmam seems to be the reading intended by the mss. (they read mostly desaya or desaye, except at 54.1 where mostly deseya or 'yam), 'may I preach the law to gods and men'. Meter seems to require - - -; Senart em. desiya (in 337.7 deseyam) which he interprets as a gerund (implausibly). Parallel with vineya, in next line, 'may I train ...', 1 sg. opt. to vineti = vinayati; vineya is telescoped for vinayeya, and probably has led to the nonceanalogical-creation deśaya (vineti : vineya = deśayati : desaya). However, the meter is questionable; there may be a corruption.

29.36. 2 sg. (e)y(y)āsi: Clearly similar to esi (§ 29.20) in origin are 2 sg. optatives in yāsi, e(y)yāsi. The first is recorded in kuryási = Skt. kuryás, Mv 1.276.8; 277.2; 278.1; 279.5 (here mss. °asi, °osi); iii.160.6 (prose, in a mā clause); 387.6; Šiks 100.16 (kuryāsi, i m.c.) and 101.2 (all but one of these in verses). Cf. siyāti = syāt, § 29.41.

29.37. The ending eyāsi is a similar extension of eyā (eya), above. It is paralleled in Pali eyyāsi, Pkt. ejjāsi (also ejjasi), but is rare in our dialect: jāneyāsi Mv ii.488.15 (prose; v.l. jānesi); smareyāsi ili.20.12 (vs), 'you are to remember'; pithiyeyāsi 25.6 (prose), so read with 1 ms. (v.l. pihi^o), 'you are to cause to be covered over'. 29.38. The same occurs with yy, as in Pali (cf. § 29.30);

samlapeyyäsi Šiks 100.16, satkareyyäsi 17 (both vss).

Non-thematic optatives

29.39. Forms of root as, 'to be'. In Pkt. this root has, according to Pischel 464, 498, only in AMg., an opt. siyā = Skt. suāt. In Pali the opt. of this root is better preserved. Geiger 141 records 1 sg. siyam and 3 sg. siya, but also 1 sg. assam and 2 and 3 sg. assa (besides plural forms). The latter type represents a Skt. *asyām, *asyās, *asyāt, with spread of the strong stem from the indicative singular.

29.40. Our language has forms corresponding to both these Pali types. So siyā as 3 sg., and siya in verses m.c. as 3 sg., 1 sg., and 3 pl.; also syā, the Skt. form but for lack of final t, and siyāti, with indicative endings as in Pali labheyyāli etc. (Geiger 127); tho siyāli is not recorded for Pali or Pkt., it has correspondents in the Asokan inscriptions, Hultzsch, lxxxii and 40 note 1. There is also an astonishing form stiya (?), on which see below. Of the other type we find asyā not only as 3 sg. but as 1 and 2 sg. and 3 pl.; also asyat, semi-Sanskrit, and asya, only in verses but, according to the mss., not exclusively m.c. (sometimes at the end of lines where meter is indifferent). No specifically 1 sg. form siyām or asyām, or the like. seems to be recorded.

29.41. siyā (only vss): 3 sg. Mv ii.353.15; iii.1.15; 365.7.

siya (only vss, and apparently a only m.c.): 3 sg. LV 271.4; Mv ii.331.10, 14; 332.4, 8; Sukh 22.7; 50.12; 3 pl. Mv ii.340.12 (Senart em. siyu); 1 sg. Mv ii.396.18; Sukh 22.5; Siks 302.3, parallel to Mv ii.375.10, where mss. (katha) yasi, read siya!

siyāti (see above): Mv i.45.1 (prose), read siyāti for Senart siyā ti (interpreted as = te, but this ti never occurs except in verses m.c.); 358.13 (vs), read siyāti snehā, with mss., for Senart sigātisneho; proved by the same vs in Pali, Sn 36, bhavanti snehā (so read); siyāti 3 pl. = Pali bhavanti.

syā: 3 sg. Mv iii.314.6 and 17 (both prose; after -e, could be interpreted as asyā, below).

stiya (1): 3 sg. Šiks 115.2, 3 (vss) yatra prapañco stiya vigraho va ... muhurtamātram stiya yatra kiešah. It seems impossible to interpret as (a) sti ya(h), and the repetition makes one hesitate to assume a corruption. No MIndic form *thiya or *thiya (or °ya; the short a may be m.c.) has been discovered. If correct, the form would seem to be a blend of opt. siya (siyā) with indic. *sti (= asti; cf. AMg. tthu, 3 sg. impv., Pischel 498). Note that in the first occurrence the form follows final -o.

asyāt: 3 sg. (all prose) Mv i.286.13 (v.l. asya); ii.67.8; 68.13; 81.17, 20; 313.17; iii.440.11, etc.

asyā: 3 sg. SP 335.10 (vs, so with Kashgar rec.); Mv i.42.10 = 53.5 (vs) = 337.1 (here printed as prose; mss. here asyān, asmān, Senart em. asyāl); Mv 1.213.14 = ii.16.19 (? the first, where mss. read asyāh, printed as prose, the second, where mss. asyd after -e, printed as vs; possibly asyāh, gen. sg. fem. of ayam, is intended); 1.277.10 (vs, 3 sg., read janādhipah with mss.); 290.3 (prose); ii.64.2 (prose, v.l. asyād); 204.17 (vs); as 1 sg. My i.42.6 = 53.3 = 336.19 (vs; in the first, mss. and ed. asya, without metrical necessity, at the end of an arya line; in the others mss. asyā, Senart em. asyām); as 2 sg. Mv ii.406.1 (vs); as 3 pl. SP 197.4 (vs).

asya: 3 sg. (only in vss, but rarely where meter demands short ultima) Mv ii.228.9; iii.1.16; 124.3.

29.42. Other athematic optatives in ya. ya. Even in prose of Mv, and in verses elsewhere, the final t or mof the 3 or 1 sg. may be lost as in MIndic; and the \bar{a} may be shortened in verses where meter requires it.

yā, 3 sg.: dadyā Mv i.80.6 (prose); iii.420.1 (vs, Pali dajjā in the same vs, Jāt. iii.355.7); sameyā Mv ii.404.8 (vs); rdhyā Mv iii.335.18; 336.7 (both prose); nigrhņiyā Mv i.347.18 (prose); in Mv i.156.7 (vs) Senart kuryā (v.l. kuryād), but as his note observes, meter requires kurya.

yā, 1 sg.: śrnuyā LV 42.19 (vs; two mss. with Calc. °yām).

ya (only vss, m.c.), 3 sg.: kurya SP 157.12 (cf. Mv

i.156.7 above); śrunuya LV 54.2; praviliya (so read with best mss. for ed. °liyu) LV 330.19 (3 sg. opt., or 'precative', for 'yddi; 'the heart of him who might be [bhavi yah] passionate would melt', pra-vi-li, intrans.); upeya Mv iii.369.21; as 1 sg., kurya Bhad 47 (v.l. dadyu, apparently for dadyām!).

29.43. Miscellaneous athematic optatives. A 2 pl. opt. dadyatha occurs in Mv ii.105.3 (vs): 'you are to give'. So the mss., with metrical correctness; Senart em. däsyätha (metrically bad). The form of the mss. is correct; it is a 2 pl. to a MIndic dajjam (= dadyām, as in Pali) with thematic short a (or merely m.c. for \bar{a} ?), exactly like Pali 2 pl. opt. assatha to assam, assa (Geiger 141).—In SP 102.12 (prose), for ed. bhaveyam, the true reading (see KN note) seems to have been probably bhûyām, a 1 sg. 'precative' analogical to bhûyās, bhûyât (Skt. bhûyā-sam).

29.44. According to Lefmann's ed. I.V 444.2 (vs) has *vadyād*, 3 sg. opt. of *vad* = Pali *vajjā* (Geiger 143 d). So ms. A; but other mss. and Calc. *dadyād*; according to

Foucaux, Tib. 'comprend', which, F. suggests (in Notes 210), points to dadhyād. The Tib. is hdzin (commonly = Skt. dhar-) byed, 'make grasp' or the like; dadhātu line 4 (Tib. 'comprendra', Foucaux) is rendered by the related gzum (bar bya), which gives some support to F.

29.45. In Gv 522.16 (prose) srnyāt may be a mere corruption for srnuyāt.

29.46. In SP 230.4 (vs) *sruno* is read with two mss.; two others *srnod* or *sunod*; meaning is 3 sg. opt.; conceivably based on the strong stem *sruno*- (in whatever phonetic guise) on the analogy of thematic optatives in -e(d). But see § 32.124.

29.47. In Mv i.69.18 (vs) the mss. present krāyur, which Senart emends to kreyur, allegedly a 3 pl. opt. (for kareyur? at any rate supposed to be from kr 'make'). Senart's form is as monstrous and unprecedented as that of the mss. Possibly read karyur, which would match Pali kayirā = *karyāt (Geiger 149 d)?

29.48. On kuryāsi for kuryās see § 29.36.

30. Imperative

30.1. Our section on the imperative is relatively brief and contains little of interest except for the spread of the 2 sg. ending hi (or m.c. hi, § 26.2) far beyond its Skt. limits, as described just below. Otherwise we shall note a few odds and ends of endings, mostly containing no surprises; but it is worth remark that the 2 sg. (originally middle) ending su, common in Pkt. and (in the form ssu) in Pali, hardly occurs here. On *i* for thematic *a*, not limited to imperatives, see § 27.10.

Spread of 2 singular ending hi

30.2. As in Pali and Prakrit, the ending hi is used much more extensively than in Sanskrit. Stems in aya (e), both causatives and denominatives and others (such as nayati), very commonly have 2 sg. impv. in ehi. (There are also, from such stems, 2 sg. forms ending in e, or m.c. i, which have injunctive force, and which I have grouped with the optative; see § 29.10 on the rather fruitless question whether it might be better to call them imperatives.) This ending also spreads to other thematic stems. But much commoner with them is the ending ahi, as in Pali and AMg., where the penultimate vowel is reported to be always long. In our dialect too it is usually long. But ahi, which Pischel records only in Ap. and regards as a shortening of ahi, is found many times in LV, tho only in verses. Elsewhere it is rare; but the prose of Mv shows it a few times, so that it can hardly be due exclusively to metrical shortening. Finally, hi is sometimes found after other vowels than e, \bar{a} , a (chiefly o, but kuruhi also occurs). Sometimes the final i is lengthened in verses m.c. All examples are from verses except in the case of Mv, from which only prose examples are cited unless the contrary is indicated by (vs).

30.3. ehi or m.c. ehi (with e for Skt. aya): janehi SP 15.12; vyapanehi (= vyapanaya) 15.12; 49.1; deiehi 167.5, 6 etc.; tārehi 167.6; 170.15; mocehi 174.4; tarpehi 190.12; dhārehi 209.7. — pālehi LV 79.17; tarpehī (ī m.c.) 164.10; bodhehi 177.22; vicintehi 184.2, 18; udīrehi 185.16; sehī or śehī (mss. mehī; ? see § 28.60) 196.10; janehi 237.1; samehi (= śamaya) 359.11; 414.13; vinehi 370.16. — Mv (all prose): utpādehi Mv i.233.10; anujānāpehi 256.14; ninesţāvehi (so read) 273.14; ārocehi 287.10, yācehi 11; bhojehi 309.10; millehi 363.14; iii.291.17 (with mss.), and mellehi ii.448.4; 454.16; preşehi 1.362.17; visarjehi 363.9; varehi ii.70.4; kārāpehi 73.18; adhyāpehi 77.16; paricārehi 103.6; pravicārehi 14; ānapehi 108.15; nehi 108.16; 428.16, etc., common.

30.4. ihi for ehi (?), probably textual error: In Mv ii.340.5 (vs) the mss. have *ullhihi* 'arise!'; but the meter requires a long penult; Senart em. to *ullhāhi*, but *ullhehi* is more likely, to present stem *ullhe(-li)*.

30.5. ehi (with e for Skt. a). Here are presented some examples of this ending analogically attached to stems which, in Skt., would not contain aya; or if they would, -ay- is preserved before *ehi* as in *dhārāpayehi* 'cause to hold' Mv ii.447.14 (prose; so mss.; Senart em. *dhārāpaye)*. Such cases are important because e in them is clearly not a MIndic phonetic resultant from *-aya*. Some of the other cases, like *vadehi*, familiarly show in

MIndic, and in our language, presents like vadeti and other form<from stems in -e-. With the qualified exception of such cases (even they, of course, also show regular Skt. presents like vadati), all examples are based on thematic presents (some MIndic) in ali. Examples not marked (vs) are from prose:

gacchehi Mv i.187.1, and paśyehi 4 (both vss; so all mss., Senart em. °āhi); vadehi 324.10; 330.12; iii.159.4; 173.18; ägamehi ii.31.20; iii.161.10; 186.17 (from 'aorist' stem āgama-); upasevehi ii.103.13; samanuśāsehi 103.14 (Pali has anusāseli beside °ali); nivartehi (not caus.) 105.8 (vs); āvasehi 405.9 (vs); dhārāpayehi 447.14; śruņehi iii.140.13 (vs); ulkanlhehi 'grieve' (an a, not aya, stem; note ulkanlhanlasya in preceding line) 167.9; prāpuņehi 172.11; praligrņņehi 211.10 (vs); prāpuņehi 270.14 (vs) (so mss., Senart em. °āhi; in line 16 Senart prāpuņāhi with one ms., the other °ehi).

30.6. ähi or (m.c.) ähi. Cf. § 30.2. Chiefly in verses except in Mv, where it is extremely common in prose and vss alike. Our Mv examples are all from prose; others are from vss except as specified. All from thematic stems:

bhavāhi SP 32.15 (and read so with Kashgar rec. in 274.8, cf. critical note and La Vallée Poussin, JRAS 1911.1077); dadāhi (to dadati) 88.8; read with Kashgar rec. bhanāhi 98.6 and 99.6; bhāsāhi 98.4; parāhanāhi 178.10; pradaršayāhi 178.11. — anugrhņāhi 'favor, have mercy on' (not 'receive' with Suzuki) Lank 4.15; vadāhi 27.6; 200.13. — bhanāhi LV 56.12; 114.22; 304.1 etc.; bhanāhi 233.1; dadāhi 94.18; 220.16; dadāhi 233.8; nişkramyāhī 162.10; smarāhi 172.17; 185.15; vrajāhī 233.12; vasāhi 241.18, 21, 22.-Mv (prose): nivartāhi Mv 1.270.14; ii.107.11, 16; yācāhi i.289.6; 349.1; parityajāhi 310.7; abhișiñcāhi 349.7, 10; gacchāhi 362.11 fl. (repeatedly); 365.13, etc.; bhāyāhi 363.1; upapadyāhi ii.2.12; snāyāhi (to snāyale) 78.3; āgacchāhi 82.12; 271.10; nisīdāhi 96.17; 107.3; anusmarāhi 103.4; krīdāhi, ramāhi 103.6, 14 etc.; abhiniskramāhi 141.2; cintayāhi 146.4; anubhavāhi 146.5; 147.6; anuprayacchāhi 146.11 etc., very common. — pai-yāhi Suv 60.3 (mss. corrupt); dadāhi 229.4; 249.4, 8; bhavāhi 232.2. — bhavāhi Kv 34.22 (prose? occurs in what may have been an anuşțubh pāda) and 24 (vs). - avabhāsayāhi Gv 54.12, anusāsayāhi 16, and a long series of similar forms in the sequel; janayāhi 485.23. — vidhunāhi, dadāhi Sādh 4.16 (prose); dadāhi 7.11 (prose); 231.6 (vs); trāyāhi 162.10; kāmāhi 162.11, 19; 163.3; 543.9.

30.7. ahi or (m.c.) ahi. Rare except in LV, where it is quite common. Chiefly in vss, where meter requires short a, but a few times in Mv prose; these are so marked; other examples from vss. In a few cases editors print hiseparately, taking it as the particle; this is usually impossible.

jinahi LV 12.19; sevahi 28.22; bhaṇahi 91.18; 141.3 (in the latter °hi seems metrically required); vibhajahi 165.8; sthapayahi 167.8; varşahi 171.4; deśayahi 175.2; upekşahi, kurvahi 178.6; muñcahi, rddhyahi (?) 178.10; samtārayahi 178.18; upekşahi 178.22; dadahi 179.4; bhavahi 221.22; rudahi 231.10; vahahi 237.1; rodahi 237.9 (most mss. rodihi); bhavahi 241.20; prekşahi 323.2; 328.16; pravarlayahi 414.22. — pramodyahi Mv i.275.9 (for pramodyasva, pass. to pramodayati: 'he delighted, take delight'); gacchahi 288.1 (prose; v.l. °ehi); upapadyahi ii.2.10 (prose); paiyahi 447.2 (prose; Senart en. paiyahi); iii.406.1 (prose); pişahi iii.3.17; chindahi 140.4. — olarahi Gv 301.5; in 485.22 and 487.2, vandahi, read with 2d ed. (prili) vindahi 'find joy'; cf. tuşli vindi 485.23; in 487.10 prili vandi is printed but corrected to vinda in 2d ed.

30.8. [ahe, for ahi, ahī ? In LV 210.4 (vs) chandakā capalu (Dict.) mā vilambahe is thus printed by Lefm., without v.l. on the last word. It is perhaps possible, but not very likely, that he (rather than hī) is m.c. for hi; at any rate, a monstrous quasi-middle imperative ending he (to hi = se : si, te : ti) seems hardly acceptable. Despite the position, at such a distance after the vocative chandakā, I think we should read vilamba he, the latter being the interjection: 'Chandaka, do not delay, ho there!' So Calc. prints it.]

30.9. ohi, or (m.c.) ohi; also uhi. Chiefly in 5th and 8th class presents, with strong stem instead of weak (\S 28.62); also *bhohi* to MIndic *bhoti* = *bhavati*. Only in verses in most texts; in Mv also in prose (examples are marked prose):

karohi SP 113.14 (v.l. karotha); Mv i.276.1; (the rest prose) i.312.1; ii.146.6; 174.1; 214.13; 215.3; 297.7 etc.; vyākarohī SP 15.12 (so read with all Nep. miss. and WT); Sukh 52.7; vyākarohi Mv ii.296.3 (prose); prakarohi LV 199.4; apāvrņohī SP 191.10; irņohi, see § 28.62; bhohi LV 237.9; Suv 12.4: Mv ii.236.16; iii.3.17; 5.14.

But also, instead of Skt. kuru, kuruhi Mv i.275.22; 277.13 = 280.1 = 281.20. And according to the mss. kruhi (!) Mv, but read brūhi (or brūmi); see Dict. s.v. kruhi.

30.10. Thi? Probably the form bhihi in mā bhihi 'fear not!' Mv ii.489.6 and 16 does not really belong here. (Senart has no note on it.) It can scarcely be from any present stem *bhī plus ending hi; no such present stem exists in Skt. or MIndic. Rather it must be related to Pkt. (AMg., Ratnachandra; and M. JM., Pischel 501) bihai 'fears' (acc. to Pischel from Skt. bhīs); the introduction, or reintroduction, of the aspirate bh, not recorded for this stem in Pkt., is doubtless due to analogical influence of forms of bhī. The ending is then not hi; h is part of the stem, and the ending is i (prob. aorist).

Other non-Sanskrit endings

30.11. 2 sg. ā. This occurs commonly but only in verse, and I believe only where meter requires a long final. Since any *a*, especially final, is freely treated thus, a few examples will suffice: prakāšayā SP 162.14; 192.10; pravarlayā 163.4; 174.1; 191.9; anumodayā LV 42.11; srunā (to śrunati) 94.12; jayā 94.17, 19; gacchā 114.22.

30.12. 1 pl. matha and 2 pl. tha (tha): see \$\$ 26.8-10, 12-14, 16.

30.13. 3 sg. and pl. (n)tū, only in verses m.c.: spriatū LV 235.19; etū LV 235.20; pravartayantū Suv 23.6; vivarjayantū Suv 24.3; āsādayantū Suv 41.14; so read the last three with the quotations in Šiks 216.15; 217.8; 219.3, respectively; Nobel's text prints °ntu with his mss.

30.14. 3 sg. to, in vss m.c. (§ 3.71): $bh\bar{a}salo = {}^{o}lu$, 'let him speak', LV 7.11 (vs), supported in sense by Tib. bśad du gsol, '(we) ask (him) to explain'; paripūryalo (for ${}^{o}lu = Skt. {}^{o}l\bar{a}m$) 'sya āsā (= āsā) spršalū (above) varabodhim LV 235.19, 'let his wish be fulfilled, let him reach enlightenment'.

30.15. 2 sg. su. This ending is common in Pkt.; Pischel's explanation (467; analogical to indicative si, like 3 sg. lu: li) seems disproved by the fact that Pall regularly has ssu, and only m.c. su according to Geiger 126; it must therefore be derived with Geiger and others from the Skt. middle ending sva. (Its frequency in dialects which 'make little use of the middle' is of no weight; most MIndic dialects 'make little use of the middle'. And surely Pali ssu and Pkt. su must have a common origin.)

30.16. In our language, however, the ending scarcely exists. I know of only a single clear case: Mv ii.55.21 (vs), where read with one ms. mā kānkşişu, 'don't hesitate' (proved by mā vicārayi, same meaning, in the same vs in Pali, Jāt. v.397.5). On *i* for a see § 27.10. Further, LV 322.20 (vs) reads in both edd. prekşasu nāiha, but this may be read prekşa sunālha, cf. 322.16 prekşa kānta, 323.2 prekşahi nālha. And LV 328.9 (vs) is read by Lefmann ramasu guvatibhih, but perhaps Calc. is right in dividing rama suguvatibhih.

30.17.1 pl. mahe (?). In SP 243.2 (prose) all Nep. mss. read vandamahe, which KN em. to ^omahai, as subjimpv.; this is kept in WT without note; Kashgar rec. vandema, opt. But an indicative would be quite possible in the context, even in normal Skt.; 'we (will) revere.'

30.18. 2 pl. dhva. Instead of *dhvam* the ending *dhva* occurs in RP 18.14 (vs) *yujyadhva*, obviously m.c. for *°dhvam*. But the same ending seems to occur in prose in Divy 374.12 *mārgadhveti*, apparently for *°dhva* plus *iti* (or for *°dhve*, transferred from the indicative, plus *li* for *iti*?).

30.19. 2 pl. dhve. The primary indicative ending *dhve* may be used instead of *dhvam* in the impv.: *abhiyujya-dhve* (v.l. °yatha) SP 79.11 (prose).

30.20. 3 pl. ntān. This is merely an orthographic variant for $nl\bar{a}m = \text{Skt. } nl\bar{a}m$; final m, from m, is constantly interchanged with n. Ex.: ramanlān (some mss. °nlāt, a mere graphic error, t written for n) LV 43.6 (vs, end of line).

31. Future

isya with thematic presents

31.1. The normal future in our dialect, indefinitely productive, is made by adding isya to thematic present stems (often MIndic) minus the thematic vowel. Examples are numerous in verses everywhere, and in the prose of My; in prose of other texts they occur sporadically. Our examples are from prose except those marked (vs). They will be arranged in alphabetic order of 'roots' (Sanskrit, so far as they exist there). Others will be found in the list of 1 sg. futures ending in am (a), §§ 31.31-33.

āp: paryāpunisyanti Mv iii.52.18 (passive in mg., 'will be mastered').

is: praticchisyati Mv ii.435.12.

kr (kirali): prākirişyanli (read pra°?) Mv iii,324.8; vikirişyase LV 335.7 (vs).

krs: krsisyanti Divy 212.17.

krī: kriņisyati (v.l. kriņī°) Mv. ii.242.1.

kşip: kşipişyali LV 145.14 (vs; so read with most mss. for Lefm. kşap°).

grah: grhnisyāmi Mv ii.450.15 (to grhnati).

ci: pravicinisyati SP 372.7.

cyu: cyavişyali Mv i.196.21; 197.1, 7; 357.3; 366.9; cyavisyasi id. ili.404.5.

chid: chindişyati Mv ii.145.8; iii.403.18.

ji: samnirjinisyāmi Mv ii.270.11.

jñā: jānişyati Mv 1.244.11; 11.437.1; 445.2.

dhā: · śraddadhişyanti SP 313.5 (vs); Mv iii.89.17 (vs); abhiśraddadhisyati Mv iii.76.6; °yanti ibid. 9.

dhov(ati): dhovişyati Mv iii.312.17.

dhyā: dhyāyişyase LV 334.21 (vs).

pattīyati, pratīyati (Dict.): pattīyişyanti SP 286.8; 312.9; and according to Kashgar rec. 36.7; 37.10, where ed. with Nep. prattyisyanti.

pad: āpadyişyanti Mv 1.360.5; °yatha iii.290.9; upapadyisyanti i.361.10; vati iii.42.20; vasi iii.43.1; 404.5. Forms recorded as *-padisyali* etc. are mere corruptions. E. g. Mv 1.50.15; 51.11 upapadisyali should be read (buddho loke) upadriyali 'appears' (with 1 ms. in 50.15, the rest upapaddiśyali) or upadiśyali, intending the same word (so most mss. in 51.11, one upaddisyati) with MIndic i for r = Pali upadissati (Skt. upadrs); possibly read -dd- by analogy as in aorist addriat etc. The extra syllable -pa- in most mss. 50.15 is mere dittography. Again Mv iii.52.8 and 9 print pratipadisyāmi and āpadisyāmi (v.l. āpagisyāmi); but in the repetition line 14 it is perfectly clear that the mss. intend pratipadyişyāmi (one here omits the syllable dyi altogether; Senart wrongly °padi°) and apadyisyami (so Senart with both mss.); these forms should be read in 8 and 9.

paś: paśyişyāmah Mv ii.74.9, 10; 315.5; °yatha ii.445.13; iii.288.3; °yasi ii.446.2; 448.20; °yāmi ii.479.3. pā: pibisyati Mv ii.242.11.

prech: prechişyate Suv 216.9 (vs)

buddh(ati): vibuddhişyati Mv ii.160.8; buddhişyanti ii.173.13; abhisambuddhişyasi ii.400.14; etc.

bhañj: bhamjişyāmi Mv ii.280.4.

bhuj 'enjoy': paribhumjisyāmah Mv i.325.6; °yāmi ii.65.16; iii.159.19; 180.9; °yasi ii.248.5; °yati iii.126.1; bhumjisyali ii.242.11; °yāmah i.28.11.

man: abhimanyişyali Mv ii.440.17. In Mv i.239.15; 331.13; 333.15 Senart reads manisyanti. Altho Pischel 457 and Ratnach. record AMg. and M. manami, mane (1st class), to which this may be a future, I am doubtful about it. In i.331.13 the mss. have sraddhatavyam anyisyanti, clearly intending manyisyanti; one ms. in 333.15 also reads manyisyati; in all three passages we have the same formula, and perhaps manyi° should be read in all.

mā, mi: abhinirmiņişyanti (or °niş°) Mv iii.288.2, 3, 5, 9, 12; °şyāmah iii.324.4.

mus: musisyāmah Divy 94.5; 101.9. mūş (= muş): mūşişyate Divy 281.8. mrd: mardişyāmi Mv ii.270.4. yaj: yajişyali Mv ii.98.7; °şyāmi ii.95.8, 9, 11; 98.6, 18. ram: ramişyati Mv ii.443.10. ric: riñcişyasi Bbh 396.20; vali AsP 190.1. ru: viravişyati Divy 276.2. ruh: āruhişyati Mv ii.90.12. lī: pratisamlayişyāmi Mv iii.428.15.

(vas: adhyāvasişyasi My ii.159.2; but vasişyati E. in BR and Whitney, Roots).

vā: vāyişyanti Mv ii.308.8 (vs).

viś: praviśisyati My 1.360.4; 361.8; upaviśisyati ii.439.15.

viş: parivişişyāmah (mss. °vif°) Mv ii.275.12. ordh: abhivardhişyanti Mv i.332.13; 333.4, 18.

śam: praśamişyati Mv 1.289.4 (Skt. śamişyati Gram.).

śru: śrunisyali LV 54.6 etc.; praliśrunisyāmi Mv il.223.20 (vs); śrunisyatha ili.297.11; ^oyati Samādh p. 68

line 18. sad: abhiprasīdişyati Mv iii.424.12; nişīdişyāmo iii.427.4.

sic: abhişimcişyanti Mv ii.207.1.

sthā: upasthihişyanti Mv 1:325.19, 20; vati ii.428.8; °yāmah iii.37.7.

sprś: sprśisyasi SP 294.10 (vs); sprsisyali LV 288.10 (vs), and with semi-MIndic orthography prsisyati (dental

s) 153.18 (vs); sprsisye LV 287.4 (vs); Mv ii.404.2 (vs). hā: jahişyāmi Mv iii.52.16; vasi iii.406.15.

hu: juhisyāmah Mv iii.161.11.

31.2. Present passive stems may form futures like any other thematic presents: bādhyişyate 'will be overcome' LV 366.2 (vs; with Tib. by em.); diyişyati Mv i.356.14, 16, 'will be given'; sthapişyati (v.l. sthapayişyati) Mv ii.69.14 (see § 37.6; read perhaps sthaplyisyati; certainly future to shapiyati, passive of shapeti); paripaccisyati (for °pacyi°) Mv II.311.8; bhajjişyati (for bhajyi°; but v.l. bhamji°) Mv ii.489.20 'will be broken'; viruddhişyati Mv ii.490.15, perhaps for virudhyio? in any case to passive virudhyate, 'will be at odds (with you)'; khajjişyatha (for khādyi°; v.l. khajjatha) Mv iii.290.17; khajjisyasi Mv ii.78.4. And with h for sy (cf. §§ 31.19, 20), kşiyihanti (mss. cited as kşipihanti, Senart kşipihanti) Mv ii.53.14 (vs; to ksiyale, Pali khiyali, 'will be lost'). Here probably belongs vikrāyişyali, which I would read in Mv ii.242.14 for Senart vikrayişyati, v.l. vikrāmāşyati (1); and cf. fut. vikrāsyati 242.11 (see below; v.l. vikrāmasyati); both from *vikrāyale 'is sold', a MIndic passive to vi-krī (§ 37.39).

(148)

Future sign affixed to stems in long vowels

31.3. But also the ya of the passive, after a long vowel (i, ā), is sometimes lost: dīşyati 'will be given' Mv i.349.18; 350.1; ii.467.8; 468.11; 469.13, 17; 471.9; 472.18; 474.13; 476.14 (cf. dīyişyali above); nīşyali 'will be taken' Mv ii.110.1 (for *nīyişyati, to nīyate); possibly sthapīşyati, § 31.2; prajāsyanti (v.l. °te) LV 398.6 (prose; jāsyati in Skt. Epic, Whitney, Roots) and probably jāsyante 'will be born' LV 371.22 (vs, text jñasyante, vv.ll. jayate and jāsyante, see Foucaux Notes 193, supported by Tib.; cf. AMg. present jāi for jāai = jāyate, and future -yāhisi for -jāsyase, etc., Pischel 487, 527 end). That my analysis of disyati, nisyati is correct seems proved by tarihasi 'you will be taken across' Mv iii.328.8, 9, with h for sy as in kşiyihanti § 31.2, from tariyati, MIndic passive to tarati 'takes across', with loss of syllable ya; in this case no other interpretation seems possible. Also vikrāsyali (\S 31.2) may belong to *vikrāyale, with loss of ya; but if the active was *vikrāti 'sells', this might be for vikrāsyale, which could have passive as well as middle function, and be derived from the quasi-root vi-kra. On svadhyasyati (: svādhyāyali) see §§ 28.54; 38.31.

31.4. Similarly from the MIndic present alliyali (Dict.), apparently not a passive, occurs a fut. alliyyalha Mv ii.253.5, 7 (in 7 v.l. alliyiyyalha). And so patişyanti Mmk 73.15 (prose) 'they will believe', certainly not passive, to *patiyali = patliyali.

31.5. Such futures from ya-presents, in which the omission of ya brings the future tense-sign into juxtaposition with a long vowel, belong in a single formal category with a group of futures which are mostly based on present stems in long vowels, to which sya (sya) is added. The distinction between sya (sya) and isya is, then, similar to that between s-aorists (after stems in long vowels) and is-aorists (added to thematic presents, §§ 32.14, 47).

31.6. Instead of sy (sy), we sometimes find h, usually followed by *i* (for Skt. ya) instead of a, after long vowel (cf. § 31.19; Pischel 520 calls h for s(s) 'ein Lautübergang, der von den auf lange Vocale und Diphthongen endigenden Wurzeln und Stämmen ausgegangen ist'). As to forms with single s, presupposed by this h, see §§ 31.26 ff. When this h is found after short vowels (§ 31.20) it is due to analogical extension.

31.7. Here belong the MIndic futures to roots $bh\bar{u}$ and kr, §§ 31.12-18. Also, to the Skt. and Pali present adhite 'reads, studies' we have the future adhisyati Mv il.77.14. From the weak 9th class present stem grhni-, futures are fairly common: grhnisyati Mv il.242.2; 435.2; grhnisyāmi il.450.3, and pari-g° il.126.6; 128.15; 129.17; ni-g° il.127.10; prati-g° il.304.16; grhnisyāmi, (1 sg., § 31.31) il.474.15; °syāmah il.449.17; 450.1. (Contrast grhnisyāmi, § 31.1, based on thematic (MIndic) grhnati.) Similarly krinisyati, if this v.l. on Mv il.242.1 is the true form.

31.8. To the strong reduplicated stem dadhā: abhiiraddadhāsyati Divy 7.29; °sye 8.1 (others, Chap. 43, s.v. dhā 1). With this cf. other bases in ā, § 31.3, vikrāsyati (to vikrāyate) and (pra)jāsyante.

31.9. To presents in *-eti*, similarly, may be formed futures in *-esyali*. These sometimes are Skt. inheritances (and therefore not included in my work), like *nesyali* to *neti* (Skt. *nayali*); or direct phonetic developments from Skt. futures, like *bhesyali* (§ 31.12) from *bhavisyati*. They may be based on MIndic causatives or denominatives, like *uthapesyali* Mv iii.3.16, to *sthapayali*, *sthapeli*; or on other MIndic stems in *-e* (or *-aya-*), like *uthesya* (1 sg.) LV 368.10, *uthesyam* (1 sg.; so with v.l., metr. also better, for monstrous *uthehisyam* of text) Mv iii.83.15,

to -(s)lheli; samvidheşyanli Mv i.354.12, to -dheli; -leşyate AsP 139.17, 18, °yante 209.3, probably to *-layati (*-leti; but Skt. Gr. leşyati); vaheşyam ($\frac{1}{2}$ sg.) Mv il.236.13, to BHS vaheli = vahati.

31.10. From the root grah, the regular Skt. future is grahisyati; the Epic records grhisyati which occurs also here (e.g. nigrhisyase LV 333.15; 334.1, both vss; pratigrhisyāmi LV 384.3, prose, all mss.). But grahesyāmi Mv iii.191.17, if correct (v.l. grhisyāmi; prose), would be based on the pres. *graheti, of which abundant evidence is found in other verbal categories (Chap. 43). So also udgrahesyanti in the Kashgar rec. of Vaj., Pargiter ap. Hoernle MR 186.8, 22 (here °yati); 187.9 (Müller's ed. based on Sanskritized mss. °hīş° 33.17; 34.14; 35.5).

31.11. Other forms of this type will be found under 1 sg. in am (a), § 31.31; any present in aya, or its MIndic equivalent e, is capable of forming a future stem in *-esya*. But such verbs also often show futures in *-isya*-, which (tho the *i* might perhaps be regarded as a MIndic change from e, § 3.49) I have treated as cases of omission of *-ay*-(or *-a* for *-aya*-), § 38.31.

31.12. The root bhu in Pali most commonly keeps the Skt. future as bhavissali, but in addition has both hessati and -bhossati, -hossati, hohili (Geiger 151.3, end, 154.2). In Prakrit (Pischel 521) hossai (also bhavissai) and hohii etc. are found, but apparently no form with e in the root syllable. Our language of course often has the regular bhavisyati, which in this case (in view of the Pali and Pkt. conditions) may be in large part ancient, and not due to late Sanskritization. Otherwise the only common and more or less regular form is bhesyati = Pali hessati. Much rarer, and in fact less than certain, is bhosyati =. Pali hossali, Pkt. hossai. There are also, as in both Pali and Pkt., forms in hoh-, usually followed by stem final i, which I believe represents Skt. -ya-, so that ho-h-iis derived from ho-s-ya-, with h resulting from intervocalic s after change of ya to i. There is an isolated form from a stem -bhūşya-, perhaps denominative. All of these fall into the pattern of this group, sya being preceded by a long vowel. In a few cases (Siks 45.12; 46.16; 47.9) s in bhesy- is written for s, cf. in the same vicinity bhavisyāmo 46.18; this is probably a merely orthographic matter, of no linguistic importance. All the following are from verses. I have not noted any occurrence of any

Such form in prose, even that of Mv.
31.13. bheşyati SP 68.1, 10; LV 48.13; 111.8; Mv
1.234.5, 11; 251.18; 277.14 = 280.2 = 281.21; ii.87.2;
iii.106.10, 11; 386.1; Samådh 8.24; 22.36; Suv 24.12;
44.14; 54.13; Dbh.g. 49(75).26; bheşyatl (1 m.c.) LV 54.5;
bheşyate (perhaps sometimes e m.c.) LV 54.12; 73.16;
210.5; bheşyasi LV 237.16; Mv i.243.1, and read so i.43.13
(såme vs) for unmetrical bhavişyasi of mss. (Senart hohisi, em.); ii.92.20; 143.2; bheşyase LV 195.16; 283.2; bheşyi
(1 sg. = bhavişye) SP 64.6; LV 92.21; Mv i.234.6; anu-bheşyāmi Mv ii.87.7; bheşyam (1 sg.) Mv i.233.18; bheşyanti SP 273.14; 274.5; Mv i.251.20, 21; ii.324.1; iii.278.20; Gv 214.5; Sukh 73.15; spelled bhesyanti Sikş 45.12; 47.9; in Lank 5.9 text bhavişyanti but meter seems to require bheşyanti; bheşyāmas or (m.c.) ^oma SP 147.14; 223.1; spelled bhesyanto Sikş 46.16; bheşyath RP 59.12.

31.14. bhosyanti SP 330.9 (both edd., no v.l.¹); bhosyi (1 sg. = bhavisye) SP 291.13 (no v.l. in KN; but WT bhesyi with their ms. K'). Since bhosy- is not recorded elsewhere, perhaps it should not be recognized in our dialect, despite the frequency of presents like bhoti.

31.15. pratibhūşyam (1 sg., \S 31.30) 'I will warrant' Mv ii.142.17. Cf. the aorist abhūşi(f), abhūşi? But perhaps \hat{u} is due directly to influence of the noun pratibhū 'surety, security', which was surely commoner than the verb. The

1. But K' bhesyanti, as Mrs. Ralph Ehret kindly

informs me from a photograph of that ms.

10*

form may even be classed as a kind of denominative. No corresponding future is noted in Pali or Pkt. (which also lack correspondents to the aorist *abhūsi*, very common here).

31.16. bhohi LV 196.4 is 2 sg., 'thou shalt be'; but probably historically 3 sg. = Pkt. hohi (for hohii, Pischel 521).

Similarly hohi ti (= iti) is probably to be read with some mss. (v.l. hohi ti) Mv i.44.3 (2 sg., Senart em. hohisi).

hohiti (= Pali id., Pkt. hohiti), sometimes with v.l. hohiti, occurs Mv i.208.5 = ii.12.14; ii.13.3; 40.20 (twice); hohisi, 2 sg., Mv i.337.9 (v.l. hohiti); in Mv ii.206.13 mss. hohiti or hohiti, 3 pl., Senart em. hohinti; meter requires long penult, but perhaps read hohiti (which also occurs as v.l. for hohiti).

31.17. Thematic a instead of penultimate *i* appears in such forms (cf. Ekt. hohāmi, hohāmo, but apparently not hoha[t]*i*, hohanti, Pischel 521): hohati Mv ii.225.5; 405.12 (mss., Senart wrongly em. hohisi; 3 sg. 'one shall become', not 2 sg.); hohanti Mv ii.224.21 (v.l. hohati).

31.18. Here belong also the MIndic futures from kr which correspond to the Pali and Pkt. forms recorded in Geiger 153, Pischel 533 (types kähiti and kähali, like hohiti and hohati, above). They look as if based immediately on Mindic *kāsiti (in which I consider the penultimate i to represent Skt. ya; otherwise Geiger 19.1) and kāsali (1 sg. kāsam Geiger 153), generally supposed to represent Skt. *karşyali (Pali also has kassāmi, kassam, Geiger 1. c.). In such forms *-rs- could have yielded ss, then single s with lengthened vowel. In our language all the following occur only in verses; at least I have failed to record any in prose, even of the Mv. We find 3 sg. kāhiti Mv ii.41.20; 42.1, 3; iii.289.7; Samādh p. 67 line 25; also kāhitī with ī m.c. Sikş 101.6 (wrongly em. in ed. to ⁹(i); and kāhi SP 154.8 (m.c. for Pkt. ^{*}kāhī = kāhii for kāhili, cf. Pischel 520; but WT kähili with v.l.); 2 sg. kähisi Mv ii.238.9, 12; 3 pl. kāhinti Mv i.256.6; ii.323.21; 353.21; 354.8 (= iii.278.12), 10, 18, 20; 355.5, 13, 19; 484.2; Samādh p. 53 line 27; with thematic a, 1 pl. kāhāma Mv ii.229.15.

Futures containing h in lieu of sy (sy)

31.19. As was noted above, § 31.6, this phonetic change certainly started after stems in long vowels (Skt. or Mindic). We have listed the forms of roots $bh\hat{u}$ (§§ 31.16, 17) and kr (§ 31.18); also the isolated *larihasi*, fut. to pass. *lariyali* (§ 31.3).

31.20. However, this formation spread by analogy to other stems, as also in Pali and Pkt. (Geiger 150, end; Pischel 520). We have mentioned kşiyihanli (§ 31.2). Here the following other cases may be noted; they are rare and troublesome. The clearest is Mmk 577.16 (vs) aham ca bhāşahe hy aira parşanmadhye sudāruņam. Here the meter is perfect, and it seems clear that bhasahe = bhasisye, I shall speak'. In Mv iii.355.14 (vs, meter imperiect) sā (sc. bodhisativasya mahākarunā) 'sya bhajahe citlasamtānam, if the text be accepted, seems to mean 'it will take possession of his mentality'; bhajahe would then function as 3 sg, future (but with ending of 1 sg, middle, as if = bhajisye). The only other case I know is even more doubtful: sampranadahe LV 338.19 (vs), apparently 1 sg., 'I shall cry out, shout' (represented by Tib. sun hbyin?); but the text is highly questionable, see note in Lefmann's Crit. App. The thematic vowel -a- in such forms must presumably be analogical to historic presents.

First singular forms in cham (?)

31.21. Senart Mv i note p. 403 cites a few forms in *ehap*, which he takes as 1 sg. optative, regarding h as a substitute for y (that is, *ehap* as a substitute for *eyap*), and stating that similar forms are found in Asokan dialects.

On the Asokan forms see now Hultzsch's ed., ye(ham)lxxxii, xcix; *âlabheham*, *yeham*, (pa)i(pāday)eham, *pațipālayeham* cix; *abhyumnāmayeham* cxxl. Senart l.c. lists from Mv three cases ('et quelques autres'), of which i.243.16 gaccheham must be dismissed (the passage is corrupt and violently emended by Senart; there is a v.l. gaccheha, and if gaccheham is right it can be understood as gacche 'ham). I shall first list all the cases I have found which could conceivably be concerned; general discussion will follow, but I note here that in every case futures would do as well as optatives, as far as the sense is concerned; and the same is true of all the Asokan forms.

yam nünäham kalpänäm iatasahasram lisiheham Mv i.51.7-8 (prose), 'suppose I should (shall) remain for 100,000 kalpas'. But half of the mss. read lisiheyam; -ham may be a mere text corruption. If it is correct it could be future, which (instead of the commoner optative) may be used after yan nüna, cf. Childers p. 603, 2d column, yan nünäham lassa sanlike dhammam sunissämi.

tasmim samaye aham anullarām samyaksambodhim abhisambudhyeham Mv i.61.5 (prose), text. But five of Senart's six mss., including the two best, read arhan or arham instead of (the first) aham. I should read (a)rhan(n), and understand abhisambudhye ham; the nom. sg. arhan goes easily in agreement with the subject.

samayato te pañca utpalâni dadeham (mss. daheham) Mv 1.233.7 (prose). Read probably dade (1 sg. pres. mid.) 'ham; there is otherwise no 1 sg. pronoun in the sentence. nāham tasya dhitām (mss. dhitā) dadyeham Mv il.73.16-

nāham (asya dhilām (mss. dhilā) dadyeham Mv ii.73.16-17 (prose), 'I will not give him my daughter'. This form is puzzling in other respects than the ending; I know of no optative from this verb in -ye- (blend of dadyām and dadeyam?); there is no v.l., but it is probably a corruption.

yan nūnāham ... āpadycham Mv ii.120.16 (prose). Cl. i.51.7-8 above; but here no v.l.

sa khalv aham bhikşavah sādhu ca susthu ca abhisamskārena ucchretvā pāmšukrtāni gātrāni pāninā parimārjeham Mv li.126.7-8 (prose). No v.l. here, but the same phrase (with slight variants) is repeated thrice later, and each time (127.12; 128.17; 130.2) parimārjeyam is read without v.l. (except in 130.2 one ms. °jayam). Perhaps read parimārjeyam also in 126.8.

sa khalv aham bhikşavo mudgayüşam (so v.l.) vikrtam (or pi krtam with v.l.?) bhumjeham Mv ii.131.8-9 (prose). Here, to be sure, the preceding clause contains an opt., ähareyam; but a future is equally possible.

yam nūnam asya ito ham udumbaraphalāni dadeham Mv ii.246.10–11 (prose). Cl. i.51.7–8 above; but here no v.l.

anupravrajyeham Mv iii.50.16 (prose) mss., Senart em. "orajeham, cf. ii.73.16–17 above, dadyeham. At the end of a very long sentence, at the beginning of which (in line 12) occurs the pronoun aham. The intervening distance is so great that it is very easy to suppose the author had forgotten the first aham, and duplicated it in 16; read then anupravrajye (or "je?) 'ham.

31.22. Discussion. All these forms could be read as containing the pronoun (a)ham, after a verb ending in -e (opt., aorist, or pres. mid.), if the reading is otherwise correct (note that in several cases it is very doubtful). True, in some of them this would duplicate a previous occurrence of aham (in some, however, such as the last case, at such a great distance that this unevenness is easily comprehensible). Such a duplication is otherwise perhaps not unprecedented; cf. Mv 1.51.14-15, where Senart assumes a pleonastic aham after a 1 sg. pres.: yam nünäham kalpänäm falasahasram sthätum icchämy aham. Note that this is a close parallel to, and shortly follows, i.51.7-8 above where Senart assumes tighheham as 'optative'. If aham may be duplicated in lines 14-15, surely it may be duplicated in lines 7-8 of the same page. (I am, to be sure, not certain of Senart's text in 14-15; he inserts icchalha after icchämy

aham, without ms. authority; perhaps this *aham* originally belonged to the following clause, corrupt in the mss.)

In most, perhaps in all, the Asokan forms cited above, (a)ham as a pronoun, following a verb form in -e, could easily be assumed.

If, however, *-eham* is really the ending of a verb form, here and in the Aśokan inscriptions, I doubt whether Senart and Hultzsch are right in taking it as an optative. Senart's assumption of a phonetic change of y to h seems implausible. I should prefer to regard such forms (if they are real) as futures in *-ham* (with *-am* instead of *-ami*, § 31.30), based on present stems in *-e-* instead of *-a-* (§ 38.2, 21), and with h instead of s(y), §§ 31.6, 19. Cf. Pischel 528 vallehāmi (from an *-aya* verb, 10th class), and 520 hasehāmi, from Skt. hasali; *haseham would be a precise parallel to the (supposed) forms here under consideration.²

Futures in sya (sya) after consonants, more or less un-Sanskritic

31.23. Mention should be made of some miscellaneous futures in sya (sya) after consonants which are not recorded in standard Skt. literature, tho some of them are formed regularly enough and some are authorized by grammarians (Gr.) or occur in the Skt. epic.

Root kruś (kroksyali Gr.): ākroksyanli Divy 38.11, 14 (prose); pratikroksyanli AsP 179.4 (prose).

krş (krakşya- B.S.; karkşyati Gr.): ākrkşyata ms., ākarkşyatah (3 dual) ed. em., MSV n.65.20 (prose).

nah (nalsyali Gr.): samnalsyāmi Šiks 283.4 (prose). lup (lopsyali, °te Gr.): vilopsyase LV 334.11.

stambh: avaştapsyate (nowhere recorded) Divy 574.18; 575.7.

han (hamsyali Epic, Whitney Roots): hansye LV 311.18 (vs).—On abhigamsye, 1 sg. fut. to abhi-gam, see § 31.29, end.

31.24. In LV 335.1 (vs), instead of Lefm. vibhartsyase (against all his mss.), a future is surely intended as in all parallel verbs in the passage, and the best mss. read vihatsyase, 'you shall be frustrated', as fut. of vi-han, which is very appropriate in meaning. I take this for *-hantsyase, corresponding to the Pali fut. hanchati (Geiger 153.2; also -hankhati and 1 sg. -hanhi). On the -l- cf. Whitney 207; and for -ls- instead of -ns- (through the medium of -nls?) cf. (e)tsu(h) as 3 pl. ending for (e)nsu(h), § 32.97; Bloch, Indo-Aryen 88. The ts may, of course, be a Sanskritization (back-formation) from Mindic ch.

31.25. More irregular and doubtful: elsye Mv i.257.18 (prosc), and by em. 257.11; seems to be a 3 or 1 sg. future of root i, for esyale or esye. Perhaps corrupt; see Senart's note. — drksye RP 46.10 (vs), for draksye (drš); corruption? — mrksyanli, or probably rather āmrksyanli Šiks 45.12 (vs; preceded by cilrā; preverb ā probably intended), 'will enjoy', fut. of Skt. (ā-)mri, cf. Pali āmasati. No fut. of mri recorded in literature; marksyali and mraksyali Gr. — anušāksyali Suv 81.6 (prose), Nobel's em., mss. anušāsisyale or upaiāksale. If correct, perhaps blend of -šāsisyali and -šikşali; cf. Pali anusikkhāmi, s.v. anusikkhati (b), 'perhaps taken for a fut. of anu plus root .šās', CPD.

Futures with s for sy

31.26. We have seen that the MIndic forms with hi (analogically sometimes ha) for sya (§ 31.12) imply MIndic

2. Bloch, MSL 23.115, regards the Asokan and Mv forms as 'une soudure du verbe [opt.] avec le pronom [*aham*] postposé.' Further: 'M. Helmer Smith me signale que dans les textes présentés à la première personne... le pali va jusqu'à postposer *aham* à des verbes déjà précédés du même pronom' (cf. several Mv cases, above). reduction of ya to i, leaving single s (after a long vowel). Some such forms with single s are actually recorded in Pall and Pkt. (e. g. Pali kāsam, Geiger 153). And in Ap. (Jacobi, Bhav. 41*, San. 17) futures in s, instead of ss, are common, and the s is not exclusively added to long vowels. In our language however they are rare, except for the single stem gamsa- (to gam) in Mv.

31.27. The only instance noted after a long vowel is prahāsale Ud xix.1, in the oldest ms. (rewritten in later mss.), fut. of pra-hā, = prahāsyale. On the dubious and obscure ākhyāsi of Mv lii.451.6, see § 32.123.

31.28. After the 'union-vowel' i: *sikşişenli* Gv 481.1 (vs), seemingly for *sikşişyanli* which is metrically impossible (e from ya? § 3.116); perhaps parirakşisanto LV 47.10 (vs; °şyanto would spoil the meter; barely possible might be parirakşi, ger., 'guarding', followed by santo nom. pl. of sant). Further, in a number of verses of LV and RP, forms in *-isya*- occur where the meter seems to demand a short syllable, suggesting that the true reading may be *-isa*-: *bhavisyatha* LV 231.20; *bhāvisyanti* 232.5 (*ā* in first syllable m., § 3.11); *vartisyante* 232.6; *bhavisyasi* 288.2, 6; sprsisyati 288.10; antarahāpayişyanti RP 17.15; *bhramisyanti* 17.17.

31.29. After consonants: pralilapssase (so text!) Gv 286.4 (vs), but this, tho not corrected in 2d ed., is probably a misprint for pratilapsyase (to labh); and pratinisraksati, v.l. for %syati Mvy 5232, is probably to be rejected. (See however § 2.23 for better authenticated instances of futures in -kşa- for Skt. -kşya-.) There are no other clear cases of futures in -s- after consonants except gamsali and its type, from gam, fairly common in Mv (not noted elsewhere), mostly in verses but twice in prose: gamsali (gansali) iii.85.16; 88.1, 9; gamsāmi ii.86.5, 18; 87.7; 240.15; 406.10; iii.15.19; 86.14; 168.8 (prose); 187.5; gamse, 1 sg. mid., iii.415.11 (prose, v.l. game); anugamsam, 1 sg., iii.270.10 (so Senart em., mss. anugamsemi, metrically impossible); gansi, v.l. gamsi, iii.45.10, seems to be 2 sg., m.c. for ganse, cf. -sye, -syi as 2 sg., § 31.35 (paralokam galo santo narakam gansi pārthiva, 'being gone to the other world, you shall go to hell, O prince'); gamsatha, 2 pl., iii.86.1. Once -gamsye occurs as 1 sg.: abhigamsye ii.144.3. On possibly related MIndic forms see Bloch, Indo-Aryen 88 (after H. Smith).

First singular futures in am (a)

31.30. Both Pali and Pkt. have well authenticated 1 sg. futures in *am* (apparently the secondary ending used as primary), by the side of the regular *dmi* (Geiger 150, Pischel 520). This *am* is very common in Mv, and occurs sporadically elsewhere. Instead of *am* we also find *a*, not recorded in Geiger or Pischel; but chiefly in verses where meter requires a short final; the few exceptions may possibly be corruptions. On the doubtful forms in *eham* see above, §§ 31.21, 22.

31.31. We shall first list separately examples of forms in esgam from presents in aya or its MIndic equivalent e (§ 31.11); all are from prose except those marked (vs): desesyam SP 336.12 (vs); sanjanesyam Mv i.145.12 (vs); pratipûresyam 145.18 (vs; so mss., Senart em. pra-pûr°, m.c.); kalpesyam 346.9, 16 (but 347.4 kalpayisyam); änåpesyam (Pali änäpeti, caus. of ä-ni) ii.103.5; ucchresyam 126.6; 127.10; 128.15; 130.1 (to ucchrayate 'arise'); mocesyam 183.7 (vs); täresyam 204.7 (vs); vahesyam 236.13

See further H. Smith, BSL 33.169 ff.; he would (170) plausibly read vahenam (Senart vahed aham, unmetr.) in Mv ii.194.2 = Pali Vv. 81.19 vahissam (so comm., text vāh^o), future! However, vahe (1 sg. mid.) 'ham may easily be understood in Mv.

(to pres. vaheti, for vahali, § 38.21); nesyam 248.4; māresyam 428.19.

31.32. Others (prose except those marked vs): udīrayişyam Mv 1.1.15; 2.4; pravekşyam 157.13 (vs); bhavişyam 362.17; 364.20; ii.480.7 etc.; samādāpayişyam 1.321.20; kārāpayisyam 325.17; jīvisyam 362.8; visarjayisyam 363.1; draksyam ii.41.7, 11 (vss); śrosgam 41.9 (vs); adhyeşişyam 108.5; nişkramişyam 141.4; abhişincişyam 158.3; chindişyam (pass., 'I shall be cut') 173.14; gamişyam 173.17; 460.17; anukşamāpayişyam 213.16; paricarişyam 214.11: abhisambudhisyam 265.6 (read °buddhi° or 214.11; abhisambudhisyam 265.6 (read ⁶budhi^o or ⁶budhyi^o?); ⁶budhyi^o, v.l. ⁶budhi^o 403.19; labhisyam 274.7, sajjişyam 8, karişyam, māpayişyam 13, kārayişyam 14; lapsyam 327.16 (vs); pariveśāpayişyam 435.11 (= °veș°, wait upon'); ulkhanāpayisyam 437.13; mārayisyam 459.16; pasisyam 463.2; 478.7 (in the latter mss. osisyam, Senart āsişyam); pradāsyam 486.16; upadaršayişyam iii.43.1; viharişyam 52.1; 53.3; ānāpayişyam 125.17 (cf. ānāpeşyam 11.103.5, § 31.31); pravrajişyam 218.3, 16; nişidişyam 225.11; acikşişyam 258.13 ff. (7 times); anveşyam 262.18; 265.6; prativiramişyam 268.11 ff.; pratişthāpayişyam 403.1; upasthāsyam Šiks 154.17; dhovisyam 154.19 (both prose, from Bhiksuprakirnaka, prose of which is like that of Mv).

31.33. Ending a, instead of am, mostly m.c.; cases where meter does not require short final are possibly corrupt and are specially noted : karisya LV 178.4; sthapesya 178.20; bhavisya 324.2; 326.17; in 327.5, read ramisya with most mss., text °şyi; in 327.15, read bhāvayişya with best ms., others 'sye (unmetr.); Lefm. 'syi; utthesya 368.10 (to uliheli, § 28.48); pravarlayişya 393.14; in 402.16, read vartisya with most mss.—In Mv i.143.10, read upesya (to upeli = Skt. upaili) with all mss. (Senart upelya; in parallel 200.6 read the same, Senart em. upeşyi); kşapayişya ii.91.10; abhisambuddhesya 404.19 (prose!; Senart em. syam; one ms. ⁹ya, the other ⁹yo; - *e* before sya is also strange, we expect ⁹buddhisyam, cf. under Mv ii.265.6, strange, we expect blanking (1, 1) there is (1, 2050)§ 31.32; but cf. §§ 38.2, 21. — proveksya ham Suv 62.1, larpisya ham 2 (so divide; ham = aham; meter guarantees a, rather than e or am). - vibhajisya RP 53.16. - prāpisya (for prāpayişyāmi) Dbh.g 12(348).11. — pūjayişya Sukh 8.12; muñcayişya 8.16; ārabhişya 8.17 (vs, but end of line, not m.c.; text with mss. ārabhidhya, taken by Müller p. xvi as gerund, certainly wrongly; read either ārabhişya °şyam). OT

31.34. Twice in RP occur forms in *isya* which seem clearly used as 3 sg. or pl.; both in verses, meter requiring short final: *mocisya* 15.8; *bhavisya* 29.12. Should they be regarded as historically 1 sg. forms used as 3 sg. or pl. (§ 25.27)? (Cf. the forms in *isye* used as 2 sg., § 31.35.) Or are they based on 3 person forms, originally with secondary endings (like the 1 sg. in *syam*), with final a m.c. for at, an?

Second singular futures in sye, isye (isyi)

31.35. In Mv iii.454.6 = 457.4 (vs) prapatisye seems certainly a 2 sg. future. Senart, note on 451.6, p. 527, calls it a future 'with optative ending'. More likely it is simply a 1 sg. (mid.) form used as 2 sg.; cf. gansi Mv iii.45.10 used as 2 sg. (§ 31.29) and the common use of 3 sg. (sometimes 3 pl., and more rarely 1 sg., § 25.27) forms in the place of other persons and numbers. Whatever the explanation may be, similar forms certainly occur elsewhere: sosisye (= iso°)... apdyān (ed. apdyda) LV 164.8 (vs) 'you shall destroy'; shāpayisye 178.12 (vs) and id. 16 (here Lefm. em. $thap^{\circ}$; his ms. A thāpayisyasi); dāsye 179.2 (vs); ksipram bhesye priliprāmodyalābhi 196.10 (vs) 'quickly you shall become an attainer of joy and gladness'. In Gv 58.2 (vs) spršisyi is also certainly 2 sg., 'you shall touch, attain', presumably m.c. for spršisye (the further change seems needed to make the line metrically

correct). [According to Foucaux's translation, LV 222.13 *śrunisyi* and 16 *ghrāyişyi* would also be 2 sg. fut.; but I consider them 1 sg. Foucaux translates the Tib. version of *śrunisyi* as 2 sg., but of *ghrāyişyi* as 1 sg.; the Tib. text does not support a 2 sg. in either case.]

Periphrastic future

31.36. Sen (55) says that the periphrastic future does not occur in 'Buddhistic Sanskrit proper' (but only in the Buddhacarita). This is an exaggeration. To be sure it is rare, and shows little of interest; such irregularities as are found are mostly duplicated among the 'epic and late' forms noted by Whitney 944b, which see on most of the following. A quite regular and normal form is kartāsmi ... pramoksam LV 231.12 (vs). First and second person forms without the copula, but with personal pronouns: bhavitāham (= bhavitāsmi) LV 324.6 (vs); thou wilt not arise from the seat-of-enlightenment'. In Mv ii.391.6 (vs) read with one ms. bhotā = bhavitā (v.l. bhogā; Senart em. bhoti). In RP 15.7 (vs) bhavitā is used with 3 pl. subject te, 'they will be'. In Šiks 46.12 (vs) kartāsma is m.c. for "smah, 'we shall do'.

Optative and imperative of the future

31.37. These ('of the utmost rarity' in Skt., Whitney 938; cf. Renou, Gr. scte. \$340b, c) occur very rarely here too. My 1.267.9 (prose) kārayisyet 'would have made'; future optative with the sense of a conditional, see \$31.40. Read anucintayisye for 'yeta SP 114.14 (vs). — gamisyatu Divy 220.10, future imperative.

Conditional

31.38. The conditional is rare, but not unknown as stated by Sen 59. Like other augmented forms, it may be deprived of the augment (below, LV 403.16 *desayisyam*; *pasyisya* Mv iii.285.19; *prapatisyam* 286.1; both with mss.).

31.39. A conditional in *-syam*, *-isyam* (*°yan*), historically 1 sg., is used occasionally as 3 sg. (all the following are prose): sa ced ... aśrosyad ājñāsyan (so text with all mss. except two ajñāsyam, certainly 3 sg.) lasmai cāham . dharmam desayisyam na ca mām sa vyahethayisyat LV 403.15-17, 'if he had heard ... he would have understood, and I would have preached the Law to him, and he would not have insulted me.' - adhyāvasisyan (mss.; Senart em. val), My iii.102.16; yadi na kumāro pravrajilo abhavisyam (Senart em. °sya) 107.3; adhyāvasisyam (Senart em. ya) 107.9; abhavişyam (Senart em. °yal) 107.10; paśyişya (3 sg., so mss.; Senart em. pasyitva, which is certainly wrong; probably read paśyişyam in accord with the fol-lowing) so jilam mūrchilbā prapalişyam (mss.; Senart em. °yad) vā...kālam akarisyam (mss.; Senart em. °yal) 285.19-286.2. All these forms are certainly 3 sg. conditional. In Mv iii.335.13 (prose) abhavişyat, followed by na, may be the true reading; however, one ms. reads abhavisyan, followed by a danda, and then na; the form is certainly 3 sg., and *abhavisyan* could be interpreted as equivalent to 'syam. Repeated in the following with the same v.l.

31.40. A future, with primary endings, but sometimes augmented, is occasionally used in the sense of a conditional: yadi tathāgato ... nābhavisyati Mv i.267.7 (so mss., Senart em. na bha°; cf. § 32.8), 'fi the Tathāgata had not become...' (the apodosis, line 9, has kārayisyet, fut. opt., § 31.37); unaugmented forms, Mv iii.102.14 mama putro yadi na pravrajisyati, rājā bhavisyati, 'fi my son had not left the world, he would have become king' (in line 16 medinīm adhyāvasisyan, mss., 'he would have dwelt in the carth').

32. Aorist, or Preterite (including imperfect)

32.1. Of the Prakrits treated by Pischel (452), none preserves the imperfect, except for the isolated asi, asi, and AMg. abbavi (assimilated to the aorist); and only AMg. preserves the aorist. In Pali the aorist is fully alive. but the imperfect hardly exists, at least independently: 'both (aor. and impf.) merge for the most part in the preterite called aorist' (Geiger 120). This was apparently equally true of our dialect originally. Except as partly Sanskritized, it does not permit treatment of the imperfect as separate from the aorist. The few remarks called for on the imperfect will be found in the section on the augment and in some later sections, especially that on the thematic aorist (§§ 32.110 ff.). Perhaps 'preterite' might be a better heading for this chapter than 'aorist'; but I have followed the example of Geiger's grammar of Pali, where the conditions appear to be similar. After all, nearly all the forms here included are historically derived from aorists.

Augment

32.2. Any clear notion of the original meaning of the augment was evidently lost in the Prakrit underlying BHS, as in MIndic generally. Most preterites in verses omit it, tho a considerable minority retain it. In the prose of My it is very commonly lacking. There is reason to believe that in the oldest form of the prose of our dialect it was omitted much more often than appears from most of our mss. and editions. Yet even these lack the augment in preterite forms not all too rarely. Striking is the form pralijñāsisam LV 418.15 (§ 1.46), in a piece of old canonical prose: no ms. records an augmented form here, tho in line 19 below some have pratyaj°. The interesting thing is that in the Pali form of the same passage, Vin. i.11.24, 29, the form (paccaññāsim) is augmented. Thus it appears that even Pali sometimes kept the augment better (or restored it more?) than our language.

32.3. Abundant illustrations of augmentless preterite forms, mostly historic aorists, which are MIndic in other respects will be found thruout this chapter. To be sure, augmentless forms are by no means unknown in (especially) Epic Skt., Whitney 587d; they are probably due to MIndic influence. It seems worth while here to quote a few samples of augmentless forms which are otherwise normal Skt., and specifically imperfects. The following are all from prose except those marked (vs): upasamkrāmal, or pl. °man (also dual °malām, SP 458.7; 461.9), occurs very commonly in the locution yena... tenopasamkrāmat, etc. (or the like), e.g.: SP 75.3; 100.4; 183.1; 428.12, 13; 464.8; 472.9; LV 4.15; 50.17, 20; 238.5; RP 54.2. And from the same root, prakrāmal LV 142.19; 245.15; 246.2; 405.3; prakrāman LV 264.21; avakrāmad LV 55.5.

32.4. Others (in 3 pl. forms final *n* sometimes written *m*): abhyavakirat SP 6.1; aparādhyam ili 104.5, for apā^o, 1 sg. impf.; the reading aparādhyāmīti, 1 sg. pres., adopted by WT from their ms. K' is probably only a lect. fac.; no v.l. in KN; sampādayetām 105.14; pratisthateti LV 20.12 (i.e. $\circ ta - iti$); kuruta (with 3 pl. subject, 'they made') 81.16 (vs); avatīryata 86.18 (text uncertain, cf. Weller 48, but this form scents probable); upapadyata

98.4 (so Lefm. with best ms., the others upapannâ); janayam (= °yan) 135.18 (vs); pramuñcat 219.20 (vs); upagacchan (v.l. upalashur) 265.2; lobhayam (= °yan) 326.5 (vs) 'they seduced'; wadam (= °dan) 326.18 (vs); in 351.22 Lefm. with mss. avabhāsyanlaḥ, cf. Weller 35, read probably avabhāsyanla with Calc. and also in 352.8 where 3 mss. with Calc. °nta; abhyudīkşan 416.12 (vs).-prajāyala Mv i.305.3 (vs); kārayat 11.115.9 (?preceded by prāsādā, perhaps understand akārayal in samdhi?); pramuhyata RP 9.6 (vs); pratikşipan Sikş 47.4 (vs); prāvişkārşīl Av i.10.5 (ms. °kāşīl); samavasthişata (to root sthā) Gv 403.10; anupraviśan 439.9.

32.5. The augment is sometimes put before a preverb or between two preverbs (cf. Renou, Gr. scte. 403, § 283c, note): upāsamkrāmat LV 108.4; upāsamkrāman 142.2; apresaye Mv i.128.16 (all prose).

32.6. As occasionally in Skt. (Whitney 579e), augmented forms are sometimes used with $m\bar{a}$ in prohibitions: $(m\bar{a}) \dots ak\bar{a}rsuh$ LV 89.10 (prose); and in vss, avaci Mv ii.482.8; avaca iii.17.14; anvagā 457.15 (these MIndic forms will be cited below).

32.7. There is even a case where we seem to find an unaugmented imperfect form with imperative force in a non-negative sentence, as in the Vedic injunctive. I doubt, however, any historic relation to the latter; if textually sound, it is probably a neologism, analogical to the similar use with ma in prohibitions: unnehi me paramilabhir arye, punyair acintyais ca vivardhayas ca Gv 346.15 (vs), " and increase me with unthinkable merits' (note parallel impv. unnehi). Note that aorist forms seem also to be used with injunctive force (§§ 32.119 ff.); but as they also may have 'potential' force, we have treated them as historic aorists with optative meaning.-If the text is rightly printed, another case would be Gv 488.16 (vs) utthiho kilamatho na suvrata, 'arise! (may there be) no fatigue (for you) ... '; utthihas, unaugmented imperfect to utthihati. But two lines below occurs utthi 'arise!' (§ 29.11), and we should probably read here utili ho (= aho).

32.8. The lack of association of the augment with past time is strikingly shown by the fact that present forms may be augmented. (Cf. § 31.40 and Senart Mv in. 374.) Especially common, in Mv, are such forms from compounds of bhāş 'speak', as adhyabhāşati Mv i.81.3; 83.2; ii.66.16; 68.10; 84.17; 85.3; 198.21; 199.10; 207.4; 244.18; 405.18; iii.28.12; 211.2; 215.17; 216.4, 16; 281.16; 451.13; adhyabhāşate i.122.10; 154.5; iii.457.7 (mss.); samadhyabhāşati (or better with v.l. samabhāşati) ii.144.1; pratyabhāşati ii.86.13; 87.1, 4; 207.7; 235.13; 236.14; 245.3; iii.14.16; 186.6, 9; 215.12; 216.1, 9; 282.1. Most of these are prose; a few have the v.l. °bhāşe (aorist); but they are so numerous (the list is not complete) and well attested that no doubt is possible. Mv i.6.9 ff. (prose) also has pratyahanyanti, six times, with hardly any variation in the mss., the elsewhere regular forms in prati-han° or pratyā-han° occur.

32.9. This phenomenon is clearly based on the use of equivalent augmented and augmentless preterite forms. Either adhibhāşe or adhyabhāşe may be used to mean 'he (or I) said', or indeed, the distinction between present and past being not rigid, 'he (or I) say'. So to adhibhāşate (°ti) is formed adhyabhāşate (°ti).

154

32.10. In other texts than Mv few such cases have been noted, but enough to help confirm the usage. In LV 122.11 Lefm. reads adhibhāşale sma with his best ms., but most mss. with Calc. adhyabhāşale sma, which may well be the true reading (as in Mv above). Gv 50.13 (prose) has pratyalabhanle, which is not an error or misprint for pratyā°, since the noun pratilābha follows and confirms it.

32.11. In Pali, an impv. *amāpaya* and a ger. *amāpelvā* are found, from an augmented aor. *amāpayim* etc. (caus. of root *mā*); see CPD s.v. (*amāpeli*); but the pres. indic. is not recorded with augment.

32.12. There is even an optative form with augment, *abhyaşiñcel*, repeatedly in Mmk 51.4; 58.19; 528.24 (all prose). Note also the transformation of augmented aorists into quasi-presents in *addriāti*, *addašāmi*, **§** 32.58.

Confusion of person and number

32.13. As with verb forms generally, forms that were originally 3 sg. are often used for any person and number; and the same is true to some extent of original 2 sg. and 3 pl. forms, as in AMg. (Pischel 516); see §§ 25.19-22, 28, 30-33.

Conspectus of preterites

32.14. The commonest aorist forms are descendants of the Skt. is a rist. Especially common are i and i, originally (2 and) 3 sg., from Skt. (is and) il; the distinctive 3 pl. isu is also common, imsu or insu less common. They are all formed from thematic present stems. Next in frequency are descendants of Skt. s (and sis) aorists. They are regularly affixed to bases in long vowels, especially \tilde{a} (often identical with Skt. roots in \tilde{a}) and e (often representing Skt. aya). Original optative forms in e(1), and sometimes others, are used as preterites. There is also a 3 pl. ending elsu(h), ensu(h) or the like. Descendants of the Skt. root-aorist are nearly limited to forms of the root bhū. Of thematic aorists (and occasionally imperfects), the commonest are from the root vac; also from gam, drs, and a few others. We shall close this chapter with some syntactic peculiarities: aorist forms are used with meanings which suggest the optative or the future.

Descendants of the Skt. is-aorist

32.15. From the is-aorist is derived the commonest aorist type in our dialect. It was evidently indefinitely productive from thematic presents in the underlying Pkt. The standard ending is i or (less often) i,¹ originally 2 and 3 sg. (Skt. is, it), but here used for all persons and numbers. Fairly common also are distinctively 3 pl. endings corresponding to Skt. isuh (here isu, imsu, etc.; details below). Other, more sporadic endings will be noted later. All these endings, like most other endings of the verb system, are normally added to thematic present stems, minus the thematic vowel a. This was apparently the case with the corresponding forms in Pali and AMg., tho it is not stated in such a formal generalization by Geiger (but see his 167) or Pischel (516). There are, howeveraside from regular Skt. is-aorist forms, which may be suspected of being secondary intrusions here in some sense -also some MIndic forms, or at least forms not known to standard Skt., which originate from either Skt. isaorist types or from other than thematic present bases.

1. On the incomplete evidence of SP and LV, I was inclined in JAOS 57.19 to regard i as the normal ending, and i as m.c. for it. The mass of evidence from Mv, particularly, disproves this. Both i and i are found, even in prose, but i is much commoner. This seems to be equally true of Pali and AMg.: Geiger 166, Pischel 516.

Some examples were mentioned in JAOS 57.19 and 27 (§§ 6, 21, 22); others will be cited below. 32.16. The ending i. Tho much less common than

32.16. The ending i. Tho much less common than i, this is well established, as also in Pali and AMg. (Geiger 166 ff.; Pischel 516) where it seems to be still rarer than in our dialect. It cannot, however, be assumed to be merely a metrical lengthening for i, since it occurs in Mv prose, and in metrically indifferent positions, as at the end of lines and pädas in SP.

1 sg. (vss): prayojayi 'I have employed' SP 90.5 (end of pāda); otarī Gv 256.13 and 257.5.

2 sg.: uddhari, LV 195.12, sampaśyi 20; khedayi 221.8 (all vss).

3 sg. (vss except as indicated): avabhāsayī SP 9.4, etc. (see list JAOS 57.22). — nāmayī LV 74.2; gacchī 81.11, etc. (see list, l. c.). — udirayī Mv 1.67.3; nimantrayī 113.4; acarī 155.13 (so with mss., Senart avacī: 'to him came a throng of gods with sweet speech as follows'); prarodī ii.32.11 (prose); āgamī iii.306.12, repeated in the sequel (sometimes with v.l. °mi). — alapī Lank 23.14. — kşipī Dbh.g. 14(350).10, ālapī 21 (so read with mss.; in both these meter is indifferent as to i or i). — daršayī Gv 240.24; upasaņkramī 255.20; udān' (for udānam) udānayī 489.11. — dešayī, vijñapī Samādh 8.22; bhāvayī 8.30 (in these three at end of lines); niścarī 8.25; upāgamī 8.26; pravrajī 8.28; àrāgayī 8.30. — upasaņkramī Suv 148 4; adhivāsayī 149.11; 150.6, read abhyokirī (to abhyavakirati); niryālayī 152.10.

3 pl. (vss except as indicated): vivardhayi SP 131.1, etc. (see list JAOS 57.23). — kşipi LV 92.16, etc. (see list ibidem). — kurvi (to kurvati, § 28.6) Mv i.223.19 = ii.26.12 (prose); phali ii.349.17. — (?)samakari Dbh.g. 48(74).11. Doubtful; if a verb form, must be 3 pl. aor. from sam + kr 'scatter'; but we should expect sam-akiri; no *karati or equivalent seems to be recorded anywhere from this root, unless JM. ukkariya (Jacobi, Ausgew. Erz. 17.9). Possibly a noun form? There is a v.l. sama-kari. — osari Gv 241.13 (end of line); samosari 258.20; sthapi 383.17 (end of rathodhatā pāda).

32.17. The ending i. This is regular for 2 and 3 sg. in Pall and AMg., and in our dialect also for other persons and both numbers, particularly 1 sg. and 3 pl. (Sometimes this occurs also in Pall, cf. Geiger 168 apassi 'I saw', and in AMg. at least in forms with s before the *i*, Pischel 516, e. g. $ak\bar{a}si$, $\bar{a}si$.) It cannot be regarded as mere metrical shortening for *i* (Skt. *is*, *it*) since it is extremely common in prose (as well as vss) of Mv. In other texts it has not been found in prose, but occurs in positions where meter is indifferent as to quantity of the final *i*. The following examples, then, are all from verses except for Mv; examples from that text, on the contrary, will be chosen entirely from prose, except for cases marked (vs).

1 sg.: bhani SP 56.6; (samkşobhi? 56.8^a;) defayi 57.14. — [?pranipati LV 361.19; but this is probably a present with *i* for $e, \S 26.3.$] — prakrami Mv 1.56.5; in i.246.19 (vs), read vivari mukham; prakrāmi 327.7, 13; uparundhi 11.124.10; nişīdi 131.14. — nirhari Bhad 6^a. kampayi Gv 231.15, pravartayi 18. — nadi Sukh 24.2.

2 sg.: labhí LV 166.14; jani (mã khu jani vişādu, 'do not become despondent') 91.17; lyaji 165.9 etc. (others, see JAOS 57.21). — paribhdşi Mv ii.93.1; mā pramādyi iii.124.18 'be not negligent' (to pramādyale); these all vss; on mā bhihi 'fear not' ii.489.6 and 16 (prose) cf. § 30.10.

2. Taken by Burnouf and Kern as a 1 sg. verb, but intrans., not caus.; if so, an instance of a caus. form (minus the aya) in non-caus. sense (§§ 38.23, 32). But I see no objection to interpretation as loc. sg. of a noun, = samksohe.

3 sg.: ādhārayi SP 25.10, etc. (see list JAOS 57.22). — cali LV 92.3, etc. (see list ibidem).—acari Mv i.3.9; 4.4: upāgami 4.3; prakrami 34.15; prakrāni 35.3; adhyabhāşi 35.13; 97.4; ādešayi 130.3, praviši 4; yāci 132.11; adhyāvasi 193.20; upasamkrami 255.10; nisīdi 319.1; 325.15; anlarahāyi ii.256.11; ayāci 264.7; praveši iii.162.5; samlarpagi, sampravāragi 257.10; etc., common. — vadi Gv 212.13; phari 240.25; nyāhari 488.26 (end of

line). — abhāşi Dbh.g. 14(350).11. — avakiri Sukh 24.18; gami 51.16.

3 pl.: darśayi SP 49.7, etc. (see list JAOS 57.23). – upagami LV 49.7, etc. (see list ibidem). – upāgami Mv i.11.16 (vs), lii.1.8 (vs); prakrāmi i.57.1; abligacchi 150.12 (vs); nisīdi ii.29.4; āgami 35.14; ośiri (Senart em. okiri) 349.16; etc., common. – vyullhihi (to vyullhihali = °lisfhali) šiks 344.1 fl. – upāgami Gv 34.16, cari 18; samosari 231.1; eşi (to eşali, °te, 'seeks') 236.12; upapadyi 285.25. – prāpuņi Sukh 52.16.

32.18. Special forms of the endings \bar{i} and \bar{i} . Inherited from regular Skt. *acārit* (aor. of root *car*), but with MIndic ending, is the 3 sg. *acāri* Ud xxxi.5 (vs). In Mv ii.272.5 (prose) text has *adrāksī*, but the regular Skt. *adrāksīn* (for °*ī*!) is recorded as v.l.

32.19. Perhaps in direct analogy to acārīl, acārī, our language shows an aorist avasāri, to avasarali, 'arrived at' or 'entered into'. But it seems more likely that this is based on the 'causative' -sārayali, used in the sense of the simplex; see § 38.23.

32.20. The inherited Skt. imperfects isis, dsil — originally formed in analogical imitation of *is*-aorists (for Vedic ds) — are to be considered *is*-aorists here; they often appear as dsi and dsi, for all persons and numbers. Thus dsi as 1 sg. SP 62.13; Mv i.269.11 (vs); ii.94.2 (vs), etc.; as 2 sg. SP 312.15; LV 167.21; Mv i.269.8; 276.10; ii.186.19 (all vss); as 3 sg. SP 27.6; I.V 76.1; 194.3; Mv i.48.16 (prose); 54.3 (prose); 113.5 (vs), etc.; as 1 pl. Mv iii.45.20 (vs); as 3 pl. Mv iii.1.11 etc. And dsi as 1 sg. Mv ii.133.3 (prose; v.l. dsil); as 2 sg. LV 168.15; 169.1, 9 (many mss. dsid each time); as 3 sg. Mv ii.94.2 (vs); 109.17 (prose, v.l. dsil); 342.4 (vs); as 1 pl. Mv iii.43.8 (prose, v.l. dsil); as 3 pl. Mv iii.1.11 (vs); SP 195.1 (vs).

32.21. Here may likewise be included *abravi* 3 sg. Mv ii.61.10 (vs; v.l. *abravil*), also a historic imperfect which resembles the aorists in our dialect (AMg. *abbavi*, Pali *abravi*).

32.22. Otherwise, most of the non-Skt. *i* and *ī* aorist forms are obviously based on thematic present stems. These latter are often MIndic, not Skt.; they will be found in my Chap. 43. They may even be Skt. passives, as Mv ii.328.2 (vs) sampravādyi (3 pl.) 'they were sounded' (to sampravādyale, pass. of °vādayati). Other examples are cited, with the thematic presents underlying them, in JAOS 57.20-21. It is also noted there that presents in -aya- may either keep or drop -ay- before aorist endings, as they may in other forms: deśayi SP 57.14, but snāpi (to snāpayati) LV 271.12; see § 38.32.

32.23. A few such forms need special attention. In LV 197.1 *lehi* must be read, with v.l., for *lekhi*; cf. *parilehati* Mv iii.144.15 and Pali *parilehisam* (Geiger 167.1) to Pali *lehati* (id. 130.5), doubtless a blend of Skt. *leqhi* and *lihati*. In Mv iii.299.11 (vs) occurs the interesting *khāyi* 'ate', based on a Prakritic **khāyati* = Skt. *khādati* (see Dict.); Pali has the past pple. *khāyita* (according to PTSD 'on analogy of s*âyita*, with which frequently combined', but rather with Geiger 36 showing Pkt. loss of intervocalic stop with *y-śruli*; § 2.32), but no finite verb forms. In LV 165.22 occurs *chini* 'cut', cf. Chap. 43, s.v. *chid* (2), to *achinal*, whence **chinati*. The aor. *anubuddhi* LV 299.12 belongs to the present *buddhali*, discussed § 28.19.

32.24 From the root vac occurs avacī LV 109.16; 165.14; 199.11; 304.3; Mv i.112.3 (all vss); and avaci LV

135.1; 329.7; 370.14; Mv i.113.15; 202.15; ii.482.8 (2 sg.); iii.111.5; 134.13; 269.7 etc. (all vss); according to the mss. also avacim as 3 pl. Mv i.247.4 (sec § 32.39). These forms, the like of which seem not recorded in any MIndic, are clearly adaptations of the aorist avaca etc. (found in our dialect, § 32.113, as well as in Pali) to the common aorist type in *I*, *i*. No thematic present vacali is noted in Pali, but our language has vaced, 3 sg. opt., SP 258.4, thus guaranteeing a present stem vaca-, abstracted from avaca which was assimilated to thematic imperfects. Sheth cites vaci, vaue, allegedly presents of vac, from the §adbhāşācandrikā; but šuch forms might belong to vad.

32.25. Somewhat similar are forms like (ad)drsi and the like, from root drs. Our language has the form addasa, identical with Pali addasā (§ 32.111). Historically it stems from thematic aorists of drs (Vedic type adrsat), whence BHS driali, § 28.12 and Chap. 43. (Cf. also addasetsuh etc., § 32.103, and addusasi etc., § 32.57.) Or we could interpret (a)drsi as a direct modification of adrsat, or its Mindic descendants, in imitation of the common aorists ending in i, i. Such forms are written; when the augment is present, with either single or doubled, but in verses the meter regularly proves that dd was pronounced. In Pali too we regularly find addasa etc., with dd. As has long been recognized, dd in such forms is due to contamination from forms like Pali addakkhi, stemming from Skt. adrāksīt. Examples of the relevant forms (all from vss) are: addusi SP 9.8 (1 sg.; mss. addarsi, metrically impossible); LV 194.12, 22 (most mss. both times adaśi, adrśi, unmetr.); Mv ii.37.9 (v.l. adarśi); 222.6 (emend. Senart, plausibly); adrśi (pronounced with dd where metrical evidence is conclusive) LV 194.15; 197.15, 17; drśi LV 230.4; Mv ii.299.9 (mss. drśi, Senart em. m.c.); drśi (or adr°) LV 194.13, 14; 197.5.

32.26. There are other *i*-aorists from the root drs. Isolated is Mv ii.54.3 (vs) anudrakşi. It is doubtless based on a Sktized present *drakşali = Pali dakkhali (also dakkhili), on which see § 28.41; Geiger 136.3; less likely a direct semi-Sanskritization of Pali (anu +)addakkhi (= adrākşil).

32.27. More numerous are forms showing guna in the root syllable: adarši, °ši, add°. With these may be compared Pali adassim Cariyāp. i.2.2 (Geiger 166), 1 sg., and addarsa, below § 32.111. They may be interpreted as based on the type (ad)drsa, °si, (ad)dasā etc., blended with normal-grade forms of the root containing dars-; but note also that adarsam is known to Skt. Spellings with both d and dd occur. But, tho they usually are found in verses, generally either the meter is indifferent, or it is in such bad shape that we cannot draw safe inferences from it. However, in Mv ii.13.5 meter requires adarsi, which Senart rightly reads, tho his mss. have addarsi or addasi. (Repetitions in ii.13.9, 13, 17 show -rs- always in the mss., which vary between dd and d.) And in Mv iii.245.1, where mss. have addarsi or adarsi, Senart is probably right in printing adarsi, the the line is in bad shape in other respects. The only prose occurrence noted is Mv ii.400.10, where addarśi and adarśi are both recorded. Others: darśi Mv i.303.15 (Senart adarśī m.c.); adarśi 305.18; adarśi ii.35.17, v.l. add° ; iii.12.6 = 17 mss. addarši, ${}^{\circ}si$, adarši (1 sg., Senart em. ${}^{\circ}sim$; meter inconclusive); vidarši (3 pl.) LV 49.10.

32.28. From the root $bh\bar{u}$ occurs, at least in composition with $pr\bar{a}dur$, a 3 sg. aor. ahi: $pr\bar{a}dur$ -ahi 'appeared' Mv ii.221.17; iii.216.7 (vss; so ms. both times; Senart wrongly em. "ahu). It apparently represents a MIndic $ahu (= ahh\bar{u}l)$, adapted to the regular aorist ending in *i*. Possibly $h\bar{i}$ in LV 230.9 (vs) stemito $h\bar{i}$ is likewise to be interpreted as $= ahi (=abh\bar{u}l)$ with \bar{i} m.c. (?) rather than the particle $h\bar{i}$, which does not seem very plausible in the context. The genuineness of the form ($pr\bar{a}dur$ -)ahi is guranteed by similar forms in Pali: $p\bar{a}lur$ -ahimsu 3 pl.

Jat. i.54.5 (v.l. °ahamsu); pālur-ahamsu ib. i.11.2; anv-abhī. to anubholi (CPD); ajjhabhi, adhibhamsu, to adhibhavali (ib.). The Pali forms -ahamsu, -bhamsu, are similarly altered to match the pattern (adā:) adamsu, Geiger 163 (or perhaps in the case of pātur-a° more precisely to forms of antar-dhā, see Chap. 43, s.v. bhū 7).

32.29. Distinctive 3 pl. forms. While i, i are often used with 3 pl. subject, we also often find MIndic 3 pl. endings related to Skt. isuh. (They are even used for other persons and singular number, e. g. § 32.31.) Like i, i they are normally added to thematic present bases minus -a. In AMg. this ending appears as imsu; in Pali both in this form and as isum. In the Pkt, underlying our dialect the common form of the ending was probably isu (as in Asokan), direct phonetic heir of Skt. isuh. In our texts it is usually isu. Unlike Pali, it (again like Asokan) never shows final anusvāra. When meter requires a long final syllable, the u is generally lengthened; we rarely find -uh or equivalent as in Skt. Infrequently s is written for s, but this means little: there is much confusion in the writing of sibilants. When meter requires a long penult (rarely otherwise, and only for special morphological reasons), i may be lengthened to i. But more often the alternative ending imsu is used, in one of its curiously various forms or spellings. With this, too, the final syllable may be lengthened m.c., usually to n, sometimes to the uh of Skt. or its samdhi equivalent. The sibilant in this ending is almost always written s, very rarely s, in contrast to isu for which isu is rare. Doubtless n for m is purely orthographic; at least we find the two often interchanging in other situations. Even itsu, with t for n or m, is recorded, at least in some mss.; on this, and on emsu, elsu (also ouh, etc.), see § 32.97, 98.

32.30. 3 pl. isu and congeners. These are nearly restricted to verses; even Mv, in strange contrast to its usual habits, seems not to use isu in prose, where elsu(h) seems to replace it (§ 32.103); even in verses Mv uses isu chiefly where elsu would be metrically bad. In the prose of LV 397.21 occurs antara-dhayisuh or (with the best ms. and others) °su, from the MIndic present anlaradhāyali 'disappears'. No prose case has been noted in the text of SP as printed, but cf. the Kashgar rec. v.l. at 181.9 udgrhnişu dhārayişu paryāpunişu, and Lüders (Hoernle MR 152) samprakampayisu for samprakampilah 270.12. The following are all in verses, except as stated:

32.31. işu: kurvişu, paśyişu, kridişu, krandişu (see JAOS 57.24) all from SP; himsisu, gacchisu, etc. (see loc. cit.) all from LV; avagāhişu Mv i.11.1; vikurvişu 64.3; parijāgrisu 76.4 (Dict.); camkramisu 189.8; pramuñcişu 219.8; ii.141.11; anubandhişu ii.198.11; prabhāsişu 307.1; abhigrhnişu (with v.l.) 307.7; janayişu 333.17; upapadyişu 350.7 = iii.274.9.—In Gv 34.21, read bhadra (= bhadrām) budhyişu; paśyişu 231.10; otarişu 240.18; vi-paśyişu 241.10; samdršyişu 'were seen' 267.2; upapadyişu 286.1. — preksisu Dbh.g. 27(53).6; (a)bhikirişu 30(56).1. – anlara-hāyişu Suv 158.1. — śikişiu Sukh 74.6. — kurvişu Samādh 22.4; prabudhyişu 22.5. In My i.304.20 the mss. read upapadyisu with a 3 sg. subject; it is unnecessary to emend with Senart to upapadyitha, § 25.32.

32.32. isu: paśyisu SP 194.7; snapayisu (v.l. °yimsu,

unmetrical) LV 234.4; gacchisu Šikş 347.2. 32.33. işū: parihāyişū SP 170.2; adhyeşişū 190.11, 191.3; paśyişū 323.14. — gacchişū, virocişū etc. LV (see JAOS 57.24). - niścarisū Gv 236.5,

32.34. isu: varlisu (v.l. vartlisu, unmetr.) LV 402.14. -Of different character are udgrhņīsu, āsīsu, § 32.37.

32.35. Işu, Işū (?): kārāpayişū SP 50.16; 51.1. But probably the true reading is °yimsū, which was read in 50.16 by one of Burnouf's sources according to a manuscript note left by him of which I have seen a photostat; while on 51.1 the Kashgar rec. reads "yimsu. - propalisu (v.l. first time, both mss. 2d time; Senart °lişu, interpreting as noun, see Dict. s.v. prapati) Mv ii.326.10; 333.15.

32.36. isuh, isus: the normal Skt. ending, in a form otherwise Prakritic, occurs in LV 282.16 ksipisus, where the meter requires a long final. LV 397.21 (prose) anlaradhāyişu(h) was noted in § 32.30. In Mv i.99.15 (vs) Senart reads pralisthihisur vanam for mss. pralisthisu vratam; but pratisthihisū would be an equally acceptable emendation. In AsP 17.7; 140.12, and elsewhere, in prose, we find uttrasisuh samtrasisuh, based on trasati, and probably MIndic, although Wh. Roots notes atrasit as authorized by grammarians.

32.37. -su, added to quasi 'stem'-final i. We have mentioned above (§ 32.30) the form udgrhuisu, recorded in the Kashgar rec. of SP, in prose. It is obviously based on the Skt. weak present stem grhūt-, with an aorist ending. Somewhat similar is āsīsu 'they were' Mv ii.305.18 (vs), if this is the true reading (note su, not su). There is a v.l. asitsu. It is evidently a plural to match the 3 sg. āsī (§ 32.20), or āsīt, on the model of such forms as gacchi: to $\hat{a}_s(x_s)$, or perhaps gacche(1): gacche(su (§ 32.98). Parallel to $\hat{a}_s(su)$, if this is the true reading, is (a)bravilsu, Mv i.186.11 (vs); Senart em. °insu. Cf. abravisi § 32.71.

32.38. 3 pl. imsu and congeners. These are much rarer than the preceding, but are less limited to occurrence in verses; they occur in prose in Mv as well as in that of Suv and of the Kashgar rec. of SP. They are written either imsu or insu; the difference I believe to be purely orthographic. Occasionally the writing ilsu also occurs; what this means must be considered together with the writing elsu(h) for ensu(h), §§ 32.95 ff. The final u may be lengthened m.c. The sibilant is almost always written s, very rarely s (indeed I have noted only one such case). Sometimes h is added to the ending. In the following, prose passages are so marked.

imsu: abhistavimsu SP 191.3; avacimsu 192.10 (cf. § 32.24; Kashgar rec. onsu, one Nep. ms. otsu); irāvauimsu 194.5. - nyasayimsu, Lefm., read probably nyasyimsu, LV 222.3 (ny-asyati, 'cast down'); abhāşayimsu 353.6 (text abhās°); stavayimsu 414.10. -- snapayimsu Mv i.99.11; pralisthihimsu (v.l. °suh) 203.4; in parallel ii.6.21 Senart pralisthihinsu, mss. °hinsuh, °hitsu; avacimsu i.207.15 = ii.12.5; same ii.227.15 (v.l. °suh, which seems better metrically); pralis/hihimsu i.2038 = ii.7.4 (vv.li. °hinsu, °hansuh, °hisuh); ninamdimsuh iii.267.8 (vš.l. °tsuh); avacimsu Suv 233.7 (Nobel with v.l. °suh, best ms. °su; end of line).

insu: prose, in Kashgar rec. of SP, bhāşinsuh La Vallée Poussin JRAS 1911.1074 for ed. 267.2 abhāșetām; and in Lüders, Hoernle MR 152, upasamkraminsu for SP 270.11 and nadinsu for 271.2; Thomas, Hoernle MR 134, pravartayinsu for SP 327.11, and 136 pradurbhavinsu for 329.2, prapatinsu for 329.3; prakraminsuh My i.212.5 (v.l. ^olsuh); udirinsuh (v.l. ^olsuh) iii.354.16.—Verses: daršinsu SP 55.12 (Nep. mss. ^osensu, ^osansu); in 50.3 fl. read probably karinsu with Kashgar rec., several times; Kashgar rec. also kārāpavinsu 51.1; ālikhinsu 51.6; samādapinsu (meter demands °sū) 51.8; niryādayinsus ca 191.2; snapinsu LV 93.21 and 22 (in 22 most miss. °tsu); upagaminsu (v.l. °tsu) 94.21; in Mv i.208.11, mss. karinsuh or °tsuh (meter demands 'su); in Suv 243.10 (prose), mss. abhistavinsuh or °lsuh (Nobel °msuh).

iņsū: raņiņsū LV 233.18; all mss. s, the only case I have noted; in the same line putimsū (with s).

imsū; samśrāvayimsū SP 194.2 (KN °su, unmetr.); ksipimsū, bhrāmayimsū LV 81.7; palimsū 233.18; slavimsū 233.19.

ilsu(h): several times noted as v.l. above, also Kashgar rec. parāhanilsu for SP 69.11 parāhananti (prose); and with t in all mss. My i.200.12 = ii.4.9, mss. pravisitsub, °varsetsuh (read the latter or °varsitsuh), and pramumcitsuh; pradhāvitsuh i.220.19.

Other non-Sanskritic personal endings of is-aorist type

32.39. Pali has a 1 sg. aorist in im, corresponding to 3 sg. *i* (Geiger 166). A very few such forms, in im or im, have been found in our language: darśayim (but vv.ll. °yi, °yi, darśi) SP 90.3 (vs); avasārim Mv ii.119.6 and 123.16 (prose) (cf. §32.19; v.l. °ri in 123.16). Senart assumes $\bar{a}sim$ 'I was' (to $\bar{a}si$, $\bar{a}si$ 'he was') in Mv i.229.3 (prose), but the mss. read corruptly $\bar{a}sin\bar{a}m$; we cannot be sure that the text did not have $\bar{a}sam$. In Mv i.247.4 (vs) mss. avacim, with 3 pl. subject, which is not particularly surprising. Senart em. to avaci, but meter requires a long final syllable (we must read sarvaikavācam avacim, or sarv' eka°).

32.40. The 1 pl. ending *imha* of Pali is pretty surely to be recognized in Mv iii.434.2 (vs), where Senart reads *akarimha* with one ms.; the other has *akāri*-followed by a corrupt akṣara; meter requires *akārimha*, which read. Cf. paricaremha (?), § 26.7.

32.41. A few times Mv seems to present an ending *ilha*, rarely *ilhā*, possibly *illhā* (once or twice at most), and once apparently *ila*, which may correspond to Pali *illha*. AMg. *illha* or *illhā* (Geiger 159, Pischel 517), regarded as a 3 sg. middle aorist. The readings are not always certain. The subject is usually 3 sg., less often 3 pl.; on one or two apparent cases of 2 pl. subjects see below. Nearly all are in verses. In several cases the form is augmented. Cf. the forms in *alha* cited below § 32.117.

32.42. The following cases (3 sg. except as indicated) are more or less plausible: jāyilha (3 pl.), 'are produced' Mv i.10.13, in sense of pres. indic. or opt., generalizing statement; avagāhitha 11.5 (3 pl.; mss. anugrāhitha, but in parallel phrase 11.1 avagāhisu); pratipadyitha (four mss. °ithe) 12.9; pravartitha (3 pl.) 13.16; upapadyitha 268.14; 305.2; ii.220.7; and so Senart reads i.304.20, but mss. upapadyişu (3 sg. subject); (samnipalilha, i.329.17 (prose), not aorist but 2 pl. impv., 'assemble yel,' for "talha; see § 27.10;) antarahāyilhā ii.240.17, v.l. °tha; prajāyilha iii.1.2; with augment, adhyabhāşilha ii.56.18; ajāyilhā iii.32.16 (end of vs, a cannot be m.c.); abhāşilha iii.438.11 (v.l. °atha). And once apparently ita: adhyabhāsita iii.135.17 (augmented; v.l. "silo which is clearly impossible); cf. abhūsita § 32.73. The forms in ittha, with double stop as in Pali and AMg., are not certain: anucariltha Mv iii.136. 15, so Senart, but mss. "illhe or "islah (the passage is corrupt and dubious, metrically and otherwise); (caratha bhikşavah cārikām) mā ca duve ekena agamittha iii.415.9 'and do not go two with one!', 2 pl. impv. (In that case read °ilha) or injunctive, cf. the impv. samnipatilha i.329.17, above; inss. here āgamiliha, agamirşya; repeated in line 15 with agamillha, v.l. agamisyam. If this 2 pl. injunctive form is not a corruption, it may be related to Skt. forms in -is-ia, in which case it would not concern the 3 sg. (or pl.) ending itha (or ittha).

32.43. Only in Mmk and only in vss, we find a number of 3 pl. aorist forms (note that $abh\bar{a}sire$ has the augment) in *ire*. based like *i*, *isu* etc. on thematic presents (note pranaigire, chindire). We may compare the Pali ending are (Geiger 159.11), which may underlie these forms (blended with *i*, *isu* etc. as to vowel of the penult). Cf. also the perfects in *ure* (*lasthure* etc., § 33.5). Twice the subject is singular; cf. the similar use of *isu* etc., § 25.32: pranasyire Mmk 172.13; abhāsire 231.20; chindire 366.13 (3 sg.); prakampire 512.19; 548.3 (3 sg. in the former); krośire 584.22; bhāsire 630.4.

32.44. Other MIndic forms with Skt. is-aorist endings. A few MIndic 3 pl. forms ending in *isub* were mentioned in § 32.36. It has seemed best to reserve for this place a few other forms the endings of which are regular in the Skt. *is*-aorist, but which are otherwise Mindic or at least not classical Skt. **32.45.** The ending il, instead of i or i, sometimes occurs in forms of this sort. Like the latter, it may be affixed to a thematic stem, which may itself be MIndic. So *sraddadhil* SP 113.10, to *śraddadhati*; pröbhaňjii 159.1, to BHS and Pali -bhañjati; abhyaştavil RP 5.6 (prose), to BHS slavati (unless we should read °ştävil, § 32.46). Probably class here nyaşil of LV 271.19, 'deposited, provided' (bhadrāsanam), from ni + as 'throw'; see discussion JAOS 57.27 and § 2.60.

32.46. In other cases we find forms which could be Skt. forms of regular *is*-aorist formation but which are doubtfully, if at all, attributable to normal Classical Skt. Thus RP 2.18 (prose) *abhyaşlāvīl* (cf. § 32.45); the form *aslāvil* is cited by Whitney Roots, as IS.S. To this, *stavisț(a)* LV 298.6 might be a regular Skt. 3 sg. middle, but for lack of augment and the fact that the subject is 3 pl.; meter, however, seems to demand *stāvisț'*. So *onamisța* LV 295.7 might be a 3 sg. mid. **avānamisța*, but Whitney records no *is*-aor. from root *nam.* Gv 334.25 has *akārişuh* 'they made' = *akārşuh*; again Whitney records no *is*-aor. from *k*, altho *akārişam* is found-in the Veda (VV I p. 190). (*mā*) *vilambişhāl* LV 217.11 (prose) could be a regular 2 sg. middle; Whitney cites *alambişla* from grammarians but not from literature.

Descendants of Skt. s and sis aorists, and similar forms

32.47. These aorist forms are typically used after bases in long vowels, as in Pali (Geiger 167, cf. 163, 165.2); Pischel 516 contains material which suggests that the same was true of AMg., though he does not formulate it thus. Some such forms were inherited from Skt., as *ajñāsi* from *ajñāsīl*; (*a)nesi* may perhaps be similarly a direct inheritance from *anaişil*, as *aśroşil* and *aśroşi* certainly are from *airauşil*. Most of the non-Skt. forms here recorded are, however, analogical creations.

32.48. Some of these are individual and sporadic, though they may be extremely common, like abhūși from bhū, which has no exact analogue (and incidentally is not precisely paralleled in any recorded MIndic dialect known to me). Sometimes however they fall into patterns which, in the underlying Prakrit, were evidently productive to a rather large extent, at any rate considerably beyond the range of inherited Skt. forms. One of these is the group in which s-endings are attached to a base in d, usually a Skt. root in ā. Skt. already had a fair number of such forms (ajñāsīt, ayāsīt, ahāsīt, etc.; lists at the end of Whitney's Roots); these are recorded in my lists when they show non-Skt. endings (ajñāsi etc.). But we also find quite a number which are not Skt., or at least not standard literary classical Skt. (e. g. adāsīl, aslhāsīl). For the 3 pl. in āsu(h) etc. see § 32.75.

32.49. Moreover other roots than those in original d are drawn into this pattern, as in Pall and Pkt. (Geiger 163.4, 165; Pischel 516); so from k_T we find (a)kdsi (so Pali and AMg.). Probably, as Geiger suggests, on the model of pairs of non-sigmatic and sigmatic aorists (akd: akāsi, adā : adāsi), Pall forms addasāsi to addasā or °sa. Our language has the same, as addaidsi, a(d)dršāsi, etc.; also agamāsi (= Pali id., cf. BHS and Pali agamā, °ma). And finally, since agama(t) etc. fall into the pattern of thematic imperfects, any thematic present may form an aorist in -āsi. Such forms are to be sure not common (āmantrayāsi, pravišāsi), but occur sporadically also in Pali (1 sg. pivāsim, 3 sg. viramāsi, Geiger) and AMg. (vayāsi, °si, to vayai = vadati, Pischel).

32.50. Our language, indeed, goes farther than Pali or AMg. (as far as recorded in Geiger and Pischel), and forms aorists in *asi* to thematic presents, with unlengthened thematic vowel. Tho virtually limited to Mv among our texts, these are numerous enough in prose as well as verses to be subject to no doubt. They may even be augmented: adhyabhāşasi 'he addressed', awacasi 'he said' (cf. thematic aorist awacas, § 32.113 = Pali id.; these asi forms very probably started with affixation of -si to awaca and the like, just as addasāsi etc., were based on addasā etc.). To this extent, the s-aorist endings transcend their usual bounds and are added to short a.

32.51. The principal other pattern which was indefinitely productive in the Pkt. underlying BHS was the type in which si was preceded by e. This type is also familiar in Pali and AMg. (Geiger 165.2; Pischel 516, c. g. kahesi). No doubt, as Geiger says, it started with such forms as (a)nesi = Skt. anaisit. Since this was associated with a Mindic present neti (= nayali), it became possible for any present in aya (e) to form such aorists; so we have e. g. aiesi, pranidhesi, ārocesi, āmantresi, etc. The Mindic character of these forms is shown by the fact that the s is almost always dental, very rarely s. And since our language, like Pali and Pkt., uses present stems in e in what are in Skt. normal a-stems, we also find such aorists as vadesi, bhavesi (to bhavali), upapadysei (to upapadyalc), etc.

32.52. In all these (and other) aorist formations, originally 3 sg. endings are freely used with subjects of other persons and numbers.

32.53. Type āsi (āsīt). This type started from Skt. forms like ajñāsil, ayāsil. With -si for sil, and with or without augment, they occur: $\bar{a}jn\bar{a}si$ Mv iii.48.19 (prose), repeated 49.6 with regular Skt. $\bar{a}jn\bar{a}sil$ or aj^2 . In iii.386.18 (vs; 1 sg.) mss. anyāsi, representing dialectic pronunciation; Senart ajānši, but probably \bar{a} -jāā is the root concerned (cf. Pali anādam = ajādam Sn 699); abhijādāsi Mv i.4.7, 9 (prose). — prāyāsi Mv iii.437.16 (prose; augmented); prayāsi i.255.3 (prose) etc.; niryāsi i.216.19 (prose) etc.; abhiyāsi ii.411.10 (vs).

32.54. Similar forms which have no correspondents in normal Skt. literature: ashdasi SP 86.4 (vs); Mv i.55.5 (prose, 3 pl.); 245.12 (prose); 327.9 (prose, 1 sg.); ii.137.2 (prose), etc.; ullhasi Mv ii.232.16 (vs; ent. but very probable); pralyashdasi Mv ii.55.15 (so with v.l.) = 56.10; iii.44.2 (prose, v.l. pralyushdasi); ii.431.2 (prose), etc.; pralyushdasi (1 for pralyushdasi); ii.431.2 (prose), etc.; pralyushdasi (1 for pralyushdasi); cf. iii.44.2 above) ii.444.7 (prose); adhişthdasi iii.53.15 (prose). — dkhydasi Mv i.9.6; 10.18; 14.4 (all vs), etc.; samdkhydasi iii.87.16 (vs). addasi Mv i.37.12; 54.3 (both prose), etc.; 53.2 (prose, 1 sg.; subject aham, not equal to ayam as Senart says; Buddha speaks of his previous existences in 1st person); anupraddasiiii.159.8 (prose). — agdasi 'sang' Mv i.130.6 (prose); Chap. 43, s.v. 2 gda. — prabhdasi Mv ii.345.8; 346.22 (vss) (or to pra-bhdas, ending -i?).

32.55. And the like with normal Skt. ending: adāsīt SP 250.2 (prose, no v.l.); samavāsthāsīt probably to be read with Kashgar rec. for samavātisthat (both edd.) SP 239.3 (prose).

32.56. Like Pali and AMg., our language has $(a)k\bar{a}si$ for Skt. $ak\bar{a}rsil$, from kr (§ 32.49). Instead of s the mss. often write \dot{s} ; this is ignored here. All but one of the following are in verses: $ak\bar{a}si$ Mv i.242.12; 262.8; 338.4 (by em.), 6, 8, 10 (the last is prose); iii.139.11; 400.15; 444.18; $k\bar{a}si$ i.267.16; 278.11; iii.106.18; 380.18; with 3 pl. subject i.267.21. Cf. § 32.74 for $(a)k\bar{a}rsil$, (a)karsil, etc.

32.57. Corresponding to Pali addasāsi (§ 32.49), from dri, we find addaiāsi or addriāsi (sometimes written with single d, but where meter is conclusive always pronounced with dd): addaiāsi, 3 sg., Mv i.237.7 (prose; by em.); 305.21 (vs); ii.91.19 (vs); 104.8 (vs, mss. addariāsi; meter confused but seems to favor addašāsi); 396.5 (vs); addriāsi (often written adr°), as 2 sg., LV 195.11, 15, 19 (vss); 3 sg., LV 196.12, 15, 19 (vss; here we seem to have general statements or predictions, as if for pres. or fut., cf. §§

32.119 ff.); Mv ii.198.5; 396.1 (vss, in both mss. °iāmi, but 3 sg.).

32.58. This ending may, of course, be 2 sg. historically, as well as 3 sg., and is actually so used in LV. As 2 sg. it was evidently assimilated to 2 sg. presents in si; and so by analogy were created a(d)dršati, 3 sg., LV 75.6; 194.20; 197.9; 241.10, and $addas\bar{a}mi$, 1 sg., but in the first passage definitely past in meaning, Mv i.318.1 (prose); iii.385.19 (vs). The latter occurs also in Pali as $addas\bar{a}mi$ (Geiger 162.3; here without being properly related to $addas\bar{a}si$ 165).

32.59. Like Pali, our language as we saw (§ 32.49) has agamāsi, based on agama(l) like addašāsi: Mv iii.12.2, 13 (vs); 79.17 (vs; 3 pl.; Senart wrongly em. agamāsu); and upagamāsi Mv iii.82.16 (vs); udgamāsi iii.246.8 (vs). And, by assimilation of such forms as agama(l) to thematic imperfects, we get forms in āsi to thematic presents: bhanāsi (to bhanali) Mv ii.221.7 (vs); ūhasāsi, probably to be read with Senart Mv i.221.20 = il.24.7 (vs, see Dict. s.v. ūhasali); āmantrayāsi iii.204.2 (prose, v.l. °yāmi, Senart em. °yesi); pravišāsi (v.l. °jāmi, Senart em. pradišāsi, wrongly) iii.390.10 (prose); and probably upanayāsi iii.429.14 (vs, so Senart m.c., mss. °yasi).

32.60. Another blend form seems to be udapāsi 'originated, was produced', apparently based on ud-apādi, normal Skt. 3 sg. aor. passive (also in Mv i.47.13; 48.18) with adaptation to aorists in āsi. It occurs repeatedly in mss. of Mv, sometimes written udupāsi in one or both mss. In both i.237.14 and 16 Senart udapāsi, mss. in 14 udupāsi; text has udapāsi in i.248.7 (prose); 351.14 (prose); 352.5 (prose); ii.130.18 (mss. udu^o); in ii.257.13 (prose) udapādi, but in repetition 258.11 udapāsi (no v.l.); also iii.50.19 (prose, v.l. udupāsi); iii.110.20; 194.1; 332.19; 333.1, 2; 407.16; 416.11; 417.9; 437.6; 440.19; 451.3 (in some of these one ms., in two both mss., udupāsi).

32.61. Equally anomalous, but isolated, is the form krtāsi, SP 51.6 (vs), discussed JAOS 57.31. It is confirmed by the quotation of the passage in Siks 93.8; WT's emendation krlavi, implausible anyhow, must be rejected, and even the Kashgar rec. ca älikhinsu may be only an old lectio facilior. It is certainly a past form, with 3 pl. subject, adapted to the numerous forms in āsi. I suggested I. c. that it is based on a 3 sg. middle (a)krta, which would be in MIndic (a)kata, or (a)katha (in accord with the frequent Pali forms like ajāyalha, Geiger 161). A few forms of this sort, with the Pali form of the ending, occur in our language (§ 32.117); and cf. pratyudgatāsuh or (more probably) °lānsuh, § 32.77. I now wonder whether forms like krlāsi, galā(n)suh may be based on the stem of the past passive participle. The Pkt. of the 'Kharosthi Documents' (Burrow § 105 ff.) has such preterites, but none, apparently, with s-aorist endings (rather, present endings).

32.62. Finally, there are even more numerous cases of aorists in asi (§ 32.50) from thematic presents. One of these is really based on a quasi-thematic aorist (avaca): avacasi 'said', 3 sg., Mv lii.386.11 (vs; Senart em. 'dsi, but better meter as mss.). That 3 sg. pasts, not 2 sg. presents, are intended is clear from the sense, and is confirmed by occasional use of the augment, in avacasi, and in adhyabhāsasi My i.55.6; ii.78.20 (both prose; these are 3d person forms, hence not augmented presents). Others: anusmarasi 'remembered' Mv i.4.8 (prose; parallel with preceding and following abhijñāsi); ārocayasi i.27.11-12 (prose); paryadhigacchasi 188.3 (vs); amantrayasi 'said' 330.11; iii.160.14, 19; 441.19 (here v.l. 'yesi; all prose); perhaps kārayasi ii.64.15 (prose, Senart with v.l. ^oyesi); āgacchasi, 1 sg., 'I came' ii.133.4 (prose); abhisarasi 'he went' ii.198.4 (so mss., better meter than Senart's em. ^oresi, which however is read in 14 below without v.l.); budhyasi 'he became enlightened' ii.336.15 (vs: four lines below budhyansu, 3 pl.); dharasi ii.356.9 (vs, v.l. °si; 3 pl.); 32.72]

antarahāyasi iii.316.10 (prose, 3 pl.; Senart em. ^ohāyensu). In Gv 259.23 (vs) sūtram nigarjasu narendrah, 'the king proclaimed (to me) a sūtra', a 3 sg. aorist is certainly intended; read nigarjasi?

32.63. Type esi (esi, esit). On this cf. § 32.51. The sibilant is usually written s; but note kirtesy (alitan. as separate word, = Tib. hdas pa, cf. Mvy 8304; so divide with WT) SP 63.9 and parigrahesid (?) SP 204.8; janesi LV 91.21; 92.4 (wrongly translated by Foucaux as if present with future meaning). The final i is almost always short; i m.c. in e. g. LV 91.21; 92.4 janesi; 169.3 thapesi. There are few forms which could be regarded as inheritances from Skt.; indeed hardly more than (a)nesi (cf. Skt. anaişit), in vinesi My i.297.13; 312.10; iii.235.8, etc., anesi id. iii.169.21; udanesi ii.286.1; iii.162.7. Doubtful is parigraheşid SP 204.8 (vs), for which WT read pari-grahişid with their ms. K' (except that it has corruptly "grhisid), which is confirmed by a ms. note of Burnouf, of which I have seen a photostat, quoting the reading pratigrahisi (cf. § 32.71). Pali has aggahi and aggahesi; should we read grahisi(d) and regard it as a blend of these two MIndic forms? But even if -grahesid is the true reading (and in any case praty-agrahesi is found My ii.38.2; 105.6), it is very doubtful whether it really is an inheritance from the Skt. aorist formation represented by agrahaisam of AB (Whitney, Roots). It seems more plausible to take it, and the equivalent Pali *aggahesi*, as MIndic creations based on the presents (Pali) gahāyali, gaheti, our *graheti (§ 35.19), as in the following cases.

32.64. Any verb with Skt. or MIndic present in -e(li, -le), or Skt. -aya(li), -aya(li, -le), is likely to form an aorist in *-esi*, most commonly 3 sg. but good for all persons and numbers. The majority are like the Skt. 10th class, chiefly causatives and denominatives; but other types occur, as: *asesi* (vv.ll. *asesi*, *asesi*) 'lay', so read twice in Mv il.86.8 (vs); to *seli* (Pali aor. *sesi*); *pranidhesi* (read °sI, m.c., cf. readings of mss. CM in 42.6 where mss. are corrupt) Mv i.42.6 = 53.3 = 336.19 (vs); i.49.16 (prose), etc.; *anupranidhesi* (135.11 (prose); to present (*anu*)*pranidheli*, § 28.48; *utlhesi* Mv ii.349.6 = iii.273.5 (vs), to *utheti*.

32.65. Examples based on presents of the 10th class type are: kirlesy (alitān, § 32.63; 2 sg., but clearly past, not = kirlayasi) SP 63.9 (vs); *thapesi* (v.l. *stha*°) 'thou didst set', (2 sg., i m.c.) LV 169.3 (vs). The rest all from Mv, prose except those marked (vs): utpādesi i.38.10; 44.13; 60.2; niryālesi i.49.16; 52.14; ārocesi i.257.15 (3 pl.); pidesi ii.175.13, 17 (vs; 2 sg., 'thou didst press'; = Pall pilesi, same vs, Jāt. iii.62.9); nirnāmesi i.228.12; dhāresi i.265.15 (vs); iii.175.9 (so with mss. for Senart vār°); āmantresi i.51.8; 267.10; 324.8; iii.107.2, etc.; kāresi i.304.12, 13 (vss), etc.; adhivāsesi i.329.8; āsvādesi i.339.10; jalpesi 'caused to say', to * jalpayali, ii.165.3; visrambhesi ii.242.15 (v.l. °eti); cilrikāresi (Senart em. wrongly cillī°) ii.268.11; vilokesi ii.281.3, 4 (four times, all em. to °eti by Senarti); vijrmbhesi ii.281.9, 10, 11 (5 times, all em. to "eli by Senart); udānesi ii.286.1; iii.162.7; prakampesi 'caused to tremble' ii.300.15; alināmesi ii.348.16; kalpesi iii.42.12; prasādesi 56.1; tāresi 77.13 (vs, 1 sg., 'I rescued'); tarpesi 249.4 (vs, 1 sg.); āņapesi 299.9 (vs); āchādesi 301.3; vīlināmesi 302.21; 303.2; yojāpesi 442.2; kārāpesi 442.3; parimocesi 446.2 (vs); kathesi 461.7.

32.66. But *csi* is also added to present stems in thematic *a* or at least to such as do not normally end in *-e*; this is doubtless part of the MIndic spread of *e*-stem verbs at the expense of *a*-stems (§§ 38.18 fl.). Particularly important are occasional forms in *-ayesi* from *aya*-presents, since here *e* cannot possibly stand for *aya*: Mv i.60.1 mānayesi pūjayesi apacāyesi (so probably read with v.l., Senart mānesi pūjesi apac³); i.49.10, 13 kārayesi; 247.6 saņmīlayesi (vs; 1 sg.); ii.284.7 abhinirnāmayesi (vi. ^eesi; repeats 283.14 where ^onāmaye); iii.106.2 (vs) prasādayesi;

iii.428.10 vinayesi; ii.233.9 dhārayesi, kalpayesi (but 234.10 dhāresi, kalpesi).

32.67. Others (all Mv; but stems in -e- are otherwise found from a number of them): upāgamesi (in 1.4.3 upāgami, same mg.; but cf. Chap. 43, s.v. gam 3) i.4.2; 246.14 (vs); okiresi adhyokiresi prakiresi (but pres. -kireti) 38.9 f.; upapadyesi 45.10 (3 sg.! cf. § 38.21; Senart wrongly em. ⁹yāmi); viharesi (there is also a pres. -hareti) 228.4, 6, 10; abhinirharesi 228.12; varlesi (not caus.) 268.17; kampesi (intrans.) ii.33.18 (vs); vrajesi 52.21 (vs, 2 sg.); 86.8 (vs); abhiprasaresi 104.6 (vs): abhisaresi 198.14 (vs); āharesi 233.8; 234.9; ācamesi 299.11 (vs); smaresi 300.12; 415.13 (vss); bhavesi 350.3; yācesi iii.97.15 (vs); bhumjesi, haresi 129.8; pravrajesi 172.16; okiresi (1 sg.) 248.6 (vs); saņsthihesi, v.l. ^osthahamsi (3 pl.; Senart em. ^osthihamsu; same line ii.351.20 saņsthihelsu) iii.276.1 (vs); ujjhesi (2 sg., perhaps present? v.l. in 9 ujhyasi) iii.295.7, 9.

32.68. Finally, the mss. at Mv ii.169.9 (prose) present *ārocalesi*, 3 sg.; if correct this would be a blend-form, 3 sg. pres. *ārocale* + aor. ending si; Senart em. *ārocayesi*.

32.69. Other si (sit, etc.) after long vowels. Directly inherited from Skt. asraugil are asrogil LV 143.13; 300.11-12 (both prose); Mv i.323.14; ii.271.14; 281.17 (3 pl.); iii.24.17; 101.19; 160.12 (ali prose), etc.; RP 45.20; 54.1 (both prose); asrogi Mv ii.272.3; 427.12; iii.101.15 (all prose, v.l. °şil); asrogi Mv i.319.15 (prose); pratyasrogi Mv i.317.18 (prose, v.l. °şil). There is also 3 pl. asroguh, (Skt. asrauguh) Mv i.258.21 (prose); and note sfraugu or sfraosu(?), § 32.101. 32.70. Mv i.59.14 f., has, in the mss., salkarosi guru-

32.70. My i.59.14 f., has, in the mss., salkarosi gurukarosi, 3 sg. past. If correct they are doubtless based on the present stem karo-, with the aor. ending -si. Senart em. Presi.

32.71. In Mv ii.54.4 (vs) occurs abravisi 'he said', a modification of abravil by affixation of the ending si, regular after a long vowel. The v.l. abravimi is of course a graphic corruption. Cf. abravitsu § 32.37. On the uncertain -grahisi(d), perhaps to be read in SP 204.8, see § 32.63; it would be based on the 'stem' grahi- as in the Skt. future, infinitive, etc.

infinitive, etc. 32.72. The regular signatic aorist of root bhû in our dialect is abhūsi (sometimes abhūsī, abhūsīt). As noted in JAOS 57.28 f., this is remarkable, for no such form seems to be recorded in any form of MIndic. (The Asokan 1 sg. aor. husam may possibly be called an exception, as for *abhūşam, perhaps also 3 pl. husu, for which however Girnar has ahumsu which resembles BHS abhunsuh. \$ 32.108.) Pali has ahosi (and ahesum), AMg. ahesi; these we should expect to appear as *abhosi and *abhesi (or °i, "It), which have not been found (unless the anomalous bhosişu contains a bare trace of the former, see § 32.78). This is the more surprising because we should except parallelism with the future, which actually is bhesyali (regularly) or bhosyati (rarely and doubtfully). Note that in the same verse My iii.373.22 has abhūsi, while Pali Jāt. v.149.29 has ahosi, Note also that the sibilant is this time always s. The form is doubtless to be explained by proportional analogy to abhû(l), on the model of such pairs as adā(l): adāsi. As to the vowel of the ending, it is clear that i is regular, in Mv prose (also in the Lüders fragment of SP, Hoernle MR 150 infra; no correspondent in SP ed.) and in verses everywhere. In Thomas ap. Hoernle MR 133-135, the Kashgar rec. of SP has (in prose) twice abhūsi (for which Thomas would read "sil) and four times abhusil; three of these six cases are 3 pl.; SP ed. 327.2 to 328.5 has instead abhūt or abhūvan. Mv i.128.13 (prose) also has abhūsi in most mss.; v.l. abhūși. In verses, where meter requires i, abhūşī occurs e.g. SP 383.11; Gv 254.10. Examples of abhūși in verses of SP (26.13 etc.) and LV (27.19 etc.) are listed JAOS 57.29. In Mv, including prose, they are common; a few, all from prose, are: i.3.13; 35.14; 36.1, 3,

32.73. Once, in Mv i.34.16 (prose), the mss. read abhūșila (one inferior ms. °ile). Senart emends to abhūșil, but abhūșila might be a blend of abhūși with the 3 sg. middle ending -ta, cl. § 32.42.

32.74. s-aorists from roots in r (r). Besides the form (a)kāsi (§ 32.56) from kr, assimilated to the type asi, we find various more or less MIndic forms from such roots which preserve r. Such a form as vyākārsi, 2 sg. 'you have explained', Mv iii.401.10 (vs), could be normal Skt. -akārsīs but for the Prakritized ending. We also find forms of this sort with shortened medial a. It is open to question (cf. JAOS 57.29-30) whether they are directly based on Skt. akārsīl, akārsus etc., with MIndic shortened a before consonant cluster (§ 3.34), or semi-Sanskritized forms based on MIndic (a)kāsi etc., influenced by forms in -ar- such as aorists kari, vyākari, udāharisu (e.g. LV 200.9; 393.12; 413.21) etc. The forms in question are vyākarşi Mv iii.139.10 (prose; v.l. vyākārşīt); vyākarşīt, perhaps the true reading for corrupt vyākarşilas LV 39.19 (prose); vyākarisi, with epenthetic i, doubtless intended by mss. vyākarsi Mv il. 193.12 (vs; Senart em. viyākarsi) .-3 pl. forms vyākarsu (but Burnouf's ms. vyākārsu) 'they predicted' SP 27.3 (vs); karşu LV 368.12 (vs; so some mss.; Lefm. em. karişu, but karşu is metrically possible; one long syllable replaces two shorts); harşur LV 271.16 (vs). Also the blend-form pravarsu 'they rained' Suv 8.8 (prose), from pra-vrs; instead of *prāvarsisus, is-aor. (avarsit V.B., Whitney, Roots), by adaptation to forms like karsu, harsu(r). There is finally a reading vyākarsa, Mv iii.386.6 (vs), extremely doubtful both as regards the ms. tradition and as to construction; it seems most likely meant for a 2 sg. aorist; read vyākarsi?

32.75. 3 pl. āsu(h), ānsu(h), amsu(h), atsu etc. We find from roots in a various forms descended from the Skt. 3 pl. s-aorist āsus (as in (a)hāsus from hā), often with intrusive nasal (written n or m), or also t (§§ 32.97, 98), before the sibilant, after the analogy of the endings imsu, insu, ensu, etsu etc. This recalls Pali forms like adamsu, from da, classed as s-aorists, Geiger 163 (actually, no doubt, analogical to add, on the model of is aorists, 3 sg. i or i: 3 pl. insu). The mss. vary seemingly at random between asu, ansu, amsu, aisu, and the same with final uh. The following are in prose except those marked vs: Wy 1.56.12, mss. asthānsu(h) or (a)sthāsu; so also iii.242.8 (vs); asthāsu 1.238.3, v.l. °ānsuh; asthānsu, v.l. asthālsuh 238.4; asthansuh, v.l. asthatsuh ii.39.2 (vs); 137.5; iii.283.15; asthātsuh, v.l. asthāsi (Senart em. sthitāsuhl) il.287.5; asthātsuh, v.l. asthāsuh (Senart em. °ānsuh) ili.64.16; 323.6; asthānsu, v.l. °suh 91.15; asthāsuh (Senart em. asthansuh) 179.5; asthamsu, v.l. asthasu 242.8 (vs); asthansu, v.l. asihāsu 244.10 (vs); asihānsuh, v.l. asihāsuh 315.10.— nirvāsu (mss., Senart nirvāmsu) Mv 1.197.6 (vs), repeated 357.8 as nirvāmsu (no v.l.); nirvānsuh, v.l. nirvāmsuh iii.161.14.

32.76. Probably on the basis of the same analogies (3 sg. i: 3 pl. insu, insu, or e: ensu, ensu, elsu etc.), a few similar 3 pl. forms are found in ansu(h), amsu, alsu etc., corresponding to 3 sg. forms in a(l), historically imperfects or thematic aorists (or forms based upon these). So lalalsuh or lalamsu (mss.) Mv ii.162.5; kurvansu (v.l. ^opetsu) Mv iii.294.11 (= akurvan); avacamsu Mv ii.233.17 (to avaca etc., § 32.113); budhyansu (v.l. ^osu) 'they became enlight-ened' Mv ii.336.19 (vs; cf. budhyasi 3 sg. four lines above); pravādayatsu Suv 8.8 (prose; the only v.l. is the obviously secondary pravādayāmāsuh; Nobel em. prāvādayimsu); pratijāgramsu (v.l. ^ojāgrsuh) Mv i.259.3 (prose), from pra-tijāgraid, see Dict.; and from the same verb a strange form

pratijāgrhansuh Mv i.231.11 (prose), in which the h is wholly anomalous; Senart suggests analogical influence of prati-grah (which does not fit here in meaning), 'probably due to copyists'.

32.77. Apparently a blend with the ending $\bar{a}suh$ or $\bar{a}nsuh$, 3 pl. aor., added to a form in ta, is pratyudgatāsuh (Senart with 1 inferior ms.), or better pratyudgatāsuh (5 mss.) Mv i.35.5 (prose), 'they went to meet'. Senart i. note 400 takes it as a periphrastic perfect = ${}^{o}gata \bar{a}suh$. But the form was apparently $-\bar{a}nsuh$; and further note kr(tāsi, § 32.61, which means 'made', not 'was made'. If kr(tāsi is based on (a)kr(ta, -gata(n)suh could also be based on a 3 sg. middle (a)gata, which actually exists in the Veda (Whitney, Roots), provided with the 3 pl. ending ānsuh. See, however, § 32.61, end. Senart reads in Mv i.222.12 (vs) udgatāsuh, which would be a form of the same sort; but it is uncertain. The mss. read udgatesuh; in the parallel ii.24.20 they read udgatesuh.

32.78. Miscellaneous s-aorists. In Sukh 7.13 (vs) bhosisu is probably corrupt; at best it might be a 3 pl. aorist of bhū based on *(a)bhosi (not recorded! cf. § 32.72) = Pali ahosi, with addition of the 3 pl. -su.

32.79. In SP 468.7 (prose), a late passage, *akşaipsil* is read (from *kşip*), a form cited by Hindu grammarians but not known in literature. Only three Nep. mss. have the word, and only one in this form (the others *akşe*°, *akşī*°), while the Kashgar rec. has *kşipalah*, which is obscure to me.

32.80. LV 136.6 (vs) reads pravekşi, from viś, which has no s-aor. in Skt.; if it existed it would be *avaikşil. LV 379.18; 380.2 have prākşur, to sprs. LV 364.11 (vs) dikşişu is not a form of diś; see Dict. s.v. udikşati.

32.81. Very anomalous is lapsi SP 190.2, to labh. Whitney, Roots, mentions alabdha, alapsata B.; lapsiya S.; lapsi might pass for a form of this type, provided with the normal ending i. Has there been contamination with the desiderative lipsate, °li?

32.82. From the same root occurs abhilabdha Mv ii.61.19 (vs; so with v.l., Senart abhilabdhe with one ms., unmetrical); alabdha (classed as 3 sg. mid. s-aor.) is found in older Skt. (B., Roots) and authorized by Panini; Pali also has aladdha (and alattha). Like the preceding would be avidha Mv i.131.15 (prose), if Senart's em. is right; mss. avidhvam; a 3 sg. from vyadh seems required.
32.83. AsP 112.17 (prose) has prananksil, from naf

32.83. AsP 112.17 (prose) has pranankşil, from nas 'perish', which has no s-aorist in Skt. (Read $t\bar{a}m$ for the following $tv\bar{a}m$.)

32.84. Several times in Mmk occurs a 2 pl. injunctive (augmentless aorist), bhais!(h)alha or bhais!ala, the equivalent of Skt. bhais!a; always with $m\bar{a}$: 'fear not!' It seems to be an extension of bhais!a by duplication of the ending la, or tha (which in our language as in Pali occurs commonly for Skt. 2 pl. ta). Most of the passages are prose, but in a verse Mmk 548.14 we find mā bhais!halha, meter confirming the form. The others are mā bhais!alu (I read surely °ta or °tha, probably the former) mārsā mā bhais!atha 4.14-15; mā bhais!atha mārsāh mā bhais!atha 182.23; mā bhais!ata bhais!ata 663.10.

Optative forms used as aorists

32.85. As was recognized JAOS 57.32-33, and much earlier by Senart Mv i note p. 401, our language uses original optative forms, or forms identical with optatives, as aorists, or at least in the meaning of past indicatives. Especially common are forms in e (or not infrequently el), and 3 pl. elsu(h), ensu(h), based on thematic presents, which in the Mv must be recognized as normal aorist forms. The form in e, and other apparent optative forms, in this sense also occur in AMg., as recognized by Pischel

466. Pischel noted, l.c. note 2, that prabruyal was used in the same way in the Hastyåyurveda. S. M. Katre (JAOS 57.316 f. and NIA 1.536) and I myself (Bull. Deccan Coll. Research Inst. 5.6; ed. of Mbh II, note in Addenda on 2.67.5d and 15d) have since observed ival (and ivam) as narrative pasts in the Skt. epic. There seems no doubt of the usage, tho it may be guessed that the (rare) occurrence of such forms in Epic and late Skt. may be due to Mindic influence. In any case many forms in e(t) in our language cannot possibly be interpreted except as pasts; they are even augmented sometimes: adhy-abhäse is very common in My. As usual they are not limited to 3 sg. subjects but may be used with any person and number. The distinctive 3 pl. forms written by Senart with ending ensu(h) are almost always written etsu(h) in the mss.; cf. My i, p. xvi. They are doubtless analogical to the form in e, like i: insu (imsu, itsu), § 32.98. Other optative forms occur as aorists more sporadically, but often enough to furnish important confirmation of this interpretation of the e-form.

32.86. Another consideration must, however, be mentioned in connexion with the preterite forms in -e(t), -etsu(h) and the like. Since in our language, as in MIndic generally, eti presents may be substituted for ati presents, it is not impossible that these aorists are in part due to this (§ 38.19). Since gacchayati or gaccheti may replace gacchali, there is no theoretic objection to regarding (a)gacche(t) as a preterite to this, as if for (a)gacchayat. (Cf. the aorists in esi from presents in ali, §§ 32.64 ff.) This can hardly be the whole story, since other optatives are used as preterites. Further, at least in the Mv, aorists in e(t), etsu(h) etc. are very much commoner than presents or other forms in eli substituted for ali; they are not likely all to be based on such eli presents.

32.87. The ending e(t) as preterite is rare except in the Mv, tho very common there. Elsewhere have been noted only sprée ca bodhim 'and he attained enlightenment' SP 190.7 (vs); possibly vyuttisthet SP 182.13 (no v.l. in KN); but WT read vyuttisthet SP 182.13 (no v.l. in KN); but WT read vyuttisthet SP 182.13 (no v.l. in KN); but WT read vyuttisthet SP 182.13 (no v.l. in KN); but WT read vyuttisthet SP 182.13 (no v.l. in KN); but WT read vyuttisthet SP 182.13 (no v.l. in KN); but WT read vyuttisthet SP 182.13 (no v.l. in KN); but WT read vyuttisthet SI 182.13 (no v.l. in KN); but WT read vyuttisthet SI 182.13 (no v.l. in KN); sut WT read vyuttisthet SI 182.13 (no v.l. in KN); sut WT read vyuttisthet SI 182.13 (no be difficult, however); snāpayet Siks 155.3 (quoted from Bhiksuprakirņaka, a work composed in language closely resembling the Mv even in prose); (1 sg.) yadā bhave (taken by ed. as yadābhave, i. e. abhave) sunetrah RP 23.16 'when I was S.'; dešayet Gv 146.3, 10, 12 etc., in straight narrative prose.

32.88. In Mv there are hundreds of cases. I shall cite a few examples chiefly from prose, taking care to illustrate augmented forms, forms in final *l* or *d*, and forms used as other than 3 sg. Forms from verses are marked vs.

32.89. Augmented forms: adhy-abhāşe i.78.2; 82.9; 108.6; 174.3, and often; apresaye (! note misplaced augment) 128.16; abhāşe 267.12 (so read with one ms.); abhyanande ii.139.22 (3 pl.); 397.6; iii.337.8 (3 pl.); alabhe ii.166.4 (vs); iii.11.3 (vs, 2 sg., read alabhe with mss.); agrahe ii.181.20 (vs); pralyabhāşe 235.8; 236.1; 237.1; abhyabhāşe 404.22 (also abhibhāşe iii.379.13); akare iii.45.18 (vs); adhyagame 191.4 (vs; in Pali ajhagamā Jāt. iii.450.22, same vs); adhyabhāşaye 327.5; aghātaye 360.17 (vs).

32.90. Forms ending in t or d: abhipidayet i.68.15 (vs); avalisihet 220.9 (vs) = avalisihe ii.22.10; samanusmaret i.228.16; alyudgacchet (Senart em. abhy?) i.229.16 = 333.7 = alyudgame 240.5; parinirväyet 267.18 (vs, v.l. 'ye); anusmaret 269.9 (vs, 1 sg.); 269.12 (vs, 2 sg.); niskramet 303.11 (vs); praviset 307.16; presayet 329.12; prajňapayet, abhināmayet, samtarpayet, sampravārayet 325.8-11; samudāgacchet 339.7; vinivarlayed ii.61.21 (vs); kārāpayet 116.1; 117.8, 10; dhārāpayet 117.5; pājayet 119.3; sthāpayet 119.4; ulpadyet 194.8 (vs); vipravāsayet iii.44.22 (vs); prajňāpayed 53.16 (1 sg.; v.l. apra⁻); prechet 242.6 (vs). 32.91. 1 sg. (besides some above): abhinigrhne abhinipiąe ii.124.2; uparundhe (mss. °ruddhe) 125.1; ähare 125.10; parivarjaye (v.l. °varje) 131.7; upapadye 133.4; prapale iii.12.6, 17 (vss); pratigrhņe 54.13; pibe 174.12 (vs); prastave 248.4 (vs); okire 248.8 ff. (vss); pratimāne 248.12 (vs; to pratimānayati); abhistave 248.17 (vs); 400.18; prastare 249.2 (vs); āchādaye 249.6 (vs); care 249.7 (vs); preche (v.l. aprecham) 369.23 (vs); jahe 431.18 (vs).

32.92. 2 sg. (besides some above): vinodaye ii.186.20 (vs); tarje iii.32.5 (vs); prasādaye 106.8; pravartaye (mss. varttaye) 381.12 (vs); jahe 431.15 (vs); kīrtaye 441.7 (vs); prakīrtaye 441.12 (vs).

32.93. 3 pl. (besides some above): paripreche i.66.6 (vs); pratigrhne 220.13 = ii.22.14 (vs); samnipate i.303.19 (vs); dhārage ii.21.6 (vs; so with mss., subject devasamphā, object bodhisattvam understood); bhāşe 300.11 (vs); prāpune 302.21 (vs); alamkare 303.8 (vs); name 303.9 (vs); abhistave 309.3 (vs); 346.18 (vs); vāye 352.14 = iii.276.16 (vs); avašire, ošire, or ^esire (so with mss.) ii.343.19; iii.273.16; pravarşe ii.344.5; upāgame, lobhaye 425.6; samprakire iii.94.20 fl. (vss); samsare 249.12 (subject bodhisattvā, with mss.); abhitande 335.9; prakrame 339.6.

32.94. 3 sg. (besides the above; a few of many cases): ulpādye i.3.4 'was produced', to ulpādyale, pass. of caus. (no 'confusion with udapādi' as suggested by Senart); parikarmaye (so with mss.) 3.12-13; nişīde 4.6; samanubudnye 4.10; kampe prakampe 40.7; alyudgame (mss., Senart em. abhyu°) 240.5; vijahe 329.6; antarahāye 'disappeared' 339.18, etc.; udānaye 351.13; iii.325.5; pralibhāye ii.122.11; 123.8; prasīde tişihe sampraskande 157.13; samdhihe (so with mss., to *samdhihali, § 28.44) 221.20 (vs; Senart em.); samslhihe (to *sthihate) 234.5; iii.65.6; 92.10, etc.; ühase 'laughed' (see Dict.) ii.268.13 fl.; anuprāpuņe iii.67.2, 4; prāduşkare 139.1; praņude 286.7 (vs, = Pali SN i.127.19 panudī); kalhe (to kalhayati, = next) 408.15; kalhaye 413.2.

32.95. 3 pl. forms in eşu, emsu, ensu, etsu(h) etc. Corresponding to the above forms in e, AMg. has 3 pl. forms in esu, emsu (Pischel 516 karesu Sūyag. 790, for which Pischel would read karimsu; Ratnachandra I p. xxvi karemsu beside karimsu). In our language eşu has been noted only once, samjalpeşu Mv i.41.8 (prose), in a formulaic passage; repeated parallels read samjānetsuh (mss.) Mv i.230.3; 240.13; iii.334.11; 341.15; samjānanti SP 163.12, etc. More often emsu or ensu occurs (below); while the mss. of Mv (rarely of other texts) very commonly present etsu(h), which Senart (see his Mv i p. xvi) invariably emends to ensu(h). For the same form in optative formation see § 29.18.

32.96. Fuller information leads me to modify the view of these forms expressed in JAOS 57.26, as follows. The rather rare endings esu and emsu(h) seem clearly analogical to e, on the pattern of 3 sg. i to 3 pl. isu and imsu, which are standard. As to elsu(h), I cannot believe that the hundreds of occurrences in Mv are all manuscript corruptions, as Senart assumes. Why would copyists introduce secondarily such a monstrous-seeming form, in such a regular and constant way? We must apparently admit that the author of Mv actually used the form; and this is confirmed by a few occurrences of elsu(h), and also of ilsu(h) and alsu(h), in mss. of other texts (§§ 32.38, 76).

32.97. The explanation may be like that of the fut. vihalsyase (§ 31.24): ensu(h) yielding *entsu(h), then etsuh; on the denasalization cf. Bloch, Indo-Aryen 88, and my § 2.67.

32.98. Alternatively, these aor. (opt.) forms (the not *vihatsyase*) could be explained analogically. We saw that the Mv uses not only the MIndic e but et, the regular Skt. 3 sg. opt. form, as a past. In the semi-Sanskritzed language which the northern Buddhists worked out for official use,

such forms evidently were well established. These forms in *et*, then, both in aorist and in optative function, might have been pluralized by addition of the 3 pl. *su*, on the pattern of *isu* to *i*. Similarly the mss. sometimes show *alsu* (\S 32.76; to thematic pasts in 3 sg. *al*) and *ilsu* (analogical to *i* for *it*; \S 32.38; rare, and perhaps created after the establishment of *elsu* to *el*). Cf. *abravisu* and perhaps *astlsu*, \S 32.37; *abhütsub* etc. \S 32.108.

32.99. There is even one case where the mss. of Mv (i.306.15, $\bar{a}ry\bar{a}vs$) show elsuh added bodily to the thematic preterite avacad. The mss. here read *le bhikşu avacatelsuh* (v.l. ^osu), which is metrically correct and which I would not enend. (Senart em. avacensuh, which compels him to adopt the further em. *bhikşu*, to correct the meter.)

32.100. The mss. vary apparently at random between final -u and -uh; often different mss. differ on this point in the same passage, in prose and verse alike. When in verses the meter requires a long final syllable, presumably uh or its samdhi equivalent should always be read; when the ultima should be short, we should read final u. In prose both are common, at least in the mss.

32.101. emsu(h) and ensu(h) are to be regarded as merely orthographic alternatives; the latter is regularly adopted by Senart instead of elsu(h). Outside of Mv we have noted LV 387.1 (vs) sādhemsu, which to be sure is Lefmann's em. but seems sound; SP 52.12 (vs) śruņensu, no v.l. in KN, but WT śruņimsu with their ms. K'. Note that Sikş 94.12 cites the line from SP, reading śruậsul (with v.l. śruotsul).

32.102. In the mss. of Mv, as distinguished from Senart's text, the form with n instead of l is rare. Both Senart's mss. apparently read parioārensu(h) ii.225.10 (vs); but such cases are extremely scarce. Somewhat more often, but still relatively rarely, one ms. has n for l, as in upagamensuh i.224.6 (vs); visodhensuh ii.199.3 (vs); vilayensu 282.15; vādyensuh 303.17; addaiensuh 344.19 (vs); prādurbhavensu iii.80.15 (prose, Senart em. $^{\circ}v$); akarensu iii.96.17 (vs); anubandhensu 96.22; 97.1 (vss); samnipatensuh 113.12 (prose); karensu(h) 365.13 (vs); gacchensu 373.21 (vs); prādurbhavensuh 430.16 (prose). Or, one ms. has vadansuh, the other vadelsu i.307.6 (prose); and so mss. sampravādyansu, $^{\circ}yetsu$ iii.96.13, 15 (vss).

32.103. The ending etsu(h) is rare outside of Mv; Suv 151.11 (vs) has in most mss. samupådagelsuh (Nobel °yitsuh, 'metrisch für samutpä°'). We shall present a few out of the very many Mv cases, all chosen from prose. They are, of course, normally based on thematic presents; sometimes from aya presents, with or without preservation of the syllable -ay- (but there is certainly, in most cases, no correlation between this element and the e of the penult in these forms). My i.37.1, 3; 39.4 manyetsuh; 40.11, 16 udirayelsuh, anuśrāvayelsuh (repeated passim, with slight vv.ll., sometimes °lsu for °lsuh, e. g. 229.12; 239.20; 336.13); 42.1-3 prapatetsuh (thrice); 52.3 tisthetsuh (v.l. °atsuh); 57.10 ff. vyākaretsuh; 80.1 (twice) vivartetsu, v.l. °suh; 194.17 kridetsu rametsu paricāretsu (cf. mss. iii.227.4); 231.2 antarhāyetsu, or antarahāyetsuh; 259.12 pratyudgacchetsu or °tsuh; 328.5, 6 adrsetsu, and addasetsu (in same sentence! v.l. "tsuh for both; cf. § 32.25); 337.2 sometimes v.l. °Isuh; 344.4 ff. prajňāyelsu (several times, sometimes v.l. °Isuh; passive); 339.15, 16; 340.8, 9; 341.2 (twice) āhārelsu or āhar°, or °lsuh; 342.11-13 samrañjelsu(h), dāşayelsuh, paiyelsu(h), ksipelsu(h) (thrice); ii.133.12 abhyanandelsuh (augmented); 161.1 gāyelsu; 215.10; iii.312.13 akarelsuh; ii.312.7 fl. addaselsuh, daselsuh (repeatedly, cl. adrietsu above); iii.69.16 addaietsuh; 409.9 paricumbetsuh rodetsuh krandetsuh iocetsuh paridevetsuh.

32.104. In a few rare cases these forms seem to be based on other than thematic presents or thematic pasts (like *adrietsu*, *addaietsu*(*h*) above). Thus Mv iii.305.6 (vs) *adetsuh* 'they gave', to *deti* 'gives'; and perhaps Mv iii.85.5

(vs) samākhyelsu 'they said', to sumākhyāli, *°khyali (analogous to -khyāmi? but v.l. samākhyāsu may be the true reading, cl. § 32.75).

32.105. Miscellaneous optative forms used as aorists, or at any rate with the meaning of past indicatives. Once augmented.

1 sg. eyam: udvīkseyam Mv ii.194.6 (vs).

3 sg. eya: prādurbhaveya 'appeared' Mv i.340.12 (prose); ādiyeya Mv i.346.11 (prose; to ādiyali, Dict.); āhaneya Mv ii.412.19 (vs); kampeya Mv ii.413.1 (vs, meter obscure; Senart em. kampe).

3 sg. iya (?), apparently equivalent to the form ia which is authorized by the grammarians in the meaning of a past in AMg. (acchia, etc.). Pischel 466 treats these forms as historically opt., quoting Lassen who described them as precatives. I do not see how they can be directly derived from any Skt. opt. or precative, unless from such 9th class pres. optatives as grhnlydl, where the *i* is part of the present stem. They seem to occur twice in Mv i.304.9 (vs) upasthiya (v.l. °yanı), 'waited upon', to upasthā (based on the pres. sthāil? § 28.51); and prechiya iii.85.2 (vs) so tām (= tān) prechīya medhāvī, 'he, the wise, asked them;' this is certainly based on the thematic present prechati. Doubtful.

1 pl. ema: āharema Mv i.344.15 (prose, v.l. āhareyāmā); tisthema Mv i.345.5, 14 (prose).

3 pl. eyuh: upasamkrameyuh Mv i.36.8 (prose); and repeatedly in Mmk, c. g. 130.19, 26 bhaveyuh, 23 āgaccheyur (all prose).

3 pl. (mid.) eran: adhyabhāşeran 'they addressed' Mv i.56.13 (prose; no important v.l.).

3 sg. (mid.) *ita*, from an *aya* stem (see § 29.6): vilakşayila 'was observed' SP 193.9 (vs).

Relatives of the Skt. root aorist

32.106. Occasionally root aorist forms from roots in \tilde{a} occur with MIndic loss of the consonant of the ending. So Mv 1.55.15 (prose) pratyasthā, for °āt; so Senart with 2 mss.; v.l. °sthāsi, § 32.54, which is read in the same phrase 56.10, while in 55.2–3 the regular Skt. °sthāt is read. In Mv ii.315.10 samayā seems to stand for °ayāt, tho the meaning seems to be optative, 'would come together'; Senart em. samaye, against both mss. In Mv iii.457.15 (vs) anvagā, 1 sg., from anu-gā; repeated several times, the mss. sometimes showing v.l. °gāh (457.20) or °gāt (458.66, still 1 sg.); in the same verse Pali anvagā Jāt. v.258.25.

32.107. The Skt. *abhût*, in that form, is (as usual) used for other persons and numbers: 1 sg. SP 22.11 (prose); LV 239.18 (prose); Kv 24.18 (prose); 2 sg. SP 64.11 (prose); 3 pl. SP 46.5; LV 74.15; 138.3; 164.1; Mv ii.231.13 (all vss). Often it appears in MIndlc form, lacking the final *t*: *abhû* as 3 sg. even before a vowel SP 157.10 (vs) and Mv iii.360.16 (vs), and otherwise Mv i.114.8 (vs); 116.17 (vs); 341.6 (prose); as 2 sg. LV 165.13, 17, 21; 166.3, 7, 11, 15; 170.19; as 1 pl. (subject *vayam*) LV 252.11; as 3 pl. (sometimes before vowel) Mv ii.91.14; 316.18; 352.5; iii.276.5, 7 (all vss). Also with shortening of the vowel m.c., in verses, *abhu* SP 204.9 (before vowel); LV 237.14; Mv ii.343.4; Gv 285.10; Samādh 8.26, 31. MIndic h for *bh* is rare, but *ahû* occurs Gv 254.23, as 1 sg., 'I was'. The Skt. 1 sg. *abhûpam* becomes *abhûpa* in a vs, m.c.: RP 25.15.

32.108. A 3 pl. to $abh\bar{u}(t)$, abhu, occurs often in Mv, written $abh\bar{u}nsu(h)$, $abh\bar{u}fsu(h)$, abhunsu(h), °tsuh, abhunsu(h); cf. Asokan ahumsu, Hultzsch lxviii, 'based on the 3 sing, ahu (= Skt. $abh\bar{u}t$), as Pali $\bar{a}simsu$ on $\bar{a}si$ (= Skt. $\bar{a}sit$)'. As to the variant with t for the nasal, cf. §§ 32.38, 96-98. The mss. often vary between these orthographies; Mv i.41.7 (prose) abhunsuh (v.l. °tsuh, °msu), in parallels 230.2; 240.12; iii.341.14 variants $abh\bar{u}nsuh$, °tsuh; i.41.10

abhunsuh (mss. mostly °msuh), in parallels 230.5; 240.14; iii.334.12; 341.17 varr. abhūnsu(h); l.41.12 abhunsuh (varr. °msu, °msuh, °tsuh), in parallels 230.7; 240.16 abhūnsuh; i.195.11-196.7, about 10 times, mostly abhunsuh, some abhūnsuh, variants with m and once t for n; l.248.9 abhūnsuh, 12 abhunsuh, 14 abhunsu (v.l. °suh), 19 abhunsuh (all prose); 249.8, 9, 11, 13 abhūnsuh, abhunsu(h) with varr. °msuh, °tsuh; 340.8 abhūnsu; 9 abhūtsuh (so mss., both prose); abhūnsuh in prose without v.l. ii.125.12, 13; 126.18; 129.8; 162.14, 16; 163.8, 19; abhūtsuh mss. in iii.199.12, prose, no v.l. And so forth.

32.109. Either to this or the next section (thematic aor.) belong one or two borderline forms from the root bhā. LV 83.13 (prose) has abhāvatām, 3 dual, for Skt. abhātām. Weller 52 would read abhuvatām, connecting it with the Pali (thematic) ahuvā (Geiger 162.2), = Vedic abhuvat; cf. AMg. bhuvi(m), Pischel 516. 1 would not emend; but the form may be a blend of the Skt. root-aorist with the MIndic thematic past. In LV 222.11 and 21 (vss) bhuv' (imu) is read by Lefm. with mss. (vv.ll. bhuvisu, \circ su) at 21; in 11 mss. all bhū-(varying as to what follows) but this is metrically impossible. We must understand a 3 sg. past, perhaps (a)bhuv(al) = Pali ahuvā. Divy 402.22 (vs), mss. aham tadābhuvan, contains a 1 sg. abhuvam, either m.c. for Skt. abhūvam, or a 1 sg. to Vedic abhuvat = Pali ahuvā. (The ed. emends to abhavam.)

Relatives of Skt. thematic aorists (and imperfects)

32.110. 3 sg. a, ā. Here belong forms of the roots drs, vac, and gam, historically thematic aorists. In general they correspond to the Pali forms addasā, agamā, and avaca, avoca (Geiger 159.II; 162.3, 4). I have hardly noted them outside of Mv, and there (strangely) only in verses. MSV has one prose instance.

32.111. The first is rare: Mv i.9.3 addasā (so Senart with one ms.), or better addasā (2 mss. addasām, 3 mss. adršā; meter requires -dd-, but long ā is not required). In Mv ii.186.8 and 14 addarsā, with full-grade root syliable as in ad(d)arsi etc., § 32.27; here the meter is indifferent as to quantity of all the syllables. For (ad)drsi and the like see § 32.25.

32.112. From gam, as in Pali (ultimately = Skt. agamal), we find as 3 sg. agamā Mv i.220.10 = ii.22.11; as 1 sg. agamā iii.12.5, 16 (so, or āgamā, one ms., the other agamo or \bar{ag}° ; Senart em. agamam); and as 3 pl., acc. to mss., agamo iii.88.5 (vs); Senart em. agamu (presumably supposing equivalence to Skt. -uh, but a more likely em. would be agamā, cf. the preceding where agamo is v.l. for agamā); also 3 pl. upāgamā ii.53.19 (vs), so read with 1 ms., v.l. upāgame; reading confirmed by same vs in Pali Jāt. v.393.4 athāgamā (Senart em. °galā); āgamā(h? before y-), 2 sg., MSV i.43.5 (prose).

 P before y-), 2 sg., MSV i.43.5 (prose).
 32.113. Commoner are the forms of vac, especially avaca (see Chap. 43), as in Pali. It is used as 3 sg. My i.143.11 = ii.4.4; ii.6.12 (parallel i.202.15 avaci, on which see § 32.24); i.225.18 = ii.28.19; i.304.6; ii.12.3 (parallel i.207.13 avaci); ii.29.11; 61.15; 134.15; 205.5 (Senart em. avacā m.c.); 222.9; 227.16; 228.16; 230.5; 327.18; 333.21; 340.4; iii.105.1, 3; 134.18; 270.16; 294.16; 355.15; as 2 sg. iii.17.14; as 3 pl. i.218.1 = ii.19.20; ii.221.9; 226.11; 228.14. (On the corresponding 3 pl. avacamsu see § 32.76.) -Senart reads avacal with one ms. ii.27.1 (v.l. avaca); in the parallel i.224.1 mss. avaca, Senart em. uvāca. In ii.71.5 the mss. have avacati, perhaps for avaca with the 3 sg. present ending (cf. avacasi 3 sg., § 32.62; addrśāti, addaśāmi, § 32.58), unless -ti represents iti, which is possible; Senart emends to avaca. And finally ii.330.5 has, according to the text, avāca, without v.l.; misprint for uvāca? Or for avaca with ā m.c.? (But the meter is confused and obscure.)

32.114. As in Pali, the regular Skt. avocat is inherited and appears as avoca Mv ii.37.12; same as 3 pl. li.204.14.

32.115. Also a number of (almost always) unaugmented thematic imperfects with Mindic ending a for al (sometimes \hat{a} , perhaps only m.c.), or occasionally for as, 2 sg., or an, 3 pl.: LV 166.5 (vs) hala bhava (= abhavas), 'thou wast struck'; 185.11 (vs), read abhistuva (metr. correct; § 28.7) for text abhivusta, Corrigenda abhitustuvu (but a 3 sg. is required; mss. vary greatly but suggest -stuva); 297.8 (vs) upadarsaya, 3 pl., 'pointed to (the Bodhisattva)'; 308.12 (vs) gaccha (= agacchal); Mv ii.205.6 (vs) parikirtaya (= paryakirtayal); Dbh.g. 18(354).8 abhikira, 3 pl. (= abhyakiran); disa (= adisat) LV 391.21.

32.116. Less numerous are such forms in \bar{a} , possibly only m.c. for a (but cf. addasā, agamā etc. above): sampraticchā, with 3 dual subject (some mss. °ccha, unmetrical) LV 92.13 (vs); samprayacchā (but this is an em., all mss. corrupt) 197.11 (end of vs).—In Mv i.182.16 (vs), read amanlrayā 'he announced', with 4 mss. (one amanlreyā; Senart em. °traye); here augmented.

32.117. 3 sg. (pl.) atha. The ending (a)tha, recognized in Pali and there classed as 3 sg. middle in origin (Geiger 158, 159.11), occurs sporadically, tho apparently only with 3 pl. subjects, and only in verses of Mv: i.176.17 lipyatha 'were stained', to lipyate; niliyathā i.219.3 = (a)bhiliyathā ii.21.5 'they flew down, flew thither' (so mss., at end of line; no metrical reason for final \bar{a} ; Senart em. °tha, °ta). Cf. §§ 32.41, 42.

32.118. 3 pl. As we have seen, imperfect and aorist forms are not clearly differentiated in our language, any more than in Pali (Geiger 158, 159.II, IV). Pali has um, corresponding to Skt. uh, in 3 pl. past forms generally (so far as they do not end in -msu). Similarly our language occasionally substitutes uh, or (m.c.) u, for an, 3 pl. thematic impf. or aor. So LV 27.22 (vs) adrsu (cddrsu, written with s) = Pali addasum, Skt. adrsan; LV 385.19 (vs) vimrşuh (v.l. with s), to vimrsah. IV 385.19 (vs) vimrşuh (v.l. with s), to vimrsah. IV 385.19 (reads desayuh SP 57.16 (vs), and 272.10 (vs); but in the latter the Kashgar rec., La Vallée Poussin JRAS 1911.1076, reads desayi, and the sense appears to be that of the optative or future (cf. below), a prophecy. According to Kern, SP Preface vi, the Kashgar rec. records avocuh for 3 pl. avocan.

Aorist forms with optative or future meaning

32.119. As we saw, many forms which historically were optatives are used in the sense of past indicatives. The reverse seems equally clear, tho less extensively attested. In JAOS 57.33-34 attention was called to 'forms in *i* [also *il*] ... which look like 3 sg. aorists, but have the meaning of optatives (either jussive or potential), or even apparently of simple futures. Since the optative approaches very closely the meaning of the future at times, all these may perhaps be classed as "optative" in meaning.' These apparent aorist forms, and others like them which have been discovered since that time, are treated here rather than under the optative, largely because some of them seem more like futures in meaning than optatives; the line is hard to draw. Sometimes, indeed, parallel forms in the context are present indicatives. See the place quoted for discussion and interpretation of the SP and LV passages.

32.120. The following seem more or less clearly optative in force, either 'jussive' (injunctive), or potential. LV verses: vyākarī 57.3; pravišī 115.4; ābhāsī 122.16, 19; tapī 122.19; niścarī 177.9; dršī 177.12.—SP verse: daršayi 291.12. — daršayī Šikş 324.13, 16; vidaršayī 325.20; all in, verses, from Vimalakirtinirdeša; cf. the parallel vinirdišet 325.19; śruņī Šikş 341.7 and paśyī 343.1, 3, 7, all in vss from Ratnolkādhāraņī, some 3 pl.

[32.121

32.121. More definitely future in mg. (promises and predictions) seem the following: uddhari LV 195.12; 196.2; sampaiyi 195.20; deiayi SP 272.5, and Kashgar rec. for deiayuh 272.10 (see § 32.118; also La Vallée Poussin l. c. prakāšayi for ed. prakāšaye 273.10); samprakāšayi SP 295.2, abhinişkramī and upasamkramī 4, pravarlayī 7, deiayī 8.

32.122. Other aorist forms similarly used: mucyişu 'let them be freed' Suv 39.4, samyujyişu 'let them be joined' 5. So Nobel, adopting the readings of the citation in Sikş 218.3, 4. The meaning is certainly injunctive; parallel verbs are imperative.

32.123. A seeming sigmatic aorist ākhyāsi seems used as a future in Mv iii.451.6 ko (so mss., Senart wrongly em. yo) me śronakam ākhyāsi, 'who will announce Ś. to me?' Paralleled in line 10 by 1 sg. aham ... ākhyāsyam (but v.l. ākhyāsi here also, perhaps read so). Same verse in Pali, Jāt. v.249.8 ko me sonakam akkhāti, but v.l. ^osi; repeated in 250.20 akkhāti, no v.l. Senart takes ākhyāsi as for ākhyāsye, a future with opt. ending; highly improbable. Occurring at the end of an anustubh pāda, the final -i cannot be metrically conditioned. I find the form very puzzling; if it is morphologically a future, it could only have s for sy (§ 31.27), but would be hard to explain otherwise (1 sg. used as 3 sg.? cf. § 31.34, and 35, 1 sg. as 2 sg.).

32.124. Finally there is a doubtful case of what might be an imperfect, with MIndic transformations, with the meaning of an optative; but the text is very uncertain: yadi siuno ekasa eta sutram SP 230.4 (vs) 'if he should hear this sutra once'. The mss. vary: sinod, siunod, siunod, siunai (which suggest siune, optative!), siune, and Kashgar rec. sidavaye (optative!); WT keep siuno without note. If it is the true reading, it may represent airnol. For another guess see § 29.46.

33. Perfect

33.1. In Prakrit the perfect, according to Pischel 518. exists only (in AMg.) in the 3 pl. forms and anamsu, both of which may also be used as 1 and 3 sg. In Pali, according to Geiger 171, the only really live forms are aha, sg., and $\bar{a}hu$, $\bar{a}ha\eta su$, 3 pl.; and 3 pl. vidu, vidū, vidu $\eta =$ viduh, 'they know'. However, other old forms, such as bubodha, susoca, are used 'in der Kunstsprache', artificially, as 'gelehrte Reminiszensen'. Probably our dialect originally resembled Pali. Actually, in our mss. and editions, regular Skt. perfect forms are not too rare, if not exactly common. It may be suspected that many of them were introduced secondarily in the course of transmission by copyists. But when Sen 54 declares that the perfect occurs in (our text of) LV only from sthā, (abhi-)stu, and ah, in SP only from vac, sthā, ah, kr, and in My only from vac, $bh\hat{u}$, $bh\hat{a}$, (vi-)hr, and $dr\hat{s}$, he is guilty of gross exaggeration. Without attempting to make a list, I can cite also (besides irregular forms cited below) neduh SP 23.13 and vineduh 190.5; pracacāla 23.15; pramumoca 23.16; babhūva 28.1, 2; yayācuh LV 7.8; upajagāma 55.13 (prose); uvāca 110.3; 371.3, provāca 110.5, and abhyuvāca 133.6; babhūva 28.19; 111.3; babhūvus 386.21; babhāşe 114.4; 199.7; 338.4; jagrāha 272.2; 387.8; menire 280.20; śekuh 329.12, mss., Lefm. em. śeku m.c.; viviśuh 385.15; pracakruh 386.2; pramumoca 412.2; cacala My i.99.1; nanade iii.24.2 (for nede).

'Irregular' perfect forms

33.2. Instead of the 3 pl. ending uh, u occurs, as in Pali and (ahu) AMg.; in prose of Mv, babhūvu i.153.16; ahu ii.96.8 (v.l. ahuh); but chiefly in verses where meter requires a short, viceru SP 331.8; pramumocu (but WT °ca with K') SP 190.4; Samādh 8.23; ahu Mv i.183.5; 220.1 (here, tho in a verse, not required by meter); iii.370.7; 423.2.

33.3. In Mv iii.395.7 the mss. read ahum; this reminds us of the regular Pali preterite 3 pl. ending um (Geiger 159); but this ending seems not to be used in our dialect, and apparently ahum is not used in Pali for ahu (= Skt. ahuh). Probably, therefore, Senart is right in emending to dhu.

33.4. Occasionally (chiefly in vss, perhaps then m.c.) a strong form of the root is used in place of the weak in dual and plural perfect forms; pramumocu SP 190.4 (but WT °ca with K'); Samādh 8.23; pramumocatur SP 460.2 (prosel); eikşepur LV 7.14.

33.5. In Mmk occurs (chiefly in verses) an ending *ure*, historically (and half the time in actual practice) 3 pl., evidently a blend of *ur* (uh) 3 pl. active and *ire* 3 pl. middle. The only common form showing it is *lasthure* Mmk 140.7; 169.10 (prose); 172.16; 202.26; and as 3 sg. (cf. below) 172.15; 381.3; 429.21; 551.18. (The regular *tasthire* is also used in Mmk 550.8.) The only other form in *ure* noted is anomalous otherwise: Mmk 503.18 papeture, 3 pl. of *pat*, = *petuh* or *petire* (with reduplication prefixed

to the weak stem in medial e; in a verse, the meter being correct, thus supporting the reading).

33.6. Another anomalous form is Mmk 381.2 roruroda, in a verse, meter being correct as it stands. Here we seem to have a double reduplication of root *rud*, the syllable *ro*- being intensive in character; indeed this could be called a sort of perfect intensive (to Skt. *rorudal*, *rorudyale*), but not of the type recognized in Skt., Whitney 1018.

33.7. Apropos of the elsewhere mentioned use of historically 3 pl. forms as singulars (\S 25.30-33), this seems particularly frequent with perfects, including regular inherited forms; so 3 sg. tasthus Mmk 470.24, cakrire 608.4. It was noted above that AMg. *āhu* and *āhaŋsu* are used as singulars. The mss. of Mv also present *āhaŋsu* as 3 sg., tho Senart emends. It will be recalled that 3 pl. 'aorists' are also sometimes used in this way, here as in AMg. Possibly the relatively greater frequency of this use of perfect forms is connected with the obsolescence of the perfect as a whole.

33.8. The Pali-AMg. form āhamsu, 3 pl. to āha, is correctly explained by Geiger 171 as modelled on such aorist forms as adamsu. It belongs to our dialect too, being extremely common in the Mv (hundreds of cases). Strangely, I have failed to note it in any other text; apparently it was discarded early in the history of our dialect. Both n and m are written before the s, at random; this is merely the usual orthographic variation between those spellings. But also *l* quite often appears instead of n or m, either in all the mss. noted by Senart (e. g. i.350.19; 351.2), or more often in some of them. This is of course an assimilation to the common, even prevalent, use of tsu(h) in the 3 pl. aorist (§§ 32.97, 98). (Senart never admits t in his text, here or in the aorist.) Another sign that these forms were felt as normal preterites may be seen in the fact that the initial vowel is often written short a, instead of \bar{a} , suggesting that it was assimilated to augmented forms. Sometimes all Senart's mss. read ah° (e. g. i.272.17); oftener it is recorded as a v.l.; Senart invariably adopts āh° in the text, but it is very probable that the authors used ah° by the side of ah°. As to the final visarga, it may be present or absent at random, to judge from the mss.; often they vacillate in an individual passage; but on the whole it seems that there are distinctly more cases where they agree on -uh than on -u. Whether we should explain this as due to secondary Sanskritization by copyists seems questionable, in view of the generally very dialectic aspect of the Mv text as the mss. present it.

33.9. The forms are so abundant thruout Mv that it seems unnecessary to quote more than a few. Usually there is some variation in the mss.; Senart's critical apparatus must always be consulted, his text being more than commonly arbitrary. The forms presented are *āhansu(h)*, *āhamsu(h)*, *āhahsu(h)*, and ah° (with any of the above medial and final spellings). Very rarely final um appears, at least as v.l., for uh or u (e. g. iii.199.2

Ĩ

(165)

11*

166

33.10. Since the perfect, so far as it survives at all, is only another preterite in meaning, it need not surprise us to find it provided with a preterite ending (cf. the shortening of \bar{a} - in *ahamsuh*, above, assimilating it to the

augment). In Mv iii.337.13 (prose) occurs uvācat (to be sure with v.l. uvāca); the aor. avocal no doubt helped to create the form.

33.11. [Senart i p. 453, note on i.99.8 dadarśatha, suggests that the 3 sg. ending tha (§ 32.117) is here attached to a perfect 3 sg. But he admits that dadarśa (a)tha is also a possible interpretation; several mss. in fact read dadarśa atha or dadarśātha. In my opinion we must certainly understand the particle (a)tha.]

34. Participles, including gerundives

Present participle suffixes affixed to past participles

34.1. Renou, Études de gr. scte. (1936) page 18, cites some bizarre forms in which pres. pple. endings, active and middle, are affixed to past pple. stems; among them, from Divy 510.8 (prose) pakvamāna 'being baked'. of pots (pakva functions as past pple. to pac). He also cites from Senart's Index baddhamānā Mv i.76.14, but this must be emended to baddhamānā (Dict.). Pali has a case noted by Geiger 191 note 1, vibhālamāna (Dhp.c. i.165.11; Geiger is wrong in questioning the form). And several others occur in our texts: (sāksi-)prşlamānam Mvy 6993 '(being) asked, questioned'; pravişlamānasya LV 7.2 (vs) 'being entered, pervaded'; vyūdhamānaś RP 55.1 (vs).

34.2. I have found only one possible case of the pres. active pple. ending so used: so 'pi (Māro) bodhisatlvasya sadvarsāni anubaddhanto alabhanto avadāram etc., Mv iii.241.5 (prose), 'he, being in pursuit ot the B. for six years, getting no opening' etc. Note samanubaddho in the preceding sentence. But there are other forms which seem to point to a present baddhati (§ 28.19 and Dict.), from which this may be derived.

māna with athematic stems

34.3. Like Pali kurumāna (Geiger 191 end) is $i_{r,numāna}$ Šiks 107.9 (vs), the only case I have recorded, except samāna = sant (Dict.).

ana with thematic stems

34.4. Found occasionally in Pali and Pkt. (Geiger 192, Pischel 562); and cf. Renou, Gr. scte. p. 414, and Études de gr. scte. p. 18, top: esāno 'seeking' Mv iii.2.9 (vs), esānah Divy 559.5 (vs); the same form in Pali, esānaGeiger I. c.; vrajāno 'going' Divy 392.20 (vs); abhisambudhāna(h) Ud xviii.18; same form in same verse in Pali Dhp. 46 (classed by CPD as 'aorist', perhaps thinking of the RV 'aorist' pple. budhānā, but I refuse to consider any possible connexion with that); note that later mss. of Ud change the text to paribudhya caiva (the patchword caiva advertises the later redactor's standardizing intent). Also anucankramānāh Sukh 58.11.

34.5. More surprising is nisīdiyāna in SP 295.5 (vs) simhāsane latra nisīdiyāno, 'sitting on that throne', which in form looks like a passive stem with ending āna instead of māna. Curiously, there is a close parallel in a stray Pali form which has also caused difficulties: paripucchiyāno Sn 696. Geiger l. c. cites this as a passive ('beſragt'); but

1. This is the second of the two possible Skt. origins envisaged by Leumann, l. c. Of course Skt. has, even in early times, pples. in *ila* which are not matched by *aga* presents. Some of them are doubtless secondary and analogical, but some may be IE., from dissyllabic bases with heavy second syllable and hence IE a = Skt. *i* in zero grade (after consonants), as Leumann suggests in his first alternative. However, his only example, *anita* 'breathed', cannot be prehistoric. After a semivowel, such to me, as to (I believe) all other interpreters, it seems certainly active in meaning, 'asking'. The PTSD apparently would emend; it cites the form as *paripucchigāna*, a gerund (§ 35.45), but apparently all mss. read "gāno. No really satisfactory explanation occurs to me, but this Pali word and the SP *nişīdiyāno* seem to support each other. The forms are alike: on the surface, present pples. in the athematic suffix āna from what appear to be passive stems (in MIndic *-iya-*, § 37.2), but with active meaning. It hardly helps to compare the active and transitive use of the past pple. in *ta* noted below, § 34.15.

Blend of perfect vāms and (la)vant (?)

34.6. In one doubtful instance, LV 368.2 (prose), the printed text has (a)valasthivantah, which if correct would be a blend of the old perf. act. pple. avalasthivānas with the later avasthilavant. Only one ms. has this form (but that the best, A); most mss. avasthilavantah, the later form.

ita added to thematic present stems

34.7. This is a common MIndic phenomenon, but is not made clear in Geiger 196, Pischel 565. A better statement in M. Leumann, IF 57.206 f. The regular Mindic pple. in ila is based on thematic presents, not on any 'present stem' as Geiger and Pischel suggest. It spread chiefly from aya verbs, where it was regular in Skt.;1 when it matches such verbs we shall not list the forms, even tho they may not exist in Skt. (as in such pples. as kārāpila, to kārāpayali, kārāpeli), since their relation to the rest of the inflexion is 'regular' even from the Skt. standpoint. We may, however, note such a special case as -dhila, in samdhila Mv ii.222.7, 17 (vss; in 7 Senart em. samdahila, presumably to correct the meter), and anlardhila Gv 444.13 (prose), presumably based on the present -dheti. (The Skt. would be -hila.) Similarly samudänita, pple. to samudänayati, Pali samudäneti, Pkt. samudānei with pple. samudānia; see §§ 3.43, 38.3.

34.8. Special interest attaches to a few forms in ayila from aya presents, in which -ay- is kept before ita. Here it is obvious that the *i* in ita is not based on aya or *e* of the present but on simple a: in LV 374.7 lätayila (as noun); 420.3 (vs) *vedayilu* = *veditam*, to *vedayate*, 'experienced by the senses'; prafamayitāni Mv 1.254.8 (vs); upanāmayito (v.1. 'nāmito) ii.168.6 (prose); prafāpayitam iii.93.3 and 4 (so Senart, for mss. prafāpāyitam, prafāpayitam idpitam; the latter may be right, § 38.56); ghosāpayita-vān SP 257.4; pratisthāpayitā(h) = 'pitā(h) Kv 18.20 (prose).

roots had a monosyllabic zero grade; IE $\bar{p} > \text{Skt.} \bar{a}$, as in $\hat{g}_{n}\bar{q}io > jd\bar{a}$. As a matter of fact I question the existence of anila as a pple. of root an in Skt. Whitney's reference (Roots) probably refers to the pple. pranila, which belongs to pranayali, not to pranili. If anila (to anili, anali) really exists, it is just as secondary and analogical as is carila (instead of old cirna) to carali. Only after consonants (other than consonantal semivowels) could IE z be preserved as Skt. *i* in such forms.

34.9. A rather special case, not guite on a par with the preceding, is anuvicintayitah SP 109.2 (prose) 'caused to meditate upon'. Note that it is not a synonym of anuvicintitāh, which would mean 'meditated upon'. It is a causative participle to a (probably denominative) verb in ava (§ 38.9), based directly on the non-causative pple. anunicintitah. One Nep. ms. has in fact the Mindic caus. pple. °cintāpitāh, and the Kashgar rec. °cintāpauamānāh (read °cintapyamanah?).

34.10. The rest will be listed in order of the Skt. 'roots', but they are all based on thematic presents (often Mindic; see Chap. 43). Those not marked prose are from verses. They are usually quoted in 'stem' form (in -la). añj: añjita (= Pali id.) Mv ii.234.17 (prose).

is (all prose): praticchila (MIndic i for i, § 3.38) Mv ii.165.19; 189.12; praticchila AsP 204.15; Gv 349.7; °lavān Gy 417.24; sampraticchita Mvy 366; yalhecchita Mv iii.24.19.

kr 'make': samalamkarila Śiks 327.16.

kr 'scatter': prakirila Mmk 76.16 (prose); abhuavakirita, perhaps read °kirita; but see § 28.34; or blended with "kirna, "kiryale, or directly based on the latter, cf. under dr below, Mmk 63.25 (prose).

khan: khanila Divy 16.4 (prose).

qūh: upaqūhila Lank 95.14.

ghrā: ghrāyita Mv ii.251.10 (prose); a-gh° Šiks 341.9. chid: chindila Mmk 708.28 (prose; text not wholly certain).

jāgr: pratijāgrita Mv i.270.13; ii.150.8; 152.1 (all prose); probably based on MIndic present -jāgrati (3 sg.), § 28.11, rather than phonetic change from -jägrla, § 34.14 (cf. § 3.94).

jñā: jānita LV 414.16 (so with best ms.; jānita sativacārī, 'known is the conduct of creatures').

taks: laksita My 1.5.8 (prose).

tap: lapyila (in tapa-lapyila-sarirām 'with bodies tormented by austerities') Mv iii.102,13 (prose; so with mss., v.l. tappita; Senart wrongly em. tapita; cf. manyita).

tr: tarita Mv ii.294.9.

trap: vyapatrapilavān 'was ashamed' (not caus.) Av i.272.9 (prose).

trā: trāgita Mv ii.177.4 (prose); Gv 301.17.

dr: dirita Mmk 143.24, for darita from darayati, the only active present in post-Vedic; if not a corruption, influenced by dirna, diryate, or directly based on the latter, cf. under kr 'scatter' above, and § 37.28.

prcch: prcchita Mv 1.232.15 (prose); 280.4; 324.5 (prose), etc.; Suv 134.3; pariprechila Suv 134.1.

bhuj: paribhuñjita My i.19.4 (prose).

man: manyila SP 63.3; a-manyila Samādh 22.12 (prose).

muc: muñcila Suv 49.1.

mlä: miläyila My ii.232.3, and pra-miläyila 4.

yaj: yajila Kv 29.21 (prose).

yam: vyāyacchila Bbh 92.16 (prose).

yā: yāyila My i.299.8 (to yāyali; Senart wrongly eni. iāta).

rabh: ārabhita SP 408.15.

ric: riñcila SP 62.2; Divy 638.10 (prose); Mmk 73.9 (prose).

ru: ravila LV 286.14 (prose; twice). Same repeatedly

in Sútrálamkāra, ed. Lévi, p. 80 (xii.9, comm.). ruc: ruccila Mv ii.427.16 (prose; with MIndic cc, to rucyali, ii.464.10).

labh: labhita LV 327.12.

vas 'dwell': vasila LV 364.3 (Skt. usila).

vij: samvijita (=Pali id.) Ud xix.1.

vr 'cover': vivarila LV 233.3, 13.

vrt: pratinivartita 'turned back' (intrans.) Mv iii.102.19 (prose).

šak: šaksila Mv iii.263.15 (prose), to *šaksali; šakkila

Av ii.23.2; 71.1 (prose), to quasi-MIndic *sakkali (Speyer em. to sakilam, wrongly).

sah: utsahitah 'he ventured', as periphrastic preterite to utsahate. My ii.90.6 (prose).

siv: sivila Mv iii.313.4 (prose).

sr: samsarila 'passed' (subject, kalpas) My 1.46.4 (prose).

stu: stavita LV 294.17; Suy 64.9 (prose).

str: an-āstarita Jm 220.14 (prose). sprš: a-spršita Bbh 111.2 (prose); upa-spršitavant Mmk 47.5 (prose).

smr: vismarila Suv 103.4 (prose).

han: vihanuila Mv 1.155.14 to vihanuale (see Dict.). hā: jahila Mv i.292.1; LV 173.20; 242.4.(Rare in Skt., chiefly Vedic.)

hu: juhita Mv iii.149.2 (prose); 431.14 = 17 = 445.7, 10.

hr: vyavaharila (mss. vyapa^o) Mv i.271.18 (prose); vyāharita Divy 397.24 (prose).

Spread of ta outside its Sanskrit bounds

34.11. Despite the prevailing trend towards ita, we find a few cases of forms in la which are not standard Skt. Some are formed from verbs which have no recorded participles in Skt. Others replace forms in ila or na. On similar trends of MIndic cf. Pischel 564-5, and M. Leumann IF 57.210. Some of our forms are identical with recorded MIndic forms. A special case is vyavasta, § 3.112. Examples are from prose except when marked vs.

as 'obtain': as/a LV 390.9 (vs). Recorded V.B., Whitney, Roots; no pple. in Class. Skt.

caks with a: acasta 'addressed' My ii.96.17. No pple. in Skt.

dham (dhmā): nirdhānta (Pali, AMg. niddhanta), Dict.; to dhumali, on pattern of kşanla: kşamali etc.

dhā: abhiśraddadhātam 'believed' Divy 16.5, repeated later in same formula. In one repetition, 17.4, mss. and ed. abhiśraddadhilam; read either °dhālam (so probably), or possibly °dhitam. Also śraddhātam 'desired' My i.187.6 (mss.; Dict. s.v. sadyam).

bhās: (Skt. only bhāsila, except Mbh. Cr. ed. 3.126.28 paribhāsiah; AMg. ābhaiiha) bhāsia Lank 283.8 (vs; note bhāşila line 10); su-bhāsta LV 230.13 (vs); abhāsta Lank 283.11; Mv ii.157.16; 234.2, 3; iii.65.3; 92.7, 8; 181.5; 313.11; 329.11; 379.14; 413.11; 430.15; pratyanubhāşta Mv iii.393.17.

mlā: (mlāta RV, otherwise Skt. mlāna; but Pali milāla) āmilāla, sammilāla, Mv ii.126.4, 5 etc. (see Dict.); the same passage in LV 254.14 reads -mlāna.

rup, quasi-root: orupta 'planted' (chiefly figuratively, with kusala or kusalamūla) Mv iii.104.18 (vs); 155.6; 432.9; avarupta id. Suv 91.8; 103.7; 113.5; 114.1 etc.; literally, of a tree. Gv 278.22. A back formation from avaropayati 'plants' (see Dict.), which is of course caus. to ruh; the formation is analogous to 'root' gup, from gopa(yali), if that is the true origin of gup (there is another theory).

vas 'clothe': (Skt only vasila; Pali nivaltha, AMg. paniyattha) nivasta LV 157.21 (vs); Mv ii.211.12; 322.10 (vs); 430.2; iii.157.1; 210.13: 267.9, 11 (these last two vss, readings questionable); Mmk 63.2; 68.25. See Dict. s.v.

vas 'dwell': (Skt. usually usida, but Epic also usia; Pali and Pkt. vullha, Geiger 195, Pischel 564) abhivusta LV 185.11 (vs); according to Senart's text samwullha Mv iii.185.9 (vs), but mss. vusto, samvutto, probably read samvusto; upavusta (mss. opa[°]) Mv iii.216.5 (vs).

34.12. A special case is palāta, as ppp. to palāyali 'flees' (Pali id., M. JM. palāa, palāya). It is probably found in Siks 347.4 (vs), where read palal' (= palalah) asurendrah, for text palato aso, Like palana, § 34.19, it seems to 34.21]

be analogical to pairs of forms pointing to presents in -dii and -dyali; or perhaps directly to such a pple. as jāta to pres. jāyate. Cf. § 28.54.

Strong form of root, for Skt. weak form, before ta

34.13. Some participles show a strong form of the root before the suffix, where normal Skt. uses the weak form. The commonest case is yaşla, from yaj, for Skt. isla; it corresponds to AMg. jallha (not recorded in Pali), and occurs (in vss except as marked prose): LV 170.12; 180.22 (prose); 223.18; 284.2; 318.4 (prose); 340.15; 341.5; 360.20; 402.9; 423.20 (prose); 437.20; Mv iii.126.10 (prose, v.l. isla); 402.13 (prose); Divy 405.6; 624.4 (prose); Gv 357.7 (prose); 539.1 (prose). — upagrahila Mv ii.247.15 (prose, no v.l.; hyper-Skt. for upaggah? see § 2.7). — praşla (= prşla) 'asked (about)' Mv iii.166.12 (prose, no v.l.). — adhyaşla (cf. Pali ajjhiljha), paryeşla (cf. Pali ajjhiljha), see Chap. 43, s.v. 1 iş, 2.

Weak form of root, for Skt. strong form, before ta

34.14. The converse of the preceding also occurs: pratijāgria (Skt. jāgarita, analogical form) Mv i.258.16; ii.111.8; 180.6; iii.324.10 (all prose); also jāgria, noun (Dict.). On pratijāgrita, see § 34.10.—prihita (= Skt. prathita) Siks 247.16 (prose); if not a textual error, perhaps hyper-Skt. for AMg. pahiya = prathita.

Participles in *la* with active meaning from transitive verbs

34.15. As occasionally in Skt. (Renou, Gr. scte. § 152, p. 197), participles in ta from transitive verbs may be used with active meaning, even governing an accusative. The usage is not noted by Sen. (ratnabhadram karena) grhilāni LV 124.4 (prose), 'having taken an excellent gem in the hand'; yasyārthe vralatapa carilas tvam 164.9 (vs). 'for the sake of which you have practised vows and penance'; balīvardhau... śakaļā prakarsitau 382.1 (prose) 'the two bulls drew the cart(?)'; so Tib.; see § 8.38; prasādam pratilabdhā(h) 382.8, followed by a- (no ms. has a gerund), 'having recovered their composure'; (lā gāvas...) sarpimaņdam pradugdhā abhūvan, 386.5 (prose), 'had milked forth cream (or butter)'; gāvah sarpimaņdam pradugdhāh 386.7 (prose). — bhagavantam abhistutah Mv i.98.2 (prose), 'praised the Bh.'; upavāsam samādatto ii.178.1 (prose), 'having taken on himself a fast'; upavāsam samādattasya id. 6; (bodhisattvo . . .) āhāram abhyavahrto 241.1 (prose) 'ate food'.

Suffix na

34.16. This suffix is very little used except in inherited Skt. forms. The only common non-Skt. form is dinna =Pali id., AMg. dinna and dinna, 'given', which is very much used in Mv, rare elsewhere: dinna Mv (prose) i.44.13; 232.7; 256.7; 287.19; 289.10; 349.12; 352.3, 5; 355.11; 359.20; 360.10; ii.64.10 etc.; KP 9.13 (vs); juxtaposed with datta, Mmk 56.21 (prose); Angadinna n. pr. Jm 192.12 (prose; the corresponding Pali name is Angali); ddinna Mv ii.387.24 (vs); upādinna Mv ii.279.14 (prose); paryādinna id. iii.429.11 (prose), 15 (vs); paryādinnavān id. 1.128.15 (prose); samādinna Mv i.211.14 = ii.15.14 (prose).

34.17. runna, pple. of rud, is familiar in both Pali and Pkt. In our language I have found it only in the cpd. runna-nayana, 'with weeping eyes', Mvy 6663 and LV 195.2 (vs, printed rūnna^o). In Pali it is used both as participial adjective and as noun, 'weeping'. In the latter sense Mv (ii.218.13 etc.) uses runda; in the sense of an adjective, 'weeping', or more often as the equivalent of a periphrastic verb expression, 'wept', it uses prarunda. On both of these see Dict.; they are related to AMg. runda (cf. Hem. 4.57). [A third occurrence of runna is presented by Senart's text of Mv iii.116.8 (prose) runnena sokena. But this is a false emendation of Senart's; see Dict. s v. ulla.]

34.18. Mvy 6983 has vidhūna, rendered Tib. bcad pa, or dgag pa 'cut off, stopped, averted'(?); Chin. and Jap. 'cut off' or 'perished'. It seems to be a pple. of vi-dhū (dhu); dhūna is authorized by Pān. 8.2.44 but not recorded in literature. I believe it exists in AMg. as vihūna, rendered in Ratnachandra by Skt. vihīna, 'deprived of'.

34.19. To palāyati is formed palāna (= JM. palāņa; Pischel 567), like palāta (\S 34.12), but occurring oftener: Mv. i.270.12; 283.7 fl.; ii.172.16; 235.2; iii.350.1; also prapalāna, Dict. That it is a past, not a present pple. (with the suffix āna, \S 34.4), is proved by several of these occurrences, notably ii.172.16 where palāno can only mean 'he has fled'. An explanation is suggested in \S 28.54. Most mss. in Mv i.71.2 read prapalīna; cf. Dh. papalīņa, Pischel 567, 110, 562; the ending reminds one of Skt. $\ddot{a}s$ -ina, and of the Aśokan ending -mīna for -māna (Hultzsch cx).

Gerundives in *itavya*

34.20. The only freely used and indefinitely productive gerundive suffix is *itavya*, added to thematic present stems. Examples are from prose except those labelled vs:

Root āp: anuprāpuņilavya Mv iii.287.17.

iş: icchilavya Mv 1.280.7, 9 (vss); iii.406.9, 10 (see Dict. s.v.).

kr 'make': satkaritavya Mv i.37.3.

krī: krīņitavya (v.l. kri°) Mv ii.274.2.

ksip: praksipitavya Mmk 695.1.

grah: samgrhnitavya Mv ii.479.1. Cf. grahetavya \S 34.21.

jñā: prajānitavya Suv 179.5; jānitavya id. 180.10 (both vss).

tras: uttrasitavya Śiks 200.14-15.

trā: paritāyitavya (MIndic t for tr) Dbh.g. 3(339).19. diś: ādiśitavya My iji.420.11.

dhā: abhiśraddadhitavya Mv iii.76.6.

nam: samnamilavya Mv ii.230.15; avanamilavya Gv 507.1.

paś: paśyitavya Mv ii.107.18.

prech: prechitavya Mv i.272.4; pariprechitavya Mv ij.297.10.

badh: bandhilavya Mv iii.55.8.

man: avamanyilavya Śikş 108.2 (vs).

mŗ: maritavya Mv iii.354.7.

ram: ramitavya Mv ii.107.17.

ruh: abhiruhilavya Mmk 701.12 (to ruhali); abhiruhyalavya (l to ruhyali, Chap. 43, s.v. ruh 3, and § 34.21) id. 719.22.

likh; abhilikhilavya Mmk 45.6; 62.23.

lī: avalīyitavya ŠsP 1456.18.

vr: āvaritavya Mv iii.55.6.

-śirati: ośiritavya Mv ii.424.20.

sad: nisīdilavya Mv i.354.14.

sic: abhișiñcilavya Mv 1.51.5.

str: samstarilavya Mv i.273.3 (not recorded in Whitney, Roots, though it might pass as regular in Skt., which has set as well as anit forms from this root).

hr: viharilavya Mv iii.265.11, 13; 266.1.

Other non-Skt. forms in lavya

34.21. After a long vowel tavya remains as such even in new (MIndic) formations: grahetavya, to *graheti, Mmk 694.10; 708.2 etc. (prose); vesfetavya, to vesfeti, Mv iii.177.16. The ger. hretavya, to hrī (Chap. 43, s.v., 5), is an easy analogiçal formation (e. g. to hrīta, as netavya to nīta). On nisetlavya see Dict. s.v. nisedyati. On svādhyātavya 'to be studied', Samādh 22.8 (prose), from the denominative svådhyåyali, see § 28.54. On selavya, to siv, see § 28.50 and Chap. 43, s.v. siv (2). Av ii.154.3 f. (prose) has sammråsjanya = Skt. °mårs°; cf. the Bråhmana future mraksyale, to mrj. On the rare sparsjavya, probably a wrong Sanskritization (blended with sparsa etc.), see Dict. Laňk 9.6 drslavya, for drasf°, is probably only an error of tradition; note drstå for the agent noun drastå in the same line. The same may be true of nigrhlavya Mv i.278.6 and samgrhilavya id. 7, for °grah°; such errors are found in the mss. of standard Skt. works. Instead of -ilavya, -lavya occurs in dlaplavya, samlaplavya, MSV ii.188.11, 12. Mmk 719.22 abhruhyalavya is likely to be an error for °hyilavya, but cl. inf. -alum, § 36.7.

Gerundives in aniya

34.22. There are a few tentative starts towards a similar application of this gerundive suffix to thermatic presents; but the trend never got far. I have noted a-spharagiya Siks 249.5, 8; pariprechaniya Mv 1.337.17; apasyaniya 'not to be looked at, hideous' Mv ii.447.9; anutapyaniya 'a regrettable thing' Mv iil.74.4 (all prose); asraddadhaniya 'unbelievable' SP 230.7 (prose, no v.l.); uposaniya Sukh 29.14 (to uposati, Dict.); and, retaining d of the old strong present stem, asraddadhāniya, Kern, SP, preface vii.

34.23. It may be noted that the secondary suffix iya enjoys a certain extension in our dialect as compared with Skt.; new formations are made with it from various noun

stems, including stems in primary ana (§ 22.20). Some such formations can not be considered gerundives; some are on the borderline. In such a list as LV 52.6 ff. = 411.8 ff. (prose) it is impossible to make any sharp division between the two categories: harsaniyäs losaniyäh premaniyäh prasādaniyā avalokaniyāh prahlādaniyā nirparnaniyā (second passage adds apralivarnaniyā) asecaniyā ... šabdāh, 'sounds delightful' etc.

Gerundives in ya

34.24. Here have been noted a couple of stray forms not known to normal Skt.: sambodheyam 'to be made known' Mv iii.74.1 (prose, no v.l.) is based on sambodheyati, sambodheti; in the latter (MIndic) form the present stem forms the base for the gdve. suffix ya (possibly vaguely supported by recollection of Skt. forms like deya etc., from roots in d?). Mv iii.318.2, 3 has visodheya, 'to be purified' (Dict.), influenced in form by the preceding vineya (cf. Pali leyya from lih, influenced by adjoining peyya, Geiger 202). Not certain is duh-frad-dadheya SP 70.2 (vs).

[34.25. Deleted.]

34.26. *a-vibandhiya* 'not to be obstructed' Dbh.g 41(67).4 is normal Skt. except for the epenthetic *i*, which in our dialect is common before any suffixal ya (\S 3.103).

34.27. a-vinīya 'not to be disciplined' .4SV ili.128.20, for a-vineya, or perh. error for avinīla; cf. Chap. 43, s.v. nī (2).

35. Gerund

35.1. The gerunds of BHS function as in Skt.¹ but formally diverge from it widely, being strikingly Middle Indic. See my article 'Gerunds in Buddhist Hybrid Sanskrit', Lang. 13.107 ff. In Skt., besides *-am*, are found the suffixes *lvā* (or *ilvā*) regularly with simple roots, and ya with compounds. On exceptions to this distinction see especially Renou Gr. sete. 124. Of compound gerunds in (*i)lvā*, the majority go with *aya* presents and end in *ayilvā*. Of uncompounded gerunds in ya, the commonest is grhya: Renou also notes shāpya, dršya, and some others.

35.2. The spread of -ilva started with the forms based on aya presents,² tho Skt. shows it in not a few others. Most of these Skt. itvā forms were certainly based on the present stem, as in MIndic and BHS. Thus carilva is based on carali, as kārayilvā on kārayali; it has practically displaced the original cirtvā. In the forms from aya presents the i before lvā was due to the analogy of the past participles in (i)la; the ppp. forms in la and gerunds in tod are regularly parallel in Ski. This parallelism ex-tends to other pairs of forms which use the present stem as 'root' for both formations: thus, dalla and dallvā, to dad, weak-grade present stem used as 'root', like carita and caritvā. The ppp. in ita from aya verbs is historic; i represents the zero grade of ay (IE *ey, as in Lat. moneo). The gerunds are unhistoric in showing -ay- before the ending: kār-ayilvā, a blend of kār-ila and kāray-ali. If the gerund were kār-ilvā, it might be considered historic; such forms are known to BHS but are not ancient inheritances; rather secondary neologisms (see § 35.24).

Gerunds in am

35.3. These are found in BHS as in Skt.; Renou 131 notes the fact, citing Divy 413.8 (vs) *nirīkṣaṃ*. In Skt. and also here, they are specially used at the end of compounds (Renou 130, 131): Mv i.339.13 (prose) ālopa-kāram 'making a morsel, a piece, of it'; i.343.18 samnidhi-kāraṃ 'making a store of it'.

35.4. Instead of indicating an action prior to the action of the main verb, the gerund in -am tends to retain its original function (cf. Renou, BSL. 41(1941).208 fl.) as an adverbial modifier of the main verb, referring to an action contemporary with, or even subsequent to, the main action. Cf. Käš. on Pān. 3.4.32 gospadapram (or °pūraṃ) vrsto devah, 'it rained so as to fill a cow's footstep'. This usage, virtually expressing purpose or result, is found also in BHS: sâli-hāraṃ gamisyāmah Mv i.343.8, 13 (prose), 'let us go rice-gathering (to gather rice)'; sâlihāraṃ galasya 346.8 (prose). Such forms could, if one chose, be called infinitives, cf. § 35.6. Even adjectival forms of corresponding a-stems are similarly used in Skt.: udahāro, 'aquam apportatūrus,' BR s.v. (from Dašak.). Cf. the similar use of aka forms in BHS, § 22.3.

1. Senart, notes on Mv i.229.14-15 and 230.5-6, wrongly supposes that the gerund may take the place of a finite verb, misunderstanding both passages. In i.230.5 delete the punctuation; this sentence goes with the preceding, and the gerund *alikramya* is quite in order; it

35.5. In our dialect, gerunds in *-am* may take the suffix ka, the form appearing as *-akam*; a usage which does not seem to be recorded in standard Skt. Thus, in the vicinity of two of the forms cited above, we find the equivalent forms *ālopa-kārakam* 'making a morsel of it' Mv i.339.16; 344.14 etc. (prose); and *śāli-hārakam galasya* 'when he had gone rice-gathering' Mv i.343.4. Others, § 22.5.

Gerunds in im (?)

35.6. I find it hard to separate $-\hat{salihara}(ka)m$ from the form udakuhārim, repeatedly used in Mv, also with forms of gam, in quite the same way, 'to fetch water'. This form occurs Mv ii.65.10, 17; 109.14; 110.15; iii.156.12; 427.13 (all prose), and ii.226.12 (vs, end of line; read "hārim with 1 ms., the other "hāri, Senart wrongly em. "hārī). Despite a few minor variants there seems no doubt of the form, nor of its essential meaning. I have found no parallel; no noun of action "hāri seems to be recorded otherwise. It would hardly help to understand the form to call it an infinitive, tho the meaning would fit such a description well enough.

Gerunds in lva and ya

35.7. An overall picture of our texts makes desirable a certain dilution of my statement Lang. 13.109: 'The suffixes lvā and ya, in their original forms (not preceded by i), occur almost exclusively in forms taken over directly from Sanskrit, and are almost always distinguished as in Sanskrit: that is, $lv\bar{a}$ is used with simple roots, ya with compound roots'. This is true of the text of SP, at least as presented in our editions, and of some other (especially late) texts. I observed l. c., however, that the distinction 'is less sharp' in some texts, including LV. Of the language as a whole, and particularly of what we may assume to be its oldest form, I should now say that the above distinction holds only as a strong tendency, and in broad outline. There are more exceptions than might be supposed from my earlier statement, particularly in Mv. Compound gerunds in tvā, and simple gerunds in ya, not preceded by i, are certainly less common than the reverse, even in Mv, and much less common-indeed relatively quite rare-in most texts, even in their verse parts. We may reasonably assume, however, that in the oldest form of our language this distinction was breaking down, and that it was later reestablished to a large extent, under the influence of Sanskrit, in BHS tradition.

35.8. Examples of compound gerunds in *tva*, or m.c. *tva* (other than those from *aya* presents, which are fairly frequent even in Skt.): SP, none recorded from text as edited; for 267.8 (prose) *utsrjya*, the Kashgar fragment of La Vallée Poussin JRAS 1911.1074 reads parilyaktod. —

should neither be altered nor interpreted as equivalent to a finite verb.

2. Hardly from dissyllabic roots as suggested by Renou 125; see Lang. l. c. 108 n. 2; even these roots historically show monosyllabic (weak) forms in the gerund, at least from roots containing semivowels. abhinirmilteā 1.V 83.22: abhistuteā 131.19; tripradaksiņikrītoā 253.21: samghāļikrīteā 267.7 (so read with mss.); āvarjanikrīteā 245.14 (? see Corr., and Lang 13.110, top); all prose: (vss) āgateā 109.14; 288.6; nihateā 195.14; etva 197.2; pragrhītea 297.21; 338.12; pratigrhīteā 387.7. – āgateā Mv i.18.11; 29.13 (both prose, et passim, common); sakrīteā gurukrīteā 44.12 (prose); abhijīteā 49.8 (prose); pratigruteā 75.5 (vs); 255.1 (prose) et passim, common); sameteā (sam $+ \bar{a} + i$) 262.17 (vs); uditeā ii.244.17 (mss.; prose); nihateā 328.8 (vs); nirgāteā 460.2; iii.114.6 (both prose); utthiteā iii.392.9 (prose); etc., many other forms. — pratigrhīteā (printed prati gr°) Divy 239.1 (prose).— In Av i.299.8 (prose), read avagateā (ms. iva°; Speyer em. avagatga). — āgateā Kv 70.12 (prose).

35.9. Examples of simple gerunds in ya. The commonest is grhya, which is known in Skt. (§ 35.1). It occurs LV 110.20; 115.20 etc. (Lang. 13.110); Mv i.146.2 = 202.17 = ii.6.14 (vs; the rest prose); i.216.3 = ii.18.18; i.289.11; 321.3, et passim (common); Divy 105.1; 251.15 (prose), et passim; common in many texts.-Others: sthapya (in Skt. Renou 124) Divy 582.24 (prose); sthāpyā, m.c., LV 109.21 (vs); drsya (in Skt. Renou 124) My i.65.1; 67.1; iii.13.16 (vss); pūjya LV 195.10 (vs; read pūjya śres/hām, construing tubhyam as acc.); Lank 4.12; vandya My i.113.4 (vs; mss. vandaya, em. Senart); Lank 4.12 .- Forms which I have recorded each in only one text: hāsya Mv i.244.5 (prose) 'causing to laugh, pleasing' (so mss., Senart em. hasya); iii.88.6 (vs) le gamya nālidūram, 'they, going not far away'. — cārya (to cārayati) Divy 43.20 (prose). labhya Suv 236.9 (vs). — stutya RP 4.7; 7.19; asya 16.1 'throwing, rejecting'; tyajya 22.5; iksya 24.7; 52.1; krtya 39.1 (all vss). - lokya Dbh.g. 44(70).13; 49(75).28; pūrya 52(78).23 (perhaps read so in 52(78).11 for pūrye). yujya Ud v.9 (vs).

35.10. The shortening of $tv\bar{a}$ to tva, and lengthening of ya to $y\bar{a}$, in verses m.c., hardly need extensive illustration; sufficient examples were given Lang. 13.110. A few from texts not included there are: *srutva* RP 5.10, 11; 12.9, 10, etc.; *krtva* 5.11; *drsiva* 12.13; *srutva* Bhad 48; *gatva* 53. And *grhyā*, read so with v.l. (m.c.) Mv ii.299.10, which must be read (with mss.) *bhrngāra grhyā ratanam vicitram*.

35.11. Sanskrit gerunds in ya may appear with epenthesis of *i*, at least in verse: $pr\bar{a}piya = pr\bar{a}pya$ SP 392.4 (vs); sanwarniya = °nya SP 25.7 (vs); ābhujiya LV 133.20 (vs); nyasiya LV 241.12 (vs); and cf. vijahiya § 35.14. From such forms originated the common suffix iya, §§ 35.37 ff.

Present stems as 'roots' before tvā and ya

35.12. Present stems (especially of thematic type) play the rôle of 'roots' before gerund suffixes chiefly in the Mindic forms -ilva, -iya, etc. There are however a few cases where $lv\bar{a}$ and ya, not preceded by *i*, are found attached to present stems, or to blend-forms showing influence of them. Indeed such formations occur even in Skt., e. g. in *bhunktvā*, *-rundhya* (Whitney 991 d, 992 b). From a Mindic present ultharati (root st_{l}) seems to be derived a ger. ultharya Kv 32.17 (printed ultarya; Dict.); from krāmati, parākrāmya Ud xi.1.

35.13. So from nasal infix presents: abhisiñcya Mv iii.353.6 (prose), to abhisiñcali; āchindya Bbh 166.23 (prose), to MIndic and BHS āchindali (= Skt. āchinatli).

35.14. From reduplicating presents: juhulvā Mv iii.426.18 (prose), to juholi; abhinirmimīya 'having fashioned by magic' Bbh 152.2 (prose), to abhinirmimīle (see Dict.); cf. abhinirminīya below; jihrīya 'being ashamed' MSV i.14.16 (prose); vijahya LV 202.6; 242.17; 211.7 (see Lang. 13.110); Šikş 106.11; pravijahya RP 59.6 (all vss); to (pra-)vijahali (= °jahāli, but recorded in Skt.); also (m.c.) vijahiya (10 § 35.42) 1.V 232.22; dada(1) $lv\bar{a}$ (? v.l. dade $lv\bar{a}$) Mv ii.205.6 (vs); if genuine, dadat $lv\bar{a}$ is a curious blend of dat $lv\bar{a}$ with dadati (= dad $\bar{a}t\bar{i}$); cf. $\bar{a}daditv\bar{a}$, § 35.28.

35.15. From ninth-class presents: abhinirminiya 'having fashioned by magle' Mv ii.411.10 (vs); to *abhinirmināti = Pali abhinimmināti; 1 have failed to note other traces of this present stem in our dialect, except in the thematized form (abhi-)nirminati (beside "minoti and "mimite); cf. abhinirmimiya above. Based on the 9th class weak-grade stem of grah are a number of forms like grhnitva, grhniya etc.; instead of n, the mss. and editions often write dental n. The LV (287.15; 288.7, vss) form grhnite \bar{a} was recorded Lang. 13.115, with expression of some doubt, but it is now confirmed by the same form (or with n) Mv ii.432.7 (prose); iii.460.6 (vs) and perhaps 9 (text grhnitvā, v.l. °itvā, vs). Also grhniya Mv ji.249.14 (prose); iii.144.9 (prose); nigrhniya (mss. n) My ii.241.14 (prose); and with Mindic -yana, grhniyana, § 35.48. In My ii.232.15 (vs, meter indifferent) Senart reads grhniug for mss. grhnişya or °şva; grhnitvā or °tva might also be considered.

35.16. From ya-presents: to the present paiyali belongs ger. paiya Nv 1.192.9 (vs; misunderstood by Senart; read with mss. CM mailryās ca; 'as good men [they attained the first fruit], beholding the king [who had attained it], and the great power of [the Buddha's] love'); and vipaiya SP 282.6 (vs). It is as if the ya of the present stem and the ya of the gerund were 'telescoped'. The limitation of Skt. pai-to the present system forbids grouping this with § 35.9.

35.17. Formally in this same category belong a couple of gerunds apparently based on present passive stems in ya. LV 194.10 (vs) utpāţya seems to mean 'being torn up', from pass. utpāţyate; derivation from the active causative utpā[ayati is at least forced and awkward. And in Mv.iii.16.2-3 (prose) read with mss. yāvad... antarpurikāhi antahpuram pravešīya (Senart em. °yati) krīdāpanako bhavisyati, 'until he, having been brought into the harem by the harem women, became (lit. should become) their playfellow'; from Mindic pass. of caus. pravešīyati (°te) 'is caused to enter.'

35.18. A blend of Skt. grhitvā and (prati-)grhya is seen in pratigrhiya LV 240.12 (vs), which might be supposed to show the persistence of the feeling against $tv\bar{a}$ with compound verbs; note that pratigrhitva, with the common shortening of final \bar{a} m.c., would be metrically equivalent. However, uncompounded grhiya is found in the mss. at Mv ii.333.22 (vs), but contrary to metrical demands; Senart em. grhiya, m.c. (perhaps read grahiya? § 35.44). **35.19.** The type of grahaya. Finally must be men-

35.19. The type of grahāya. Finally must be mentioned the gerund grahāya, and one or two forms which seem to be based upon it.⁵It is clearly MIndic, tho found repeatedly in the Harivamás (pw, s.v.). It is probably a Sanskritization of Pali, AMg. and JM. gahāya. With Pischel 591 end (otherwise Renou 128, top), I regard it as belonging to the denominative verb which P. cites as "gahāai, Skt. "grahāyati; it actually exists in Pali as -gahāyati (derived by Geiger 186.5 from Vedic grbhāyati), also "gaheti (Geiger 139.2), and in Ap. as gaheti (Jacobi, San. 629.9). The gerund is virtually a postposition, meaning 'with' (as in Pali etc.): SP 211.6 (prose); Divy 327.14; 369.21; 380.21; 391.27; 393.18; 402.3; 431.19 (all prose); Mmk 676.22; 680.1; 682.20, 27; 698.6 (all prose). C; grahetuē etc., § 35.25, based on the same present.

35.20. Analogical to grahāya is nišrāya (cf. also upanišrāya, Dict.), originally 'dwelling in' or the like, but normally postposition, 'in, at, with, near; because of (see Dict.); SP 12.15 (vs); Mv i.144.3 f. (four times; prose); 277.3 (vs); ii.260.9; 261.10; iii.53.13 (prose); 271.7 (vs); Mvy 6368 (prose); Karmay 161.1 (vs). It = Pali nisādņa,

AMg. nissãe, and is based on a proportional analogy involving Mindic presents in -eii (-ei), and/or past participles in ita, thus: Pali gaheti and ppp. gahita : gahāya (our grahāya) = *nisseti (Pali nissayati) and Pali nissita : x (Pali nissāya, our niśrāya). Geiger 212, end, points out that Pali anvāya is based on anveti (or anvita), on the analogy of gahāya to gaheti. He might have added Pali pațihāya. I have not noted anvāya in our dialect; but Jm 35.5 presents adhiśrāya 'having put on (the fire)', which is clearly an imitation of niśrāya, above.

35.21. Once a gerund vikreya occurs to the present vikreti, Chap. 43, s.v. krī (3).

Gerunds in *ilvā*

35.22. This is the commonert BHS gerund suffix. Like the others, to be treated later, which show *i* before (or use *i* as) the suffix proper, it is essentially MIndic, and the 'root' to which it is affixed is regularly abstracted from the present stem; specifically from a thematic present stem, itself often MIndic. In this respect it parallels most other verb forms in BHS (\S 24.1 ff.). The same is true, to be sure, of the Skt. analogical forms in *ibo*d, which only means that this essentially MIndic process starts, or perhaps better is reflected, in what passes for Skt. Even in the rare cases when no parallel thematic stem has been discovered in BHS or MIndic, we are generally justified in assuming that such a stem must have existed.

35.23. The MIndic character of gorunds in itvā, iya, itvāna, iyāna, and i (i) is strikingly indicated by the absence of any tendency in them to distinguish between simple and compound roots. They are all used equally freely with both. Consequently, when we find compound gerunds in ilva, even if Sanskrit uses the corresponding form in itvd from the simple root, we should regard compound itvā forms as directly made from the corresponding compound presents, not as inheritances of the (simple) Skt. gerund in ilva extended to compounds. Take as examples such forms as vyāyamitvā SP 109.5, pravrajitvā SP 465.3 (both prose). Even in verse, and a fortiori in prose, SP does not seem to use gerunds in *lvā* (not preceded by *i*) from compound roots. Therefore, altho Sanskrit has uncompounded gerunds yamitvā, vrajitvā, I believe the above forms have no real connexion with them, but are MIndic formations based on vyāyamali and pravrajati.

35.24. The like may even be true of gerunds in ayiloā from compound presents in ayai, which are very common: praharşayitoā SP 25.11; paripūrayitoā 152.7; samlarpayitoā Av i.368.11 (prose), etc. Here, however, it is proper to recall that even Sanskrit compound verbs in aya occasionally form gerunds in ayitoā (§ 35.1). Our language also often shows gerunds in *itoā*, minus -ay-, from aya presents, both simple and compound: janitoā LV 234.7; māritoā Mv ii.248.5; prajāapitoā SP 194.3, etc. These are not historic inheritances (cf. § 35.2); like many other forms in which -ay- is lost in such verbs (and conversely forms in which -ay- is inserted unhistorically in other verbs), they are late (MIndic) creations. On the whole matter see Lang. 13.113 with note 4, and §§ 35.43 f.

35.25. We should expect to find contractions of -ayitoā to -etvā, and I noted (l. c.) with surprise that none appeared to occur in SP or LV except shāpetu(a) SP 91.8 = Pali thapetvā 'except' (much more commonly shāpayitvā, see Dict. s.v.). I have now found others in Mv, but not many; and none in other texts. Quite common in Mv is graheivā, to *grahāyati, *graheti (above, § 35.19) i.236.6; 297.20; ii.317.9 ff.; 320.9; 322.9, 12; 329.13; etc. In most of these occurs a v.l., either grhitvā or grhetvā (the latter is even put into the text ii.322.12 by Senart); but there is no v.l. at (e. g.) i.293.22; ii.342.6; iii.89.5; 111.1. Others, from aya presents, arc vijodhetvā Mv i.4.7; nišāmetvā

(= $nis\bar{a}mya$) i.151.16 = 225.16 = ii.28.17; pravicāretvā ii.111.16; okrāmetvā iii.318.15, to avakrāmayati LV 400.14 (non-caus.). An 'aya present' (the not of the Skt. 10th class) is nayati, also; and *netvā* occurs Mv ii.168.8, samānetvā i.185.1, tho in both there is a v.l. (samā-)nitvā. Similar forms under -*lvāna*, below.

35.26. But even where no present in *-ayati* or *-eti* is regular, or even known to occur, there are occasionally gerunds in *-etvā*; these are doubtless to be associated with the spread of presents in *-eti* (for *-ati*, § 38.2). I have noted uddharetvā Mv i.7.14, to uddharati (which I have assigned to root hr; cf. pres. āhareti; in any case not to ud-dhārayati!); niksipetvā il.65.4 (so mss., Senart em. °itvā), to niksipali; and pragrhņetvā ili.131.4 (v.l. °grhītvā), to (BHS) pragrhņati (an impv. -grhņetni is found). All these are from prose. The last may have been specially influenced by grahetvā, as if a blend of this and grhņītvā. And, according to one variant, from an aya present but retaining the ay-, prali-jāgarayetvā (v.l. °garitvā) Mv i.324.8 (prose).

35.27. In Lang. 13.114 ff? I presented what I believed to be a full list of roots forming MIndic gerunds in *itvā* (also *iya* etc.) found in SP and LV, so far as they do not occur in Skt. (except the common forms from *aya* presents); included were some *itvā* gerunds that occur in Epic or other limited departments of 'Sanskrit'. Many of these, and others like them, are common in other texts, especially Mv, and in the verses (mostly) of such texts as Suv, Gv, Bhad, Samādh, RP; also sometimes in prose, especially that of KP (as well as Mv), and in the Kashgar fragments found in Hoernle MR. Note e. g. *pratisțhitivă* MR 180.12 (prose), for Vaj 21.9 *pratisțhitena; abhisamharitvā* MR 191.21 (prose), for Vaj 42.3 *abhisamhriya*.

35.28. The following is a brief selected supplement to that list, with notation of the corresponding thematic presents (often MIndic). I have taken pains to include specially interesting or problematic cases, besides enough normal cases to illustrate standard practice. References to other than standard Skt. presents either will be given below, or may be found in other parts of this grammar which are cited, or in Chap. 43; often also in my Dictionary (here sometimes referred to, by D), and in the familiar Pali, Prakrit, and Ardhamágadhi lexicons.

Root āp: (paryāpuņati D) paryāpuņitvā Samādh 8.29 (vs).

kr 'make': (karati) vyākaritvā Mv i.357.9 (prose).

kr 'scatter': prakiritvā Mv i.211.20 (prose).

krs 'drag': apakarsilvā Mv ii.75.5 (prose).

kri: (krinali; § 28.5) krinilvā Mv 1.335.10 (so mss., Senart em. kri^o).

khid: (khijjati, °te D = khidyate) khijjitvā Mv ii.252.14 (prose).

gam: (gacchati) gacchitvā Mv iii.142.16 (prose); (gamati, Skt. agamat) āgamitvā Gv 190.4 (prose).

grah: (grhņati) grhņitoā Mv iii.144.16 (prose); pratigrhņitoā Mv iii.304.17 (prose).

ghrā: (jighrati) upajighritvā Mv i.353.8; ii.137.8; 139.4 (all prose); (ghrāyati) ghrāyitvā Mv ii.251.10; 295.12; iii.52.10.

ci: (vicinati) vicinitva Mv ii.465.10 (prose); (Pali -ceti, opt. niccheyya; Pkt. cei = cayati) niścayitva SP 48.6 (? WT with K' niśrayitva).

chid: (chindati) chinditvā Mv ii.236.3; Suv 63.5 (vs); (*chinati, § 28.12 and Chap. 43, s.v. chid 2; presupposed by) chinitvā LV 195.12: 196.2, and apr. chini.

by) chinilvā LV 195.12; 196.2, and aor. chini. jāgr: (jāgarati) pralijāgarilvā Mv iii.141.15 (prose).

ji: (jayali) jayilvā Mv 1.264.6; (jinali) abhinirjinilvā Mv i.52.9 (prose; two inferior mss. °jilvā; in the parallel i.49.8 Senart reads abhijilvā, most mss. °jilvā); a-jinilvā Mv ii.315.13.

ju: (anu-javati Pali and BHS, Mv ii.212.17) anujavitvā Mv i.321.6 (prose). dā: (duduli) ādadilvā (so divide) SP 283.9; (samādiyali D) samādiyilvā Mv i.128.9 (prose).

dhā: (dadhali) abhiśraddadhilvā SP 80.5 (prose); (Pali dahali) dahilvā Mv ii.377.6 (vs); (praņidheli D) praņidhilvā Mv i.96.10 (prose), as mārilvā etc. to mārayali

pragidhilvā Mv i.96.10 (prose), as mārilvā etc. to mārayali (māreli) etc., § 35.24; or to °dhali? Chap. 43, s.v. dhā (5, 6). dhu: (dhunali) apadhunilvā Mv i.321.6 (prose).

nī: (nayali) ānayilpā Mv li.211.15 (prose). In the same line occurs āneti, which has led to ger. nelvā (§ 35.25). pā: (pibali) pibilvā Mv li.96.19; ili.144.14 (both prose).

budh: (budhyale, 'li) budhyilva Gv 315.20 (vs); also in some mss. of Mv for next in some passages; (buddhati, § 28.19) buddhilvā or °lva Mv il.323.19; 354.4; iil.273.1, etc., common, and often without v.l.: vibuddhilvā iii.191.12; abhisambuddhilvā Mv iii.111.17: 272.18, etc.; in ii.419.1; iii.246.3 and elsewhere v.l. °budhy°, sometimes adopted in text by Senart; in iii.114.10 mss. °buddhilvā or °bodhilvā (to bodhati, °le), Senart em. °budhilvā, wrongly. These forms must be evaluated in connexion with other forms which point to the present buddhati. In Mv it seems that the mss. rarely give clear support to ger. (-)budhyilvā, but often are unanimous for (-)buddhilvā.

bhid: (bhindali) bhinditvā Mv ii.239.16 (vs).

bhuj 'enjoy': (bhuñjali) bhuñjilvā Mvi.306.19; ii.263.17; 300.2 (vss), etc.; (bhujali?) bhujilva LV 74.6 (vs); see Chap. 43, s.v. 1 bhuj (2).

bhuj 'bend': (dbhujadi, with paryaṅkam) dbhujitva LV 259.5. But in Mv usually dbhumjitva, i.144.11; 213.8 = ii.16.12 (mss. corrupt in i.213.8; in ii.16.12 v.l. abhujitva); ii.131.15; 268.4 (Senart with v.l. dbhujitva); 313.14, all prose. See Chap. 43, s.v. 2 bhuj.

man: (manyate) avamanyitvä Mv i.309.11 (prose).

mā, mi: (nirmiņali) nīrmiņilvā Mv i.227.7 (prose); iii.282.15 (prose), cl. §§ 35.14, 15.

yuj: (*yujati? yuñjati) an-anuyujilvā, printed at Mv iii.160.6; but later, 165.11, referring back to 160.6, ananuyunjilvā; and so 170.7. All prose, no v.l. noted. It seems probable that *yujilvā in 160.6 is a mere misprint (loss of anusvāra) for *yunj° (*yuňj°), to yuñjati. ric: (riñcati) riñciivā KP 90.3 (prose); and doubtless

ric: (riñcati) riñcitvá KP 90.3 (prose); and doubtless read so for ricitvá RP 34.12 (prose; as in the preceding case, anusvára lost by misprint or corruption).

rudh: (rundhali) orundhilvā Mv ii.75.10 (prose).

viş: (parivişali D) parivişildü (so I would read; Senart "silvü; mss. "śilvü) Mv iii.145.12 (prose); in iii.130.12 Senart "vişildü (mss. "viś").

 v_{f} : (prāvarati) prāvaritvā Suv 69.12; 102.5 (both prose). In the second, no v.l.; in the first, most mss. prāvriya but the best prāvaritvā. This illustrates the way in which original MIndic forms are eliminated in the course of transmission.

vyadh: (vidhyali, MIndic vijjhali) viddhilvā (?) Mv i.12.5 (vs). So text with 3 mss.; 2 mss. viddhalvā; one (inferior) ms. vichilvā (intending vijhilvā, vijjhilvā?); same verse vijjhilvā Mv iii.456.11; cf. apavijjhiyāna, § 35.47. Senart suggests that viddhilvā is 'based on the analogy of the past pple. viddha'; perhaps rightly, cf. § 28.19. Yet I suspect that vidhyilvā or vijjhilvā should be read.

sio; (*sivati?) sivilvā Mv ili.313.7 (prose; mss. šivilvā).

sthā: (lişthadi) lişthilvā SP 253.9; (sthihati) sthihilvā very common in vss of most texts (e. g. Bhad 45, Gv 258.1, Samādh 8.22) and prose of Mv (e. g. ii.294.2; iii.42.16); (-sthahati) abhyutthahilva (mss. abhyustha[°]) Mv ii.38.14.

snā: (Epic Skt. snāyale) snāyitvā Mv i.4.6; iii.313.8 (both prose).

Gerunds in (i)tvāna (°tvanā, °tvana)

35.29. In principle, little needs to be changed in my statement on this form, based on SP and LV, in Lang.

13.117 f. In the form *lvāna* it seems to exist only in Pali (Geiger 208); and the (m.c.) variants *lvanā*, *lvana* are nowhere recorded. We may compare the suffix *lvānam*, quoted by Skt. grammarians as Vedic (Whitney 993 c) but not recorded in literature. AMg. also has *llāņam* and m.c. *llāņa* (Pischel 583); likewise (*l)ūņa(m*), (in this case in free variation with or without final nasalization.) as in general MIndic including Pali (*lūna*, Geiger 211). Our dialect shows no form with final nasal, and probably no *lūna* (one or two doubtful cases below, § 35.36). When meter requires a long final, a is lengthened. The penultimate ā may be shortened m.c., so that in verses *tvanā* and *lvana* are found (no **lvānā*, doubtless by accident).

35.30. Not a single occurrence has been found in prose except that of Mv. and even there it is extremely rare. I have noted only drstvāna i.3.13; kledayitvāna iii.7.19; samnipatilvāna 42.10; veditvāna 61.11 (so read with mss., to vedayati; Senart em. vadi°); viditvāna 71.6; galvāna 176.16. While I may have missed some others, certainly there is a striking and unusual difference between the prose and verses of Mv on this point. For in Mv verses it is quite common; I have counted over 150 instances, without claiming completeness.

35.31. Even in the verses of most texts it is not too common, but SP and LV have not far from a score each (listed l. c.). Others (not a complete list) are prawrajitwāna Samādh 8.29; drstvāna Suv 238.3; krtvāna Sukh 54.1; śrutvāna Karmav 161.3. And tvana: karitvana Samādh 22.4; nihanitvana 213.14; cyavitvana 215.5; spharitvana Gv 213.11; jinitvana 213.14; cyavitvana 215.5; spharitvana Gv 213.12, Also tvanā: karitvanā Gv 34.8; spharitvanā 34.11, 23; 240.22; 241.6; in LV 54.15 (vs) read with v.l. upasthapitvanā for text "pisva nā. The forms tvanā and tvana are used only m.c.; but it is remarkable that they seem not to occur at all in the verses of Mv, where tvāna is so common.

35.32. That the suffix is analyzable as an extension in *na* of Skt. *lvā* is obvious, and is confirmed by the fact that a sizable minority of the instances consist of regular Skt. gerunds in *lvā* (not preceded by *i*), thus extended. Examples from SP and LV were cited l. c.; others are cited above; we may add from Mv (vss) *chiltvāna* ii.357.15; *da(t)tvāna* 365.18, 20; *labdhvāna* 370.3; *pitvāna* iii.292.12; etc.

35.33. Considerably more numerous are gerunds of MIndic type in *ilvāna*, based regularly on thematic presents, like those in *ilvān*, § 35.28. Examples from SP and LV were given Lang. 13.118. In addition to those mentioned in §§ 35.30, 31, a few of the many found in Mv (vss) may be noted: bharitvāna i.216.6; karitvāna 268.6; vijahitvāna 268.21; hanitvāna ii.223.12; fruņitvāna 227.14; dahitvāna (to dahati = dadhāti = dadhāti) 365.6; ośiritvāna (avaširati D) 367.22; sthihitvāna iii.45.6 (cf. sthihitvā, above); abhiruhitvāna 289.1, etc.

35.34. Like the *-etoā* forms (§ 35.25), forms also occur in *-etoāna*, from aya-presents, contracted in MIndic fashion (ayi to e), or based directly on *-e*-stems. The commonest is grahetvāna, parallel to grahetvā, q.v., l. c.: Mv i.226.10 = il.29.13; ii.34.20; 39.1; 86.12 etc. (at least a dozen times; also pragrahetvāna i.252.3, v.l. pragrhitvāna). Also samjanetvāna ii.205.1; janetvāna ii.318.13 (but janayitvāna iii.77.8); udvejetvāna ii.77.8 (same line as preceding); pranāmetvāna iii.97.14 (so Senart em., mss. pranetvāna, impossible in meter and sense); ogāhetvāna iii.289.1 (Pali ogāheti, with ger. ogāhetvā, = ogāhati = Skt. avagāhati); upalobhetvāna iii.294.15. In Mv i.117.7 mss. saminivetvāna seems corruptly to represent some form of this sort, but what? (Senart em. saminiveseti, which is too violent; saminivešetvāna is metrically impossible.)

35.35. Of quite different nature is bhellvana Mv ii.12.16, from bhid, 'having split'. It is clearly Mindic, and corresponds to Pali bhetvā, AMg. bhettā (Geiger 209, Pischel 582; there is also chetvā, chettā from chid); an explanation is suggested for chetvā and related forms, which could equally apply to bhetvāna, in § 28.50. Cf. bhi(t)tvāna Mv i.11.8, to Skt. bhittvā.

Gerunds in tūna, m.c. tuna (?)

35.36. It is surprising not to find in our dialect the common MIndic suffix represented by Pali tāna, Pkt. ($l\bar{\lambda}aa(m)$). But no reliable case seems to occur. The nearest approach to one is Mv ii.12.17, an āryā verse, in a bad state of preservation, where the mss. read *śrupiluna* (v.l. ^onu), which is metrically as good as Senart's em. *śrupilva*, and might stand for **śrupilūna*, m.c. In Mv i.191.2 Senart would read paśyilāna 'having seen'; mss. le paśyi nāna (one good ms. nūnam), which might be interpreted as (a)paśyi, 3 sg. aor. as 3 pl., 'they saw', followed by the particle nūna(m).

Gerunds in iya (iyā, ia)

35.37. This suffix is obviously only a Mindic form of Skt. ya; it appears in Pali as iya (Geiger 213), and in Pkt. as *ia*, *iya* (Pischel 590 f.). It originates, of course, from Skt. ya with epenthetic *i* (cf. § 35.11).

35.38. The spelling *ia* has been noted twice in Siks, 299.8 karia and 343.18 parivāria. The difference between this and *iya* may be little more than orthographic in our texts.

35.39. Like other MIndic gerund suffixes, this is attached as freely to simple bases as to compounds; and the base is always a thematic present stem (often MIndic), minus the final a. As with ya (§ 35.10), the final a may be lengthened, usually where meter requires a long, but in this case also once or twice in metrically indifferent positions: thus at the end of padas in SP 59.5 vijāniyā, or even of lines, SP 125.8 (ed. vijāniya but Kashgar rec. and one Nep. ms. "yā). According to Pischel 590, Pkt. $i(y)\bar{a}$ occurs only 'in verses', that is presumably m.c., altho AMg. (ib. 587) has a gerund suffix $cc\bar{a}$ (Skt. * $ty\bar{a}$) with regular a, doubtless influenced by the vocalism of AMg. ttā (Skt. tvā). It is possible that the ā in iyā forms is similarly influenced by that of itvā; cf. iyāna (§ 35.45), obviously based on iya (iya) on the analogy of itvana to itvā. I agree with Pischel in doubting any connexion of such forms with the Vedic gerunds in ya, but am not convinced by his view that to became cc (thru *ty) by a phonetic process in AMg. cca.

35.40. In Lang. 13.119 are cited iyd forms from LV, all with d m.c. (*jiniyâ*, *bhaviyâ*, *avaširiyâ*, *janiyâ*, etc.). Others are kşipiyâ Mv ii.391.1; samjaniyâ Gv 301.5; samciyâ Dbh.g. 50(76).17 (cf. § 35.44).

35.41. In the same place are also cited examples of regular *iya* from SP and LV. Here a few others will be added, mostly from Mv, where they are common in prose as well as verse; our examples will be chosen from prose of Mv except when stated to be from verses. All examples from other texts are from verses. As was noted 1. c., *iya* seems to be specially common with *aya* presents, which will be listed separately. In most cases it will not be necessary to quote the thematic present which serves as base for the form.

35.42. From a (not aya) presents: paśyiya Mv 1.312.14; ii.170.17; praviśiya i.361.5; gacchiya 362.13; äruhiya ii.90.7; paribhuñjiya 96.18; parilchiya 194.5 (vs) (to lehali, § 28.40); kriņiya 241.17 (root krī, see § 35.28); nidahiya 320.16 (vs) (so with mss., Senart em. °hya; Pali nidahali = nidadhāti); prāvariya 441.14; aņviya (aņvali Dict.) iii.145.1. — otariya (avalarati) Gv 286. 2. — pramuñciya Šikş 327.6. — labhiya Dbh.g. 8(344).15. — vidhuniya (dhunali, § 28.18) Sukh 23.8. 35.43. From aya presents: arūpiya (arūpayali Dict.) Mv i.352.20; sabdāpiya il.78.3 (so read with one ms.); avalokiya 164.12; nivuļtāpiya 172.7; lambāviya (so with mss.) 172.14; niskāsiya 174.11; anvesiya 179.2; posiya 241.15; slhapiya 244.11. — janiya Sukh 23.7. — vinivartiya Bhad 57. — vidaršiya Šiks 329.3.

35.44. With these belong a few special cases like grahiya (LV 297.3 etc.; Mv ii.317.5, vs) to *grahāyati, *graheti, see § 35.19; samciyā (§ 35.40) Dbh.g. 50(76).17, to Pali samceti (not directly connected with RV cayal(i) or ceti, but with a MIndic ceti for cinoti, see § 28.48, Geiger 131, and cf. Pali nicchiya to pres. niccheti, Geiger 213); apaniya Sukh 23.9 (to Pali apaneti = apanayati, Geiger 131; however, this gerund might be m.c. for apaniya).

Gerunds in iyāna

35.45. As in Pali (Geiger 214) and AMg. (Pischel 592), this MIndic suffix occurs, tho not very commonly, in our dialect. Pischel correctly explains it as a case of proportional analogy; $(i)to\hat{a} : (i)to\hat{a}na = (i)ya$, metrically (and otherwise? § 35.39) $iy\hat{a}$? $iy\hat{a}na$. Since the forms are relatively rare, I shall list all that I have recorded, only referring to Lang. 13.119 f. for the occurrences in SP and LV (17 in all, from 10 different roots; note that Siks 353.3, quoting SP 283.12, confirms WT's reading nişidiyāna, adopted by me l. c.). I have found hardly any instances outside of SP, LV, and Mv.

35.46. The forms practically always contain the stem of a thematic present plus iyana. The chief exception I have found is (varnan) udiryana (jinasya) Gv 255.9 (vs), 'having uttered' (so 2d ed.; 1st ed. prints udirya na, but na as a separate word makes no possible sense; *udiriyana would spoil the meter). We seem here, for once, to have -na added to a regular Skt. gerund (udirya, to udirayati); in other words, a form in -yana (not preceded by i), comparable to the forms in -tvana noted § 35.32.

35.47. Others, of what may be called the regular type: Mv (prose except those marked vs) vijahiyāna i.208.4 (here mss. °hyāna) = ii.12.13 (vs); upagrahiyāņa (to *-graheti, § 35.19) i.227.16; ksipiyān(a) 241.2 (vs); samikšiyāņa 278.20 (vs); ujjhiyāna ii.13.2 (vs); omuňciyāna 73.1; apavijjhiyāna 104.4 (vs; MIndic, to apavidhyati); śabdāpiyāna 173.1 (v.l. °vijāna, Prakritic; this latter, without v.l., occurs ii.453.13 and iii.287.16); supiyāna ii.201.19 (vs); paiyiyāna 249.13; anāprechiyāna 272.12; kariyāna 388.4 (vs); dadiyāna 388.8 (vs, repeated in 12 etc.); grhaiyāna 452.13; nirdhāviyāna 471.19; 475.6; veļhiyāna 485.14; bhuājiyāna iii.120.18 (vs). — In Sukh 53.3

35.48. In Mv ii.241.16; 492.6 grhniyāna has i and belongs with grhnibā, grhniya, § 35.15. Contrast grhniyāna in ii.452.13, above.

Gerunds in *i*, *i*

35.49. This suffix has been recorded only in Ap.; see Hemacandra 4.439, Pischel 594, Jacobi, Bhavisat-takahā 39, Sanatkumāracarita 27, Alsdorf, Kumārapālapratibodha 39. Jacobi does not recognize it as actually occurring in San. but notes five cases in Bhav., and there is at least one case in San., suni 445.5 ('having heard that the sun, the friend of the world, had gone to rest'); and Alsdorf quotes three forms — kari, gacchi, suni — from the Kum. The forms of our dialect are, as we should expect, always based on thematic present stems. The extraordinary ambiguity of the ending i(i) makes it often difficult to be sure of the gerundial nature of the forms. Sometimes they may be interpreted as aorists, as optatives, or even as noun forms (nom. or acc. sg. or pl. of *i*, *i*, or *i* n stems). Nevertheless in some cases it seems scarcely

possible to take them otherwise than as gerunds. I have assembled here some of the most convincing cases, avoiding most of those which could easily be explained otherwise. My list is therefore probably far from complete, since there are not a few others where the interpretation as gerund would be as plausible as any other, tho possibly not more so. I derive the ending from ya quite simply by samprasāraņa, cf. § 3.115. Otherwise Pischel 594, and yet more implausibly Gray, BSOS 8.575. The forms occur practically only in verses, and in most cases meter determines the quantity of the final vowel. There are very few occurrences in metrically indifferent positions (e. g. at the end of padas in SP), and the ambiguity of the form makes it unsafe to argue from them as to the original quantity, since there are few cases where it would be absolutely impossible to assume some other verb or noun form.

35.50. All the examples listed below are from verses except a single prose case from Mv (i.3.3), where, by the way, clear cases are extremely rare, even in verses.

abhyokiri SP 228.15 (here might be taken as optative); (devā 'bhihananti tūryān mandāravarşam ca visarjayanti,) mamaiva abhyokiri...325.4 'sprinkling me myself...'.

stāpāms ca kārāpayi nirvrtānām puspehi gandhehi ca pājayişyati SP 152.5: 'and after causing stūpas to be made for them when they have entered nirvāņa, he will honor them' etc.

upasamkrami SP 11.11; 191.1. In the latter passage, the word occurs in a clause that is bracketed between two precisely parallel clauses, both of which contain gerunds (viditva, and abhyokiriyāna; for the latter, to be sure, Kashgar rec. reads abhyokiri, but this would also best be taken as a gerund).

punyamtato bahutaru tasya hī(so,m.c) syāt, paramparātah śruni ekagāthām, anumodi vā... SP 351.11-12: 'greater than that would be the merit of him, having heard a single gāthā from line of succession (of teachers), or having joyfully received it...' The forms in *i* would have no subject if taken as finite verbs; and in the next stanza *srutoā* is used in a closely parallel phrase.

varşi LV 294.5. Seems clearly parallel to sphuritva in line 4: 'having spread abroad (or, 'filled'; sphurati has this meaning in BHS) the cloud of the law... and having rained the good law... he will cut off...' This appears to be confirmed by Tib., which has char pa hbebs, exactly parallel to khyab = sphuritva in the preceding line; Foucaux renders by a future, but the future of hbebs pa is dbab according to Jäschke, and it seems that both khyab and hbebs must be understood as gerunds, lacking the 'ending' as often in gäthäs. na uttari prārthayi nāpi cintayī SP 213.10, perhaps:

na uttari prārthayi nāpi cintayi SP 213.10, perhaps: '(for we were satisfied with mere nirvāņa, or a small part of nirvāņa,) not asking for, nor even thinking of, anything further.' But these forms might be taken as aorists (3 sg. with 1 pl. subject), and apparently Tib. so understands them (sems so = cintayl).

lajji hirotrapatta (ablative) muninah prapatisu carane (read so with vv.ll. and meter) LV 329.13: 'being ashamed, thru modesty and shame they fell at the muni's feet.' Confirmed in Tib.

tyaji LV 362.7. — vayam api anušikši tasyā mune(r) vīryasthāmodgalam, kšipra bhavema . . . lokottamā LV 422.5; see Dict. s.v. anušikšati.

samirāvayi SP 384.1 (parallel with pres. pple. sahantu, 'having spoken...enduring...'); Tib. renders by gerund in na. Interpretation as aorist would be less plausible.

(te tam stuvanti gunabhûta yathārthadarši,) dhyāyī gunām vigataklešatamonudasya LV 116.10. — upanayi LV 166.10; might be 2 sg. aor., but better as gerund; Tib. khrid de, gerund. — smari LV 167.2, as prec.; Tib. bsams te, gerund. — codayi LV 169.22; better as gerund than as 3 sg. aor. with pl. subject; in the same line ravisu has 3 pl. form, and *codayi* is rather dependent upon it than parallel with it.

magadhapuri praveśi (§ 38.23) lokanātho LV 243.11; followed by parallel phrases containing gerunds, vihariya, and arthu kariya; then upagami, the aorist main verbi: 'the Lord of the World, having entered . . ., having dwelt there, . . . having done the welfare of . . ., went to . . .' Tib. zhugs te, gerund, 'having entered'. Tho based on the causative form pravefayali (Pali paveseti), it has the meaning of the simple verb, as not infrequently in our dialect; metrical requirements may be concerned.

Irst tanum ca kari chindali putrasneham LV 200.9, 'and diminishing (controlling, cf. Fali tanu-karana) his desire, he cut off his love for his son' (i. e. violated it, acted in spite of it). Tib. (misrendered by Foucaux) sred pa bsrabs te (= tanu-kr, cf. Mvy 1145) bu la chags pa yons su gcod (essentially = my rendering).

upāgami Mv i.3.3 (prose), 'having approached'. So mss.; Senart em. upāgamiya. Seems clearly gerund; parallel with drslpa. — vimoksayi Mv i.72.14. — śruni Mv i.207.13 = ii.12.3; so doubtless read both times (with mss. ii.12.3; at i.207.13 mss. srnu, which is senseless). Senart has different emendations, both violent and implausible, in the two passages; he understands a gerund.

kurvi Mv i.236.12 (vs; mss. kurva, kurve, em. Senart; could be aor., but in same line i.241.6 mss. krtvā, Senart m.c. krtva). — avakiri Mv ii.333.11: nabhe sthinitvā avakiri puspacārnaih, bodhisattvam pramudita kiranti (mss., Senart em. okiranti), 'standing in the sky and sprinkling with flower-dust, they joyously besprinkled the B.' — pravarşi Mv ii.413.17; so mss., apparently ger., dependent on udīrayanti; Senart em. pravarşensu. The meter is incomprehensible to me (as to Senart, see his note on 412.18).

35.51. Šikş contains (quoted from Ratnolkādhāraņī) a number of probable cases. Quite certain seems pramuñei 327.14, evidently m.c., corresponding to gerund pramuñciya in closely parallel lines 6, 8, 10, 12; an additional syllable in 14 results in the shortening to pramuñei. (This is recognized by Bendall.) The same form 334.1 (by em., mss. pramuñciya, unmetrical); 334.7; 335.3; and pramuñci (at the end of a dodhaka pāda) 334.16; 335.9, 13; 336.7. (But some of these could be aorists, like osari 336.3, 15.) In the same passage the following seem to me probably gerunds: daršayi 'showing' 328.17; dešayi 'preaching' 329.6; ācari 'doing' 330.3; nidaršayi 330.8; 333.17; vidarśayi 333.17; anusmari 339.2; samgrahi 340.5; samstari 342.5; bhawī 345.13 (I may be m.c.: eka bhawī bahu bionti; cf. next line, bhūtva bahuḥ puna eka bhavitvā dhyāyati, etc.).

paíyi Dbh.g. 4(340).19: labdhvā samādhišata buddhašatam ca pašyi, 'having won a hundred samādhis and seen a hundred Buddhas.' A certain case. daršayi Gv 240.26; 253.15. The latter reads: (pra-

daršayi Gv 240.26; 253.15. The latter reads: (prapañcarala) bâlân dharmasvabhāva daršayi vinesi, 'thou trainest simpletons, showing them the true nature of the Dharma.'

vindi Gv 485.23 tușți vindi, both edd., 'finding satisfaction'. But possibly 2 sg. opt., 'mayst thou find', or read vinda, impv., with 487.10 in 2d ed. (1st ed. corruptly vandi).

Gerunds in tā(?)

35.52. In AMg., and also in JŚ. and JM. (Pischel 582), we frequently find the ending $lt\tilde{a}$ (after consonants $lt\tilde{a}$, as gantā) as phonetic representative of Skt. $lv\tilde{a}$. No $lt\tilde{a}$ is recorded in Pali, and cases of $l\tilde{a}$, or other forms lacking -v-, after consonants are rare; but manlā occurs at least once, and laddhā for labdhvā is more frequent (Geiger 209). A half-Sanskritized form of this laddhā may be found in Mv ii.334.2, 4, 10, 18; 335.2, 6, 10 (vss), where the mss. read unanimously labdhā. It is certainly a gerund; Senart em. labdhvā, perhaps rightly (but is it not strange

that the same corruption occurs seven times?). Again, in Mv iii.325.11 (prose), the mss. read vustā, which may be a half-Sanskritized form of a Mindic *vulthā or *vulthā, representing Skt. ustvā; but Senart em. vusto (past pple. of the same verb).

35.53. It is possible, perhaps even likely, that in Suv 36.11 (vs) we should read *lulella* instead of Nobel's en. *lulelva*; the best ms. (G) reads *lulellā* (since meter demands short final, Nobel is right in shortening \bar{a} to a). I know of no other case where the AMg, ending *ltā* may be as plausibly assumed; the lack of clear support elsewhere leaves me in doubt. But see Lang. 13.122 for *krlā*, possibly m.c. for **krltā*, in LV 356.6 pūjām krtā vrajala ksetra svaka-svakāni, 'having performed homage, go to your several fields.' Tib. seems to support this: mchod pa byas kyis (lit. 'by reason of homage performed') rań raň dag gi zhin du deň.

Gerunds in lu(m)

35.54. In AMg. and JM. forms in *-lum* (Pischel 576), perhaps also in *-(l)lu* (? ib. 577), are used in the mg. of gerunds; *-lu* is regular in Aśokan. The usage occurs, if very rarely, in BHS. A quite clear case is LV 245.19 (prose) (*lac chramanena*) Gautamenālpakrchrenādhigantum sāk-sātkrlam, 'this by the monk Gautama, after comprehending it with little difficulty, has been manifested.' So Tib., khoń du chud ciń (ger., 'having thoroughly understood')

mnon du byas la ('manifested'); note that forms of adhigacchali and sākşātkaroli, or its equivalent (see Dict. s.vv. sākşā-, sākşī-karoti) are elsewhere parallel; also in Pali (e. g. SN iv.337.16-17). 35.55. On two cases (SP 330.10 labdhu, LV 74.3

35.55. On two cases (SP 330.10 labdhu, LV 74.3 bhuklu) which have been wrongly interpreted as gerunds in tu for lvā, see Lang. 13.122; the forms are participles, not gerunds. As I said l.c., tu for tvā (by samprasāraņa) is phonetically as possible as i for ya, § 35.49; and, in verses, tu might stand for tum (prec.) m.c. But I doubt whether any such form is to be recognized. In Divy 476.18 (prose) text reads (sacen mana...) nayanam grhilvā ulpāļayitu punar muñca; no v.l. reported; the passage seems certainly corrupt; there is no verb with sacet, unless ulpāļayitu or the impv. muñca conceals one. Lines 13 and 15, just above, and closely parallel, suggest that we should read ulpāļayilvā and muñcasi. In Suv 36.14 (vs) Nobel reads gaņetu, with some mss.; there is a v.l. gaṇailva, intending ganetva; a gerund seems required, and the final syllable must be short.

Gerund (falsely alleged) in dhya

35.56. Certainly baseless is Max Müller's suggestion (p. xvi) that *ārabhidhya*, reported as the reading of the mss. at Sukh 8.17 (vs), contains a gerund suffix *dhya* related to Vedic *dhyai*. Read *ārabhisya*, 1 sg. future.

36. Infinitive

36.1. The standard infinitive of our dialect has the ending *itum*, added to thematic present stems minus the vowel *a*. Chiefly in verses, the final *m* to (*itum* may be lost. There seem to be cases of *atum* instead of *itum*; and *etum* occurs, chiefly from presents in -e-. The Vedic and Pali ending *tave* (also *tavai*) occurs, but rarely. Commoner are dative forms of other noun stems, especially *anāya* (*anāye*). Other non-Sanskrit infinitive forms are miscelaneous and scattering. For the inf. form in the sense of a gerund see § 35.54.

Ending itum

36.2. The favorite and indefinitely productive infinitive suffix is *ilum*, added as usual to thematic present stems (sometimes stems in *-e*- or *-aya-*). It is found more or less all over, even in prose, including that of other texts than Mv. Most of our examples are prose; those from verses are so marked.

Root ap: paryapunitum KP 139.9.

i: adhiyilum Mv ii.77.13, to Pali adhiyali. There is a v.l. adhilum, probably only a corruption; but Senart reads so in a subsequent occurrence in the same line, where both his mss. read adhilum; this also I think is a mere corruption for adhiyilum. Cf. however § 36.9.

iș: praticchitum Mv i.310.4.

kr 'make': samskaritum Divy 207.24.

krī: vikriņitum Mv ii.87.15 (v.l. vikrīņītum). Senart reads kriņitum ii.88.6 (vs), but the true reading is vikretum (§ 36.8; mss. ketum).

kşi 'destroy': *kşayilu* (for °*lum*, m.c., § 36.4) Dbh.g. 41(67).8.

kşip: nikşipitum Mv iii.428.4.

gam: gacchitum Samādh p. 30, line 31.

gā 'sing': gāyitum MSV iv.240.22.

grah: nigrhnitum Mv iii.393.5, 10.

ji: parājayilum Divy 285.20, parājayilu-kāma Suv 94.17.

jus: jusilu-kāma Mvy 6345.

 $j\hbar a$; jānitum SP 31.10; ājānitum Mv i.316.18; iii.318.10; iānitum Mv ii.379.14 (vs); iii.34.1 (see § 36.3); Samādh 22.13, 45 (vss); prajānitum id. 22.43 (vs); vijānitum Vaj 43.9 (vs).

dī: uddīyilum Mv ii.243.6.

lyaj: lyajilum Mv iii.151.12, 19; Divy 416.26 (vs); parityajilum Mv iii.166.4.

diś: vyavadiśitum Mv iii.50.8.

dru: abhidravitum MSV ii.34.4.

dhā: śraddadhitum Mv i.185.11 (vs); antarahāyitum Dict.) Mv ii.97.10, 20; (uttaram) samdhitum 'to concoct make, a reply)' Mv iii.390.6 (to samdheti, or *samdhati; Chap. 43, s.v. dhā 6 or 5).

dhr: dharitum LV 29.9 (vs).

dhyā: avadhyāyitum Divy 494.8; Av ii.139.8 (the atter by em., see Speyer's note).

pad: upapadyitum Mv iii.404.8.

paś: paśyitu-kāma Mv ii.446.4, 7; 448.9; paśyitum Mv ii.10.9; Samādh 22.17 (vs).

pā: pivitum LV 309.15 (vs).

prech: prechitu-kāma Mv iii.366.10 (vs); prechitum Divy 345.1; SP 280.1.

bundh: bandhitum Mv ii.484.5 (vs).

budh: budhyilum 'to become enlightened' LV 270.21; 271.2 (in both most mss. °alum); RP 19.9, 15; 59.3 (all vss).

bhid: bhinditum Samādh p. 17, line 4. bhuj 'enjoy': bhunjitum Ny ii.56.3; iii.129.4 (both

vss); paribhumjilum Mv iii.131.13; 158.18. muc (pass.): mucyilum 'to be set free' Mv ii.223.13 (vs); adhimucyilum KP 139.9.

muş: muşitum Divy 94.29; 101.6, 11; also müşitukāma (Dict. s.v. mūşati) Divy 276.24.

mr: marilu-kāma Mv ii.83.2 (vs).

mlā: mlāyitum Divy 570.25.

yam: anuprayacchilum Mv ii.147.2, 3.

rādh: aparādhyilum Mv iii.126.18; 127.2.

li: pratiliyalu-kāma (so mss., see § 36.7) Mv iii.428.10.

vah: udvahitum Mv iii.252.16, 20; 253.15 (all vss). vā: parinirvāpayitu-kāma SP 250.12 (§ 38.58).

viś: praviśitum My iii.151.8; MSV i.106.11 (prose).

śram: viśramitum Mv iii.350.21.

śru: śrunitum Suv 157.5 (vs).

sic: ulsiñcitum Mv ii.91.1.

sr: samsaritum My iii.253.3 (vs).

sthā: upasthihitum Mv i.295.16; iii.24.15; utthihitum Mv ii.428.1.

snā: snāpilum Mv iii.12.5 (vs; intrans., 'to bathe', despite causative stem; Senart em. snāyilum, but see § 38.58).

han: hanilum Mv i.312.6; āhanilum Mv ii.454.15.

hr: viharitum Mv i.34.13; samuddharitum Mv i.98.14 (vs); samanvāharitum Divy 357.20.

tu for tum

36.3. Not very rare in verses are infinitive forms (either Sanskrit or dialectic) lacking the final nasal. It is natural to assume that they are only m.c.; and as a rule, certainly, they occur where the meter requires a short final. However, the mss. at Mv iii.34.1, prose, read *jānitu*, emended by Senart to °tum (next word is *na*; may we suppose that the nasal *n* motivated the corruption in tradition,—that is, that some one failed to hear *-m* before *n*-?). And the form *vodhū* KP 82.9 (vs), § 36.5, suggests independent existence of the form lacking the nasal. Cf. the AMg. JM. forms in (*l*)(*u*, used, to be sure, as gerunds but acc. to Pischel 577 derived from infinitives. In Ap., moreover, forms in (*l*)*u* are unquestionably used as infinitives (Jacobi, San. 18; Alsdorf, Kum. 64).

36.4. Instances of (i)lu for (i)lum in verses, presumably m.c.: jñalu SP 30.12; pratikartu 119.4. — naditu LV 46.6; in 166.13, read: vyavasitu kşapayitu svaya (so v.l.) maha-udadhim = vyavasitam kşapayitum svayam mahodadhim; in 304.15-16 kā tasya sakti mama drstva (so with v.l., Lefm. drsti) sapārişadyam, notthātu mahya carane sirasā prapatum, 'what power has he (= how can he), on seeing me with my followers, not (to) arise and fall down with his head at my foot?' — samanugantu Mv i.65.4; upetu 204.18; kartu ii.286.5. — kşayitu, see § 36.2, under kşi.

$l\bar{u}$ for lu = lum

36.5. Much more strange is the form $vodh\bar{u}$, apparently with metrical lengthening for vodhu = vodhum, KP 82.9 (vs); see § 36.3. Is it a case of the 'law for morae', long for nasalized vowel (§ 3.3)?

itum for itum

36.6. Once the vowel *i* before *tum*, in a dialectic infinitive from ksi 'destroy' (cf. ksayitu, § 36.2), is lengthened in a verse m.c.: ksayitum Siks 345.10 (vs. from Ratnolkādhārani).

atum for itum

36.7. Repeatedly the mss. (and sometimes the printed editions) present what are certainly infinitive forms ending in alum rather than ilum. Senart regularly emends to itum in Mv; whether rightly, I am not sure. There are many cases where at least one of his mss. reads atum, and doubtless more than I have recorded in which all of them read so. If the ending is textually sound, it would seem to involve carrying over of the thematic a of the present stem into the infinitive. My pupil Mr. A. H. Yarrow has discovered avisatum in Mbh. Cr. ed. 1.222,10d, which supports the ending. na . . . śakya aksarebhih pravisatu LV 393.5 (vs), 'it cannot be penetrated by syllables (words)'.—In Mv ii.484.15 (vs), mss. paśyałum or ⁹lu; prcchatum iii.394.17 (mss.; vs); pratiliyału-kāmo 428.10 (prose), (§ 36.2; to pratiliyate; Senart em. pratisamliyitu°, the sam certainly wrong). But in Mv ii.32.1 (prose), where mss. jāgratum, an infin. cannot be construed; read jāgratam (acc. sg. pres. pple.; not jagritam with Senart) .- Divy (all prose): vyavatokayatum 179.17; cāratum 180.13; viharatum 498.3; avamardatum 622.17.

Ending elum

36.8. This occurs not very often, and chiefly, as we should expect, from verbs which otherwise show stems in e(li). So parākrametum LV 134.2 (BHS -krameti = -kramati); upametum (see § 28.48) SP 304.10 (vs); nigrahetum Mv iii.322.4 (prose); in Mv ii.88.6 (vs) read vikretum (regular Skt. form) for mss. ketum, Senart em. kriņitum (but the meaning must be 'sell', not 'buy'!); -padyetum Mmk 514.24; 540.28 (both prose: cf. aor. upapadyesi Mv i.45.10, mss.); pivāpayetu-kāma, Kashgar rec. for SP 322.2 (prose).

Miscellaneous non-Skt. forms in tum

36.9. A few times a weak form of the root is used instead of the regular strong form. Twice r thus occurs for ar where metrical considerations may be responsible; viorlum, 'to open up' LV 46.7 and krtum = kartum Suv 156.4 (both in vss where meter requires a short syllable). But in the rest there is no such reason: utksiptum Divy 279.2 (prose); udgrhitum Divy 579.21 (prose); nirdislum Mmk 4.18 and anupravisjum id. 94.1 (both prose); adhītum in Mv ii.77.13 is Senart's em. for mss. adhitum, but occurs once as a v.l., see § 36.2.

36.10. Av i.182.8 (prose) has syotu-kāma 'desiring to sew'; and MSV ii.50.16 setum (§ 28.50); no infinitive of root siv is recorded in Whitney, Roots.

36.11. śraddadhālum Divy 6.21 (prose) is formed on the reduplicated present śraddadhāli; pratisamlātu-kāma MSV ii.128.5 on the noun pratisamlāna (Dict.; cf. also caus. lāpayati, Skt.).

36.12. More anomalous are the forms paryādalum and °dallum, KP 33.2 and 3 (prose), infinitives to pary-ā-dā 'conquer'. Should °dallum be read for both (influenced by dalla, dalloā)? Or °dalum, cf. § 36.7? Or, finally, the regular Skt. form °dālum? **36.13.** Yet more monstrous is Mmk 74.16 (prose) prārabhantum, infin. from prā-rabh. Should we assume an error of tradition (misplaced anusvāra) and read prārambhalum (or °itum), from the present -rambhali (E +, Whitney, Roots)? Or adaptation to the infin. of some (what?) root in final nasal?

Ending tave (tavai)

36.14. The Vedic ending *lave* lives on in Pali (Geiger 204); and see Pischel 578 on relatives in AMg. In our dialect it is extremely rare. Twice *allave*, an actual Vedic form from *ad*, occurs in Mv (both vss) iii.295.12 *evam jänelk' anallave*, 'so understand, for non-eating' (i. e. that you may not be eaten by the ogresses); 299.13 (one ogress sends a message to her sisters) *kşipram āgacchath' allave*, 'come quickly to eat' (the men). Senart fails to recognize the form. In Ud xxxi.2 the ending is written *lavai*, in *prahātavai*, probably a hyper-Sanskritism (or error?). The same verse in Pali, Dhp.34, has *pahātave*.

Ending anaya

36.15. As in Pali (Geiger 204.3; cf. Sen 12-13), the dative of *a*-stem nouns is used in our dialect rather extensively as an infinitive, even governing an accus. object. Especially is this true (again as in Pali) of stems in ana; the ending anāya in this function is fairly frequent. The stems in ana are sometimes new formations based on thematic present stems; such as budhyanāya, paiyanāya, juhanāya, (pibanāye, below). It is to be noted that such new stems in ana are quite common in our dialect, and often show other than dative forms and infinitive function as infinitive is questionable, are in general not included here. (Sen 1. c. includes a number of cases with dependent genitive; these seem to me dubious.) Our examples are from prose except those marked vs.

lam eva yam icchali bhāşanāya SP 28.6 (vs) 'just that which he wishes to say'. Others, cited by Sen 12–13: gamanāya 187.5; daršanāya vandanāya paryupāsanāya 425.2–3; śravaņāya ... daršanāya 431.1; śravaņāya 459.1. no šaklā siya budhyanāya LV 271.4 (vs); grahaņāya

no śaklā siya budhyanāya LV 271.4 (vs); grahaņāya icchet 338.6 (vs); šaknuvanti ima (with v.l.) dharma vijānanāya 420.22 (vs).

prakrami bhagavanlam darśanāya upasamkramanāya paryupāsanāya Mv 1.255.4-5 'he went there to see' etc.; gamanāya 1.257.12, 18, etc.; ii.101.9, with forms of anujād 'permit to go', also iii.287.9 with avakāšam karoli, 'gives opportunity to go', and i.361.20-21 mrgīye vāro ... gamanāya 'the turn of the doe to go'; na ... śakyo (mss. śakyam) ... cālanāya ii.329.20 (vs), 'cannot be moved' śakyām kṣapanāya 362.8 'capable of being destroyed'; paśyanāya 450.14; 451.1; 453.3, 7 'to see'; pariprechanāya iii.48.8; daršanāya 102.1, 3, 7; 364.4 fl., 'in order to see' (object accus.); agniholram juhanāya 161.9; na pratisamharanāya jakyam 252.8 (vs) 'cannot be turned back'.

lam paroalam abhirohanāya Divy 113.20, 'to elimb the mountain'; daršanāya vandanāya 203.20; kārvaļikam samnāmanāya 446.20, 'to conquer the k.' — pūjanāya ... munim RP 5.10 (vs). — (bodhisatīvo) klešair na šakyam vinipālanāya KP 48.7 (vs) 'cannot be destroyed by depravities'. — no śakyu ... nivarlanāya Dbh.g. 17(353).9.

Ending anaye

36.16. Only in Mv, such infinitive forms not infrequently end in \bar{aye} instead of \bar{aya} . This is not to be interpreted as dat. of a fem. stem in \bar{a} ; the Mv otherwise has not a few cases of indubitable short a stems showing datlves in \bar{aye} (§ 8.45). Nor is meter concerned; these forms occur freely in prose, and in metrically indifferent positions.

12

No record of the like has been found elsewhere. Examples (prose except as indicated): varo...gamanāye Mv i.362.15 'turn to go'; (anujānāhi tām) gamanāye ii.101.1 (cf. line 9 gamanāya in close parallel); eşo na śakyo (mss. śakyam) maye dharşanāye 320.8 (vs) 'he cannot be hurt by me' (metr. indifferent); pibanāye 432.4, 5; paśyanāye 456.8, 14 (also 454.20, 455.2, in both of which v.l. °nāya); darśanāye iii.366.20 (vs, but metr. indifferent; cf. 364.4, above, °nāya; same phrase).

Ending ana-tāyai

36.17. Enlargements of nouns in ana by addition of the suffix $l\bar{a}$ (§ 22.41, 42) also may furnish dative infinitives: *cakru anuttaru vartanatāyai* Bhad 10 (vs), 'to turn the supreme wheel'.

Ending anā (ana?)

36.18. LV 282.8 (vs) reads: hantā gacchatha pūjanā hitakaram, etc., 'Hol Go ye to honor the Beneficent One.' There seems no doubt of the reading. In Ap. an infinitive ending ana is recognized by Hem. 4.441 (Pischel 579), and is recorded in texts (Tagare, Hist. Gr. of Ap., 321 ff.). Our form pūjanā may perhaps be identified with this form. The a could be m.c., if ana is the original form (cf. §§ 8.3 ff. for use of the bare 'stem' in lieu of oblique case forms of a-stems). Or we could assume a fem. stem pūjanā, of which our form could be dative (or any other oblique

case, §§ 9.64 ff.); the Ap. and would then contain shortening of the final vowel, as very often in Ap. Tagare calls and a nom. sg. (= Skt. -anam), which seems implausible.

Other dative infinitives

36.19. As in Pali (Geiger 204.3), we may doubtless admit that the dative of almost any noun of action could be similarly used. But in no other category is the usage anything like as clear-cut as with ana nouns; most of the cases could be described as ordinary nominal datives. The following may perhaps be regarded as a reasonably clear case: My i.61.15 (vs) yatra care brahmacaryam (mss. add bhagaván, unmetr.) sarvajňatām abhilāsāya. This seems to mean 'under whom he led the religious life, in order to seek for omniscience'. The alternative of taking abhilāsāya as abl. of a fem. stem *abhilāsā is unattractive, since such a stem does not exist and its type is hardly common (though the meaning, 'thru desire for omniscience', would be perhaps easier).

am, akam, im (?)

36.20. On the forms $-h\bar{a}ra(ka)m$, $-h\bar{a}rim$, the meaning of which would fit classification as infinitives, see §§ 35.4–6. I have treated them under the gerunds; it may be recalled that, in Mindic, historic gerunds and infinitives are confused (see e.g. Pischel 577).

37.1. Regular passives were formed in our dialect by the suffixes iya, iya, from present stems in either e (= aya) or thematic a. Besides this common MIndic type, there were passives in ya inherited from Skt., sometimes retaining in our texts MIndic phonology, and capable (as in all MIndic) of inflection outside of the present system. Active instead of middle endings are widely used. In fact, as in MIndic generally, the middle endings are moribund, except perhaps the pple. ending mana, but even for this (a)nl(a) is often substituted in both passive and middle forms; doubtless it no longer had any difference of meaning in any MIndic dialect. On the other hand, middle endings (at least those in final e) are often substituted for active endings (in i), but apparently only in verses and only, or chiefly, for metrical convenience. Finally some miscellaneous irregularities, or at least formations not noted in normal Skt., will be found.

Passives in *îya*, *iya*

37.2. A regular passive is formed from thematic stems or from present stems in e(aya) with the suffix *iya*, or less commonly *iya*. Both formations are found in Pall, along with *iyya*, equivalent to *iya* by the 'law of morae' (Geiger 175, 176). In Prakrit (Pischel 535) only equivalents of our *iya* seem to be recorded (*ia*, *ijja*, *iyya*), no **iya*. According to Geiger, Pali uses this formation (with both i and \bar{i}) especially from e(aya) presents; the same might perhaps be said of our language, altho in it (at least in Mv) it is common enough from ordinary thematic presents (as indeed it is in Pali and Pkt. also). The endings are much more commonly active than mediopassive (cf. §§ 37.6 ff.).

37.3. The *iya* formation has been plausibly explained by M. Leumann, IF 57.233.¹ It started, he suggests, with Skt. passives like *niyate*, *diyate*, *slhiyate*, *dhiyate*, inherited as Pali *niyati*, *diyati*, *-lhiyati*, *dhiyati* (and the like in Pkt.). By the side of these, Pali had presents *neti*, *deti*, *-lheti*, *-dheti* (in part probably analogical creations to participles in *ita*, Leumann 1. c. 211). Hence to any present in MIndic *eti* (primarily = Skt. *ayati*) a passive in *iyati* came to be formed analogically. Then, since *aya* and *a* presents came to be frequently interchanged (§§ 38.1 ff.), like any other verb form originally belonging to *aya* (*e*) presents, the passive in *iya* was made from any *a*-present.

37.4. On the other hand, in the passive in *iya*, the short *i* is nothing but the ordinary epenthetic *i*, as was seen by Leumann (l. c. 234 f.); cf. § 3.102. In some cases, to be sure, we must probably assume with Leumann analogical lengthening of the vowel of the preceding syllable; so Pali māriyati for *mariyati, to match māreti. For MIndic epenthesis of *i* before *y* was later than shortening of a long vowel in a closed syllable, at least before the combination *-ry*-; that is, māryate yielded first *maryate (or °*ti*), and then *mariyati; the actual Pali māriyati has analogical ā. It is not clear that this was true of all combi-

1. Anticipated in principle by Tedesco, JAOS 43.390, who cites only *deti*: **dita*: *diyate*. — A special and very peculiar case is *sanniyate* Karmay 27.27, 'is known', for

nations of consonant plus y; cf. e. g. Pali $S\bar{a}kiya$ (our $S\bar{a}kiya$) for $S\bar{a}kya$. In our dialect, most passives in *iya* are to be regarded as containing quite normal instances of epenthetic *i*. When a long vowel appears in the preceding syllable, this may or may not be analogical (as it probably is in Pali *māriyati*, but probably not in Pali *Sākiya*); it may, of course, also be due to phonetic semi-Sanskritization (and sometimes certainly is that).

37.5. We shall try to separate examples of passives of aya (e) presents from those belonging to ordinary *a*-presents; it must be remembered that the distinction is often far from clear. All the following are prose except those indicated as (vs).

37.6. īya passives from aya (e) presents: kuf[iyantāMv i.6.5 'being pounded or crushed'; piceīyanti 24.3, 5; pravešiyati 223.10; vardhīyanti 274.2; ohāriyatu 295.18 'let (a tax) be levied'; kathiyati ii.17.4 'is told'; sthapiyatu 69.12, thapiyatu 13, sthapişyati 14 (v.l. sthāpayişyati, read perhaps sthapiyişyati 'will be established', fut. of pass.); mārgiyanti 'are searched', loliyanti 'are agitated, upset' 167.12; nivāriyanti 174.12 (fem.) 'being restrained'; prajīapīyanti 274.4 '(seats) are provided'; šekhīyati 423.15 'is taught' (°yanti 434.10); vardhīyanti 433.14 (v.l. vaddhī°) 'are raised'; upanāmiyanti 465.10 (v.l. °iya°); dhāriyāmi 489.15; vāriyati 493.1 'is restrained'; samkiyanto iii.37.9 (? = šaňk°, to caus. of śaňk) 'being made doubtful, anxious'; allīpīyati 127.4 (cf. allīpiyanti, § 37.8), °piyantam (pple.) 5; pravešiyantehi 127.12; cārīyanti 176.10; sampardhīyati 390.3; 405.9; visrānīyati, pravāhīyati 405.2. ardīyamānā(h) Divy 39.7; pratipādīyati 226.21 'is handed over' (to pratipādayati; so mss.; ed. em. pratipradīyate); abhyarthīyase 249.30 'you are begged of'. — dešiyate Suv 155.8 (vs) 'is taught'. — pithīyate Ud xvi.9, 10 'is covered'.

37.7. iya passives from a-presents: bhāsīyamāņe 'being spoken', Kashgar rec. (Thomas ap. Hoernle MR 133) for SP 327.1, ed. nirdišyamāne; otariyali (see Dict.) 358.12 (vs) 'is conquered', probably for utlar⁶ from Skt utlarati 'conquers' (ot- for utl-, § 3.73). — kaļīyantiye Mv i.217.7 (gen. sg. f. pres. pple., see Dict. kaļīati); karīyati 269.3 (vs) 'is made'; karīyatu iil.360.1 (vs); prechīyanti i.272.12, 14, 15; 350.18; ii.110.14; iil.393.7 etc., prechīyati ii.13.21; 112.4; iii.177.3; 392.17 etc.; prechīyamāna iii.131.9; apakarsīyate (mālyam odiena) i.302.15 'is carried off' (mss. °yanlam); frunīyati 'is heard' ii.97.7; iii.297.8; apāvuriyati (see Dict.) ii.158.1; parivisīyati 26.4 'is served'; samharīyati 371.7 'is destroyed or damaged'; paricarīyamāno 423.9 'being tended' (in same line upasthihiyamāno); dhovīyanti 466.4, 5 'are washed'; upasthihīyati iii.23.17 'is attended'; āsiyati (to ās 'sit') 86.3; šāsīyema 166.14 (1 pl. opt. pass. to šās 'punisħ'); anujānīyati 176.16 'is permitted'; raksīyati 298.11; yācīyanti 315.18; parigrhņīyeya 393.13 (mss. °yam; 3 sg.; here ī may be influenced by Skt. stem "grhnī', but °grhnāti with suffix Iya would give the same result).—In Suv 157.2 (vs), read: yadra sātram śurujyate, 'where the sūtra

samifinity on the analogy of dhivate etc. to $dh\bar{a}$ etc., with Prakritic nn for $(m)j\bar{n}$.

(181)

is heard', with most mss.; Nobel with one ms. yas tam for yatra, leaving unconstruable *śruniyate*, which is read by all mss. except one *śruni*°. – nirdiśiyati Samādh p. 23, line 1 (vs?).

37.8. iya passives from aya (e) presents: daršiyali I.V 184.3 and 4 (vs) 'is shown'; dhyāpiyantānām Mv 1.126.2 (mss. dhyāniy°, em. Senart) 'of (Buddhas) being cremated', to dhyāpayali (?); ānāpiyalu 1.310.14 (v.1. °piy°) 'be it commanded'; višrāņiyantaisya 76.10 'being given away' (= višrāņyamānesu); pūriyantasya 76.10 'being drawn' (of a bow; = pūryamānasya); posiyantā 223.9 (mss., vs) 'being nourished'; vāriyanto 274.1 (v.1. vāri°; both mss. "nlaņi) 'being restrained'; ovāhiyali (to avavāhayati), sajjiyali 274.16; saņvardhiyamāno 423.14 (v.1. °dhīy°); mūrchiyāmi 428.1 '1 am stupefied'; ulkhanāpiyamānāni 439.6; mārgiyamānā 460.4; upanāmiyanti 463.10, 464.9 (in 464.9 v.1. °iyanti); allīpiyanti iii.68.11; 405.15; sekhiyanti 184.6.

37.9. iya passives from a-presents: kariyati LV 185.6 (vs) 'is made' (so Lefm. em.; seems required by meter; mss. karişyati). — In Mv i.16.13, 15, read takşiyanti (Senart takşi^o, all the corrupt mss. end in -iyanti in 13, and two of them in 15); anugacchiyamānā ii.101.15 'being followed'; prechiyati 207.3: iii.39.19 (v.l. both times precht^o); upashihiyamāno ii.423.9 'being waited upon' (in same line paricarīyamāno).

Active for middle (passive) endings

37.10. Even in Skt. (at least pre-classical and epic; Whitney 774) out-and-out passive forms appear with active endings. And confusion between active and middle is even more widespread in Skt. (Renou, Gr. scte. p. 392 f.). As Renou points out, metrical considerations often play a part; cf. § 26.3. In Mindic (Geiger 120, Pischel 452) these tendencies are carried much farther; the middle is moribund, and the passive has very largely active endings. Only the pple. in māna is found to be still fully alive in both Pail and 'all dialects' of Pkt.; even it may, however, be replaced by active forms in anl(a), al(a). Naturally, then, our language, like Mindic in general, shows pretty complete confusion in regard to voice, in comparison with Sanskrit.¹ As we saw, even the specifically passive forms in iya (iya) more often than not have active endings.

37.11. A few (out of many) examples of active forms from verbs which in Skt. are middle only: muhārtam āgamaya 'wait a moment' LV 103.3 (prose), to Skt. āgamaya i utpadyati 'is produced, arises' Mv i.81.2, utpadyanti i.121.14 (twice; all prose); samutpadyati SP 46.13 (vs, could be m.c. for "le); āpadsyatha 'you will attain' SP 73.7 (prose; below in 9, āpadyante, and in 13 āpadsyāmahe).

37.12. When the only difference in ending consists in substitution of final *i* for *e*, in verses, one may often suppose that metrical convenience is concerned, as in the following (some known to Epic Skt.): driyanti 'are seen' SP 9.7; 16.6; 24.4, 6; ucyati 'is called' LV 28.8; parihiyati 'is lost' LV 175.4; chidyanti 'are cut off' LV 223.15; *irûyati* 'is heard' LV 416.11; Samādh 19.23, 24; upajāyati 'is produced' LV 420.4; anunīyati, pratihanyati, Samādh 19.33. Sometimes before a word in initial vowel the *i* of the ending becomes consonantal: dahyady (amano^o) 'is burned' SP 85.14 (vs); driyanty ekam yathā 'were seen as one' LV 280.12 (vs; in both the preceding and following lines driyante occurs). Noteworthy is pûryati 'is filled' Ud xvii.5, twice in one verse, in the oldest ms.; one of the two is changed to pûryate in a later ms. The Pali parallel (Dhp. 121) has pûrati.

2. Note, however, that the passive value of such an ending as *-le* is still vigorous enough, at times, to give birth to a corresponding active in *-li*: samdrsyati 'sees'

37.13. But exactly similar forms are also found commonly in the prose of Mv, as: upajājati i.7.6; pratieidhyanti 'are penetrated' 7.13; dahyanti 'are burned' 25.4; prajāāgati 'is observed' 27.6; vipacyali 'is matured' 104.6 (mss. "nti); vādyanti 'are sounded, played' 144.15; 214.7; ghātayisyasi (so with mss., Senart em. wrongly) 'you will be killed' 244.2; kriyanti 'are made' 352.17, 18; khādyati 'is eaten' ii.152.16; hanyanti 'are slain' 213.7; and many others.

37.14. And other active endings are common enough in passive forms, where meter can not be a controlling factor: dhakşyalha 'you will be burned' (for 'yadhve) SP 73.7 (prose); samlayatha (2 mss. °yadhoe) 74.4 (prose); dahyeyu 'they would be burned' 86.8 (vs). — diyatu 'let be given' LV 109.20 (vs); paripäryatu 'let be fulfilled' 200.11 (vs; in 235.19 pariparyalo, m.c. for °tu = °tām, § 30.14); abhiyujyatha (impv., to pass. abhiyujyate which in our language seems to replace a middle abhiyunkte etc.) 'apply yourselves' 203.4 (cf. abhiynjyadhve, v.l. °yatha, as impy., for presumptive Skt. abhiyungdhvam, SP 79.11, prose; abhiyujyante SP 80.5 and 7, in the latter v.l. onti); read with v.l. upanāmye (for $^{\circ}yel$, 3 sg. opt. pass.) 'yam 'let him be offered (food)' LV 386.17. — dršyatu 'be it regarded' Mv i.150.4 = 218.5 = ii.20.4 (vs); diyatu, mucyalu, hanişyali (all passives) i.311.15 (prose); chindişyam 'I shall be cut' (for °şye) ii.173.14 (prose); vadhişyam 'I shall be slain' iii.352.14 (vs); vacyāmi 'I am told' (for ucye) i.362.19 (prose); mā . . . vihanyāhi (for °yasva) 'be not downcast, subject to disappointment' ii.405.16 (vs). - paridahyāmi 'l am tormented' (for 'ye) Divy 420.6 .(prose). --- ucchosyatu, vikiryatu, vicchidyatu, prapūryatu, visudhyalu (all 3 sg. impv. pass.) Suv 53.10-14 (vss). kşaņyema Šikş 92.1; kşaņyati 92.3 (both prose).

37.15. A few examples of present participles with active ending (otherwise mostly normal Skt.) may be listed separately, in view of the fact noted § 37.1 that the middle ending manu is still alive in MIndic generally (as it is, of course, also here): dahyanto (so with mss., or with WT adahyanto) 'being burned' = dahyantanah(nom. sg. masc.) SP 254.8 (vs); dahyalah = °mānān (acc. pl. masc.) 321.3 (prose); ākruiyatah = ākruiyamānasya 'of him being reviled' 378.12 (prose). — prohyatas (so with v.l. for Lefm. prodgatas) = prohyamānān, acc. pl. pple. pass. of pra-vah 'being carried away' LV 358.6 (vs); uhyalah (so read with best mss., others °tāh, Leim. em. ⁹tā) 361.4 (vs). — (All Mv cases prose:) upanīyantā = ⁹yamānāḥ (nom. pl. masc.) 'being led' Mv i.132.9; satkriyanto = °yamānah (nom. sg. masc.) 'being revered' 308.17; aprajñāyantehi = aprajñāyanāneşu (which is actually used below in line 5) 'not being known' 339.2 ff.; amueyantiye = °yamānayā (instr. sg. fem.) 'not being released 363.7; niyanto = niyamānaķ ii.155.17; 168.13; labhyanto (mss.) = labhyamānah iii.166.9. — abhinirmimiyantam (acc. sg. masc.) 'being magically created' Gv 444.15 (prose). Note reduplicated stem; cf. § 37.37.

37.16. Yet more bizarre seem some cases in which the passive sign ya is not found, and what looks like a plain active form is used in passive meaning. The connecting link is clearly the use of middle forms which had meanings indistinguishable from passives. Thus in Skt. *muñcali* means 'releases', but *muñcate*, mid., 'frees himself' or 'is freed', virtually = *mucyale*, pass. Now since MIndic and BHS freely use active for middle endings, they may also say *muñcati* for *muñcate*, 'is freed'; hence Mv i.308.8-9 (prose) *vyādhitā vyādhito muñcanti* 'the sick are freed from sickness'. Perhaps of similar origin, and in any case found in Pali, is *bhañjati* 'breaks', intransitive, LV 175.17.

Gv 523.21 (prose), formed as a direct pendant to samdrsyate 'is seen'; likewise drsyati (see Dict. s.vv.). Cf. § 37.23. This meaning might attach to Skt. *bhajyale* (pass.); no middle seems to be recorded in Skt. except in the perfect, and the active is always transitive acc. to BR except for a single form *babhañja*, for which BR would read *babhañje*. In Pali, *bhañjati* is both trans. and intrans.; perhaps when intrans. it represents **bhañjale*, or a more Sanskritic *bhaňkte*, middle.

37.17. A quite normal-seeming 1 sg. optative active is used with definitely passive meaning in SP 113.3 (vs, end of line, meter not concerned in ending): *veştim pi ca kārayeyam*, 'and I might be made to perform forced labor.'

37.18. In this light, doubtless, are to be understood such participles as pldayantā 'being tormented' My i.5.10 (prose), which represents pidayamana 'undergoing torment', mid., instead of pass. pidyamāna. Senart emends this to pidiyanta (cf. § 37.8), but inconsistently keeps the precisely similar prarthayanti of Mv ii.69.7 (prose), 'being asked for' (yasodharā . . . devadattena); the mss. are recorded as reading °yanli, which certainly intends nom. sg. fem.; repeated in line 9 with even worse corruptions in the mss., but intending the same form, while between the two, in line 8, occurs devadultena ca prarthayamana, middle (not passive!); the same prarthayanti, correctly written, as passive, in ii.72.12 and 13. Similarly samjñāpayamānāh My i.23.11 (prose; kept by Senart) 'being signalled to' is a middle pple. in passive sense. And from the noncausative of the same verb, a middle finite form is used in passive sense, LV 289.4-5 (prose) ckatkaś ca bodhisattvo devaputrais caivam samjanile sma, 'and each several Bodhisattva was thus imagined by the gods'.

37.19. I hesitate to class here *stuvantam* (acc. sg. pres. pple.) 'being praised', on which see § 37.35.

37.20. On the other hand, *paśyeta* LV 29.8 (vs) 'she would be seen' is irregular only in that in Skt. *drśyate* is used as the passive to *paś*, while both *paśyati* and *paśyate* are only active in sense, 'sees'. In our language, however, there is even with active ending a *paśyati* 'is seen' LV 419.12 (vs, perhaps "li m.c. for "le). Perhaps wahyanli 'flow' (subject 'rivers') LV 398.6 (prose) is to be judged similarly; see Chap. 43, s.v. vah (3).

37.21. Like other *a*- (and *ya*-) presents, passive forms in *ya* may be enlarged by a second syllable *ya*, on the model of causatives that are indistinguishable in meaning from transitive *a*-presents. So ghusyayante = ghusyante 'are resounded'; samvidyayante = samvidyante 'are on hand'. See § 38.21.

37.21 a. Contrariwise, the y(a) of the passive sign is lost, after long vowels, in a group of future forms treated in § 31.3.

Seemingly passive forms with active meaning, and usually active endings

37.22. There is one well-known case, at least, in Epic Sanskrit, of a seemingly out-and-out passive form with active meaning. Skt. *šakyale* is regularly passive to *šak* 'be able', and is either used impersonally, or if it has a personal subject, is then accompanied by an infinitive which we translate by a passive (Speyer, Skt. Syntax § 387). But in the epic, *šakyale* may mean 'he can', and may then also have active endings, *šakyali* etc. Such forms are fairly common in BHS (examples Chap. 43, s.v. *fak* 3); and so Pali *sakkali*, Pkt. *sakkali*.

37.23. Our language shows other forms which seem comparable with this:³ *ādīyati* (also *ādiyati*) 'takes' (as

3. Or are they, and perhaps even Epic Skt. and BHS *śakyali*, to be regarded as newly created actives, based on Skt. passives (*ādīyate* etc.), from which they differ only in having active endings? So we have explained *drśyali*

also Pali ādiyali), cf. Skt. adiyale, passive of a- $d\bar{a}$, 'is taken'; $\bar{a}h\bar{u}yali$ 'calls on, challenges'; $\bar{a}khy\bar{a}yali$ 'tells' (Pali akkhāyali, usually passive like Skt. $\bar{a}khy\bar{a}yale$, but at least once active, see Dict.). These are included as ya-presents in my classification of non-Sanskritic presents, §§ 28.24, 25. Some of those listed in § 28.28 probably also belong here.

Medio-passive for active endings

37.24. The confusion of medio-passive and active endings also shows itself in substitution of the former for the latter. So far as I have noted, however, this occurs only in endings (present or future indic.) which show final e instead of i, and only in verses. It seems to be almost exclusively occasioned by metrical convenience; in one or two cases it occurs at the end of lines but it is doubtful whether even these are exceptions. (In LV 54.15 read upasthapilvanā, ger., for upasthapisva nā of ed.) - bhavate bhavati SP 63.1; LV 190.2; bhavisyate SP 149.6, 10; bhesyale LV 54.12 (cf. bhesyali, with i m.c. for i, 54.5, and below); darsayute SP 24.10; 127.10; vadate 25.12; vinesyase 67.12; spršisyase = spraksyasi 67.14; cintayante (also Epic Skt.) 87.2; samśrāvayate 127.11; paśyate 355.8; sthāpite for sthāpayati (see § 3.48) LV 74.22; śosayate 174.19; jāyale for jayali (Dict.) LV 262.12 and 13 (nāšūro jāyale senām ... śūrus lu jāyale senām); kārayale Mv i.117.6 (certainly m.c.; note parallel kārayati in preceding line). Others, § 3.61.

37.25. Twice at the end of rathoddhatā lines, in the same passage, LV has final e for i: *janeşyase* (one ms. ⁹*si*; cannot be taken as pass. or intrans.; 'you will produce' [in such persons great faith and joy]) LV 54.2; and *bhesyale* 54.14 (one ms. ⁹*ii*). At the end of lines of this meter a long syllable is usual in LV, and was perhaps originally required; note however the exception *srunisyati* 54.6. These are the only cases I have noted of final e for i in verb endings in positions where the meter may, perhaps, not demand a long syllable; but probably others occur.

Non-present forms from passive stems

37.26. As in MIndic generally, the passive stem is inflected independently: aorists, futures, infinitives, and even (occasionally) past participles are formed on it. Thus, aorists: dr_{sysu} 'were 'sen' LV 74.12 (vs); $v\bar{a}dy_{isu}$ 'were made to sound' LV 194.2 (vs); $v\bar{a}dy_{isu}$ 'were 'sen' LV 74.12 (vs); $v\bar{a}dy_{isu}$ 'were 'sen' LV 74.12 (vs); $v\bar{a}dy_{isu}$ 'were 'sen' LV 194.2 (vs); $v\bar{a}dy_{isu}$ 'due'se 'sen' LV 21.4 (prose); $samprav\bar{a}dy_i$ (3 pl.) Mv ii.328.2 (vs); $v\bar{a}dhy_i$ (for $b\bar{a}^\circ$, to pass. of $b\bar{a}dh$) 'were oppressed' LV 22.6 (vs); $bhij_{je}$ (so Senart, mss. $bhoj_{je}$; with jj for dy, to $bhidy_{ade}$) 'was split' Mv ii.412.11 (prose).—Futures: see § 31.2.—Infinitives: mucyitum 'to be released' Mv ii.223.13 (vs); adhimucyitum KP 139.9 (prose). — Past pple.: kim ti vihanyitena Mv i.155.14 (vs) 'what's the use of your being depressed?' (either pple. or substantive, according as ti for te be taken as lnstr. or gen.). (Cf. Pali chijjita, ppp. to passive chijjati = Skt. chidyate.)

Middle Indic phonology in passive forms

37.27. Definitely MIndic phonology occurs not infrequently in passive forms inherited from Skt. Thus we find forms with MIndic assimilation of y to a preceding consonant, like (v)uccali for ucyale, khajjali etc. for khādyale, bhajjali etc. for bhajyale, -paccale, ⁹li, etc. for pacyale;

'sees' in § 37.10, n. 2. In Epic Skt., however, and at least once (Divy 374.19) in BHS, even *sakyate*, with passive endings, may have active meaning. see § 2.8; on sanniyate see § 37.3 n. 1. The prothetic v in *vuccati* and the like is, of course, also a direct retention from MIndic: see §§ 2.50–55. In Mv ii.79.4 ff. (vss) occurs eleven times a form which the mss. mostly write *mryati* or 2e, but which meter proves must have had a long first syllable. Senart writes *mryyati* or *mryyate* (the latter seems to be the reading of both mss. at 80.3); the same verse in Pali, Jät. iv.53.9, has *miyare*. Probably something like Senart's reading was intended.

37.28. The participles *nistirila*- (§ 38.10) and *dirita*-(§ 34.10) imply MIndic presents passive **lirati* and **dirati* (= Skt. *liryale*, *diryate*). Formally similar, but active in meaning, is *kirati* = *kirati* 'scatters'.

Miscellaneous non-Sanskrit Passives

37.29. There remain a rather miscellaneous number of passive formations not known to normal Skt.; some, at least, are definitely MIndic in character. On the anomalous *jivali* for *jiyale* see Dict. and $\S 2.31$.

37.30. The root *kşan* (*kşan*) 'wound, injure' seems to have no passive recorded in Skt. In Sikş 187.8 (prose) occurs *kşanyale* 'is injured'; and with active endings (§ 37.14) *kşanyema*, *kşanyali*, Sikş 92.1 and 3.

37.31. The Skt. passive of *sams* 'praise' is *sasyale*. In Mv i.88.8 (vs) all mss. have *prasamsyale* 'is praised', and Senart reads so also in 11 below, where however all mss. lack the anusvāra. If the form is not a mere corruption, it is a passive based on the thematic present, but hardly to be described as normal MIndic (which would give *prasamsiyali* or the like). Similarly *samslaryetsu(h)* Mv i.235.17 (mss.) may be understood as 3 pl. aor. to **samslaryale*, a passive to *samstarali*. Senart em. *samslar rayensu*, but a passive seems called for; meter obscure,

37.32. Perhaps phonetic corruptions, or else simplifications of consonant clusters involving MIndic assimilation of y after consonants, are prechasi 'thou art asked', for prechyase, Mv i.280.5 (vs); and adhyeşanti 'they are invited (to give instruction)', for adhyeşante (or more strictly 'correct' Skt. adhişyante), Mv i.106.12 (vs); Senart em. adhyeşyanti, but the form may represent a MIndic *ajjhēsati for *ajjhessali = *adhyeşyale. It may also be a mere corruption, as Senart assumes; or even an instance of a middle form (used in passive sense) with active ending (but no metrical compulsion can be involved here).

37.33. For *janeya*, conceivably for *janiyeya 'would be produced' (cf. Pischel 535 *vejja* etc.), see § 8.36, where another interpretation is given.

37.34. Lank 24.17 (vs; and similarly 33.8; 203.13) samādhyate seems to be a quasi-denominative, passive, to the noun samādhi, which immediately precedes the

Voice

phrase samādhyale. kalham cillam, 'how is the thought concentrated?'.

37.35. In LV 195.20 (vs) mām sampašyi sarvalokai(h) sluvanlam, 'you shall see me being praised by all the worlds (people), sluvanlam certainly has passive meaning (confirmed by Tib., hjig rien kun gyis na la bslod pa milhon, 'see praise to me by all the world'). The Pkt. passive of slu is thuwvai but AMg. at least once records thuvanti, for which Pischel 536 (see also 473) would read thuvo². Is *v* for *vv* m.c.? Or have we a blend of MIndic (Pkt.) thuvaai and the alternative thuvijiai? I should hesitate to class the form baldly as an active used for passive, though formally it can be called a quite normal active pres. pple.. acc. sg. masc.; cf. § 37.19. In any case the meter confirms the correctness of the LV reading.

37.36. On the other hand, I am inclined to regard the spelling *śruyali*, sometimes (e. g. ii.31.3; 427.14; iii.67.6, all prose, no v.l.) recorded in Mv for *śrūyale* (Pali suyyali, Pkt. suvvai), as a mere miswriting for *śrū*°; it seems to me that in Senart's place I should have emended.

37.37. On the passive suffix ya added to a reduplicated present stem see abhinirmimiyanlam, \S 37.15. More questionable is anudadhyamāna (see Dict.), perhaps a passive pple. to anu-dhā.

37.38. Definitely Mindic is håyate as passive to hā, instead of hiyate, 'wanes, declines, becomes deficient' (= Pali håyati beside hiyati): kim häyate tava 'what is lacking to you?' Mv i.147.1, 5 = 203.16, 204.3 = ii.7.12, 18 (vs); häyante SP 177.2, 4 (vs), so Kashgar rec. and 1 Nep. ms., others hiyante; fut. häyisyanti Mv i.241.18 (vs); häyisyati Mv i.44.4 (vs). It is analogous in formation to Skt. jääyate, khyäyate, dhmäyate, ghräyate, and a few others. Of different and problematic origin, is nirdhäyate or 'li (root dham?); Dict.

37.39. Somewhat similar to hayate for hiyate is vikrāyale instead of vikrīyale, a passive to vi-krī 'sell' (possibly patterned on häyate = hiyate? no other possible origin has occurred to me). It seems to have existed in AMg., which records a present passive pple. vikkāyamāņa 'being sold'. In Mv ii.241.20 we must apparently assume an opt. of this: vikrāyetsuh 'would be sold'. So one ms.; the other vikrāmayensuh; Senart em. vikrīyensuh. And in Mv ii.242.14 we should probably read vikrāyişyati 'will be sold' (Senart vikrayisyali with one ms., the other vikrāmāsyati); in Mv ii.242.11 vikrāsyati (v.l. vikrāmasyati), on which see § 31.3. In Pkt. (JM.) there is even recorded a pres. act. pple. vikkanta (Sheth), possibly pointing to an active *vikrāli, or *vikrali; the former could be a back formation from vikrāyate or vikrāsyati, the latter might be modelled on vikreti (§ 28.49), a case of an ali present instead of an eli present (§§ 38.3, 25 ff.). But I have found no such active form in our language.

38. Stems in aya (e); Causative, Denominative, etc.

38.1. It is well recognized (Pischel 472) that in Prakrit, presents in e(l)i are freely formed by the side of presents in a(l)i, without difference of meaning. The same process occurs in Pali (Geiger 139.2), the less commonly.

38.2. The most important source of this phenomenon was, I believe, first suggested by Tedesco, JAOS 43.388, and later independently by M. Leumann, IF 57.212. Past participles in *ila*, to some extent even in Sanskrit (for whatever historic reasons), and much more regularly in MIndic (§ 34.7 fl.), are associated with presents in *ati* (as well as with those in *ayati*). So, even in Vedic, we find such forms as ppp. *raksila*, *palila*, *likhila*, to *raksali*, *palati*, *likhati*; and in later Skt. e. g. *vadila* to *vadati*, and many others. But since the *ila* ppp. was regularly associated with presents in *ayali* or MIndic e(l) are analogically formed to such participles. A familiar example is *vadeli* 'speaks', found all over the MIndic field and in BHS; it means precisely the same as *vadali*, and was surely based on *vadila*. A list of such BHS forms is given below, §§ 38.18 fl.

38.3. Leumann (l. c.) pointed out that the converse substitution of ali for historic eli forms also occurs quite commonly in MIndic. This fact has not been so generally recognized. It is however a natural, almost predictable, result of the frequent occurrence of the equivalent types padeii: vadali. Hence e. g. AMg. māranli 'they kill', by the side of mārenli = Skt. mārayanti. BHS has many such forms, including mārase = mārayasi. See §§ 38.25 ff. The role of the ppp, here is interestingly illustrated by samudānita, ppp. of samudānayati (°neti), replacing Skt. (samud)ānīta. This was obviously formed to the present samudaneti or onayali by analogy with the many presents in eti or ayati with ppp. in ita. There is also a ger. samudāniya (as well as apaniya); see § 3.43. I have found no a-present like *samudānati; but some such form seems presupposed by the noun samudanana, 'acquisition' (Dict.).

38.4. Not a few Skt. aya forms, often called 'causatives', are apparently indistinguishable in meaning from corresponding ati presents. Some of these are also indistinguishable in form from them, except that -ayatakes the place of -a. So by the side of raksati 'guards' (V +), Classical Skt. has (rarely) raksayati, called 'causative', but meaning also simply 'guards'. This rare form seems almost certainly based on the pp. raksita (V +), just like the MIndic *eti* presents. So with many so-called 'causatives used in the sense of the simple verb'.

38.5. But not with all. Sometimes other formal criteria distinguish historic causatives from non-causatives, in Skt. One: the non-causative present may not have a stem in -a-. Two: it may have a different vowel in the root syllable. So kārayali, originally or at least usually 'causes to make' or 'to be made', is distinguished in both these ways from the regular Skt. present karoli. Even this kārayali, however, is sometimes used in Skt. without causative meaning, that is like the simple verb. See Lang. 22.94-5, with notes 1 and 3, where such forms are discussed.

38.6. A causative based on a simple transitive may take an acc. either of the person or of the thing (or of both). That is, $k\bar{a}rayati$ may mean 'causes (some one) to do

(something)', or 'causes (something) to be done, gets (it) done'; the last must sometimes have been hard to distinguish from 'does (that thing)'. Perhaps that is why kārayati, even in Skt. (BR s.v. 7), sometimes cannot be distinguished in meaning from karoli. On the other hand, some such forms may not be historic causatives at all, but at best homonyms of causatives. They may, for instance, be denominatives. Everyone recognizes that ghātayali (B+) must have been a denominative to the noun ghāla; yet it functions as causative to hanli (root han). Its origin is recognizable because it happens not to coincide in pattern with bases of the standard causative type. But a noun kāra (very common in Skt. as final in compounds, hardly in use otherwise) might have formed a denominative kārayati, which no one could distinguish from the causative in form, but which might well have the same meaning as karoli - or also the meaning of a causative to it. It actually has both meanings in Sanskrit. (I am not, of course, expressing any belief that kārayati actually originated as a denominative; I only say that such a form might have had that origin.)

38.7. And finally, analogical influence from forms which, for any of the above reasons, have the formal aspect of causatives but no causative meaning, may have led to the use of other (genuinely historic) causatives with non-causative meaning. A list of apparent formal causatives with the meaning of the simplex in BHS will be given below, § 38.23.

38.8. Whatever the reasons may be, BHS shows an even more confused and intricate interplay of stems in -a, on the one hand, and -aya- (or -e-) on the other, than is suggested by the statements about Pali and Prakrit in the standard treatments. Whether more searching studies in MIndic dialects would make them seem closer to BHS, I cannot say with confidence.

Causatives in ayali

38.9. To begin with, the suffix aya (or e) still lives as a productive causative-forming suffix, as in Skt. (despite the far greater prevalence of $\bar{a}paya$, which we shall treat below). Only, in BHS this aya is, like other verbal morphemes, added to the thematic present stem (minus final a), not to the Skt. 'root'. So also, at least occasionally, in Pali (Geiger 179, end: nacceti, taggeti, to naccali, laggati). Hence:

Root kram: abhiniskramayati, implied by ppp. °mila, to °mali, na kadācin mayā kumāra udyānabhūmim abhiniskramilah LV 187.3-4 (prose), 'I have never allowed the prince to go out into the pleasure-park'. (Skt. has simple kramayali, to be sure, tho krāmayati is commoner; but it has no caus. of abhi-nis-kram, and our form is surely a non-Skt. creation.)

kşubhayanli LV 339.13 (vs), caus. to Pali (sam-)khubhali (cf. ppp. Skt. kşubhila, AMg. khuhia).

grhnayām āsa, 'caused to grasp', Mmk 599.10 (vs), implying grhnayati, caus. to grhnati (§ 28.5).

ghr: parighareti, 'moistens round about', Mv iii.301.17 (prose; so with mss., Senart em. °ghār°), caus. to -gharali (see Chap. 43). cint: anuvicintayitāḥ, see § 34.9.

chid: chindayati, 'causes to be cut off', Mmk 395.14, 18 (prose). Not = chindati 'cuts off' (cf. § 28.4), but caus. to it, as shown by parallel verbs in 18, dālayati pācayati iosayati utsādayati (all caus.).

jalp: jalpesi Mv ii.165.3 'ordered to say'.

jāgr: pralijāgarayelvā (one ms., § 35.26) Mv i.324.8, 'having had (food) prepared'.

pai: pațila, 'torn', My 1.5.8 (prose), certainly caus. in meaning (cf. vipățiyanlān, pres. pass. pple. of caus., same line). Skt. has only caus. pățayati, 'rends', except a single instance of pațati, 'sich spalten' (pw). A pațayati, caus. to this, seems implied by pațila; perhaps the same word Desin. 6.12 pația, defined by vighațila.

pasyayanti, 'they cause to be seen, display', Dbh.g. 48(74).26, to pasyati, 'sees'.

bharayitvā, 'having caused to be hired', SP 105.11 (prose), to bharati, 'hires'.

muñcayişya, 'I shall cause to send forth', Sukh 8.16 (vs), to muñcali (but not certainly caus., perhaps 'I shall send forth', in that case = muñcali).

vid, 'find': vindayanti, 'they cause to find', LV 242.7 (vs; text °nli, but certainly 3 pl.; best ms. °nli), to vindati.

vr, 'cover': prāvarayanlyā, 'by the (nun) who is making (the novice) dress', Bhīk 11a.5 (prose), to BHS prāvarati, 'puts on (clothes)'.

śi: for śayāyilo read śayāpilo, § 38.53.

sad: nişīdayati, 'causes to sit', Divy 295.14 (prose); LV 70.21 (prose); nişīdayişyāmaļ Divy 420.25 (prose); to nişīdati, 'sits down'.

sphr: parispharayati Mvy 1649, caus. to -spharati (Pali parippharati).

svap: supeti, 'puts to sleep', Mv ii.423.7; 433.12; supiya, ger., 'having put to sleep', LV 50.4; to BHS supiti, Pali supati, 'sleeps'.

hā: vijahayya, ger., 'having caused to leave', Bhik 15a.5, implying vijahayali (cf. Whitney 1051d), caus. to vijahali.

Causatives in aya from passive stems

38.10. Based on a historic present passive are Pali *lireti* 'accomplishes', BHS ppp. *nis-lirila* 'accomplished' Dbh 3.20 (prose), implying *lirayali*, caus. to **lirali*, not recorded in Pali or BHS but = Pkt. *lirai*, clearly = Skt. *liryale*, pass. to *larali* (Geiger 175.2; Pischel 537). Skt. *nistarati* means 'accomplishes'; Skt. *nistiryale*, BHS (MIndic) **nistirali* 'is accomplished'; caus. *nistirayali* (implied by *nistirila*), 'causes to be accomplished', thus finally = *nistarati*. The form defles any other interpretation (PTSD calls *lireli* 'caus. to *larali*', which of course is unsatisfactory).

38.11. This is confirmed by other cases, such as kāma alabhamāna dahyayante, LV 242.7 (vs), 'desires when not obtained cause (a person) to be burned'. Here dahyayante can only be a caus. to dahyate, pass., 'is burned'. It is true that Whitney, Roots, cites as Epic a 4th class present dahyati 'burns'. But he undoubtedly was relying on BR's citation (under dah, pass.) dahyema, trans., 'verbrennen', Mbh. 4.799. But the Crit. ed. 4.22.7 shows that this is a mere corruption, not found in any ms. there cited; Crit. ed. reads dahyeta 'would be burned', a true passive.

38.12. Another such case is *pilhiyeyāsi* (mss.) My iii.25.6 'you are to cause to be covered', 2 sg. opt. caus. (**pilhiyeli*) to *pilhiyati*, pass. to *pilhayati* or *pilhati* 'covers' (Dict.).

38.13. The MIndic caus. suffix -*āpaya*- (below) may similarly be added to a passive stem, if (as I believe) we may adopt the reading *uddhvasyāpayilvā* Šikş 57.6 (Dict.), 'having caused to be insulted', to *uddhvasyate* 'is insulted'. And a stray Pali form noted in Geiger 181 seems to belong here: Milp. 90.11 *chejjapessāmi* (read surely chejjā^o; Geiger says 'zu erwarten wäre chijjā^o') seems probably based on some sort of passive, whether chejjali = Skt. chedyale (pass. to caus. chedayali), or, following Geiger's hint, chijjali = Skt. chidyale.

Causative with long \ddot{a} for a in root-syllable

38.14. In a few cases, BHS shows causatives containing \bar{a} in the root syllable, where Skt. either shows short a, or has no such form at all. These seem to be analogical to Skt. causatives with long radical a, which are the regular type: e. g. vādayali, caus. to vadali. It is worth noting that MIndic and BHS have also vadeli (non-caus.) = vadali (§ 38.21); this differs formally from vādeli (= vādayali), caus., only in the quantity of the root vowel. The existence of pairs of this sort seems, in a few forms, to have brought about lengthening of a to \bar{a} in the rootsyllable, as a mark of the causative. I can see no other way of explaining nispālāyilā Bhik 28b.1, 'she that has been caused to flee' or 'banished', ppp. to a *nispālāyali, 'banishes', caus. to nispaläyali, 'flees, runs away'. Pali has paleti by the side of palāyaii (Geiger 139); this would be to caus. *påleti (*pālāyali) as vadeti to caus. rādeti.

38.15. In Skt., *janayati*, 'begets', and its compounds have only short *a* in the root syllable. So far as I can discover, only *janei* (and the like) is recorded in Pkt., too; and according to Geiger 178.2 b, only *janeti* in Pali. But this last statement is questionable. An aor. *samjānesum* occurs (to be sure, with v.l. °*jan*°) in Dhp. comm. iii.251.8. And a noun of agent, nom. sg., *samjānetā* occurs SN iii.66.16 (repeated in Index), tho PTSD would emend to *samjanetā*, which is in fact read in the same passage SN i.191.2. In any event, there is no doubt that BHS repeatedly shows *jānayati* for Skt. *jan*°: *samjānayati* Gv 502.23 (prose); *jānayi* (1 sg. opt.) Bhad 45; *upajānaya* (impv.) Mv ii.135.4 (vs); *samjānayisyanti* LV 439.4 (prose; v.l. °*jan*°); gerunds *jānayitā* Mv ii.131.10 (prose), *sam-jānayua* LV 137.19 (prose); ppp. *jānitah* Gv 381.17.

jānayya LV 137.19 (prose); ppp. jānilah Gv 381.17. **38.16.** Note also ger. vighāļya, 'having opened' (muşim, 'the clenched fist'), Divy 280.9 (prose); the caus. of vi-ghal in Skt. seems recorded only as vighaļayali (pw 'zerreissen, trennen, zerstreuen: vereiteln, zu Nichte machen'; meanings somewhat different from that of our word), while AMg. vihādei 'opens', is probably identical with the BHS verb. To be sure Skt. has udghāļayali, 'opens', from the same root with another preverb.

38.17. Divy 77.20 (mss.) and 466.16 ucchrāyilah should doubtless be read ucchrāpilah (Chap. 43, s.v. śri).

Verbs in ayati (eli) = ati

38.18. It will be convenient to list next the instances noted in which *eli*, or its Sanskritized equivalent *ayali*, and forms based thereon, occur apparently as precise synonyms of forms based on *ali* (§ 38.2). Any sort of *ali* (or middle *ale*) present may thus be replaced by *ayali* or *eli*. Even passives are so treated, sometimes with relention of medio-passive endings: *ghusyayante*, *sampidyayante* (below). In BHS, forms in *-aya-* are, at least in the present system, rather more common than those in *-e-*. But it is impossible to separate them: the same list will include both.

38.19. Most of the examples presented below are forms belonging to the old Skt. present system: present indicatives, imperatives, participles, and (of the *aya* type) optatives. There are included also a few gerunds, gerundives, and infinitives. Of futures of this type I have noted below *abhisambuddhesya*, *muncayisya*, *vahesyam* (all 1 sg. fut.); for others see § 31.9. Aorists in esi certainly belong here in so far as they are not based on old presents in eti (*ayadi*), but rather on original *ali* presents; such cases are

38.21]

fairly numerous and will be found in §§ 32.66, 67. Whether, or to what extent, the substitution of eli for ali presents was influential in the establishment of the aorist type ending in -e(1), -etsu(h) and the like, is not so clear; these forms seem primarily borrowed from the optative: see § 32.86.

38.20. There are even cases in which a stem retaining aya in that form is provided with an ending containing stem-final e, yielding a blended stem-form -uye-. Thus dhārāpayehi 'cause to hold' My ii.447.14 (prose; so mss., Senart em. "paye); § 30.5. Similarly aorists in ayesi will be found in § 32.66, and past participles in ayila in § 34.8.

38.21. The forms are listed alphabetically by 'roots'. Those occurring in prose are so marked; it will be noted that they are fairly numerous. Moreover, in many of those occurring in verses the e or and was certainly not, or not chiefly, due to metrical convenience.

āp: prāpuņehi Mv iii.270.14, to prāpuņati. So the mss.; Senart em. °āhi. In line 16 Senart prāpuņāhi with one ms.; the other "ehi.

ās: adhyāsayali, 'sits upon' = adhyāsle, LV 18.8 (prose); samupāsayante, 'they revere' Dbh.g. 41(67).9.

is; icchesi My ii.483.8 (in the same line kāmasi for kāmuyasi or kāmesi, § 38.28; both may be m.c.); pralīcchayanti Mmk 65.12 (prose). Acc. to BR 5.1145, icchayāmi once = icchāmi, Rām. (but in the late Book 7). - adhyesayāmi = adhyeşāmi, 'I ask for instruction', Lank 7.13, 15 etc., see Dict.

iks: niriksayālha, 'look!' LV 50.16; samudiksayanto LV 134.4.

ulkanth(-ali, denom.): ulkanthehi, 'grieve' My iii.167.9 (prose). Not an aya present regularly; cf. ulkanthanlasya in preceding line.

kamp: kampayali, 'quakes' (intrans., subject 'the earth') My iii.341.7 (prose; thrice in this line; in line 9 kampali, same mg.).

kānks: ākānksayatha (impv.) Sukh 99.6 (prose)

kr 'make': vikurvayalo Gy 267.3 (= vikurvanlah). kr 'scatter': prakireli My ii.252.1, 16 (prose); prakirentasya (pres. pple.) id. ii.253.10 (prose); prakirenti id. i.211.20 (prose; in the context, and in the parallel ii.15.17 fl., prakiranli repeatedly); okirenti id. ii.304.3; vikirayamāņān, 'cleaving' Gv 87.8; 89.11 (both prose). Pali also has pakireli, wrongly called 'Caus.' in PTSD; it means exactly the same as pakirali.

krs: awakarşayanti, 'they remove' SP 159.12 (prose); parikarşayati, 'carries around' Divy 475.14 (prose). In Skt. karşayati (simplex) seems used in the same sense as the active karsali.

klp: kalpenti Mv i.135.1, 10 (prose) 'are adapted, serve', intrans., = kalpante.

kram: ākramesi My ii.408.18 (mss. parākramesi, hypermetric: the form might, however, be opt.); upakramemi My ii.459.16 (prose): parākrametum LV 134.2: alikrametum LN 134.7; avakrāmayali = avakrāmati, 'enters into (a mental state)' LV 400.14 (prose; same passage okrāmelpā Mv iii.318.15).

kri: krinenti Mv i.91.10 (so 5 mss., Senart with 1 ms. krinanti) = krinanti.

krīd: krīdayanti LN 75.17 = krīdanti; krīdayantam LV 157.7 (prose).

ksip: niksipayilvā (kamandalum) = niksipya My i.241.1 (nikșipetva would be metrically better); nikșipetvă id. ii.65.4 (prose; so mss., Senart eni. °pilva).

khan: anukhaneli parikhaneli, 'digs up to,-around' Mv iii.301.16 (prose).

gam: agametu My iii.263.17 (prose; based on stem of what in Skt. is 'thematic aorist', here blended with thematic pres.); āgamehi Mv ii.31.20; iii.161.10; 186.17 (all prose); ayamaya 'come!' Divy 76.10 = 465.10 (prose); MSV i.3.12; 24.8, etc.; agamayala Divy 534.12 (prose); gacchayamī = gacchāmi LV 223.1; gacchehi Mv i.187.1 (Senart em. °āhi).

garh: vigarhayitavyā(h) Divy 115.7 (prose).

uah; viuahayamanah Bhad 39. This and vipasyayamanah (below, s.v. pas) are parallel to visodhayamanah and vimocayamanah, and are made to fit them in meter and rhythm.

But vigāhayamāna, for vigāhamāna, occurs also Gy 476.4. grah: pragrhnetva My iii.131.4 (prose; v.l. pragrhitva), to pragrhņali; pratigrhņchi My iii.211.10.

ghus: ghusyayante LN 92.6 'are resounded', to passive ghuşyate.

car: caresi = carasi My ii.187.23; vicaresi = ²rasi id. ii.321.21 (both these might be m.c.); carayesi (2 sg. opt., § 29.21) brahmacaryam, 'live thou the religious life' (= cara or careh) My ii.42.18 (mss. $vara^{\circ}$).

chid: chindaya, 'cleave?' Suv 133.8 (= chinda, Skt, chinddhi).

jrmbh: vijrmbheli, 'yawns' = °bholi My ii.281.7 ff. (prose).

iña: anujanaye My 1.257.11 (prose) 'I authorize'. anujāne; in repetition of same, line 18, anujāneyam, opt, of anujānali; Pali anujāneti recorded only as caus, parijānayeyā 1.N 330.17 = parijāneyā, Skt. [°]jāniyāl. prajānayanti Sukh 9.8 = prajānanti.

dham: vidhamaye (2 sg. opt.) LV 117.6 (Pali also vidhameti = vidhamati).

dhr: dhuresi (2 sg. pres.) My i.210.1 = ii.13.21 (prose); i.210.9 = ii.14.8 (vs): cl. dhāremi, response to i.210.1, in verse i.210.3. To BHS and Pali dharati (Skt. dhārayati, which may have influenced this form-a sort of blend of dharasi and dhārayasi?).

nam: (cillam) namayali LN 398.1 (prose) (the Tathagata's) 'mind turned' (lowards indifference); = namale or namali, intrans. — pranamayya, 'bowing' (? = pranamya; see Dict. s.v. pranamati) Av i.347.4; ii.3.9; Divy 463.22 (all prose: the last doubtful).

namasya (denom.): namasyayanli, 'pay homage' LV 119.2 = namasuanti.

pad: samāpadyelum Mink 514.24; pratipadyelum id. 540.28 (both prose); upapadyesi, § 32.67.

paś: vipaśyayamānāh Bhad 39; see above s.v. gāh; paśyehi Mv i.187.4 (Senart em. °āhi).

prech: prechenti My i.256.10 (prose); pariprechesi id. ii.298.3 (prose, v.l. °asi). [Note: precheti SP 16.7 is taken by Burnouf and Kern as = prcchali. But as WT say it is rather precha-lit: precheli mailreya jinasya putra sprhenli te ... 'Saying, "Ask!" Maitreya, son of the Jina, these ... are cager' (to hear the answer given to your question).]

bandh: bandhesi Mv ii.483.7 = iii.18.10; bandhayilva Mv ii.168.6 (prose; subject is a policeman, object a suspected criminal); anubandheli = °ati, 'follows' Mv iii.294.19; (sam)anubandhayatām, 'let him follow' LV 46.2 and 12 (probably m.c., since anubandhatām occurs elsewhere in the context, e.g. line 8); anubandhayāmah LV 47.7, 21 and 48.7.

budh: abhisambuddhesya (1 sg. fut.) My ii.404.19 (ms.; prose), to present °buddhali (§ 31.33).

bhañj: prabhañjayitvā, 'having broken' SP 159.1 (prose).

bhan: bhanesi = bhanasi My ii.222.19 (also Pkt. bhaņei).

bhās: adhyabhāsaye Mv iii.327.5 (prose) = the much commoner adhyabhase (3 sg. aor.); see further Chap. 43, s.v. bhāş (2).

bhid: bhindayilvanā (so read with v.l. for text bhindiy°), ger., LV 73.21.

man: abhimanyayāmah, 'we desire' = °manyāmah LV 49.11; manesi, manayase (?), see Chap. 43, s.v. man (2).

mīmāmsa (desid.): mīmāmsaye (li) Av i.94.3 (prose). So divide, with Feer, supported by Tib.; 1 sg. pres. mid.; li = ili; misunderstood and wrongly em. by Speyer. So also Av i.189.4 (prose), read with mss. mimamsayeyam, 1 sg. opt.

muc: muñcayantah Megh 308.13 (prose); muñcayisya (1 sg. fut.) Sukh 8.16.

mürch: sammürchauilvä, 'having fainted' LV 234.2. mī: marenti (= Pali, BHS maranti) 'die' Gv 213.20. yāc: yācesi (= yācasi) Mv ii.185.9; yācintyā(ħ) (i for

e, § 3.49) = yācentyāh = Skt. yācantyāh (gen. sg. pres. pple.) Mv i.133.1 (prose).

ruc: virocenti, 'shine' = °canti Mv i.78.12 (prose); just below. line 14, abhirocante is used with transitive (causative) mg., 'illumine'.

rud: rodayāmi, 'I weep' Mv iii.351.9 (prose, no v.l.;

below in line 13 rodāmi). lap: ālapeti, 'speaks' LV 236.2 (and see Dict. s.v. lapayati, 'boasts').

Iabh; Iabheni Šikš 336.8 (from Ratnolkādhāraņi). vad: vadayati: 'says' Mv ii.237.18 (prose); vadeti SP 197.4; KP 81.10; vademi SP 59.2; LV 335.15; Mv i.292.16; iii.110.13 (mss.); 371.21; 372.2 (all vss; but note vadāmi iii.371.17); i.8.14; ii.418.8, 9; 430.15; iii.51.17; 447.16 (all prose); vadesi SP 93.11; Mv iii.110.21; 195.1 (this prose); 371.9 fl.; vadenti i.256.11 (prose); vadehi i.324.10; 330.12; iii.159.4; 173.18; 205.10 (all prose). Common in Pali and most Prakrits, as well as in BHS. vand: vandenti Mv i.256.10 (prose).

vas: vasayathā, 'rest' (= vasata, impv.) LV 202.5 (see Dict.); vasayāmi, 'I dwell' (= vasāmi) Divy 212.4, 6 (prose); āvasehi Mv ii.405.9.

vah: vaheti Mv ii.228.1 (v.l. vahati, but meter seems to support vaheli); vaheşyam (1 sg. fut.) Mv ii.236.13 (prose).

vid: passive samvidyayanta (ime), i.e. onte, = samvidyante, 'are on hand', 'are provided' LV 115.9. vrt: nivartayasva, 'turn backl' (intrans.) LV 225.12

(prose); mā nivartayatha, 'do not return!' LV 228.8 (prose); nivarlehi My ii.105.8.

vedh (MIndic): pravedhayamánah (so with best mss.), 'trembling' LV 188.3 (prose). Both this cpd. and the simplex vedhali regularly show ali, not ayali, forms in Pali and BHS; and this is certainly not caus.

śās: śāsayilavyah, 'to be punished' Divy 243.28 (prose); samanuśāsehi Mv ii.103.14 (prose). Pali anusāseti, beside °ati.

śru: śrunehi Mv iii.140.13, to śrunali.

sad: nişīdayed, 'would sit down' SP 344.1; nisīdeli galamohā Mv i.306.12 (so read with mss.; Senart em. niside vigatamohã).

sev: upasevehi Mv ii.103.13 (prose).

stu: stavayimsu, 'they praised' LV 414.10 (vs), = (aor. to) BHS slavali.

sthā: tişthayet = lişthet Sādh 411.18 (vs); anulisthayema Mv iii.290.1 (prose; Senart with v.l. °s/hiy°; °ma is Senart's em. for mss. 'sa or 'ia, and anu for ava), 'we would (may) experience or carry out'.

sprs: sprsayilum, 'to touch' Mv ii.65.14 (prose); inf. to spršayati = spršati.

sphr, sphur: sphurayilvā, 'pervading, suffusing' LV 113.2 (prose) = sphuritvā (to sphurali).

hr: āhareli Mv iii.201.5 (prose) = āharali, next line. In corresponding Pali passage ahareli occurs repeatedly, e.g. DN ii.223.27. The object in both is aharam, and PTSD regards ahareli as denominative to this noun; ähareti could not be this. - uddharetvā My i.7.14 (prose).

Historic causatives with non-causative meaning

38.22. In the preceding list there is nothing but the infix -aya- (or MIndic -e-) to suggest association with causatives; and all such forms have the meaning of non-causatives in -a-. It is, however, worth noting that some of them are identical in form (allowing for MIndic phonological or morphological features which are here irrelevant) with aya-formations which in Sanskrit are recorded with causative meanings. This is true of -ikşayali, kampayali, -karsayali, -kramayali and -krāmayali, kalpeli (Skt. kalpayali), -gamayali, -garhayali, -gāhayali, namayali, bandheli (Skt. bandhayali), -bhāşayali, mūrchayali, yāceli (Skt. yācayali), -roceli (rocayati), rodayati, -vartayati. Cf., even in Skt., such cases as rakşayali = rakşati (§ 38.4). Other influences which may have resulted in non-caus. function with what look like causatives were discussed in §§ 38.5-7.

38.23. Whatever the reasons may be, it seems clear, that most, if not all, of the following list were historically causatives. They all differ from the preceding list in that other formal features than -aya- (-e-) identify them as such; usually the vowel of the root syllable. It is at most possible that in one or two cases historic denominatives are concerned (cf. § 38.6). All of them have non-causative meaning. (For a few Skt. causative stems in -paya- used in non-causative meaning see below, § 38.58.)

āp: prāpayanli 'they get' Mv i.30.15; 31.15 (both prose). Skt. prāpayati and Pali pūpayati, pāpeti seem recorded only as causatives. However, AMg. appears to have both pavei and pavai as non-caus. (Ratnach. 'to obtain, to get, to acquire'). If we could assume that *prāpali (AMg. pāvai) was older than prāpayali (AMg. pāvei) as non-caus., 'gets', then the latter would be based on the former (which could be a Mindic analogical present), and this example would belong in the preceding list. It seems to me more likely that AMg. pāvai is secondary to pāvei, like the cases in §§ 38.25 ff.

dis: uddesenti 'they point out (the way)' Mv i.21.3; 23.9 (both prose). Formally *uddesayati would be expected to have causative meaning. To be sure, it does not seem to be recorded anywhere; I find only Ap. uddesai (= *uddesati), Jacobi, Bhav. 162.9, used in the sense of the simplex uddišali. If our My form is based on that Ap. form, it would belong in the preceding section.

nī: (timiram) apanāyayişyase LV 365.3, 'you will dispel', in the sense of apanayati. Cf. Skt. caus. nāyayati. Does this mean literally 'you will cause to be dispelled'? Cf. § 38.5.

nud: pralivinodayali (= Pali pațivinodeli), 'dispels', is the regular form (and only one known to Pali); "nudati in one Divy verse, see Dict.

pat (?): see Dict. s.v. vinipātayati.

bhū: abhisambhāvayilvā 'having attained' (object tad-artham) Mv i.4.12 (prose), in a formula found elsewhere, Mv ii.115.6; iii.90.18; 382.8. The sense is that of pre-Class. Skt. and Pali abhisambhavati, Pali and ISHS abhisambhunāli (°nali). The causative is recorded in BR only once in mg. 'begrüssen' from BhägP.

vad: vivādenti 'quarrel' Mv i.10.12; = Skt. and Pali vivadali; but Pali also vivādayali, usually written vivādiyali and regarded by PTSD, perhaps rightly, as denom. to vivada. This might be the explanation of our form; vioādagali is recorded in BR only as 'start a lawsuit' i. e. 'cause to dispute' (Yājñ.). But the form is probably a mere corruption; read pi pāțenti with iii.455.6.

vah: (ājñām) pralivāhayāmi 'I disobey' Gv 122.8

(prose); = prativahali, q.v. (Dict). viś: praveši (ger.), 'having entered' (= pravišya) LV 243.11; nivešayi (aor.; buddhadharme), 'entered into' or 'rested in' LV 158.13; praveśi (aor.) Mv iii.162.5 (prose). In Skt. and Pali -veśayali is caus. only; acc. to Ratnach., AMg. pavesei is used in the same meaning as pavisai, 'to enter, to penetrate'.

vraj: pravrājagegam Gv 417.15 (prose) gan nv aham bhagavalo 'nlike pravrājagegam, 'suppose I now ... enter the religious life'. For the converse, pravrajali = caus. pravrājayati, sec § 38.24.

38.27]

śru: pratiśrāvayitum, 'to heed, give ear to' (= pratiśrołum) Divy 207.7 (prose). ślis: samślesayel, app. 'would embrace', SP 114.9;

samslisyel. So Burnouf; Kern 'conciliate'. Dhātup. gives samsleşayate 'ālingane'.

sr: avasāri Mv i.319.14, 16, 20; 323.13, 16 (in 16 mss. anusāri); iii.47.12; as 1 sg. ii.117.20; 120.17; avasārim (1 sg.) My ii.119.6; 123.16, 'came to, arrived at', apparently for avasari(m), which is read in corresponding passages in Pali, e. g. Vin. iii.11.30 (CPD s.v. avasarali). No present *sārati seems to be recorded, and the forms (all in prose) certainly have no caus. mg., yet seem to be based on ava-sārayali. Perhaps acāri = acāril may have influenced this form; § 32.19. But note also anusāre in prose Mv iii.101.16; this is not so easily understood as influenced by acarit or the like.

On nivāsayati see Dict.; it does not belong here, tho PTSD wrongly defines nivāseti as if it were used as an intransitive. It has the same meaning as the same form in Skt.

Non-causative forms with causative meanings

38.24. The converse of the preceding group occurs more rarely. These must apparently due be to some analogy; perhaps to the much commoner replacement of -ayali (-eti) by -ati forms (below), which often means in effect replacement of causatives by their underlying noncausative forms (while retaining causative meanings). Hence, possibly, other primary non-causatives, which differ formally from their causatives in other respects than lack of -aya- (-e-), came to be used with causative meaning:

alliyati, regularly 'comes', but also 'brings', see § 38.67. apadyati Mv i.181.10 (vs), apparently 'fetches' (object bhojanam), $= \bar{a}p\bar{a}dayali$; but only two mss. read so; the other four are clearly corrupt; perhaps the true reading was different.

ulpadyema Mv i.365.10 (prose), apparently 'produce, cause', as if from *ulpādayali*: ye ... heļhām ulp^o. No v.l. cited; but in parallels Mv ii.447.17; 448.7; 486.1; 489.3, forms of the caus. utpād(ay)ati are always used.

jahāti SP 61.8 (vs), kānksām ca šokam ca jahāti prāninām (subject sugatāna ghosah, 'the voice of the Buddhas'). The most natural meaning seems to be 'dispels' or 'removes (doubt and sorrow of, i. e. for, or from, living beings)'; so Burnouf and Kern; and so Tib., bsol. This implies a sort of causative to the usual meaning, 'abandons'.

pravrajāhi mām Mv iii.386.13 (vs), 'make me a monk!' So one ms.; the other prajahi, evidently a corruption for this. Senart em. pravrājehi, which is not only violent, but metrically impossible. Similarly, tho closer in form to the Skt. caus., pravrajehi Mv iii.268.5 (mss.; prose; caus., Senart em. °vrāj°); the impv. ending ehi is however often found in verbs which otherwise show no aya (e) forms (§ 30.5). And in My iii.379.11, for Senart's pravrājehi, one ms. has pravrajehi (the other pravājehi). Further support is afforded by pravrajişyam Mv iii.268.6 (prose) 'I shall make a monk' (so mss.; Senart em. pravrājayişyam); pravrajelsuh (mss.; Senart em. pravrājensuh; 3 pl. aor.) Mv i.323.9 (prose).

vilapisyase LV 335.3 (vs), 'you shall be made to howl', and amlayisyase LV 335.5 (vs), 'you shall be caused to wither', both with instr. bodhisallvena, apparently caus. in meaning. In the same context with vivarjişyase, vitrāsişyase, prapālişyase, višoşişyase, etc. (§ 38.31), which are normal causative forms except for lack of -ay-.

pralilisthe Mv iii.294.10 (vs), aorist; as Senart observes, must apparently be causative in mg., 'set, placed'. This gives support to SP 63.10 (vs) pratisthihitva, which seems to mean 'firmly establishing' (so Kern), the object being upāyakaušalya; Burnouf 'grāce à leur habileté ...', perhaps understanding upayakausalya in the sense of a locative? (This would be possible, § 8.11.)

Verbs in ati = ayati (eti)

38.25. As was pointed out above (§ 38.3), MIndic and BHS not infrequently replace stems in ayati by ati, without change of meaning. This is the converse of the substitution of ayali for ali, §§ 38.18 ff. Many of the forms in ayali thus displaced were historically causatives, and sometimes the substitute in ali has the form of the primary verb on which they were based. Such cases could be associated with the immediately preceding list, but since here the only formal difference is replacement of -aya-(-e-) by -a-, it seems best to list them with this much more extensive list, which includes denominatives, and presents in -aya- (-e-) of still other origins, for which -a- is substituted.

38.26. Tho paralleled in Prakrit, as we saw, this substitution is less common in our language than the opposite substitution of ayati for ati, except in certain formal categories such as the gerund and especially the optative. In the optative the mode-sign regularly contains e, and this leads to frequent telescoping of preceding e (aya) of the base (cf. § 29.4), resulting in forms like tarpet for tarpayel, deseyam for desayeyam, etc. For such reasons as this it will be more illuminating here to group examples by formal categories instead of by 'roots'. The list of optatives offers only a sampling and could be considerably extended; some other examples will be found in our chapter on the optative. The list of gerunds (on which cf. Lang. 13.113 note 4; also § 35.24, with references; they are MIndic new creations, not ancient inheritances) in ilvā instead of ayilvā is also by no means exhaustive. This is, in fact, true of other sections of the following list. Prose cases are marked as such.

38.27. Optatives: Ending e(1): tarpet 'would gratify' SP 126.14 = tarpayet; dhare SP 255.10 (in same line with dhāreti = dhārayati) and 293.8; śrāved 'should recite' SP 229.7 = śrāvayed; upasthape (Kashgar rec. and WT) SP 88.12; vācet SP 342.4.

Ending i (m.c. for e): prasami Sukh 23.5, and praśāmi id. 23.6.

Ending eta: dhāreta SP 255.14, so read with La Vallée Poussin's Kashgar fragment JRAS 1911.1073; deseta SP 282.6; janeta SP 250.16.

Ending ema: vijňāpema My i.360.5 (prose).

Ending eyuh: kalheyuh (denom.) Sukh 74.18. Ending eran: ghāteran 'they would kill' Lank 252.16 (prose) = ghātayeran.

Ending eyam: vineyam (= vinayeyam) Mv i.337.8; deśeyam Mv i.37.8; 60.6 (v.l. deśayeyam); pravarteyam Mv i.39.1 (v.l. °layeyam); 330.4; 331.3; tāreyam, āśvāseyam id. i.39.5 (in both v.l. °ayeyam; the same sentence contains mocayeyam and parinirvapayeyam; all the preceding are prose); sthapeyam SP 128.2 (v.l. sthapisye); praphreyam Mv i.53.10 (in parallels i.42.9 and i.337.3 mss. prapūraueuam which Senart keeps in the latter, regarding this version as prose); pramoceyam Mv i.42.17 (repeated 53.14 where mss. unmetrically °cayeyam; so, unmetrically, all mss. also read in i.43.2 = 53.16; vicaleyam, see Dict. s.v. vicalayati.

Endings eya, eyā: (verses) kşapeya SP 111.14; dhāreya 229.6; prakāšeyā 237.2; vijnapeyā 284.8; tarpeya 351.4; jareya LV 74.3; janeyā 152.6; daršeya 284.5; dešeya 393.2; šoșeya Mv ii.329.7; grāheya ii.330 18; dešeya, moceya, tarpeya, pureya (read pūr°), śameya Suv 37.7 ff.; (the rest are prose) dāleya sampradāleya Mv ii.125.4; vipravešeyā ii.146.14; vihelheya ii.178.7; šobheya iii.10.16.—Denom-inatives: (verses) kalheyā SP 283.15; gaņeya LV 338.7. —

From vi-nayati: vineya (for vinayeya) Mv i.43.4 = 54.2 = 337.8 (vs; printed as prose in the last place, where Senart moreover prints vineyam, but one ms. reads vineya).

Ending eyyä: prakāśeyyā My 1.336.20; niveśeyyā i.53.6 (mss.).

38.28. Present Indicatives: kāmasi = kāmayasi My ji.483.8; in the same line, conversely, icclusi for icchasi (§ 38.21); both favored by meter.

kilāmasi = klūmavasi My ji.483.9 (may be m.c.).

ghātanti = ghātayanti My i.27.8 (prose: so all mss.; Senart em. ghātenti). Cf. AMg. ghāyae, ghāyāvaha, ghāyamāna (Ratnach.).

chadante (read °ti, m.c.) = chadayanti 'cover' Sukh 50.16.

desate = desayati My i.187.3 (perhaps m.c.); desati Suv 19.4 (vs. but metr. indifferent): desantu Suv 23.8 (cited as desentu Siks 217.1).

nāmanti = nāmayanti Šiks 324.18.

paripācati = °pācayati Siks 325.2.

payanti 'they give to drink' My i.8.5 (prose), mss. for payayanti which Senart adopts by em.; Pali normally has payeti, but records a pres. pple. payamana as if from pāyate (°ti). In a verse parallel to this passage, i.12.7, mss. āyācanti (except one °centi); I would read apāyanti (Senart °yenti).

paripūrati (v.l. °rayati, but meter seems to demand °rati) Mv iii.356.6.

moce = mocaye (mocayāmi) LV 219.4.

pratisammodale (see Dict.) Gv 53.14; °dante Bhike 26b.4, 5 (all prose).

mārase = mārayase My i.179.18; AMg. māranti and the like.

rocanti 'they accept, approve' (sāsanam) SP 306.4, rocayanti, Pali rocenti; but AMg. royai etc. beside roei etc.; abhirocante 'they illumine' My i.78.14 = °cayanti; in line 12, just before, *birocenti* 'shine', intrans. (§ 38.21). *lālati* 'coddles' = *lālayati* Mmk 35.5.

varņanti (denom.) = varņayanti LV 29.12. vādatah (3 dual, = vādayatah) MSV iii.16.6.

vedati 'experiences, feels' = vedayati, Pali vedeti, Av i.243.12 (vs, but metr. indifferent); in later repetitions. e.g. 247.10, vindati is used. AMg. has veyai, veyanti, vedanti by the side of eti forms.

visesanti 'distinguish' = visesayanti Lank 357.1; 371.3 (in 371.2 the regular visesayet).

38.29. Present Imperative (rare, except such cases as \$30.5): samtāra 'take across!' RP 49.15 (vs) = samtāra 'take across 'take ac lāraya.

38.30. Present Participles: lada, m.c. for ladan = tādayan, LV 341.13.

vyäyämanlena 'exerting oneself' Mv ii.30.11 (prose). The long radical a seems to prove that vyayamayati is the present concerned (not vyāyamali). It is not common; BR cite vyāyāmya, ger., from Manu, and Sheth cites vāyāmento (pres. pple.). It may be a denominative from vyäyäma.

iranta(h) Gv 372.13 = irayantah with i for i perhaps m.c. (but cf. pres. iryali; note also that irentah would be as good metrically).

a-vilomanta (°mata) Dbh.g. 27(53).3; see Chap. 43, s.v. vilomayali.

sūcata Šiks 342.10; I understand this as for sūcatā (with a m.c.), instr. sg. pres. pple. to the denom. sūcayati.

38.31. Futures: SP (all vss) darsisye = darsayisye 237.9; uccārisye 237.10; adhivāsisyāma 271.10. — LV (all vss) vivarjisyase 333.14; vilrāsisyase 334.7; prapālisyase 334.9; viśosisyase 334.13; udbhrāmisyase 334.17; rosisyase 334.19; tárişye 361.4; tárişyase 361.6. — parājişyāmi Mv il.314.17 (prose, no v.l.), 'I shall overcome', if textually sound, can only stand for parajeşyāmi, or rather presumably MIndic (and sporadic Skt.) °jayişyāmi. — pravrājişye

Divy 260.20 (prose; so mss., ed. cm. °jayişye). --- prāpişya (= prāpayisyāmi) Dbh.g. 12(348).11. -- svādhyāsyati (and the same form repeatedly in this text) AsP 52.5, 21 (prose). The denom. svådhyayali 'studies, repeats aloud' is common (see § 38.35 and Chap. 43), and all its forms are based on the stem svādhyāya- except this future, perhaps a passive opt. svadhyayeta SP 391.7, and the gerundive svādhyālavya Samādh 22.8 (prose). It corresponds to Pali saiihāuati. To the latter, Pali has two causatives (lieiger 188.1), sajjhāpayati (°peti) and sajjhāyāpeti, both meaning 'causes to study, teaches'. The first is formed according to our § 38.56, the second according to our § 38.53 (sujjhāvali being treated as an *a*-present). In our language this pair of equivalent causatives appears to have led, by proportional analogy, to the creation of an irregular future svādhyāsyali (and corresponding gerundive svādhyātavya; the passive opt., if sound, would perhaps be a further development of these forms, as if on a present svädhyäti), beside the regular svädhyäyisyanti (SP 478.7, prose). The formula is *svädhyäyäpeti (sajjhäyäpeti): *svädhyäpeti (sajjhāpeti) = svādhyāyişyati: x (srādhyāsyati).

38.32. Aorists: samādapinsu SP 51.8 (Kashgar rec.). -- snapinsu 1.N 93.21, 22. -- pranāmi Mv iii.147.19; prati-māne (to pratimānayati) iii.248.12; kathe iii.408.15 (prose; to denom. kathayati; cf. kathaye iii.413.2). — vijňapi Samādh 8.22. — darši 'displayed' Gy 254.25 (to daršayali); nidarśi Gv 255.6; sthapi Gv 383.17. -- toşi Śiks 346.6.

38.33. Gerunds: SP (all vss) harsitva 25.7; pățitva 85.12; bhāvitva 92.8; 116.4; vivarjitva 99.4; cintitva 117.4 (text cintelva); tārilva 149.12; prajňapitvā 194.3; pūjitva 217.9; choritva 251.5. - LV (all vss) janitva 50.14; 422.4; janitvā 234.7; upusthapitvanā, 54.15, so read with v.l. for text °pisva nā. — My (all but the first prose) parivarjitvā My i.12.14; pralisammodilvā i.273.10 (here Senart em. °etvā); ii.443.17; iii.450.20; mārilvā ii.248.5; veditvāna (so with mss.; to vedayali) iii.61.11. - pūritvana KP 20.23 (vs; to pūrayali). - And many similar forms.

Denominatives

38.34. Denominative formations present little that is different in principle from Sanskrit denominatives. Attention may first be called to a considerable number of onomatopoetic formations, like the 'quasi-denominatives' noted in Whitney 1066c. Most of them are reduplicated: khalakhalāyali, gudāyali (gudugudāyale, gulugulāyali, gu-luguluyali), ghuņaghuņāyale, ciļiciļāyali (°le, ciļac°, ciļicifiyati), culuculāyati, tuņatuņāyati (cf. tuņa, noun), pilipalipäyeti (note pres. eti after -äy-), pharapharäyate, sarusarāvate, huduhuduvati.

38.35. Of denominatives in simple -ali, without -ya-(Wh. 1054), among the few clear cases noted are sisyate 'learns' (Dict.), prusnati 'questions', duhkhati 'hurts', and svādhyāyali 'studies' (§ 38.31), apart from forms based on past passive participles (§ 28,19). But in view of Pali sukkhali, the inf. suskilum may be regarded as based on **suskale* (or °*li*) 'get dry' rather than **suskayale.* -- On the anomalous hyper-Skt. *nisedyali* see Dict.

38.36. The usual Skt. formation in -avali or -avali (°te) is abundantly represented. The formal and semantic distinction between -ayali, active, and -ayale, middle (Wh. 1059c), is probably extinct; since our language commonly substitutes active for medio-passive endings, we find intransitive denominatives in -āya- with active endings: avasyāyanta ūşmāyanto bāspāyantah LN 251.8 and 12 (prose), 'turning to hoar-frost, to steam, to vapor'. Sometimes the denom, appears with the Mindic ending -eli, which may represent either -ayali or -ayali ("te): abhyanyeli. pratibhägeti, sekheti. And sometimes it is recorded only in extra-present forms which are ambiguous: nepallhilaor nipacchita-, ppp.

38.37. Some of the following denominatives are more or less obscure in form and origin: but the same is true of some Sanskrit denominatives. For denominatives based on ppp. stems see § 28.19. Examples of forms in -ayali are: ārāgayali (see Dict.) and virāgayati, ārogyayati, udānayati, *vijaļayati, vilomayati, *svastyayati (to svasti); bharayati 'fills' is a Mindic creation to the Skt. bharita, a denom. ppp. from bhara, see Chap. 43, s.v. bhr (2).

38.38. Examples of -āyali (°le) forms are: *kāikṣāyali (ppp. °āyila = Pali kaikhāyila, 'doubted'), to kāikṣā; kelāyali (kela°?) 'cares for' = Pali id., Pkt. kelāyai (etymology uncertain): dhandhāyali to BHS dhandha, Pali dandha; prajhāyale, to prajhā: mamāyale, °li 'cherishes', to mama, 'my'; maitrāyale, °li, 'shows love', to BHS maitrā, Pali meltā; vrstāyate 'rains': rahāyali 'is hidden' = Pali id. (Skt. Gr. ^cle), related to rahas; hrijāyali (hrij[°]) 'is ashamed', related in some way to hrī, and if mss. of Divy 484.8 are right, prasarāyilā (see Dict.), 'delivered (of a child)'.

38.39. There are also a few forms in *-iyali*, in which the origin of the *i*-vowel is sometimes obscure. Besides the onomatopoetic *citicitiyali*, § 38.34. I have noted *artiyali* (and various other related forms or alternative spellings, see Chap. 43) 'is grieved', connected with *ārti*; *baliyali* 'grows strong, prevails', to *bala*, very likely influenced in form by the comparative *baliyas*; *sāliyati* or (v.l.) *sādiyati*, also *svādiyati*, 'takes pleasure in', cf. *šāla*, Pali *sādu*, *sātiya*, adj., 'pleasant', and *sādiyati* 'agrees to', interpreted by Childers and PTSD as connected with root *svad*, while they connect *sāta* with the (BHS and) somewhat dubious Skt. *šātu*, lexical only except for *atišāta* Gīt. 10.9: *sābhiyati*, apparently 'is slow, delays' (of obscure origin).

38.40. On *adattādāti (fut. °dāsyatha), anomalous if denom. from adattādāna, see § 28.53.

Causatives in apayati (apeli, aveli)

38.41. Like MIndic generally, BHS has a freely productive suffix $\bar{a}payati$ (or the like), which unlike the forms in *ayati* has definitely causative meaning. It has no other function, and the cases where it seems to have lost it, that is to be used in the sense of the simplex, are minimally small in number (§ 38.58). Some of them I consider formal blends of two other forms. It seems certain that its very extensive development in MIndic, where it is the only causative formation that is productive on a large scale, is due to this semantic precision, which gave it a great advantage over its older but ambiguous rival *ayati*. For occurrences in Skt. of these 'bastard stems with $\bar{a}p$ added (as in Prakrit)', as Whitney calls them, see his Hoots, p. 238.

38.42. On the origin and use of this suffix see my article in Language 22(1946).94-101. Following M. Leumann, IF 57.224 (somewhat similarly, and earlier, Tedesco, JAOS 43.389), I believe that it started from pairs of participles in -ita, noncaus., and -apita, caus., from certain old bases containing rootfinal Skt. å (heavy bases); especially sthå and its cpds., also certain cpds, of dhå, då, and må 'measure'. Such roots regularly formed causatives in Skt. -payali, which in Skt. is nearly limited to roots in final ā. So Skt. had ppp. uthila 'arisen', and caus. ulthapita 'raised, lifted up'. By analogy with such pairs, other participles in ita, of whatever origin, formed (first, apparently) causative participles in apita. Such participles in ita were regular with presents in ayati (= MIndic eti), especially causatives and denominatives. But also, from early times, pples. in ita are associated with presents in ati, e.g. RV raksita to raksati, AV patita to patati, likhita to likhati. And any ppp. in ita could form an analogical causative ppp. in apita. So, on the model of utthapila (etc.) to utthita, such forms arose as BHS pradipāpila 'caused to be lighted, ordered lighted', to pradipila 'lighted' (ppp. of pradipagali 'lights', originally caus. of pradipyate 'is alight'), a causative to a causative; or dandapita 'caused to be punished', to dandita 'punished' (ppp. of denominative dandayati 'punishes'); or likhāpita 'caused to be (ordered) written', to likhita 'written' (ppp. of likhati 'writes'). From such participles in āpita it was a short and inevitable step to new causative presents like pradīpāpayati, daņdāpayati, likhāpayati, and a host of similar Mindle forms, with full inflections.

38.43. Tedesco, i.e., assumed as starting points deti : dila : dāpayali, and jeii (Skt. jayali) : jita : jāpayati. But jāpayati, tho recorded in VS, itself needs explanation as much as the Mindic forms, of which it is in fact only an early-recorded example. And the pairs of participles clearly furnish a better basis than the presents for the proportional analogy. They were much used, for example, even in Skt., and still more in Mindic (in Pkt. except AMg., exclusively), in the place of preterite verb forms (cf. Renou, Gr. sete, p. 509, § 377); e.g. utfhilah 'he arose', Mbh. (Calc.) 13.192.

38.44. Note particularly the following case, instructive as proving philologically the importance of the ppp.; Leumann i. c. 224, 226 hardly does justice to it. In Classical Skt. the ppp. ucchrita 'clevated', as adj. 'lofty', and the gerund ucchritya 'raising, lifting up', are the only common forms of ud plus root *sri*. Finite forms are rare. The active ucchragati 'raises' (trans.) does, indeed, occur; the middle ucchragate 'rises, stands upright' is however only Vedic, dead in Classical Skt. (tho quoted grammatically). Even more striking are cpds. of ud-*sri*: according to BR and pw they seem to show no finite forms whatever, only the ppp. abhyucchrita, procchrita, samuechrita.

38.45. In Pali also ussita 'high, lofty', and samussita 'clevated', or (much oftener) figuratively 'haughty, arrogant', are common forms. In normal Pali usage (as to some extent even in Skt.) they are adjectives, not participles. That is, they were hardly associated any longer with the verbs to which they once belonged systematically. These verbs, in fact, have only a shadowy existence in Pali, it would seem. PTSD cites usseli only twice, both times with object yāpa (Jāt. iv.302.5; vi.203.9), and samusseli just once in opt. samusseya (AN i.199.15; PTSD 'to be grandiloquent', not very felicitously), in a situation which clearly proves that it was a back-formation from samussila (used in the first line of this passage, AN i.199.1).

38.46. But Pali has a participle ussāpila, and a present ussāpeli, 'lifts up, raises' (c. g. a banner, a parasol, an elephant's trunk), also 'heaps up, makes a pile of' (sand, wood, bones), which is common; and even a nom. act. ussāpana 'raising' (of a banner; cf. samudānana to samudānita, § 38.3). It seems to me certain that this group was based on ussila = Skt. ucchrila, not on the rare usseti = ucchrayati. It is, of course, not certain that in this case the caus. pp. ussāpila was formed before the caus. pres. ussāpeti (once started, the analogy could have led directly to the latter): but this present, too, was surely based, and meant 'makes high' (ussila). The meaning, besides the rarity of usseti, makes it much less likely that it was based on usseti 'raises'; if it had been, it should have meant rather 'causes to raise' or 'to be raised'.

39.47. BHS has ucchrāpayati, a Sanskritized form of ussāpeti. Indeed this form is reflected even in the (late) Vedic language: VS 23.26 ucchrāpaya 'lift up!' This, like many other Vedic forms, is to be understood as a Mindle borrowing.

38.48. Once this process was established, however, it is obvious that presents in *āpayati* could be made as causatives to any presents in *ayati*; and, since we have seen that these are interchangeable in Mindic with *ati* presents, they could be made as causatives even to these. Hence such forms as BHS *āhvāpayati* 'causes to summon', to Skt. *āhvayati* 'summons' (Pali *auheti*); and sporadically in Skt. or even Vedic (VS. *jāpayati* to *jayati*, and *ucchrāpaya*, above), under Mindie influence.

38.49. As noted just above, BHS and general Mindic causatives in $\bar{a}payali$ (or equivalent) may be based on presents in either ayali (Mindic also eli), of any of several different types, or ali. In the following lists an attempt will be made to sort the $\bar{a}payali$ forms according to different categories to which the presumed simplex seems to belong.

38.50. In some cases doubt may exist, or may at first sight seem to exist, as to just what primary verb was the base of the apaya-causative. The doubt is often more apparent than real. So $ksam \bar{a}payati$ 'asks pardon' is, I believe, certainly a fundamentally MIndic caus. (tho it

occurs also in Skt.) to kşamali (°le) 'endures. is patient'. It is true that there is a Skt. ksamayali, having the same (caus.) meaning as ksamāpayali, and some might ask, why could it not be the base of the latter? The answer is double. First, there is no proof that ksamayati survived in MIndic; no Pali *khameli or Pkt. *khamei seems recorded. Second, and more important: as I showed in Language 22.94-101, and as will be more fully shown below. the meaning is against that theory. If based on kşamayali, ksamāpayati should mean 'causes to ask pardon', or the like. We should not expect it to be, as it is, a synonymous substitute for kşamayali. Hence, even when a Skt. ayacausative persists in MIndic, as in Pali nivalleli = Skt. nivarlayali 'sends back, makes to (re)turn', the new MIndic caus, Pali nivallapeli 'sends back' is based not on nivalteti but on nivallali = Skt. nivarlali (°te) 'returns' (intrans.).

38.51. Our lists will include all BHS examples (so far as noted by me) formed in MIndic fashion in apayali (āpeli, āveti), even if they are recorded in Skt. (as ksamāpayati, ucchrāpayati, jāpayati, above), since these are clearly of the same type, and I think they are best regarded as MIndicisms in Skt. I shall not, of course, record regular Skt. forms like *dā-pavati*: nor such Skt. forms as ropayali, to root ruh, which are indeed aberrant in Skt. but do not show the same pattern with the MIndic apayali forms (nor any MIndic pattern; they are stray anomalies which concern Skt. alone).

38.52. Forms in apayati from roots in a., Before. listing the apayali forms, however, we must refer briefly to the fact that, as in Pali and Pkt. (Geiger 180, Pischel 551), the root-final a of some roots may be shortened before -paya-. This occurs, in fact, in Skt., notably with jñā, where jñapayali is recorded even in AV, jñāpayali only from the Brähmanas according to Whitney, Roots; also snapayati (AV +) for snap° (RV +); glapayati for (older) glāp°; mlapayali (late and rare) for mlāp°. (Leumann's attempt to explain these short a forms. IF 57.227. seems to me unconvincing.) Presumably by analogy with the very common Skt. equivalents jñapayali: jñāpayali (and perhaps also snapayali: snapayali), MIndic has such forms (generally paralleled by forms in -ap-) from other roots in d which in Skt. never shorten to -ap-. The BHS examples (in addition to jñapayali and snapayati, inherited from Skt.) are parallel to Pali forms: sthapemi (also thap-; thap-), and cpds. utthap-, upasthap-, pratisthap-, vilhap-, to sthā (Pali thapeli, and cpds.); samādapeti (Pali id.) to da. The ppp. nidhyapta (-citta, Siks 130.13; also noun nidhyapti), apparently to dhya, corresponds to Pali nijjhatta; it is paralleled by Skt. jñapta (B+, Whitney, Roots) from ina. On all these see Chap. 43. We proceed to list the BHS forms in causative apayali (or equivalents), and first those based on present stems in ali.

38.53. Causatives in apaya (ape) to present stems in a.

adhiyāpita (= Pkt. ahijjāviya) 'caused to study' Mv iii.394.9 (prose), to adhiyate 'studies'.

icchāpita (Pali icchāpeli, Childers; AMg. icchāvei) 'caused to desire' Divy 256.1, to icchali; pralicchāpaya (Pali pațicchăpeti, Pkt. padicchāvei) Mmk 27.24, to praticchali.

karāpayet, KP 158.7 (prose), to karali (Dict.). Possibly, however, this is a corruption for karap^o (§ 38.57), which occurs shortly after in KP 159.12, and which is the usual form.

kartāpayel (not in Pali or Pkt.) 'shall cause to twist' Mmk 57.7, to *karlali (Dict.) 'twists'.

ākramāpilo (7 by em.) Mv iii.363.9, 'caused to be buried', to ākramati, 'buries'; see Dict. kridāpayati (Pali kiļāpeti) 'causes to play' Mmk

458.20; °petha Mv i.227.12; ii.151.13; 153.8 etc. (all prose);

to kridati. Skt. kridauali is apparently not recorded in Mindie

ksamāpavali (late Skt.; Pali khamāpeli, Pkt. khamāvei) 'asks pardon: takes leave of, says farewell to': "payanti LV 379.6 (prose); [°]penli Mv iii.359.10 (prose); [°]payām āsa Mmk 640.19 (vs); [°]peyam (1 sg. opt., v.l. [°]payeam) Mv i.37.8 (prose); [°]payilvā LV 38.14 (prose), etc., to ksamati. No *khameli (Skt. kşamayali) seems recorded in MIndic. kşipāpayet (Pali khipāpeli) 'would cause to be thrown',

Mmk 528.27 (prose); ksipāpauilavua id. 49.19 (prose); to ksipali.

khanāpayilvā (AMg. khanāvai, and Pali khanāpeti. Childers) 'having had dug' Mv i.352.21 (prose), to khanati; utkhanāpayişyam Mv ii.437.13 (prose); utkhanāpita Mv ii.437.16 (prose), iii.363.11 (prose); pass. ulkhanāpiyamāna My ii.439.6. The Skt. caus. is khānavali: khanavām āsuh is recorded in BR once, Ram, 2.80.12, but modern editions read khān°. Cf. Asokan khānāpeti, blend of khānavati and khanāpayati; also khānāpāpila, § 38.63. khādāpaya (Pali khādāpeti) 'cause to eat' Mmk 56.26

(prose); khādāpila Mv i.24.16; ii.479.10 (both prose); to khādali. No *khādeli (Skt. khādayali) seems recorded in MIndic.

aāvāpavalah 'cause (dual) to sing' MSV iii.16.6: to qayali.

grhnāpayali (Pali ganhāpeli) 'causes to take' Mmk 28.9; 29.13; 564.10; grhnäpelvä My iii.393.15 (prose); to BHS grhnati.

grathnapayatah 'cause (dual) to twine, wind' MSV iii.16.5 (prose), after grathnilah; to *grathnati, grathnäli.

carāpayeyam, opt. (Pali carāpeti) 'I would cause to move, proceed' SP 53.4 (vs); to carati.

chindāpayişyāmah (Pali chindāpeli) 'we shall cause to be cut off' Mv iii.403.9 (prose; so mss., ed. em. vāmi); to BHS and Pali chindati.

jalpāpayati (not in Pali or Pkt.) 'causes to speak' Mmk 421.21; to jalpati.

pratijāgarāpetha (not in Pali or Pkt.) 'cause to be cared for, looked after, prepared' Mv ii.156.9 (prose), to pratijāgarati (Dict.).

jīvāpayali (also late Skt.; Pkt. jīvāvei; apparently not in Pali) 'causes to live' Mmk 574.24; 696.20 (both prose); to jivati. Skt. jivayati has no recorded descendant in Pali or Pkt.

anujānāpehi (Pali anujānāpeli) 'cause to consent' i. e. 'ask permission of' Mv i.256.14 (prose); anujānāpetvā

'having got permission' My ii.174.17 (prose); to anujānali. dasāpayati (Pali dasāpeti, Childers) 'causes to bite' Mmk 462.13 (prose); to dasati. Cf. next.

damiāpayet (AMg. damsāvei) 'shall cause to bite' Mmk 463.6 (prose); to damiali 'bites' (Chap. 43); cf. preceding.

Skt. damiayali seems not to be preserved in MIndic. dahāpaye (not in Pali or Pkt.) Mv iii.20.8, dahāpelvā 9 (vss) 'cause to burn, have burned'; to dahali. Cl. dahapayati § 38.61.

duhkhāpayasi (= Pali dukkhāpeti) Mv i.179.19 'afflict'; to duhkhati (Dict.), properly denom., but no *duhkhayali is recorded.

uddhvasyāpayilvā, to uddhvasyale 'is insulted', pass. of uddhvamsati, § 38.13.

palliyāpayişyāmi (cl. Pkt. paltiāvei) 'I shall cause to believe' SP 288.5 (prose); to paltiyati (Chap. 43) 'believes'.

pivapayet 'should make drink', and pivapayelu-(kama), Kashgar rec. for text pāyayeyam, pāyayilu-, SP 322.2 (prose), to pibali.

opunāpayilavya (Pali opunāpeli) 'to be caused to be winnowed' Mv iii.178.5, to *opunati = *ava-punāti, see Dict.

bandhāpetha, impv. (Pali bandhāpeti) 'have bound' My 1.258.9 (prose; mss. ^oyelha); ^opelvā My ii.175.6 (prose); ^opayitavya Mmk 529.20 (prose); to bandhati (= badhnāti).

(budhyapayati, 'causes to become enlightened', to budhyate, is implied by the nouns budhyapaka, opana. which may be genuine forms; see Dict.) paribhumjāpelvā (not in Pali; but acc. to Childers

bhuñjāpeli; cf. AMg. bhumjāvei) 'having caused to enjoy

or eat' Mv iii.148.15 (prose), to paribhuñjali. muñcāpelha (Pali muñcāpeli) 'cause to be freed' Mv iii.297.12 (prose); ^opayali Mmk 640.18 (vs); 704.10 (prose); to muñcati.

ramapetha (Pali ramapeti) 'cause to take enjoyment' Mv ii.151.13; 153.9 (both prose) etc.; ^opehi id. 430.4 (prose; in same line *rāmehi*!); iii.5.16; 6.3 ff. (all vss); ramāpayate Mink 571.20 (prose); abhiramāpetha My ii.103.8; 144.15 (both prose); "penti Mv iii.70.14 (prose); to ramali.

abhiruhāpayitvā (not in Pali or Pkt.) 'having caused to mount' My iii.39.6 (prose); to abhirnhali (cf. next).

rohāpayali (not in Pali or Pkt.) 'causes to grow' KP 30.1 (prose); to rohali 'grows' (not to the early Skt. caus. rohayati which was superseded by ropayati). Cf. prec., and ropāpayasi § 38.57.

likhāpayati (Pali likhāpeti, etc.; also late Skt.) 'causes to write': "payed SP 51.4 (vs); 226.4 (prose); 342.5 (vs); KP 159.19 (prose); [°]payilbā Divy 547.6 (prose); abhili-khāpayilavya Mmk 68.9 (prose); to likhali.

lavāpayilavya (Pali lavāpeti) 'to be caused to be mown' My iii.178.4 (prose); to *lavali, Pkt. lavai.

vandāpayisyāmo (Pali vandāpeli) 'we shall cause to pay homage to' Mv ii.26.6 (prose); vandāpayelsuh Mv i.223.13 = ii.26.9 (vs); to vandai.

upaviśāpila (not in Pali or Pkt.) 'caused to sit down' My ii.103.4; 180.15; 446.14, 15; iii.36.7; 406.2, 4 (all prose); °payilvā My ii.479.10 (prose); to upavišati. — samvišāpehi (not in Pali or Pkt.) 'cause (allow) to cohabit' Mv ii.430.6 (prose); to samvisali.

parivisāpita (not in Pali or Pkt.) 'caused to be waited on' Mv ii.439.4 (prose); to parivisali.

nivullāpiya (ger.; not in Pali or Pkt.) 'holding down, causing to sink down' Mv ii.172.7 (prose); to nivuțlati (Dict.), of which the ger. nivulliya occurs in preceding line.

nivarlāpayilum (Pali nivallāpeli) 'to cause to return or turn back' Mmk 86.3 (prose); to nivartate (also Pali nivalleli = nivarlayati); pratinivartāpayitu (read [°]paya tu) Divy 346.10 (prose). (Pali only paţinivatleti = [°]vartayati.)

vardhāpayilvā 'having wished success to' (jayena, in addressing a king) Mv i.310.2 (prose); vaddhāpayilvā Mv ii.421.11; vaddhāpita (mss.; ed. em. vardh°) Mv i.287.16 (prose); aor. vardhāpaye Mv ii.38.1; to vardhale (in such phrases as disiyā vardhase, etc.). Cf. vardhayati, Dict. Pali has vaddhapeli and Pkt. vaddhavemi, but whether they are used in this same sense I am unable to say

varşāpayati (Pali vassāpeti) 'causes to rain' Mmk 464.1 (prose); to varsali. (Skt. varsayali; acc. to Childers Pali

(prose); to varşalı. (Skt. varşayau; acc. to chuers Fan also vasseli, and PTSD cites caus. pple. vassila.) śikşāpayati (Pali sikkhāpeli, Pkt. °ānei) 'causes to learn', i. e. 'teaches': °payasi LV 125.19 (prose); °peli Mv ili.362.5, °pehi id. 4, °payişyāmi id. 1 (all prose); °pila Mv ili.394.9; Divy 198.3; 369.27; 391.8 (all prose); Mmk 57.8 (-arcshi te fibeauti (Stt fibeauti hut no Pali or Pkt (prose); to siksati (Skt. siksayati, but no Pali or Pkt.

*sikkheti, except once Pkt. sikkhayanta, pr. pple., Sheth). sisyāpayali = prec., but may be caus. to sisyale, see Dict.: ^opayisyasi LV 126.12.

śayāpito (Pali sayāpeti) 'caused to lie down', to śayati. So I think we must read instead of sayayito Mv iii.39.6; 40.2 (prose); Senart reports no v.l. except sayito once in one ms. The graphic error y for p is commonplace; I cannot find any possible analogy for the form printed by Senart and cannot believe that it was originally intended.

nişîdāpayitvā (Pali nisīdāpeti) 'having caused to sit down' Mv ii.435.13; 446.9; iii.70.5 (all prose); nisīdāpetvā Mv iii.298.18 (prose); to nisidati.

?prasavāpitā (em., mss. °vāyitā, see § 38.38), 'caused to bring forth, bear' (a child): Divy 484.8 (see Dict.).

pralinisrjāpayisyanli 'will make renounce' iv.140.7; to °srjati. MSV

harāpayasi 'you let take' MSV iv.207.2; to harati,

38.54. Causatives in apaya (ape) to presents in aya (e). The aya (e) is lost before the causative suffix.

38.55. Causatives from aya presents which are neither causative nor denominative. There is, of course, no reason in principle for separating these from the rest, from the point of view of our language or of MIndic generally. The separation is made here solely for the convenience of Sanskritists, who may be interested in the different historic backgrounds of various aya (e) presents which were provided with apaya (ape) causatives.

ānāpeşyam (Pali ānāpeli) 'I shall have (cause to be) fetched' Mv ii.103.5 (prose); "payisyam Mv iii.125.17 (prose); "pita, pple., Mv ii.76.4; 103.3; iii.38.7 (all prose); to anayati (Pali aneti) 'fetches'.

ucchrāpayanti (Pali ussāpeti) 'they raise' (banners) Mv ii.112.18 (prose); °payetsu (aor.) Mv ii.343.22; to ppp. ucchrita (Pali ussita), see §§ 38.44-47.

pardhāpayala, 'cause to be cut off', to Skt. Dhātup. vardhayali, 'cuts': MSV i.119.14; 120.5. āhvāpayale (not in Pali or Pkt.) 'has (causes to be)

summoned' Divy 323.1 (prose); °pya (ger.) Av ii.53.5 (prose); to ahvayati (Pali avhayati, avheti).

38.56. Causatives from denominatives: cf. duhkhāpayasi, šişyāpayali, \$ 38.53, to denom. duhkhali, (?)šişyale; perhaps baddhāpayitavya, below, should be entered in that section, since the pres. seems to be baddhati.

andhakārāpita-(tva), caus. pple. to Skt. denom. pple. andhakārila (to andhakāra), 'made darkened', see Dict.

avadvārāpayilvā 'having had (the city gate, nagaradvāram) closed' Mv ii.490.1 (prose); to a denom. *avadvārayali (not found), presumably lit. 'gates off', i.e. 'shuts' (a gate).

ārogyāpaya (not in Pali or Pkt.) 'convey a greeting to, cause to be greeted' Divy 128.25; to arogyayati 'salutes' (Dict.).

kelāpayitvā, or kelap°, Mmk 708.28 (printed kaila°), to kelāyali (?Dict.).

cintāpayisyāmaļi (not in Pali or Pkt.) 'we shall cause to think (on)' SP 271.4 (prose); anuvicintāpayamānāļi 'being caused to ponder' SP 109.2 (prose), v.l. of Kashgar rec.; to (anu-vi-)cintayati.

citrāpayilavya (not in Pali or Pkt.) 'to be caused to be painted' Mmk 61.16; 68.13, 22 (all prose), etc.; to citrayati, citrita.

daņdāpayanli (Pkt. pple. daņdāvia) 'cause to be fined or punished' Siks 63.13 (prose); "payed 67.10 (vs); to dandayati.

prajňaplāpilam (so one ms.; v.l. prajňāpāyilam; Senart em. prajňāpayilom), with āsanam, 'a seat was caused to be provided' (for himself), Mv iii.93.3 and 4. The form would be ppp. to caus. of a denom. from prajñapla, 'provided' (§ 28.19).

baddhāpayilavya 'to be caused to be bound' Mmk 50.16; to baddha(ya)li(?).

bhaksāpayet (cf. Pkt. bhakkhāvaņa, noun) 'shall cause to eat, feed' Mmk 82.21 (prose); to bhakşayali (or bhakşati) 'eats'.

mantrāpayati (not in Pali or Pkt.) 'causes to announce (etc.)' Mmk 307.5; 566.24 f. (prose); to manirayati.

mundapayitva 'having had shaved' Divy 261.15 AMg. mundavei; to Skt. mundayati (Pali ppp. mundita) 'shaves'.

vijatāpayed 'shall cause to comb (card, untangle; object, wool)' Prāt 498.1; to Pali vijaleli, denom. to Skt. vijaļa.

veşlāpayitavyāni 'to be caused to be supervised' Mv

iii.178.1; to vesteti, denom. to vesti = Skt. visti, see Dict.

sabdāpitu (sabdāpayati allegedly 'summons', but rather 'orders summoned'; evidently MIndic; repeatedly in Rām., but chiefly in the late Book 7, see BR; cf. AMg. saddāvei, "viya) 'caused to be summoned' Mv i.272.14; ii.100.17 (both prose) etc.; "piya (ger.) Mv ii.110.14 (prose); "payitvā Mv ii.435.4, 13 (prose); to Pali saddāyati 'summons' (Skt. sabdāyati 'makes a noise, cries; calls, summons').

sukhāpaye (with mss.) 'I make happy' Mv iii.355.8; sukhāpita, °payitavya, °payitu- (Dict.); to Skt. and Pali sukhāyate (°ti) 'is happy'.

samhārāpayilavya, see next §.

hriyāpayanti (not in Pali or Pkt.) 'make ashamed' Bhīk 11a.5; to hriyāyati (Dict.) 'feels shame'.

38.57. Causatives from aya forms which are themselves originally causatives, or at least secondary-conjugation or '10th class' forms in -aya-, and apparently not denominatives. Note however that these causatives in dpaya (dpe) are not synonyms of the underlying 'causatives' in aya, but have the meaning of causatives in relation to them; they are so-called 'double causatives'. Cf. Lang. 22.97 f.

niskāsāpayilum 'to have (cause to be) expelled', caus. to niskāsayali 'expels', Divy 339.6 (prose). kārāpayali, °peli 'has (causes to be) effected or per-

formed', caus. to Pali kāreti 'performs', also Skt. kārayati, 'not infrequently in the meaning of the simplex' BR s.v. 1 kar, caus., 7. Pali kāreti apparently never has genuine caus. meaning; PTSD, tho entering it under the heading Caus., adds the inconsistent and confusing parenthesis "(Denom. to kāra)". Cf. §§ 38.5, 6; Lang. 22.95, 97. -kārāp-ayanti SP 15.3 (vs); -ayet KP 159.12 (prose); -aye Mmk 86.25 (prose); -ayi (aor.) SP 114.11 (vs); Mv i.118.5 (vs); -ayişū (or better -ayimsū), § 32.35, SP 50.16; 51.1; -ayisyam (fut.) Mv i.325.17 (prose); -ayamāna- Mv i.96.6 (prose); -ayi (ger.) SP 152.5 (vs); -ayilvā Mv i.52.13, 63.9; Mmk 49.17 (all prose); -enti Mv i.26.15 (prose); -etha Mv i.258.11; ii.111.5; 156.7 (all prose); -ehi Mv ii.73.18 (prose); -ita- SP 50.14 (vs); 110.6 (prose); Mv i.18.2; 19.11; 349.17 (all prose), etc.; Divy 375.3; common. Cf. karāpeti, above, § 38.53; also § 38.62.

To the cpd. alamkaroti, or rather MIndic alamkarati (e. g. Pali alamkara, 2 sg. impv., CPD), is found a caus. alamkārāpeti 'causes to adorn', found also in Pali (^opelha Mv ii.103.7; 111.4; 153.14; ^opehi Mv iii.441.20, all prose); no *alamkārayati or ^okāreti seems recorded anywhere; presumably alamkārāpeti is analogical to the uncompounded causative kārāpeti. The proportion would be karoti, karati : kārāpeti = alamkaroti, ^okarati : x.

kalpāpayitvā (Pali kappāpeti) 'having caused (hair and beard) to be trimmed', Mv ii 489.8 (prose), caus. to Pali kappeti 'trims'. Skt. also uses the pple. kļpta of the simplex kalpati in the meaning 'trimmed', of hair.

ghosāpita 'caused to be proclaimed' Mv iii.390.19 (prose); ghosāpayitavān (§ 34.8) 'he ordered proclaimed' SP 257.4 (prose); = Pali ghosāpeti, caus. to Pall ghoseti 'proclaims'; Skt. both ghosali and ghosayati in same mg., 'proclaims'.

olārāpila 'ordered brought down or out' Mv i.273.16 (prose); caus. to olāreli (Mv i.272.11 olārehi 'bring down!') = Pali id., Skt. avatārayali 'brings down', itself caus. to avalarati 'comes down'.

pradipāpila (not in Pali or Pkt.) 'ordered lighted,

1. Vin. iv.259, last line, and 260.2 fl.; bhikkhuniyo sambådhe lomam samharåpetvä ... vesiyähi saddhim naggä ... nhåyanli: misinterpreted in PTSD 'make gather' or 'grow', where it obviously means just the opposite, 'remove' (the pubic hair, in the manner of courtesaus). caused to be lighted' Mv iii.177.8 (prose); to pradipayati (Pali padipeti) 'lights', caus. to pradipyate 'is alight'. dhārāpayet (aor.) 'caused to hold' (a parasol) Mv

dhārāpayet (aor.) 'caused to hold' (a parasol) Mv ii.117.5 (prose); dhārāpayehi (so with mss.) id. ii.447.14 (prose); caus. to Pali dhāreti = dhārayali 'holds'.

bhīsāpayet 'shall cause to he frightened' Prāt 518.10 (prose); = Pkt. bhīsāvei, caus. to Pkt. bhīsei = bhīsayali 'frightens'; Lang. 22.97.

mârăpemi 'I cause to be killed' Mv ii.247.3 (prose); = Pali id., caus. to Pali māreti = mārayati 'kills'.

yojāpayanti 'they cause to be yoked' Mv iii.101.20 (prose); yojāpehi iii.441.20; °pesi (aor.) iii.442.2; °payitvā i.259.8; iii.443.9 (all prose); = Pali yojāpeti, caus. to Pali yojeti = yojayali 'yokes'; Lang. 22.98.

ārocāpila 'caused to be stated' My i.307.13 (prose); = Pali ārocāpeti, caus. to Pali āroceti = BHS ārocayati, ^ceti 'states'.

ropápayasi 'you cause to be planted' Mv ii.486.6 (prose); = Pali ropápeti, caus. to Pali ropeti = Skt. ropayali 'plants'. Cf. rohápayati, § 38.53.

vādāpita 'caused to be played' SP 51.11; 52.1; °piya (ger.) id. 52.4 (all vss); = Pali vādāpeti, caus. to Pali vādeti = vādayati 'plays (mus. instruments)'.

vārāpeya 'would cause to distribute' KP 158.3 (corrupt); 159.6-7; caus. to vārayali (Dict.).

parivesāpayisyam (text $\circ vei^{\circ}$) 'I shall cause to be waited upon' Mv ii.435.11 (prose); to either parivesati (rare in Skt. for parivisati; Pali only parivisati, but Pkt. both $\circ visai$ and $\circ vesai$) or parivesayati (not in Pali, but Pkt. pr. pple. parivesayanta), both meaning 'waits upon'; the rarity of parivesali suggests that parivesayati is the more likely source.

prative flåpehi 'have enclosed, surrounded' Mv ii.171.12; ve flåpita id. 15; to ve fleti = ves fayati 'wraps, encloses'; Lang. 22.98.

sodhāpayet 'would cause to clean up' SP 107.6 (prose); sodhāpita 'caused to be cleaned' Mv iii.298.8 (prose); = Pali sodhāpeti, caus. to Pali sodheti = sodhayati 'cleans'.

sobhāpayali 'causes to be adorned' (vasumdharām), Mmk 644.17; = Pkt. sohāvei (Sheth saphā karānā), caus. to Pali sobheti = sobhayati 'adorns'.

ghātāpayitum 'to cause to be killed' Mv i.132.8 (prose); ghātāpita 'caused to be killed' Mv ii.170.15; iii.166.8 (both prose); = Pali ghātāpeti, caus. to Pali ghāteti = ghātayati 'kills'; Lang. 22.98.

samhārāpayitavya 'to be caused to be gathered' (dhānyajālāni, khalahāneşu) Mv ili.178.5 (prose). This is not simple. One would at first sight connect it with samharali. But to this root Pali has no *samhāreli nor *samhārapeli; only samharapeli, which is recorded only with 'hair'' as the object; this (or 'hair and nails') is also the only recorded object of Skt. samhārayati; both mean 'remove' (hair, etc.). We are dealing here with a different word. With it I would connect a gerundive samhāraņijja cited by Sheth, which he associates with sambharei 'gathers'. What we have in samhārāpayitavya appears to be gdve. of a causative to this, in a Prakritic form (as in the Pkt. gdve.). The caus. is recorded nowhere else; sambhärei is otherwise known in Pkt., but not in Pali, and (as sam-bhārayati) doubtfully in Skt.: in BR (5.209) only cited from Ram. 1.11.3° sambhārayāśu nrpale sambhārān yajñasādhakān. As this passage suggests, sambhārayali (= sambhārei, samhārei), if it is the true reading, is probably a denom. to sambhāra, not a caus. to sambharali as PW takes it.

2. So Schlegel and Gorresio; others 1.12.3; NWRām. 1.9.40. The reading is not at all certain; more modern editions generally seem to read sambhārāh sambhriyantām te for the first pāda (so also NWRām.). **38.58.** Formal causatives in (ā)paya without caus. meaning. Renou, Gr. setc. 469, states that even the Skt. suffix -paya- is sometimes 'deprived of causative force', but cites no examples. (See Lang. 22.95, n. 5 for other but unfounded allegations as to such use of Skt. -paya-.) In standard Skt. such cases, if they occur, are surely extremely rare. They are very rare in BHS also. Nevertheless I have noted a few forms, inherited from Skt., which seem to me quite clearly non-causative in meaning (perhaps by analogy with the more frequent similar use of original causatives in aya; above § 38.23): parinirvāpayeyam Divy 90.10, 'may I enter complete

parinirvāpayeyam Divy 90.10, 'may I enter complete nirvāņa'. Here parinirvāpayati is certainly (unless we emend) used precisely in the sense of parinirvāyati. In SP 250.12 read parinirvāpayitu-kāmo with mss. (except two °vātu-), for text em. °vāyitu-. utthāpayet LV 335.15, 'let him set forth' = uttisihet.

ulthāpayet LV 335.15, 'let him set forth' = ultiş/hel. Similarly upasthāpayilum, 'to wait upon' LV 100.13; upasthapetvā, "waiting on' Mv ii.220.18 (see Dict. s.v. upasthāpayati 4).

Several such forms occur from snāpayali or snapayali, both regularly used in Skt. and BHS as causatives to snā, but in BHS also as intransitives, like the simplex: snāpi (aor.) LV 271.12 'bathed (himself), took a bath'; snapayanlam Mv ii.77.18 (prose), 'while bathing' (intrans.), and in the next sentence, ii.77.19, as well as ii.78.1, snāpayali, 'bathes' (intrans.); snāpilum (inf.; Senart em. snāyilum) Mv iii.12.5, 'to bathe' (intrans.).

samādāpayati (a MIndic creation tho its form is regular for Skt.), ordinarily causative, twice means 'assumes, takes on oneself' (= samādiyati, the BHS and Pali simplex); see Dict. s.v. (5).

38.59. Mindic forms in *āpagali* are rarely used except as causatives to the simplex on which they are based (even when, as in the long list above § 38.57, that 'simplex' is itself a causative, so that the result is a 'double causative', Geiger 182). I have suggested in Lang. 22.98 ff. that most, possibly all, the real exceptions may be explained as blend formations, as follows.

38.60. We have seen that the old Skt. causatives in ayati (and even some new MIndic forms of the same type) are still freely used, with caus. meaning, in MIndic. They are sometimes closely juxtaposed in recorded texts with new MIndic causatives in *āpayati* (*āpeti*, etc.). E. g., Mv ii.430.4 presents in the same line, and obviously as synonyms, rāmehi (from Skt. rāmayati, caus.) and ramāpehi (MIndic caus. based on ramati). Such collocations may easily be supposed to have led to blend forms like *rāmāpeti (this particular blend has not been noted), which would have the aspect of a form in *āpeti* based on rāmeti but having the same meaning. The only forms of this type which I have noted in BHS, and most if not all similar forms in Pali and Pkt., can be so understood without difficulty. Cf. M. Leumann, IF 57.223, top, and my article cited above.

38.61. So chedāpayati, 'causes to be cut off' (= ehedayati, id., still used in Pali, Mahāvaṃsa 21.18 and 35.43), occurs in BHS only in Mv iii.403.13 (prose) chedāpayisyati. Here it paraphrases, in free quotation, the equivalent chindāpayisyāmaḥ (so read with mss., Senart "yāmi), a normal MIndic caus. to MIndic chindati, 'cuts'. I believe it is surely a blend of this form with chedayati; and the collocation in Mv suggests how such forms must often have originated. (Of course they are not limited to such collocations; cf. Lang. 22.99, note 10.)

So also dāhāpayati, 'causes to be burned', Mmk 634.9, is a blend of the regular Skt. caus. dāhayati with the regular BHS (MIndic) caus. dahāpayati, § 38.53, to dahati 'burns'.

And bhojāpehi My i.305.10, if it means 'feed, cause to eat' (cf. Lang. 22.99 with note 11), is a blend of Skt. bhojayati with BHS (MIndic) bhuñjāpeti, both having the same meaning.

Analogical in a different way is *alamkārāpeti* 'causes to adorn', on which see § 38.57.

 $i\bar{a}y\bar{a}pilaka$, on which see Dict., if textually correct, seems to be a -ka extension of a ppp. of * $i\bar{a}y\bar{a}payati$, 'makes go to sleep', = Skt. $i\bar{a}yayati$, caus. of *iete*. It occurs in prose, MSV i.117.13, and perhaps should be emended to $isay\bar{a}^{\circ}$ (§ 38.53). The text later, i.120.6, replaces it by $i\bar{a}yayitah$, the regular Skt. form.

38.62. And finally, in LV 215.9, 12, 16 (vss) the ppp. kārāpitam, formally to kārāpeti (see § 38.57), seems to imply non-causative meaning like that which kārayati, kāreli regularly has in MIndic (and sometimes in Skt.), equivalent to karoli (or MIndle karali) 'effects, carries out', with synonyms of rajyam: aisvarya (adhipatyam, māreśvaratvam) kārāpitam, 'rulership was carried on'. No other interpretation seems to me possible; the Bodhisattva speaks of himself (in the instr.) as the person by whom, in past lives, rulership was 'exercised, performed'; he can hardly have meant that rulership was (by him) 'caused to be performed' (by someone else). Another MIndic form in -p- similarly used may be allipitavyam, § 38.67 below. Compare the rare, but undeniable, cases of Skt. forms in -paya- which in BHS are used as non-causative intransitives (§ 38.58), and the commoner cases of Skt. causatives in -aya- which in BHS are similarly used (§ 38.23).

38.63. On similar cases in MIndic see my article cited above. Note that many of the alleged cases of equivalence between \bar{apeti} and eti forms are simply misinterpretations; the \bar{apeti} form is really caus. to the eti form. So notably the Asokan forms (ppp.) likhāpāpila, khānāpāpila, with double $-\bar{ap}$ -, which have never been properly interpreted. They are genuine causatives to likhāpila, khānāpila, as I showed Lang. 22.100. (So now J. Bloch, Asoka [1950], 91 n. 4, 172.)

38.64. Miscellaneous and irregular forms in payati. A few sporadic irregularities need to be noted. $p\bar{a}payati$ 'causes to drink', Divy 398.17 (prose), fits the general pattern of Skt. roots in \hat{a} (sth $\bar{a}payati$, d $\bar{a}payati$, etc.), but the normal Skt. caus. from this root is $p\bar{a}yagati$ (Pali also $p\bar{a}yeti$), and I have found no * $p\bar{a}peti$ or descendant in Mindic. It should however be noted that $p\bar{a}payati$ is occasionally recorded in Skt.; see BR, who regard such forms as mere corruptions for $p\bar{a}y^{\circ}$. If they are right, the same might be assumed for the isolated Divy occurrence.

38.65. ucchrepayati 'lifts up, sets up' occurs repeatedly (see Dict.), and even forms a noun ucchrepana (so, not 'na). No such form seems to be recorded in Pali or Pkt. It is used exactly like ucchrāpayati, on which see above, §§ 38.44-47, 55, and seems to be a blend of that form with *ucchreti (Skt. ucchrayati) = Pali usseti, which has the same meaning.

38.66. Our language has a MIndic present alliyati (also alli²) 'comes, approaches', which however seems to be used also (Mv ii.190.5 ff.) in a causative sense, 'brings'. It corresponds to Pali alliyati (defined CPD 'to stick to, to lean against, to be attached to, to be fond off; caus. alliyāpeti 'to attach to, to apply, sew on'), and Pkt. alliai (alli⁰) 'come; enter; join' (jornā); 'resort to' (diraya karnā); 'embrace; be united'; AMg. alliai 'to resort to; to restrain or control the senses' (Ratnach.). It is obviously derived from Skt. áliyate (§ 3.4 a). To it, in our language but nowhere else as far as I know, are formed two causatives. One, allāpayati, I have found only in the pple. allāpita 'brought, caused to come', or perhaps 'ordered brought, caused to be brought' (either would be possible), Mv iii.362.3 (prose). It seems clearly based on the pattern ānīyate (passive) 'is brought': ānāpita 'caused to be brought' (§ 38.54) = alliyate (°ti) 'comes' (substantially = 'is brought'), or also 'brings': allāpita. There are other

13

verbs, such as $(\hat{a})diyale : (\hat{a})d\hat{a}pila$, which have intransi-

tives (originally medio-passives) in iya(li, °te) and causa-

tives in apaya; but the close semantic resemblance.

amounting to synonymity, between ani- and alli- seems

to point to ani- : anap- as the specific source of allap-

in our texts. I can only understand it as a blend of allāpayali, just mentioned, with the simplex allīyali. It is significant and important that this latter is used

with 'causalive' (transitive) meaning in our language, as we saw (see § 38.24 and Dict.). Thus alliyati and

allāpayali (if this meant 'brings') were or could be

synonyms; this made the blend-form allipayali all the

easier. The process is a sort of inverse of that which

gave rise to blends like chedāpayati § 38.61. Forms noted

are (all prose) : allipayati Mv ii.435.14, 'causes to be

brought'; 'peya iii.127.17 and 'peti iii.144.12 (in these two, the mss. have y for p, and the first should probably

be read "yeya and classed as opt. to alliyali as trans.;

while for the second, one ms. has alliyanti; read alliyati,

trans., 'brings'?); "pita, pple., i.311.2; ii.471.1; 472.11; iii.24.4; 408.4; 421.8; "piyanti, passive, iii.68.11; 405.15; "piyati, id., iii.127.4. In Mv ii.107.5 allipitāni seems to

mean 'were caused to be brought', as if caus. to alliyali

in caus. sense, 'bring'. On the other hand, in iii.288.9-10 allipitavyam, despite the p, seems to be intransitive, 'to be approached, gone up to'; perhaps read alliyitavyam?

38.68. Prakritic forms in aveti. By the side of

38.67. The other causative, allipayati, is commoner

forms in *āpayali*, *āpeti*, sporadic instances occur (hardly except in Mv; once in Sādh) of Prakritic *āveti*, with v for p. There are such forms in Pali (Geiger 38.5; *ubbiliāvita* beside °*pila*; *visīveti* = **visīgāpayali*), where they must be loans from other (Prakrit) dialects; I assume the same for BHS, where they are certainly not common. (To be sure, the mss., or some of them, record cases which are not taken into Senart's text.) There is no proof that the underlying Prakrit had *āveti* regularly and that p was substituted for v in the process of Sanskritization.

38.69. The following are examples of *āveti* forms based on presents in *ati*:

grhņāveli 'causes to be taken' Mv ii.467.3 (prose) = grhņāpayati § 38.53, to grhņati.

bhumjāvenli 'they cause to eat' Mv i.12.6 (vs) = (pari)bhumjāpeli § 38.53, to bhuñjali.

lambāvaget (Pali lambāpeti) 'he shall hang' (trans.) Sādh 170.20 (prose); lambāvija (ger., with mss.) Mv ii.172.14 (prose); to lambate, °li.

38.70. The rest are causatives to aya presents:

ghātāvita 'caused to be killed' Mv i.19.2 (prose) = ghātāpita 38.57, to ghātayati.

niveşfāvehi (not in Pali or Pkt.) 'cause to be wrapped' Mv i.273.14 (prose; mss. nices/h°; Senart em. vesf°, omitting ni), to Skt. nivesfayali 'wraps'.

sabdāvitvā 'having had summoned' Mv ii.421.5; 442.2; 491.9; iii.36.11; 390.15; ^oviyāna, v.l. for ^opiyāna (which Senart reads), ger., Mv ii.173.1; (no v.l.) ii.453.13; iii.287.16; ^ovita, pple., Mv iii.36.2 (in same line occurs sabdāpehi); 406.3; = sabdāp-, § 38.56, to sabdāyati.

from alli-.

But cf. § 38.62.

39. Intensive

39.1. Our language has cańkramati (= Pali cańkamati; AMg. pple. cańkamanta), by the side of the regular cańkramyale (which is found e. g. LV 377.4, 8; in the first, v.l. cańkramate). Examples: anucańkramāmi SP 55.1; °manti SP 81.8; cańkrameta SP 98.10; cańkrami (opt.) SP 344.1; cańkrami (aor.) LV 368.16; cańkramet LV 369.1 (all vss). Others, see s.vv. (anu)cańkramati, Dict., and Chap. 43, s.v. kram (5). Vedic language has a future cańkramisyali, implied in dat. sg. pple. cańkramisyate TS 7.1.19.3, KSA 1.10 (154.17), associated with cańkramyamānāga and cankramitāya. Perhaps the MIndic cankramali (cankamati) is analogical to the future (§ 28:13) or the ppp. cankramita (§ 28:30).

39.2. Similarly, SP 171.11 *jājvalanti* shows a thematic present active instead of the middle *jājvalyale*, intensive of *jval*. Such forms are, in this case, recorded in the Mbh (BR).

39.3. An intensive to hrī, not recorded in Skt. or MIndic, is *jehrīyate* Mvy 1829; *jehrīyamāņa* SP 108.6; *jehrīyante* Divy 39.7.

13*

40. Desiderative

40.1. Beside regular sussüsati (Skt. iuirūșate, °ti), Pali 'often' (PTSD) writes sussüyati, which Geiger 184 note 3 regards as a corruption. But Gv 520.23 (prose), 1st ed., reads iuiruyantam (read iuirū??), acc. sg. pres. pple., which would be a Sanskritization of the Pali form.

Is it somehow assimilated to a ya-present? However, Gv

15 it somenow assimilated to a garphesent? Provever, GV
2d ed. *iuirūsanlam* (by em.?).
40.2. Unreduplicated forms, ppp. (*pari-)bhuksila*, adj. *bhuksa*, and noun *bhuksā*, for Skt. *bubhuks*°, occur as in
Pkt. *bhukkhia*, *bhukkhā*; see Dict.

41. Syntactic addenda to the verb

Special uses of *as* and *bhū*; periphrastic verbal expressions

41.1. Occasionally forms of as and bhū are used as quasi-particles; cf. the Skt. usages mentioned by Speyer, Skt. Synt. p. 234. So $\dot{a}sit$ in SP 135.12 (prose) pūrvam cāham alpaprajāo 'lpapralisamvedy andhabhūto 'smy $\ddot{a}sit$, 'formerly I was of little intelligence or experience, blind'; lit. 'I am ..., it was'; and (a)bhūt in LV 193.11 mā bhūc chākyakuloditasya gamane chidyela vamšo hy ayam, 'may it not happen that, on the departure of the scion of the Sākya race, this family shall be cut off' (the verb introduced by mā appears to be chidyeta; (a)bhūt seems virtually a constructionless particle; Tib. seems to render by gran, 'perhaps').

41.2. The periphrastic verb expressions, involving use of participles with or without forms of as and bha as copulas, mentioned in Sen 56-7, are on the whole hardly characteristic of BHS in distinction from Skt. (See Renou, Etudes de gran. skte., Paris, 1936; on periphrastic use of present pples., especially 22-24.) The few stray examples which follow are very unpretentious. All of them could probably be paralleled from Skt., and most of them no doubt have many other parallels in BHS.

41.3. Present participles, with or without auxiliary, in place of finite verbs: (latru dhātryaś ca ceļivargāš ca sthāpilā abhūvan, parišesāh) šākyāh šuddhodanapramukhāh prakrāmantah LV 125.16 (prose), 'then the nurses and slave-women were halted; the other Sākyas including Suddhodana proceeded' (is abhūvān felt as carried over to serve as auxiliary with prakrāmantah?); na kadācid vayam vidhāryamānās Divy 177.4, 'we are never kept out'; ily evamādicaritāni pūrva caranta duşkarakrtāni RP 27.11 (vs), 'such-like actions I was performing of old, difficult tasks'; āsi (= āsil, as 1 sg.) caranta pure ahu (= aham) nityam RP 27.18 (vs), 'I was always performing of old' (here āsi as auxiliary). 41.4. Forms of bhavali with past pples. are frequent as periphrastic expressions, equivalent to preterites; Sen 56 mentions only the aorist (which he calls imperfect) abhūl. Examples of the present bhavali: (yehi ...) paribhuñjilāni bhavanli, 'were enjoyed (in life)' Mv i.19.4; (yāva...) karma vyanlikŗlam na bhavali i.20.5, 'until their action has been completed'; many others in this context, e. g. i.20.8; 21.14; 22.5-6, 7, 14; 23.3, 4; also i.233.3 katham te bhavati (note sg. for pl.) imd utpalāni kritāni, 'how did you get these utpalas bought?'; yathāham ... snānašālām pravisļo bhavāmi Divy 420.19-20, 'when I am (have, shall have) entered into the bathroom'.

41.5. Perfect babhūva: sambādham āpanno babhūva Mv i.131.6, 'came into dire straits'.

41.6. Aorist abhūl (common; a few others in Sen); cillam abhyupagalah ladā abhūl Mv i.64.15, 'then arrived at the thought'; asmābhir apy anye bodhisaltoā avavadilā abhūvan... na ca ... sprhācillam ulpādilam abhūt SP 101.4 (prose).

41.7. The future bhavişyanli is used with the perfect active pple. in lavanl, in a meaning which I find hard to distinguish from the future, in bhavişyanli khalu devasyālyayāt paścimā janapadāh pariprşiavanto Divy 224.3-4, 'you see, after your majesty's death, later people are going to ask' (hardly 'there are going to be people who ask', since the pple. in tavant could scarcely bear this meaning). The same future is also used with a nounformation in aka: katham etāni pravešakāni bhavişyanti Divy 249.4, 'how will they get to enter (lit. become enterers)?'

41.8. In *abhibhūya bhūto* Mv iii.286.3 (prose), 'has conquered', lit. 'has become having conquered', we seem to have a periphrasis consisting of a gerund with the past pple. *bhūta*, taking the place of an auxiliary.

41.9. Sen 61 cites *upeyivān* SP 257.8 (prose) as the only periphrastic use of the perf. pple. in *vāņs* noted by him in BHS (cf. Renou, Ét. de gram. skte. 25).

42. Constructions with $m\bar{a}$

42.1. The use of ma in BHS differs both formally and semantically from that of standard Skt. (on which see Speyer, Skt. Synt. §§ 353-4, 405, VSS §§ 193, 237; Renou, Gr. scte. pp. 412, 439, 451. 462, 513, 525). Normally Skt. uses $m\bar{a}$ with the augmentless aor., less often with the imperfect and impv., in prohibitions, also sometimes in negative purpose clauses ('in order that not; lest'). There are stray occurrences, particularly in the epic (see Kulkarni, ABORI. 24.83-97), of its use with other verb forms, such as opt., fut., and pres. indic., but they are abnormal.

42.2. In BHS it is very freely used with the opt. (for which the regular Skt. negative is na), with the fut. and pres. indic., and with no finite verb at all. As to meaning, while still common in prohibitions and lestclauses, it is also much used in expressions of strong negative wish or hope, often translatable by expressions of fear. These most commonly relate to future events, but may also relate to the present or past. This use shades over into expressions of dubiety, uncertainty, or worry (almost invariably about something deprecated or regarded as undesirable; rarely only about something the speaker regards as improbable). Such expressions are sometimes clearly interrogative in tone: 'I hope ... not?' 'it isn't so, is it?' 'isn't there reason to fear that ...?' 'isn't (regrettably)...?' Presumably in actual speech these were distinguished by interrogative intonation, if in no other way. In default of any way of indicating this in the text, it is often difficult or impossible to be sure whether it was present or not.

42.3. All these meanings of $m\bar{a}$ occur freely with the opt., fut. and pres. indic., and with no finite verb at all. The non-Skt. meanings occur rarely with the augmentless aorist, and so far as I have noted never with the imperative; these two forms are nearly or quite restricted to their Skt. uses. Augmented preterites are only rarely used (as also in Skt.).

42.4. The negative $m\bar{a}$ is very often accompanied by a particle, most commonly haiva (ha-eva); frequently khalu, which seems interchangeable with haiva (cf. Gv 116.20-25 where both occur in parallel expressions); also tāvat and nāma. But no particle is necessary; all the forms and meanings listed may occur without any. In prohibitive expressions sma is sometimes used (LV 192.22) as in Skt.; other particles rarely occur with mā in this sense.

Prohibitions

42.5. No examples of augmentless aorist or imperative are quoted, since these are normal Skt.

Augmented aor.: mā ānanda tathāgatam (so read, Foucaux, Notes p. 113) prāmāņikam akārsuh LV 89.9-10, 'let them not make (consider) the T. measurable!'.

Opt.: mā tvam svayam gacche, cetim presehi Mv ii.169.17, 'don't go yourself; send a maidservant'.

Fut. with haira: luām samdišāmī ... mā haira luam bālajanasya agrato bhāşişyase sūtram ... SP 97.5-6, '1 command you, do not recite this sūtra before fools'.

Fut. with sma: mā sma kadācil samgītim vichetsyatha LV 192.22 (prose), 'do not ever interrupt your singing' (in verse parallel 193.12 samgili ma chetsyatha, without particle).

Fut. (in the first also present) without particle: mā prāņinam ghātayisyatha mādaltādāsyatha mā kāmesu mithyā carişyatha mā mṛṣām vakṣyatha, yāvan mā me vijile adharmam ulpudyale (pres.), mādharmacāriņo rocetha (= rocayatha, pres.) LV 15.9-11, 'do not kill, steal, be unchaste, lie, and so on, in short, let no wrong take place in my kingdom; do not tolerate evil-doers'; ihaiva tvam kurma kurusva mā bhūvo 'nyalra gamisvasi SP 106.8, 'do your work right here; go no more anywhere else'; pūrvam-gamo bhava tvam mā bhesyasi pašcimas tesām LV 185.14, be thou the first of them! be not the last!'; mā dāsyathā jīvitam LN 335.13, 'don't grant him life'; mā esu bhoksyatha bhājaneşu LV 383.12, 'don't cat in these dishes'; mā lumhe camum apaharisyatha My ii.173.14, 'don't take away the coffin'; similarly mā ... daršayisyasi 173.18-174.1; mā śocisyatha mä utkanthisyatha 215.1–2; mā atra allīsyatha 253.5, 'don't go here!'; mā dāsyasi Divy 12.22, 'don't give'; ārya listha mā praveksyasi 82.3, 'Sir, stop, don't enter!'; mā ... mahāsamudram avalarisyasi Av i.199.8, 'don't embark on the high seal'

Pres. indic., without particle: $m\bar{a}yuktayogina vadesi$ etal SP 93.11, 'do not say this to those who have not practised discipline'; $m\bar{a}$ me bhūyo āgacchasi Mv i.244.7, 'do not come to see mc any morel'; Senart, failing to recognize this common use of the present, interprets the verb as aor., which is possible (§ 32.62) but by no means necessary; devi mā bhūyo padminim pasyanāya nirdhāvasi Mv ii.451.1, 'queen, don't go out again to visit the pool'; mā marmabhedini vacāņsi vaksi Gv 407.1, 'do not speek ...'

Gerundive without particle: mā me kalhamcid upasamkramilavyam anyena Mv iii.225.12, 'let no one else approach me by any means.'

Strong determination

42.6. The prohibitive $m\bar{a}$ shades off, first, into expressions of strong determination, the subject being third or first person. They are hardly what we should describe as commands, yet there is more than mere wish or hope in them.

Fut. with khalu: mā khalu kumāro 'nabhijnāta evābhiniskramisyatīti LV 186.14-15, (the king established strong guards around the Bodhisattva's dwelling) 'thinking, The prince is not going to (I'm determined that he shall not) depart (from worldly life) quite unrecognized!'

Fut. without particle: $m\bar{a}$ bodhisalloo 'bhiniskramisyalili LV 192.21-2. (same situation; the king stationed men to watch him night and day.) 'thinking, The Bodhisattva is not going to (I won't let him) depart'.

Fut. 1st person, without particle: ma vahesyam etam mrgam pāšato Mv ii.236.13, '1 (a hunter who has trapped a deer, but is moved to pity) mustn't (certainly won't) take this deer from the thong' (i. e. 4'll let him go free).

'Lest, in order that not' clauses .

42.7. The augmentless aorist, and sporadically other forms, are so used in Skt. (Speyer, Skt. Synt. § 405 Rem. 1). Aorist (common); a single example, without particle:

mā vo bhūd dirgharātram arthāya hilāya sukhāya LV 409.7, text. But Tib. (which Foucaux mistranslates) obviously read anarthāya ahilāya asukhāya, and the original must have had this (cf. the familiar Pali phrase anatthāya ahilāya dukkhāya, or asukhāya). '(Do not address the Tathāgata with āyuşmant,) lest it result in harm, disadvantage, and unhappiness for you for a long time.'

Opt. with haioa: $m\bar{a}$ haiva... sattbah... kinikrtasamjnā (so for KN kili^o, see Dict.) bhaveyur na ca tathāgale durlabhasamjnām utpādayeyur... viryam nārabheyus... na ca... utpādayeyuh SP 319.6-9, 'lest creatures... should get their fancies calloused, and should not get the idea that the T. is hard to find', etc. (note na in neg. of mā-clause, 'lest...not').

Opt. without particle: lasmād bhavanlam na yācāmi, mā me vidvesanā bhavel Mv iii.419.8 (vs), 'therefore I do not beg of you, lest I become subject to hatred'.

Fut. with haiva: mā haiva ksepsyanti šruņitva dharmam, 'lest on hearing the dharma they reject it' SP 291.12; mā haivelah kālam krivā narakesūpapatsyasa (ms. °tsyala) iti Av i.272.12, 'lest dying from this life you be reborn in (one of) the hells'.

Fut. without particle: mā laksanā akuśalā vikalpayisyanli dvijasamghā Mv i.224.4 = ii.27.4, (Maheśvaragods came to cast the child's horoscope) 'lest the throngs of brahmans should erroneously distinguish unfavorable signs' (cf. vikalpa, Dict.); mā tena udakarāksasena khajjisyasi ii.78.4, 'lest you be eaten by that water-ogre'; devi talo bhaişajyagudikāto pibanāye na lubhati, mā devī pāpakam putram janayisyati ii.432.4-5, 'the queen didn't get (wasn't allowed) to drink from that medicinal pill, that the queen might not bear an ugly son' (as had been predicted); mā iha anāhāro marişyali iii.132.1, 'lest he die of hunger here'; mādharmeņa rājyam kāraya, mā narakaparāyaņo bhavisyasili Divy 59.5, 'don't rule unjustly, lest you become headed for hell'; maitam kuścid drstvā śāsane 'prasādam pravedayişyatiti Divy 190.13, 'lest someone, seeing him (a drunken monk), should say unseemly things about the Doctrine'.

Pres. indic. without particle: (Māra says to the Bodhisattva: Do as I bid you,) mā veņuyasi haritām va chinadmi te 'dya LV 338.14, 'less I cut you off today like a green bamboo stalk'; mā paritapyāmi Mv ii.248.7 (so with v.l. for text °yāsi, § 27.8), 'lest I suffer for it'; mā vāņijakā sāgare vinasyante (= vinaš°) Mv iii.355.11, (the Bodhisattva in a previous birth resolves to give up his life) 'that the merchants (his companions) may not perish in the sea'.

Strong negative wishes or hopes for the future

42.8. The line is not easy to draw between this and the last two divisions, between negative determination or purpose, and negative wish or hope. The problem here becomes, in fact, hardly more than one of translation; and sometimes different choices seem about equally good.

Opt. without particle: māsmād vrajet sūrato, mā bhūc chākyakuloditasya gamane chidyeta vamśo hy ayam LV 193.10-11, 'may the Gracious One not depart from here; may this lineage not be cut off on the departure, perchance ($bh\bar{u}t$, § 41.1), of the scion of the Sākya family'. The second mā clause is here assumed to be coördinate with the first; but it might also be a 'lest' clause, subordinate to it.

Fut. with haiva: mā haiva kālum kurişyali sokaprāplā I.V 234.5, 'Oh, I hope she won't die in her grief!' (said of Gopā by her attendants, after the Bodhisattva's departure).

Fut. with khalu: mā khalv ayam... kālam karişyaty akrtārtha eva LN 252.9-10, 'I do hope he will not die without attaining his goal!' (said by gods on observing the Bodhisattva in a trance).

Fut. with no particle: nivarlanam ma ca bhavisyalili SP 197.7, 'and in the wish (ili, with the thought), may (let) there be no turning back!'; ma maivamulpa... bhiksavo bhavisyanti LV 88.13, (Buddha has just predicted the future existence of evil monks who will not keep the law; Ananda exclaims:) 'Oh, I hope that there will never come to be monks of that sort!' (God forbid!).

Pres. indic. with no particle: $m\bar{a}$ ca kadācid bhūtaguru nātho anlarahāyale Mv i.175.9 (so, or °hāyante, which is unmetr., all six mss.; Senart em. °hāyalul; in a series of stanzas in praise of Buddha, spoken by a god at the dharmacakrapravartana), 'and may the Lord, the Guru of beings, never disappear!'

Fears for the future (also referring to present and past)

42.9. The preceding usage leads almost imperceptibly into another and very large group of cases in which disquietude or fear (already present in some of the preceding) becomes the dominant emotion; the negative wish or hope is strongly colored by dread lest it be disappointed: 'I am afraid that...' or 'I hope that... not'. In a number of these, it is quite possible that questioning intonation was present (cf. below).

Opt. with haiva: $m\bar{a}$ haiva mama kālakriyā bhavel, sarvam idam aparibhuklam vinašyel SP 102.11, '(I am old and have no son;) I'm afraid I may die and all this (wealth) may be lost before it is enjoyed'; $m\bar{a}$ haivetvarapunya devamanujā... no šaklā siya... LV 271.3-4, (the Bodhisattva resolves to go now to the Bodhi-tree;) 'I fear that (otherwise)... gods and men would not be able...' (could also be interpreted as a 'lest' clause); $m\bar{a}$ haiva (so v.l., text heva) me cyāvaye āsanālo Mv ii.319.14, 'I fear he may cause me to fall from my seat'.

Opt. with haiva läval: na me putro 'sti, mā haiva tāvad aham aputro kālam kareya Mv 11.424.17, 'I have no son; I fear I may die sonless'.

Opt. with laval: mā lāvad aham vadhyo dandyo bhaveyam SP 104.7, 'I am afraid I may be subject to execution or punishment' (or question? 'isn't there danger that I may ...'?)

Opt. with khalu: mā khalv ime tapasvinas tādriam mahāratnadvīpam na gaccheyur iti SP 187.10, 'I'm afraid these wretches may not go to such a great jewel-island'; mā khalv ima ekam eva buddhajāānam śrutvā dravenatva pratinivartayeyur naivopasamkrameyuh SP 189.1, 'I'm afraid they...would turn back and not draw near' (parallel to the preceding).

Opt. with no particle: $m\tilde{a}$ dani doşam pi labheyam atra SP 113.3, 'I'm afraid I may now get even some disaster here' (or question, 'may I not..., isn't there danger that I may...'?)

Fut. with haiva: (once, haiva tāval Mv ii.144.7) mā haiva...dhaksyathānayavyasanam āpatsyatha SP 73.6-7, 'I'm afraid you will be burned, will get into trouble and disaster'; here translation by 'lest...' would be possible, but not in the closely parallel 73.13 mā haivāham ceme ca...anayavyasanam āpatsyāmahe, 'I fear that both I and they will...'; hā dhik śākyakulasya rddhi vipulo mā haiva samdhaksyate LV 194.6, 'woe is me! I fear that the abundant fortune of the Śākya family may be burnt up!'; yadi kumārasya śāntehi dhyānehi citlam abhiramati, mā haiva tāvad asitasya rşisya salyam vyākaranam bhavisyati Mv ii.144.7, 'if the prince's mind takes pleasure in calm meditations, I fear that the sage Asita's prediction may come true'; so, rājño bhavati: mā haiva yathā asitena... vyākrto tathā bhavisyati ii.151.11-12; likewise ii.153.7 (read mā haiva ...); mā haiva (so read with v.l.) aham pi tathā eva hanisyāmi ii.170.18, 'I fear I too may be slain in the same way'; so rājā... lrasto samjāto, mā haiva me sanagarajanapadam šāpena bhasmīkarisyati (mss. bhasmam[°]), ii.213.13-15, 'the king was frightened (thinking): I fear he will reduce my land, including city and country, to ashes by his curse'; mā haiva (so read with v.l.) śramaņo abhibhavisyati 276.19, (Māra says) 'I fear the monk will conquer me'.

Fut. with khu (= khalu): $mathbf{a}$ has a state of has a ha

42.10. In the following the fear applies to a present situation, instead of to future events, altho in the first example one of the verbs is a future of probability:

Fut. with haiva, then infin. plus kama with haiva and with khalu (all parallel): mā haivāyam māro bhavisyali...mā haiva mamānlarāyam karlukāmah sarvajāatāyām, mā khalu mām visameņa pathā praņetukāmah, mā khalu me dharmamukhānlarāyam karlukāmo...Gv 116.20-25, 'I am afraid that this is Māra', etc. (or perhaps question, 'isn't this perchance Māra' and doesn't he want' etc.)

In the rest either the pres. indic. is used, or there is no finite verb form; in both cases with or without khalu kācid vipralipatlih LV 104.3-5, (the king speaks to Asita) 'Why are you weeping, O sage?... I hope there is nothing going wrong with (no disaster predictable for) the child'; in verse parallel to this, kim brähmano roditi, mā vighnam khalu paśyale 'yam asilah sarvārlhasiddhasya me LV 111.3-4, '... I hope this Asita doesn't see any danger; same situation in My 11.32.13 kim bhagavan tam kumāram drstvā rodasi, mā kumārasya kāmeid vipattim pašyasi, ... I hope you don't see ...'; asaigasya mā khalu visaye sajjate manah, mā khalv amarair asamcodito vismarati pürvapralijñäm LV 209.5-7, (gods, seeing the Bodhisattva intently gazing on the women of the harem, say:) 'I hope that, unattached (as he is), his mind is not attaching itself to the world of sense-objects; I hope that, un-prompted by the gods, he is not forgetting his former undertaking'; mä me bhiksu imehi adhisthänehi (so mss.; loc. in sense) düsayallti Mv i.244.14, (he became afraid,) 'thinking, I'm afraid the monk is ruining me in these quarters!' (the speaker then proceeds to murder the monk); kim idam ... šokārdilo upavisto 'si? mā kimcit šarīrasya pratipidām pašyasi, mā dhanaksayam upalaksasi, mā paracakrabhayam upasihilam My ii.145.12-14, (Suddhodana to the Bodhisattva:) 'Why are you sitting thus in sorrow? I hope you don't see any bodily affliction, or notice any loss of money, or danger from a hostile power drawn nigh' (but this ma clause may be a question, 'you don't, do you?'...).

Interesting and somewhat exceptional is $r\tilde{a}j\tilde{a}$ kuśo durāsado dusprasaho, mā imam muhūrtam anayavyasanam āpadyasi Mv ii.493.1-2, (the doorkeepers speak to King Kuśa, whom they do not recognize, as he tries to enter his own palace:) 'King Kuśa is hard to get at, dangerous; you're in danger of getting into trouble and disaster this moment!' One might render 'I'm afraid you are...', but the speakers do not really participate in the emotion of fear.

42.11. There are also cases referring to past events; the rendering may be 'I only hope...not (but am afraid...)'. The verb, or substitute for it, is always preterite in effect; but the opt., or future of probability, of the roots as or *bhū* may be used with past participles as a sort of periphrastic conjectural preterite (or equivalent of 'future perfect').

Ppp. with haiva (and with or without auxiliary): mā haiva syāmakasiri simhena vyāghrena vā... vihethito bhūd iti Mv ii.215.15 (prose), 'I only hope Ś. has not been harmed by a lion or tiger...! (but am afraid he has)'; in the corresponding verse account, aho kaslam, mā haiva rsikumāro viheļhilol 225.16-17 (without bhūt); mā khalv anenaisām vastrāņām arthe kumāro jīvildā vyaparopilah syāt LV 228.15, 'I'm very much afraid the prince has (or, I hope he hasn't) been deprived of life by this person, for the sake of these (the prince's) garments (which the person is wearing)'; repeated substantially 228.20-21; rsisya elad abhūsi: mā haiva me mātulakā kenacid viheļhilā bhavisyanti Mv iii.431.7-8 (fut. of probability), 'the rşi thought: I fear my uncles must have been harmed hy some one'.

With no verb form, not even a participle: deva ma maya durnriyam Divy 553.17-18; a king sees a fatal omen as he is accompanying his queen's dancing on a via; he drops the via; she says: 'Sire, I hope I haven't danced badly!' (or perhaps a question, 'I haven't danced badly, have I?')

Ouestions

42.12. In quite a few of the preceding, interrogative intonation may have been present, as I have suggested in several instances. There is no doubt that some $m\bar{\alpha}$ clauses were questions. In other words, the concern, perturbation, or deprecation of a feared condition, which we have found so common, is certainly at times cast in interrogative form. In actual speech this was no doubt made objectively clear by the intonation. Since the writing cannot show this, we frequently cannot be sure whether to assume it or not.

In the following the verb prechati proves it: raja lam sārthavāham prechati: dharmabhrāte, mā śrānto 'si klānto vāsi' viśrama... Mv iii.350.17-18, 'the king asked the (visiting) merchant: Brother in religion, aren't you tired or weary? Rest...' In view of this, we may probably assume a question in the same situation where no verb of asking is used: svāgatam śrona māsi trşito bubhukşito vä Divy 9.16-17, 'welcome, Śrona! aren't you thirsty or hungry?'; so also Divy 11.4; 335.8; similarly svāgatam... aren't you tired with going on your pilgrimage?'; māsi klānto LV 352.20, (Bodhisattvas to the Buddha just after his enlightenment) 'aren't you tired?' (there follows immediately reference to his strenuous fight with Māra's host).

Another case of a clear question is māsi dāsī Bhīk 16a.4, 'you are not a slave-woman, are you?' (in the formal ritual of questioning the candidate for initiation to the order; she must reply na hi, 'no!'; there follows mā āhrlikā, and a long list of such terms, asi being omitted; all require negative answers; in default of them, the candidate could not be initiated, hence the deprecating mā). Doubtless a question is also meant in mayā snuṣābhihitā: vadhuke mā toam pranītāni praheņakāni bhakṣayituāsmākam lāhāny upanāmayasi, Divy 14.2-3, 'I said to my daughter-in-law: Daughter-in-law, aren't you eating the fine presents-offood and giving me poor ones?' and similarly 14.12.

42.13. Two or three times md, with or without haiva (the verb, if any, being pres. indic.), is used in negative phrases precisely parallel to positive phrases containing kaccid. In Skt., and in BHS, kaccid marks a question to which the speaker hopes for an affirmative response (BR and Speyer Skt. Synt. §§ 412, 413 are not happy in using the Latin terms nonne and num, which are not relevant; a better statement in Renou, Gr. scte. p. 511). Its negative in Skt. is kaccin na; see e. g. Mbh. (Crit. ed.) 2.5.7 fl., a long series of such questions, some positive, some negative. So also in BHS, e. g. kaccin na parilapyase Divy 559.7. But in BHS a question 'I hope...not?', in other words the negative of a kaccid clause, may be expressed rather by md (sometimes with haira; but not with kaccid!): kaccid bhagavan saltbdh sväkärdh...md haiva bhagavatah khedam ulpādayanti SP 301.1-2, 'I hope, Lord, your creatures are...? and that they do not cause you trouble?'; so in the verse version, mā haiva khedam janayanti (pronounced janenti) 301.7; in SP 429.5 (and similarly 6, 7) mālīva rāgacaritāh (sc. le saltvāh), as the preceding, also associated with positive kaccid phrases; mā kiņcil parihīgate MSV üi.67.21, 'I hope nothing is lacking?'

42.14. In all the preceding the verb, if any, was pres. indic.; in most of them no particle was used, tho haiva was sometimes found. The opt. with haiva (or nāma) is also used in what seem clearly interrogative expressions; it implies more dubiety or conjecture. The condition inquired about is regularly regarded as something to be feared or deprecated: mā haiva paribhrameyuh SP 73.4, 'isn't there danger that they would wander around (go astray)?'; tat kim manyase sāriputra, mā haiva tasya puruşasya mṛṣāvādah syāt SP 76.5, 'so what think you, Sāriputra? Isn't there danger (or, likelihood) that that man would be guilty of lying?' (the answer is 'nol'); mā nāma dahyeyur SP 87.4, 'isn't it likely (to be feared) that they would be burned?'; tat kim manyadhve kulaputrā mā haiva tasya ... kašcin mṛṣāvādena samcodayet SP 322.13 (cf. 76.5 above), 'so what think you, gentlemen? isn't there danger that some one would accuse him of lying?'

42.15. Finally, in some questions astonishment, verging on incredulity, seems to be the dominant emotion, rather than deprecation or fear, tho these may also be present, as in the first:

Pres. indic. with tāvat: rājāha, mā iāvan mamāpicchasi ghātayitum Divy 380.11, 'you don't mean to say you want to kill me too!?' (clearly a startled question; he has just received a hint of this astonishing and terrifying fact). Mere incredulous amazement, without fear or deprecation, seems im, lied by the opt. with haiva in LV 130.7-8: ko nv ayam nisanno? mā haiva vaišravaņo dhanādhipatir bhavel? āho svil... (The rşis, seeing the young Bodhisattva in meditation, reflect.) 'who is this that is seated here? Surely it can't be Vaišravaṇa? Or else (āho svil, marking an alternative question!)...' (there follows a series of other supernatural beings; finally in line 11 bhavisyati, fut. of probability, 'will it be...?', without negative).

With the particle *tāval*, and a past pple. or pres. indic., astonishment (in this case somewhat ironical) is also dominant in mā *tāval tavaikajanmikasya maranabhayāt tava rājaśriyam prāpya harşo notpanna*h Divy 422.7, 'you don't mean to say that thru fear of death for you in one single life you, having obtained royal majesty, felt no joy?'; mā *tāvad*... tava na jāyate (similarly) 422.18.

With khalu and no verb (in an equational sentence), astonishment (tinged with annoyance) seems to be suggested by $m\bar{a}$ in LV 330.7-8 $m\bar{a}$ khalu mūdhah ajno 'tha yuşmākam rūpākrlim na pasyati, (Māra speaks to his daughters:) 'Isn't he (the Bodhisattva) mad, or a fool? or doesn't he see your beautiful forms?'

na for mā with imperative(?)

42.16. Instead of $m\bar{a}$, the regular negative with the imperative, na is used at least once, if Senart's em. is right, in *na tuvam manasikara* (but mss. *manasi tathā*) Mv iii.4.2, 'do not concern yourself with it!' I doubt the em.

43. Synopsis of verb forms

This synopsis is meant to include typical examples of (1) forms of verbs which are not known to exist in standard Sanskrit, and (2) non-Sanskritic forms of verbs which do exist there. The first class will be found entered in the Dictionary, the second in the appropriate sections of the Grammar, to which references are often given when it might be troublesome to find them. For the verb, this list, with the detailed table of contents at the beginning, will, it is hoped, be a satisfactory substitute for an Index (which has not been provided for reasons set forth in the Preface).

accha-ti, sit: Pali id., Pkt. acchai Pres. acchati Mv ii.379.5; Šikš 298.4; impv. acchā LV 343.2

añja-li, anoint: Pali id.; § 28.4 Ppp. añjita- Mv ii.234.17 (= Pali id.)

anva-li, go: acc. to Senart, Mv i.381, from Vedic rnvati

Pres. anvati Mv ii.457.15; anvāmas SP 168.9 (with Kashgar rec.); pple. anvanto Mv ii.212.16; anvamāna- Mv ii.441.6 Aor. anvi SP 258.10 (so read, Dict.) Ppp. anvita- Mv iii.144.15 Ger. anviya Mv iii.145.1

Inf. anvitu-kāma- Mv ii.211.10 Gdve. anvitavya- Mv iii.178.7

alliya-li, -le, see arli°

ad-, eat (Skt.) Inf. attave Mv iii.295.12; 299.13 (= Ved. id.); § 36.14

*adattādā-li(?), steal, denom. Fut. adattādāsyatha LV 15.9 (§ 28.53)

adhiya-le, adhi-le, see i- with adhi

*andhakārayali, Skt. ppp. °kārita, denom. pple. Caus. pple. andhakārāpita-, see § 38.56 and Dict.

abhidakşiņa-li, or °ņīya-li, pass around to the right, denom.

Aor. abhidakşiņetsuh or °ņīyetsuh (mss.) Mv ii.416.11

abhyange-li, anoint, denom. to abhyanga, unction: AMg. abbhangei, °gai

Pres. abhvangeti My iii.154.14; 156.7 (here v.l. abhvangi, which could stand as ger.)

Ger. abhyangitvå, v.l. °getvå Mv iii.23.16 (so with mss.)

?abhra-li, or MIndic abbha-li, go: sautra root in Skt. and Pali

Ger. abhritvā or abbhitvā, perhaps intended by mss. Mv i.187.7 (Dict.)

arcaya-li, honor (Skt.) Aor. arcayetsuh Mv i.211.18 f.

arliya-li, -le, be grieved; also ar(t)ti°, atti°, atti°, once ārti" (? v.l. atti"; no other occurrence has unambiguously a-); ästiryati (false hyper-Skt.); see also ardiyati (s.v. ardaya-li) and rliyate: Pali atțiyati (and other spellings, see CPD). Denom. from arti-, with MIndic shorten-ing to arti- (this is not normal in Skt., see BR)

- Pres. ārtiyate (read prob. with v.l. atti°) MadhK 297.2; arttiyati Mv ii.242.13 (mss.); arttiyante Mv i.219.17 = ar(t)tīyanti ii.21.19; hyper-Skt. āstiryati, only in Karmav, 47.26; 49.2 et alibi; nārtīyanti Mv i.89.18; nārtīyate Mvy 1830; opt. -mokșeņāritiyeran Bbh 282.8 (read, or understand—with MIndic epenthesis?—°arti°), repeated in 23 as -mokşeņa ritīyeran (read as before, or rtī°, see rtīyate); pple. attiyantā (or °to, mss.) Mv i.343.1; arttīyanto Mv ii.161.7, 12 f.; -cintayārtīyantam SP 108.6
- Ppp. sokārtita- Mv ii.480.7; jarārtita- LV 174.14

arthaya-ti, seek (denom., Skt.)

Pres. opt. prārthaye (2 sg.) Mv ii.483.15; pple. fem. prārthayantī, in pass. mg. (§ 37.18), Mv ii.69.7 Ger. prārthayi SP 213.10

Pass. abhyarthiyase Divy 249.30

ardaya-ti, distress (Skt.)

Pass. pple. ardiyamāna- Divy 39.7 (= Pali addiyati; cf. artīyati)

alliya-li, come to, approach: Pali alliyati (not in this sense), AMg. and other Pkt. alliai, allio (in this sense); doubtless to Skt. a-liyate; see §§ 3.4a; 38.66, 67

- Pres. alliyati Mv ii.210.19; (caus. sense) alliyanti Mv ii.190.5 ff. (or em. to allipenti?); opt. alliyeya (caus. sense, Senart em. allipeya) My iii.127.17
- Fut. allişyatha Mv ii.253.5, 7 (in 7 v.l. alliyişyatha); § 31.4
- Ppp. allina- Mv ii.32.1 (also in mg. attached to, in love with, cf. Pali above, Mv ii.100.1); cf. Skt. lina

Caus. (a) Ppp. allapita- Mv iii.362.3 (§ 38.66) (b) allipayati (§ 38.67) Mv ii.435.14; (°peti iii.144.12, read prob. ⁹yati, in caus. sense, see Dict.;) Ppp. ⁹pita i.311.2; Gdve. ⁹pitavya- iii.288.9-10; Pass. allīpīyati iii.127.4; ⁹piyanti iii.68.11; 405.15

1 aś-, eat: (1) aśno-li; § 28.58; not in Whitney's Roots, but occurs Brh. Ar. Up. (Mādhy.) 3.8.8 (K. aśnāti), and asnute Boehtl. Sprüche 3067; it probably was substituted for asnati by influence of the homonym as, ohtain

Pres. impv. aśnutha Mv i.196.14 (prose)

(2) asa-le (asa-le): cf. Pali asamana; formed on nonpresents like asisyati, asita, asitvā (Pali asissati, asita, asitvā), perhaps specifically on the future, § 28.14

Pres. ase (for ase, 1 sg. mid.) Mv ii.56.10 (read with mss. no ase)

2 aś-, obtain (Skt.)

Ppp. asta- LV 390.9 (Ved. in cpds.); § 31.11

(204)

as-, be (Skt.)

- Pres. si (after cons.) Mv iii.164.15; 207.8; smi Gv 231.5; § 28.64, and cf. § 4.7; asmo (1 pl.) SP 147.9; LV 115.15 (but both doubtful; § 28.60); opt. siyā Mv ii.353.15; siya LV 271.4; siyāti Mv i.45.1; syā Mv iii.314.6, 17; stiya(?) Siks 115.2, 3; asyāt Mv i.286.13; asyā Mv i.42.10; asya Mv ii.228.9; see § 29.41; pple. samāna- (= Pali id.) SP 72.15; Mv i.311.19 (§ 34.3; Dict. s.v.)
- Pret. āsi, āsi (1, 2, 3 sg., 3 pl.) SP 62.13 etc. (§ 32.20); āsīsu (v.l. °tsu) Mv ii.305.18 (§ 32.37)

-asa-li, throw (Skt.); also ny-asayati(?). Cf. Ap. nir-asahī (3 pl.), Sanatk. (ed. Jacobi) 495.2 Pres. opt. nyaseya LV 151.8

Aor. nyasi (3 pl.) LV 236.14; nyaşit (§§ 2.60; 32.45) LV 271.19; nyasayimsu LV 222.3 (perhaps read nyasyimsu, to ny-asyati, § 32.38)

Ppp. nirașța-, doubtful reading for nirasta-, Dict.

ah-: áha, say (Skt.)

(Preterite) 3 pl. āhansu(h), āhamsu(h), āhatsu(h), and with ah- for āh-, common in Mv (i.8.1; 18.7, etc., § 33.9)

āņape-li, āņāpe-li (and āņapa-li?), command: MIndic forms of ā-jñā (§ 2.15); Pali āņāpeti, Pkt. āņāvei and āņavei, āņavai; here both are found, but only in Mv; āņa- seems commoner, but Senart usually em. to āņā-; the mss. are very confused Pres. ānapeti Mv iii.125.21; 264.2; °pesi ii.490.8, 16;

āņapati, so probably read ii.25.15; impv. āņapehi i.362.2, 5 (so read); ii.108.15; āņāpehi iii.125.19

Aor. ānapesi Mv iii.299.9; probably read ānapasi i.223.4 Ppp. āņatta- Mv i.258.7

- Pass. (a) read anapyati Mv i.364.20; 365.4, 5
 - (b) read āņapiyanti Mv i.362.17; impv. āņāpiyatu (or °piy°) i.310.14

åp-, get, attain: (1) - äpuna-ti, also - äpune-ti: Pali -āpunati; to MIndic 3 pl. -āpunanti (9th class), replacing Skt. āpnoti, āpnuvanti; §§ 3.114; 28.17

- Pres. prāpuņeti Mv iii.418.11 (?taken by Senart as caus., but mss. corrupt and text dubious); opt. anuprāpuneyāma (with Kashgar rec.) SP 163.2; paryāpuņeyā Šiks 189.5; impv. prāpuņehi (mss.) Mv iii.270.14 Fut. paryāpunişyanti Mv iii.52.18
- Aor. prāpuņe Mv ii.302.21 (3 pl.); °ņetsuh (mss.; 3 sg.!) Mv iii.338.20; anuprāpuņi Mv iii.67.2,4; prāpuņi (3 pl.) Sukh 52.16; paryāpunisu SP 181.9 (prose; Kashgar rec.)
- Ger. prāpuņitvā Sukh 24.12; paryāpunitvā KP 130.4
- Inf. paryāpunitum KP 139.9; anuprāpunitum (so, or °puņ°, read for °puritum) LV 46.5

Gdve. anuprāpuņitavya- Mv iii.287.17

- (2) -āpaya-ti, caus. (Skt.)
- Pres. in non-caus. meaning, prāpayanti, they get, Mv i.30.15; 31 15 (§ 38.23)
- Fut. in non-caus. mg., prāpisya (1 sg.) Dbh.g. 12(348).11 (§ 38.31)
- Aor. prāpaye (v.l. °yet) Mv ii.37.15; anuprāvetsuh (for °petsuh, § 2.30) Mv iii.52.8

ārāgaya-li, win, acquire, propitiate (see Dict.)

- °yati Mvy 2394; 7602; Divy 173.4; ārāgeti Mv Pres. ii.363.4; opt. ārāgaye Mv ii.120.1 Fut. ārāgayişyati LV 434.6, 7
- Aor. ārāgayī SP 27.12; 384.6 Ppp. ārāgita- SP 393.5
- Ger. ārāgayitvā SP 385.6, ārāgetvā Mv iii.415.4
- Inf. ärägavitu-kāma- Šiks 244.3

ārūdhaya-li, make mounted, cause to mount (denom. to ārūḍha)

Ger. °yitvå Mv iii.146.14

ārogyaya-li, (a) salute; (b) send greetings to (grüssen lassen, caus. to a): denom. to ārogya Pres. °yayati Divy 129.5; 273.25; impv. °yaya Divy

- 273.19 (all these b)
- Ger. ărogyayitvā Divy 259.11 (a)

Caus. (= b, above): impv. ārogyāpaya Divy 128.25

ārtīya-li, sec artīyati

āsa-le, -li, sit (Skt.); also -āsaya-li, -le, non-caus. Pres. adhyāsayati LV 18.8; samupāsayante Dbh.g. 41(67).9; opt. āse Mv iii.422.16

Inf. paryupāsanāya SP 425.3

Pass. āsīyati Mv iii.86.3

āstīryati, hyper-Skt., secartīyati

āha, see ah-

i-, go: (1) e-li (Skt.)

- Pres. upeti Mv iii.110.1, 6; enti (3 pl., § 28.60, as in Pali and Pkt.) LV 74.11; opt. sameyā (§ 29.42) Mv ii.404.8; upeya (ib.) Mv iii.369.21; impv. entu Mv iii.206.15; samabhyupaintu Divy 399.9
- Fut. etsye (? § 31.25) Mv i.257.18 and (em.) 11
- (2) *ya-li: based on 3 pl. yanti. Implied by:
 Pres. opt. 3 sg. yeya (or, with mss., yeyam) Mv ii.233.3 (prose; §§ 28.7, 29.31)

i- with adhi: (1) adhiya-le (rare in Skt.) and adhite (regular), study (Skt.): Pali adhiyati, Pkt. ahijjai

- Fut. adhīsyati Mv ii.77.14 (§ 31.7)
- Inf. adhiyitum Mv ii.77.13 (§ 36.2); [adhitum, see ib.] Caus. pple. adhlyāpita- Mv iii.394.9
- (2) Caus. adhyāpaya-ti (Skt.)
- Pres. impv. adhyāpehi Mv ii.77.16

iñja-le, -li, move (once recorded iñca-li): Pali id. (Cf. Geiger 41.1)

- Pres. iñjate LV 259.7, 20; opt. iñjeya Mv ii.408.5; pple. an-iñjamāna- SP 5.10; iñcati Mahāsamāj. Waldschmidt Kl. Skt. Texte 4.195.4 (Pali correspondent iñjayum)
- Ppp. injita-(tva-) Mv ii.411.8 (to caus.?)
- Inf. injitum Mv ii.340.13, 17; 341.11 (transitive; to caus.?) Caus. inf. injayitum Divy 185.10
 - irya-li, see ir-

1 is-, seek: (1) iccha-li (Skt.), also icchayati, iccheli (non-caus.)

Pres. icchesi Mv ii.483.8; opt. icche Mv ii.427.7; iccheyā SP 278.8; pple. praticchayanti Mmk 65.12

- Fut. praticchisyati Mv ii.435.12
- Ppp. icchita- Mv iii.24.19; praticchita- Gv 349.7; praticchita- Mv ii.165.19
- Inf. praticchitum Mv i.310.4
- Gdve. icchitavya- Mv i.280.7
- Caus. impv. praticchāpaya Mmk 27.24; Ppp. icchāpita-Divy 256.1
- (2) esa-li (Skt., at least Epic, in certain cpds. as anv-eș°)
- Pres. eşati LV 364.9; impv. eşatha (text °ta) LV 379.11; pple. eşan LV 242.10; eşamāņa- RP 5.14; eşāņa-(§ 34.4) Mv iii.2.9; Divy 559.5
- l'ut. adhyeşişyam (1 sg.) Mv ii.108.5

Aor. adhyesi (3 pl.) LV 416.3; adhyesisu LV 414.10 (ed. 'tu)

Ppp. paryeșța- SP 101.8 (?); adhyeșța- LV 48.11; 412.7: § 34.13

Ger. anveşiya Mv ii.179.2

Pass. adhyesanti Mv i.106.12 (§ 37,32); adhyesyante LV 397.5

(3) -esaya-ti, with adhy-, pary-, in same meaning as eşali

Pres. adhyeşayāmi Lank 7.13, 15; °yasi Divy 329.3, 7; opt. °yeyam LV 415.22; °yet Mmk 38.14; °yeyuh Suv 123.9; Siks 354.5

Fut. paryeşayişyāmi Mv i.232.3

Ger. paryeşayitvā SP 105.13

(4) isya-ti, -te (Skt. in cpds., as anvisyati)

- Pres. işye (1 sg.) Divy 476.16 (prose); işyasi Divy 560.8 (vs) (5) -esyali, with adhy-, pary-, in same meaning as -eşa(ya)ti. Not recorded elsewhere; blend of -eşati and -isyali? or phonetic corruption?
- Pres. paryeşyāmah Mv iii.59.15 (v.l. °şāmah); °şyanti Suv 227.2 (textually doubtful); opt. adhyeşyeyam LV 394.2; °yemahi LV 394.7

2 iş-, send: preşaya-ti (Skt.)

Pres. opt. preşayi SP 113.6

Aor. preșayi Mv ii.55.9; apreșaye Mv i.128.16 (§§ 32.5, 89)

ikş-, see: (1) ikşa-li (Skt.), also ikşaya-li, non-caus., and ud-iksati (Dict.)

- Pres. prekšima (ed. [°]şiya) LV 367.13 (§ 27.10); opf. parikşeya My ii.67.7; impy. upekşahl LV 178.6; prekşasu (? § 30.16) LV 322.20; nirikşayātha LV 50.10; pple. samudikşayanto LV 134.4
- Aor. upekşi LV 172.18 (2 sg.); nirikşi LV 230.4; udikşişu LV 114.18; udvikşeyam (1 sg.) Mv ii.194.6 (§ 32.105); udikşişu (Dict.)
- Ger. pratyavekşitvā Mv i.275.19; samīkşiyāna Mv i.278.20 (2) -Iksua-li, -le: not recorded elsewhere; phonetic corruption?
- Pres. vikşyasi RP 7.2; upaparikşyanti Divy 220.9, 15; parikşyate Divy 407.5; opt. upaparikşyeta Lank 214.13; pple. nirikşyamanah Divy 408.8

ir- (ir-), move (intrans.), act: (1) irya-ti: Pali iriyati (cf. Vedic Ir-te; ya-present due to influence of noun iryāpatha, or irya-?)

Pres. iryasi Mv iii.118.18 (mss.)

(2) caus. (transitive) Iraya-li (Skt.), also ira-li

Pres. pple.: irantah Gv 372.13 (§ 38.30)

- opt. udirayed (2 sg.) Mv i.84.16; impv. udirehi LV 185.16
- Aor. udīrayī Mv i.67.3; °yi Mv i.205.1; °ye Mv ii.54.4; vetsuh Mv i.229.12; udirinsuh Mv iii.354.16

Ger. udiryana Gv 255.9 (§ 35.46)

ujjha-li, abandon (Skt.)

Aor. (or Pres.?) ujjhesi Mv iii.295.7, 9 (in 9 v.l. ujhyasi) Ger. ujjhitvā Mv ii.197.13; ujjhiya SP 286.12; ujjhiyāna Mv ii.13.2

utkaniha-ti, long for (Skt.), denom.; also utkanihe-ti Pres. opt. utkantheyā (could be derived from either) My ii.429.8; impv. utkaņțhehi Mv iii.167.9; utkaņțhāhi Mv ii.274.12

utkrośaya-ti, exalt (denom.; Dict.) Pres. °ti AsP 419.4

udānaya-ti, make a joyful utterance, denom.: Pali udāneti

Pres. udānayati LV 103.13

Aor. udánayi Gv 489.11; 'ye Mv i.351.13; 'yetsu(h) Mv i.340.14; udānesi Mv ii.286.1

uş-, burn (Skt.) Ppp. uşita- SP 85.9 (Dict.)

ūsmāya-ti, steam: denom. (Skt.) Aor. üsmäyetsu(h) Mv ii.124.4. 5. 8

ūha-li, remove (Skt.)

Aor. samühetsuh Mv ii.125.6

r-: Caus. arpaya-li, deliver, hand over (Skt.); appe-li Pres. impv. appehi Mv iii.295.2 (? uncertain: if correct, MIndic for arp-)

rliya-le, or riliya-le = arliyali, above: perhaps false Sktization from a MIndic form like Pali atțiyati. Only in Bbh; usually printed rit°

Pres. ritiyate Bbh 168.23; opt. (a)ritiyeran, see artiyati; pple. rtīvamāna- (after -ā) Bbh 166.11; ritīvamāna-(after -am) Bbh 171.9

rdh-, succeed: (1) rdhya-te, -li (Skt.), also spelled ridh°

- Pres. ridhyate RP 59.4; opt. rdhyeya Mv i.299.14; rdhyā (§ 29.42) Mv iii.335.18; 336.7; impv. r(d)dhyahi (?) LV 178.10; ridhyatu (v.l. rdh°) Bhad 15
- (2) rddha-li, denom. to ppp. rddha- (§ 28.19) Pres. mid. prārdhadhve Divy 146.11 (? see Dict.)

Fut. rddhişyati (v.l. riddh°), will succeed, Divy 102.9

odde-ti, (a) fasten, especially (b) lay (a snare): = Pali uddeti in sense (a), oddeti in sense (b). Sense (a) only in Mv ii.75.21

Pres. oddeti Mv ii.252.16; pple. oddentasya Mv ii.253.9 Ppp. oddita- Mv ii.75.21 (read odditā for occhritā with v.l., see Senart's Crit. App. on 74.8); ii.235.1, 6; 251.1, 8 ff.; 253.11 Ger. odditvā Mv ii.252.17

kalla-li, draw: corresponds to Pali kaddhati, Skt. krs; regularly emended to kaddh- by Senart in Mv; once

- (SP 84.1) apparently kaidh- occurs
 Pres. okaţtati Mv iii.29.13, 15, 16; parikaţtati Mv iii.255.17, 18; ākaţtāmi Megh 306.17; opt. okaţtiyā (mss.) Mv iii.30.8; pple. vikaddhamāna- (so, no v.l.) SP 84.1
- Ppp. kațțita- Mv ii.75.3, 11; okațțita- Mv iii.29.17; 34.10 (here mss. °tum); utkațțita- (mss. utkanțh°, so Senart, or utkarita-) Mv ii.249.11; ukkațțita-(mss., for utk°) Mv iii.431.7
- Ger. samkattitvā Mv i.302.15; ukkattetvā (for utk°) Mv iii.158.13; nişkaţţiya (so, see Dict.) Mv ii.242.12

Inf. avakatitum Mv iii.30.6; okatitum Mv iii.30.1; 34.8 Pass. kattiyanti Mv ii.241.18; okattiyanti Mv iii.30.3; pple. kattiyantiye Mv i.217.7; okattiyanto (mss.) Mv iii.30.4; kațțiyamāni (v.l. kațți°) Mv ii.429.15

kathaya-li, say, denom. (Skt.)

Pres. opt. katheyā SP 283.15; katheyuh Sukh 74.18; kathayesi (2 sg.) Mv ii.200.1; impv. kathayāhi SP 209.8 (v.l. °yesi) 209.8 (v.l.

Aor. kathe Mv iii.408.15; kathesi Mv iii.461.7; kathaye Mv i.312.9; kathayetsuh Mv iii.356.2

Pass, kathivati My ii.17.4

kamp-, shake, intrans.: (1) kampa-li (Skt.), also kampaya-ti, non-caus.

Pres. kampayati Mv iii.341.7 (§ 38.21); impv. anukampāhi Mv ii.230.14

Aor. kampi Mv i.64.10; samprakampayişu SP 270.12

(Kashgar rec.; § 32.30); kampe Mv i.40.7; kampeya (inss.) Mv ii.413.1 (§ 32.105); kampesi Mv ii.33.18; prakampire Mmk 512.19 (3 sg.); 548.3 (3 pl.), § 32.43

(2) Caus. kampaya-li, shake (trans., Skt.)

- Aor. kampayi (1 sg.) Gv 231.15; prakampesi Mv ii.300.15
- kas-, go (Skt., Gr. except in some cpds.): (1) -kasa-ti Ppp. okasta-, gone down, entered into, Mv i.245.4 etc. (cf. Ved. vikasta-) (2) -kāsa-li: recorded nowhere else

- Pres. nişkāsati, goes forth, Mv i.361.3 (§ 28.33; Dict.) (3) Caus. (nis-)kāsaya-li, expel (Skt.)
- Ger. nişkāsiya Mv ii.174.11
- Caus. Inf. niskāsāpayitum, to cause to be expelled, Divy 339.6

kānks-, doubt (in Skt. and here in cpds., desire): (1) kānksa-li: = Pali kankhati

Pres. kānksanti Mv i.85.12; impv. (§ 30.16) kānksisu Mv ii.55.21

(2) -kāńkşaya-li = kāńkşati

- Pres. impv. ākāńkşayatha, doubt, Sukh 99.6; abhikāńksayāhi, desire, Mv ii.147.12 (so mss.)
 - (3) *kānksāya-li, doubt, implied in:
- Ppp. kānkṣāyita-(tva-, tā) = Pali kankhāyita-; -tva- Av i.228.6 ff.; AsP 454.9; -tā- AsP 454.10; § 38.38

kāma-li, love: Skt. kāmayati, Pali kāmeti

Pres. kāmasi Mv ii.483.8; impv. kāmāhi Sādh 162.11

kāś-, also spelled kās-, appear: (1) kāśa-te, -li (Skt.) Aor. vitikāsetsuh Mv ii.125.14

(2) Caus. kāśayali (Skt.) Pres. opt. prakāšeyā SP 237.2; °šeyyā Mv i.336.20; prakāśayeyā SP 284.11; prakāśayesi (2 sg.) SP 98.2 Aor. prakāšavī ŠP 23.10; 193.4; °ye Mv i.188.7

kāsa-te, -li, cough (Skt.)

Aor. ukkāsi (for utk°) Mv ii.281.13, 18 (v.l. in both °se); ukkāse id. 14 (v.l. °si), 15

- kīrtaya-ti, celebrate, denom. (Skt.) Pres. opt. anukirtaye (2 sg.) Mv i.77.8
- Aor. kirteşy (2 sg., § 32.63, 65) SP 63.9; kirtaye (2 sg.) Mv iii.441.7

kuccha-fi, abuse (MIndic for Skt. kutsati, which is used rarely for kutsayati). Pres. kucchanti Mv i.106.9

kuttaya-ti, bruise (Skt.) Pass. pple. kuțțiyantă Mv i.6.5

(ut-)kumbha-ti, rears (of a horse; denom.?) Pres. °ti KP 108.2

kr- make, do: (1) karo-ti, kuru-te (Skt.)

- Pres. kurumi (§ 28.64) LV 56.11; kurumi LV 221.1; karoma (§ 28.62) SP 116.9; karonti LV 195.10; opt. kurya (§ 29.42) SP 157.12; kuryāsi (2 sg.; § 29.36) Mv 1.276.8; impv. karohi (v.l. karotha) SP 113.14; kuruhi (§ 30.9) Mv i.275.22; karotha Mv i.245.7; karontu Mv i.295.1; pple. karontāš SP 83.14
- Aor. satkarosi gurukarosi (3 sg.) Mv i.59.14 f. (mss.); \$ 32.70

(2) kara-li: Pali id., Pkt. karai; based on fut. karişyati (Pali karissati), § 28.13

Pres. opt. satkare Mv ii.370.21; satkareta SP 228.14; kari (1 sg.) Bhad 58; kareya LV 335.19; kareyyā Mv ii.144.10; satkareyyasi (2 sg.; § 29.38) Šiks 100.17; pple. alamkaratā (instr. sg.) Šiks 341.18; karamāna- Bhad 38.57

- Fut. 1 sg. karişyam Mv ii.274.13; °şya LV 178.4
- Aor. akari Mv i.268.7; akare Mv iii.45.18; (a)karensu(h) Mv iii.96.17; 365.13; vyākaretsuh Mv i.57.10; karinsu (with Kashgar rec.) SP 50.3 ff.; akarimha Mv iii.434.2 (§ 32.40)
- Ppp. samalamkarita- Siks 327.16
- Ger. karitvā SP 229.8; °tvāna Mv iii.97.20; °tvana Samādh 22.4; karia (§ 35.38) Šiks 299.8; kariya LV 243.12; kariyāņa (°na) SP 222.1; Mv ii.388.4; kari LV 200.9 Inf. samskaritum Divy 207.24
- Gdve. satkaritavya- Mv i.37.3
- Pass. kariyati Mv i.269.3; kariyati LV 185.6; impv. kariyatu Mv iii.360.1
- Caus. opt. karāpayet KP 158.7 (? § 38.53)

(3) kurva-li, also -kurvaya-li (non-caus.): Pali kubbati, AMg. kuvvai; § 28.6

- Pres. kurvāmi SP 125.14; kurvamī LV 195.22; kurvasi Sukh 52.6; kurvati SP 28.8; KP 7.7; kurvāmah Kv 47.18; kurvatha Mv iii.293.16; kurvate (3 sg.) Suv 144.8; kurvāmahe LV 276.2; impv. vikurva Mmk 55.26; kurvahi LV 178.6; kurvatu Suv 57.12; pple. jihmikurvamāna- Gv 87.5
- Pres. in -aya- (§ 38.21), pple. vikurvayato (= vikurvantah) Gv 267.3
- Aor. kurvī Mv i.223.19; vikurvī Šiks 347.6; kurvisu SP 10.6; vikurvişu Mv i.64.3; kurvansu Mv iii.294.11 (§ 32.76)
- Ppp. vikurvita- (as noun, miracle) LV 119.8 etc.
- Ger. kurvi Mv i.236.12 (? § 35.50)

Inf. vikurvitu-kāma- Samādh p. 6, line 23 (4) kāraya-li, in sense of simplex, not usually caus.;

- so regularly Pali kāreti, and often Skt. (BR 2.87); § 38.57 Pres. opt. kāraye (2 sg.) Mv iii.20.9; in sense of causative
- passive (§ 37.17), kārayeyam SP 113.3
- Fut. opt. kārayisyet Mv i.267.9 (§ 31.37) Aor. akārimha (1 pl.) Mv iii.434.2 (§ 32.40); kārayesi, v.l. °yasi (§ 32.62; 3 sg.) Mv ii.64.15; kāresi Mv i.304.12;
- kārayesi Mv i.49.10, 13; kārayetsu(h) Mv i.259.6 Caus. kārāpayanti (§ 38.57) SP 15.3; [°]penti Mv i.26.15; (2 sg.) Mv ii.486.5; impv. °pehi Mv ii.73.18; pple. °payamāna-Mvi.96.6;--Fut. °payişyam Mvi.325.17; -Aor. °payi SP 114.11; °payişû or °payimşû (**§** 32.35) SP 50.16; °pesi Mv iii.442.3; °payet Mv ii.116.1;-Ppp. °pita- SP 50.14 etc.; in non-caus. mg., LV 215.9, § 38.62;-Ger. °payitvä Mv 1.52.13; °payi SP 152.5

(5) Miscellaneous forms

- Pres. opt. krāyur Mv i.69.18 (? § 29.47); (impv. kruhi, and similar forms, probably errors of tradition, see Dict. s.v. kruhi)
- Fut. (§ 31.18) kähiti Mv ii.41.20; °ti (ms.) Šikş 101.6;
 kähi (3 sg.; v.l. kähiti) SP 154.8; kähisi (2 sg.)
 Mv ii.238.9; kähinti Mv i.256.6; kähäma Mv ii.229.15
- Aor. akārisuh Gv 334.25 (§ 32.46); (a)kāsi (§§ 32.49, 56) Mv 1.242.12; vyākārši (§ 32.74; 2 sg.) Mv ill.401.10; vyākarši (3 sg.) Mv ill.139.10; °şīt (? text vyākarșitaś) LV 39.19; vyākariși (mss. vyāka-rși, Senart vlyākarsi) Mv il.193.12; vyākarsu (3 pl.) SP 27.3; karsu (ed. em. karisu) LV 368.12; vyākarsa (? 2 sg., § 32.74) Mv iii.386.6; krtāsi (§ 32.61) SP 51.6
- Ger. -kārakam Mv i.339.16 (§ 35.5); krtā ? LV 356.6 (§ 35.53)
- Inf. krtum Suv 156.4 (§ 36.9)

kr-, scatter: (1) kira-ti (Skt.)

Fut. prākirişyanti (read pra°?) Mv iii.324.8; vikirişyase LV 335.7

- Aor. avakiri Sukh 24.18; abhyokiri (so read) Suv 150.6: okire (1 sg.) Mv iii.248.8; abhikirişu Dbh.g. 30(56).1; okiresi (but cf. -kireti) Mv 1.38.9; prakiretsuh Mv i.211.18
- Ppp. prakirita- Mmk 76.16 Ger. prakiritvā Mv i.211.20; abhyokiriyāņa SP 191.2, 7; abhyokiri SP 325.4; 228.15 (? perhaps opt.); avakiri Mv ii.333.11
- (2) -kire-ti, -kiraya-ti, non-caus.: Pali also pakireti = pakirati (wrongły called caus. in PTSD) Pres. prakireti Mv ii.252.1, 16; °enti Mv i.211.20; okirenti
- Mv ii.304.3; pple. prakirentasya Mv ii.253.10; vikirayamāņa- Gv 87.8; 89.11

(3) -kira-li or -le: so, rarely, in Pali, see CPD s.v. abhikirati; §§ 28.34; 34.10; 37.28

Pres. pple. abhyavakiramāna-, strewing (flowers), Gv 430.3 Ppp. abhyavakirita-, bestrewn, Mmk 63.25

(4) *-kara-ti?

Aor. samakari Dbh.g. 48(74).11 (? 3 pl.; very uncertain; § 32.16)

krl-: *karla-li, twist: AMg. kattai, pres. pple. kattanta-; in Skt., chiefly Vedic, krņatti, Pali kantati; Skt. ppp. kartita, ger. kartya, once each in Hariv., BR 5.1259. Are our and the AMg. forms based on the like of these? Fut. kartişyanti Divy 213.5 Ppp. su-kartita- Mmk 57.7

Inf. kartitum Divy 213.2

Caus. opt. kartāpayet Mmk 57.7

krs-, drag: (1) karsa-ti (Skt.) and (non-caus.) karsaya-ti (which in Skt. too appears to be used in the sense of karsati, trans.)

Pres. parikarşayati Divy 475.14; avakarşayanti SP 159.12 Ger. apakarşitvā Mv ii.75.5

Pass. apakarsiyate Mv i.302.15

(2) krsa-ti (Skt.) Fut. krsisyanti Divy 212.17

(3) Miscellaneous

Fut. äkrkşyata (ms., em. äkarkşyatah) MSV ii.65.20 (§ 31.23)

klp-: kalpa-li, -le, and kalpaya-li, kalpe-li, be adapted (intrans., in Skt. so only the simplex); arrange (in Skt. so virtually only the caus.); trim (hair, etc.; caus.); in BHS both aya and a forms occur with both meanings

Pres. kalpenti Mv i.135.1, 10 (intrans., are adapted); opt. kalpeya SP 344.2 (trans., would arrange) Aor. (trans.) kalpesi Mv iii.42.12; ii.234.10; kalpayesi Mv ii.233.9; prakalpaye (v.l. [°]yet) Mv iii.45.2

Caus. ger. kalpāpayitvā, having had trimmed, Mv il.489.8

keläya-ti (also recorded as kelay°), care for, cherish, tend, clean up: Pali id., Pkt. kelāyai; Dict.

keläyanti Divy 631.5; opt. keläyeyuh AsP 253.13; pple. keläyanti (mss. °nti) Mv-iii.154.13 Pres.

Ppp. kelāyita- Mmk 52.14; 81.26; Dbh 39.22; AsP 253.13 Inf. kelayitum (so text, read kelā°?) LV 100.9 (prose) Gdve. kelāyitavya- Mvy 2591

Caus. ger. kelāpayitvā Mmk 708.28 (so, or kela°, read for text kailapayitvā; prose)

kojaya-ti, strike (once recorded -koth°): Pali (ā)koteti; cf. Skt. kuţţayati, kuţa-

- Pres. koțayati Divy 115.27; opt. ākoțayet Mvy 9260; ākoțhayed (so recorded) Prāt 491.8; utkoțayet (Dict.) Prāt 503.4; impv. ākoțaya Av i.18.8
- koțita- Divy 115.27; LV 236.9; äkotita- Divy 335.13 Ppp. ākoțayitvā AsP 289.7 Ger.
- Gdve. koţayitavya- Divy 114.22

Pass. impy. ākotyatām Av i.258.9; pple. ākotyamāna- LV 299.10

kranda-li, cry (Skt.)

Aor. krandişu SP 84.12; vikrande Mv i.347.9; krandetsuh My iii.409.9

kram-, stride: (1) krama-li (Skt.)

- Pres. opt. upasamkrami (1 sg.) Bhad 35; krame SP 293.5; nişkrameyā LV 201.21; ākrameyyā LV 199.15; impv. abhinişkramāhi Mv ii.141.2
- niskrami (followed by ti = iti, so divide) LV 133.13; Aor. prakrami Mv i.255.4; nişkrami LV 92.12; nişkramet Mv i.303.11; prakrame (3 pl.) Mv iii.339.6; upasamkraminsu Hoernle MR 152 for SP 270.11: upasamkrameyuh Mv i.36.8 (§ 32.105); prakramensuh (v.l. °tsuh) Mv i.212.13
- abhinişkramitvā SP 68.8; Mv i.358.7; upasamkramī Ger. SP 11.11; 191.1
- Caus. (Ppp. abhinişkramita- LV 187.4, § 38.9;) Ppp. åkramāpita- (? em.) or ākrāmāpita- (Dict.) Mv iii.363.9

(2) -krame-li, non-caus. = kramati

- Pres. äkramesi Mv ii.408.18; upakramemi Mv ii.459.16
- parākrametum LV 134.2; atikrametum LV 134.7 Inf. (§ 36.8)

(3) krāma-ti (Skt.), also krāmaya-ti, *krāme-ti, noncaus. (In Gv 72.3, prose, read with 2d ed. upasamkrāmanti for 1st ed. °kranti.)

Pres. avakrāmayati LV 400.14

Impf. upāsamkrāmat, °man, see § 32.5

- prakrāmi Mv i.256.16; 323.11 Aor.
- Ger. okrāmetvā Mv ili.318.15; parākrāmya (§ 35.12)
- Ud xi.1 (4) -kramya-li, -le: Pali 2 sg. impv. pațikamma, Geiger 136.1 (cl. Skt. Gr. krămyati)

- Pres. ākramyate Divy 399.25; impv. upasamkramya (mss.) Mv i.330.13; nişkramyā LV 163.20; 164.11; ākramyā LV 163.21; nişkramyāhī LV 161.22; 162.10
- Aor. nişkramye Mv ii.198.13 (v.l. nişkrame; meter equally good)

(5) cańkrama-ti, intens.: Pali cańkamati (Skt. cańkra-myate, very rarely °mati; for Mbh. Calc. 1.7919 cańkra-mamāņau, cited BR, the Crit. ed. reads cańkramya°, 1.211.14); # 28.13, 30

- Pres. anucańkramāmi SP 55.1; °manti SP 84.8; opt. cankrami SP 344.1; °med LV 369.1; °meta SP 98.10; anucankramanto, pr. pple. n. sg., SP 61.10 etc.
- cańkrami LV 368.16; °mişu Mv i.189.8; °me Mv AOF. i.317.9; anu-°metsuh Mv iii.282.5
- Ger. anucańkramitvā Mv ii.455.16; 456.21

kri-, buy: (1) krina-li: apparently not in Pali, but Pkt. has forms in ki- beside more usual ki-; based on Skt. 3 pl. kriņanti

vikrinati Mv ii.241.15; 274.1 (both prose, no v.l.; Pres. in 274.1 Senart em. °nāti)

Fut. krinişyati Mv ii.242.1 (v.l. krinişyati) Gdve. krinitavya- (v.l. kri^o) Mv ii.274.2 (prose)

(2) krina-ti, once krine-ti: Pali kinati, § 28.5

- Pres. kriņāmi, vikriņāmi Mv ii.49.15; kriņenti Mv i.91.10 (so 5 mss.; Senart, krinanti with one inferior ms.); impv. krinātu Mv ii.87.16 (prose)
- kriņitvā Mv i.335.10 (prose; Senart em. krl°); Ger. kriniya Mv ii.241.17 (prose)

vikriņitum Mv ii.87.15 (prose; v.l. vikriņitum!) Inf. Gdve., see under krinati (1)

- (3) (vi-)kre-ti: AMg. vikkei (vikkeha); § 28.49 Pres. vikrenti SP 280.4
- vikreya MSV ii.143.7 (ms.; ed. em. vikriya); Ger. \$ 35.21
- (Inf. vikretum, Skt.:) so read Mv li.88.6 (§ 36.8); Pali also has vikketum

(4) Pass. vi-krāya-ti (or -le?), be sold: AMg. pple. vikkāyamāņa-; cf. also .IM. active pple. vikkanta-, selling (Sheth); origin doubtful, § 37.39

Pres. opt. vikrayetsuh, would be sold, My ii.241.20

Fut. vikrāyisyati (vikra°? see Dict. s.v. vikrāyati) and vikrāsyati, will be sold, My ii.242.14 and 11 (§ 31.2, 3)

krida-li, play (Skt.), also kridaya-li, non-caus.

- Pres. kriditah Divy 404.18 (3 dual; § 27.10); opt. krideyyā Mv ii.144.9; impv. krīdāhi Mv ii.103.6 kridayanti LV 75.17; pple. kridayantam LV 157.7
- kridişu SP 50.12; kridetsu Mv i.194.17 Aor. kriditvä Mv ii.111.16 Ger.
- Caus. kridāpayati Mmk 458.20; impv. kridāpetha Mv i.227.12

kruś-, cry out: (1) krośa-li (Skt.)

- krośire Mmk 584.22 (§ 32.43); vikrośe Mv i.347.9 (2) Miscellaneous (see also utkrośaya-ti)
- Fut. (§ 31.23; krokşyati, Skt. Gr.) ākrokşyanti Divy 38.11, 14; pratikrokşyanti AsP 179.4

krūd-, roast (read kūd-?)

- Pass. parikrūdyante, are roasted, LV 207.10 (Dict.)
- klam-, grow weary: (1) klama-li = Pali kilamati Pres. pariklamanti MPS 31.56
- (2) kilamya-ti, blend of klämyati and prec.
- Pres. impv. kilamyatu Mv iii.175.3
- (3) Caus. klāmaya-ti, weary (Skt.), also klāma-ti (caus. or non-caus. mg.)
- Pres. kilāmasi for klāmasi = klāmayasi Mv ii.483.9; opt. kilāmaye (2 sg.) Mv iii.19.15; non-caus. mg., opt. klāmed MSV ii.85.1 (= Skt. klāmyed)

(klambh-, false reading, see Dict. s.v. viklambh-)

klid-, caus. kledaya-li, moisten (Skt.) Ger. kledayitvāna Mv iii.7.19

ksan-, injure (Skt.)

Pass. (§ 37.30) kşanyate Siks 187.8; kşanyati, and opt. kşanyema Sikş 92.3 and 1

kşama-li, endure (Skt.)

- Pres. opt. kşame SP 334.15; kşameyā Mv i.285.1; impv. kşamāhi Mv ili.23.8
- Caus. kşamāpayanti LV 379.6; °penti Mv iii.359.10; opt. °peyam (v.l. °payeyam) Mv i.37.8; Ger. °payitvā LV 38.14

kși-, destroy: (1) kșaya-li (once in Rām.): Sheth cites Pkt. (Gram.) khaai as intrans.; § 28.39

- Inf. kşayitu Dbh.g. 41(67).8, and kşayitum Šikş 345.10, both m.c. for kşayitum (§§ 36.2, 4, 6)
- (2) Pass. kşīya-te (Skt.); kşīyati (? Dict.) My ii.480.2, 4 kşiyihanti (§ 31.2) Mv ii.53.14 Fut.

(3) Caus. kşapaya-li (Skt.)

Pres. opt. kşapaye Mv ii.93.13; kşapeya SP 111.14

kşapaņāya Mv ii.362.8 Inf.

kșip-, throw: (1) kșipa-ti (Skt.), also *kșipaya-ti, *kșipe-ti, non-caus. (see Ger. below)

- Pres. opt. upanikșipe Mv il.365.1; nikșipi SP 158.1; abhyutksipeyā Mv ii.315.9
- kşipişyati LV 145.14 (so with most mss., ed. kşap[°]) kşipi LV 387.7; parikşipi Mv iii.301.6; kşipe Mv Fut.
- Aor. ii.337.3; 338.5; ksipisus (§ 32.36) LV 282.16; ksipimsū LV 81.7; ksipetsu(h) Mv i.342.13

ksipitvā SP 58.14; °tva SP 55.7; ksipitvāna (mss.) Ger.

Mv ii.92.5; abhyutkşipitvāna(?) Mv ii.329.2; kşipiyā Mv ii.391.1; kşipiyāna Mv i.241.2; nikşipayitvā (? mss.; read °petva?) Mv i.241.1; niksipetvā Mv ii.65.4 (§ 35.26; Senart em. °itvā)

Inf. niksipitum Mv iii.428.4

Gdve. praksipitavya- Mmk 695.1

- Caus. opt. ksipāpayet Mmk 528.27; Gdve. ksipāpayitavya-Mmk 49.19
 - (2) Miscellaneous forms
- Aor. akşaipsit (v.l. kşipatah; § 32.79) SP 468.7 Perf. cikşepur I.V 7.14 (§ 33.4)
- utksiptum Divy 279.2 (§ 36.9) Inf.

kșnyu-le or kșū°, shake (trans.) Pres. pple. samkșuyamăno, or °kșū° Mv ii.268.12 ff. (?see Dict.)

kşubh-, quake: (1) kşubhya-ti (Skt.)

Aor. ksubhye Mv ii.162.5

(2) *kşubha-ti; Pali (sam)khubhati; cf. Ppp. Skt. kşubhita-, AMg. khuhia-

Caus. kşubhayanti LV 339.13

khana-ti, dig (Skt.), also -khane-ti, non-caus.; once khan-, as in Pali (below under Caus. Ppp.)

- Pres. anukhaneti, parikhaneti Mv iii.301.16; opt. nikhaneyā (1 sg.) Mv ii.436.3
- khanita- Divy 16.4 PDD.
- Ger. parikhanya (or pali°) Mv iii.284.18
- Caus. opt. utkhanāpeyā (so read) Mv ii.436.4; Fut. utkhanāpayişyam Mv ii.437.13; Ppp. utkhanāpita- Mv ii.437.16; su-khanapita-, mss., read °khanā°, Divy 71.5; Ger. khanāpayitvā Mv i.352.21; Pass. pple. utkhanāpiyamāna- Mv ii.439.6

khara-li, flow, MIndic for kşar-: AMg. kharai; cf. also abr-

Pres. pple. prākharantah (v.l. pra°) SP 126.8

khala-ti, see skhala-ti

khalakhalāya-li, rattle (onomat.): cf. JM. khalakhalei (Jacobi Erz. 48.9)

Pres. °yati Mv i.172.3

khād-, eat: (1) khāda-li (Skt.)

- Pres. opt. khādesi (2 sg.) Mv iii.7.4, 8; impv. khādāhi Mv iii.159.9
- khāditva (Epic Skt. °tvā) SP 84.5; khādiya Mv Ger. iii.4.16
- Caus, impv. khādāpaya Mmk 56.26; Ppp. khādāpita- Mv i.24.16

(2) *khāya-li* (MIndic): Pkt. khāyai, and Pali ppp. khāyita- (§§ 2.32; 32.23)

Aor. khāyi Mv ili.299.11

- Ppp. khāyita- Mv iii.84.17
- (3) Pass. khajja-li, MIndic for khādyate: Pali id. (§ 2.14)
- Pres. khajjāmi Mv ii.450.6, 8; 452.8, 11; opt. khajje Mv iii.252.15; khajjeya Mv ii.78.1
- Fut. khajjişyasi Mv ii.78.4; °şyatha Mv iii.290.17

khid-, be afflicted: (1) khijja-ti, MIndic for khidyate: not in PTSD, cited in Childers from Clough; Pkt. khijjai (§ 2.14)

Pres. khijjante Mv ii.457.13; pple. a-khijjantam Mv ii.252.15; khijjantasya Mv ii.253.9, 13

khijjitvā Mv ii.252.14 Ger.

(2) Caus. khedaya-ti (Skt.)

Aor. khedayi (2 sg.) LV 221.8

khurda-ti, leap, or sport, play: Skt. kurd-, for which Dhatup, khurd- or khurd-, (Prob. not related to AMg. khuddai, break)

Pres. impv. khurda Mmk 28.15

(ā-)khyā-, tell: (1) ākhyāya-ti, act., tell; § 37.23; Dict. Pres. impv. äkhyäyatha, relate! Mv ii.437.3 (prose)

(2) ākhyā-li, Skt. (*ākhya-li? cf. Skt. ākhyat, imperf.) ākhyāsi (§ 32.54) Mv i.9.6; as future(?) Mv iii.451.6 (§ 32.123); samākhyetsu Mv iii.85.5 (§ 32.104)

gad- = gal-, with vi-, disappear, fall down (vi-gal-, Skt.)

Ppp. vigadita- SP 72.5

gaņaya-li, reckon (Skt.)

Pres. opt. gaņeya LV 338.7; impv. gaņayāhi Mv iii.265.5 [Ger. ganetu? § 35.55]

gam-, go: (1) gaccha-li, also gacchaya-li, gacche-li, non-caus.

- Pres. gacchayami LV 223.1; opt. gacche Mv i.274.18; ägacche (2 sg.) Mv ii.90.2; gaccheyā LV 201.14; gaccheya Mv i.258.13; gacchesi (2 sg.) Mv ii.42.17; gacchāsi (2 sg., supposed 'subjunctive', § 27.8) Mv i.157.3; ii.194.11; impv. gacchehi (mss.) Mv i.187.1; gacchāhi Mv i.362.11; gacchahi (v.1. °ehi) Mv i.288.1
- gacchi LV 133.13; gacchi LV 81.11; gacchisu LV 74.6; gacchisu Šikş 347.2; āgacchasi (1 sg.; § 32.62) Mv ii.133.4; paryadhigacchasi (3 sg.) Mv i.188.3; Aor. atyudgacchet (ed. em. abhy°) Mv i.229.16; ägacche Mv i.221.7; ágaccheyur (§ 32.105) Mmk 130.23; gacchensu Mv iii.373.21; pratyudgacchetsu(h) Mv i.259.12

Ger. gacchitvā Mv iii.142.16; gacchiya Mv i.362.13

gacchitum Samādh p. 30 line 31 Inf.

- Pass. Pple. anugacchiyamana- Mv ii.101.15
- (2) gama-ti: AMg. gamai; § 28.12 Pres. opt. āgame Mv ii.141.7; °meyā LV 232.18; °mesi (2 sg.) Mv ii.199.20; upagami Mv ii.387.15 Fut. impv. gamisyatu Divy 220.10 (§ 31.37)
- Aor. agamā (§ 32.112) Mv i.220.10 (3 sg.); iii.12.5 (so read; 1 sg.); upāgamā Mv ii.53.19 (so read; 3 pl.); gami Sukh 51.16; agami Mv iii.306.12; adhyagame Mv iii.191.4; upagaminsu (v.l. °tsu) LV 94.21; agamittha (? § 32.42) Mv iii.415.9; agamāsi (§32.59) Mv iii.12.2; upagamensuh (°metsuh) Mv i.224.6 = ii.27.6
- āgamitvā Gv 190.4; gamiyāna SP 157.13; upágami Ger. Mv i.3.3 (mss.)
- gamanāya SP 187.5; gamanāye Mv i.362.15 Inf.

(3) gamaya-ti, game-ti, non-caus., = gamati

- Pres. impv. āgamaya Divy 76.10 = 465.10; °yata Divy 534.12; agamehi (could be entered under gamati) My ii.31.20; °metu My iii.263.17
- upägamesi Mv i.4.2; 246.14 (could be entered under Aor. gamati, cf. § 32.67)
- (4) Miscellaneous forms gamsati (gans°) Mv iii.85.16 etc. (§ 31.29); once Fut. abhigamsye (1 sg.) Mv ii.144.3
- (? § 32.77) pratyudgatāsuh, or °tānsuh Mv i.35.5; udgatāsuh Mv i.222.12 Aor.
- Ger. adhigantum LV 245.19 (§ 35.54)

garja-li, -le, roar (Skt.)

nigarjasu (3 sg., read °si? § 32.62) Gv 259.23; abhigarje Mv ii.270.1, 19; abhigarjetsuh Mv ii.293.12 Aor. Caus. (not in Wh. Roots) pple. nigarjayamana- Gv 272.26

garhaya-li, censure, non-caus. (in Skt. also = garhati) Gdve. vigarhayitavya- Divy 115.7

gal- (or gr-, gar-), swallow: -gala-li (or -gara-li), for normal Skt. gilati, girati (garati Dhātup., very rare and doubtful in literature); AMg. galai

Pres. pple. nirgaranto, or (best mss.) nirgalanto LV 306.21 (prose), spitting forth

gaveş-, seek: (1) gaveşa-le, gaveşaya-ti (both Skt.) Aor. samagaveşi (v.l. °şī) Mv ii.205.2; gaveşi (so with

mss., followed by yam) Mv ii.220.14 (2) gavesya(ya)-li; cf. Pali gavessatha Dhp. 146 Fut. gaveşyayişyati AsP 113.6

1 gā-, go (Skt.) No present *gāti is recorded; Skt. agāt Aor. anvagā (§ 32.106; 1 sg.) Mv iii.457.15

2 gā-, sing: gāya-li (Skt.); rarely Skt. gāli; Pali gāhi, Jāt. iii.507.25

- gāye Mv ii.268.10; gāyetsu Mv ii.161.1 Aor.
- Inf. gavitum MSV iv.240.22
- gāyāpayati MSV iii.16.6 Caus.
- (Aor. Vedic agăsit:) agăsi Mv i.130.6 (§§ 28.55; 32.54)

gāha-te, -ti (Skt.), plunge; also -gāhaya-te, -ti, non-caus. Pres. pple. vigāhayamāna- Gv 476.4; Bhad 39

- Aor. avagāhi Mv ii.222.4; °hisu Mv i.11.1; °hitha (3 pl., § 32.42) My i.11.5
- a-paryavagāhitvā Mv iii.153.2; 167.5; °hetvā Mv Ger. iii.165.11; 170.7 (mss.); ogähetväna Mv iii.289.1
- Pass. with active meaning, or 4th class present: samanu gāhyamāna- Prāt 482.7

gila-ti, swallow (Skt.)

Pres. opt. gile (2 sg.) Ud xxxi.31

gudāya-li, thunder, and similar onomat. forms

Pres. gudāyati (v.l. gud°; subject deva) LV 308.4; pple. gudugudāyamāna- Megh 288.15; gulugulāyanti Mmk 294.12; pple. guluguluyanto Megh 304.19

gumugumunli (! 3 pl.), hum (of lutes; onomat.) Mv iii.267.3; § 28.58

gulugul-, see s.v. gudáya-ti

guh-, see güha-ti

guhm-, semi-Mindic for gumph-, twine (§ 2.35) . guhmita-, strung together, RP 47.3; = Skt. gumph-Ppp. ita-, cf. Saur. gumhadu (Sheth)

güha-ti, and gühaya-ti, wind (garlands); once recorded as guh- (cf. Pali Cp. i.9.18 nappațiguhāmi, but I suspect a corruption here; meter favors °gühāmi); on etymology see Dict.

- Pres. gühasi (impv. sense) Mv iii.4.1; gühayati Mv ii.426.19; impv. guhāhi Mv ii.427.4 (prose; read gúh°?)
- Aor. (a)vagühayet Mv i.304.15

gr-, eat, see gal-

grath-, twine, wind: grathnāti (Skt.), *grathnati Caus. grathnāpayati MSV iii.16.5

grah-, take: (1) grhna-li (sporadic in early Skt.): Pali ganhati; n often for n; also *-grhne-li (see under impv., Ger.)

grhņati Mv ii.242.8; °si Mv iii.169.8; grhņāmah Pres. Karmav 52.7; opt. nigrhne, pragrhne Mv i.278.6; °neyā Mv i.347.18; impv. udgrhņa Divy 613.25; anugrhnähi Lank 4.15; parigrhnatha SP 90.10; pratigrhņehi Mv iii.211.10; pple. grhņamāna-Karmav 164.24

- grhnişyāmi Mv ii.450.15 Fut.
- abhinigrhne (1 sg.) Mv ii.124.2; abhigrhnişu (mss.) Aor. Mv ii.307.7
- grhņitvā Mv iii.144.16; °tva SP 113.3; pragrhņetvā Ger. (§ 35.26) Mv iii.131.4; grhņiyāna Mv ii.452.13 nigrhņitum Mv iii.393.5, 10
- Inf.
- Gdve, samgrhuitavya- Mv ii.479.1
- Pass. opt. parigrhnlyeya Mv iii.393.13 (mss. °yeyam, ovetam)
- Caus. (a) grhņāpayati Mmk 28.9; Ger. °petvā Mv iii.393.19 (§ 38.53)
 - (b) grhņāveti Mv ii.467.3 (§ 38.69)
 - (c) *grhnayati, implied by perf. grhnayam asa Mmk 599.10 (§ 38.9)
 - (2) grhņi-le (Skt.)
- Pres. opt. nigrhņīyā Mv i.347.18
- grhņişyati (§ 31.7) Mv ii.242.2, etc. Fut.
- udgrhnisu SP 181.9 (Kashgar rec.; § 32.37) Aor.
- (§ 35.15) grhnītvā LV 287.15; 288.7; grhnīya Mv ii.249.14; iii.144.9; grhnīyāna (§ 35.48) Mv ii.241.16; Ger. 492,6
- (3) *grahe-ti: Pali *gaheti, Ap. gahei (Sanatk. 629.9; Pres. not recorded here, and only -gahāyati in Pali, cf. Tedesco JAOS 43.389; see § 35.19)
- graheşyāmi Mv iii.191.17 (prose, v.l. grhi°); ud-Fut. graheşyanti Vaj. in Hoernle MR 186.8 (§ 31.10)
- agrahe Mv ii.181.20; parigraheşīd SP 204.8 (§ 32.63); pratyagrahesi Mv ii.38.2 (ib.); pratya-Aor. grahetsu Mv i.220.15
- grahāya (§ 35.19) SP 211.6; Divy 327.14 etc.; grahetvā (Senart [°]tva, m.c.; § 35.25) Mv 1.236.6; [°]tvāna (§ 35.34) Mv iii.5.18; grahiya (§ 35.44) LV Ger. 297.3; Mv ii.317.5; upagrahiyāņa Mv i.227.16; samgrahi Siks 340.5
- nigrahetum Mv iii.322.4 (§ 36.8) Inf.
- Gdve. grahetavya- (§ 34.21) Mmk 79.25; 83.16; 694.10; 708.2
- (4) Miscellaneous forms
- Fut. pratigrhīsyāmi LV 384.3 (§ 31.10)
- Аог. -grahisi(d) SP 204.8 (? see §§ 32.63, 71)
- upagrahita- Mv ii.247.15; § 34.13 Ppp.
- grhya (§ 35.9) LV 110.20 etc.; pratigrhiya LV Ger. 240.12; grhiya Mv ii.333.22 (? § 35.18)
- udgrhitum (§ 36.9) Divy 579.21; grahanāya LV Inf. 338.6
- Gdve. nigrhitavya- Mv i.278.6; samgrhitavya- id. 7 (§ 34.21)
- Pass. opt. nigrhyeya Mv iii.391.17
- Caus. opt. grāheya Mv ii.330.18; ppp. pratigrāhita- Mvy 8461

(vi-)ghala-ti, caus. (vi-)ghālaya-ti, open (Skt. only vighațayati; cf. Pali vighāțita-, AMg. vihādiya-) Caus. ger. vighāțya Divy 280.9 (§ 38.16)

ghalla-li, strike (Skt.)

samghattetsuh Mv i.235.16 Аог.

ghunaghunāya-le, buzz (of insects; onomat.); cf. Pkt. ghunahuniā (noun)

Pres. pple. ghuņaghuņāyamāna- (printed °ņa-) Kv 46.21

ghuş-, sound: (1) ghoşa-ti (Skt.)

- Caus. Ppp. ghosāpita- Mv iii.390.19; ghosāpayita-vān (§§ 34.8; 38.57) SP 257.4
 - (2) Pass. ghuşyayale = Skt. ghuşyate

Pres. ghuşyayante (= ghuşyante) LV 92.6 (§§ 37.21; 38.21)

ghr-: ghara-li, -le, drip, flow: Skt. only Gr.; Pali paggharati; cf. kharati

- Pres. gharati Mv iii.408.1; pragharati Mvy 6967 (v.l. praggh°, so Mironov); Divy 57.21; 409.1; Šiks 249.7: Av i.202.13: Karmay 66.4: gharanti My i.227.18 (separate from lala preceding)
- Caus. parighareti Mv iii.301.17 (prose; so mss., Senart em. °ghār°; Skt. only -ghār°)

ghrș-, rub, crush: *ghrșa-li: seems implied by Pkt. oghasiya- (Sheth) = *avaghrșita-

Ger. oghrsitvā Mv iii.7.4; nighrsitvā Mv iii.7.7

ghrā-, smell: (1) jighra-li (Skt.)

- upajighritvā My i.353.8; ii.137.8; 139.4 Ger.
- (2) ghrāya-li, -le (not pass.; trans., with object gandhaor the like): Pali ghāyati
- Pres. ghrāyati SP 360.5 fl.; 364.4; ghrāyate SP 362.2; 364.1, 2; opt. ghrāyi Šikş 341.9; pple. ghrāyanto (n. sg. m.) Mv ii.150.15; 152.9
- Fut. ghrāyişyi (1 sg. mid., i for e m.c.) LV 222.16 Ppp. ghrāyita- Mv ii.251.10
- Ger. ghrāyitvā Mv ii.251.10; 295.12; iii.52.10

(ā-)cakşa-li, -le, say, tell: (Epic Skt.)

Ppp. ācaksita- SP 230.8; ācasta- (§ 34.11) My ii.96.17

-caggha-ti, -cagghaya-ti, laugh; cf. Pali (uj-)jagghati

- Pres. samcagghati Bbh 169.4; uccagghantl Siks 13.1; opt. samcagghet Bhik 28a.3; pple. uccagghantah (n. pl.) Šikš 12.15; 49.12; °ghayanto AsP 232.13; uccagghayamāna- AsP 232.18
- Fut. uccagghayişyati AsP 385.13; 388.19
- Ppp. uccagghita-vantah SP 382.12

cama-ti, sip (Skt.)

Aor. ācamesi Mv ii.299.11

car-, move: (1) cara-li (Skt.), also caraya-li, care-li, non-caus.

- Pres. caresi Mv ii.187.23; vicaresi Mv ii.321.21; opt. cari (2 sg., ms.) RP 39.4; ācare (3 sg.) Mv i.12.14; samācareyā Mv iii.373.16; carayesi (2 sg.) Mv ii.42.18; (paricaremha? see § 26.7)
- acari Mv i.3.9; cari Gv 34.18; niścari LV 179.5; care Mv i.61.15; niścaretsuh Mv ii.32.15; acări (§ Aor. 32.18) Ud xxxi.5; anucarittha (§ 32.42) Mv iil.136.15 caritvāna Mv ii.42.15; ācari Siks 330.3 Ger.
- Pass. pple. paricariyamāna- Mv ii.423.9
- Caus. opt. carāpayeyam SP 53.4 (2) Caus. cārava-ti (Skt.)
- Pres. opt. pravicāreyyā Mv ii.144.10; impv. paricārehi Mv ii.103.6
- Fut: uccārişye SP 237.10
- paricāretsu(h) Mv i.194.17 (mss.) Aor.
- Inf. cāratum Divy 180.13 (§ 36.7)
- Pass. cārīvanti Mv iii.176.10

cala-li, move (Skt.). See also pracalāya-li Pres. opt. cale LV 335.17; caleya LV 330.3

Aor. cali LV 92.3

Caus. inf. cālanāya Mv ii.329.20

(apa-)cāy-, revere: (1) -cāya-ti (Brāhmaņas and Pall id.) apacāyanti Mv ii.259.10; 260.3; opt. apacāyet AsP Pres. 57.9

- Fut. apacāyişyanti Mv ili.424.14
- apacāyesi Mv i.60.1 Aor.
- apacāyita- SP 5.8; Mv ii.139.10; Šiks 147.11 Ppp.
- Ger. apacāyitvā Mv i.44.12 (in same formula Mv iii.138.8 apacavitvā, by misprint or other error)
- Gdve. apacāyanīya- AsP 57.2
 - (2) -cā-ti (§ 28.54)

Pres. impy. apacahi My ii.335.12 (so mss.); see Dict. s.v. apacāvati

ci-, gather, note, etc.: (1) -cina-ti (for Skt. cinoti): Pali -cinati (and -cināti)

- Pres. vicinanti Dbh.g. 11(347).2; 40(66).1; samcinanti Dbh.g. 31(57).19; pple. uccinanti Mv ii.452.3, 4 (em.); ācinam (= ācinan) Ud xvii.5 (text unmetr. ācinam)
- Fut. pravicinişyati SP 372.7
- vicinitvā Mv ii.465.10 Ger.

(2) -caya-ti, -ce-li, cf. cayate 'V.B.' in Wh. Roots; but see § 28.48; Pali opt. niccheyya Sn 785, implying *niccheti (*niśceti or *niścayati)

- niścayitvā SP 48.6 (uncertain text, see § 35.28 and Dict.; cf. Epic cayitvā, Wh. Roots); samciyā (§ 35.44) Dbh.g. 50(76).17; udviciya Mv ii.244.7 (? Ger. see Dict.)
 - (3) cino-li (Skt.)
- Pres. 3 pl. upacinonti Siks 4.5; read upacinonti in 4.4 for upacinvanti, as meter proves

(ā-, abhyā-)eikşa-ti, tell: Pali (abbh-)ācikkhati, Pkt. (abbh-)āikkhai, see Dict. s.v. ācikşati

- Pres. āciksati Mv i.243.6; ii.408.7; °anti ii.132.3; opt. abhyācikseyam Mv i.45.1; impv. āciksa Mv ii.57.17; 58.15; °āhi iii.192.13
- ācikşişyanti Mv i.272.5; °şyam (1 sg.) Mv iii.258.13 Fut. Ppp. āciksita- Mv i.355.2, 5; abhyā° Mv iii.27.18

cilicitāya-li (and variants), crackle, onomat. (of sparks of fire and the like)

Pres. ciţiciţāyanti Mmk 298.2; 719.1; ciţaciţāyati Mmk 687.9; ciţiciţīyanti (read °ciţāy°?) Mmk 295.6; pple. citicitāvamāna- Divy 606.1

citraya-ti, decorate, paint (denom., Skt.) Caus. gdve, citrāpayitavya- Mmk 61.16

cintaya-ti, think (Skt.)

- Pres. opt. anuvicintaye (2 sg.) Lańk 11.13; cintayi SP 343.13; impv. vicintehi LV 184.2; cintayāhi Mv ii.146.4
- Aor. cintayi SP 99.3 (Kashgar rec.) Ger. cintitva (mss.) SP 117.4; cintayi SP 213.10
- Caus. (a) as if *cintayayati: ppp. anuvicintayita-, caused to ponder, SP 109.2 (§ 34.9); see next
 - (b) cintāpayati: anuvicintāpayamāna-, v.l. Kashgar rec. SP 109.2 (cf. prec.); Fut. cintāpayişyāmaņ SP 271.4

cipya-te (pass.), be crushed: Pali cippiyate or °ti. See also Dict. s.v. nada-cippitikam Pres. pple. cipyamāna- Šikş 182.2 (cf. Pali cippiyamāna-

Miln. 261.28, 30)

cud-, impel: caus. codaya-ti (Skt.) codayi LV 169.22 Ger.

cumba-ti, kiss Aor. paricumbetsuh Mv iii.409.9

culuculāya-li (onomat.), quiver (? Dict.) °vati Mmk 319.11, 15; 324.1 Pres.

cetaya-ti, buy (see Dict.): cf. Pali caus. cetāpeti, cause to be bought

Ger. cetayitvā Prāt 492.13

cyava-ti, fall (Skt.) Fut. cyavisyati Mv i.196.21

cyavitvā LV 56.7; °tvāna SP 95.1; °tvana Gv Ger. 215.5; cyaviya Mv i.203.4 = ii.6.21

chadde-ti, abandon; Pali id. Pres. °ti My ii.170.18; iii.291.14

chad: chādaya-li, cover (Skt.), also chāda-le Pres. châdaute Sukh 50.16; opt. abhichâdaye Mv ii.367.17;

- chādayeyā SP 228.15; chādayetsuh Mv ii.287.2 chādayi LV 158.19 (3 pl.); āchādesi Mv iii.301.3; Aor.
- āchādaye (1 sg.) Mv iii.249.6 chāditavya- Mv iii.55.4, chādavitavya- ib. 5, 7, 8, 10 Ger.

chambhu-li, also chamba-li, be frightened: Pali chambhati, °eti, derived from Skt. skambh-. The spelling chambseems recorded only in the Dhvajāgrasūtra, Waldschmidt, Kl. Skt. Texte 4; perhaps only a false recording, but repeated several times

- Pres. chambati Dhvaj. (above) 49.23. Only finite form recorded
- Ppp. (a-)chambhita- Mv ii.281.15; chambhita-tva- SP 63.5; Gv 159.24; chambita-tva- Dhvaj. 47.6, 16, 24, ctc.

chid-, split: (1) chinda-ti, and once chindaya (impv., non-caus.): Pali id.; based on Skt. 3 pl. chinoanti

- Pres. chindămi Gv 231.20; chindati LV 200.9; Mv iii.362.10; RP 8.19; Ud x.6 (a later ms. chinatti); samucchindāmah Divy 22.20; chindate Mv i.169.14; opt. chindeya (mss. cha°) Mv ii.125.5; chindetsuh My iii.258.15, 18; impv. chinda My iii.384.16; chindaya (= chinda) Suv 133.8; chindāhi Mv ii.482.16; chindahi Mv iii.140.4; chindatha Suv 61.14
- chindişyati Mv ii.145.8; iii.403.18; chindişyam Mv Fut. ii.173.14 (1 sg., passive)
- Aor. chindire (3 sg.) Mmk 366.13 (§ 32.43)
- Ppp. chindita- Mınk 708.28
- Ger. chinditvā Mv ii.236.3; āchindya Bbh 166.23 (§ 35.13)
 - (a) chindayati Mmk 395.14, 18 Caus.
 - (b) Fut. chindāpayişyāmah Mv iii.403.9

(c) Fut. chedāpavisvati Mv iii.403.13 (§ 38.61)

(2) *china-li, based on Skt.-3 sg. imperf. acchinat, to which an analogical 2 sg. is authorized (optionally) by Pāņ. 8.2.75 (Kāś. cites acchinas as an example). While this acchinas has not been noted in Skt. literature, a further analogical creation, acchinani 1 sg., actually occurs in Mbh. (Crit. ed.) 3.22.3. The 'Prakrit Dhammapada' has china = MIndic chinda, Senart JA IX.12 (1898) 248 (and regularly in other cases, ibid. 212); but this may be orthographic for chinna, with nn from nd by phonetic change (so J. Bloch JA 1912, 1, 332, 334). My earlier explanation of the BHS forms (Lang. 13.115) is hereby withdrawn. See § 28.12

- ? Pres. chinnāmi Divy 417.8 (vs); if not an error, possibly a blend of *chināmi with chindāmi; assimilation of nd to nn, cf. above, has no parallel in our dialect chini LV 165.22 Aor.
- chinitvā LV 195.12; 196.2 (§ 35.28) Ger.

(3) *che-ti: seems implied by the following forms (one of which however is more than doubtful) and by AMg.

acche; see § 28.50. [Pres. impv. 2 pl. ucchethā LV 335.13 (vs), perhaps, destroy! All mss. and both edd. read so (not ucchetthā); followed by hanathā vilumpatha. But Tib. slon's, which would render Pali usseti = Skt. ucchrayati, raise (ucchrayate, rise, Vedic), and I incline to take ucchetha as MIndic for ucchretha, with intransitive meaning, arisel]

Ger. acchetvā Mv iii.285.6, 9 (for ā-ch°, so Senart em., cf. Pali acchindati)

(4) Pass. chidya-te (Skt.)

Pres. opt. chidyeya Sukh 45.13

chodaya-li (rare), choraya-li, leave, abandon (cf. Skt. chut-, chur-): AMg. chodiya-, JM. chodei, Ap. chodai chorayanti LV 321.2; prachodayati Kv 58.7, 8,

- Pres. yāmi 12; impv. chorayantu Divy 353.23
- chorayişyamah Mvy 8595-6; Prat 534.13 Fut.
- chorita- Mvy 2553; Divy 6.6; 82.23; 86.23; 185.28; Ppp. Kv 57.15, 16
- choditvā (v.l. °dayitvā) Mv iii.20.2; prachodayitvā Ger. Kv 55.16; 56.19; choritva SP 251.5; chorayitvā Mv ii.483.3; Divy 166.1; Av i.245.4; Kv 31.24; Mmk 107.27 (text sthor°); choretva Mv ii.391.22; 394.6.10
- Pass. choryate Divy 84.21

jan-, jā-, beget, be born: (1) jāya-te, be born (Skt.) Fut. jāsyante LV 371.22; prajāsyanti LV 398.6 (§ 31.3)

jāyitha (? 3 pl.) Mv i.10.13; prajāyitha (3 sg.) Mv iii.1.2; ajāyithā (3 sg.) Mv iii.32.16 (on these three Aor.

see § 32.42); jāye Mv ii.298.16 (so mss.) (2) janaya-ti, beget (Skt.)

- (2) Janaya (1, Diget (Sec.)
 Pres. opt. jane (2 sg.) Mv ii.336.7; janaye (3 sg.) Mv ii.295.14; janeyā LV 152.6 (§ 8.36); janeta SP 250.16; impv. samjanayāhi Mv ii.134.3; janehi SP 15.12
- jani (2 sg.) LV 91.17; janayi (3 pl., or ger.?) My Aor. ii.328.6; janayisu Mv ii.333.17; janaye Mv i.220.7; jane Mv ii.299.13; janeşī LV 91.21 (§ 32.63); janayetsuh Mv iii.359.9
- janitvā LV 234.7; janayitvāna Mv iii.77.8; sam-Ger. janetvāna Mv ii.205.1; samjaniya (or °yā) Mv i.146.9; upajenitvā (! read °jān°? Dict.) LV 36.20 (3) jānaya-li (= Skt. janayati). Not mentioned in
- Pischel or Geiger. See § 38.15 Pres. samjānayati Gv 502.23; opt. jānayi (1 sg.; not to
- jñā!) Bhad 45; impv. upajānaya Mv ii.135.4 samjānayişyanti LV 439.4 Fut.
- jānita- Gv 381.17 Ppp.
- jānayitvā Mv ii.131.10; samjānayitvā Mv i.238.1 Ger. (prose; so mss.; see Dict.); samjānayya LV 137.19

*jantaya-li or *jante-li, impel, make go; denom. to janta- (MIndic for yantra-, see Dict.) Ppp. jantita- LV 372.15 = Pali yantita-

jalpa-ti, speak (Skt.)

- Pres. impv. jalpāhi Mv ii.248.2
- samjalpeşu Mv i.41.8 (§ 32.95) Aor.
- (a) *jalpayati (Skt. Gr.): Aor. jalpesi Mv il.165.3, caused, ordered to say (§§ 32.65, 38.9) Caus. (b) jalpāpayati Mmk 421.21

jägr-, watch: (1) jägar-ti (Skt.). Regular Skt. forms occur (Pres. pratijāgarti Divy 124.9; °rmi Divy 306.12; impv. °grhi Av ii.145.2; Fut. °garişyāmah Mv iii.323.10). Also:

- pratijägrhansuh Mv i.231.11, anomalous (see Dict. Aor. and § 32.76)
- pratijägrta-, formed regularly but seemingly not recorded in Skt. (which has jägarita-) Mv i.258.16; Ppp. ii.111.8; 180.6; iii.324.10; § 34.14

(2) jāgara-li (cited in BR once from Mbh 12); Pali id.; based on

- Fut. jägarişyati, Ppp. jägarita-
- pratijāgaranti Mv i.147.11; parijāgaranti Mv i.210.17 Pres. = ii.14.16; opt. pratijāgareta (3 sg. mid.) Ud xvi.1; °retha (2 pl.) Mv ii.153.14

- Ger. pratijägaritvä Mv iii.141.15
- Inf pratijāgaritum Av i.257.14

Caus. impv. pratijāgarāpetha Mv ii.156.9; ger. perhaps (3) jāgra-li ('S.E.' in Wh., Roots): Pali jaggati;

§ 28.11

- Pres. opt. pratijägreta Ud v.15 (see Dict.); impv. pratijāgrāhi Mv ii.274.12 (so Senart em.; mss. °grehi, grāmi); pratijāgratha (v.l. °grthā) Mv ii.150.2
- parijāgrisu Mv i.76.4; pratijāgramsu (v.l. °grsuh) Aor. Mv i.259.3 (§ 32.76)
- Ppp. pratijāgrita- Mv i.270.13; ii.150.8 (both prose); \$ 34.10
- ji-, win: (1) jaya-li (Skt.)

Pres. opt. jaye (2 sg.) Mv ii.322.7

- Fut. parājisyāmi (§ 38.31) Mv ii.314.17
- jayitvā Mv i.264.6 Ger.
- parājayitum Divy 285.20 Inf.
- Caus. jāpayati (Vedic, on MIndic pattern); Pali and BHS, e. g. MSV ii.184.1, read ajāpayan (ms. °yah; ed. em. ahāp° with Pali). Pali shows ms. variation between jap° and hap° (PTSD)

 (2) *finā-ti* and *fina-ti*: Pali id. (Vedic jināti, commonly assigned to 'root jyā, ji'; in MIndic entirely equivalent to ji-, jayati; § 28.5

- Pres. nirjināti ŠP 289.3; jināti Mv iii.91.19; parājinanti Mv ii.260.11; 261.11; opt. jineya (1 sg.) Suv 37.9; impv. jinā LV 285.10; jinahi LV 12.19
- samnirjinişyāmi Mv ii.270.11 Fut.
- a-jinitvā Mv ii.315.13; abhinirjinitvā Mv i.52.9; jinitvana LV 364.9; jiniyā LV 326.17 Ger.

 (3) jāya-le (unparalleled; §§ 28.24, 33, and Dict.)
 Prés. jāyate, conquers, LV 262.12, 13 (ā not metrically required)

(4) jivali for jiyale, passive (§ 2.31 and Dict.) Pres. jīvati Mv iii.91.19

jīva-li, live (Skt.)

Pres. opt. jīve Mv iii.436.4; jīveya Mv ii.184.10 Caus. jīvāpayati Mmk 574.24

ju -: - java-ti, hurry, run, chiefly with anu: Pali (anu)javati; perhaps inheritance from Ved. javate; there is an Epic ppp. prajavita-; no Skt. *javişyati or other obvious source of an analogical creation seems recorded

Pres. anujavati Mv ii.212.17; nirjavāmi Gv 434.4; pple. anujavanto Mv ii.217.6

- anujavitvā Mv i.321.6 Ger.
- Inf. anujavitum Gv 400.7

juşa-ti, enjoy (Skt.) Inf. / juşitu-kāma- Mvy 6345

jr-: jaraya-li, digest (Skt.) Pres. opt. jareva LV 74.3

jrmbhe-li, yawn = Skt. jrmbhati Pres. vijrmbheti Mv ii.281.7 f. vijrmbhesi Mv ii.281.9 ff. (mss.) Aor.

jñā- know: (1) jāna-ti (Skt. 'U.E.', Wh. Roots): Pali id.

- Pres. jānati Bhad 55; °si Mv ii.493.1; samjānīte with pass. mg. (§ 37.18) LV 289.4-5; opt. parijāni SP 32.4; jāniyāt (§3.39) SP 31.5; jāne (2 sg.) Mv ii.184.8; anujāneyā Mv ii.119.15; anujāneyyā Mv i.351.18; jānesi (2 sg.) Mv i.275.17; jāneyāsi (2 sg.; § 29.37) Mv ii.488.15; impv. (cf. § 28.63) prajānāhi SP 91.7; jānātha LV 314.16
- jānişyati Mv i.244.11 Fut.
- abhijāni (3 pl.) LV 342.3; samjāne Mv ii.353.3; Aor. samjānetsuh Mv i.230.3

- Ppp. jānita- LN 414.16 (§ 34.10)
- vijānitvā Mv i.192.6; jāniya SP 326.11; vijāniyāna Ger. SP 116.1
- jānitum SP 31.10; jānitu Mv iii.34.1 (§ 36.3); Inf. vijānanāva LV 420.22 Gdve. jānitavya- Suv 180.10; prajān° Suv 179.5
- Pass. anujānīyati Mv ili.176.16
- Caus. impv. anujānāpehi Mv 1.256.14; Ger. anujānāpetvā My ii.174.17
- (2) jānaya-ti, also (m.c.) -janeti, non-caus. = jānati Pres. anujānaye Mv i.257.11; prajānayanti Sukh 9.8; pratijaneti (a m.c.; = °jānāti) Gv 481.22 (vs); opt. parijānayeyā LV 330.17
- (3) Caus. jhāpaya-ti (Skt.)
 Pres. opt. vijňāpema Mv i.360.5; pple. mid. in pass. mg. (§ 37.18) samjnāpayamānāh Mv i.23.11
- Aor.
- prajñāpayed (1 sg.) Mv iii.53.16 prajñāpayita- Mv iii.93.3, 4 (§ 34.8) Ppp.
- (4) Caus. jñapaya-li (Skt.)
- Pres. opt. vijňapeyā SP 284.8
- Aor. vijňapi Samadh 8.22; prajňapayet Mv i.325.8 (mss.)
- Ger. prajňapitvä SP 194.3
- Pass. prajňaplyanti Mv ii.274.4
- (5) Pass. jñāya-te (Skt.)
- prajñāyetsu(h) Mv i.339.2 Aor.
- (6) Miscellaneous forms (on anapeli, anapeli, see s.v.) Aor. ājnāsi Mv iii.48.19; ajnāsi, Senart em. for mss.
- anyāsi (§ 2.15) ili.386.18; §§ 32.48, 53 Pass. sanniyate, is known, Karmav 27.27 (§§ 2.15; 37.3 n. 1; Dict.)
- jvala-li, blaze (Skt.)

Pres. opt. jvaleyā Mv iii.317.9

- Intens. jäjvalanti SP 171.11 (Epic Skt. °lati)
- jhāşaya-li, destroy: not in MIndic; cf. Dhātup. jhaş-, destroy, and Ved. jāsayati id. Pres. jhāşayati Šikş 59.12; 60.12; 61.2; 89.3

thap-, see stha- (9)

(ud-)diya-le, fly (Skt.), cf. also 2 liuddīyitum Mv ii.243.6 Inf.

*dhala-li, dangle: Pkt. (Deśi) dhalai (pple. dhalanta, Sheth)

- Ppp. dhalita- LV 371.17
- Caus. pple. dhālayantam Sādh 569.19; 575.7
- taksa-ti, fashion (Skt.); also tacch- (MIndic, § 2.18). Note nistakșņuvanti Jm 196.4 (takșņuvanti cited only as 'B.S.' in Wh. Roots)
- Ppp. taksita- Mv i.5.8 (Skt. only tasta-); tacchita- Mv i.16.14 (Pali Lex. id., Childers), to Pali tacchati, taccheti
- Pass. takşīyanti (em. Senart) Mv i.16.13, 15

tanga-ti, stumble (Dhātup.; not in MIndic) Pres. impv. (in an etymology: mā) tanga Divy 632.1

- tana-li, extend: vi-t°, see Dict.; cf. Pali ger. vitanitvāna, Pass. vitaniyyati; AMg. Pass. taņijjae; based on tanoti, § 28.17
- Pres. pple. vitanatā MSV ii.156.1 (text vitar°)

Aor. vitane Mv ii.221.20 Ppp. vitanita- MSV ii.159.4 (text vitar^o)

lapya-li, be heated, grieved (Skt.)

- Pres. impv. anutapyähi Mv ii.428.3
- Ppp. tapyita- Mv iii.102.13 (so, or tappita-, mss.; Senart em. tapita-)

Gdve. anutapyaniya- Mv iii.74.4

larja-li, larjaya-li, threaten (Skt.)

Aor. tarje (2 sg.) Mv iii.32.5; samtarjayetsuh Mv iii.194.17

(pari-)lasya-li, be exhausted (see Dict.)

- Pres. paritasyati Mvy 6813; Bbh 193.6; pple. a-paritasyamāna- Bbh 218.9
- lädaya-ti, beat (Skt.), also läda-ti Pres. pple. tāda LV 341.13 (= tādan = tādayan), § 38.30
- tij-, Caus. lejaya-li, inflame (Skt.) Aor. samuttejaye (v.l. °yet) Mv i.334.1
- limaya-li, make wet (cf. Skt. tim-; timita, wet, Kaut. Arth. 2.15, Shamash. p. 95, line 3 from foot); Ap. ppp. timia- (Sheth) Ger. tīmayitvā Divy 285.25

lunalunaya-li, onomat., of the sound of the kokila; cf. tuna (Dict.); not in Mindic Pres. °yanti Mv iii.256.2 (v.l. bhunabhunā°)

lulaya-li, weigh (Skt.) Pres. impv. samtulehi Mv i.278.19

Ger. tulettä, °tta Suv 36.11 (? § 35.53)

lus-: Caus. losaya-li, gratify (Skt.) Pres. opt. atitoşaye Mv iii.388.9 Aor. toşi Sikş 346.6

tr-, pass; (1) tara-li (Skt.)

- Pres. opt. otari (1 sg.) Bhad 29; uttare LV 331.18; impv. otarāhi Mv ii.249.17; otarahi Gv 301.5
- Aor. otari (1 sg.) Gv 256.13; otarisu Gv 240.18; samuttaretsuh My ii.125.3
- Ppp. tarita- Mv ii.294.9
- taritvā LV 216.10; uttaritvāna Mv ili.301.5; otariya Ger. Gv 286.2
- Pass. (tariyati:) otariyati SP 358.12 (Dict.; to ut-tar-, conquer; ot- for utt-, § 3.73); Fut. tarihasi Mv iii.328.8, 9 (§ 31.3)
- (2) Caus. tāraya-ti (Skt.) Pres. opt. tārayeyā LV 300.17; tāreyam Mv i.39.5; impv. samtāra RP 49.15; tārehi SP 170.15; samtārayahī LV 178.18
- Fut. tārişye LV 361.4; [°]şyase LV 361.6 Aor. tārayi (3 pl.) SP 51.8; tāresi (1 sg.) Mv iii.77.13 Ger. tāritva SP 149.12; otāriyāna Mv ii.249.13
- Caus. ppp. otārāpita- Mv 1.273.16

(3) Pass. *-lira-li, MIndic = Skt. tiryate, with nis-, be accomplished

Caus. (*nistirayati:) Ppp. nistirita- Dbh 3.20 (§§ 37.28; 38.10)

trp-, be gratified: (1) trpya-ti (Skt.)

trpyetsuh Mv iii.425.12 Aor. (2) Caus. larpaya-li (Skt.)

- Pres. opt. tarpet SP 126.14; tarpeya SP 351.4; impv. tarpehi SP 190.12
- samtarpayi Mv iii.257.10; tarpesi (1 sg.) Mv iii.249.4; Aor. samtarpayet Mv i.325.11

lyaj-: (1) lyaja-li, abandon (Skt.)

- Pres. opt. tyaje Mv i.310.12; impv. parityajāhi Mv i.310.7
- Aor. tyaji (2 sg.) LV 165.9
- parityajitvā SP 408.15; tyaji LV 362.7 Ger.
- tyajitum Mv iii.151.12; Divy 416.26 Inf.
 - (2) lyajya-li, nowhere recorded (or pass. with active

ending and mg.? § 37.23) Imperf. atyajyat Gv 362.18

trapa-te, -ti, be abashed (Skt.) Ppp. vyapatrapita-vān (not caus.) Av i.272.9 Gdve. apatrapitavya- (not caus.) Bbh 223.12

tras-: (1) trasa-li, be frightened (Skt.) Pres. impv. trasāhi (em.) Mv ii.462.18 Aor. uttrasişuh samtrasişuh Asl² 17.7 (§ 32.36) Ppp. an-u(t)trașța- LV 12.16 (§ 2.61) Gdve. uttrasitavya- Sikș 200.14-15 (2) Caus. Irasaya-li (Skt.) Fut. vitrāsisyase LV 334.7

trāya-le (Skt.), -ti, save; also Mindic tāy-Pres. impv. trāyāhi Sādh 162.10 Ppp. trāyita- Mv ii.177.4; Gv 301.17 Gdve. paritäyitavya- (MIndic for °trã°) Dbh.g. 3(339).19

dandaya-li (denom.) punish (Skt.) Caus. dandāpayati Šiks 63.13; 67.10

damaya-li, subdue (Skt.) Ger. damiyāna SP 118.4

(vi-)darbhaya-li (denom.?), intertwine Pres. opt. vidarbhayet Sādh 532.18; 584.7 Ppp. vidarbhita- Sādh 125.6 Ger. vidarbhya Sādh 357.17

dal-, caus. dālaya-li, cause to burst or split (Skt.) Pres. opt. dāleya sampradāleya Mv ii.125.4; sampradāletsuh Mv iii.258.15

das-, bite: (1) dasa-li (Skt.)

Caus. daśāpayati Mmk 462.13

(2) damsa-li (once in Skt., pw): Pali damsati, AMg. damsai

Caus. opt. damśāpayet Mmk 463.6

dah-, burn: (1) daha-li (Skt.)

Ger. dahitvā Mv i.8.6

- Caus. (a) Pres. opt. dahāpaye (2 sg.) Mv iii.20.8; Ger. dahāpetvā Mv iii.20.9
 - (b) dāhāpayati Mmk 634.9 (§ 38.61)
- (2) Pass. dahya-le (Skt.) Caus. dahyayante LV 242.7 (§ 38.11)

dā-, give: (1) dada-li (Ved. and Epic Skt.): Pali id.; § 28.9. A few regular Skt. present forms are included as examples.

- Pres. dadāmo LV 368.22; dadanti LV 72.19; opt. daded SP 119.8; dadeya SP 229.4; impv. dada LV 43.1; dadahi LV 78.15; dadāhi (§ 28.61) Mv i.131.3; dadasva SP 321.2
- Fut. dadeham? Mv ii.246.11 (§ 31.21)
- ādade Mv ii.157.14 Aor.
- ādaditvā (so with WT) SP 283.9; daditvāna SP 334.10; dadiya LV 241.7; dadiyāna Mv ii.388.8; Ger. dada(t)tvā (? § 35.14) Mv ii.205.6 daditum LV 288.3

Inf.

(2) de-li: Pali id.; divergent theories as to origin, Geiger 143; Tedesco, JAOS 43.388; either the latter is right (to ppp. dita-), or these forms are analogical to stheti etc. (root sthā), § 28.48

Pres. demi LV 109.20; Mv i.93.6 etc.; desi Mv ii.428.19; deti LV 384.13; dema Mv i.299.20; denti LV 75.17; Mv i.23.12 (mss.); impv. detu Mv i.349.11; deth(a) LV 123.7; Mv i.46.10; dentu Sādh 2.2

adetsuh Mv iii.305.6 (§ 32.104) Aor.

(3) -daya-li (equivalent to -deti): § 28.48. But the forms are all dubious; see s.v. samādayali (Dict.), for which prob. read always either °dap°, caus., or °diy'

- Pres. samādayanti RP 15.6; pple. samādayanto SP 12.11, Nep. mss. (Kashgar rec. °dapenti, KN em. °dapento, kept by WT without note); dayanto (mss. °nte) Mv i.176.7 (prose; uncertain)
- (? or Ger.) samādayi Šiks 337.1, 11 Aor.
- samādayitvā Šiks 304.2, = Mv ii.382.14 where Ger. °diyitvä
- (4) da-li: Ap. dai (Jacobi, Bhav. 205.10); based on aor. adāt, Pali adā. Sce also parindāmi
- Pres. dāmi LV 58.20; 59.1, 5; dāsi Mv i.128.18; pradanti (with MIndic short a, § 3.34) Mv i.177.4
- adāsit (§ 32.55) SP 250.2; adāsi (§ 32.54) Mv i.37.12 Aor. (5) Pass. diya-le (Skt.)
- Fut. diyişyati Mv i.356.14, 16; dişyati Mv i.349.18 etc. (§§ 31.2, 3)
- (6) a-dīya-li, -diya-li, take (also upā°, paryā°, samā°):
- Pali adiyati. Spelling with i seems not recorded in Pali
- (but pariyādiyyati occurs, PTSD). See Dict. and § 37.23 ādīyati Mv iii.447.12; upādīyasi (mss.) Mv i.233.7; Pres. upādiyāmi Mv i.233.11; 234.8; samādiyāmi Mv i.145.8 = 202.5 = ii.6.2; opt. ādiyeyam (v.l. ādi°) Mv i.346.10; impv. upādiyāhi Mv i.233.17; ādiyatu Mv iii.217.17; 218.12; pple. ādiyantam (v.l. ādī°) Mv i.346.12; Passive in mg.: paryādīyante KP 5.2,
- see Dict. ādiyeya Mv i.346.11 (mss. °yam or ādīyeya; 3 sg.), Aor. \$ 32.105
- -adinna-, see below Ppp.
- samādiyitvā Mv i.128.9; ii.382.14; upādiyitvā (em.) Ger. Mv 1.89.17
- (7) Caus. -dāpaya-ti (Skt.); also as non-caus. = -diyali, see § 32.58
- Aor. samādāpaye Mv i.321.18
- (8) Caus. -dape-ti: Pali id. (§ 38.52)
- Pres. samādapetī (with all mss.) SP 23.6; °pemī SP 323.9;
- °pemo SP 56.4; pple. °pento SP 64.7 samādapinsu SP 51.8 (so with Kashgar rec.) Aor. (9) Miscellaneous forms
- Pres. opt. dadyā Mv i.80.6; dadyatha (mss.) Mv ii.105.3 (§ 29.43)
- [Fut.? dadyeham Mv ii.73.17, prob. corrupt; § 31.21]
- Ppp. dinna- (= Pali id.) Mv i.44.13, and often; § 34.16; KP 9.13; Mmk 56.21; samādinna- (to -ādiyati, above) My i.211.14 = ii.15.14; paryādinna- (to adiyati) Mv iii.429.11, 15 paryādatum and °dattum KP 33.2 and 3 (§ 36.12)
- Inf.
- dií- point, indicate: (1) diśa-li (Skt.) Pres. opt. nirdišeya SP 284.1
- Preterite diśa (= adiśat) LV 391.21 (§ 32.115)
- Ger. samdiśitvā Mv i.354.14
- vyavadiśitum Mv iii.50.8 Inf.
- Gdve. adiśitavya- Mv iii.420.11
- Pass. nirdiślyati Samādh p. 23 line 1
- (2) -diśya-li (not recorded)
- Pres. pple. uddiśyadbhir Divy 191.3 (Dict.) (3) -deśa-te (not recorded; denom.?) See also under (4) Pres. pple. vyapadeśamāna- Prāt 525.9, making represen-

tations, speaking; perh. denom., to vyapadeśa (4) Caus. deśaya-ti (Skt.), also as non-caus., § 38.23, and also desa-li, -le as caus.

Pres. deśati Suv 19.4; deśate Mv i.187.3; impv. deśantu Suv 23.8.—opt. desaya (? 1 sg.; § 29.35) Mv i.43.3; deśayi (1 sg.) Suv 51.14; deśeta SP 282.6; deśeyam Mv i.37.8; 60.6; deseya LV 393.2; impv. desehi Mv iii.124.14; desayahi LV 175.2—In non-caus. mg., uddeśenti, they point out, Mv i.21.3; 23.9

desayet Gv 146.3 (§ 32.87); desayuh or °yi (§§ 32.118, Aor. 121) SP 272.10 (5) Miscellaneous

nirdiştum (§ 36.9) Mınk 4.18 Inf.

dipaya-ti (caus.), light, trans. (Skt.) Caus. pple. pradipāpita- Mv iii.177.8

duhkha-li (denom.), be painful: Pali dukkhati Pres. duhkhati SP 352.7; duhkhanti SP 100.12 Caus. duhkhāpayasi Mv i.179.19

duş-, düş-: (1) düşaya-li, defile (Skt.) Aor. düşayetsuh Mv i.342.12

(2) dusya-ti. -te, dusya-te, become malicious or hateful (Dict.; this mg. acquired through association with dosa, hatred = Pali dosa, Skt. dveşa) Pres. duşyati KP 97.4; düşyate MadhK 457.4; opt.

duşyeta MadhK 143.1

dr-, split: Pass. dirya-le (Skt.), see § 34.10 Ppp. dirita- Mmk 143.24 (§ 37.28)

drś-, see: (1) drśa-li, and MIndic daś-, chiefly in pret. forms: based on Pali addasa etc. = Ved. aor. adrsat (dd by contamination with Pali addakkhi = Skt. adrāksit); \$\$ 28.12; 32.25

- Pres. drśanti Dbh.g. 8(344).16; opt. drśema, or with v.l. indic. drśama, Mv iii.427.6; opt. drśi Śikş 341.5
- addaśā (§ 32.111) Mv i.9.3; addaśi, addrśi (§ 32.25) Aor. LV 194.12, 22 (mss. d for dd); Mv ii.37.9 etc.; śāti (ib.) LV 75.6 etc.; addašensuh Mv ii.344.19; Setsuh (§ 32.103) Mv ii.312.7; adršetsuh Mv iii.409.13; adrsetsu, addasetsu Mv i.328.5, 6; adrsu (for °su) LV 27.22 (§ 32.118)

- (2) darśa-li, -le; see § 32.27 Pres.(?) darśe LV 222.15; if correct, 1 sg. pres. mid. with fut. mg.; but some mss. drakse, intending draksye, fut.
- vidarśi LV 49.10 (3 pl.); a(d)darśi, °śi, darśi (§ 32.27) Aor. Mv i.303.15 etc.; addarśā (§ 32.111) Mv ii.186.8, 14
- darśanāya SP 425.2; 431.2 etc.; darśanāye Mv Inf: iii.366.20

(3) Pret. -draksi; may be semi-MIndic for Skt. adrāksīt (or semi-Skt. for Pali ad(d)akkhi); or more likely a new aor. to Pres. *drakşati = Pali dakkhati, §§ 28.41; 32.26; Geiger 136.3 (who bases it on the fut., cf. saksati, root sak; but cf. also aor. Pali addakkhi); anudraksi Mv ii.54.3

(4) Pass. drsya-le (Skt.) Pres. active (§ 37.10 n. 2), drsyanti Lank 268.14 (vs), they

see, and samdrsyati, sees, Gv 523.21 (based on pass.; Dict.); passive, opt. drsye Mv iii.10.16 drsyisu LV 74.12; drsyetsu(h) Mv iii.94.15 Aor.

(5) Caus. darśaya-li (Skt.)

- Pres. opt. darśeya LV 284.5; darśaye (2 sg.) Mv iii.2.17; upadarśayesi SP 98.8; impv. upadarśehi Mv ii.426.10; pradarśayāhi SP 178.11 (Kashgar rec.) darśisye SP 237.9
- Fut.
- daršayi SP 49.7 (3 pl.); °yīm (§ 32.39) SP 90.3; darši Gv 254.25; daršinsu SP 55.12; daršaye Mv i.297.14; Aor. darsetsu Mv i.192.15
- Ger. vidarśiya Śikş 329.3; darśayi Śikş 328.17; Gv 240.26; 253.15; nidarsayi Šiks 330.8; 333.17; vidarśayi Śiks 333.17
- Pass. darśiyati LV 184.3, 4.-Uncertain is darśyanti LV 222.4 (vs); perh. for Skt. °te, are shown; but Tib. 'Sakra and Brahma show (ston, as if darsayanti!)

... saying, This is the way.' Ms. A drsyanti, are secn. Cf. § 28.26

(6) Miscellaneous

Fut. drksye RP 46.10 (§ 31.25)

Gdve. a-deksiya- and a-drāksaniya-(?), see Dict.

(pari-)deva-ti, mourn (Skt.) parideve Mv ii.33.8; paridevetsuh Mv ii.217.11 Aor. paridevitvā Mv ii.189.15 Ger.

dru, run: dravati (Skt.) abhidravitum MSV ii.34.4 (prose) inf.

*druha-li, be hostile: Skt. aor. (a)druhat; AMg. duhati (Ratnach.), duhai (Sheth, Addenda, iv.1263) Aor. druhi LV 165.18 (less likely ger.)

*(uva-)dvāraya-ti, close a door (denom.) Caus. Ger. avadvārāpayitvā Mv ii.490.1

(vi-)dvesa-le, hate: Pali viddesate (read so in Therig. 418, most mss. vindesate; Sdhp. 82 viddesaniya-); prob. denom. from vidveşa = Pali viddesa Pres. pple. vidveşamāņa- SP 94.8

dhandhāya-ti, -te (denom.), be slow, be dull: cf. Pali dandha-, adj. Usually spelled dhanv- (Dict.)

- Pres. dhandhāvati AsP 284.4; 326.4; pple. a-dhandhāyamäna- Bbh 284.7
- dhandhäyişyati AsP 176.9 Fut.
- dhandhāyita-tva, -tā AsP 31.16; 454.9, 11; Gv 451.7 Ppp.

dhama-ti, blow (Skt.), also dhamaya-ti, non-caus. Pres. opt. vidhamaye (2 sg.) LV 117.6; vidhameyā Mv ii.404.7

Special form of ppp. nirdhanta = Pali niddhanta; prob. also Pass. nirdhāyate, °ti, pres. pple. °yantāni; sec Dict. s.vv.; §§ 34.11; 37.38

dhā-, place: (1) dudhā-ti (Skt.); non-Skt. forms on the strong redupl. stem, as in MIndic, c. g. Pali gdve. saddahātabba- DN ii.346.22; § 28.52

- Pres. impv. śraddadhādhvam SP. 44.3 (prose, both edd., no v.l.); § 28.61
- (§ 31.8) abhisraddadhāsyasi Divy 7.29; °sye id. 8.1; Fut. śraddadhāsyanti SP 38.3, 5 (vss; in corresponding prose śraddhāsyanti), and Kashgar rec. at SP 231.4 (prose, text with 2 Nep. mss. sraddadhişyanti); abhisraddadhāsyanti Divy 14.24; Kv 23.14
- Ppp. abhiśraddadhāta- Divy 16.5 (§ 34.11)
- Inf. śraddadhātum Divy 6.21 (§ 36.11) Gdve. a-śraddadhāniya- (§ 34.22), acc. to Kern, Preface, vii, is read somewhere in SP by Kashgar rec. for Nep. a-śraddheya-; duh-śraddadheya SP 70.2 (vs; reading not certain; § 34.24)

(2) dadha-ti, -le (§ 28.9; such forms occur in RV and later, esp. in Epic; in MIndic they, or daha-, are regular)

- Pres. (a few samples; very common) abhisraddadhasi Divy 538.3; śraddadhanti I.V 89.21; 393.6; opt. śraddadhed SP 255.2; abhiśraddadheta SP 93.4; impv. avadadhata LV 409.10; pple. a-śraddadhantā(h) SP 95.6
- śraddadhişyanti SP 313.5; abhiśraddadhişyanti Mv Fut. iii.76.9
- śraddadhi and °dhit (§ 32.45) SP 113.10; samādadhi Aor. Mv ii.204.11 (vs; so read with one ms.)
- abhiśraddadhitvā SP 80.5 Ger.
- śraddadhitum Mv i.185.11 Inf.
- Gdve. abhiśraddadhitavya- Mv iii.76.6; a-śraddadhaniya-SP 230.7 (prose)

Pass. pple. anu-dadhyamāna- (?), see Dict. and § 37.37 (3) *daha-li: Pali id.; = prec. with MIndic h for dh (§ 2.35)

- dahitvā Mv ii.377.6; °tva id. 379.7; °tvāna id. Ger. 365.6; nidahiya Mv ii.320.16 (Senart em. nidahya) Caus. samnidähayanti MSV iv.87.8 (Dict.)
- (4) -dhā-ti: AMg. and other Pkt. dhāi (apparently not in Pali); based on Skt. aor. adhāt (?with fut. dhāsyati, inf. dhātum); § 28.51
- Pres. antardhāmi Gv 70.9; upanidhāti Ud ix.6 (reading not certain but probable); śraddhāti SP 336.2
- Ppp. sraddhāta- Mv i.187.6 (mss.), see § 34.11 and Dict. s.v. sadyam
- (5) -dha-le: AMg. -dhai (Pischel 500), -dhati, dhac (Ratnach. iv.654); analogical to -dhāmi (above, 4) on pattern bhavāmi : bhavate etc.; § 28.10

Pres. sa na śraddhate Divy 279.12 (prose)

(6) -dhe-ti: Pali id., AMg. -dhei; acc. to Geiger 142.2 based on impv. dhehi; in some cpds. occurs ppp. -dhita, from which this might easily be a back-formation; spellings are found with i for c, usually m.c. but sometimes, perhaps, as in MIndic for ě in closed syllable; § 28.48

- Pres. apidheti Mv ii.493.8; pidheti Mv ii.493.17; pranidheti Mv i.83.8; °dhenti Mv i.120.5; patisamdheti Mv iii.65.9, °dhenti 12, pratisamdheti 14; pranidhisi (2 sg., i m.c.; § 3.48) Gv 57.15, 20; pple. pranidhintāna (mss.; gen. pl.; § 3.49) Mv 1.124.18 samvidheşyanti Mv 1.354.12 Fut.
- Aor. praņidhesi Mv i.42.6; anupraņidhesi Mv i.335.11; \$ 32.64
- pidhita- Šikş 81.1 (cf. pithita s.v. pithayati); antardhita- Gv 444.13; samdhita- Mv ii.222.7 Ppp. (mss.), 17; § 34.7
- prauidhitva Mv i.96.10 (or to 5 above), § 35.28 Ger. samdhitum Mv iii.390.6 (or to 5) Inf.

(7) -dhaya-li, -le: § 28.48; cf. -dheti (above, 6); most of the forms recorded belong to srad-dhā, and might be regarded as denom. to the noun sraddha

- abhiśraddhayanti Suv 192.8 (prose; v.l. °dadhanti); Pres. śraddhayate, and pple. śraddhayamāna- Šiks 2.16 śraddhayita- Divy 437.1 (prose) Ppp.
- pidhayitvā Mmk 566.21, having closed (a door or Ger. gate); cf. pidhāyitvā (8, below) just above in 18. One of these may be an error of tradition for the other

(8) -dhāya-li, -le; chiefly in comp. with antara-, intrans., disappear; prob. based on old Skt. aor. pass. antar-adhāyi (this is a common form in Pali), 'was concealed', then, 'disappeared', which was reinterpreted in MIndic as antara-dhāyi, hence pres. antara-dhāyati (i being the regular MIndic aorist ending to stems in a); § 28.22. For this antara-hāvati is usually substituted in our language (9 below). The isolated pidhāyitvā is suspicious (below). In Pali cf. ppp. saddhāyita-, 'believed', Pv ii.8.5 (glossed in comm. 109.10 saddhāyitabba, 'credible'); this may be a denom. from saddhā, cf. (7) above Aor. antaradhāyişuh (most mss. °şu; § 32.30) LV 397.21-22 Ppp. antarahita-, see under -hāyati (9)

pidhāyitvā Mmk 566.18, see under (7) Ger.

(9) -haya-li, chiefly with antara-; = -dhayati (8); § 2.35; once also pari-hapita-. Not recorded in Pali or Pkt. except in ppp. antara-hita (cf. Skt. hita). Once antar-hā°; antara-hā° is much commoner in BHS than °dhā°

- Pres. antarahāyati Mv iii.410.12; °te Mv i.175.9 (mss.); impv. °tu My iii.346.21
- antarahāyi (= antar-adhāyi, see under 8) Mv Aor. ii.256.11; °hāye Mv i.339.18; 340.11; °hāyithā (v.l. °tha; 3 sg.; § 32.42) Mv ii.240.17; °hāyetsuh Mv i.231.2 (so one ms.); ii.258.3; 259.2; iii.93.2; °hāyişu Suv 158.1; °hāyasi (§ 32.62) Mv iii.316.10

- antarahāyita- (v.l. antarh°) Mv iii.424.16; antara-Ppp. hita- (= Skt. antarhita-) Mv i.50.14 etc.; Gv 325.16 antarahāvitum Mv ii.97.10, 20 Inf
- Pass. pple. antarahiyanto Mv ii.178.3 (mss. onti, onti)
- Caus. impv. antarahāpaya (em.?) Mv i.75.14; Fut. antarahāpayişyanti RP 17.15; Ppp. parihāpita- Mv ii.170.12; Ger. antarhāpayitvā Divy 329.12; in Mv iii.6.9 prob. read antarahāpayitvā (mss. °hāyatvā, Senart °hāyitvā, but trans. mg., having made to disappear)

(10) *-dhiha-li; nowhere recorded; analogous to sthihati from slhā-, q.v.; § 28.44

- Aor. 3 sg. samdhihe Mv ii 221.20 (Senart em. samdahe) (11) Miscellaneous
- samādhyate Lank 24.17; 33.8; 203.13 (denom. to samādhi? § 37.34; or false Sktization of Pali passive samādhiyate?)

dhāva-ti, run (Skt.)

Pres. impv. nirdhāvāhi Mv ii.454.20

- pradhāvitsuh Mv i.220.19; samdhāvetsuh Mv Аог. i.347.16
- samdhāvitvāna Mv i.55.8; 56.14 (by Senart's Ger. plausible em.); nirdhāviya Mv ii.31.15; nirdhāviyāna Mv ii.471.19; 475.6

dhu-, dhū-, shake: (1) dhuna-ti: Pali id.; based on Skt. dhunoti, rarely dhunāti; §§ 28.15, 17

- Pres. impv. dhuna LV 285.10 (read dhunā m.c.); vidhunāhi Sādh 4.16; pple. nidhunantah RP 29.1
- vidhunesyase LV 333.12 (passive; to 'Caus.', 3 Fut. below? but v.l. vidhunişyase, which perhaps read); vidhunişyasi (v.l. vidhū°) Mv ii.307.15
- Ppp. vidhūna- Mvy 6983 (see § 34.18 and Dict.)
- apadhunitvā Mv i.321.6; vidhuniya Sukh 23.8 Ger. (2) Pass. dhūya-le (Skt.)

(2) rass unigert (Sht.)
 (3) 'Caus.' dhūnaya-ti (Skt.; in mg. = simplex);
 perh. also vidhunayati implied by vidhuneşyase, above?

vidhūnitvā Mv ii.315.8 Ger.

dhūpaya-ti (denom.), perfume (Skt.) Pres. opt. dhūpeya Mv ii.373.17

- dhr-, hold: (1) dhāraya-ti (Skt.)
 Pres. opt. dhāre SP 255.10; 293.8; dhāraye SP 98.12; dhāreta (\$ 38.27) SP 255.14; dhāreya SP 229.6; dhārayi SP 393.8; impv. dhārehi SP 209.7
- dhārayi Mv i.235.6; °yişu SP 181.9 (Kashgar rec.); dhāresi Mv i.265.15; ii.234.10; dhārayesi Mv ii.233.9; Aor. dhāraye (mss.; 3 pl.) Mv ii.21.6; dhārayetsuh Mv i.220.14 (? so mss.); dhāretsuḥ Mv i.265.1 Pass. dhāriyāmi Mv ii.489.15 (v.l. dhāri°)
- Caus. impv. dhārāpayehi (so with mss.) Mv ii.447.14 (§ 38.20); Aor. dhārāpayet Mv ii.117.5

(2) dhara-ti, also dhare-ti, non-caus.: Pali dharati; § 28.13

Pres. opt. abhyuddhareyā Mv ii.393.8; pple. dharanto (gen. sg.) LV 191.7.-indic. 2 sg. dharesi Mv i.210.1 = ii.13.21 (prose; in reply to this, in vs, i.210.3 = ii.14.2 dhāremi); i.210.9 = ii.14.8 (vs) dharasi (v.l. °sī; § 32.62) Mv ii.356.9 Aor.

dharitum LV 29.9 Inf.

dhrs-, violate

dharşanāye Mv ii.320.8 Inf.

dhova-ti, wash, MIndic: Pali id.; once dhop-; related to Skt. dhāv- (o influenced by ppp. Pali dhota-, Geiger 34; less plausibly Pischel 482)

Pres. dhovati Mv ii.466.7; Šiks 155.1; dhovāmi Mv

ii.244.15; Šikş 154.18 (text dhop°, perhaps to be kept, see Dict.); Jm (App.) 241.1; dhovanti Mv i.168.12; impv. dhovāhi Mv ii.430.3; dhova Šikş 154.20; dhovatu Siks 154.19

- dhovişyati Mv iii.312.17; dhovişyam Sikş 154.19-20 su-dhovita- Mv ii.466.9; samdhovita- Mmk 60.7 Fut. Ppp. (see Dict.); also paridhauta- (cf. Skt. dhauta-) My
- iii.144.11 (Dict.) dhovitvā Mv ii.244.13; iii.313.7 Ger.
- dhovitu-kāma- Mv iii.312.15 Inf.
- Pass. dhovlyanti Mv ii.466.4, 5

1 dhyā-, meditate: (1) dhyāya-li, -le (Skt.); once -dhyeli (MIndic contraction?)

- Pres. pranidhyenti Gv 371.24 (Dict.)
- dhyāyişyase LV 334.21 Fut.
- dhyāyi SP 23.12; dhyāyi LV 133.21; pradhyāye Aor. Mv ii.283.3
- Ger.
- dhyāyitvā LV 164.9; dhyāyi LV 116.10 avadhyāyitum Divy 494.8; Av ii.139.8 Inf.
- Caus. nidhyāpayanti LV 187.19 (Pali nijjhāpeti; Skt. dhyāpayati only Gr.); dhyāpayati Mv iii.388.9 (? v.I. dhyāyeti (it text doubtful); Ppp. nidhyapta-Šikş 130.13 (= Pali nijjhatta-, Dict.), § 38.52
 (2) -dhyā-ti (Skt.), or -dhya-ti? § 28.10
- Pres. pple. avadhyanto (v.l. °tā) Mv i.9.4, musing upon, with acc.; so mss.; Senart em. avajjanto. The form may, however, belong in § 3.34

2 dhyā-, burn: dhyāya-li, false hyper-Skt.`= = Pali jhāyati (intrans., burn, be on fire), caus. jhāpeti (trans., burn). Here mostly trans., even when (in the mss.) containing no -p- (such forms should perhaps be emended by reading -p- for -y-, as caus.) Pres. abhidhyāyati Mv iii.341.6, was aflame, intrans.

- (see Dict.)
- dhyāyita- Mv ii.78.15; 174.11, burned, cremated Ppp. (read dhyāpita- ?)
- dhyāyetvā Mv i.302.12; 304.12 (mss.; Senart Ger. °tvāna) having burned, cremated (read dhyāpetvā ?)
- Caus. Ppp. dhyāpita-, cremated, Mv i.357.17; RP 57.2; Pass. pres. pple. dhyāpiyantānām (so Senart for mss. dhyāniy[°]) Mv i.126.2

(ud)-dhvasya-le, be insulted, pass. of ud-dhvams-; cf. Pali uddhamsati, AMg. uddhamsei, insult; § 38.13

- Pres. pple. uddhvasyamāna-, so read for text udvasya° Šiks 57.8
- Caus. ger. uddhvasyāpayitvā, so read for text udvasyā° Siks 57.6, having caused to be insulted

nad-, sound: (1) nada-ti (Skt.)

- Pres. opt. anunade Mv ii.282.12
- Fut.
- sampranadahe (? 1 sg.; § 31.20) LV 338.19 nadi (1 sg.) Sukh 24.2; nadinsu Hoernle MR 152 Aor. for SP 271.2; anunade Mv ii.282.10; ninadetsuh Mv i.71.7 (so read)
- naditvā Mv iii.410.17; °tvāna Mv iii.24.3; nadiya Ger. Mv 1.206.5
- Inf. naditu LV 46.6 (2) *-nadya-ti, nowhere recorded
- Aor. praņadyetsuh Mv ii.160.19

nanda-li, rejoice (Skt.)

- Pres. opt. abhinandevā Mv iii.387.18 (Senart em. °ya); impv. nandāhi Mv ii.187.9
- abhinandi Mv ii.226.10; nandişu LV 76.2; ninan-Aor. dimsuh (v.l. °tsuh) Mv iii.267.8; abhyanande (3 pl.) Mv ii.139.22; abhinandetsuh Mv i.335.7

abhinanditvā Mv ii.443.17; abhinandiya Mv i.149.18 Ger. [Caus. aor.: in Mv ii.62.11 probably abhinandayet or °ye

is false reading for abhinadaye(t), made resound; cf. corresponding Pali Jat. v.409.4 abhinādayittha]

nam-, bend, bow: (1) nama-ti (Skt.), also namaya-ti, non-caus.

- namayati I.V 398.1; opt. samname (1 sg.) Mv ii.93.9; name (3 sg.) Mv ii.395.11 Pres.
- onami (3 pl.) LV 356.19; namișu LV 92.8; onamișța Aor. LV 295.7 (§ 32.46); name Mv i.322.20
- onamitvā Mv iii.98.9; praņamayya Av i.347.4; Ger. ii.3.9; Divy 463.22 (?)
- Gdve. samnamitavya- Mv ii.230.15; avanamitavya- Gv 507.1

(2) -nāma-li, nowhere recorded: denom.? or modelled on pairs of equivalent presents such as kramati and krāmati? § 28.33. Or to §§ 38.23 and 27? Cf. (3) below

Pres. opt. pariņāmet, would be digested, LV 64.20 (prose; all mss. reported with ā, only Calc. a); cf. pariņāma-, noun (?)

(3) Caus. nāmaya-li (Skt.), and rarely nāma-li (§ 38.28); once nam-, probably m.c.; on pariņāmayati (sometimes intrans., may be denom., or to § 38.23) see Dict. and (2) above

- Pres. nāmanti Šiks 324.18; upanāmemi Mv iii.135.13; opt. nāmaye Mv ii.367.13; impv. upanāmehi Mv iii.352.6
- praņāmi Mv iii.147.19; nāmayī LV 74.2; upanāmayi Aor. Mv ii.38.12; praņāmaye Mv i.303.14; nirņāmesi Mv i.228.12; abhinirnāmayesi Mv ii.284.7; abhināmayet Mv i.325.8; abhiprāņamayetsuh Mv i.65.18 (probably m.c. for °nām°)
- Ppp.
- upanāmayita- Mv ii.168.6 (§ 34.8) praņāmetvā Mv i.255.12; °tvāna Mv iii.97.14 (by Ger. em.)
- Inf.
- samnāmanāya Divy 446.20 upanāmīyanti (v.l. °miy°) Mv ii.465.10; upanāmi-Pass. yanti Mv ii.463.10; 464.9 (here v.l. °miy°); opt. (read with best mss.) upanāmye 'yam LV 386.17

namasya-li (denom., Skt.) and namasyaya-li, non-caus. Pres. namasyayanti LV 119.2

Aor. namasyi (mss.) Mv ii.298.10

- naś-, perish: (1) naśya-ti (Skt.) pranaśyire Mmk 172.13 (§ 32.43) Аог.
- (2) Miscellaneous
- Aor. pranańksit AsP 112.17 (§ 32.83); cf. Skt. fut. nańkşyati

nah-, bind: (1) -naha-ti; not recorded, unless implied by Pali ger. samnahitvä (beside samnayhitvä) Jät. i.273.23 Pres. onahati, or (v.l.) °te SP 126.2

- Aor. paryavanahe Mv i.346.2
- samnahitvā (= Pali, above) Mv i.30.6, 7; ii.410.3 Ger. (2) Miscellaneous
- samnatsyāmi Šikş 283.4 (natsyate Skt. Gr.); Fut. § 31.23

ninda-li, reproach (Skt.)

Pres. impv. nindāhi Mv il.448.4

nipacchita, see nepatthita

nī-, lead: (1) naya-li (Skt.), ne-li: Pali both

- Pres. opt. vineyam Mv i.337.8; vineya Mv i.43.4; apanaye Mv ii.393.13; impv. vyapanehi SP 15.12; anayahi Mv ii.170.6
- upanayi (1 sg.) Sukh 23.10; upanaye Mv ii.299.12; Aor. nayetsuh My i.346.6; upanayāsi Mv iii.429.14 (§ 32.59); vinesi Mv i.297.13 etc. (§ 32.63); vinayesi Mv iii.428.10

- samudānita- SP 178.13; LV 414.16 (§§ 3.43, 34.7, Ppp. 38.3); cf. samudānana, Dict.
- apanayitvā SP 106.4; netvā (§ 35.25) Mv ii.168.8; Ger. samudānayitvanā LV 178.15: samudāniya (§§ 3.43, 38.3) LV 394.17; apaniya (§ 35.44) Sukh 23.9; upanayi IN 116.10
- Caus. Fut. anapayisyam Mv iii.125.17; °pesyam Mv ii.103.5; Ppp. ānāpita- Mv ii.76.4

(2) ni-le, act. or pass. in meaning; § 24.12

- Pres. samudānīmahe Divy 92.9 (prose), we assemble or procure (merchandise). Could be taken as indic. or opt. Hardly to be connected with Vedic forms like anitam. In line 12 same expression with samudanayanti, 3 pl. Prob. analogical to ppp. nita- on some such pattern as adhita- : adhite, or more remotely yāta- : yāti etc.
- Fut. nişyati, pass., will be taken, Mv ii.110.1 (§ 31.3) Gdve. a-viniya-, not to be disciplined (if not error for
- avineya or avinita), MSV iii.128.20; § 34.27
- (3) Caus. nāyaya-le (Skt.), in non-caus. mg. (§ 38.23) apanāyayişyase LV 365.3, you will dispel Fut.

nuda-li, impel (Skt.)

Aor. pranude Mv iii.286.7

(Caus. nodayati:) Aor. vinodayc (2 sg.) Mv ii.186.20; vinodetsu Mv i.250.10. In non-caus. mg., prativinodayati, dispel (§ 38.23), see Dict.

nepatthila-, nepacchila-, or the like, clothed; perhaps read nev- (MIndic, § 2.30); ppp. of denom. from Skt. nepathya-, raiment, see Lüders, Kl. Skt. Texte ii.44

Ppp. nipacchita-, Kalpanāmaņditikā (Lūders l.c.); corruptly Divy 48.24; 49.16; 159.12; 342.2; 463.26; 558.22; text naiva sthita-, nopasthita-, or te pan-thita-; Av i.284.11; 347.9, ms. neva sthita-, nepathvasthita-

pamsaya-li, -le, besmirch (Skt. Gr.): AMg. pamsei

- Pres. pamsayati Siks 158.3; Bbh 107.10; 174.21; prati-p° KP 107.3; pamsayanti Siks 62.3; 63.6; 90.14, 16; °yate KP 107.24; opt. °yema Sikş 98.17; °yeyur 186.6
- Ppp. pamsita- KP 107.2, 4; °ta-vant- Siks 68.16

[paksa-li, adhere to (a party), denom. (Skt. Gr.; nowhere else recorded)

Pres. pakşanti LV 88.15; prob. error, see Dict.]

puc-, cook: (1) paca-li (Skt.)

pacitvā Mv ii.65.9 Ger.

- (2) Pass. pacya-te (Skt.), MIndic pacca-
- Pres. pple. paccamāna- (v.l. ya°) Mv iii.45.14; sampaccamāna- (mss. samya[°]) Mv iii.43.3 paripaccişyati Mv ii.311.8 (pass.)
- Fut.
- (3) Caus. pācaya-ti (Skt.), also pāca-ti
- Pres. paripācati Šiks 325.2 (§ 38.28) Aor. paripācayī SP 204.13
- Ger. prapācayitvā Gv 411.20
- (4) Miscellaneous
- Ppp. pakva-māna-, being baked, Divy 510.8 (§ 34.1)

(pața-li,) pāțaya-li, tear (Skt.)

Ppp. pațita- Mv i.5.8 (§ 38.9)

- Ger. pățitva SP 85.12; [utpățayitu? § 35.55] Pass. Ger. utpățya, being torn up, I.V 194.10 (contrast
- utpāțya, tearing out, Divy 417.6), § 35.17

pat-, fly, fall: (1) pata-ti (Skt.)

Pres. pranipati (1 sg. mid.) LV 361.19; opt. prapate Mv ii.334.12; pateyā Mv ii.324.21; impv. samnipatitha Mv i.329.17 (§ 27.10)

Perfect papeture Mmk 593,18 (§ 33.5)

- Aor. nipati LV 282.3; prapatisu LV 329.13; prapatisu or °tişu Mv ii.326.10; 333.15 (Dict. s.v. prapati); prapatinsu Hoernle MR 136 for SP 329.3; patimsū LV 233.18; samnipate (3 pl.) Mv i.303.19; prapatetsuh Mv i.42.1 ff.; samnipatensuh Mv iii.113.12 nipatitva LV 119.2; samnipatitvāna Mv iii.42.10 Ger.
- (2) *ut-patta-ti, go forth, fly up; see Dict.
- Ppp. utpattitā (mss. ^oto) Mv i.26.3 Ger. utpattitvā Mv i.6.8 ff.; ii.243.11; utpattitvāna Mv iii.456.5 (parallel to i.11.15 where mss. are confused, Senart utpatitvāna) (3) Caus. pålaya-li (Skt.)
- Pres. impv. prapātehi Mv iii.168.5; in non-caus. mg. (§ 38.23; or denom.? Dict.), pplc. vinipātentā My i.30.3
- Fut. prapătișyase LV 334.9
- Aor. pātavetsuh My iii.428.7

vinipätanāva KP 48.7 (vs) Inf.

palliya-ti, paliya-ti (rare), pratiya-ti, believe; cf. Pali pattiyāyati, Saur. Mg. pattiāadi, but (closer to our form) AMg. JM. pattiyai, pattiai; derived by Pischel (487), Sheth, Ratnach., from prati with yā or i, which may be supported by our occasional pratiy°, if this is not false Hyperskt. But PTSD takes the Pali verb as denom. to -pattiya in para-pattiya, believing someone else (so correct PTSD; always adj.; there is no noun meaning 'belief in someone else')

- Pres. pattīyati Mv ii.110.9; 208.6; iii.189.14; KP 16.8; vyasi Šikş 174.7, 18; vyanti Šikş 174.15; pratiyasi Kv 54.12, 16; impv. pattiyata SP 44.3; prativatha Sukh 97.5; 99.6; pratīya Mv ii.223.20, mss., Senart em. pattīya, perhaps rightly (metr. better) pattīyişyanti SP 286.8; 312.9; pratīyişyanti SP
- Fut. 36.7; 37.10 (Kashgar rec. patti° both times); patisyanti Mmk 73.15 (§ 31.4)
- Ppp. pattiyita- Mv ii.249.14
- Caus. Fut. pattīyāpayişyāmy SP 288.5 (prose; Kashgar rec. pattīyāpayāmi)

pad-, attain: (1) -padya-le, -li (Skt.); also -padyesi (Aor.), -padye-tum (Inf., helow), § 38.21

- Pres. āpadyati (in caus. mg., § 38.24) Mv i.181.10; opt. (caus. mg., § 38.24) utpadyema Mv i.365.10; (not caus.) utpadye Mv iii.6.19; impv. upapadyāhi Mv ii.2.12; upapadyahi Mv ii.2.10
- āpadyişyanti Mv i.360.5; upapadyişyati Mv iii.42.20; Fut. apady isyant in v houses a papady isyant in the initial space of the initial same pady is a second state of the initial same pady is a second state of the initial second
- Aor. samupapadye, also aor.); upapadyişu Mv ii.350.7; utpadye Mv i.59.14; °yet Mv ii.194.8; pratipadyetsuh Mv iii.289.11; pratipadyitha Mv i.12.9 (§ 32.42); upapadyitha (id.) i.268.14; 305.2; ii.220.7; upapadyesi (mss.; 3 sg.) Mv 1.45.10 (§ 32.67) samāpadyitvā Mv i.357.16; upapadyitvā Mv iii.404.7
- Ger.
- Inf. upapadyitum Mv iii.404.8; samāpadyetum Mmk 514.24; pratipadyetum Mmk 540.28 (§ 36.8) (2) Caus. -pādaya-li (Skt.) Pres. opt. utpādayi SP 284.3; nispādayi (3 pl.) SP 131.2;
- impv. utpādehi Mv i.233.10
- Aor. udapādayi Mv i.190.16; utpādesi Mv i.38.10; utpādaye (mss. °yo) Mv i.181.7; samupādayetsuņ (§§ 2.88; 32.103) Suv 151.11; utpādetsuņ Mv i.44.7
- Gdve. samupādanīya- (§ 2.88) Šikş 107.18 Pass. (a) (utpādyate, Skt.) Aor. utpādye Mv i.3.4 (b) pratipādīyati Divy 226.21 (mss.) (3) Miscellaneous

Pres. pple. -utpadanta-, §§ 18.5; 28.30

(Fut. -padisyati, only by error, see § 31.1)

udapāsi, originated, was produced, My i.248.7 etc. Aor. (§ 32.60)

parikarmaya-li, adorn, make fit (Skt.), denom. parikarmaye (mss.) Mv i.3.12 Aor.

parinda-li, parinda-li, present (also anup-°), not recorded elsewhere; taken by Burnouf, Lotus 417, as connected with da-; see Dict.

- Pres. parindāmi SP 484.5; LV 443.9 (so with v.l., text paridāmi); anuparindāmi SP 410.10; 420.12; 484.5; I.V 91.8; 443.9; Dbh.g. 56(82).5; parindāmy anu-parindāmi AsP 460.14; 461.12
- Fut. anuparindișyāmo LV 90.22
- parindita- Mv i.141.13 (°tārtha-); parindita- Vaj Ppp. 20.4, 10 (but in Hoernie MR 179 parin°); AsP 218.18

palây-, palāya-li, flee (Skt.)

palāta- Šikş 347.4 (§ 34.12); palāna- Mv ii.172.16; nispalāna- MSV i.288.10; prapalīna- (? most mss.) Ppp. Mv i.71.2; § 34.19

Caus. pple. nispālāyita- Bhīk 28b.1 (§ 38.14)

pallāna-li, saddle, harness: JM. pallānei, Skt. paryānayati (§ 2.16), denom.

Pres. pple. pallanatena (so with mss.; instr.) My ii.160.4

paśya-ti, see (Skt.); once -paśyaya-te, non-caus. (pres. pple.), also pasye-hi (impv.), § 38.21

Pres. opt. paśye Mv ii.360.22; paśyeyā, °ya Mv ii.150.6, 13; 151.19; 153.18; paśyiya Suv 52.5; paśyesi (2 sg.) SP 98.4; impv. paśyāhi Mv ii.28.4; paśyahi Mv ii.447.2 (mss.); paśychi Mv i.187.4 (mss.); pple. vipasyayamāna- Bhad 39

paśyişyāmi Mv ii.479.3 Fut.

- sampaśyi (§ 32.121; 2 sg.) LV 195.20; paśyi (3 pl.) Aor. LV 75.15; paśyisu SP 15.5; su SP 194.7; paśyetsu(h) Mv i.342.12
- paśya Mv i.192.9; vipaśya SP 282.6 (§ 35.16); Ger. paśyitvā Mv i.302.10; paśyiya Mv i.312.14; paśyiyāna Mv ii.225.11; paśyi Dbh.g. 4(340).19; [paśyitūna, see § 35.36]
- paśyitum Samādh 22.17; paśyatum (or °tu, mss.) Inf Mv ii.484.15 (§ 36.7); paśyanāya Mv ii.450.14; paśyanāye Mv ii.456.8, 14

Pass. (§ 37.20) paśyati LV 419.12; opt. paśyeta LV 29.8 Caus. paśyayanti Dbh.g. 48(74).26 (§ 38.9)

 $p\bar{d}$ -, drink: (1) piba-ti (Skt.), also with v for b Pres. impv. pibāhi Mv ii.430.16

- Fut. plbişyati Mv ii.242.11
- pibe Mv iii.173.7 Aor.
- pibitvā Mv iii.144.14 Ger.
- pivitum LV 309.15; pibanāye Mv ii.432.4, 5 Inf.
- Caus. opt. pivāpayet, and Inf. pivāpayetu-(kāma-), § 36.8, Kashgar rec. of SP for pāyayeyam, pāyayitu-, 322.2 (prose).

(2) Miscellaneous forms

- pitvā; so both edd. LV 242.14 (?MIndic i for i; Ger. cf. § 3.38; Pali records pitvā and pitvā). But Lefm. Crit. App. cites pitvā as v.l. of several mss. and Calc., implying that he meant to print a different reading (presumably pitvā) in his text Caus. pāyanti (for pāyayanti, § 38.28) Mv i.8.5 (mss.)
- pāpayati Divy 398.17 (§ 38.64)

pālaya-li, protect (Skt.), denom. Pres. impv. pālehi LV 79.17

*piccaya-li (Skt. Gr.) or *picca-li, squeeze, torment

(piccita-, Suśr.); not in Pali, Pkt. Ppp. piccita- Mv i.21.14; 24.4

- Pass. picciyanti Mv i.24.3, 5

pilhaya-li, rarely pilha-li, cover, shut, block; MIndic, see Dict.

- Pres. pithayati LV 257.2; pithanti Gv 63.15 (prose; 2d ed. em. wrongly); opt. pithayet Mmk 287.1; 362.15; impv. pithetha LV 201.17; pithayata LV 228.2; pple. pithad Siks 348.11
- pithayişyati Gv 112.20 (text pithapayisyati, which Fut. might perhaps be fut. to a caus. pithapayati, like dapeti to da, etc.)
- Ppp. pithita- SP 260.9; Siks 307.3; su-pithita- (§ 3.18) Gv 55.1
- pithayitvā Mink 560.27 Ger.
- pithitum LV 46.7 Inf.
- Pass. pithiyate Ud xvi.9, 10; Caus. (to pithiyati) 2 sg. opt. pithiyevāsi (so, or pihi°, mss.), you are to cause to be covered Mv iii.25.6 (§§ 29.37; 38.12)

pilipalipāye-li, rattle (onomat.); nowhere recorded Pres. °pāyeti, and impv. °pāyehi Mv iii.179.16

pīdaya-li, oppress (Skt.)

Pres. pple. °yantā, with pass. mg. (§ 37.18), Mv i.5.10

pidesi (2 sg.) Mv ii.175.13; abhipidayet Mv i.68.15; Aor. abhinipīde (1 sg.) Mv ii.124.2

Ger. nipīdiya Mv i.113.15

pīşaya-li, pīşa-li, crush: AMg. pīsai; cf. ápīşan AV 4.6.7, praty-apīşan Mbh 2.42.27 (Crit. ed.); Pali has, besides pimsati, pass. pisiyati and ppp. pisita- which point to a Skt. *pisati (epic piseyam etc., for regular pinașți); see § 28.4

Pres. pişayanti Mv ii.273.15; opt. pişayet Mmk 82.15; Suv 105.4; impv. pişehi Mv ii.427.9; pişahi Mv iii.3.17; pple. nispīsanto MSV ii.76.7

٦

- pişetsuh Mv ii.275.6 Aor.
- pişayitvä Mınk 81.29; Divy 409.18 Ger.

pu-, cleanse, winnow: *puna-li, § 28.5 Caus. Gdve. opunāpayitavya- Mv iii.178.5

punch-, see ponch-

pus-, Caus. posaya-li, make thrive (Skt.) Ger. poșiya Mv ii.241.15

Pass. pple. poșiyantă Mv ii.223.9 (mss.)

pū-, see pu-

pūjaya-li, revere (Skt.), denom.

- pūjima (§ 27.10) Dbh.g. 19(355).9; opt. pūjeva Mv i.270.1; impv. pūjehi Mv i.234.3 Pres.
- pūjayi Mv i.61.16; pūjayesi Mv i.60.1 (see § 32.66); Aor. pujayet Mv ii.119.3; vetsuh Mv ii.286.12
- pūjitva SP 217.9 Ger.
- pūjanāya RP 5.10; pūjanā LV 282.8 (§ 36.18) Inf.

pr-, fill: 'Caus.' pūraya-ti (Skt.), also pūra-ti, pura(ya)ti, pürya-ti

paripūrati (v.l. °rayati, apparently unmetr., § 38.28) Pres. Mv iii.356.6; opt. pureya (u m.c. ? Dict.) Suv 37.13; paripūrye Šiks 34.1 (prose; Dict.); prapūraye (1 sg.) Suv 30.4; pūrayi (1 sg.) Bhad 41; prapūreyam Mv i.53.10

pūrişu LV 387.3 Aor.

pūritvana KP 20.23 (vs) Ger.

Pass. pple. puriyantasya Mv ii.76.10; impv. (3 sg., § 30.14) paripūryato LV 235.19

220

prech-, ask: (1) precha-li (Skt.), also preche-li, noncaus., § 38.21

- Pres. prechenti Mv i.256.10; pariprechesi Mv ii.298.3 (v.l. °asi); opt. preche (2 sg.) LV 231.17; prechesi (2 sg.) Mv i.254.17; impv. prechāhi Mv ii.155.5 prechişyate Suv 216.9 Fut.
- prcchi (3 pl.) LV 421.18; prcchet Mv iii.242.6; Aor. prechetsu(h) Mv ii.271.1; prechiya (§ 32.105) Mv iii.85.2
- prcchita- Mv i.232.15; Suv 134.3 Ppp.
- a-prcchitvā Suv 142.6; an-āprcchiyāna Mv ii.272.12 Ger.
- prechitum SP 280.1; prechatum (mss.; § 36.7) Mv Inf.
- iii.394.17; pariprechanāya Mv iii.48.8
- Gdve. prcchitavya- Mv i.272.4; pariprcchaniya- Mv i.337.17
- Pass. prcchiyati Mv ii.13.21; prcchiyati (v.l. prcchi°) Mv ii.207.3; iii.39.19; prcchasi (§ 37.32) Mv i.280.5 (2) Miscellaneous forms
- Ppp. prasta- Mv iii.166.12; § 34.13

Blend of ppp. and pres. pple. prstamana (§ 34.1) Mvy 6993

prś-, prs-, see sprś-

poñcha-le (also puñch-?), wipe oil, clean (MIndic): Pali puñchati, Pkt. pumchai (Skt. proñchati)

Pres. poñchate Divy 491.12; impv. poñchasva Divy 491.7 puñchitum (? so mss. in 7) or poñchitum Divy Inf. 491.7, 8

pracalāya-ti, -le, nod (the head, in sleep): Pali pacalāyati (Skt. ppp. pracalāyita- and n. act. °yana-)

Pres. pple. pracalāyatah (gen. sg.) Jm 233.8; °yamāna-Gv 405.9

*prajñapla(ya)-li, provide (scats), denom. to prajňapta Caus. ppp. prajňaptāpita- Mv iii.93.3, 4 (§ 38.56)

prajňāya-ti (denom.), know, be conscious of (from prajñā, noun); also -prajñāya-te, there is consciousness (of . . .)

Pres. prajňāyanti Mv i.52.10; (jāti-, etc.) -prajňāyate LV 19.12 (wrongly Weller 18)

pralibhāge-ti (denom.), give a share in

Pres. impv. pratibhāgehi Mv il.426.8 (by em.; mss. ^obhā-nehi, ^obhāņehi)

pratiya-li, see pattiya-li

prath-, spread (Skt.) Ppp. prthita- Siks 247.16 (§ 34.14)

praśna-li, -le, interrogate (denom.): Pali paripañhati AN v.16.2 (Skt. praśnayati, which also occurs here); \$ 38.35

Pres. praśnasi Mink 666.23 (prose); paripraśnase Mmk 218.17 (prose)

Gdve: paripraśnitavya- Mink 229.16 (prose)

prasavali, prasavāyati, deliver (of a child); prob. denom. to Skt. prasava

Ppp. prasavitā Divy 485.12; prasavāyitā 484.8 (see Dict. and § 38.38)

1 phara-li, suffuse, fill (also 1 phala-, sphr-, qq.v.): Pali pharati

pharanti Gv 236.6: opt. pharetha, pharema Mv Pres. iii.374.11, 16 (mss. har°)

phari Gv 240.25 Aor.

Ger. pharitvā Gv 236.8

2 pharali, jump (= Skt. phalati?) Ger. utpharitvā(?) Mv ii.249.16 (see Dict.)

pharapharaya-le, crash(?), onomat.: Ap. pharaharantu (Dict.)

Pres. °yate Mmk 674.4

1 phala-ti, suffuse, fill (= 1 phara-, sphr-)

- Aor. phali Mv ii.349.17
- phalitva (mss.) Mv iii.124.15 Ger.
- 2 phala-li, burst (Skt.)

Pres. opt. phaleyā LV 81.19

bandh-, bind: (1) bandha-ti (very rare in Skt.): Pali id.; § 28.14

- Pres. bandhati Mv ii.90.16; opt. bandheya Mv ii.178.16; impv. ābandhāhi Mv iii.25.5; bandhatha Mv i.329.17
- anubandhişu Mv ii.198.11; anubandhensu Mv Aor. ili.96.22; °tsuh Mv iii.323.2
- bandhitvä SP 273.10; prabandhitväna Mv iii.281.19; Ger. bandhiya Mv i.272.11, 12
- Inf. bandhitum (reported once from Ram., BR) Mv ii.484.5 Gdve. bandhitavya- Mv iii.55.8; a-vibandhiya- Dbh.g
- 41(67).4 (§ 34.26) Caus. impv. bandhāpetha Mv i.258.9; ger. °petvā Mv
 - iii.175.6; gdve. bandhāpayitavya- Mmk 529.20
- (2) bandhaya-li, bandhe-li, non-caus. § 38.21
 Pres. bandhesi Mv ii.483.7 = iii.18.10; anubandheti Mv iii.294.19; anubandhayāmah LV 47.7, 21; 48.7; impv. anubandhayatām LV 46.2, 12
- bandhayitvā Mv ii.168.6 Ger.
- (3) baddha-li (*baddhayati?), denom. to baddha-, \$ 28.19
- Pres. impv. abaddhahi Mv ii.492.8; anubaddhitam LV 45.4 (§ 27.10); pple. anubaddhanto Mv ii.241.5 (but cf. § 34.2); [baddhamäna-, see § 34.1]
- baddhitvā Mv iii.7.5 Ger.
- Caus. Gdve. baddhāpayitavya- Mmk 50.16

baliya-ti (denom.), grow strong, prevail: Pali id.

Pres. °ti Mv i.275.10; ii.423.10 (here text bahviyati; see Dict.)

bådh-, oppress: (1) bådha-ti (Skt.)

- Aor. bādhişu LV 357.6 (read °şū)
- (2) Pass. bādhya-te (Skt.)
- Fut. bādhyişyate LV 366.2 (? by em.)
- Aor. vādhyī LV 222.6 (3 pl.)
- (3) ? -bādhya-li (active, trans.) Pres. sambādhyati Mv ii.146.1; so text with one ms.; the other bodhati, perhaps indicating that the true reading was (sam-)bådhati

bāspāya-li, -le (denom.), steam, turn to vapor Pres. pple. bāspāyantah LV 251.8, 12; vāspāyamāna- Divy 462.2 (here text with mss. vapya°), 6

Aor. bāspāyetsu(h) Mv ii.124.4, 5, 8

bāhaya-ti, remove, put out or aside: Pali bāheti (cf. Senart's note Mv i.431; to bahis, bāhya-) Ger. vähetvä Ud xi.12 (see Dict.)

Ppp. bāhita- LV 353.14 etc.

bukka-li, bark (only Dhātup., and not in MIndic) . Pres. bukkati Karmav 22.7; 26.1

budh-, become aware, be enlightened: (1) budhya-le, -li (Skt.)

Pres. paribudhyāmi Mv ii.66.15; 444.16, 21; opt. budhyeya Mv ii.160.5; vibudhyiya (1 sg.) Bhad 41 (but prob. read with v.l. vibuddhiya); impv. budhyāhi Mv ii.158.12

- budhyisyate SP 343.14 (2 mss. buddhisyate, sic! Fut. with s)
- samanubudhye Mv i.4.10; budhyişu (read bhadra Aor. bu°) Gv 34.21: budhyasi (3 sg., § 32.62) Mv ii.336.15, budhyansu 19 (§ 32.76)
- budhyitva Gv 315.20; abhisambudhyitvā or °tva Ger. Mv ii.419.1; iii.246.3 (in both v.l. °ddhi°)
- budhyitum LV 270.21; RP 19.9, 15; budhyanāya Inf. LV 271.4
- (Caus. budhyāpayati: see budhyāpaka, °pana, Dict.) (2) buddha-li, denom. to buddha- (§ 28.19)
- Pres. opt. buddhe Mv il.377.5; 378.20; vibuddhi Mv i.155.19; buddhema LV 361.10; sambuddheyuh Suv 107.12 (Nobel em. °budhe°); abhisambuddheyam Sādh 225.8; impv. pratibuddhatha Ud xv.7, so oldest ms., changed later to "budhyata (ms. "tah)
- vibuddhişyati Mv ii.160.8; iii.39.10, 11, 12; buddhi-Fut şyanti Mv ii.173.13; abhisambuddhişyati Mv ii.313.11; °si Mv ii.400.14; °şyāmi Mv ii.265.17; 314.16; abhisambuddhesya (1 sg.) Mv ii.404.19 (\$ 31.33)
- buddhişū LV 220.12; abhisambuddhe Mv ii.285.3; Aor. anubuddhi LV 299.12 (§ 32.23)
- buddhitvā (§ 35.28) LV 163.19; 164.12; Mv iii.273.1, Ger. etc.; °tva LV 231.19; 355.15; Mv ii.323.19 etc.
- buddhita- Mv i.209.23; so mss., Senart em. budhita-Ppp. § 2.89, which meter requires; same line ii.13.20 text with mss. bodhita- despite meter; read budhitaboth times, prob. m.c. for buddhita-, but cf. (4) below
 - (3) bodha-li (Skt.)
- abhisambodhitvā Mv i.312.18; ii.348.15 (v.l. here Ger. °buddhitvā)
 - (4) -budhati? § 28.30
- Pres. abhisambudhati, v.l. for °budhyati Mvy 6907; pple. abhisambudhāna- Ud xviii.18 (§ 34.4)
- abhisambudhisyam Mv ii.265.6 (so text, no v.l.; Fut. prose; § 31.32)
 - (5) Caus. bodhaya-li (Skt.)
- Pres. impv. bodhehi LV 177.22
- Aor. pratibodhayişu LV 173.9
- Pass. ger. bodhyitvå, having been made enlightened, Mv i.170.9 (v.l. buddhyitvā) (6) Miscellaneous
- Pres. pple. buddhyāyamāna- Divy 574.8 (Dict.; prob. denom.)
- Gdve. sambodheya- Mv iii.74.1 (§ 34.24)

brmhaya-li, promote (Skt.) Pres. opt. brmhaye (vr°) Ud xviii.5

(ā-)brh-, vrh-, tear, pluck: °hali (Skt.). See also (đ-)vrh-

Ppp. åbrdha-, åbridha-, åbūdha- (Dict.)

brū-, speak: bravi-ti (Skt.)

- Pres. brūmi, § 28.64, to be read in Mv i.280.14 etc. (so Senart sometimes, as ii.50.2); see Dict. s.v. [kruhi]; impv. bravihi (§ 28.60) SP 93.11 (Kashgar rec.) Preterite abravi (v.l. °vit; § 32.21) Mv 11.61.10; abravitsu
- (§ 32.37) Mv i.186.11; abravīsi (3 sg.; § 32.71) Mv ii.54.4

bhakşaya-li or bhakşa-li, eat (both Skt.) Caus. opt. bhaksapayet Mmk 82.21

1 bhaj-: bhaja-li, share, revere (Skt.)

Pres. opt. bhaje Mv ii.66.21; sambhajeya Mv ii.67.2; impv. vibhajahi LV 165.8

- Fut. bhajahe (? 3 sg., § 31.20) My iii.355.14
 - 2 bhaj-, bhañj-, break: (1) bhañja-ti: Pali id. (trans. and intrans.); § 28.4; once ger. of -bhanjaya-ti, non-caus.
- bhañjati LV 175.17 (intrans., is broken; § 37.16); Pres. bhañjante Divy 343.16 (so mss.; trans., they break) Fut. bhañjişyāmi Mv ii.280.4
- prābhaňjit SP 159.1 (prose), § 32.45 Aor.
- bhañjitvā Mv i.30.8; °tva Mv ii.308.18; prabhañja-Ger. yitvä (not caus.) SP 159.1 (prose)
- (2) bhaja-li? Not recorded elsewhere, and doubtful: §§ 2.89; 28.31
- Pres. prabhajanti Mv i.30.7 (prose); so ed. without v.l. But in next line bhamjitvā. Misprint, or error of mss., for °bhamj°?
- bhajişyati Mv iii.23.19, passive in mg., will be Fut. broken. So Senart with one ms.; v.l. bhamj°, which is metrically bad; might be m.c. for bhamjo or bhajj° (bhajy°, see next) (3) bhajja-li, passive, MIndic for bhajyate
- Pres. bhajjati Mv iii.65.9 (v.l. bhajyati)
- Fut. bhajjişyati Mv ii.489.20 (v.l. bhamj°)

bhana-li, speak (Skt.), also bhanc-li, non-caus., § 38.21

- Pres. bhanesi My ii.222.19; bhanāti My iii.268.8 (prose; § 27.4); opt. bhane Mv ii.176.6; bhaneya SP 58.6; bhanesi Mv ii.224.7; impv. bhanāhi LV 56.12; bhanahi LV 91.18 (ed. bhana hi)
- bhani LV 166.17; bhane Mv ii.238.14; prabhanişu Аог. LV 164.20; bhanāsi Mv ii.221.7 (§ 32.59); samprabhanişu LV 167.10 (Dict.; perh. m.c. for °bhān°, caus.)
- (Ppp. pratibhānita- Mv ii.37.17; if correct, prob. ā m.c.; but v.l. ^obhan^o)

bhandaya-li, quarrel (Skt. Gr.): Pali bhandati; with vi-, distort (the face)

- pratibhandayati KP 107.3; Bbh 170.20; bhandayate Pres. KP 107.25; vibhandayati Divy 263.14, °yasi 15
- bhandita- Mvy 8711; KP 107.3, 5; Bbh 170.20; Ppp. Bhik 31a.4
- Gdve. pratibhaņditavya- Mvy 8711; Bhīk 31a.4

bharts-, abuse: bhaccha(ya)-li, bhatsa(ya)-li (= Skt. bharts°; §§ 2.17, 18; respectively MIndic, and imperfect Sktization): Pali (nib)bhaccheti (aor. °esi Jāt. ii.338.5); see also Dict. s.v. nirbhatsanā (noun) Pres. pple. avabhatsayanto SP 114.6 (so all Nep. mss.;

ed. em. "bharts"; Kashgar rec. wholly different) Ppp. nirbhacchita- RP 32.14; nirbhätsita- Mvy 7183 (?

- see Dict.)
- nirbhatsya LV 319.10 (prose; both edd., no v.l.); Ger. bhatsiya (most mss.) or bhartsiya (one ms. and WT) SP 114.9 (KN em. bhartsayi)

1 bhā-, shine: (1) bhāya-li (§ 38.24): Pali (pați-)bhāyissati, fut. (PTSD s.v. pațibhāti), Pkt. (apparently not AMg.) bhàyai, padihàyai Pres. pratibhàyati LV 222.18; Mv iii.9.11 = 10.2; 38.3;

- °si Mv ii.297.11; °nti Mv ii.297.12; 480.13; impv. °yatu Mv iii.48.9
- Aor. bhāyi (v.l. hāyi) Mv ii.308.16; pratibhāye Mv ii.122.11; °yetsuh Mv ii.121.1

(2) Miscellaneous

prabhāsi Mv ii.345.8 (or from pra-bhās with ending Aor. i; § 32.54; cf. also obhāsi under bhās)

2 bhā-li, fear, see under bhi-

bhāş-, speak: (1) bhāşa-li (Skt.)

Pres. adhyabhāşati Mv i.81.3 (etc.; augmented; § 32.8); opt. bhāşe SP 98.7; bhāşi SP 236.8; bhāşeya (mss.) Mv ii.258.15; bhāșesi (2 sg.) Mv iii.384.16; impv. bhāșato (3 sg., § 30.14) LV 7.11

Fut.

- bhāşahe (1 sg., § 31.20) Mmk 577.16 abhāşi (3 pl.) SP 194.3; bhāşe Mv i.29.4; bhāşişu Aor. LV 352.20; abhāşitha Mv ili.438.11; adhyabhāsita Mv iii.135.17 (§ 32.42); abhāşire Mmk 231.20 (§ 32.43); adhyabhāşasi (3 sg.) Mv i.55.6 (§ 32.62); abhyabhāşeran Mv i.56.13 (§ 32.105)
- (§ 34.11) bhāsta- Lank 283.8; su-bhāsta- LV 230.13; Ppp. ābhāsta-Mv ii.157.16, etc.; pratyanubhāstaiii.393.17
- Ger. bhāşitva (so mss., m.c.) Mv ii.299.16; bhāşiya SP 23.11

bhāşaņāya SP 28.6 Inf.

(2) bhāşaya-li, non-caus. (so rarely in Skt., bhāşayeta

- Mbh. Crit. ed. 5.44.11); note Pali paribhāseti = paribhāsati Jāt. iv.285.8
- ābhāşayimsu (text ābhās°) LV 353.6; adhyabhāşaye Aor. My iii.327.5
- Ger. paribhāşayitvā SP 213.4

Inf. paribhāşayitum Bhik 5b.3

- Gdve. paribhāşayitavya- Bhik 5b.3
- (3) bhāşya-ti, nowhere recorded

Pres. impv. bhāşya Lank 351.12

- bhās-, shine: (1) bhāsa-ti (Skt.) obhāsi Mv ii.335.17 (or to avabhāti; cf. also pra-Aor. bhāsi under bhā); prabhāsisu Mv ii.307.1; bhāsetsu Mv ii.306.14
- obhāsitvā My iii.94.11 Ger.
- (2) Caus. bhāsaya-li (Skt.)
- Pres. impv. avabhāsayāhi Gv 54.12
- Aor. avabhāsayī SP 9.4
- Pass. (? or ya-present from simplex?) avabhāsyante LV 352.8 (perhaps, are made to be resplendent ?); pple. avabhāsyantah (n. pl.) LV 351.22

bhid-, split: (1) bhinda-ti: Pali id.; § 28.4. Also (ger.) bhindaya-ti, non-caus., § 38.21

- Pres. bhindāmi My i.335.1; vyatibhindati Ud xxxi.17-22; opt. bhinde Mv ii.330.3; impv. bhindata LV 306.5; pple. bhindanto (n. sg. m.) Mv i.335.1 bhindi (2 sg.) LV 370.22
- Aor.
- Ger. bhinditvä My ii.239.16; bhindayitvanā LV 73.21 (so with v.l., text bhindiy°)
- Inf. bhinditum Samādh p. 17 line 4

(2) Pass. bhidya-te (Skt.), and MIndic bhijja-te (§ 2.14); once -bhidya-ti, active (4th class?)

Pres. opt. bhidyeya Sukh 45.13; nirbhidyet Lank 24.14, active (4th class?) in form and mg.

bhijje Mv ii.412.11 Aor.

(3) Miscellaneous

(Pres. bhinanmi 1 sg. LV 309.10, 20, both edd.; best mss. bhinadmi, bhinatmi. Cf. Whitney 161b; Wackernagel I §§ 164, 176 for similar stray forms in Skt., regarded by Whitney as false readings. Against this Bartholomae ZDMG 50.712 f.)

Ger. bhetvāna Mv ii.12.16 (§ 35.35)

bhi- fear: (1) bhāya-li; Pali id.; analogical to quasiantonym träyate, Pali täyati, rescue; cf. (2) bhä-ti and \$ 28.23

Pres. bhāyati My ii.236.9; °si My i.163.18; Ud ix.3 (later ms. bibheși); °se Mv ii.58.19; impv. bhāya Mv ii.134.3; Šikš 154.17; bhāyāhi Mv i.363.1; iii.408.11 (here v.l. bhāhi); bhāyatha or °thā SP 26.4; 197.1; Mv i.361.15; iii.303.17

Aor. bhāyi Mv ii.308.16

501), or more specifically with quasi-antonym trati (cf. under bhāya-ti)

- Pres. impv. bhāhi LV 232.3, so read for tāhi, allegedly with nearly all mss. (but v.l. bhāhi); Tib. hjigs, fear. Same Mv iii.403.17, and v.l. iii.408.11 (above) (3) Miscellaneous forms
- (injunctive) bhihi Mv ii.489.6, 16, § 30.10; bhaistatha, Aor. °ta Mmk 4.14-15 (text °tu); 182.23 etc. (§ 32.84)

bhīşaya-ti, frighten (Skt., caus. to bhī)

Caus. opt. bhīşāpayet Prāt 518.10

1 bhuj-, enjoy: (1) bhuñja-li (Skt. sporadically, 'U +' Wh. Roots): Pali id.; § 28.4 Pres. bhuñjati Mv ii.1.10; opt. bhuñje (1 sg.) Mv ii.164.14;

- paribhuñjeyā Mv iii.312.2; paribhuñjesi (2 sg.) Mv iii.311.10; impv. bhuñjāhi My ii.55.21
- bhuñjişyati Mv ii.242.11; bhuñjeham (1 sg., § 31.21) Fut. Mv ii.131.9
- paribhuñji LV 387.7; bhuñjesi Mv iii.129.8; upa-Aor. bhuñje Mv li.204.19; paribhuñjetsu(h) Mv i.343.18 Ppp. paribhuñjita- Mv i.19.4
- Ger. bhuñjitvā Mv i.306.19; paribhuñjiya Mv ii.96.18; bhuñjiyāna Mv iii.120.18
- bhuñjitum Mv ii.56.3 Inf.
- Caus. bhuñjāventi (§ 38.69) Mv i.12.6; ger. paribhuñ-jāpetvā (§ 38.53) Mv iii.148.15

(2) bhuja-ti, perhaps by adaptation to homonymous root bhui, bend: Ap. bhujantu (pres. pple. n. sg. m.) Jacobi, Sanatk. 665.7; and Sheth cites from a gram. source (Sadbhāşācandrikā) bhuai, bhuae. The converse

- adaptation leads to bhuñjati, bends, q.v.
 Pres. ? in Mv i.8.5 (prose) Senart bhumjantām (3 pl. impv.), mss. mostly bhūjānta, °te, °to; also bhūmj°; uncertain; in Mv i.27.5 (prose) Senart em. paribhujantăm (acc. pl. m. pple.), most mss. parivrajantăm (seems meaningless; one inferior ms. °bhumjatām); in Mv i.88.7 (vs) Senart bhumjanti, metrically correct, but with one inferior ms.; the rest bhujanti, metrically poor, yet not worse than many other metrical passages as recorded in Mv.
- Ger. bhujitva LV 74.6 (vs; could be understood as m.c. for bhuñjitvā)

(3) Caus. bhojaya-li (Skt.)

Pres. impv. bhojchi Mv i.309.10

Caus., blend form (or true caus. to bhojeti? § 38,61) bhojāpehi Mv i.305.10

(4) Desid. (unredupl.) ppp. bhukșita-, see § 40.2, Dict., also s.v. bhukşa

2 bhuj-, bend: (1) bhuja-li (Skt.)

ābhujitvā (Pali id.) LV 259.5); Mv ii.268.4 (v.l. Ger. ābhumj°); ābhujiya LV 133.20

(2) -bhuñja-li, perhaps by adaptation to homonymous root bhuj-, enjoy; cf. the reverse adaptation bhuja-ti, enjoy. The v.l. abhumjitva for abhujitva is recorded in Pali MN i.56.13

Pres. (?) ābhumjasi Mv ii.403.6, text; but meter is better with v.l. abhujasi (-bhuja-, two shorts = one long)

ābhumjitvā Mv i.144.11 (prose); ii.131.15; 313.14; Ger. and text in i.213.8 (prose, mss. corrupt) = ii.16.12 (here with one ms., v.l. ābhujitvā).

bhū-, become: (1) bhava-li (Skt.); m.c. also bhāvati (§ 3.11), q.v. Dict.

Pres. opt. bhave LV 56.1; bhavi SP 175.7; bhaveyā SP 313.7; °ya SP 53.3; bhaveyyā Mv i.331.8; bhaviya ,LV 78.11; bhavesi (2 sg.) Mv i.233.8; impv. bhavāhi Mv ii.230.6; bhavahi LV 241.20; imperf. bhava LV 166.5 (§ 32.115)

⁽²⁾ bhā-li: M.bhāi, 'by analogy with roots in ā' (Pischel

- bhayisya RP 29.12 (§ 31.34); conditional abhavi-Fut. syati Mv i.267.7 (mss.; § 31.40); abhavişyam (3 sg., § 31.39) Mv iii.107.10 (mss.)
- bhavi RP 27.12 (read tatra bhavi); bhave (§ 32.87) Aor. RP 23.16 (1 sg.; § 32.87; read yadā bhave, or (a)bhave); prādurbhave Mv i.221.6 (v.l. °vet) = ^obhavet ii.23.11; prādurbhaveya Mv i.340.12 (§ 32.105); bhaveyuh (id.) Mmk 130.19, 26; prādurbhavetsuh (or °vensu; so mss.) Mv iii.180.15; prādurbhavinsu Hoernle MR 136 for SP 329.2; bhavesi Mv ii.350.3
- bhavitvā LV 79.19; °tvāna Mv iii.2.2; bhaviya Ger. LV 297.9; bhavi Šiks 345.13

(2) bho-li, ho-li: Pali hoti. Cf. § 1.29. In BHS the bhforms are much commoner than the h- forms; yet bh- in such contracted forms (with o for ava or avi) seems not recorded in Pali; but it occurs in western Asokan and in later Prakrits (Pischel 475)

- Pres. (bh-forms common) bhoti SP 92.8; Mv i.16.1: Suv 4.10; Samādh 22.22; Bhad 51; bhomi SP 89.11; prabhomi LV 56.1; bhonti SP 90.7; hoti (rare) Mv i.208.2 = ii.12.11; impv. bhohi LV 237.9
- Fut. bhoşyanti SP 330.9; bhoşyi (1 sg. mid.) SP 291.13 (these forms are questionable; § 31.14); bhohi (2 sg.) LV 196.4; hohī (2 sg.; read hohī ti with some mss.) Mv i.44.3; hohisi Mv i.337.9; hohiti (sometimes v.l. hohiti) Mv i.208.5 = ii.12.14; ii.13.3; 40.20 (twice); as 3 pl. (mss.) Mv ii.206.13; hohati Mv ii.225.5; 405.12 (ms.; 3, not 2 sg.); hohanti Mv ii.224.21. On all these see §§ 31.16, 17. bhosisu (? 3 pl.; § 32.78) Sukh 7.13 (vs) Aor.
- (3) stem bhe- (for bhavi-) Fut. bhesyati SP 68.1; LV 48.13; Mv i.234.5, etc., common. The regular form (beside bhavişyati); = Pali hessati; apparently not in Pkt.

(4) stem bhū-, hū-. The aor. abhū(t) was inherited from Skt.; the other forms, notably the standard aor. abhūşi, are doubtless based on it by proportional analogy, on such models as $ad\bar{a}(t)$: adāsi. The fut. -bhūşyam could be similarly explained

Fut. pratibhūşyam (1 sg.; § 31.15) Mv ii.142.17

- Root Aor. (Skt. abhūt, Pali ahū, ahu) and related forms: abhū SP 157.10; abhu SP 204.9; ahū Gv 254.23 (§ 32.107); abhūva (id.; 1 sg.) RP 25.15; abhūvatām (3 dual) LV 83.13 (§ 32.109, q.v. for other, doubtful forms); abhūnsu(h), abhūtsu(h), abhunsu(h), $^{\circ}msu(h)$, $^{\circ}tsu(h)$, 3 pl., Mv i.41.7 etc. (§ 32.108)
- s-Aor. abhūsi (the regular form, very common; § 32.72) SP 26.13; LV 27.19; Mv i.3.13 etc.; abhūsi SP 383.11; Gv 254.10; abhūsīt Hoernle MR 133-5 (§ 32.72); abhūsita Mv i.34.16 (§ 32.73)

(5) (sam-, abhisam-)-bhuna-li, attain; also -bhuno-li: Pali id.; § 28.17; acc. to CPD 'formed after analogy of pû : punāti'; but since it is 'nearly synonymous with pāpunāti', as CPD also says, it seems clear that this, thematized as p(r)apunati, is rather the source. Note domal ņ (also in Pali pāpuņāti); note also -bhuņoti in one passage, which confirms my theory (Skt. prāpnoti); § 28.17

- Pres. sambhunanti Mv ii.107.14; abhisambh° Mv iii.334.9; opt. abhisambhunema Mv iii.374.7, °netha 12 abhisambhunoti Mv iii.206.7-10 (four times)
- sambhunişyasi Mv iii.264.14; abhisambhunişyāmah Fut. Mv iii.265.3
- abhisambhunitvā Mv iii.265.3 (so mss.; Senart em. Ger. °nitam)

(6) Caus. bhāvaya-ti (Skt.); also as non-caus.

- Pres. opt. bhāvaye Mv iii.373.11; bhāvayi Samādh 8.30
- bhāvaye Mv i.192.4 (developed, = expounded; AOT. Senart misunderstands); vibhāvayetsu(h) Mv i.340.17
- Ger. bhāvitva SP 92.8; bhāvayitvāna Mv ii.239.15;

bhāviya (mss.) Mv i.357.18; in non-caus. mg. abhisambhāvayitvā (§ 38.23), having attained (cf. abhisam-bhunati, just above) Mv i.4.12

- (7) Miscelianeous forms
- Pres. bhaviti LV 325.5 (3 sg.); modelled on braviti? or corrupt for bhavāti, m.c. for bhavati? Calc. glosses bhavati; Tib. implies this by its (mi gan hdod pas bud med rnams kyi khol por) gyur pa (dan), 'what man because of desire becomes the slave of women', indicating a text kāmā (abl.) dāsu bhavīti (°vāti? two ms. °vati, unmetr.) yo narah (so with v.l., metr. required) pramadānām; prādurbhāmi, I appear, Gv 70.9 (prose); follows antardhāmi, I disappear, its antonym, and may be modelled on it; cf. Pali aor. pätur-ahamsu, appeared, also modelled on forms of antar(a)-dhā; but 2d ed. °bhavāmi (by em.?)
- hī (= ahi = abhu?) LV 230.9 (§ 32.28); prādur-ahi Aor. Mv ii.221.17; iii.216.7 (§ 32.28)

bhr-: (1) bhara-ti, carry, hire (Skt.)

Ger. bharitvāna, carrying, Mv i.216.6

Caus. bharayitvā (§ 38.9), having caused to be hired, SP 105.11

(2) bharaya-li (also bhara-li?), fill: AMg. bharei, bharai; back-formation from Skt. bharita-, loaded, filled, which

- is a denom. pple. to bhara-, load; § 38.37 Aor. bharayetsuh Mv iii.427.16 (prose; so one ms.; v.l. bhavetsuh; Senart em. bharensuh)
- bharitvā Mv i.231.5 (Senart em. °tva, m.c.); ii.295.9 Ger. bharayitum Mv iii.427.14 (prose; mss. hara°, cf. bha-Inf. rayetsuh, above, two lines later)

bhram-, wander: (1) bhrama-li (Skt.)

bhrametsuh Mv i.222.7 Aor.

- (2) Caus. bhrāmaya-ti (Skt.)
- Fut. udbhrāmişyase LV 334.17
- bhrāmayimsū LV 81.7; bhrāmayetsuh Mv ii.415.1; Аог. bhrāmetsu Mv ii.342.22

matha-ti, shake (Skt.)

Ger. mathiya Mv i.72.16

mad-: mādya-li, be exhilarated (Skt.); also madya-li(?) Pres. opt. pramadyeyā Mv il.147.13. Blend of -madati

and -mādyati? Or false Skt. for MIndic (Pali) pamajjati? Or error (misprint)?

pramādyi (2 sg.) Mv iii.124.18 Aor.

man-, think: (1) manya-le, -li (Skt.); also -manyaya-li, non-caus., § 38.21

- abhimanyayāmah LV 49.11; opt. manye LV 208.22; Pres. impv. manyāhi (v.l. manyahi) Mv il.487.5 abhimanyişyati Mv il.440.17
- Fut.
- manyetsu (v.l. °suh) Mv i.302.4 Aor.
- manyita- SP 63.3 Ppp.
- Ger. avamanyitvā Mv i.309.11
- Gdve. avamanyitavya- Śiks 108.2
- (2) mana-li, manaya-le, mane-li: AMg. M. manai; §§ 28.17; 38.21
- Pres. manesi Mv ii.249.3 (prose; v.l. manasi); manayase Bbh 226.9 (prose, which in this work rarely shows non-Skt. morphology; misprint for manyase? not noted in Glossary, but not corrected in Corr.)
- manişyati, so Senart Mv i.239.15; 331.13; 333.15 Fut. (but see § 31.1)

(3) Caus. mānaya-li (Skt.)

pratimāne Mv iii 248.12; mānesi or mānayesi Mv Aor. i.60.1

(4) Desid. mimāmsaya-li, non-caus. = Skt. mimāmsati; cf. Pali vimamseti beside °sati

Pres. mimāmsaye (see § 38.21) Av 1.94.3; opt. mimāmsayeyam (so with mss.) Av i.189.4

mantraya-ti, announce (denom.; Skt.)

- nimantrayi Mv i.113.4; āmantrayāsi Mv iii.204.2 Aor. (§ 32.59); °yasi (§ 32.62) Mv i.330.11; āmantresi Mv i.51.8; mantrayetsuh Mv i.344.1
- āmantrayitvāna My ii.60.8 Ger.
- Caus. mantrāpayati Mmk 307.5

mamāya-le, -li, cherish: Pali °ti (Skt. once °te in Mbh., see Dict.); denom. from mama

- Pres. mamāyate Šiks 269.6; opt. mamāyeyur AsP 254.2; pple. mamāyamāna- Bimbisārasūtra (Waldschmidt Kl. Skt. Texte 4) 125.10
- mamāyitum LV 100.9 Inf.
- Ppp. mamāyita- Dbh 39.23; Ud xxxii.18(17)

marga-li, see mārga-li

mā-, measure: (1) (nir-, abhinir-)mimī-le (Skt.), create by magic

- Pres. (formally normal Skt.) nirmimite SP 245.11; LV 290.20; abhinir^o SP 189.3; LV 70.20; 293.2; Divy 166.6; opt. abhinirmimīyāt SP 188.1 (all these prose)
- abhinirmāya (normal Skt.) LV 77.20 etc.; abhinir-Ger. mimīya Bbh 152.2 (prose), § 35.14
- Pass. abhinirmimiyantam (acc. sg. pres. pple.) Gv 444.15 (prose), being magically created, §§ 37.15, 37
- (2) (nir-, abhinir-)mino-li (Skt.), = prec. Not recorded in Pall; occurs here only in prose, cf. next Pres. abhinirminoti Divy 251.19; °miuvanti LV 350.19;
- opt. nirmiņuyām Divy 50.3

(3) (abhinir-)mināli, = prec.: Pali id.; not recorded in Skt. Here represented only by thematized forms (see next), except for the isolated ger .:

abhinirminiya Mv ii.411.10 (§§ 28.16; 35.15) Ger.

(4) (nir-, abhinir-)mina-li, = prec.: Pali id.; § 28.17

- Pres. nirminati My i.179.7 (mss. °miśati, em. Senart); opt. abhinirmineyam SP 196.7, 9; in mg. exchange, nirmineya, and another (uncertain) form My ii.176.12, 14
- Fut. abhinirmiņișyanti Mv iii.288.2-12 (five times); °şyāmah Mv iii.324.4
- nirmiņi LV 219.19; °ne Mv i.189.6; abhinirmiņi Mv i.185.4; °ne Mv i.183.10; in Sikş 346.11, 16 text Aor. nirmaņi, intending nirmiņi
- (Ppp. nirmita- and abhinirmita-, regular Skt., common)
- Ger. nirmiņitvā Mv iii.282.15; abhinirmiņitvā SP 63.6; Mv ii.49.11; nirmiņiya Mv i.366.10 (5) *-me-ti; § 28.48
- Inf. upametum SP 304.10 (§ 36.8)
- (6) māya-li: Pkt. māai, beside māi (Skt. māti); § 28.24
- Pres. sammäyati Mv ii.173.12 (prose; so read with 1 ms.); opt. mäyet AsP 157.19
 - (7) Caus. māpaya-ti (Skt.)
- Ger. māpayitvāna Mv ili.45,2

mārga-ti, mārgaya-ti, hunt, seek (both Skt.); also marga-li

- Pres. margati, recorded several times in KP, e. g. 125.7; semi-MIndic for mārgati? (§ 3.34) or corruption?;
- mārgadhva Divy 374.12 (§ 30.18) Pass. mārgiyanti Mv ii.167.12; mārgiyamāna- Mv ii.460.4

-miñjaya-ti, draw, jerk (or the like): Pali (sam)miñjeti (and °ati); see Dict. s.v. samminjavati

Pres. opt. sammiñjayet Mv i.55.1; Mmk 3.26; °jaye Mv iii.425.16

Ppp. samminjita- Mv i.55.1 etc.; also un-, ni-minjita-(Dict.)

mille-ti, leave, abandon, reject: AMg. millai. Also melleti, q.v.

Pres. impv. millehi Mv i.363.14; iii.291.17 (here Senart em. mellehi)

milaya-li, close (Skt.)

Aor. sammilayesi Mv i.247.6

muc-, release: (1) muñca-li (Skt.), also in passive mg., § 37.16

- Pres. opt. muñceya Mv ii.331.1; impv. muñcāhi Mv ii.459.15; muñcahi LV 178.10
- muñci LV 92.20; pramuñce Mv ii.340.10; muñcisu Aor. LV 193.17; pramuñcitsuh Mv ii.4.9; muñcetsu Mv i.266.16
- Ppp. muñcita- Suv 49.1
- muñcitvá My ii.82.19; omuñciya My ii.72.20: Ger.
- omuñciyāna Mv ii.73.1; pramuñci Šiks 327.14 Caus. muñcāpayati Mmk 640.18; impv. muñcāpetha Mv iii.297.12
- (2) muñcaya-li, non-caus., § 38.21. Not recorded in Pali or Pkt.
- Pres. pple. muñcayantah Megh 308.13
- Fut. muñcayişya (1 sg.) Sukh 8.16
- (3) muca-li: AMg. mucai, muyai, cf. Pischel 485: § 28.31
- Pres. mucanti SP 85.8 (could be m.c. for muñc°) (4) Pass. mucya-te (Skt.)
- Pres. opt. mucyeyā Mv i.364.2
- Aor. mucyişu Suv 39.4 (§ 32.122) Inf. mucyitum Mv ii.223.13; adhimucyitum KP 139.9 (5) Caus. mocaya-li (Skt.)
- Pres. moce (1 sg. mid.) LV 219.4; opt. pramoceyam Mv i.42.17; moceya Suv 37.8; vimocayi (1 sg.) Suv 51.15; impv. mocehi SP 174.4 Fut. mocisya RP 15.8 (§ 31.34)
- mocaye Mv ii.187.5; parimocesi Mv iii.446.2 Aor.
- (6) Miscellanous forms Perf. (§§ 33.2, 4) pramumocatur (3 dual) SP 460.2; pra-
- mumocu (3 pl.) SP 190.4; Samādh 8.23

mundaya-ti, shave (denom.; Skt.)

Caus. ger. mundāpayitvā Divy 261.15

mud-, rejoice: (1) moda-li (Skt.) Pres. opt. anumodi SP 58.6; impv. anumodāhi Mv ii.98.6 pratimoditvā Mv ii.248.10; (anumodya Kv 29.15;) Ger.

anumodi SP 351.12 (2) 'Caus.' modaya-ti (Skt.), here (and even in Skt.)

hard to distinguish in mg. from the simplex (but pratisammodayati seems oftener to have -aya-, Dict.)

- Pres. pratisammodate Gv 53.14; °dante Bhik 26b.4, 5
- anumodaye Mv iii.426.6 Aor.

pratisammoditvā Mv i.273.10 (Senart em. °etvā); Ger. ii.443.17; iii.450.20

Pass. impv. pramodyahi Mv i.275.9

muşa-ti (Epic Skt.): Pali musati, which means steals, with 'eyes' or 'sight' as the object; correct PTSD; so also Skt. mus-, BR 5.836, last line; and AMg. musanta-, pres. pple.; § 28.37. Cf. also mūsati

Fut. musisyāmah Divy 94.5, 10, 27 (prose)

musitum Divy 94.29 Inf.

muhya-li, go astray (Skt.)

Pres. opt. vimuhye Bhad 19

mürcha-ti, become stiff (Skt.), also mürchaya-ti, both caus. (as in Skt.) and non-caus.

murchitvā My iii.167.6; sammurchayitvā (non-Ger. caus.) LV 234.2

Pass, of caus. mürchiyāmi Mv ii.428.1, I am made stiff

mūsa-ti, steal = musati, q.v.: Pkt. mūsai (Sheth); § 28.36

Fut. mūşişyate Divy 281.8 (prose)

mūşitu-kāma- Divy 276.24 (prose) Inf.

mr-, die: (1) mara-ti = Pali id.; based on fut. marisyati, § 28.13 (not connected with alleged Vedic marati); also mare-ti, non-caus.

Pres. marati Mv ii.242.4; marāmi Mmk 704.15; maranti My i.253.6; 359.23; 360.3; 361.5; marenti (§ 38.21) Gv 213.20; opt. mareya Mv iii.158.10

maritu-kāma- Mv ii.83.2 Inf.

Gdve. maritavya- Mv iii.354.7

(2) mryya-li, -le, semi-MIndic for Skt. mriyate (§ 37.27) Present °ti, °te Mv ii.79.4 ff.

(3) Caus. māraya-ti (Skt.)

Pres. mārase Mv i.179.18 (§ 38.28) Ger. māritvā Mv ii.248.5

Caus. mārāpemi Mv ii.247.3 (§ 38.57)

mṛj-: mārja-li, wipe (Skt.) Aor. anuparimārje Mv ii.282.6-7 Gdve. sammrästavya- (Skt. °märst°) Av ii.154.3 f. (§ 34.21)

mrd-: marda-ti, crush (Skt., chiefly epic)

Pres. impv. avamardāhl Mv iii.25.11

Fut. mardişyāmi Mv ii.270.4

avamardatum Divy 622.17 (§ 36.7) Inf.

mŗśa-ti, stroke (Skt.) Pres. opt. parāmrse (1 sg.) Mv iii.8.7, 9 Fut. āmŗkşyanti Šikş 45.12 (§ 31.25) Aor. vimṛṣuh (for °śuh) LV 385.19 (§ 32.118)

mrs-: marsaya-li, pardon, let pass (Skt.) Pres. impv. marsehi Mv iii.192.1

melle-ti, abandon, reject (= milleti): Pkt. mellei, mellai

Pres. impv. mellehi Mv ii.448.4; 454.16

Ger. mellitvā Mv ii.463.17; melletvā Mv ii.463.15

maitrāya-te, -ti (denom.), show love (from maitrā, Dict.)

Pres. maitrāyate Bbh 369.15; pple. °yatā (instr.) Divy 105.17; 123.7; °yamāņa- AsP 395.12

mokşaya-ti, free (Skt., denom.) Pres. impv. moksehi Mv ii.179.4 Aor. moksaye Mv iii.13.12 vimokşayi Mv i.72.14 Ger.

mlāya-ti, fade (Skt.); also MIndic milā°: Pali milāyati Pres. opt. sammiläyetsuh Mv ii.178.8 Fut. pass., in caus. mg. (§ 38.24), āmlāyişyase LV 335.5 Ppp. milāyita- Mv ii.232.3; āmilāta-, sammilāta- Mv ii.126.4, 5 (§ 34.11; Dict.)

mlāyitum Divy 570.25 Inf.

yaja-li, sacrifice (Skt.) Pres. impv. yajāhi Mv ii.237.19 Fut. yajişyati Mv ii.98.7 yajita- Kv 29.21; yaşta- LV 170.12 etc. (the com-Ppp. mon form: § 34.13)

Ger. yajitvā Mv ii.237.20; °tvāna Mv ii.405.12

yat-, stretch: Caus. (nir-)yātaya-ti, -te, once perhaps -yād° (as in Pali): Pali niyyādeti

Pres. niryātayi (1 sg. mid.) SP 115.9

- niryātayī Suv 152.10; niryātesi Mv i.49.16; niryā-Аог. dayinsus, Kashgar rec. for SP 191.2
- niryātetvā Mv i.63.2; °itvā Mv i.117.8 Ger.

yam-, hold: (1) yaccha-ti (Skt.)

Pres. impv. anuprayacchāhi Mv ii.146.11

Aor. prayacche (mss. pranacche) My ii.268.11

Ppp. vyáyacchita- Bbh 92.16

anuprayacchitum Mv ii.147.2 Inf.

(2) yama-li; cited as 'E +' in Whitney, Roots, but very rare and irregular in Skt.; regular in Pali in all cpds. (Geiger 133); also AMg. samjamai, beside Pkt. jacchai (Sheth). Once vyāyāma-, see below

Pres. pple. vyāyamentena Mv i.246.4 (so Senart by em.); vyäyämantena Mv ii.30.11 (same mg., exerting himself; not caus.; § 38.30; cf. Skt. vyäyämya Manu 7.216, Jolly, without v.l., but Bombay ed. of 1886 vyāyamya, with several comms. Is the Mv reading also a corruption, or influenced by the noun vyayāma?)

vyāyamitvā SP 109.5 Ger.

yā-, go: (1) yā-ti (Skt.)

Aor. prayāsi (§ 32.53) Mv i.255.3; niryānsuh (°msuh, 3 pl.) Mv iii.161.14 (§ 32.75); samayā (mss., for °yat) Mv ii.315.10 (§ 32.106)

(2) yāya-ti: Pali id. (Geiger 138); § 28.24 Pres. opt. niryāyeyuh SP 78.12 (prose, no v.l.); pple. prayāyatah (gen. sg.) Mv iii.120.2; yāyena (mss., instr., = yāyatā; § 18.61) Mv i.232.6; prodyayāmāna, m.c. for prodyāyamānah, §§ 3.31; 27.1, setting out, Dbh.g. 51(77).25

Ppp. yāyita- Mv i.299.8: read yāyitā, they have gone to ..., with one ms. (v.l. yācitā; Senart wrongly em. jātā)

yāca-li, beg (Skt.), also yāce-li, non-caus., § 38.21

- Pres. indic. yācesi Mv ii.185.9; opt. yāci (mss., 2 sg.) Mv iii.419.5; yācesi (2 sg.) Mv ii.406.1; impv. yācāhi Mv i.289.6; yācehi Mv i.287.11; pple. yācintyāḥ (i for e, § 3.49; gen. sg. fem.) Mv i.133.1 Aor. yāci Mv i.132.11; yāce Mv ii.399.1; yācesi Mv
- . iii.97.15
- Pass. yācīyanti Mv iii.315.18

yuj-, join: (1) yuñja-li, -le ('U.E.+', Whitney, Roots); once -yuñjya-le (?)

- Pres. opt. prayunji (see note) Šiks 343.2. In Divy 512.12 samanuyuñjyāmahe may be a textual error (or else a phonetic corruption) for °yuñjāmahe, or otherwise a 4th class pres. stem in -ya-, cf. below (2); § 28.26
- an-anuyujitvā Mv iii.160.6, prob. misprint, or Ger. corruption, for vunjitvā, see § 35.28(2) Pass. yujya-te (Skt.); also with active mg., or
 - 4th class pres., § 28.28
- Pres. opt. samyujyeyā (by em.) Mv i.295.18; with active mg., or 4th class pres. (cf. samanuyuñjyāmahe above) samanuyujyamāna- Prāt 482.4, 6; impv. yujyadhva RP 18.14 (§ 30.18); abhiyujyadhve SP 79.11 (§ 30.19)
- . samyujyişu Suv 39.5 (injunctive), § 32.122 (3) Caus. *yojaya-ti* (Skt.) Aor.

Pres. opt. prayojeyyā Mv i.279.16

- Aor. prayojayi (1 sg.) SP 90.5; udyojayi Mv i.322.7
- udyojayitvāna Mv ii.320.3 Ger.
- Caus. yojāpayanti Mv iii.101.20; impv. °pehi Mv iii.441.20; Aor. °pesi Mv iii.442.2; Ger. °payitvā Mv i.259.8

? yul-: udyotita-, ppp., see Dict.

-yūha-li (with ā-, nir-, once m-), exert oneself (or the like), MIndic: Pali ayuhati, samyuhati, AMg. nijjuh-. Perhaps historically related to Skt. vyūh-, cf. āvyūhati nirvyūhati MadhK 298.13; 517.20 (but these are doubtless secondary attempts at Sanskritization)

- Pres. āyuhāmi inryūhāmi Gv 83.12; pple. instr. āyūhatā niryūhatā Gv 69.24; gen. Gv 80.25; loc. fem. Gv 199.24; āyūhantī viyūhantī Gv 222.15
- Pass. äyühyamänam näyühyate, niryühyamänam na niryūhyate Lank 115.13 f.

rakşa-li and rakşaya-li, protect (both Skt. but the latter very rare there)

Pres. opt. abhirakşaye Mv iji.388.6; parirakşeyā (2 sg.) Mv i.277.8; rakşesi (2 sg.) Mv iii.387.7 Fut. pple. parirakşisanto LV 47.10 (? § 31.28)

Pass. rakşiyati Mv iii.298.11

raj-, be charmed: (1) rajya-li (Skt.) Pres. opt. rajyeyā Mv ii.147.13

(2) Caus. rañjaya-li (Skt.)

Aor. samrañjetsu(h) Mv i.342.11

raņa-li, sound (Skt.)

Pres. opt. anurane Mv ii.282.12

Aor. rani LV 236.10; rane Mv ii.342.7; ranişū LV 222.7; ranimsū LV 233.18

rabha-ti, -te, take hold (Skt.)

- ārabhişya (1 sg., so read, § 31.33) Sukh 8.17 (not Fut. ger. ārabhidhya, § 35.56)
- samārabhe My ii.308.1 Aor.
- ārabhita- SP 408.15 Ppp.
- Inf. prārabhantum (! § 36.13) Mmk 74.16

- ram-, rejoice: (1) rama-li (Skt.) Pres. ramimo LV 321.20 (§ 27.10); opt. rameyyā (v.l. °eyā) Mv ii.144.9; impv. ramāhi Mv ii.103.6; ramasu (? § 30.16) LV 328.9
- ramişyati Mv ii.443.10 Fut.
- rametsu Mv i.194.17; °tsuh Mv ii.301.13 ramitvā Mv ii.111.16 Aor.
- Ger.
- Gdve. ramitavya- Mv ii.107.17
- Caus. ramāpayate Mmk 571.20; impv. ramāpehi Mv ii.430.4

(2) Caus. rāmaya-li (Skt.)

Pres. impv. rāmehi Mv ii.430.4 (in same line with ramāpehi, above; they are obviously used as synonyms)

rahāya-ti, be hidden (denom.; Skt. Gr. °te): Pali id. Pres. rahāyati Šikş 4.20 (vs; i could be m.c. for e)

rādhya-li, succeed (Skt.)

- aparādhyetsuh Mv ii.137.9; so mss., and prob. a Aor. form of this root is intended at ii.139.4, see Dict., and under root 1 vyadh (1)
- aparādhyitum Mv iii.126.18; 127.2 Inf.

riñca-li, leave: Pali id.; § 28.4

Pres. riñcati Mvy 2552; Śiks 53.9; impv. riñcata LV 18.12, rincatha (in version of the same incident) Mv i.197.1, 4; also i.357.4, 6

Fut. riñcişyasi Bbh 396.20

- Ppp.
- rincita- SP 62.2; Divy 638.10; Mmk 73.9 rincitvā KP 90.3; ASP 237.7; read so, probably, Ger. for ricitva RP 34.12 (prose), despite Whitney Roots 'aricat C'

Gdve. riñcitavya- AsP 243.6

riliya-le, see rli°

ridhya-le, see rdh°

- ru-: rava-li, cry (rare in Skt.): Pali id. Pres. oravanti Mv ii.100.17; impv. ravähi Mv ii.462.18 Fut. viravişyati Divy 276.2
- anuravi LV 164.22 (3 pl.; so read for text tatu Aor. ravi); ravisu (or ravi su- with v.l.; ed. em. °sû) LV 167.12
- Ppp. ravita- LV 286.14; Sutrāl. comm. on xil.9

ruc-, shine: (1) roca-ti, -le (Skt.); here also in caus. mg., see (3) rocayati

Pres. opt. viroce Mv i.70.7

- Aor. virocişū LV 122.22
- (2) rucya-li, be pleasing, also rucca-(li) with MIndic phonology: Pali ruccati; an analogical passive or 4th class pres. to Skt. ppp. rucita-, or to rocate which (with caus. pass. rocyate, rare) is used in the same mg. in Skt. Pres. rucyati Mv ii.464.10; rucyanti SP 98.11; ruccati Mv ii.485.6
- ruccita- Mv ii.427.16 Ppp.
- (3) Caus. -rocaya-li (Skt.), here also roca-li in caus.
- mg. (§ 38.28); conversely (vi-)roce-li as non-caus. Pres. virocenti (non-caus.) Mv 1.78.12; rocanti (caus.) SP 306.4; abhirocante (caus.) Mv i.78.14; impv. ārocehi Mv i.287.10
- arocayasi (3 sg., § 32.62) Mv i.27.11-12; ārocesi (3 pl.) Mv i.257.15; ārocatesi (3 sg., § 32.68) Mv il.169.9; ārocaye Mv ii.198.16 (v.l. °cate); ārocetsuh Aor. Mv i.258.2
- Caus. ppp. ārocāpita- Mv i.307.13

rud-, weep: (1) roda-ti (Skt.), also rodaya-ti, noncaus., § 38.21

- Pres. rodayāmi Mv iii.351.9 (non-caus.; prose; in response to question: kim rodasi?); impv. rodāhi Mv ii.218.13; rodahi (most mss. rodihi) LV 237.9
- prarodi Mv ii.38.16; prarodi Mv ii.32.11; rodetsuh Aor. Mv ii.225.12
- Ger. roditvā (once in Epic Skt., Mbh. Calc. 13.5410) Mv ii.189.15; 215.5
- (2) ruda-ti (Skt.) Pres. impv. rudahi LV 231.10
- (3) Miscellaneous forms
- Perf. (intensive) roruroda Mmk 381.2 (§ 33.6)
- Ppp. ruņņa- (= Pali and Pkt. id.; § 34.17) Mvy 6663; LV 195.2 (text rūņņa)

rudh-, hold back: (1) rundha-ti (Epic Skt.): Pali id.; § 28.4

- Aor. uparundhi (1 sg.) Mv ii.124.10
- Ger. / orundhitvā My ii.75.10
- (2) -ruddha-ti; § 28.19
- Aor. uparuddhe (mss. u-ru°; 1 sg.) Mv ii.125.1 (Senart em. uparundhe)
- viruddhişyati, will obstruct, My ii.490.15 Fut. (3) Caus. rodhaya-ti (Skt.)
- Aor. nirodhayi LV 236.5

(ava-)rup-, plant: only in ppp., planted, back-formation from avaropayati (Skt. and BHS), plants, caus. of (ava-)ruh, q.v.; see Dict. and § 34.11

Ppp. avarupta- Suv 91.8; Gv 278.22; Mv ii.314.11; orupta- Mv iii.104.18 etc.

ruș-, be angry: (1) rușa-li (Skt.) Ger. rușitvă RP 23.3

- (2) Caus. roşaya-li (Skt.)

Fut. (pass.) roșișyase LV 334.19

ruh- mount, grow, etc. (see also ārūdhayali): (1) roha-li (Skt.)

Aor. rohişu LV 75.20

- abhirohanāya Divy 113.20 lnf.
- Caus. rohāpayati KP 30.1
- (2) ruha-li (Skt.)
- Pres. opt. āruheyā LV 80.10; āruhi Mv i.201.11 (1 sg.; or ger.? so read) = ii.5.6
- āruhişyati Mv ii.90.12 Fut.
- āruhī LV 236.15 Aor.
- āruhitvā SP 128.12; abhiruhitvāna Mv iii.289.1; Ger. āruhiya Mv ii.90.7
- Gdve. abhiruhitavya- Mmk 701.12
- Caus. ger. abhiruhāpayitvā Mv iii.39.6
- (3) ruhya-li: not in Skt. and seems not recorded in Geiger or Pischel, but occurs in Pali: Jāt. iv.408.9 ruyhamānāni, growing (to aor. rūhimsu, just preceding)
- Pres. ruhyati, heals, gets well, Mv i.5.9; viruhyati (mss. °nți; v.l. virühyanti) Mv iii.405.11; pple. oruhyantam Mv iii.34.6, 8, 10
- Gdve. abhiruhyatavya- (read °hyitavya- ? § 34.21) Mmk 719.22
- (4) Caus. ropaya-li (Skt.). See also -rup-, plant (backformation from this)
- Pres. opt. āropaye (2 sg.) Mv iii.20.10 Aor. āropayi (3 pl.) SP 331.11
- Caus. ropāpayasi Mv ii.486.6
- (5) Caus. -rūpaya-ti, with ū for o (or u), § 3.55, as in Pali rūhati = rohati (ruhati), and Pali rūpeti, makes grow, Sn. comm. 142.24, 26; 143.14 (this = Skt. ropayati, and is a caus. to Pali rūhati; it has nothing to do with the noun rūpa as PTSD suggests)
- Pres. orūpayanti (keśān), cut off, Mv i.169.14 (= Skt. and BHS avaropayanti); opt. abhinirupayed Sikş 16.6 (= Pali °ropeti; see Dict.)
- ārūpita- Mv iii.68.19 Ppp.
- ārūpayitvā Mv iii.169.20; 393.18; ārūpetvā Mv Ger. iii.160.7; ārūpitvā Mv iii.150.1; ārūpiya Mv i.352.20

(vi-)rolaya-li, agitate: Pkt. virolai (Hem. 4.121), ppp. virolia-; Skt. vilodayati

- Ppp. virolita- Mvy 5339
- Ger. virolayitvā Mmk 711.2

lakşaya-li, mark (Skt., denom.) vilakşayita SP 193.9 (§§ 29.6; 32.105) Aor.

lagna-ti, stick; also (?) lagga-ti, lagnaya-ti, lagne-ti; \$ 28.19

- Pres. lagnati Mv iii.148.2 (twice); laggati (text) or lagneti (v.l.) Mv ii.429.13; opt. lagneyam (1 sg.) Mv iii. 128.3; impv. lagnatha Mv iii.354.10
- anulagnişyati Mv iii.73.4; and in preceding line 3 Fut. (for Senart avalambişyati) mss. anulagnayişyanti, anulambişyanti; read anulagn(ay)işyati or °yanti
- Ppp. vilagnita- Mv ii.266.13

- lańgh-, leap, jump over: (1) lańghaya-li (Skt.) lańghitvā Mv ii.75.11 (v.l. lańghayitvā, which is Ger. read ii.75.2); langhiya Mv ii.40.21 (Senart °yā m.c.)
- (2) -lańghya-li, nowhere recorded Pres. vilańghyāmi Divy 593.8 (vs; for vilaṅghayāmi,
- which would not fit meter)
 - lajja-li, be ashamed (Skt.)

Ger. lajji LV 329.13

Gdve. lajjitavya- (= Pali lajjitabba-) Ud xvi.4

lapa-ti, speak (Skt.); also lapaya-li, lape-li, the latter at least non-caus.

Pres. ālapeti (non-caus.) LV 236.2; opt. ālape samlapeyyāsi (both 2 sg.; § 29.38) Šiks 100.16

Fut. pass. with caus. mg. (§ 38.24) vilapisyase LV 335.3

alapi Lańk 23.14; ālape My iii.386.15 Aor.

- Gdve. ālaptavya-, samlap°, MSV ii.188.11, 12 (§ 34.21)
- Ger. ālapitvā Mv ii.42.11
- Caus. (? see Dict.) lapayati (= Pali lapeti) Ud xxx.52; more likely non-caus., = lapati, § 38.21

- labh-, take: (1) labha-li (Skt.), also labhe-li, non-caus. Pres. labhenti Siks 336.8; praty-alabhante Gy 50.13 (§ 32.10); opt. labhe Mv ii.365.8; labheya Mv ii.184.11
- Fut. labhişyanı (1 sg.) Mv ii.274.7
- labhi LV 53.4; alabhe Mv ii.166.4; pratilabhetsu Aor. Mv ii.343.11
- Ppp. labhita- LV 327.12
- labhitvā LV 38.2; labhiya Dbh.g. 8(344).15; labhi-Ger. yāna SP 118.7

(2) Miscellaneous forms

- (Fut. pratilapssase Gv 286.4, prob. misprint, see § 31.29) lapsi SP 190.2 (? § 32.81); abhilabdha Mv ii.61.19 Aor.
- (? § 32.82) labdhā Mv ii.334.2 ft. (? § 35.52) Ger.

- lamb-, hang: (1) lamba-le, -li (Skt.) [Pres. impv. vilamba-he? LV 210.4; § 30.8] Aor. vilambişthān (2 sg.) LV 217.11 (§ 32.46) Ger. adhyālambitva SP 253.5
- Caus. opt. lambāvayet Sādlı 170.20; Ger. lambāviya (mss.) Mv ii.172.14 (§ 38.69)
- (2) -lambya-te, nowhere recorded
- Pres. pple. avalambyamānā Mv i.220.8 = ii.22.9 (in the latter mss. agree on "mbva"; in the former they are corrupt but seem to point in the same direction)

lardaya-li, load: Pkt. laddeum, Hindi ladna and other New Indic forms

- Pres. pple. lardayantam Divy 5.22
- Ger. lardayitvā Divy 5.26; 334.19
- Inf. lardayitum Divy 5.23
 - lal-, sport: (1) lala-li (Skt.)
- lalatsuh or lalamsu (mss.) Mv ii.162.5 (§ 32.76) (2) Caus. *lālaya-ti* (Skt.) Aor.
- Pres. lālati Mmk 35.5 (§ 38.28)
- Ppp., as noun, lalayita- LV 374.7 (Dict.; § 34.8)

las-, desire (Skt.)

Inf. abhilāşāya Mv i.61.15 (? § 36.19)

likha-ti, write (Skt.)

- Pres. opt. abhilikhe Mmk 63.25; likheya SP 229.6
- Aor. alikhinsu SP 51.6, Kashgar rec.
- Gdve. abhilikhitavya- Mmk 45.6; 62.23
- Caus. opt. likhāpayed SP 51.4; Ger. °payitvā Divy 547.6; Gdve. abhilikhāpayitavya- Mmk 68.9

lip-, smear: (1) limpa-li (Skt.) Pres. impv. vilimpāhi Mv ii.103.13

Aor. vilimpetsu Mv iii.118.16

(2) *lipa-ti: nowhere recorded; in KSS 24.93; 41.50 is recorded alipat, classed as aorist (cf. Pān. 3.1.53 f.); on this our form may be based, § 28.12

Ger. anulipitvå Mv ii.173.6

- (3) Pass. lipya-le (Skt.); lipya-le(?)
- Pres. lipyase Gv 389.2, misprint or error for lipyase? Aor. lipyatha Mv 1.176.17 (3 pl., § 32.117)

lih-, lick: leha-li: Pali id.; apparently blend of ledhi and lihati (the latter also Skt. but based on 3 pl. lihanti), § 28.39

Pres. parilehati Mv iii.144.15

Aor. lehi LV 197.1 (so with v.l., text lekhi), § 32.23

Ger. parilehiya Mv ii.194.5 (so em. Senart, certainly rightly)

1 li-, cling, lie (see also alliyati): (1) liya-le, -ti (Skt.) Pres. (? or Precative) opt. praviliya (so for text °yu) LV

- 330.19 (§ 29.42) pratiliyatu-kāma- (so mss.; § 36.7) Mv iii.428.10; Inf. pratisamlātu-kāma- MSV ii.128.5 (§ 36.11)
- Gdve. avaliyitavya- SsP 1456.18
- (2) *-laya-ti (Gr., and layate 'V.B.', Whitney, Roots); and cf. next
- Fut. pratisamlayişyāmi Mv iii.428.15
- vilayensu (v.l. °tsu) Mv ii.282.15
- (3) Miscellaneous (probably phonetic equivalent to preceding)
- (leşyati, Skt. Gr.) avaleşyate, samleşyate AsP Fut. 139.17, 18; °şyante id. 209.3

2 li-, fly (= Skt. and BHS di-): -liya-li, cf. Pali Gr. II- (and dI-), PTSD s.v. deti

- Pres. oliyanti Mv i.216.5; praliyanti Mv i.216.11 = ii.19.8, also ii.19.2 (= i.216.5); Senart em. pradiº in ii.19.2, 8, keeping -liyanti in i.216.5, 11
- (§ 32.117) abhiliyathā (mss.) Mv ii.21.5; nilīyathā Aor. (mss.) Mv i.219.3
- lu-, lū-, cut, reap: (1) lūnā-ti (= Skt. lunāti) [Pres. lūnāti LV 286.4 (? see Dict.)]
- (2) *lava-li: Pkt. lavai; Pali *lavati implied by caus. lavāpeti
- Caus. Gdve. lavāpayitavya-, to be caused to be reaped, Mv iii.178.4

luj-, break (= Skt. ruj-): lujya-le, lujja-te (and -li) is broken = Pali lujjati, Skt. rujyate Pres. lujyate Mvy 3061; vilujyati Mv ii.423.13 (so read,

- see Senart's note); lujyante, pralu° AsP 256.7, 8; pple. pralujjam Mvii.354.13 (so read for text puluvam) and perhaps iii.278.17 (or here pralujyanto, °yato?); lujjanti or lujyanti (loc. sg.) Mv ii.371.3 (ms.), also (mss.) pra-lu° Mv ii.370.22; lujyamāna- or lujja", or pra-lu (various forms) Mv ii.356.9; 371.4, 13, 17, etc.; 373.4; Šikş 17.3 lujje, pralujje Mv ii.412.11, 12
- Aor.
- Ppp. lugna-pralugna- Mv ii.429.18

luda-li, stir (Skt. Gr., = lulati): Pali lul- and lut-Pres. ludanti LV 308.4

Ppp. ludita- Mvy 6819; Gv 202.21; a-l° LV 181.12; a-samludita- Gv 402.14

lup-, break (Skt.) vilopsyase LV 334.11 (lopsyati, -te, Skt. Gr.); Fut. § 31.23

lubh-, desire: Caus. lobhaya-li (Skt.)

- Pres. impv. lobhehi Mv iii.291.2
- Aor. lobhaye (3 pl.) Mv ii.425.6; lobhe (so read; mss. loke; 3 pl.) Mv iii.1.9

upalobhetvāna Mv iii.294.15 Ger.

lokaya-li, regard (Skt.)

- Pres. impv. avalokayāhi Mv ii.164.8
- avalokayi Mv i.157.12; vilokaye Mv i.305.17; Aor.
- vilokesi (mss.) Mv ii.281.3; vītiloketsuh Mv ii.125.14 avalokiya Mv ii.164.12 Ger.
- Inf. vyavalokayatum Divy 179.17 (§ 36.7)

lolaya-li, agitate (Skt.) Pass. loliyanti Mv ii.167.12

vac-, speak: (1) vaca-ti: based on Pali aor. avaca, and chiefly limited to aor. forms. Geiger does not explain avaca; it seems to me clearly a blend-form, based primarily on avocat, with root-vowel by influence of forms in vac-: § 28.12

Pres. opt. vaced SP 258.4

- avaci LV 135.1 (§ 32.24); avacī LV 109.16 (id.); Aor. avacim (mss.; 3 pl.; § 32.39) Mv i.247.4; avacimsu (§ 32.38) SP 192.10; avaca (§ 32.113) Mv i.143.11 etc.; avacatetsuh Mv i.306.15 (§ 32.99); avacasi (3 sg., § 32.62) Mv iii.386.11; avacati (mss.; § 32.113) Mv ii.71.5; avacamsu Mv ii.233.17 (§ 32.76); avāca (? § 32.113) Mv ii.330.5
- Perf. uvācat (§ 33.10) Mv iii.337.13; ovācu (§ 3.71) Gv 212.16 (ending of 3 pl.)
 - (2) Aor. Skt. avocal
- avoca Mv ii.37.12 (§ 32.114); avocuh, see § 32.118 Aor. (3) Pass. uccati (cf. next), rare
- Pres. uccati (v.l. ucyati) Mv ii.101.2

(4) Pass. vucca-li, -le, etc. (MIndic; Skt. ucyate); see § 2.51

(5) Caus. vácaya-li (Skt.)

Pres. opt. vacet SP 342.4 (§ 38.27)

vad-, speak: (1) vada-li (Skt.); but most forms could be derived from vade-ti, below

- Pres. opt. vade (2 sg.) SP 116.4; vadi SP 98.7; vadeya SP 229.10; vadeyya (or °yā) Mv ii.233.13, 18; vadesi (2 sg.) SP 93.11; 98.4, etc.; impv. vadāhi Lank 200.13; vadahi LV 56.11
- vadetsu (v.l. vadansuh) Mv i.307.6 (§ 32.102) Аог.
- vaditvāna Mv iii.328.12 Ger.
- . (2) vade-li, vadaya-li: Pali vadeti (see also under prec.); § 38.21
- Pres. vadeti SP 197.4; vademi SP 59.2; LV 335.15; Mv i.8.14 (prose; in same context vadāmi i.30.14); vadenti Mv i.256.11; vadayati Mv ii.237.18; impv. vadehi Mv i.324.10
- (3) Caus. vādaya-li (Skt.), vāde-li, vādati (§ 38.28), 'play (instruments)' etc.
- Pres. [vivädenti Mv i.10.12, but see § 38.23] vädatah (dual) MSV iii.16.6; impv. abhivādehi Mv iii.174.9
- vādisu LV 75.13; pravādayatsu Suv 8.8 (mss.), Aor. § 32.76
- Pass. aor. sampravādyi (3 pl.) Mv ii.328.2; vādyisu LV 194.2; vādyensuh Mv ii.303.17; sampravādyansu (v.l. [°]yetsu; mss.) Mv iii.96.13 (§ 32.102); sampravādyetsuh Mv ii.160.19
- Caus. vädäpayatah (dual) MSV iii.16.7; Ppp. vädäpita-SP 51.11; 52.1; Ger. vädäpiya SP 52.4 (4) Miscellaneous forms

[Pres. (Prec.?) opt. vadyād LV 444.2 (§ 29.44)?]

['Pass.' (apparent) vadyase Mv ii.58.8 (here mss. vād°); 59.10; so interpreted by Senart, but really false Skt. for MIndic (Pali) vajjasi = Skt. varjyase, you are rejected, shunned (Dict.)]

vadh-, also (MIndic) vah-, slay: vadha-ti: Pali id.; no pres. in normal Skt.; § 28.14

- Pres. impv. vadhatha Mv ii.282.3 (? mss. vadha, bandhatha)
- Fut. vahişyāma (so read) Mv i.17.2
- vadhitvā Mv iii.78.5; °tvāna Mv ii.236.6 Ger.

vadhraya-li, castrate, denom. to vadhri-Pres. vadhrayanti Mv i.96.8

vanda-le, greet (Skt.), also vande-li, non-caus. (§ 38.21) Pres. vandenti Mv i.256.10; vandima LV 364.6 (§ 27.10)

- abhivandişu LV 116.4 Aor.
- vandiya (mss.) Mv iii.93.8 vandanāya SP 425.3 Ger.
- Inf.

- Caus. Fut. vandāpavisvāmo Mv ii.26.6; Aor. °payetsuh Mv i.223.13
- rap. sow, scatter (Skt.) Ppp. vutta (§§ 2.10, 53) Mv iii.360.14

varņaya-li, describe (denom., Skt.), also varņa-li (§ 38.28)

Pres. varņanti LV 29.12

Aor. samvarnaye Mv i.299.17

pardh-, cut (Skt. Dhätup, vardhayati) Caus. vardhāpayata MSV i.119.14 (§ 38.55)

1 vas-, dwell: (1) vasa-li (Skt.)

- Pres, opt. samvaseyā Mv ii.67.5; impv. vasāhi I.V 241.18 adhyāvasişyasi (Epic Skt.) Mv ii.159.2; vasişyam Fut. Mv ii.463.2
- adhvävasi Mv i.193.20; vipravasetsu(h) Mv i.343.1-2 Aor. Ppp. vasita- LV 364.3
- vasitvā Mv ii.462.18; °tvāna Mv iii.15.19 Ger.
- (2) vasaya-li, vase-li, non-caus. (§ 38.21) Pres. vasayāmi Divy 212.4, 6; impv. āvasehi Mv ii.405.9; vasayathā LV 202.5
- (3) *-uşa-li, in upoşa-li, keep (Sabbath) = Pali upavasati, and really only a phonetic Middle-Indicism for that form, with o for ava
- Pres. (upoşadham) upoşati Mv ii.177.20
- ? Gdve. upoşaniya-(prabha-) Sukh 29.14, Dict.

(4) Caus. vāsaya-li (Skt.)

Pres, impy, adhivāsehi My ii.272.2

- Fut. adhivāsisyāma SP 271.10
- Aor. adhivāsavī Suv 149.11; adhivāsesi Mv i.329.8; vipravāsayet Mv iii.44.22
- (5) Semi-MIndic ppp. vusta, etc., see §§ 2.54, 62; 34.11; and perhaps ger. vustā, § 35.52

Ppp. nivasta- (= Pali nivattha-) LV 157.21 etc. (§ 34.11)

vasya-li, cry: § 3.34 and Dict. Pres. vasyati (mss.) Mv ii.450.8

1 vah-, carry; intrans. move: (1) vaha-li (Skt.); most of the forms could go with vaheli, see next

- Pres. opt. āvahe Mv ii.176.5; pravaheya Mv iii.358.3; Impy. vahahl LV 237.1
- udvahetsuh (so with mss.; intrans.) Mv ii.39.5 Aor.
- vahitvā (mss. vān°) Mv ii.434.9 Ger.

udvahitum Mv iii.252.16 Inf.

(2) vahe-li, non-caus. (cf. under prec.), § 38.21

Pres. vaheti Mv ii.228.1

Fut. vaheşyam Mv ii.236.13

(3) vahya-ti, flow (intrans.); in this sense Skt. may use vahati, vahate, and pass. uhyate; is this a blend of these, or a passive irregularly formed on vah- instead of uh-? Or merely a 4th Class present, as often in other roots (§ 28.28)? Cf. § 37.20

Pres. vahyanti LV 398.6 (prose)

- (4) Caus. vähaya-ti (Skt.)
- Pres., non-caus. (§ 38.23), prativāhayāmi Gv 122.8
- Pass. pravählyanti Mv iii.405.2; ovähiyati Mv il.274.16 (5) Pass. vuhyati (= uhyate), etc., see § 2.52

(6) Miscellaneous

- Participle vyūdha-māna RP 55.1 (§ 34.1)
- Inf. vodhū (? \$\$ 3.3; 36.5) KP 82.9 (vs)

2 vah-, see vadh-

1 vā-, weave: (1) *vāya-li, for Skt. vayati; aside from the caus. (below), only in Ppp. väyita- (= Pali id.)

Divy 276.11, and em. våyitum (? sec 2). Geiger 196 seems to regard this form (he alleges also an equivalent vata-, not recorded in the dictionaries, but quotes no passage) as based on a pres. pass. *väyate, but I find no record of such a form. See §§ 28.24, 33

- Caus. vāpayanti Divy 213.10; Fut. vāpavisyante Divy 213.11: Inf. vapayitum Divy 213.8
- (2) vayati (Skt.) vayitum MSV i.82.14 (= Divy 83.23 em. vâyitum, Inf. mss. vāsitum)
 - (3) Miscellaneous

Pass, pple. uyamāna- Mvy 8410, see Dict. s.v. uvate

2 vá-, blow: (1) váya-li (Skt.)

Pres. opt. nirvāyi Mv ii.375.8; vāyeyā Mv iii.317.7

Fut. vāyisyanti Mv ii.308.8

parinirvâyi (? mss. °ya) Mv i.304.11; vâye (3 pl.) Aor. Mv ii.352.14; parinirvāvet Mv i.267.18; nirvāvetsu(h) Mv ii.23.2

(2) vå-li (Skt.)

Aor. nirvāsu, nirvāmsu (3 pl.) Mv i.197.6 = 357.8 (§ 32.75)

(3) Caus. vāpaya-ti (Skt.); vāpe-ti; also as non-caus. (\$ 38.58)

- Pres. impv. uirväpehi Mv iii.390.11 In non-caus. mg.: opt. parinirvāpayeyam Divy
- 90.10 parinirvāpayitu-kāma- SP 250.12 (mss., except two °vātu-; KN em. °vāyitu-) Inf.

vāhaya-li, sec bāh°, and vah- (4)

vij-, shake: (1) vija-li (Skt.)

- Ppp. samvijita- Ud xix.1
- (2) Caus. vejaya-ti (Skt.)
- Ger. udvejetvāna Mv ili.77.8

*vijalaya-li, untangle, card (wool): Pali vijateti; denom. from vijata-

Caus. opt. vijatåpayed Prät 498.1

villale-li, make impure: Pkt. (Deśi) id.; in Pañc. Rec. 2 § 130 read vițțălități, as noted in Add. and Corr. Pres. vittäleti Mv iii.126.16

1 vid-, know: (1) vida-ti, § 28.7

- Pres. vidāmy Gv 340.9; vidasi Gv 316.21
- viditvāna Mv iii.71.6; 83.18 Ger.

(2) Caus. vedaya-li (Skt.), also veda-li

Pres. vedati (§ 38.28) Av i.243.12; opt. nivedaye Mv ii.37.12; impv. prativedehi Mv iii.2.5 Ppp. vedayita- (§ 34.8) LV 420.3

Ger. veditvāna (mss.) Mv iii.61.11

(3) Miscellaneous

Pres. 2 pl. vettha LV 125.7 (? § 28.60)

2 vid-, find: (1) vinda-li (Skt.)

- Pres. opt. vindeyā LV 201.22; impv. vindahi (? text vandahi) Gv 485.22; 487.2 (here 2d ed. vi°) Ger. vinditvā Mv ii.67.6; vindi (?) Gv 485.23; 487.10
- (§ 35.51)
- Caus. vindayanti or °nti (§ 38.9) LV 242.7 (2) vida-li ?

Pres. vidanti SP 305.11 (§ 28.31)

(3) Pass. vidya-te (Skt.), and -vidyaya-le, non-caus.

- Pres. samvidyayante LV 115.9 (§§ 37.21; 38.18, 21); opt. vidyetsuh Mv i.273.2
 - (4) (Caus.? denom.?) -veda(ya)-li ?
- Ger. nirveditvā Mv ii.198.1 (prose), becoming disgusted; denom. from nirveda-? Form uncertain; v.l. nirvinitvā; read nirvinditvā? Pali pres. nibbindati, but

² vaş-, clothe (Skt.)

also opt. nibbide, v.l. °je (Jät. v.368.8; comm. nibbindeyya), and ger. nibbijjitvā, implying nibbijjate = Skt. nirvidyate; Pkt. only nivvijjai (apparently)

vip-: vepa-ti, -le, tremble (Skt.) Ger. vepitvā Mv ii.42.11

virāgaya-ti, be averse, offend (denom.; rare in Skt.); for regular Skt. forms see Dict.

Pres. opt. virāgayi (1 sg.) Bhad 24

vilomaya-li, go contrary to (denom.; Skt. ppp. vilomita-), also viloma-ti

Pres. vilomayanti Gv 493.3; pple. a-vilomayan Dbh 47.18; a-vilomanta (read °mata, m.c.; n. pi. m.) Dbh.g. 27(53).3 (§ 38.30)

viś-, enter: (1) viśa-ti (Skt.)

Pres. impv. pravišāhi Mv iii.25.8

praviśisyati Mv i.360.4 Fut.

- upāviši Mv ii.199.9; pravišitsuķ (mss. °şitsuķ, pravarsetsuķ) Mv i.200.12; pravišāsi Mv iii.390.10 Aor. (§ 32.59); pravišet Mv i.307.16; pravišetsuh Mv i.151.15
- praviśitvā SP 236.7; praviśiya Mv i.361.5 Ger.
- praviśitum Mv iii.151.8; praviśatu LV 393.5 (§ 36.7) Inf. Caus. impv. samviśāpehi Mv ii.430.6; Ppp. upavišāpita-
- Mv ii.103.4; Ger. upavišāpayitvā Mv ii.479.10 (2) -visya-li: Pali pavissāmi, occurs twice, inter-
- preted by Geiger 65.2 as haplological for pavisissami.
- But it can quite as well represent pravisyami, as here
- Pres. praviśyāmi Mv ii.492.19; praviśyanti Divy 196.20
- (both prose) (3) Caus. veśaya-ti (Skt.), here sometimes in noncaus. mg.
- Pres. opt. vipraveśeyā Mv ii.146.14; niveśeyyā Mv i.53.6 (mss.); impv. samveschi Mv ii.431.11, substituted
- for samvišāpehi (above, 1) in same phrase nivešaye Mv i.312.12; (non-caus., § 38.23) nivešayi Aor. (3 pl.) LV 158.13; praveši Mv iii.162.5 praveši (non-caus.) LV 243.11
- Ger.
- iii.123.10; pple. pravešiyantehi Mv
 iii.127.12; Ger. pravešiya (§ 35.17) Mv
 iii.16.3
 (4) Miscellaneous forms Pass.
- Aor. praveksi LV 136.6 (§ 32.80)
- Ppp.
- with -māna-: pravistamāna- (§ 34.1) LV 7.2 anupravistum (§ 36.9) Mmk 94.1 Inf.
- vis-: (1) (pari-)visa-ti, serve = Pali id.; in Cl. Skt. only pariveşayati (Epic rarely °veşati), in same mg., and pass. °vişyate; is °vişati a back-formation from one of these forms? For s, both s and s are written in mss. (and sometimes printed by Senart in Mv)
- Pres. parivișati Mv ii.211.17 (so read; Senart praticarati, em.); 276.9; iii.145.10; opt. °vişeyam Mv ii.276.6 parivişişyāmah Mv ii.275.12 Fut.
- parivișitvă Mv iii.130.12; 145.12 Ger.

Pass. parivişiyati Mv ii.276.4

Caus. Ppp. parivişāpita- Mv ii.439.4

(2) (pari-)vesaya-li, serve (Skt.; cf. above)

Caus. Fut. parivesāpayişyam (1 sg.; text with mss. °veś°) Mv ii.435.11

*(ni-)vulla-li, sink down; MIndic form (§§ 2.11; 3.57) of nivartati; cf. nivarteti Mv ii.172.4, just before these forms

Ger. nivuttiya Mv ii.172.6

Caus. Ger. nivuțțăpiya Mv ii.172.7

1 vr-, cover: (1) -vrni-le, 9th class (§ 28.16) Pres. impv. vivrnisva LV 398.17

(2) -vara-ti: Pali id. (as vi-, sam-varati); cf. § 28.13

Pres. prāvaranti SP 11.12; Divy 216.3 (but 215.29 prāvrņvanti); vivarāmo SP 110.1; vivarāmi Gv 67.3 (prose); impv. prāvaritu (§ 27.10) Divy 215.29; prāvaradhvam Divy 216.4; pple. vivaran LV 414.22 (so with v.l. "ram for text vicaram: disclosing = bestowing treasures) vivari LV 236.17; vivarctsu(h) Mv iii.262.6

Aor.

- vivarita- LV 233.3 Ppp.
- prāvaritvā SP 114.5; Suv 69.12; vivaritvā SP Ger. 176.11; prāvariya Mv ii.441.14; prāvariyāņa SP 272.3

Gdve. ávaritavya- Mv iii.55.6; prāvaritavya- MSV ii.68.9 Caus. pple. prāvarayantyā Bhīk 11a.5 (§ 38.9)

(3) -mura-li, cf. Pali (apā-)purati; for apā-varati (cf. prec.) with u for a after labial consonant, § 3.57, cf. Geiger 19.2; but I think Geiger wrong in suggesting that the position of the accent is concerned. I doubt that the form is old as implied by CPD s.v. apapurati (with reference to Wackernagel I § 21) Pass. apāvurīyati (Dict.) Mv ii.158.1

(4) Caus. vāraya-li (Skt.)

- Aor. sampravārayi Mv iii.257.10; °rayet Mv i.325.11; parivārensu(h) Mv ii.225.10 (§ 32.102)
- pravāritvā Mv ii.489.10 (prose), having clothed Ger. (another); parivāria (§ 35.38) Šiks 343.18
- vārīvati Mv ii.493.1; pple. vārivanto (v.l. vārī^o) Mv ii.274.1; nivārīvantī (n. sg. fem.) Mv ii.174.12 Pass. (5) Miscellaneous forms

Pres. impv. apāvrņohi SP 191.10; prāvrņothā LV 80.2 (§ 28.62)

vivrtum LV 46.7 (§ 36.9) Inf.

2 vr- choose: (1) 'Caus.' varaya-li (Skt.; usually indistinguishable from simplex in mg.) Pres. impv. varehi Mv ii.70.4

(2) vāraya-li, hand out, present; in this sense app. unrecorded; Pali and Epic Skt. have the form in mg. choose = varayati, vrnite; may be denom. to Skt. vāra. see Dict.

Pres. vāreti (v.l. vārayati; Dict.) Mv li.442.2

Caus. opt. vārāpeya KP 158.3 (text corrupt); 159.7

vrj-: varjaya-ti, avoid (Skt.)

Pres. opt. varjeya (mss.) Mv ii.299.15

vivarjişyase LV 333.14 Fut.

- vivarjayi (3 pl.) LV 158.13; parivarjaye (v.l. °varje; 1 sg.) Mv ii.131.7 vivarjitva SP 99.4; parivarjitvā Mv i.12.14 Aor.
- Ger.

vrt-, turn [see also (ni-)vuff-]: (1) varta-ti (Skt.)

- Pres. opt. pravarti LV 48.11; impv. nivartahi Mv i.270.14 Aor. parivartişu LV 173.16; vartisu LV 402.14; pravartitha (3 pl., § 32.42) Mv i.13.16; vivartetsu(h) Mv i.80.1; vartesi Mv i.268.17
- pratinivartita- Mv iii.102.19 (intrans.) Ppp.
- Ger. nivartitvā Mv iii.102.18
- nivartanāya Dbh.g. 17(353).9; vartanatāyai Bhad Inf. 10 (§ 36.17)
- impv. pratinivartāpaya (tu, separate word) Divy Caus. 346.10; Inf. nivartāpayitum Mmk 86.3

(2) Caus. vartaya-ti (Skt.), varte-ti, also in non-caus. mg. (§ 38.21)

- Pres. opt. pravarteyam Mv i.330.4; parivarteya Mv iii.32.2; nivartaye (2 sg.) Mv i.75.19; impv. nivartehi Mv ii.151.8; pravartayahI LV 414.22; (non-caus.) nivartayatha LV 228.8; nivartehi Mv ii.105.8; nivartayasva LV 225.12
- (a)vartayi LV 220.22; pravartayi (1 sg.) Gv 231.18; Aor. pravartayinsu Hoernle MR 134 for SP 327.11; vinivartayet Mv ii.61.21

vrdh-, grow: (1) vardha-li (Skt.); also MIndic vaddhfor vardh-

- Fut. abhivardhişyanti Mv i.332.13
- Caus. Aor. vardhāpaye Mv il.38.1; Ppp. vaddhāpita- (mss.) Mv i.287.16; Ger. vardhāpayitvā Mv i.310.2; vaddhāpayitvā Mv ii.421.11; vaddhāpetvā (mss.) My i.289.8
 - (2) Caus. vardhaya-li (Skt.)
- Pres. opt. vardhayi (2 sg.) Gv 488.14; impv. abhivardhehi Mv iii.168.1
- vivardhayi (3 pl.) SP 131.1 Aor.
- Pass. samvardhīvati Mv iii.390.3; 405.9; vardhīyanti Mv i.274.2; ii.433.14; pple. samvardhiyamāna- (v.l. °dhī°) Mv ii.423.14

vrs-, rain: (1) varşa-li (Skt.)

- Pres. impv. varşahī LV 171.4 Aor. abhivarşi LV 301.6; pravarşi (3 pl.) LV 285.6; varşişu LV 222.7; prāvarşu (3 pl.; § 32.74) Suv 8.8; pravarșe Mv ii.339.2; pravarșetsuh (one ms.) My i.200.12
- varşitva LV 117.11; varşi LV 294.5; pravarşi Mv Ger. ii.413.17 (mss.)
- Caus. varşāpayati Mmk 464.1
- (2) vrsya-le (§ 28.26)
- Pres. (devo) vrşyate, rains, Divy 71.5 (prose)

prstayale, rains (denom. to Skt. vrsta or vrsti) Pres. pple. °yamāne MSV i.36.8 Ppp. vrstita LV 283.10

1 (ā-)vrħ-, = brħ-, q.v., tear: (ā-)vrṃħa-li (§ 28.32) Pres. pple. āvrṃħato Mv i.13.2 (see Dict.) Ppp. avrmhita- Mv i.18.12 (prose)

2 (ā-)vrh-, brh-, brah-, free, restore (a penalized monk); blended with prec., see Dict. s.v. abrhati: (1) abrha-ti

(*ābraha-li?)

Pres. äbrhyät Prät 488.2

- Ppp. äbrhita- Prät 488.3
- Gdve. abrahitavya- Prat 488.1
- (2) āvarha-li
- Pres. opt. ävarhet MSV iii.53.10; impv. °hata 49.11; °hatu 51.9
- Ppp. āvarhita- ib. 57.4; 58.18
- Gdve. avarhitavya- ib. 49.17
- āvarhitum ib. 57.2 Inf.
- (3) Miscellaneous
- Ppp. āvrīdha- (as ābrīdha- to ā-brh-) MSV iii.74.6 etc.
- veth-, wrap: vethe-li-, vethaya-ti = Pali id.; MIndic for Skt. 1 vest-, q.v., § 2.12
- Pres. impv. vethetha (mss. ved°) Mv ii.451.9

Ppp. vethita- Mv ii.82.14

- vețhitvā Mv il.173.6; vethayitvā Mv ili.390.8; Ger. vethiyāna Mv ii.485.14
- Gdve. nirvethayitavya- Bhik 16a.2
- Caus. impv. prativețhāpehi Mv ii.171.12; Ppp. vețhāpita-My ii.171.15

vedha-ti, also vedhaya-ti (non-caus.), § 38.21, shake, tremble: Pali vedhati; MIndic from vyath-, with Prakritic voicing of th, § 2.28; on e cf. § 3.116. See also 2 vyadh-

- Pres. vedhati sampravedhati Divy 479.11, 12; pple. avedhamāna- SP 24.15; pravedhayamāna- (so with best mss.) LV 188.3
- Imperf. avedhat prāvedhat samprāvedhat LV 352.2; 411.1; avedhanta prāvedhanta samprāvedhanta Dbh 98.31
- vedhe sampravedhe Mv iii.334.2; 341.5 Aor.
- Ppp. vedhita-, pravedhita-, sampravedhita- Mvy 3007-9; Samādh 19.6

vell-, shake: Caus. vellaya-li (simplex once in Skt. in mg. knead; not recorded in Pali or Pkt.)

Fut. parivellayişyanti AsP 215.13

1 vest-, wrap (see also veth-): vestaya-ti (Skt.) Caus. impv. nivestāvehi Mv i.273.14 (so read. § 38,70)

2 veşt-, labor; caus. veşte-ti, supervise, keep working (servants and domestic animals); to denom. from vesti-(Dict.) = Skt. visti-, forced labor

Gdve. vestetavya- (§ 34.21) Mv iii.177.16, to be supervised Caus. gdve. vestāpayitavya- Mv iii.178.1, to be caused to be supervised

1 vyadh-, pierce: (1) vijjha-ti = Pali id., MIndic for Skt. vidhyati, § 2.14

(Aor. apavijhinsuh Mv ii.139.4; so Senart; mss. aparijhimsuh; in ii.137.9 Senart reads apavidhyinsuh, to Skt. apavidhyati, but mss. aparadhyetsuh. Form and mg. both dubious; see under root radh-)

Ppp. vijjhita- Mv i.22.5

vijjhitvā Mv ili.456.11 (= viddhitvā i.12.5); apa-Ger. vijjhiyāna Mv ii.104.4

(2) *viddha-ti; if correct, denom. to ppp. viddha-, § 28.19; see § 35.28

Ger. viddhitvā Mv i.12.5 (= vijjhitvā Mv iii.456.11)

(3) vindha-ti, vindhaya-ti: AMg. vindhai, vindhemāņa-, cf. Pischel 489; § 28.32

- Pres. opt. vindhet Sādh 357.14 (prose; vv.ll. vivandhayet, bandhayet); vindhayet Sādh 384.15 (prose); pple. vindhantim Sādh 298.4 (vs); vindhayantim Sādh 301.1 (prose)
 - (4) Miscellaneous

Aor. aviddha (? by em.; § 31.82) Mv i.131.15 Gdve. vedhanīya Šikş 42.15. Cf. Skt. vedhana, vedha, etc.

2 vyadh- : vyadha-li, shake, tremble; doubtless false Skt. for vedh-, q.v. (or may be regarded as blend of this and Skt. vyathati)

Pres. vyadhati pravyadhati sampravyadhati Divy 46.7 vyadhita-, pravyadhita-, sampravyadhita- Divy Ppp. 327.9

vraj-, proceed: (1) vraja-ti (Skt.)

- Pres. opt. vraje Mv ii.328.22; vraji SP 111.2; vrajeyā LV 201.18; pravrajiyāmah (mss.) Mv ii.234.1 (§ 29.34); impv. pravrajāhi Mv ii.210.1; (with caus. mg., § 38.24) Mv iii.386.13; pravrajehi (caus. mg., Senart em. °vrāj°) Mv iii.268.5; pple. vrajāna- (§ 34.4) Divy 392.20
- Fut.
- pravrajişyam (caus. mg., § 38.24) Mv iii.268.6 vrajī LV 236.18; pravraji Mv ii.166.3; vrajişû LV Aor. 169.12; pravraje Mv iii.191.6; vrajesi Mv ii.86.8; pravrajetsuh (mss.; caus. mg.; § 38.24) Mv i.323.9
- pravrajitvā SP 465.3; °tvāna Samādh 8.29; pra-Ger. vrajiya Mv ii.43.1; °yâna LV 240.9 (2) -prajya-li; in some Pkts. vajjai, Pischel 488
- Pres. (both prose) pravrajyāmi Divy 574.1; impv. pravraj-
- yähi Mv ii.147.5 (kept by Senart) Fut. (allegedly opt., § 31.21) anupravrajyeham Mv iii.50.16 (so mss., Senart em. °jeham)

(3) Caus. pravrājaya-ti (Skt.)

- Pres. opt. (non-caus. mg.) pravrājayeyam Gv 417.15 (§ 38.23)
- pravrājayişye Divy 260.20 Fut.
- pravrājayi (3 pl.) Mv iii.271.5 Aor.

-samsa-ti, praise (Skt.)

- Aor. prasamse Mv ii.119.3
- Pass. praśamsyate Mv i.88.8 (§ 37.31)

suk-, be able: (1) sakno-li (Skt.)

Pres. śaknoma (followed by ā-; one ms. śaknoma-r-ā°) Mv iii.427.2; śaknotha Mv i.313.9; śaknonti Mv iii.74.6 (all prose; § 28.62)

(2) Pass. sakya-le (Skt.)

Pres. opt. śakycya Mv iii.392.15

(3) sakya-li, -le as active and personal, 'can' (so Epic Skt., see BR); also semi-MIndic sakk-: Pali sakkati, AMg. sakkei (both apparently rare), Pkt. sakkai (commoner); in BHS fairly common. Looks like use of passive form as active, cf. § 37.22

- Pres. śakyati Mv iii.151.15; 159.13; °asi LV 143.22; 147.14; Mv iii.390.16; Divy 303.30; 397.24; 490.18; °āmi LV 143.22; 147.16; Mv ii.247.17; iii.390.17; Divy 358.10; 486.11; °āmo Mv ii.175.10; °āma Mv ii.126.3 (prose; in same line šaknoti); °atha Mv ii.175.9; šakyase Divy 374.19; opt. šakyeya Mv ii.126.18; °yã (1 sg.) Mv ii.249.1; °yam Mv ii.171.18
 Ppp. šakkitam, it was possible, Av ii.23.2; 71.1 (Speyer
- em. śakitam both times) (4) *śakşa-li (§ 28.41), false Skt. to Pali sakkhati,

(4) "sakknati, which derives from fut. saksyati but came to be used also as pres. even in Pali, as is shown by the fact that a new fut. sakkhissati is formed on it. In Sn 319,834 the forms sakkhati, sagghasi are to be taken as presents

Ppp. śaksitam Mv iii.263.15 (prose: asmābhi na śa°... kartum, we could not do...)

(5) Miscellaneous

śakyā (uninflected) = Pali, Pkt. sakkā, (it) is possible, Mv i.351.10 etc. (Dict.)

śańk-, doubt: (śańka-ti, Caus. śańkaya-ti, both Skt.)
 Pass. of caus. (?), pres. pple. sańkiyanto (for śa°) Mv
 iii.37.9 (see Dict.), lit., being made anxious? (§ 37.6)

śa[-: *šā*[aya-li, *šā*[e-li, destroy: Pali sāţeti (Skt. *šā*tayati). Note that Kalpanāmaņģitikā fol. 196³ V 2 (Lūders, Kl. Skt. Texte 2, 177, cf. 63) has the regular Skt. *šā*t- in a passage corresponding to a Pali passage containing sāţ-

Pres. śātayati Mvy 2423; śāteti Mv iii.385.17

sabdaya-ti (Skt.) or śabda-ti (cf. Pali samsaddati), sound (denom.); Caus. śabdāpaya-ti (Epic Skt.), śabdāvayati, cause to be called, have summoned, §§ 38.56, 70 Pres. °payati Divy 31.23 f.; opt. °payethāh Divy 484.6;

- Pres. °payati Divy 31.23 f.; opt. °payethāh Divy 484.6; impv. śabdāpehi Mv iii.36.2
- Fut. śabdāpayişyati Divy 5.24 f.
- Ppp. śabdāpita- Mv l.272.14 etc.; śabdāvita- Mv iii.36.2
 Ger. śabdāpayitvā Mv ii.435.4; 441.1; °piya Mv ii.78.3 (so read); 110.14; śabdāvitvā Mv ii.421.5; 442.2; śabdāviyāna Mv ii.453.13; iii.287.16 (and v.l. for °piyāna ii.173.1)

Inf. sabdāpayitum Divy 171.5

sam-, be quiet: (1) sama-li (also sa°). Not in Skt. unless in the very questionable $\delta m.\lambda \epsilon \gamma$. prasamet (Rām. Gorr., see BR); not recognized for Pali by Geiger or PTSD but occurs there at least once (see below); in AMg. and other Pkt. occurs as samai. Back formation from the Skt. causative, esp. its ppp. samita; § 28.30

Pres. vopasamanti, are pacified, Mv iii.371.5; cf. § 3.71; would be vupasamanti in Pali, but the same vs Jāt. v.143.2 reads upasamanti (queried by Fausböll, but now supported by, and supporting, our form)
Fut. praśamisyati Mv i.289.4 (Skt. Gr.)

(2) Caus. samaya-ti (Skt.)

- Pres. opt. praśami (§ 38.27) Sukh 23.5; śameya Suv 37.16; impv. samehi LV 359.11
- Ppp. praśamayita- Mv i.254.8 (§ 34.8)
 - (3) Caus. šāmaya-li (Skt.)

Pres. opt. prašāmi (§ 38.27) Sukh 23.6; impv. nišāmehi Mv ii.88.13

šal-, shake (Skt. Gr. śalate, Pali salayati) Ppp. salita-, stirred up, Jm 89.23

sallaya-ti, afflict, injure: cf. AMg. sallai

Inf. sallayitum Mmk 462.19

śās-, order: (1) śāsa-li (Skt.); also śāsaya-li, jāse-li, non-caus. (§ 38.21)

- Pres. impv. anuśāsayāhi Gv 54.16; samanušāsehi Mv ii.103.14
- Aor. prāsāsi (mss. prās°) Mv ii.394.18; anusāse Mv iii.104.14
- Gdve. śāsayitavya- Divy 243.28
- Pass. opt. śāsiyema Mv iii.166.14

(2) Miscellaneous

Fut. anuśāksyati (? § 31.25) Suv 81.6

śiksa-li, learn (Skt.)

- Fut. siksisenti Gv 481.1 (§§ 3.116; 31.28)
- Aor. śiksisu Sukh 74.6
- Ger. anuśiksi LV 422.5
- Caus. śikşāpayasi LV 125.19; °peti Mv iii.362.5; impv. °pehi Mv iii.362.4; Fut. °payişyāmi Mv iii.362.1; Ppp. °pita- Mv iii.394.9 etc.

(ava-, o-)śira-li, also -śire-li, sometimes -sara-li, abandon, etc.; MIndic root (also spelled °sir°); see Dict. Pres. ośirati Mv il.426.7; iii.1.6; 2.11; osiranti Mv il.24.1;

- avasirasi Mv ii.165.12; osiresi (2 sg.) Mv iii.165.19; opt. osireyā Mv ii.315.6; osire (v.l. oś°) Mv ii.383.7; impv. osirāhi Mv ii.272.13; pple. osiratu (n. sg. m.) Gv 255.14
- Fut. ośirisyati Mv ii.426.18 (v.l. osar^o); 427.10; osarisyasi (mss.) Mv ii.459.15
- Aor. avašire (3 pl.; v.l. °sire; Senart em. °kire) Mv ii.343.19; osire (3 pl., v.l. oś°) Mv iii.273.16; avaširi Mv ii.344.15 (so read, mss. avašiti, avati, Senart avasrjati); oširi Mv ii.349.16 (so mss., Senart em. °kiri); ošarI(-r) LV 357.3 (Dict. s.v. avaširati 3)
- Ger. ośiritvă Mv i.143.13 (read °tva m.c.; no v.l.) = 200.9 (here mss. osaritva) = ii.4.6 (here mss. okiritvā); ii.298.6 (mss. ośiritvā or okir°); ii.334.22; 335.4; 367.19; 452.16; ośiritvāna Mv ii.367.22; avaširiyā LV 240.15
- Gdve. ośiritavya- Mv ii.424.20; 425.16

śiş-, leave: (1) Caus. (vi-)śeşa-li for -śeşaya-li (which is Skt.)

- Pres. viścsanti Lańk 357.1; 371.3 (§ 38.28) (2) Miscellaneous
- [Inf. ucchestum (? v.l. ucchre°), to send forth, Divy 186.5; see Dict.]

sisya-le, denom., learn (? Dict.); cf. also next Pres. *sisyate LV* 127.4

Caus. śişyāpayişyasi LV 126.12

sisyaya-li, -le, instruct, make into a pupil (denom., or caus. to prec.)

Fut. śişyayişyāmi LV 127.2; °şye LV 126.20

ši-, lie: (1) *še-le* (Skt.)

- Pres. 3 pl. šenti (analog. to šeti for šete; or to šayati, § 28.60) Ud xvii.4; impv. šehi (or sehi; mss. mehi) LV 196.10
- Aor. aśesi (so read, § 32.64) Mv ii.86.8
- (2) śaya-te, -ti (Skt.)
- Caus. Ppp. sayāpita- (so read for sayāyita-, § 38.53) Mv iii.39.6; 40.2

Synopsis of verb forms

(3) Caus. śdyayati (Skt.) Ppp. śāvita- MSV iii.142.1 Caus. ppp. śāyāpita-(ka-), see § 38.61 and Dict.

śuc-: śoca-ti, burn, grieve (Skt.) Pres. impv. śocahi Mv ii.218.13 Aor. socetsuh Mv iii.69.7

sudh-: Caus. sodhaya-li, purify (Skt.) Aor. samsodhayi SP 114.1; sodhaye Mv i.336.15; visodhensuh Mv ii.199.3 Ger. visodhetvā (§ 35.25) Mv i.4.7

Gdve. visodheya- (§ 34.24) Mv iii.318.2, 3 Caus. opt. sodhapayet SP 107.6; Ppp. sodhapita- Mv iii.298.8

subh-, be beautiful: sobha-li, -le (Skt.) Pres. opt. sobheya Mv iii.10.16 Aor. sobhe Mv ii.306.16 Caus. sobhāpayati Mmk 644.17

śus-, get dry: (1) śusya-ti (Skt.) Pres. opt. susye LV 330.3 (2) Caus. śosaya-li (Skt.) Pres. opt. upaśosaye My ii.239.2; śoseya My ii.329.7 Fut. (pass.) višoslsyase LV 334.13

*suska-ti or suskaya-ti (-te), get dry (denom.): Pali sukkhati

suskitum Divy 397.27 Inf.

sr-, crush: (1) Pass. sirya-te (Skt.) siryetsuh Mv ii.126.9 Aor. (2) Caus. sāraya-ti (Skt. Gr. only) Ppp. viśārita- LV 351.6

śekhe-li, train; denom. to Pali se(k)kha-, AMg. seha-, BHS śaiksa- (Dict.); cf. AMg. sehai; also miswritten śes-; and with s for \$

Pres. sekheti Mv iii.162.10

- sekhita- Mv iii.390.4 (for se°); seșita- (miswriting Ppp. for sekh°) Mv ii.73.10, 15
- Pass. sekhiyati (v.l. śe°) Mv ii.423.15; śekhiyanti Mv ii.434.10; sekhiyanti Mv iii.184.6

śran-: (vi-)śrānaya-li, give (Skt.) Pass. viśrāņīyati Mv iii.405.2; pple. viśrāņiyantehi Mv ii.72.17

śrama-ti, be weary (Skt.) viśramitvā Mv ii.106.2 Ger. Inf. viśramitum Mv iii.350.21

srambh-, srabh-, (with vi-) trust; (with pra-, prali-pra-), be still, (trans.) still; also with s for \$:(1) (pra-, prati-pra-)srabhya-ti, -te, also with s-, be abated; ger. (trans.) still: apparently not in Pali, which only has (patip)passambhati, is calmed, and "bheti, calms (caus.); these cpds. seem not to exist at all in Pkt.

- Pres. (intrans., or passive?) prasrabhyate Mvy 1587; pratiprasrabhyati Mv i.253.10 (and in 8 read °śrabhyeyā? for mss. °śasteyā, Senart em. °śāmyeyā); pratiprasrabhyante Divy 68.3; 138.10; 367.1 (mss. °śra°); 568.15; Mmk 183.6 (°śra°)
- prasrabhya (trans.) Mvy 1179 f., 1185 f.; prati-prasrabhya (or °śrabhya; trans.) Divy 161.11; Ger. 163.3; 190.22; 494.22; 549.10; Mmk 182.3

(2) Caus. -śrambhaya-li (or °sra°; Skt.)

Pres. opt. viśrambhayi SP 114.12

visrambhesi Mv ii.242.15 Aor.

Inf. pratiprasrambhayitum Mv i.254.2

śri-, resort: (1) śraya-li (Skt.), śre-li; Skt. ucchrita. ppp. (see §§ 38.44 ff.)

Pres. impv. ucchethā, for ucchrethā LV 335.13 (vs), see s.v. chid (3)

Fut. ucchreşyam Mv ii.126.6

- niśrayitvā (Skt. śrayitvā 'E +', Wh. Roots) SP Ger. 335.11 (? so Nep. mss., text nişevitvā with Kashgar rec.), and SP 48.6 with WT.
- (a) ucchrāpayanti Mv ii.112.18; Aor. ucchrāpayetsu Mv ii.343.22; ppp. ucchrāpitah MSV i.75.8, in same Caus. passage ucchräyitah Divy 77.20 (mss., here em. pitah); 466.16; prob. by error; § 38.17 (b) blend of prec. with ucchreti (§ 38.65), impv. ucchrepaya LV 399.19; Ppp. ucchrepita- LV 213.18;

351.7; Suv 62.8 (2) Miscellaneous forms

Ger. niśrāva (§ 35.20) SP 12.15 etc.; adhi-śrāva Jm 35.5. In Divy 264.18 (prose) ucchrivitvā, having raised (trans.), surely corrupt; read ucchrepitvă or ucchrāpitvā or ^opayitvā (to 1, above), or even ucchrayitvā = normal Skt. ucchritya

śru-, hear: (1) śrno-li, śrnu-le (Skt.), also semi-MIndic śruno-ti etc.

Pres. (§ 28.62) srnoma (before vowel) Mv i.289.1; (before consonant) Mv ii.79.9; śrņotha SP 324.8; śrņonti SP 325.7; śrņonte (°ta, before u-) Gv 236.11; śrūņute (§§ 3.26, 96) LV 74.9; opt. śrnuyā LV 42.19; śruņuya (§ 29.42) LV 54.2; śruno (3 sg. opt.? § 29.46, or imperf., § 32.124) SP 230.4; impv. srnohi SP 352.6; pple. śrnumāna- Šiks 107.9 (§ 34.3); śrnotu LV 364.5 (§ 18.35)

(2) śrnva-li, § 28.6; apparently not recorded elsewhere

Pres. 3 sg. śrnvati SP 359.8 (one ms. śrnoti, metrically impossible); Mv iii.83.1 (v.l. pasyati! with object sabdam); impv. 2 pl. srnvata Siks 1.6 (but here meter requires short first syllable); srnvadhvam Mmk 104.26; 105.10

(3) śruna-ti (rarely written śu°, su°, but in vss never makes preceding syllable long in cpds.); Pali sunati; §§ 3.96; 28.17. (n is often written for n)

- Pres. śrunati Mv ii.201.6; Śiks 258.3; pratiśrunāmi Mv ii.214.10; śrunanti LV 97.9; 233.10; opt. śrune Mv ii.387.5; śruneya Suv 51.13 (here 1 sg.); KP 159.17; read śruneyā in LV 80.12 for ed. sunesyā (all mss. śru-); śruņiyāma (§ 29.34) LV 364.11; impv. śruņā LV 94.12; pple. śrunanto Mv iii.71.14; 373.4 (v.l. in both śrnvanto); śruņamāna- Mv iii.83.8; 264.14; Gv 34.18; 215.15, 16
- śrunisyati LV 54.6; °yanti SP 236.4; °yi (1 sg. mid., §§ 26.3; 31.35) LV 222.13 Fut.
- Aor. śruni (3 pl.) SP 90.4; sunetsu(?) Mv i.247.7; śrunensu (WT with K' °nimsu) SP 52.12 (? § 32.101; guoted
- as śrnůsu or śrnotsu); šrunetsu (v i.337.2 śrunitvā LV 230.7; 232.13 etc.: śunitvā LV 57.5, 15; śrunitvāna Mv ii.227.14; °tvanā LV 235.21; Ger. śruniya Mv i.204.7; śruniyāna SP 61.9; śruni SP 351.11; Mv ii.12.3 (mss.); śruņituna (? § 35.36) Mv ii.12.17

Inf. śrunitum Suv 157.5

Gdve. śrunitavya- SP 308.11

Pass. srunjvati Mv ii.97.7; iii.297.8; °te Suv 157.2 (4) srune-ti: MIndic sunei ('In JM. AMg. überwiegt aber die Flexion nach der e-Conjugation,' Pischel 503). Rare in BHS

Pres. impv. śruņehi Mv iii.140.13 (§ 38.21)

(5) Pass. *sruya-ti*, etc., prob. miswriting for $sru^{\circ} =$ Skt. śrūyate, § 37.36)

Pres. śruyanti Mv ii.31.3, etc.

(6) Caus. śrāvaya-li (Skt.), also in non-caus. mg.

Pres. opt. śrāved SP 229.7; impv. śrāvehi Mv ii.256.7

- Aor. śrāvayimsu SP 194.5; anuśrāvayetsuh Mv i.40.11, 16
- samśrāvavi SP 384.1 Ger.
- (non-caus. mg.) pratiśrāvayitum Divy 207.7 (§ 38.23) Inf. (7) Desid. śuśrnya-li?
- Pres. pple. acc, sg. śuśruyantam Gv 520.23 (prose); § 40.1; cf. Pali sussūyati
 - (8) Miscellaneous forms
- Opt. srnyāt Gv 522.16 (? § 29.45)
- aśroșit LV 143.13; aśroși Mv ii.272.3; °și Mv i.319.15; Aor. °şuh (3 pl.) Mv i.258.21; § 32.69
- Prec. śruyāt, prob. m.c. for śrūyāt (Wh. Roots) Mmk 207.3 tam ca sabdam sruyāt ksipram (fitting regular scheme of anustubh); § 3.46
- śravanāya SP 431.1; 459.1 Inf.

ślis-, embrace: Caus. -ślesayati, in non-caus. mg. (§ 38.23)

Pres. opt. samślesayet SP 114.9

svas-, breathe: Caus. -svāsaya-ti (Skt.)

Pres. opt. āśvāseyam Mv i.39.5; āśvaseyā LV 300.19 (vs; a for a m.c.)

*samhāraya-li, gather (prob. for sambhāravati, denom. to sambhāra, § 38.57)

Caus. Gdve. samhārāpayitavya- Mv iii.178.5

sakka-li, go (MIndic): Pali ava-, opa-, o-sakkati: see Dict. s.v.

- Pres. avasakkanti Mv i.23.10 (mss. °sappanti); impv. osakka Mv ii.83.2; pple. parisakkantam Mv ii.254.3
- Ppp. osakkita- Mv 1.353.14
- anusak(k)ya, following(?), Mv i.23.11; Dict. s.v. Ger. anuśakya

*saja-li, send forth: Pali id., MIndic for Skt. srjati, § 3.90. Pali also has forms in sanj-, Dict. Ppp. sajita- (v.l. samj°) Mv i.37.1 Ger. utsajitva Gv 481.25

sajjaya-li (Skt.), sajje-li, prepare Pres. impv. sajjehi Mv i.148.7 Fut. sajjişyam Mv ii.274.8 Pass. sajjiyati Mv ii.274.16

sad-, sit: (1) sīda-li (Skt.)

- Pres. opt. praside Mv ii.45.1; prasideya Mv iii.437.7; impv. nişidāhi Mv ii.96.17; pple. nişidiyāna- SP 295.5 (§ 34.5)
- Fut. nişidişyam Mv iii.225.11; °şyāmo Mv iii.427.4

nisidi (3 pl.) Mv ii.29.4; niside Mv i.4.6; nisidetsuh Aor. My i.152.7

nişiditvā SP 236.8; nişīdiyāna SP 283.12 (§ 35.45) Ger. Gdve. nişīditavya- Mv i.354.14

Caus. (on nişidayati as caus. see below) Ger. nişidāpayitvā Mv ii.435.13; ^opetvā Mv iii.298.18 (2) (*ni*)sedyāmi, Pres., Fut. nişetsyāmi, and Gdve.

(ni)şettavya-, hyper-Skt. quasi-denom. to Pkt. nisejjä, seat, = BHS nişadyā; see Dict. s.v. nişedyati

(3) (ni)şīdaya-li, (ni)şīde-li, MIndic caus. to nişīdati

- (§ 38.9); also as non-caus. (§ 38.21) Pres. (caus.) nisIdayati LV 70.21; Divy 295.14; (non-caus.) nișideti Mv i.306.12 (with mss.); opt. nișidayed (non-caus.) SP 344.1
- Fut. (caus.) nişīdayişyāmah Divy 420.25

(4) Caus. sādaya-ti (Skt.)

- Pres. impv. prasādehi Mv iii.106.9
- prasādaye (2 sg.) Mv ili.106.8; prasādayesi Mv ili.106.2; prasādesi Mv ili.56.1 Aor.

(sura-ti, sec sr-, smr-, and sira-ti)

sarasuráya-le, onomat., prob. rustle: pple. °vamāna-Kv 30.13 (Dict.)

saha-li, prevail (Skt.)

Ppp. utsahita- Mv ii.90.6

Ger. sahitva RP 21.5

sātīya-ti or sādī°, take pleasure in (denom.); see Dict. and cf. svådiyati Pres. °yati SP 277.8

sādhaya-ti, accomplish (Skt.) Aor. sādhemsu LV 387.1 (§ 32.101)

sābhīya-li, be slow (see Dict.) Pres. °yati Śiks 152.7 (mss.)

sārajya-li, be attached = Pali sārajjati (Skt. samrajvate); § 3.3

Pres. °yanti My iii.295.15 (Senart's em. but plausible)

si-, bind (Skt. -syati)

Ppp. vyavasta- Divy 416.26 (for vyavasita-, § 3.112)

sic-, pour: (1) siñca-li (Skt.)

Pres. opt. utsimci (1 sg.) My ii.92.6; abhy-asimcet Mnrk 51.4 (§ 32.12); impv. abhişimcāhi Mv i.349.7

abhişimcişyam Mv ii.158.3; °şyanti Mv iii.207.1 Fut.

siñcitvā SP 147.6; abhişiñcya Mv iii.353.6 (§ 35.13) Ger. Gdve. abhişiñcitavya- Mv i.51.5

(2) sica-li; § 28.12; not recorded elsewhere. In all the following mss. lack the nasal (they mostly write un- for ut-); Senart writes "simc" (which in ii.92.6 is favored by meter; the others are prose) except in ii.90.17 where he keeps °sic°

Pres. utsicāmi Mv ii.90.15, 17

utsici Mv ii.92.6 Aor.

utsicitum Mv ii.91.1 Inf.

sīv-, sew: (1) *sīva-li (§ 28.29) Ppp. sīvita- My iii.313.4

Ger. sīvitvā Mv iii.313.7

- Ger. sivilya My m. s.s., (2) *se-li, see § 28.50 Gdve. setavya- MSV ii.50.13, 15; perh. to be read, with Roth's note, 152 f., on Nirukta 11.31, with the shorter version (p. 206), for sevitavya of both edd., which Sarup derives from sev-
- setum (ms. sentum) MSV ii.50.16 (§ 36.10) Inf.

(3) Miscellaneous

Inf. syotu-kāma- Av i.182.8 (§ 36.10)

sukhāya-te, be happy (denom., Skt.)

Caus. sukhāpaye (1 sg. mid.; so with mss.) Mv iii.355.8; Ppp. sukhāpita- Suv 96.2; Inf. sukhāpayitu-kāma-Suv 94.16; Gdve. sukhāpayitavya- Suv 79.6

subha-li, smite (§ 28.31): Pali only sumbhati (also with pati-, pari-), and AMg, visumbh-Pres. subhanti Mv i.14.2, 12; patisubhanti Mv i.20.3;

visubhanti Mv i.27.8

sūcaya-li, indicate (denom., Skt.) Pres. pple. súcata Siks 342.10 (§ 38.30)

sūtraya-li, cut into shreds (strings; denom.; Skt. in other senses)

Ppp. sūtrita- Mv i.5.7

Ger. sūtrayitvāna Mv i.12.16

sr-, move (see also s.vv. -śirati, smr-): (1) sara-li (Skt.) Aor. samosari (3 pl.) Gv 231.1; osare Mv ii.222.1; abhisaresi Mv ii.198.14; abhisarasi Mv ii.198.4 (3 sg., § 32.62)

- samsarita- Mv i.46.4 Ppp.
- samsaritvā Mvy 6628; Mv i.244.19 Ger.

Inf. samsaritum Mv iii.253.3

(2) -sarya-ti, nowhere recorded

Pres. opt. upasaryet AsP 390.9 (prose)

- (3) Caus. sāraya-ti (Skt.), also in non-caus. mg. Pres. opt. prasāraye Mv iii.422.17; impv. prasārehi Mv 111.401.15
- Aor. (non-caus.) avasāri (§§ 32.19; 38.23) Mv i.319.16 etc.; avasārim (§ 32.39) Mv ii.119.6; anusāre Mv iij.101.16 (§ 38.23)
- vyatisārayitvā Mv iii.47.18 (vīti°); 60.11 (°sāretvā); Ger. 206.1; 208.13; 325.14; 443.19
- Pass. utsaryati, for utsāryate, Mv ii.92.18 (§ 3.34)

srj-, send out, away (see also *saja-li): (1) srjya-li, -te; cf. AMg. sajjai, Pali ussajjitvā AN iv.191.11; these are prob. originally based on 'caus.' Skt. forms like (vi)sarjayati (see next) = (vi)srjati; even Skt. has very rarely sarjati. Our srjyati may be false Skt. for these Mindic forms (§ 28.26)

Pres. utsrjyate (Mironov °ti) Mvy 2558 (Tib. gton ba, abandon); pple. pratisrjyantau (dual) Mv i.181.1, Dict.

(2) srjali (Skt.)

- Caus. pratinisrjāpayisyanti MSV iv.140.7
- (3) 'Caus.' -sarjaya-li (Skt.)
- Pres. impv. visarjehi Mv i.363.9 (even in Skt. = visrjati) (4) Miscellaneous
- Fut. -srakşati (? § 31.29) Mvy 5232

sekh-, see sekh-

seva-li, -le, serve (Skt.); perhaps also seve-li, non-caus., \$ 38.21

- Pres. opt. seveyā Mv i.366.1; impv. upasevehi (non-caus.) Mv ii.103.13 (but cf. § 30.5); sevahi LV 28.22
- sevitva My ii.327.8; nisevitvā SP 335.11 (acc. to Ger. text). (sevitvā 'E', Whitney, Roots)

skanda-li, leap (Skt.)

Aor. sampraskande Mv ii.157.13

skhala-li, stumble (Skt.); MIndic khala-li = Pali id. Pres. pple. prakhalamāna- Mv ii.151.3 Aor. khali (v.l. skhali; 2 sg.) LV 362.11

(anu-)slanaya-ti (?stanāya-ti), mourn: cf. Pali anut-

thunāti, id., connected with Skt. stanati anustanayetsu(h) Mv i.341.8 ff. (mss. °stanā°, in Aor. 11 °strănă°)

stambh-, stabh-, hold firm (Skt.) Fut. avaştapsyate (§ 31.23) Divy 574.18; 575.7

slu-, praise: (1) slava-li (RV stavate), also slavaya-li, non-caus.: Pali thavati. Vedic inheritance conceivable; more likely analogical new formation, but specific source not clear; fut. stavisyati is only Vedic Pres. stavami Bhad 4; stavanti Šiks 339.11; abhistavati

- Mv ii.157.10; °vanti SP 12.6; opt. samstaveyā SP 229.11
- Aor. stavișu LV 329.14; stavimsū LV 233.19; abhistavinsuh Suv 243.10 (prose); stavayimsu LV 414.10; abhyastavit RP 5.6 (§ 32.45); abhistave Mv ii.266.2; abhistavetsuh Mv ii.187.6
- Ppp. stavita- LV 294.17; Suv 64.9
- stavitvā LV 225.4; staviya Mv ii.372.15 (read with Ger. mss. guņa staviya)

(2) stuvati: § 28.7; AMg. thuvai (Ratnach.)

Imperfect abhistuva LV 185.11 (§ 32.115)

- Ger. stuviya LV 50.4 (vs), see Dict.
- (3) stuti
- Pres., ed. stuti m.c. for ms. stauti Siks 341.11 (vs); see Dict. (4) Miscellaneous forms
- Aor. abhyastāvīt RP 2.18; stavist(a) LV 298.6; § 32.46 Pass. pres. pple. stuvantam, being praised, LV 195.20 (§ 37.35)

str-: stara-li, strew; sporadic in Skt.

Pres. impv. samstarāhi (v.l. °ehi) Mv ji.236.5

- Aor. prastare (1 sg.) Mv iii.249.2 Ppp. an-āstarita- Jm 220.14 (prose)
- Ger. samstaritvā LV 133.19; samstari Šiks 342.5; uttharya (?printed uttarya) Kv 32.17 (§§ 2.12; 35.12) Gdve. samstaritavya- Mv i.273.3
- Pass. Aor. samstaryetsu(h) My i.235.17 (mss.), see § 37.31
- stomaya-ti, praise (Skt. Gr.): Pali thometi; denom. from stoma-
- Ppp. stomita- (Pali thomita-) Mvy 2614; LV 7.22 (so read); Bhik 24a.4

slhā-, stand: (1) liştha-li (Skt.), also lişthaya-li, noncaus. (§ 38.21)

- Pres. opt. tisthayet Sādh 411.18; anutisthayema (Senart with v.l. "sthiyema) Mv iii.290.1; tisthe Mv iii.53.13; tiştheyā Mv ii.150.11; tişthiya (§ 29.34) SP 251.6; impv. tişthāhi Mv ii.449.8
- vyuttisthet (§ 32.87) SP 182.13; avatisthet Mv Aor. i.220.9 = avatisthe ii.22.10; tisthetsuh Mv i.52.3; tisthema (§ 32.105) Mv i.345.5, 14; (in caus. mg., § 38.24) pratitisthe Mv iii.294.10 tişthitvā SP 253.9 Ger.

(2) -the-ti: Pali -theti, Pkt. -thei (cf. also -sthayati, 3); \$ 28.48

- Pres. opt. or impv. (§ 29.11) utthi (only in vss, 2 sg., m.c. for utthe) LV 236.4; 322.3; Gv 488.18, and probably 16; impv. utthehi Mv ii.229.17; iii.6.7; 315.14 (in same vs LV 397.10 uttistha)
- utthesya (1 sg.) LV 368.10; utthesyam (so with v.l.; followed by ti = iti) Mv ii.83.15 utthesi (\S 32.64; 3 sg.) Mv ii.349.6 Fut.

Лог.

(3) -sthaya-ti: cf. -theti (2), and Pkt. utthai (Pischel 483, end); § 28.48

Pres. opt. adhisthayet Sādh 46.6; anusthayet Sādh 355.11 (both prose)

(4) *-lhāya-li: Pali thāyati (Geiger 132.3); § 28.24 Aor. pratyutthäyetsuh Mv iii.329.7

(5) shā-ti: Pali thâti, Pkt. thâi. Geiger 132.2 says analog. to yāti; but rather see § 28.51

- Pres. sthāti Mv ii.147.16 (here mss. sthāsi, sthāhi); iii.160.2; samsthāti Mv i.308.3; utthāti Mmk 157.20; adhişthāmi Gv 72.2 (prose); Sādh 268.8; impv. samupasthāhi Mv il.225.6; utthātu Mmk 585.7; pple. sthātu (gen. sg.), sthāti (loc. sg.) Mv i.236.12, 13 = 241.6, 7
- pratyasthā (§ 32.106; v.l. °sthāsi) Mv i.55.15; asthāsi (§ 32.54) SP 86.4; samavāsthāsit (§ 32.55; Aor. Kashgar rec.) SP 239.3; asthāsu(h), asthāmsu(h), asthānsu(h), asthātsu(h) Mv i.56.12 etc. (§ 32.75); upasthiya (? § 32.105) Mv i.304.9
- (6) -sthaha-ti: Pali thahati; § 28.42 Pres. adhisthahāmi SP 323.13; pratisthahanti SP 129.5; opt. adhisthahet Prät 506.12 (prose; same passage in Pali, Vin. iv.47.25, adhitthaheyya)

utthahitvā Mv ii.159.11 (v.l.; text utthih°); 345.1; Ger. vyutthahitvā SP 295.7; abhyutthahitvā Mv ii.38.14

(7) sthiha-li, -le (sometimes written with thi- for sthi-); unrecorded elsewhere, but very common here, in simplex and all cpds.; prob. the regular pres. of the dialect (tisthati being largely Sktization); § 28.43. Cf. s.v. dhā- (10)

- Pres. sthihanti Suv 49.20; 236.13; sthihate LV 29.1 136.7; samsthihati Mv i.236.10 = 241.4; ii.262.13; samsthihanti Mv i.101.2; samsthihate KP 47.6 (vs; in prose 47.2 samtişthante); upasthihati Mv ii.211.17; 212.2 (here mss. °hiti), 7; upasthihasi Mv i.302.7 (?text °hisi); upasthihanti Mv i.227.8; adhisthihanti Mv ii.263.12; visthihati Mvy 6821 (mss. vary; Mironov vişthirati); Sikş 18.6 (ms.; prose); utthihate LV 139.14; pratyutthihati Mv iii.114.11; opt. sthihi Mv ii.329.18; sthihed LV 369.1; sthiheyā LV 42.18; upasthiheya Mv ii.214.6; pratisthihe Vaj. Hoernle MR 180.14 (prose; in ed. Vaj 21.11 pratitisthet; upasthihesi (2 sg.) Mv ii.491.12; impv. sthihantu (v.l. thi^o) Bhad 11; utthihantu Gv 34.14; pple. upasthihantasya Mv ii.212.3; a-visthihanta Dbh.g. 4(340).8
- upasthihişyanti Mv i.325.19, 20; °şyāmi Mv ii.218.2 Fut.
- sthihetsuh Mv ii.302.8, 11; samsthihe Mv ii.234.5 = Aor. iii.65.6; 92.10; pratisthihisur (? § 32.36) Mv i.99.15; upasthihetsu (v.l. °suh) My iii.120.20; upasthihi Mv ii.489.11 (mss.); samsthihi Sukh 50.14; upasthihişu Gv 254.17; pratisthihimsu Mv i.203.4; samsthihesi (v.l. °sthahamsi; 3 pl.) Mv iii.276.1 (mss.)
- sthihitva LV 93.1; pratisthihitvā SP 63.10 (here prob. caus. mg., § 38.24); LV 134.9, etc.; sthihitvāna Ger. Mv ii.308.1; sthihiya LV 240.17
- upasthihitum Mv i.295.16; iii.24.15; utthihitum Mv Inf. ii.428.1
- Pass. upasthihīyati Mv iii.23.17; pple. upasthihiyamāna-Mv ii.423.9

(8) Caus. sthāpaya-li (Skt.), also in non-caus. mg.; also thāp°

- Pres. sthāpite for sthāpayati LV 74.22 (§§ 3.48; 37.24); opt. utthapayet, let him set forth, nou-caus. (§ 38.58) LV 335.15; impv. utthapehi Mv ii.430.6 Fut.
- thāpayişye LV 178.16 (most mss. sthā°)
- sthäpayet Mv ii.119.4; upasthäpe (so read for °ye, Aor. provided) Mv i.336.15; sthāpayetsu(h) Mv i.259.5 vithāpita- LV 324.8 (Dict.); pratisthāpayita-Ppp. (§ 34.8) Ky 18.20
- (9) Caus. sthapaya-li, sthape-li; also tha-, tha-; the usual BHS and Mindic caus. (§ 38.52): Pali thapeti
- sthapemi SP 323.9; pratisthapemi (Kashgar rcc.) Pres. SP 47.2; vithapayati KP 32.1 (prose); opt. sthapeyam SP 128.2; pratisthapeyam SP 47.4; upasthape (Kashgar rec. and WT) SP 88.12; upasthapesi (2 sg.) Mv ii.247.10; impv. sthapaya LV 12.15; 184.22; sthapayahi LV 167.8; sthapehi Gv 55.22; thapehi Mv ii.448.4; thapetu Gv 34.24; upasthapetha Mv ii.95.11
- Fut. utthapeşyati Mv iii.3.16; sthapeşya (1 sg.) LV 178.20; upasthapeşyāmah Mv ii.218.16; sthapeşye LV 216.10
- Aor. thapesi (§§ 32.63, 65; 2 sg.) LV 169.3; sthapi Gv 383.17
- sthapita- LV 166.9; vithapita- SsP 1534.7; thapita-Ppp. Mv iii.36.12 etc.
- Ger. sthapetvā My i.353.1; sthapitvā (so for text sthayitvā) Gv 122.12; upasthapetvā (non-caus., § 38.58) Mv ii.220.18; upasthapitvanā (so with v.l. for °pisva nā) LV 54.15; pratisthapiya Mv i.108.15
- Gdve. thapaniya- LV 127.17
- Pass. impv. sthapiyatu Mv ii.69.12; thapiyatu 13; Fut. sthapişyati or sthapiyişyati (§§ 31.2, 3; 37.6) Mv ii.69.14
 - (10) Miscellaneous forms
- Perfect 3 pl. tasthure (§ 33.5) Mmk 140.7 etc.; pple. avatasthivantah (? § 34.6) LV 368.2

snā-, bathe: (1) snāya-le (Epic Skt.), -li

- Pres. impv. snáváhi Mv ii.78.3
- snāyetsuh Mv iii.439.2 Aor.
- snāyitvā Mv i.4.6 Ger.
- snāyitum (?but mss. snāpitum, see below) Mv Inf. iii.12.5

(2) Caus. snāpaya-li (Skt.), also in non-caus. mg. (§ 38.58)

- Pres. snäpayati (non-caus.) Mv ii.77.19; 78.1
- snāpayet (§ 32.87), Šiks 155.3; snāpi (non-caus.) LV Aor. 271.12
- snäpitum (mss., non-caus., Senart em. °vitum) Mv Inf. iii.12.5

(3) Caus. snapaya-ti (Skt.), also in non-caus. mg. (§ 38.58)

- Pres. pple. snapayantam (non-caus.) My ii.77.18
- visnapī (3 pl.) LV 93.2; snapinsu LV 93.21, 22; Аог. snapayisu LV 234.4; °yimsu Mv i.99.11

sprś-, touch: (1) sprśa-ti (Skt.), also sprsaya-ti, noncaus. (§ 38.21): semi-Mindic pr°; and with s for s

- Pres. opt. sprse Mv ii.399.2; impv. sprsähi (v.l. °ehi) Mv ii.199.19
- spršisyasi SP 294.10; °ti LV 288.10 (s for š); prsi-syati (for sprš°) LV 153.18; spršisye LV 287.4; Fut. Mv ii.404.2 (? °şya a-); sprśişyi (2 sg.; § 31.35) Gv 58.2
- sprse SP 190.7 (§ 32.87) Aor.
- a-sprsita- Bbh 111.2 (prose); upasprsita-(vant-) Ppp. Mmk 47.5 (prose) spršitvā LV 301.2 etc.
- Ger.
- sprsayitum Mv ii.65.14 Inf.

(2) sparsa-li (Skt. once BhagP, see BR s.v. 11): Pali phassati (phasse Sn, Geiger 134; samphassamāna-, PTSD s.v. samphusati); AMg. samphase, cf. Pischel 486. Origin

- not clear to me; back-formation from caus. sparsayati?
- Pres. samsparšati Gv 399.22 (prose); opt. visparšeyus Siks 182.11 (prose); impv. sparsehi Mv iii.268.4 (3) Miscellaneous forms
- Aor. prākşur, 3 pl., LV 379.18; 380.2 (§ 32.80)
- prsta- LV 122.7; (by error prstha-) Divy 190.11 prastum Mv ii.427.7; iii.158.5; Divy 519.18 Ppp.
- Inf.
- Gdve. (as substantive) sparstavya- Siks 198.9, and perhaps intended by mss. My ii.391.16; § 34.21 and Dict.

sphāla-li (sphāle-li?), burst (intrans.): cf. Pali phāleti, trans. and intrans.; cf. Skt. sphalati (intrans., Gr.) and ā-sphālayati (caus., trans.)

- Pres. opt. sphāleyā, would burst (intrans.; mss.) Mv iii.114.12
- Caus. sphālayāmi, I burst (trans.), Av i.339.9 (mss. corrupt)

sphr-, suffuse, fill: (1) sphara-ti (Skt. Gr.), °te = Pali pharati; back-formation to caus. sphārayati? (perhaps on analogy of dhāravati: dharati?). See also pharati, phalati

- Pres. spharati I.V 250.20; Sikş 187.7; 216.5; spharanti Mv i.230.2; 240.12 (in parallels iii.334.9; 341.14 sphuranti, once v.l. spha^o); Gv 43.6; spharate LV 196.21; pple. spharan Gv 10.14
- spharitva Divy 106.4; Bbh 263.10; 332.4; spharit-Ger. vanā Gv 34.11
- Gdve. a-spharaniya- Śiks 249.5, 8
- Caus. parispharayati Mvy 1649; § 38.9. (In Skt. only sphārayati)

(2) sphura-li (Skt.), also *sphuraya-li, non-caus. (§ 38.21)

Fut. sphurisyati LV 145.3 (cited BR from Calc. as sphurayisyati)

Synopsis of verb forms

sphuri LV 357.4 Aor.

sphuritvā Mv ii.410.6; °tva LV 294.4; sphurayitvā Ger. LV 113.2 (prose)

smr-, be mindful: (1) smara-li (Skt.)

- Pres. opt. samanusmareya Mv ii.103.8; smareyāsi (2 sg.; § 29.37) Mv iii.20.12; impv. smarāhi LV 172.17 samanusmaret Mv i.228.16; °re Mv ii.284.8; anu-
- Aor. smarasi (3 sg.) Mv i.4.8 (§ 32.62); smaresi Mv ii.300.12 Ppp. vismarita- Suv 103.4
- smaritvā SP 115.11; smaritvana Gv 213.11; smari Ger. LV 167.2; anusmari Siks 339.2
- (2) sara-li (MIndic) = Pali id. (\S 2.16)
- Pres. sarati (= smarati) Mv iii.284.5 (vs)

srabh-, srambh-, see sr°

sru-, flow: sruva-le, not recorded elsewhere, unless we count śruvat RV 1.127.3; see §§ 3.57; 28.37 Pres. sruvate Mmk 456.17

svad-, 'caus.' (a-)svādaya-li, taste, eat (Skt.)

Aor. āsvādesi My 1.339.10; āsvādavetsu(h) My 1.339.12

svap-, sleep: (1) svapa-ti (Skt.)

- Ppp, svapita- Gv 254.22 (vs: svapiti, m.c. for °te, nrpe, the king being asleep)
- (2) *supa-li, supi-li: Pali supati, Pkt. suvai; directly
- from Skt. svapati, svapiti, § 3.117 Pres. supiti Sikş 257.19
- Ger. supiyāna Mv ii.201.19 (not caus.) Caus. supeti (§ 38.9) Mv ii.423.7; 433.12; Ger. supiya,
- having put to sleep, LV 50.4 (? but see Dict.) (3) sopa-li: Pkt. sovai; not in Pali; § 3.117
- Pres. osopati Mv ii.31.21 (prose)
- Ppp. sopita- Mv i.155.20
 - (4) svapya-ti: Pali suppati, soppati (Geiger 136.2); \$ 28.28

Pres. svapyanti Megh 294.11

*svaslyaya-li, thank, or reward; denom. from svasti-Ger. svastyayitvā Mmk 60.15

spādiya-li, take pleasure in (cf. sāliya-li, sādi^o) = Pali sadiyati; perhaps influenced secondarily by forms of svad-? Or MIndic pass. of caus. svadayati, is made to enjoy (so PTSD)?

Pres. °yati Mv ii.145.3

svādhyāya-, study (denom.): (1) svādhyāya-li: Pali sajjhāyati (§ 38.35)

- Pres. svådhyäyati Šikş 7.8; 197.5; °yanti Divy 631.22; impv. °yata LV 440.15; pple. svådhyäyanto SP 237.7; °yatah (gen. sg.) Divy_491.13; °yamāna-Divy 489.28; Suv 148.12 (so read); Gv 521.1 svådhyäyişyanti SP 478.7 Fut.
- Ppp. svādhyāyita- SP 420.1; Divy 55.7; 233.24; 464.18; Av 1.287.8
- Ger. svādhyāyya AsP 224.13
- Inf. svädhyäyitum Divy 339.22
- Pass. pple. svådhyäyyamäna- AsP 223.3
- (2) *svādhyā-li (see §§ 28.54; 38.31)
- Fut. svådhyåsyati Samådh p. 67 line 8; AsP 52.5, 21 Gdve. svädhyätavya- Samädh 22.8 (prose)
- Pass. opt. svädhyåyeta, would be studied, SP 391.7

han-, strike, slay: (1) hana-li (Epic and late Skt.): Pali id.; \$ 28.14

Pres. parähanati LV 318.16; hananti Mv i.10.6; opt. parāhaņe (1 sg.) Mv i.42.8; parāhaneya Mv ii.342.6; impv. parāhanāhi SP 178.10 (Kashgar rec.)

- abhihanî (3 pl.) LV 329.4; parāhanitsu SP 69.11 Aor. (Kashgar rec.); hane My ii.342.8; āhaneya My ii.412.19 (§ 32.105)
- nihanitvā LV 287.4; hanitvāna Mv ii.223.12: Ger. nihanitvana Samādh 22.5; parāhaniya LV 343.1
- hanitum Mv i.312.6 Inf. (2) Pass. hanya-le (Skt.)
- Pres. praty-ahanyanti (§ 32.8) Mv i.6.9 fl.; opt. hanyeyā Mv i.364.2; impv. vihanyähi Mv ii.405.16
- vihanyita- Mv i.155.14 (§ 37.26) Ppp.
- (3) Caus. ghālaya-li (St.), also ghāla-li (§ 38.28)
 s. ghātanti (mss.) Mv i.27.8; opt. ghāteran Lauk 252.16; ghātaye (2 sg.) Mv iii.387.13 Pres.
- aghātaye Mv iii.360.17 Aor.
- Caus. Ppp. 'ghātāpita- (§ 38.57) Mv ii.170.15; ghātāvita-(§ 38.70) Mv i.19.2; Inf. ghātāpayitum Mv i.132.8 (4) Miscellaneous forms
- Fut. hansye (§ 31.23) LV 311.18; read vihatsyase (§ 31.24) LV 335.1

hasa-li, laugh (Skt.)

Aor. ūhasāsi (Dict. and § 32.59) Mv i.221.20 = ii.24.7; ūhase My ii.268.13 fl.; in My i.218.18 = ii.20.20 read prob. ühasi for text ühati, which (like Senart, see his note) I cannot interpret

hā-, leave: (1) jaha-li (Skt. 'E+'): Pali id.; § 28.9

- Pres. opt. jahe Mv ii.390.21; viprajaheya Ud xx.1; impv.
- Fut
- yiprajahāhi Mv II.142.18 jahişyāmi ('E+' in Whitney, Roots) Mv iii.52.16 jahī LV 134.17; vijahe Mv i.329.6 Aor.
- Ppp.
- jahita LV 173.20; 242.4 (§ 34.10) jahitvā SP 13.1; °tvanā LV 325.8; vijahitvāna Mv Ger. i.268.21; vijahya (§ 35.14) LV 202.6; vijahiya LV 232.22; vijahiyāna Mv i.208.4 (here mss. °hyāna) = ii.12.13
- Caus. *vijahayati: Ger. vijahayya Bhik 15a.5 (§ 38.9)
- (2) hāya-le, -li = Pali id.; irregular passive (§ 37.38) Pres. hāyate Mv i.147.1, 5 etc.; parihāyante Gv 462.9; opt. parihāye Mv ii.204.14
- Fut. häyisyati Mv i.44.4; °yanti Mv 1.241.18
- Aor. parihāyişū SP 170.2
- (3) Caus. hāpaya-li (Skt.)
- Pres. impy. parihāpehi Mv iii.168.1
- (4) Miscellaneous forms

- Pres. in caus. mg. (§ 38.24) jahāti SP 61.8 Fut. prahāsate Ud xix.1 (§ 31.27) Inf. prahātavai (intending °ve?) Ud xxxi.2 (§ 36.14)
 - himsa-li, injure, also himsaya-li, id. (both Skt.)
- Pres. opt. himsi (2 sg.) Mv ii.236.6; vihimsaye (3 sg.) Mv iii.322.10; vihimseyā Mv ii.10.20
- vihimsi (3 pl.) LV 75.16; himsisu LV 28.6: vihimse Aor. My iii.369.10

hind-, wander: (1) (chiefly with a-, anva-)hinda-li, -le, MIndic = Pali (ā-, anvā-)hiņdati; also hiņdaya-li, non-caus. Pres. āhiņdati Māy 242.31; anvāhiņdati Divy 249.11;

- °anti Mmk 139.19 (so read for andā hi°); 249.10; °ate Av ii.116.10; ähindase Divy 165.3; pple. hindan LV 140.7 (see Dict.); anvāhiņdatā (instr.) Divy 237.24; °dantā (n. pl.) Mv i.20.3; paryāhiņdanto (u. sg. m.) Mv i.353.4; °dantiye (gen. sg. fem.) Mv iii.155.1; anvāhiņdamāna- Av i.242.6; upāhiņdamäna- Divy 264.19
- Ger. anvāhiņdayitvā Mvy 6942; °hiņdya Mvy 5116; LV 16.4; Divy 68.23; 139.3; Av i.5.10
 - (2) (ā-, anvā-)hindya-le, not recorded elsewhere
- Pres. anvāhiņdyante Divy 175.12, 15; pple. āhiņdyamāna-Divy 141.22

hiş-, hiş-, neigh (cf. Skt. heşati): (1) hişa-li, nowhere recorded

- Pres. hisati I.V 236.7 (vs)
- (2) *hişa-li, nowhere recorded; § 28.36
- Ppp. hişita- Mv ii.160.7
- hisitvā Mv iii.76.2, 12 Ger.
- (3) hisya-ti, nowhere recorded
- Pres. hisyanti (v.l. hresvanti, cf. Skt. hresati) Mv i.308.6
 - hu-, pour: (1) juho-li, juhu-mas (Skt.)
- juhutvā (on weak pres, stem) Mv iii.426.18 (§ 35.14) Ger. (2) juhva-li, 3 sg. (§ 28.11)
- Pres. juhvati Mv iii.426.12 (prose; so one ms., v.l. junati which is obviously corrupt; certainly 3 sg.) (3) *iuha-li: § 28.18
- Fut. juhisyāmah Mv iii.161.11
- juhita- Mv iii.149.2; 431.14 = 17 = 445.7, 10 Ppp.
- juhanāya Mv iii.161.9 Inf

huduhuduva-ti, onomat, (of nagas sending rain); cf. huduk, huluhuli

Pres. impv. °yantu Megh 304.18

- hū-, call: (1) hvaya-li (Skt.)
- Caus. āhvāpayate (§ 38.55) Divý 323.1; Ger. āhvāpya Av ii.53.5

(2) hvāya-ti, nowhere recorded; § 28.33; Senart compares bhāyati from bhī. The ā is probably due to influence

- from nominal forms, cf. Pali ppp. avhāta for Skt. āhūta ähväyanti Mv i.177.6 (vs. but metrically indifferent Pres.
- in 2d syllable)
 - (3) hūya-li, cf. Skt. āhūyate, passive only
- Pres. āhūyati, calls upon, challenges, Divy 275.9 (prose), \$ 37.23

hr-, take: (1) hara-li (Skt.), also hare-li, non-caus. (§ 38.21)

- Pres. ähareti My iii.201.5: opt. abhinirhari (1 sg.) Bhad 34; udāhare (2 sg.) Mv i.77.6; hareyā Mv ii.146.13
- uddhari (§ 32.121; 2 sg.) LV 195.12; nirhari (1 sg.) Bhad 6^a; udäharişu LV 413.21; uddharetsuh Mv Aor. iii.78.18; viharesi (3 sg.) Mv i.228.4, 6, 10; abhinir-haresi Mv i.228.12; äharesi Mv ii.233.8; 234.9; ähare Mv i.343.6, 16; äharema (1 pl.) Mv i.344.15 (§ 32.105)
- Ppp. vyavaharita- Mv i.271.18; vyāharita- Divy 397.24 haritvā Mv ii.93.6; samharitvāna Mv i.304.15; ud-Ger.
- dharetvå (§ 35.26) Mv i.7.14; vihariya LV 243.11 viharitum Mv 1.34.13; pratisamharanāya Mv iii.252.8; viharatum (§ 36.7) Divy 498.3 Inf.

Gdve. viharitavya- Mv iii.265.11 Caus. harāpayasl MSV iv.207.2

(2) -hara-li, -le, prob. denom.; § 28.33. Recorded only once in Pali; see below, and under (4)

- Pres. pple. samvyavahāramāņa- Divy 259.10 (prose); Dict.; same form in Pali samvohāramāna- AN ii.188.10; samvohāra occurs in the same Pali context (but samvyavahāra not in the Divy context) (3) -haro-li; § 28.18
- Pres. abhyuddharoti Dbh.g. 21(357).12. No v.l. reported; perhaps read °dhareti (cf. above under 1). But Kondo's ed. also has °dharoti.

(4) Caus, hārava-li (Skt.). On samhāravali, prob. not belonging here, see s.v.

- [Aor. aharetsu(h) My i.340.8, 9; in 9 v.l. ahar°; object āhāram; prob. denom., cf. (2)]
- Pass, impy, ohārīvatu My 1,295,18
- (5) Miscellaneous forms (on hrivati, °te, see s.v. hri 2) Aor. harşur (3 pl.) LV 271.16 (§ 32.74)
- -hārakam Mv i.343.4; -hārim Mv ii.65.10 etc. Ger. (\$\$ 35.5, 6)

hrs-, thrill: (1) *harsa-li; cf. Ved. harsata, Skt. once samharsamana- (MarkP., see BR); Pali hamsati and perhaps hassati (see PTSD s.v. hasati), both derivable from Skt. hrsyati; is harsati a false Sktization of one of these Mindic forms? § 28.40 Aor. harşi LV 232.16

- (2) Caus. harşaya-li (Skt.)
- sampraharsaye (v.l. °yet) Mv i.334.1 Aor.

Ger. harsitva SP 25.7

hethaya-li, hethe-li, rarely hetha-li, injure; MIndic: Pali hetheti, hethayati. In large measure takes the place of hims-

- Pres. vihethāma Mv iii.131.19; hetheti Mv iii.215.14; vihethayati Dbh 91.1; °yasi Divy 42.17; °yanti LV 76.18; opt. vihethayeta LV 338.10; 'yet LV 403.7: vihetheya Mv ii.178.7; °thaye Mv iii.322.16, 19; pple. a-vihethayatā Divy 104.13; 105.19
- vihethayişyanti LV 404.11; °şyati Mv i.361.16 Fut
- vihețhi (3 pl.) LV 75.16; vihețhetsuh Mv ii.359.19 vihețhita- Mv ii.225.17 Aor.
- Ppp.
- Ger. hethavitvā Mv iii.369.8
- Gdve. vihethayitavya- Mv i.360.11

hri-, be ashamed: (1) hriyāya-ti, hriy°(?), denom.(?); cf. Pali hirlyati, hiriyati, harāyati (is our form a blend of these?); Pkt. hiriāmi

Pres. pple. fem. hriväyanti (in 8 v.l. hri°) Mv ii.48.8. 11: hriyåyanti (so text, but v.l. hriyayåyanti, intending hriyāyantī) Mv li.64.5

Caus. hriyāpayanti Bhīk 11a.5; § 38.56

(2) hriya-li, -le Mv 11.365.13; 366.6, 10. If from this root, cf. Pali hiriyati (above). But context suggests that passive forms of hr may be meant: 'is carried away' = Skt. hriyate, with 1 perhaps m.c.; cf. Pali hariyati (to harati). There are dependent instrumentals dosena, rägena, etc. Cf. § 28.25

(3) jihriya-11: blend of Skt. jihreti and Pali hiriyati? § 28.25

- Pres. jihriyati, v.l. for jihreti Karmav 47.26; in 49.2 text jihriyati without support of mss., one of which reads hriyati, the other omits; in 49.10, 16 text with mss. jihriyati (v.l. in 16 ^ete) jihriya MSV i.14.16 (but this is doubtless based on
- Ger. the Skt. weak pres. stem jihri-, § 35.14)

(4) Intens, jehriya-le; recorded nowhere else, not even in Pali

°yate Mvy 1829; °yante Divy 39.7; pple. °yamāņa-Pres. SP 108.6

(5) Miscellaneous

Gdve. hretavya- Bbh 223.9 (§ 34.21)

F. エジャートン『仏教混淆梵語 文法および辞典』

F. EDGERTON :	BUDDHIST	HYBRII) SANSKRIT	
	GRAMMAR	AND 1	DICTIONARY	Vol.1

1985年9月30日 複製第1刷発行	定価19,000円		
	 発行者 片 岡 英 三 印 刷 明 信 印 刷 所 製 本 新生製本株式会社 		
孫行所 株式 臨 川 書 店 京都市左京区今出川通川端東入 (〒 606) 郵便振持京都7-800番 電話(075)721-7111	RINSEN BOOK CO. P.O. BOX: SAKYO &/IMADEGAWA KAWABATA, SAKYO-KU, KYOTO 666, JAPAN		

ISBN4-653-01245-8 C3015 ¥19000E